United States Environmental Protection Agency Solid Waste And Emergency Response (OS-240) OERR 9200.0-10A January 1991



Superfund Removal Procedures

Directives Notebook Volume 1



Printed on Recycled Paper

OERR Directive 9200.0-10A September 1991

SUPERFUND REMOVAL PROGRAM DIRECTIVES NOTEBOOK

VOLUME ONE

Office of Emergency and Remedial Response U.S. Environmental Protection Agency Washington, DC 20460

TABLE OF CONTENTS

VOLUME I

AUGUST 1991

REMOVAL POLICY AND GUIDANCE	DATES	COMMENTS
Removal Completion Date Definition	10/24/86	
Guidance on Implementation of the "Contribute to Remedial Performance" Provision (OSWER Dir. 9360.0-13)	04/06/87	Supplements OSWER Dir. 9200.3-02
Outline of Engineering Evaluation/Cost Analysis (EE/CA) Guidance	03/30/88	
Removal Program Priorities (OSWER Dir. 9360.0-18)	03/31/88	
Draft Guidance for Conducting Federal- Lead UST Corrective Actions on Indian Lands (OSWER Dir. 9610.9)	07/18/88	Supplements OSWER Dir. 9360.0-16A
RCRA		
Definition of "Liquid Waste" (OSWER Dir. 9432.01-81	06/28/81	Clarifies 40 CFR 260.10(a) (25) and 261.21
Determining If the Soils from Missouri Dioxin Sites are Hazardous (OSWER Dir. 9441.01-84)	01/06/84	201.21
Definition of "Treatment" (OSWER Dir. 9432,01-85)	06/26/85	Clarifies 40 CFR 260.10
Interim RCRA/CERCLA Guidance on Non-Contiguous Sites and On-Site Management of Waste and Treatment Residue (OSWER Dir. 9347.0-01)	03/03/86	
Responsibilities of Regional RCRA Off- Site Disposal Contacts	03/24/86	
Use of Liquids for Wind Dispersal Control at Hazardous Waste Landfills (OSWER Dir. 9487.00-01A)	04/21/86	

RCRA (continued)	DATES	COMMENTS
Consideration of RCRA Requirements in Performing CERCLA Responses at Mining Waste Sites (OSWER Dir. 9234.0-04)	08/19/86	
Joint EPA/NRC Guidance on the Definition and Identification of Commercial Mixed Low-Level Radioactive and Hazardous Waste (OSWER Dir. 9432.00-2)	03/02/87	
Land Disposal Restrictions	08/11/87	
Revised Procedures for Implementing Off- Site Response Actions (OSWER Dir. 9834.11)	11/13/87	Supersedes OSWER Dir. 9330.2-5 & 9330.2-6
Transmission of Guidance for Conducting Federal-Lead Underground Storage Tank Corrective Actions (OSWER 9360.0-16A)	07/25/88	Supersedes 9360.0-16
Policy for Superfund Compliance with the RCRA Land Disposal Restrictions (OSWER Dir. 9347.1-02)	04/17/89	
Land Disposal Restrictions as Relevant and Appropriate Requirements for CERCLA Contaminated Soil and Debris (OSWER Dir. 9347.2-01)	06/05/89	
Notification of Out-Of-State Shipments of Superfund Site Wastes (OSWER Dir. 9330.2-07)	09/14/89	
Analysis of Treatability Data for Soil and Debris: Evaluation of Land Ban Impact on Use of Superfund Treatment Technologies (OSWER Dir. 9380.3-04)	11 <i>İ</i> 30/89	
ALTERNATE TECHNOLOGIES		
Draft Guidance on Differentiating Alternative Technologies for the Removal Program (OSWER Dir. 9380.2-05)	09/30/87	

ALTERNATE_TECHNOLOGIES (continued)	DATE	COMMENTS
Superfund Innovative Technology Evaluation (SITE) Program Requirements Technology (OSWER Dir. 9380.2-06)	03/22/88	Updates Superfund Innovative Program 12/01/86
Factors to Consider When Selecting Incinerators as Alternative Technology for Superfund Cleanups	Not Specified	
CERCLA/SARA INTERPRETATIONS	DATE	COMMENTS.
CERCLA Removal Actions at Methane Release Sites (OSWER Dir. 9360.0-08)	01/23/86	
Implementation Strategy for Reauthorized Superfund: Short Term Priorities for Action (OSWER Dirs. 9200.3-02)	10/24/86	Supplemented by (OSWER Dir. 9360.0- 12 & 9360.0-13)
Guidance on Implementation of the "Contribute to Remedial Performance" Provision (OSWER Dir. 9360.0-13)	04/06/87	
Scope of the CERCLA Petroleum Exclusion Under Sections 101 (14) and 104 (a)(2) (OSWER Dir. 9838.1)	07/31/87	
Guidance on Non-NPL Removal Actions Involving Nationally Significant or Precedent- Setting Issues (OSWER Dir. 9360.0-19)	03/03/89	
Final Guidance on Implementation of the "Consistency" Exemption to the Statutory Limits on Removal Actions (OSWER 9360.0-12A)	06/12/89	Supersedes 9360.0- 12
ADMINISTRATIVE/COST MANAGEMENT		T WRAT I
Clarification of Eligibility and Approval for Hazardous Duty Pay (Office of the Comptroller Transmittal No. 85-5)	09/16/85	- manazara i aktori ostati u sitati
Superfund Indirect Cost Manual for Cost Recovery Purposes (FY 1983 through FY 1986)	03/00/86	Updated by 01/05/87

ADMINISTRATIVE/COST MANAGEMENT (continued)	DATE	<u>COMMENTS</u>
Superfund Indirect Cost Update	01/05/87	Updates Indirect Cost Manual
Superfund Site Allowances (Comptroller Policy Announcement No. 87-07)	01/13/87	
Interim Concurrence Procedures for Removal Actions Under Delegated Regional Authority Requiring EPA Headquarters Concurrence	03/06/87	
New Method for Determination of Indirect Costs in Superfund Removal Project Ceilings (Comptroller Policy Announcement No. 87-15)	07/15/87	Replaces draft; Supplements Indirect Cost Manual
Timing of \$2 Million Exemption Request Submissions	04/04/88	
Model Program for Removal Site File Management (OSWER Dir. 9360.2-01)	07/18/88	
Site Specific Contracting for Removals (OSWER Dir 9242.2-02)	04/10/89	
Need for Contract Officers Authorization Before Contractor Activation (OSWER Dir. 9242.6-03)	08/09/89	

Removal Completion Date Definition



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

OCT 2 4 1986

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Removal Completion Date Definition

FROM: Timothy Fields, Jr., Director Emergency Response Division

TO: OHM Coordinators Regions I - X

During the recent FY 86 SCAP review, inconsistencies were noted among Regions in determining CERCLA removal action closing dates. This memorandum is intended to clarify how the completion dates of removal actions are to be determined when temporary storage of hazardous substances is involved. The determination of removal completions was originally described in Superfund Removal Procedures -- Revision Number Two, date: August 20, 1984, and subsequently has been discussed in the various SPMS and SCAP methodologies.

Temporary demobilization and temporary storage onsite are not considered completions, unless temporary storage is the only action identified in the Action Memorandum to mitigate threats to public health, welfare and the environment. Likewise, temporary <u>off-site</u> storage of hazardous substances at a storage, treatment, and disposal (TSD) facility other than the facility of ultimate disposal is a continuation of the removal action, <u>not a completion</u>.

A removal action would not be considered complete if:

- Hazardous substances stored onsite are being monitored by the ERCS contractor or if any additional ERCS expenditures are anticipated, or
- Hazardous substances are being stored at an off-site facility, other than the ultimate TSD facility.

A removal action would be considered complete if:

The scope of work for the removal action does not specify final off-site disposal of hazardous substances, the substances have been stabilized and are stored onsite and no additional CERCLA removal program funds are anticipated to be expended at the site. In this case, hazardous substances may be expected to undergo long-term storage onsite due to circumstances such as the unavailability of a final treatment/disposal remedy (e.g., Missouri dioxin sites). In this instance, no CERCLA removal program funds will be expended for long-term site operation and maintenance. Any long-term (greater than 6 months) site operation and maintenance will be performed by the responsible party or another agency (e.g., State).

Hazardous substances are being stored off-site at the location of final disposal, and no additional ERCS expenditures are anticipated.

If any further clarification of how to determine the closing date of a project is required, please contact Jim Jowett or Mark Mjoness at FTS 382-2188.

cc: GAMS Hans Crump Terry Ouverson

Guidance on Implementation of the "Contribute to Remedial Performance" Provision

United States	
Environmental	Protection
Agency	

Office of Solid Weste and Emergency Response

₽EPA

DIRECTIVE NUMBER: 9360.0-13

TITLE: Guidance on Implementation of the "Contribute to Remedial Performance" Provision

APPROVAL DATE: April 6, 1987

EFFECTIVE DATE: April 6, 1987

ORIGINATING OFFICE:

I FINAL

STATUS:

REFERENCE (other documents):

OSWER Directive 9200.3-02 Implementation Strategy for Reauthorized Superfund: Short Term Priorities for Action

OSWER OSWER OSWER VE DIRECTIVE DIRECTIVE

GUIDANCE ON IMPLEMENTATION OF THE

"CONTRIBUTE TO REMEDIAL PERFORMANCE" PROVISION

1. INTRODUCTION

Section 104(b) of the Superfund Amendments and Reauthorization Act of 1986 (SARA) amends section 104(a) of CERCLA to include the statement that any removal action undertaken by the President, or by any other person referred to in section 122* of the new law, should, to the extent practicable, contribute to the efficient performance of any long-term remedial action with respect to the release or threatened release concerned. This guidance document explains how to implement this provision, and includes guidelines on the applicability of the requirements, the definition of "contribute to efficient performance," exceptions, documentation and coordination. This document should be used in conjunction with the general removal procedures described in the Superfund Removal Procedures -- Revision Number Two, August 20, 1984, or, as may be amended.

2. APPLICABILITY

This provision will be applicable to removal actions at all sites -final National Priorities List (NPL), proposed NPL, and non-NPL. The term "long-term remedial action" as used in this provision will therefore refer to a remedial action to be taken by the EPA, State, or a private party.

3. DEFINITION OF "CONTRIBUTE TO EFFICIENT PERFORMANCE"

3.1 Purpose

This provision promotes the performance of removal actions that address threats more efficiently by considering the overall site cleanup before the start of the action. To the maximum extent practicable, removal actions should be designed to avoid wasteful, repetitive, short-term actions that do not contribute to the efficient, cost-effective performance of long-term remedial actions to be taken by the EPA, State, or other party. The major objective of this requirement is to provide maximum protection of public health and the environment at minimal cost by avoidance of removal restarts. The focus of this provision is on avoidance of restarts that are due to recurring threats that were not adequately abated in the original removal action, and threats from deteriorating site conditions that should have been foreseen.

There are other circumstances, however, where removal restarts may be necessary to meet program goals. For example, a removal action may be a phased response. The first removal action might involve site stabilization and waste

^{*}Section 122 refers to potentially responsible parties (PRPs) who have entered into settlements with EPA.

characterization. The site may then be demobilized and closed out to allow removal personnel to prepare an analysis of waste treatment/disposal options. Once an option is selected, a removal restart would be implemented to complete the waste disposition phase. In this case, the removal restart would actually contribute to achieving a more efficient cleanup. Removal restarts may also occur in an attempt to meet other program goals, such as pursuing responsible party (RP) cleanups or State assumption of removal action operation and maintenance requirements. An RP may take over a removal action from EPA, but EPA may have to initiate a restart if the RP is not performing an adequate cleanup. The "contribute to efficient performance" provision was not intended to conflict with these other program goals. As stated above, the provision was intended to reduce removal restarts due to inadequate planning at the start of the action.

3.2 Implementation

To meet the goal of avoiding removal restarts, response personnel must adequately assess the threats posed by the hazardous substances on a site and. consider how the removal action would most effectively contribute to the longterm remedy. The following questions should be considered:

1. What is the long-term cleanup plan for the site?

This provision requires removal actions to contribute to the performance of the "long-term remedial action." At an NPL site, if the Record of Decision has already been signed, then comparing the removal action to the remedial cleanup plan is a straightforward task. However, for proposed NPL sites and for many final NPL sites, the remedial action may not have been selected when the removal action is implemented. In these cases, response personnel will be limited to identifying a range of feasible remedial alternatives. Response personnel need only review existing site information and use their best professional judgment. Removal and remedial personnel in the Region must coordinate with each other in this effort. It is the responsibility of the Region to establish appropriate coordination mechanisms.

At non-NPL sites, response personnel should, where practicable, consult with the party performing the long-term response action at the site (e.g., State, RP) to determine the proposed approach for the long-term cleanup. It is recognized that it may be more difficult to ascertain the remedial action at non-NPL sites. Response personnel should use their best efforts to coordinate with the party performing the long-term remedy. At many non-NPL sites, there may be no plans for another party to conduct a remedial action.

2. Which threats will require attention prior to the start of the long-term action?

The February, 1986 National Contingency Plan (NCP) broadened removal authority by allowing removal actions to be taken in response to "threats" rather than just "immediate and significant" threats. This expanded authority will allow a removal action to address any near-term threats that may arise prior to the start of the long-term action, provided the threats meet the removal criteria established in section 300.65 of the current NCP. Potential threats should be identified when the first removal action at a site is implemented to avoid the need for future removal restarts. Therefore, in addition to identifying immediate threats, response personnel should also identify potential near-term threats from contaminant migration, deteriorating site conditions, etc. This assessment is particularly important if a decision is made to leave surface hazardous substances on site after the removal action is completed,

Response personnel must identify threats that may arise prior to the start of long-term actions, but the length of time before longterm actions will begin will vary from site to site. For example, for NPL sites where a ROD has been signed, the time frame that response personnel must consider will be shorter than for NPL sites where the Remedial Investigation/Feasibility Study (RI/FS) has just been initiated. Of course, at some NPL sites, the remedial program may plan to conduct an operable unit during the RI/FS if time permits. In this case, the time period to consider would again be shorter. Response personnel should consult with the party performing longterm action at the site to determine when such action will begin, and use their best professional judgment.

At non-NPL sites where there are no plans for another party to undertake a long-term cleanup, all threats and potential threats that meet the removal criteria in the current NCP should be identified.

3. How far should the removal action go to assure that the threats are adequately abated?

The expanded authority in the 1986 NCP will allow more complete removal actions to to be taken. Removal actions no longer have to stop when emergency situations are mitigated, but can continue, or be initiated, where needed to ensure that near-term threats are adequately abated. Measures that provide only temporary protection, insufficient to last until long-term actions begin, should be avoided to the extent possible. However, as noted above, consideration must also be given to the availability of other response mechanisms (e.g., State action, remedial operable unit) to initiate long-term action in a timely manner.

Whether or not the removal action should address all surface hazardous substances must be decided on a site-by-site basis. A removal action would be appropriate whenever surface hazardous substances may present a threat (as established in section 300.65 of the current NCP) before the start of long-term action. How the removal action should address the surface hazardous substances will also depend on site-specific conditions and the long-term cleanup plan. With the increased emphasis on using alternative technologies and new restrictions on land disposal, remedial actions may often include onsite treatment if surface contamination is extensive. In this case, the removal action may consist of consolidating and stabilizing the substances on site to await treatment. It is important to design the removal action to ensure that the materials are adequately stabilized. At other sites, surface hazardous substances may constitute only a small part of the problem; may not be safely stabilized for a long period of time; or may be more efficiently addressed as one unit by immediate treatment or disposal. In these situations, it may be more appropriate for a removal action to include final disposition of all surface hazardous substances. The conditions at the site and the longterm cleanup plan will determine the appropriate scope of the removal response.

At non-NPL sites where there are no plans for another party to perform long-term remedial action, the threats that meet the removal criteria in the current NCP should be completely cleaned up, if possible. The avoidance of removal restarts due to recurring threats is the ultimate goal. If mitigation of the threats that meet the NCP removal criteria results in complete site cleanup (i.e., no further Federal response required), the "contribute to efficient performance" provision is fully satisfied.

In considering all of the factors described above, the major determinant of how far the removal action should go to assure threats are adequately abated will be the statutory limits on removal actions. Removal actions should contribute to the efficient performance of remedial actions to the maximum extent practicable given the \$2 million/ 12 month limits on removal actions. (An exemption to the limits may be granted where the site qualifies under the "emergency" or "consistency" waivers.)

With regard to cleanup standards, this provision does not compel the removal program to lower its cleanup standards. Rather, the purpose of this provision is to improve the design of removal actions such that after cleanup standards are established for a removal site. the chosen removal action will address those substances targeted for cleanup in a manner that avoids the need for removal restarts. For example, the removal program has historically used 50 ppm as a benchmark in determining the appropriate extent of cleanup of PCBcontaminated soil. The "contribute to efficient performance" provision would not affect this number, but would direct that the method chosen to address soil contaminated above 50 ppm should be designed to avoid the need for removal restarts to the extent practicable.

4. Is the proposed removal action consistent with the long-term remedy?

The removal action that is chosen should be consistent with long-term actions at those sites where further cleanup will be taken. "Consistent" is defined in its broadest sense and may be characterized as a range of possible approaches. At one end of the spectrum, removal actions may be found consistent if they do not hinder or interfere with the remedial action to be taken. At the other end of the spectrum, removal actions may be found consistent because they contribute in a positive way to the long-term cleanup plan. For example, a removal action to provide carbon filters to homes with contaminated drinking water as an interim measure would not interfere with a long-term remedial plan to clean up the contaminated aquifer. A removal action to solidify sludge could, however, hinder a long-term plan to incinerate the waste and should, therefore, be avoided if other approaches are feasible. A removal action to remove surface drums from a landfill could contribute in a positive way to a remedial plan to clean up the site.

Removal actions may be found consistent if they fall anywhere within this range; the most appropriate approach will depend on site-specific factors. It is recognized that in some cases, the removal action may create additional work for the remedial action and yet still be the most appropriate approach for the site. For example, a common removal action is capping contaminated soil to prevent migration and human contact in the time period before remedial actions begin. Although the cap would have to be removed to implement a long-term plan to excavate and treat the soil, it may still be the most effective method to mitigate the threatin the short-term. Protection of public health and the environment, as well as technical feasibility, must always be considered. If such an action is selected, the rationale for selection should be explained in the Action Memorandum. (See Section 5.)

The answers to these four questions will help determine what type of removal action is needed and how it can be designed to contribute to the efficient performance of long-term remedial actions. These questions are provided as general guidelines to indicate the various factors that should be considered in implementing this provision of SARA. A written analysis of the answers to each of these questions is not required. The conclusions should be documented in the Action Memorandum. (See Section 5.)

EXCEPTION

The only situation where it may not be feasible to <u>consider</u> how the proposed removal action relates to the long-term remedy is in an emergency. In such cases, response personnel may need to take whatever immediate measures are required to protect the public health, welfare, and the environment.

5. DOCUMENTATION AND COORDINATION

The Action Memorandum should specifically cite the "contribute to efficient performance" requirement and briefly discuss how the proposed removal action relates to long-term remedial actions, to the extent practicable. [See the Superfund Removal Procedures for information on the preparation of Action Memoranda.] If the proposed removal action completes the cleanup and no further action is required, this should be so noted. If only minimal information is available about long-term actions, this should also be explained. If an emergency existed that precluded an analysis of how the removal related to long-term actions, this should be noted. Finally, if compliance with this provision would conflict with other program goals (e.g., pursuit of RP cleanup), this shoud be explained. Compliance with this requirement does not require special approval; the Action_Memorandum should be approved by the established concurrence chain in the Region or in Headquarters, if appropriate. In making the determination, however, it will be the responsibility of the OSC to coordinate with the party that will undertake the long-term remedy (for those sites where additional cleanup measures will be taken).

Outline of Engineering Evaluation/ Cost Analysis (EE/CA) Guidance



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C. 20460

MAR 3 0 1000

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: OUTLINE OF EE/CA GUIDANCE

FROM: Timothy Fields, Jr., Director Emergency Response Division

TO: Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X

In June 1987, the Emergency Response Division (ERD) issued the first draft guidance on Engineering Evaluations/Cost Analyses (EE/CA) for non-time-critical removal actions. The goals of the EE/CA are to: 1) satisfy environmental review requirements for removal actions; 2) satisfy administrative record requirements for improved documentation of removal action selection; and 3) provide a framework for evaluating and selecting alternative technologies.

ERD delayed issuance of a second draft EE/CA guidance pending the outcome of several issues related to the upcoming National Contingency Plan (NCP) revisions. Most of these issues have now been resolved and ERD is preparing a new draft EE/CA guidance for Regional review this spring.

Attached is an outline of the EE/CA guidance ERD is developing. Because there have been a number of questions from the Regions on EE/CAs, we thought it would be helpful to provide an outline at this time to assist the Regions in preparing EE/CAs until the new draft is available. Note that the EE/CA process no longer includes an initial screening of the alternatives and that the selection criteria have changed somewhat.

If you have questions on the attached outline, please call Jean Schumann of my staff at FTS 382-4671.

Attachment

cc:	Hans Crump	Earl Salo
	Paul Nadeau	Lee Tyner
	Bill Hanson	Joe LaFornara
	Don White	Bruce Engelbert
	Lloyd Guerci	John Riley
	Frank Russo	Cristina Griffin
	John Cross	Jean Schumann

I. INTRODUCTION

II. EE/CÁ REPORT

- A. Site Characterization
 - 1. Site description
 - 2. Site background
 - 3. Analytical data
 - 4. Site conditions that justify a removal action
- B. Identification of Removal Action Objectives
 - 1. Statutory limits on removal actions
 - 2. Removal action scope
 - 3. Removal action schedule
 - 4. Applicable or relevant and appropriate requirements (ARARs)
- C. Identification of Removal Action Alternatives
- D. Analysis of Removal Action Alternatives
 - 1. Effectiveness
 - a. Protectiveness
 - b. Use of alternatives to land disposal

2. Implementability

- a. Technical feasibility
- b. Availability
- c. Administrative feasibility
- 3. Cost
 - a. Total costb.- Statutory limits on removal actions
- E. Comparative Analysis
- F. Proposed Removal Action
- III. CONTRACTING CONSIDERATIONS
 - IV. COST MANAGEMENT
 - V. EE/CA FUNDING
 - VI. ENFORCEMENT-LEAD ACTIONS

- I. INTRODUCTION
- * EE/CAs are required only for non-time-critical removal actions/Expedited Response Actions (ERAs). The Regions may choose to prepare an EE/CA for other actions.
 - Non-time-critical removal action: Those releases or threats of releases not requiring initiation of on-site activity within 6 months after the lead agency determines, based on the site evaluation, that a removal action is appropriate. (In other words, based on threat, there is at least a 6 month lead-time available before cleanup action must begin.)
- * Steps in the EE/CA process (apply only to actions that are determined at the outset to be non-time-critical):
 - A. <u>Site evaluation</u>. Removal PA/SI results indicate that the site meets the criteria for initiating a removal action and that the threat is non-time-critical. (At an NPL site, RPMs should continually evaluate site conditions to determine if a removal action is appropriate.)
 - B. <u>Issue PRP notice</u>. General notice required; special notice discretionary.
 - C. <u>EE/CA Approval Memorandum</u>. Documents that the site meets the criteria for initiating a removal action and secures management approval to conduct the EE/CA. (To be resolved: Format and approving official)
 - ° OSC/RPM should notify the community relations staff of the upcoming EE/CA.
 - Designate site spokesperson.
 - ° Open Administrative Record (AR) and publish notice of availability.
 - * Begin community interviews and preparation of Community Relations Plan (CRP).
 - D. Contract for EE/CA preparation. TAT, REAC, REM, site-specific.
 - E. <u>EE/CA study and report preparation</u>. See II below for more detail. May include on-site activities to better define site and characterize waste (104(b) activities), but may not include cleanup measures.
 - F. EE/CA completed.
 - Place EE/CA in AR and publish notice of EE/CA availability plus brief summary of EE/CA.
 - CRP should be completed by this time.

- F. Thirty-day public comment period. On EE/CA and other documents in AR.
- G. Action Memorandum, attaching responsiveness summary. Describes proposed removal action and secures management approval to conduct the action. Responsiveness summary is a summary of significant public comments and EPA's response to these comments.
 - * AR closes when Action Memorandum is included. At this point, all information relating to the selection of the removal action must be included in the AR. EPA may add documents generated after the Action Memorandum is signed only if they concern issues which were specifically reserved for future action or if they support an amended Action Memorandum.
- H. Implementation of removal action. \$2 million/12 month statutory limits apply only to the implementation of the removal action, not to previous 104(b) activities.
- Note that an EE/CA and public comment period are not required if a removal action will be used to implement a signed ROD. In that case, the RI/FS and remedial public participation procedures fulfill the EE/CA requirements.

II. EE/CA REPORT

The EE/CA report should follow the format below.

- A. Site Characterization
 - 1. Site Description

Location, facility type, surrounding land use, hydrology, nature and extent of contamination, etc.

2. Site Background

Prior site use, operational history, regulatory involvement, etc. (Confidential information must be placed in confidential portion of AR.)

3. Analytical Data

Summary of results of analytical data (considering the quality of that data).

4. Site Conditions That Justify a Removal Action

Information contained in the EE/CA Approval Memorandum should be used here.

B. Identification of Removal Action Objectives

1. Statutory Limits on Removal Actions

Brief explanation for the public of the \$2 million/12 month limits on removal actions and two types of statutory exemptions available ("emergency" and "consistency"). Stated objective should be to remain within these limits, unless site qualifies for one of the statutory exemptions.

2. Removal Action Scope

Description of the scope of the project, e.g., total site cleanup, site stabilization, completion of operable unit (NPL sites), surface cleanup only. Include description of principal threats to be addressed. Particularly important to clearly define scope if removal action will not address the entire universe of threats at the site.

3. Removal Action Schedule

General scheduling objectives for the removal action, identifying any time constraints (e.g., must complete action prior to winter, threat requires initiation of action within 1 year).

4. Applicable or Relevant and Appropriate Requirements (ARARs)

Identification of chemical-specific and location-specific Federal and State ARARs for the site. (<u>Compliance</u> with identified ARARs will be discussed in the analysis of removal alternatives in section D below.) States are required to identify promulgated State ARARs in a timely manner.

Removal actions should attain ARARs to the extent practicable.

C. Identification of Removal Action Alternatives

Description of appropriate removal action alternatives for site (including description of necessary equipment, personnel, etc.). Based on OSC/RPM experience and best professional judgment.

A "no action" alternative is not required.

Additional resources available to assist in identifying appropriate technologies: ERT, SITE program, Superfund Regional Technology Transfer contacts, industry publications, best demonstrated available technologies (BDATs) identified in the land disposal restriction rules.

D. Analysis of Removal Alternatives

Each alternative should be evaluated individually based on the criteria below.

- 1. Effectiveness
 - a. Protectiveness
 - Protection of the community during the removal action

Description of threats that may result from implementing the removal action, such as air quality impacts from an incinerator that may affect human health, and mitigative measures that can be taken.

Protection of workers during the removal action

Description of threats that may result from implementing the removal action, such as dust from excavation, and mitigative measures that can be taken.

* Threat reduction

Evaluation of the extent to which the completed action will reduce risk or mitigate the threats identified in the description of removal scope (8.2). Measured qualitatively or quantitatively (e.g., cleanup levels or cancer risk levels achieved), as appropriate.

* Time until protection achieved

Determination of the time until protection is achieved for the principal threats at the site, compared to the removal action schedule (B.3) where appropriate.

Compliance with chemical- and location-specific ARARs

Determination of whether ARARs identified in section 8.4 can be met or whether a waiver may be appropriate.

Compliance with criteria, advisories, guidances

Description of compliance with other criteria, advisories or guidances that are not ARAR, but could appropriately be applied to the site. For example, if PCB-contaminated soil will be excavated in the alternative, EE/CA may compare the cleanup level the alternative will achieve (the level described under "threat reduction" above) to the appropriate cleanup levels in the EPA PCB Spill Cleanup Policy.

* Environmental impacts

Description of the potential adverse environmental impacts

that may result from implementing the removal action and mitigative measures that can be taken. (If overlap with ARARs evaluation occurs, simply refer reader to the appropriate ARARs discussion in the EE/CA report.)

Potential exposure to remaining risks

Assessment of potential for future exposure to residuals remaining on-site.

* Long-term reliability for providing continued protection

Assessment of potential for failure of the alternative and need for replacement, and description of potential threats from such failure or replacement. Should address the reliability of engineered components of the alternative (cap, treatment system), non-engineered components (fences), and any institutional controls (deed notices), as appropriate.

b. Use of Alternatives to Land Disposal

Description of the degree to which the alternative utilizes treatment or recycling. Removal program policy encourages the use of alternatives to land disposal where practicable.

- 2. Implementability
 - a. Technical Feasibility
 - Ability to construct and operate technology

Description of the ability to construct the technology and to keep it running during operation, considering difficulties and unknowns that may lead to schedule delays. Compare to removal action schedule (B.3) where appropriate.

Compliance with action-specific ARARs

Identification of Federal and State action-specific ARARs and determination of whether ARARs can be met or whether a waiver is appropriate.

Ability to meet process efficiencies or performance goals

If overlap with ARARs evaluation occurs, simply refer reader to appropriate ARARs discussion in the EE/CA report.

^o Demonstrated performance

Evaluation of maturity of technology and whether it has been used under similar conditions for similar wastes.

Environmental conditions

Evaluation of impact of environmental conditions, such as terrain and climate. For example, a generally reliable oil/water separator may be inoperable in freezing temperatures without the use of heaters. A site located in a valley may pose a problem for a technology if surrounding air currents provide insufficient dispersion of particulates.

Compliance with SARA requirement that removal actions should contribute to the efficient performance of long-term remedial action to the extent practicable

Is the action designed to prevent the need for removal restarts to address the same threats? Is the action consistent with the long-term remedy for the site?

- b. Availability
 - * Availability of necessary equipment, materials, and personnel

Compare to removal action schedule (B.3) where appropriate.

* Availability of adequate offsite treatment, storage, and disposal capacity, if appropriate

Compare to removal action schedule (B.3) where appropriate.

Post-removal site control (PRSC)

Description of any PRSC measures that will be required at completion of the action, including monitoring, and availability of another party to assume these activities at the end of the removal action.

- c. Administrative Feasibility
 - Likelihood of public acceptance of the alternative, including State and local concerns
 - Activities needed to coordinate with other agencies
 - Ability to obtain any necessary approvals or permits (permits are not required for actions conducted on-site)
- 3. Cost
 - a. Total Cost (Present Worth) of the Alternative

Include direct capital costs, indirect capital costs, and any post-removal site control costs. The draft NCP recommends use of discount rate of 5 percent before taxes and after inflation.

b. Statutory Limits

Comparison of total cost to the \$2 million statutory limits on removal actions.

E. Comparative Analysis

Qualitative assessment of strengths and weaknesses of each alternative relative to the others. Summary tables would be helpful, with alternatives along one axis and evaluation criteria along the other axis. (Include post-removal site control costs when comparing costs of alternatives.)

F. Proposed Removal Action

Identification of the proposed removal action. If proposed action will exceed \$2 million, include justification of need to exceed the statutory limits.

3.0 CONTRACTING CONSIDERATIONS

To avoid potential conflict-of-interest, the contractor who conducts the EE/CA may not be used to perform the site cleanup.

- 4.0 COST MANAGEMENT
- 5.0 EE/CA FUNDING
- 6.0 ENFORCEMENT-LEAD ACTIONS

Removal Program Priorities



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

MAR 3 1 1988

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE OSWER Directive No. 9360.0-18

MEMORANDUM

SUBJECT:	Removal Program Priorities
FROM:	J. Winston Porter Assistant Administrator

TO: Regional Administrators, Regions I-X

 ATTN: Director, Waste Management Division, Regions I, IV, V, VII, VIII
 Director, Emergency & Remedial Response Division, Region II
 Director, Hazardous Waste Management Division, Regions III, VI
 Director, Toxics and Waste Management Division, Region IX
 Director, Hazardous Waste Division, Region X
 Director, Environmental Services Division, Regions I, VI, VII

PURPOSE

The purpose of this memorandum is to confirm general policy regarding removal program priorities.

BACKGROUND

The enactment of the Superfund Amendments and Reauthorization Act of 1986 (SARA) and other recent developments have expanded the potential scope of the removal program. Removal resource levels, however, require us to set priorities for the removal program, which address health and environmental threats to the maximum extent possible within the confines of our resource limitations. Funding levels and SCAP targets for FY 88 were contained in a memorandum of August 31, 1987, entitled Final 1987 SCAP, from Henry L. Longest and Gene Lucero to the Regions.

The national extramural cleanup budget for removals in FY 88 is about \$90 million: this will fund approximately 190 removal actions. All of the \$90 million is being distributed to the Regions; Headquarters is not keeping any contingency reserve. The rate of removal obligations in the first and second quarters of FY 88 suggests that there may not be enough removal funds to last through the entire year if the current pace of program activity is continued. Each Region is responsible for planning and conducting a removal program consistent with its annual resource allocation.

REMOVAL PRIORITY POLICY

Removal resources must be used to address the most serious public health and environmental threats. In all cases, Regions should aggressively pursue cleanup by the responsible party (RP), if time permits, before initiating any Fund-lead removals. Top priority for removal resources should be given to time-critical actions as follows:

- Classic emergencies involving incidents (e.g., threats of fire or explosion) where response is generally necessary within a matter of hours.
- Time-critical removals at sites on the National Priorities List (NPL). Regions should always give due consideration to the fact that removals at NPL sites may result in a loss of the 10 percent State cost-share; and
- * Time-critical removals at non-NPL sites posing major health and environmental threats, which cannot be addressed by other authorities.

These three categories of time-critical removals should be the primary focus for the \$90 million allocated to the removal program for cleanup contractor support.

As resources permit, there are other <u>non-time-critical</u> removals at NPL sites which we may be able to conduct. These removals will usually be at sites already funded in the remedial SCAP to:

- Completely cleanup NPL sites, resulting in deletion; or
- ° Conduct interim actions at NPL sites.

Generally, we intend to use remedial funds to pay for non-time-critical actions at NPL sites. The principal source of these remedial funds will be proceeds from RP settlements and funds allocated for remedial projects in the SCAP. My staff is examining ways to create additional funding flexibility in the Superfund SCAP and budget processes to facilitate this new direction.

Regions should strive to maintain consistency with remedial actions, but actions taken should remain within the scope of a removal. Stabilization at NPL sites is usually the most appropriate removal action, unless complete cleanup can be done within the Region's resource allocation.

States should be strongly encouraged to conduct non-time-critical removals at sites which will not score high enough to be on the NPL. Preference should be given to response alternatives which contain and control the source of contamination and prevent off-site migration. Removal personnel should also continue to provide full cooperation with the enforcement program in pursuing potential RP cleanup at these sites.

Draft Guidance for Conducting Federal Lead UST Corrective Actions on Indian Lands

OTHER RESPONSE MECHANISMS

Regions should actively pursue response by States and potential responsible parties (PRPs). Enforcement authorities should be vigorously applied to encourage PRP response. This means conducting a complete PRP search, issuing general notices or special notice letters (if time permits), negotiating with PRPs, and using administrative orders on consent to formalize settlements. Unilateral administrative orders should be considered in every case where voluntary settlement is not achieved.

Regions should also pursue response under other authorities, and give priority to those incidents posing threats which can only be addressed by Federal removal authority. For example, installation of new water supply systems should generally be a State or local responsibility. The new drinking water action levels are only one of many site-specific factors to be considered in deciding whether Federally-funded removal action is appropriate. More specific guidance in this area is under development. In the interim, Regions should contact the appropriate Regional coordinator in the Emergency Response Division whenever there are questions about removal priorities.

CONCLUSION

The importance of managing the removal program within the boundaries of its resources cannot be overemphasized. We have provided these national guidelines on removal priorities to assist Regions in this effort; however, Regions maintain the discretion to conduct other types of removals within program authorities, if site-specific conditions necessitate such action and if Regional resource levels permit. We recognize that the intrinsic nature of the removal program is such that even with the most careful planning, unanticipated events may occur.

I hope this memorandum is helpful as we all attempt to use our Superfund resources to address the most significant environmental priorities.

cc: Superfund Branch Chiefs, Regions I-X
Oil and Hazardous Materials Coordinators, Regions I-X
Jack McGraw
Henry Longest
Gene Lucero
Tim Fields
Paul Nadeau
Lloyd Guerci



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUL 18 1988

OFFICE OF SOLID WASTE AND EMERGENCY PESHUNGE

MEMORANDUM

- SUBJECT: Guidance for Conducting Federal-Lead UST Corrective Actions on Indian Lands
- FROM: Ron Brand, Director Ra Buel Office of Underground Storage Tanks
- TO: Hazardous Waste Division Directors, Regions 1-3, 5-9 Water Division Directors, Regions 4 and 10 Superfund Branch Chiefs, Regions 1-10 OHM Coordinators, Regions 1-10 Timothy Fields Jr., Emergency Response Division, OERR (WH-548B) Lisa Friedman, Office of General Counsel (LE-132S)

I have enclosed a copy of our draft guidance for conducting Federal-lead underground storage tank corrective actions in response to petroleum releases on Indian Lands. The guidance discusses the procedures and documentation that are necessary for approving and carrying out Federal response to UST releases on Indian Lands. The guidance supplements the soon to be distributed final guidance on Federal-lead UST corrective actions for States (OSWER directives 9360.0-16A), which does not address Indian Land releases.

Please submit your comments on the enclosed draft to Mark Waiwada (WH-562A) of my staff. We would appreciate receiving all comments by August 15, 1988. Should you have any questions, please call Mark at FTS-475-9727.

cc: UST_Regional Program Managers Jobs Riley, ERD (WH-548B) Bruth Englebert, ERD (WH-548B) Betty Zeller, ERD (WH-548B) Carrie Wehling, OGC (LE-132S) Mark Waiwada, OUST United States Environmental Protection Agency Office of Solid Waste and Emergency Response



DIRECTIVE NUMBER: 9610.9

TITLE: Draft Interim Guidance on Conducting Federal-Lead Underground Storage Tank Corrective Actions for Petroleum Releases on Indian Lands APPROVAL DATE:

EFFECTIVE DATE:

ORIGINATING OFFICE: Office of Underground Storage Tanks

DRAFT

STATUS: For Review and Comment

REFERENCE (other documents):

OSWER Directive 9360.0-16A Guidance for Conducting Federal-Lead UST Corrective Actions

OSWER OSWER OSWER 'E DIRECTIVE DIRECTIVE D

DRAFT INTERIM GUIDANCE ON CONDUCTING FEDERAL-LEAD UNDERGROUND STORAGE TANK CORRECTIVE ACTIONS FOR PETROLEUM RELEASES ON INDIAN LANDS

1.0 INTRODUCTION

1.1 Purpose

This guidance document is designed to provide direction to On-Scene Coordinators, UST Regional Program Managers, and other Regional personnel for the initiation and coordination of Federal-lead corrective actions on Indian Lands in response to petroleum releases from underground storage tank (UST) systems. The guidance discusses the procedures to be followed, including how to determine whether Federal-lead corrective action is justified, and what documentation is necessary for undertaking Federal-lead UST corrective actions on Indian Lands.

This interim document supplements the guidance on Federal-lead corrective action for States (OSWER directive 9360.0-16A) which does not address releases on Indian Lands. The final guidance document on Federal-lead UST corrective action on Indian Lands will be issued in FY 1989, following the completion of Indian Lands pilot projects being conducted in Regions 8 and 9. The pilot projects will gather information about USTs on Indian Lands to assess the extent of the UST problem and the capacity of Indian Tribes to address the problem. EPA will assess the information gathered through the pilot projects, and other information gathered in responses initiated under this interim guidance, to develop the final guidance document.

1.2 Background

The Superfund Amendments and Reauthorization Act (SARA) contains provisions in section 205 amending Subtitle I of the Solid Waste Disposal Act (SWDA) which gives EPA and States under cooperative agreement the authority to conduct corrective actions in response to petroleum releases from USTs, using monies from the \$500 million Leaking Underground Storage Tank (LUST) Trust Fund. SARA does not provide EPA the authority to enter into LUST Trust Fund cooperative agreements with Indian Tribes.

EPA expects Federal-lead corrective actions on Indian Lands to occur only in a limited number of instances. Preliminary assessments indicate that the total number of underground storage tanks on Indian Lands is small (0.2 percent of the regulated universe). Further assessments, including those based on the information gathered during the Indian Lands pilot projects, will generate a more accurate estimate of the total number of USTs on Indian Lands. While this document is limited to "time-critical" responses on Indian Lands (i.e., releases that require response within six months), guidance for longerterm corrective actions will be issued when the extent of the problem on

* * * DRAFT * * *

Indian Lands is more fully understood.

1.3 Policy

The EPA LUST Trust Fund Program is State implemented. States under cooperative agreement carry out Fund-financed responses to releases from leaking USTs, except in rare instances. EPA's role is to provide guidance and support to the States in their development of State LUST programs. EPA will lead a Fund-financed response in a State only in limited instances when there is a major public health or environmental emergency, the owner or operator is unwilling or unable to respond, and the State is unable to respond.

Because of the State emphasis of the program, Indian Lands present a unique situation for the EPA LUST program. EPA does not have the statutory authority to enter directly into LUST Trust Fund cooperative agreements with Indian Tribes, and most States do not have the authority to run environmental programs on Indian Lands. EPA's goal is to treat Indian Lands as States, allowing Indian Tribes to implement programs similar to those carried out by States. A legal mechanism for allowing Indian Tribes to establish cleanup programs and directly access the LUST Trust Fund is being sought, and the willingness and capability of Indian Tribes to develop and implement such programs is being examined. During this period EPA is prepared to respond to emergency LUST releases on Indian Lands.

To trigger Federal-lead action, an Indian Lands site must pose a timecritical, either immediate or near-term, substantial threat to human health and the environment, the Indian Tribe is unable to respond, and the owner or operator must be unable or unwilling to provide adequate and timely response. Considering that many Indian Tribes may lack the capability to oversee or conduct corrective actions, the criteria for initiating a Federal response to releases on Indian Lands has been expanded from the Federal-lead corrective action guidance for States. Federal response in States is limited to classic emergencies (i.e., the release immediately threatens human health and the environment and requires response within hours or days), while Federal response on Indian Lands may occur in situations where immediate action is not necessary but the release is considered time-critical and will require response within six months.

Federal-lead UST corrective actions will be performed by the same EPA emergency response and contractor personnel that conduct oil and hazardous substance removal actions under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) and section 311 of the Clean Water Act. Removal actions appropriate to a Federal response to an UST release on Indian Lands include, but are not limited to, the following:

 Site investigations or exposure assessments to determine potential health effects of a leak and to establish corrective action priorities;

2
- Ventilation of fumes from a residence or other building;
- Construction of fences, warning signs, or other security or site control precautions;
- Emptying of a leaking UST or removal of contaminated soils;
- Construction of an intercepting ditch;
- Provision of alternative water supplies; and
- Temporary relocation of affected residents.

Federal-lead corrective action on an Indian Lands site will be terminated when the threat to human health and the environment has been mitigated. Release situations for which longer-term corrective actions are necessary, such as groundwater cleanup, may require remedial actions that are beyond the capabilities of EPA's removal program. OUST will address long-term corrective action under separate guidance when the need for such action is established.

Efforts will be made to involve Indian Tribal authorities in cleanup decisions. On-Scene Coordinators and UST Regional personnel should rely on Indian Tribal authorities to gather as much information as possible about the site and owners and operators prior to initiating a Federal response action.

2.0 CRITERIA FOR FEDERAL-LEAD RESPONSE

To qualify for Federal-lead response, an Indian Lands site must meet the legislative criteria specified in Section 9003(h) of Subtitle I of the SWDA for the periods before and after the effective date of the final regulations, and the site must pose a time-critical (i.e., an immediate or near-term substantial threat to human health and the environment). More specifically, the site must meet at least one of the following criteria:

- The release poses a time-critical, either immediate or near-term, substantial threat of direct human, animal, or food chain exposure to petroleum;
- The release poses a time-critical, either immediate or near-term, threat of fire or explosion;
- The release poses a time-critical, either immediate or near-term, substantial threat to public drinking water supplies; or
- The release poses a time-critical, either immediate or near-term, threat to a significant population or substantial amounts of property, or poses substantial threats to natural resources.

These criteria have been expanded from those requiring a "major" emergency for Federal-lead response in States. This allows EPA to respond to a wider range of releases on Indian Lands. Specifically, near-term has been added to the criteria to allow for Federal-lead response on sites that pose a less immediate, but still substantial threat. Immediate and near-term threats are defined as follows:

- <u>Immediate threat</u>: A release requires response within hours or days of discovery. Examples include potential or actual UST explosions or fires, or a release that is contaminating a public well.
- Near-term threat: The release is time-critical, and requires response within six months. An example includes a slow release of petroleum spreading at a rate which will contaminate residences within 6 months.

Since the OSC must give priority to responding to "classical" emergencies, both in the removal program under CERCLA as well as the LUST program, response to less urgent UST removals will depend on the availability of staff and resources.

3.0 DELEGATIONS

The procedures for initiating Federal-lead corrective actions on Indian Lands are covered by the delegations specified in the Federal-lead guidance on UST corrective actions in States. Federal UST corrective actions that initially cost over \$250,000, require approval of the Assistant Administrator (AA), Office of Solid Waste and Emergency Response (OSWER). The Office Director (OD) of the Office of Emergency and Remedial Response (OERR) will approve actions that initially cost up to \$250,000 and ceiling increases that bring the cost of an action up to \$250,000, with concurrence from the OD, Office of Underground Storage Tanks (OUST).

In addition, Regional Administrators (RAs) may approve actions costing up to \$50,000 in acute, life-threatening situations where response must be initiated before Headquarters can be contacted. This authority may be redelegated to Division Directors and On-Scene Coordinators.

4.0 APPROVAL PROCESS

The **approval** process for Federal-lead corrective actions on Indian Lands is essentially the same as that for Federal-lead UST corrective actions taken in States. However, on Indian Lands EPA's Regional personnel are also responsible for assisting in the initial evaluation of the site and gathering of information necessary to support a request for Federal action. To qualify for Federal-lead action a site must pose an immediate or near-term substantial threat to human health or the environment, the Indian Tribe is unable to respond, and the owner or operator must be unwilling or unable to carry out corrective action properly and in a timely manner. Adherence to the

* * * DRAFT * * *

4

procedures set forth in this approval process ensures that Federal-lead corrective actions on Indian Lands will be limited to sites that meet the criteria in section 2.0 of this guidance.

To request Federal-lead corrective action, an individual designated by Tribal authorities must notify the Regional Oil and Hazardous Materials (OHM) Coordinator, or other designated Regional management official, that a release has occurred. The UST Regional Program Manager will gather information about the site through Indian Tribes and other Federal agencies, such as the Bureau of Indian Affairs and the Indian Health Services, and will evaluate all attempts made by Indian Tribes to locate the responsible party. The assigned On-Scene Coordinator (OSC) and the UST Regional Program Manager will determine whether the site qualifies for Federal response. If insufficient information about the site is available from outside sources, the OSC, in coordination with the UST Regional Program Manager, may conduct a site assessment if available information provides sufficient reason to warrant such a visit. To use the Technical Assistance Team (TAT) on a site assessment, the OSC must obtain Headquarters approval as prescribed in OSWER directive 9360.0-16A (Appendix A, section 6.2.1).

Requests for Federal-lead corrective action can either be oral, for sites requiring action within hours or days, or written, for those sites that do not require immediate action.

4.1 Oral Request

Oral requests should be used in case of an emergency in which there exists an immediate threat of death, injury, or catastrophic environmental damage due to a petroleum release from an UST. The OSC may take immediate action to address the situation, utilizing the RA's authority to approve emergency obligations of up to \$50,000.

The OSC, in coordination with the UST Regional Program Manager, when possible, will get approval from the RA or the RA's representative to request approval from Headquarters as prescribed in OSWER directive 9360.0-16A. The OSC provides oral request information to Headquarters (see Appendix A). This information is similar to that required for an oral request for Federal-lead corrective action in States, except that additional information is required on the Tribal authority reporting the release and having jurisdiction over the site. If the request is approved, the UST Regional Program Manager, in conjunction with the OSC, prepares an action memorandum within 10 days of initiating a response, which contains more detailed information about the site (see Appendix B).

4.2 Written Request

The UST Regional Program Manager, in conjunction with the OSC, will gather all information necessary to support a request for Federal-lead

corrective action and will prepare an action memorandum (see Appendix B). This action memorandum is similar to that required for a written request for Federal-lead corrective action in States except that it includes sections requiring information on the Indian Tribe's capacity to oversee or conduct corrective actions and information on why the site meets the criteria for Federal response on Indian Lands. The action memorandum is reviewed by the Emergency Response Division (ERD) Regional Coordinator and other appropriate Headquarters official(s) prior to approval as prescribed in the Federal-lead UST corrective action guidance document for States. The ERD Regional Coordinator will communicate to the Region as quickly as possible the decision to approve or deny the action.

5.0 ENFORCEMENT ACTIONS

Enforcement actions may be required prior to the initiation of Federallead corrective actions to ensure that every attempt is made to compel the owner or operator to respond to the release. Potential responsible party (PRP) searches and negotiations to secure owner or operator action should be undertaken in most instances by Indian Tribes. Depending on the availability of time and resources, the OSC shall undertake initial PRP searches and preliminary discussions with PRPs, if necessary, to secure owner or operator action.

Subsequent enforcement actions against owners or operators, including cost recovery, should be coordinated by Regional UST personnel in conjunction with other responsible EPA Offices. Due to the limited number of releases expected to occur on Indian Lands, and the requirements for encouraging owners or operators to respond to such releases prior to using Trust Fund monies, cost recovery by Regional personnel will be necessary only in limited instances.

6.0 INITIATING AND MANAGING FEDERAL-LEAD UST CORRECTIVE ACTION

The procedures to be followed for initiating and managing Federal-lead corrective action on Indian Lands (including information required) are the same as those specified in OSWER directive 9360.0-16A.

Procedures for initiating Federal-lead corrective action are outlined in section 6.0 of OSWER directive 9360.0-16A. These include procedures for accounting information, procuring EPA contractor and other services, and obtaining assistance from other agencies.

Procedures for managing Federal-lead UST corrective actions are outlined in section 7.0 of OSWER directive 9360.0-16A. These include information and procedures for allowable costs, stabilization standards, ceiling increases, reporting requirements, and operation and maintenance.

6

APPENDIX A: ORAL REQUEST INFORMATION

- 1. What Indian Tribal authority has jurisdiction over the site?
- 2. Is the person requesting Federal-response an official representative of the Indian Tribe? If so, in what capacity does he or she serve?
- 3 Does the Indian Tribe have its own legislative authority that covers this release?
- 4. Does the Indian Tribe have any formal agreement/relationship with the State regarding UST cleanups or UST related programs?
- 5. Is the release from a leaking underground storage tank (see definition in 9001(1) of SWDA)?
- 6. Is the released material a petroleum substance (see definition in Section 9001(8) of SWDA)?
- 7. Location of release.
- 8. Time and date release was discovered.
- 9. Name, description, and general condition of facility.
- 10. Is the release a long-term or chronic problem?
- 11. Type and estimated amount of petroleum released to the environment.
- 12. Which of the criteria for Federal-lead response on Indian Lands in section 2.0 does the release meet?
- Number and proximity of persons potentially affected.
- 14. Increased threat to human health or the environment if response is delayed or denied?
- 15. Ongoing efforts to respond to release?
- 16. Tribal authority's ability and willingness to provide response, with specific reasons for inability to respond (e.g., lack of authority, technical expertise, qualified staff, or funding).
- 17. Efforts undertaken to locate owner/operator and pursue an owner/operator-financed cleanup.
- 18. Type of action needed to mitigate or stabilize emergency (if known)

APPENDIX B: ACTION MEMORANDUM

The Regional UST Manager, in conjunction with the OSC, must submit an action memorandum to initiate approval of a written request for Federal action on Indian Lands and within 10 days of an oral request. The action memorandum should address all of the topics outlined in Appendix C of OSWER Directive 9360.0-16A, "State Request Letter Format," in addition to the topics specified below. References to a State or local implementing agency in Appendix C of OSWER Directive 9360.0-16A should be changed to Indian Tribal authority.

The Regional UST Manager and the OSC should attempt to gather the necessary information through Indian Tribal authorities and other sources. If the information necessary to substantially complete the action memorandum is unavailable from these sources, an on-site evaluation/investigation may be undertaken if resources permit.

- I. HEADING
 - DATE: [Month/Day/Year]
 - SUBJECT: Request for Emergency Federal-Lead UST Corrective Action on Indian Lands [Site] ACTION MEMORANDUM
 - FROM: [Regional Administrator]
 - TO: [Director, OERR]

II. BACKGROUND

The background section should contain information on the location of the site, the nature of the incident (including the history of the site, general character of the site, and issues relevant to petroleum management), quantity and types of petroleum substances present, Indian Tribal authority's role, the cleanup time frame, and actions to date, including previous and current actions to abate the threat. For specific instructions on these sections see OSWER Directive 9360.0-16A, Appendix C

- A. <u>Location Description</u>
- B. <u>Site and Incident Characteristics</u>

C. Quantity and Types of Petroleum Substances Present

D. Indian Tribal Authority Capacity

- 1. Describe the capacity of the appropriate Indian Tribal authority to oversee/conduct corrective actions.
- Discuss the appropriate Tribal authority's past experience with UST corrective actions. Has the Tribal authority ever overseen/conducted an UST cleanup or petroleum spill? If so, describe the incident and cleanup activities undertaken.
- Discuss the Indian Tribal authority's experience with other EPA, State, or local environmental programs
- Discuss the Indian Tribal authority's experience in overseeing contractors.

E. <u>Actions to Date</u>

IV. CRITERIA MET FOR TIME-CRITICAL FEDERAL-LEAD CORRECTIVE ACTION ON INDIAN LANDS

Strict criteria must be met to justify Federal response to a petroleum release from an underground storage tank on Indian Lands. The Indian Tribe must be unable to respond, the owner or operator must be unable or unwilling to provide adequate response, and, in addition, the release must pose an immediate or near-term (within six months) substantial threat to ``uman health and the environment, thus indicating a time-critical situation. I time-critical situation exists if:

- The release poses an immediate or near-term substantial threat of direct human, animal, or food chain exposure to petroleum;
- The release poses an immediate or near-term threat of fire or explosion;
- The release poses an immediate or near-term substantial threat to public drinking-water supplies; or
- The release poses and immediate or near-term threat to a significant population or substantial amounts of property, or poses substantial threats to natural resources.

Specific reasons why the site meets the above criteria should be discussed in this section.

- V. ATTEMPTS MADE AT SECURING OWNER/OPERATOR CLEANUP See OSWER Directive 9360.0-16A, Appendix C.
- VI. EXPECTED CHANGE IN THE SITUATION SHOULD NO ACTION BE TAKEN OR SHOULD ACTIONS BE DELAYED.

See OSWER Directive 9360.0-16A, Appendix C.

VII. TYPE OF ACTION REQUESTED

See OSWER Directive 9360.0-16A, Appendix C.

Definition of "Liquid Waste"

Key Words: Waste Analysis, Liquid Waste, Free Liquids

Regulations: 40 CFR 260.10(a)(25), 261.21

Subject: Definition of "Liquid Waste"

Addressee: Chris Howell, Chemical Processors, Inc, 5501 Airport Way south Seattle, WA.

Originator: David Friedman, Manager, Waste Analysis Program, Hazardous and Industiral Waste Division

Source Doc: #9432.01(81)

Date: 6-28-81

Summary:

A liquid is any material that will pass through a 0.45 micron filter at a pressure differential of 75 psi. If the material to be evaluated consists of two or more phases, then the phases should be separated by centrifugation or other means prior to evaluating whether any of the phases meet the above definition. Free liquids as defined in §260.10 (a)(25) are defined as any liquid which passes through the Paint Filter Test (method 9095).



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON. D.C. 20460 OFFICE OF SOLID WASTE

JUN 2 C 13C.

OFFICE OF WATER AND WASTE MANAGEMENT

Mr. Chris Howell Chemical Processors, Inc. 5501 Airport Way south Seattle, WA 98108

Dear Mr. Howell:

I am in receipt of your letter of June 1, 1981 to Mr. Ken Schuster regarding your request for a working definition of a "liquid" waste.

As you are aware, the Agency is actively working to develop improved laboratory procedures for defining both "ignitable" and "liquid." In the interim you may employ the following working definition of a "liquid" when evaluating wastes:

A liquid is any material that will pass through a 0.45 micron filter at a pressure differential of 75 psi. If the material to be evaluated consists of 2 or more phases then the phases should be separated by centrifugation or other means prior to evaluating whether any of the phases meet the above definition. "Free liquids" as defined in \$260.10(a)(25) are a subset of this broader class of liquids.

Any waste on phases of a waste found to meet the above definition of a "liquid" should then be evaluated for ignitability using the procedures in §261.21. All such wastes which contain or consist of liquids which have a flash point below 60°C are to be considered as ignitable wastes.

I hope this response serves to answer your question. If you would like further information please feel free to give me a call at (202) 755-9187.

Sincerely,

David Friedman Manager Waste Annlysis Program Hazardous and Industrial Waste Division (WH-565)

Determining If the Soils from Missouri Dioxin Sites are Hazardous

DOC: 9441.01(84)

Regulations: Subject: Determining if the Soils from Missouri Dioxin Sites are Hazardow Addressee: David Wagoner, Director, Air and Waste Management Division, Region VII Originator: John H. Skinner, Director, Office of Solid Waste Source Doc: \$\mathcal{P}9441.01(84) Date: 1-6-84	Key Words:	Contaminated Soil, Dioxin
Subject:Determining if the Soils from Missouri Dioxin Sites are HazardowAddressee:David Wagoner, Director, Air and Waste Management Division, Region VIIOriginator:John H. Skinner, Director, Office of Solid WasteSource Doc:#9441.01(84)Date:1-6-84	Regulations:	
Addressee: David Wagoner, Director, Air and Waste Management Division, Region VII Originator: John H. Skinner, Director, Office of Solid Waste Source Doc: #9441.01(84) Date: 1-6-84	Subject:	Determining if the Soils from Missouri Dioxin Sites are Hazardous
Originator: John H. Skinner, Director, Office of Solid Waste Source Doc: #9441.01(84) Date: 1-6-84	Addressee:	David Wagoner, Director, Air and Waste Management Division, Region VII
Source Doc: #9441.01(84) Date: 1-6-84	Originator:	John H. Skinner, Director, Office of Solid Waste
Date: 1-6-84	Source Doc:	#9441.01(84)
	Date:	1-6-84

Summary:

To determine if a soil, in which toxic compounds are present, is a RCRA hazardous waste, the origin of the toxicants must be known. If the exact origin of the toxicants is unknown, the soil is not considered RCRA hazardous unless it exhibits one or more of the characteristics of RCRA hazardous waste.

JAN - 6 1934

HEMORANCUM

- SUBJECT: Soils from Missouri Dioxin Sites
- FROM: John H. Skinner, Director Office of Solid Waste
- TO: David Wagoner, Director Air and Waste Management Division, Region VII

We have reviewed the results of the analytical program for soils from Missouri dioxin sites, in response to your request for an interpretation is to whether or not these soils are RCRA hazardous wastes.

The analyses indicate the presence of a number of toxic compounds in many of the soil samples taken from various sites. However, the presence of these toxicants in the soil does not automaticall; make the soil a RCRA hazardous waste. The origin of the toxicants must be known in order to determine that they are derived from a listed nazardous waste(s). If the exact origin of the toxicants is not known, the soils cannot be considered RCRA hazardous wastes unless they exhibit one or more of the characteristics of hazardous waste (i.e., ignitability, corrosivity, reactivity, or extraction procedure toxicity).

If there are any questions, please contact Matt Straus in the Waste Identification Branch (PTS 382-4770).

DFagan:dmf:1/5/84:disk Pagan 6

Definition of "Treatment"

•

DOC: 9432.01(85)

.ey Words: Treatment, Containers, Authorized States

Regulations: 40 CFR 260.10

Subject: Definition of Treatment

Addressee: Robert F. Walter, Office of Environmental Health and Safety, Virginia Commonwealth University, Box 112, Richmond, VA 23298

.

Originator: John H. Skinner, Director, Office of Solid Waste

Source Doc: #9432.01(85)

Date: 6-26-85

Summary:

The letter clarifies the definition of "treatment" under RCRA. In light of that definition, the puncturing, crushing, or shredding of containers holding hazardous waste does not constitute treatment as defined under RCRA.

/margaret/rm 82348/3824

Mr. Robert F. Walter Office of Environmental Health and Safety Virginia Commonwealth University Box 112 Richmond, Virginia 23298

Dear Mr. Walter:

locure

This is in response to your letter of June 17, 1965, in which you question whether shredding of containers holding toluene constitutes "treatment" of a waste under REM.

The Environmental Protection Agency (EPA), has taken the Socilion that panoturing, drushing, or shredding of containers dues not constitute treatment. As defined under BER, treating dues with changing the state of the water the shredding operation dhanger the physical character of the container, and the watethis interpretation is further explained in the enclosed letter to Mr. Bierisin, Counsel for Safe Transportation of Harardour Articles, responding to questions on the same subject. Of course, this interpretation applies only to those States where the Federal MCRA regulations apply. It is intended that the States take is operation, that State's rules and regulations prevail is lies of the Federal program. Virginia was granted final authorization that supply to your wester.

If you have any further questions, please contact Alan Corson. of an staffy he may be reached at 202/382-4770.

Sincerely,

John H. Skinner Director Office of Solid Waste

Interim RCRA/CERCLA Guidance on Non-Contiguous Sites and On-Site Management of Waste and Treatment Residue



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C. 20460

MAR 2 7 1986

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

9347.0-1

MEMORANDUM

SUBJECT: Interim RCRA/CERCLA Guidance on Non-Contiguous Sites and On-Site Magagement of Waste and Treatment Residue FROM: J. Winston Porter Assistant Administrator

TO: Regional Administrators Regions I - X

Region VI has recently raised several RCRA/CERCLA interface issues that have broad implications for remedial actions at many other Superfund sites. The purpose of this memorandum is to lay out EPA policy on several of these issues, including:

- Combined treatment of CERCLA waste from non-contiguous locations;
- 2. On-site disposal of treatment residue;
- 3. Limitations on the construction of hazardous waste incinerators for on-site CERCLA use; and
- 4. Off-site treatment of waste and redisposal on-site.

This memorandum and attachment represent interim guidance which should be used now, but will be refined following regional review. Please submit your comments on this interim guidance to Betsy Shaw (FTS 382-3304) of the Hazardous Site Control Division, Office of Emergency and Remedial Response by April 28, 1986. We are particularly interested in comments which address the implications of this guidance for Superfund removal actions at both NPL and non-NPL sites.

Select RCRA/CERCLA Issues:

1. <u>Combined treatment and/or disposal of CERCLA waste from</u> non-contiguous NPL sites

NPL sites may be combined for remedial action if the following statutory criteria are met: the sites must be geographically close or pose similar threats to public health and the environment (CERCLA \$104 (d)(4)). If combined remedial actions will involve the transport of waste from one site to another site, the wastes must be compatible for the selected treatment or disposal method and managed in a manner that is part of the highly reliable long-term remedy selected for that site or group of sites. Combined remedies must be cost -effective and should not result in any significant additional short-term impacts on public health and the environment at the receiving site. As in every case, CERCLA waste which is transported must be manifested. The Record of Decision (ROD) for a remedial action that involves more than one site should state that several sites are being treated as one and that their combined treatment constitutes on-site action. (See attachment.)

2. On-site management of waste and treatment residue

EPA interprets CERCLA to require that off-site treatment, storage and disposal of hazardous wastes comply with all RCRA requirements, including permitting. With respect to on-site disposal, the National Contingency Plan (50 FR 47912, November 20, 1985) requires that CERCLA activities meet the technical requirements of RCRA (and other Federal environmental requirements) that are applicable or relevant and appropriate¹ while the procedural requirements, such as permitting, need not be met.

Waste and treatment residues may be managed on-site in several ways. The approach selected will depend on the cost-effectiveness analysis at each site. One approach is to remove the waste (and treat if desired) and dispose of the waste and/or treatment residue in a new on-site land disposal unit. This unit would meet the technical RCRA Subtitle C land disposal requirements of 40 CFR Part 264 (e.g. §264.301 design and operating requirements, and land disposal closure and post closure care requirements in §264.310).

^{1 &}quot;Applicable requirements" are those Federal requirements that would be legally applicable if the response actions were not undertaken pursuant to CERCLA \$104 and \$106. "Relevant and appropriate requirements" are those Federal requirements that, while not applicable, are designed to apply to problems sufficiently similar to those encountered at CERCLA sites that their application is appropriate.

The second approach allows waste to be removed, treated and the residuals to be replaced in the area from which they originated. The area would then be capped and monitored consistent with the technical requirements of land disposal closure (\$264.310). Under this approach, a double liner/ leachate collection system would not be required if the wastes are removed during closure for the purpose of treating them to enhance the effectiveness of the closure.

A third approach requires no further management of waste or treatment residue if the waste can be evaluated, determined to be non-hazardous and delisted. This would normally entail preparing a delisting analysis using the Vertical and Horizontal Spread (VHS) model (50 FR 48886, November 27, 1985) or other similar generic models that do not consider site specific factors. A delisting petition is not required for on-site CERCLA actions.

Finally, the National Contingency Plan (40 FR 47947 -47948) provides for selection of a remedy that does not attain applicable or relevant and appropriate requirements if: 1) the alternative is only an interim remedy; 2) the need to use the Fund at other sites outweighs the need to implement a remedy that fully attains all requirements; 3) it is technically impractical to implement a remedy that meets all applicable or relevant and appropriate requirements; 4) meeting all such requirements will result in an unacceptable environmental impact; or 5) there is an overriding public interest related to enforcement.

The determination that RCRA requirements for treatment, storage and disposal will be met should be made during the Remedial Investigation and Feasibility Study (RI/FS). In the case of incinerator residue, a waste analysis should be conducted during the RI to provide the necessary data. Subsequent analyses, including a test burn, may be conducted during Remedial Design (RD) as appropriate on a case by case basis. Assurance of the consistency of the remedy with RCRA and other applicable or relevant and appropriate Federal requirements should be presented in the ROD, and, if appropriate, reviewed again during RD.

3. Limitations on the construction of hazardous waste incinerators for on-site CERCLA use

If an incinerator is to be constructed for on-site remedial action, there should be a clear intent to dismantle or remove the unit after the CERCLA action is completed. Dismantling or removal should be a part of the remedy presented in the ROD and funds should be included in the financial or contractual documents. Should there be plans to accept commercial waste at the facility after the CERCLA wastes have been treated or destroyed, it is EPA policy that a RCRA permit be obtained before the unit is constructed. (See attachment.)

.

4. Off-site treatment of waste and redisposal on-site

On-site disposal may involve transport of waste off-site for treatment or storage if the CERCLA waste or treatment residue is ultimately disposed of at the site of waste origin. For this activity, the CERCLA waste is manifested to and from the site and maintained separately throughout all off-site activities.

If you have any questions regarding this memorandum or attachment, please call Betsy Shaw or Bill Hanson (FTS 382-2345).

Attachment

Attachment: Interim RCRA/CERCLA Guidance on Non-Contiguous Sites and On-Site Management of Waste and Treatment Residue

Combining Hazardous Waste Sites for Remedial Action

Background:

Several situations have arisen where it may be advantageous to combine several NPL sites together for the purpose of conducting a more effective remedial action. Subject to the requirements in CERCLA §104 (d)(4), sites in proximity to one another, sites with similar wastes, and sites with the same PRPs may be good candidates for combined remedial actions. A treatment system or incinerator, for example, may be more efficient treating wastes from several sites. Expected economies of scale would lower the unit costs and favor more reliable technologies. Overall, protection of public health and the environment may increase if the waste of several smaller sites are combined at a central treatment or disposal location.

<u>Legislative Authority:</u> Section 104(d)(4) of CERCLA states that non-contiguous sites may be treated as one site when the separate sites are reasonably related on the basis of:

- 1) Geography; or
- 2) Threat or potential threat to public health and the environment.

Cost-Effective Reasons for Combining NPL Sites for Remedial Action

Several different circumstances may occur that favor combining site remedial actions.

- Example 1: Incineration is effective for destroying wastes at several closely arrayed sites. One alternative is to use a mobile incinerator at each site. Another alternative that may be cost effective is to incinerate the wastes of several sites at one location. The residue could be disposed at the original site but, again, it would probably be more cost-effective to dispose of all ash at the same location.
- Example 2: Construction of a new on-site land disposal facility has been found to be cost effective at site A. Wastes at nearby site B are similar in character and a small quantity needs to be managed.

Site B wastes could be managed on-site but it could be less expensive and more effective to dispose of the waste at Site A. Example 3: Site A and Site B have similar wastes and are close to one another. RCRA closure with a cap has been found to be cost effective at both sites. It may be cost effective to design and remediate both sites at the same time. Therefore, the State or Region would like to contract with one design firm and one construction company to undertake both remedies.

Regions should identify opportunities to combine RI/FSs for several NPL sites in the Site Management Plan or other preremedial activities. Combining RI/FSs may improve the timing and effectiveness of remedial actions and should be shown in the Superfund Comprehensive Accomplishments Plan (SCAP).

Criteria for Treating Non-Contiguous Sites as One

The September 21, 1984 NPL listing (40 FR 37076) provides the flexibility to respond to several sites listed separately on the NPL with a single response if the statutory factors are met and it appears cost-effective to do so.

The following criteria would be used to treat non-contiguous sites as one when transportation of the waste is involved:

- 1. Sites are reasonably close to one another;
- 2. Wastes must be compatible for the selected treatment or disposal approach;
- 3. Wastes that are transported to another site need to be managed in a manner that is part of a highly reliable, long-term remedy; 1 and
- Incremental short-term impacts (e.g. sudden releases, 4. fugitive dust and fumes) to public health and the environment at the receiving site will be minimal. (This factor is important when the receiving site is located near a residential community.)

Of course, the remedy must also be cost-effective by either costing less or by providing increased or more reliable protection of public health and environment than two separate remedies. _

When short-term impacts are found to be significant, combining sites may be determined to be inappropriate and the remedy may be reconfigured. Options include but are not limited to:

This type of remedy generally is defined as: 1

- a. Requiring little or no long-term active O/M;

b. Relatively low probability of release to the environment;c. If a release did occur, it would not endanger public health or the environment.

- Use another hazardous waste site where there would be fewer impacts;
- Pretreat wastes at the original site locations
 (e.g., metal extraction) or improve materials handling
 procedures;
- Dispose of treated residuals (e.g., incineration ash) at originating sites.

If incremental short-term impacts are significant and cannot be mitigated, then non-contiguous sites should not be treated as one for the purpose of combined treatment or disposal regardless of cost-effectiveness.

CERCLA Compliance with Other Environmental Laws

Under response actions occuring at non-contiguous sites which are treated as on-site actions, Superfund or PRPs under an EPA approved enforcement action would:

- Manifest hazardous wastes transported to another site;
- Meet the applicable or relevant and appropriate technical requirements of RCRA TSD facilities but would not be required to obtain RCRA permits.
- Limitation: The cost of dismantling or removing a treatment or storage unit constructed as part of an on-site remedy should be factored into the determination of the tost-effectiveness of that remedy. If that alternative is selected, funds for the dismantling of the unit should be included in the remedy obligation. Should there be plans for a treatment or storage unit constructed as part of an on-site remedy to accept commercial wastes after the CERCLA waste has been processed, it is EPA policy that a RCRA permit be obtained before the unit is constructed. The cost and scheduling implications of obtaining a permit should also be factored into the analysis of cost-effectiveness.

Proposed Implementation Process:

- 1. Initial evaluation of NPL sites to determine if the RI/FSs of several sites should be combined. Show combined RI/FSs on SCAP.
- Feasibility Study recommends that a combined site action would be cost-effective. Further, the Feasibility Study shows that the selected remedy meets the necessary criteria of this policy. (The NPL need not be amended.)

.

- 3. A joint public comment period is held to seek comment from all interested parties on the proposed consolidation of sites and a responsiveness summary is written.
- 4. Regional Administrator or Assistant Administrator signs Record of Decision for non-contiguous site action.
- 5. A new Record of Decision, public comment period and responsiveness summary would be required if additional sites are added to the response plan after the first Record of Decision.

Responsibilities of Regional RCRA Off-Site Disposal Contacts



OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

- SUBJECT: Responsibilities of Regional RCRA Off-Site Disposal Contacts
- FROM: Jerry Kotas, Chief Compliance & Implementation Branch
- TO: RCRA Enforcement Branch Chiefs Regions I-X

It has come to our attention that CERCLA personnel have experienced some difficulty in obtaining timely facility compliance status information from Regional RCRA Off-Site Disposal Contacts (RROCs). CERCLA off-site disposal decisions related to emergency cleanups were delayed because the RROC was on travel, on vacation, or otheswise unavailable, and no other person in that Region had been designated to act as a substitute for the RROC. On one occasion, it took a CERCLA On-Scene Coordinator several days to make contact with an RROC. In light of this, please designate an alternate RROC and provide me with the names, addresses, and phone numbers of both your lead and alternate RROCs within 2 weeks. Please note that, as the names of designated RROCs or their alternates change, you should notify this office as soon as possible.

Please be reminded that each Region is responsible for providing timely facility compliance status information to CERCLA response personnel. If the designated RROC is unavailable, the alternate should be available to respond to such requests for information. Secretarial staff should be informed that such requests should be addressed to the alternate. This will assure that off-site disposal of CERCLA wastes will proceed as quickly as possible, especially in the case of emergency actions, and in keeping with the mandates of the CERCLA and RCRA programs.

In addition to the above, an upcoming initiative related to the adoption of the "Procedure for Planning and Implementing Off-Site Response Actions" (Off-Site Policy) by other Federal agencies is expected to increase the frequency of inquiries that RROCs will receive regarding facility compliance status. This will make the designation of an alternate, to respond in the absence of the RROC, even more critical. As many of you know, GAO recommended in a recent report that EPA encourage other Federal agencies to adopt the Off-Site Policy for disposal of response-related and other hazardous wastes. That recommendation is based upon the premise that no Federal wastes should be sent to facilities which may pose current or future threats to health or the environment.

The Assistant Administrator for OSWER has endorsed that recommendation, and discussions with other agencies are underway on this topic. Representatives from other agencies will be given the names of the RROCs and alternates as the source of information on "eligibility" of facilities within the Region. It is important that the RROC or the alternate be available for representatives from other Federal agencies, as well as for EPA personnel managing CERCLA response actions.

There are several other activities related to the Off-Site Policy in progress, and I would like to briefly summarize them for you. In the very near term, interim guidance will likely be issued from the Assistant Administrator on whether and how to provide notice, and an opportunity to confer, to facilities which are deemed ineligible under the Policy. Final revisions to the policy (including any changes to the notice procedures) will be issued in May, 1986 or therabouts. OSWER will also prepare guidance and forms for maintaining and reporting information related to both facility status and disposal location for CERCLA wastes.

We will try to keep you informed as to the status of each of these efforts. If you need any further information on the new initiatives mentioned above, please contact Mike Kilpatrick at (FTS) 382-4812.

Thank you for your cooperation in this matter.

cc: Regional RCRA Off-Site Contacts, Regions I-X
Lee Herwig, OFA
Mike Kilpatrick, OWPE
Tim Fields, ERD
Superfund Branch Chiefs, Regions I-X

Use of Liquids for Wind Dispersal Control at Hazardous Waste Landfills

United States Environmental Protection Agency Office of Solid Waste and Emergency Resource

•

DIRECTIVE NUMBER: 9487.00-1A			
TITLE: Use of Liquids for Wind at Hazardous Waste Land	Dispersal Control fills		
APPROVAL DATE:			
EFFECTIVE DATE:			
ORIGINATING OFFICE:			
🖾 FINAL			
C DRAFT			
STATUS: -			
REFERENCE (other documents):			



.

Contraction of the second

UNITED STATES ENVIR INMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

: APR 2 1 1986

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

OSWER POLICY DIRECTIVE 19487.00-1A

MEMORANDUM

- SUBJECT: Use of Liquids for Wind Dispersal Control at Hazardous Waste Landfills
- FROM: Marcia Williams, Director Muuch Office of Solid Waste (WH-562)
- TO: Robert L. Duprey, Director EPA Region VIII Waste Management Division (8-HWM)

This is in response to your request received in November 1985 for guidance concerning the question of under what conditions, if any, it is acceptable to use water or other liquid chemical stabilizers to control wind dispersal of waste in a landfill cell. We believe that the use of nonhazardous liquids for wind dispersal control at hazardous waste landfills should not be subject to the restrictions under Section 3004(c)(3) of HSWA. This use must, of course, be limited to amounts necessary to comply with wind dispersal control requirements. Such amounts should be determined by regulatory authorities on a case-by-case basis.

As stated in your memorandum, Sections 264.301(f) and 265.302(d) require the owner or operator of a landfill containing hazardous waste that is subject to wind dispersal to cover or otherwise manage the landfill to control such dispersal. Since the liquids that are used to control wind cispersal are usually nonhazardous (e.g., water), a response to your question is contained in a guidance we have drafted concerning Section 3004(c)(3) (which addresses the placement - --of nonhazardous liquids in hazardous waste landfills) of the 1984 Hazardous and Solid Waste Amendments (HSWA) to RCRA. This draft guidance will be sent to the regional offices in the near future. OSWER POLICY DIRECTIVE \$9487.00-1A

We believe that the language and legislative history of Section 3004(c)(3) specifically, and of Section 3004(c) generally, indicate that Congress' primary concern in banning liquids was to control the placement of liquids in landfills for treatment, storage, and disposal. We believe, further, that Congress did not intend to require owners and operators to apply for an exemption for uses of nonhazardous liquids in or near a landfill that are necessary in order to comply with the technical requirements of the RCRA regulations.

You also inquired in your memorandum how wind dispersal control, including the use of liquid agents, was being managed at other sites nationally, and under what conditions. It is unfortunate that we have little information concerning the national management of wind dispersal. We do know, however, that the use of water appears to be a common management practice for control of wind dispersal for dust and particulate matter. Other methods include waste containerization, use of cover material (soil and other waste), and waste treatment before disposal (e.g., chemical fixation, carbon adsorption).

We hope this response clarifies the issue. If you should have any additional comments or concerns, please contact Paul Cassidy, of my staff, at FTS-382-4682.

Let me say finally that we were very impressed with your full and perceptive analysis of the issue of the limited use of liquids for controlling wind dispersal.

cc: EPA Regions I - VII and IX - X

3

2

Consideration of RCRA Requirements in Performing CERCLA Responses at Mining Waste Sites

UNITED STATES ENVIRONMENTAL PROTECT ON AGENC

WASHINGTON DC 20160

9234.0-4

- : -

AUG 19 1986

MEMORANDUM

- SUBJECT: Consideration of RCRA Requirements in Performing CERCLA Responses at Mining Waste Sites
- FROM: Henry L. Longest II, Director Watter Hand

TO: Waste Management Division Directors Regions I - X

As you know, on July 3, 1986, the Agency issued a final determination on whether mining waste would be regulated under Subtitle C of RCRA (copy attached). This determination was based on a report to Congress mandated by RCRA Section 3001(b) (3)(C) and subsequent public comments. The determination is that mining wastes will not be regulated under Subtitle C at this time. This conclusion is based on the belief that several aspects of EPA's current hazardous waste management standards if applied universally to mining sites, are likely to be environmentally unnecessary, technically infeasible, or economically impractical.

However, given the concern about actual and potential mining waste problems, the Agency intends to develop a program for regulating mining waste under Subtitle D. The current Subtitle D program establishes criteria principally aimed at municipal and industrial solid waste which focus on standards related to surface water discharges, groundwater contamination and endangered species. Modifications to this program will focus on identifying environmental problems, setting priorities for applying controls at sites with a high potential for risk, and employing a risk management approach in the development of appropriate standards to protect human health and the environment, as necessary, including closure options, tailored controls, pretreatment of wastes prior to disposal, and cleanup options. Revisions to Subtitle D criteria are expected to be proposed in mid-1988; however, EPA has reserved the option to reexamine a modified Subtitle C in the future if this approach is unworkable or insufficient.

In the interim, Superfund will continue to address mining waste problems through the RI/FS and ROD/EDD processes taking into account current Subtitle D requirements as well as options for addressing risks not addressed by Subtitle D requirements. To address such remaining risks, you may wish to consider the technical requirements of Subtitle C regulations during the initial review of remedial alternatives. If these requirements seem to be technically infeasible, they may be rejected early in the screening process. If Subtitle C approaches appear to satisfy the criteria found in Section 300.68 (g), Initial Screening of Alternatives, of the NCP, they should be considered in the detailed analysis. Other remedial alternatives should be evaluated in a risk management analysis. In some cases, a combination of Subtitle C and risk analysis approaches may be used to address a discrete phase of response. All data generated during remedial planning, including the basis for selection of specific remedies, should be forwarded to my office as it becomes available so that the information can be transmitted to OSW to assist that office in its development of standards for mining wastes.

Attachment

cc: Marcia Williams, OSW Gene Lucero, OWPE Dan Berry, OGC


Thursday July 3, 1986

Part V

Environmental Protection Agency

40 CFR Part 261 Regulatory Determination For Wastes From the Extraction and Beneficiation of Ores and Minerals



ENVIRONMENTAL PROTECTION AGENCY

40 CFR Part 261

(FRL 3033-7)

Regulatory Determination for Wastes from the Extraction and Beneficiation of Ores and Minerals

•

AGENCY: Environmental Protection Agency_ _ -۰. _ ACTION: Regulatory determination.

SUMMARY: This is the regulatory determination for solid waste from the extraction and beneficiation of ores and minerais required by section 3001(b)(3)(C) of the Resource Conservation and Recovery Act (RCRA). This section of RCRA requires the Administrator to datermine whether to oromugate regulations under Subutle C of the Act for these wastes or determine that such regulations are unwarranted: the Administrator must make this determination no later than six months after completing a Report to Congress on these wastes and after public nearings and the opportunity to comment on the report_After completing these activities and reviewing the information available, the Agency has determined that regulation of the wastes studied in the Report to Congress, i.e., wastes from the extraction and beneficiation of ores and minerals. under Subtitle C is not warranted at this time.

ADDRESS: The address for the Headquarters docket is: United States Environmental Protection Agency, EPA RCRA docket (Sub-basement), 401 M street SW., Washington DC, 20460, (202) 475-5327. For further details on what the EPA RCRA docket contains. see Section VIL of this presmble, utiled "EPA RCRA_ Docket" under "SUPPLEMENTARY INFORMATION.".

FOR FURTHER INFORMATION CONTACT: RCRA/Superfund Hotline at (800) 424-9346 or (202) 382-3000 or Dan Derkics at (202) 382-2791.

SUPPLEMENTARY INFORMATION:

Praymaic Outline

L Summary of Decision

- IL Background
- III. Legal Automity
- IV. Report to Congress

V. Application of Subtitle C to Mining Waste VI. Application of Subtitle D to Mining Waste VIL EPA RCRA Docket

Supplementary Information

I. Summery

Based on the Report to Congress. comments on the report. and other

available information, EPA has determined that regulation of mining ... waste under Subtitle C of the Resource Conservation and Recovery Act (RCRA) is not warranted at this time.

This conclusion is based on EPA s pener that several aspects of EPA s current hazardous waste management standards are likely to be environmentally unnecessary. imoractical when applied to mining waste. While under existing law EPA: would have some flexibility to modify its standards for nazardous waste ÷ . management as appued to these wastes. there are substantial questions about - - -----whether the flexibility innerent in the ---statute coupled with the Agency's current data on these wastes provide a . sufficient pasis for EPA to develop a mining waste program under Subtitle C that addresses the risks presented by mining waste while remaining sensitive to the unique practical demands of mining operations. Given these uncertainties. EPA does not intend to impose Subtitle C controis on mining waste at this time.

The Agency, nowever, is concerned . about certain actual and potential mining waste problems, and therefore ... plans to develop a program for mining waste under Subtitle D of RCRA. The long-term effectiveness of this program depends on available State resources for designing and implementing a program tailored to the needs of each State, and on EPA's ability to oversee and enforce the program. As noted below in section VI. EPA will be working with the States to determine the specific nature of their current mining waste activities and their future plans to administer such programs. The Administration will work with Congress to develop expanded Subtitle D authority (i.e., Federal oversight and enforcement) to support an effective State-implemented program for mining waste. EPA has already made preliminary contacts with Congress and intends to hold detailed discussions on the specifics of the Subtitle D program in the coming year. In the interim, EPA will use RCRA secuon 7003 and CERCLA sections 104 and 106 to protect against suostantial threats and imminent hazards. If EPA's unable to develop an effective mining waste program under Subutle D. the Agency may find it necessary to use Subtitle C authority in the future.

II. Background

Section 8002(f) of the Resource Conservation and Recovery Act of 1978 cirected EPA to conquet:

A detailed and comprehensive study on the adverse effects of soud wastes from active

and apandoned surface and underground mines on the environment, including, but not limited to, the effects of such wastes on numans, water, air, nealth, welfare, and natural resources, and on the adequacy of means and measures currently employed by the mining industry. Government agencies, and others to dispose of and utilize such solid wastes to prevent or substantially mitigate suca adverse effects.

do

1. The Sources and volume of ¹ discarded material generated per yearfrom muning

2. Present disposal practices:

3. Potential danger to human health and the environment from surface runoff of leachate and air collution by dust

4. Alternatives to current disposal methods

5. The cost of those alternatives in terms of the unpact on mine product costs: and

6. Potential for use of discarded material as a secondary source of the mine product

On May 19, 1980, EPA promuigated regulations under Subtitle C of RCRA which covered, among other things. "solid waste from the extraction," beneficiation, and processing of ores and minerals." i.e., mining weste. On October 21, 1980, just before these Subtitle C regulations became effective. Congress enacted the Solid Waste Disposal Act of 1980 (Pub. L 96-482) which added section 3001(b)(3)(A)(ii) to RCRA. This section prohibits EPA from reguiating "solid waste from the extraction, beneficiation, and processing of ores and minerais, including phosphate rock and overburden from the mining of uranium ore" as hazardous waste under Subtitle C of RCRA until at least six months after the Agency completes and submits to Congress the studies required by section 8002(f), and by section 8002(p) (which was also added to RCRA by the 1980 amendments).

Section 8002(p) required EPA to perform a comprehensive study on the disposal and utilization of the waste excluded from regulation. i.s., solid waste from the extraction, beneficiation. and processing of ores and minerals. including phosphate rock and overburgen from the mining of uranium cre. This new study, to be conducted in conjunction with the section 8002(f) study, mandated an analysis of:

1. The source and volumes of such materials generated per year.

2. Present disposal and utilization stactices:

2 Potential danger, if any, to human nealth and the environment from the disposal and reuse of such materials:

 Documented crises in which danger to human health or the environment has been proved:

5. Alternatives to current disposal methods:

6. The costs of such alternatives;

7. The impact of these alternatives on the use of prospnate rock and uranium ore, and other natural resources; and

8. The current and potential utilization of such materials.

The 1980 amendments also added section 3001(b)(3)(C), which requires the Administrator to make a "regulatory determination" regarding the waste excluded from Subttle C regulator. Soecifically within six months after submitting the Report to Congress, and after holding public hearings and taking public comment on the report, the Administrator must "determine to promulgate regulations" under Subttle C of RCRA for mining waste or determine that such regulations are unwarranted."

EPA was required to complete the study and submit it to Congress by October 16, 1983. In 1984, the Concerned Citizens of Adamstown and the Environmental Defense Fund sued EPA for failing to complete the section 8002 studies and the regulatory determination by the statutory deadlines. The District Court for the District of Columbia ordered EPA to complete the studies by December 31, 1985, and to publish the regulatory determination by june 30, 1986.

EPA submitted its Report to Congress on mining waste on December 31, 1985. A notice announcing the availability of the report, and the dates and locations of public hearings, was published January 8. 1986 (51 FR 777). EPA heid public nearings on the report in Tucson, Arizona on Marca 6, 1986: Washington. DC on Marca 11, 1986; and Denver. Colorado on Marca 13. 1986. The comment period on the report closed March 31. 1986. This nonce constitutes the Agency's regulatory determination for the wastes covered by the Report to Congress, Le, wastes from the extraction and beneficiation of ores and minerals.

On October 2, 1986, EPA proposed to narrow the scope of the mining waste exclusion in RCRA section 2001(b)(3)[A)(n), as it about to processing wastes (50 FR 40292). Under this proposal, wastes that would no ionger be covered by the mining waste exclusion would be subject to Subtile C f they are nazardous. These

" reinterpreted" wastes were not studied in the mining waste Report to Concress and therefore, are not covered by this regulatory determination.

III. Legal Authority

EPA has concluded that its decision whether to regulate mining waste under Subtitle C should be based not just on whether mining waste is hazardous (as currently defined by EPA regulations) out also should consider the other factors that section 8002 required EPA to study. The basis of this conclusion is the language of section 3001(b)(3)(A) which states that the regulatory determination must be "based on information" ceveloped or accumulated pursuant to (the section 8002 studies), public hearings, and comment. . . ." Clearly, Congress envisioned that the determination would be based on all the factors enumerated in sections 8002 (f) and (p). Congress already knew that some mining waste was hazardous. since the RCRA Subtitle C regulations which were promutgated on May 19. 1980 were to apply to hazardous (both characteristic and listed) mining waste. Congress apparently believed, however, that EPA should obtain and consider accitional information, not just data on which types of mining waste are hazardous, before imposing Subutle C regulation on these wastes. Accordingly, this regulatory determination is based on consideration of the factors listed in sections 8002 (f) and (p).

In reviewing the factors to be studied which are listed in sections 8002 (f) and (2), and the legislative nistory of these and other mining waste provisions. EPA has concluded that Congress believed that certain factors are particularly important to consider in making the Subtile C regulatory determination. First. Congress instructed EPA to study the potential dangers to human health and the environment from mining waste. indicating that the decision to regulata under Subutle C must be based on a finding of such a danger. Second. section 8002(p) required EPA to review the actions of other Federal and State agencies which deal with mining waste with a view toward avoiding duplication of effort." From this provision. EPA concludes that Congress believed Subtitle C regulation might not be necessary if other Federal or State programs control any risks associated with mining waste. Third. Congress expected EPA to analyze fully the cisposal practices of the mining industry which, when resc in comunction with the legislative history of this provision. indicates concern about the feasibility of Subtitle C controis for mining waste. Finally, Congress instructed EPA to look at the costs of various alternative mernous for mining waste management.

as well as the impact of those alternatives on the use of natural resources. Therefore, EPA must consider both the cost and impact of any Subtitle C regulations in deciding whether they are warranted. Clearly, Congress believed that it was important to maintain a viable mining industry. Therefore, any Subtitle C regulations which would cause widespread closures in the industry would be unwarranted.

IV. Report to Congress

EPA's Report to Congress provides information on sources and volumes of waste, disposal and utilization practices. potential danger to human health and the environment from mining practices. and evidence of damages. EPA received more than 60 written comments on the report and neard testimony at the hearings from more than 30 individuals. A complete summary of all the comments presented at the hearings and submitted in writing is available (ICF. 1986a see VII No. 6]; (see "EPA RCRA Docket"). This section summarizes the information contained in the Report to Congress, public comments received on the report and EPA's response to the comments.

A. Summary of Report to Congress

1. Structure and Location of Mines - -

EPA focused on segments producing and concentrating metallic ores, phosonate rock, and asbestos, totalling fewer than 500 active sites during 1985. These sites, which are predominantly located in sparsely populated areas westrof the Mississippi River, vary widely in terms of size, product value, and volumes of material handled. Several segments are concentrated primarily in one state: The iron segment is mainly concentrated in Minnesota, lead in Missouri, copper in Arizona, asoestos in California, and phosphate in Fiorida.

2 Waste Quantities

The Report to Congress estimated that 1.3 and 2 billion metric tons per year of nonfuel mining waste were generated in 1982 and 1980, respectively. The accumulated waste volume since 1910 from nonfuel mining is estimated to be approximately 50 billion metric tons. The large volume of annual and accumulated nonfuel mining waste results from the sign waste-to-product ratios associated with mining. The fact that most of the material handled in mining is waste and not marketaole product distinguishes mining from many other process industries where waste materials make up a relatively small portion of the materials used to produce

24497

a final product. Consequently, some of the larger mining operations nancie more material and generate more waste than many entire industries.

3. Waste Management Practices

The report indicated that site selection for mines, as well as associated beneficiation and waste disposal facilities, is the single most important factor affecting environmental quality in the mining industry. Most - mine waste is disposed of in piles, and "... most tailings in impoundments. Mine water is often recycled through the mill and used for other purposes onsite. Offsite utilization of mine waste and mill tailings is limited (i.e., 2 to 4 percent of all mining waste generated). Some waste management measures (e.g., source separation, treatment of acids or cvanides, and waste stabilization) now used at some facilities within a narrow segment of the mining industry could be more widely used. Other measures applied to nazardous waste in nonmining industries may not be appropriate. For example, soil cover from surrounding terrain may create aggitional reclamation problems in and . . .regions.

4. Potential Hazard Characteristics

Of the 1.3 billion metric tons of nonfuei mining waste generated by extraction and beneficiation in 1985. about 61 million metric tons (5 percent) exhibit the characteristics of corrosivity and/or EP (Extraction Procedure) toxicity. as defined by 40 CFR 251.22 and 261.24, respectively. Another 23 million metric tons (2 percent) are _ contaminated with cyanide (greater than 10 mg/1). Further, there are 182 million metric tons (14 percent) of copper leach dump material and 95 million metric tons (7 percent) of copper mill tailings with the potential for release of acidic and toxic liquid, i.e., acid formation. There are 443 million metric tons (34 percent) of weste from the phosphate and uranium segments with radioactivity content greater than 5 picocuries per gram: a total of 93 million metric tons (7 percent) has radioacuvity content greater than 20 picocuries per gram. Finally, aspestos mines generated acout 5 million metric tons (less than 1 percent) of waste with a carysotile content greater than 5 percent.

5. Evidence of Damages

To determine what damage might be caused by mining waste. EPA conducted ground-water monitoring and examined documented damage cases. During short-ferm monitoring studies at eight sites. EPA detected seepage from tailings impoundments, a copper leach

dump, and a uranium mine water pond. The EP toxic metals of concern. however, did not appear to have migrated during the 6- to 9-month monitoring period. Other ground-water monitoring studies, nowever...detected sulfates, cyanides, and other contaminants from mine runoff, tailings pond seepage, and leaching operations. The actual numan health and ----environmental threat posed by any of - these receases is largely dependent upon site-specific factors, including a site's _ proximity to human populations or sensitive ecosystems. Sites well removed from population centers. counking water supplies, and surface waters are not likely to pose high risks.

incidents of damage (e.g., contamination of drinking water aquiers. degradation of aquatic ecosystems, fish kills, and related certagation of environmental quality) nave also been documented in the phosphate, gold, silver, cooper, lead, and uranium segments. As of Septemper 1985, there were 39 extraction, peneficiation, and processing sites included or proposed for inclusion on . . the National Priorities List under . -CERCLA (Superfund), including five 😳 gold/siver, three copper, three aspestos, and two lead/zinc mines. The asbestos Superfund sites differ from other sites in that these wastes pose a hazard via aircorne exposure.

6. Potential Costs of Regulation

The Report to Congress presented for five metal mining segments, total annualized costs ranging from S7 million per year (for a scenario that emphasizes primarily basic maintenance and . . monitoring for wastes that are hazardous under the current RCRA criteria) to over \$800 million per year (for an unlikely scenario that approximates a full RCRA Subtitle C regulatory approach. emphasizing cap and liner containment for all wastes considered hazardous under the current criteria, plus cyanide and acid formation wastes). About 60 percent of the total projected annualized cost at active facilities can be attributed to the management of waste accumulated from past production. Those segments with no nazardous waste (e.g., iron) would incur no costs. Within a segment. incremental costs would vary greatly from facility to facility, depending on current requirements of state laws. ore grade, geography, past waste accumulation. percentage of waste which is hazardous, and other factors.

8 Comments Received on the Report to Congress and EPA's Response

1. Potential Hazard Characteristics

EPA received several comments accreasing the magnitude of the wastes generated by the mining industry, and the amount that is nazardous. Many agreed with the report's conclusion that there are substantial volumes of waste, but questioned EPA's estimates of the amount of "hazardous" waste.

Many commenters noted that they believed the EP (Extraction Procedure) - I test is inappropriate for mining waste because the municipal landfill mismangement scenario on which the test is based is not relevant to mining waste. They further noted that the corrosivity charactenstic is not appropriate because it does not address the ouffering capacity of the environment at certain mining sites. Finally, several commenters noted that leaching operations are processes, rather than wastes and are thus outside the purview of RCRA.

The Agency agrees that dump and neao leach piles are not wastes: rather they are raw materials used in the production process. Similarly, the leach liquor that is captured and processed to recover metal values is a product, and . not a waste. Only the leach liquor which escapes from the production process and abandoned heap and dump leach . 1 piles are wastes. Since the report identified 50 million metric tons of heap and dump leach materials as RCRA corrosive wastes. EPA has accordingly reduced its esumate of mining waste volumes which meet the current definition of hazardous waste. The Agency currently estimates that out of the 61 million metric tons per year of mining waste identified as hazardous in the Report to Congress, only 11 million metric tons of mining waste generated annually are hazardous because they exhibit EP toxicity, and an unknown amount of escaped leach liquor is corrosive. EPA has also concluded that potential problems from substantial quantities of mining waste which have other properties. Le., radioacuvity. asoestos, cyanide, or acid generation potential will not be identified by the current RCRA characteristics. EPA. therefore, believes that entirely different criteria may more appropriately identify the mining wastes most likely to be of cancer.

2. Evidence of Damages

EPA received many comments on whether the Report to Congress cemonstrates that mining waste pose a threat to human health and the environment. Many commenters alleged that the report does not demonstrate conclusively that such wastes do pose a threat. They claimed that EPA did not adequately consider the site-specific nature of mining waste management problems. They pointed out that the environmental settings of sites vary widely, as do management practices, and that all these factors influence risk. Also, several commenters noted that the

- report fails to distinguish between the threat from past practices and the threat, if any, from current practices. Based on these ooservations, many of
- these commenters urged EPA to postpone regulations bending additional analysis. However, other commenters noted that they believed there is sufficient evidence that mining waste poses a threat to numan health and the environment and asked for immediate regulatory action, noting that the time for study was over.

The Agency agrees that adverse effects to the public and the environment from the disposal of mining waste is not likely at sites well-removed from population centers, chinking water supplies, surface water, or other receptors. However, for other sites. analyses of contaminant plumes released by leaching operations and releases of other contaminants (e.g., acids, metals, dusts, radioactivity) demonstrate adverse effects. Moreover. the Agency recognizes, as evidenced by the mining waste sites on the National Priorities List, the potential for problems from mining sites. It is apparent that some of the problems at Superfund or other abandoned sites are attributable to waste discosal practices not currently used by the mining industry. However, :: is not clear from the analysis of damage cases and Superiung sites, whether current waste management practices can prevent damage from seepage or sudden releases. EPA is concerned that a large exposure potential exists at some sites generating mining waste. particularly the sites that are close to population centers of in locations _ . conductive to high exposure and risk to human nealth and the environment.

3. Potential Costs of Regulation

EPA received a large number of _________ comments pertaining to the cost of ________ complying with regulations for mining _______ waste, and the effects these compliance costs would have on the mining industry. Many commenters claimed ________ industry. Many commenters claimed ________ that regulating the mining industry would impose costs much greater than those EPA estimated in its Report to Congress. They also noted that the mining industry was depressed, and that for many mines, increased compliance costs would be greater than the profits. leaoing to forced closures.

Many commenters also pointed out that there are current Federal and State regulations which already apply to mining, which impose costs. They noted that EPA needs to review the existing Federal and State regulatory structure before adding to it, thereby imposing additional costs. Others did not agree, commenting that existing Federal and State regulations are inadequate, and that additional EPA regulation is necessary.

EPA is sensitive to the potential costs to the industry associated with mining waste regulations under Suptitle C. The Agency is also cognizant that many EPA programs aiready affect the mining incustry such as the Clean Water Act which, among other things, control surface water discharge via national Pollutant Discharge Elimination system (NPDES) permits. Other Federal agencies, including the Bureau of Land Management, the Forest Service, and the National Parx Service, also exercise oversight and impose regulatory controis (CRA_ 1986b see VII no. 3). The Fegeral waste disposal requirements generally call for practices that will prevent unnecessary and unque cegradation. Federal reclamation surgennes are somewhat more detailed. requiring approval of a land management operating plan and an environmental assessment. Also these agencies generally require compliance with all applicable state and local laws ano ordinances.

A number of states have their own statutes and implementing regulations for mining waste. Some states have comprenensive and well-integrated programs: other States have newer. partially developed programs (CRA. - 1988c see VII no. 4). Although there is great variation in programs, many states have siting and permitting requirements. and require financial assurance, groundwater and surface water protection, and closure standards. EPA agrees that any requirements necessary to protect human nealth and the environment snould consider the existing Federal and State mining waste programs with a -_view toward avoiding duplication of eliort ----٠.

C. Mining Waste Conclusions

Source and Volume

 The waste volume generated by smining and peneficiation is considerably larger than the volume of waste generated by other industries currently subject to nazarcous waste controls. The mining industry alone generates over one billion metric tons of waste per year compared to 250 million metric tons generated annually by all other hazardous waste industries. The average mining waste facility manages about three million metric tons of waste annually while the typical facility subject to Subtitle C controls manages about 50 thousand metric tons of waste per year.

 In general, mining waste disposal facilities are considerably larger than incustrial hazardous waste disposal facilities: most of the largest industrial hazardous waste land disposal facilities are (tens of acres) in size, while typical mining waste disposal facilities are (hundreds of acres) in size. Agency stucies indicate that mining waste tailings impoundments average about 500 acres: the largest is over 5000 acres. Mining waste piles average 128 acres: the largest exceeds 500 acres. 11-1 Hazarcous waste impoundments. however, average only about 6 acres and nazardous waste landfills average only about 10 acres. Consequently, EPA beneves that many traditional .- ---hazarcous waste controls may be technically infeasible or economically · unoractical to implement at mining .. waste sites because of their size ... 1

Waste Management Practices

• EPA estimates indicate that most hazardous waste generators (about 70 percent) snip all of their waste off-site, however, no mines snip all of their waste off-site. In addition, nearly all mining waste is land disposed, while less than half of all industrial hazardous waste is land disposed.

- Evidence of Damage

 In general, environmental ______ conditions and exposure potential associated with mining waste are different than those associated with inquetral hazardous waste streams. Agency studies suggest that mining waste streams generally have lower exposure and nsk potential for severalreasons. -First, mining weste management : facilities are generally in drier and 42.0 climates than hazardous waste management facilities, thereby reducing the leaching potential. Over 80 percent of the mining sites are located west of the Mississippi River. which generally has other cumates. whereas industrial hazardous waste ... landfills are more evening distributed nationally. In addition, the Agency estimates that more than sixty percent

of all mines have annual net recharge between 0-2 incnes, and only ten percent have net recharge greater than ten incnes. However, about 80 percent of the hazardous waste iand disposal facilities nave net recharge greater than five inches, and over one-third exceed 15 incnes.

Second. EPA studies indicate that hazardous waste land disposal facilities are closer to ground water, than mining waste sites. Over 70 percent of hazardous waste sites have a depth to ground water of 30 feet or less, while about 70 percent of mining sites have ground water depths greater than 30 feet.

Third. Subtile C facilities tend to be located in more densely populated areas. EPA estimates that mining waste sites have average populations of less than 200 within one mile of the site, while hazardous waste sites average over 2.000 people at the same distance. Within five miles of the mining waste sites, the average population is almost 3.000, while nazardous waste sites average nearly 60.000 people.

-Courth. Agency studies suggest that, compared to mining weste sites. hazardous waste sites tend to be located closer to drinking water receptors and serve larger populations. Almost 70 percent of the hazardous waste sites are located within five miles of a drinking water receptor serving an average population of over 18.000 and as many as 400.000 people. Almost half as many mining sites are located within this same distance, and they serve considerably smaller populations (averaging 3.000 but ranging as high as 20.000.)

 Although the Agency believes that the human exposure and risk potential appears to be lower for mining waste sites than for industrial hazardous waste sites, many mines are located in sensitive environmental settings. EPA estimates that about 50 percent of the mines are located in areas that have resident populations of threatened or endangered species or species of other special concern. (often the case for industrial sites). In addition, mining sites are typically located in relatively remote and otherwise undisturbed natural environments.

Cost and Economic impacts

 EPA believes that many traditional waste management controls designed onnoipally for industrial hazardous waste management facilities may be economically impractical to implement at mining sites and could impose substantial costs to the industry. resulting in potential mine closures. Full Subtitle C controls for mining sites could impose as much as \$850 million per year in compliance costs. Such costs could be greater than profiles resulting in mine closures.

V. Application of Subtitle C to Mining "Waste

EPA believes that it needs maximum flexibility to develop an appropriate program for mining waste which abcresses the technical feasibility, the environmental necessity, and the economic oracticality of mining waste controls. The program should consist of a tailored risk-based approach which addresses the diversity and unique characteristics of mining waste problems.

The current Subutle C program is designed principally for controlling propiems created by industrial westes. Based on information available, the Agency believes that many controls required under the current Subutle C program. if applied universally to mining sites, would be either unnecessary to protect numan health and the environment, technically infeasible, or economically impractical to implement. For instance, certain Subtitle C requirements such as single and double liner system requirements which provide liquid management and closure and capping standards to minimize infiltration, may be technically inteasible or economically impractical to implement for mining wastes because of the quantity and nature of waste involved. In addition. for many mining sites located in remote areas, such controis may be necessary to protect human health and the environment. For example, liquid releases to the ground water can be minimized and controlled using cutoff walls or interceptor wells (i.e., controlled release) as well as through inter systems, and alternate capping requirements designed to address site-specific concerns such as . direct human contact or wind erosion. . are likely to be feasible and practical. thus providing better long-term protection of human health and the environment

Section 3004(x) of RCRA does provide flexibility for regulating mining waste. This section gives EPA the authority to modify certain Subtitle C requirements for mining waste which were imposed by the Hazarcous and Solid Waste Amendments of 1984 (HSWA) which

relate to tioulos in landfills, prohibitions On land disposal, minimum tecanoiogical requirements. continuing receases at permitted facilities, and retroitting interim status surface moountements with liners. In modifying inese requirements. EPA may consider site-specific characteristics as well as the practical difficulties associated with implementing such requirements. In addition. EPA has general authority under RCRA section J004(a) to modify remaining Subtitle C requirements, such as administrative standards, financial requirements, and closure and capping requirements, if a waste poses different risks or the existing standards are technically infeasible. However, in modifying such requirements. section 🗠 3004(a) does not provide EPA the same agree of flexibility to consider the economic impact of regulation that is found in section 3004(x).

As described earlier in this notice. EPA believes that the decision whether to regulate mining waste under Subtitle, C must consider the factors listed in RCRA sections 8002 (f) and (p). inclucing the risks associated with mining waste, the cost of such regulation, and the effect regulation might have on the use of natural resources. EPA has concluded that in order to meet that objective, it would want to develop a program that has maximum flexibility to develop an effective control strategy for individual facilities based on site-specific . conditions. The existing Subtille C regulatory program would probably nave to be changed substantially for mining waste to provide that type of flexibility.

Given these general conclusions about what would be needed to make the ... Subtille C system appropriate for mining waste, there are substantial uncertainties about whether that program is the right mechanism to address mining waste. First, it is unclear whether the legal authorities under which EPA would be acting (i.e. sections 3004(a) and 3004(x)) give EPA sufficient flexibility to craft a program for "hazardous" mining waste given the statutory and regulatory approach established for other hazardous wastes. Second, and closely related, there are -substanual questions about whether the Agency's current data on mining waste management provide a basis for substantial modifications to the existing Subtitle C regulatory program. With the mining waste study and the supplementary information collection efforts associated with looay s notice. EPA has greatly expanded its understanding of mining waste

24500

manugement practices. At the same time, auditional data collection and analysis would probably be necessary to support specific modifications of multiple provisions in the existing hazardous waste regulations before those regulations would provide the type of flexibility we currently believe might be necessary. These uncertainties have led us to the conclusion that Subtille C does not provide an appropriate template for a mining waste management program.

VI. Application of Subtitle D to Mining Waste

Solid waste that is not hazardous waste is subject to regulation under Subtitle D. Therefore, mining waste. which is included in the RCRA definition of solid waste is currently covered oy Subtitle D. EPA believes that it can design and implement a program specific to mining waste under Subtitle D that addresses the risks associated with such waste. The current Subtitle D program establishes criteria which are. for the most part, environmental performance standards that are used by States to identify unacceptable solid waste disposal practices or facilities (See 40 CFR Part 257.) These criteria include, among other things, standards related to surface water discharges. ground-water contamination. and endangered species. Because the program s criteria are aimed principally at municipal and industrial solid waste. EPA believes they do not now fully address mining waste concerns. In addition, many of these criteria, such as control of disease vectors and bird hazards, are not appropriate for mining waste.

The Agency is currently revising these criteria for facilities that may receive hazardous household waste and small quantity generator hazardous waste: these revisions will not apply to mining waste which are generally not codisposed with such wastes. However. the Agency intends to further augment the Subtitle D program by developing appropriate standards and taking other actions appropriate for mining waste . problems. EPA will focus on identifying environmental problems and setting priorities for applying controls at mining sites with such potential problems as high acid-generation potential, radioactivity, aspestos and cvanice wastes. EPA will also develop a riskmanagement framework to develop

appropriate standards as necessary to protect human health and the environment. EPA will consider requirements such as: (1) A range of closure options to accommodate variable problems such as infiltration to ground water and exposure from fugitive dust: (2) options to define tailored controls, including those established by the Clean Water Act. to address problems from runoff to surface water. (3) options for liquid management controls such as pretreatment of wastes prior to disposal, controlled release, or liner systems: (4) ground-water monitoring options that accommodate site-specific variability; and (5) a range of clean-up options.

in developing such a program, EPA will use its RCRA Section 3007 authority to collect additional information on the nature of mining waste, mining waste management practices, and mining waste exposure potential. EPA believes this authority does not limit information collection to "hazardous" waste icentified under Subtitle C but also authorizes the collection of information on any solid waste that the Agency reasonably believes may pose a hazard when improperly managed. (EPA may also use this authority in preparing enforcement actions.) Initially, EPA will use this information to develop a program under Subutie D. The information, however, may indicate the need to reconsider Subtitle C for certain mining wastes.

In specifying the appropriate standards. EPA also will further analyze existing Federal and State authorities and programs and determine future plans for administering their mining waste programs. Additionally. EPA will perform analyses of costs. impacts. and benefits and will comply fully with Executive Orders 12291 and 12498, the Regulatory Flexibility Act. and the Paperwork Reduction Act.

EPA is concerned that the lack of Federal oversight and enforcement authority over mining waste controls under Subtitle D of RCRA and inadequate.State resources to develop and implement mining waste programs may jeopardize the effectiveness of the program. The Administration therefore will work with Congress to develop the necessary authority. In the interim. EPA will use section 7003 of RCRA and sections 104 and 106 of CERCLA to seek relief in those cases where wastes from

mining sites pose substantial threats or imminent hazards to human health and the environment. Mining waste problems can also be addressed under RCRA Section 7002 which authorizes citizen lawsuits for violations of Subtitle D requirements in 40 CFR Part 257.

As EPA develops this program for regulating human health and environmental risks associated with mining waste, the Agency may find that the Subtitle D approach is unworkable. pernaps because there is insufficient authority to implement an effective program (i.e., the Agency does not obtain oversight and enforcement authority under Subtitle D). or that States lack adequate resources to develop and implement the program. In such an event. EPA may find it necessary to reexamine use of Subtitle C authority with modified mining waste standards in the future.

EPA has already made preliminary contacts with Congress to discuss the best approach for an effective mining waste program. The Agency intends to immediately begin collecting additional technical, economic, and other relevant information needed for program development, and to complete its data analysis by late 1987. EPA hopes to propose revisions to the Subtitle D criteria that are specific to mining waste by mid-1988.

VII. EPA RCRA Docket

The EPA RCRA docket is located at. United States Environmental

Protection Agency.

EPA RCRA Docket (Sub-basement). 401 M Street, SW.

Washington. DC 20460. The docket is open from 9:30 to 3:30 Monday through Finday, except for Federal holidays. The public must make an appointment to review docket materials. Call Mia Zmud at (202) 475– 9327 or Kate Blow at (202) 382–4675 for appointments.

Copies of the following documents are available for viewing only in the EPA docket room:

1. Buc & Associates Inc., 1988. Location of -Mines and Factors Affecting Exposure.

2. Charles River Associates. 1986a. Estimated Costs to the U.S. Uranium and Phosphate Mining industry for Management of Radioactive Solid Wastes.

3. Charles River Associates. 1986b. Federal Non-EPA Regulations Addressing Mining Waste Practices. S. Frontier Technical Automatical States, Groundwater Monitoring Data on Ore Mining and Milling Solid Waste Disposel. 8. ICF. 1986a. Summary of Comments on the

.

- Report to Congress.
- 7. ICF. 1986b. Overview of Superfund Mine Sites
- Sites. 6. Mendian 1988. Statistical Analysis of .
- Maridian 1988. Statistical Analysis of Mining Weste Data.
 Verser, 1988. Quantities of Cyanide-bearing and Acid Caserating Wastes.
 Supporting the Mining Waste Regulatory Determination.

The public may copy a maximum of 50 pages of material from any one regulatory docket at no cost. Additional .

Copies cost \$20/page. -Lee M. Thomas. Administrator. -

[FR Doc. 86-15166 Filed 7-2-66: 8:45 am] 31LLING CODE 6000-00-0

Joint EPA/NRC Guidance on the Definition and Identification of Commercial Mixed Low-Level Radioactive and Hazardous Waste

9432.00-2



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON DIG 12143

MAR 2 1987

MEMORANDUM

- SUBJECT: Joint EPA/NRC Guidance on the Definition and Identification of Commercial Mixed Low-Level Radioactive and Hazardous Waste /
- FRCM: Marcia E. Williams, Director Office of Solid Waste (WH-562)
- TO: Hazardous Waste Division Directors Regions I-X

As you know, EPA announced in the <u>Federal Register</u> on July 3, 1986 (51 <u>FR</u> 24504) that in order to obtain and maintain authorization to administer and enforce a RCRA Subtitle C hazardous waste program, States must apply for authorization to regulate the hazardous components of radioactive mixed waste. States which received final authorization prior to July 3 must revise their program by July 1, 1988 (or July 1, 1989 if a statutory amendment is required) to demonstrate authority to regulate the hazardous components of radioactive mixed waste. States initially applying for final authorization after July 3, 1987 must incorporate this provision in their application for final authorization. To date, only one State (Colorado) has been authorized for mixed waste.

Following publication of the July 3 notice and subsequent dialogue with the Nuclear Regulatory Commission (NRC) and others regarding the universe of affected-wastes, it became apparent that generators of commercial low-level radioactive waste (LLW) needed guidance to facilitate delineation of whether their LLW contained a hazardous waste subject to RCRA regulation. Accordingly NRC and EPA jointly developed the attached guidance to (1) clarify the definition of commercial mixed low-level radioactive and hazardous waste (Mixed LLW) (2) assist generators of commercial low-level radioactive waste (LLW) in determining if their LLW are radioactive mixed wastes, and (3) answer anticipated questions about mixed low-level wastes.

You should feel free to consult with the contacts identified in the guidance regarding specific technical considerations or your may contact Betty Shackleford, OSW Mixed Waste Coordinator on (FTS) 475-9565.

Attachment

9402.00-2

TO ALL NRC LICENSEES:

SUBJECT: GUIDANCE ON THE DEFINITION AND IDENTIFICATION OF COMMERCIAL MIXED LOW-LEVEL RADIOACTIVE AND HAZARCOUS WASTE AND ANSWERS TO ANTICIPATED QUESTIONS

Under the Resource Conservation and Recovery Act (RCRA), the U.S. Environmental Protection Agency (EPA) has jurisdiction over the disposal of solid wastes with the exception of source, byproduct, and special nuclear material, which are regulated by the U.S. Nuclear Regulatory Commission (NRC) under the Atomic Energy Act (AEA). Low-Level Radioactive Wastes (LLW) contain source, byproduct, or special nuclear materials, but they may also contain chemical constituents which are hazardous under EPA regulations in 40 CFR Part 261. Such wastes are commonly referred to as Mixed Low-Level Radioactive and Hazardous Waste (Mixed LLW).

NRC regulations exist to control the byproduct, source, and special nuclear material components of the Mixed LLW; EPA has the authority and continues to develop regulations to control the hazardous component of the Mixed LLW. Thus, all of the individual constituents of Mixed LLW are subject to either NRC or EPA regulations. However, when the components are combined to become Mixed LLW, neither agency has exclusive jurisdiction under current Federal law. This had led to a situation of dual regulation where both agencies, NRC and EPA, regulate the same waste.

The enclosed document, "Guidance on the Definition and Identification of Commercial Mixed Low-Level Radicactive and Hazardous Waste," was developed jointly by the NRC and EPA to aid commercial LLW generators in assessing whether they are currently generating Mixed LLW. This guidance is based on NRC and EPA regulations in effect on December 1, 1986. In addition to the

2

definition and the methodology for identifying Mixed LLW, which we hereby endorse, the staff has prepared answers to anticipated questions from generators which are also included.

Sincerely,

1 Jonn G. Davis, Director Office of Nuclear Material Safety and Safeguards

U. S. Nuclear Regulatory Commission

J. Winston Porter Assistant Administrator Office of Solid Waste and Emergency Response U.S. Environmental Protection Agency

Enclosures: As Stated

.



9

ധ

2 •

00-

2

9432.00-2

I

GUIDANCE ON THE DEFINITION AND IDENTIFICATION OF COMMERCIAL MIXED LOW-LEVEL RADIOACTIVE AND HAZARDOUS WASTE

(37/01/05)

Definition

Mixed Low-Level Radioactive and Hazardous Waste (Mixed LLW) is defined as waste that satisfies the definition of low-level radioactive waste (LLW) in the Low-Level Radioactive Waste Policy Amendments Act of 1985 (LLRWPAA) and contains hazardous waste that either (1) is listed as a hazardous waste in Subpart D of 40 CFR Part 251 or (2) causes the LLW to exhibit any of the hazardous waste characteristics identified in Subpart C of 40 CFR Part 251.

Identification

The policy provided in this guidance is developed for commercial LLW jointly by the U.S. Nuclear Regulatory Commission (NRC) and the U.S. Environmental Protection Agency (EPA). LLW that contains mazarcous wastes defined under the Resource Conservation and Recovery Act (RCRA) is Mixed LLW. Under current Federal law, such waste is subject to regulation by NRC under the Atomic Energy Act (AEA), as amended, and by EPA under the AEA and RCRA, as amended. In the absence of legislation to the contrary, management and disposal of this waste must be conducted in compliance with NRC and EPA or equivalent state regulations.

This guidance presents a methodology (Figure 1) that may be used by generators of commercial LLW to identify Mixed LLW. Implementation of the methodology should identify Mixed LLW and aid generators in assessing whether they are currently generating Mixed LLW. Generators are cautioned, however, that application of the methodology does not affect the need to comply with applicable NRC and EPA regulations. Because EPA's regulations for hazardous waste are currently changing, generators should use applicable regulations that are in effect at the time of implementation of the methodology. This guidance has been prepared based on NRC and EPA regulations in effect on December 1, 1986.

Application of this methodology to identify Mixed LLW will reveal the complexities of the definition of Mixed LLW. If generators have specific questions about whether LLW is Mixed LLW, they should promotly contact the agencies by writing to the persons listed below.

9432.00-2

2

For questions about whether the waste is low-level radioactive waste, contact:

Dr. Sher Bahadur Division of Waste Management Mail Stop 623-SS U. S. Nuclear Regulatory Commission Wasnington, OC 20555 For questions about whether the waste is hazardous waste, contact:

Mr. Alan Corson Deputy Director Characterization and Assessment Division Mail Code WH-5628 U. S. Environmental Protection Agency 401 M Street, S.W. Washington, DC 20460

Methodology

Step 1. Identify LLW

Step 1 in the methocology requires that the generator cetermine whether the waste is LLW as defined in the LLRWPAA. This Act defines LLW as radioactive material that (A) is not high-level radioactive waste, spent nuclear fuel, or byproduct material as defined in section 11e.(2) of the AEA (i.e., uranium or thorium mill tailings) and (B) the NRC classifies as LLW consistent with existing law and in accordance with (A). If the generator determines that the waste is LLW, the generator should proceed to step 2. If the determination is negative, then the waste cannot be Mixed LLW because it is not LLW. However, the waste may be another radioactive or hazardous waste regulated under AEA, RCRA, or both statutes.

Step 2. Identify Listed Hazardous Waste

In step 2, the generator determines whether the LLW contains any hazardous wastes listed in Subcart D of 40 CFR Part 251. Subpart D of Part 261 is reproduced in Appendix I of this guidance. LLW is Mixed LLW if it contains any hazardous wastes specifically listed in Subpart D of 40 CFR Part 261. Listed hazardous wastes include hazardous waste streams from specific and non-specific sources listed in 40 CFR Parts 261.31 and 261.32 and discarded commercial chemical products listed in 40 CFR Part 251.33. The generator is responsible for determining whether LLW contains listed hazardous wastes. The determination should be based on knowledge of the process that generates the waste. For example, if a process produces LLW that contains spent solvents that are specifically listed in the tables of Subpart D of Part 261, the ______

_material, and (2) the tailings or wastes produced by the extraction or concentration of uranium or thorium from any ore processed primarily for its source material content.

Source, special nuclear, and byproduct materials, however, may be mixed with other radioactive or non-radioactive materials that are not source, special nuclear, or byproduct materials. For example, tritium may be contained in toluene, a nonhalogenated arcmatic solvent. Consistent with the definition of byproduct material, the tritium may be considered a byproduct material, while the toluene that contains the tritium would not be byproduct material. Mixtures of toluene and tritium could satisfy the definition of Mixed LLW because they contain listed hazardous waste (spent toluene) and tritium that may qualify as LLW if it has been produced by activities regulated by NRC under the AEA.

2. What are some examples of Mixed LLW?

A preliminary survey performed for the NRC identified two potential types of Mixed LLW:

- * LLW containing organic liquids, such as scintillation liquids and vials; organic lab liquids; sludges; and cleaning, degreasing, and miscellaneous solvents.
- ^e LLW containing heavy metals, such as discarded lead shielding, discarded lined containers, and lead oxide dross containing uranium oxide; light water reactor (LWR) process wastes containing chromate and LWR decontamination resins containing chromium; and mercury amalgam in trash.

The preliminary survey concluded that potential Mixed LLW comprises a small percentage of all LLW. For example, LLW containing organic liquids accounted for approximately 2.3% by volume of LLW reported in the preliminary survey (Bowerman, et al, 1985).

An earlier survey identified a more diverse universe of potential Mixed LLW including wastes that contained aldehydes, aliphatic halogenated hydrocarbons, alkanes, alkenes, amino acids, aromatic hydrocarbons, chelating agents, esters, ethers, ketones, nitrosamines, nucleotides, pesticides, phenolic compounds, purines, resins, steroids, and vitamins (General Research Corporation, 1980). NRC also anticipates that additional LLW may be identified as Mixed LLW in the future, as generators implement the definition of Mixed LLW and as EPA revises the definition of hazardous waste.

9432.00=2 =

3. Could some "below regulatory concern" wastes be considered Mixed LLW?

A determination that radioactive wastes are below regulatory concern (BRC) for radioactivity may affect how the wastes are managed or discarded, but it does not affect the legal status of the wastes. Specifically, their status with respect to the definition of Mixed LLW does not change. BRC waste is still LLW because it satisfies the definition of LLW in the LLRWPAA and is within the NRC's jurisdictional authority under the AEA.

When radioactive waste contains sufficiently low concentrations or quantities of radionuclides, NRC may find that they do not need to be managed or disposed of as radioactive wastes. For NRC to make such a finding, management and disposal of the waste must not pose an undue radiological risk to the public and the environment. However, NRC's determination that the radioactive content of the wastes is below NRC regulatory concern does not relieve licensees from compliance with applicable rules of other agencies governing non-radiological hazards (e.g., regulations of EPA or the Department of Transportation).

Therefore, some ERC wastes may still be considered Mixed LLW if they contain hazardous wastes that have been listed in Subpart D of 40 CFR Part 261 or that cause the LLW to exhibit any of the hazardous characteristics described in Subpart C of 40 CFR Part 261. BRC Mixed LLW may be managed without regard to its radioactivity (but it must still be managed as a nazardous waste in compliance with EPA's regulations for hazardous waste generation, storage, transportation, treatment, and disposal (cf. 40 CFR Parts 262 through 256)).

4. If I use chemicals in my process that are identified by EPA as hazardous constituents, should I assume that my LLW is Mixed LLW?

No. Low-level radioactive waste that contains hazardous constituents may not necessarily be Mixed LLW. As defined above, Mixed LLW is LLW that contains a known hazardous wasta (i.e., a listed hazardous wasta) or that exhibits one or more of the hazardous characteristics because it contains non-AEA materials. For wastes that are not listed in Suppart D of 40 CFR Part 261, testing is not necessarily required to "determine" whether the LLW exhibits any of the hazardous characteristics. A generator may be able to determine whether the LLW is Mixed LLW based on knowledge of the waste characteristics or the process that generates the LLW.

Furthermore, if the generator normally segregates LLW from hazardous and other types of wastas, there is no need to assume that hazardous wastes may have been inadvertently mixed with LLW or to inspect each container or receptacle to ensure that inadvertent mixing has not occurred. Although the generator is subject to RCRA inspections and must follow the manifest, pre-transport, and other requirements of 40 CFR Part 262, the generator is not required to demonstrate that every LLW container does not contain hazardous waste.

.

3

5. What are EPA and NRC currently doing to address the Mixed LLW issue, and what should generators do in the interim before a regulatory program for Mixed LLW is established?

An incentive exists for generators to minimize the generation of Mixed LLW because Mixed LLW must currently be managed and disposed of in compliance with the regulatory controls of both EPA and NRC. These dual regulatory controls complicate management and disposal of the waste. NRC and EPA are presently working together to develop guidance for generators and disposal site operators on the management, treatment, and disposal of Mixed LLW. In the interim. generators are encouraged to minimize the generation of Mixed LLW through management practices such as waste segregation and materials tracking. Generators and waste handlers are also encouraged to consider treatment techniques to reduce the amount and hazards of Mixed LLW requiring licensed land disposal. Kempf et al (1985) prepared a preliminary evaluation of current practices and potential management options for Mixed LLW. Current disposal site operators must develop and operate facilities to dispose of Mixed LLW in compliance with both NRC and EPA requirements or cease disposing of Mixed LLW. Licensees should recognize that all of these activities must be performed in compliance with applicable NRC requirements in 10 CFR Parts 20, 30, 40, 50, 61, and 70, and applicable EPA requirements in 40 CFR Parts 124, and 260 through 270, or applicable State requirements.

6. What should I do if I believe that the RCRA regulations are inconsistent with the AEA regulations?

Section 1006 of RCRA states that, "Nothing in this Act shall be construed to apply to (or to authorize any state, interstate, or local authority to regulate any activity or substance which is subject to . . . the Atomic Energy Act of 1954 (42 U. S. C. 2011 and following) except to the extent that such application (or regulation) is not inconsistent with the requirements of such [Act]." This provision allows the modification of the RCRA requirements when they are found to be inconsistent with the AEA requirements. "Inconsistent" includes situations where satisfying both sets of regulations (RCRA and AEA regulations) would increase the radiation hazard, would be technically infeasible, or would violate national security interests. Variances from the RCRA requirements may be granted to generators, transporters, and facilities that treat, store, or dispose of Mixed LLW.

NRC licensees may petition for variances from RCRA requirements when they believe that application of one or more of these requirements would be inconsistent with the AEA. NRC licensees should first discuss the inconsistency with NRC prior to preparing the petition. NRC's review will ensure that the licensees' interpretations of the AEA requirements are correct and that the reasons for the variance petition are technically sound.

4

7. How can I cotain representative samples of heterogeneous trasm included in LLW to perform the mazarcous characteristics tests?

Sefore discussing the collection of representative samples of waste, generators are reminded that they are not required to test all LLW to determine if the waste contains hazardous wastes that cause the LLW to exhibit the hazardous waste characteristics. Such comprehensive testing of all LLW would likely violate the principle of keeping radiological exposures as low as is reasonably achievable. Generators should select testing as a basis for determining. whether the LLW exhibits any of the hazardous waste characteristics if they cannot make the determination based on their knowledge of the process that generates the LLW.

Representative samples of wasta should be collected for testing in accordance with EPA's regulations in 40 CFR Part 261.20(c), which state that waste samples collected using applicable methods specified in Appendix I of Part 261 will be considered as representative samples for hazardous characteristics testing. This appendix has been included in its entirety in Appendix II of this guidance. The sampling techniques described in Appendix I of Part 261 apply to extremely viscous liquids, fly ashritke material, containerized liquid wastes, and liquid wastes in pits, ponds, lagoons, and similar reservoirs. In the absence of guidance about sampling methods described in Appendix I of Part 261 in compination with other methods to collect, to the maximum extent practicable, representative samples of the waste to be tested.

References

•

- Bowerman, B. S., Kemof, C. R., MacKenzie, D. R., Siskind, B. and P. L. Piciulo, 1985, "An Analysis of Low-Level Wastes: Review of Hazardous Waste Regulations and Identification of Radioactive Mixed Wastes," NUREG/CR-4406, U. S. Nuclear Regulatory Commission.
- General Research Corporation, 1980, "Study of Chemical Toxicity of Low-Level Wastes," NUREG/CR-1793, U. S. Nuclear Regulatory Commission.
- Kempf, C. R., MacKenzie, D. R., and B. S. Bowerman, 1936, "Management of Radioactive Mixed Wastes in Commercial Low-Level Wastes," NUREG/CR-4450, U. S. Nuclear Regulatory Commission.

Land Disposal Restrictions



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C. 20460

AUG 7 1097

OFFICE OF SOLID WASTE AND EMERGENCY RESPON

MEMORANDUM

SUBJECT: Land Disposal Restrictions

FROM: Timothy Fields, Jr., Director Emergency Response Division V

fields for.

TO: Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X

The Office of Emergency and Remedial Response has prepared the attached summary of the land disposal restrictions mandated under the 1984 amendments to the Resource Conservation and Recovery Act (RCRA). This summary is still being reviewed within Headquarters, but I am sending the Regions an advance copy of the document because this information is critical to the ongoing operations of the Superfund program, especially for removal actions. In particular, I want to bring your attention to page 15 of the summary. EPA expects that the <u>Federal Register</u> notice discussed on that page will be published next week, and it is important that the Regions review the notice and provide comments to Headquarters. We will be contacting the Regions next week to notify them of the exact publication date.

The final, signed summary of the land disposal restrictions should be sent to the Regions shortly. I do not expect it to contain major changes from this advance draft.

Attachment

cc: Clem Rastatter Russ Wyer John Riley Mark Greenwood Lloyd Guerci Elaine Stanley John Cunningham Frank Russo Jean Schumann



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

WASHINGTON, D.C. 20460

ALG | | 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Land Disposal Restrictions

FROM: Henry L. Longest II, Director MIN Office of Emergency and Remedial Response

> Gene Lucero, Director () (ml. () Mlvo Office of Waste Programs Enforcement

TO: Waste Management Division Directors, Regions I-X Environmental Services Division Directors, Regions I, VI, and VII

As you know, in November 1986, EPA promulgated regulations restricting the land disposal of certain solvent and dioxin wastes pursuant to the 1984 amendments of the Resource Conservation and Recovery Act (RCRA). The effective date of these regulations for Superfund response action wastes, however, was extended until November 1988. On July 8, 1987, EPA published the second set of land disposal restriction (LDR) regulations, which addresses the "California list" wastes. The California list wastes include liquid and non-liquid wastes containing halogenated organic compounds (HOCs), and liquid wastes containing polychlorinated biphenyls (PCBs), corrosives, free cyanides, and certain metals. As of July 8, 1987, land disposal prohibitions took effect for Superfund for the following California list wastes: dilute HOC wastewaters, PCBs, corrosives, free cyanides and the metals. An extension of the effective date has been granted for the remaining HOC wastes.

The purpose of this memorandum is to summarize the LDR provisions that took effect on that date, and to inform you of the issues that are still being resolved in Headquarters regarding Superfund implementation of LDR requirements. More explicit guidance on Superfund compliance with the solvent, dioxin, California list, and other LDR provisions is forthcoming. Separate guidance will also be provided on LDR issues encountered in PRP negotiations. Until that time, the Regions should contact Headquarters if questions arise regarding compliance with the restrictions on land disposal of the California list wastes. In addition, this memorandum requests that the Regions review and comment on a Notice of Data Availability and Request for Comment which will be published in the <u>Federal</u> <u>Register</u> in the near future, and will address free cyanide and metal wastes. Due to the significant impact LDR will have on the Superfund program, the Regions should give high priority to this review. ATTACHMENT A

-

.

SUMMARY OF MAJOR LDR PROVISIONS

AND

CALIFORNIA LIST PROHIBITIONS

Section I of Attachment A summarizes the major LDR provisions and indicates issues that must still be resolved regarding Superfund implementation of those provisions. Section II of Attachment A describes the specific prohibitions on the California list wastes. Attachment B is a copy of the November 1986 Federal Register that promulgates the overall LDR framework and the solvent and dioxin land disposal restrictions. Attachment C is a copy of a Federal Register correction notice published in June 1987, which modifies the 1986 rule. Attachment D is a copy of the July 1987 California list regulations.

Questions regarding removal program compliance with the land disposal restrictions should be directed to Jean Schumann of the Emergency Response Division (FTS 382-4671). Questions regarding remedial program compliance should be directed to John Cunningham of the Hazardous Site Control Division (FTS 382-2446), and enforcement-related questions should be directed to Frank Russo of the CERCLA Enforcement Division (FTS 382-4838).

Attachments: A - Summary of Major LDR Provisions and California List Prohibitions

- B November 7, 1986 Federal Register (LDR Framework and Solvent/Dioxin Regulations)
- C June 4, 1987 Federal Register (Correction Notice)
- D July 8, 1987 Federal Register (California List Regulations)
- cc: (without Attachments B, C, and D):

Marcia Williams Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X Clem Rastatter **Russ Wyer** Timothy Fields Mark Greenwood Lloyd Guerci Elaine Stanley Sylvia Lowrance Joe Carra Arthur Weissman Phil Jalbert Steve Silverman Gary Jonest John Cunningham Frank Russo Jean Schumann

.

CONTENTS

Ι.	su	MMARY OF MAJOR LDR PROVISIONS	Pag	
	Α.	Introduction	1	
	8.	Applicability	1	
	с.	Schedule for Land Disposal Restrictions	2	
	D.	Exemptions from LDR Schedule	2	
	٤.	Treatment Standards	5	
	F.	Single vs. Multiple Constituent LDR Wastes	5	
	G.	Single Constituent Addressed Under Multiple Rulemakings	7	
	н.	Alternatives to Meeting the Treatment Standard 1. Treatability Variance i1. Equivalent Treatment Method Petition iii. Case-by-Case Extension iv. No Migration Petition v. Delist the Waste	7	
	Ι.	Storage of Restricted Wastes	9	
	J.	Exemption for Treatment in a Surface Impoundment	11	
	К.	Dilution and Mixing Prohibited	11	
	٤.	Disposal of Lab Packs Containing Restricted Wastes	11	
	Μ.	Testing and Recordkeeping	11	
II.	CALIFORNIA LIST PROHIBITIONS			
	Α.	Overview	13	
	8.	Free Cyanides and Metals	14	
	c.	Corrosives	16	
_	٥.	Polychlorinated Biphenyls (PCBs)	16	
	ε.	Halogenated Organic Compounds (HOCs)	17	
	F.	Summary of California List Prohibition Levels	18	

I. SUMMARY OF MAJOR LDR PROVISIONS

A. Introduction

The Hazardous and Solid Waste Amendments to the Resource Conservation and Recovery Act (RCRA) were enacted on November 8, 1984. Among other things, these amendments require EPA to evaluate all hazardous wastes according to a strict schedule to detarmine whether land disposal of these wastes is protective of human health and the environment. For wastes that are restricted from land disposal, the amendments require EPA to set levels or methods of treatment which substantially diminish a waste's toxicity or reduce the likelihood that a waste's hazardous constituents will migrate. Beyond specified dates, restricted wastes which do not meet these treatment standards are prohibited from land disposal (unless the waste is subject to an exemption or variance, discussed later in this document). In accordance with the amendments, if EPA fails to set treatment standards for a particular waste by the scheduled deadline, that waste is automatically prohibited from land disposal.

On November 7, 1986, EPA promulgated the first set of land disposal restriction (LDR) regulations. This rule establishes the framework for implementing the LDR program. Section I of this document summarizes the major provisions of the RCRA amendments and the November 1986 regulations.

8. Applicability

The land disposal regulations apply to hazardous wastes placed in land disposal units. "Land disposal" is defined as placement in or on the land and includes, but is not limited to: placement in a landfill, surface impoundment, waste pile, land treatment facility, salt dome or salt bed formation, underground cave or mine. Also included is placement in a concrete vault or bunker intended for disposal purposes (40 CFR Section 268.2(a)). For the purposes of this definition, open detonation and open burning are not considered "land disposal," but rather, are methods of thermal treatment. At a later date, prohibitions will also be imposed on the disposal of hazardous wastes in injection wells.

LDR applies only to hazardous wastes placed in land disposal units <u>after</u> the effective dates of the prohibitions. Wastes placed in the land prior to the effective dates of the prohibitions (and not removed) are not covered by the regulations. For example, if wastes placed in the land prior to the effective date are later treated in-situ, they are not subject to the requirements of LDR. However, if wastes are removed from land disposal after the effective date, the wastes must meet the applicable treatment standards before subsequent new placement in the land. To date, Headquarters has determined that the following scenarios do not constitute new "placement," and therefore, that the land disposal restrictions are not applicable:

- Waste is consolidated within a unit or area of contamination.
- * Waste is capped in place, including grading prior to capping.
- Waste is treated in-situ.

Other scenarios have been identified that do constitute placement, and therefore, trigger the land disposal restrictions:

- * Waste from different units or areas of contamination is consolidated into one unit or area of contamination.
- Waste is treated outside a unit or area of contamination, and redeposited into the same unit or area of contamination.

(Note that in many cases there are no defined "units" at Superfund sites, but rather, areas of contamination which could be considered units under RCRA.) Future guidance will provide more detail on the definition of placement.

The land disposal prohibitions apply to wastes placed in the land as well as to waste treatment residues placed in the land. Therefore, if the recycling or other treatment of a restricted waste includes land disposal of treatment residues, these residuals must meet applicable treatment standards.

Wastes which are placed into storage before the prohibition effective date are not subject to the LDR restrictions on storage. However, wastes taken out of storage after the effective date must meet applicable treatment standards prior to land disposal.

The land disposal restrictions apply to both interim status and permitted facilities.

C. Schedule for Land Disposal Restrictions

The effective dates for the land disposal restrictions under the 1984 RCRA amendments are set forth in Table 1 (page 3). The listed wastes are banned from land disposal on the effective dates unless the waste stream or residue from treatment of the waste stream meets specified treatment standards. (Exemptions from this schedule are discussed in Section I.D, and alternatives to meeting treatment standards are explained in Section I.H.)

D. Exemptions from LDR Schedule

The 1984 RCRA amendments and the LDR regulations provide exemptions and variances from the LDR schedule as set forth in Table 2 (page 4).

TABLE 1

-

:

LDR Schedule

WASTE	BAN EFFECTIVE DATE
Spent solvent wastes (F001, F002, F003, F004, F005)	November 8, 1986
Dioxin-containing wastes (F020, F021, F022, F023, F026, F027, F028)	November 8, 1986
California list wastes	July 8, 1987
First third of all ranked and listed RCRA hazardous wastes	August 8, 1988
Second third of all ranked and listed RCRA hazardous wastes	June 8, 1989
All remaining ranked and listed RCRA hazardous waste and all RCRA characteristic hazardous wastes	May 8, 1990
Any RCRA hazardous waste listed or identified under RCRA 3001 after November 8, 1984	Within six months of listing or identification

1

TABLE 2

Exemptions from LDR Schedule

WASTE

All soil and debris wastes from from CERCLA response* and RCRA corrective actions

All RCRA-listed dioxin wastes

All RCRA-listed solvent wastes from CERCLA response and RCRA corrective actions (non-soil and debris)

Small-quantity generators (100 kg -1000 kg per month) of RCRA solvent wastes

Solvent-water mixtures, solventcontaining sludges, or solventcontaminated soil or solids (non-CERCLA or RCRA corrective action) containing less than 1 percent total F001-F005 solvent constituents as initially generated

Liquid and non-liquid hazardous wastes containing HOCs in total concentration greater than or equal to 1,000 mg/l, or 1,000 mg/kg, respectively (except for dilute HOC wastewaters, defined in Section II)

EXEMPTION/VARIANCE

Statutory two-year exemption from effective date until November 8, 1988

Regulatory two-year national variance until July 8, 1989

^{*} Soil and debris have been defined to include soil, dirt, and rock as well as natural and manufactured materials, such as contaminated wood, stumps, clothing, equipment, building materials, storage containers, and liners. The definition does not include bulk wastes that are not contaminated soil or debris. Only soil and debris resulting from CERCLA 104 and 106 (enforcement) actions are included in this exemption. State-ordered, State-funded, or private-party funded response wastes are not entitled to this exemption if CERCLA section 104 or 106 authority is not utilized, even if section 107 is used for purposes of invoking liability.

E. Treatment Standards

LDR prohibits land disposal of restricted wastes after the effective date unless specified treatment standards are met. (Alternatives to meeting the treatment standards are discussed in Section I.H.) Treatment standards may be expressed as: 1) concentration-based performance standards (concentrations in the total waste or in a waste extract), or 2) specified methods of treatment. Treatment standards are generally based on the best demonstrated available technology (BDAT) for treatment of that waste.

If the treatment standard is expressed as a waste concentration or a concentration in a waste extract, any treatment technology that meets this performance standard may be used to treat the waste prior to land disposal. If the standard is expressed as a specific technology, that technology must generally be used to treat the waste prior to land disposal.

In the November 7, 1986 LDR regulations, treatment standards were established for the RCRA-listed solvent and dioxin wastes (see Table 3 on the next page). These treatment standards are expressed as concentrations in a waste extract (40 CFR Section 268.41). The July 8, 1987 rule establishes treatment standards for the California list PCB and HOC wastes (except for dilute HOC wastewaters). These treatment standards, discussed in more detail in Section II of this document, are expressed as specific technologies (e.g., thermal treatment). Treatment standards for the remaining wastes will be evaluated according to the schedule set forth in Table 1.

Treatment standards are to be used only for disposal purposes, not as cleanup standards. In fact, sites may be cleaned to levels lower than the treatment standards.

F. Single vs. Multiple Constituent LDR Wastes

Single constituent LDR wastes must meet the treatment standard specified for that waste. For example, a wastewater containing 6 percent solvents by weight, with chlorobenzene the only LDR constituent at a concentration of 1.63 milligrams per liter (mg/l), must be treated with a technology capable of achieving a chlorobenzene concentration of less than 0.15 mg/l, the chlorobenzene treatment standard (see Table 3).

Multiple constituent LDR wastes must meet the treatment standards for each restricted waste. For example, assume a spent solvent waste contained the following:

LDR	Constituent	Concentration	Treatment Standard
(1)	acetone	0.75 mg/1	0.59 mg/l
(2)	methylene chloride	1.20 mg/l	0.96 mg/1
(3)	trichloroethylene (TCE)	0.87 mg/1	0.091 mg/l

Table 3

Solvent and Dioxin Treatment Standards

.

F001 - F005	Extract Concentrations (mg/1)		
Solvent			
Constituents	Wastewater	Other	
Acetone	0.05	0.59	
n-Butyl alcohol	5.0	5.0	
Carbon disulfide	1.05	4.81	
Carbon tetrachloride	0.05	0.96	
Chlorobenzene	0.15	0.05	
Cresols (and cresylic acid)	2.82	0.75	
Cyclohexanone	0.125	0.75	
1,2-Dichlorobenzene	0.65	0.125	
Ethyl acetate	0.05	0.75	
Ethyl benzene	0.05	0.053	
Ethyl ether	0.05	0.75	
Isobutanol	5.0	5.0	
Methanol *	0.25	0.75	
Methylene chloride	0.20	0.96	
Methylene chloride (from			
obacmaceutical industry)	12.7	0, 96	
Methyl ethyl ketone	0.05	0.75	
Methyl isobutyl ketone	0.05	0.33	
Nitcohanzana	0.66	0,125	
Rucidine	1 12	0.33	
ryr iuine Tataachlasaathulasa	0.079	0.05	
Teluce	1 12	0.33	
	1.12	0.33	
1,1,1-Irichloroethane	1.03	0.41	
1,1,2-171011070-1,2,2-	1.05	0.06	
trifiuoroethane	1_05	0.90	
Inichioroethylene	0.062	0.091	
irichiorofiuoromethane	0.05	0.90	
Xylene	0.05	0.15	
	Ext	ract	
LDR Dioxins	Concentra	ation (ppb)	
Hexachlorodibenzo-p-dioxins	< 1.	.0	
Hexachlorodibenzofurans	< 1.	.0	
Pentachlorodibenzo-p-dioxins	< 1.	.0	
Pentachlorodibenzofurans	< 1.	.0	
Tetrachlorodibenzo-o-dioxins	< 1.	.0	
Tetrachlorodibenzofurans	< 1.	.0	
	Ext	iract (
LDR Chlorophenols	Concentr	ration (pom)	
2,4,5-Trichlorophenol	< 0.	.05	
2,4,6-Trichlorophenol	< 0.	.05	
2,3,4,6-Tetrachlorophenol	< 0.	. 10	
Pertachlorophenol	< 0.	.01	

2,4,5-Irichlorophenol 2,3,4,6-Tetrachlorophenol Pentachlorophenol

The waste must then be treated using a treatment technology capable of meeting the treatment standards for acetone, methylene chloride, and TCE. Similarly, if a hazardous waste contained prohibited levels of HOCs and a California list metal, the waste would be prohibited from land disposal until it was in compliance with LDR requirements for both HOCs and the restricted metal.

Note, however, that for multiple constituent wastes the prohibition effective date for each individual constituent would also be applicable. Thus, for a mixed non-liquid HOC/California list metal waste, the statutory prohibition on the metal is effective in 1987, but the HOC restriction is not effective until 1989. Until 1989, therefore, the only applicable restriction is the metal prohibition.

G. Single Constituent Addressed Under Multiple Rulemakings

Where a single constituent is addressed under more than one rulemaking, the applicable treatment standard is that for the more specific waste stream. For example, an HOC-containing waste that is also covered by the FOOL spent solvent listing is prohibited from land disposal according to the effective date for the solvent regulation and must be treated to the level specified in that rule for FOOL waste.

H. Alternatives to Meeting the Treatment Standard

If the treatment standard for an LDR waste cannot be met, several alternatives are available:

(i) Obtain a Treatability Variance (§ 268.44)

In some cases, it may not be possible to treat a waste to the specified level, or a specified technology may not be appropriate for a waste. A treatability variance allows the regulatory standard to be modified on a waste-specific basis for wastes that are significantly different from the wastes evaluated by EPA in setting the treatment standards and, therefore, cannot be treated to the applicable standard. Table 4 lists the major information requirements required in a petition for a variance from a treatment standard. Headquarters is currently developing treatability variance procedures specifically for Superfund wastes.

(ii) Obtain an Equivalent Treatment Method Petition (§ 268.42(b))

This alternative is available for LDR wastes where the treatment standard is expressed as a specific technology. Section 268.42(b) provides that any person may submit a petition to EPA demonstrating that an alternative treatment method can achieve a measure of performance equivalent to that achieved by the specified treatment technology. Headquarters is currently developing petition procedures specifically for Superfund wastes.

Table 4

Major Information Requirements for a Petition

for a Variance from a Treatment Standard

- A detailed description of the petitioner's waste (including data and information on the physical and chemical characteristics of the waste) that EPA can use to compare the petitioner's waste to the wastes considered by EPA in developing BDAT.
- If the waste has been treated, a description of the treatment system, including the process design, operating conditions, and an explanation of why the treatment standard cannot be achieved using the treatment system, or an explanation of why the specified treatment technology is inappropriate for the petitioner's waste.
- If the waste has not been treated, an explanation of why the petitioner believes the waste will react to treatment differently from the wastes evaluated by EPA in developing the treatment standard.
- * A description of any alternative treatment systems examined by the petitioner, and, as appropriate, the concentrations that can be achieved by applying such treatment to the waste.
- The dates of the sampling and testing.
- A description of the methodologies and equipment used to obtain representative samples.
- A description of the sample handling and preparation techniques.
- A description of the tests performed (including results).

(111) Obtain a Case-by-Case Extension (§ 268.5)

In cases where the treatment or disposal capacity cannot reasonably be made available by the effective date of the land disposal prohibitions, the the effective date can be extended for up to one year. This extension may be renewed one time, for a total possible extension of two years. The petition for the extension must demonstrate, among other things, that the generator has made a good faith effort to locate adequate treatment, recovery, or disposal capacity and that he has entered into a binding contract to provide such capacity at a future time. During the extension period, wastes may be disposed in a landfill or surface impoundment if the facility meets minimum specified technological requirements.

Headquarters is currently evaluating whether use of such extensions is appropriate for Superfund wastes.

(iv) Obtain a No Migration Petition (§ 268.6)

EPA will consider allowing land disposal of restricted wastes if a petitioner can demonstrate, to a reasonable degree of certainty, that such disposal will not allow migration of hazardous constituents from the disposal unit or injection zone for as long as the wastes remain hazardous. In general, a successful "no migration" petition will allow land disposal only of a specific waste at a specific facility. Table 5 lists the major information requirements for a "no migration" petition. Headquarters is currently developing petition procedures specifically for Superfund wastes.

(v) Delist the Waste (Sections 260.20, 260.22)

A hazardous waste under RCRA can be formally delisted, however, this process may be too time-consuming for Superfund response actions, especially for removal actions. The current delisting process may take from six to twenty-four months, depending on the type of material and the completeness of data on the material.

I. Storage of Restricted Wastes

The 1984 RCRA amendments provide that LDR waste cannot be stored unless such storage is solely for the purpose of accumulating sufficient quantities for proper treatment, recovery, or disposal. Although no time limit has been established on such storage, in the event of an enforcement action, the storage facility bears the burden of demonstrating compliance with this purpose where storage of LDR wastes exceeds one year. For periods less than one year, EPA has the burden of proving that such storage is not in compliance with this purpose. The regulation also provides that a transporter of restricted wastes may only store such wastes at a transfer facility for ten days.

Headquarters is currently working on the issue of temporary storage of restricted Superfund wastes while awaiting implementation of on-site treatment/disposal or procurement of off-site treatment/disposal.

Table 5

Major Information Requirements for a

"No Migration" Petition

- The identification and a full characterization of the specific waste, including a comprehensive chemical and physical characterization.
- The identification and a comprehensive characterization of the disposal unit, including background air, soil, and water quality.
- A demonstration that all waste and environmental sampling, test, and analysis data are accurate and reproducible.
- A demonstration that EPA-approved sampling, testing, and estimation techniques were used.
- A demonstration that all simulation models for the specific waste and disposal site conditions were calibrated, and that the models were verified by actual measurements.
- Analyses performed to identify and quantify any aspects that could contribute significantly to uncertainty regarding the suitability of the site, including the potential for damage from earthquakes, floods, severe storms, droughts, or other natural phenomena.
- A quality assurance and quality control plan that addresses all aspects of the "no migration" demonstration.

J. Exemption for Treatment in a Surface Impoundment

Treatment of LDR-subject wastes in surface impoundments is allowed under specific provisions in 40 CFR Section 258.4. These provisions state that treatment of the restricted waste must occur in the impoundment; representative samples of sludge and supernatent must be analyzed separately to determine if they meet the applicable treatment standards in Section 268.41 (or prohibition levels, where treatment standards have not been established); treatment residues not meeting applicable standards or levels must be removed annually and not placed in any other surface impoundment; the impoundment must meet the operation and design requirements of Section 264.221 or 265.221, or be exempted under those Sections; and the owner or operator must submit to the Regional Administrator a certification that these requirements have been met, with a copy of the waste analysis plan.

K. Dilution and Mixing Prohibited

Dilution of restricted wastes with liquids or non-liquid agents (e.g., flyash, sawdust) to reach the desired concentration levels is prohibited under 40 CFR Section 258.3. Such dilution may not be used as a substitute for adequate treatment. (Note that solidification may be appropriate treatment for some California list wastes at this time. See Section II of this document.) Dilution as a part of proper treatment (e.g., addition of fixing agents, addition of lime to neutralize) is allowed. Addition of such substances produces physical or chemical changes and does not merely dilute the hazardous constituent into a larger volume of waste so as to lower the constituent concentration.

The prohibition on dilution also applies to residuals after treatment. In particular, wastes meeting Subpart D treatment standards must not be mixed with wastes that do not meet such standards in order to achieve the treatment standard (or prohibition level) for the mixture.

Restricted wastes may not be mixed unless they are mixed with wastes that require treatment or facilitate treatment. Where wastes are mixed in accordance with this principle, the lowest treatment standard (or prohibition level) must be achieved for the constituent of concern.

L. Disposal of Lab Packs Containing Restricted Wastes

If a lab pack contains restricted wastes, the entire lab pack is subject to LDR. The lab pack may not be land disposed unless the restricted wastes are removed prior to land disposal, the restricted wastes in the lab pack meet the treatment standards (or prohibition levels), or a successful "no migration" petition has been made.

M. Testing and Recordkeeping

Generator requirements. Generators are responsible for determining whether their wastes are restricted from land disposal. Such determinations can be made based on knowledge of the waste, testing, or both. When the determination is based solely on knowledge, the generator must keep all supporting documentation in his files.
A generator who determines that a waste <u>requires treatment prior to</u> <u>land disposal</u> must notify the treatment facility in writing of the appropriate treatment standard (or prohibition level) for the waste. The notification must specify the EPA Hazardous Waste Number, the applicable treatment standard (or prohibition level), the manifest number associated with the shipment of waste, and include the waste analysis data (if available).

A generator who determines that a waste <u>can be land disposed without</u> <u>treatment</u> must submit to the disposal facility a notice which contains the EPA Hazardous Waste Number, the manifest number, the applicable treatment standard (or prohibition level), and the waste analysis data (if available) or cross references to relevant data submitted at an earlier time, plus a certification statement. See Section 268.7(a)(2)(ii) for the appropriate wording for the certification.

If a generator's waste is subject to a case-by-case extension, a "no migration" petition, or a nationwide variance from the effective date, he must submit a notice to the land disposal facility receiving the waste, stating that the waste is exempt from the land disposal restrictions.

For CERCLA actions where the waste has been granted an exemption or variance specific to CERCLA waste (e.g., soil and debris, RCRA-listed solvents) and the waste does not meet LDR treatment standards, the OSC/RPM must provide to the land disposal facility a notice that the waste is a CERCLA waste and exempt from LDR, along with any waste analysis data available. The notice may read:

"I notify (facility) that the waste transported under this manifest number is a waste generated from a (remedial or) removal action taken under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA) and is exempt from prohibitions on land disposal under 40 CFR Part 268."

Treatment facility requirements. An off-site treatment facility must obtain the required data from the generator prior to treatment and place that data in the operating record. Treatment residues must be tested prior to land disposal according to the frequency specified in the treatment facility's waste analysis plan to determine if treatment has achieved the required levels. For example, if the waste analysis plan calls for testing of each batch of waste from an incineration process, these data must be submitted to the land disposal facility.

Each waste shipment to a disposal facility must be accompanied by a certification statement including cross references to any relevant data submitted at an earlier time, and a notice which includes the EPA Hazardous Waste Number, the manifest number, the applicable treatment standard (or prohibition level), and the waste analysis data (if available).

Where the treatment standard is performance-based, the certification must be worded according to Section 268.7(b)(2)(1). Where the treatment standard for an applicable waste is a specified method of treatment, the last facility to treat the waste must certify to the land disposal facility that the waste has been treated using the specified technology. See Section 268.7(b)(2)(ii) for the appropriate wording.

Land disposal facility requirements. The disposal facility, which is ultimately responsible for verifying that only wastes meeting the treatment standards (or prohibition levels) are land disposed, must maintain all documentation that the waste has been treated in accordance with the standards. Land disposal facilities must test their wastes to determine compliance with treatment standards as frequently as specified in the facility's waste analysis plan. If generation, treatment, and disposal all occur at the same site, all testing records must be placed in the operating record.

II. CALIFORNIA LIST PROHIBITIONS

A. Overview

The California list wastes include five categories of waste: free cyanides, metals, corrosives, polychlorinated biphenyls (PCBs), and halogenated organic compounds (HOCs). With the exception of HOCs, the California list addresses only liquid wastes; HOCs are regulated under LDR in both liquid and non-liquid forms. Land disposal prohibitions went into effect on July 8, 1987, for the <u>free cyanides, metals, corrosives, PCBs, and dilute HOC wastewaters</u>. The effective date for the other HOC wastes was extended.

The 1984 RCRA amendments (section 3004(d)) established prohibition levels for the land disposal of California list wastes that would go into effect on July 8, 1987, unless EPA promulgated more protective treatment levels by that date. In the July 8 California list rule, EPA does not set treatment levels for the free cyanides and metals, but instead allows the statutory prohibitions to take effect at this time. (More stringent levels may be established, however, in the coming months.) For corrosive wastes, the California list rule codifies the statutory prohibition levels. For PCBs and HOCs (except dilute HOC wastewaters), the California list rule establishes treatment standards.

Note that the California list addresses fairly broad categories of waste. Many of these wastes have been, or will be, covered by other LDR rulemakings, such as the spent solvent restrictions. Because the other regulations address more specific waste streams, where a California list waste is also covered by another rulemaking, the restrictions in the other regulation will take precedence over the California list prohibitions.

In order to be subject to the California list land disposal prohibitions, a given waste must meet four criteria: 1) the waste must contain a constituent specified in the California list provisions or have a pH less than or equal to 2 (RCRA section 3004(d)); 2) the physical form of the waste must be a liquid (except for HOCs); 3) the wastes containing the California list constituent must be listed or identified as hazardous under RCRA section 3001 (as implemented in 40 CFR Part 261); and 4) the waste must contain a concentration of one or more California list constituents at or above the prohibition levels specified in RCRA section 3004(d), or more stringent levels that may be specified by EPA in the LDR regulations.

With regard to criterion two, (the waste must be a liquid), note that the rule generally requires use of the Paint Filter Liquids Test to determine if a given waste is a liquid (Method 9095 in EPA Publication SW-846). Once a waste is determined to be a liquid and to exceed the applicable prohibition levels, however, the entire waste is restricted from land disposal, not just the liquid portion.

With regard to criterion three (the waste must be a RCRA hazardous waste), in some cases it may not be possible to determine if a Superfund hazardous waste is also a RCRA hazardous waste. For example, it may be possible to determine that a given waste contains trichloroethylene, but it may not be possible to determine whether the source of the trichloroethylene is one of the processes regulated under RCRA. Headquarters is continuing to work on developing a policy for Superfund compliance with LDR for such "unknowns."

These four criteria are discussed in greater detail below for each of the California list wastes.

B. Free Cyanides and Metals

Definition. To determine if a given waste contains free cyanides for purposes of RCRA section 3004(d), the rule requires that the filtrate from the Paint Filter Liquids Test be analyzed for free cyanides. The rule does not require use of a particular analytical test, but recommends use of the Cyanides Amenable to Chlorination test in Method 9010 (EPA Publication SW-846) for determining free cyanide concentrations.

The California list metals are: arsenic, cadmium, chromium VI, lead, mercury, nickel, selenium, and thallium. These metals are defined with reference to the periodic table of elements. The LDR provisions apply both to individual constituents and to the relevant metal portion of any compounds containing such metals.

<u>Physical Form Requirement</u>. To determine if a given free cyanide or metal waste is a liquid for the purposes of RCRA section 3004(d), the rule requires use of the Paint Filter Liquids Test.

<u>Hazardous Waste Requirement</u>. To be subject to LDR provisions, the waste must be regulated as hazardous under RCRA. Thus, a given waste must contain the specified cyanides or metals, and also be either listed as hazardous under 40 CFR Part 261 or exhibit one or more characteristics of hazardous waste identified in Part 261 (i.e., ignitability, corrosivity, reactivity, or EP toxicity). <u>Prohibited Levels</u>. The rule does not establish treatment standards for cyanides and metals at this time. Instead, EPA is allowing the statutory "hammer" under RCRA section 3004(d) to fall on July 8. The statute prohibits land disposal of the following:

- "Liquid hazardous wastes, including free liquids associated with any solid or sludge, containing free cyanides at concentrations greater than or equal to 1,000 mg/l."
- "Liquid hazardous wastes, including free liquids associated with any solid or sludge, containing the following metals (or elements) or compounds of these metals (or elements) at concentrations greater than or equal to those specified below:

(1) arsenic and/or compounds (as As) 500 mg/l; (ii) cadmium and/or compounds (as Cd) 100 mg/l; (iii) chromium VI and/or compounds (as Cr VI) 500 mg/l; (iv) lead and/or compounds (as Pb) 500 mg/l; (v) mercury and/or compounds (as Hg) 20 mg/l; (vi) nickel and/or compounds (as Ni) 134 mg/l; (vii) selenium and/or compounds (as Se) 100 mg/l; and (viii) thallium and/or compounds (as Th) 130 mg/l."

To determine if a waste contains free cyanides or metals that exceed these statutory prohibitions, a generator may use knowledge of his waste or must test the filtrate from the Paint Filter Liquids Test. The rule recommends use of the applicable methods in EPA Publication No. SW-846 to test the filtrate.

Although EPA is not promulgating treatment standards for these wastes at this time, the Agency is planning to publish in the near future a Notice of Data Availability and Request for Comment. This notice will outline the Agency's approach to substituting more stringent prohibition levels for the cyanide and metal wastes. Because this notice had not been written at the time of this memorandum, we are not able to provide you with a draft for review in advance of the publication date. Instead. we will telephone the Regions when the publication date is known, and request that each Region obtain a copy of that Federal Register to review the proposed approach. Comments should then be provided to Headquarters at least one week prior to the close of the public comment period for this notice to allow adequate time for us to compile Regional comments. Removal personnel should submit comments to Jean Schumann by telephone (FTS 382-4671), E-Mail, or in writing (WH-548/B, Emergency Response Division). Remedial personnel should submit comments to John Cunningham (FTS 382-2446, WH-548/E, Hazardous Site Control Division).

<u>Treatment of Restricted Metals</u>. Section IV.D of the preamble to the California list rule states that legitimate solidification technologies are appropriate for use on California list metal-bearing wastes, at least until treatment standards are established for such wastes. The preamble also notes that the liquids in landfills prohibitions set forth in Sections 264.314 and 265.314, which place certain prohibitions on the use of absorbents, remain applicable.

C. Corrosives

Definition. The California list corrosive wastes are defined as having a pH less than or equal to 2.

<u>Physical Form and Hazardous Waste Requirement</u>. The statute applies to liquid (acidic) corrosive wastes. The rule requires the use of the Paint Filter Liquids Test to determine if the given waste is a liquid. Acidic wastes are defined as hazardous based on the characteristic of corrosivity found at 40 CFR Section 261.22(a)(1) when the pH is less than or equal to 2. If these wastes are treated to a pH greater than 2, they are no longer characteristic hazardous wastes and may be land disposed in a Subtitle D facility. In addition, because the regulation does not establish a technology-based treatment standard, corrosive wastes may be rendered nonliquid by chemical fixation or other treatment methods and be eligible for land disposal. If a waste is hazardous solely because of the characteristic of corrosivity, rendering it non-liquid also renders it nonhazardous because the characteristic of corrosivity based on low pH applies only to aqueous wastes.

<u>Prohibition Levels</u>. The rule codifies the statutory prohibition level (pH less than or equal to 2), but does not establish a treatment standard for corrosive wastes. The Agency plans to address the issue of appropriate treatment standards for corrosive wastes when it promulgates regulations for the remaining listed wastes.

To determine if a waste has a pH which exceeds the prohibition level, a generator may use knowledge of his waste, or must test the total waste (not an extract or filtrate) using the existing method for corrosivity specified in 40 CFR Section 261.22(a)(1).

D. Polychlorinated Biphenyls (PCBs)

Definition. PCBs are defined in accordance with the Toxic Substances Control Act (TSCA) regulations as "any chemical substance that is limited to the biphenyl molecule that has been chorinated to varying degrees or any combination of substances which contain such substance." In addition, inadvertently generated non-Aroclor PCBs are defined as "the total PCBs calculated following division of the quantity of monochlorinated biphenyls by 50 and dichlorinated biphenyls by 5." (40 CFR Section 761.3)

<u>Physical Form Requirement</u>. To determine if a given waste is a liquid for purposes of this rule, the Paint Filter Liquids Test is required.

Hazardous Waste Requirement. Liquid PCB-containing wastes are subject to the California list prohibitions only if they are mixed with or otherwise contained in wastes which are listed as hazardous under 40 CFR Part 251, or if the mixture exhibits one or more of the characteristics of hazardous waste identified in that Part. <u>Prohibition Levels</u>. The rule codifies the 50 ppm prohibition level in section 3004(d) of the RCRA amendments. Therefore, liquid hazardous wastes containing greater than or equal to 50 ppm may not be land disposed. The rule requires that once a waste is determined to be a hazardous liquid, then the total waste (not an extract or filtrate) must be analyzed to determine if the 50 ppm PC3 level is exceeded.

Treatment Standards. The rule requires thermal destruction (i.e., treatment in incinerators or high efficiency boilers) of restricted PCB wastes pursuant to the operating standards set forth in 40 CFR Sections 761.60 and 761.70.

The TSCA regulations state that where there is an inconsistency between TSCA and RCRA standards, the more stringent regulations govern. Therefore, although TSCA allows land disposal in certain cases of wastes containing between 50 and 500 ppm PC3s, such land disposal would not be allowed if the PCB waste is also restricted under the California list rule. Additionally, because TSCA does not allow any land disposal of liquid PCB wastes at concentrations greater than or equal to 500 ppm, such wastes that are also covered by the California list rule are not eligible for a "no migration" petition under LDR (see Section I of this document).

The LDR framework provides that where a treatment standard is expressed as a specific technology, any person may submit a petition demonstrating that an alternative technology can achieve an equivalent measure of performance. Headquarters is currently developing such petition procedures specifically for Superfund wastes.

<u>Effective Date</u>. The effective date of the California list prohibitions for liquid PCB wastes is July 8, 1987.

E. Halogenated Organic Compounds (HOCs)

Definition. HOCs are defined as those compounds having a carbonhalogen bond which are listed under Appendix III to Part 268 (added by the July 8 rule).

<u>Physical Form Requirement</u>. The rule regulates HOC wastes in both liquid and non-liquid form. (However, soil and debris from CERCLA response actions are exempted from LDR until November 1988.) The Paint Filter Liquids Test is required to determine if the waste is a liquid. With regard to liquid HOCs, the rule also establishes dilute HOC wastewater as a separate treatability group. Dilute HOC wastewaters are defined as liquid hazardous wastes that are primarily water and contain HOCs in total concentration greater than or equal to 1,000 mg/l and less than 10,000 mg/l.

Hazardous Waste Requirement. Wastes containing HOCs are only subject to the California list prohibitions if the waste is listed as hazardous under 40 CFR Part 261 or exhibits one or more of the characteristics of hazardous waste identified in that Part. <u>Prohibition Levels</u>. The land disposal prohibition in the California list rule applies only to hazardous wastes containing HOCs in total concentration greater than or equal to 1,000 mg/l (liquids) or 1,000 mg/kg (non-liquids). In determining whether a waste contains HOCs above these prohibition levels, only those HOCs which are listed in Part 258 Appendix III must be considered. The generator may use knowledge of his waste or must test the total waste (not an extract or filtrate) to determine HOC concentration. The rule does not require use of a specific test.

Relationship to PCB Restriction. PCBs are also HOCs, but because PCBs are the more specific waste stream, the PCB prohibition level (50 ppm) would apply for <u>liquid</u> hazardous wastes containing PCBs. Therefore, a liquid HOC waste that contains greater than or equal to 50 ppm PCBs may not be land disposed, even if total HOCs are below 1,000 mg/l. However, a non-liquid HOC waste that contains greater than or equal to 50 ppm PCBs may be land disposed under this rule as long as total HOCs are below 1,000 mg/kg. If total HOCs in a liquid or non-liquid hazardous waste are greater than or equal to 1,000 mg/kg, the waste may not be land disposed even if PCB concentrations are below 50 ppm.

<u>Treatment Standards</u>. The rule establishes <u>incineration</u> (in accordance with 40 CFR Part 254 Subpart 0 or 40 CFR Part 255 Suppart 0) as the treatment standard for all hazardous wastes containing HOCs in total concentration greater than or equal to 1,000 mg/kg (non-liquids) or 1,000 mg/l (liquids) except dilute HOC wastewaters. However, as noted earlier, if an HOC-containing waste is already subject to a treatment standard for a specific HOC (e.g., FOOI spent solvent), the treatment standard for the more specific HOC waste would take precedence. Thus, the waste would have to meet the treatment standard for the solvent waste, but need not be incinerated.

The LDR framework provides that where a treatment standard is expressed as a specific technology, any person may submit a petition demonstrating that an alternative technology can achieve an equivalent level of performance. Headquarters is currently developing such petition procedures specifically for Superfund wastes.

With regard to <u>dilute HOC wastewaters</u>, these wastes must be treated below the 1,000 mg/l prohibition level prior to land disposal, but no particular technology is required under this rule.

Effective Dates. Due to a lack of incineration capacity, the rule provides a 2-year nationwide variance from the statutory effective date for most HOC wastes until July 8, 1989. However, the prohibitions on land disposal of dilute HOC wastewaters take effect on July 8, 1987.

Note again, that if an HOC waste is also covered under a more wastespecific rulemaking, the treatment standard <u>and</u> effective date for the more waste-specific rule are applicable.

F. Summary of California List Prohibition Levels

The following table summarizes the prohibition levels for the California list wastes.

Table 6

.

•

Waste	Prohibition Level
- Free cyanides	1000 mg/l
Metals:	
Arsenic Cadmium Chromium VI	500 mg/1 100 mg/1 500 mg/1
Lead Mercury Nickel Selenium	500 mg/1 20 mg/1 134 mg/1 100 mg/1
Thalltum	130 mg/1
Corrosives	ph <u><</u> 2
PCBs	50 ppm
HOCs	
Non-liquids Liquids Dilute Wastewaters	1000 mg/kg 1000 mg/l 1000 mg/l

California List Prohibition Levels

Revised, Procedures for Implementing Off-Site Response Actions

United States	•
Enterenting (13)	LLOLOLOLO
Agency	

Unice or Solid Waste and Emergency Response

€EPA DIRECTIVE NUMBER: 9834.11 TITLE: Revised Procedures for Implementing Off-Site Response Actions November 13, 1987 APPROVAL DATE: November 13, 1987 EFFECTIVE DATE: ORIGINATING OFFICE: Office of Waste Programs Enforcement FINAL (Interim) LEVEL OF DRAFT A - Signed by AA or DAA B — Signed by Office Director C - Review & Comment **REFERENCE** (other documents):



	Washington, DC 20	460	1 Directive Number
OSWER I	Directive Init	lation Reque	9834.11
	2. Originator Inform	ation	
Name of Contact Person	Mail Code	Office	Telephone Code
NANCY BROWNE	WH-527	RM 2830	475-9326
Revised Procedures for :	Implementing Off-S	lite Response Acti	.ons
4. Summary of Directive (include bief state	ment of purpose)		
This Directive describes under CERCIA or Section disposal of CERCIA waste	s procedures that 7003 of RCRA inv 2.	should be observe olves the off-sit	d when a response action e treatment, storage or
5. Keywords Off-Site Policy, Off-Sit	e Waste, CERCLA W	aste, Off-Site Ma	nagement of Waste
5a. Does This Directive Superseds Previou	s Directive(s)?	• XX Yes 1) What off	drective (number. title) Declures for Planning & In -site Response Actions.
b. Ooes It Supplement Previous Directive(s)?	2) Pro	viding Notice to Faciliti
		"3) Elu	Tibility of Facilities
		Ass	essment Monitoring (9130
A - Signed by AA/DAA	B - Signed by Office Direc	tor C - For Revit	w & Comment D - In Develop
8. Document to be di	stributed to States	by Headquarters	? Yes x No
8. Document to be di This Request Meets OSWER Directives :	stributed to States System Format Standards.) by Headquarters	? Yes X No
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord	stributed to States System Formet Standards.	by Headquarters	? Yes X No
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Waldhum M- Wald	System Format Standards.	by Headquarters	? Yes X No Dete 11/13/87
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Manual Manual 0. Name and Trile of Approving Official Walk Manual C. Name and Trile of Approving Official	System Format Standards. Ornator Unator Unator	by Headquarters	? Yes x No Date 11/13/87 Date 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Marchan M- War 0. Name and Title of Approving Official Control Marchan PA Form 10/5-17 (Rev. 5-57) Previous et	System Format Standards. dinator <i>k i an</i> <i>A</i> ditions are obsciete.	by Headquarters	? Yes X No Dete 11/13/87 Dete 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord	System Format Standards. dinator <i>Le rans</i> <i>A</i> dibons are obsolete.	by Headquarters	? Yes x No Date 11/13/87 Date 11/13/8
8. Document to be di This Request Meets OSWER Directives : Signature of Lead Office Directives Coort Manual Manual O. Name and Trile of Approving Official Manual Manual PA Form 10/5-17 (Rev. 5-87) Previous et	System Formet Standards. dinator du rans Market A 2 dibons are obsolete.	by Headquarters	? Yes x No Dete 11/13/87 Dete 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord March M. M. 0. Name and Title of Approving Official March M. PA Form 1975-17 (Rev. 5-87) Previous et	System Format Standards. dinator & rane A A ditions are obsolete.	by Headquarters	? Ves x No Dete 11/13/87 Date 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Manual Manual O. Name and This of Approving Official Manual Manual PA Form 1075-17 (Rev. 5-87) Previous et	System Format Standards. dinator the rank A J ditions are obsolete.	by Headquarters	? Yes x No Date 11/13/87 Date 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Martin M. With 0. Name and Trile of Approving Official Martin Martin PA Form 1975-17 (Rev. 5-87) Previous et	System Formet Standards. dinator du rans Marso dibons are obsolete.	by Headquarters	? Yes x No Dete 11/13/87 Dete 11/13/8
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord March M. Work 0. Name and Title of Approving Official March March March PA Form 1075-17 (Rev. 5-87) Previous en SWER	SWER	by Headquarters	? Yes x No Dete 11/13/87 Date 11/13/8 WER
8. Document to be di This Request Meets OSWER Directives 1 Signature of Lead Office Directives Coord Marthur M. War 0. Name and This of Approving Official Marthur M. Marthur PA Form 1075-17 (Rev. 5-87) Previous et SWER O	System Format Standards. dinator <i>Li ran</i> ditions are obsolete. SWER	by Headquarters	? Yes x No Date 11/13/87 Date 11/13/8 WER

,



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

November 13, 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Revised Procedures for Planning and Implementing Off-site Response Actions FROM: J. Winston Porter Assistant Administrator

TO: Regional Administrators Regions I-X

With this memo I am transmitting the revised procedures for planning and implementing off-site response actions (the "off-site policy"). These procedures should be observed when a response action under the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) or Section 7003 of the Resource Conservation and Recovery Act (RCRA) involves off-site treatment, storage or disposal of CERCLA waste.

This policy incorporates all of the mandates of CERCLA as amended by the Superfund Amendments and Reauthorization Act (SARA) and expands several of the more stringent requirements when applying them to wastes resulting from CERCLA decision documents signed, and RCRA section 7003 actions initiated, after the enactment of SARA. This revised policy also reinterprets the original off-site policy, issued in May 1985, as it applies to CERCLA wastes resulting from decision documents signed, and RCRA section 7003 actions initiated, before the enactment of SARA.

This revised policy is effective immediately upon issuance. It is considered to be an interim final policy as key elements of the policy will be incorporated in a proposed rule to be published in the <u>Federal Register</u>. As part of that rulemaking, the policy will be subject to public comment. Comments received during that period may cause additional revisions to the policy. If you have comments regarding this revised policy, please contact Gene Lucero, Director, Office of Waste Programs Enforcement.

.

cc: Waste Management Division Directors Regions I-X

REVISED PROCEDURES FOR IMPLEMENTING OFF-SITE RESPONSE ACTIONS

I. INTRODUCTION

The off-site policy describes procedures that should be observed when a response action under the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) or Section 7003 of RCRA involves off-site storage, treatment or disposal of CERCLA waste. The procedures also apply to actions taken jointly under CERCLA and another statute.

The purpose of the off-site policy is to avoid having CERCLA wastes contribute to present or future environmental problems by directing these wastes to facilities determined to be environmentally sound. It is EPA's responsibility to ensure that the criteria for governing off-site transfer of CERCLA waste result in decisions that are environmentally sensible and that reflect sound public policy. Therefore, in developing acceptability criteria, the Agency has applied environmental, standards and other sound management practices to ensure that CERCLA waste will be appropriately managed.

EPA issued the original off-site policy in May 1985. See "Procedures for Planning and Implementing Off-Site Response Actions", memorandum from Jack W. McGraw to the Regional Administrators. That policy was published in the Federal Register on November 5, 1985. The 1986 amendments to CERCLA, the Superfund Amendments and Reauthorization Act (SARA), adopted EPA's policy for off-site transfer of CERCLA wastes, with some modifications. CERCLA §121(d)(3) requires that hazardous substances, pollutants or contaminants transferred off-site for treatment, storage or disposal during a CERCLA response action be transferred to a facility operating in compliance with §§3004 and 3005 of RCRA and other applicable laws or regulations. The statute also requires that receiving units at land disposal facilities have no releases of hazardous wastes or hazardous constituents. Any releases from other units at a land disposal facility must also be controlled by a RCRA or equivalent corrective action program. While the original policy required compliance with RCRA and other applicable laws, SARA goes beyond the original policy, primarily by prohibiting disposal at units at a land disposal facility with releases, rather than allowing the Agency to judge whether the releases constituted environmental conditions that affected the satisfactory operation of a facility.

The off-site policy has been revised in light of the mandates of SARA. This revised policy also extends the SARA concepts to certain situations not specifically covered by the statute. These requirements apply to CERCLA decision documents signed, and RCRA §7003 actions taken, after enactment of SARA. Specifically, this policy covers:

- Extending SARA's "no release" requirement to all RCRA units receiving CERCLA waste, not just units at RCRA land disposal facilities;
- Expanding SARA's release prohibition to include releases of CERCLA hazardous substances, in addition to releases of RCRA hazardous waste and hazardous constituents;
- Addressing releases from other units at RCRA treatment and storage facilities; and
- o Addressing off-site transfer to non-RCRA facilities.

The revised policy also reinterprets the May 1985 policy as it now applies to CERCLA decision documents signed, and RCRA §7003 actions taken, prior to the enactment of SARA.

The revised off-site policy is effective immediately upon issuance. It is considered to be an interim policy as key elements of the policy will be incorporated in a proposed rule to be published in the <u>Federal Register</u>. As part of that rulemaking, the policy will be subject to public comment. Comments received during that period may cause additional revisions to the policy. The final rule will reflect the final policy under CERCLA §121(d)(3) and EPA will issue a revised implementation policy memorandum if necessary.

II. APPLICABILITY

There are a number of variables which will determine whether and how the off-site policy applies: waste type, authority, funding source, and whether the decision document or order supporting the clean-up was signed before or after the enactment of SARA (i.e., before or after October 17, 1986). In order to determine which elements of the policy apply to a specific CERCLA cleanup each factor must be considered.

The first factor to consider is the type of waste to be transferred. The revised policy applies to the off-site treatment, storage or disposal of all CERCLA waste. CERCLA wastes include RCRA hazardous wastes and other CERCLA hazardous substances, pollutants and contaminants. RCRA hazardous wastes are either listed or defined by characteristic in 40 CFR Part 261. CERCLA hazardous substances are defined in 40 CFR 300.6.

Because RCRA permits and interim status apply to specific wastes and specific storage, treatment or disposal processes, the Remedial Project Manager (RPM) or On-Scene Coordinator (OSC) must determine that the facility's permit or interim status authorizes receipt of the wastes that would be transported to the facility and the type of process contemplated for the wastes. Therefore, it is important that facility selection be coordinated with RCRA personnel.

A CERCLA hazardous substance that is not a RCRA hazardous waste or hazardous constituent (i.e., non-RCRA waste) may be taken to a RCRA facility if it is not otherwise incompatible with the RCRA waste, even though receipt of that waste is not expressly authorized under interim status or in the permit. Non-RCRA wastes can also be managed at non-RCRA facilities. Criteria applicable to CERCLA wastes that can be disposed of at non-Subtitle C facilities are discussed later in this revised policy.

The second factor to consider in determining whether this revised policy applies is the statutory authority for the action. This revised off-site policy applies to any remedial or removal action involving the off-site transfer of any hazardous substance, pollutant, or contaminant under any CERCLA authority or under RCRA §7003. This policy also applies to response actions taken under §311 of the Clean Water Act, except for cleanups of petroleum products. The policy also covers cleanups at Federal facilities under §120 of SARA.

The third factor to assess is the source of funding. The revised policy applies to all Fund-financed response actions, whether EPA or the State is the lead agency. The policy does not apply to State-lead enforcement actions (even at NPL sites) if no CERCLA funds are involved. It does apply to State-lead enforcement actions where EPA provides any site-specific funding through a Cooperative Agreement or Multi-Site Cooperative Agreement, even though the State may be using its own enforcement authorities to compel the cleanup. Similarly, non-NPL sites are covered by this policy only where there is an expenditure of Fund money or where the cleanup is undertaken under CERCLA authority.

The final factor that affects how this revised policy applies is the date of the decision document. As noted earlier, there are two classes of actions subject to slightly different procedures governing off-site transfer: first, those actions resulting from pre-SARA decision documents or RCRA §7003 orders issued prior to October 17, 1986, are subject to the May 1985 policy as updated by this revised policy; and second, those actions resulting from post-SARA decision documents or RCRA §7003 orders issued after October 17, 1986, are subject to the requirements of SARA as interpreted and expanded by this revised policy. Although the procedures in this policy are similar for these two classes of actions, there are important differences (e.g., the requirements pertaining to releases from other units at a facility) that will be highlighted throughout this document.

Compliance with the revised procedures is mandatory for removal and remedial actions. However, there is an emergency exemption for removals if the OSC determines that the exigencies of the situation require off-site treatment, storage or disposal without following the requirements. This exception may be used when the OSC believes that the threat posed by the substances makes it imperative to remove the substances immediately and there is insufficient time to observe these procedures without endangering public health, welfare or the environment. In such cases, the OSC should consider temporary solutions (e.g., interim storage) to allow time to locate an acceptable facility. The OSC must provide a written explanation of his or her decision to use this emergency exemption to the Regional Administrator within 60 days of taking the action. In Regions in which authority to make removal decisions has not been fully delegated by the Regional Administrator to the OSC, the decisions discussed above must be made by the Regional official to whom removal authority has been delegated. This emergency exemption is also available to OSC's taking response actions under §311 of the Clean Water Act.

III. DEFINITIONS

<u>A. Release</u>

For the purposes of this policy, the term "release" is defined here as it is defined by $\S101(22)$ of CERCLA, which is repeated in 40 CFR 300.6 of the NCP, and the RCRA $\S3008(h)$ guidance ("Interpretation of Section 3008(h) of the Solid Waste Disposal Act", memorandum from J. Winston Porter and Courtney M. Price to the Regional Administrators, <u>et al</u>, December 16, 1985). To summarize, a release is any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injection, escaping, leaching, dumping or disposing to the environment. This includes releases to surface water, ground water, land surface, soil and air.

A release also includes a substantial threat of a release. In determining whether a substantial threat of release exists, both the imminence of the threat and the potential magnitude of the release should be considered. Examples of situations where a substantial threat of a release may exist include a weakened or inadequately engineered dike wall at a surface impoundment, or a severely rusted treatment or storage tank.

De minimis releases from receiving units are exempt; that is, they are not considered to be releases under the off-site policy. <u>De minimis</u> releases are those that do not adversely affect public health or the environment, such as releases to the air from temporary opening and closing of bungs, releases between landfill liners of 1 gallon/acre/day or less, or stack emissions from incinerators not otherwise subject to Clean Air Act permits. Releases that need to be addressed by implementing a contingency plan would not normally be considered <u>de minimis</u> releases.

Federally-permitted releases, as defined by CERCLA §101(10) and 40 CFR 300.6, are also exempt. These include discharges or releases in compliance with applicable permits under RCRA, the Clean Water Act, Clean Air Act, Safe Drinking Water Act, Marine Protection, Research and Sanctuaries Act, and Atomic Energy Act or analogous State authorities.

For purposes of this policy, an interim status unit in RCRA ground-water assessment monitoring (under 40 CFR 265.93) or a permitted unit in compliance monitoring (under 40 CFR 264.99) is not presumed to have a release. EPA will evaluate available information, including the data which led to a determination of the need for assessment or compliance monitoring, data gathered during assessment monitoring, and any other relevant data, including that gathered from applicable compliance inspections. A determination of unacceptability should be made when information will support the conclusion that there is a probable release to ground water from the receiving unit. Finding a release can happen at any time before, during or after an assessment or compliance monitoring program.

On the other hand, it is not necessary to have actual sampling data to determine that there is a release. An inspector may find other evidence that a release has occurred, such as a broken dike or feed line at a surface impoundment. Less obvious indications of a release might also be adequate to make the determination. For example, EPA could have sufficient information on the contents of a land disposal unit, the design and operating characteristics of the unit, or the hydrogeology of the area in which the unit is located to conclude that there is or has been a release to the environment.

B. Receiving Unit

The receiving unit is any unit that receives off-site CERCLA waste:

- for treatment using BDAT, including any pretreatment or storage units used prior to treatment;
- (2) for treatment to substantially reduce its mobility,

toxicity or persistence in the absence of a defined BDAT; or

(3) for storage or ultimate disposal of waste not treated to the previous criteria.

Note that the acceptability criteria may vary from unit to unit, and that the receiving unit may vary from transfer to transfer.

C. Other Units

Other units are all other regulated units and solid waste management units (SWMU's) at a facility that are not receiving units.

D. Controlled Release

In order to be considered a controlled release, the release must be addressed by a RCRA corrective action program (incorporated in a permit or order) or a corrective action program approved and enforceable under another applicable Federal or delegated State authority.

E. Relevant Violations

Relevant violations include Class I violations as defined by the RCRA Enforcement Response Policy (December 21, 1984, and subsequent revisions) at or affecting a receiving unit. A Class I violation is a significant deviation from regulations, compliance order provisions or permit conditions designed to:

- Ensure that hazardous waste is destined for and delivered to authorized facilities;
- o Prevent releases of hazardous waste or constituents to the environment;
- o Ensure early detection of such releases; or
- o Compel corrective action for releases.

Recordkeeping and reporting requirements (such as failure to submit the biennial report or failure to maintain a copy of the closure plan at the facility) are generally not considered to be Class I violations.

Violations affecting a receiving unit include all ground-water monitoring violations unless the receiving unit is outside the waste management area which the ground-water monitoring system was designed to monitor. Facility-wide Class I violations (such as failure to comply with financial responsibility requirements, inadequate closure plan, inadequate waste analysis plan, inadequate inspection plan, etc.) that affect the receiving unit are also relevant violations.

Violations of State or other Federal laws should also be examined for relevance, considering the significance of the requirement that is being violated; the extent of deviation from the requirement; and the potential or actual threat to human health or the environment.

F. Relevant Release

- A relevant release under this revised policy includes:
- Any release or significant threat of release of a hazardous substance (defined in 40 CFR 300.6) not previously excluded (i.e., <u>de minimis</u> releases or permitted releases) at all units of a RCRA Subtitle C land disposal facility and at receiving units of a' RCRA Subtitle C treatment or storage facility; and
- Environmentally significant releases of any hazardous substance not previously excluded at non-receiving units at RCRA Subtitle C treatment and storage facilities and at all units at other facilities.

G. Relevant Conditions

Relevant conditions include any environmental conditions (besides a relevant violation) at a facility that pose a significant threat to public health, welfare or the environment or that otherwise affect the satisfactory operation of the facility.

H. Responsible Agency

Determinations of acceptability to receive an off-site transfer of CERCLA waste will be made by EPA or by States authorized for corrective action under §3004(u) of RCRA. References in this document to the "responsible Agency" refer only to EPA Regions or to States with this authority.

I. Responsible Government Official

The responsible government official is that person authorized in the responsible Agency to make acceptability determinations under this revised policy.

A. Acceptability Criteria for Wastes Generated Under Pre-SARA Decision Documents

CERCLA wastes from actions resulting from pre-SARA decision documents and pre-SARA RCRA §7003 orders may go to a facility meeting the following criteria:

- o There are no relevant violations at or affecting the receiving unit; and
- o There are no relevant conditions at the facility (i.e., other environmental conditions that pose a significant threat to public health, welfare or the environment or otherwise affect the satisfactory operation of the facility).

In order to determine if there is a relevant violation, an appropriate compliance inspection must be conducted no more than six months before the expected date of receipt of CERCLA waste. This inspection, at a minimum, must address all regulated units. This inspection may be conducted by EPA, a State or an authorized representative. When a State conducts the inspection, it should determine the facility's compliance status. Where a violation or potential violation comes to EPA's attention (e.g., through a citizen complaint or a facility visit by permit staff), the Region or State is expected to investigate whether a violation occurred as soon as is reasonably possible.

The May 1985 policy does not refer specifically to releases. Rather, a corrective action plan is required for relevant conditions. Therefore, in some cases, a facility receiving CERCLA wastes from an action subject to a pre-SARA decision document may not need to institute a program to control releases. Releases will be evaluated by the responsible Agency to determine whether such releases constitute relevant conditions under this policy.

The activities related to determining acceptability, providing notice to facilities, regaining acceptability and implementation procedures are discussed in the "Implementation" section of this document, and apply to off-site transfers of waste generated under pre-SARA and post-SARA decision documents.

B. <u>Acceptability Criteria for Wastes Generated Under Post-SARA</u> Decision Documents

Under this revised policy, there are three basic criteria that are used to determine the acceptability of a facility to receive off-site transfers of CERCLA waste generated under a post-SARA decision document or post-SARA RCRA §7003 cleanup. The criteria are:

- o There must be no relevant violations at or affecting the receiving unit;
- There must be <u>no</u> releases from receiving units and contamination from prior releases at receiving units must be addressed as appropriate; and
- Releases at other units must be addressed as appropriate.

The last two criteria are applied somewhat differently, depending on the type of facility. These differences are described below.

1. Criteria Applicable to All RCRA Subtitle C Treatment. Storage and Disposal Facilities. The first criterion that applies to all Subtitle C facilities is that there can be no relevant violations at or affecting the receiving unit. As discussed earlier, this determination must be based on an inspection conducted no more than six months prior to receipt of CERCLA waste.

A second element that applies to all Subtitle C facilities is that there must be <u>no</u> releases at receiving units. Releases from receiving units, except for <u>de minimis</u> releases and Stateand Federally-permitted releases, must be eliminated and any prior contamination from the release must be controlled by a corrective action permit or order under Subtitle C, as described in the next section.

The final criterion that applies to all Subtitle C facilities, is that the facility must have undergone a RCRA Facility Assessment (RFA) or equivalent facility-wide investigation. This investigation addresses EPA's affirmative duty under CERCLA §121(d)(3) to determine that there are no releases at the facility.

Releases of RCRA hazardous waste or hazardous constituents and CERCLA hazardous substances are all included under the policy. While the RFA need not focus on identifying releases of hazardous substances that are not RCRA hazardous wastes or hazardous constituents, to the extent such releases are discovered in an RFA or through other means, they will be considered the same as a release of hazardous waste or hazardous constituents.

o <u>Additional Criteria Applicable to RCRA Subtitle C Land</u> <u>Disposal Facilities</u>. Land disposal facilities must meet additional requirements imposed by SARA and this policy. The term "land disposal facility" means any RCRA facility at which a land disposal unit is located, regardless of whether the land disposal unit is the receiving unit. Land disposal units include surface impoundments, landfills, land treatment units and waste piles.

As stated earlier, there must be no releases at or from receiving units. In addition, releases from other units at a land disposal facility must be controlled under a corrective action program. The RFA will help determine whether there is a release. In addition, land disposal facilities must have received a comprehensive ground-water monitoring evaluation (CME) or an operation and maintenance (O&M) inspection within the last year.

Units at RCRA Subtitle C land disposal facilities receiving CERCLA waste that is also RCRA hazardous waste must meet the RCRA minimum technology requirements of RCRA §3004(0). Only where a facility has been granted a waiver can a land disposal unit not meeting the minimum technology requirements be considered acceptable for off-site disposal of CERCLA waste that is RCRA hazardous waste.

o <u>Criteria Applicable to Subtitle C Treatment and Storage</u> <u>Facilities</u>. The criterion for controlling releases from other units does not apply to all releases at treatment and storage facilities, as it does at land disposal facilities. Releases from other units at treatment and storage facilities must be evaluated for environmental significance and their effect on the satisfactory operation of the facility. If determined by the responsible Agency to be environmentally significant, releases must be controlled by a corrective action program under an applicable authority. Releases from other units at treatment and storage facilities determined not to be environmentally significant do not affect the acceptability of the facility for receipt of CERCLA waste.

2. <u>Criteria Applicable to RCRA Permit-by-Rule Facilities</u>. This revised policy is also applicable to facilities subject to the RCRA permit-by-rule provisions in 40 CFR 270.60. These include ocean disposal barges or vessels, injection wells and publicly owned treatment works (POTWs). Permit-by-rule facilities receiving RCRA hazardous waste must have a RCRA permit or RCRA interim status. RCRA permit-by-rule facilities must also receive an inspection for compliance with applicable RCRA permit or interim status requirements. In addition, these facilities (and other non-RCRA facilities) should be inspected by the appropriate inspectors for other applicable laws.

In general, except for POTWs (discussed below), these facilities will be subject to the same requirements as RCRA treatment and storage facilities. That is, there can be no releases of hazardous waste, hazardous constituents or hazardous substances from receiving units. There also can be no relevant violations at or affecting the receiving unit, as confirmed by an inspection conducted no more than six months prior to the receipt of CERCLA waste. Releases from other units determined by the responsible Agency to be environmentally significant must be controlled by an enforceable agreement under the applicable authority.

Criteria for discharge of wastewater from CERCLA sites to POTWs can be found in a memorandum titled, "Discharge of Wastewater from CERCLA Sites into POTWs," dated April 15, 1986. That memorandum requires an evaluation during the RI/FS process for the CERCLA site to consider such points as:

- o the quantity and quality of the CERCLA wastewater and its compatibility with the POTW;
- o the ability of the POTW to ensure compliance with applicable pretreatment standards;
- o the POTWs record of compliance with its NPDES permit; and
- o the potential for ground-water contamination from transport to or impoundment of CERCLA wastewater at the POTW.

Based on a consideration of these and other points listed in the memorandum, the POTW may be deemed appropriate or inappropriate for receipt of CERCLA waste.

3. <u>Criteria Applicable to Non-Subtitle C Facilities</u>. In some instances, it may be appropriate to use a non-Subtitle C facility for off-site transfer: for example, PCB disposal is regulated under the Toxic Substances Control Act (TSCA); nonhazardous waste disposal is regulated under Subtitle D of RCRA and applicable State laws; and disposal of radionuclides is regulated under the Atomic Energy Act. At such facilities, all releases are treated in the same manner as releases from other units at Subtitle C treatment and storage facilities. That is, the responsible Agency should make a determination as to whether the release is environmentally significant and, if so, the release should be controlled by a corrective action program under the applicable Federal or State authority.

١

Requirements for the disposal of PCBs are established in 40 CFR 761.60. Generally, these regulations require that whenever disposal of PCBs is undertaken, they must be incinerated, unless the concentrations are less than 50 ppm. If the concentrations are between 50 and 500 ppm, the rule provides for certain exceptions that provide alternatives to the incineration requirements. The principal alternative is disposal in a TSCA-permitted landfill for PCBs. If a TSCA landfill is the receiving unit for PCBs, then that facility is subject to the same criteria applicable if a RCRA land disposal unit is the receiving unit; i.e., no relevant violations, no releases at the receiving unit and controlled releases at other units. PCBs at levels less than 50 ppm may be transported to acceptable Subtitle D facilities as discussed previously.

V. IMPLEMENTATION

A. Determining Acceptability

Acceptability determinations under the off-site policy will be made by EPA or by States authorized for corrective action under §3004(u) of RCRA. Where States have such authority, the State may make acceptability determinations for facilities in the State in consultation with EPA. Regardless of a State's authorization status, the Region and States should establish, in the Superfund Memorandum of Agreement, mechanisms to ensure timely exchange of information, notification of facilities and coordination of activities related to the acceptability of facilities and potential selection of facilities for off-site transfer. The Regions and States also need to establish or enhance coordination mechanisms with their respective RCRA program staffs in order to ensure timely receipt of information on inspections, violations and releases. These agreements can be embodied in State authorization Memoranda of Agreement, State grant agreements, or State-EPA enforcement agreements.

The responsible government official in the Region or State in which a hazardous waste facility is located will determine whether the facility has relevant violations or releases which may preclude its use for off-site transfer of CERCLA wastes. Each Region and State should have a designated off-site coordinator responsible for ensuring effective communication between CERCLA response program staff and RCRA enforcement staff within the Regional Offices, with States, and with other Regions and States.

The off-site coordinator should maintain a file of all information on the compliance and release status of each commercial facility in the Region or State. This information should be updated based on the results of State- or EPA-conducted compliance inspections or other information on these facilities.

CERCLA response program staff should identify potential off-site facilities early in the removal action or the remedial design process and check with the appropriate Regional and/or State off-site coordinator(s) regarding the acceptability status of the facilities. If one or more facilities is identified that has not received an inspection within the last six months, the Regional off-site coordinator(s) should arrange to have such inspection(s) conducted within a timeframe dictated by the project schedule. The CERCLA REM/FIT contractor may conduct the inspection under the direction of the Deputy Project Officer. If contractor personnel are used, the Region should ensure that such personnel are adequately trained to conduct the inspections.

Responsible Agencies should base their acceptability determinations on an evaluation of a facility's compliance status and, as appropriate, whether the facility has releases or other environmental conditions that affect the satisfactory operation of the facility. States not authorized for HSWA corrective action may assist EPA in making the acceptability determination by determining a facility's compliance status (based on a State inspection) and providing this information to EPA. Regions and States should use the following types of information to make acceptability determinations:

- State- or EPA-conducted inspections. EPA will continue to assign high priority to conducting inspections at commercial land disposal, treatment and storage facilities. Facilities designated to receive CERCLA waste must be inspected within six months of the planned receipt of the waste. In addition, land disposal facilities must have received a comprehensive ground-water monitoring inspection (CME) or an operation and maintenance (O&M) inspection within the last year, in accordance with the timeframes specified in the RCRA Implementation Plan (RIP).
- o <u>RCRA Facility Assessments (RFAs)</u>. To be eligible under this policy, a RCRA Subtitle C facility must have had an RFA or equivalent facility-wide investigation. The RFA or its equivalent must be designed to identify existing and potential releases of hazardous waste and hazardous constituents from solid waste management units at the facility.
- o <u>Other data sources</u>. Other documents such as the facility's permit application, permit, Ground Water Task Force report, ground-water monitoring data or

ground-water assessment report can contain information on violations, releases or other conditions. Relevant information from these documents should also be used to determine a facility's acceptability to receive waste under the off-site policy.

B. Notice Procedures

EPA expects that Regions and States will take timely and appropriate enforcement action on determining that a violation has occurred. Where a responsible Agency performs an inspection that identifies a <u>relevant</u> violation at a commercial facility likely to accept CERCLA wastes, within five working days of the violation determination, the responsible Agency must provide written notice to the facility of the violation and the effects of applying this policy. States not authorized for HSWA corrective action should inform EPA of the violation so that EPA can notify the facility of the effect of the violation under this policy. (See RCRA Enforcement Response Policy for a discussion of appropriate enforcement responses and timeframes for Class I violations.)

When the responsible Agency determines that a relevant release has occurred, or that relevant conditions exist, the responsible Agency must notify the facility in writing within five working days of that determination. The notice must also state the effect of the determination under this policy. A copy of any notice must also be provided to the non-issuing Region or State in which the facility is located. States not authorized for HSWA corrective action should provide EPA with information on releases so that EPA can determine whether a relevant release has occurred.

Private parties conducting a response action subject to this policy will need to obtain information on the acceptability of commercial facilities. The responsible Agency must respond with respect to <u>both</u> pre-SARA and post-SARA wastes. In addition, the responsible Agency should indicate whether the facility is currently undergoing a review of acceptability and the date the review is expected to be completed. No enforcement sensitive or predecisional information should be released.

A facility may submit a bid for receipt of CERCLA waste during a period of unacceptability. However, a facility must be acceptable in order to be awarded a contract for receipt of CERCLA waste.

<u>Scope and Contents of the Notice</u>. The responsible Agency must send the notice to the facility owner/operator by certified and first-class mail, return receipt requested. The certified notice, if not acknowledged by the receipt return card, will be considered to have been received by the addressee if properly sent by first-class mail to the last address known to the responsible Agency. The notice should contain the following:

- A finding that the facility may have conditions that render it unacceptable for receipt of off-site waste, based upon available information from an RFA, an inspection, or other data sources;
- A description of the specific acts, omissions or conditions that form the basis of the findings;
- o Notice that the facility owner/operator has the opportunity to request an informal conference with the responsible government official to discuss the basis for the facility's unacceptability determination under this revised policy, provided that such a request is made within 10 calendar days from the date of the notice. The owner/operator may submit written comments within 30 calendar days from the date of the notice in lieu of holding the conference.
- Notice that failure to request an informal meeting or submit written comments will result in no further consideration of the determination by the responsible Agency during the 60 calendar days after issuance of the notice. The responsible Agency will cease any transport of CERCLA waste to the facility on the 60th calendar day after issuance of the notice.
- Notice that the owner/operator may request, within 10 calendar days of hearing from the responsible government official after the informal conference or the submittal of written comments, a reconsideration of the determination by the Regional Administrator or appropriate State official. The Regional Administrator or State official may agree to review the determination at his or her discretion; and
- Notice that such a review by the Regional Administrator or appropriate State official, if agreed to, will be conducted within 60 calendar days of the initial notice, if possible, but that the review will not stay the determination.

The facility may continue to receive CERCLA waste for 60 calendar days after issuance of the initial notice. As indicated above, facility owners or operators may request an informal conference with the responsible government official

within 10 calendar days from the date of issuance of the notice, to discuss the basis for a violation or release determination and its relevance to the facility's acceptability to receive CERCLA wastes. Any such meeting should take place within 30 calendar days of the date the initial notice is issued. If unacceptability is based on a State inspection or enforcement action, a representative of the State should attend the meeting. If the State does not attend, EPA will notify the State of the outcome of the meeting. The owner/opeator may submit written comments within 30 calendar days from the date of the notice in lieu of holding the conference. If the responsible Agency does not find that the information submitted at the informal conference or in comments is sufficient to support a finding of acceptability to receive CERCLA wastes, it should so inform the facility orally or in writing.

Within 10 calendar days of hearing from the responsible government official after the informal conference or the submittal of written comments, the facility owner or operator may request a reconsideration of the determination by the Regional Administrator or appropriate State official. The Regional Administrator or appropriate State official may use his or her discretion in deciding whether to conduct a review of the determination. Such a review, if granted, should be conducted within the 60 day period (originating with the notice) to the extent possible. The review will not stay the determination.

The RPM, OSC or equivalent site manager must stop transfer of waste to a facility on the 60th calendar day after issuance of a notice. The facility then remains unacceptable until such time as the responsible Agency notifies the owner or operator otherwise. The off-site coordinator and the OSC/RPM should maintain close coordination throughout the 60-day period.

In limited cases, the responsible Agency may use its discretion to extend the 60 day period if it requires more time to review a submission. The facility should be notified of any extension, and it remains acceptable during any extension.

The responsible Agency may also use its discretion to determine that a facility's unacceptability is immediately effective upon receipt of a notice to that effect. This may occur in situations such as, but not limited to, emergencies (e.g., fire or explosion) or egregious violations (e.g., criminal violations or chronic recalcitrance) or other situations that render the facility incapable of safely handling CERCLA waste.

Implementation of this notice provision does not relieve the Regions or States from taking appropriate enforcement action under RCRA or CERCLA.

<u>C. Procedures for Facilities with Outstanding Unacceptability</u> <u>Determinations</u>

Under the original May 1985 off-site policy, facilities determined to be unacceptable to receive CERCLA wastes were provided with written notice and were generally afforded informal opportunities to comment on the determination (the latter step was not required by the policy). Although the Agency believes that these steps represented adequate procedural safeguards for facilities seeking to receive CERCLA wastes, EPA has decided to provide an additional opportunity for review, in light of this revised policy, for facilities with unacceptability determinations already in place on the effective date of the revised policy.

Any such facility that wishes to meet with the responsible Agency to discuss the basis for a violation or release determination and its relevance to the facility's ability to receive CERCLA wastes, may request an informal conference with or submit written comments to the responsible Agency at any point up to the 60th day after the publication of the proposed rule on the off-site policy in the Federal Register. Such a meeting should take place within 30 calendar days of the request. If the responsible government Agency does not find the information presented to be sufficient to support a finding of acceptability to receive CERCLA wastes, then it should inform the facility orally or in writing that the unacceptability determination will continue to be in force. The facility may, within 10 calendar days of hearing from the responsible government official after the informal conference or submittal of written comments, petition the EPA Regional Administrator or appropriate State official for reconsideration. The Regional Administrator or State official may use his or her discretion in deciding whether to grant reconsideration.

These procedures for review of unacceptability determinations that were already in place on the effective date of this revised policy will not act to stay the effect of the underlying unacceptability determinations during the period of review.

D. Re-evaluating Unacceptability

An unacceptable facility can be reconsidered for management of CERCLA wastes whenever the responsible Agency finds that the facility meets the criteria described in the "Acceptability Criteria" section of this policy.

For the purposes of this policy, releases will be considered controlled upon issuance of an order or permit that initiates and requires completion of one or more of the following: a facility-wide RCRA Facility Investigation (RFI); a Corrective Measures Study (CMS); or Corrective Measures Implementation (CMI). The facility must comply with the permit or order to remain acceptable to receive CERCLA waste. At the completion of any such phase of the corrective action process, the responsible Agency should again review the facility for acceptability under the off-site policy using the criteria listed in this document, and as necessary and appropriate, make new acceptability determinations, and issue additional orders or modify permit conditions to control identified releases. Releases that require a determination of environmental significance will be considered controlled upon issuance of an order or permit to conduct an RFI, CMS or CMI, or upon completion of an RFI which concludes that the release is not

environmentally significant. Again, the facility must comply with the permit or order to remain acceptable to receive CERCLA waste.

If the facility is determined to be unacceptable as a result of relevant violations at or affecting the receiving unit, the State (if it made the initial determination) or EPA must determine that the receiving unit is in full physical compliance with all applicable requirements. Where a State not authorized for HSWA corrective action makes this determination, it should notify EPA immediately of the facility's return to compliance, so that the Agency can expeditiously inform the facility that it is once again acceptable to receive CERCLA wastes.

The responsible Agency will notify the facility of its return to acceptability by certified and first-class mail, return receipt requested.

E. Implementation Procedures

All remedial decision documents must discuss compliance with this policy for alternatives involving off-site management of CERCLA wastes. Decision documents for removal actions also should include such a discussion.

Provisions requiring compliance with this policy should be included in all contracts for response action, Cooperative Agreements with States undertaking Superfund response actions, and enforcement agreements. For ongoing projects, these provisions will be implemented as follows, taking into consideration the differences in applicable requirements for pre- and post-SARA decision documents:

> o <u>RI/FS</u>: The Regions shall immediately notify Agency contractors and States that alternatives for off-site

management of wastes must be evaluated against the provisions of this policy.

- <u>Remedial Design</u>: The Regions shall immediately notify Agency contractors, the States, and the U.S. Army Corps of Engineers that all remedies that include off-site disposal of CERCLA waste must comply with the provisions of this policy.
- <u>Remedial Action</u>: The Regions shall immediately assess the status of compliance, releases and other environmental conditions at facilities receiving CERCLA waste from ongoing projects. If a facility is found not to be acceptable, the responsible Agency should notify the facility of its unacceptability.
- <u>Enforcement</u>: Cleanups by responsible parties under enforcement actions currently under negotiation and all future actions must comply with this policy. Existing agreements need not be amended. However, EPA reserves the right to apply these procedures to existing agreements, to the extent it is consistent with the release and reopener clauses in the settlement agreement.

If the response action is proceeding under a Federal lead, the Regions should work with the Corps of Engineers or EPA Contracts Officer to negotiate a contracts modification to an existing contract, if necessary. If the response action is proceeding under a State lead, the Regions should amend the Cooperative Agreement.

Transmission of Guidance for Conducting Federal-Lead Underground Storage Tank Corrective Actions



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUL 2 5 ::::

OFFICE OF SOLID WASTE AND EMERGENCY RESPOR

OSWER DIRECTIVE 9360.0-16A

MEMORANDUM

SUBJECT: Transmission of Guidance for Conducting Federal-Lead Underground Storage Tank-Corrective Actions

FROM: J. Winston Porter. MAssistant Administrate TO: Regional Administrators Regions I-X

ATTN: Regional Waste Management Division Directors Regions I-X

Attached is the final guidance for removal program personnel conducting Federal-lead corrective actions for petroleum leaks from underground storage tanks (USTs). This guidance supersedes the interim guidance of June 4, 1987 (OSWER directive # 9360.0-16). Based on comments offered by reviewers, changes have been made to clarify definitions and procedures throughout the guidance. A separate guidance is under development regarding corrective actions on Indian lands.

In response to comments, further emphasis has been placed on the responsibility of State governments to conduct responses in non-emergency situations and enforcement activities. In several places, the guidance now stresses EPA policy requiring the State to perform any remedial action after the Federal response stabilizes a situation. The role of the Regional UST Coordinator in the process of implementing and overseeing Federal-lead UST corrective actions has been greatly increased. Finally, specific emphasis has been placed on the necessity for contacting owners or operators before initiating corrective action.

If there are any questions regarding this guidance, please contact Elizabeth Zeller at FTS 382-7735.

Attachment

cc: Superfund Branch Chiefs, Regions I-X Henry Longest Ron Brand

OSWER DIRECTIVE 9360.0-16A

•

GUIDANCE FOR CONDUCTING FEDERAL-LEAD UNDERGROUND STORAGE TANK CORRECTIVE ACTIONS

-

June, 1988

Prepared by the Emergency Response Division Office of Emergency and Remedial Response

•

DISCLAIMER

The policies and procedures established in this document are intended solely for the guidance of government personnel. They are not intended, and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. The Agency reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.
Section

Page

1.0	Introduction				
	1.1	Purpose			
	1.2	Background			
	1.3	Policy			
	1.4	Definitions			
	T • A				
2.0	Criteria for Corrective Action4				
	2.1	Legislative Standards and Criteria4			
	2.2	EPA Criteria			
3.0	Fund	Utilization			
	3.1	Eligible Activities6			
	3.2	State Cost Share6			
4.0	wbbr	Delegation of hythemitics			
	4.1	Delegation of Authorities			
	4.2	Oral Request from State			
	4.3	written Request from State10			
•	4.4	Regional UST Coordinator/Enforcement11			
5.0	Accounting Information13				
6.0	Init	iation of On-Site Actions			
	6.1	Procuring Cleanup Services			
	6.2	Procuring Technical Assistance			
	6.3	Assistance from Other Agencies.			
	6.4	Stabilization Standards			
	•••	,			
7.0	Managing Federal-Lead UST Corrective Actions				
	7.1	Oversight of Contractors			
	7.2	Allowable Costs			
	7.3	Non-Allowable Costs			
	7.4	Cost Management			
	7.5	Ceiling Increases/Scope Modifications22			
	7.6	Access Agreements			
	7.7	Worker and Visitor Health and Site Safety25			
	7.8	Community Relations and Public Information26			
	7.9	Record-Keeping Requirements			
	7.10	Reporting Requirements			
	7.11	Post-Corrective Action Control			

Appendices

А.	Delegations		
в.	Oral Request	Information	Outline

- State Request Letter Format c.
- Action Memorandum Format D.
- E.
- Ceiling Increase Request Format Federal-Lead UST Corrective Action Accident Report Format F.
- Delivery Order Preparation and Processing G.
- Technical Direction Document Preparation and Processing H.
- Interim Financial Procedures Governing Use of the LUST Trust F" I.

1.0 INTRODUCTION

1.1 Purpose

This guidance document is designed to provide direction to On-Scene Coordinators (OSCs), other removal program personnel, and Regional Underground Storage Tank (UST) Coordinators for the initiation and continuation of Federal-lead corrective actions in response to petroleum releases from leaking underground storage tanks. The guidance presents procedures to be followed, including required justifications and the documentation necessary for undertaking a Federal-lead UST corrective action. A separate guidance document is being developed for corrective actions on Indian lands.

1.2 Background

The Superfund Amendments and Reauthorization Act of 1986 (SARA) amends Subtitle I of the Solid Waste Disposal Act (SWDA) to provide the authority for Federal response to petroleum releases from leaking USTs. The amendments also establish a \$500 million Leaking Underground Storage Tank (LUST) Trust Fund to finance both Federal and State UST program activities. Federallead UST corrective actions will be performed by the same EPA emergency response and contractor personnel that conduct oil and hazardous substance removal actions.

Note: As used in this guidance, "SWDA" refers to the Solid Waste Disposal Act as amended by any other legislation, including SARA, the Resource Conservation and Recovery Act of 1976 (RCRA), and the Hazardous and Solid Waste Amendments of 1984 (HSWA).

1.3 Policy

It is EPA's policy that LUST Trust Fund-financed responses at petroleum releases from underground storage tanks will be conducted by States under cooperative agreement with EPA, except in rare circumstances. Most States will have broad cooperative agreements to address emergency response and to perform cleanups; in the absence of such agreements, the Region and State are encouraged to develop site-specific cooperative agreements, under which the State will conduct corrective actions at individual sites. EPA will undertake a corrective action only in instances where there is a major public health or environmental emergency, and the State is unable to respond. In addition, the State must demonstrate that it has contacted or attempted to contact owners or operators, and the State is convinced that owners or operators are unable or unwilling to provide adequate, timely response. Federal-lead corrective action will be limited to stabilization of the immediate situation, with the expectation that further cleanup will be performed by the State under an appropriate cooperative agreement.

-1-

1.4 Definitions

For the purposes of this guidance, the following definitions apply:

Exposure Assessment - As defined in Section 9003(h)(10) of SWDA, "the term 'exposure assessment' means an assessment to determine the extent of exposure of, or potential for exposure of, individuals to petroleum from a release from an underground storage tank based on such factors as the nature and extent of contamination and the existence of or potential for pathways of human exposure (including ground or 'surface water contamination, air emissions, and food chain contamination), the size of the community within the likely pathways of exposure, and the comparison of expected human exposure levels to the short-term and long-term health effects associated with identified contaminants and any available recommended exposure or tolerance limits for such contaminants. Such assessment shall not delay corrective action to abate immediate hazards or reduce exposure."

<u>Maior Public Health or Environmental Emergency</u> - To qualify for Federal response action, an UST site must be deemed a major public health or environmental emergency. (This definition is more strict than that for current hazardous substance removal actions and is intended to significantly limit the number of Federal-lead UST responses, so that only health or environmental emergencies are addressed). Such an emergency exists if the following criteria are met:

- o The release poses an immediate and substantial threat of direct human, animal, or food chain exposure to petroleum; or
- o The release poses an immediate threat of fire and/or explosion; or
- The release poses an immediate and substantial threat to public drinking water supplies; or
- o The release poses an immediate threat to human health or substantial amounts of property, or poses an immediate and substantial threat to natural resources.

<u>Petroleum</u> - As defined in Section 9001(8) of SWDA, "the term 'petroleum' means petroleum, including crude oil or any fraction thereof which is liquid at standard conditions of temperature and pressure (60 degrees Fahrenheit and 14.7 pounds per square inch absolute)." This term includes, but is not limited to, gasoline, diesel fuel, and jet fuel. <u>Underground Storage Tank</u> - As defined in Section 9001(1) of SWDA, "the term 'underground storage tank' means any one or combination of tanks (including underground pipes connected thereto) which is used to contain an accumulation of regulated substances, and the volume of which (including the volume of the underground pipes connected thereto) is 10 per centum or more beneath the surface of the ground. Such term does not include any:

- Farm or residential tank of 1,100 gallons or less capacity used for storing motor fuel for noncommercial purposes,
- B) Tank used for storing heating oil for consumptive use on the premises where stored,
- C) Septic tank,
- D) Pipeline facility (including gathering lines) regulated under
 - i) the Natural Gas Pipeline Safety Act of 1968 (49 U.S.C. App. 1671, et seq.),
 - ii) the Hazardous Liquid Pipeline Safety Act of 1979 (49 U.S.C. App. 2001, et seq.), or
 - iii) which is an intrastate pipeline facility regulated under State laws comparable to the provisions of law referred to in clause (i) or (ii) of this subparagraph,
- E) Surface impoundment, pit, pond, or lagoon,
- F) Storm water or waste water collection system,
- G) Flow-through process tank,
- H) Liquid trap or associated gathering lines directly related to oil or gas production and gathering operations, or
- I) Storage tank situated in an underground area (such as a basement, cellar, mineworking, drift, shaft, or tunnel) if the storage tank is situated upon or above the surface of the floor.

The term `underground storage tank' shall not include any pipes connected to any tank which is described in subparagraphs (A) through (I)."

Final clarification of these terms and definitions will be found in the regulations (and preamble) for underground storage tanks, scheduled for publication later in 1988.

2.0 CRITERIA FOR CORRECTIVE ACTION

2.1 Legislative Standards and Criteria

This guidance is designed to provide direction for undertaking Federal-lead corrective action at petroleum leaks from underground storage tanks. The basic criteria for a Federal response are found in Section 9003 of the amended SWDA; Subsection (c) specifies the release detection, prevention, and correction requirements for USTs to be promulgated by EPA, and Subsection (h) provides two sets of criteria for Federal-lead UST responses.

Before the effective date of the regulations promulgated under Section 9003(c) of SWDA, a corrective action may be undertaken if the Administrator (or State under cooperative agreement) deems it necessary to protect human health and the environment, as authorized in Section 9003(h)(1) of SWDA. The EPA (or the responding State) must give priority to sites where no owner or operator is able to undertake proper corrective action, and to UST releases posing the greatest threat to human health and the environment.

After the regulations become effective, a response may be conducted only if corrective action is necessary to protect human health and the environment and one or more of the following criteria, presented in Section 9003(h)(2) of SWDA, are met:

- No financially solvent tank owner or operator, capable of carrying out proper corrective action, is found;
- o The situation requires prompt action to protect human health and the environment;
- Corrective action costs exceed the financial responsibility requirements established for the tank, and expenditures from the LUST Trust Fund are necessary to ensure effective action; or
- The tank owner or operator has failed, or refused, to comply with an administrative order to perform corrective action.

Priority must be given to releases posing the greatest threat to human health and the environment.

\$

2.2 EPA Criteria

In addition to the basic policy guidelines discussed in Section 1.3, and the legislative restrictions presented in Section 2.1, EPA will undertake corrective action only if a major public health or environmental emergency exists, as defined in Section 1.4.

EPA will take only those corrective actions that stabilize the emergency situation, allowing State or other responders to implement permanent cleanup remedies. If EPA's mitigative measures require continuing post-corrective action control (operations and maintenance), the affected State will be responsible for these measures as soon as possible.

3.0 FUND UTILIZATION

In accordance with SWDA Section 9003(h), the statute that describes Federal response under the LUST Trust Fund, and in compliance with EPA policy for Federal-lead actions, the general activities listed in Section 3.1 may be conducted with Fund monies. See Sections 7.2 and 7.3 for more detail on allowable and non-allowable costs related to performing corrective actions.

3.1 Eligible Activities

- Site investigations or exposure assessments (see
 Section 1.4) to determine potential health effects of a leak, and to establish corrective action priorities.
- Corrective actions such as stabilization of the site, temporary relocation of affected residents, or provision of alternate household water supplies.
- Enforcement activities (see Section 4.4) including development, implementation, and oversight of enforcement orders directed to responsible tank owners/operators.
- Cost recovery activities to seek reimbursement of LUST Trust Fund expenditures from liable parties.

3.2 State Cost Share

Until EPA's final regulations for release detection, prevention, and correction become effective later in 1988, there is no requirement for States to cost share or match Trust Fund monies. After the effective date of the regulations, States must pay a 10 percent share of the cost of LUST Trust Fund corrective actions, as required by Section 9003(h)(7)(B) of SWDA. However, failure to pay this required cost share will not prevent EPA from conducting a response if immediate action is necessary to mitigate an imminent and substantial endangerment to human health or the environment.

4.0 APPROVAL PROCESS

Federal-lead UST corrective actions may be approved only if the definitions and criteria outlined in Sections 1.0 and 2.0 are met. In particular, the site must pose a major public health or environmental emergency to which the owner or operator or the State is unable to respond in an adequate and timely manner. In addition, no Federal-lead corrective action will be conducted unless an appropriate request is received from a State.

The Federal-lead UST delegations of authority, as well as two approval processes (based on response time considerations), are presented below.

4.1 Delegation of Authorities

All obligations for Federal-lead UST corrective actions (with limited exceptions for emergency situations, as noted below in section 4.1.1) must be approved by the Assistant Administrator (AA), Office of Solid Waste and Emergency Response (OSWER), as provided by Delegation 8-33 (see Appendix A). This authority may be redelegated to the Office Director (OD), Office of Emergency and Remedial Response (OERR). If redelegated, the OD, OERR, must obtain the concurrence of the OD, Office of Underground Storage Tanks (OUST).

4.1.1 In accordance with Delegation 8-33, Regional Administrators (RAs) may approve emergency obligations for Federal-lead UST corrective actions of up to \$50,000 per site. This authority may be be used only

1) during non-duty hours (after 5:00 p.m. EST [local time in Washington, D.C.] on weekdays, or on Saturday, Sunday, or federal holidays), or if the situation precludes contacting Headquarters prior to initiating necessary response actions, and

2) if there is risk of death, injury, or catastrophic environmental damage, due to a petroleum release from an UST. Such an emergency would be posed by imminent or actual events such as:

- o Fire and/or explosion; or
- Substantial or irreversible damage to a sensitive ecosystem or significant natural resource.

RAs may redelegate this limited authority to the Division Director and OSC level, as provided in Delegation 8-33. If redelegated, the OSC's \$50,000 authority is included in the RA's \$50,000 authority; it is not in addition to that authority. This authority may be used to initiate response, and may be used to initiate project restarts should new and unforeseen emergency conditions occur which meet the above criteria. This authority cannot be used for continuations of work in progress. The costs that are applied toward the RA's authority to obligate funds up to \$50,000, and the costs that are excluded from this limit, are similar to those defined in EPA's <u>Superfund Removal Procedures --</u> <u>Revision Number Three</u>.

4.2 Oral Request from State

When the response time demands of the situation preclude implementing the formal written approval process described below in Section 4.3, the following oral approval process may be implemented.

4.2.1 The State evaluates the site and gathers all the necessary information to support an oral request (see Appendix B). The State UST Coordinator orally transmits the request and information to the Regional Oil and Hazardous Materials (OHM) Coordinator, or other designated Regional management official.

4.2.2 The assigned OSC consults with the Regional UST Coordinator, where possible (see Section 4.4).

4.2.3 The OSC (jointly with the Regional UST Coordinator, where possible) determines if the criteria for corrective action discussed in Section 2.0 are met, and if an immediate response (within hours or days) is necessary. If so, the following approval process is implemented.

4.2.4 Approval process for corrective action

4.2.4.1 During regular working hours:

- a) The OSC or other Regional official, after consultation with the Regional UST Coordinator, notifies the appropriate Headquarters Emergency Response Division (ERD) Regional Coordinator of the Region's intent to request oral approval from Headquarters to initiate an UST corrective action (phone # 8-382-2188, Magnafax # 755-2155, TWX # 710-822-9269, E-mail # EPA 5511).
 - b) The OSC gets oral approval from the RA or delegatee.
 - c) The OSC or other Regional official provides the information set forth in Appendix B to Headquarters.
 - d) The ERD Regional Coordinator reviews and processes the request, and relays the request and a recommendation for approval/denial to the appropriate Headquarters official.

- e) The ERD Regional Coordinator communicates the decision and the appropriate accounting information (see Section 5.0) to the OSC or other Regional official as quickly as possible, and confirms the decision in writing by the end of the next work day. The OSC must notify the Regional UST Coordinator of the decision.
- f) The RA transmits a formal action memorandum (see Appendix D), along with a formal State request letter (see Appendix C) to ERD within 10 days.

4.2.4.2 During non-duty hours (after 5:00 p.m. EST on weekdays, on Saturday, Sunday, and holidays), or if the situation precludes contacting Headquarters prior to initiating necessary response actions:

- a) The OSC obtains oral approval from the RA, or from his/her delegatee (see Section 4.1, or Appendix A). This authority may only be used to obligate funds up to \$50,000, for corrective actions which meet the criteria in section 4.1.1.
- b) The OSC notifies the Regional UST Coordinator (see Section 4.4) and ERD Regional Coordinator of this action as soon as possible.
- c) The RA transmits a formal action memorandum (see Appendix D), along with a State request letter (see Appendix C) to ERD within 10 working days. A ceiling increase request (see Appendix E) may be incorporated within this memorandum, if needed.

4.2.4.3 During non-duty hours, if a ceiling greater than \$50K is needed:

- a) The OSC telephones the National Response Center (NRC), 800-424-8802, identifies himself/ herself, and asks to be put in contact with the ERD Duty Officer.
- b) The NRC contacts the ERD Duty Officer.
- c) The ERD Duty Officer contacts the OSC and asks for the information contained in Appendix
 E. The ERD Duty Officer notifies the supervisory duty officer, the OD/OERR, the OD/OUST, and the AA/OSWER (if appropriate), who approves or denies the request.

- d) The ERD Duty Officer communicates the decision to the OSC as quickly as possible, and confirms the decision in writing by the end of the next work day.
- e) The RA sends a formal action memorandum (see Appendix D) along with the State request letter (see Appendix C) to ERD within 10 days.

4.3 Written Request from State

This approval process must be used whenever response time demands allow. Typically, this will mean that Federal response can be delayed for several days.

4.3.1 Receive written request from State

The affected State shall perform a thorough site evaluation to clearly determine the extent of release, source, substance(s) released, and threats posed by the release. The State shall also make every reasonable attempt to locate owners or operators and compel corrective actions. The State must then prepare a formal written request for Federal-lead UST corrective action that presents all of the pertinent site information, using the State request letter format in Appendix C. The request letter should be sent to the Regional OHM Coordinator, or other designated Regional management official, from the Director of the State UST agency designated by the Governor of the affected State.

4.3.2 Notify Regional UST Coordinator

When a letter requesting UST corrective action is received from the State, the Regional UST Coordinator must be consulted and provided a copy of the letter (see Section 4.4).

4.3.3 Evaluate State request

The OSC and Regional UST Coordinator jointly evaluate the State request to determine whether:

- Additional information is required. If the request letter (see Appendix C) is deficient, the Region should contact the State and obtain additional information, or instruct the State to resubmit the request letter incorporating the Region's comments. If necessary, the OSC may perform a perfunctory site inspection to gather needed data.
- o Federal-lead UST corrective action is not justified. If the situation does not meet the criteria discussed in Section 2.0, the OSC and the Regional UST Coordinator must recommend to the OHM Coordinator that the request be denied.

 Federal-lead UST corrective action is justified.
 If the situation appears to meet the criteria discussed in Section 2.0, the following approval procedures should be implemented.

4.3.4 Regional role

The OSC notifies the appropriate ERD Regional Coordinator of the Region's intent to request Headquarters' approval to initiate a Federal-lead UST corrective action (phone # 8-382-2188, Magnafax # 755-2155, E-mail # EPA 5511, TWX # 710-822-9269). The OSC should also forward a copy of the State request letter, or a draft copy of the action memorandum to allow the ERD Regional Coordinator to expedite the Headquarters approval process.

The OSC should provide the information set forth in the State request letter (see Appendix C) by submitting an action memorandum, using the format presented in Appendix D. The action memorandum must be signed by the RA and addressed to the AA, OSWER through the OD, OERR, to the attention of the Director, ERD.

4.3.5 Headquarters role

The ERD Regional Coordinator reviews the action memorandum, gets concurrences from other offices as necessary (e.g., OUST, OGC), and sends it with a recommendation for approval/denial to the appropriate Headquarters official for final determination and signature.

The ERD Regional Coordinator then communicates the OSWER decision to the Region as quickly as possible, along with the appropriate accounting information (see Section 5.0). Written confirmation of the decision is forwarded to the OSC and the Regional UST Coordinator as soon as practicable.

4.4 Regional UST Coordinator/Enforcement

The Regional UST Coordinator is a Regionally-designated employee who acts as a liaison between emergency response personnel and Regional UST program management. The Regional UST Coordinator may have information about a specific site, the status of a State's UST program, and the State's eligibility for a cooperative agreement under the LUST Trust Fund, that could prove useful in weighing the State's response capabilities against the characteristics and magnitude of an UST emergency. The Regional UST Coordinator should be informed as soon as possible whenever the Region receives a State request for Federal-lead UST corrective action; whenever practicable, the Regional UST Coordinator, together with the OSC, should evaluate the appropriateness of Federal-lead corrective action prior to seeking approval. The OSC should keep the Regional UST Coordinator informed of all significant events during Federallead UST corrective actions.

States, under cooperative agreements, will be expected to initiate and pursue enforcement action. The Regional UST Coordinator is responsible for coordinating with affected State agencies, particularly in identifying owners or operators and pursuing enforcement actions. EPA has the authority to issue several different types of administrative orders under SWDA: Section 9005 warning letters can be issued to compel tank testing and investigation to detect suspected leaks; Section 9003(h) orders can be issued to require site assessment, development of a response plan, and implementation of corrective action pursuant to the EPA-approved plan; and Section 9006(a) orders can be issued to enforce compliance with a previous order. States are expected to have or develop similar enforcement authorities. Enforcement efforts to secure response from owners or operators must not delay Federal-lead corrective action if site conditions meet the criteria in Section 4.1.1 of this guidance.

5.0 ACCOUNTING INFORMATION

The LUST Trust Fund appropriation number is 68-20X8153; the program element is FPYV2B. For all actions given prior Headquarters approval, account and Document Control numbers will be issued to the ERD Regional Coordinator by the Headquarters Financial and Administrative Management Support Staff (FAMS). The ERD Regional Coordinator will then inform the OSC of the assigned numbers.

Each Region has approval authority up to \$50,000 per site to use in responding to certain UST emergencies (see Section 4.1.1 for more detail). For actions initiated by the Region using its \$50,000 authority, account and Document Control numbers should be issued by the Regional Financial Management Officer.

Detailed information describing Trust Fund appropriation, account number structure, activity codes, and other relevant matters has been issued by the Comptroller General to all Financial Management Officers in Comptroller Policy Announcement Number 87-13, "Interim Financial Policies and Procedures Governing Use of the Leaking Underground Storage Tank (LUST) Trust Fund" (see Appendix I).

The Policy Announcement establishes policies and procedures for use of the Fund by EPA and State governments. Each site will be assigned a ten-digit account number in the Financial Management System (FMS), which enables tracking of site-specific costs for cost management and cost recovery purposes. The Policy Announcement details methods of charging time and other direct costs to the.Fund.

6.0 INITIATION OF ON-SITE ACTIONS

Mechanisms available to the OSC for initiating an UST response include:

- o Procurement of cleanup services;
- o Procurement of technical assistance; and
- o Assistance through other agencies.

6.1 Procuring Cleanup Services

For site tracking purposes, the Agency has defined the "start" of the action to be the date on which a cleanup contractor (e.g., ERCS) mobilizes on the site, not the date on which a site assessment is performed, the action is approved, or the first obligation occurs.

6.1.1 Emergency Response Cleanup Services (ERCS) contracts

The ERCS zone contracts, supplemented by the Regional ERCS contracts, form the core of EPA's emergency response resources. These contracts provide 24-hour, immediate delivery of cleanup contractor personnel, services, and material for response to CERCLA hazardous substance releases. The contracts also include provisions for UST response. Procedures to mobilize ERCS contractors are identical to current removal program procedures, as outlined in Superfund Removal Procedures -- Revision Number Three, ERCS Users' Manual, and EPA Superfund Emergency Contracting Procedures. An Ordering Officer should fill out a Delivery Order (DO) and Procurement Request (PR) using the accounting information provided by the ERD Regional Coordinator or the Region's Financial Management Officer. Instructions for completing and processing DOs can be found in Appendix G.

6.1.2 Other commercial contracts

When use of the ERCS contractor is inappropriate due to conflict of interest, response time considerations, or other unusual or unforeseen circumstances, other contractors may be used. For information and procedures on procuring other contractor services, OSCs should contact the appropriate Headquarters Procurement and Contracts Management Division (PCMD) Contracting Officer (CO).

6.2 Procuring Technical Assistance

6.2.1 Technical Assistance Team (TAT) contracts

The TAT contracts provide technical and management assistance to OSCs. TAT services include sampling, cleanup monitoring, documentation of site conditions and activities, project planning, health and safety planning, cost tracking,

quality assurance, and related tasks. The two TAT zone contracts include provisions for UST response. Procedures for implementing these contracts are similar to current removal program procedures as outlined by Superfund Removal Procedures -- Revision Number Three and the TAT Contract User's Manual, except that the Project Officer must give approval. The Region's Deputy Project Officer (DPO) should complete a Technical Direction Document (TDD) form, being sure to specify that the source of funding is the LUST Trust Fund (instructions for completing and processing TDDs can be found in Appendix H). The DPO must obtain the Headquarters Project Officer's (PO) approval of the TDD; the PO will contact the contractor Zone Program Manager (ZPM) as soon as possible to confirm approval. The ZPM will then communicate the PO's approval to the appropriate TAT Leader, who may only then implement the task requested by the TDD. Appendix H provides a detailed description of the process for activating TAT.

6.2.2 Environmental Response Team (ERT)

The EPA Environmental Response Team (ERT) provides a wide range of technical, analytical, investigative, and planning services. To obtain ERT assistance, contact the ERD Regional Coordinator.

6.3 Assistance from Other Agencies

Other Federal agencies may have personnel with specific expertise and experience that could be useful to OSCs in performing Federal-lead UST corrective actions. EPA has a Memorandum of Understanding (MOU) or site-specific Interagency Agreements (IAGs) with each of several agencies to facilitate their direct participation in hazardous substance responses using CERCLA monies. Although there is no provision for their participation in Federal-lead UST corrective actions, which are funded under SWDA, OSCs may contact other agencies to seek informal advice or assistance as appropriate. The following agencies may be particularly helpful.

6.3.1 Federal Emergency Management Agency (FEMA)

FEMA is often used to provide assistance with temporary and permanent relocation of affected residents and businesses during removal actions.

6.3.2 Agency for Toxic Substances and Disease Registry (ATSDR)

ATSDR can provide advice concerning exposure effects of certain substances, calculate risks to the public and the environment from releases and issue health advisories where appropriate, and can recommend cleanup levels. ATSDR may be able to provide assistance or guidance in performing exposure assessments (as defined in Section 1.4) at Federallead UST corrective action sites.

6.3.3 United States Coast Guard (USCG)

USCG is experienced in responding to a wide range of release incidents involving petroleum. USCG may have sitespecific information concerning UST releases in coastal zones or affecting surface waters, and should be notified whenever Federal-lead UST corrective action is considered at such a release to determine whether response is more appropriate under the Clean Water Act (CWA).

6.4 Stabilization Standards

Federal-lead UST corrective action will generally be limited to stabilization of the emergency conditions that justified the initiation of the response; occasionally, however, more complete cleanup may be appropriate. Whenever possible, further cleanup after stabilization should be conducted, as needed, by the State under an appropriate cooperative agreement with EPA.

When conducting stabilization actions, OSCs should consider relevant state standards and other applicable guidelines, as may be provided by the Regional UST Coordinator. The OSC should stabilize the site to a level that protects human health and the environment by mitigating the immediate threats.

7.0 MANAGING FEDERAL-LEAD UST CORRECTIVE ACTIONS

This section provides guidance on the roles and responsibilities of OSCs when managing UST response projects. A variety of topics is addressed, including oversight of contractors, allowable costs, cost management, ceiling increases and scope modifications, access agreements, health and safety, community relations, reporting requirements, and post-corrective action site control considerations.

OSCs have complete responsibility for directing response operations. This means that they must ensure that all on-site activities are consistent with Subtitle I of SWDA, as amended by Section 205 of SARA, and program policies and procedures; that all expenditures of funds are appropriate and reasonable; and that subsequent cost recovery actions will be supportable.

7.1 Oversight of Contractors

A major OSC role is oversight of the contractors performing response activities. Examples of oversight activities include:

- Preparation of the work report;
- Review and certification of the Contractor Cost Report (EPA Form 1900-55), which lists daily contractor costs;
- Daily monitoring of contractor personnel and equipment to verify satisfactory completion of the work;
- Determination of the overall project status.

In conducting these oversight activities, OSCs may request support from the TAT. Examples of such support are maintaining entry/exit logs of all contractor personnel and equipment, communicating oral or written messages from the OSC to the cleanup contractor, and maintaining logs related to project costs. The TAT staff may not, however, assume the OSC's responsibilities for directing site activity, verifying satisfactory completion of work, or approving 1900-55 forms.

Compelling circumstances, such as another response incident, may require the OSC to leave the site for more than 24 hours. The OSC may designate capable persons from Federal, State, and local agencies to act as OSC representatives to supervise response operations. TAT staff, because of their nongovernmental status, may not be designated OSC representatives.

Response program policy dictates the following guidelines for, and limitations on, the designation and activities of OSC representatives: <u>Federal employees</u> - The preferred designee would be another Federal employee because such a designee would have authority to direct, not merely oversee, contractors; to initiate activities involving expenditures of money; and to certify completion of work and costs.

Examples of appropriate Federal employees are another OSC, a non-OSC Superfund employee, or a non-Superfund employee from within the Region; an OSC or other employee from another Region; or Headquarters personnel or employees from another Federal agency such as USCG.

As a practical matter, reassignment of staff from other duties may be difficult, particularly with respect to personnel outside the program, Region, or Agency. Moreover, designation of a Federal employee who does not have adequate training in program procedures and response operations could endanger the response.

- <u>State/local staff</u> State and local officials are precluded from taking any actions involving expenditures of LUST Trust Fund monies, unless an appropriate cooperative agreement has been executed. In practice, this means that State and local representatives may transmit and supervise the implementation of the OSC's work orders but may not provide new instructions.
- <u>TAT</u> TAT personnel may not serve as OSC designees; they may, however, continue to provide support services at the site and monitor cleanup contractor performance in the absence of the OSC.

Because of the practical difficulties in designating an OSC representative who can assume full on-site responsibilities, OSCs are discouraged from leaving the site except in very limited circumstances. Examples of such circumstances are when EPA has an agreement with a State or local government to provide water hookups, or when the site clearly has "insignificant" activity (e.g., a pump running).

7.2 Allowable Costs

In general, EPA will expend LUST Trust Fund monies at a Federal-lead UST corrective action only to stabilize the situation and mitigate those problems that are directly responsible for creating the major public health or environmental emergency. Because of limitations on funds, long-term remediation and post-corrective action site control (operation and maintenance) activities must be performed by the State. During an UST corrective action, the OSC is authorized to incur only those costs that qualify as appropriate uses of the LUST Trust Fund. These costs must be directly allocable to a particular response, reasonable, and necessary to accomplish the response.

This section summarizes both extramural and intramural costs that are allowable. To assist OSCs in tracking indirect costs, the Financial Management Division issues provisional EPA indirect cost rates. These rates should be used to estimate indirect costs incurred during the action.

In addition to the items specified below, an exposure assessment, as defined in Section 1.4 of this guidance, may also be performed at a Federal-lead UST corrective action. This assessment is a brief version of the detailed and complex risk assessment often performed to estimate exposure potential near RCRA and CERCLA hazardous substance sites. Although the exposure assessment is an allowable cost, it may rarely be appropriate due to time and financial restrictions imposed by the response criteria presented in Section 2.0 of this guidance. The ERD Regional Coordinator should be notified if an exposure assessment is being considered.

- 7.2.1 Extramural costs
 - <u>Cleanup contractor and consulting costs</u>, including waste transportation and disposal, now provided principally under the ERCS contractor system and supplemented as needed by non-ERCS commercial contractors.
 - <u>Support contractor costs</u>, including TAT.
 - <u>Subcontractor costs</u>.
 - <u>Other costs</u>, such as EPA leasing or rental of equipment; incremental costs for EPA-owned equipment; supplies, materials, and equipment (including transportation costs) procured for the specific corrective action and fully expended during the corrective action; and utilities.
 - <u>15% contingency allowance</u>, for unforeseen extramural costs.
- 7.2.2 Intramural costs
 - <u>EPA direct costs</u>, including the salaries, overtime, travel, and per diem of on-site EPA personnel.
 - <u>Direct costs</u> incurred by ERT.

- <u>Direct costs</u> incurred by Headquarters and Regional technical and legal staff.
- EPA Regional laboratory costs.
- <u>Indirect costs</u>, including EPA Regional and Headquarters management and administrative costs and fringe benefits.
- 7.3 Non-Allowable Costs

Corrective action costs not allowed under the LUST Trust Fund include (but are not limited to):

- State and local costs for which prior authorization was not specifically given by the OSC or addressed in a cooperative agreement (e.g., municipal services such as use of police or fire departments and State personnel who are on-scene performing tasks not specifically requested by the OSC).
- Costs for the research and development of equipment and response technologies used in conjunction with an UST corrective action (e.g., emerging alternative disposal technologies). Funding may be available, however, through other sources; contact the ERD Regional Coordinator for more information.
- Costs incurred by a contractor to provide response measures, for which that contractor is later found to be liable.

7.4 Cost Management

During Federal-lead UST corrective actions, all Regions must implement an effective system for managing response costs. This management system must ensure the efficient use of public monies, enable all response costs to be tracked against dollar ceilings, and provide the necessary information to support cost recovery actions.

Ultimate responsibility for cost management rests with the OSC. Detailed guidelines for the OSC are in the <u>Removal Cost</u> <u>Management Manual</u>. That manual identifies four components of cost management (i.e., cost projection, cost control, cost documentation, and cost recovery) which are applicable to Federal-lead UST corrective actions, and are summarized briefly below.

7.4.1 Cost projection

The key to effective cost management is through cost projection prior to the start of a response, as well as during a response. Pre-response estimates of costs form the basis for establishing a tetal project ceiling recorded in the action memorandum; cost projection during a response allows the OSC to anticipate the need for increases in the project ceiling. To estimate indirect costs, OSCs should use the provisional rates provided by the Headquarters Financial Management Division through the Region's Financial Management Officer.

7.4.2 Cost control

Cost control consists of cost planning and monitoring as well as verification of costs. OSCs are in the most advantageous position to control response costs if they stay informed on the availability of cost-effective resources. Thus, OSCs should: identify non-commercial support services and response equipment available to the Region; familiarize themselves with cost-effective cleanup services in the event contracting outside the ERCS network is required; maintain information on the cost of obtaining, operating, and maintaining safety equipment; and review final UST response reports of costs at past responses.

In addition, OSCs are responsible for monitoring site work and verifying that the contractor has provided the personnel, equipment, expendables, and subcontractors for which it has charged the government. OSCs should note the strict limitations under the EPA contract management policies for delegating these responsibilities to non-Federal personnel such as TAT, or to State officials not operating under a cooperative agreement (see Section 7.1).

7.4.3 Cost documentation

Cost documentation refers to the specific set of procedures that OSCs use to maintain a record of all on-site activities and associated costs. The method of cost documentation should be consistent from day to day at a specific response but may vary from site to site and Region to Region. The method an OSC selects must ensure thorough record-keeping on the following six information items:

- Chronology of events and decisions;
- Site conditions;
- Movement of personnel and equipment (e.g., site entry and exit);
- Contractor planned and authorized work compared to actual accomplishments;
- Contractor costs (e.g., commercial cleanup contracts); and

÷

• Other costs (e.g., TAT, ERT, Regional laboratory services, direct Headquarters and Regional intramural obligations, site access/acquisition).

The <u>Removal Cost Management Manual</u> provides applicable guidance on methods OSCs can use to determine each category of direct cost, and includes examples of the various types of documents OSCs can use to record information (e.g., POLREPs, entry/exit log, incident obligation log).

7.4.4 Cost recovery

Because of the possibility of a cost recovery action for any case involving the expenditure of LUST funds, OSCs have a responsibility to observe, document, and preserve critical evidence relating to the response and its costs. The cost documentation efforts described above are designed to ensure that facts concerning the release and owners or operators are recorded before response activity or the passage of time obscures or eliminates the evidence; that physical evidence essential for a trial is collected and preserved in a manner that will withstand judicial scrutiny; and that the government has maintained sufficient evidence of actual costs and substantiation of the need to incur those costs.

The essential elements of a cost recovery action are:

- Evidence of a release or threat of release of petroleum from an UST;
- •. Evidence that the defendant(s) is owner or operator of the UST;
- Evidence that the corrective action taken was necessary to protect human health and the environment; and
- Proof of incurred costs.

7.5 Ceiling Increases/Scope Modifications

OSCs should anticipate the need for ceiling increases or scope changes as early in the corrective action as possible so that the approval process does not interrupt the continuity of a project. The OSC should notify the appropriate ERD Regional Coordinator of such changes prior to submitting a formal request for approval; OSCs are encouraged to send a draft of the impending formal request to the ERD Regional Coordinator, to expedite concurrence and approval.

7.5.1 Ceiling increases

The initial action memorandum approving the corrective action establishes a project ceiling. This ceiling

represents the total funding approved for the corrective action. OSCs cannot incur costs in excess of the initial project ceiling unless a ceiling increase is approved.

The request for a ceiling increase should contain the information shown in Appendix E. The level of detail required will vary according to site-specific circumstances. In general, the request should include information on the current site conditions, actions taken to date, costs to date, and the reasons why the ceiling increase is required (e.g., changed site conditions or increased disposal costs). It is important to explain whether the increase is 1) to perform more work to mitigate the threat in the original scope of work, or 2) to respond to an additional threat to human health, welfare or the environment, not previously documented, requiring additional corrective measures. If the ceiling increase is needed to complete actions previously approved, the request should detail the reasons why additional funds are required for those actions. If the ceiling increase is needed to address additional threats, the request should contain a new finding that the threats pose a major public health or environmental emergency. Ceiling increase requests must be submitted to Headquarters under the signature of the RA and must include the information outlined in Appendix E.

The RA should send the request for a ceiling increase to the the AA, OSWER. In order to ensure an expeditious, smooth processing of the request, a final copy should also be sent to the ERD Regional Coordinator who is responsible for coordinating the Headquarters concurrence process.

7.5.2 Change in scope - no ceiling increase

UST corrective actions are approved by the AA, OSWER with a complete scope of work. In order to expand this scope (e.g., to address an additional threat), a formal change of scope request, specifying the additional corrective actions to be performed, must be approved by the official who approved the initial request.

The request should include information on current site conditions, actions taken to date, costs to date, and the additional proposed corrective actions as well as the additional threat.

7.6 Access Agreements

Gaining access to sites, where response actions require the use of adjoining property or property within the site boundaries owned by parties who are not owners or operators of the UST, may require obtaining access agreements or negotiating for rights-ofway with the property owners. Such agreements may be needed in order to establish a new road, to allow for the use of a private road, or to establish a command post.

7.6.1 Obtaining access

Primary responsibility for arranging for site access rests with the State. However, the OSC is ultimately responsible for obtaining site access agreements. The OSC may need to work with the State to ensure that arrangements are executed. Typically, the State will approach the property owner and the final access agreement will be drawn up either between the landowner and the State, or directly between the landowner and the OSC or the response contractor. Property access agreements must cover the duration of the response action and any associated postcorrective action control measures. The OSC is responsible for overseeing all site access negotiations and agreements, regardless of whether they are obtained through Federal or State channels. If gaining access voluntarily is a problem, the OSC should consult the Office of Regional Counsel. The OSC should also consult the Regional Counsel to obtain legal advice on gaining access to property for which the State has no authority, such as rights-of-way for public utilities, railroads, and Federal lands.

7.6.2 Payments for property

The Agency will not, as a rule, pay property owners for rights-of-way or easements for property adjacent to the site or within the site boundaries. Cases where payment becomes an issue in arranging for site access should be referred to the Regional Counsel or the ERD Regional Coordinator for assistance.

A written agreement signed by the OSC or EPA contractor with an owner/operator of an UST or a property owner who is not an owner/operator but whose property is contaminated or threatened cannot promise to compensate the owner for use of the property, to indemnify the owner for potential third party liability, or to pay for damages. Any written agreements offered by any property owner must be reviewed by Regional Counsel.

Generally, the OSC should attempt to restore the property, to the extent practicable, to its pre-response condition (e.g., regrading and reseeding a temporary site access road). The OSC may recommend fair payment to the owner as compensation, or assess the extent of any contamination and arrange for disposal, if necessary. In cases where the amount or type of compensation becomes an issue, the OSC should consult with the Regional Counsel or ERD Regional Coordinator.

7.7 Worker and Visitor Health and Site Safety

Response actions are subject to all applicable Federal, State, and local Occupational Safety and Health (OSH) laws. Standards of the Occupational Safety and Health Administration (OSHA) form the basis for the safety and health protection of workers involved in Federal UST corrective actions. Where State OSH laws exist, these laws may also apply to response actions. The safety and health requirements of other Federal agencies may also apply.

For LUST Trust Fund-financed corrective actions, all Federal, State, and contractor personnel involved are required to comply with the lead agency's overall occupational safety and health policies and with a site-specific safety plan. All visitors to the site are also subject to the same health and safety requirements.

Because response activities associated with each specific incident are unique, standard procedures will often have to be adapted or modified to meet the incident-specific requirements. For this reason, a written safety plan must be prepared for each incident, distributed, and posted in the command post. This should be done before response operations begin on the site, or as soon as possible thereafter. The plan must cover all phases of incident operations and identify key personnel and must be updated or modified as needed or as conditions change. As a minimum requirement, the safety plan should address the following:

- Establishment of the number of personnel permitted to enter the contamination zones.
- Establishment of entry and escape routes.
- Establishment of procedures to identify, locate, and alert off-site medical personnel.
- Determination of physical, chemical, and biological properties of known contaminants.
- Establishment of decontamination protocol.
- Establishment of levels of protection.

:

- Establishment of air-monitoring protocol.
- Establishment of general safety rules and equipment.

\$

- Scheduling of daily safety meetings.
- Posting of key agency and emergency contacts.

OSCs are responsible for ensuring that workers and visitors are informed of on-site hazards and the provisions of the site health and safety plan. The OSC shall ensure that all individuals entering the site (e.g., EPA, TAT, contractors, press) have read the plan and understand its contents.

Throughout the response action, the OSC may call upon OSHA to advise on worker health and safety issues. When needed, the OSC may request that the National Institute of Occupational Safety and Health (NIOSH) provide assistance to OSHA in testing worker protection equipment and gathering information for guidance on safety issues. The OSC must be cognizant of on-site health and safety activities and is responsible for monitoring Federal and contractor compliance with EPA and site safety requirements and applicable Federal and State laws and regulations. However, OSCs are directly responsible only for their own staff; each government agency and private employer is responsible for the health and safety of its own employees and for ensuring compliance with OSHA requirements, applicable State OSH laws, and Agency health and safety programs. The Agency will not assume responsibility for other government or contractor personnel.

OSHA has authority for enforcing compliance with Federal OSH regulations. Response actions are subject to OSHA inspections. Where State OSH laws exist, State inspections may take place. If the OSC discovers an infraction of safety requirements, the OSC must remind all site personnel and visitors of the requirements. Should the infraction continue, the OSC may call in OSHA or State OSH inspectors to review practices to ensure compliance.

7.8 Community Relations and Public Information

Community relations is a communication network between response officials and the community. The objectives of community relations are: to identify community concerns about the site; to encourage citizens to express concerns and provide information; to provide information to the community on health and environmental effects of the release and proposed response action; and to incorporate citizen comments and concerns (including those of the owners or operators) into the decisionmaking process.

Community relations activities will be important at Federallead UST corrective action sites because a major public health or environmental emergency exists. However, the time and extent of Federal involvement may not warrant complex community relations plans. The EPA's <u>Community Relations Policy</u> (May 1983) and <u>Community Relations in Superfund: A Handbook</u> (March 1986) can be used as applicable guidance for developing and implementing community relations activities. Although no specific activities ' currently required at Federal-lead UST corrective actions, a community relations profile should be prepared for any action that lasts longer than five days. The profile should provide notice to owners/operators that the record is available for review. The profile explains how program and community relations staff intend to plan for and implement community relations activities at the site. It should contain a brief outline of the nature of community concerns, the key site issues, the objectives of community relations activities, and the communications activities considered for the site..

Some petroleum releases may require short-term corrective action lasting no longer than a few days. These kinds of actions may not involve substantial community relations planning because of the nature of the emergency and the quick response time required. At these sites, the focus of community relations is to provide information about the site and its risks to the community; information can be channelled through the media and local officials.

7.9 Record-Keeping Requirements

OSCs must ensure that they document and record all decisions and determinations they make prior to and during responses. OSCs must also include in the files any significant comments received from the owners/operators and their response to these comments. Structured site and Regional files are the sole repositories for site records. Care must be taken to ensure their completeness and long-term security. Both site management and financial management records are critical when cost recovery is involved. Complete and precise OSC records of oral and written communication with owners and operators, contractors, and participating Federal, State, and local agencies must be maintained should litigation arise at some later time.

7.9.1 Pre-response records

Prior to the initiation of an UST response (when time allows), the OSC must maintain documentation regarding decisions and determinations relating to issues such as:

- The appropriateness of a Federal-lead UST corrective action;
- Contact with, and comments received from, the owner or operator;
- Contact with, and comments received from, the State;
- Planning response actions;
- Developing contractual arrangements; and
- Complying with relevant environmental statutes.

7.9.2 UST response records

During the course of a LUST Trust Fund-financed corrective action, the OSC is responsible for generating and maintaining site-specific documents such as action memoranda and daily and periodic cost control reports. It is critical that the OSC maintain a log of on-site activities and record all communications with the contractor, the owners or operators, and participating Federal, State, and local agencies. The <u>Removal Cost Management Manual</u> outlines the OSC's cost control record-keeping responsibilities. The <u>ERCS Contract Users' Manual</u> and the <u>TAT Contract User's</u> <u>Manual</u> prescribe guidance for contractor-related recordkeeping. Any cost control record-keeping requirements in conjunction with Federal, State, and local agencies may be prescribed as necessary.

7.10 Reporting Requirements

The OSC is responsible for documenting and reporting all response activities taken at a site. Reporting requirements include preparing and submitting to ERD a series of POLREPs and a final UST response report. POLREPs consist of initial, progress, and final reports. This section provides guidance on when the various reports should be submitted and the types of information each report should contain. All site information developed by the OSC must be made available to the Regional UST Coordinator, who will coordinate with the State to ensure that an effective, final resolution to a release will be accomplished by the State.

7.10.1. Pollution reports (POLREPs)

POLREPs provide factual operational data surrounding the incident and a current accounting of the total funds allocated in an incident. POLREPs should also detail measures to ensure that the affected community is properly and fully informed of all response activities. The Regions should bear in mind that POLREPs are a method of alerting Headquarters that critical events may be pending and that requests/action are about to be initiated. However, all requests for Headquarters decisions must be formally submitted in accordance with Section 4.0 of this guidance. To properly assist Headquarters management, routine POLREPs are sent to ERD at (202) 755-2155 (Telefax), 710-822-9269 (TWX), or EPA 5511 (E-Mail).

7.10.1.1 Initial POLREP

The OSC should prepare and send to Headquarters and the Regional UST Coordinator an initial POLREP for each new Federal-lead UST corrective action. This report should give the start date, describe the incident (including the outcome of any site evaluation), give the status of actions (including enforcement), and describe future plans. 7.10.1.2 Progress reports

.

Routine progress reports should be submitted to ERD and the Regional UST Coordinator a minimum of once a week and daily, where practicable, when events are rapidly occurring. Progress POLREPs should identify the following:

- a) Situation -- present status of ongoing response activities;
- b) Actions Taken -- activities undertaken since the last POLREP;
- c) Future Plans -- planned actions by the OSC;
- Project Costs -- estimated funds obligated thus far, including a breakdown of the cost categories as noted in the following example:

Extramural	<u>Current Ceiling</u>	<u>Obligations To Date</u>	
Cleanup contractors TAT	\$20,000 5,000	\$10,000 2,000	
15% Contingency	3,750	N/A	
Intramural (both HQ and Regio	on)		
Direct Indirect (estimate based on , provisional rates	3,000 n 6,000 s)	1,500 3,000	

- TOTAL PROJECT CEILING \$37,750 \$16,500
 - e) Any other pertinent information such as status of efforts to obtain cleanup by responsible parties.

Also, POLREPS should be provided to ERD and the Regional UST Coordinator on all major unanticipated developments of interest at approved corrective actions (e.g., fires, explosions, and all accidents even if no damage or injury has been caused) not included in other progress reports. In addition to reporting accidents to ERD via POLREPs, a corrective action accident report (see Appendix F) should also be completed and submitted to ERD. This form was developed to provide more detailed documentation of circumstances surrounding accidents during the course of response actions. The information it contains is vital should litigation occur.

\$

7.10.1.3 Final POLREP

When a Federal-lead UST corrective action has been completed, a final POLREP (e.g., POLREP #15 and FINAL) should be submitted that describes the final actions taken at the release, results achieved, detailed final costs and date of completion, the demobilization date, and future actions planned and who will perform them.

7.10.2 Final UST reports

Within 60 days after the conclusion of a Federal-lead UST corrective action, the OSC should prepare and submit to ERD a final UST report. It is necessary that ERD have these final UST reports on hand to respond to inquiries from the public, Congress, the Office of the Inspector General, and the General Accounting Office.

The final UST response report is a complete report on the response operation and the action taken. It should include a summary of events, an analysis of the effectiveness of corrective actions, a list of problems affecting the response, if applicable, and OSC recommendations. The completion date is the date on which any wastes are shipped for ultimate disposal or the site is demobilized, whichever is later. Temporary demobilization and temporary on-site storage are not considered completions unless they are the final actions approved (i.e., off-site disposal is not approved).

7.11 Post-Corrective Action Site Control

The State or local government must assume responsibility for operations and maintenance performance and costs after the system is proven to be operational. Examples of operations and maintenance at a corrective action include running pumps or operating a ventilation system.

If the State recommends response options involving operations and maintenance in the oral request (see Appendix B) or State request letter (see Appendix C), the State must explain how it intends to assume resource and financial responsibility for these options. Some States may have cooperative agreements that provide for operations and maintenance; the Regional UST Coordinator should be contacted for this information, as detailed in Section 4.4.

Some situations may require operations and maintenance as part of all response options. If no owner or operator agrees to assume responsibility, the State must take over these actions under an UST cooperative agreement.

SOLID WASTE DISPOSAL ACT (SWDA)

8-33. <u>Selection and Performance of Federal Corrective Actions</u> <u>at Leaking Underground Storage Tanks</u>

1. <u>AUTHORITY</u>. Pursuant to Subtitle I of the Solid Waste Disposal Act (SWDA), to determine the necessity of and to select Federal corrective actions at leaking underground storage tanks containing petroleum, to enter property and to perform such corrective actions.

2. <u>TO WHOM DELEGATED</u>. Assistant Administrator for Solid Waste and Emergency Response and Regional Administrators.

3. LIMITATIONS.

a. Regional Administrators or their delegatees may exercise this authority only for requests for obligations not exceeding \$50,000, and must comply with Federal-lead UST corrective action guidance issued by the Assistant Administrator for Solid Waste and Emergency Response (AA/OSWER).

b. If this authority is redelegated, the Director, Office of Emergency and Remedial Response, must obtain the advance concurrence of the Director, Office of Underground Storage Tanks.

4. REDELEGATION AUTHORITY.

a. This authority may be redelegated by the Assistant Administrator for Solid Waste and Emergency Response to the Director, Office of Emergency and Remedial Response.

b. Regional Administrators may redelegate this authority to Division Directors, who may then redelegate to On-Scene Coordinators.

5. ADDITIONAL REFERENCES.

- a. Superfund Removal directives.
- b. Section 9005(a) of SWDA.
- c. Section 9003(h) of SWDA.
- d. UST corrective action procedures.

- 1. State official requesting assistance (should be authorized by Director of State UST agency designated by Governor).
- 2. Is the release from a leaking underground storage tank? (see Section 1.4 of this guidance for definition)
- 3. Is the released substance petroleum? (see Section 1.4 of this guidance for definition)
- 4. Location of release.
- 5. Time and date of release.
- 6. Name, description, and general condition of facility.
- 7. Is the release a long-term or chronic problem?
- 8. Types and amounts of petroleum substances involved, and quantities released to environment.
- 9. Major public health or environmental emergency conditions. (see Sections 1.4 and 2.2 of this guidance for detail)
- 10. Number and proximity of persons potentially affected.
- 11. Increased threat to human health or the environment if response is delayed or denied.
- 12. Ongoing efforts to respond to release.
- 13. State/local/owner-operator ability and willingness to provide response, with specific reasons for inability to respond (e.g., lack of authority, technical expertise, qualified staff, or funding).
- 14. Specific enforcement actions undertaken by the State.
- 15. Type of action needed to mitigate or stabilize emergency (if known).

APPENDIX C: State Request Letter Format

Federal-lead UST corrective action will not be initiated unless specifically requested by the appropriate State via the State UST Coordinator. A letter requesting Federal-lead corrective action should be signed by the Director of the State UST agency designated by the Governor and addressed to the Regional OHM Coordinator, or other designated Regional management official.

Below is a recommended format for the State to follow in preparing an UST corrective action request letter. While use of this model format is not mandatory, the letter should address all of the topics presented in the model in order to demonstrate that the incident meets the Federal-lead UST response criteria (particularly that it poses a major public health or environmental emergency and that neither the State, nor the owner or operator can provide adequate response) and that all actions in the scope of work are consistent with SWDA. Because this letter will provide much of the source material to be used by the Region in preparing an action memorandum (see Appendix D), adherence to this format is strongly encouraged.

I. HEADING

DATE:	[Month/Day/Year]	
SUBJECT:	Request for Federal-lead UST Corrective Actic [Site, City, State] STATE REQUEST LETTER	
FROM: ·	[Director of Designated State UST Agency]	
TO:	[Regional OHM Coordinator]	

II. BACKGROUND

The background section should contain information on the location of the site, the nature of the incident (including the history of the site, general character of the site, and issues relevant to petroleum management), quantity and types of petroleum substances present, State and local authorities' role, the cleanup time frame, and actions to date, including previous and current actions to abate the threat.

A. Location Description

1. Describe the site's physical location.

Give distances from nearest populations and points of reference, as appropriate. Also state the population size. For example, "A school is within 1/4 of a mile and there are 1,000 residences within a mile of the site; the area is mainly suburban residential with some industrial areas." Describe the areas adjacent to the incidents or site in terms of nearby vulnerable or sensitive populations, habitats, and natural resources. For example, "The site is adjacent to wetlands and a tributary to the river flows nearby. The area's primary drinking water aquifer underlies the site."

Describe any areas protected by statute, such as parks, historic sites, and sensitive ecosystems. This may include areas such as the New Jersey Pinelands, wetlands areas, or wild and scenic rivers.

2. Provide supporting documentation.

Providing pictures, diagrams, maps, and/or sketches is encouraged. They may be included as attachments or incorporated into the text. This documentation may help to substantiate the threat at the site.

B. <u>Site and Incident Characteristics</u>

1. Discuss the history of the incident or situation that poses the major public health or environmental emergency.

Describe the type of incident that has occurred (e.g., a classic release) and why it occurred. For example, "A corroded storage tank failed during refilling." Be sure to list all of the site's key problem areas (e.g., tanks, associated pipe lines, free product on ground water, or contaminated soils).

5

Describe the exact location of the incident at the site. 'For example, "The release occurred at a corroded tank in the south corner of the site." Include the time and date (if known) of the incident. State whether the release is new, just recently discovered, or a chronic problem that has deteriorated. Also describe when and how the incident was discovered; for example, "The town fire marshal received complaints from residents of strong gasoline fumes in their basements."

2. Discuss the general character of the site; show that it meets the definition of "underground storage tank" (see Section 1.4 of this guidance).

Describe the current use of the site (e.g., active facility, vacant lot, recreational area). Describe the previous nature and type of the facility, as well as the activities historically undertaken at the site; for example, a tank farm used for storing aviation fuels, or an oil refinery with numerous underground tanks and pipelines. State the site characteristics that qualify it as an "underground storage tank" site.

Include any pertiment information on the site owners or operators, past and present. This information should reflect whether the current or previous owners or operators contributed to the conditions on the site.

3. Present all findings on the extent of the release to date.

First, note all pathways of release (e.g., surface water, ground water, air emissions/vapors, soil). Then, indicate whether the release is confined to the site or has migrated off the site. Where possible, present the results of any on-site or off-site monitoring. With respect to drinking water contamination, note the number of municipal or private wells contaminated or threatened.

4. Discuss the relevant issues relating to current petroleum management practices.

Describe any existing structures, measures, or conditions that would either mitigate or accelerate the release of any materials on site (e.g., an unstable dike, a temporary containment system, adverse weather conditions, site security, fencing, condition of containers and similar situations). State whether the release is widespread and/or is migrating rapidly.

Indicate the extent to which the petroleum wastes can be treated or are amenable to alternatives to land disposal. State whether the site is to be stabilized or cleaned up, if known.

C. <u>Ouantity and Types of Petroleum Substances Present</u>

•

1. List all petroleum substances known to be on site at the time of the approval request.

Describe briefly the results of the sampling (e.g., "most affected residences exceed the lower explosive limit for gasoline vapors in their basements") and give estimates of quantities of the classes of petroleum substances if they are available (for example, "four inches of diesel fuel was found floating on the underlying ground water").

2. Describe the sampling methodology.

Briefly describe the sampling methodology as well as methods for maintaining consistency, reliability, and quality control. Mention who performed the data collection and the lab analysis (e.g., EPA, contractor, local health authorities).
D. <u>State Role</u>

· Discuss the State request for Federal-lead UST response.

List and describe the reasons why the State cannot provide adequate response to mitigate the major public health or environmental emergency posed by this UST petroleum release. Be detailed; indicate whether the State cannot respond due to a lack of authority, expertise, or resources. Estimate the resources required for response and specify which resources the State lacks, and why. Discuss why a site-specific cooperative agreement under the LUST Trust Fund would be an inappropriate response mechanism for this site or incident.

E. <u>Actions to Date</u>

1. Discuss any previous actions to abate the threat.

Describe any Federal, State, local, owner or operator, or privately sponsored activities that have been performed. Give the dates, costs, and effectiveness of such actions. Actions to abate the threat may include issuance of an advisory from the locality advising residents not to drink their water, or the provision of bottled water.

2. Discuss any current actions to abate the threat.

Any Federal, State, local, owner or operator, or privately-sponsored activities that are currently being conducted should be described. This information should include estimated costs and completion dates of these activities.

III. MAJOR PUBLIC HEALTH OR ENVIRONMENTAL EMERGENCY

Strict criteria must be met to justify Federal response to a petroleum release from an underground storage tank. These criteria require that neither the State nor the owner or operator be able to provide adequate response. In addition, the release must pose a major public health or environmental emergency, as indicated by:

- An immediate and substantial threat of direct human, animal, or food chain exposure to petroleum; or
- o An immediate threat of fire and/or explosion; or
- An immediate and substantial threat to public drinking water supplies; or
- An immediate threat to human health or substantial amounts of property, or an immediate and substantial threat to natural resources.

The State should ensure that their request letter discusses how the release meets these criteria. Sections A and B below provide some specific examples of the types of material to include that demonstrate a major public health or environmental emergency.

A. Threats to Public Health and Welfare

Describe the threats to public health and welfare.

List all of the threats, starting with the most serious, that adversely affect human health or welfare (e.g., drinking water contamination, fire/explosion). Identify any human exposure that already has occurred, the kind of exposure (e.g., inhalation, ingestion, dermal contact) and the exposure pathway (e.g., water supply, indoor air pollution). Describe any reports of illness, injury, or death that appear to be linked to the exposure. Be as specific as possible about the number of individuals exposed, the proximity of sensitive populations (e.g., hospitals, schools), the geographic area affected, and whether exposure was acute or chronic. Also, describe any anticipated exposure and whether it is imminent, especially with regard to public drinking water supplies.

Compare the amounts or concentrations of substances shown to background levels or health standards as appropriate. If a health advisory is given, or if an exposure assessment has been performed, include it as an attachment to the letter, and reference it in the discussion.

B. Threats to the Environment

Describe threats to the environment.

List all the threats, starting with the most serious, that adversely affect the environment (e.g., damage to sensitive ecosystem, animals, ground water). Identify any natural resource or environmental damage that already has occurred and the extent of exposure (e.g., acute or chronic). Indicate whether there have been reports of deaths of flora or fauna (e.g., fish kills). If so, state how much environmental damage has occurred (e.g., 20,000 acres of wetland contaminated, one million fish killed).

Discuss all actual and potential impacts on the affected area. Describe any anticipated exposure and whether it is imminent. Indicate whether the release threatens endangered species, critical wetlands, or other resources protected under law.

IV. ENFORCEMENT

The purpose of this section is to meet the statutory requirement that priority be given to cases where the owner or operator is unable or unwilling to undertake necessary action. This information should be referenced here as "see attachment" and placed on a separate page entitled "Enforcement Sensitive." This section includes information on the enforcement strategy (summarized), the status of notice letters and/or negotiations, the available enforcement authority, owners or operators, previous enforcement actions, the probability of recovering costs, and the recommended enforcement strategy. This section also should contain information on the potential for a non-Federal response. This information is required before Federal corrective action is started, unless the release meets the conditions in Section 4.1.1.

A. Enforcement Strategy

1. Briefly summarize the enforcement strategy.

Summarize the enforcement strategy for notifying, negotiating with, and litigating against owners or operators. Indicate whether the State or Federal enforcement attorneys are actively pursuing informal negotiations, are actively pursuing litigation, or have decided to postpone or not pursue litigation. For example, "no enforcement actions currently underway," or "the owner or operator has agreed to conduct a cleanup." Describe what actions are planned (e.g., negotiations or administrative orders).

2. Briefly summarize the enforcement actions.

Indicate if litigation is proceeding or is contemplated. Cite under what statutory authority the action will be or is being taken (e.g., SWDA Section 9003(h)).

B. <u>Status of Enforcement Actions</u>

1. Owners or operators

Note the names of owners or operators. Indicate whether the owner or operator has taken action; if so, describe the action and explain why it was inadequate.

Describe any efforts being undertaken to obtain additional owner/operator response. Describe the attempts that have been made to locate owners or operators (e.g., oral inquiries both on and off site). Include whether the owner or operator has been notified (e.g., contacted in person or by telephone, written follow-up). Give the dates that notice letters were sent and a summary of the responses of the recipients (e.g., the owner agreed to clean up the site, or denied involvement at the site). If negotiations are underway, describe the activities under discussion.

2. Provide information on the status of notice letters and/or negotiations.

Describe any previous State or Federal enforcement actions taken to date. These actions may include notice letters or demand letters.

3. State the probability of recovering costs.

Estimate the solvency of the owner or operator. Evaluate the ability to obtain the necessary actions in a timely fashion through litigation. This should be included if it explains why actions are being requested of EPA when owners and operators are financially able to undertake these actions.

V. EXPECTED CHANGE IN THE SITUATION SHOULD NO ACTION BE TAKEN OR SHOULD ACTION BE DELAYED

Describe any expected changes in the situation should no action be taken or should action be delayed. Include a description of the worst case that could possibly occur should no action be taken. These changes may include:

- o Spread of contamination. For example, the ground water contaminant.plume may spread through a larger area.
- o Change in nature of contamination. For example, gasoline vapors may seep into structures, producing the added threat of fire/explosion.
- Increased threat to human health and the environment if action is delayed or denied. For example, the contaminant plume may soon reach drinking water wells.
- Impact on future response actions if action is delayed or denied. For example, the tanks will deteriorate further, leaking additional petroleum into the ground.
- VI. IMPORTANT POLICY ISSUES (only as necessary and applicable)

Include a separate section on important policy issues that are significant to this request. These issues may include:

 Cost sharing (State cost sharing is applicable after regulations promulgated under SWDA Section 9003(C) become effective).

- The division of responsibilities among Federal and/or State agencies.
- o Owners and operators.

. .

- o Off-site disposal availability.
- o Special coordination and similar issues.
- o Concurrent CERCLA action or the presence of hazardous materials on site.
- Contiguous sites (if multiple locations are recommended for consideration as one site, give justification for such consideration).

Issues should be fully explained, including a discussion on the efforts being made to resolve the issue and/or decisions that must be made before a resolution is reached.

VII. STATE REQUEST FOR EPA ACTION

The State should specify the corrective actions it is requesting that EPA perform. These might include

- o Recovery of free material;
- o Temporary water supply; or
- o Temporary relocation of residents.

The Region should prepare an action memorandum based on the State letter requesting Federal-lead UST corrective action (see Appendix C). The action memorandum should address <u>all of the</u> <u>topics outlined in Appendix C</u>, along with the additional items presented below. If the State request letter adequately follows the format suggested in Appendix C, then the Region's action memorandum may simply consist of the following sections with a copy of the State letter referenced and attached. If the State letter is deficient, the Region must ensure that the action memorandum addresses and corrects all of the deficiencies, including all of the information requested in Appendix C and in the additional sections described below. In any case, a copy of the State request letter should always be attached to the action memorandum for reference.

- I. HEADING
 - DATE: [Month/Day/Year]
 - SUBJECT: Request for Federal-Lead UST Corrective Action [Site, City, State] ACTION MEMORANDUM
 - FROM: [Regional Administrator]
 - TO: [see Appendix A]
- II. PROPOSED ACTIONS AND COSTS

OSCs should consider cost as one of many factors when proposing UST corrective actions. However, there are no statutory or regulatory requirements for cost-effectiveness. The purpose of this section is to provide guidelines for presenting information on proposed and alternative actions, estimated costs, and the project schedule.

- A. <u>Proposed Actions</u>
 - 1. Describe the proposed actions.

Describe the specific tasks involved and the results sought by the corrective actions as they pertain to the threat(s) discussed in Appendix C. For example, "The primary objective of the proposed action is the mitigation of the threat to public health by provision of alternate potable water supplies to the affected homes."

Indicate whether any further information is needed before all response actions can be identified (e.g., sampling to address ground water or drinking water contamination).

State why the proposed actions are appropriate for this situation and explain the rationale for choosing the proposed option. Describe the technical feasibility and probable effectiveness of the proposed action. Address response time requirements, intermedia relationships, temporary versus long-term solutions, institutional considerations, and environmental impacts.

Describe the procedures to be undertaken in the proposed actions. For example, "Corrective action will include ventilation of affected structures and provision of bottled water to impacted residents." Describe any impediments to the proposed action (e.g., lack of public acceptance or problems with access).

2. Describe the project schedule.

Give the time frame needed to stabilize or clean up the site and include how quickly response can begin. Give the estimated period of performance. Indicate whether postcorrective action site control (operation and maintenance) will be required, and how the State intends to perform it.

3. State the estimated costs.

Estimate the total project ceiling with an itemized breakout of the following cost categories that comprise that ceiling: extramural costs, which include cleanup contractor costs, TAT costs, and a 15% contingency allowance; and EPA intramural costs, both direct and indirect. For example:

Extramural

••

. •

<u>Estimated Costs</u>

Cleanup contractors TAT	\$20,000 5,000
15% Contingency	3,750
termural (both UC and Dogion)	

<u>Intramural</u> (both HQ and Region)

Direct						3,000
Indirect	(estimate	based	on	provisional	rates)	6,000

TOTAL PROJECT CEILING

\$37,750

If LUST Trust Fund monies have been obligated for past actions, then indicate the obligations (in both Procurement Requests and Delivery Orders).

B. <u>Alternative Actions</u>

• Briefly describe the alternative actions, explaining the decision rationale used to select the proposed corrective action plan.

Describe what other actions were considered (e.g., providing hookups to city water instead of providing bottled water). Briefly describe the technical feasibility and probable effectiveness of each option. Address response time requirements, intermedia relationships, temporary vs. long-term solutions, institutional considerations, environmental impacts, and estimated costs of each option. State any impediments to the options (e.g., public acceptance or access to the site). State why the selected alternative was chosen.

III. RECOMMENDATION

The purpose of this section is to briefly present the Region's recommendations, rationale, and projected costs for the action. For example, "Because conditions at [site name] meet the criteria for Federal-lead UST response, I recommend your approval of the proposed UST corrective action. The estimated total project costs are \$37,750 of which \$20,000 are for extramural cleanup contractor costs. This site meets the criteria for a major public health (and/or environmental) emergency, and no State, local, or other agency, owner or operator, or other party can or will provide adequate and timely response."

Approve:	[Name and Title]	Date:
Disapprove:	[Name and Title]	Date:
Concur:	[Name and Title]	Date:

Attachment[s]

The following action memorandum format is recommended for requesting approval for ceiling increases. The purpose of this format is to provide the OSC with a blueprint that can be easily followed when substantiating the need for exceeding the current ceiling on Federal-lead UST corrective action costs.

- I. HEADING
 - DATE: [Month/Day/Year]
 - SUBJECT: Request for a Ceiling Increase [Site, City, State] ACTION MEMORANDUM
 - FROM: [Regional Administrator]
 - TO: [see Appendix A]

II. ISSUE

Briefly explain why the ceiling increase is being requested. Indicate what the new project ceiling will be if the ceiling increase is approved. For example, "A ceiling increase of \$16,000 for a new total of \$54,550 is being requested to continue a Federal-lead UST corrective action at this site."

III. BACKGROUND

The primary purpose of this section is to identify the key characteristics of the release in order to lay the foundation for demonstrating that a major public health or environmental emergency exists. These characteristics include site location and pre-release use, the type of release, and its scope. In compiling this information, the OSC should consider the availability of pictures, diagrams, maps, and/or sketches that may assist in describing the site.

A. <u>Incident/Response History</u>

Discuss the background/history of the site, including the following: current conditions at the site; who initiated actions and when; the date Federal-lead UST corrective actions were approved and by whom; approved actions; and actions taken to date (e.g., ventilation of structures, provision of bottled water).

B. <u>Site Conditions</u>

Briefly describe the site conditions and the reasons for a ceiling increase request. State whether the additional funds are needed due to increased costs for previously approved actions, or to expand the scope of work beyond that approved in the original action memorandum. If the ceiling increase is needed to complete previously approved actions, explain the need for additional funding (e.g., disposal costs higher than anticipated). If the ceiling increase is needed to expand the scope of work, discuss how the site conditions meet the criterion of presenting a major public health or environmental emergency; also explain why neither the owner nor operator is able to conduct the corrective action. Include a description of a worst-case scenario should the ceiling increase not be granted.

Discuss the present status of the UST corrective action (e.g., gasoline release contained and soil awaiting excavation).

Include any information that may help substantiate the need for a ceiling increase. Attach any enforcement status information, ATSDR health advice memoranda, and other pertinent information such as pictures, drawings, and other materials to the back of this action memorandum.

IV. PROPOSED ACTIONS

Describe the proposed actions to be undertaken if a ceiling increase is approved. Briefly describe the actions that are required to complete the response; for example, "Bottled water will be provided until water main hookups are completed." Indicate the objective of the proposed actions or the threats these actions are to abate, minimize, or limit.

V. SUMMARY OF COSTS

Provide a summary of costs, including a breakdown of costs for both the current ceiling and the proposed ceiling. Detail the estimated total project ceiling with an itemized breakdown of the following cost categories that comprise the ceiling: extramural costs, including cleanup contractor costs, TAT costs, and a 15% contingency allowance; and EPA intramural costs, both direct and indirect. For example, the total project ceiling should be broken down as follows:

<u>Extramural</u>	<u>Current Ceiling</u>	Proposed Ceiling
Cleanup contractors TAT 15% Contingency	\$20,000 5,000 3,750	\$30,000 7,000 5,550
Intramural (both HQ and Regio)	
Direct Indirect (estimate based or provisional rates	3,000 6,000 \$)	4,000 8,000
TOTAL PROJECT CEILING	\$37,750	\$54,550

VI. RECOMMENDATION

The purpose of this section is to briefly present the Region's recommendations, rationale, and the project costs. For example, "To eliminate the major health emergency posed to the nearby public and the environment consistent with the Federallead UST response criteria, I recommend you approve this \$16,800 ceiling increase request." Briefly summarize what the additional funds will be used for. Briefly state how the approval would increase the current total project ceiling. For example, "Your approval would raise the extramural cleanup contractor ceiling to \$30,000 and bring the total project ceiling from \$37,750 to \$54,550. You may indicate your approval or disapproval by signing below."

Approve:	[Name and Title]	Date:	
Disapprove:	[Name and Title]	Date:	
Concur:	[Name and Title]	Date:	

Attachment[s]

APPENDIX F: Federal-Lead UST Corrective Action Accident Report Format

- 1. Date and Time of Incident:
- 2. Site Name and Location:
- 3. OSC:
- 4. Description of Incident:

.•

- 5. Factors Leading Up to Incident:
- 6. Site Work Related to Incident (OSC Orders, TAT Oversight, Foreman's Orders):
- 7. Weather Conditions During Incident (Temperature, Humidity, Wind Direction and Speed, Precipitation):
- 8a. Injuries (Person, Role of Person On Site, Description of Injury):
- 8b. Exposure (Person Exposed, Substances Involved, Type of Exposure - Inhalation, Ingestion, Dermal):
- 8c. Medical Treatment (Paramedic, Physician, Hospital, Length of Stay, Estimated Cost):
- 9. Property Damage (Owner, Location, Description of Damages, Estimated Cost):
- 10a. Other Persons On Site: 10b. Other Person's Roles/ Activities On Site on Day of Incident:

\$

F-1

Date of Report:

Prepared By:

Preparer's Phone #:

APPENDIX G: Delivery Order Preparation and Processing

To activate the ERCS contractor and initiate cleanup at a Federal-lead UST corrective action, a Delivery Order (DO) must be prepared and issued. All DOS will be issued by Ordering Officers (e.g., Regional Deputy Project Officers (DPOS), Headquarters Contracting Officers (COS), or EPA On-Scene Coordinators (OSCS); new warrants are being issued enabling OSCs to order services at UST cleanups) for individual corrective actions. These DOS will be issued on a fixed rate, indefinite quantity basis, with time and material provisions.

This appendix on DO preparation and processing describes procedures which address:

- o Oral Delivery Orders;
- o Delivery Order completion and processing; and
- o Delivery Order modifications.

The procedures described in this appendix are applicable to both the ERCS zone contracts and the separate ERCS Regional contracts.

A. <u>Oral Delivery Orders</u>

As indicated, DOs can be issued orally to the ERCS contractors. This flexibility is designed to enhance response capabilities under the ERCS contract network. However, any oral order must be confirmed by a written DO within 48 hours.

When the ERCS contractor is contacted by telephone for purposes of orally issuing a DO, the Ordering Officer should simultaneously complete a Delivery Order form (see Exhibit G-1) noting:

- Date and time of the order (all references to time specified in the DO should reflect the local time of the location where services are to be provided);
- Contractor representative contacted (should be a person authorized to commit the contractor) and telephone number;
- o Response location;
- o Date and time the contractor is required on site;
- o Date and time the contractor agrees to be on site; and
- Brief narrative of the services (e.g., personnel, equipment, and materials) and level of protection (e.g., health and safety) required.

Some of these items can be completed before the call is made to the contractor. For example, the response location or the level. of safety protection required may be known ahead of time and could be completed in advance.

By completing the Delivery Order form during (or for some items directly before) the discussion with the contractor, the Ordering Officer will:

- Provide the contractor clear direction on the services needed;
- Document personnel, equipment, and materials expected on site;
- Establish the OSC-Response Manager site management relationship, and
- Facilitate completion of the written DO issued to the contractor at the site.

The information recorded during issuance of the oral Delivery Order will be important should there be any dispute concerning the contractor's ability to provide the services within the required response times.

B. <u>Delivery Order Completion and Processing Instructions</u>

To formally initiate services performed by the ERCS contractor to conduct UST corrective actions, Ordering Officers must prepare written Delivery Orders. The DO specifies the services to be performed by the ERCS contractor in executing a specific corrective action. Each DO establishes a ceiling amount that constitutes the maximum amount for which the government shall be liable. Instructions for the completion and processing of DOs are outlined below.

1. Delivery Order completion.

:

Delivery Order preparation is the responsibility of the Federal Ordering Officer. A standard format is used for the DO; all items in the DO should be completed as explained below and illustrated by the sample in Exhibit G-1.

- o <u>Date of Order (Box 1)</u>: Enter the date of issuance of the DO to the contractor.
- <u>Contract Number (Box 2)</u>: Enter the contract number (e.g., 68-01-___) of the ERCS contract under which services are being ordered. The final four digits of this number vary; call the CO to obtain the current contract number.

I. CALLUP OF	ROER	1 2 CONTRACT NO.		J. ORDER NO.	
01/	06/87	68-01-		03	
A TIME OF IN	TUAL ORDER	I & DELIVERY ORDER C	ELING AMOUNT		
				. 20,000	
	8 00			(Chilgered An	
		C. ACCOUNTING AND	Document Control 4	Account # -	Class
	Time Zone	68 -20X8153			
	CONTRACTOR (Name, Antres	and Zie Coder	A ISSUED BY: ORDER	ING OFFICER (Name, A	Heress and Zip Ca
UST 2200 Roci	Cleanups, Inc. O Mudtrack Lane hester, NY 14609		Vincent Bra Environment 841 Chestnu Philadelphi	dford al Protection at Street a, PA 19107	Agency
b. PROGRAM	MANAGER (Name and Phone No.	.,	B. EPA REGIONUSCO		e. ZONE
Jose	ph Davidson (716)	482-2811	03		1
AESPONSE	MANAGER (Name and Prone No.	.,	A. ON-SCENE COCRDIN	LATCR INSME SHE Phon	e Ne.i
Fred	1 Davidson (716) 52	2-3093	Janet Smit	h (215) 597-0	807
AESPONSE L	CC17ION ISke Name and/or Le	tress and Zip Code	10. CONTRACTOR RECU	IRED ON SITE OAR AN	e Timei
ABC 2927 Arli	Service Station North 2nd Street Ington, VA 22201		01 / 07 / 87		EST
	•				
L STATEMENT nocossory h 1)	CF WCRK The Cantractor snew for or members to the performance Establish & Secur visibility fence	i furmism the mecessary is of the worm set form is ity perimeter a and installing	eranne. melenets. serve enew tround the site waraing signs a	by installing directed by	high- OSC.
2)	Drain two 10,000- recycling facilit	gallon tanks of y in compliance	diesel oil and with RCRA.	transport to	• <u>-</u> .
	Excavate grossly	contaminated so	il and securely	store ou site	•
3)			wardlarian an	signent for end	closed
3) 4)	Provide and instal structures, as new	ll non-sparking cessary.	, ventiation eq.	•	
3) 4) 5)	Provide and instal structures, as new Collect samples of	ll non-sparking cessary. f soil and dies	el oil as direct	ed by OSC.	
3) 4) 5)	Provide and instal structures, as new Collect samples of	ll non-sparking cessary. f soil and dies	el oil as direct	ed by OSC.	

- Order Number (Box 3): Enter a nine (9) digit DO number
 which sequentially consists of:
 - Last four digits of the contract number (see Box 2 above);
 - EPA Region (e.g., 01, 02, . . .10); and
 - Three digit number representing the sequence of the DO being issued in the Region or Agency.
- <u>Time of Initial Order (Box 4)</u>: Enter the time of issuance of the DO. All references to time on the DO should reflect the local time at the site where the services are to be provided.
- Delivery Order Ceiling Amount (Box 5): Enter the total estimated cost of contractor personnel, equipment, and materials for which the order is being placed. The ceiling amount represents the amount obligated by the government for the corrective action. The OSC's/ Ordering Officer's authority to obligate the government for UST response is limited to \$50,000. All initial DOs or Order modifications for amounts greater than the initial \$50,000 must be obligated by the CO. Under no circumstances may the ERCS contractor develop the cost estimate used here. This is the sole responsibility of the Federal Ordering Officer who may, however, with Project Officer approval, seek the assistance of the Technical Assistance Team (TAT) contractor or EPA's Environmental Response Team (ERT) in developing the estimate.
- <u>Accounting and Appropriation Data (Box 6)</u>: Accounting and appropriation data consist of four numbers, which should be entered as follows for UST actions:
 - <u>Appropriation Number</u>: **68-20X8153** (does not change).
 - <u>Account Number and Document Control Number (DCN)</u>: Represented by ten (10) and six (6) character numbers, respectively. These numbers will be obtained by the ERD Regional Coordinator from the Headquarters Financial and Administrative Management Support Staff (FAMS) for all actions given prior Headquarters approval. If the Region's \$50,000 allocation is being used (i.e., the emergency criteria in section 4.1.1 have been met), the Region should use its own numbering system developed by the Regional Financial Management Officer.
 - <u>Object Class Code</u>: 25.35 (for all EPA program contracts; does not change).

- Issued To: Contractor (Box 7A): Self-explanatory.
- Program Manager (Box 7B): Enter the name and phone number of the contractor representative authorized to receive the DO and commit contractor resources to provide the services and supplies required to complete the Statement of Work.
- <u>Response Manager (Box 7C)</u>: Enter the name and phone number of the contractor representative designated by the Program Manager as the single point of contact for on-scene coordination and responsible for management and execution of cleanup activities as specified by the OSC or other designated Federal officials.
- <u>Issued By: Ordering Officer (Box 8A)</u>: Selfexplanatory.
- <u>EPA Region/USCG District (Box 8B)</u>: Enter the number for the EPA Region; the USCG District does not apply for Federal-lead UST corrective action.
- <u>Zone (Box 8C)</u>: Enter the number of the ERCS zone where the site is located:

Zone 1 - Regions I-III Zone 2 - Region IV Zone 3 - Region V Zone 4 - Regions VI - X

.*

- <u>On-Scene Coordinator (Box 8D)</u>: Self-explanatory.
- <u>Response Location (Box 9)</u>: Enter the location of the release or site where services are to be performed by the contractor.
- <u>Contractor Required on Site (Box 10)</u>: Enter the date and time contractor personnel, equipment, and materials are required on site to implement the corrective action.
- <u>Required Work Completion Date (Box 11)</u>: Enter the anticipated date by which contractor services are to be completed. Estimates are acceptable and dates may be revised through modifications issued by the CO.

- <u>Statement of Work (Box 12)</u>: This block contains the description of the services to be performed by the ERCS contractor. The Statement of Work should not be so narrow as to restrict the contractor's effort nor so broad as to permit the contractor to explore areas having little relationship to the desired work. The block should either contain, or refer to attachments that contain:
 - Statement of Work, including a task breakdown and schedule;
 - Site-specific institutional requirements or clearances that must be obtained by the contractor (e.g., permits for transportation and disposal of wastes or right-of-way clearances); and
 - Any plans, including a site operations plan, health and safety plan, or quality assurance plan developed for the specific corrective action.
- Ordering Officer (Box 13): Self-explanatory.
- 2. Delivery Order processing.

The completed DO is signed by the Ordering Officer and issued to the contractor Program Manager or designee (e.g., onsite Response Manager). The contractor is required to acknowledge receipt of the DO in writing within one week or one half of the time specified for performance of the order, whichever is less, following receipt. The acknowledgment of receipt of the DO must be submitted to the Ordering Officer, with a copy forwarded to the CO.

It is the Ordering Officer's responsibility to make sure that the contractor submits an acknowledgment of receipt each time a DO is issued. The acknowledgment of receipt will help preclude misunderstandings between the Ordering Officer and the ERCS contractor over the terms and conditions of the DO. It will also serve as documented evidence when potential contractual actions are required to enforce the terms of a work order.

C. <u>Delivery Order Modifications</u>

During the course of an UST corrective action, it may become necessary to modify the Statement of Work, completion date, or ceiling amount specified in the DO. All such changes must be authorized in a written amendment of the DO using Standard Form 30 (see Exhibit G-2). If increased funding is required for a project, the existing DO should be amended; a new DO should not be issued.

EXHIBIT G-2

		05.000170.007	11	
AMENOMENT OF SOLICITAT	ION/MODIFICATION	UP CONTRACT		
LAMENOWENT. WOBIFICATION NG.	T. CALECTINE ONLE	A REGUISI MONIDUA	C-456 463 40	1 340.25 NO Ill concerns
	1	I ARMINISTERED AN		
			, 115 A 1994 5 1996 1996 199 64 1	
. NAME AND ADDRESS OF CONTRACTOR	Ma., serves, county, State and	ZIP Com	I'd) SA. AMENOM	ENT OF SOLICITATION NO.
			140. SATES /3	
				THIS PART AND THE TRANSPORT
			NO.	
			108. DATED 4	188 ITEM 131
12CE	FACILITY CODE		1	
11 THISI	TEN ONLY APPLIES TO	AMENOMENTS OF S	OLICITATIONS	
	and the second section of the second			
		and and any Subrat lat		
	ent prior to the hour and date	southing in the sourcestion	A GE AN APPRICAGE, BY A	ne of the following Methods:
A By comparing theme & and 15, and celumine	costs of the amend	ment: (b) Ev acanowater		comment as such spar of the off
Consisted, or let By mearate letter or telegram			ment numbers, FALL	IRE OF YOUR ACKNOWLED
AENT TO BE RECEIVED AT THE PLACE DES	IGNATED FOR THE RECEI	T OF OFFERS PRIOR TO	THE HOUR AND	ATE SPECIFIED MAY RESUL
		the amendment, and is rea		we new and date stapfied.
	A Ill meneral)			
IJ. THIS ITEM	Applies only to mod Is the contraction		THACTS/UNDER	A ,
- A. THIS CHANGE CROEN IS ISSUED AU	ASUANT TO: (Specify eutre	NITI THE CHANGES SET	PORTH IN ITEM 14	
TRACT ORDER NO. IN ITEN LOA.				
		EPLECT THE ADMINIST	BATINE CHANGES	
aproprietan dest. est. SET PORTIN IN	TEN 14. PURSUANT TO T	HE AUTHORITY OF FAL	43.103(8).	
C. THIS SUPPLEMENTAL AGREEMENT	S ENTERED INTO PURSUA	NT TO AUTHORITY OF.		
O. JTHER (Specify 17 pe of moutherness of	16 au (10/17)			
IMPORTANT: Contractor in not.	is required to sigh th	IS COCUMENT AND RETURN	COUHS	
	ATION (Organises by UCF a	enon nearings, including i		i Meet wet so mene terment.)
		:		
				•
accet as provides herein, at terms and condition	te of the decument referenced	In 110m 94 or 104, 45 hor	nerere changes, rema	ins unchanges and in full force
A. MANE AND TITLE OF SIGNER (Type of P	MA()	ILA. NAME AND TITLE	OF CONTRACTING	OFFICER (TTPO OF PRAL)
SE. CONTRACTOR/OFFERDE	ISC SATE SOME	ISA. UNITED STATES	AMERICA	LINC DATE -
	· 1	18474044	Conmenne Officer	,

Requests for amendments should be prepared by the Ordering Officer or the OSC and forwarded to the CO for approval. In most cases, requests and approvals can be handled over the telephone. The CO will prepare and sign the amendment and issue it to the contractor. Minor changes, such as adjustments in quantities of labor and equipment which will not result in an increase to the DO ceiling, can be provided directly to the contractor by the OSC. Such changes, however, should be noted in the written site documentation kept by the OSC, and should be forwarded in writing to the contractor and the CO.

APPENDIX H: Technical Direction Document Preparation and Processing

When the need arises for technical services during a Federal-lead UST corrective action, a Technical Direction Document (TDD) should be issued to activate the Technical Assistance Team (TAT) contractor. This appendix presents a detailed description of the process, including an explanation of the TDD form used to document the initiation of services and a description of EPA and contractor roles and responsibilities in managing and implementing the process. The following are addressed:

- o TDD completion and processing;
- o TDD amendments; and
- o Special Project TDDs.

The procedures described in this appendix apply to both TAT zone contracts; however, the specific instructions on completing the TDD form directly relate to the sample TDD forms used in TAT zone 1 (EPA Regions I-V) and in TAT zone 2 (EPA Regions VI-X).

A. <u>TDD Completion and Processing Instructions</u>

When a need for TAT contractor services arises, the Headquarters Project Officer (PO), or Regional Deputy Project Officer (DPO) with PO concurrence, prepares and issues a written TDD, which serves as the principal mechanism for initiating these services. Where practicable, the DPO should confer with the Regional UST Coordinator prior to issuance of the TDD. An oral TDD may be issued, but must always be followed by a written TDD within five (5) calendar days. The TDD includes background data, the scope of work to be performed, a schedule of deliverables, an estimate of time and costs required to perform the work, and other related information.

Only the DPO or PO is authorized to prepare and issue TDDs to the contractor TAT Leader. The TAT may not act on an UST corrective action without the specific concurrence of the PO. Each DPO can issue TDDs only to the TAT Leader assigned to that particular DPO's Region. For example, the Region I DPO can issue TDDs only to the TAT Leader assigned to Region I. TDDs can also be issued by the PO to the appropriate TAT contractor Zone Program Manager (ZPM) for special ZPM efforts within the contract Statement of Work (SOW).

1. TDD completion.

A sample TDD form that has been used in TAT zone 1 is shown in Exhibit H-1; Exhibit H-2 shows a TDD form used in TAT zone 2. Each of the elements on the TDD should be completed by the DPO as explained below.

		EXHIBIT H-1		
I. COST CENTER:	TAT - CONTRACT 68-01-7367 2 NO TECHNICAL DIRECTION DOCUMENT (TDD) OHM EMERGENCY RESPONSE AND SPILL PREVENTION PROGRAM			2 MO:
	ROY F. WESTON INC.			ZA. TYPE:
3. PRIORITY:	4. Source of funds:	S. EPA SITE IO:	& COMPLETION DATE:	E. AEFERENCE INF
- HIGH (1) MEDIUM (2) COW (3)		SA." EPA NAME:	T. OVERTIME APPROVE	DI TYES
9. GENERAL TASK DESCRIPT	70N:			
SA, ESTIMATED COST: 8			ESTIMATED HO	IURS:
10. SPECIFIC ELEMENTS:			11. INTERIM C	
	F0			
OTHER (SPECIFY):			· .	· ·
3. COMMENTS:	•			
			:	
4. AUTHORIZING OPO:	_ ·	· <u>·</u> ··································	18. (ATE:
-	· •			
•	(SIGNATU	A ()		
ACCEPTED	ACCEPTED WITH EXC	EPTIONS	REJECTED	
_	17A71 818MA			
L DESCRIPTOR:	[I.C.] 2 312/A			· · · · · · · · · · · · · · · · · · ·
 1 titles - 870 Coin there is 747, Annual - 2 Gran - 747, Con 3 Gran - 747, Con 3 Gran - Anna Coin 4 Nus - Anna Gran Gran 4 Nus - Anna Coin 	** (Drama) • 907	E-2		

1A. Cost Conter:, , 1B. Account No.:		TAT ZONE II CONTRACT ONTRACT NO. 68-01-7368 AL DIRECTION DOCUMENT (TDD) OGY AND ENVIRONMENT, INC.	Z. No.: T
3A. Prienty	4A. Estimate of Total Hours:	SA. EPA Site Name:	Amendment 7. Completion Det
I Low 38. Key EPA Contact: Name: Phone:	48. Overame Approved:	SB. SSID No.: SC. City/County/State: 6. Source of Funds: Other CERCLA I 311 UST	8. Reference info: Yee Atta No Pick EA. Subrask Code:
9. Type of Activity: CWA SPCC On-Scene M Spill Clean-	L-311CE Lonitoring Site A Lonitoring Remov Lon Funded Remov On-Site	RCLA AS SPECIFIE ssessment Soeciel Project val Funded Analytical Project val PRP (AO/CO) TITLE III te Monitoring UST FEMA	D ABOVE Quality Assurance Training Program Manegement Technical Assistance Information Manegeme
10. General Task Descriptio	on:		I1. Desired Report Form: Form: Formal Report Latter Report Formal * 'vitil Other H
12. Specific Elements:	•	•	
			13. Interim Deedlin
	· · · · · ·	·····	
			15. Dete:
16. Received by: Acc	epted 🗌 Accep	(TATL Signature)	cted 17. Date:
Sheet 2 Blue - TATL Sheet 2 Blue - TATL Sheet 3 Green - 2PM Sheet 4 Canery - PO Ci Sheet 5 Pins - CO C	Coev . Coev Coev Sov Sov Sov	E-3	π

- <u>Cost Center (Box 1A)</u>: Enter the number of the TAT
 Region that is being issued the TDD; in the sample, 07
 is entered to indicate Region VII.
- o <u>Account Number (Box 1B)</u>: Leave blank; to be filled in by the TAT contractor.
- O <u>TDD Number (Box 2)</u>: Enter this serial number which identifies the TAT Region in which the TDD is issued, the calendar year and month, and the sequence number of the TDD issued that month, For example, the number 07-8701-03 is interpreted as follows: 07 refers to the TAT location in Region VII; 87 refers to the calendar year in which the TDD is issued; 01 is the month the TDD is issued; and 03 indicates that this is the third TDD issued in the month of January. If the TDD amends an earlier TDD, check the "Amendment" blank.
- o <u>Priority (Box 3A)</u>: Check the box corresponding to the priority of the request. The priority should reflect whether an immediate action is required (high priority), an action is to be taken within a definite time period (medium priority), or an action is to be taken within a flexible time period (low priority).
- <u>Key EPA Contact (Box 3B)</u>: Enter the name and telephone number of the EPA person directly involved with overseeing and managing contractor performance of the task. This will typically be the OSC.
- <u>Estimate of Total Hours (Box 4A)</u>: Enter an estimate of the technical labor hours needed to accomplish the TDD assignment. The estimate should be based on best engineering judgment considering knowledge of the project or task requirements, data needs, and previous experience on similar projects.
- <u>Estimate of Total Costs (Box 4A)</u>: Enter an estimate of the total cost including direct labor, travel, expenses, and subcontracting.
- Overtime Approved (Box 4B): The DPO must specify
 whether overtime hours are allowable for the task
 outlined in the TDD by checking the appropriate box.
- o <u>EPA Site Name (Box 5A)</u>: Self-explanatory; if not applicable enter N/A.
- <u>SSID No. (Box 5B)</u>: UST sites must be numbered to allow cost recovery; see page 2 of Attachment I to the Comptroller's Policy Announcement No. 87-13 (Appendix I of this guidance). Contact the Regional Financial Management Officer and ERD Regional Coordinator for more information on assigning a site-specific ID number.

:

- <u>City/County/State (Box 5C)</u>: Enter the location of the UST site; if not known, this can be entered by the contractor. If not applicable enter N/A.
- o <u>Source of Funds (Box 6)</u>: Check the UST box.
- <u>Completion Date (Box 7)</u>: Enter the date specified by the DPO for the TAT contractor submittal of the completed TDD assignment. Completion dates must not be left open.
- <u>Reference Info (Box 8)</u>: This item allows the DPO to provide the TAT contractor with any reference materials or supplemental information necessary to expedite TDD completion. At the same time, accountable control of such information is maintained. Check the most applicable box.
- o <u>Type of Activity (Box 9)</u>: Check the UST box.
- <u>General Task Description (Box 10)</u>: In this section, enter a description of the task requirements that indicates the following:
 - The type of support desired;
 - The level of intensity required (the depth to which certain issues should be pursued); and
 - Any other pertinent information.

The task description must clearly delineate the goals and objectives of the activity and the desired products and/or deliverables. Ambiguous phrases such as "assist OSC" are not sufficient. Where no interim task objectives are present, continuation of the general task description can be made into Box 12. Where interim task objectives are present, continuation of the general task description should be made on a separate sheet of paper that can be attached to the TDD.

- <u>Desired Report Form (Box 11)</u>: The DPO should indicate the type of end product desired for the TDD assignment.
 Additional information can be provided in Box 12 or by attaching additional sheets to the TDD.
- O <u>Specific Elements (Box 12)</u>: In this section, elaborate on the general task description and define interim task objectives. This section should provide the TAT Leader with a clear understanding of the objectives, expected results, and required deliverables and/or reports. If more space is needed, the continuation of the description of specific task elements should be made on a separate sheet of paper that can be attached to the TDD.

- <u>Interim Deadlines (Box 13)</u>: Denote completion dates, where applicable, for the interim task objectives that
 are specified on corresponding lines in Box 12.
- Signatures and Dates (Boxes 14-17): These sections are self-explanatory. The TDDs are to be issued and signed by the PO (or DPO with PO concurrence) and received and signed by the contractor TAT Leader. If the TAT Leader judges the TDD to be out of scope, or for other reasons unacceptable, the appropriate box must be checked to show the action. This is necessary to bring the matter to the attention of the DPO, PO, Regional UST Coordinator, and Headquarters Contracting Officer (CO). An appropriate explanation can be noted on the right-hand margin of the form.
- o <u>Distribution</u>: Copies of the TDD should be distributed as specified on the bottom of the form, plus an additional copy to the Regional UST Coordinator.

The DPO should keep in mind three important points while preparing a TDD. First, the DPO must be as specific as possible in describing the scope of work to be performed by the TAT. Second, the DPO must clearly indicate contractor reporting requirements. And, finally, it is extremely important that the DPO include in the TDD estimates for technical hours and costs needed to accomplish the assignment.

Although exact tasks and response activities, particularly those involving emergency situations, may be somewhat difficult to define, it is very important to the extent possible to provide specific guidance to the contractor on the expected work scope (see Exhibit H-1, Boxes 4A and 10 - 13). The task description should be as detailed as available information permits so that there is a clear understanding by the TAT contractor of the activity objective, yet allow some flexibility for alternative and innovative actions by the contractor as conditions warrant. Examples of candidate tasks which could be performed for different types of response, corrective action, or prevention work can be extracted directly from the TAT contract Statement of Work (SOW).

Care must be taken by the DPO and TAT Leader to ensure that all tasks requested and performed are within the bounds of the TAT contract SOW. Particular attention must be given to ensure that TAT TDDs do not involve personal services. For example, the DPO must not direct the TAT to provide clerical support, travel arrangements, or other personal services for EPA staff. The DPO is also forbidden from specifying an individual TAT member to perform a task. Specific TAT staffing decisions are the responsibility of the TAT contractor.

The tasks included in any one activity will depend upon the project-specific conditions. It may prove useful for the Region to develop standardized work scopes for an activity and to use

this scope (modified as necessary for a specific situation) as an attachment to the TDD.

In addition, the TDD reporting requirements (Box 11) must by spelled out such that the work product provides the degree of detail desired by the DPO and therefore facilitates accomplishment of the activity objectives. The DPO may wish to develop standard report outlines for various types of activities (i.e., facility inspection reports) and to attach these to the TDD as appropriate.

Finally, the DPO is responsible for ensuring that an estimate of technical hours and costs needed to accomplish the TDD assignment is included in the TDD. These estimates may be used in three important ways to ensure that the TAT performs efficiently. First, these estimates can serve as the basis for individual TDDs. The actual hours and costs can then be compared with the estimates. Any large differences between the figures may be indicative of problems and should be reviewed by the DPO. Second, the estimated hours, used in combination with the estimated completion date, can help to identify the need for overtime to complete the task. Since the contract limits the overtime which may be charged, overtime must be monitored.

The third use for the estimate of technical hours is to provide the basis for comparison of total estimated hours for all active TDDs with the total technical hours available. Total technical hours available can be computed from the information in the contract. The difference between the two figures will tell the DPO to what extent available contractor resources are being utilized. The TAT contractor can also provide this information on a regular basis (e.g., biweekly).

2. TDD processing.

After the TDD is completed, it is signed by the DPO or the DPO's designee and forwarded to the contractor TAT Leader. (When the DPO plans to be out of the office and unavailable to perform contract management functions, the DPO must submit a written request, or telephone the PO and CO, for approval to appoint a temporary assistant DPO; this request must be for a discrete period of time.) The PO must also be notified by the DPO for concurrence with the TDD. The PO will convey concurrence to the ZPM, who will communicate this concurrence to the TAT Leader as soon as possible. The contractor TAT Leader can take one of three actions on the TDD:

 If the TDD is acceptable to the contractor TAT Leader as issued by the DPO, the TAT Leader signs the TDD and proceeds with performance of the assigned activities. In general, the TAT contractor should be willing to accept any assignment within the bounds of the TAT contract SOW. However, as stated earlier, conducting personal services for EPA staff is not allowed.

- The TAT leader can accept the TDD with exceptions. 0 In this case, the exceptions are to be noted on the form • • and a copy is returned to the DPO. The DPO and TAT Leader are to resolve these exceptions prior to the contractor commencing work. If necessary, a revised TDD should be issued reflecting resolution of any exceptions. If the issue cannot be resolved on this level, the DPO should consult with the PO in Headquarters for guidance. If necessary, the PO will consult with the contractor ZPM to try to resolve the situation. If it is resolved, the DPO may issue a revised TDD reflecting the issue resolution and the contractor will proceed with task performance. If the PO and contractor ZPM cannot resolve the issue, the matter is referred to the CO for final resolution. Where possible, and with concurrence of the DPO, the contractor should begin work on the acceptable work elements of the TDD, pending resolution of the points at issue.
- o If the contractor TAT Leader rejects the TDD, it is returned to the DPO with an explanation of the reasons for rejection. The DPO and TAT Leader are to confer to resolve the situation. If a resolution is reached, a revised TDD can be issued, if necessary, and the contractor proceeds with the task. If the DPO and the TAT Leader cannot resolve the problem, the DPO should contact the PO for guidance as explained above.

When the TDD is acceptable to both the TAT Leader and the DPO, it should be signed and dated. Appropriate copies should be placed in the contractor's and EPA's central TDD files. The DPO should then send copies of the TDD to the PO, CO, and the Regional UST Coordinator.

B. <u>TDD Amendments</u>

In the event that an issued TDD needs revision, the DPO must issue a TDD amendment containing the appropriate changes. Events that require TDD amendments may include, for example, an original underestimation of a project's magnitude or an acceleration of a project's period of performance. The original TDD's number will be used, with a check in the "Amendment" blank (Box 2). The revised TDD should include the material contained in the original TDD and the revised information so as to be capable of standing alone. Amendments are to be processed in the same manner as described above.

C. <u>Special Project TDDs</u>

Funding has been obligated for the performance of special projects under the TAT contracts. All TAT Special Projects TDDs for Federal-lead UST corrective actions must be signed and issued by the appropriate PO, after conferring with the Regional UST Coordinator. If a special project is required, the PO may request that the ZPM prepare and submit a work plan and cost estimate for completing the TDD. The TAT contractor's work plan should include a detailed technical approach, schedule for completion, and cost breakdown. It should be sufficiently detailed to provide a base for guiding work, measuring progress, and controlling budgets. The contractor work plan should address the elements specified below:

- Objectives and scope of the proposed activity and its relationship to other ongoing or planned activities.
- Technical approach for the activity, anticipated problems, and proposed solutions.
- o Detailed breakdown of tasks to be performed.
- Description of work to be performed under each task, including objectives and scope, information sources, and methods to be used.
- o Analysis of resources (level) expected for each task.
- o Anticipated total labor cost for each task.
- o Identification and tabulation of total direct costs for each major work element.
- o Schedule, including critical path and milestones.
- o Listing and schedule of deliverables.
- o Safety and contingency measures.

·.

Upon review of the plan, the PO will approve and issue the TDD.

APPENDIX I: Interim Financial Procedures Governing Use of the LUST Trust Fund

Comptroller's Policy Announcement No. 87-13

Available from OUST Program -202 - 475-9810

ì

8-475-9810

I-10

Policy for Superfund Compliance with the RCRA Land Disposal Restrictions



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

APR 1 7 1989

OFFICE OF 2010 1 ASTE AND EMERCENCY POSPOT

CSWER Directive 9347.1-02

MEMORANDUM

SUBJECT:	Policy for Superfund Compliance With the RCRA Land Disposal
	Restrictions
	MALAIL I DANAAL
FROM:	Jonathan 2. Campon
	Acting Assistant Administrator
TO:	Regional Administrators, Regions I-X

Purpose

To transmit the Superfund policy for complying with the RCRA land disposal restrictions (LDRs) at Superfund sites.

Background

CERCLA section 121(d) requires on-site Superfund remedial actions to comply with Federal, and more stringent State, environmental requirements that are determined to be applicable or relevant and appropriate requirements (ARARs). Section 121 also identifies six ARAR waivers: 1) interim remedy; 2) greater risk to human health and the environment; 3) technical impracticability; 4) equivalent standard of performance; 5) inconsistent application of State standard; and 6) Fund-balancing.

With regard to Superfund removal actions, the current NCP requires on-site removal actions to comply with Federal ARARs to the extent practicable, considering the exigencies of the situation. The preamble to the proposed NCP contains guidance on how to determine whether compliance is "practicable."

On-site removal and remedial actions must comply with <u>substantive</u> aspects of both **applicable and relevant and appropriate** requirements. Off-site removal and remedial actions must comply with both <u>substantive</u> and <u>administrative</u> aspects of applicable requirements only.

The RCRA land disposal restrictions are a potential ARAR for Superfund actions. As you may know, OERR is developing a guidance document to assist the Regions in complying with the LDRs. Although several issues must be resolved before this guidance is issued, this memorandum will summarire one of the major issues that has been decided, namely, how to determine whether the LDRs are "applicable" to a Superfund response action. This policy will be discussed in greater detail in the guidance document.

Objective

In order to assist Regional removal and remedial staff in making current site decisions about the LDRs, this memorandum will explain: 1) how to determine when the LDRs are "applicable" to a Superfund removal or remedial action, and 2) the Superfund approach for complying with the LDRs when they are determined to be applicable. (This memorandum does not address how to make "relevant and appropriate" determinations.)

Implementation

Section A below explains how site managers (OSCs, RPMs) should determine whether the LDRs are "applicable" to a Superfund response action. Section B explains how Superfund intends to comply with the LDRs when they are determined to be applicable.

A. Application of the LDRs to CERCLA response actions

To determine if the LDRs are applicable to a given response action at a Superfund site, the site manager must answer three questions. The answer to each question must be "yes" for the LDRs to be applicable.

1. Does the CERCLA action constitute "placement"?

The LDRs are triggered as applicable requirements by "placement" of restricted RCRA hazardous wastes in land-based units.¹ Placement occurs when wastes are land disposed (or placed) in land-based RCRA units, such as landfills, surface impoundments, waste piles, and land treatment facilities. Placement does not occur if wastes are moved within a unit or are left in place (e.g., capping, in-situ treatment, consolidation within a unit). Placement does occur when wastes are moved from one unit and placed in another unit. For example, if wastes from a CERCLA site are disposed at an off-site landfill, this action constitutes placement.

However, the concept of a RCRA unit may be less useful for uncontrolled hazardous waste sites, which often involve widespread and dispersed contamination. Therefore, to assist in defining when placement occurs for onsite disposal at Superfund sites, the Agency has developed the concept of an

¹ Several LDR requirements (the storage restrictions, dilution prohibition, and off-site notification requirements, in particular) are triggered when restricted wastes are generated, or picked up, rather than when the wastes are "placed." However, the major LDR restrictions discussed in the remainder of this memorandum are triggered only if wastes are "placed."

"area of contamination" (AOC). An AOC is delineated by the extent of continuous contamination, although one AOC may contain varying types and concentrations of contamination. For example, a waste pit with the surrounding contaminated soil is one AOC and may be viewed as a single "unit," e.g., a single landfill. For the purposes of the LDRs, therefore, AOCs are equivalent to RCRA units.

Movement of waste within the AOC does not constitute placement, but movement of waste out of the AOC into another unit will trigger placement. Placement would occur if wastes from different AOCs are consolidated into one AOC or if wastes are removed and treated outside the AOC and returned to the same or a different AOC. Placement would also occur if wastes are excavated from the AOC, placed in an incinerator or tank located within the AOC, and then redeposited into the AOC, because the incinerator and tank are considered separate units from the AOC.

2. Is the CERCLA waste also a RCRA hazardous waste?

The LDRs are applicable only to RCRA hazardous wastes (i.e., listed and characteristic wastes identified under §261). However, not all wastes at Superfund sites are RCRA hazardous wastes. Therefore, the site manager must decide if it is reasonably ascertainable, within the scope of the Superfund site investigation, that the CERCLA waste is also a RCRA hazardous waste. Reasonable efforts must be used to collect the information needed to determine if a waste is a RCRA listed or characteristic waste. (It is expected that current data collection efforts at Superfund sites should be sufficient for this purpose.) The site manager should have affirmative evidence (e.g., manifests, records, knowledge of process) to demonstrate that the Superfund waste is a RCRA hazardous waste for the LDRs to be potentially applicable.

To determine whether a CERCLA waste is a RCRA <u>characteristic</u> waste, site managers may test the waste or use their knowledge of the properties of the waste. To determine if a waste is a <u>listed</u> waste, sampling alone will not be sufficient. The RCRA listing descriptions will generally require that the site manager have knowledge about the source of the waste (for example, did the sludge on site result from a wastewater treatment operation?) or its prior use (e.g., was the waste unused when it was discarded?).

If the site manager determines that the site waste is a RCRA hazardous waste, he/she must also determine if that waste is a "California list" waste. The California list wastes are a distinct category of RCRA hazardous wastes regulated under the LDRs. The LDR regulations describe the California list wastes and they will be discussed in the forthcoming guidance document.

3. Is the RCRA waste restricted under the LDRs at the time of placement?

The land disposal restrictions are being phased in for the RCRA hazardous wastes over a period of time. Attachment 1 presents the LDR statutory deadlines established by section 3004 of the 1984 RCRA amendments. A RCRA waste becomes a restricted waste under the LDRs on its statutory deadline, or earlier if EPA chooses to promulgate treatment standards for a waste prior to this deadline. Note that after May 1990, <u>all</u> RCRA hazardous wastes (that were

-3-

listed or characteristic as of the 1984 RCRA amendments) will be restricted under the LDRs.

To determine if the LDRs are applicable, site managers should determine if the RCRA waste will be restricted under the LDRs at the time the waste is to be placed.

To summarize Section A, the LDRs are applicable when three conditions are met: 1) the CERCLA action constitutes placement, 2) the CERCLA waste is a RCRA hazardous waste, and 3) the RCRA waste is restricted at the time of placement. If these conditions are met, the CERCLA action must comply with the LDRs, unless an ARAR waiver is granted (remedial actions) or compliance with the LDRs is determined not to be "practicable" (removal actions).

B. Superfund compliance with the LDRs

Section B briefly describes the different types of LDR requirements and provides an overview of the Superfund approach for complying with these LDR requirements when they are determined to be "applicable." Section B describes only the major LDR restrictions; the upcoming guidance document will give a complete description of all LDR provisions.

1. Summary of the major LDR requirements

When a waste becomes "restricted" on its statutory deadline (or possibly earlier), one of four types of restrictions will take effect:

<u>Treatment standard</u> (§268.40-43) - The RCRA amendments direct EPA to promulgate treatment standards for all RCRA hazardous wastes by the statutory deadlines. To date, most of the standards set by EPA are concentration levels that must be achieved prior to land disposal. (The regulations specify whether a total waste analysis or the Toxicity Characteristic Leaching Procedure (TCLP) must be used to measure the concentration levels.) For concentration-based treatment standards, any technology may be used to achieve these standards. However, in limited cases, EPA has also promulgated a specific technology as a treatment standard, or has established a "no land disposal" treatment standard where a waste was no longer generated, no longer being land disposed, or was capable of being totally recycled.

National capacity extension (§268.30-33) - When EPA sets a treatment standard for a waste, it must also determine if there is sufficient capacity available nationwide to treat the waste to that standard. If not, EPA may grant a nationwide capacity extension for the waste for up to two years. During the extension, the waste does not have to meet the treatment standard. However, if waste that does not meet the standard is disposed in a landfill or surface impoundment, the receiving unit must meet the RCRA §3004(0) minimum technology requirements (e.g., double liner, leachate collection system, ground water monitoring). Because of these limitations on disposal, wastes are still considered "restricted" during national capacity extensions. Attachment 2 highlights the national capacity extensions that EPA has granted to date for CERCLA soil and debris wastes that are contaminated with RCRA restricted wastes.

<u>Soft hammer</u> (§268.8) - If EPA fails to set a treatment standard for a First or Second Third waste on the statutory deadline, the soft hammer goes into effect automatically. The soft hammer places two requirements on the disposal of wastes in landfills and surface impoundments: 1) the receiving unit must meet the RCRA minimum technology requirements, and 2) the generator must demonstrate and certify that he has investigated treatment options for the waste, and, where treatment is practically available, that the waste has been treated using the best practically available treatment method. The soft hammer remains in effect until EPA sets a treatment standard for the waste, or until the hard hammer falls in May 1990, whichever comes first.

<u>Hard hammer</u> (RCRA §3004(g)(6)(C)) - If EPA fails to set a treatment standard for a solvent, dioxin, or California list waste by the statutory deadlines for these wastes, or for any "Third" waste by May 1990, the hard hammer falls. The hard hammer prohibits all land disposal of the affected waste.

Compliance with RCRA and the LDRs may also be obtained through several options other than meeting the restrictions above. It is important to note that these options constitute compliance with RCRA; they do not require an ARAR waiver under CERCLA.

A <u>Treatability Variance</u> (§268.44) is available when a treatment standard has been set for a waste. The variance can be used where, because the site manager's waste is significantly different from the waste used by EPA to set the treatment standard, the standard cannot be met or the BDAT technology is inappropriate. The variance can be granted either administratively, for a particular waste at a particular site, or through a rule-making procedure, which establishes a new nationwide waste category and associated treatment standard.

An Equivalent Treatment Method Petition (§268.42) can be used where a treatment standard is a specified technology, but the site manager can demonstrate that another technology can achieve an equivalent measure of performance.

A <u>No-Migration Petition</u> (§268.6) can be used as an alternative to any of the four restrictions above. The site manager must demonstrate that there will be no migration of hazardous constituents above health-based levels from the disposal unit or injection zone for as long as the waste remains hazardous.

<u>Delisting</u> (§260.20 and §260.22) can be used as an alternative to any of the four restrictions above, when the RCRA hazardous waste is a listed waste. The site manager must demonstrate that: 1) the waste does not meet any of the criteria under which the waste was listed, and 2) other factors
(including additional constituents) would not cause the waste to be hazardous.

2. Superfund approach for complying with the LDR requirements

The present Superfund approach for complying with the LDRs when they are applicable requirements is illustrated below:

CASE A: CERCLA liquid or sludge wastes that are also RCRA restricted hazardous wastes

CERCLA liquid + RCRA restricted + Placement = LDR is applicable. Must or sludge hazardous waste comply (unless CERCLA ARAR waiver is granted). If the LDR restriction is a treatment standard, evaluate whether it can be met. If not, determine if a Treatability Variance or other RCRA option is appropriate.

CASE B: CERCLA soil or debris wastes that contain RCRA restricted hazardous wastes

CERCLA soil + RCRA restricted + Placement = LDR is applicable. Must or debris hazardous waste comply (unless CERCLA ARAR waiver is granted). If LDR restriction is a treatment standard, will generally be appropriate to seek a Treatability Variance. Other RCRA options may also be appropriate.

CERCLA response actions often address waste matrices, such as contaminated soil and debris, that are different from the RCRA industrial wastes used to set the LDR treatment standards. Therefore, the Agency is undertaking a rulemaking that will set LDR treatment standards specifically for contaminated soil and debris. Until that rulemaking is completed, site managers should use the data collected during the removal and remedial site investigations to support a Treatability Variance for soil and debris where necessary. As part of this interim approach, the Agency is developing specific guidance for obtaining a Treatability Variance for soil and debris, which establishes alternate treatment levels or methods for soil and debris. If you have further questions, you may call the Headquarters Superfund Regional Coordinators, Carolyn Offutt of the CERCLA program (FTS 475-9760), or Michaelle Wilson of the RCRA land disposal restrictions program (FTS 382-4770).

Attachments

cc: Regional Counsel, Regions I-X Director, Waste Management Division, Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division, Region II Director, Hazardous Waste Management Division, Regions III and VI Director, Toxics and Waste Management Division, Region IX Director, Hazardous Waste Division, Region X Environmental Services Division Directors, Regions I, VI, and VII Henry Longest Sylvia Lowrance Bruce Diamond Lisa Friedman Superfund Branch Chiefs, Regions I-X Oil and Hazardous Materials Coordinators, Regions I-X Bettie Van Epps, OERR Document Coordinator

•

Attachment 1

LDR STATUTORY DEADLINES

RCRA HAZARDOUS WASTE	STATUTORY DEADLINE*
Spent solvent wastes (F001-F005)	November 8, 1986
Dioxin wastes (F020-F023 and F026-F028)	November 8, 1986
California list wastes - Any RCRA hazardous waste; and - Liquid (except for HOCs); and - Exceeds statutory prohibition level for certain cyanides, metals, corrosives, PCBs or HOCs	July 8, 1987
CERCLA/RCRA corrective action soil and debris (Solvent-containing, dioxin-containing, and California list wastes only)	November 8, 1988
First Third wastes (listed RCRA hazardous wastes)	August 8, 1988
Second Third wastes (listed RCRA hazardous wastes)	June 8, 1989
Third Third wastes (listed and characteristic RCRA hazardous wastes)	May 8, 1990
New RCRA wastes (any RCRA hazardous waste listed or identified under RCRA 3001 after November 8, 1984)	Within 6 months of listing or identification**

* These dates are statutory deadlines in HSWA. On this date, some type of LDR restriction will apply (i.e., treatment standard, minimum requirement during national capacity extension, soft hammer, hard hammer). However, the Agency also has the authority to restrict a waste earlier than its statutory deadline. Currently, the Agency is planning to restrict certain Third Third wastes in the June 1989 Second Third rule, so individual regulations must be checked.

** If EPA misses the 6 month deadline, the waste will not be restricted under the LDRs because HSWA contained no hammer provisions for newly identified wastes.

Attachment 2

LDR NATIONAL CAPACITY EXTENSIONS FOR CERCLA SOIL AND DEBRIS

Waste Category	Statutory Deadline	Treatment Standard Effective Date
Solvent (F001-F005)	November 8, 1988	November 8, 1990*
Dioxin (F020-F023 and F026-F028)	November 8, 1988	November 8, 1990*
California list (HOCs)	November 8, 1988	November 8, 1990*
First Third:		
Wastes where BDAT is incineration	August 8, 1988	August 8, 1990*
Wastes where BDAT is other than incineration	August 8, 1988	August 8, 1988**
Soft hammer wastes - treatment standard not set; must meet soft hammer restrictions as of 8/8/88	August 8, 1988	N/A

* The effective date is based on the granting of a national capacity extension. During the capacity extension, the soil and debris do not have to meet the promulgated treatment standards. However, if soil or debris that does not meet the standard is disposed in a landfill or surface impoundment, the receiving unit must meet the RCRA minimum technology requirements (double liner, leachate collection system, ground water monitoring).

** Except for K048-K052 and K071, which were granted capacity extensions until August 8, 1990.

Land Disposal Restrictions as Relevant and Appropriate Requirements for CERCLA Contaminated Soil and Debris



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

JUN - 5 1989

OSWER Directive No. 9347.2-01

MEMORANDUM

SUBJECT: Land Disposal Restrictions as Relevant and Appropriate Requirements for CERCLA contaminated Soil and Debris
FROM: Henry L. Longest II, Director Wether Handing Particle of Emergency and Remedial Response for Office of Emergency and Remedial Response for Office of Waste Programs Enforcement
TO: Directors, Waste Management Division Regions I, IV, V, VII, VIII
Director, Emergency and Remedial Response Division Region II
Directors, Hazardous Waste Management Division Regions III, VI

Director, Toxic and Waste Management Division Region IX Director, Hazardous Waste Division Region X

PURPOSE

To transmit OSWER policy on the relevance and appropriateness of the Land Disposal Restrictions (LDRs) to CERCLA responses involving contaminated soil and debris.

BACKGROUND ·

As clarified in OSWER Directive 9347.1-02 (see attachment), the LDRs are applicable to CERCLA responses only when such actions constitute placement of a restricted RCRA waste. Therefore, if no restricted RCRA wastes are identified in a Superfund waste that is being placed, the LDRs would not be applicable. Site-specific questions have arisen, however, as to the relevance and appropriateness of the LDRs to soil and debris that do not contain RCRA restricted wastes. In particular, Region II (having determined that the contaminated soil and debris to be treated and "placed" at the 93rd Street site did not contain RCRA hazardous wastes) sought consultation with Headquarters on whether LDRs should be considered relevant and appropriate given that the Agency is in the process of developing treatment standards for soil and debris wastes separate from the treatment standards developed for industrial process wastes.

OSWER POLICY

OSWER has concluded that until a rulemaking is completed that establishes treatment standards for soil and debris, the LDRs generally should not be considered as relevant and appropriate for soil or debris that does not contain restricted RCRA wastes. The following language should be incorporated into feasibility study ARAR discussions, proposed plans, and the "Compliance with ARARs" section of future RODs for situations similar to the above example:

The Agency is undertaking a rulemaking that will specifically apply to soil and debris. Since that rulemaking is not yet complete, EPA does not consider LDR to be relevant and appropriate at this site to soil and debris that does not contain RCRA restricted wastes.

Should you have any questions regarding this policy, please contact your Regional Coordinators in the Hazardous Site Control Division, the CERCLA Enforcement Division, or Steve Golian (FTS 475-9750) in the Site Policy and Guidance Branch.

Attachment

cc: Sylvia Lowrance, OSW

Notification of Out-Of-State Shipments of Superfund Site Wastes

JUNER U	rective initi	ation Request	OSWER 9330.2-07
	2. Criginator Inform	stion	
Name of Contact Person	Man Code	Cifice	Telenore Cace
Hugo Fleischman	<u> </u>	OERR/HSCD/SLCB	382-5734
Notification of Out-of	-State Shipment	s of Superfund Site	Wastes
Summary of Crocine include oner statemen The purpose of this memo i shipment of Superfund wast Regional personnel will pr	totournesses s to implement es to an out-of ovide notice to	EPA's policy that pr state waste managem that State's enviro	ior to the off-site ent facility, EPA rmental officials.
Keywords Superfund, CERCLA, SA	RA		
a. Does This Cirective Supersede Previous Cir	COVE(S)	. Ves What drect	re (number. Lile)
). Does it Supplement Prevous Grectivets)?	X	Yes What dreck	ive (number, tise)
Sraft Lave 8	- Signed by Ciffice Direct	car C - For Ronow & C	comment 0 - In Oevecom
8. Document to be distr	ibuted to States	by Headquarters?	Y Yes No
his Request Meets OSWER Directives Syste	em Formet Slandarda.		
Signature of Lead Office Directives Coordinate	07		Cate
Betti VanEpps, OERR Publica	tions Coordina	Lor	
Name and Title of Approving Cificial			Cate
Jonathan Cannon, Act. Asst.	Admin., OSWER		

EPA Form 1315-17 (Rev. 5-67) Previous eatons are assored.

OSWER OSWER OSWER O E DIRECTIVE DIRECTIVE DIRECTIVE



SEP | 4 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

OSWER Directive 9330.2-07

MEMORANDUM

SUBJECT:	Notification of Out-of-State Shipments of Superfund Site
•	Wastes y
FROM:	Jonathan 2 Cannon, Acting Assistant Administrator
	Office of Solid Waste and Emergency Response
TO:	Regional Administrators Regions I- X

PURPOSE

The purpose of this memorandum is to implement EPA's policy that prior to the off-site shipment of Superfund wastes to an out-of-State waste management facility, EPA Regional personnel will provide notice to that State's environmental officials.

BACKGROUND

A number of States and localities have expressed the concern that they are not formally notified before unusually large amounts of Superfund wastes are transferred to permitted facilities within their States for treatment, storage or disposal. The Agency believes that such notice may be appropriate, and that indeed, such notice may be helpful in facilitating the safe and timely accomplishment of Superfund waste shipments. Thus, EPA has decided to adopt a policy of providing States with prior notification of off-site shipments of Superfund wastes. (Because the State in which the site is located participates in the remedy selection process, and thus is already aware of Superfund remedies within that State, this policy will apply only to out-of-State waste shipments.)

OBJECTIVES

The objectives of this policy are to alert States to shipments of wastes from out-of-State Superfund sites, so that the State may take any steps necessary to facilitate the safe transfer of waste, and to respond to any public inquiries concerning the waste movements.

IMPLEMENTATION

Effective immediately, the Region should implement the notification of Superfund waste shipments. The Agency intends that notice under this policy should be routinely provided to State environmental officials for all remedial actions and "nontime critical" removal actions involving the off-site shipment of Superfund wastes that are known to Regional officials, including waste shipments arising from Fund-lead responses, State-lead responses, Federal facility responses, and responses conducted by PRPs (emergency and time-critical removals are not covered by this policy). This policy pertains to all such offsite shipments, not merely response actions involving "unusually large" shipments; however, you may, in your discretion, decide that notice is unnecessary for shipments of small amounts of wastes (e.g., 10 cubic yards).

The notification should be in writing, and should set out the following information, where available: (1) the name and location of the facility to which the wastes are to be shipped; (2) the type and quantity of waste to be shipped; (3) the expected schedule for the waste shipments; and (4) the method of transportation. In addition, the Region should notify the State of major changes in the shipment plan, such as a decision to ship the wastes to another facility within the same State, or to a facility in another State.

The identity of the receiving facility and State will be determined by the lead agency following the award of the response action contract. The Region should provide relevant information on the off-site shipments as soon as practicable after the information becomes available to the Region: unless the information is unavailable, it should be provided before the wastes are actually shipped.

• Because CERCLA actions may be carried out under a number of mechanisms and by a number of parties (e.g., lead State agencies, other Federal agencies, PRPs), OSWER plans to issue additional guidance to help Regions to implement this notification policy. That additional guidance will address: modification of contracting procedures and contract terms in order to provide for routine notification under Fund-lead response actions; modification to cooperative agreements in order to provide for routine notification under State-lead response actions; model provisions for orders and settlement agreements with PRPs in order to provide for routine notification under PRP response actions; and model provisions for Interagency Agreements to provide for notification in Federal facility response actions. The guidance may also attempt to establish levels below which such notice would not be necessary. OSWER will minimize the burden on the Regions by placing the responsibility for prior notification on the response action contractor where appropriate.

Should you have any questions concerning this policy or its implementation, please contact William O. Ross of my staff at FTS 382-4645.

.

Analysis of Treatability Data for Soil and Debris: Evaluation of Land Ban Impact on Use of Superfund Treatment Technologies

OSWER DI	rective Initia	ation Request	9380.3-04
	2. Originator Informati	lon	
Name of Contact Person Henry L. Longest	Mail Code OS-200	OERR	Telephone Code 382-2180
1 Tile Analysis of Treatabi Evaluation of Land B Treatment Technologi	lity Data for So: an Impact on Use es: A Superfund !	il and Debris: of Superfund Management Review: Re	commendation 34A
4. Summary of Directive (include bief statemen The purpose of this memo of treatment technologies to the recommendation in evaluate impact of RCRA	nt of purpose) is to transmit a s for contaminate the Superfund Ma land ban and othe	an analysis of the ef ed soil and debris in anagement Review to " er rules on use of al	fectiveness response carefully ternative technolog
S. Keywords			
6a. Daes This Directive Supersode Previous D	irective(s)?	Yes What directly	e (number, tile) .
b. Daes it Supplement Previous Directive(s)?	No	Yes What directly	f (fiumber, 113e)
7 Draft Lovel A - Signed by AA/DAA	I – Signed by Office Directo	C - For Review & Co	mment D-ti Develo;
8. Document to be dist	ributed to States	by Headquarters?	Yes No
This Request Meets OSWER Directives Sys	tem Format Slandards.		
9. Signature of Land Cffice Orectives Coordina	·		Date
Betti VanEpps			11-30-89
10. Name and Title of Approving Official			Date
Henry L. Longest II	•		11-30-89

EPA Form 1315-17 (Rev. 5-67) Previous editions are obsidete.

OSWER OSWER OSWER (VE DIRECTIVE DIRECTIVE DIRECTIVE

.



NOV 30 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

OSWER Directive 9380.3-04

MEMORANDUM

SUBJECT: Analysis of Treatability Data for Soil and Debris: Evaluation of Land Ban Impact on Use of Superfund Treatment Technologies

Superfund Management Review: Recommendation 34A

FROM: Henry L. Longest II, Director Office of Emergency and Remedial/Revolutionse (OS-200)

TO: Addressees

Purpose

The purpose of this memo is to transmit an analysis of the effectiveness of treatment technologies for contaminated soil and debris in response to the recommendation in the Superfund Management Review to "carefully evaluate impact of RCRA land ban and other rules on use of alternative technologies." This analysis will provide support to Regional decisions to employ treatability variances for complying with the RCRA Land Disposal Restrictions as applicable or relevant and appropriate requirements for Superfund actions involving contaminated soil and debris.

Background

The Superfund Management Review acknowledged that Superfund response actions may not be able to meet treatment standards based on "best demonstrated available technology" (BDAT) under the Land Disposal Restrictions (LDR). This may limit the potential treatment technologies available for Superfund Cleanups, with technologies such as soil washing, stabilization, and biological treatment being precluded because they may not meet the highest level of performance required by LDRs. In contrast, the study encouraged the greater use of innovative technologies and urged the reduction of non-technical barriers, such as regulatory and policy constraints, that inhibit use of treatment technologies, while preserving the intent and spirit of applicable RCRA regulations.

OSWER program offices recognized the potential limitation (treatment technologies for Superfund actions and developed a process to use LDR treatability variances for soil and debris. Guidance was issued to the Regions in July 1989 through the Superfund LDR Guide 6A, "Obtaining a Soil and Debris Treatability Variance for Remedial Actions" (OSWER Directive 9347.3-06FS). Superfund LDR Guide 6B, "Obtaining a Soil and Debris Treatability Variance for Removal Actions," is scheduled to be issued in December 1989. These guides describe the treatability variance process, include alternate treatment levels to be obtained under treatability variances, and identify treatment technologies which have achieved the recommended levels. OSWER recognizes that the use of treatability variances represents an interim approach and is currently in the process of acquiring additional data for developing a regulation on treatment standards for contaminated soil and debris.

The following analysis summarizes the effectiveness of treatment technologies applied to soils and other environmental wastes and provides support for decisions by the Regions to use treatability variances, when appropriate. The analysis identifies some of the key technical considerations to be evaluated in obtaining a treatability variance when there is a reasonable doubt that a technology operated at full scale cannot consistently meet the BDAT treatment standards for the soil and debris to be treated.

Analysis of Treatment Effectiveness

An extensive effort was undertaken during 1987 and 1988 to collect existing data on treatment of soil, sludge, debris, and related environmental media. The results from several hundred studies were collected and reviewed. All applicable treatment information from 67 studies was extracted, loaded into a data base, and analyzed to determine the effectiveness of technologies to treat different chemical groups (Summary of Treatment Technology Effectiveness for Contaminated Soil, U.S. EPA, EPA/540/2-89/053).

Although some of the data on which the analysis is based have limited quality assurance information, the data, nevertheless, do indicate potential effectiveness (at least 90% to 99% reduction of concentration or mobility of hazardous constituents) of treatment technologies to treat Superfund wastes. Some reductions in organic concentrations or organic mobility of more volatile compounds may actually represent the removal of those compounds as a direct result of volatilization during dechlorination, bioremediation, soil washing, or immobilization, which requires consideration of appropriate emission controls. Percentage removal reductions are not always a good measure of effectiveness, especially when high concentrations remain in the residuals. Some of the performanc summarized below is based upon a relatively small number of data points and may not extrapolate well to the broad array of soils requiring treatment.

Based on this analysis, a number of technologies commonly used in the Superfund program provide substantial reduction in mobility and toxicity of wastes as required in Section 121 of the Superfund Reauthorization and Amendments Act of 1986. For example:

- Thermal destruction has been proven effective on all organics compounds, usually accomplishing well over 99% reduction of organics.
- Although the data indicate that PCBs, dioxins, furans, and other aromatic compounds have been dechlorinated to approximately 80%, more recent data indicating that removal efficiencies may approach 99.9%.
- Bioremediation successfully treats many halogenated aliphatic compounds, non-halogenated aromatics, heterocyclics, and other polar compounds with reduction efficiencies in excess of 99%.
- Removal efficiencies for low temperature thermal desorption have been demonstrated with averages up to 99% for non-polar halogenated aromatics and with treatment often exceeding 90% for other polar organics.
- Soil washing data on organic compounds indicate average removal efficiencies of approximately 90% for polar nonhalogenated organics and 99% for halogenated aromatics, with treatment often exceeding 90% for polynuclear aromatics. The chemical extraction process, with optimized solvent selection, has demonstrated removal efficiencies often exceeding 90% for volatile and nonvolatile metals.
- Immobilization processes, while not actually destroying the organic compounds, reduce the mobility of contaminants an average of 99% for polynuclear aromatic compounds. Immobilization may not effectively stabilize some organic compounds, such as volatile organics, and the long-term effectiveness of immobilization of organics is under evaluation. Immobilization can achieve average reductions in mobility of 93% for volatile metals, with reductions in mobility often exceeding 90% for non-volatile metals.

The attachment contains a more detailed summary of the data which is extracted from the "Summary of Treatment Technology Effectiveness for Contaminated Soil."

Technology Limitations to be Considered

The data available suggest that treatment of soil and debris with organic contamination by technologies other than thermal destruction will not be able to consistently achieve BDAT standards. Therefore, other technologies should be used for those wastes, only if approved under a treatability variance.

The residual concentrations in contaminated soil treated by technologies other than thermal destruction is highly dependent upon the concentrations in the untreated soil. Therefore, when evaluating technologies other than thermal destruction, the ability of those technologies to treat high concentrations of organics should be considered.

The Regions need to carefully review the site conditions and characteristics in designing and operating materials handling, pretreatment, and treatment requirements. High variability in contaminant concentrations of untreated soil may have an adverse effect on the ability to achieve treatment levels at higher concentrations using technologies other than thermal destruction. Consideration should be given to the need for blending wastes.

In selecting technologies for contaminated soils and sludges, the number and types of contaminants must be carefully screened, and, in some cases, different technologies may be necessary for soils and sludges.

Implementation

The data indicate potential limitations of treatment technologies to meet BDAT standards for Superfund wastes. Superfund LDR Guide 6A outlines the treatability variance process for Superfund soil and debris and identifies alternate treatment levels. Guide 6A should be followed, when appropriate, until OSWER completes a regulation with treatment standards for contaminated soil and debris. The limitations on technologies identified in this memorandum should be taken into account when evaluating, selecting, designing, and implementing Superfund response actions.

Attachment

Addressees:

Sylvia Lowrance, Director Office of Solid Waste

Bruce Diamond, Director Office of Waste Programs Enforcement

Directors, Waste Management Division Regions I, IV, V, VII, VIII Director, Emergency and Remedial Response Division Region II Directors, Hazardous Waste Management Division Regions III, VI Director, Toxic and Waste Management Division Region IX Director, Hazardous Waste Division .. Region X Attachment

TECHNOLOGY CONCLUSIONS ON SOIL TREATMENT

Extracted from "Summary of Treatment Technology Effectiveness for Contaminated Soil" EPA /540/2-89/053

For each treatability group, the effectiveness of various technologies was evaluated and is summarized in Figure 1. The following ratings were used:.

- o Demonstrated Effectiveness: A significant percentage of the data, 20%, are from pilot or full scale operations, the average removal efficiency for all of the data exceeds 90%, and there are at least ten data pairs.
- Potential Effectiveness: The average removal efficiency for all of the data exceeds 70%.
- No Expected Effectiveness: The average removal efficiency for all of the data is less than 70% and no interference is expected to this process as a result of this group.

No Expected Effectiveness: Fotential adverse effects to the environment or the treatment process may occur. For example, high concentrations of metals may interfere with biological treatment.

In some cases, a different rating was selected when additional qualitative information and engineering judgment warranted.

Two symbols were used if the compounds within a treatability group were so variable that a range of conclusions could be drawn for a particular technology.

TECHNOLOGY CONCLUSIONS ON SOIL TREATMENT

Extracted from "Summary of Treatment Technology Effectiveness for Contaminated Soil" EPA /540/2-89/053

Thermal Destruction (See Figure 2)

Principle of Operation

- o Thermal destruction uses high temperatures to incinerate and destroy hazardous vastes, usually by converting the contaminants to carbon
- dioxide, water, and other combustion products in the presence of oxygen.
- Effectiveness on Organics
 - This technology has been proven effective on all organic compounds, usually accomplishing well over 99% removal.
 - Thermal destruction technologies are equally effective on halogenated, non-halogenated, nitrated, aliphatic, aromatic, and polynuclear compounds.
 - o Incineration of nitrated compounds such as trinitrotoluene (TNT) may generate large quantities of nitrous oxides.

Effectiveness on Inorganics

- o Thermal destruction is not an effective technology for treating soils contaminated with high concentrations of some metals.
- o High concentrations of volatile metal compounds (lead) present a significant emissions problem, which cannot be effectively contained by conventional scrubbers or electrostatic precipitators due to the small particle size of metal-containing particulates.
- o Non-volatile metals (copper) tend to remain in the soil when exposed to thermal destruction; however, they may slag and foul the equipment

Dechlorination (See Figure 3)

Principle of Operation

o Dechlorination is a destruction process that uses a chemical reaction to replace chlorine atoms in the chlorinated aromatic molecules with an ether or hydroxyl group. This reaction converts the more toxic compounds into less toxic, more water-soluble products. The transformation of contaminants within the soil produces compounds that are more readily removed from the soil. An evaluation of the end products is necessary to determine whether further treatment is required.

Effectiveness on Organics

- PCBs, dioxins; furans, and other aromatic compounds (such as pentachlorophenol) have been dechlorinated to approximately 80% removal, with more recent data indicating that removal efficiencies may approach 99.9%.
- Other limited laboratory data suggest potential applicability to other halogenated compounds including straight-chain aliphatics (such as 1,2-dichloroethane). The removal indicated by the data may be due in part to volatilization.
- Although no data were available for halogenated cyclic aliphatics (such as dieldrin), it is expected that dechlorination will be effective on these compounds as well.
- When non-halogenated compounds are subjected to this process, volatilization may occur.

Effectiveness on Inorganics

o Dechlorination is not effective on metals, and high concentrations of reactive metals (such as aluminum), under very alkaline conditions, hinder the dechlorination process.

Bioremediation (See Figure 4)

Principle of Operation

 Bioremediation is a destruction process that uses soil microorganisms including bacteria, fungi, and yeasts to chemically degrade organic contaminants.

Effectiveness on Organics

- Bioremediation appears to successfully treat many halogenated aliphatic compounds (1,1-dichloroethane), non-halogenated aromatics (benzene), heterocyclics (pyridine), and other polar compounds (phenol) with removal efficiencies in excess of 992; however, the high removal implied by the available data may be a result of volatilization in addition to bioremediation.
- o Nore complex halogenated (4-4'DDT), nitrated (triazine), and polynuclear aromatic (phenanthrene) compounds exhibited lover removal efficiencies, ranging from approximately 50% to 87%.

o Poly-halogenated compounds may be toxic to many microorganisms.

Effectiveness on Inorganics

- o Bioremediation is not effective on metals.
- e Metal salts may be inhibitory or toxic to many microorganisms.

Low Temperature Thermal Description (See Figure 5)

Principle of Operation

o Low temperature thermal desorption is a physical transfer process that uses air, heat, and/or mechanical agitation to volatilize contaminants into a gas stream, where the contaminants are then subjected to further treatment. The degree of volatility of the compound rather than the type of substituted group is the limiting factor in this process.

Effectiveness on Organics

o Removal efficiencies have been demonstrated by these units at bench, pilot, and full scales, ranging from approximately 65% for polynuclear aromatics (naphthalene), to 82% for other polar organics (acetone) and 99% for non-polar halogenated aromatics (chlorobenzene).

Effectiveness on Inorganics

- o Low temperature thermal desorption is not effective on metals.
- o Only mercury has the potential to be volatilized at the operating temperatures of this technology.

Chemical Extraction and Soil Vashing (See Figure 6)

Principle of Operation

o Chemical extraction and soil vashing are physical transfer processes in which contaminants are disassociated from the soil, becoming dissolved or suspended in a liquid solvent. This liquid vaste stream then undergoes subsequent treatment to remove the contaminants and the solvent is recycled, if possible.

- o Soil vashing uses water as the solvent to separate the clay particles, which contain the majority of the contaminants, from the sand fraction.
- Chemical extraction processes use a solvent which separates the contaminants from the soil particles and dissolves the contaminant i the solvent.

Effectiveness on Organics

- o The majority of the available soil vashing data on organic compounds indicates removal efficiencies of approximately 90% for polar non-halogenated organics (phenol) to 99% for halogenated aromatics (chlorobenzene). with lower values of approximately 71% for PCBs to
- 82% for polynuclear aromatics (anthracene).
- o The reported effectiveness for these compounds could be due in part to volatilization for compounds with higher vapor pressures (such as acetone).
- o This process is least effective for some of the less volatile and less water soluble aromatic compounds.

Effectiveness on Inorganics

o The chemical extraction process, with optimized solvent selection, has demonstrated removal efficiencies of 85% to 89% for volatile metals (lead) and non-volatile metals (copper), respectively.

Ismobilization (See Figure 7)

Principle of Operation

- o Immobilization processes reduce the mobility of contaminants by stabilizing them within the soil matrix, without causing significant contaminant destruction or transfer to another medium.
- Volatile organics will often volatilize during treatment, therefore a effort should be made to drive off these compounds in conjunction wit an emission control system.

Effectiveness on Organics

- Reductions in mobility for organics range from 61% for halogenated phenols (pentachlorophenol) to 99% for polynuclear aromatic compounds (anthracene).
- o Immobilization is also effective (84% reduction) on halogenated aliphatics (1,2-dichloroethane).
- o Some organic mobility reductions of more volatile compounds may actually be removals as a direct result of volatilization during the exothermic mixing process and throughout the curing period.
- o The immobilization of organics is currently under investigation, including an evaluation of the applicability of analytical protocols (EP, TCLP, total analysis) for predicting long-term effectiveness of immobilization of organics. The preliminary available data indicate that significant bonding takes place between some organic contaminants and certain organophilic species in the binding matrix; however, immobilization may not effectively stabilize some organic compounds, such as volatile organics.

Effectiveness on Inorganics

o Immobilization can accomplish reductions in mobility of 81% for non-volatile metals (nickel) to 93% for volatile metals (lead).

FIGURE 1: PREDICTED TREATMENT EFFECTIVENESS FOR C \MINATED SOIL

TECHNICLOGY	THERMAL DESTRUCTION	DECHLORIKATION	BIOREME DIA TION	LOW TEMPERATURE THERMAL DESCRIPTION	CHEMICAL EXTRACTION AND BOIL WASHING	
NON-POLAR HALOGENATED AROMATICE (MDI)	•	•	ο,	• •	•	•
PCB, MALODE MATED DIOXING, FURANS, AND THEIR PRECURSORD (MGR)	•	•	θ,	0'	•	Q ¹
HALOGENATED PHENOLS, CRESOLS, MINES, THOLS, AND OTHER POLAR ARCHATICS (MCB)	• 3	•	•	•	•	θ,
HALODENATED ALIPHATIC COMPOLINDI (MOI)	•	0,	•	•	•	Q ²
HALOGENATED CYCLIC ALIPHATICS, ETHERS, ESTERS, AND KETCHES (MOS)	•	θ'	•	0'	۰'	Q ¹
NITRATED COMPOLINDS (NOS)	•	0'	•	· o'	•	e'
HE TEROCYCLICS AND SIMPLE NON HALOGENATED AROMATICS (M07)	٠	0²	0 ²	· ●	•	Q² ·
POL YMLICLE AR ARCMATICS (MOR)	•	0²	•	Ο	Θ.	•
OTHER POLAR NON HALODENATED ORGANIC COMPOLINDS (MOI)	•	O ²	O 2	•	Q ·	Θ²
NON-VOLATEE METALS (W10)	0'	0'	0 % '	0'	. •	• 1
' VOLATILE METALB (W11)	X ¹	0'	0 x ¹	0'	θ	•

11/29/00

Demonstrated Effectiveness

- Potential Effectiveness
- O No Expected Effectiveness (no expected interference to process)
- X No Expected Effectiveness (potential adverse effects to environment or process)
- 1 Data were not evaluable for this treatability group. Conclusions are drewn from data for compounds with similar physical and chemical charactenstics.
- 2 High removel elliciencies implied by the data may be due to volatilization or soil washing
- 3 The predicted effectiveness may be different than the data imply, due to limitations in the jest conditions.
- 4 These technologies may have limited annihrability to both levels of recentre

FIGURE 2: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY THERMAL DESTRUCTION

	MANDER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (pm) AND % REMOVALS	GÉ MERAL OBSERVATIONS
MON POLAR HALOGENATED AROMATICS (MOI)	22 PAIRS % NENCH N PLOT N FULL	AVERADE AVERADE CONCENTRATIONS REMOVAL OPTIG EFFICIENCY NOTLIENT <u>900</u> 200 %	 This technology works vary well at optimum operating conditions on a variety of initial concentrations. Brominated compounds will inhibit flame propagation. High levels of acid gases produced in the presence of oxygen will attack the refractory walls and exposed metal surfaces.
PCB., HALOCEHATED DICKINS, FURANS, AND THEIR PRECUBCING (MOZ)	% DENCH % DENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SPINS EFFICENCY SUFLUENT <u>1,100</u> <u>200</u> % EFFILIENT <u>0.005</u>	 This technology works very well at optimum operating conditions on a variety of initial concentrations. High levels of acid gases produced in the presence of oxygen will attack the refractory wells and exposed metal surfaces.
HAL CIGENATED PHENCLB, CREBCLB, AMPEB, THICLS, AND OTHER POLAR ARCMATICS (WOD)	B1 PAIRS % BENCH % PLOT % FUAL	AVE RAQE CONCENTRATIONS BUTO DIFLUENT 550 06 % EFFLUENT 070	 This technology works well at optimum operating conditions on a variety of initial concentrations. Oxides of nitrogen and suffur can create potential serious cross media impacts if not removed from gas emissions. High concentrations of acid gases produced in the presence of oxygen will attack the refractory wells and exposed metal surfaces.
HALOGENATED ALPHATIC CONFOLMOS (W64)	02 PAIR0 % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL STORE BUTLIENT 41 200 % EFFLUENT 0.016	 This technology works well at optimum operating conditions on a variety of initial concentrations. If this is the only treatability: group present, low temperature thermal description may be more cost effective. High levels of acid gases produced in the presence of oxygen will attack the retractory waits and exposed metal surfaces.
HALOGENATED GYOL IC ALIPHATICA, ETHERI, ESTERI, AMD KETOMES (WOR)	<u>118</u> PARE <u>67</u> % BENCH <u>30</u> % PLOT <u>0</u> % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BUPILENT	• This technology works well at optimum operating conditions on a variety of initial concentrations.
NITRATED COMPOLINDS (NOS)	<u>142</u> PAIRS <u>73</u> % BENCH <u>27</u> % PEOT <u>0</u> % FULL	AVE RAGE CONCENTRATIONS SUPITI EFFICIENCY INFLUENT 90,000 99 % EFFLIENT 200	 This technology works well at optimum operating conditions on a variety of initial concentrations. High amounts of nitrous gases may be released into the atmosphere if not controlled by a nitrous oxide burner.

FIGURE 2: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY THERMAL DESTRUCTION (CONT.)

TREATABLITY GROUP	MANIBER AND BCALE OF AVAL ABLE DATA	AVERAGE CONCENTRATIONS (PPH) AND % RENOVALS	QENERAL OBBERVATIONS
HETEROCYCIICS AND SIMPLE NON HALOOENATED AROMATICS (W07)	12 PAIRS 12 BENCH 12 PROT 12 FUSS	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (1997) EFFICENCY INFLLENT 740 300 %	 This technology works very well at optimum operating conditions on a variety of initial concentrations. Low temperature thermal desorption may be more cost effective.
POLYMJCIEAR ARIXMATICS (W08)	_ 24 PARB _ 33 % BENCH _ 50 % PROT _ 8 % FIAL	AVE RAGE AVERAGE CONCENTRATIONS REMOVAL (ypm) EFFICIENCY INFLUENT <u>1,000</u> <u>>00</u> % EFFLUENT <u>0.32</u>	This technology works very well at optimum operating conditions on a variety of initial concentrations.
OTHER POLAR NON HALOGENATED ORQANC COMPOUNDS (W00)	35 % BENCH 35 % BENCH 65 % PLOT 0 % FUIL	AVERAGE AVERACIF CONCENTRATIONS RENDVA igym) EFFICENCY INFLUENT <u>690</u> <u>98</u> %	 This technology works well at optimum operating conditions on a variety of initial concentrations.
NON VOLATE METALE (N) (0)	Q PARES Q % BENCH Q % PILOT Q % FLAL	AVERADE AVERADE CONCENTRATIONS REMOVAL UPM) EFFICIENCY INFLUENT00 %	 The physical and/or chemical characteristics of the constituents of this treatability group indicate that this technology would <u>not</u> be effective. Pyrolysis and intrared thermal destruction of wastee with metal concentrations over 500 ppm may possibly reduce the mobility of these metals by binding the metals into the solid residue.
VOLATRE METALE (W11)	0 % BENCH 0 % BENCH 0 % PLOT 0 % FULL	AVERADE AVERADE CONCENTRATIONS REMOVAL (ppm) EFFICIENCY INFLUENT O N EFFLUENTO	 This technology is not recommended if the waste contains high concentrations of volatile metals, due to potential volatilization of these metals with subsequent cross media impacts. Pyrolysis and intrared thermal destruction may reduce the mobility of these metals by binding the metals into the solid residue.

FIGURE 3: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY DECHLORINATION

	MEMBER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (1991) AND % REMOVALS	GENERAL COSERVATIONS
NON POLAR HALOGENATED ARCMATICS (NO1)	PAIRS 0 % BENCH 0 % PLOT 0 % FLEL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SPM) EFFICIENCY DIFLUENT 190 96 %	• Data ware for chlorobenzene only. These data suggest that this technology is potentially effective in certain situations
PCBN, HALOGENATED DIDHING, FURANS, AND THEIR PRE CURSORIS (W02)	21 PAIRS % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Sprt) EFFICENCY INFLUENT 180 83 %	 This technology is potentially effective, especially for sandy soils. Data on sludges show better removal due to more uniform distribution of contaminants and better reagent contact. Lower initial concentrations give lower removal efficiencies. Moisture content over 4 to 7% deactivates the NePEG reagent. Particle size and soil matrix allect reagent penetration and process effectiveness. Recent data indicate that greater than 99% of PCBs and lurans can be destroyed (des Roslers, 1988).
HALOGENATED Fre HOLS, CRESCLS, AMINES, THIOLS, AND OTHER POLAR AROMATICS (MOD)	®PAIRS % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BOTS EFFICIENCY DIFLUENT <u>98</u> <u>95</u> %	 Data were for pentachlorophenol only These data suggest that this technology is potentially effective in certain situations. Recent data indicate that greater than 99% of contaminants can be destroyed (des Rosiers, 1968).
HALOGENATED ALIPHATIC COMPOUNDS (WD4)	<u>16</u> PAIRS <u>100</u> % BENCH <u>0</u> % PLOT <u>0</u> % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SPM) EFFICIENCY DIFLUENT 044	 These data suggest that this technology is potentially effective in certain situations. Some halogenated allphatics react with the APEG reagents to form explosive compounds, especially in the presence of heavy metals. The potential for this to occur should be evaluated in the laboratory before dechtorination treatment is selected. The high removal efficiency may be the result of volatilization or the APEG process acting as a soil washing process.
HALOGENATED CVCLIC ALIPHATICS, ETHERS, ESTERS, AMD KETCINES (NOL)	9 PAIRS 9 % BENCH 9 % PLOT 9 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BOPMS EFFICIENCY BIFLUENTOO %	 Data were not evaluable for this treatability group. Data for compounds with similar physical and chemical characteristics suggest that this technology is potentially effective in certain situations. Treatability studies will be needed to confirm the technology's effectiveness.
MITRATED COMPOLINDS (NOS)	QPAIRS Q % BENCH Q % PEOT Q % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL OPTS EFFICIENCY INFLUENT 0 0 0 %	 Data were not available available for this treatability group. The physical and/or chemical characteristics of the constituents of this treatability group indicate that this technology would <u>not</u> be effective.

FIGURE 3: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY DECHLORINATION (CONT.)

	NUMBER AND BCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (ppm) AND % REMOVALS	GENERAL OB SERVATIONS
IE IEROCYCLICS AND SIMPLE NOM HALCOE NATED AROMATICS (NO7)	24 FARB 100 % BENCH 10 % PROT % FIRE	AVERAGE AVERAGE CONCENTINATIONS REMOVAL (APPN) EFFICENCY NOFILENT	 The physical and/or chamical characteristics of the constituents of this treatability group suggest that this technology would not be attactive. The high removal elificiency may be the result of volatilization or the APEG process acting as a soil washing process.
POLYNLICI EAR ARIONATICE (WOS)	0 PARS 0 % BENCH 0 % PLOT 0 % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (VVV) EFFICIENCY INFLUENT <u>3.000</u> <u>01</u> %	 The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would <u>not</u> be effective. The high removal efficiency may be the result of volatilization or the APEQ process acting as a soil washing process.
OIHER POLAR NON HALDOENAIED ORDANC COMPOLNOS (WOI)	16 PAIRS 100 % BENCH 0 % PROF 0 % FUL	AVERAGE AVERAGE CONCENTRATIONS REMOVA UPPA) EFFICIENCY INFLUENT 1,700 06 % EFFLUENT 30	 The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would <u>not</u> be affective. The high removal efficiency may be the result of volstilization or the APEQ process acting as a soil washing process.
NON VOLATRE METALB (W10)	0 PAIRS 0 % BENCH 0 % PROT 0 % FURL	AVERAGE AVERAGE CONCENTRATIONS RELIOVAL SPOT EFFICENCY DELLENTO	 The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would <u>not</u> be effective.
VCI ATRE METALA (W11)	PAIRE % BENCH % PR.DT % FURL	AVERADE AVERADE CONCENTRATIONS REMOVAL Upm) EFFICIENCY INFLUENTOO &	 The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would <u>not</u> be effective.

FIGURE 4: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY BIOREMEDIATION

	NUMBER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (rpm) AND % REMOVALS	GENERAL GROLAVATIONS
, NON POLAR HALOCIENATED ARCMATICS (WG1)	66 PAIRS 16 BENCH 16 N. PLOT 16 FULL	AVERAGE AVERAGE CONCENTRATIONS NEMOVAL OPPY) EFFICENCY OFLUENT	 This technology is not effective for all contaminants in this class; however, there is potential for effectiveness for low initial concentrations with further development. The presence of these contaminants at low concentrations is not expected to interfere with the treatment of applicable wastes. The effectiveness of this technology may be different than the data imply, because the initial concentrations in these tests were so low.
PCBL HALCIGENATED DICHINS, FURMIS, AND THEIR PRECURSORS (MD)] PAIRS 0 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BOPMS EFFICENCY REFLUENT 0 12	 The lone data pair is PCBs. Ongoing research suggests that this technology may be potentially effective for this group.
HAI COENATED PHENICLE, CRESCLE, AMID ES, THICLE, AND OTHER POLAR AROMATICE (MOD)	3 PAIRS % BENCH % PLOT % FIAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Sprit EFFICIENCY DELLENT	 This technology is potentially effective for low initial concentrations. Bioremediaton requires uniformly mixed media with small particle sizes. Toxic compounds such as cyanides, arsenic, heavy metals, and some organics adversely affect the treatment. Preprocessing includes mixing and nutrient and organism addition. Bioremediation is a slow process. Bioremediation has low costs relative to other technologies.
HALOGENATED ALPHATIC COMPOLINOS (MOI)	<u>27</u> PAIRS 0 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL OPPO EFFICIENCY DELLENT 0027	 This technology is potentially effective for low initial concentrations. Bioremediation requires uniformly mixed media with small particle sizes. Toxic compounds such as cyanides, arsenic, heavy metals, and some organics adversely affect the treatment. Preprocessing includes mixing and nutrient and organism addition. Bioremediation has low process. Bioremediation has low costs relative to other technologies. Removal may actually represent volatilization during preprocessing and treatment.
HALOGENATED CYCLIC ALIPHATICS, ETHERS, ESTERS, AND KETCHES (MOS)	0 % BENCH 0 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Oprit EFFICENCY DIFLIENT 0 0 %	 Data were not available for this treatability group. Data for compounds with similar physical and chemical characteristics suggest that this technology may be potentially effective in certain situations with low initial concentrations.
NITRATED COMPOLINOS (WOS)	<u>22</u> PAIRS <u>0</u> % BENCH 100 % PLOT <u>0</u> % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (ppm) EFFICIENCY INFLUENT 13 000 62 % EFFLUENT	 This technology is potentially effective on these contaminants, especially at low concentrations. Some of the available data for this treatability group were based on very high initial concentrations; however consideration should be given to the ability of the technology to treat high initial concentrations. Bioremediation requires uniformly mixed media with small particle sizes. Toxic compounds such as cyanides, arsenic, heavy metals, and some organics adversely affect the stand. Stand. /ul>

FIGURE 4: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY BIOREMEDIATION (CONT.)

	MANDER AND BCALS OF AVALABLE DATA	AVERAGE CONCENTRATE Y'S (ppm) AND & REMOVALS	CIE ME RAL COOLERVATIONS
HE TEROCYCLICS AND SIMPLE NON HALCOENATED AROMATICS (W07)	0 % BENCH 0 % BENCH 0 % PLOT 0 % FLLL	AVEHAGE AVERAGE CONCENTRATIONS REMOVAL Upped EFFICIENCY INFLUENT 220 300 % EFFLUENT 0.025	 This technology is potentially effective for low initial concentrations. Bioremediation requires uniformly mixed media with small particle sizes Toxic compounds such as cyanidus, arsenic, heavy metals, and some organic compounds adversely effect treatment Preprocessing includes mixing and nutrient and organism addition. Bioremediation has low processa Bioremediation has tow costs relative to other technologies. Removal may actually represent volatilization during preprocessing and treatment.
POL VNJCLEAR ARKMATICE (WOS)	<u>37 PARE</u> <u>10 % BENCH</u> <u>01 % PLOT</u> <u>0 % FIAL</u>	AVERAGE AVERAGE CONCENTRATIONS REMOVAL GOM EFFICIENCY INFLUENT 120 07 %	 This technology is potentially effective for low initial concentrations. Bioremediation requires uniformly mixed media with small particle sizes. Toxic compounds such as cyanides, arsenic, heavy metals, and some organic compounds adversely affect treatment. Preprocessing includes mixing and nutrient and organism addition. Bioremediation is a slow process. Eloremediation has low costs relative to other technologies.
DTHER POLAR NON HALODENATED ORGANG COMPOLINDS (WOS)	22 PAIRS 0 % BENCH 100 % PLOT 0 % FLAL	AVERAGE AVERAGE CONCENTRATIONS RELIQVAL (IV)MI EFFICENCY INFLUENT <u>012</u>	 This technology is potentially affective for low initial concentrations. Bioremediation requires uniformly mixed media with small particle sizes. Toxic compounds such as cyanides, arsenic, heavy metals, and some organic compounds adversely allect treatment. Preprocessing includes mixing and nutrient and organism addition. Bioremediation has low costs relative to other technologies. Removal may actually represent volatilization during preprocessing and treatment.
NON-VOLATILE METALB (W10)	0 PAIRS 0 % BENCH 0 % PROT 0 % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (OPP) EFFICIENCY INFLUENT 0 %	 High concentrations of heavy metals may adversely affect particular organisms. The physical and/or chemical characteristics of the constituents of this treatability group suggest that the technology would not be effective.
VOLATRE METALB (Writ)	0 PAIRS 0 % BENCH 0 % PLOT 0 % FUIL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (ppm) EFFICIENCY INFLUENT 0 0 % EFFLUENT0	 High concentrations of heavy metals may adversely affect particular organisms. The physical and/or chemical characteristics of the constituents of this treatability group suggest that the techny logy would not be effective.

FIGURE 5: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY LOW TEMPERATURE THERMAL DESORPTION

TREATA BILITY GROUP	MANDER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (ppm) ANG % RE MOVALS	GENERAL OBSERVATIONS
NON-POLAN HALOGENATED AROMATICS (NOI)	20 PAIRS 40 % BENCH 4 % PLOT 4 % FULL	AVERAGE AVERAGE CONCENTRATIONS RELICVAL SPOTO EFFICENCY INFLUENT 130 90 % EFFLUENT 007	 Although this technology was not expected to perform well on this treatability group, the data from studies which utilized higher operating temperatures and longer residence times indicate that many of the compounds in this group may be treated by this technology with potential effectiveness. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled. This technology has demonstrated effectiveness on some of the more volatile contaminants in this group, and it is potentially effective on the remaining contaminants.
PCRI, HALOGENATED DIDIRIS, FURANS, AND THE IR PRECUSIORS (WID)	PAIRS % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BOMI EFFLENCY DIFLIENT 0 0 8	 No data were available The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would not be effective. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
HAL COENATED PHENCLE, CRESCLS, AMD 65, THICLE, AND OTHER POLAR AROMATICS (WOD)	14 PAIRS % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL BOM EFFICENCY DIFLIENT 250 72 %	 Although the data suggest that this technology is not as effective with this treatability group, the technology, if operated at higher temperatures and tesidence times, may successfully treat many of the compounds in this group. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
HALOGENATED ALPHATIC COMPOUNDS (WOI)	<u>132</u> PAIRS <u>27</u> % BENCH <u>50</u> % PLOT <u>23</u> % FULL	AVE RAQE CONCENTRATIONS BUTUENT BOD B4 BUFLUENT 10	 This technology works well on this treatability group. Removal efficiencies are not as high with soils having extremely elevated concentrations. A longer residence time may remedy this situation. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
HALOGENATED CYCLIC ALIPHATICS, ETHERS, ESTERS, AND KETONES (MOS)	@ PAIRS 0 % GENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SUPID EFFICIENCY INFLUENTOO %	 No data were available for this treatability group. The physical and/or chemical characteristics of the constituents of this treatability group suggest that this technology would not be effective. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
NITRATED COMPOUNDS (NOI)	QPAIRS Q%BENCH Q%PLOT Q%FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL UPPT) EFFICIENCY INFLUENT 0 0 0 1	 No data were available for this treatability group. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.

FIGURE 5: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY LOW TEMPERATURE THERMAL DESORPTION (CONT.)

THE ATABLITY GROUP	MANDER AND BCALS OF AVAL ABLE DATA	AVE RAGE CONCENTRATIONS (ppm) AND % REMOVALE	QE HERAL CONSTANTIONS
HETEROCYCLICS AND SIMPLE HON HALCOENATED AROMATICS (W67)	111 PAIRS 37 WBLINCH 30 WPROT 24 WFIAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Uppe) EFFICIENCY INFLUENT <u>B20</u> <u>D6</u> % EFFLUENT <u>17</u>	 This technology works well on this treatability group. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled
POL WAICLEAR ARDMATICE (Wee)	<u>62</u> PAIRS <u>27</u> % SENCH <u>60</u> % PAOT <u>13</u> % FIAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Upmi EFFICENCY DEFLUENT 1,400 65 %	 This technology is not generally effective as a treatment for this group, but individual compounds may be treated effectively at higher operating temperatures and longer residence times. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
OTIER POLAR NON IVI ODEHATED ORGANG COMPOLNDS (W96)	24 PAIHS 62 % BENCH 6 % PROT 12 % FUEL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (upm) EFFICIENCY INFLUENT 1.000 02 10 EFFLUENT 170	 This technology is potentially effective on some contaminants in this group. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
NOH VOLATILE Lietale (W10)	PARS 9 % BENCH 9 % PK OT 9 % FLLL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (ppm) EFFICENCY DEFILENT 0 0 % EFFILENT0	 The physical and/or chemical characteristics of the constituents of this treatability group indicate that this technology would <u>not</u> be effective. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.
VOLATRE METALE (W11)	Q PARB Q % BENCH Q % PLOT Q % FUAL	AVERAGE AVERAGE CONCENTRATIONS AVERAGE (OPPN) EFFICIENCY INFLUENT 0 % EFFLUENT0	 The physical and/or chemical characteristics of the constituents of this treatability group indicate that this technology would not be effective. This technology is not recommended for the treatment of waste mixtures which contain high concentrations of metallic and/or organic forms of mercury, unless emissions are controlled.

FIGURE 6: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY CHEMICAL EXTRACTION AND SOIL WASHING

TREATABILITY GROUP	GRANDER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRATIONS (ppm) AND % REMOVALS	GE HE RAL CHISER VATIONS
NCN POLAR HALOGENATED AROMATICS (W01)	20 PAIRS 100 % BENCH % PLOT % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SPMI EFFICIENCY DELLIENT 170 >00 %	 This technology is potentially effective on these contaminants but all data are from bench scale. Surfactants may adhere to the soil and reduce soil permeability. Possible volatile emission losses may occur during treatment.
PCB, HALOGENATED DIOXINS, FURANS, AND THE IR PRECURSORS (M02)	PAIRS % BENCH % PLOT % FULL	AVERACIE AVERACE CONCENTRATIONS REMOVAL SPINI EFFICIENCY DELLIENT <u>9900</u> 71 % EFFLUENT <u>4,000</u>	 This technology is potentially effective on these contaminants with further development. Some of the available data for this treatability group were based on very high initial concentrations; however consideration should be given to the ability of the technology to treat high initial concentrations. The presence of oil in the matrix enhances removal. The removal efficiency decreases as the percent of clays and clayey sits increases. Surfactants may adhere to the soil and reduce soil permeability.
HAL OGENATED PHENCLS, CRESCLS, AMPRES, THICLS, AND OTHER POLAR AROMATICS (WICH)	20 PAIRS 100 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE . CONCENTRATIONS REMOVAL Sport EFFICENCY BUFLUENT <u>87</u> <u>72</u> %	 Data were from pentachlorophenol only. This technology is potentially effective on these contaminants, especially for treating sandy soils. Surfactants may adhere to the soil and reduce soil permeability.
HALOGENATED ALFHATIC COMPOLIN OS (MOI)	40 PAINS 100 % BENCH % PLOT % FULL	AVE RAGE AVE RAGE CONCENTRATIONS REMOVAL Spirit STFLENCY DIFLUENT 200 >00 % EFFLUENT 022	 This technology is potentially effective on these contaminants, but all data are from bench scale. This technology may be more applicable to sendy soils. Surfactants may adhere to the soil and reduce soil permeability. Volatile emissions may occur during treatment.
HALOGENATED CYCLIC AL PHATICS, ETHERS, ESTERS, AND KETONES (NOL)	Q PARS Q % BENCH Q % PLOT Q % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SUPPO EFFICIENCY DIFLUENT 0 0 %	 Data were not available for this treatability group. Data for compounds with similar 'physical and chemical characteristics suggest that this technology is potentially effective in certain situations. Surfactants may achieve to the soil and veduce soil permeability.
NITRATED COMPOLINDS (NOS)	<u> </u>	AVERAGE AVERAGE CONCENTRATIONS REMOVAL SPIMS EFFICIENCY DIFLUENT <u>6,000</u> >99 % EFFLUENT <u>47</u>	This technology is potentially effective on these contaminants. However, data are limited and testing was conditided at bench scale.

FIGURE 6: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY CHEMICAL EXTRACTION AND SOIL WASHING (CONT.)

	NUMBER AND SCALE OF AVAILABLE DATA	AVERAGE CONCENTRA TIONS (pm) AND % REMOVALS	GENERAL OBSERVATIONS
HE TEROCYCLICS AND SIMPLE NON HALOGENATED ARCHATICS (MO7)	<u>55</u> PAIRS <u>08</u> % BENCH <u>0</u> % PLOT <u>2</u> % FULL	AVE RAGE AVE RAGE CONCENTRATIONS REMOVAL (ppm) EFFECENCY DFLUENT 1,700 >99 % EFFLUENT 30	 This technology is potentially effective on these contaminants but nearly all data are from bench scale. Volable emissions may occur during treatment Surfactants may adhere to the soil and reduce soil permeability.
POL YNJCLEAR AROMATICS (WOB)	<u>24</u> PAIRS <u>71</u> % BENCH 0 % PLOT 20 % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (ppm) EFFICENCY DELUENT 1600 82 %	 This technology is potentially effective on these contaminants with further development. Some of the available data for this treatability group were based on very high initial concentrations; however, consuleration should be given to the ability of the technology to treat high initial concentrations. Surfactants may adhere to the solil and reduce solil permeability.
OTHER POLAR NON HAI OGENATED ORGANIC COMPOLINDS (MOR)	50 pains 6 % BENCH 6 % PLOT 6 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL UPPN EFFICENCY DELUENT 70,000BI & EFFLUENT 13,000	 This technology is potentially effective on these contaminants. Some of the available data for this treatability group were based on very high initial concentrations; however, consideration should be given to the ability of the technology to treat high initial concentrations. Treatment effectiveness should be evaluated on a case-by-case basis. Surfactants may adhere to the soil and reduce soil permeability. Volatile emissions may occur during treatment
NON VOLATLE METALS (WIQ)	34 PAIRS % BENCH % PLOT % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Upm) EFFICENCY OFLUENT	 This technology is potentially effective on these contaminants. Water and H₂SO₄ at a pH of 1.0 and a 3.1 moler ratio of EDTA at a pH of 12.0 can both achieve good levels of extraction. Iron (1-2%) may cause solvent regeneration problems.
VCLATRE METALS (W11)	% PAIRS % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS RELIVOVAL (GPM) EFFICIENCY DEFLUENT	 This technology is potentially effective on these contaminants, especially for sandy soils Slity and clayey soils are not as effectively treated. Arsenic may be difficult to extract due to low solubility.

FIGURE 7: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY IMMOBILIZATION

	MAREA AND BOALS OF AVALABLE DATA	AVERAGE CONCENTRATIONS (ppm) AND % REMOVALS	GENERAL GEOGRYATIONS
NON POLAR HALOGENATED AHOMATICE (W61)	4 PARS 100 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVI ITALE CONCENTRATIONS REMOVA Upped EFFICIENCY NELUENT <u>21</u> <u>83</u> % EFFLUENT <u>065</u>	 Data were for chlorobenzene anly. These data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is low. The treatment mechanism for the more volatile compounds may be volatilization as apposed to immobilization. All pollution control systems may be necessary to minimize cross media impacts from these volatile emissions. It is not recommended that this technology be selected if this is the only treatability group present.
PCBa, Hul Ogenated Dioxina, Furana, And The IR Precusiona (WGJ)	PAIRB % BENCH % PLQT % FLAL	AVERAGE AVERAGE CONCENTRATIONS REMOVA (opm) SFFICENCY BRUENT	 Incomplete quantitative date were available to avaluate treatment effectiveness. These quantitative data and additional qualitative information suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is low. It is not recommended that this technology be selected if this is the only treatability group present.
HAI OGENATED Prencis, crescis, Annes, throis, And other polar Ardmatics (Wgg)	4 PARS 100 % BENCH 0 % PLOT 0 % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL Upper) EFFICIENCY INFLUENT	 Data were from pentachlorophenol only. These data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is fow. The effectiveness of this technology on these contaminants may be different than the data imply, due to limitations in the test conditions. It is not recommended that this technology be selected if this is the only treatability group present.
	PARS % SENCH % PLOT % FULL	AVERADE AVERADE CONCENTANTIONS RELIOVAL UPPN EFFICIENCY DELUENT 11	 Though these data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is low. The reductions in mobility may be due to volatilization of the volatilis compounds during treatment. As politrion control systems may be necessary to minimize cross media impacts from these volatilis emissions. It is not recommended that this technology be selected if this is the only treatability group present.
HALOGENATED CYCLIC ALIPHATICS, ETHERB, ESTERB, AND KETONES (WOS)	% BENCH % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (UPIN) EFFICIENCY INFLUENTOO & EFFLUENTO	 Data were not available for this treatability group. Data for compounds with similar physical and chemical characteristics suggest that this technology is potentially effective in certain elusitons, particularly where the initial concentration is fow. It is not recommended that this technology be selected if this is the only treatability group present.
MIRATED COMPOLINDS (WOL)	PAIRE % BENCH % PLOT % FULL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (ppm) EFFICIENCY NFLUENT 0 0 1	 Data were not available for this treatability group. Data for compounds with similar physical and chemical characteristics suggest that this technology is potentially effective in certain situations, particularly where the initial concentrations are low.

4

FIGURE 7: FINAL CONCLUSIONS BY TREATMENT TECHNOLOGY IMMOBILIZATION (CONT.)

THE AT ABE IT Y GROUP		AVERAGE CONCENTRATIONS (ppm) AND % REMOVAL 8 *	- GEMERAL DESERVATIONS
HE TE ROCYCLICS AND SIMPLE NON HAL COENATED AROMATICS (W07)	12 PARS 100 % BENCH % PLOT % FILL	AVERAGE AVERAGE CONCENTRATIONS REMOVAL (00%) EFFICIENCY*	 Though these data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is tow, the reductions in mobility may be due to the volatilization of volatile organic compounds during treatment. Air pollution control systems may be necessary to minimize cross media impacts from these volatile emissions. It is not recommended that this technology be selected if this is the only treatability group present.
POLYNJELEAR AROMATICE (WOO)	2 PARS 100 % BENCH 0 % PLOT 0 % FLAL	AVERADE AVERADE CONCENTRATIONS PEMOVAL BOMS EFFICIENCY" OFFLUENT 20 90 %	 These limited data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is low
GIHER POLAR NON HALOGENATED ORDANC COMPOLINDS (W00)		AVERAGE AVERAGE CONCENTRATIONS REMOVAL BYTM) EFFICENCY*	 These limited data suggest that this technology is potentially effective in certain situations, particularly where the initial concentration is low. The treatment mechanism for the more volatile compounds may be volatilization as opposed to immobilization. Air pollution control systems may be necessary to minimize cross media impacts from these volatile emissions. It is not recommended that this technology be selected if this is the only treatability group present.
NON VOLATILE METALB (NTO)	24 PARS 57 % BENCH 33 % PLOT 0 % FLEL	AVEHADE CONCENTRATIONS GOMO EFFICIENCY NELLENT 034	 This technology works well on these contaminants. High levels of oB and grease may interfere with the process. Soluble salts of Mg, Sb, Zn, Cu, and Pb may interfere with the pozzolan reaction. High levels of suffates may interfere with the process. Pretreatment may be required to increase pit
VOLATUE METALO (W11)	% PAIRS % BENCH % PLOT % FLAL	AVERAGE CONCENTRATIONS Upper) EFFICIENCY INFLUENT	 Based on the pilot scale data this technology works well on these contaminants. Some banch scale data was not representative of optimum conditions. High levels of all and grease may interfere with the process. Soluble saits of Mg, Sb, Zn, Cu, and Pb may interfere with the pozzolan reaction. High levels of sulfates may interfere with the process. Protreatment may be required to increase pH.
Draft Guidance on Differentiating Alternative Technologies for the Removal Program



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

SEP 30 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPON:

MEMORANDUM

SUBJECT: Draft Guidance on Differentiating Alternative Technologies for the Removal Program (9380.2-05)

FROM: Timothy Fields Jr., Director Emergency Response Division

TO: Oil and Hazardous Materials Coordinators EPA Regions I - X

> Superfund Branch Chiefs EPA Regions I - X

Captain Robert L. Storch Jr., Chief Marine Environmental Response Division, USCG

The attached document provides draft guidance to removal field personnel to differentiate among alternative technologies, identifying them as available, innovative, or emerging.

The guidance describes criteria used for determining the developmental status of a particular technology. Among the topics addressed are: definitions of alternative technology and the terms - available, innovative, and emerging; identification of Alternative Technologies; and a Summary of Alternative Technology Contracts Procurement. It is expected that this guidance will become part of the forthcoming Alternative Technology Guidance.

Please review the document and provide comments by **Getterber** 25, 1987, to Elizabeth Zeller of my staff. If you have any questions, Elizabeth may be reached at, FTS 382-2190.

Thank you for your attention to this request.

Attachment

cc:	Russ Weyer, HSCD	George Turner, USCG
	Art Weissman, PAS	Hans Crump, ERD
	Lee Tyner, OGC	(Linda Garczynski, ROB
	Joe Freedman, OGC	Colleen Carruthers, EOS
	Lloyd Guerci, OWPE	Mark Mjoness, WOS
	Frank Biros, OWPE	Dick Mueller, PCMD
	Alan Corson, OSWER	Tom Sullivan, PCMD
	John Kingscott, OERR	Joe Lafornara, ERT
	,	Royal Nadeau, ERT
		John Riley, RSCB

	United States Enviro Washing	Inmental Protection Agency Iton, DC 20460	1 Derective Number							
SEPA os	WER Directiv	e Initiation Request	9380.2-05							
	2. Origina	tor information	(
Name of Contact Person	Mail Code	Office	Telephone Number							
Elizabeth Zeller	WH-548/B	OERR/ERD	382-2190							
J. Time Guidance on Differe	ntiating Alterna	ative Technologies								
4 Summary of Directive (Include Enel Statement of purpose) The purpose of this document is to provide removal program field personnel with guidance on identifying alternative technologies as either available, innovative, or emerging. This document presents general criteria based on developmental status. Use of the classification criteria will assist field personnel in technology selection and procurement. In addition, this document will direct field personnel on how to obtain the necessary concurrences and approvals for the 5. Keywords use of an alternative technology. Superfund, CERCLA, SARA, Alternative Technology Ba. Does this Directive Supersede Provide Directives!										
Does it Supplement Previous Directive(s)? XI Yes INo What Directive (number, inte) 9380.2-1 Administrative Guidance for Removal Program Use of Alternatives to Land Disposal Oraft Level A - Signed by AA/DAA IS - Signed by Office Director IS C - For Review & Comment In Development										
his Aequest Meets OSWER Ovectives System Format										
Name and Title of Approving Official Timothy Fields, Jr., Director/END THE TROUCH N. 930.07										
•										

OSWER OSWER OSWER DIRECTIVE DIRECTIVE

GUIDANCE ON DIFFERENTIATING ALTERNATIVE TECHNOLOGIES FOR THE REMOVAL PROGRAM

I. PURPOSE

The purpose of this document is to provide removal program field personnel with guidance on identifying alternative technologies as either available, innovative, or emerging. This document presents general criteria based on developmental status. Use of the classification criteria will assist field personnel in technology selection and procurement. In addition, this document will direct field personnel on how to obtain the necessary concurrences and approvals for the use of an alternative technology.

II. BACKGROUND

There is concern in the scientific community and the general public over the longterm reliability of containment technologies (e.g., landfills, containment lagoons) used as disposal options at Superfund removal sites. Because these and similar disposal options do not render the waste non-hazardous, they are not permanent cleanup solutions. In response to these concerns, the Superfund Amendments and Reauthorization Act of 1986 (SARA) explicitly addresses, promotes, and sponsors the use of alternative technologies to effect permanent cleanups.

An effort is under way by EPA to maximize the use of alternative technologies at Superfund removal sites through amendments to the National Contingency Plan (NCP); development of policies and documents such as the Alternative Technology Guidance, Land Disposal Restriction Guidance, and the Engineering Evaluation/Cost Analysis Guidance; compliance with the Resource Conservation and Recovery Act (RCRA) and the Toxic Substances Control Act (TSCA); and implementation of the Superfund Technology Transfer program. The alternative technology provisions in these policies and programs were developed to encourage permanent cleanups in a safe, cost-effective manner.

III. DEFINITIONS

The Superfund Amendments and Reauthorization Act of 1986 (SARA) defines an alternative technology as a "series of unit operations or any unit operation that permanently alters the composition of hazardous waste through chemical, biological, or physical means so as to significantly reduce toxicity, mobility, and/or volume of the hazardous waste or contaminated materials being treated." For the purpose of this document, alternative technologies are defined by their development status. The Superfund Innovative Technology Evaluation (SITE) Strategy and Program Plan identifies three levels of development

(1) Available Alternative Technology - A technology, such as several forms of incineration, that is fully proven and in routine commercial or private use;

(2) Innovative Alternative Technology - Any fully developed technology for which cost or performance information is incomplete, thus hindering routine use at CERCLA hazardous waste sites. An innovative alternative technology requires full-scale field testing before it is considered proven and available for routine use; and

(3) Emerging Alternative Technology - An alternative technology in an early stage of development; research has not yet successfully passed laboratory- or pilot-scale testing.

IV. CRITERIA FOR DIFFERENTIATING ALTERNATIVE TECHNOLOGIES

The criteria for classifying a technology as available, innovative, or emerging are based on the developmental status (including performance evaluation of the process, i.e., types of tests performed and the results). The criteria listed in this section are subjective; all technologies may not meet all of the criteria in a particular category based on the performance evaluation. However, by incorporating good engineering foresight and past performance records (particularly from other CERCLAfunded sites), one should be able to decide which category is most appropriate for that technology. The number of uncertainties will ultimately be the deciding factor between available and innovative or innovative and emerging. Since emerging technologies are usually in the early stages of development, it will not be difficult to distinguish between available and emerging technologies. In general, any data available for emerging technologies are based on bench- or pilot-scale tests, whereas

2

performance and reliability records for available technologies at the full-scale stage are obtainable for review.

The OSC should be fully cognizant of the following information prior to determining the category of the alternative technology:

- 1. The types of tests required before full-scale implementation;
- 2. Modifications to the process or technology necessary before implementation; and
- 3. The performance record of the technology.

It should be noted that although a technology may be classified as available or innovative, some modification of standard or past operating configuration is usually required to achieve successful results. There will be a different set of concerns with each technology, depending on site conditions and the contaminant present. The attainability of the technology (e.g., in a particular Region or State) has no bearing on its classification as available, innovative, or emerging. The provision of various levels of cost data, however, may allow for the upscale of innovative and emerging technologies.

Generally, available technologies should meet all the criteria listed below; innovative technologies would meet most of the criteria, in whole or in part, while emerging technologies would fail to meet any of the criteria sufficiently. Application of the criteria is shown in Exhibit 1. Good engineering foresight and professional judgment in conjunction with the criteria will be required to make an effective determination of the appropriate category.

The criteria for differentiating alternative technologies are:

- 1) The technology has successfully passed bench-scale testing.
- 2) The technology has successfully passed pilot-scale testing.
- 3) The technology has accomplished full-scale waste treatment under site conditions which did not differ significantly from site conditions of previous applications.

- 4) The technology has been used successfully to treat the same concentration of contaminants under site conditions which did not differ significantly from site conditions for previous applications.
- 5) The technology is routinely used at CERCLA hazardous waste sites.
- 6) Operation and maintenance records are available for review prior to implementing the technology.
- 7) The technology is generally available in the commercial/industrial market.
- 8) The technology will produce emissions/effluents that are easy to manage from environmental, cost, and health standpoints.
- 9) The technology is a valid comparison to previous methods of disposal in terms of providing better treatment or destruction of hazardous waste.
- 10) Demonstration and performance data are available for review.

V. IDENTIFICATION OF ALTERNATIVE TECHNOLOGIES

Information on potential alternative technologies for removals may be obtained from various sources, such as: the Removal Alternative Technology List, ORD Publications Announcement and Technology Transfer Newsletter, ERT Technology Bulletins; the Superfund Regional Technology Transfer Contact, the Superfund Technology Transfer Program; the Superfund Innovative Technology Evaluation Program; various industry • publications (trade journals); or discussions with OSCs.

Alternative Technology List

An initial compilation of known technologies developed by ORD is shown in Exhibit 2. The technologies are identified by their treatment process and development status, and were constructed by interpreting scientific data and information currently available. These tables are presented in draft form and will be replaced with a final version as soon as they become available. Technologies are always evolving as development continues and improvements are made. Because the developmental status of a particular technology may change, the technology's original classification may also change. For example, technologies originally classified as emerging may continue to develop through pilot- and fullscale testing. If further tests are successful, the technology may be classified as innovative. It is important to note that although a technology may be classified as available for the treatment of a particular waste stream or media; it may, however, be classified as innovative for another waste stream since its performance for treating the latter may not be fully proven. As technology development and improvement occur, 'the information in Exhibit 2 will be updated and periodically distributed to all Regions.

VI. SUMMARY OF ALTERNATIVE TECHNOLOGY CONTRACTS PROCUREMENT

The approval, procurement, and use of an alternative technology is dependent upon the technology's development status. This section describes contract procurement procedures for available, innovative, and emerging technologies. Exhibit 3 summarizes the concurrence procedures for all of the technologies.

A. Available Technologies

Generally, the use of available technologies is supported by Headquarters and procurement/implementation is straightforward. The majority of removal actions involving available alternative technologies are conducted through cleanup contracts (ERCS, Regional, and Site-Specific). Non-competitive subcontracts, fully documented by the contractor to justify sole source procurement and to verify reasonable costs, may be used.

B. Innovative Technologies

To use innovative technologies, (or a technology for which the development status is uncertain), Regions must consult with ERD and obtain Headquarters concurrence. The Regions are required to send a memorandum to the Director of ERD, outlining the technology. The memorandum should contain as much relevant ... technical and site information as possible. Technical information necessary to fully review a technology may include, but is not limited to: performance evaluation of full-scale testing; results from past usage at CERCLA-funded removals (if applicable); bench- or pilot-scale test results specific to the site conditions and contaminants; and potential and/or known modifications

5

September 4, 1987

necessary before implementation. Copies of the memorandum should be forwarded to the Chief of the Environmental Response Team (ERT), the Procurement and Contracts Management Division (PCMD), the Regional Coordinator, and the ERCS Project Officer. The ERT will review and evaluate the technology before approval is given for its use. Reasonable costs must be established for the proposed innovative technology. When the cost of the technology exceeds S500,000, consent by PCMD will be required, and an audit of the vendor's proposal must be performed by the Office of the Inspector General.

C. Emerging Technology

It is important to note that ERD and PCMD must be consulted before deciding to use an emerging technology as an option. The Superfund program generally does not sponsor emerging technologies with the possible exception of demonstration projects and performance-based contracts. These contracts require specific cleanup and/or treatment achievements as a basis for payment (performance-based contracts may also be considered for available and innovative technologies.) Only where stated goals are met do contractors receive payment.

An engineering evaluation/cost analysis (EE/CA) is required for non-time-critical removal actions (actions which allow at least a 6-month planning period before onsite action must be initiated) regardless of the technology's status. Procedures for drafting and submitting such an analysis are described in the upcoming <u>EE/CA</u> <u>Guidance for Non-Time-Critical Removal Actions.</u>

Procurement and contracting considerations are detailed in the forthcoming Alternative Technology Guidance. I. BIOLOGICAL TI

INT TECHNOLOGIES

TAAA

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Activated Sludge	Soluble organics in dilute aqueous waste streams (< 1% suspended solids).	BOD < 11,000 ppm. Re- quires low concentra- tions of heavy metals, PCBs, pesticides, oil and grease. Output sludge contains heavy metals and refractory organics which require further treatment.	A	X .	7,8,10
Aerated Lagoons	Industrial wastewater, organics with slow biodegradation potential, soluble organics in dilute aqueous waste streams. /	Requires large area. Unsuitable for solids. Requires a temperate climate. Output sludge contains heavy metals and refractory organics which require further treatment.	A		7,8
Aerobic Treatment (sequential batch reactor, fluidized bed, fixed film fluidized bed with/without activated carbon, aerated biofilm reactor, membrane reactor)	Aqueous waste with low levels (BOD < 10,000 ppm) of nonhalogenated organics and certain halogenated organics (i.e., phenols, formaldehyde, PCP).	Requires consistent, stable operating conditions.	A,I	Χ.	1,10,15
Anaerobic Treatment (fluidized bed, fixed film fluidized bed with/without activated carbon)	Aqueous slurry with low to moderate levels of non-chlorinated organic compounds con- taining less than 7% solids.	Requires consistent stable operating conditions. Unsuitable for oil and grease, aromatics and long chain hydrocarbons. Output sludge requires inciner- ation	A,I	x	1,2,8,10

•

I. BIOLOGICAL TI

'NT TECHNOLOGIES

DRA I

•

TECHNULOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Composting	Aqueous sludge with less than 50% solids, non-chlorinated hydrocarbons, high organic wastes including oils, tars, and industrial processing sludges.	Requires nutrient supplementation. Output sludge contains heavy metals.	A	x	7,8
Enzyme Treatment	Soluble organics in dilute aqueous waste streams.	Requires stable influent concentration.	E		
Genetically Engineered Organisms:		Requires pretreatment removal of heavy metals, oil, grease and VOCs.			
- Mycorrhizal	Soil-entrained hazardous waste constituents.		E		2
- Organo- mercurial Detoxifying Bacteria	Organomercurial-contaminated water, soil, and secure burial sites.		E		2,12
- PCB Degraders	PCB-contaminated soils.		Е		2
- - Soil Detoxifying Microorganisms	Various organic compounds in soils.		E		2
- Yeast Strain	Halogenated organics.		Е		5
- White Rot Fungus (Phanerochaete Chrysosporium)	Toxic or refractory halogenated organics in soil (i.e., 2,3,7,8-TCDD, DDT, mirex, lindane, hexachlorobenzene).		E		2,4,12

I. BIOLOGICAL TH NT TECHNOLOGIES

DRA:

.

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Rotating Biologic Contactor	Biodegradable dilute aqueous organic waste including solvents and halogenated organics.	Limited to low concentrations of heavy metals and concentrated refractory organics. Unsuitable for studges or solids.	A	x	8,10
Trickling Filter	Soluble organics in dilute aqueous waste streams with less than 1% suspended solids including solvents and halogenated organics.	BOD < 5000 ppm. Output sludge contains heavy metals and refractory organics which require further treatment.	A	x	8,10
Waste Stabiliza- tion Pond	Soluble organics in dilute aqueous waste / streams with less than 0.1% solids.	Not known.	· A		
:					

II. CHEMICAL TREATE / TECHNOLOGIES

Concentrated liquid chlorinated organic waste streams with low contrations of sulfur and oxygen.	Unsuitable for solids and tars. Unsuitable for benzene and aromatics. Output carbon , tetrachloride can be recovered.	Ι	-	8
Halogenated organics in soils and sludges that are partially dehydrated (i.e., PCBs, dioxins).	Requires heat and ex- cess reagent.	I i	х	1,2,4,5,10
Ilalogenated organics (i.e., PCBs).	Not known.	E		13
High concentrations cyanide (10%) and metals wastes.	Suitable for low solid content waste.	A		1,10
Solids, soils, sludges, slurries, or liquids contaminated with organic compounds.	Requires careful handlin of strong acids and alkalines. Reaction is performed a high temperatures and pressures requiring close monitoring.	A	х	1,10
Aqueous organic or inorganic waste streams, principally metals.	Suitable for liquid waste only.	A	x	1,7,8,10
Aqueous organic or inorganic waste streams.	Not known.	E		2,5
Organic liquids or vapors.	Suitable for small quantities of highly toxic waste and material recovery.	Е		5
	 Streams with low contrations of sulfur and oxygen. Halogenated organics in soils and sludges that are partially dehydrated (i.e., PCBs, dioxins). Halogenated organics (i.e., PCBs). High concentrations cyanide (10%) and metals wastes. Solids, soils, sludges, slurries, or liquids contaminated with organic compounds. Aqueous organic or inorganic waste streams, principally metals. Aqueous organic or inorganic waste streams. Organic liquids or vapors. 	Concentrated input chiofinated organic waste streams with low contrations of sulfur and oxygen. Ilalogenated organics in soils and sludges that are partially dehydrated (i.e., PCBs, dioxins). Ilalogenated organics (i.e., PCBs). High concentrations cyanide (10%) and metals wastes. Solids, soils, sludges, slurries, or liquids contaminated with organic compounds. Aqueous organic or inorganic waste streams, principally metals. Aqueous organic or inorganic waste streams. Organic liquids or vapors. High concentrations cyanide streams. Aqueous organic or inorganic waste streams. Organic liquids or vapors.	Concentrated induc culorinated organic waste streams with low contrations of sulfur and oxygen. Halogenated organics in soils and sludges that are partially dehydrated (i.e., PCBs, dioxins). Halogenated organics (i.e., PCBs). High concentrations cyanide (10%) and metals wastes. Solids, soils, sludges, slurries, or liquids contaminated with organic compounds. Aqueous organic or inorganic waste streams, principally metals. Aqueous organic or inorganic waste streams. Organic liquids or vapors. High convertional content waste streams. Organic liquids or vapors.	Concentration contrations of sulfur and oxygen. Istreams with low contrations of sulfur and organics (i.e., PCBs). Istreams wates. Solids, soils, sludges, slurries, or liquids contaminated with organic compounds. Aqueous organic or inorganic waste streams, principally metals. Aqueous organic or inorganic waste streams. Organic liquids or vapors. Istreams waste and material recovery.

II. CHEMICAL TRF FTECHNOLOGIES

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Neutralization	Corrosive liquid wastes both acids and bases.	Unsuitable for sludges and solids. Requires corrosion resistant equipment.	A	x	1,7,8,10
Oxidation (chlorination, ozonation, hydrogen peroxide, potassium permanganate, chlorine dioxide, hypochlorites)	Dilute aqueous waste (< 1% waste) containing organic/inorganic compounds. \ \	Requires controlled reaction conditions. Suitable for liquids and sludges only.	Α	x	1,7,8,10
Photooxidation using irradiated semiconductor oxides	Organics and inorganics present in leachates and industrial sludges.	Not known.	E		2,5
Polymerization	Organic compounds such as aromatics, aliphatics, and oxygenated monomers.	Application is limited to spills.	1	ļ	1,10
Precipitation .	Aqueous organic and inorganic waste containing metals.	Requires optimization of the reaction pll for the specific mix of metals present. Output sludge requires further treat- ment. Cross-reactivity may occur for mixed-metals content waste. Unsuitable for sludges, tars, and slurries.	A	x	1,7,8,10

II. CHEMICAL TRE..... i TECHNOLOGIES

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Reduction (sulfur dioxide, sodium borohydride, sulfite salts, ruthenium tetraoxide)	Diluté aqueous waste stream containing inorganic compounds especially metals (<1% heavy metal concentration).	Applicable to inorganic waste only. Suitable for liquid waste only.	I	х	1,7,8,10
UY/Photolysis	Liquid waste containing dioxins.	Suitable for liquids only.	E		1,4,5,10,12
Ultrasonic Dechlorination		Requires controlled temperature conditions.	Е	·x	14

III. PHYSICAL TREATMENT TECH

IES (COMPONENT SEPARATION)

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE ?	REFERENCE
Air Flotation (dissolved or induced)	Liquid waste containing oils or light suspended solids.	Liquid effluent may require further treat- ment.	A	x	8
Belt Filter Press	Biological and industrial sludges.	Filter cake may require further treatment.	^	x	1,8
Centrifugation (bowl, basket, disk)	Organic/inorganic liquids, slurries, and sludges containing suspended or dissolved solids or liquids where one component is nonvolatile. For example, wastewater sludge, wastes containing immiscible liquids, or wastes containing three distinct phases.	Unsuitable for tars, solids, dry powders, or gases. Not applicable for small size or low density particles.	A	x	1,8,10
Chamber Pressure Filtration (pressure leaf, tube element, plate and frame, horizontal plate)	Wastewater sludges, or sludges with a / flocculated or adhesive nature.	Dewatering technology. Unsuitable for sticky or gelatinous sludges.	A	x	7,8,10
Ġranular Media Filtration	Liquid waste containing suspended solids and/or oil.	Requires pretreatment fo suspended solids concen- tration < 100 mg/l. Re- quires frequent back- washing.	۸	x	1,8,10
Gravity Separation (coagulation, flocculation, sedimentation)	Liquids containing settleable suspended solids, oil, grease. :	Liquid effluent may require further treat- ment. Unsuitable for heavy slurries, sludges, or tars.	A	x	8
1	1				1

III. PHYSICAL TREATMENT TECHNOLIES (COMPONENT SEPARATION)

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE	MOBILE '	REFERENCE
High Gradient Magnetic Separation	Magnetic or paramagnetic particles in slurry.	Requires low concen- tration of magnetic materials. Limited to a 15% maximum total solid content.	E	x	10
In situ Soil Extraction	Soils with low levels of organics or inorganics/metals contamination.	Unsuitable for dry or organic-rich soils.	E	х	1,2
Oil Separation	Immiscible oily liquids in water that form an emulsion.	Not known.	A	x	
Vacuum Filtration (fixed media, rotary drum)	Organic or inorganic chemical sludges, metals and cyanides bound up in hydroxide sludges.	Dewatering technology. Unsuitable for sticky or gelatinous sludges.	A	x	1,8,10
	·				

•

III. PHYSICAL TREATMENT TECH

JGIES (PHASE SEPARATION)

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Air Stripping	Aqueous and adsorbed organic and inorganic wastes with relatively high volatility and low water solubility such as chlorinated organics, aromatics, and ammonia.	Limited to VOC concen- tration < 100 ppm. Suspended solids may clog tower.	A	x	1,2,7,8,10
Artificial Freezing	Sludges and soils contaminated with organics or metals.	Not known.	I	x	2
Carbon Adsorption	Aqueous organic wastes (containing <1% total organics and < 50 ppm solids) with high molecular weight and boiling point, and low water solubility, polarity and ionization.	Unsuitable for metals.	A	x	1,7,8,10
Colloidal Gas Aphrons (CGAs) (enhances air stripping and biodegradation)	Soils contaminated with phenols, phthalate / esters, aromatic hydrocarbons, aliphatic hydrocarbons, chlorinated hydrocarbons, amines and alcohols.	Hydraulic conductivity of the soil must be > 10 ⁻⁴ cm/sec.	E		2
Distillation	Liquid organic mixtures with low viscosity that can be separated due to molecular weight/ volatility differences.	Unsuitable for thick polymeric materials, slurries, sludges, or tars.	A	x	1,8,10

III. PHYSICAL TREATMENT TEL...

JGIES (PHASE SEPARATION)

T't m	70 8
	- e

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE	MOBILE '	REFERENCE
Electrokinetics	Soils contaminated with organic or inorganic waste.	Soil matrix must be relatively permeable and saturated.	1		2
Evaporation	Organic/inorganic liquid solvents contaminated with nonvolatile impurities (i.e., oil, grease, paint solids, polymeric resins).	Liquids must be volatile. Unsuitable for tars, sotid, dry powders, or gases. Energy-intensive process	A	х	1,8,10
Freeze Crystallization	Dilute aqueous organic/inorganic waste solutions containing < 10% total dis- solved solids.	Unsuitable for foamy, viscous or high solid content waste streams.	E		2
Mechanical Soil Aeration	Volatile organics in sludge and soil. /	Effluent may require further treatment.	· A	x	7
Metal Binding	Metal-contaminated aqueous waste streams, leachate or groundwater.	Limited to metal con- centrations between 500-1000 ppm.	E		2
Resin Adsorption	Aqueous waste streams containing soluble organics particularly phenols and explosive materials.	Limited to low concen- trations of organics (<8%) and suspended solids (<50 ppm).	A		8
Reverse Osmosis	Aqueous waste streams containing <400 ppm heavy metals, high molecular weight organics, and dissolved gases.	Unsuitable for oxidants. Requires controlled pH, low concentration of sus- pended solids, and no strong oxidants.	I	x	2,5,8

III. PHYSICAL TREATMENT TE

DGIES (PHASE SEPARATION)

DRA.

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Solvent Extraction	Aqueous stream contaminated with single or multi-component dissolved organic wastes. Sludge contaminated with oil, toxic organics, and heavy metals.	Extracting solvent must be immiscible in the liquid and differ in density so gravity separation is possible. Suitable for studges containing < 20 wt % oil/organics and < 20 wt % solids.	I, A	x	8,10,12,15
Steam Stripping	Aqueous solutions of volatile orgnics.	Effluent may require further treatment. Suitable for waste streams with low metal concentration.	A	x	1,2,7,8,10
Supercritical Extraction	Sludges, solids, or liquids contaminated with organics.	Effluent may require further treatment,	E	x	2,10,15
Ultrafiltration	Removes oils, metals and proteins from aqueous solutions with dissolved organics, emulsions, and colloidal particles.	Limited to low concen- trations of suspended soils.	A	x	8

IV. STABILIZATION/SOLIDIFICATION/Enc. JLATION TREATMENT TECHNOLOGIES

DIJT

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE ,	REFERENCE
Cement-based Fixation	Treated sludges and soils containing metal cations, radioactive wastes, and solid organics (i.e., plastics, resins, tars).	Long term stability/ leachability is unknown. Lignite, silt or clay increase setting time. Dissolved sulfate salts, borates, and arsenates must be limited.	A	x	1,6,10,15
Macro- Encapsulation, Overpacking, Thermoplastic and Thermoset- ting Techniques	Chemically or mechanically stabilized organic, inorganic, and radioactive wastes.	Encapsulating matrix must be campatible with waste. Long term leach- ability unknown, there- fore, waste storage must be considered. Requires specialized equipment.	•	x	1,6,10
Pozzolanic- based Fixation (flyash, lime- based)	Treated sludges and soils containing heavy metals waste oils, solvents, and low level radioactive waste.	Borates, sulfates, and carbohydrates interfere with the process. Long term stability/ leachability is unknown.	^	х	1,6,10
Sorptive Clays (treated, chemically modified).	Halogenated organic compounds and heavy metals.	Long term leaching is a problem, therefore, waste storage must be considered.	1	х	2,5,6
Vitrification	Soils contaminated with organic, inorganic, and radioactive wastes.	Limited to soils with high silica content.	A/I	x	1,2,4,10

	V. THERMAL TRE	T TECHNOLOGIES	1. 94. 4			
TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE	
Electric Reactor	Soils contaminated with solids and liquid organics and inorganics.	Contaminated soil must be finely divided and dry.	1 🕴	x	1,2,4,10,11, 15	
Electric Pyrolysis	Viscous liquids, sludges, solids, high ash materials, salts and metals and halogenated waste.	Requires homogeneous waste input. Metals and salts in the residue can be leachable.	1	x	1,10,15	
Fixed Hearth	Bulky solids, liquids and sludges.	Particle size must be large enough not to fall through grate.	A		8	
Fluidized Bed	Organic solids, liquids, and sludges.	Requires low water and inert solid content.	A	х	1,2,4,8,10,12,15	
Industrial Boiler	Granulated solids, liquids, and sludges.	Requires low chlorine and sulfur content. Ash content clogs system. Particle size.	Α'		1,8,10	
Industrial Kilns	Spent pot lining, nonhalogenated oils and PCB-contaminated liquids and sludge.	Chlorine and sulfur content must be limited. Particle size.	A		1,8,10,12	
Infrared Incineration	Soils, solids, and sludges contaminated with chlorinated organic compounds (i.e., PCBs, dioxins, explosives).	Primarily for solid organic waste. Heavy metals are not fixed in ash.	A	x	1,4,10,15	

.....

•

÷

.

V.	TH	ER	MA	L	TR	EA	7
----	----	----	----	---	----	----	---

TECHNOLOGIE8

38.		١.	•	14	1256.44
þ	١	Ľ	:		3

TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFERENCE
Liquid Injection	Pumpable liquid organic waste.	High inorganic content and heavy metal content must be limited. Chlorinated solvents cause accelerated corrosion rates	, A	x	1,5,8,10,12
Microwave Plasma	PCBs	Not known.	1		
Molten Glass	Organic solids, liquids, gases, sludges (i.e., (i.e., plastics, PCBs, asphalt, pesticides).	Sodium sulfates must be limited to <1% content. Inappropriate for soils and high ash waste.	1		1,2,10,11
Molten Salt	Low ash, low water content liquid or solid / waste.	Corrosion problems. Requires frequent bed re- placement.	, I ,	x	1,7,10
Multiple Hea rth	Granulated solids, sludges, tar, liquid, and and gaseous combustible waste.	Water, salt and metal content must be limited. Particle size must be small enough to pass through injector nozzles. Not recommended for hazardous wastes.	A		1,7,8,10,12
Plasma Systems	Liquid organic wastes (i.e., pesticides, dioxins, PCBs, halogenated organics).	Liquids only.	I	x	1,2,7,10,15
Pure Oxygen Burner	Liquid wastes which require high temperatures for destruction or have low heating values.	Requires specially engineered nozzles to atomize the liquid waste.	I	x	15
Pure Oxygen Burner	Liquid wastes which require high temperatures for destruction or have low heating values.	Requires specially engineered nozzles to atomize the liquid waste.	Ĩ	x	15

	V. THERMAL TREA'	TECHNOLOGIES				
TECHNOLOGY	APPLICABLE CONTAMINANTS	QUALIFYING FACTORS	PHASE '	MOBILE '	REFEREN	
Radio Frequency Thermal Heating	Volatile, low boiling point, or easily de- composed organic compounds in soil.	Not known.	1	x	1,10	
Rotary Kiln	Solid, liquid, or gaseous organic waste.	Containerized wastes are difficult to handle. High inorganic salt or heavy metal content wastes require special consider- ation. Fine particulate matter must be limited.	A	х `	1,2,4,7,8,10, 12,15	
Supercritical Water Oxidation	Aqueous organic solution/slurry or mixed organic/inorganic waste.	'Not known.	I	x	1,4,10,11,12	
Wet Air Oxidation	Aqueous waste streams (<5%) with dissolved or suspended volatile organic substances.	Unsuitable for solids, viscous liquids, or highly halogenated organic compounds. Not economi- cal for dilute or concen- trated waste.	A	x	1,7,10,11	

.

- -

BIBLIOGRAPHY

¹U.S. EPA, Treatment Technology Briefs: Alternatives to Hazardous Waste Landfills, EPA 600/8-86/017, HWERL, Cincinnati, Ohio, June, 1986.

²Pheiffer, Tom. Memorandum: One Pagers on Innovative Technologies, OSWER, --- December 11, 1985.

³Klee, A. and Dial C. Memorandum: One Pagers on Innovative Technologies, ORD, HWERL, July 10, 1986.

⁴Technical Resources, Inc. Dioxins-Containing Waste Streams, EPA Contract No. 5W-6242-NASX, July 31, 1985.

- ⁵U.S. EPA, Alternative Technology Division: Summary of Projects, HWERL, Cincinnati, OH, April 1986.
- ⁶U.S. EPA, Superfund: Handbook for Stabilization/Solidification of Hazardous Wastes, EPA 540/2-86/001, HWERL, Cincinnati, OH, June 1986.
- ⁷Versar, Inc. The Use of Treatment Technologies at Superfund Sites-Slide Presentation, EPA Contract No. 68-01-7053; June 25, 1986.
- ⁸Versar, Inc. Alternative Technologies Briefing, 1986.
- ⁹Camp Dresser and McKee, Inc. Mobile Treatment Technologies for Superfund Wastes-Interim Edition; U.S. EPA 540/2-86/003, July 1986.
- ¹⁰U.S. EPA, RCRA/CERCLA: Alternative Treatment Technology Seminars, ORD, HWERL, CERI, OSWER, Boston, MA, June 1986.
- ¹¹U.S. EPA, Project Summary: Innovative Thermal Hazardous Wasie Treatment Processes, EPA 600/S2-85/049, HWERL, Cincinnati, OH, June 1985.
- ¹²Key Technologies for the Treatment of Dioxin-Contaminated Wastes. (one page no reference)
- ¹³Massey, Dr. M.J., and Dr. F.M. Walsh, An Electrochemical Process for Decontaminating PCB-containing Transformer Coolants, presented at EPRI Conference, October 1985.
- ¹⁴Hazardous Waste Management, Inc., Detoxification of Substances by Utilization of Ukrasonic Energy, October 16, 1984.
- ¹⁵U.S. EPA, SITE Program Memorandum: SITE Participant Technology Descriptions, HWERL, Cincinnati, OH, October 7, 1987.
- 16Office of Technology Assessment, Superfund Strategy Summary, Congress of the United States. 1985.
- ¹⁷Olexsey, Robert A., Memorandum: List of Alternative Technology Unit Processes, ORD, HWERL, July 10, 1986.
- 18U.S. EPA, SITE Program: Draft Strategy and Program Plan; ORD and OSWER, June, 1986.

Superfund Innovative Technology Evaluation (SITE) Program Requirements



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C 20460

. A 2 2 1988

OFFICE OF SOLID WASTE AND EMEPGENCY RESPONSE

MEMORANDUM

SUBJECT: Superfund Innovative Technology Evaluation (SITE) Program Requirements FROM: J. Winston Porter Assistant Administrator TO: Addressees

PURPOSE

The purpose of this memorandum is to describe the regulatory requirements and procedures for implementing treatment technology demonstration projects in the SITE program.

BACKGROUND

The SITE program was established under section 209(b) of the Superfund Amendments and Reauthorization Act (SARA) (section 311(b) of CERCLA, as amended) to evaluate technologies for the treatment of hazardous waste. The purpose of the program is to expedite the development and commercialization of technologies to fulfill the cleanup requirements under section 121 of SARA. This is primarily accomplished by conducting technology demonstrations. These demonstrations provide performance, cost and reliability data so that potential future users have sufficient information to make sound judgements as to the applicability of the technology for a specific site or to compare it to other alternatives. Demonstrations will take place under conditions that either duplicate or closely simulate actual wastes and conditions found at Superfund sites. The demonstrations will simulate a commercial scale application and will be of sufficient size to generate data applicable to full-scale operation. The results of these demonstrations will be broadly applicable to all hazardous waste cleanup efforts.

The program is focused on conducting technology evaluations as partnerships between EPA and developers. After acceptance into the program, the technology developers are responsible for providing and operating their equipment, while the Agency is responsible for monitoring and evaluating performance. New technologies are accepted into the program based on responses to a Request for Proposal (RFP), which is distributed annually. Once a developer is accepted, a Superfund waste is chosen for treatment demonstration based on site nominations provided by the Regions. At the present time, there are 20 developers participating in the program with projects at various stages of completion. Several field demonstrations are taking place for the first group of technologies; Regions are considering Superfund site nominations for the second group; and a third RFP will be distributed in January.

In the course of implementing these projects, several issues have arisen concerning the relationship to Superfund and RCRA requirements. Specifically, issues concerning permitting, ARARS (applicable or relevant and appropriate requirements), community relations, and cost recovery are addressed below.

DISCUSSION

Section 311 of CERCLA authorizes EPA to carry out the SITE program and sets forth detailed standards governing the program. Projects taken under the SITE program are not remedial actions and are not governed by the provisions of section 121 of CERCLA. Rather, section 311(b)(4) requires that SITE projects be carried out:

under such terms and conditions as the Administrator shall require to assure the protection of human health and the environment and to assure adequate control by the Administrator of the research, testing, evaluation, development, and demonstration activities at the site.

In addition, section 311(b)(7) provides that in selecting technologies to be demonstrated, EPA "shall, consistent with the protection of human health and the environment, consider..." several enumerated criteria.

EPA believes that section 311 gives the Agency authority to determine what is necessary to protect human health and the environment, and that SITE program activities are not subject to environmental permitting requirements under other Federal and State laws. However, in order to assure protection of human health and the environment, SITE demonstration projects taking place at Superfund sites should comply with the substantive requirements of all applicable or relevant and appropriate State and Federal environmental laws, except where a waiver similar to one of those provided in section 121(d) (4) is appropriate. For the same reason, off-site demonstration projects should be limited to facilities having all appropriate State and Federal permits.

Occasionally, it will be necessary to conduct laboratory bench-scale treatability tests prior to the actual technology demonstration. Bench-scale tests are primarily required to assess the effectiveness of and establish operating parameters for biological and solidification technologies. In order to assure protection of human health and the environment, EPA will prefer to use permitted facilities to conduct bench-scale treatability tests. However, if this is not practicable, the tests may be performed at non-permitted off-site facilities. It is anticipated that unpermitted facilities would only be used in a limited number of circumstances. Any usee of off-site facilities should be restricted to wastes in limited quantities necessary to perform the tests (e.g., 50 kg). Laboratories conducting treatability tests should be required to submit a test plan and to have a health and safety plan. No public comment period is necessary for the small-scale laboratory testing. Unused samples, treated residues, and by-products should be returned to the original site or disposed of at permitted facilities. Manifests should be used for shipment of waste to Subtitle C facilities. However, shipments from a Superfund site to a laboratory for study and then back to the Superfund site need not be manifested. Materials should be transported in accordance with relevant DOT or postal service regulations and all shipping receipts should be kept in the project files. In addition, sample handling requirements in 40 CFR 261.4(d) should be met.

With respect to community relations, CERCLA section 311(b)(5)(e) requires that the Agency give notice and opportunity for public comment on SITE demonstrations. The primary purpose of this provision is to solicit comments on the proposed matching of technologies with sites for the purpose of conducting field demonstrations. The SITE program will use the community relations procedures established for CERCLA remedial actions when carrying out demonstrations at Superfund sites. Off-site demonstrations should comply with relevant permit requirements, and the associated public comment process will satisfy the SITE program requirements.

Finally, since the SITE demonstrations are not part of the Superfund response action, the costs are not recoverable from responsible parties.

Attachment

Addressees:

Director, Office of Emergency & Remedial Response, Region II Director, Hazardous Waste Management Division, Regions III & VI Director, Air and Waste Management Division, Region II Director, Waste Management Division, Regions I, IV, V, VII & VIII Director, Toxics and Waste Management Division, Region IX Director, Hazardous Waste Division, Region X Director, Environmental Services Division, Region I-X

cc: Director, ORD/OEETD Director, QRD/HWERL Director, OERR Regional SITE Coordinators

Factors to Consider When Selecting Incinerators as Alternative Technology for Superfund Cleanups

BACKGROUND PAPER

SUPERFUND INNOVATIVE TECHNOLOGY EVALUATION PROGRAM

INTRODUCTION

The Fnvironmental Protection Agency's Office of Research and Development (ORD), joining with the Office of Solid Waste and Emergency Response (OSWFR), has initiated the Superfund Innovative Technology Evaluation (SITE) Program. The SITF Program Will help EPA find, test, and encourage the use of new ways to destroy, stabilize or otherwise treat hazardous wastes, rather than just hurying them in the ground.

The overall goal of the SITE Program is to maximize the use of alternatives to land disposal and containment at Superfund sites. To accomplish this goal, the program will provide reliable cost and performance information on technologies that offer an alternative to land disposal. This information will be generated by conducting pilot-scale or full-scale demonstrations of alternative technologies at Superfund sites.

BACKGROUND

EPA's Superfund program is designed to clean up hazardous waste sites around the country. It has been underway for six years with action taken at over 450 Superfund sites to address long-term problems created by hazardous wastes. More than half of these sites have involved burying the hazardous wastes in specially prepared landfills -- a process called <u>land disposal</u>. While land disposal can be a good way to handle wastes, in some instances it does not provide a permanent solution to the problem. The wastes in the landfill may still be dangerous, and may potentially leak through the confines of the burial site.

Recently, members of the scientific community, the public, and Congress expressed concern that it was time to move away from reliance on land disposal for handling hazardous wastes. These views are reflected in the Hazardous and Solid Waste Amendments of 1985 (HSWA) and the Superfund Amendments and Reauthorization Act of 1986 (SARA). The Agency responded by altering certain policies and developing regulations regarding hazardous waste disposal in landfills and the expanded use of alternative technologies.

More specifically, the Land Disposal Restriction provisions of HSWA stipulate that certain wastes cannot be land disposed unless they meet specific concentration-based treatment standards that represent best demonstrated available technology. Similarly, the new SARA sets a clear preference for the use of cost-effective

DECEMBER 1, 1986

treatment remedies which permanently and significantly reduce the toxicity, mobility or volume of wastes at Superfund sites. In addition to these provisions, SAPA directs EPA to carry out a program of research, evaluation, testing, development and demonstration of alternative or innovative technologies to achieve more permanent protection of human health and welfare and the environment. These laws, regulations, and policies clearly provide the impetus toward the use of treatment alternatives for the management of hazardous wastes. Technical and policy guidance on the use and availability of alternative technologies in both the Superfund removal and remedial programs is expected soon.

SITE PROGRAM PHASES

The four phases of the SITE Program listed below are being conducted simultaneously. They will be integrated so that information from one phase can be used in another. Each phase is designed to meet a particular objective of the SITE Program. The SITE program will be periodically reviewed to determine how information generated from all phases may be used to encourage the selection of new technologies at Superfund sites.

o Phase I: Identify and Remove Impediments

Trained engineers, economists and policy analysts will identify the obstacles to the orderly development and use of alternative technologies to manage hazardous wastes. Resolving issues dealing with permit requirements treatment residue disposal, liabilities, procurement requirements, and public interest will help pave the way to increased use of treatment alternatives. Recommendations on how EPA can remove or limit obstacles to the use of alternative technologies will be complete by mid 1987.

o Phase II: Conduct a Demonstration Program

EPA will conduct a demonstration program for innovative technologies at selected Superfund sites. From these demonstrations, EPA will provide performance and cost information for the new technologies. This information will help EPA decide when, and under what circumstances, to use the technology at other hazardous waste sites.

The demonstration progam will be an ongoing effort. EPA plans to start several demonstrations as soon as possible to obtain information on the most advanced technologies. Demonstrations will then occur on a yearly schedule. An annual SITE demonstration plan will be published for public comment and will discuss the proposed technologies to be demonstrated and the Superfund sites identified for use that year.

o Phase III: Applications Analyses

Applications analyses will be conducted for technologies that warrant further investigation after completion of the demonstration. Such analyses will examine the applicability of the technology to other Superfund sites and include factors such as cost, appropriate site conditions, efficiency characteristics, and waste types. Policies and procedures that need to be taken into account in applying promising new technologies to Superfund sites will be examined.

o Phase IV: Fmerging Technologies Development

ORD will establish a research program to continually evaluate and encourage the development of emerging alternative technologies through pilot testing to full scale demonstrations. This phase will be initiated in the spring of 1987 and will be ongoing thereafter.

THE DEMONSTRATION PHASE: A SUMMARY

Phase II, the Demonstration Phase, will be the most publicly visible of the four phases. Actual work will begin at Superfund sites during the demonstration phase and new technologies will be evaluated. Key events in the demonstration phase are summarized:

o Advertise in Commerce Business Daily

A notice requesting proposals will be placed in the <u>Commerce Business Daily (CBD)</u> on or before January 15, 1987, and annually thereafter. Developers of new and innovative technologies are asked to obtain from EPA the information necessary to submit proposals for demonstrating their technology. Proposals submitted in response to the CBD announcement can be made through February. Information about the Request for Proposals (RFP) can be obtained from Mr. Steve James (OFD) by calling 513-569-7877 (commercial) or 684-7877 (FTS).

In addition to choosing technologies based on the CBD advertisement, EPA has set up a second mechanism where innovative technologies may also be demonstrated as part of routine response actions. Superfund or other cleanup sites on which EPA is planning to use treatment technologies as part of the response action may be adapted for SITE demonstrations.

o Select Technologies

Potential new technologies will be carefully screened by OSWER and ORD. The screening process will be competitive and must be completed within 90 days of receiving the completed application. In selecting technologies to be demonstrated, EPA must consider, at a minimum, the following criteria:

- The potential for contributing to solutions to those waste problems which pose the greatest threat to human health, which cannot be adequately controlled under present technologies, or which otherwise pose significant management difficulties.
- 2. The availability of technologies which have been sufficiently developed for field demonstration and which are likely to be cost-effective and reliable.
- 3. The availability and suitability of sites for demonstrating such technologies, and the capability to conduct demonstration projects in such a manner as to assure the protection of human health and the environment.
- The likelihood that the data to be generated from the demonstration project at the site will be applicable to other sites.

o <u>Match Technologies to Sites</u>

Many of the new technologies may only be suited for specific kinds of wastes at a specific site. Using the expertise and knowledge in EPA Headquarters, laboratories, and the EPA Regions, technologies will be matched with wastes at available Superfund sites. The new Superfund law requires EPA to conduct a minimum of ten technology demonstrations each year. While most demonstrations will take place at Superfund sites, some technologies may be tested at commercial hazardous waste sites or at special test and evaluation facilities operated by EPA or the developer. Factors such as risk, public interest, expense, disposal of residues and involvement of potentially responsible parties will be considered when matching sites and wastes with technologies. o Develop Annual SITF Demonstration Plan

The annual SITE demonstration plan will discuss the technology demonstration projects planned, how and when the projects will be carried out, the tentative sites selected, the resource requirements for conducting the demonstrations, and the means for public involvement and input. This plan will be published for public review and comment.

o Implement Site-Specific Community Relations Programs

Once a site and technology have been tentatively selected, EPA will develop and implement a community relations plan to seek the ideas and suggestions of local residents. In accordance with the plan, EPA will introduce the community to the SITE Program and its role at their local site, and will provide information on the specific technologies. Public comment will be sought throughout the demonstration project and community relations activities will continue for the duration of the demonstration.

o Conduct Demonstration

After reviewing and incorporating comments on the SITE demonstration plan, EPA will enter into a contract, letter agreement, or cooperative agreement with the developer of the technology. The demonstration will begin and probably last several months, depending upon the time required to gather information on the effectiveness and reliability of the technology. The developer will provide the equipment and he responsible for set-up and operation at the demonstration site. The technology will treat Superfund wastes during the test, but since it is only a demonstration, it will not be expected to If clean up all of the wastes at the Superfund site. necessary, another standard technology will be used to handle the remaining wastes. EPA's role will be to ensure credible results by providing the testing protocols and procedures and preparing the analytical and quality assurance/quality control work plans so the performance data can be consistently and accurately interpreted.

o Complete Evaluation and Distribute Information

At the close of each demonstration, EPA will evaluate the results of all tests. If the results are positive, the technology could be applied at other similar Superfund sites. EPA will modify Agency policies and procedures to encourage the use of these and other alternative technologies. EPA will make available the results of the evaluation of each new technology to the hazardous waste cleanup industry, regulatory agencies and the public. The Agency is developing a technology transfer program to ensure the distribution of accurate information as quickly as possible.

THE PROGRAM TO DATE

The Superfund reauthorization calls for a minimum of ten field demonstrations per year through 1940. Congress authorized \$20 million per year for the demonstration program, including demonstrations of innovative monitoring equipment. The alternative technology demonstration program is limited to \$10 million per year, with a \$3 million cap for a single demonstration.

In anticipation of reauthorization, EPA began setting up the SITE Program in 1986. In March of 1986, the first notice requesting proposals was placed in the <u>Commerce Business Daily</u>. In response to RFP SITE-001, twenty proposals were received representing the following technologies:

- 10 Incineration/Thermal
- 3 Biological Treatment
- 2 Containerization
- 1 Solidification/Stabilization
- 1 In-Situ Vapor Extraction
- 1 Chemical Detoxification
- 1 Robotics
- 1 Vapor Condensation

The proposals were evaluated and ranked according to the specified criteria. In addition, EPA identified several potential demonstrations through the second mechanism -- routine response actions. A listing of firms, tentative sites (where determined), and a brief description of the technologies which are potentially proposed for demonstration during 1987 is attached.

CONTACTS

- OSWER: Don White 202-475-8600 (commercial) or 475-8600 (FTS) For information on Superfund wastes and sites and other general information.
 ORD: Ron Hill 513-569-7861 (commercial) or 684-7861 (FTS) or Steve James 513-569-7877 (commercial) or 684-7877 (FTS) For information related to applications of treatment technologies, the <u>Commerce</u> <u>Business Daily</u> announcement, and the Request for Proposals.
- <u>Superfund/RCRA Hotline</u> 800-424-9346 202-382-3000
SUPERFUND INNOVATIVE INTHNOLOGY EVALUATION (SITE) PROGRAM: PROPOSED DEMONSTRATION PROJECTS AS OF NOVEMBER 25, 1986

Main Track (CRD)		Tentative
Developer	Technology	Demonstration
Waste-Tech Services, Inc. 18400 West 10th Avenue Golden, Colorado 80401 Contact: Eliot Cooper 303-279-9712	Mobile thermal combustor- fluidized bed	Coors Company Golden, Colorado
Advanced Combustion Technologies Inc. P.O. Box 940498 Atlanta, Georgia 30340 Contact: Thomas McGowan 404-662-5360	Pure oxygen hurner	EPA ORD Test and Evaluation Facility
Shirco Infrared Systems, Inc. 1195 Empire Central Dallas, Texas 75247 Contact: Scott Berdine 214-630-7511	Electric infrared incinerator	To be determined
Hazcon, Inc. P.O. Box 947 Katy, Texas 77492 Contact: Ray Funderburk 713-391-1085	Solidification/ stabilization process	To be determined
Westinghouse Electric Corp. Waste Technology Services Division P.O. Box 286		
Madison, Pennsylvania 15663 Contact: Carrie Penman 412-722-5709	(1) Pyroplasm system	To he determined
Contact: William Reed 412-722-5303	(2) Electric pyrolyzer	To be determined
Terra Vac, Inc. P.O. Box 550 Corado, Puerto Rico 00646 Contact: James Malot 809-723-9171	In-situ vacuum extraction	To be determined

Through Response Actions		Tentative
Developer	Technology	Demonstration
New York State Department of Environmental Conservation (Plasma Systems, Inc.) 50 Wolf Road Albany, New York 12233 Contact: Norman Nosenchuck 518-262-1338	Plasma pyrolysis thermal unit	Love Canal Niagara Falls, New York
GA Technologies, Inc. P.O. Box 85608 San Diego, California 92138 Contact: Harold Diot 619-455-2383	Circulating bed combustor	(A Technologies' facility; test on waste from McColl, Fullerton, CA (remedial action)
Resources Conservation Company 3101 N.E. Northup Way Bellevue, Washington 98004 Contact: Paul McGough 206-828-2455	Basic Extraction Sludge Treatment (BEST): solvent extraction technology	General Refinery, Inc. Savannah, Georgia (removal action)
General Electric One River Road Schenectady, New York 12345 Contact: John Harrsen 518-385-0045 and International Waste Technologies 807 North Waco, Suite 31 Wichita, Kansas 67203 Contact: Jeff Newton 316-262-1338	In-situ solidification/ fixation process	General Electric Raileah, Florida

FACTORS TO CONSIDER WHEN SELECTING INCINERATORS AS ALTERNATIVE TECHNOLOGY FOR SUPERFUND CLEANUPS

- Facility Capacity Many incinerators are no longer accepting Superfund wastes because they are not in compliance with the RCRA Offsite Disposal Policy. Since the number of incinerators available to accept wastes is limited, it is critical to consider the chemical and physical characteristics of the waste before selecting a disposal option. Otherwise, because of limited incinerator capacity, hazardous wastes that must be incinerated (e.g., high level PCBs, pesticides, cyanide wastes, etc.) will have to be stored in a temporary storage facility, or staged onsite until a RCRA approved incinerator has the capacity to receive the waste.
- ^o Mobility of the Hazardous Material Certain hazardous materials will tend to leach more quickly than others. This should be considered prior to landfilling.
- Incinerable Characteristics Some wastes, particularly liquids with a high flash point, must be incinerated. Other wastes, especially bulk soils and sludges, or materials with high water or ash contents because they require a high energy input, are difficult and much more expensive to incinerate.
- * Hazardous Constituent Concentration Highly contaminated wastes are generally better candidates for incineration than low level materials, because of the higher long term public health and environmental risks associated with landfilling this material instead of destroying it.
- Cost According to the Offsite Disposal Policy, alternative technology can be eliminated only if the cost of the alternative far exceeds the cost of others (e.g., by an order of magnitude) and does not provide substantially greater public health and environmental benefit.

Other Alternatives

- Reuse or recycle
- Onsite treatment or pretreatment (e.g., neutralize acids, remove lead from soil)
- Encapsulation
- Treat at commercial or public waste treatment facilities
- Incinerate in a cement kiln or other industrial boilers/furnaces

CERCLA Removal Actions at Methane Release Sites

•

•

•

United States Environmental Protection Agency Office of Solid Waste and Emergency Response

DIRECTIVE NUMBER: 9360.0-8
TITLE: CERCLA Removal Actions at Methane Release Sites
APPROVAL DATE: January 23, 1986
EFFECTIVE DATE: January 23, 1986
ORIGINATING OFFICE: Superfund
FINAL
DRAFT
STATUS:
Memorandum from Henry Longest II, Director, OERR
To: Basil G. Constantelos, Director, Waste Management Division, Region V
REFERENCE (other documents):

OSWER OSWER OSWER VE DIRECTIVE DIRECTIVE L

2



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JAN 23 1986

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: CERCLA Removal Actions at Methane Veloce Sites

FROM: Henry L. Longest II, Director Office of Emergency and Remedia/Decremse

TO: Basil G. Constantelos, Director Waste Management Division Region V

This memorandum clarifies Office of Emergency and Remedial Response policy on the appropriateness of removal actions at methane gas release sites under authority of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA). This issue was recently brought to our attention as a result of response actions which were being considered at the Industrial Excess Landfill (IEL) site in Lake Township, Ohio.

Because methane gas is not listed or designated under any of the statutory provisions in section 101(14) of CERCLA, it is not a "hazardous substance." However, response actions under CERCLA section 104 are not limited to hazardous substances. Section 104(a)(1) authorizes responses to actual or potential releases of "pollutants or contaminants." The definition of hazardous substance in section 101(14) and pollutant or contaminant in section 104(a)(2) excludes certain types of natural gas and petroleum. Naturally occurring methane gas found in or associated with petroleum deposits is a type of natural gas and is therefore exempted from CERCLA coverage. However, methane gas emanating from a landfill is not considered to be "natural gas" and such releases may therefore be eligible for response under section 104(a)(1) if methane gas otherwise meets the definition of pollutant or contaminant under section 104(a)(2).

With respect to the response authority for pollutants or contaminants, response action under section 104(a)(1) is authorized only if there is a threatened or actual release of such substances which may present an "imminent and substantial danger to the public health or welfare." The potentially explosive gas levels, detected during daily monitoring at the perimeter of the landfill and nearby homes and businesses, appear to meet the criterion of imminent and substantial danger. Therefore, because the methane gas is not excluded as a "natural gas," if it is determined to be a "pollutant or contaminant" as defined in section 104(a)(2) and to "pose an "imminent and substantial danger to the public health and welfare," response under CERCIA is authorized. Although the proposed removal action is authorized under CERCLA section 1C4(a)(1), the responsible party may not be liable under section 107 for removal action costs since liability under that section is limited to releases of hazardous substances.

As a matter of policy, CERCIA responses to methane gas releases should be carefully evaluated on a case-by-case basis. Approval of the removal action at IEL should not be considered as setting a general precedent for future actions involving methane gas. Methane gas is produced in most landfills, and response authorized under CERCIA section 104(a)(1) for release of a "pollutant or contaminant" must be carefully evaluated and documented for the presence of an "imminent and substantial danger to the public health or welfare."

In a related issue, please note that methane control measures may be taken as part of a response action that controls a release of a hazardous substance. These measures would be taken to protect public health and the environment and to ensure the integrity of the remedy.

cc: Waste Management Division Directors, Regions I-IV and VI-X Environmental Services Division Directors, Regions I,VI and VII Regional Branch Chiefs Regional OHM-Coordinators Gene Lucero Russ Wyer Steve Lingle Jim Lounsbury Tim Fields Hans Crump Sherr, Hawkins

Implementation Strategy for Reauthorized Superfund: Short Term Priorities for Action



OCT 24 1986

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE 9200.3-02

MEMORANDUM

SUBJECT: Implementation Strategy for Reauthorized Superfund: Short Term Priorities for Action Winston Porter FROM: J.` Assistant Administrator Regional Administrator, Regions I - X TO: Regional Counsel, Regions I - X Director, Waste Management Division Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division Region II Director, Hazardous Waste Management Division Regions III and VI Director, Toxics and Waste Management Division Region IX Director, Hazardous Waste Division Region X Environmental Services Division Directors Regions I, VI, and VII

On October 17, 1986, the President signed the Superfund Amendments and Reauthorization Act of 1986 (SARA) amending the current "Superfund" law and enacting certain additional provisions. The attached transition guidance provides basic interpretations and instructions with respect to SARA.

SARA continues the process and program that was put in place with the revised National Contingency Plan (NCP) in November 1985. It contains a number of new provisions, however, that give statutory emphasis to some aspects of the existing program, or that add important new considerations. In addition, the new law requires the Agency to meet mandatory schedules for initiating and completing various remedial activities, and challenges us to efficiently manage a program that is much larger in size and scope.

As we move ahead with the new Superfund program, I want us to proceed along two tracks. The first involves strong emphasis on finishing work currently in the pipeline, particularly those projects in the final implementation phases. Completion of this and other ongoing work must incorporate careful consideration of the impact of SARA in a manner designed to minimize program disruption. The other track will involve initiating new work, as well as further refinement of our understanding of the SARA provisions.

This is the first in a series of memos that will provide direction for implementing the new Superfund program. Many of you will encounter policy issues in advance of guidance. Questions you face in the field will help drive our priorities for providing such quidance. I ask that you move forward aggressively to implement the new program and not wait until all the questions have been answered.

When you encounter an unresolved issue that may have national policy implications, please consult with the Headquarters contact appropriate to that issue. I have attached a list of contacts to assist in key areas.

The new Superfund will be a great challenge for all of us. It will require close working relationships within EPA, as well as with other Federal agencies, State and local governments, citizens groups, contractors, and industry. It will be interfative that we show major <u>results</u> with the program and funds entrusted to us. I look forward very much to working with all of you on this very important endeavor.

Attachment

ec:	Administrator
-	Deputy Administrator
	Associate Administrator for
	International Activities
	Associate Administrator for
	Regional Operations
	Assistant Administrator for
	Administration & Resources Management
	Assistant Administrator for
	Enforcement & Compliance Monitoring
	General Counsel
	Assistant Administrator for
	Policy, Planning & Evaluation
	Assistant Administrator for
	External Affairs
	Inspector General
	Assistant Administrator for
	Locistant Administrator for
	Addition
	ALL & ROUIDLINH Logistant Administrator for
	Decheides (Teris Substances
	Pesticides & Toxic Substances
	Kesearch & Tenetobmeur

CERCLA REAUTHORIZATION TRANSITION GUIDANCE

I. SCOPE OF GUIDANCE

This guidance specifically addresses the management of ongoing response actions (remedial and removal, Fund and enforcement) affected by the SARA. Although SARA includes a number of other new authorities (such as Emergency Planning and Community Right to Know) that are effective immediately, these provisions will be largely addressed in separate memoranda and guidance documents.

This document paraphrases some sections of the new law. Such paraphrasing is not meant to be interpretive of legislative language nor does it summarize complete sections of the law that may themselves be very lengthy. Please carefully review the attached summary and the actual provisions in order to understand fully the legislation.

This guidance is organized as follows:

- I. Scope of Guidance
- II. Effective Date of Provisions
- III. Removal Program Provisions and Priorities
- IV. Remedial Program Provisions and Priorities
- V. Enforcement Program Provisions and Priorities
- VI. Cross Cutting Provisions
- VII. Impact on Delegations

L. EFFECTIVE DATE OF PROVISIONS

All provisions of the new statute took effect on the date of enactment (October 17, 1986) unless otherwise specified by law. One area, however, where previous Agency decisions are "grandfathered" is with respect to records of decision (RODs) and consent decrees [\$121(b)(1)].

A. <u>Signed RODs and Consent Decrees - Prior to and within 30 Days</u> of SARA

RODS signed or consent decrees lodged <u>prior</u> to the date of enactment are not required to meet <u>new</u> requirements of \$121 (Clean-up Standards) unless the record of decision is reopened after the date of enactment to modify the remedy [\$121(b)(1)]. An Enforcement Decision Document (EDD) is the functional equivalent of a ROD; therefore the effective date provision applies to all signed EDDs, as well as RODs, where public participation was equivalent to that provided for RODs. (Where the EDD resulted in a consent degree, the grandfathering provision, of course, applies.) In the future, all selections of remedies for fund and enforcement lead sites will follow the ROD process. For RODs signed or consent decrees lodged within 30 days of date of enactment, EPA must certify in writing that the portion of the remedial action covered by the ROD or consent decree complies to the maximum extent practicable with §121 of the new law [§121 (b)(2)]. The certification responsibility is given by statute to the Administrator. Regions should consult with their Headquarters' contacts and take active measures to ensure that this certification can occur at the time of ROD signature.

B. RODs signed or Consent Decrees Lodged 30 days after SARA

RODs not signed within 30 days of enactment are required to comply fully with all new SARA provisions. In considering the new provisions, Regions should recognize that while cost effective remedies which protect human health and the environment continue to be required, the statute places a greater emphasis on the performance, long-term protectiveness and reliability of remedial actions. [See Section IV; ial Program Provisions.]

C. Implementation Considerations: Signed RODs for Operable Units

Projects in the design and construction phase at the time of enactment are the highest priority for Agency actions (see Section IV; Remedial Program). However, before proceeding, Regions should examine whether additional RODs are planned and assess the overall remedial strategy to ensure that future operable units are consistent with the new SARA requirements.

III. REMOVAL PROGRAM CONSIDERATIONS AND PRIORITIES

A. <u>New Provisions</u>

Three significant provisions of the new law are effective immediately and may have an impact on on-going and future removal operations.

1. Time and Dollar Limits

The new law raises the time and dollar limits for removal operations from six months and \$1 million to twelve months and \$2 million [\$104(e)(1)]. Although new time and dollar limits are effective immediately, the Regions are not currently delegated the authority to sign Action Memoranda above \$1 million. Until delegation to the Regions of additional authority, Headquarters' approval of ceiling increases and exemption requests above \$1 million will be required. Regions are already delegated the authority to approve extensions of any time limits. Any findings by the Region that an extension of time is needed must be made as early as practicable, and at least before expiration of the new statutory time limit of 12 months. Headquarters' review of non-delegated dollar limit extensions will be expeditious and will focus on consistency with criteria for removal actions and, in this immediate post-enactment period, on the availability of limited dollars to complete emergency actions. Regions have the discretion to re-evaluate on-going removals to determine if the scope should be changed under the new limits. At some sites, it is possible that a more efficient approach could be designed given the additional fund/time limits available.

2. Consistency Waiver

The new statute provides for an additional waiver to statutory limits which allows EPA to continue a removal action beyond \$2 million and 12 months where such action is "appropriate and consistent" with future remedial actions [\$104(e)(2)]. This waiver is available at both proposed and final NPL sites. Our current position is that it will not be used at non-NPL sites.

3. Contribution to Efficient Performance

Removal actions that take place after SARA are to be conducted in such a manner as to "contribute to the efficient performance" of long-term remedial measures "to the extent the President deems practicable" [\$104(b)(2)]. This provision promotes the performance of removal actions that more efficiently address threats by considering the overall site clean-up before the start of the action. The goal of this requirement is to reduce the need for removal restarts. The responsibility under this provision is effective immediately.

The Action Memorandum must include a specific discussion on how the proposed removal action meets this criterion. One situation where it may not be feasible to consider how the removal action contributes to the performance of the remedial action is in an emergency involving an immediate threat. In such cases, response personnel may need to take whatever immediate measures are required to protect the public health, welfare and the environment, and should document the reasons for taking the action without having first considered this criterion.

For on-going removals, response personnel should keep in mind the requirement that removals contribute to the efficient performance of long-term remedial measures, and take whatever steps are practicable under site-specific field circumstances to meet this requirement. Changes to on-going removal actions that take place in the course of exercising this responsibility should be documented in an amended Action Memorandum. This documentation should occur as soon as possible¹.

^{1/} As it is existing policy to ensure that removal actions contribute to the efficient performance of long-term measures to the extent practicable, this provision may have very little practical impact on signed Action Memoranda or on-going actions.

- If an Action Memorandum has been signed, the removal is on-going, and a new Action Memorandum is necessary to go beyond statutory limits. The new memo should address the degree to which this requirement has been addressed.
- If an Action Memorandum has been signed but a removal action not yet initiated, consideration should be given to amending the Action Memorandum, if this requirement is not already addressed prior to initiating the response action.
- If an Action Memorandum has not been signed, it must address this requirement.

B. Removal Program Priorities

Until SARA funding is available, we must continue to conserve funding for removal actions. We plan to provide \$2 million per month nationally to respond to the most serious emergency situations.

Regions should use this period to carry out preliminary activities that will allow on-site work to begin promptly when new CERCLA funding becomes available. Such activities include:

- Continue to conduct preliminary assessments under CERCLA section 104(b) authority at sites where removal action may be necessary.
- Coordinate and prepare Action Memoranda and secure the Regional Administrator's informal approval (not signature) for potential removal actions of less than \$1 million so that on-site activities can begin promptly when SARA funding becomes available.
- Prepare and submit to Headquarters draft ceiling increase requests (between \$1 million and \$2 million) and exemption requests (above \$2 million), so that coordination and informal approval can take place during the period of restricted funding. Review of draft requests will ensure that on-site actions can be initiated promptly and will be particularly important in obtaining timely Headquarters' approval of exemption requests based on the new "remedial consistency" waiver.
- Refer sites needing action and having responsible parties to Regional enforcement programs.

During fiscal year 1986, many removal actions were not initiated, others were demobilized and some actions were conducted at a reduced pace. As the removal program gears up with SARA funds, removal site priorities will have to be established by each Region. Available personnel and funding resources will have to be considered in setting these priorities.

IV. REMEDIAL PROGRAM PROVISIONS AND PRIORITIES

Highlighted below are <u>some</u> of the considerations that you will want to keep in mind as you proceed to incorporate the SARA requirements into on-going work. Paraphrasing of the statutory language -- particularly the cleanup standards section -- was necessary due to length. The statutory language and the Regional Counsel should be consulted for a more complete description of SARA's impact on the program.

A. Major Provisions

Clearly, the most important section of the law relating to the remedial program is §121, cleanup standards. This section codifies many of the existing requirements under the National Contingency Plan (NCP) but also adds new requirements, additional detail and direction. Some of the areas addressed include emphasis on treatment technologies in selection of remedies, meeting State standards, and formalizing the role of States j the cleanup process. The following discussion highlights the areas that should receive particular attention and consideration during the conduct of RI/PSs and development of RODs.

1. Applicable, Relevant and Appropriate Federal and State Requirements.

Section 121(a) and (d) establish the requirements for the degree of cleanup for remedial actions. The new amendments require that remedial actions conducted on-site shall meet the "applicable or relevant and appropriate standards, limitations, criteria, and requirements" (ARAR) of State and Federal environmental laws. Specific Federal environmental laws including, but not limited to, TSCA, SDWA, CWA, RCRA or MPRSA, are listed as potentially applicable to on-site clean ups. In addition, remedial actions are required to attain specifically identified standards, such as maximum contaminant level goals or MCLGs, formerly known as recommended maximum contaminant levels (RMCLs), established under the Safe Drinking Water Act and water quality criteria established under the Clean Water Act, when relevant and appropriate under the circumstances. (See statutory language, Section 121(d), for a more definitive description.)

The new law basically builds upon EPA's site-specific approach to cleanup standards (found in the NCP and in the CERCLA Compliance Policy) which requires remedial actions to meet the applicable or relevant and appropriate requirements of other Federal environmental statutes. Additional RCRA regulations that become effective both before and after reauthorization will themselves expand the specific requirements that SARA clean-ups have to meet. Some of the most significant requirements which can be applicable or relevant and appropriate to Superfund remedial actions are the land disposal ban provisions of HSWA. The land ban requirements could potential? have significant impacts on the cleanup levels, treatment technologies and the decisionmaking processes Superfund uses in remediating sites. Many substantive issues pertaining to these regulations and their impact on Superfund remedial actions and RCRA corrective actions remain to be resolved. It is clear, however, that these regulations can affect all projects in every stage of the remedial process. Headquarters will keep the Regions closely informed on policy development in this area.

The new law expands the list of potentially applicable or relevant and appropriate requirements to include promulgated State standards, requirements, criteria, or limitations. These State requirements should be addressed in the same manner that Federal requirements are currently. Under certain circumstances State ARARs need not be met [\$121(d)(2)(c)(ii), and (d)(4)(E)]. These circumstances include inconsistent application of State requirements, lack of formal promulgation of the requirement, and requirements that would effectively result in a statewide prohibition of land disposal.

The addition of State requirements, criteria, standards, and limitations as applicable, relevant and appropriate requirements requires EPA to obtain a complete picture of State requirements early in the RI/FS process. The Regions should develop their own process for obtaining information from the States on applicable, relevant and appropriate requirements. However, it is recommended that this include:

- A request to the State to notify EPA of the specific requirements that they think will be applicable or relevant and appropriate to each alternative under examination in the feasibility study.
- The above request made in writing, as early as possible, but not later than the time when the remedial investigation is 25 percent complete.
- Give the State a fixed time period for review of alternatives for which they are to identify ARARs and ask for an offical documented response.

2. <u>Preference for Permanent Solutions and Alternative</u> Treatment Technologies.

While the new provisions continue to require cost-effective remedies which protect human health and the environment, the statute places a greater emphasis on the long-term protection and reliability of remedial actions.

9200.3-02

The language calls for remedial actions which utilize permanent solutions and alternative treatment or resource recovery technologies to the maximum extent practicable. It establishes a preference for remedies in which treatment which permanently and significantly reduces the mobility, toxicity, or volume of waste comprises the principal element. If a remedy in accordance with the preference for treatment and permanent solutions is not selected, an explanation must be published [Section 121(b)(1)(G)].

Regions should collect sufficient data during the RI/FS to assess and compare treatment performance, reliability, and other operating parameters. As a goal, information should also be collected sufficient to attain an accuracy on costs of +50/-30 percent at the time of the ROD. Treatability studies frequently will be necessary prior to the ROD to properly evaluate treatment technologies and estimate costs. As appropriate, this would entail pilot scale testing (e.g., confirmation burns) or, in some situations, bench scale testing (e.g., for fixation).

In evaluating alternatives, Regions should closely examin.. 1) how effectively and significantly each alternative reduces the toxicity, mobility, and volume of the waste; 2) the degree of protectiveness and long-term reliability each alternative affords (including the uncertainties associated with land disposal); 3) the effect of applicable or relevant and appropriate land ban requirements on performance targets; 4) short-term impacts posed by each option; and 5) short-term and long-term costs of the alternatives, including capital, operation and maintenance, and replacement costs over the life of the remedial action [\$121(b)]. Present worth costs would then be calculated, as currently practiced.

3. State Involvement

SARA provides broad authority and an extensive list of requirements for State involvement in every phase of the Superfund program [\$121(f)]. Over the long term the Agency will develop and issue regulations concerning State involvement which could go beyond current Agency policy. In the interim, Regions should proceed to work with the States to inform them of new requirements and priorities as outlined in this memo.

The amended law generally does not alter State cost-sharing provisions.² Given the preference for treatment established in the new law, and the likely increase in the cost of remedies, the value of the State cost share of the remedial response action is likely to increase in many cases. Regions should begin discussions with their States now to alert them to the likely impact of the new law so that the States may begin to consider how to raise the additional funds that are likely to be necessary.

^{2/} Section 104(f) has the effect of requiring a 50 percent State match for State <u>operated</u> facilities.

States should be made aware that pre-emption by the original CERCLA on State creation of trust funds similar to "Superfund" has been removed in the new law [\$114(a)].

4. Waivers

The new law adopts many of the waivers contained in the existing NCP, but also adds some new ones. Compliance with ARARs can be waived when:

- The remedial action is an interim measure where the final remedy will attain the ARAR upon completion;
- Compliance will result in greater risk to human health and the environment than other options;
- Compliance is technically impracticable;
- Other remedial actions will attain "an equivalent standard of performance to that required under the otherwise applicable requirement, through use of another method or approach";
- For State requirements, the State has not consistently applied the State requirement, or demonstrated the intent to apply such requirements, at similar remedial actions; or
- For \$104 remedial actions where compliance will not provide a balance between the need to protect human health and the environment at a facility and the availability of Fund money for response at other facilities [\$121 (d)(4)].

The waivers for fund balancing, technical impracticality, and interim remedies remain essentially the same as in the NCP, while the waiver for enforcement cases with strong public interest has been eliminated. The waiver for unacceptable environmental impacts has been expanded and redefined as a waiver where compliance would result in "greater risk to human health and the environment." A new waiver is added for actions whose "standard of performance" is "equivalent" to a requirement through use of another method or approach.³ And another new waiver is added for State requirements that have not been consistently applied.

5. Health Assessments

Under the new law, a health assessment must be conducted by the Agency for Toxic Substances and Disease Registry (ATSDR) for every site on the National Priorities List (NPL) on the following schedule:

 By 12/10/88 for facilities proposed for the NPL prior to SARA; and

^{3/} The Conference Report states that this waiver allows "flexibility in the choice of technology, but does not allow any lesser stand or other basis such as risk-based calculation"... unless "the original standard is risk based."

 Within one year after proposal for facilities proposed for the NPL after SARA [\$110].

These assessments will assist EPA and ATSDR in determining whether action is required to be taken to reduce human exposure to hazardous substances, and whether additional information on human exposure and associated health risks (i.e. epidemiological studies by ATSDR) is needed. The assessments will evaluate the current and potential risk to human health posed by individual sites and facilities [\$110(3)(G)].

While ATSDR will provide an assessment of whether existing exposure provides a risk to public health, EPA will continue its risk assessment (public health evaluation and endangerment assessment) activities as part of its risk management responsibilities -determining and selecting the remedy.

Health assessments are not required to be completed before a project moves forward. However, ATSDR is required to complete health assessments "to the maximum extent practicable" before the RI/FS is completed. The highest priority for completion of health assessments should be any RODs the Region expects to sign within the next several months. Regions should coordinate with ATSDR to establish overall priorities and schedules for health assessments as well as work closely with ATSDR during the RI phases and development of alternatives.

.B. Remedial Program Priorities

As previously mentioned in Section II (C) of this guidance, projects in the construction phase and design projects at the time of enactment should proceed on schedule to the degree possible. One of the Regions' highest priority activities should be to examine these projects and assess their consistency with subsequent operable units given that these subsequent units will have to comply fully with the new law.

In addition, priorities for funding remedial projects will be determined according to the phase that the projects were in when SARA was passed. Priorities for remedial work should focus on the need to fund the construction pipeline. Projects nearest completion (remedial actions) will receive funding and staffing first, followed by remedial designs, and on-going RI/FSs. Although we recognize that some Regions will have a need to start new RI/FSs, for the time being the backlog of work to be done in ensuring that on-going projects meet or exceed the SARA requirements may cause new starts to receive a lower priority.

V. ENFORCEMENT PROGRAM PROVISIONS AND PRIORITIES

SARA includes a number of changes to the existing enforcementrelated provisions and adds a new section (§122) on settlement procedures. In general, the enforcement-related amendments adopt many of the provisions of the Interim CERCLA Settlement Policy and other existing enforcement program activities. The purpose of this section is to outline the significant new enforcement procedures and requirements and how these procedures will affect ongoing CERCLA activities.

A. SETTLEMENT PROCEDURES

1. RI/FS Special Notice Procedures

Section 122(e) establishes a new negotiation procedure for RI/FSs and RD/RAs which EPA may in its discretion choose to follow. This procedure involves issuance of "special notice" to PRPs followed by a moratorium on EPA action for a set time period. Although EPA may continue past notice and negotiation practices, it is expected that in most instances the negotiation procedure under \$122(e) will be followed.

EPA may issue "special notice" if it determines that negotiations would facilitate an agreement with potentially responsible parties (PRPs) to either undertake or finance an RI/FS. Special notice is required to include the following information, to the extent it is available:

- o Names and addresses of PRPs;
- o Volume and nature of substances; and
- o Ranking by volume of substances [\$122(e)(1)].

EPA must also provide notice to the State of negotiations with PRPs and provide an opportunity for State participation in the negotiations [\$121(f)(1)(F)]. If the release or threat of release at the site in question may have resulted in damages to natural resources, EPA must notify the Federal Trustee and provide an opportunity for the Trustee to participate in the negotiations [\$122(j)(1)]. To simplify the notification of Federal Trustees, the Agency plans to provide a list of projects in the SCAP to the Trustees as notice to participate in negotiations. Additional guidance on coordination with Federal Trustees will be developed.

The PRPs who receive special notice have 60 days to submit a proposal to undertake or finance the RI/FS [$\frac{22}{2}(2)$]. During this 60 day period, EPA may not initiate the RI/FS [$\frac{22}{2}(2)$]. Additional studies or investigations authorized under $\frac{104}{10}$ may be initiated and nothing precludes EPA's authority to undertake response or enforcement activity regarding a significant threat to the public health or the environment [$\frac{22}{2}(2)$]. The Regions may, under forward planning, initiate a scope of work or a negotiations support document⁴. The scope of work or negotiations

4/ Under forward planning, using TES or REM contractors, at a cost of up to \$50K per site and estimating a work period of approximately one month, the Regions may develop a site specific "negot' tions support document." In general, this work would include collecting background information, conducting a site visit and developing a scope of work. More detailed guidance on the scope and use of these documents will be forthcoming. support document should be provided to PRPs when notice is given so they can prepare an adequate proposal. Initiating the scope of work or negotiations support document will not constitute starting the RI/FS under the moratorium.

If a good faith proposal⁵ is submitted within 60 days of notice, the moratorium limiting initiation of the RI/FS continues for a total of 90 days from the date of notice.

If settlement is achieved, the agreement for the conduct or financing of the RI/FS must be in the form of an Administrative Order on Consent or a Consent Decree [122(d)(3)]. Administrative Orders on Consent are the preferred format for RI/FS agreements, except where an action has been filed in court.

The agreement may authorize the PRPs to conduct the RI/FS only when the following conditions are met:

- o EPA determines that the PRPs are qualified to do the RI/FS;
- EPA arranges or contracts for a qualified person to assist in overseeing the conduct of the RI/FS⁶; and
- o The PRPs agree to reimburse EPA for the cost of such oversight [\$104(a)].

The settlement agreement for the RI/FS need not contain a finding of imminent and substantial endangerment to the public health or the environment [\$122(d)(1)(A)].

If a good faith proposal is not submitted within sixty days of notice, or negotiations fail after submittal of a proposal, EPA may initiate the RI/FS [122(e)(4)].

- 5/ In general, a "good faith" proposal is a proposal in writing, in which the PRPs make a showing of their qualifications and willingness to conduct or finance the RI/FS which, and at least, addresses the major elements of the workplan or statement of work.
- 6/ The Statement of Managers refers to a "qualified person" as "someone with the professional qualifications, expertise and experience necessary to provide additional assurance that the President is conducting meaningful oversight." This person could be a State employee, employee of another Federal agency or any other "qualified person" EPA may contract with to perform the oversight. This provision does not replace the need for in-house EPA oversight, but rather supplements it, much like the current use of TES/REM contractors or the COE for oversight. Moreover, it is still EPA policy to require reimbursement for the cost of its oversight.

2. Ongoing RI/FS Activities

If notice letters have already been sent for RI/FS negotiation and Regions are prepared to negotiate or have inititiated negotiations, parties do not have to receive another notification regarding their liability for the conduct of the RI/FS unless the Regions desire to use the moratorium procedures in \$122(e). The Regions must bear in mind, however, that the remedy eventually selected for these sites must comply with \$121, cleanup standards.

a. Ongoing Negotiations for RI/PS

In ongoing negotiations for RI/PSs, PRPs should be informed of the requirements of \$121 for cleanup standards. In particular these include, but are not limited to, consideration of alternatives that meet ARAR requirements, the statutory preference for permanent remedies, and notification and involvement of States in determining ARARs and concurring on remedy selection. EPA must also arrange for a qualified person to assist with the oversight of the RI/FS and the PRPs must agree to reimburse EPA for that oversight.

b. Ongoing RI/FS

An RI/FS currently being conducted by PRPs should be reviewed to assure that alternatives evaluated include those that comply with §121. This review will be similar to the evaluation that will be conducted for ongoing fund-financed RI/FSs. (See section on cleanup standards.) Sections of Administrative Orders on Consent or Consent Decrees may need to be revised to reflect that the final remedy must meet the requirements of §121. A letter should be sent to the PRPs informing them of the new provisions and providing them with an opportunity to discuss the changes to the Administrative Order or Consent Decree. (A sample letter is being developed and will be sent to the Regions shortly.)

c. New Negotiations for RI/FS

As EPA's current policy suggests, all notice letters should be issued to the PRPs as early as possible. Responsible party searches should be conducted concurrently with the expanded site inspections (ESI) and notice letters should generally follow shortly after proposal on the NPL. If the Region chooses to invoke the §122(e) "special notice" and negotiation procedure, the notice letter should specifically reference that fact and explain that the 60-day timeframe in which the PRP must make an offer begins with receipt of the notice letter. Using the notice letter as the vehicle to begin the moratorium on initiation of the RI/FS should avoid any potential delays when EPA is actually ready to commence the RI/FS. A model administrative order on consent and a detailed scope of work for the RI/FS should be sent to the PRPs at the time notice is given. If the Region chooses not to invoke the \$122(e) procedure, the notice letter should state the reasons why the procedure is inappropriate [\$122(a)].

- - -

3. RD/RA Procedures

If EPA decides to invoke the negotiation procedures in §122(e) for the remedial action, EPA must again provide special notice to the PRPs and provide information on volume, nature and ranking of wastes. The Statement of Managers indicates that this information "should be routinely made available at this time". This is a separate notice and information release from the RI/FS notice. State and Federal Natural Resource Trustees must also be notified and provided an opportunity to participate in the negotiations. Notice for RD/RA negotiations should generally be given as early as possible, but no later than when EPA has identified a "preferred" remedy.

Again, as with the RI/FS procedures, the PRPs have sixty days to make a good faith proposal to conduct or finance the remedial action. A good faith proposal is a proposal in writing, in which the PRPs make a showing of their qualifications and willingness to conduct or finance the major elements of the ROD. During these sixty days, EPA may not initiate remedial action under \$104(a) or under \$106. Additional studies authorized under \$104(b), however, including remedial design may be initiated during the negotiation period.

If a good faith proposal is submitted, the moratorium on initiation of a \$104(a) response action or the issuance of \$106 Administrative Order or the filing of a \$106 civil action continues for 120 days from the date of notice.

An agreement with PRPs for remedial action must be in the form of a consent decree. Several sections of the new statute affect the terms which can be included in such a decree. First, \$121(e)(2) requires that the decree contain stipulated penalties. These penalties are in addition to the penalties which can be collected under \$122(e). Second, because \$113(j)(2) establishes that EPA decisions relating to remedial measures are to be judged under the arbitrary and capricious standard, all dispute resolution provisions must provide that in any dispute concerning the response action the court will uphold the EPA response decision unless the objecting party can demonstrate on the administrative record that the decision was arbitrary and capricious or otherwise not in accordance with law. Finally, in \$122(f) Congress has established the conditions upon which a covenant not to sue can be granted. A separate section below explains how this statutory provision affects settlement terms.

The proposed agreement must be filed with the Court at least 30-days before final approval is sought from the Court and persons not party to the agreement must have an opportunity to comment before final judgment. [\$122(d)(2)(B)]. This requirement is consistent with current Department of Justice (DOJ) practice [28 CFR 50.7]

It is important to note here that under the new statute, EPA retains its authority to issue unilateral administrative orders for RD/RAs, if the settlement fails or EPA has decided not to pursue a settlement.

a. Application to Ongoing Negotiations

Except for the RODs/EDDs which are subject to the "grandfather" provision, any remedies which are the subject of ongoing negotiations for RD/RA must also be evaluated for compliance with \$121. Particular attention should be paid to the application of State standards, preference for permanent remedy, use of alternate concentration limits (ACLs) and cost effectiveness.

Any ongoing negotiations for RD/RA that were proceeding in the form of an Administrative Order on Consent must be redirected to a Consent Decree. These Consent Decrees must incorporate the new statute's provisions on covenants not to sue, stipulated penalties, and deference to Agency decisionmaking. The Department of Justice should be notified of these cases and brought into the negotiations.

b. Application to New Negotiations

If "special notice" is given to provide the PRPs an opportunity to undertake the RD/RA, EPA anticipates that initiation of the remedial design during the first 60-day period will only occur in exceptional circumstances. Initiation of remedial design during this period must have advance concurrence from Headquarters. If the PRPs submit a good faith proposal, initiation of the design during the second 60-day period should again be the exception and requires concurrence from Headquarters.

4. Releases or Covenants not to Sue

Section 122(f) authorizes EPA to provide to PRPs, in certain circumstances, covenants not to sue for any liability, including future liability, under CERCLA for a release or a threatened release of a hazardous substance addressed by a remedial action. This provision adopts with only a few changes, the guidance on this subject set forth in the Interim CERCLA Settlement Policy. Releases from liability are to be in the form of a covenant not to sue.

- 1) Effectiveness and reliability of remedy;
- 2) Nature of risks remaining;
- 3) Extent performance standards are in decree;
- 4) Extent action provides complete remedy;
- 5) Extent technology used is demonstrated to be effective;
- 6) Whether source of funding would be available for any additional remedial actions; and
- 7) Whether action will be carried out in whole or significant part by PRPs.

Section 122(f)(2) makes it mandatory that covenants not to sue for future liability be provided under certain circumstances for redisposal liability and for the portion of the remedial action which involves the permanent destruction or treatment of the hazardous waste. No covenants not to sue for future liability, however, may take effect until EPA certifies that the remedial action has been completed and any covenant issued under \$122(f) is conditioned upon satisfactory performance of the remedial action.

If a covenant not to sue for future liability is not 'mandated under 122(f)(2), the covenant must include a reopener clause which allows EPA to pursue PRPs concerning a release or threat of release that arises out of conditions which were unknown at the time EPA certified that the remedial action was completed [122(f)(6)(A)]. The reopener should not be conditioned on the presence of an imminent and substantial endangerment. EPA is also authorized to include any other terms in the reopener necessary to protect public health, welfare, and the environment [122(f)(6)(C)]. In "extraordinary circumstances," the reopener clause may be omitted if the terms of the agreement are sufficient to provide assurance that public health and the environment will be protected from future releases [122(f)(6)(B)]. Regions should still include the second reopener for "new scientific information" contained in the Interim Settlement Policy.

The new language on covenants not to sue is effective immediately and should be included in all consent decrees involving remedial action. At a minimum, consent decrees must specify that any covenant not to sue for future liability does not take effect until the remedial action has been completed, that the covenant not to sue is predicated upon satisfactory performance of the remedial work, and that the reopener is not limited to imminent and substantial endangerment situations. Additional guidance on covenants not to sue is being developed. Until such guidance is available, Regions must consult with Headquarters in advance of any agreement that will provide a mandatory covenant not to sue under \$122(f)(2) or contain a no reopener clause on the basis of the "extraordinary circumstances" provision.

5. Additional Parties

If during the course of "special notice" negotiations, additional PRPs are identified, EPA may bring those parties into the negotiations [\$122(e)(2)(C)]. However, the addition of new parties does not affect the original date of notice and commencement of the moratorium provision. (This applies to RI/FS negotiations as well.)

B. Additional Enforcement Related Amendments

1. Contribution Protection

Section 113 of CERCLA has been amended to provide contribution protection statutorily to PRPs for matters addressed in an administrative or judicially approved settlement. The settlement reduces the potential liablity of other PRPs by the amount of the settlement. Since contribution protection is now provided by law, it is no longer appropriate to include such a provision in the consent decree.

2. Pre-enforcement Review

Section 113(h) has been amended to include language on preenforcement review. The amendments state that Federal court jurisdiction to review challenges to removal and remedial actions is limited to the following cases:

- o Actions under \$107;
- o Actions to enforce an order under \$106(a);
- o Action for reimbursement under \$106(b)(2)⁷;
- o Action under \$106 where U.S. has moved to compel remedial action; and
- o Action under \$310 (Citizen Suits) that alleges the response action was in violation of CERCLA.

^{7/} Any person who receives and complies with an Administrative Order may petition the Agency for reimbursement within 60 days after completion of the required action. Reimbursement may be obtained if a party shows by a preponderance of the evidence that it is not liable under \$107 or if the party can demonstrate that the action ordered was arbitrary and capricious.

With the exception of the new citizen suit provision and the new provision on reimbursement, this new language confirms the Agency's current position that parties may not take any action challenging the Agency's selection of remedy before an enforcement action is taken. If the parties challenge the selection of remedy following an enforcement action, that challenge is limited to judicial review of the administrative record.

3. Judicial Review/Administrative Record: \$113(j)

SARA limits the judicial review of adequacy of any response action taken to the administrative record. Judicial review, including review of RODs signed before the date of enactment, will be on the administrative record. EPA is required to establish an administrative record which is to be the basis for the selection of any response action. An administrative record is required for <u>all</u> response actions: removal and remedial, fund-financed and enforcement. The record must also be available for public review.

Until regulations on the administrative record are promulgated under \$113(j), the administrative record consists of all items considered by the Agency in selecting the response actions and includes items developed and received under the current procedures for selection of the response action including public participation procedures. These existing procedures are set out in the June 1985 RI/FS Guidance and the February 1985 Draft ROD Guidance.

The Agency will be developing detailed guidance on what documents need to be maintained in the record and develop, where necessary, a process for organizing the record for sites where the response action has been selected.

Because the record is the only basis for review of the selection of the response action by PRPs or citizen suits, it is extremely important that the Agency properly document and maintain all the information it uses for making the selection of response action. The Regions must closely adhere to the procedures outlined in the RI/PS and ROD Guidance.

C. Priorities for Enforcement Activities

The Regions, in consultation with Headquarters, will need to re-evaluate ongoing enforcement activities and develop priorities for assuring compliance with the new amendments. The following should be the first priorities for Regions in re-evaluating their enforcement activities:

 Review of sites scheduled for RI/FS and RD funding in the first and second quarters of FY 87. Regions must make sure that proper notice and information exchange with PRPs has occurred. (This should be consistent with SCAP targets.)

- 2. Review draft settlement documents to ensure incorporation of the provisions discussed in V.A.3.
- 3. Review Agency position in ongoing negotiations for RD/RA (where consent decree has not been lodged or ROD/EDD has not been signed) and assure consistency with new requirements.
- Review of PRP conducted RI/FSs that are nearing completion. PRPs must be informed of the new statutory requirements and Regions must assure that selection of remedy will be consistent with \$121.
- 5. Review all other RI/FS being conducted by PRPs under administrative Orders or Consent Decrees. PRPs must be notified of new requirements and appropriate revisions to the Order or Decree to reflect new requirements should be made.

D. Additional Enforcement Provisions

SARA includes several other provisions that affect enforcement activities. In general, however, these additional provisions will not be employed by the Agency until they have been delegated from the President to the Agency and the Agency has developed policies and guidelines for their use. A brief description of these provisions has been provided.

1. Response Action Contractor (RAC) Indemnification

Section 119 gives EPA discretionary authority to indemnify RACs against liability (including the expenses of litigation or settlement) for negligence arising out of the RAC's performance in carrying out response action activities under CERCLA [\$19(c)(1)]. The amendment does not allow EPA to indemnify RACs whose conduct or activities are deemed to involve gross negligence, intentional misconduct, or for conduct for which they are strictly liable under State law [\$119(c)(1)]. RCRA facility owners and operators and publicly owned treatment works (POTWs) are also precluded from EPA indemnification [\$119(c)(5)(D)].

2. Non-Binding Preliminary Allocation of Responsiblity

Section 122(e)(3) of SARA requires the Agency to develop guidelines for preparing non-binding preliminary allocations of responsiblity (NBARs) for PRPs. These guidelines may include such factors as volume, toxicity, mobility, strength of evidence, ability to pay, litigative risk, etc. The Agency, in its discretion, may, after the RI/FS, provide an NBAR to the PRPs. NBARs are not admissable as evidence, are not subject to judicial review and do not constitute an apportionment or other statement on the divisibility of harm or causation. Working with the Regions, Headquarters is currently developing the guidelines for NBARs and plans to develop several pilot projects beginning early next year.

3. De Minimis Settlements

Section 122(g) of SARA authorizes EPA to reach final settlements with PRPs if the settlement involves a minor portion of the response costs and the waste sent to the site by the PRP is minimal in comparison to the other hazardous substance at the facility in terms of amount and toxicity. Final settlements also may be entered with landowner PRPs if the landowner did not conduct or permit the disposal of hazardous waste at the site, did not contribute to the release of hazardous substances by an act or omission, and did not buy the property with the knowledge that waste had been disposed of at the site. PRPs claiming a defense to liability as opposed to being less culpable than other PRPs must meet the new requirements of \$101(f) to establish that defense.

These "de minimis" settlements may be in the form of an Administrative Order or a Consent Decree. Administrative Orders for facilities where total response costs exceed \$500,000 must have the written approval of DOJ.

The Agency has several "de minimis" pilot projects underway and is concurrently developing Agency-wide guidance for "de minimis" settlements.

4. Cost Recovery Settlements

Section 122(h) authorizes any agency with authority to respond, to compromise and settle claims under \$107, if the claim has not been referred to DOJ. DOJ must give written approval of any claim compromised where the total response costs exceed \$500,000.

Procedures and guidance on compromising claims are under development.

VI. CROSS-CUTTING PROVISIONS

A. Scope of Response Action

SARA contains three prohibitions on response actions that generally reflect existing agency policy. Fund-eligible response actions are generally prohibited with respect to:

- Releases of naturally occurring substances;
- Releases from products which are part of buildings or structures and result in exposure therein; and
- Contamination of drinking water supplies due to normal deterioration of the system.

The Agency may respond to these situations in cases of emergency where no other party can respond in a timely manner [\$104(c)].

B. Off-Site Policy

The new statute reflects the current off-site policy in most respects. The existing off-site policy remains in place as is, except as altered by the Statute. If aspects of the off-site policy are more stringent than statutory requirements, these provisions remain in effect.

The statute requires that hazardous substances, pollutants and contaminants be disposed of off-site only at facilities in compliance with (i.e. having no significant violations) RCRA Subtitle C or TSCA or other applicable Federal laws where appropriate, and applicable State requirements [§121(d)(3)].

Disposal at off-site land disposal facilities is further restricted in that:

- The unit receiving the waste must have no release into ground water, surface water, or soil (other than de minimus releases into soil); ⁸ and
- Any releases from other units must be controlled under an approved corrective action program (either through a permit or administrative order).

With respect to the above statutory conditions, Regions will need to examine commercial facilities to determine if there are significant violations of State standards, or if there are releases from the proposed receiving unit, in order to determine whether the off-site facility is eligible to receive CERCLA waste. (Please note, as per previous guidance, the fact that the facility is in assessment monitoring does not mean that the facility is ineligible. Evidence of a release determines eligibility/ineligibility.) In addition, Regions must examine whether there are releases from other units (regulated units or solid waste management units). If such releases are present, the facility must be under an order or permit schedule of compliance to correct such releases in order to be eligible to receive CERCLA waste.

The statute also requires that notice of ineligibility determinations made under the Off-Site Policy be given to facilities. Until regulations are developed, notice must be given per the statutory requirements. Those requirements are met by following the notice procedures set forth in OSWER Directive Number 9330.2-05 (CERCLA Off-Site Policy: Providing Notice to Pacilities; May 12, 1986).

^{8/} The Conference Report states that this language "is intended to preclude transfer or disposal of hazardous waste or constituents thereof into unlined units and lined units with releases other than de minimis releases into soil."

C. Technical Assistance Grants

The new amendments provide authority to issue technical assistance grants of up to \$50,000 to "any group of individuals which may be affected by a release or threatened release at any facility which is listed on the NPL..." [Section 117(e)]. SARA requires that rules be issued governing these grants. These rules are under development and we plan to issue them as interim final regulations. Information on how citizens can apply for the grants will also be issued at that time.

D. Leaking Underground Storage Tank Trust Fund

SARA also contains a section amending Subtitle I of RCRA to establish a Leaking Underground Storage Tank Trust Fund which is to pay costs incurred for corrective action and enforcement action resulting from responses to leaking underground petroleum storage tanks. These amendments also require establishment of financial responsibility by private parties for purposes of corrective action and compensation resulting from accidental tank releases [§205].

In general, responsible parties will provide the first line of response action for releases from leaking tanks. When responsible parties are not available to conduct the response, States, acting under cooperative agreements will determine the need for and type of response actions. The Federal government will conduct responses only where the release constitutes a major public health emergency and no State authority or responsible party is able to respond in a timely manner. We anticipate that a Federal response will be minimal. In addition to responding to emergencies, a major priority for the Regions will be to negotiate cooperative agreements with the States.

Because the removal contracts and program personnel are experienced in conducting response actions, responsibility to undertake these few Federal actions will be assigned to them. The new Emergency Response Clean-up Services contracts and Technical Assistance Team contracts have been modified to include responses under Subtitle I of RCRA using UST Trust Fund appropriations. These contracts will be available for use later this fall.

UST program authorities are new authorities that are currently not delegated to the Regions. These provisions [Section 205 of SARA] amend the Solid Waste Disposal Act and are given, by statute, to the Administrator. Headquarters' sign-off will therefore be required on response actions in this area.

VII. IMPACT ON DELEGATIONS

A. Executive Delegations

Like CERCLA, the new law provides direct authority, in most cases, to the President rather than directly to EPA or another Federal agency. Some authorities in the new law will, therefore, have to be delegated to EPA and other Federal agencies through a revision to Executive Order 12316 before they can be implemented.

The new law affects current delegation of authority as well as future delegations of new authority. A list of authorities that have been delegated, as well as those that are new or changed, will be provided shortly.

B. Internal Delegations

Existing internal delegations also remain in effect unless they are in conflict with the new law, and/or are unavailable for delegation until the Executive Order is revised. The language of each existing internal delegation will be evaluated.

If an existing internal delegation is very specific, and a new provision is beyond the scope of the delegation, it is not automatically delegated to the Region and must be delegated.

LIST OF CONTACTS

TRANSITION GUIDANCE

OVISIONS	NAME & TITLE	PHONE
Office of Emergency & Remedial Response		
Overview: Program Implementation/Guidance	Clem L. Rastatter Executive Officer	382-2 180
Clean up Standards/ Permanent Remedy	Tom Sheckells, Chief Remedial Analysis Branch	382- 2339
Applicable Appropriate & Relevant Standards	Arthur Weissman, Acting Chief Policy Analysis Staff	382-2182
State Issues	Sam Morekas, Chief State & Regional Coordination Branch	382-2443
Health Authorities	Elaine Stanley, Deputy Director Hazardous Sites Control Division	382-4632
Grandfathering of Section 121	Elaine Stanley, Deputy Director Hazardous Sites Control Division	382-4632
vmoval Provisions	Hans Crump, Chief Response Operations Branch	382-2188
UST Romoval Actions	Hans Crump Response Operations Branch	382-2188
Office of Waste Programs Enforcement		
Overview: Program Implementation/Guidance	John Cross, Chief Guidance & Oversight Branch	475-6770
Settlements/Notifications	Janet Farella, Chief Oversight & Documentation Section	382-2034
Administrative Records	Janet Farella, Chief Oversight & Documentation Section	382-2034
Indemnification & Response Action Contracts	Bob Mason, Chief Guidance Section	382-4 015
Non-binding Allocation of Responsibility	Debbie Wood Policy Coordinator	475-8715

Guidance on Implementation of the "Contribute to Remedial Performance" Provision

	wasning Discosting	non, us 2000.	• 9360.0-13			
Jean Schumann	Mail Code WH-548/B	Office OSWER, OERR, ERD	Telephone Number 382-2190			
Guidance on Implementation of the "Contribute to Remedial Performance" Provision						
Immery of Directive (Include brief statement of purpose) Provides guidance to Regions on implementation of the SARA provision that requires removal actions to contribute to the efficient performance of long-term remedial actions.						
Superfund, CERCLA, SA	RA, Removal	Action, Contribute to	Remedial Performance			
es It Supplement Previous Directive(s)? A Yes No What Directive (number, tn/e) Implementation Strategy for Reauthorized Superfund: Short Term Priorities for Action						
ett Level A — Signed by AA/DAA B — Signed by Office Director C — For Review & Comment In Development						
Request Meets OSWER Directives System F	18/mnc		· · ·			
Insture term Office Directives foordinato			4.3.87			
. Winston Porter, Assistant	Administrat	cor/OSWER	Dete 4/6 /37			

SWEROSWEROSWERDIRECTIVEDIRECTIVE

GUIDANCE ON IMPLEMENTATION OF THE

"CONTRIBUTE TO REMEDIAL PERFORMANCE" PROVISION

1. INTRODUCTION

Section 104(b) of the Superfund Amendments and Reauthorization Act of 1986 (SARA) amends section 104(a) of CERCLA to include the statement that any removal action undertaken by the President, or by any other person referred to in section 122* of the new law, should, to the extent practicable, contribute to the efficient performance of any long-term remedial action with respect to the release or threatened release concerned. This guidance document explains how to implement this provision, and includes guidelines on the applicability of the requirements, the definition of "contribute to efficient performance," exceptions, documentation and coordination. This document should be used in conjunction with the general removal procedures described in the Superfund Removal Procedures -- Revision Number Two, August 20, 1984, or, as may be amended.

2. APPLICABILITY

This provision will be applicable to removal actions at all sites -final National Priorities List (NPL), proposed NPL, and non-NPL: The term "long-term remedial action" as used in this provision will therefore refer ") a remedial action to be taken by the EPA. State, or a private party.

J. DEFINITION OF "CONTRIBUTE TO EFFICIENT PERFORMANCE"

3.1 Purpose

This provision promotes the performance of removal actions that address threats more efficiently by considering the overall site cleanup before the start of the action. To the maximum extent practicable, removal actions should be designed to avoid wasteful, repetitive, short-term actions that do not contribute to the efficient, cost-effective performance of long-term remedial actions to be taken by the EPA, State, or other party. The major objective of this requirement is to provide maximum protection of public health and the environment at minimal cost by avoidance of removal restarts. The focus of this provision is on avoidance of restarts that are due to recurring threats that were not adequately abated in the original removal action, and threats from deteriorating site conditions that should have been foreseen.

There are other circumstances, however, where removal restarts may be necessary to meet program goals. For example, a removal action may be a phased response. The first removal action might involve site stabilization and waste

Section 122 refers to potentially responsible parties (PRPs) who have entered nto settlements with EPA.
characterization. The site may then be demobilized and closed out to allow removal personnel to prepare an analysis of waste treatment/disposal options. Once an option is selected, a removal restart would be implemented to complete the waste disposition phase. In this case, the removal restart would actually contribute to achieving a more efficient cleanup. Removal restarts may also occur in an attempt to meet other program goals, such as pursuing responsible party (RP) cleanups or State assumption of removal action operation and maintenance requirements. An RP may take over a removal action from EPA, but EPA may have to initiate a restart if the RP is not performing an adequate cleanup. The "contribute to efficient performance" provision was not intended to conflict with these other program goals. As stated above, the provision was intended to reduce removal restarts due to inadequate planning at the start of the action.

3.2 Implementation

To meet the goal of avoiding removal restarts, response personnel must adequately assess the threats posed by the hazardous substances on a site and consider how the removal action would most effectively contribute to the longterm remedy. The following questions should be considered:

1. What is the long-term cleanup plan for the site?

This provision requires removal actions to contribute to the performance of the "long-term remedial action." At an NPL site, if the Record of Decision has already been signed, then comparing the removal action to the remedial cleanup plan is a straightforward task. However, for proposed NPL sites and for many final NPL sites, the remedial action may not have been selected when the removal action is implemented. In these cases, response personnel will be limited to identifying a range of feasible remedial alternatives. Response personnel need only review existing site information and use their best professional judgment. Removal and remedial personnel in the Region must coordinate with each other in this effort. It is the responsibility of the Region to establish appropriate coordination mechanisms.

At non-NPL sites, response personnel should, where practicable, consult with the party performing the long-term response action at the site (e.g., State, RP) to determine the proposed approach for the long-term cleanup. It is recognized that it may be more difficult to ascertain the remedial action at non-NPL sites. Response personnel should use their best efforts to coordinate with the party performing the long-term remedy. At many non-NPL sites, there may be no plans for another party to conduct a remedial action.

2. Which threats will require attention prior to the start of the long-term action?

The February, 1986 National Contingency Plan (NCP) broadened removal authority by allowing removal actions to be taken in response to "threats" rather than just "immediate and significant" threats. This expanded authority will allow a removal action to address any near-term threats that may arise prior to the start of the long-term action, provided the threats meet the removal criteria established in section 300.65 of the current NCP. Potential threats should be identified when the first removal action at a site is implemented to avoid the need for future removal restarts. Therefore, in addition to identifying immediate threats, response personnel should also identify potential near-term threats from contaminant migration, deteriorating site conditions, etc. This assessment is particularly important if a decision is made to leave surface hazardous substances on site after the removal action is completed.

- 1-

UNITED BUILDENTE

Response personnel must identify threats that may arise prior to the start of long-term actions, but the length of time before longterm actions will begin will vary from site to site. For example, for NPL sites where a ROD has been signed, the time frame that response personnel must consider will be shorter than for NPL sites where the Remedial Investigation/Feasibility Study (RI/FS) has just been initiated. Of course, at some NPL sites, the remedial program may plan to conduct an operable unit during the RI/FS if time permits. In this case, the time period to consider would again be shorter. Response personnel should consult with the party performing longterm action at the site to determine when such action will begin, and use their best professional judgment.

At non-NPL sites where there are no plans for another party to undertake a long-term cleanup, all threats and potential threats that meet the removal criteria in the current NCP should be identified.

3. How far should the removal action go to assure that the threats are adequately abated?

The expanded authority in the 1986 NCP will allow more complete removal actions to to be taken. Removal actions no longer have to stop when emergency situations are mitigated, but can continue, or be initiated, where needed to ensure that near-term threats are adequately abated. Measures that provide only temporary protection, insufficient to last until long-term actions begin, should be avoided to the extent possible. However, as noted above, consideration must also be given to the availability of other response mechanisms (e.g., State action, remedial operable unit) to initiate long-term action in a timely manner.

Whether or not the removal action should address all surface hazardous substances must be decided on a site-by-site basis. A removal action would be appropriate whenever surface hazardous substances may present a threat (as established in section 300.65 of the current NCP) before the start of long-term action. How the removal action should address the surface hazardous substances will also depend on site-specific conditions and the long-term cleanup plan. With the increased emphasis on using alternative technologies and new restrictions on land disposal, remedial actions may often include onsite treatment if surface contamination is extensive. In this case, the removal action may consist of consolidating and stabilizing the substances on site to await treatment. It is important to design the removal action to ensure that the materials are adequately stabilized. At other sites, surface hazardous substances may constitute only a small part of the problem: may not be safely stabilized for a long period of time; or may be more efficiently addressed as one unit by immediate treatment or disposal. In these situations, it may be more appropriate for a removal action to include final disposition of all surface hazardous substances. The conditions at the site and the longterm cleanup plan will determine the appropriate scope of the removal response.

At non-NPL sites where there are no plans for another party to perform long-term remedial action, the threats that meet the removal criteria in the current NCP should be completely cleaned up, if possible. The avoidance of removal restarts due to recurring threats is the ultimate goal. If mitigation of the threats that meet the NCP removal criteria results in complete site cleanup (i.e., no further Federal response required), the "contribute to efficient performance" provision is fully satisfied.

In considering all of the factors described above, the major determinant of how far the removal action should go to assure threats are adequately abated will be the statutory limits on removal actions. Removal actions should contribute to the efficient performance of remedial actions to the maximum extent practicable given the \$2 million/ 12 month limits on removal actions. (An exemption to the limits may be granted where the site qualifies under the "emergency" or "consistency" waivers.)

With regard to cleanup standards, this provision does not compel the removal program to lower its cleanup standards. Rather, the purpose of this provision is to improve the design of removal actions such that after cleanup standards are established for a removal site, the chosen removal action will address those substances targeted for cleanup in a manner that avoids the need for removal restarts. For example, the removal program has historically used 50 ppm as a benchmark in determining the appropriate extent of cleanup of PCBcontaminated soil. The "contribute to efficient performance" provision would not affect this number, but would direct that the method chosen to address soil contaminated above 50 ppm should be designed to avoid the need for removal restarts to the extent practicable.

4. Is the proposed removal action consistent with the long-term remedy?

The removal action that is chosen should be consistent with long-term actions at those sites where further cleanup will be taken. "Consistent" is defined in its broadest sense and may be characterized as a range of possible approaches. At one end of the spectrum, removal actions may be found consistent if they do not hinder or interfere with the remedial action to be taken. At the other end of the spectrum, removal actions may be found consistent because they contribute in a positive way to the long-term cleanup plan. For example, a removal action to provide carbon filters to homes with contaminated drinking water as an interim measure would not interfere with a long-term remedial plan to clean up the contaminated aquifer. A removal action to solidify sludge could, however, hinder a long-term plan to incinerate the waste and should, therefore, be avoided <u>if other approaches are feasible</u>. A removal action to remove surface drums from a landfill could contribute in a positive way to a remedial plan to clean up the site.

- 2 -

Removal actions may be found consistent if they fall anywhere within this range; the most appropriate approach will depend on site-specific factors. It is recognized that in some cases, the removal action may create additional work for the remedial action and yet still be the most appropriate approach for the site. For example, a common removal action is capping contaminated soil to prevent migration and human contact in the time period before remedial actions begin. Although the cap would have to be removed to implement a long-term plan to excavate and treat the soil, it may still be the most effective method to mitigate the threat in the short-term. Protection of public health and the environment, as well as technical feasibility, must always be considered. If such an action is selected, the rationale for selection should be explained in the Action Memorandum. (See Section 5.)

The answers to these four questions will help determine what type of removal action is needed and how it can be designed to contribute to the efficient performance of long-term remedial actions. These questions are

ided as general guidelines to indicate the various factors that should considered in implementing this provision of SARA. A written analysis of the answers to each of these questions is not required. The <u>conclusions</u> should be documented in the Action Memorandum. (See Section 5.)

4. EXCEPTION

The only situation where it may not be feasible to <u>consider</u> how the proposed removal action relates to the long-term remedy is in an emergency. In such cases, response personnel may need to take whatever immediate measures are required to protect the public health, welfare, and the environment.

5. DOCUMENTATION AND COORDINATION

The Action Memorandum should specifically cite the "contribute to efficient performance" requirement and briefly discuss how the proposed removal action relates to long-term remedial actions, to the extent practicable. (See the Superfund Removal Procedures for information on the preparation of Action Memoranda.) If the proposed removal action completes the cleanup and no further action is required, this should be so noted. If only minimal information is available about long-term actions, this should also be explained. If an emergency existed that precluded an analysis of how the removal related to long-term actions, this should be noted. Finally, if "pliance with this provision would conflict with other program goals (e.g., suit of RP cleanup), this shoud be explained. Compliance with this requirement does not require special approval; the Action Memorandum should be approved by the established concurrence chain in the Region or in Headquarters, if appropriate. In making the determination, however, it will be the responsibility of the OSC to coordinate with the party that will undertake the long-term remedy (for those sites where additional cleanup measures will be taken).

Scope of the CERCLA Petroleum Exclusion Under Sections 101(14) and 104(a)(2)



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON DC 20460

JUL 3 1 1987

OFFICE OF GENERAL COUNSEL

MEMORANDUM

SUBJECT:	CERCLA Petroleum Exclusion
FROM:	Lisa K. Friedman Associate Genera Solid Waste & Emergency Response Division

TO: Addressees

Attached is OGC's long-awaited memo on the scope of the petroleum exclusion under CERCLA. If you have any questions about it, please call me (382-7706), Mark Greenwood (382-7703), or Carrie Wehling (382-7706).

Attachment

Addressees: Gene Lucero (WH-527) Lloyd Guerci (WH-527) Tom Adams (LE-133) Steve Leifer (LE-134S0 Regional Counsels, Regions 1-X Dave Buente, DOJ Peggy Strand, DOJ



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY . WASHINGTON D.C. 20160

JUL 31 1987

SFFICE OF SEVERAL COUNSEL

MEMORANDUM

SUBJECT: Scope of the CERCLA Petroleum Exclusion Under Sections 101(14) and 104(a)(2)

- FROM: Francis S. Blake F. S. Blake General Counsel (LE-130)
- TO: J. Winston Porter Assistant Administrator for Solid Waste and Emergency Response (WH-562A)

One critical and recurring issue arising in the context of a Superfund response activities has been the scope of the petroleum exclusion under CERCLA. Specifically, you have asked whether used oil which is contaminated by hazardous substances is considered "petroleum" under CERCLA and thus excluded from CERCLA response authority and liability unless specifically listed under RCRA or some other statute. For the reasons discussed below, we believe that the contaminants present in used oil or any other petroleum substance are not within the petroleum exclusion. "Contaminants", as discussed below, are substances not normally found in refined petroleum fractions or present at levels which exceed those normally found in such fractions. If these contaminants are CERCLA hazardous substances, they are subject to CERCLA response authority and liability.

Background

Under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 as amended (CERCLA), governmental response authority, release notification requirements, and liability are largely tied to a release of a "hazardous substance." Section 104 authorizes government response to releases or threatened releases of hazardous substances, or "pollutants or contaminants." Similarly, liability for response costs and damages under Section 107 attaches to persons who generate, transport or dispose of hazardous substances at a site from which there is a release or threatened release of such substances. Under Section 103, a release of a reportable quantity of a hazardous substance triggers notification to the National Response Center.

The term "hazardous substance" is defined under CERCLA Section 101(14) to include approximately 714 toxic substances listed under four other environmental statutes, including RCRA. Both the definition of hazardous substance and the definition of "pollutant or contaminant" under Section 104(a)(2) exclude "petroleum, including crude oil or any fraction thereof", unless specifically listed under those statutes. 1/ Accordingly, no petroleum substance, including used oil, can be a "hazardous substance" except to the extent it is listed as a hazardous waste under RCRA or under one of the other statutes. Thus two critical issues in assessing whether a substance is subject to CERCLA is whether or not, and to what extent, a substance is "petroleum." - This memorandum discusses the second type of petroleum exclusion issue. The question, therefore, is not whether used oil is "petroleum" and thus exempted from CERCLA jurisdiction, but to what extent substances found in used oil which are not found incrude oil or refined petroleum fractions are also "petroleum". -If such substances are not "petroleum" then a release of used oil containing such substances may trigger CERCLA response actions, not to the release of used oil, but to the contaminants present in the oil.

1/ The full texts of these provisions are as follows:

Section 101(14)

The term [hazardous substance] does not include petroleum, including crude oil or any fraction thereof which is not otherwise specifically listed or designated as a hazardous substance under subparagraphs (A) through (F) of this paragraph, and the term does not include natural gas, natural gas liquids, liquefied natural gas, or synthetic gas usable for fuel (or mixtures of matural gas and such synthetic gas).

Section $104^{-}(a)(2)$

The term [pollutant or contaminant] does not include petroleum, including crude oil and any fraction thereof which is not otherwise specifically listed or designated as hazardous substances under section 101(14)(A) through (F) of this title, nor does it include natural gas, liquefied natural gas, or synthetic gas of pipeline quality (or mixtures of natural gas and such synthetic gas). Although the term "hazardous substance" is defined by statu there is no CERCLA definition of "petroleum" and very little direct legislative history explaining the purpose or intended scope of this exclusion. None of the four early Superfund bills originally excluded responses to oil, although the apparent precursor to Section 101(14), found in S. 1480, excluded "petroleum" without explanation in all versions except that introduced. The legislative debates on the final compromise indicate only that Congress intended to enact later, separate superfund-type legislation to cover "oil spills." See generally 126 Cong. Rec. H11793-11802 (December 3, 1980).

Since the enactment of CERCLA, the Agency has provided some interpretations of the nature and scope of the petroleum exclusion. In providing guidance in 1981 on the notification required under Section 103 for non-RCRA hazardous waste sites the Agency stated that petroleum wastes, including waste oil, which are not specifically listed under RCRA are excluded from the definition of "hazardous substance" under 101(14). 46 Fed. Reg. 22145 (April 15, 1981). 2/

In 1982 and in 1983, the General Counsel issued two opinions on the CERCLA petroleum exclusion. In the first opinion, the General Counsel distinguished under the petroleum exclusion between hazardous substances which are inherent in petroleum, such as benzene, and hazardous substances which are added to or mixed with petroleum products. The General Counsel concluded that the petroleum exclusion includes those hazardous substances which are inherent in petroleum but not those added to or mixed with petroleum products. Thus, the exclusion of diesel oil as "petroleum" includes its hazardous substance constituents, such as benzene and toulene, but PCB's mixed with oil would not be excluded. Moreover, if the petroleum product and an added hazardous substance are so commingled that, as a practical matter, they cannot be separated, then the entire oil spill is subject to CERCLA response authority.

In the second opinion, the General Counsel concluded that the petroleum exclusion as applied to crude oil "fractions" includes blended gasoline as well as raw gasoline, even though refined or blended gasoline contains higher levels of hazardous

^{2/} In the notice the Agency used the term "waste oil" without stating whether it was intended to include all waste oil or only unadulterated waste oil. The Agency has subsequently interpreted the reference to "waste oil" in this notice to include only unadulterated waste oil. 50 Fed. Reg. 13460 (April 4, 1985).

substances. The increased level of hazardous substances results from the blending of raw gasoline with other petroleum fractions to increase its octane levels. Because virtually all gasoline which leaves the refinery is blended gasoline, the petroleum exclusion would include virtually none of this fraction if the increased concentration of hazardous substances due only to its processing made it subject to CERCLA.

Finally, the Agency has interpreted the petroleum exclusion in two recent <u>Federal Register</u> notices. In the April 4, 1985 final rule adjusting reportable quantities under Section 102, the Agency provided its general interpretation of the exclusion:

> EPA interprets the petroleum exclusion to apply to materials such as crude oil, petroleum feedstocks, and refined petroleum products, even if a specifically listed or designated hazardous substance is present in such products. However, EPA does not, consider materials such as waste oil to which listed CERCLA substances have been added to be within the petroleum exclusion. Similarly, pesticides are not within the petroleum exclusion, even though the active ingredients of the pesticide may be contained in a petroleum distillate: when an RQ of a listed pesticide is released, the release must be reported.

50 Fed. Reg. 13460 (April 4, 1985).

In March 10, 1986, the Agency published a notice of data availability and request for comments on the proposed used oil listing under RCRA. 51 Fed. Reg. 8206. In that notice, the Agency responded to commenters who had argued that the RCRA listing would discourage used oil recycling because it would subject generators, transporters, processors, and users to Superfund liability. The Agency stated that used oil which contains hazardous substances at levels which exceed those normally found in petroleum are currently subject to CERCLA. 51 Fed. Reg. 8206 (March 10, 1986). Although the fact that the used oil is contaminated does not remove it from the protection of the petroleum exclusion, the contaminants in the used oil are subject to CERCLA response authority if they are hazardous substances. Accordingly, most used oil, even without a specific listing, would not be fully within the petroleum exclusion, irrespective of the listing.

Discussion

Because there is no definition of "petroleum" in CERCLA or any legislative history which clearly expresses the intended scope of this exclusion, there are several possible interpretations which could be given to this provision. However, we believe that our current interpretation, under which "petroleum" includes hazardous substances normally found in refined petroleum fractions but does not include either hazardous substances found at levels which exceed those normally found in such fractions or substances not normally found in such fractions, is most consistent with the statute and the relevant legislative history. Under this interpretation, the source of the contamination, whether intentional addition of hazardous substances to the petroleum or addition of hazardous substances by use of the petroleum, is not relevant to the applicability of the petroleum exclusion. The remainder of this memorandum explains in greater detail this interpretation and its legal basis, and responds to arguments raised in opposition to this interpretation.

The following is our interpretation of "petroleum" under CERCLA 101(14) and 104(a)(2), which we believe to be consistent with Congressional intent and the position which the Agency has taken on the scope of the petroleum exclusion thus far. First, we interpret this provision to exclude from CERCLA response and liability crude oil and fractions of crude oil, including the hazardous substances, such as benzene, which are indigenous in those petroleum substances. Because these hazardous substances are found naturally in all crude oil and its fractions, they must be included in the term "petroleum," for that provision to have any meaning.

Secondly, "petroleum" under CERCLA also includes hazardous substances which are normally mixed with or added to crude oil or crude oil fractions during the refining process. This includes hazardous, substances the levels of which are increased during refining. These substances are also part of "petroleum" since their addition is part of the normal oil separation and processing operations at a refinery in order to produce the product commonly understance the "petroleum."

Finally, hazardous substances which are added to petroleum or which increase in concentration solely as a result of contamination of the petroleum during use are <u>not</u> part of the "petroleum" and thus are not excluded from CERCLA under the exclusion. <u>3</u>/ In such cases, EPA may respond to releases of the added hazardous substance, but not the oil itself.

We believe that an interpretation of "petroleum" to include only indigenous, refinery-added hazardous substances is the interpretation of this provision which is most consistent with Congressional intent. The language of the provision, its explanation in the legislative history, and the Congressional debates on the final Superfund bill clearly indicate that Congress had no intention of shielding from Superfund response and liability hazardous substances merely because they are added, intentionally or by use, to petroleum products.

The language of the petroleum exclusion describes "petroleum" principally in terms of crude oil and crude oil fractions. This language is virtually identical to the language used in an earlier Superfund bill to define "oil." 4/ There is no indication in the statute or legislative history that the term "petroleum" was to be given any meaning other than its ordinary, everyday meaning. <u>See Malat</u> v. <u>Riddell</u>, 383 U.S. 569, 571 (1966) (words of a statute should be interpreted where possible in their ordinary, everyday sense). Petroleum is defined in a standard dictionary as

3/ The mixing of two or more excluded petroleum substances, such as blending of fuels, would not be considered contamination by use, and the mixture would thus also be an excluded substance.

4/ See H.R. 85, 96th Cong., 2d Sess. §101(s) (as passed by the House, September 1980) (""Oil" means petroleum, including crude oil or any fraction or residue therefrom"). H.R. 85 was designed principally to provide compensation and assess liability for oil tanker spills in navigable waters. As discussed below, the omission of this "oil spill" coverage under the petroleum exclusion was believed to be the most significant emission in terms of response to environmental releases upder the final Superfund bill.

Although the bill containing the precursor to Section 101(14), S. 1480, does not have a definition of "petroleum", its accompanying report did explain the term "petroleum oil" in the context of the taxing provisions:

The term "petroleum oil" as used in subsection 5 means petroleum, including crude petroleum and any of its fractions or residues other than carbon black.

S. Rep. No. 96-848, 96th Cong., 2d Sess. 70 (1980).

- 6 -

an oily flammable bituminous liquid that may vary from almost colorless to black, occurs in many places in the upper strata of the earth, is a complex mixture of hydrocarbons with small amounts of other substances, and is prepared for use as gasoline, naphtha, or other products by various refining processes.

Webster's Ninth New Collegiate Dictionary 880 (1985). Thus, an interpretation of the phrase "petroleum, including crude oil or any fraction thereof" to include only crude oil, crude oil fractions, and refined petroleum fractions is consistent with the plain language of the statute. 5/

The only legislative history which specifically discusses this provision states that

petroleum, including crude oil and including fractions of crude oil which are not otherwise specifically listed or designated as hazardous substances under subparagraphs (A) through (F) of the definition, is excluded from the definition of a hazardous substance. The reported bill does not cover spills or other releases strictly of oil.

S. Rep. No. 96-848, 96th Cong., 2d Sess. 29-30 (1980) (emphasis added). Thus, the petroleum exclusion is explained as an exclusion from CERCLA for spills or releases <u>only</u> of oil. The legislative history clearly contemplates that the petroleum

This distinction under the exclusion in Title I of 5/ CERCLA between petroleum as the substance that leaves the refinery and the hazardous substances which are added to it prior to, during or after use was also made by Congress in Title II, the revenue provisions or CERCLA. In Title II, Congress mide a distinction between "chemicals", petrochemical feedstocks and inorganic substances, taxed in Subchapter B of Chapter 38 of Internal Revenue Code, and "petroleum", crude oil and petroleum products, taxed in Subchapter A. Section 211 of CERCLA. The list of taxed chemicals includes many of the contaminant hazardous substances typically found in used oil: arsenic, cadmium, chromium, lead oxide, and mercury. The term "petroleum products" was explained in the legislative history as including essentially crude oil and its refined fractions. H. Rep. No. 96-172, Part III, 96th Cong., 2d Sess. 5 (1980) (to accompany H.R. 85).

exclusion will not apply to mixtures of petroleum and other toxic materials since these would not be releases "strictly of oil".

The Congressional debates on the final compromise Superfund legislation provides further clarification of Congressional intent concerning the scope of the petroleum exclusion, both in terms of what this provision deleted from the bill and what it did not. First, the major concern expressed with respect to the final compromise bill was the omission of its oil spill jurisdiction due to the petroleum exclusion. See e.g. 126 Cong. Rec. H11787 (Rep. Florio) (daily ed. December 3, 1980); id. at H11790 (Rep. Broyhill); id. at H11792 (Rep. Madigan); id. at H11793 (Rep. Studds); id. at H11795 (Rep. Biaggi); id. at H11796 (Rep. Snyder). This omission was of concern because it was believed to leave coastal areas and fisheries vunerable to tanker spills of crude and refined oil, such as the wreck of the Argo Merchant. and offshore oil well accidents. 126 Cong. Rec. H11793 (Rep. Studds) (daily ed. December 3, 1980). See also 126 Cong. Rec. S10578 (proposed amendment to S1480 by Sen. Magnuson) (daily ed. August 1, 1980); id. at S10845 (proposed amendment to S1480 by Sen. Gravel) (daily ed. August 5, 1980). The omitted coverage . of oil spills was believed to include approximately 500 spills per year, 126 Cong. Rec. H11796 (Rep. Snyder) (daily ed. December 3, 1980), far less than the number of contaminated oil releases each year.

However, it was clear that the omission of oil coverage was intended to include spills of oil only, and there was no intent to exclude from the bill mixtures of oil and hazardous substances. The remarks of Rep. Mikulski are typical of the general understanding of the effect of the petroleum exclusion in the final bill:

The Senate bill is substantially similar to the House measure, with the exception that there is no oil title. I realize that it is disappointing to see no oilrelated provision in the bill, but we must also realize that this is our only chance to get hazardous waste dump site cleanup legislation enacted. . .

Moreover, there is already a mechanism in place that is designed to deal with spills in navigable waterways. There is not, however, any provision currently in our law that addresses the potentially ruinous situation of abandoned toxic dump sites.

I, therefore, believe that it is imperative that we pass the Senate bill as a very important beginning in our attempt to defuse the ticking environmental time bomb of abandoned toxic waste sites.

Id. at H11796.

In addition, several speakers specifically identified suc mixtures as releases not only covered by the legislation but releases to which the bill was addressed.

Mr. Edgar ...

In my State, hazardous substances problems have been discovered at an alarming rate in recent years. In the summer of 1979, an oil slick appeared on the Susquehanna River near Pittston, Pa. When EPA officials responded under section 311 of the Clean Water Act, they learned that the slick contained a variety of highly poisonous chemicals in addition to the oil.

Officials estimate that more than 300,000 gallons of acids, cyanide compounds, industrial solvents, waste oil and other chemicals remain at this site where they could be washed to the surface anywhere in a 10-square mile surface.

Id. at H11798. See also 126 Cong. Rec. S14963 (daily ed. November 24, 1980) (Sen. Randolph) (contaminated oil slick). Other petroleum products containing hazardous substance additives intended to be addressed by the legislation include PCB's in transformer fluid, id. at S14963 (Sen. Randolph) and S14967 (Sen. Stafford), dioxin in motor fuel used as a dust suppressant, id. at S14974 (Sen. Mitchell), PCB's in waste oil, id. (Sen. Mitchell) 6/ and contaminated waste oil, id. at S14980 (Sen. Cohen). Accordingly, Congress understood the petroleum exclusion to remove from CERCLA jurisdiction spills only of oil, not releases of hazardous substances mixed with the oil.

There are two principal arguments which have been raised in opposition to this interpretation. First, the argument has been made that this interpretation narrows the petroleum exclusion to the extent that it has became virtually meaningless. As we have noted in previous opinions on this issue, an interpretation which emasculates a provision of a statute is strongly disfavored. <u>Marsano v. Laird</u>, 412 F.2d 65, 70 (2d Cir. 1969). Hewever, this interpretation leaves a significant number of petroleum spills outside the reach of CERCLA. Spills or releases of gasoline remain excluded from CERCLA under the petroleum exclusion. As indicated by the legislative history for the 1984 underground storage tank

^{6/} The illegal disposal of PCB's in North Carolina described by Senator Mitchell was a result of the spraying of 131,000 gallons of PCB-contaminated waste oil along a roadway. See 126 Cong. Rec. H9448 (daily ed. September 23, 1980).

legislation, leakage of gasoline from underground tanks appears to be the greatest source of groundwater contamination in the United States. 130 Cong. Rec. S2027, 2028 (daily ed. February 29, 1984) (Sen. Durenberger). In addition, spills of crude or refined petroleum are not subject to Superfund, as was frequently noted prior to its passage. See generally 126 Cong. Rec. H11786-H11802 (daily ed. December 5, 1980). Moreover, under this interpretation not all releases of used oil will be subject to CERCLA since used oil does not necessarily contain non-indigenous hazardous substances or hazardous substances in elevated levels. 7/ Although used oil is generally "contaminated" by definition, see e.g., RCRA Section 1005 (36), the impurities added by use may not be CERCLA hazardous substances.

A second argument which has been made opposing this interpretation is that Congress intended to include in the term "petroleum" all hazardous substances added through normal use of the petroleum substance. However, even if it were possible to determine in a response situation whether a hazardous substance was added intentionally or only through normal use or to determine what additions are "intentional", the legislative history is contrary to such a distinction. As noted above, the Senate Report explaining this provision states that it excludes releases or spills <u>strictly</u> of oil. This explanation expresses Congressional intent that releases of mixtures of oil and toxic chemicals, <u>i.e.</u> releases which are not <u>strictly</u> of oil, would be subject to CERCLA response authority. Releases of contaminated oil even if contaminated due to "normal use" are not releases strictly of oil.

Furthermore, the Congressional debates prior to passage clearly indicate an intent that contaminated oil would be subject to Superfund as several such releases were discussed

as the focus of the legislation. Congress was concerned with the environmental and health effect of abandoned toxic waste sites, not whether the presence of such hazards was intentienal or due to normal practices. In fact, one of the petroletim-hazardous substance mixtures most often mentioned during the debates was that of PCB contaminated oil, which is a type of contamination arguably resulting from the "normal use" of the oil in transformers. Accordingly, an interpretation of the petroleum exclusion which includes as "petroleum" hazardous substances added during use of the petroleum would not be consistent with Congressional intent.

^{7/} Data submitted to EPA by the Utility Solid Waste Activities Group et al. in Appendix C of their comments on the RCRA Used Oil listing, February 11, 1986.

Finally, although the Superfund Amendments and Reauthoriza Act of 1986 (SARA) contains several provisions related to oil and oil releases, it did not amend the petroleum exclusion under CERCLA. Moreover, the new provisions concerning oil and oil releases and their legislative history do not indicate a Congressional intent inconsistent with this opinion.

The only discussion of "petroleum" in the Conference Report for SARA is in the context of defining the scope of the new petroleum response fund for leaking underground storage tanks under Subtitle I of the Resource Conservation and Recovery Act (RCRA). Subtitle I defines "petroleum" in a manner nearly identical to CERCLA. The Conference Report specifies that used oil would be subject to the response fund notwithstanding its contamination with hazardous substances. H. Rep. No. 99-962, 99th Cong., 2d Sess. 228 (1986). The Conference Report is not inconsistent with the Agency's position on "petroleum" under CERCLA since it merely specifies that the leaking underground storage tank (UST) response fund is applicable to tanks containing certain mixtures of oil and hazardous substances. as well as to tanks containing uncontaminated petroleum. In fact, the Report further states that the UST response fund must cover releases of used oil from tanks since "releases from tanks containing used oil would not rise to the priority necessary...for CERCLA response", id. (emphasis added), not because such releases would be entirely excluded from CERCLA jurisdiction. See also 132 Cong. Rec. S14928 (daily ed. October 3, 1986) (Senator Chaffee) (Nothing in Section 114, pertaining to liability for releases of recycled oil, "shall affect or impair the authority of the President to take a response action pursuant to Section 104 or 106 of CERCLA with respect to any release...of used oil or recycled oil"); 132 Cong. Rec. H9611 (daily ed. October 8, 1986) (Rep. Schneider) ("...the oil companies are rightfully assessed a significant share of the Superfund tax...Waste oils laced with contaminants have been identified at at least 153 Superfund sites in 32 States.").

Guidance on Non-NPL Removal Actions Involving Nationally Significant or Precedent-Setting Issues



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

MAR -3 1989

MEMORANDUM

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

- SUBJECT: Guidance on Non-NPL Removal Actions Involving Nationally Significant or Precedent-Setting Issues (OSWER Directive 9360.0-19)
- FROM: Henry L. Longest II, Director Office of Emergency and Remedial Reports
- TO: Director, Waste Management Division Regions I, IV, V, VII, VIII Director, Hazardous Waste Management Division Regions III, VI Director, Emergency and Remedial Response Division Region II Director, Toxics and Waste Management Division Region IX Director, Hazardous Waste Division, Region X Director, Environmental Services Division Regions I, VI, VII

Purpose:

This memorandum transmits guidance for identifying non-NPL removal actions that may be nationally significant or precedent-setting and establishes procedures for requesting Headquarters (HQ) concurrence. The guidance also outlines procedural requirements for five categories of removals which are of special interest from a national perspective, but which are **not** subject to the HQ concurrence requirement for nationally significant or precedent-setting removals.

Background:

Delegation 14-1-A (February 1987) and OSWER Directive 9360.0-12 (April 1987) require the concurrence of the Assistant Administrator for Solid Waste and Emergency Response (AA, OSWER) prior to initiation of removal actions taken at non-NPL sites where the proposed action is of national significance or precedent-setting. Redelegation R-14-1-A transfers authority to concur to the Director of the Office of Emergency and Remedial Response (OD, OERR); authority to non-concur remains with the AA, OSWER. The purpose of the concurrence requirement is to promote national consistency in the implementation of the Superfund removal program.

It is not anticipated that a large number of removal actions will pose issues requiring HQ concurrence. Assessment of the potential long-term implications of initiating certain removal actions is largely interpretive, however, and Regional personnel should consult this guidance whenever considering a removal action at a non-NPL site. -2-

Objective:

The objective of this guidance is to ensure Regional compliance with HQ concurrence requirement for non-NPL removal actions involving nationally significant or precedent-setting issues. This document identifies categories of potential removal situations which have been determined to be of national significance or precedent-setting and specifies procedures for requesting HQ concurrence on these actions. The guidance also identifies categories of removals subject to special procedural requirements but not to the HQ concurrence requirement.

The types of removals subject to the concurrence requirement are not limited to those categories identified in the guidance. These categories are to be used by the Regions as a guide for screening proposed removals at non-NPL sites that may require HQ concurrence. Since evaluation of these sites is largely interpretive, final determinations regarding removals of a nationally significant or precedent-setting nature should involve consultation with Emergency Response Division (ERD) Regional Coordinators.

This interim final guidance is effective immediately. Additional revisions to the guidance will be considered as experience is gained and/or further policies are established that may affect the established categories and the HQ concurrence mechanisms.

Implementation:

I. NATIONALLY SIGNIFICANT OR PRECEDENT-SETTING CATEGORIES

Six categories of removals have been designated as nationally significant or precedent-setting. The list is not exhaustive and early consultation with the Emergency Response Division (ERD) is recommended where there are questions. In making the determination, the key considerations are:

- (a) whether Fund-financed response to a particular incident will establish a precedent for when or how future response actions must be taken; or
- (b) whether a response will commit EPA to a course of action that could have a significant impact on future resources, due to the widespread occurrence of a particular problem.

The categories identified and the rationale for identification are as follows:

 Removal actions at sites within the United States or its territories involving contamination or response actions that may affect other sovereign nations, including Indian tribes. -3-

<u>Rationale:</u> HQ concurrence will facilitate the execution of proper diplomatic protocol by the Department of State, and proper coordination with Indian tribes, the Bureau of Indian Affairs, the Indian Health Service, and other appropriate organizations, where applicable.

- 2. Removals involving pesticide contamination arising from:
 - improper storage of pesticide products awaiting indemnification
 - lawful application of pesticides, including special local use pesticides
 - grain fumigation operations.

<u>Rationale</u>: HQ concurrence will ensure that the Agency avoids commitment to cleanup of widespread contamination beyond the intended scope of CERCLA.

3. Removal actions at sites involving any form of dioxin when it is one of the principal contaminants of concern.

Rationale: HQ concurrence will ensure national consistency in dioxin cleanup. The Dioxin Disposal Advisory Group (DDAG) in HQ must review all dioxin removal actions to verify that the proposed action will provide an acceptable level of protection from dioxin exposure.

 Removal actions at sites involving releases from consumer products in consumer use (e.g., lead-contaminated soil resulting from peeling leadbased paint on houses).

<u>Rationale:</u> HQ concurrence will ensure that the Agency avoids a commitment to the cleanup of widespread non-point source contamination that is beyond the intended scope of CERCLA.

5. Removals involving asbestos when it is the principal contaminant of concern.

Rationale: HQ concurrence remains necessary because action levels for response have not yet been set and these determinations are being made on a case-by-case basis.

6. Removal actions involving substances or releases which may be subject to statutory exclusions or limitations in CERCLA. These include:

-4-

- substances excluded from Fund-financed response under the SARA section 101(14) definition of "hazardous substance" (e.g., petroleum products including crude oil, and natural gas or synthetic gas usable for fuel);
- releases excluded from Fund-financed response under the SARA section 101(22) definition of "release" (e.g., emissions from the engine exhaust of motor vehicles; releases of radioactive material from a nuclear incident; and releases caused by normally applied fertilizer);
- releases excluded from Fund-financed response under SARA section 104(a)(3) including releases of a naturally occurring substances; releases from products that are part of a structure and result in exposures within the structure; and releases in public or private drinking water supplies due to system deterioration from ordinary use.

Specific examples of substances or releases that have raised statutory interpretation or related policy issues with respect to their eligibility for CERCLA removal action include radon contamination in building structures, pentachlorophenol (PCP) contamination in log cabins, releases from coal gasification facilities, methane gas releases, and asbestos in building materials in homes.

<u>Rationale</u>: HQ concurrence will ensure that statutory exclusions and <u>limitations</u> are interpreted in a consistent manner. HQ concurrence will also ensure consistent application of EPA's authority under CERCLA section 104(a)(4) to respond to any release or threat of release if it constitutes a public health or environmental emergency and no other person will respond in a timely manner.

Concurrence Procedures

Early screening for issues of a nationally significant or precedentsetting nature is essential to ensure timely HQ concurrence when necessary. OSCs should contact the appropriate ERD Regional Coordinator when a possible nationally significant or precedent-setting removal action is first identified, to alert the Regional Coordinator that a request for HQ concurrence will be forthcoming. OSCs should also call the Regional Coordinator for advice on actions that are not specifically listed in the guidance, but which may be nationally significant or precedent-setting. Some nationally significant removal actions may require special coordination and oversight by the National Incident Coordination Team (NICT). These types of removal actions are discussed in a November 10, 1986, memorandum from the AA, OSWER entitled "Relationship between Preparedness Staff and Office of Emergency and Remedial Response during a Nationally Significant Incident," which states that OSCs should inform the Regional Coordinator when these types of incidents occur.

For those removal actions where HQ concurrence is required, written concurrence must be received <u>prior</u> to the Regional Administrator's (RA) formal approval of the Action Memorandum, except in cases of emergencies (i.e., situations where a response must be initiated within hours after completion of a site evaluation). HQ concurrence procedures for non-emergency removal actions at dioxin sites have been modified to streamline procedures. These non-emergency, emergency, and special dioxin concurrence procedures are discussed below.

Non-Emergency Removal Concurrence Procedures

All non-emergency concurrences must be requested through an Action Memorandum with a Request for Concurrence form attached. The Action Memorandum should be in final draft form, except that it should <u>not</u> be signed by the RA. The request form must be addressed from the RA to the OD, OERR and should describe the nationally significant or precedent-setting issue. This form has been developed in an effort to minimize the additional paperwork associated with obtaining HQ concurrence. A copy of the form is attached.

The RA may approve the Action Memorandum for a nationally significant or precedent-setting removal action once the action has been concurred upon by HQ. Additional HQ concurrence is required only if the scope of work described within the Action Memorandum changes significantly. In this case, HQ concurrence on the amended Action Memorandum is required, as discussed above, prior to any additional actions at the site. HQ concurrence is not required on requests for ceiling increases or time exemptions, unless the scope of work changes significantly. Most \$2 million exemption requests require approval by the AA, OSWER, unless the consistency exemption authority for that site has been delegated to the RA.

Emergency Removal Concurrence Procedures

In cases where emergency removal actions, as defined above, involve nationally significant or precedent-setting issues, Regions may initiate a removal action without HQ concurrence. In these cases, however, OSCs must take only those actions necessary to mitigate the emergency or stabilize the site, and then inform the appropriate ERD Regional Coordinator on the next working day after the removal action was initiated.

If the response is determined to be nationally significant or precedentsetting but no further actions are required beyond the emergency mitigation, the Regions must send to the Director, OERR a copy of the Action Memorandum submitted to the RA for that removal. The Action Memorandum should clearly describe the nationally significant or precedent-setting issues involved. A request for HQ concurrence is not necessary when the incident does not require actions beyond the initial emergency measures.

For those nationally significant or precedent-setting sites where further response is required beyond the emergency measures, HQ concurrence must be obtained before taking any further action. These concurrence requests are subject to the non-emergency procedural requirements described above. HQ will expedite the review of these requests to avoid delaying on-going removal actions.

Special Dioxin Concurrence Procedures

To reduce the administrative burden that the HQ concurrence procedures place on Regions with large numbers of dioxin sites, the non-emergency concurrence procedures have been modified. This modification permits the concurrence on a single dioxin site Action Memorandum to be used for multiple dioxin sites in the same Region. To qualify for this special concurrence procedure, the additional dioxin sites must have identical forms of dioxin present, and identical cleanup measures must be employed to achieve identical cleanup goals. Regions with multiple dioxin sites meeting these criteria may obtain concurrence for them all on a single Action Memorandum if supplementary information is supplied as described below.

The additional sites should be listed on the concurrence form if they are known at the time the original Action Memorandum is submitted. It should be specifically stated that the sites are identical in nature and that identical cleanup measures will be employed. If additional dioxin sites meeting the above criteria are discovered after receipt of the original HQ concurrence, the Regions are required to inform the appropriate ERD Regional Coordinator of the location of the additional removal actions. The Regions must also note within the Action Memorandum that previous concurrence on the cleanup approach has been provided.

II. REMOVAL ACTIONS SUBJECT TO SPECIAL PROCEDURAL REQUIREMENTS

The requirements established below apply to five removal categories that do **not** present nationally significant or precedent-setting issues requiring HQ concurrence, but instead involve issues that require special Regional procedures. -7-

The five categories of removal actions and the policy for handling each are as follows:

1. Removals involving mining sites.

<u>Procedures:</u> OSCs must consult with their ERD Regional Coordinator and demonstrate within the Action Memorandum that they have investigated other potential cleanup authorities (e.g., the Surface Mining Act) but found that a response could not be initiated under such authorities within the time frame required to protect human health, welfare, or the environment, or that these authorities do not apply to the particular response situation.

2. Removals involving Federal facilities.

<u>Procedures</u>: Guidance on conducting removals at Federal facilities is under development. Until this guidance is effective, OSCs must confer with the ERD Regional Coordinators to ensure that the roles and responsibilities of the various agencies are assigned appropriately.

3. Removals involving site-specific contracts.

Procedures: OSCs must coordinate with the HQ Procurement and Contracts Management Division (PCMD) to confirm that the contract Statement of Work (SOW) is consistent with the Action Memorandum and the SOW conforms with CERCLA and the NCP.

4. Removals involving radiation sites.

<u>Procedures:</u> OSCs must contact the HQ Office of Radiation Programs for guidance on health and safety in conducting radiation cleanup activities.

5. Removals involving business relocations.

<u>Procedures:</u> Action Memoranda for removals involving business relocations may be approved by the Regional Administrators, and other response activities comprising the removal may be initiated; however, until specific guidance is developed, OSCs must confer with ERD Regional Coordinators on business relocations prior to initiating the specific business relocation activities. This is to ensure national consistency in the criteria used to determine the need for business relocations, and the specific expenses incurred.

•

-8-

Comments and questions on this guidance should be directed to Betty Zeller in the Emergency Response Division, FTS 382-7735.

Attachment

cc: Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X Betti Van Epps Tim Fields Betty Zeller

Subject	Request for Concurrence on Proposed Nationally Significant or
-	Precedent-Setting Removal

From: Regional Administrator

To: Director Office of Emergency and Remedial Response

The OSC has discussed this proposed removal with staff of the HQ Emergency Response Division. ERD has advised the OSC that this removal is considered nationally significant or precedent-setting because

The action memorandum is attached for your review. My approval awaits your concurrence.

<u>Concur</u>

Director, Office of Emergency and Remedial Response

According to the redelegation, authority to non-concur remains with the Assistant Administrator. If you choose not to concur on this action, please forward this memo to the Assistant Administrator.

Non-Concur:

Assistant Administrator for Solid Waste and Emergency Response

Concur:

Assistant Administrator for Solid Waste and Emergency Response Date

Date

Date

Final Guidance on Implementation of the "Consistency" Exemption to the Statutory Limits on Removal Actions



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUN 1 2 1989

DFFICE CF SOLID WASTE AND EMERGENCY PEER CLEE

MEMORANDUM

Final Guidance on Implementation of the "Consistency" Exemption SUBJECT: to the Statutory Limits on Removal Actions (OSWER Directive 9360.0-12A) FROM: Jonathan Z. Cannor Acting Assistant Administrator TO: Director, Waste Management Division Regions I, IV, V, VII, VIII Director, Hazardous Waste Management Division Regions III, VI Director, Emergency and Remedial Response Division Region II Director, Toxic and Waste Management Division Region IX Director, Hazardous Waste Division, Region X Superfund Branch Chiefs, Regions I-X Oil and Hazardous Materials Coordinators, Regions I-X

Purpose:

The purpose of this memorandum is to transmit final guidance on use of the exemption from the statutory limits on removals for actions that are otherwise appropriate and consistent with the remedial action to be taken.

Background:

On April 6, 1987, interim final guidance was issued on implementation of the revised statutory limits on removal actions which discussed procedures for using the new exemption contained in the Superfund Amendments and Reauthorization Act of 1986 (SARA). This exemption allows removals to exceed the statutory time and money limits of one year and \$2 million where necessary to achieve consistency with the remedial action to be taken. This guidance is final and supersedes the interim final version of April 1987.

-2-

Objective:

The final guidance elaborates on the approach adopted in the proposed National Contingency Plan. Except in limited circumstances, use of the exemption from the statutory limits will be restricted to sites on the National Priorities List. Justification for use of the exemption will require that the removal action be "consistent" with the remedial action as defined in the guidance, and fall into at least one of the four categories of activities that are listed as "appropriate." Included with the guidance is a sample action memorandum demonstrating proper documentation of the justification.

Implementation:

1.0 Introduction

Section 104(e) of the Superfund Amendments and Reauthorization Act of 1986 (SARA) amends section 104(c) of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA) to raise the statutory limits on removal actions and establish a new exemption from those limits. Under SARA, the limits on removals increase from \$1 million and six months to \$2 million and 12 months.

The new exemption may be used if "continued response action is otherwise appropriate and consistent with the remedial action to be taken." It applies to any Fund-financed removal and thus encompasses State-lead as well as EPAlead responses. Actions where the Agency has the lead, but is to be reimbursed by private parties or other Federal agencies, are still subject to the statutory limits and provisions for exemption.

Regional Administrators (RAs) are authorized to approve requests for exemption from the 12-month limit. The Assistant Administrator (AA), Office of Solid Waste and Emergency Response (OSWER) retains authority to approve requests for exemption from the \$2 million limit, but may delegate that authority to RAs on a case-by-case basis.

2.0 Purpose of the Exemption

The "consistency" exemption in CERCLA 104(c) supports the new provision in CERCLA 104(a)(2) requiring removal actions to "contribute to the efficient performance of any long-term remedial action" (see OSWER Directive 9360.0-13). Together, the new CERCLA 104(a) provision and the "consistency" exemption in 104(c) are intended to promote and enhance efficiency and continuity in the Superfund program as a whole.

The 104(a) provision does this by ensuring that the removal program attempts to anticipate remedial action that will be needed and avoids taking

-3-

response actions that will impede the remedial action or result in wasteful restarts. The "consistency" exemption promotes efficiency by allowing removals to exceed the statutory limits for time and cost when to do so will result in lower overall cleanup cost as well as enhanced protection of public health and the environment.

3.0 Application of the "Consistency" Exemption

3.1 Criteria for Eligible Activities

As stated above, removal actions should take into account efficiency of the Superfund program as a whole. If there is no efficiency to be gained from continuing a removal action beyond the statutory limits, then the "consistency" exemption should not be used. In addition, in order to show that a proposed removal is "appropriate and consistent with the remedial action to be taken" it must be shown to meet the criteria for consistency in (a) and for appropriateness in (b) below:

(a) <u>Consistency</u>: At a minimum, the removal does not foreclose the remedial action.

This criterion is necessary to ensure that planned or expected remedies are not precluded by the removal. The "remedial action to be taken" is the remedial action that, prior to the <u>start</u> of the removal action, was planned or could reasonably have been expected to be taken. Certainly, the actual performance of the activities that are part of a planned or expected remedial action are consistent with that action. It may turn out that after a removal done under a "consistency" exemption, the Agency will decide not to take any further response action.

(b) <u>Appropriateness</u>: The activity is necessary for any <u>one</u> of the four following reasons:

1. To avoid a foreseeable threat.

This is an action that permanently abates a threat, as opposed to a temporary measure that, of necessity, will have to be repeated periodically, until the permanent remedy is performed.

2. To prevent further migration of contaminants.

This is an action taken to minimize the scope of the cleanup and the potential for harm to human health and the environment.

-4-

3. To use an alternative to land disposal.1

This criterion recognizes that procurement of alternative technology is more time-consuming and expensive than that of land disposal. CERCLA expresses preference for alternative technologies over land disposal.

4. To comply with the Off-site Policy.

This criterion recognizes that the standards required of facilities at which Superfund wastes may be disposed of may limit the number of available facilities. This in turn may cause delay in, or increase the cost of, disposing of site wastes.

3.2 Extension of Statutory Limits

For eligible activities, use of the "consistency" exemption to exceed the statutory limits of \$2 million and 12 months will be considered in the following manner:

(a) <u>Cost</u>: Only reasonable increases will be granted. Generally, this means not more than \$1 to \$2 million above the statutory limits.

(b) <u>Time</u>: Limits on duration will be decided, based on the particular circumstances at the site.

3.3 Sites at Which Use of the Exemption is Appropriate

This exemption will be used primarily at sites listed on the National Priorities List (NPL). However, there may be limited circumstances when use of this exemption will be appropriate for non-NPL sites. Those instances are expected to occur only rarely, and will be determined by the AA, OSWER, on a case-by-case basis. In addition to the above criteria, the AA will generally consider the following factors when making that determination:

(a) the magnitude of the contamination and the threat to human health and the environment;

(b) the status of negotiations with potentially responsible parties;

¹ Procedures for analysis, justification, and documentation for emergency and time-critical actions can be found in the "Administrative Guidance for Removal Program Use of Alternatives to Land Disposal," August 1988, OSWER Directive 9380.2-1; for non-timecritical actions use the EE/CA Guidance memo from Tim Fields, March 30, 1988.

-5-

- (c) the opportunity for widespread technology transfer; and
- (d) whether the site is likely to be proposed for the NPL.

4.0 Approval Procedures

4.1 Documentation

The action memo requesting approval of the "consistency" exemption should document that the proposed activities meet the requirements under section 3.0, above.

4.2 Concurrences

In addition to any concurrences ordinarily obtained, where the site in question is proposed for or listed on the NPL, the appropriate official in the Region's remedial program must concur.

4.3 Approval

Regional Administrators (RAs) are authorized to approve requests for exemption from the 12-month limit for both NPL and non-NPL sites. The Assistant Administrator (AA), Office of Solid Waste and Emergency Response (OSWER) retains authority to approve all requests for exemption from the \$2 million limit, but may delegate that authority to RAs on a case-by-case basis.

cc: Henry Longest Bruce Diamond Tim Fields Russ Wyer Lloyd Guerci

Clarification of Eligibility and Approval for Hazardous Duty Pay (Office of the Comptroller Transmittal No. 85-05)



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

.... . .

CC! Han

FFICL W SOLIO WASTE O EMERGENC

MEMORANDUM

SUBJECT: Hazardous Duty Pay FROM: Thad Juszczak

The attached Office cf the Comptroller Transmittal Number 85-5 provides procedures for authorizing hizardous duty pay and recording hazardous duty hours. Please ensure that the appropriate personnel are made aware of these policies and procedures. If you have any questions, please see me. Thank you.

Attachment

cc: Jack McGraw Bruce Engelbert Tina Parker Dave Chamberlin Rachel Hughes


UNITED STATES ENVIRONMENTAL PROTECTION AGE!'CY WASHINGTON, D C. 20460

SEP 1 6 1985

OFFICE Or ADMINISTRATION AND RESOURCES MANAGEMENT

- -- - -

OFFICE OF THE COMPTROLLER TRANSMITTAL NUMBER 85-5

SUBJECT: Clarification of Eligibility and Approval for Hazardous Duby Pay FROM: John J. Sacdy, Acting Director Financial Management Division

TO: Financial Management Officers

Recently, a number of questions have been received concerning the hazardous duty pay differential. This Transmittal clarifies: 1) procedures for authorizing hazardous duty pay and 2) recording of hazardous duty pay on the Time and Attendance Report.

PROCEDURES FOR AUTHORIZING HAZARDOUS DUTY PAY

The following procedures guide the authorization of hazardous duty pay:

- Supervisors who believe that an employee has performed or vill perform hazardous duty (as defined in the EPA Pay Administration Manual) are required to submit a written recommendation for hazardous duty pay to the operating personnel officer.
- Supervisors must ensure that work for which a hazardous duty premium is requested falls strictly within the criteria of the EPA Pay Administration Manual before recommending approval.
- The personnel officer will coordinate the recommendation with the local safety officer and the Director, Personnel Management Division, where necessary.
- If the hazardous duty pay differential is determined to be appropriate, the personnel officer will indicate approval by signing the recommendation and returning it to the supervisor.

A hazardous duty premium may not be paid until an approved recommendation is received from the personnel officer.

Detailed procedures for the authorization of hazardous duty payments are contained in Chapter 9, Section 5 of the EPA Pay Administration Manual.

RECORDING HAZARDOUS DUTY HOURS

The following procedures apply to the recording of hazardous duty hours;

- Only employees for which a hazardous duty pay differential has been authorized are eligible for hazardous duty pay.
- The Time and Attendance Report must reflect total hours for which the hazardous pay differential is authorized. The differential is payable for the total hours during which the employee is in pay status on a calendar day, including overtime hours. When the tour of duty extends into another calendar day, the hazardous duty pay is charged to the day work began.
- * Hazardous duty pay should be listed on the Time and Attendance Report as "Hours in Pay Status" under the heading "HAZ".

Chapters 3 and 4 of the EPA Timekeeping Manual provide additional information on the recording of hazardous duty hours.

INQUIRIES

If you have any questions please contact Joe Nemargut of the Fiscal Policies and Procedures Branch on FTS 382-5113.

cc: Kym Davis (PM 212) Management Division Directors Senior Budget Officers

Addressees:

Chris O'Connor, Financial Management Officer, Region I Ron Ghegardi, Financial Management Officer, Region II Bill Hoffman, Financial Management Officer, Region III Conny Chandler, Financial Management Officer, Region IV Richard Walker, Financial Management Officer, Region V Richard Kenyon, Financial Management Officer, Region VI Mary Jo Wallerstedt, Acting Financial Management Officer, Region VII

Alfred Vigil, Financial Management Officer, Region VIII William Anning, Financial Management Officer, Region IX Mildred Martin, Financial Management Officer, Region X Alan Lewis, Financial Management Officer, Las Vegas Rich Ruhe, Financial Management Officer, Cincinnati Bill Laxton, Financial Management Officer, RTP-Durham Vincette L. Goerl, HQ Accounting Operations Branch

Superfund Indirect Cost Manual for Cost Recovery Purposes (FY 1983 - FY 1986)

ENVIRONMENTAL PROTECTION AGENCY



SUPERFUND INDIRECT COST MANUAL FOR COST RECOVERY PURPOSES FY 1983 through FY 1986

OFFICE OF THE COMPTROLLER

OFFICE OF ADMINISTRATION AND RESOURCES MANAGEMENT

March 1986

Subject: Recovery of Superfund Indirect Costs

Recovering the costs of Superfund site clean-up from parties responsible for the contamination will be one of the major sources of replenishment of the Hazardous Substance Response Trust Fund (the Superfund) in the years ahead. In that regard, it is critical that EPA seek to recover all costs associated with clean-up. These costs should include all direct and <u>indirect</u> costs related to site clean-up.

Indirect costs are the costs necessary to operate the program but which cannot be attributed directly to specific sites. Examples include program management, indirect salaries and fringe benefits, administrative support, rent, and utilities. EPA has developed an indirect cost allocation system which allocates these indirect personnel and program overhead costs down EPA's organizational structure to Superfund sites. Determining the appropriate charges for each site is the ultimate objective for cost allocation.

While indirect costs are generally understood and accepted in the business community, they are not normally used in the government environment. Accordingly, I have directed that this manual be prepared to (1) provide an explanation of what indirect costs are and how EPA allocated them, and (2) to provide instructions to regional financial management personnel for calculating the amount of indirect costs which should be claimed in cost recovery actions.

The Superfund Accounting Branch, Financial Management Division, at EPA Headquarters will calculate indirect cost rates for each region for each fiscal year, beginning with fiscal year 1983. As rates for succeeding years are calculated, the Financial Management Division will issue transmittals to keep this manual up to date.

This manual has been provided to regional Financial Management Officers, Regional Counsels, Headquarters Legal Offices (OECM and -OGC) and all Headquarters and regional Superfund Program Division Directors.

Suggestions for improvement or comments should be referred to George Alapas, Chief, Superfund Accounting Branch at FTS 382-2268. The address is:

EPA Headquarters Superfund Accounting Branch, PM-226 401 M Street, S.W. Washington, D.C. 20460 c. Morgan Ki Comptroller

Table of Contents

	Page
Preface	i
Table Of Contents	ii
Introduction	iii
Allocation Methodology	I-1
What Are Indirect Costs?	I-1
What Is Cost Allocation?	I-2
Why Did EPA Develop An Indirect Cost Allocation Methodology For Superfund Sites?	I-2
How Was That Methodology Developed?	I-2
What Costs Are Allocated?	I-4
What Is The Conceptual Form of EPA's Methodology?	I-5
How Does EPA Apoly The Methodology To Determine Indirect Cost Rates For Superfund Sites?	I-6
What Does The Indirect Cost Rate Represent?	I-7
EPA Headquarters Organization Chart	I-8
EPA Regional Organization Chart	I-9
EPA Indirect Cost Rates And Explanation Of Use I	1-1
Rules For Applying The Rates I	I-1
Indirect Costs And The Cost Documentation PackageI	1-2
The Indirect Cost Rate Reference Sheet And Worksheet I	I-3 ·

Introduction

The purpose of this manual is twofold:

- (1) To provide an explanation of how EPA's indirect cost rates were developed, and
- (2) To explain how those rates should be used to calculate indirect costs allocable to individual Superfund sites.

The first section includes an explanation of what indirect costs are, and why and how they are allocated so that an indirect cost rate can be developed. The purpose of the section is to provide an understanding of how this occurs so that Agency representatives can become comfortable enough with the concepts to defend them in negotiations and/or litigation.

The second section provides those rates that have been calculated for each region and an explanation of how they should be applied to derive indirect costs allocable to a given site. These costs should be pursued in cost recovery actions with the same intensity as direct costs.

Allocation Methodology

The purpose of this section is to present EPA's methodology for indirect cost allocation to Superfund sites in an understandable format for non-accountants. It is intended to assure the reader that the indirect costs claimed in cost recovery actions have been derived from the most accurate and defensible methodology available using generally accepted accounting principles. As such these indirect costs are valid and should be recognized as part of the total recoverable costs incurred by the Government in clean-up actions under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA).

This section answers relevant questions about Superfund site indirect costs. The specific questions this section addresses are as follows:

- What are indirect costs?
- What is cost allocation?
- Why and how did EPA develop an indirect cost allocation methodology for Superfund sites?
- What costs are allocated?
- What is the conceptual form of EPA's methodology?
- How does EPA apply the methodology to determine indirect cost rates for Superfund sites?
- What does an indirect cost rate represent?

The answers to these questions should provide a sufficient under-.standing of the indirect costs of a Superfund site that the reader will accept their validity as recoverable costs.

What are indirect costs?

Indirect costs are those costs which are necessary to the operation of the program and support of site clean-up efforts, but which cannot be directly identified to the efforts at any one site.

They range from costs closely related to site work -- pay earned by on-scene-coordinators while in training or awaiting the next clean-up assignment -- to costs not so closely related to site work -- a portion of the Administrator's time. While these costs are very different in their relationship to any particular site, both are necessary and are incurred in support of the site clean-up program.

Other examples of indirect costs are:

- costs of site clean-up personnel while they are not cleaning specific sites, e.g., training, vacation, sick and holiday pay.
- all costs of non-site personnel, e.g., program managers; clerical support; and personnel, finance, contracts and other administrative support staff.
- office space costs such as rent and utilities for both site and non-site staff.

In summary, indirect costs are all costs attributable to the Superfund program that cannot be directly identified to a specific site.

What is cost allocation?

Cost allocation is the method by which indirect costs are assigned to one or more cost objectives. A cost objective is any activity for which a separate measurement of costs is desired. Examples include departments or services, such as the clean-up of Superfund sites. Cost allocation is done to develop the full cost of a cost objective including direct and indirect costs for any number of reasons -- for EPA the reason is cost reimbursement.

Why did EPA develop an indirect cost allocation methodology for Superfund sites?

It is universally acknowledged by the business world that indirect costs are a real part of the total costs of any product or service. In that context EPA is concerned with indirect costs because they are a real part of the total cost of response at hazardous waste sites. As such, it is EPA policy to seek recovery of "all costs...incurred by the United States Government," as stated in Section 107 (a)(4)(A) of CERCLA.

How was that methodolocy developed?

To provide a sound and defensible basis for determining indirect costs, EPA decided to develop a formal indirect cost allocation methodology. To ensure that the methodology would be based on appropriate accounting principles, EPA contracted with the international accounting firm of Ernst and Whinney. Ernst and Whinney was tasked with developing an indirect cost allocation methodology which would most accurately reflect the level of indirect support provided to Superfund sites by EPA organizations. This section describes the decision making process EPA and Ernst and Whinney went through to decide upon the most appropriate methodology.

Development of a cost allocation methodology requires 1) selecting a method, 2) selecting how to accumulate costs, and 3) selecting the allocation bases that link the cost objectives to the costs themselves. An allocation base is that which defines the various recipients of the support provided by an organization as well as the proportionate amount of that organization's costs which should be allocated to each of those various recipients.

Selecting the method

Selection of a particular method of allocation is dictated by concerns about accuracy and practicality. The various levels of accuracy provided by each of the generally accepted alternative methods is dictated by the degree to which each method recognizes the concept of support department interdependency. The following is an illustration of this concept.



Example of Support Department Interdependency

In this example, the Comptroller provides support to Data Processing, but the support is mutual -- in reality each supports the other. Both support departments, in turn, support the Assistant Administrator for OSWER and OERR, separately.

Only the cross allocation method adequately recognizes the type of support department interdependency displayed in this example. The cross allocation method is defined as explicitly recognizing the mutual services rendered among all departments. For this reason, the cross allocation method is universally considered the most accurate and defensible of all methods. EPA's need for just such a method led to the adoption of this method over less accurate methods.

Selecting how to accumulate costs

EPA's Financial Management System (FMS) is designed to accumulate costs by allowance holder. Allowance holders (AHs) are responsible for the day to day management of EPA resources. Allowance holders are generally identified with the Assistant Administrator (AA) organizational level. Responsibility for control of resources is further delegated to and accounted for by responsibility centers (RCs). Responsibility centers are usually assigned to division level organizations. This AH/RC ⁴ design allows identification of costs to organization units down to division level. The ability to accumulate costs by division enabled Ernst and Whinney and EPA to study EPA's organizational units at the division level for support relationships to other organizational units; to the Superfund program; and to Superfund sites, the final cost objectives.

Selecting the allocation bases

Cost accounting literature provides guidance and principles governing the selection of bases used for distributing support costs. The guidance and principles are guite technical and are not presented here. However, Ernst and Whinney adhered to those principles in selecting the allocation bases that best reflect the support provided by each of those organizations which are allocable within the principles of cost allocation.

What costs are allocated?

EPA's funding generally falls into appropriations for Salaries and Expenses, Research and Development, Abatement and Control, Construction Grants, Buildings and Facilities, and Superfund. Indirect costs that are allowable for allocation -i.e., costs that support Superfund -- are expended from EPA's appropriations for Salaries and Expenses and for Superfund. Therefore, only costs paid from those two appropriations were reviewed for possible allocation. Within those appropriations, costs specifically excluded from allocation are 1) capital costs, which are considered investments rather than operating expenses related to a given fiscal year, and 2) costs charged directly to Superfund sites, i.e., direct costs. Specifically, the indirect costs which are allocated are as follows:

- salaries and fringe benefits
- EPA travel and transportation
- rent, communications, and utilities
- printing and reproduction
- supplies and materials
- other contractual services.

What is the conceptual form of EPA's methodology?

EPA's cost allocation methodology is a three stage process which allocates the indirect costs from higher level organizations down EPA's organizational structure through two intermediate levels to the final cost objectives -- Superfund sites. The first intermediate level is the Assistant Administrator organizational level, and the second intermediate level is the regional program division organizational level -- that which works most closely with the Superfund sites themselves.

Stace 1

The first stage's basic function is to allocate <u>Headquarters</u> support costs to the Acency's major "Assistant Administrator (AA)" organizational level, i.e. Headquarter's Program AAs -- Pesticides and Toxic Substances: Air, Noise and Radiation; Water; Research and Development; and Solid Waste and Emergency Response (OSWER) -- and Regional Administrators (RAs). An organization chart for EPA Headquarters is presented at the end of this section (page I-R) to aid this discussion. Temporary cost pools are created to receive the allocation to the "AA" level. Dollars allocated to the cost pools for the Headquarters program AAs for Pesticides and Toxics; Air, Noise and Radiation; Water; and Research and Development are not brought forward to Stage 2 for allocation to lower organizational levels because they are non-allocable.

They are considered non-allocable to Superfund because with the exception of Research and Development, they do not support Superfund. While Research and Development does support Superfund, the beneficial relationship to actual site work is often tenuous. Accordingly, we did not allocate it. The cost pools formed in Stage 1 to accumulate costs that are allocated to the AA for OSWER and the Recional Administrators are broucht forward to Stage 2 for further allocation because OSWER supports Superfund and the Regional Administrators support their regions.

Stage 2

The second stage's basic functions are (1) to allocate the allocable cost pools formed in Stage 1 (OSWER and the Regional Administrators) to lower organizational levels (division level) within OSWER and the Regions, (2) to allocate OSWER administrative support organizations to lower level OSWER organizations and (3) to allocate Regional administrative office costs (the Regional Administrator, Regional Counsel, Policy and Management, etc.) support costs to regional program divisions (Air Management, Waste Management, Water Management, and Environmental Services). The authorized regional organization chart is presented at the end of this section (page I-9) to provide a visual representation of regional organizations.

Stace 3

The third stage's basic functions are 1) to allocate lower level OSWER organizations to three final cost pools -- Superfund sites, Superfund in general, and non-Superfund; and 2) to allocate regional program division costs to those same three final cost pools. Regional indirect cost rates can be calculated after costs have been allocated in the third and final stage of the allocation.

Automating the allocation of costs

EPA chose to automate the allocation because the crossallocation methodology adopted by EPA involves many repetitive calculations. EPA chose to use Ernst and Whinney's cross-allocation software (GOVCOST). This software is also used by several State and local governments to perform their indirect cost allocations. In that context, the GOVCOST software has been approved by several Federal government agencies for use by State and local governments in determining indirect costs that are allocable to Federal programs and allowable for reimbursement.

How does EPA apply the methodology to determine indirect cost rates for Superfund sites?

The basis on which costs are allocated from regional program divisions to Superfund sites, Superfund in general, and non-Superfund cost pools is according to the proportion of regional program division employee hours charged to each of those categories. This allocation basis is used because Ernst and Whinney and EPA studied regional program divisions and determined that where the employees of those divisions charged their time was the best measure of support provided to each of those categories by those divisions. Once costs have been allocated to the final three cost pools mentioned above, EPA calculates an indirect cost rate for each region. Using their individually derived rate, each region can apply indirect costs to each of its individual Superfund sites.

The rate for any region is calculated by dividing the costs allocated to that region's Superfund site cost pool by the number of regional program division hours charged to sites. For example, if the total cost allocated down EPA's organizational structure to a region's Superfund site cost pool is \$1,500,000, and the total regional program division hours charged to that region's sites is 21,000 hours, the indirect cost rate for that region is \$71 per hour of regional program division labor.

> $S_{1,500,000} = S71.43$, or S71 rounded to the nearest dollar. 21,000

What does the indirect cost rate represent?

Using the above example, assume that the region had only three sites, and the 21,000 regional program division hours were charged to sites as follows:

Site A = 1,000 hours Site B = 15,000 hours Site C = 5,000 hours Total = 21,000 hours.

Indirect costs for those sites would be as follows:

Site A indirect costs = $$71 \times 1,000$ hours = \$71,000Site B indirect costs = $$71 \times 15,000$ hours = \$1,065,000Site C indirect costs = $$71 \times 5,000$ hours = \$355,000

Total

-

\$1,500,000.

The rate is merely a means of determining an individual site's share of the indirect costs allocated to the Superfund sites cost pool.



AUTHORIZED EPA REGIONAL ORGANIZATION



*Alternative Regional organization may reflect Management Division concept.

** Alternative Regional organization may reflect a single Division including Air and Waste Management functions.

This section has four purposes:

- provide rules to follow in applying regional indirect rates to individual Superfund sites;
- issue instructions for including indirect costs in a site's cost documentation package;
- provide the regional indirect cost rates for each region and fiscal year;
- provide a worksheet to calculate the indirect costs to be claimed in a Superfund site's cost recovery action.

Rules for Apolving the Rates

Rule #1

A recion's indirect cost rate must only be applied to hours charged to a site by regional program division personnel. This is because that rate has been derived by dividing the indirect costs allocated to a region's Superfund site cost pool by only the number of hours charged to sites by regional program division personnel. For example, expanding the example used in Section I, assume the total number of hours charged to sites is as follows:

	Regional Program Divisions	Regional Administrative Divisions	Headquarters Offices
Site A	1,000	200	50
Site B	15,000	3,500	2,000
Site C	5,000	1,500	500
Total	21,000	5,200	2,550.

If the S71 rate was applied to all regional hours charged to sites in the region, the total calculated indirect costs for sites would be:

Site A: (1,000 hours + 200 hours) x 571 = 5 85,200 Site B: (15,000 hours + 3,500 hours) x 571 = 51,313,500Site C: (5,000 hours + 1,500 hours) x 571 = 5 461,500 Total: (21,000 hours + 5,200 hours) x 571 = 51,860,200.

II-1

This calculation of indirect costs is obviously incorrect because the total indirect costs allocated to the region's cost pool for Superfund sites was only \$1,500,000 as developed in Section I. This example reflects the critical importance of this rule.

Exception to Rule #1

There is one and only one exception to rule \$1. The exception applies <u>only</u> to regions 3 and 4. The exception is required because the accounting structure of regions 3 and 4 do not permit distinction between program division and administrative division hours. Therefore, the total regional division hour charges was the allocation basis used to allocate costs from regions 3 and 4 program divisions to the Superfund site, Superfund in general, and non-Superfund cost pools.

Additionally, the indirect cost rates for regions 3 and 4 were derived by dividing costs allocated to the Superfund site cost pool by the number of hours charged to sites by all regional divisions (program and administrative). Therefore, regions 3 and 4 <u>must</u> apply their indirect cost rates to all regional hours charged to a site in order to calculate the correct amount of indirect costs for that site.

Rule #2

The indirect cost rate for a fiscal year must only be applied to hours that relate to that fiscal year. Cost accounting principles dictate that indirect cost rates should be calculated for a specific period of time, generally a fiscal year. EPA calculates indirect cost rates for each fiscal year. EPA uses the latest calculated rates as provisional rates until the next fiscal year's rates can be calculated.

Indirect costs and the cost documentation package

This manual is intended to support and defend the indirect costs that are being claimed in cost recovery actions. It has been written for the non-accountant to best serve that role. Therefore, the indirect cost portion of the cost documentation package should consist of a copy of this manual with only the applicable regional rate reference sheet and calculation worksheet. Any requirements for a more detailed discussion of EPA's indirect costs claimed should be directed to the Superfund Accounting Branch.

Indirect cost rate reference sheet and worksheet

The remainder of this section provides, by region, the indirect cost rates for each fiscal year and a worksheet that will display the application of a region's indirect cost rates which produces the amount of indirect costs claimed in cost recovery for a Superfund site. The completed worksheet and a copy of this manual should serve as the support and documentation of the amount of indirect costs claimed. The fully documented regional program division hours on which a site's indirect costs are calculated are, and will continue to be, part of the documentation for direct site costs. UPDATED REGIONAL REFERENCE AND WORKSHEETS FOR CALCULATING INDIRECT COSTS FOR COST RECOVERY PURPOSES ARE ATTACHED TO THE SUPERFUND INDIRECT COST UPDATE MEMORANDUM ISSUED BY VINCETTE L. GOERL, DIRECTOR OF FINANCIAL MANAGEMENT DIVISION (PM-226)

•

.

Superfund Indirect Cost Update



OFFICE OF ADMINISTRATION AND RESOURCES MANAGEMENT

MEMORANDUM

SUBJECT: Superfund Indirect Cost Update

- FROM: Vincette L. Goerl, Director Financial Management Division (PM-226)
- TO: Assistant Regional Administrators Regional Management Division Directors

Gene Lucero, Director Office of Waste Programs Enforcement

Edward E. Reich Associate Enforcement Counsel For Hazardous Waste Enforcement

In March, 1986, the Comptroller issued the Superfund Indirect Cost Manual for Fiscal Years (FY) 1983, 1984, 1985, and 1986 which provided indirect cost rates to be used in Superfund cost recovery actions. The manual also furnished guidance on the application of the rates along with an explanation of the methodology employed.

The purpose of this memorandum is to update information contained in the manual by (1) providing new rates for FYs 1983-86 and a provisional rate for FY 1987, (2) clarifying the distinctions between provisional rates and final rates, and (3) apprising you of our plans regarding indirect costs during FY 1987.

Attached are revised pages to the Manual providing the new rates which should be used by the regional financial management offices in computing indirect costs for cost recovery actions. The rates should be applied in the same manner as explained in Section II of the Manual. You will note that the rates for FYs 1983 and 1984 are labelled "Final" while the FY 1985-87 rates are termed "Provisional." The general distinction is that final rates are based on the actual costs incurred during the year, whereas provisional rates are interim, temporary rates to be used until actual costs are known and final rates can be computed. EPA's policy for determining provisional rates is to use the rates from the latest fiscal year for which rates based on actual costs have been computed. Since FY 1984 is the latest such year, the FY 1984 rates are used as provisional rates for 1985-87. In addition to being based on the actual incurred costs for the year, final rates also reflect adjustments recommended by auditors from the Office of the Inspector General (OIG). The OIG has completed most of the fieldwork on audits of FY 1983 and 1984. They have recently begun the FY 1985 and FY 1986 audit.

Ideally, final rates should be the basis for determining the indirect costs allocable to a site since they are based on actual costs. In reality, many cost recovery actions are scheduled for completion before final rates are known. In such cases, it is acceptable to use the provisional rates. EPA's policy of provisional rate determination based on latest known final rates is a common and accepted practice utilized by numerous government contractors and grantees.

Our goals regarding indirect costs for FY 1987 include the following:

- 1. Computation and issuance of FY 1985 and 1986 final rates.
- 2. Analysis of the rate computation methodology for possible revisions.
- 3. Evaluation of the existing indirect cost documentation with an eye towards revision (e.g. the Manual) and/or issuance of new material.
- 4. Conducting training courses or workshops on indirect costs for Regional and Headquarters personnel.
- 5. Examining the feasibility of incorporating indirect costs into the Superfund accounting and financial reporting process.

The attached pages should be substituted for the corresponding pages in the Manual. Please direct any questions or comments to William Cooke of the Superfund Accounting Branch on (202) 382-2880.

Attachments

cc: David P. Ryan Budget Division Regional Comptrollers

REGION I

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final	Provisional
Region I's Indirect Cost Rate Per Direct Labor Hour:	<u>FY 1983</u> <u>\$ 62</u> <u>\$ 60</u> <u>FY 1984</u> <u>\$ 60</u>	<u>FY 1985</u> FY 1986 FY 1987 \$ 60 \$ 60 \$ 60

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

Code ·	Title	Year Applicable (X)						
		FY 1983	FY 1984	FY 1985	FY 1986	<u>FY 1987</u>		
01C	Air Division	x	x	x	x	, x		
01L	Water Division	x	x	x	x	x		
01N	Environmental Service Division	x	x	x	x	x		
01R	Waste Management Division	x	x	x	x	x		

REGION I

SITE #: SITE NAME:

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

.

.

	Direct Hours (i.e.,	Total	Regional II	ours)	Total Hours	X	Rate	=	Subtotal
	01C	01L	01N	01R					
FY 1983							\$62		
FY 1984							\$60		
FY 1985							\$60		
FY 1986							\$60		
FY 1987							\$60		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

.

REGION II

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final	<u>-</u>	Provisional	
Region II's Indirect Cost Rate Fer Direct Labor Hour:	\$ <u>68</u>	\$ <u>50</u>	\$ <u>560</u> \$ <u>60</u> \$ <u>68</u>	<u>FY 1987</u> \$ 68

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

.

•

•

Code	Title	Year Applicable (X)						
		FY 1983	FY 1984	FY 1985	FY 1986	FY 1987		
02C	Caribbean Field Office	x	x	x	x	X		
02D	Emergency and Remedial Response Division	x	x	x	x	x		
02M	Air and Waste Management Division	x	x	x	x	x		
02N	Water Mangement Division	x	x	x	x	x		
02P	Environmental Services Division	x	x	x	x	x		

REGION III

SITE #:______SITE NAME:

Norksheet For Calculating Indirect Costs For Cost Recovery Purposes

		Direct Hours (i.e., Total Regional Hours)	Total Hours	X	Rate	=	Subtotal
		•					
FY	1983	(Final)			\$52		
FY	1984	(Final)			\$ 52		
FY	1985	(Provisional)			\$ 52		
FY	1986	(Provisional)			\$52		
FY	1987	(Provisional)			\$52		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

REGION II

SITE #:_____ SITE NAME:

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

.

•

	Direct Hours	(1.e., To	otal Regional	licurs)		Total Hours	X	Rate	8	Subtotal
	02C	[.] 02D	02M	02N	02P					
FY 1983								\$68		
FY 1984								\$68		
FY 1985	;							\$68		
FY 1906	•							\$63		•
FY 1987	,					<i>.</i>		\$68		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

REGION IV

SITE #: SITE NAME:

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

i

;

		Direct Hours (i.e., Total Regional Hours)	Total Hours	X	Rate	=	Subtotal
		•					
FY	1983	(Final)			\$59		
FY	1984	(Final)			\$54		
FY	1985	(Provisional)			\$5 4		
FY	1986	(Provisional)			\$ 54		
FY	1987	(Provisional)			\$54		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

REGIO.

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Fi	nal	Provisional
Region V's Indirect Cost Rate	<u>FY 1983</u>	FY 1984	<u>FY 1985</u> <u>FY 1986</u> <u>FY 1987</u>
Per Direct Labor Hour:	\$71	\$ 61	\$ 61

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

.

.

Cocle	Title	Year Applicable (X)					
		FY 1983	FY 1984	<u>FY 1985</u>	FY 1986	<u>FY 1987</u>	
05E	Dir. Office Waste Mgmt/Emerg. Response	x	x	x	x	x	
05F*	Waste Management Division - Superfund	x	x	x	x	x	
05G	Nazardous Waste Enforcement Branch	x	x	x	x	`x	
0511	Great Lakes Coordinator	x	x	x	x	x	
05L	Air Division	x	x	x	x	X	
05P	Water Division	x	x	x	x	x	
05Q	Waste Management Division - Non-Superfund	x	x	x	x	x	
05R	Environmental Services Division	x	x	x	x	x	
05W	Central Regional Lab	x	x	x	x	x	
05Y	Eastern Office	x	x	x	x	x	
05z	S&A Central District Office	x	x	x	x	x	

* .05F was Emergency and Remedial Response Branch in FYs 1983-86 but nonetheless should be used in calculating indirect costs for those years.

I.

REGION V

SITE #: SITE NAME:

•

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

	Direct Hours (i.e., Total Regional Hours)			<u>)</u>	Total Hours X				Subtotal	
	05E	05F	05G	0511	05L					
FY 1983								\$71		
FY 1984								\$61		
FY 1985								\$61		
FY 1986								\$61		·
FY 1987								\$G1		

SUB-TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY (transfer to page 2)

REGION V

SITE #:_____SITE NAME:

.

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

	Direct Hours (i.e., Total Regional Hours)					Total Hours X	Rate	=	Subtotal	
	05P	05Q ·	05R	05W	05Y	05Z				
FY 1983								\$71		
FY 1984							•	\$61		
FY 1985								\$61		
FY 1986								\$61		·
FY 1987								\$61		

Subtotal this page

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

.

REGION VI

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final	L	Provisional			
Region VI's Indirect Cost Rate Per Direct Labor Hour:	<u>FY 1983</u> \$ 66	\$ <u>60</u>	\$ <u>500</u> \$	<u>Y 1986</u> <u>FY 1987</u> 60 \$ 60		

'Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

•

Code	Title	Year Applicable (X)						
		<u>FY 1983</u>	FY 1984	FY 1985	<u>FY 1986</u>	<u>FY 1987</u>		
061*	Houston Lab	-	-	x	x	· -		
06J	Nir, Pesticides & Toxics Division	-	-	-	x	x		
06K	Water Management Division	x	x	x	x	x		
06L**	liazardous Waste Management Division	x	x	x	x	x		
OGM	Environmental Services Division	x	x	x	x	x		

* OGN was Management Division in FYs 1983 and 1984 and should not be used to calculate indirect costs for those years.

** OGL was Air & Waste Management Division in FYs 1983-1985; but, nontheless, should be used in calculating indirect costs for those years.

REGION VI

SITE #:_____ SITE NAME:

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

	Direct Hours (i.e., Total Regional Hours)				Rate	=	Subtotal		
	OGH	06 J	06K	06L	OGM				
FY 1983							\$66		
FY 1904							\$60		
FY 1985							\$60		
FY 1906							\$60		
FY 1987							\$60		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

REGION VII

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final	L	Provisional			
Region Vil's Indirect Cost Rate Per Direct Labor Hour:	FY 1983 \$ 68	<u>FY 1984</u> \$ 65	\$ <u>65</u> \$ <u>65</u> \$ <u>65</u> \$ <u>65</u> \$ <u>65</u>			

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

٠

Code	Title	Year Applicable (X)							
		FY 1983	FY 1984	FY 1985	FY 1986	FY 1987			
07C	Air and Water Management	x	x	x	. X	· x			
07K	Water Management Division	x	x	x	x	x			
07L	Air and Toxics Division	-	-	x	x	x			
07M	Environmental Services Division	x	x	x	x	x			
07w	Waste Management Division	-	-	x	x	x			
REGION VII

SITE #:_____SITE NAME:_____

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

•

	Direct Hours	(i.e., Tot	al Regiona	1 Hours)		Total Hours	X	Rate	=	Subtotal
	07C	07K	071.	07M	07 W					
FY 1983								\$68		
FY 1984								\$65		
FY 1985								\$65		
FY 1986								\$65		
FY 1987								\$65		

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

EXECTION 273

١

REGION VIII

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final		Prov	isional
Region VIII's Indirect Cost Rate Per Direct Labor Hour:	\$ <u>60</u>	\$ <u>52</u>	\$ <u>52</u>	\$ <u>562</u> \$ <u>62</u> \$ <u>62</u>

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

.

Code	Title	Year Applicable (X)							
		FY 1983	<u>FY 1984</u>	FY 1985	<u>FY 1986</u>	FY 1987			
OBK	Water Division	x	x	x	x	· x			
08L	Air and Hazardous Material Division	x	x	x	x	x			
08M*	Environmental Services Division	x	x	x	X .	x			
09R	Montana	x	x	x	x	x			
085	Air and Hazardous Material Branch	x	x	x	x	-			

*00M was Surveillance and Analysis Division in FYs 1983-1986; but, nontheless, should be used in calculating indirect costs for those year

REGION VIII

SITE #:______SITE NAME:_____

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

.

	<u> </u>	Direct Hours	(i.e., '	Total Regio	nal llours)		Total Hours	X	Rate	-	Subtotal
		00K	08L	MBO	08R	085					
FY 19	83								\$60		
FY 19	984								\$62		
FY 19	985								\$62		
FY 19	986								\$62		
FY 1	987								\$62		
•											

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

REGION IX

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Final		Provisional			
Region IX's Indirect Cost Rate	FY 1983	FY 1984	FY 1985	FY 1986	FY 1987	
Per Direct Labor Hour:	\$ 65	\$ 63	\$ 63	\$ 63	\$ 63	

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should be Applied:

•

Cocle	Title	Year Applicable (X)							
		FY 1983	FY 1984	FY 1985	FY 1986	FY 1987			
09K	Toxics and Waste Management Division	x	x	x	x	· x			
09L	Water Management Division	x	x	x	x	x			
09M	Air Management Division	x	x	x	x	x			

REGION IX

SITE #: SITE NAME:

•

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

	Direct Hours	(1.e., Tot	al Regional Hours)	Total Hours	X	Rate	Subtotal
	09K	09L	09M				
FY 1983						\$65	
FY 1984						\$63	
FY 1985						\$63	
FY 1986						\$63	
FY 1987						\$63	

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

۱

REGION X

Reference Sheet For Calculating Indirect Costs For Cost Recovery

	Fin	al	Provisional		
Region X's Indirect Cost Rate	FY 1983	<u>FY 1984</u>	FY 1985	<u>FY 1986</u>	FY 1987
Fer Direct Labor Hour:	\$ 64	\$ 61	\$ 61	\$ 61	\$ 61

Allowance Holder/Responsibility Centers To Which The Indirect Rate Should Be Applied:

Code	Title	Year Applicable (X)							
		FY 1983	FY 1984	FY 1985	FY 1986	FY 1987			
100*	Air and Toxics Division	x	x	x	x	x			
10G	Alaska Operations Office	x	x	x	x	x			
104**	Water Division	x	x	x	x	· x			
10N	Hazardous Waste Division	x	x	x	x	х			
10P***	Environmental Services Division	×	x	x	x	x			
10R	Oregon Operations Office	x	x	x	x	x			
10 r	Idaho Operations Office	x	x	x	x	x			
10w	Washington Operations Office	x .	x	x	X ·	x			
*10C was	Emergency Response in FYs 1983 and 198 calculating indirect costs for those y	4; but, nonethe ears.	eless, shou	ild be used	i in				
**10M was	Air and Water Programs Division in FYs	1983 and 1984;	but, none	etheless, s	should be a	used in			

calculating indirect costs for those years

***10p was Surveillance and Amlysis Division in FYs 1983 and 1984; but, monetheless, should be used in



Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

	Direct	llours (1.	e., Total	Regional I	lours)	Total Hours	X	Rate	=	Subtotal
	100	10G	. 10 M	10N	10P					
FY 1983								\$64		
FY 1984								\$61		
FY 1985								\$61		
FY 1986								\$61		
FY 1987								\$61		

SUB-TOTAL INDIRECT' COSTS TO CLAIM IN COST RECOVERY (transfer to page 2) REGION X

SITE #: SITE NAME:

Worksheet For Calculating Indirect Costs For Cost Recovery Purposes

• .

		Direct Hours	(1.e.	, Total Re	gional Hours)	Total Hours	. X	Nate	=	Subtotal
		101	ર	10 T	10w					
FY 1	.983							\$64		
FY 1	.984							\$61		
FY I	1985							\$61		
FY]	1986							\$61		
FY 1	1987							\$61		

Subtotal this page

TOTAL INDIRECT COSTS TO CLAIM IN COST RECOVERY

Superfund Site Allowances (Comptroller Policy Announcement Number 87-07)

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20480 213 COMPTROLLER POLICY ANNOUNCEMENT No. 97-07 CFFITE OF MEMORANDUM JANIS : 757 SUBJECT: Superfund 2 te Al lowand David P.\Ayan FROM: Comptroll T0: Assistant Regional Administrators Management Division Directors

ADMINISTRATICN ANC RESOURCES MANAGEMENT

Senior Budget Officers Regional Comptrollers

This policy announcement introduces a separate Advice of Allowance for certain Superfund extramural site activities and describes the policies and procedures necessary to implement this new allowance. Because the Regions are the only current recipients of site-specific response extramural funding, these new Site Allowances will be issued only to the Regional Offices. They will provide a separate extramural dollar allowance for Superfund Removals, Remedial Designs (RD) and Remedial Actions (RA), and will distribute the Design and Action allowances by site.

BACKGROUND

This policy announcement is one of a continuing series of improvements being implemented to strengthen the Agency's funds control procedures. The new Site Allowances will contribute to improved funds control in a number of ways: (1) they provide a fiscal management structure that ensures the Superfund program is executed according to Congressional and Agency plans, (2) they separate the dollars into reasonable management units without the loss of flexibility afforded by the large site response program element, and (3) they augment tracking and planning activities conducted through the Superfund Comprehensive Accomplishments Plan (SCAP) by imposing formal Agency funds control requirements on key activities.

POLICY

In the past, each Region received all of its Superfund dollars in one allowance. Effective with implementation of this policy, each Region will receive two separate Superfund allowances: a Site Allowance and a Regular' Superfund Allowance.

The Site Allowance will be divided into three categories:

1) Removal Actions - Extramural Only. This portion of the allowance will provide funding for emergency as well as non-emergency removals, and will not be issued site specifically.

2) Remedial Design - Extramural Only. This portion of the allowance will be issued site specifically, listing the funding provided for each design by site name, site ID number and dollar amount.

3) Remedial Action - Extramural Only. This portion of the allowance will also be issued site specifically, listing the funding provided for each remedial action by site name, site ID number and dollar amount.

The <u>Regular Superfund Allowance</u> will contain all of the remaining Superfund dollars for the Region. This includes all intramural dollars and all other Superfund extramural funds (e.g., site inspections, investigations and feasibility studies, community relations, management assistance, etc.). This allowance will not be issued site specifically.

IDENTIFICATION OF THE SITE ALLOWANCE

The Site Allowance will be referred to as the "Alpha" allowance since it will include an "A" in the Allowance Holder field. For example, Region V's regular Superfund allowance is issued to Allowance Holder "05", while the Site . Allowance will be issued to Allowance Holder "5A". Region X's Site Allowance will be identified as "0A" (zero, A). In this way, the allowance and the account numbers established from it will be uniquely identified as intended for site-specific purposes only. This will allow obligating officials to ensure that funds issued for RD/RA and Removal activities in the Alpha allowance are - not used for other activities.

REPROGRAMMINGS

The initial quarterly Superfund Site Allowance will be developed based on the program's SCAP. Any modifications to that initial plan would be submitted through both the SCAP adjustment/amendment process and through the standard change request process, thus revising the next allowance to be issued. Change requests will be required to reprogram dollars among Superfund activities on the Site Allowance (even if the site remains the same), between sites under the same Superfund activity, and between any portion of the Site Allowance and the regular Superfund allowance. Each change request should equal a net change of zero dollars and should provide a detailed explanation as 'to why resources are being shifted among activities and/or sites. Requests for additional resources should be submitted directly to the Office of Solid Waste and Emergency Response (OSWER).

SUBMISSION OF CHANGE REDUESTS

Change requests from the Regions should be submitted to the Budget Division with a copy to OSWER's Resource Management Staff. OSWER will review the requests to ensure that the change request is consistent with the SCAP and will then forward the request to the Budget Division for final approval. Notification of approval will be part of the normal change request and SCAP update process.

In order to reduce paperwork, electronic systems for your use in transmitting SCAP adjustments/amendments and change requests will be established. -Instructions on these systems will be issued in the near future.

MONITORING COMPLIANCE

The Budget Division will monitor obligations against the Site Allowance on a monthly basis. If an Allowance Holder exceeds the Removal allocation or any one of the Remedial Design or Remedial Action site allocations, the Budget Division will notify the Allowance Holder and request resolution of the overobligation. If the Allowance Holder does not submit a change request, decommit or deobligate funds, or effect corrections in the Financial Management System as necessary, the Budget Division will initiate reprogramming from the Allowance Holder's regular Superfund allowance. Repeated violations of site or activity allocations may result in partial or total withdrawal of the Allowance Holder's Site Allowance. As is standard Agency policy, if an Allowance Holder exceeds either the Regular or Site Allowance, the Budget Division will withdraw obligational authority in accordance with existing procedures.

During the last quarter of the year, the Budget Division will work directly with the Allowance Holders and OSWER as necessary to ensure that all allowances and obligations are aligned prior to year-end closing.

EFFECTIVE DATE

This policy is effective January 1, 1987 and will affect all Regional Superfund allowances issued thereafter.

OUESTIONS

If you have any questions about the policy and change request procedures outlined above, please call Ron Bachand on 382-4165 or Becky Kennedy on 382-2995. Questions concerning the SCAP should be addressed to Terry Quverson on 382-2447.

cc: J. Winston Porter Henry Longest Thad Juszczak Terry Ouverson

Interim Concurrence Procedures for Removal Actions Under Delegated Regional Authority Requiring EPA Headquarters Concurrence

Interim Concurrence Procedures for Removal Actions Under Delegated Regional Authority Requiring EPA Headquarters Concurrence

At the present time, there are three types of removal actions under the delegated Regional Authority which require EPA Headquarters concurrence. They are:

removals using the consistency waiver;

-

- removals with a ceiling of S1 million to S2 million; and
- removals that are nationally significant or precedent-setting.

Outlined below are interim concurrence procedures to be used by the OSCs and RAs to obtain this concurrence. Attached to these interim concurrence procedures are examples of nationally significant or precedent-setting removal scenarios that require EPA Headquarters concurrence. For a Removal Action under delegated Regional Authority requiring Headquarters concurrence, the OSC shall prepare an Action Memorandum that follows the appropriate guidance for the type of action planned (e.g., compliance with the consistency waiver guidance).

The OSC shall notify the appropriate Regional Coordinator in the Guidance and Monitoring Section, ERD, of the Region's intent to request HQ concurrence to initiate a removal (phone: FTS 8-382-2118, Magnafax: 202-755-2155, TWX: 710-822-9269).

The OSC should contact his or her designated Regional Coordinator as soon as possible to alert ERD that an Action Memorandum is being sent for concurrence. Advance notice of a request is important because the Regional Coordinator must send the Action Memorandum through several additional offices for concurrence.

The OSC shall then provide the required information in the Action Memorandum, with a cover memorandum detailing why the Removal Action under delegated Regional authority requires HQ concurrence. The request must be signed by the RA, and addressed to the AA/OSWER through the Director/OERR to the attention of the Director/ERD.

OSCs are encouraged to submit draft Action Memoranda and cover memoranda to their designated Regional Coordinator to ensure that all requests requiring HQ concurrence are complete. In addition, OSCs should send all final Action Memoranda requiring HQ concurrence to their designated Regional Coordinator who will immediately begin the concurrence process.

The Guidance and Monitoring Section, ERD, will review the Action Memorandum, coordinate and gain concurrence from other divisions and offices as necessary (e.g., the Office of Waste Programs Enforcement, Office of General Counsel, and Hazardous Site Control Division), and relay the request and a recommendation for approval/denial to the Director, OERR. The Director/OERR will review the request and forward it with a recommendation to the AA/OSWER for final concurrence. ERD will communicate OSWER's decision to the OSC as quickly as possible. Written confirmation of the decision will be forwarded to the O' by ERD as soon as practicable.

In an emergency, requests may be made verbally. Within 24 hours, the OSC should follow up with an Action Memorandum and cover memorandum signed by the RA.

.

•

.

.

•

•

.

New-Method for Determination of Indirect Costs in Superfund Removal Project Ceiling (Comptroller Policy Announcement Number 87-15)



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUL 15 1987

COMPTROLLER POLICY ANNOUNCEMENT No. 87-15 OFFICE OF ADMINISTRATION AND RESOURCES MANAGEMENT

MEMORANDUM

New Method for Determination of Indirect Costs in SUBJECT: Superfund Removat Project Ceilings - Turkand Baska Dávid P. Ryan FROM: **C**omptroller Assistant Regional Administrators TO: Management Division Directors Senior Budget Officers Regional Comptrollers Waste Management Division Directors Environmental Services Division Directors Regions I, VI and VII Director, Office of Emergency and Remedial Response Director, Office of Waste Programs Enforcement Financial Management Officers

This memorandum announces Agency policy regarding determination of indirect costs in Superfund removal project ceilings.

BACKGROUND

Except in emergency situations, the On-Scene Coordinator (OSC) must prepare a Removal Action Memorandum before initiating removal cleanup action. This memorandum must be signed by the Regional Administrator (RA) or his/her designee. In the event of an emergency action of \$50,000 or less, the OSC, when delegated the authority, may initiate the removal first, and then may prepare an Action Memorandum to be signed by the RA or his/her designee to document that the removal has been approved.

The estimated project cost of the removal must be included in the Removal Action Memorandum and this estimate becomes the project ceiling. Under the Superfund Amendments and Reauthorization Act (SARA), this project ceiling cannot exceed \$2 million or have a duration of more than 12 months unless an exemption is granted. Prior to SARA, the statutory limits were \$1 million and six months,---respectively. With the exception of certain legal and site investigatory costs, which are covered under Section 104(b) of CERCLA the estimated project ceiling covers all direct and indirect costs of work to be performed for the removal project. Indirect costs are those costs necessary to operate the program, but which cannot be attributed directly to specific sites. Examples include program management, indirect salaries and fringe benefits, administrative support, rent and utilities.

In March 1986, the Office of the Comptroller distributed the Superfund Indirect Cost Manual, which provided instructions for calculating the amount of indirect costs to be claimed in cost recovery actions. It also established provisional rates (dollars per hour of Regional direct labor hours) for estimating indirect costs allocable to a specific site. However, the Indirect Cost Manual did not specifically address the use of provisional indirect cost rates in the development of estimated project costs for removal actions. Therefore, On-Scene Coordinators continued to use the methodology contained in the January 1985 EPA Removal Cost Management Manual to project both indirect and direct costs for Superfund removal projects.

POLICY

To allow a uniform recording method, the application of the provisional rates contained in the Indirect Cost Manual should also be used in determining indirect rates for removal costs, rather than the method described in the January 1985 Removal Cost Management Manual, which is currently under revision, and will incorporate this new policy. Accordingly, we are establishing the following policy. For clarification purposes, it restates our existing policy concerning intramural and extramural direct costs, as well as provides the new policy for indirect cost projection.

Extramural Costs

Project ceilings for CERCLA removal actions will include an estimated cost unit for anticipated direct extramural costs at the site. These direct costs include, but are not limited to, cleanup contractor costs (ERCS), subcontractors, letter contracts with State and local governments, orderfor services, notice to proceed contracts, alternative technology contracts, and interagency agreements (IAGs) with other Federal agencies. The "Extramural Cleanup Contractor" costs include a contingency factor that is determined by the OSC, normally 10-20%, as discussed in the Removal Cost Management Manual.

- Other extramural items can include the projected costs for the Technical Assistance Teams (TAT), Contract Laboratory Program (CLP) analytical costs, and the Environmental Response Team (ERT) contracts.
- After reviewing the projected total extramural costs, the OSC will utilize a 15% contingency factor, (again as discussed in the Removal Cost Management Manual). This 15% contingency allows for unforeseen or unanticipated contractor expenses -- which can include delays resulting from poor weather conditions, or equipment failure. Therefore, the total projected direct extramural costs should be increased by this 15% contingency factor. Attachment I, "Removal Project Ceiling Estimate" illustrates this guidance.

Intramural Costs

- The OSC will estimate the cost of EPA personnel, Regional as well as Headquarters, who will charge hours to the site.
- While the OSC can estimate the number of Regional hours planned for the site, it is difficult to project the number of Headquarters hours planned. Therefore, it has been determined that a percentage (10%) of the planned Regional hours be used to estimate Headquarters hours. For instance, if there are 300 Regional hours projected - there would be 30 hours (10% of 300) planned for Headquarters.
- Tracking of actual hours can be accomplished through FMS reports obtained through the Financial Management Office.

Attachment II provides the latest provisional indirect cost rates for FY 1987. Any changes to these rates will be provided to the OSCs by the Regional Financial Management Officer (FMO). The FMO is also available to provide assistance to the OSC in the calculation of the indirect costs. In addition, the Financial Management Division will notify the Emergency Response Division (ERD), Office of Emergency and Remedial Response, when changes in the provisional rates occur, so that ERD can modify the software package currently being used by the OSCs.

EFFECTIVE DATE

This policy is effective for all current removal projects that have not had a Removal Action Memorandum approved as of the date of this policy and for all future site-specific removal activity.

SUNSET DATE

This policy will be incorporated in a forthcoming section of the Resources Management Directives System. When that section is issued in final, you will be notified that this policy announcement is superseded by that document. The next update of the EPA Removal Cost Management Manual will also be revised to reflect this policy.

INQUIRIES

If you have any questions concerning this announcement, please contact Liz Milstead, Fiscal Policies and Procedures Branch, on 382-4205. Questions regarding the EPA Removal Cost Management Manual should be addressed to Jean Wright, OERR, ERD, on 382-2342.

Attachments

cc:J. Richard BasharFMD Branch ChiefsAlvin PesachowitzTim FieldsJohn J. SandyOHM Coordinators, Regions I-XTony MusickTAT/ERCS DPO's, Regions I-XCarole AnshelesERT

Attachment I

<u>SAMPLE</u>

REMOVAL PROJECT CEILING ESTIMATE

EXTRAMURAL COSTS

Extramural Cleanup Contractor \$862,500 (This cost category includes OSCs estimates for: ERCS, Mini-ERCS, subcontractors, Letter Contracts, Alternative Technology Contracts, IAGs with other Federal agencies, etc. Also includes 10-20% contingency) CLP Analytical Costs \$100,000 ERT Contract (EERU) \$100,000 Subtotal, Extramural Direct Costs..... \$1,112,500 15% Contingency of Above Costs \$167,000 (round to nearest thousand) TOTAL, EXTRAMURAL COSTS \$1,279,500 INTRAMURAL COSTS _____ Intramural Direct Costs \$9,900 (\$30 X 330 Labor Hours (300 - Region/30 Hdqtrs) Intramural Indirect Costs \$18,000 (See instructions below) TOTAL, INTRAMURAL COSTS \$27,900 · -TOTAL REMOVAL PROJECT CEILING ESTIMATE \$1,307,400 Formula for Calculating Indirect Costs: - . -.--... Region Specific Estimated Regional Indirect Cost Rate X Direct Labor Hours = Indirect Costs \$60 300 \$18,000

PROVISIONAL' FY 1987 INDIRECT COST RATES

.

I Air, Water, Environmental Services, and Waste Management Division \$60 per hour II Caribbean Field Office; Emergency and Response, Air and Waste Management, Water Management, and Environmental Services Divisions \$68 per hour III All Regional Organizations \$52 per hour IV All Regional Organizations \$54 per hour V Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,	Region	Organizations to Which the Rates Apply	FY 87 Provisional Rates
IICaribbean Field Office; Emergency and Response, Air and Waste Management, Water Management, and Environmental Services Divisions\$68 per hourIIIAll Regional Organizations\$52 per hourIVAll Regional Organizations\$54 per hourVDirectors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,	I	Air, Water, Environmental Services,	S60 per hour
Environmental Services Divisions \$68 per hour III All Regional Organizations \$52 per hour IV All Regional Organizations \$54 per hour V Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,	II	and waste Management Division Caribbean Field Office; Emergency and Response, Air and Waste Management, Water Management, and	yoo per nour
<pre>III All Regional Organizations \$52 per hour IV All Regional Organizations \$54 per hour V Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,</pre>		Environmental Services Divisions	\$68 per hour
<pre>IV All Regional Organizations \$54 per hour V Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,</pre>	III	All Regional Organizations	\$52 per hour
V Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air,	IV	All Regional Organizations	\$54 per hour
Water, and Environmental Services	v	Directors Office - Waste Management/ Emergency Response; Hazardous Waste Enforcement Branch, Great Lakes Coordinator; Central Regional Lab; Eastern and S&A Central District Offices; and Waste Management, Air, Water, and Environmental Services	
Divisions \$61 per hour		Divisions	\$61 per hour
VI Houston Lab; and Air, Pesticides and Toxics, Water Management, Hazardous Waste Management, and Environmental Services Divisions S60 per hour	VI	Houston Lab; and Air, Pesticides and Toxics, Water Management, Hazardous Waste Management, and Environmental Services Divisions	S60 per hour
VII Water Management, Air and Toxics, Environmental Services, and	VII	Water Management, Air and Toxics, Environmental Services, and	
Waste Management Divisions \$65 per hour VIII Air and Hazardous Material Branch; Montana Office; and Water, Air and Hazardous Material, and	VIII	Waste Management Divisions Air and Hazardous Material Branch; Montana Office; and Water, Air and Hazardous Material, and	\$65 per nour
Environmental Services Divisions 562 per hour IX Toxics and Waste Management, Water Management, and Air Management	IX	Environmental Services Divisions Toxics and Waste Management, Water Management, and Air Management	\$62 per hour
Divisions \$63 per hour		Divisions	\$63 per hour
X Alaska, Oregon, Idaho, and Washington Operations Offices; and Air and Toxics, Water, Hazardous Waste, and	x	Alaska, Oregon, Idaho, and Washington Operations Offices; and Air and Toxics, Water, Hazardous Waste, and	
Environmental Services Divisions \$61 per hour		Environmental Services Divisions	Ş61 per hour

-

Timing of \$2 Million Exemption Request Submission



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

APR 4 1988

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Timing of \$2 Million Exemption Request Submissions

FROM: Henry L. Longest II, Director Office of Emergency and Remedial Temponse

TO: Addressees

PURPOSE:

This memorandum is to inform you of the requirements for submitting removal \$2 million exemption or ceiling increase requests to Headquarters for approval.

BACKGROUND:

Over the past six months, Regions have been submitting removal exemption and criling increase requests that have not provided Headquarters with adequate time for review. Several of these requests have been received only a matter of days before Headquarters approval is needed to avoid bringing the work to a stop. The review process at Headquarters involves several offices, including the Office of General Counsel and Office of Waste Programs Enforcement. We need at least three weeks to process a complete, well-justified request.

PROCEDURES:

The procedures outlined in this memorandum apply to planned and ongoing removal actions that may require exemptions from the \$2 million statutory ceiling. You must prepare a \$2 million exemption request as soon as you estimate that a project ceiling will exceed \$2 million. If you estimate the project cost will exceed \$2 million before you start an action, Headquarters approval must be received before the removal action can begin if it is practical to do so. Exemption requests for ongoing sites where additional funds will make the project ceiling exceed \$2 million or ceiling increase requests for sites that have already received exemptions to the \$2 million statutory criteria must be submitted three weeks before Headquarters approval is needed.

CONCLUSION:

Exemption requests signed by the Regional Administrator that are incomplete or insufficiently justified will be returned to the Regions to be rewritten. This will probably result in a substantial delay in obtaining Headquarters approval. You can avoid this delay by starting as early as possible and working closely with the appropriate Headquarters Regional Coordinator throughout the justification and review process. I think your personal attention to these types of requests is critical to their quality and timeliness and will appreciate your help on future submissions. Superfund Branch and Section Chiefs were provided with guidance on October 22, 1987, on the preparation of \$2 million exemption requests that you should find useful.

Addressees:

Director, Environmental Services Division Regions I, VI, and VII Director, Waste Management Division, Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division, Region II Director, Hazardous Waste Management Division, Regions III and VI Director, Hazardous Waste Division Region X Director, Toxics and Waste Management Division, Region IX Earl Salo, OGC cc: Timothy Fields Jr., ERD OHM Coordinators, Regions I-X

Model Program for Removal Site File Management

MODEL PROGRAM FOR REMOVAL SITE FILE MANAGEMENT

.

July, 1988

Office of Solid Waste and Emergency Response and the Office of Administration and Resources Management U.S. Environmental Protection Agency

MODEL PROGRAM FOR REMOVAL SITE MANAGEMENT

•

. •

•

OSWER Directive 9360.2-01

July 1988

Unites States Environmental Protection Agency Office of Emergency and Remedial Response Emergency Response Division Washington D. C. 20460

NOTICE

Development of this document was funded, wholly or in part, by the United States Environmental Protection Agency under contract No. 68-01-7281 to American Management Systems, Inc. It has been subjected to the Agency's review process and approved for publication as an EPA document.

TABLE OF CONTENTS

Page

1.	Overview	1
2.	Site File Structure	3
3.	Site File Kit	. 8
4.	File Management Processes	i1
5.	Supporting the Administrative Record	14

MODEL PROGRAM FOR REMOVAL SITE FILE MANAGEMENT

Overview

Effective file management in the Superfund Emergency Removal Program requires performance of several important functions, particularly:

- Securing necessary file management supplies (e.g., site file kit)
- Transporting file materials to the site
- Setting up the site file
- Completing appropriate documentation in a timely manner
- · Filing documentation in the site file
- Distributing (copies of) appropriate documentation to the regional office
- Replenishing file management supplies
- Periodically reviewing the site file (while action is on-going) for completeness and accuracy
- Determining whether site file documents belong in the administrative record
- Distributing (copies of) appropriate documentation to the administrative record.

These functions have typically been the responsibility of the On-Scene Coordinator (OSC), with assistance from other EPA Regional staff, contractors (e.g., TAT) and the USCG Strike Teams. It is critical that each Region clearly assign responsibility for each of these OSC functions.

The Model File Management Program described in this document was prepared to guide the OSCs, Regional administrative support staff, and other responsible personnel in performance of good site records management. The Model Program consists of four parts:

• Site File Structure - A model for classifying all the documentary material generated in the course of a removal action.

- Site File Kit A recommended kit containing items necessary to support structured collection and storage of information at the site.
- File Management Processes A step-by-step review of the stages of file management for a site.
- Supporting the Administrative Record A recommended approach to ensuring that appropriate site files are placed in the administrative record.

The remainder of this document presents more information on each of these four areas.

2. Site File Structure

One of the OSC's important responsibilities during a removal action is to establish and maintain site files. These files include documentation of all aspects of the removal -- technical, contractual, and financial. Good documentation is vital for proper management of the ongoing removal action and subsequent cost recovery.

Individual Regions currently use varied site file structures. Some of the variety stems from the different levels of complexity involved in removal actions. The individuality of emergency response organizations in the Regions, and of the OSCs within them, also contributes to the variety of site file structures.

There are two important guiding principles for the site file structure:

- Each Region should have a site file structure that can be used consistently.
- All site file structures in use in the Regions should include the same minimum set of information.

The minimum set of information^{*} to be included in all site file structures consists of the following:

- · Chronology of events and decisions
- Entry and exit of personnel and equipment
- Contractor work planned/authorized and contractor work accomplished
- Contractor costs
- Site conditions, such as weather
- Cumulative intramural and extramural project costs.

^{*} Removal Cost Management Manual, Chapter 5, Exhibit 5-1.

4

This minimum set of information is generally documented using the following forms:

- OSC Log
- POLREPS
- Site Entry/Exit Log
- Hot Zone Entry/Exit Log
- Work Reports
- 1900-55s
- Incident Obligation Log.

Since certain Regions already have site file structures that satisfy the minimum requirements, we do not believe that all Regions should be required to use the same detailed site file structure. However, each Region must ensure that:

- It adopts a comprehensive, workable site file structure
- Its Regional OSCs use the structure consistently.

Exhibit 1 presents the model site file structure which meets the minimum requirements outlined above and combines aspects of several Regional site file structures currently in use. This model should be the basis for a file structure in Regions which have not already developed a satisfactory file structure. The model structure consists of five major groupings of documentation, each of which contains a number of specific document types, as follows:

(1) Operational documents are those used in general management of the removal project. These documents are mainly type-written or computergenerated and are generally on letter-size (8.5 by 11 inch) paper. Documents in this category include:

- POLREPS
- Action Memos
- Site Safety Plan
- Contractor Workplans

MODEL SITE FILE STRUCTURE



- Contractor Progress Reports
- OSC Lcg Books
- Performance Summanes
- Performance Evaluation Reports
- Notice Letters
- Section 106 Orders or Other Enforcement Orders

(2) Legal documents include those issued by a court of law as well as other binding documents. These documents are mainly type-written and are generally on letter-size (8.5 by 17 inch) or legal-size (8.5 by 14 inch) paper. Examples of legal documents include:

- Administrative Orders
- Access Agreements
- Warranty Deed
- Release of Liability
- Consent Agréement

(3) Financial documents include 1900-55s, timesheets, travel vouchers, and other items used to document EPA costs for potential recovery actions. These documents may be forms filled out by hand or by computer and may include carbon or tissue copies, as backup (e.g., hotel bills). The forms are generally letter-size, and the backup documentation may be any size. A full list of financial documents includes:

- Procurement Requests
- Delivery Orders
- 1900-55s
- Timesheets
- Travel Vouchers
- Contract Invoices
- Personnel/Equipment Logs
- Daily Work Reports
- IG Audit
- IAGs Between EPA and Other Federal Agencies
- Subcontracting Justification Documentation
- Order for Services Contracts
- State Letter Contracts

Many of these originals (e.g., timesheets, travel vouchers, contract invoices) should be regularly forwarded to the appropriate finance office for processing. OSCs may wish to include copies of these documents in the site files.

(4) **Public relations** documents cover interactions with the local community, the media, public officials, and the public at large. They include such items as media articles, records of communication, and community relations plans. These documents may be typewritten or computer-generated reports on various size paper or they may be originals or copies of newspaper articles or handwritten notes.

(5) Technical documents are more scientific in nature and may include sampling and analysis data, investigations, and waste profiles. These documents may be typewritten or computer-generated and are generally on lettersize paper. Technical documents also include *photographs and maps*. Photographs may be black and white or color, any size from snapshot to enlargement. Maps may be originals or copies on, usually, oversize stock. Separate, oversize binders may be needed to store such items.

3. Site File Kit

Use of Site File Kits will make it easier for OSCs and Regional administrative support staff to establish and maintain the necessary site files. In the Model Program, the Regional office would distribute Site File Kits to OSCs to maintain for use when they initiate new removal actions. Alternatively, OSCs may delegate responsibility for the Site File Kits to the Regional administrative support person or the TAT.

Components of the Site File Kit should be expendable, with each new site receiving a new, complete kit. The Kit should be streamlined and light, making it easy for the OSC (or designee) to hand-carry it, along with personal luggage, to the new site. There should be a single inventory control point in each Region for the site file kits.

Exhibit 2 illustrates the contents of a sample Site File Kit. It consists of a sample case (or other sturdy, portable container) which can be outfitted with pre-labelled folders, blank forms, and basic office supplies. The pre-labelled folders and blank forms conform to the site file structure established in the Region or the model site file structure outlined above. In this example, the Kit contains six folders with the following labels:

- Operational Documents
- Legal Documents
- Financial Documents
- Public Relations Documents
- Technical Documents
- Photographs/Maps

Blank forms include timesheets, travel vouchers, and personnel/equipment logs.

The Kit also contains office supplies "#cessary to "begin business" at the removal site. Such office supplies may ...c.uce:



- Stapler, staples, and staple-puller
- Variety of pencils and pencil sharpener
- Variety of pens (including felt-tip markers)
- Paper clips
- Push pins
- Masking and transparent tape
- Scissors
- Inkless stamps (e.g., ORIGINAL, DRAFT)
- · Post-it notes
- Opaquing fluid
- Calendar
- Document-numbering stamp.

A logical supplement to the Kit is a portable personal computer and a copy of the Removal Cost Management System software which can be used to produce reports and analyze data. With a modern, the computer could be used to transmit computer-generated reports (e.g., POLREPs) to the Regional office and headquarters. The computer can also be used to access reference materials stored on diskette. Such reference materials might include:

- Statement of Work
- Users' manuals
- Directory of ERCS contractors
- Applicable sections of ERCS contract
- · Guidance/procedures on invoices, dailies, etc.
- EPA phone book (both HQ and Region).

While the sample Site File Kit is fairly inclusive, the OSC may require some bulky items which would be better purchased near the site. Such items might include a bulletin board, graphics supplies, 3-ring binders, and blank video tape.

4. File Management Processes

OSCs have the primary responsibility for maintaining site files while the removal site is active. OSCs may be assisted in this responsibility by Regional administrative support staff. TAT and other contractors, and USCG Strike Team personnel. The OSC (or his/her delegate) must complete the appropriate forms and reports in a timely fashion and file them according to the site file structure endorsed by the Region.

1.1

Physical location of the various site file components depends on both the nature of the removal project and the organization of the Regional office. In general, the OSC must have direct access to complete documentation (either originals or copies) from the time that a removal action is initiated until the final OSC report is published.

The major steps involved in site file management follow the life-cycle of the site-specific removal action:

- Project initiation
- · Short-term actions with no command post
- · Long-term actions with command post
- Preparation of the final OSC report.

Each of these steps is described in greater detail below.

a. Project Initiation

When a project begins, the OSC (or delegated person) should bring a new site file kit to the site when he/she first visits it. The Regional administrative support person may later bring additional site filing and office supplies. It is most important, however, that some form of site file kit, even abbreviated, be available on site at project start.

5. Supporting the Administrative Record

Detailed agency policy and procedures on handling the administrative record requirements of SARA are issued separately by OWPE. It is recommended that the Regions establish a regular process for reviewing active site documents against the current criteria for inclusion in the administrative record.

The suggested steps in this process are:

- The OSC (or designated person) brings all new site documents to the Regional office on a regular basis for copying and/or storage in Regional files.
- At the same time, the OSC or other Regional staff responsible for file management reviews the removal action materials for possible inclusion in the administrative record.
- The OSC or other designated Regional staff (or contractors) prepares copies of selected items for the Region's administrative record and, if appropriate, the local administrative record maintained on or near the site.

A flow chart summarizing this process is in Exhibit 3.





Site Specific Contracting for Removals

OSWER Directive In	itiation Requ	est 9242.2-02
2. Criginator Info	Imation	
Name of Contact Person Max Code	CHICE /POP	. Telephore Cade
Bruce Engelbert		382-2188
Site Specific Contracting fro Re	movals	
Provides direction on use of sit	e specific contra	cts for removal actions.
Keywords Superfund, CERCLA, SARA		
E. Does This Cirective Sugersede Previous Grective(s)?	No. Yes wh	et directive (number, title)
L Coes it Supplement Previous Onective(s)?	No Yes We	el droctive (number, tille)
Draft Level	rector 🗌 C – Far Re.	new & Comment D - in Developin
8. Document to be distributed to Stat	es by Headquarter	67 X Yes
his Request Meets OSWER Directives System Format Standard		
Betti VanEpps OERR Directives	oordinator	4/10/89
Name and Title of Approving Official		Cate
Henry L. Longest II, Director,	OERR	4/10/89
Form 1315-17 (Rev. S-47) Provide anticas are consults		

OSWER OSWER OSWER O <u>E DIRECTIVE DIRECTIVE</u> DIRECTIVE



APR 1 0 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPO

OSWER Directive #9242.2-02

MEMORANDUM

SUBJECT: Site-Specific Contracting for Removals

FROM: Henry L. Longest II, Director

David J. O'Connor, Director Walled J. C'linnor Procurement and Contracts Management Division

TO: Environmental Services Division Directors Regions I, VI, VII Waste Management Division Directors Regions I, IV, V, VI, VIII Emergency and Remedial Response Division Director Region II Hazardous Waste Management Division Directors Regions III, VI Toxics and Waste Management Division Director Region IX Hazardous Waste Division Director Region X

Purpose:

This memorandum provides direction on the use of site-specific contracts for removal actions.

Background:

A key component of the removal response strategy is site-specific contracting. Site-specific contracts are desirable because they promote competition, which leads to getting the best cleanup price possible for removal actions. The removal program has made minimal use of site-specific contracts. In FY'88, no site-specific contracting was done. In FY'87, only 4 of the 256 removals started used this approach. In FY'86, after an Inspector General audit of the Emergency Response Cleanup Services (ERCS) contracts and congressional hearings, a new removal contracting strategy was developed. The plan called for:

- Large zone ERCS. These contracts would ensure adequate capability to respond to emergency and time-critical releases.
- Regional ERCS (mini-ERCS). These contracts would supplement the zone ERCS, but would not have such strict minimum response times. The smaller size of these contracts would help to open up competition and get more companies involved in the program.
- Site-specific contracts. These contracts would get the government the best cost in those situations where the time for competition is available. In order to help reduce the necessary lead time.
 a strategy of technically pre-qualifying contractors (PQOPS) is being established. Again, the small and well defined nature of site-specific contracts will help to increase the number of companies participating in the program.

Objective:

This memorandum is intended to encourage the use of site-specific contracting in as many removal actions as the circumstances allow. Right now, many good opportunities for site-specific contracting are being overlooked. In each of these cases, we could be saving valuable funds, as well as ERCS capacity, that could be used for other removal actions. In non-emergency situations, a Region should always consider whether there will be time and resources to do a site-specific contract. The Regional removal program should work with its contracting officer (CO) to decide the best contracting approach.

Site-Specific Criteria:

Site-specific contracts are contracts, competed through an EPA CO, to obtain carefully defined cleanup services. Regions should consider the following factors when deciding whether or not to do a site-specific contract.

 Lead Time -- Generally, a site-specific contract will take about four months to compete. Many removal actions are emergencies or must be started in less than four months. But 15 to 20 percent of removal actions, including most of those involving an alternative technology, have a sufficient planning horizon or will last long enough so that some portion of the work can be done through a site-specific contract. During the update of the removal Superfund Comprehensive Accomplishments Plan (SCAP) each quarter, you should give serious consideration to possible site-specific contracting candidates. We understand that not all sites in the SCAP, with a scheduled start date beyond four months, are good candidates, since they may be low priority projects that are likely to be deferred or not done at all. In some cases, it will make sense to do site-specific contracts at actions that start out as emergencies. If the cleanup work will take a substantial amount of time, as is often the situation when a \$2 million exemption is granted, perhaps it will be practical to do the latter stages of the work site-specifically. Except in very unusual circumstances, any removal where an Engineering Evaluation/Cost Analysis is necessary should be done through a site-specific contract.

- 2. <u>Cost</u> -- The larger the action, the greater the potential for savings if a site-specific contract is used. Actions that will cost less than \$500 thousand are probably not good candidates for site-specific competition. Actions costing over \$2 million should generally have some component of the work that is done on a site-specific basis.
- 3. <u>Complexity</u> -- Actions, or parts of large actions, that are relatively straightforward and uncomplicated, are the best candidates for site-specific contracting. Since site specific contracting requires being able to write a clear statement of work, unusual or poorly defined situations should be handled through the ERCS mechanisms. For example, a site that has a lot of unidentified materials should probably be done through ERCS, at least up to the transportation and disposal phase of the work. On the other hand, if the action is basically excavation and disposal or incineration, and the contamination type and amount is well understood, a site-specific contract may make sense.
- 4. <u>Management</u> -- Site-specific contract preparation and management will require CO assistance and a small amount of program FTE beyond what an ERCS action would require. Consequently, the projected dollar savings should outweigh the extra staff time that must be invested. This determination should be made in consultation with the appropriate CO, who should always be in on major contracting choices in non-emergency situations. In FY'90, Regions will get a small increment of FTE just for site-specific contracting. Also, the Environmental Response Team (ERT) and its Response Engineering and Analytical Contract are usually available to provide support in developing the statement of work for a site-specific contract.

Implementation:

In non-emergency situations, Regions should start with the assumption that a site-specific contract is the way to proceed. Site conditions, resource constraints or other factors may quickly convince the On-Scene Coordinator (OSC) that a zone or Regional ERCS contractor is the better response route. Where there is a site-specific potential, an early discussion with a CO is important. The CO can help sort through the schedule and steps to see if the approach is feasible. All of the Regions already have or soon will have their own CO, so this consultation will be relatively easy. If the response will be done through a site-specific contract, the OSC is responsible for writing the scope of work, organizing any necessary technical evaluation of the offerors, and serving as project officer on the contract. The CO will take care of managing the advertisement, bid review, negotiations and actual award of the contract. Clearly, the process will take some additional OSC time (a rough estimate is 40 to 120 hours beyond what the project would probably take under ERCS). The ERT is often available to provide assistance in preparing the scope of work. And once the PQOPS pools of technically qualified contractors for incineration and fixation are in place later this summer, there will be time savings in the technical review process. In many cases, particularly those involving alternative technologies, the small incremental OSC time commitment is well worth the overall cost savings on the project.

Conclusion:

Site-specific contracts, under the right circumstances, can save considerable funds that can be used for other removal actions. There are no mandatory requirements or quotas for site-specific contracts. This approach should only be used where Regions believe that the removal action schedule and work requirements can be met. There are many instances where this will be the case, so the removal program needs to make a greater commitment to using sitespecific contracts in the future.

Need for Contract Officers Authorization Before Contractor Activation

EPA	DIRECTIVE NUMBER: 9242.6-03
	TITLE: Need for Contract Officers Authorization Before Contractor Activation
	APPROVAL DATE: 8/9/89
	EFFECTIVE DATE: 8/9/89
	ORIGINATING OFFICE: OERR/IO
	FINAL
	DRAFT
	STATUS:[]A- Pending OMB approval[]B- Pending AA-OSWER approval[]C- For review &/or comment[]D- In development or circulating
	REFERENCE (other documents): headquarters

Journal State

OSWFR OSWFR OSWFR CORECTIVE DIRECTIVE D

United States Environmental Protection Agency Washington DC 20450	1 Directive Number
OSWER Directive Initiation Request	9747 6-03
2. Originator Information	
Name of Contact Person Mail Code Office	Telephore Coae
Sherry Hawkins OS-200 OERR/10	202-382-21.90
Need For Contract Officers Authorization Before Contractor Act	ivation
4 Summary of Orective (include onel statement of purpose)	
This memo is to remind all OERR staff that contractor perform <u>not</u> be initiated until the contracting officer has given form	ed work must al authorization.
5. Keywords Superfund, CERCLA, SARA Contractor Activation	
6a. Ones This Directive Supersede Previous Directive(s)?	(number, title)
b. Does It Supplement Previous Directive(s)? No Yes What directive	(number, ttie)
7 Draft Level A - Signed by AA/DAA X 8 - Signed by Office Director C - For Review & Corr	iment D - In Oevelopment
8. Document to be distributed to States by Headquarters?	Yes Na
This Request Meets OSWER Directives System Format Slandards.	
9. Signature of Lead Office Directives Coordinator	Dete
0 Name and Title of Approving Official	Date
PA Form 1315-17 (Rev. S-87) Previous editions are obsolete.	

OSWER OSWER OSWER O VE DIRECTIVE DIRECTIVE



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

AUG 9 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

OERR Directive No. 9242.6-03

MEMORANDUM

- SUBJECT: Need for Contract Officers Authorization Before Contractor Activation
 - FROM: Henry L. Longest II, Director Office of Emergency and Remedial Free Pense
 - TO: All OERR Staff

Purpose

This memo is to remind all OERR Staff that contractor performed work must <u>not</u> be initiated until the contracting officer has given formal authorization.

Background

As you know, the issue of Superfund contracts management has been intensely analyzed by Congressional staff, GAO, the IG, and others. We must assure strict adherence to <u>all</u> contract procedures to avoid any appearance or occasion of mismanagement or conflict or interest. Furthermore, a contractor is at risk for any work performed before the authorization date.

Implementation

Until the contract is approved by the contracting officer, discussions concerning the project with the firm or potential firm must be related to preliminary scoping matters only; no work may commence before authorization from PCMD. To speed the procurement process, specify a date by which PCMD approval is needed. PCMD will always strive to accommodate specific data requirements; "ASAP" requests are processed after data specific requests.

cc: Cannon (OSWER O'Connor (PCMD) Harper (PCMD) Dietrich (CORAS) **United States Environmental Protection** Agency .

Solid Waste And Emergency Response (OS-240)

•

OERR 9200.0-10B January 1991



Superfund Removal Procedures

Directives Notebook Volume 2





OERR Directive 9200.0-10B September 1991

SUPERFUND REMOVAL PROGRAM DIRECTIVES NOTEBOOK

VOLUME TWO

Office of Emergency and Remedial Response U.S. Environmental Protection Agency Washington, DC 20460

TABLE OF CONTENTS

VOLUME II

AUGUST 1991

REMOVAL/REMEDIAL INTERFACE	DATE	COMMENTS
Relationship of the Removal and Remedial Programs Under the Revised NCP (OSWER Dir. 9360.0-06A)	03/17/86	Supersedes OSWER 9360.0-6
Use of Expanded Removal Authority to Address NPL and Proposed NPL Sites (OSWER Dir. 9360.0-14)	02/07/87	
Unaddressed NPL Sites (OSWER Dir. 9200.2-01)	07/06/89	
Use of Removal Approaches to Speed Up Remedial Action Projects (OSWER Dir. 9355.0-25A)	07/06/89	
Accelerated Response at NPL sites Guidance (OSWER Dir. 9200.2-02)	12/15/89	
Update to the "Procedures for Completion and Deletion of NPL Sites" Guidance Regarding the Performance of the Five- Year Reviews (OSWER Dir. 9320.2-3B)	12/29/89	Supersedes OSWER Dirs. 9320.2-02, 02A & 9320.2-03
Interim Guidance on Addressing Immediate Threats at NPL Sites (OSWER Dir. 9200.2-03)	01/30/90	
<u>STATES</u>		
Guidance for State-Lead Removal Actions (OSWER Dir. 9375.1-4-W)	07/10/87	Supplements OSWER Dir. 9360.0-6
Clarification on Allowability of Management Assistance to States for ERAs and Removals (OSWER Dir. 9375.1-13)	06/10/88	

ENFORCEMENT	DATE	COMMENTS
Issuance of Administrative Orders For Immediate Removal Actions (OSWER 9833.1)	2/21/84	
Procedures for Authorizing Application for Civil Search Warrants Under CERCLA	04/03/87	
Entry and Continued Access Under CERCLA	05/05/87	
Interim Guidance on Notice Letters, Negotiations, and Information Exchange (OSWER Dir. 9834.10)	10/19/87	
Draft Guidance on CERCLA 106 Administrative Orders for Removal Actions	06/19/89	
DRINKING WATER		
Interim Final Guidance on Removal Action Levels at Contaminated Drinking Water Sites (OSWER Dir. 9360.1-10)	10/6/87	Updated by 04/19/88 Memo on Drinking Water Action Levels
Information on Drinking Water Action Levels	04/19/88	Supplements OSWER Dir. 9360.1-10
ARARs		
Policy on Flood Plains and Wetlands Assessments (OSWER Dir. 9280.0-02)	08/06/85	Supplements OSWER Dir. 9280.0-1
CERCLA Compliance with other Environmental Statutes (OSWER Dir. 9234.0-02)	1 0/02/85	Supplemented by OSWER Dirs. 9234.0-3 and 9330.2-1
Discharge of Wastewater from CERCLA sites into POTWs (OSWER Dir. 9330.2-04)	04/15/86	
Occupational Safety and Health Administration: Labor Federal Register Vol 51, No. 244	12/19/86	
Interim Guidance on Compliance with Applicable or Relevant and Appropriate Requirements (OSWER Dir. 9234.0-05)	07/09/87	Supplements OSWER Dir. 9234.0-02

MISCELLANEOUS	DATE	COMMENTS
Notification of Restrictions on Reimbursement of Private Party Costs for Removal Actions (OSWER 9225.0-3)	11/25/85	
Analytical Support for Superfund (OSWER 9240.0-2)	03/20/86	
Guidance on Producing Superfund Guidance Documents (OSWER Dir. 9200.4-1)	02/09/87	
Comparison of EPA and Commonwealth of Kentucky PCB Cleanup Standards	03/31/87	
Concurrence on Payment of Relocation Costs for Business During Removal	06/15/87	
Employee Occupational Health and Safety (OSWER Dir. 9285.3-02)	07/07/87	
EPA Interim Guidance on Indemnification of Superfund Response Action Contractors Under Section 119 of SARA (OSWER Dir. 9835.5)	10/06/87	
Significant New Legislation- Federal Employees Liability Reform and Tort Compensation Act of 1988	12/30/88	
OSHA Final Rule, Hazardous Waste Operations and Emergency Response (29 CFR 1910) Docket No. S-760A	03/06/89	Supersedes 12/19/86 Interim Final Rule; See 07/07/87 OSWER Dir. 9285.3-02
Treatability Studies Contractor Work Assignments (OSWER Dir. 9380.3-01)	07/12/89	
Final Guidance on Administrative Records for selecting CERCLA Response Actions (OSWER 9833.3A-1)	12/03/90	Supersedes 03/01/89 OSWER Dir. 9833.3A
Furthering the Use of Innovative Treatment Technologies in OSWER Programs (OSWER 9380.0-17)	06/10/91	

Relationship of the Removal and Remedial Programs Under the Revised NCP

	<u> </u>	nited States Environmi	ental Protection Agency		Interim Girective Number
SEPA	OSWE	R Directive	Initiation Reg	uest	9360.0-6A
		L May Cade		Telenone Numbe	,
Cheryl Ha	wkins	WH-5488		382-5650	•
sad Cilice			1 1 -00000	WBINB R 101	
		Signatu gol Office C	reflar U. M.		Date 1 last
asw	CI AL-OSWER	Finol	ly pello	fro H.L.	3/18/86
Title	-				
"Relation	iship of The Rend	oval and. Remed	ital ProgramsUnde	er the Revise	d NCP"
		-			
Suggery of Directive	·····				
Pavisions	to the NCP rade	afine removal a	and remedial acti	ons to exped	ite
cleanup a	ctivities. This	document addr	resses management	issues that	may
arise bet	ween the two pro	ograms in Headd	quarters and the	Regions when	-
· implement	ing the new prog	gram definition	15.		
				•	
	•				
The of Dungting Ma	WAL POLICY DIRECTIVE AND			Status	•
· · · · · · · · · · · · · · · · · · ·			.		
Poincy UI	rective	•	• •	EV Fierd	
			:	1	
Sees this Directive Su		The size of the si		ent Frevious Directi	
IT "Yes" to Either Que	- 	Superce	edes draft documer	it (Same Titl	e) 9360.6
				· · · · · · · · · · · · · · · · · · ·	(4360.0-6)
Review Plan					
			Cifter (Specify)		
KA OERR	CT OWPE				•
	L Aegians				• • •
This Acquest Meets	SV/ER Directives System	Format		<u> </u>	
	Ca Anacines Onicer				1.0.44
NYMAN /					010 2/01
10010010	Uncotons				2/2.3/86
ignature of OSWER	WMGHJONS Girecziver Ollicer			<u>-</u>	2/2.3/86
ignature of OSWER	Umantons Girective Otticer	<u> </u>			2/2.3/86
ignalure of OSWER	WMGHJON Girecuver Ollicer				0 are 2/2 <i>:</i> 3/86 0 are



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

MAR 17 1986

MEMORANDUM

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

- SUBJECT: Relationship of the Removal and Remedial Programs Under the Revised NCP
- FROM: Henry L. Longest II, Director **H** I Office of Emergency and Remedial Response
- TO: Environmental Services Division Directors, Regions I, VI, and VII Waste Management Division Directors, Regions I, IV, V, VI, VII and VIII Emergency and Remedial Response Division Director, Region II Hazardous Waste Management Division Director, Region III Toxics and Waste Management Division Director, Region IX Hazardous Waste Division Director, Region X

Attached is the final issue paper entitled "Relationship of the Removal and Remedial Programs Under the Revised NCP". This paper has been revised to reflect comments and suggestions submitted by the Regions during the review of the draft paper. The comments also raised several additional issues, which we will defer for incorporation into future procedural guidance. Among these issues are program specific requirements for decision documentation, community relations activities, State involvement, public review and NEPA compliance, deletion activities, enforcement, and contract execution.

As I indicated in my previous memo on the topic, there are other policy issues, e.g., operation and maintenance, which we are examining as a result of this change in removal criteria. We intend to involve the Regions as we make decisions on all of these issues.

Attachment

cc: J. Winston Porter, OSWER Jack McGraw, OSWER Gene Lucaro, OWPE Walt Kovalick, OERR Russ Wyer, HSCD Tim Fields, ERD Steve Lingle, HRSD Sherry Hawkins, ERD Margie Russell, OSWER Dan Berry, OGC Pam John, PCMD Jim Lounsbury, PAS

PURPOSE :

Revisions to the National Contingency Plan (NCP) redefine the response categories of "removal actions" and "remedial actions" so that removals now include all activities formerly considered immediate removals, planned removals, and initial remedial measures (IRMs). While these changes in response categories will expedite many cleanup activities by avoiding previous remedial requirements for remedial investigations/feasibility studies (RI/FSs) and full cost effectiveness analyses, this expanded definition of removal actions may raise questions regarding the relationship between the two programs in both Headquarters and the Regions. This memo addresses some of these questions and provides overall information on the interface of the removal and remedial programs under the new NCP.

BACKGROUND:

Earlier CERCLA program implementation tended to clearly differentiate certain activities as removal or remedial, and those activities fell accordingly into the removal or remedial program organizations. Subsequent experience in CERCLA program implementation and new program directions being implemented via the revised NCP indicate that a higher degree of program integration and flexibility will enable us to stabilize or cleanup a greater number of sites in a more expedited manner. In the case of cleanups, this will allow us to delete more sites from the National Priorities List (NPL). While this goal is seen as highly desirable, it creates some "gray" areas with regard to managing actions that ware termed "initial remedial measures," but that are now defined as removals.

Despite the MCP changes, certain types of response actions continue to fall into discrete programmatic areas. In the view of OERR, the following actions are still logically located in the removal and remedial programs, respectively.

Removals are taken to abate or mitigate threats to public health, welfare, or the environment and are generally surface cleanups. Actions considered to be within the area of removal program responsibility are:

- all response actions at non-NPL sites and releases
- stabilization actions at NPL sites prior to initiating remedial activities
- classic emergencies arising at ongoing remedial actions that require On-Scene Coordinator (OSC) expertise and fast-track contractor activation
- emergency provision of an alternate water supply
- complete cleanup of NPL sites, where consistent with remedial program scheduling/activities, the public interest, and within the purview of a defensible removal action.

Remedial actions are generally complete site cleanups that address ground water impacts, where necessary. Actions within the area of remedial program responsibility are:

- traditional, pre-planned remedial activities
- removal actions determined to be necessary during the course of a remedial action, within the expertise of the remedial project manager (RPM), and for which sufficient time exists for competitive contracting procedures.

Certain activities or phases of a response action, however, may not lend themselves to classification into these specific program areas, but rather require effective program integration and management flexibility to implement successfully. Examples of these situations are:

- A massive drum removal at an NPL site resulting in nearly complete site cleanup which creates a question of whether the removal or remedial program should complete the cleanup.
- A completed removal at an NPL site which raises the question of whether the removal or remedial program should continue action and perform the remedial investigation under the CERCLA 104(b) authority to document that the removal action has cleaned up the site, in accordance with the NPL site deletion guidance and the NCP.

In situations such as these, there appears to be a need for a management prerogative to assign program responsibility on a case-by-case basis. A manager may determine that an IRM-type removal is necessary and that it should be handled by the remedial program for the sake of continuity in an on-going remedial action, or the manager may determine that the action should be performed by the removal program to expedite a time-sensitive response action.

DISCUSSION:

OERR supports the concept of organizational flexibility in this area so that IRM-type removals or certain other removal activities may be undertaken using either remedial or removal program resources. It seems reasonable to deal with these programmatic overlaps on a case-by-case basis that allows flexibility to 1) use appropriate program expertise, 2) use the most advantageous contract mechanism, 3) use realistic time and urgency factors when planning a response, and 4) assure consistency of removals with longer term remedial actions. The implementation of this concept would be the responsibility of Regional program managers who, when making actual response decisions, need the flexibility to assign available and appropriate expertise (OSC or RPM) and dollar resources to specific cleanup situations that may overlap traditional definitions of removal and remedial.

Since the goal of the program is to clean up sites as quickly as possible in the most cost effective manner, Fund expenditures must continue to be carefully controlled to assure expeditious cleanup or mitigation at reasonable cost to the public. Therefore, it is essential that the most appropriate contracting mechanisms are used to assure maximum return for expenditures. The remedial construction program, through the Corps of Engineers (COE) and the REM contracts, uses the invitation for bid procedures to secure lump sum or fixed unit price contracts. The removal program uses Emergency Response Cleanup Services (ERCS) contracts to allow quick response on a pre-negotiated and pre-competed time and materials basis. In addition, the removal program may procure fixed price contracts on a case-by-case basis, if necessary, and if time permits.

IRM-type response activities may lend themselves, depending upon the circumstances, to any of these procurement mechanisms. The ERCS contract system provides expedited and knowledgeable response capability. REM contracts and COE contracts provide cleanup capability when time allows for full competition. ERCS should generally be reserved for removal and - IRM-type situations requiring rapid response such as drum removals or lagoon drawdowns. Other IRM-type removals with 4-5 months lead time have probably been identified in the FY 86 SCAP and could be pursued through REM or remedial construction contracts. Thus, it may often be desirable to pursue fixed-price contracts to address certain removal actions where urgency is not a critical factor. Where urgency is a factor, ERCS is the preferred contract mode. It must be noted, however, that a contracting warrant and appropriate training is a prerequisite to an OSC or RPM using the ERCS contracts.

A test program is being developed by HSCD to facilitate removals at NPL sites where remedial actions are underway and time exists for competitive fixed price contracting. Removal actions taken under this pilot program will be performed by remedial contractors to ensure continuity with RI/FS activities and schedules as well as consistency with the final remedial alternative. The approach will entail the use of an Engineering Evaluation/Cost Analysis (EE/CA) procedure, plans and specifications development, competitive bidding and construction management to plan, design, and implement the project. Initial pilot cases will include corrections to a landfill and provision of alternate water supplies. Guidance for performing EE/CAs is under development.

Further, for certain "non-urgent" removals, it may be desirable to perform a limited cost analysis that is consistent with the Guidance Document for Cleanup of Surface Tank and Drum Sites and draft guidance on EE/CAs. Where time allows, such an analysis could help assure selection of the best technical option at the best price. Such an analysis would not be appropriate for an urgent removal situation.

Response personnel are also reminded that it is EPA policy to provide potentially responsible parties (PRPs) the opportunity to perform the response actions described in this memo pursuant to a CERCLA §106 Administrative Order on consent. Where PRPs have been identified and do not take appropriate or timely actions, EPA will issue a unilateral order where appropriate. Where the decision is made to initiate a Fund-financed action because of the exigencies of the situation, the Agency will also pursue cost recovery for all costs incurred in conducting the removal action. EPA will also pursue treble damages where PRPs fail to comply with an administrative order without sufficient cause. More specifically, once a site has been identified for removal action the Region must determine the immediacy and seriousness of the release situation.

-3-

At sites presenting an extremely urgent situation (e.g., delay of start-up of on-site work cannot exceed one week following determination of the need for a removal), the Region should make a reasonable effort to identify PRPs and notify the parties verbally as to their potential liability followed by a notice letter as soon as possible. The parties should also be given a limited time to respond to the request for conducting the removal action. In the event the negotiations are successful and PRPs agree to undertake the removal action, the agreement should be embodied in an Administrative Order under \$106 of CERCLA. It is also EPA policy to proceed with a unilateral administrative order if PRPs fail to respond appropriately to the request provided necessary criteria are met. For non-urgent removals, procedures for obtaining PRP response should be essentially the same as those for remedial actions. Notice letters should be issued to PRPs, negotiations should be scheduled quickly in order to secure private party cleanup within an established timeframe consistent with the conditions presented by the site. After negotiations, the procedures for issuing administrative orders are the same as those described above.

CONCLUSIONS

Experience has demonstrated that all removals are not necessarily urgent and that all remedial actions are not necessarily deferrable. Having program flexibility to allow certain IRM-type measures to be performed under the supervision of remedial staff and be deferred (because of longer contracting procedures) or to have these actions performed by removal staff on an expedited schedule, gives managers a means of realistic and justifiable scheduling of response actions. This flexibility also allows for management continuity and accountability within programs, and for the use of appropriate expertise.

Use of Expanded Removal Authority to Address NPL and Proposed NPL Sites



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

_FEB -7 1987

/ OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Use of Expanded Removal Authority to Address NPL and Proposed NPL Sites

FROM: Henry L. Longest II, Director Office of Emergency and Remedial Response

TO: Waste Management Division Directors, Regions I-X Environmental Services Division Directors, Regions I, VI, and VII

The Superfund Amendments and Reauthorization Act of 1986 (SARA) and 'the recent National Contingency Plan (NCP) revisions expand removal authority: in several ways. The Office of Solid Waste and Emergency Response (OSWER) is exploring methods to use this increased authority innovatively to enhance efforts to clean up sites on the proposed and final National Priorities List (NPL).

A number of provisions in SARA broaden removal authority. First, the amendments revise the statutory limits on removal actions from \$1 million and 6 months to \$2 million and 12 months. The new law also provides for a waiver of these limits where "continued response action is otherwise appropriate and consistent with the remedial action to be taken." The three "emergency criteria" contained in the prior law still exist. The new waiver, however, is independent of the original three exemption criteria. This means that an exemption may be granted if the proposed removal action meets the original three criteria, or the fourth criterion alone. Another new provision in SARA requires removal actions to "contribute to the efficient performance of any long-term remedial action" to the extent practicable. Finally, SARA establishes a preference for using alternative technologies and resource recovery techniques to achieve more permanent solutions.

In addition, the NCP revisions expanded removal authority by authorizing removal actions to be taken in response to threats, rather than just immediate and significant threats. This increased authority allows remedial operable units, e.g., the former initial remedial measures, to be implemented under noval authority. OSWER would like to examine whether these new legislative and regulatory removal authorities can be used to clean up, or substantially clean up, any of the final or proposed NPL sites. It may be possible to delete certain sites from the NPL, or to achieve substantial cleanup, by performing a remedial operable unit under the expanded removal authority. Each Region is, therefore, requested to evaluate its final and proposed NPL sites and to identify those sites that may be appropriately addressed using removal authority. Such sites must meet the criteria for initiation of a removal action in Section 300.65 of the NCP, and the potential response action should generally remain within the new \$2 million/12 month removal limits. OSWER will consider approval of actions that exceed the \$2 million limit by a reasonable amount if the additional funds are needed to conduct an efficient response. Complicated response actions that require extensive study, such as the cleanup of a contaminated aquifer, would more properly be addressed using remedial authority.

Each Region is requested to provide us with the following information:

- * List of potential NPL sites for removal action.
- * For each site, a brief description of
 - Site background
 - Threat
 - Proposed removal action
 - Estimated cleanup cost.

The appropriate removal, remedial, and enforcement personnel in the Region should coordinate with each other in this effort. Please submit this information to Tim Fields, Director, Emergency Response Division, no later than February 27, 1987.

cc: J. Winston Porter Jack W. McGraw Tim Fields Russ Wyer Superfund Branch Chiefs, Regions I-X

Unaddressed NPL Sites



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C 20460

COLIC WASTE AND EMERGENCY RESPONS

175,02.05

ຸຫຼາ 7 1989

MEMORANDUM

- SUBJECT: Unaddressed NPL Sites
- FROM: Bruce A. Engelbert, Chief Response Operations Branch
- TO: OHM Coordinators Regions I-X

I have attached the memorandum from Henry Longest, which requests that information be supplied regarding unaddressed NPL sites. As we discussed in Kansas City, you need to develop realistic schedules for conducting the removal site assessments and provide this schedule and any other corrections or information regarding these sites to us by July 21, 1989. CERCLIS will be used to track the completion of removal site assessments and this information should be entered into the new CERCLIS field so that we can accurately track progress in this area. For the Federal facilities included in the attached listing, CERCLA fund money may not be used to conduct the removal site assessments. Other government agencies are responsible for Federal facilities. If you have any additional questions, please feel free to call me on 382-2188.

attachment

cc: Tim Fields, ERD ROB Staff John Riley, RSCB Christina Griffin, RSCB



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

WASHINGTON, D.C. 20460

sten

JUL 6 1989

CEFICE OF SOLID MASTE AND EVERGENCY RESPO

OSWER Directive = 9200.2-

MEMORANDUM

SUBJECT: Unaddressed NPL Sites

FROM: Henry L. Longest II, Director Green (05-200)

TO:

Director, Waste Management Division Regions I, IV, V, VII, VIII
Director, Emergency and Remedial Response Division Region II
Director, Hazardous Waste Management Division Regions III, VI
Director, Toxics and Waste Management Division Region IX
Director, Hazardous Waste Division, Region X
Director, Environmental Services Division Regions I, VI, VII

<u>Purpose:</u> The purpose of this memo is to request submission, by July 21st, 1989, of a status report on unaddressed NPL sites, and to reiterate our intention to complete removal assessments at all unaddressed sites by September 30, 1989.

Background: Through the recently released Superfund Management Review, the Administrator has directed us to quickly assess NPL sites to determine whether actions (fund-lead or enforcement) are needed to render sites safe from immediate hazards to public health and the environment. In implementing this initiative, we have set a goal to conduct removal assessments at all unaddressed NPL sites by the end of FY 1989 (unaddressed sites are sites which have had neither removal nor remedial activity).

As you may be aware, I have recently instructed my staff to begin an investigation of these unaddressed NPL sites. This effort was initiated with a March 6th memo to the Division Directors requesting a status report on all unaddressed, final NPL sites. Your responses to that memo provided much insight into the nature of these sites. Perhaps most importantly, your replies indicated that many of these sites <u>have</u> received some substantive action either through State efforts or the efforts of other government agencies (as in the case of Federal facilities).

<u>Implementation</u>: To immediately implement the recommendations of the Management Review, I am expanding our investigation to includsites proposed for addition to the NPL as well as those added to the final list since the March memo and I am including the requirement that removal assessments be conducted at these unaddressed NPL sites. Since the removal assessment requirement was not part of the original March exercise, I have attached the March 6th list of unaddressed, final NPL sites (Attachment A) for your review. The attachment listing the new group of sites (Attachment B) is broken into two categories; those sites which were proposed to the NPL prior to Update 7 (June 1988) and those sites proposed in Updates 7 and 8.

As with the March exercise, I am requesting that you review your site lists (Attachment B only) and determine if the CERCLIS data showing no removal, remedial, or enforcement activity is correct. Please pay particular attention to whether State enforcement lead or Federal facility activity has or should be indicated. If the data is correct, provide me with your planned course of action for the site in question including current site status and plans for any upcoming RI/FS or removal activity or associated enforcement activity.

You will need to use both Attachments A and B for the removal assessment exercise. Sites which we believe have received an adequate removal assessment are marked with an * beside the EPA I.D. on the attachments. As removal assessments are completed at the remaining sites, enter this data into CERCLIS. To allow for this, we have created a new event in the database for you to record the completion of the removal assessment. We will monitor progress towards our September 30th goal through the database. In conducting the removal assessment, Regions should be sure to draw on the wealth of data gathered during the course of the listing process. This data can also be used in determining if a removal assessment has already been conducted for the site in question.

If you have any questions regarding the site status reports or the attached lists, please contact Tom Sheckells at 8-382-2466. Questions concerning the removal assessments should be referred to Bruce Engelbert at 8-382-2188. A recap of the implementation schedule for this memo follows:

Action

Timeframe

Review attached list of unaddressed Status reports/CERCLIS sites (Attachment B). Correct CERCLIS corrections due COB 7/21 if data is incorrect, submit site status report to HQ on legitimate unaddressed sites.

Conduct removal assessments at unaddressed NPL sites (Attachments A and B). Code completion dates into CERCLIS.

September 30, 1989

- 2 -
Attachments

cc: Superfund Branch Chiefs, Regions I-X Jon Cannon OERR Division Directors Bruce Engelbert Penny Hansen

.

Attachment A (For use in removal assessment analysis only)

<u>Region</u>	Sitename	EPA I.D.
1	Davis (GSR) Landfill	RID980731459*
	Ravernili Municipal Dr Revers Textile Prints	CTD004533610
	Shnack Landfill	MAD980503973
	Subsex Danartit	1210500303313
2	Anchor Chemicals	NYD001485226
	Griffiss Air Force Base	NY4571924451
	Jones Sanitation	NYD980534556
	Naval Air Engineering	NJ7170023744
	Rocket Fuel Site	NYD980535124
	Vestal Water Supply 4-2	NYD980652267
3	Follandsbee Site	WVD004336749
	Metal Banks	PAD046557096
	Stanley Kessler	PAD014269971
٨	Konners Co Inc	SCD003353026
-	Leonard Chem Co.	SCD991279324
	Olin Corp/McIntosh Plant	ALD008188708
	Palmetto Recycling Inc.	SCD037398120
5	Boise Cascade/Onan/Medtro	MND053417515
5	Delevan Muni Well #4	WTD980820062
	East Bethel Demolition	MND981088180
	Freeway Sanitary Landfill	MND038384004
	Joliet Army Ammo Plant Mfg	IL7213820460
	Joslyn Mfg and Supply	MND044799856
	Kent City Mobile Home	MID981089915
	Koppers Co/Galesburg	TLD990817991
	Koppers Coke	MND000819359
	Lauer I San Landfill	WID058735994
	McGraw-Edison Company	MID005339676
	Nutting Truck & Caster Co.	MND006154017
	Olmsted County San Ldfl	MND000874354
	Perham Arsenic	MND980609572
	SCA Independent Landfill	MID000724930
	Scrap Processing Inc.	WID046536785
	Southwest Ottawa County	MID980608780
	Sparta Landfill	MID000268136
	Spartan Chem Co.	MID079300125
	St. Augusta Ldfl Engen.	MND981002256
	St. Regis Paper Co.	MND057597940
	Tomah Armory	WID980610299*
	Tomah Fairground Area	WID980616841*
	Waite Park Wells	MND981002249

•

Region	<u>Sitename</u>	EPA I.D.
5	Waste Mgmt. of Michigan Waste Research and Reclam. Windom Muni Dump	MID060179587 WID990829475 MND980034516
6	None	
7	Obee Road Site	KSD980631766
8	Mouat Industries	MTD021997689*
9	Mather AFB (AC & W Disp.) McClellan AFB (36 Areas)	CA8570024143 CA4570024337
10	Bangor Ordinance Disposal Kaiser Aluminum Mead Work McChord AFB Pesticide Lab - Yakima Umatilla Army Depot	WA7170027265 WAD000065508 WA8570024200 WAD120513957 OR6213820917

.

.

Attachment B

Unaddressed Pre-Update 7 Sites

<u>Final Sites</u>

Rl	Nutmeg Valley Road	CTD980669261*
R2	W.R. Grace/Wayne Interim Storage	NJ1891837980
R3	USA Letterkenny Southeast	PA6213820503
R5	Douglas Road Uniroyal Inc.	IND980607881
	Opene Hills Newth Tandfill	
	Omega Hills North Landlill	WID000808568
	Savanna Army Depor	113210020803
	Southistue Santtary Landilli	WTD980610307+
	Toman Municipal Sanitary Lantill	WID98001030/-
	TRW INC. MINErva Plant	000004175339
R6	Lone Star Army Ammunition	TX72138218319
	Castle Air Force Base	CA3570024551
	Moffett Naval Air Station	CA2170090078
	Norton Air Force Base (LF #2)	CA4570024245
Pro	posed Sites	
R2	Naval Weapons Sta. Earle-Site A	NJ0170022172
R3	Buckingham County Landfill	VAD089027973
	Culpepper Wood Preservers	VAD059165282
	Dover Gas Light Co.	DED980693550*
	IBM Corp. (Manassas Plant)	VAD064872575
	Mobay, New Martinsville	WVD056866312
	Pigeon Point Landfill	DED980494603
	River Road Landfill	PAD000439083
	Rohm and Haas Landfill	PAD091637975
	Transicoil Inc.	PAD057152365
	Tyler Refrigeration Pit	DED980705545
	USA Aberdeen - Edgewood	MD2210020036*
	USA Aberdeen, Michaelsvill	MD3210021355*
	USN Naval Air Development Center	PA6170C2+545*
34	Diamond Shamrook Corn	CAD9907/1092*
R4	Olin Chamical Crown	CAD040690737
	Dinn Chemical Group	SAD040050737
	Piper Aircrait Corp.	FLD004054284*
	Roumester Property	200200840028=
RS	Barrels Inc.	MID017188673
	Firestone Industrial Products	IND980605877
	H O D Landfill	ILD980605836
	Hooker Montague Plant	MID006014906
	Kerr-McGee Kress Creek	ILD980823991

Unaddressed Pre-Update 7 Sites

Proposed Sites

R5	Kysor Industrial Corp.	MID043681840
	Lacks Industries Inc.	MID006014666
	Stauffer Chem Co. Chicago	ILD005110143
R6	Air Force Plant #4	TX7572024605*
R7	Frit Industries	IAD041103193
	US Nameplate	IAD054758958*
R8	Midvale Slag	UTD081834277*
R9	Kunia Wells I	HID980894943
	Kunia Wells II	HID980895049
	Marley Cooling Tower Co.	CAD009140120
	Mililani Wells	HID980895031
	Waiwa Shaft	HT7170090016
	Wainahu Wells	HTD980895023
	Wainin Wainte Walls IT	WTD980895015
	wathin vetaves wells II	1119300932012
R10	Naval Air Station Ault Fld	WA5170090059
	Naval Air Station Seaplane Base	WA6170090058
	Naval Undersea Warfare	WA1170023419

Old Inland Pit

•

WAD980982557*

Unaddressed NPL Sites (Updates 7 and 8)

Final Sites

- R1 Bennington Landfill Burgess Brothers Landfill Holton Circle
- R2 Action Anodizing & Plating Global Landfill Sydney Landfill
- R3Berkley Prod Co DumpPAD980538649Boarhead FarmsPAD047726161*Bush Valley LandfillMDD980504195*Elizabethtown LandfillPAD980539712
- ABC One Hour CleanersNCD024644494*Cedartown Municipal LandfillGAD980495402*Elmore Waste DisposalSCD980839542New Hanover County AirportNCD981021157Red-Penn Sanitation Co.KYD981469794T.H. Agriculture and NutritionGAD042101261Wilson Concepts of FloridaFLD041184383 R4 ABC One Hour Cleaners
- Carter Lee Lumber Co.IND016395899Continental Steel Corp.IND001213503Hechimovich LandfillWID052906088Interstate Pollution ControlILT180011975Parsons Chemical WorksMID980476907*SE Rockford GroundwaterILD981000417Yeoman Creek LandfillILD980500102 R5 Carter Lee Lumber Co.
- R6 Cal West Metals (SBA) Cleveland Mill Fourth Street Abandoned Refinery OKD980696470* Gulf Coast Vacuum Service LAD980750137* PAB Oil and Chemical Service LAD980749139*
- Hydro-Flex Inc.KSD007135429Mid-America Tanning Co.IAD085824688*Pester RefineryKSD000829846* R7 Hydro-Flex Inc. Pester Refinery
- R9 Old Minot Landfill
- R9 Modesto Groundwater Contamination CAD981997752 Newmark Wellfield CAD981434517 Valley Wood Preserving
- R10 Fairchild Air Force Base Joseph Forest Products Yakima Flating Co.

VTD981064223* VTD003965415* NHD981063860 NYD072366453 NJD063160667 NYD980507677

NMD097960272* NMD981155930* LAD980749139*-

NDD980959548*

CAD063020143

WA9571924647 ORD068782820* WADC40187890*

Proposed Sites

- R1 Atlas Tack Co. Atlas Tack Co. Barkhamsted-New Hartford BFI/Rockingham Cheshire Associates Property Darling Will Dump Darling Hill Dump Durham Meadows Gallup's Quarry Parker Landfill Precision PlatingCTD051316313Rose Hill Regional LandfillRID980521025Saco Municipal LandfillMED980504393*_Tansitor Electronics Inc.VTD000509174 Tansitor Electronics Inc.
- R2Higgins Disposal Service Inc.NJD053102232Jones Chemical Inc.NYD000813428Naval Security GroupPR4170027383Niagara Mohawk/Operations HQNYD980664361Tri-City BarrelNYD980509285Witco Chemical Corp.NJD045653854
- R3 AIW Frank Anne Arundel Co. Landfill Bell Landfill Berks Landfill Jack's Creek/Sitkin Smelt Kent City Landfill Occidental Chem/Firestone Occidental chem/filesconePAD002353969Recticon/Allied SteelPAD002353969Saegertown Industrial AreaPAD980692487Suffolk City LandfillVAD980917983Sussex Co. Landfill #5DED980494637
- R4 Agrico Chemical Site Anodyne, Inc. Beaunit Corp./Circular KnitFLD981014368Benfield Industries, Inc.SCD000447268Beulah LandfillNCD981026479 Leather Co. Leather Co. KYD980501019 KYD980501019 KYD045738291 TND044062222 Cedartown Industries Inc. Chem-Form Inc. Firestone Tire and Rubber Firestone Tire and Rubber GaD990855074 Fort Hartford Coal Stone Ganral Tire and Rubber Company KYD980844625* Genral Tire and Rubber Company KYD006371074 Green River Disposal Land Hevi-Duty Electric Co. NCD0205 Beulah Landfill

MAD001026319* CTD980732333* VTD980520092 CTD981067317 VTD980520118 CTD001452093 CTD108960972* VTD981062441 PAD004351003 MDD980705057* PAD980705107* PAD000651810* PAD980829493 DED980705727 PAD980229298

FLD980221857 FLD981014368 FLD980494660 Proposed Sites

- R4 JFD Electronics/Channel Master Lexington County Landfill Madison County Sanitary Landfill Murray Ohio Manufacturing Redwing Carriers Inc. T. H. Agriculture Nutrition Co. Townsend Saw Chain Co. USMC Camp Lejuene Wingate Road Municipal Incineration FLD981021470 Woodbury Chemical Co. Woolfolk Chemical Works
- R5 Adams County Quincy Landfill Albion Sheridan Township Landfill Allied Paper/Portage Creek Amoco Chemical Corp. Beloit Corp. Central Illinois Public Service Co. Dupage County Landfill Fort Howard Paper Co. Hinco Dump Lenz Oil Service Inc. Madison Metro L ILD005451711Muskegon Chem Co.MID078934403Sauk County LandfillWID072569510Sauk County LandfillWID980610141State Disposal LandfillMID980609341Tippecance Sanitary LandfillIND980997639USAF Wright Patterson BaseOH7571724312Warner Electric Brake and ClutchILD006114151Waste Mgmt. of Wis-BrookfieldWID980901225Whiteford Sales and Serv NetWebsel Woodstock Municipal Landfill
- R6 D. L. Mud Inc. Lee Acres Landfill (USDOI) Mosley Road Sanitary Landfill Pagano Salvage Prewitt Abandoned Refinery Rio Grande Oil Co. Refinery Sunray Oil Co. Refinery Tex-Tin Corp.
- E. I. Dupont De NemoursIAD980685804Electro-Coatings Inc.IAD005279039John Deere (Ottumwa Works)IAD005291182Lehigh Portland Cement Co.IAD005288634Northwestern States Portland CementIAD980852461 R7 Lehigh Portland Cement Co.

NCD122263825 SCD980558043 FLD981019235 TND981014954 ALD980844385* ALD007454085* SCD980558050 NC6170022580 FLD004146346* GAD003269578* ILD980607055 MID980504450 MID006007306 ILD002994259 ILD021440375 ILD981781065 ILD980606305 WID006136659 IND980500292 MID006016703 ILD005451711 ILD980605943 LAD981058019*

NMD980750020* 0KD980620868* NMD980749980* NMD980622773* TXD980795736* OKD000764357* TXD062113329*

Proposed Sites

R7	Oronogo-Duenwig Mining	MOD980686281
	Peoples Matural Gas Co.	IAD980832378*
	Sheller-Globe Corp. Disposal	TAD980630750
	St. Louis Airport/His/Futura	MOD380633176
	29th and Mead Groundwater	KSD007241656
R8	Comet Oil Co.	MTD980403554*
R9	Advanced Micro Devices #9	CAT080034234
	Brown & Bryant Inc.	CAD052384021*
	Concord Naval Weapons Station	CA7170024528
	Crazy Horse Sanitary Landfill	CAD980498455
	CTS Printex	CAD009212838
	El Toro Marine Corps Air Station	CA6170023208
	Fresno Sanitary Landfill	CAD980636914
	GBF & Pittsburg Dumps	CAD980498562*
	Hewlett Packard I Palo Alto	CAD980884209
	Hexcel Corp.	CAD058783952
	Intersil	CAD041472341
	Kaiser Steel Corp.	CAD008274938*
	Kearney KPF	CAD981429715
	Pacific Coast Pipe Lines	CAD980636781
	Riverbank Army Ammunition	CA7210020759
	Sola Optical USA Inc.	CAD981171523
	Solvent Service Inc.	CAD059494310
	Sulphur Bank Mercury Mine	CAD980893275*
	Synertek # (Bldg 1)	CAD990832735
	TRW Microwave	CAD009159088
	Yuma Marine Corps Station	AZ0971590062
R10	Centralia Municipal Landfill	WAD980836662
	Eastern Michaud Flats Contamination	IDD984665610
	Hanford 100-Area (DOE)	WA3890090076
	Hanford 1100-Area (DOE)	WA4890090075
	Hanford 200-Area (DOE)	WA1890090078
	Hanford 300-Area (DOE)	WA2890090077
	Kerr-Mcgee Chem Corp.	IDD041310707
	Monsanto Soda Springs Plant	IDD081830994
	Northwest Trans. (S. Harkness St.)	WAD027315621*
	Pacific Car & Foundry Co.	WAD0092;9210
	Pasco Sanitary Landfill	WAD991281874

Use of Removal Approaches to Speed Up Remedial Action Projects



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

WASHINGTON, DC 20460

JUL 6 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPO

OSWER Directive No. 9355.0-25A

MEMORANDUM

SUBJECT:	Use of Removal Approaches to Speed Up Remedial Action Projects
FROM:	Jonatilian 12 Cannon in
	Acting Assistant Administrator
TO:	Environmental Services Division Directors
	Regions I, VI, VII
	Waste Management Division Directors
	Regions I, IV, V, VI, VII, VIII
	Emergency and Remedial Response Division Director
	Region II
	Hazardous Waste Management Division Directors
	Regions III, VI
	Toxics and Waste Management Division Director
	Region IX
	Hazardous Waste Division Director
	Region X

Purpose:

This memorandum defines the basic requirements to be met when a Region chooses to use removal authorities and contracting methods to speed up remedial projects.

Background:

Several Regions have expressed interest in the use of removal authorities and contracting methods to speed up remedial actions on National Priorities List (NPL) sites where surface cleanups may result in a total site cleanup or completion of a major portion of the site. Memoranda containing conditions for this approach were forwarded to Region IV on March 18, 1988, and July 11, 1988. (OSWER Directive No. 9355.0-25). Region IV has used this approach on eight sites. Three of these sites are now scheduled to be deleted from the NPL this fiscal year. Region III has also used this approach at two NPL sites. Several issues arose in defining and implementing this approach, which required coordination of policies and procedures between both the remedial an removal programs. This coordination resulted in a unique set of requirements which each Region must meet when undertaking the use of removal authorities to achieve early action at NPL sites.

Objective:

This policy is aimed at speeding up response at some NPL sites. The purpose is not to establish an alternative to remedial contracting methods but to provide an additional response option until such time as rapid remedial contracting alternatives are fully developed. This document supersedes previous memoranda on this topic. This memorandum describes the criteria and other considerations for effective utilization of this approach. The requirements outlined in this memorandum do not apply to the usual 40 to 50 emergency and time-critical removal actions conducted by the removal program at NPL sites each year.

Implementation:

The following are the key requirements which must be met before removal authorities or removal contractors can be used to perform remedial actions at NPL sites.

- All sites must have a signed Record of Decision (ROD). Should the proposed response activities entail a substantive change from the remedy specified in the ROD, the Region must either amend the ROD or publish an explanation of significant differences, whichever is appropriate, prior to commencing the cleanup.
- 2) Sufficient time and enforcement resources must be allocated to the extent feasible to conduct a potential responsible party (PRP) search and obtain information about PRP's through Section 104(e) actions. Follow-up PRP Search activities should be conducted where necessary. If PRP's are identified during this process, the Region should send a notice letter, and, consistent with CERCLA Section 122(a) and (e), issue special notice or advise the PRP that such procedures will not be utilized. If site exigencies require early response, the Region may chose to issue notice orally and follow up in writing. Assuming that there is sufficient time, the Region should conclude Section 106 negotiations with the PRP prior to initiation of any response action.
- 3) These cases draw upon removal authorities, but are considered remedial actions. Consequently, if the Region takes an enforcement action, it must utilize a consent decree or issue a Unilateral Administrative Order. If compliance is not achieved and time is critical (typical of many removals) the Region should take over the response and pursue cost recovery, seeking treble damages and/or other penalties.
- 4) All activities must be well documented for cost recovery.

- 5) A signed State Superfund Contract must be obtained from the State. prior to the start of the action, providing the Section 104(c) assurances for cost sharing, operation and maintenance, off-site disposal and, when applicable, 20-year waste capacity.
- 6) The proposed response action at the site must meet the National Contingency Plan (NCP) criteria for removal actions in section 300.65.
- 7) All proposed response activities must be described in a signed removal action memorandum. If the response should exceed the statutory limits of 12 months or \$2 million, it will be necessary for the Region to prepare an exemption request. In the-case where site costs are expected to exceed the \$2 million limit, Headquarters approval must be obtained prior to commencing the removal action. These exemption requests must be prepared as early as possible in the process. Only Regional approval is required for exemption requests where activities are expected to exceed the 12-month limit.
- 8) All funding, activity codes, account numbers, SCAP and CERCLIS data will use remedial codes. This will ensure that these activities are reported on and tracked as remedial actions. Funds will come from the remedial portion of the Region's Advice of Allowance (AOA). All projected starts should be entered into CERCLIS with their projected obligations data for the appropriate quarter of the fiscal year.
- 9) Community relations requirements must be met and an administrative record must be established for each site. The public comment period must be observed in accordance with NCP requirements for both the remedial and removal programs.
- 10. Generally at the completion of this cleanup work, the site should be ready for deletion. OSWER Directive 9320.2-3A procedures must be followed to delete the site from the NPL. When a major portion (significant operable unit) of work is undertaken using these procedures and the site does not qualify for deletion. unusual circumstances (e.g., emergency) must preclude the use of remedial contracting mechanisms.

Future Plans

The remedial program has two initiatives underway to provide the contractual mechanisms and construction management systems needed to expedite projects within the remedial process and authorities. One is the Corps of Engineers new rapid response contracts used to expedite smaller pieces of some large site cleanups they manage. The second is the subcontracting provision of the Alternative Remedial Contracts Strategy (ARCS). ARCS will more typically be used for smaller construction projects such as those anticipated for coverage under this policy. With the inclusion of experienced engineering and construction management contractors in the ARCS program, opportunities open up for the use of more expeditious construction subcontracts for some sites with plans for early initiation of construction based on limited design. Alternatively, other subcontracting vehicles such as basic ordering agreement and other methods of bidder prequalification might be used to reduce procurement lead time. More guidance on the use of these mechanisms will be issued. It is expected that as the Regions gain experience with the ARCS program and these subcontracting mechanisms, the use of removal authorities and program mechanisms to speed up remedial projects will be phased out.

Other Considerations:

If removal authorities are going to be used, careful consideration must be given to the type of contract that is selected for the work. Obtaining the best price and maximizing competition are always major goals, as in using competitive contracting mechanisms to the maximum extent practicable. The Emergency Response Cleanup Services (ERCS) contracts may be the most appropriate vehicle where rapid response is necessary under emergency and timecritical circumstances. However, a Region should always consider using the Prequalified Offerors Procurement Strategy (PQOPS) or other site-specific contracting mechanisms. This is especially true if the project is a nonemergency situation where the consistency waiver to the \$2 million limit is used.

When use of alternative technologies is specified in the ROD, PQOPS should be considered. The use of this arrangement is most appropriate when a 3 to 5 month lead time is available. This procurement strategy is about to be implemented for mobile incineration. PQOPS for other technologies will follow. When the lead time is approximately 4 months and PQOPS is unavailable, use of site-specific subcontracts under the ERCS contracts may be feasible. This approach may be used only where the prime contractor has not proposed rates for the site's particular cleanup activity. In addition, many Regions have Regional ERCS contracts with 24 to 72 hour response times which may be more cost-effective than the Zone ERCS contracts. In all cases, the Region should keep in mind that achieving the maximum competition is a primary goal of both the removal and remedial programs, taking into account the need for rapid response and the magnitude of the risks posed.

A final factor to consider is removal contracts capacity. Generally, this policy should not be used to do expensive remedial work. It is intended to help expedite deletion from the NPL of projects of modest scope. This approach cannot be used where the adequacy of removal contracts capacity is jeopardized. Being able to always promptly and fully respond to the normal removal workload is a higher priority than doing the remedial work that is the subject of this policy.

In summary, use of removal authorities or removal (e.g., ERCS) contracts to take early action at NPL sites is an alternative in certain limited situations. The site must meet the criteria for a removal action as well as fulfill all the regular remedial requirements. This strategy will enable the Regions to complete cleanup at certain NPL sites in a more expeditious and efficient way and to start the necessary deletion process. cc: Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X Henry Longest, OERR Bruce Diamond, OWPE Timothy Fields, ERD Russel Wyer, HSCD Clem Rastatter, OPM Sally Mansbach, OWPE Frank Russo, OWPE Linda Boornazian, OWPE Earl Salo, OGC Kirsten Engel, OGC -Karen Clark, OGC Arthur Weissman, OPM Betti Van Epps, PAS Bruce Engelbert, ERD John Riley, ERD Mark Mjoness, ERD Linda Garczynski, ERD Dave O'Connor, PCMD Sallyanne Harper, PCMD Pat Patterson, PCMD

•

Accelerated Response at NPL Sites Guidance



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C 20460

DEC | 5 1989 OSWER Directive No. 9200.2-02

MEMORANDUM

CF+ CE C+ SOLID 4-ST(AND EVERDENCY RES-DNSE

SUBJECT: Accelerated Response at NPL Sites Guidance (Superfund Management Review: Recommendation No. 22) mun i jen . Don R. Clay FROM: Assistant Administrator TO: Director, Waste Management Division Regions I, IV, V, VII, VIII Director, Emergency and Remedial Response Division Region II Director, Hazardous Waste Management Division Regions IIi, VI Director, Toxic and Hazardous Waste Management Division Region IX Director, Hazardous Waste Division **Pegion X** Director, Environmental Services Division Regions I, VI, VII Regional Counsels, Regions I-X

PURPOSE

The purpose of this memorandum is to transmit Agency guidance on accelerating responses at National Priorities List (NPL) sites.

BACKGROUND

Pursuant to the Superfund Management Review, a workgroup was formed to develop guidance to assist the Regions in taking expedited approaches to site cleanups and in making NPL sites "safer." After evaluation of Regional comments, the guidance was split into two separate documents. The attached guidance, the first of the two documents, describes available procedures and contract mechanisms to allow the Regions to take action at NPL sites more quickly under both removal and remedial authority.

IMPLEMENTATION

Specifically, the attached guidance requires that you:

 Ensure that all pre-remedial, removal, remedial, and enforcement staff are familiar with the need to accelerate responses at NPL sites;

- o Use Superfund removal and remedial authority, as appropriate, to take accelerated actions at those NPL sites where feasible and prudent;
- Employ enforcement authority promptly at NPL sites to encourage increased PRP involvement in site cleanup;
- o Establish mechanisms to ensure proper coordination and funding of accelerated responses within the Regions; and
- Promote the operation of Superfund as "one program" through use of elements such as improved interoffice communication and cross-training of Agency personnel.

If you have any questions on this guidance, please contact Hans Crump-Wiesner, Acting Director, Emergency Response Division, at FTS 475-8720, or Scott Maid at FTS 382-4671.

Attachment

cc: Henry Longest II, OERR Lloyd Guerci, OWPE Lisa Friedman, OGC Hans Crump-Wiesner, ERD Clem Rastatter, OPM Larry Reed, HSED Russ Wyer, HSCD Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X OSWER Directive #9200.2-02

Accelerated Response at National Priorities List Sites¹

1.0 INTRODUCTION

1.1 Background

In June 1989, the Environmental Protection Agency (EPA) completed a study entitled "A Management Review of the Superfund Program" (Superfund Management Review). This document outlined a new long-term strategy for management of Superfund and described the need for EPA guidance on expediting response at National Priorities List (NPL) sites. The report also emphasized elimination of immediate risk to public health and safety and the minimization of long-term risk from hazardous substances at NPL sites as new measures of program success.

The Superfund Management Review specifically recommended that EPA "take expedited approaches to site cleanup whenever possible" (p. 3-13). The report also emphasized the need for creative or alternative approaches for improving the effectiveness and timeliness of remediation at NPL sites. Recommendations from the Superfund Management Review were further discussed in a September 18, 1989, memorandum from F. Henry Habicht to the Regional Administrators, entitled "Immediate Actions to Implement the Superfund Management Review."

1.2 Purpose

This document focuses on accelerating responses at NPL sites and coordinating available removal, remedial, and enforcement procedures and contract mechanisms in order to accomplish this. This guidance is intended for Regional site managers, including On-Scene Coordinators (OSCs), Site Assessment Managers (SAMs), and Remedial Project Managers (RPMs), enforcement staff, and other Regional and Headquarters Superfund personnel. By implementing these procedures, we may accelerate all types of response actions, and encourage management of NPL sites under "one program."

1.3 Scope

Specifically, this guidance addresses the following areas:

o What is an accelerated response? (Section 2.1)

¹ The policies and procedures established in this document are intended solely for the guidance of EPA personnel. They are not intended, and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. EPA reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

- What are the available mechanisms to accelerate responses at NPL sites? (Section 2.2)
- What are the enforcement aspects of accelerated response? (Section 3.0)
- o What additional factors should be considered in an accelerated response? (Section 4.0)
- o How may the various Superfund program offices wor: as "one program" to accelerate responses? (Section 5.0)

2.0 ACCELERATED RESPONSE

Accelerated responses may be used in many situations where site managers want to act on sites quickly. Site managers have access to a variety of mechanisms for accelerating responses to threats at NPL sites. In most cases, the tools are modifications of established response options that have been in common use in the Superfund program. Regions should follow the provisions, described below, whenever practicable to expedite cleanups at NPL sites.

2.1 What Is An Accelerated Response?

An accelerated response is an action taken at an NPL site using streamlined response mechanisms, with the purpose of acting quickly to reduce acute risk to human health and the environment. Accelerated responses can help Regions reduce risk from these sites, and can allow for more efficient use of EPA resources.

If evaluation of a site indicates that an accelerated response is warranted, then appropriate action should be taken, by:

- Conducting a removal action in accordance with section 300.65 of the National Oil and Hazardous Substances Pollution Contingency Plan (NCP)² (proposed NCP section 300.415); or
- Preparing and executing an early action operable unit Record of Decision (ROD) based on existing data or a limited data gathering effort; or

² For ease of use, references to both the old (1985) NCP and the proposed (1988) NCP sections are provided in the text. It is important to note, however, that the 1985 NCP remains in full effect until a revised NCP is promulgated. The revised NCP, which was proposed on December 21, 1988 (53 <u>FR</u> 51394) is expected to be finalized in 1990, at which point the revised section numbers will become effective.

OSWER Directive #9200.2-02 - 3 -

o Implementing techniques to expedite the planning and design process prior to remedial construction.

2.2 What Are The Available Mechanisms?

Removal Actions

Removal actions are used to prevent, abate, minimize, stabilize, or mitigate releases or threats of releases of hazardous substances, pollutants, or contaminants that pose a threat to public health or the environment. Section 300.65 of the NCP (proposed NCP section 300.415) describes factors for determining that a removal action is appropriate (<u>e.g.</u>, contamination of drinking water, threat of fire or explosion, potential for migration) and the types of removal actions that are appropriate in certain situations. Removal actions are performed at NPL and non-NPL sites. Approximately 40 removal actions have been conducted annually at NPL sites.

A site manager may, in certain situations, choose to use removal authorities and contracting methods to accelerate response at NPL sites. Actions with a planning period of less than six months are generally (but not always) performed prior to the development of the ROD. If a removal action is indicated at an NPL site, and adequate planning time (<u>i.e.</u>, greater than six months) is available before the start of the removal, an engineering evaluation/cost analysis (EE/CA) should be conducted as part of the non-timecritical removal. Alternatively, a remedial investigation/feasibility study (RI/FS) can be conducted. EE/CAs contain evaluations of possible alternative technologies, selection of the response, and document the decisionmaking process. The EE/CA must be made available for public comment as part of the administrative record, in compliance with the public participation procedures for non-time-critical removal actions described in §300.820 of the proposed NCP.

For any category of removal action, the appropriateness of the action is not limited to the factors explicitly described in section 300.65 of the NCP, nor does the NCP limit the responses EPA may take to the examples given in the NCP. Action Memoranda must be completed for all removal actions and must include all information described in the "Superfund Removal Procedures Manual," Chapter III-C.8. Expedited enforcement activities, such as expedited PRP searches and use of model administrative orders, are appropriate for these actions.

Continuation Of Removal Action. As an acute threat at an NPL site is being addressed by a removal action, it may be possible that an incremental expansion in the scope of the removal action would help to further protect human health and the environment and lead to expediting overall cleanup of the site. Regions may determine on a case-by-case basis whether this is appropriate at a site. The following factors should be considered in such case-by-case evaluations: OSWER Directive #9200.2-02 - 4 -

Scope of Continued Action

 After an acute threat at an NPL site is addressed by a removal action, Regions should consider whether there is any appropriate action that would allow further protection of human health and the environment. This may require employing a removal or remedial action to complete an operable unit. The aim of the accelerated response is to reduce risk to human health and the environment at the site.

Concurrence

- o The decision to accelerate response through use of removal authority must be made in consultation with pre-remedial, remedial, removal, and enforcement program managers. The State should also be involved whenever the State is expected to play a role in the action. If the action does not meet removal criteria, however, the accelerated response option chosen must be performed as a remedial action.
- o The Region must weigh the loss of cost-share against the need for, and efficiency of, accelerating the response at a site on a case-by-case basis. Regions should attempt to obtain agreement from States on the proposed course of action before proceeding with any accelerated response option.
- o If the cleanup will exceed the \$2 million statutory limit for a removal action, then Headquarters approval of an emergency or consistency exemption is necessary. If no exemption applies, the accelerated response option chosen must be performed as a remedial action.
- o If the removal response will be nationally significant (<u>e.g.</u>, involving dioxins or Indian lands), Headquarters concurrence will be necessary. Headquarters concurrence will also be necessary if the action employs innovative or emerging alternative technology.
- o The public's interest and concern in the site should be taken into account when deciding what the response should be.

Restrictions

- o The State, remedial program, PRP, or other authority must be willing to conduct post-removal site control (PRSC) where needed following a Fund-financed removal action (see section 4.4). The Region may pursue unilateral enforcement action, including judicial action if needed, to obtain PRSC. If arrangements for PRSC cannot be made, the accelerated response must be performed as a remedial action.
- o If the action will require extensive, long-term response, such as restoration of a contaminated aquifer, the response should be performed as a remedial action.

OSWER Directive #9200.2-02 - 5 -

Contract Resources

- o Regional resources (including resources that may be transferred from the remedial program) must be adequate to meet the requirements for an accelerated removal response at the site without compromising emergency response capability in the Region. If resources for accelerated responses at NPL sites cannot be provided without compromising emergency and time-critical response capabilities in the Region, the accelerated response must be performed and funded as a remedial action.
- An assessment of removal and remedial contract capacity should be performed in order to determine the capacity, availability, and suitability of response contractors to the site in question. Regions must evaluate relative contract capacity before an accelerated response can be continued at the site.

Removal Approaches To Remedial Actions. Remedial action may be conducted using removal contracting methods where the action complies with all removal as well as remedial requirements. In these circumstances, remedial funding is used to implement a ROD at an accelerated rate. Guidance on this response option was issued to the Regions on July 6, 1989 (see "Use of Removal Approaches to Speed Up Remedial Action Projects," OSWER Directive #9355.0-25A). These actions are remedial actions. The term "removial," which has been used informally to characterize these actions, should not be used.

The response must meet both remedial and removal program requirements. Time is saved by using removal contractors, and through the use of an abbreviated and less formal design procedure. Remedial funding is provided through a ROD and a state cost-share is provided through a Superfund state contract. For all purposes, including enforcement, these actions are remedial actions. All agreements with PRPs must be embodied in a consent decree. Since this response approach uses large amounts of limited Emergency Response Cleanup Services (ERCS) capacity, it should only be used in unusual emergency or timecritical circumstances. New and streamlined remedial alternatives should obviate the need for this course of action in most cases.

Remedial Actions

The purpose of the remedial action process is to implement remedies that reduce, control, or eliminate risks to human health and the environment. Only those sites included on the NPL are eligible for Fund-financed remedial action (NCP section 300.68). The remedial process generally includes an RI/FS, a proposed plan, a ROD, engineering design, and implementation of the remedial action. All remedial actions must comply with the requirements of §300.68 of the NCP (§§300.430 and 300.435 of the proposed NCP).

A site manager may, in certain situations, choose to use remedial authorities and contracting methods to accelerate response at an NPL site. This may be accomplished through the implementation of an early action operable OSWER Directive #9200.2-02 - 6 -

unit ROD (for example, to remove drums) and the use of streamlined remedial contracts. Early actions may make it possible to provide earlier protection of public health and the environment, and the actions may also help provide information that may be used to improve the phasing and design of later remedial stages.

Site managers can break actions into distinct portions, which are known as "operable units," to achieve quicker response. An operable unit is "a discrete part of the entire response action that decreases a release, threat of release, or pathway of exposure" (NCP section 300.6; see also proposed NCP Subpart A). Operable units can be designated to accelerate remediation for portions of the site, but all operable units conducted as remedial actions must have RODs. Separate enforcement agreements may be reached for individual operable units.

Expediting Remedial Actions. After signing a ROD, accelerated responses may be implemented under remedial authority. The most obvious method to accelerate remedial action is to initiate construction sooner, <u>i.e.</u>, speed up the planning and design process. Once this is achieved, contracting and construction options can be explored to best enhance site remediation. This section briefly describes techniques for expediting remedial construction. (The techniques are covered in greater detail in OSWER Directive #9355.5-02/FS, "Expediting Remedial Construction.") These techniques are applicable to all Superfund projects; however, they are geared toward small (less than \$5 million), well defined projects using proven technologies.

Remedial Management Strategy. The remedial management strategy (RMS) is a systematic approach used to identify and establish the preferred contracting strategies to be used in the implementation of a remedial action. The objective is to look at each of the operable units that are part of the remedy described in the ROD and lay out a strategy for construction that meets all of the constraints imposed on the project. The RMS establishes the overall course of action for the project. It is at this point that decisions are made about phasing portions of the project, fast-tracking design and construction, employing limited designs for specific elements, or utilizing alternative procurement methods.

Phasing Remedial Design and Construction. An analysis of remedial design/remedial action (RD/RA) project elements results in the determination that some can be effectively phased or time-sequenced to accelerate them through the design and remediation process. Phasing may achieve an overall fast-track schedule and thereby mitigate the continuing threat of the site to the environment and public safety. Large, complex projects (or operable units) may be broken down into smaller, more manageable response elements. Elements may be worked in unison, but each individual element has its own schedule and moves at its own rate through the remediation process.

Fast-Tracking RD/RA. Fast-tracking might be considered a subset of phasing. Where phasing breaks large complex projects into smaller more manageable units, fast-tracking is a method to accelerate the implementation of those individual elements. There are several techniques that can be used to fast-track RD/RA:

OSWER Directive #9200.2-02 - 7 -

- o Expedite RD Discretionary steps in the RD process may be eliminated or shortened. Site managers must realize, however, that shortcutting the process involves some risk. For example, deciding to use only data collected during the RI/FS for design is one method of expediting. However, the design risks being delayed if the RI/FS data turns out to be marginal or incomplete.
- o Use of Removal Authority. As mentioned in the section on Removal Approaches to Remedial Actions, removal contracting methods and remedial funding can be used to implement RD/RA on an accelerated basis.
- o Optimize RD Optimization is the rearrangement of the sequence in which RD elements are performed to enhance the overall schedule. For example, the site access portion of a design could be completed and construction initiated while the rest of the design is still ongoing.
- o Fast-Track Construction. Many large projects can be divided into separate stages of construction. This is generally accomplished by awarding each stage of work for construction as soon as the design effort on that particular stage of work has been completed. This approach has the advantage that the project will be started and completed sooner than would be possible if it were necessary to wait until all design work had been completed. Another aspect of fasttrack construction is ordering items that require long lead-times in advance of the time they will be needed on the job.

Preplaced and Pre-Qualified Contracts. One method to expedite initiation of remedial construction is to use preplaced contracts or pre-qualified contractors. There are several options currently available for use. These methods require approximately 30-60 days to initiate construction activities by eliminating the solicitation and audit requirements of site-specific contracts, thus reducing the time from design completion to construction initiation.

The U.S. Army Corps of Engineers (USACE) has developed methods to expedite the initiation of remedial action at Superfund sites by implementing two innovative contracting strategies: Preplaced Remedial Action (PRA) and Rapid Response (RR) contracts. Both may be used for projects when delaying the remedial action for normal procurement actions may result in detrimental effects on human health or the environment. PRA contracts are structured to implement full-scale remedial actions. RR contracts are for demolition actions, closures, point source contamination control, and site stabilization. They are limited to \$2 million per delivery order and may be used for projects where it is necessary to abate, stabilize, mitigate, or eliminate hazardous or contaminated materials or structures.

The Pre-qualified Offerors Procurement Strategy (PQOPS), when completely in place, will provide a list of prequalified contractors that have the capability of performing a specified technology (<u>i.e.</u>, incineration, fixation). All contractors on the list will have been technically evaluated and deemed OSWER Directive #9200.2-02 - 8 -

qualified to perform the specified work. However, they are limited to providing the equipment for a specific technology and do not include broad response support (e.g., site access, excavation, site closure) to fully implement the remedy. The transportable incineration system (TIS) PQOP is in place and the fixation/solidification system (FSS) PQOP will be in place during FY 90.

3.0 WHAT ARE THE ENFORCEMENT ASPECTS OF ACCELERATED RESPONSE?

The Superfund Management Review placed great emphasis on the prompt use of enforcement authority at NPL sites. At sites where there is accelerated response, enforcement and program staff must anticipate each other's needs. Good communications are essential. For example, those evaluating a NPL site, who discover the probable need for accelerated action, need to contact enforcement personnel promptly so that this change can be incorporated into the enforcement strategy for the site. Conversely, enforcement staff must appreciate how delays in performing enforcement activities may affect timing of site response.

Site managers must take advantage of enforcement authorities whenever possible. The enforcement authorities that are available to EPA include strong liability provisions, administrative order authority, judicial enforcement authority, and the authority and funding to take direct action to clean up sites and subsequently recover costs. When developing an accelerated response action, the following enforcement activities should be taken into account.

Enforcement Strategy

Enforcement personnel should take a site-specific approach when developing enforcement strategy. The approach should generally cover the iters discussed here (<u>e.g.</u>, PRP search, notice to PRPs and States). If enforcement authority is not used, site managers must document why.

PRP Search

If the site is on the NPL, an expedited PRP search can be conducted by focusing on owners and operators that are known and generators that are readily identifiable. PRP searches are discussed in detail in the "Enforcement Project Management Handbook," OSWER Directive #9837.2 (July 1989); see also the "PRP Search Manual," OSWER Directive #9834.3-1A and the "PRP Search Supplemental Guidance for Sites in the Remedial Program," OSWER Directive #9834.3-2A.

Notice to PRPs

Where possible, it is usually advantageous to notify PRPs of their potential liability before transmitting to the PRPs a draft administrative order on consent. Moreover, except for emergencies, PRPs should be notified prior to issuance of a unilateral administrative order. If PRPs have not been notified, a notice letter should be issued. For additional information on enforcement activities, see Section 6.0, Bibliography, for a listing of applicable OSWER Directives. OSWER Directive #9200.2-02 - 9 -

Notification of the State

Prior to issuing an administrative order, EPA must notify the State. In situations where there is little time available before initiation of site activity, the State may be notified by telephone, followed by written confirmation.

Administrative Order on Consent (AOC)

If the response is an accelerated removal action and PRPs are willing to perform the action, the PRPs' conduct of an accelerated response should be pursuant to an AOC (CERCLA §106). If the accelerated response follows a ROD and is a remedial action, PRP conduct of the action should be pursuant to a consent decree (CERCLA §122). Moreover, settlements that include owners must include an agreement for access to the site. If, during negotiations, site conditions dictate the need for immediate response, the site manager should discontinue negotiations and initiate on-site response. Whenever appropriate, a unilateral order should be issued to allow EPA to seek treble damages and/or possibly convince the PRPs to take over the response effort.

Unilateral Administrative Order (AOU)

Generally, when negotiations become protracted or in critical situations (including some emergencies where time allows), EPA policy is to proceed with a CERCLA §106(a) AOU to viable PRPs before Fund activation. There are exceptions, such as: sites where there is an immediate need to respond; where PRP liability is very uncertain; where there are unique technical problems; or where there are problems with the technical capability of the PRP to conduct the removal action.

4.0 WHAT ADDITIONAL FACTORS SHOULD BE CONSIDERED?

4.1 ARARS

The Superfund Amendments and Reauthorization Act of 1986 (SARA) required that on-site remedial actions comply with applicable or relevant and appropriate requirements (ARARs) of other federal and state environmental laws. Although CERCLA only requires compliance with ARARs for remedial actions, the current NCP requires removals to comply with Federal ARARs to the extent practicable. EPA policy under the proposed NCP (§300.415) requires removal actions to comply with State and Federal ARARs to the extent practicable. Until this policy is promulgated by regulation, compliance with State ARARs during removal actions must be justified based upon protectiveness. Factors used in determining whether removal compliance with ARARs is practicable include: (1) the urgency of the situation, and (2) the scope of the removal action to be conducted, which includes consideration of the statutory limits for removals. Off-site actions must always comply with applicable requirements. (For a statement of EPA's off-site policy, see 50 FR 45933, November 5, 1985, as revised November 13, 1986 in OSWER Directive #9834.11.) Remedial actions, including those discussed in the section on p. 5, Removal Approaches

OSWER Directive #9200.2-02 - 10 -

to Remedial Actions, must comply with all ARARs identified in the ROD, unless an ARAR is waived.

Waivers of ARARS (CERCLA section 121(d)(4)) also may be used for removal as well as remedial actions where they apply. See the document "CERCLA Compliance With Other Laws Manual" (OSWER Directive #9234.1-01) for additional information.

4.2 Public Participation

Informed public involvement in the decision-making process is a key element in the Superfund program. The Superfund Management Review identified that the public wants greater and earlier involvement in the process. As a steward of the environment, EPA must be fully responsive to the concerns of the public if it wishes to retain the public's confidence.

Before a ROD can be signed for an early remedial action, a proposed plan must be circulated, and a 30-day public comment period must be held. An opportunity for public hearing must also be provided. The current NCP provides for a 21-day comment period. However, the proposed NCP provides for a minimum of 30 days for public comment. Adequate information on the proposed action and a limited number of alternatives must also be available to the public along with the proposed plan. This information may, however, be presented in any type of document, including but not limited to an RI/FS or a focused feasibility study.

Public participation requirements for removal actions are set forth in the proposed NCP sections 300.415 and 300.820, and the "Superfund Removal Procedures Manual," Chapter III-F.6. Remedial action requirements are set forth in the proposed NCP sections 300.430 and 300.435, and the Community Relations Handbook.

4.3 <u>Alternative Technologies</u>

As noted in the Superfund Management Review, EPA should continue to encourage the employment of alternative technologies to treat hazardous substances, pollutants, or contaminants at Superfund sites. It is important that technologies selected for removal actions at NPL sites be consistent with planned remedial work and contribute to permanent remedies. OSWER Directive #9355.0-26 (February 1989) reaffirms the use of treatment technologies at Superfund sites and summarizes guidance documents and activities that encourage and support the use of innovative treatment technologies.

Also, OSWER Directive #9380.3-01 (July 12, 1989) describes a treatability data base which is being developed by the Office of Research and Development (ORD) to aid in expediting technology selection on a site-specific basis for the removal and remedial programs.

4.4 Post-Removal Site Control (PRSC)

Provisions for PRSC must be made before removal action initiation. PRSC may be a removal or remedial response under the statute. For remedial actions

.

OSWER Directive #9200.2-02 - 11 -

a state contract or Superfund cooperative agreement must be in place prior to remedial action initiation in order to assure any State operation and maintenance responsibilities. Information and guidelines on PRSC may be found in the proposed NCP section 300.415 and the "Superfund Removal Procedures Manual," Chapter III-H.2.

4.5 Documentation

The various offices responsible for NPL sites in the Regions should work together to ensure that documentation for sites is adequate to support decision-making and, if appropriate, cost recovery. This is very important at every Superfund site, but it will be especially important if a site is selected for accelerated response. The response action must be sufficiently documented in order to fully justify the rationale for the Region's actions; that is, to explain why a certain activity at an NPL site is being conducted on an accelerated basis and to specify the authority under which the response is being conducted. See NCP section 300.69, and "Interim Guidance on Administrative Records for Selection of CERCLA Response Actions," OSWER Directive #9833.3A.

It is EPA's policy to develop decision documents for responses at sites in order to support the decision and remedy selection and to completely document costs to support cost recovery. Documentation of cleanups must also show that human health and the environment have been protected along all possible pathways of exposure. If a removal response cannot provide sufficient documentation to support the eventual deletion of the site from the NPL, then the site may be completed as a remedial action. Every removal action must demonstrate how it will contribute to any long-term remedial action to be taken at the given site.

5.0 HOW MAY SUPERFUND WORK AS "ONE PROGRAM?"

5.1 Promoting Communication

In order to foster the development of Superfund as "one program," EPA must encourage an increased level of cooperation among the various program offices that administer and support Superfund. The Superfund Management Review states that many of the difficulties Superfund has encountered in the past may be traced to the lack of proper communication between programs. We must institute procedures to improve coordination of site activities among the different Superfund program offices, <u>i.e.</u>, pre-remedial, remedial, removal, and enforcement, to provide the internal support necessary for implementing Superfund as "one program." It is also important to ensure that EPA coordination with appropriate authorities located outside the Superfund program (<u>e.g.</u>, Agency for Toxic Substances and Disease Registry [ATSDR]) takes place in a consistent manner.

Information on sites should be shared freely among programs, and changes in site status likewise should be communicated to all affected offices. For example, the removal staff may be asked by the remedial staff to conduct a removal site evaluation whenever a new site is proposed for inclusion on the

OSWER Directive #9200.2-02 - 12 -

NPL, and enforcement staff may work with pre-remedial staff on identification of PRPs. Pre-remedial reviewers should share with remedial and removal staff any PA/SI or Hazard Ranking System (HRS) information that would indicate a need for early action. It is important for accelerated responses that technical and professional concerns of all four program offices about NPL sites be identified and addressed early in the response process.

5.2 Training

Superfund managers should encourage cross-training for pre-remedial, remedial, enforcement, and removal staff to allow SAMs, RPMs, and OSCs to learn how the entire Superfund response system works. It is important for site personnel to have a working knowledge of all programs. For example, preremedial and remedial staff should understand the capabilities of the removal program so that they can help ensure that removal action is taken where appropriate, and removal staff should know what remedial criteria will be considered before sites can be deleted from the NPL. Regional managers should encourage rotational assignments. OSWER Directive #9200.2-02 - 13 -

6.0 BIBLIOGRAPHY

<u>Guidance</u>

"A Management Review of the Superfund Program," Report from U.S. EPA, Washington, D.C. (6/89)

"Immediate Actions to Implement the Superfund Management Review," Memorandum from F. Henry Habicht to Regional Administrators (9/18/89)

OSWER Directive No. 9200.2-01, "Unaddressed NPL Sites," Memorandum from Henry L. Longest, II (7/6/89)

OSWER Directive No. 9355.0-25A, "Use of Removal Approaches to Speed Up Remedial Action Projects," Memorandum from Jonathan Z. Cannon (7/6/89)

OSWER Directive No. 9355.0-26, "Advancing the Use of Treatment Technologies for Superfund Remedies" (2/89)

OSWER Directive No. 9355.5-02/FS, "Expediting Remedial Construction" (10/89)

OSWER Directive No. 9380.3-01, "Treatability Studies Contractor Work Assignments," Memorandum from Henry L. Longest, II (7/12/89)

OSWER Directive No. 9833.0, "Guidance on the Use and Issuance of Administrative Orders Under Section 106" (9/8/83)

OSWER Directive No. 9833.1, "Issuance of Administrative Orders for Immediate Removal Actions" (2/21/84)

OSWER Directive No. 9833.3A, "Interim Guidance on Administrative Records for Selection of CERCLA Response Actions" (3/1/89)

OSWER Directive No. 9834.2, "Timely Initiation of Responsible Party Searches, Issuance of Notice Letters, and Release of Information" (10/9/85)

OSWER Directive No. 9834.3-2A, "PRP Search Supplemental Guidance for Sites in the Remedial Program" (6/16/89)

OSWER Directive No. 9834.4-A, "Guidance on Use and Enforcement of CERCLA Information Requests and Administrative Subpoenas" (8/25/88)

OSWER Directive No. 9834.10, "Interim Guidance on Notice Letters Negotiations, and Information Exchange - 53 FR 5298, February 23, 1988" (10/19/87)

OSWER Directive No. 9834.11, "Revised Procedures for Implementing Off-Site Response Actions," Memorandum from J. Winston Porter to EPA Regional Administrators (November 13, 1987) OSWER Directive #9200.2-02 - 14 -

OSWER Directive No. 9835.4, "Interim Guidance: Streamlining the CERCLA Settlement Decision Process" (2/12/87)

Manuals

OSWER Directive No. 9230.0-03B, Community Relations Handbook (1988)

OSWER Directive No. 9234.1-01, <u>CERCLA Compliance With Other Laws Manual</u> (8/8/88)*

OSWER Directive No. 9360.0-03B, <u>Superfund Removal Procedures Manual</u>, Revision No. 3 (2/88)

OSWER Directive No. 9834.3-1A, PRP Search Manual (8/87)

OSWER Directive No. 9837.2, Enforcement Project Management Handbook (7/89)

Statutes and Regulations

The Comprehensive Environmental Response, Compensation, and Liability Act of 1980 as amended, 42 U.S.C. 9601-9657

The National Oil and Hazardous Substances Pollution Contingency Plan, 40 CFR Part 300 (11/20/85)

The National Oil and Hazardous Substances Pollution Contingency Plan, 40 CFR Part 300, 53 Federal Register 51394-51520 (12/21/88)*

.

* draft document

Procedures for Completion and Deletion of NPL Sites: Five Year Review



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

DEC 2 5 1989

OSWER Directive 9320.2-3B

MEMORANDUM

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

SUBJECT: Update to the "Procedures for Completion and Deletion of National Priorities List Sites" Guidance Document Regarding the Performance of Five-Year Reviews (Superfund Management Review: Recommendation No. 2)
FROM: Henry L. Longest II, Director Hyp. The American State of Sta

Office of Emergency and Remedial Response Bruce M. Diamond, Director Office of Waste Programs Enforcement

TO: Director, Waste Management Division Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division Region II Director, Hazardous Waste Management Division Regions III, VI, and IX Director, Hazardous Waste Division Region X

PURPOSE

This memorandum incorporates into the "Procedures for Completion and Deletion of National Priorities List Sites" guidance document (OSWER Directive 9320.2-3A) EPA's policy to conduct at least one Five-Year Review prior to deleting sites from the National Priorities List (NPL). This memorandum: (1) implements Recommendation No. 2 contained in the Administrator's Management Review; (2) is a necessary follow-up to the October 30, 1989 Jonathan Cannon, Acting Assistant Administrator, policy directive to EPA Regional Administrators which explains which sites will require five-year reviews, and how the policy will affect deletions; and (3) identifies how EPA will administratively amend the deletion process to account for this policy directive.

BACKGROUND

On October 30, 1989, the Acting Assistant Administrator for the Office of Solid Waste and Emergency Response (OSWER) issued a policy directive concerning the performance of CERCLA 121(c) Fiveyear reviews and the relationship of such reviews to the deletion of sites from the NPL. This policy directive noted that EPA will ensure that five-year reviews are conducted for all remedial actions which result in hazardous substances, pollutants, or

contaminants remaining at the site above levels that allow for unlimited use and unrestricted exposure. This means that EPA will conduct reviews of a remedial action unless the site has been cleaned to at least health-protective levels and such levels allow for unlimited use and unrestricted exposure. Consequently, EPA will conduct five-year reviews of all remedies requiring any access or land-use restrictions or control, including remedies that attain health-protective levels for the current use, but which include restrictions on activities due to limits on exposure. Reviews will begin no more than five years after the initiation of a remedial The directive set out the policy that a site subject to action. five-year reviews should generally not be deleted from the NPL until at least one such review has been conducted following completion of all remedial actions at a site (except operation and maintenance).

Although SARA provides that CERCLA Section 121 (including 121 (c)) applies only to actions resulting from RODs signed post-SARA, the policy directive also notes that EPA believes as a matter of policy that it would be inappropriate to distinguish between pre- and post-SARA RODs in determining whether to conduct five-year reviews. Furthermore, also as a matter of policy, EPA will examine previously deleted sites to assess the appropriateness of conducting five-year reviews for those remedial actions which result in hazardous substances, pollutants, or contaminants remaining above levels that allow for unlimited use and unrestricted exposure. The purpose of such an examination would be to determine whether such remedies remain protective.

EPA is also currently developing guidance on the nature and extent of five-year reviews. EPA will revise and reissue the guidance on deletion/completion of NPL sites upon the issuance of the guidance on five-year reviews, which is expected in 1990.

IMPLEMENTATION

The following update of the April 1989 OSWER Directive 9320.2-3A, "Procedures for Completion and Deletion of NPL Sites", provides the administrative requirements which should be followed prior to deletion of sites from the NPL as a result of EPA's October 30, 1989 Five-Year Review policy directive. Effective immediately, these procedures should be followed for all sites affected by the Five-Year Review policy. Any questions regarding the attached update may be directed to Ed Hanlon of OSWER's Hazardous Site Control Division (HSCD) at FTS: 475-9753. Until the completion/deletion guidance is fully revised and reissued, please contact Allen Dotson, HSCD, at FTS: 382-5755, to determine the current policy on five-year reviews.

Attachment

CC: Regional Superfund Branch Chiefs Offices of Regional Counsel - Regional Branch Chiefs

Attachment

<u>12/29/89 Amendment to the April 1989 OSWER Directive 9320.2-3A.</u> <u>"Procedures for Completion and Deletion of</u> <u>National Priorities List Sites (NPL)</u>

1. <u>Disclaimer, "Notice," Amendment:</u>

a) Page ii. Add the following as the second paragraph:

"The policies set out in this memorandum are intended solely for the guidance of Government personnel. They are not intended, nor can they be relied upon, to create any rights enforceable by any party in litigation with the United States. EPA officials may decide to follow the guidance provided in this memorandum, or to act at variance with the guidance, based on an analysis of specific site circumstances. The Agency also reserves the right to change this guidance at any time without public notice."

- 2. Chapter 1, "Introduction," Amendment:
- a) Page 2. Add the following as the fifth paragraph under <u>Introduction</u>:

"EPA will ensure that five-year reviews are conducted at all sites at which a selected remedial action results in hazardous substances, pollutants, or contaminants remaining at a site above levels that allow for unlimited use and unrestricted exposure. EPA will generally not delete a site for which five-year reviews are required until one such review has been conducted following completion of all remedial actions at a site (except operation and maintenance). EPA Headquarters also intends to revise and reissue this guidance (OSWER Directive 9320.2-3A, as amended December 29, 1989) when the final policy on when and how to conduct five-year reviews is released. Until the reissuance of this completion/deletion guidance, EPA Regions should consult with EPA Headquarter's Hazardous Site Control Division to determine when and how the five-year reviews should be considered and conducted."

- 3. Chapter 2. "Site Completion." Amendments:
- a) Page 3. Add the following as the second paragraph under the sub-heading: "Final Operable Unit Remedial Actions":
"For Fund-financed remedial actions, the lead and support agencies should conduct a joint inspection at the conclusion of construction of the remedial action and concur through a joint memorandum that (a) the remedy has been constructed in accordance with the ROD and with the remedial design, and (b) a period for evaluating the operation of the remedy commences at that time, and should continue until the completion of any activities necessary to ensure that the remedy is fully operational and functional. Once the remedy is considered operational and functional by the party contracting for construction, a Remedial Action Report should be prepared by the party contracting for construction to officially provide its assurance that the work was performed within desired specifications, and is considered operational and functional. The lead and support agencies should then conduct a joint inspection and execute a joint memorandum accepting the Remedial Action Report."

b) Page 3. Change the second sentence under the subheading "No Action Sites" as follows:

"It does not include sites with RODs requiring only monitoring or institutional controls; these types of sites will be considered "Limited Action Sites" which will require five-year reviews to ensure protection of human health and the environment".

c) Page 3. Add the following to the third sentence under the subheading "No Action Sites":

"...have been addressed (e.g., O&M assurances, need for five-year reviews, and institutional controls)."

d) Page 5. The following new text will supersede the old text of the "LTRA Sites" subsection:

"Long Term Response Action (LTRA) Sites

An "Interim Close Out Report for LTRA Sites", prepared by the Region and approved by the RA, will be required of all LTRA sites. This report will contain final information for all completed operable units at the site and describe the LTRA activities to be performed, the cleanup levels to be achieved for the LTRA portion of the site, and any five-year review responsibilities (as discussed in the next section). This report will act as the determining factor for designating sites as LTRAs on the NPL and for internal Superfund tracking. In addition, once a ground or surface water restoration LTRA operable unit is operating

•

as designed, States may assume responsibility for operation of the LTRA.

The "Interim Close Out Report for LTRA Sites" will be amended when cleanup levels are achieved to include final information for the LTRA operable units of the site in order to satisfy completion requirements. The "Interim Close Out Report for LTRA Sites" and the amendment together will constitute the final Close Out Report for the site. The LTRA site will then be recategorized on the NPL as either a "Site Awaiting Deletion" or a "Five-Year Review Site""

e) Page 5. The following new text will be added as a separate subsection after the "LTRA Sites" subsection:

"Five-Year Review Sites

An "Interim Close Out Report for Five-Year Review Sites", prepared by the Region and approved by the RA, will be required of all Five-Year Review sites (this may incorporate by reference interim or final Close Out Reports already prepared). This report will contain final information for all completed operable units at the site and describe the Five-Year Review activities to be performed. This report will also act as the determining factor for designating sites as Five-Year Review sites on the NPL and for internal Superfund tracking. This report will be amended when at least one five-year review has been conducted following the completion of the remedial action (except operation and maintenance), and any appropriate actions have been taken to ensure that the site remains protective of human health and the environment. The "Interim Close Out Report for Five-Year Review Sites", and the amendment, together will constitute the final Close Out Report for the site. States may conduct five-year reviews under/pursuant to Cooperative Agreements or Superfund State Contracts with EPA, and submit five-year review reports to EPA.

For LTRA's such as bioremediation, flushing, and groundwater pump and treat where health-based levels may not be achieved on site for an extended period of time during and/or after site remediation, EPA will conduct five-year reviews from the date on which the first contract is awarded for work to install, construct, or implement the LTRA operable unit. Even at sites that are expected to achieve health-based levels at the completion of remedial action, EPA will, as a matter of policy, assure the conduct of five-year reviews when the remedial action will require more than five years to complete. An Interim Close Out Report for Five Year Review Sites would be required, for example, for a landfill closure site which is not an LTRA site. However, one Interim Close Out Report may be prepared for those sites which are designated both as an LTRA as well as a fiveyear review site. In these cases, the Interim Close Out Report will be amended twice, as follows: (a) when at least one five-year review has been conducted following the completion of the remedial action (except operation and maintenance), and any appropriate actions have been taken to ensure that the site remains protective of human health and the environment; and (b) when the LTRA cleanup levels are achieved, to include final information for the LTRA operable units of the site in order to satisfy completion requirements."

4. Chapter 3. "The Close Out Report," Amendments:

a) Page 7. Add the following separate category (as component 6) to the listed components which are necessary to be addressed in the Close Out Report:

***6)** Five-Year Review

- Statement explaining: (a) that at least one 0 five-year review has been conducted following completion of all remedial actions at the site (except operation and maintenance), and that any appropriate actions have been taken to ensure that the site remains protective of human health and the environment; or (b) why no five-year review was required. (EPA Headquarters will revise and reissue this completion/deletion guidance when the final policy on when and how to conduct five-year reviews is released. Until the reissuance of this guidance, EPA Regions should consult with EPA Headquarter's Hazardous Site Control Division to identify when and how the five-year reviews should be considered and conducted.)
- Assurance that, where appropriate, an acceptable and detailed workplan is in place for the performance of future five-year reviews, and is sufficient to determine whether the protectiveness of the remedy(s) for each operable unit, and of the site as a whole, is maintained.
 (A five-year review workplan may be incorporated into the operation and maintenance assurance agreements and workplans.)

 b) Page 8, Exhibit 2. Add the following item to the "Contribution to Close-Out Report" section across from the "Remedial Action" cleanup activity:

"o Five-year review plan, where appropriate"

5. Chapter 4. "NPL Deletion Criteria." Amendment:

a) Page 10. Add the following paragraphs after the third deletion criteria:

"In addition to the above, for all remedial actions which result in hazardous substances, pollutants, or contaminants remaining at the site above levels that allow for unlimited use and unrestricted exposure, it is EPA's policy that sites should generally not be deleted from the NPL until at least one five-year review has been conducted following completion of all remedial actions at a site (except operation and maintenance), any appropriate actions have been taken to ensure that the site remains protective of public health and the environment, and the site meets EPA's deletion criteria as outlined above. EPA must also assure that five-year reviews will continue to be conducted at the site until no hazardous substances, pollutants, or contaminants remain above levels that allow for unlimited use and unrestricted exposure. States may conduct five-year reviews under/pursuant to Cooperative Agreements or Superfund State Contracts with EPA, and submit five-year review reports to EPA.

An exception to this requirement involves situations where a Consent Decree contained language specifically committing EPA to delete a site from the NPL upon completion of certain response activities. In such cases, EPA Regions must consult with EPA Headquarters prior to initiation of any deletion activities. However, such an exception would apply only to the general policy of not deleting sites before completion of the first five-year review, not to the requirement to conduct reviews. EPA would still need to assure that five-year reviews will be conducted at the site. Given the October 30, 1989 policy directive from the Acting Assistant Administrator for OSWER regarding the performance of five-year reviews and their relationship to the deletion process, Consent Decrees should now require one five-year review following the completion of the remedial action (except operation and maintenance) before deletion."

6. Chapter 5. "The Deletion Process." Amendments:

a) Page 10. Revise the first sentence of the first paragraph as follows:

"The deletion process may begin after approval of the Close Out Report by the RA, and after RA and/or the State's approval of at least one five-year review at those sites which require five-year reviews."

b) Page 11, Exhibit 3. Add the following step immediately under the "Approved Close Out Report" step:

"Where Appropriate, Conduct At Least One Five-Year Review"

c) Page 12. Add the following immediately under "Close Out Report" in the suggested list of documents for the deletion docket:

"Initial Five-Year Review report, where appropriate"

 d) Page 13. Add the following separate bullet item to the "Supplementary Information: Item IV - Basis for Intended Site Deletion(s)" section, immediately under the description of O&M procedures:

"Description of the results of the initial five-year review, where appropriate, as well as reasoning for the need for future five-year reviews, and plans for performance of such reviews, in accordance with EPA's requirements for protectiveness at the time of each future review."

- 7. Appendix A, "Completion Process Diagrams," Amendment:
- a) Pages A-1 through A-4. Add the following immediately above the "NPL Deletion" item in the "Completion Scenario" charts for Remedial Sites, LTRA Sites, No Action Sites, and Removal Sites:

"Where Appropriate, Conduct At Least One Five-Year Review"

8. Appendix B. "Sample Close Out Report," Amendment:

a) Page B-6. Add the following separate chapter, as the new Chapter V, prior to the "<u>PROTECTIVENESS</u>" Chapter, to provide a summary of the Five-Year Review which, if appropriate, was conducted, and what actions, if any, were taken as a result of that Review, as follows:

"V. <u>SUMMARY OF FIVE YEAR REVIEW STATUS</u>

Consistent with the requirements of the October 30, 1989 policy directive from the Acting Assistant Administrator for OSWER which describes EPA's general policy of not deleting sites before completion of the first five year review following completion of all remedial actions at a site (except operation and maintenance), a five year review was completed and signed by the EPA Region IX Office on _____. Based on the findings of this five year review, EPA and the State of California have determined that all remedial actions conducted at the site remain protective of public health, welfare, and the environment.

EPA Region IX entered into a Superfund State Contract with the State of California on ______ to assure the performance of future five-year reviews at this site by the State. An acceptable and detailed workplan is in place for the performance of future fiveyear reviews. This workplan has been incorporated into the operation and maintenance plan already in place. If necessary, it will be revised at the time of each fiveyear review."

9. Appendix C. "Sample Notice of Intent To Delete." Amendments:

a) Page C-3. Add the following paragraphs after the third deletion criteria under Chapter II:

"In addition to the above, for all remedial actions which result in hazardous substances, pollutants, or contaminants remaining at the site above levels that allow for unlimited use and unrestricted exposure, it is EPA's policy that sites should generally not be deleted from the NPL until at least one five-year review has been conducted following completion of all remedial actions at a site (except operation and maintenance), any appropriate actions have been taken to ensure that the site remains protective of public health and the environment, and the site meets EPA's deletion criteria as outlined above. EPA must also assure that five-year reviews will continue to be conducted at the site until no hazardous substances, pollutants, or contaminants remain above levels that allow for unlimited use and unrestricted exposure. States may conduct five-year reviews under/pursuant to Cooperative Agreements or Superfund State Contracts with EPA, and submit fiv. year review reports to EPA.

An exception to this requirement involves situations where a Consent Decree contained language specifically committing EPA to delete a site from the NPL upon completion of certain response activities. In such cases, EPA Regions must consult with EPA Headquarters prior to initiation of any deletion activities. However, such an exception would apply only to the general policy of not deleting sites before completion of the first five-year review, not to the requirement to conduct reviews. EPA would still need to assure that five-year reviews will be conducted at the site. Given the October 30, 1989 policy directive from the Acting Assistant Administrator for OSWER regarding the performance of five-year reviews and their relationship to the deletion process, Consent Decrees should now require one five-year review following completion of the remedial action (except operation and maintenance) before deletion."

b) Page C-3. Add the following as the new procedure 1. under Chapter III:

"1. EPA Region II entered into a Superfund State Contract with the State of New Jersey to conduct fiveyear reviews at this site. New Jersey conducted the first five-year review on _____. EPA and the State find that the remedy continues to provide adequate protection of human health and the environment.

c) Page C-5. Add the following after the sentence beginning with "A five year..." in the paragraph beginning with "The institutional controls...", and delete the existing last sentence which begins "That Program...":

"EPA Region II entered into a Superfund State Contract with the State of New Jersey to conduct five-year reviews at this site. New Jersey conducted the first five-year review on _____. EPA and the State find that the remedy continues to provide adequate protection of human health and the environment.

An acceptable and detailed workplan is in place for the performance of future five-year reviews. This workplan has been incorporated into the operation and maintenance plan already in place, and has been sufficiently prepared to allow the EPA and the State of New Jersey to determine whether the protectiveness of the remedy for the site will be maintained over time. If necessary, it will be revised at the time of each five-year review."

10. Appendix E. "Sample Notice of Deletion." Amendment:

a) Page E-1. Change the last sentence of the SUMMARY section as follows:

"Moreover, EPA and the State of _____ have determined that remedial actions conducted at the site to date remain protective of public health, welfare, and the environment."

•

Interim Guidance on Addressing Immediate Threats at NPL Sites



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C 20460

JAN 30 1990

OSWER Directive No. 9200.2-03

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Interim Guidance on Addressing Immediate Threats at NPL Sites (Superfund Management Review: Recommendation No. 22) FROM: Don R. Clay Assistant Administrator

TO: Director, Waste Management Division Regions I, IV, V, VII, VIII
Director, Emergency and Remedial Response Division Region II
Director Hazardous Waste Management Division Regions III, VI
Director, Toxic and Hazardous Waste Management Division Region IX
Director, Hazardous Waste Division Region X
Director, Environmental Services Division Regions I, VI, VII
Regional Counsel, Regions I-X

PURPOSE

The purpose of this memorandum is to transmit Agency guidance on addressing immediate threats at National Priorities List (NPL) sites.

BACKGROUND

As a result of the Superfund Management Review, the Administrator committed that by September 30, 1990, all NPL sites would be free from immediate threats. Subsequently, this commitment became a Presidential-level "Management-By-Objective." The Office of Solid Waste and Emergency Response (OSWER) formed a workgroup to develop guidance to assist the Regions in taking expedited approaches to site cleanups and in making NPL sites "safer." On December 15, 1989, EPA issued the guidance "Accelerated Response at National Priorities List Sites" (OSWER Directive #9200.2-02) to address expediting cleanup. The question of making NPL sites "safer" is being addressed by the following directive, which provides procedures to help the Regions identify, document, and eliminate to the extent possible, immediate threats at proposed and final NPL sites. This document on addressing immediate threats and the previous OSWER Directive #9200.2-02 on accelerated response should be used together as companion pieces.

IMPLEMENTATION

Specifically, the attached OSWER directive requires that you:

- Establish Region-wide procedures to routinely review all NPL sites (the guidance outlines examples of procedures);
- By September 30, 1990, review all final NPL sites for the presence of immediate threats, document the findings of the review, and take action at all NPL sites that have immediate threats (the guidance provides documentation statements);
- o Thereafter, review and document the status of every final NPL site at least once every two years, at minimum reviewing half of the final NPL sites one year, the other half the next. The documentation memorandum will be due on January 1 of each year starting with January 1, 1992;
- Conduct an initial removal site evaluation at each newly proposed NPL site within three months of the date of proposal (the guidance explains some exceptions);
- o Take action at all proposed and final NPL sites that have immediate threats; and
- o Complete an environmental indicator form for all removal actions completed at NPL sites this fiscal year.

This directive is effective immediately and Regions should begin developing procedures and reviewing their sites. However, this is being issued as interim guidance to consider any comments you may have. For example, you may wish to suggest a due date different than January 1 (see third bullet above). Please send your comments to: Hans Crump, Acting Director, Emergency Response Division (OS-210), EPA, 401 M Street SW, Washington, D.C. 20460 by March 1, 1990.

If you have any questions Hans Crump may be reached at FTS 475-8720, or you may call Scott Maid at FTS 382-4671.

Attachment

cc:	Henry Longest II, OERR	Larry Reed, HSED
	Lloyd Guerci, OWPE	Russ Wyer, HSCD
	Lisa Friedman, OGC	Superfund Branch Chiefs, Regions I-X
	Hans Crump, ERD	OHM Coordinators, Regions I-X
	Clem Rastatter, OPM	Pre-Remedial Section Chiefs, Regions I-X

Addressing Immediate Threats At National Priorities List Sites¹

1.0 INTRODUCTION

1.1 Background

Two major recommendations of the study commissioned by the Environmental Protection Agency (EPA) in June 1989, entitled "A Management Review of the Superfund Program" (Superfund Management Review), were to expedite response at National Priorities List (NPL) sites and to make these sites safer. On December 15, 1989, EPA issued the guidance "Accelerated Response at National Priorities List Sites" (OSWER Directive #9200.2-02) to address the first recommendation. The directive described removal, remedial, and enforcement procedures and contract mechanisms for use by the Regions to accelerate CERCLA response actions at NPL sites. The following directive addresses the recommendation to make NPL sites safer. Because it refers to some sections of the accelerated response guidance, it should be used as a companion piece to that guidance.

1.2 Purpose

This document provides detailed procedures and guidance for evaluating and addressing immediate threats at NPL sites this year and in the following years. It provides that proposed new additions to the NPL undergo a removal site evaluation to identify the presence of immediate threats. It also provides that final NPL sites be reviewed at least once every two years to ensure, to the extent possible, that all NPL sites are free from immediate threats. This guidance is intended for Regional site managers, including On-Scene Coordinators, Site Assessment Managers, Remedial Project Managers, enforcement staff, and other Regional and Headquarters Superfund personnel and managers.²

¹ The policies and procedures established in this document are intended solely for the guidance of EPA personnel. They are not intended, and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. EPA reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

² Federal agencies with facilities listed on the NPL are encouraged to follow this guidance. Federal agencies with facilities on the NPL have primary responsibility for evaluating and documenting threats at their sites. (EPA may respond to emergencies at some Federal facilities; for more information see E.O. 12580.)

1.3 Scope Of Guidance

Specifically, the guidance will discuss the following topics:

- (1) Identifying and addressing immediate threats at NPL sites (Section 2.0); and
- (2) Procedures for reviewing and documenting that NPL sites do not pose immediate threats (Section 3.0).

2.0 IDENTIFYING AND ADDRESSING IMMEDIATE THREATS AT NPL SITES

The Agency's goal is to protect public health and the environment as much as possible from the risks posed by NPL sites, both short-term and long-term. Public health and the environment can be protected from short-term risks at NPL sites by stabilizing or mitigating immediate threats. Immediate threats to human health and the environment that result from deterioration of NPL site conditions before the completion of remedial action must be addressed as soon as possible.

The sections below discuss how the Agency will work to identify and address immediate threats. Section 2.1 explains what we mean by immediate threat and gives examples of problems that the Regions should be looking for. Sections 2.2 and 2.3 explain the evaluation and review process. Section 2.4 briefly discusses the mechanisms available for addressing immediate threats identified at NPL sites.

2.1 <u>Considerations During The Review And Evaluation Process</u>

The goal of the review and evaluation process discussed below (in sections 2.2 and 2.3) is to identify, document, and eliminate to the extent possible immediate threats that may be posed by NPL sites.³ For example, the review and evaluation process should attempt to identify threats of fire or explosion, direct contact threats, significant threats of near-term migration, and other relatively predictable threats. Direct contact threats might include (but are not limited to) situations such as uncontrolled waste piles, overflowing lagoons.

³ Documentation that there are no immediate threats at an NPL site is not related to evidence of possible imminent and substantial endangerment. An endangerment is a threatened or potential harm. An endangerment is imminent if the conditions that give rise to it are present, even though the harm might not be realized for years. An endangerment is substantial if there is reasonable cause to believe that someone or something may be exposed to a risk of harm from a release or threatened release. The mere threat of harm or potential harm to public health, public welfare, or the environment is sufficient. The endangerment need not be immediate to be imminent.

contaminated drinking water, and uncontrolled access to sites where conditions on the surface pose health threats. Significant threats of near-term migration might include extensive contamination on the soil surface that would be spread by a heavy rain or snow. Generally, these situations should be addressed as quickly as possible to alleviate the threat. In addition, NPL sites should not have, on the surface, tanks and drums containing hazardous substances. In most cases, such waste on NPL sites should be addressed by a removal or remedial action (see section 2.4 below on mechanisms for addressing immediate threats). The reviewer should also consider whether conditions might have worsened at the site subsequent to the initial site evaluation as a result of weather, physical plant deterioration, vandalism, or other causes that would indicate the need for additional evaluations or a response action.

It is important that all Superfund site managers (including remedial project managers and site assessment managers), staff, and site contractors (such as the Field Investigation Team) be knowledgeable about the capabilities of the removal program, including the factors in §300.65 of the NCP (section 300.415 of the proposed NCP).⁴ The removal criteria in the NCP should be considered whenever a site is investigated at the Preliminary Assessment and/or Site Inspection (PA/SI) stage and when NPL sites are reviewed or evaluated for immediate threats. In addition, information collected as part of the PA/SI and removal site evaluations at NPL sites should be used, as appropriate, to develop long-term plans for remedial action for the sites.

The reviews and evaluations of NPL sites can generally be based on easily obtainable information for each site, as judged by the Region. In past discussions, Regions have asked to what extent they will be expected to sample ground water as part of the review process. If sampling wells exist, it may be appropriate to sample the water. However, drilling new sampling wells is generally not appropriate (i.e., usually it should be done as part of the Remedial Investigation/Feasibility Study (RI/FS) process, not during an NPL site review or removal site evaluation).

2.2 Evaluating Newly Proposed NPL Sites

Because conditions at sites newly proposed for inclusion on the NPL may pose immediate threats to human health or the environment, it is important that a removal site evaluation be routinely conducted within a short period after a site has been proposed (if the site has not been evaluated recently before proposal). A removal site evaluation helps ascertain the current condition of

⁴ For ease of use, references to both the old (1985) NCP and the proposed (1988) NCP sections are provided in the text. It is important to note, however, that the 1985 NCP remains in full effect until a revised NCP is promulgated. The revised NCP, which was proposed on December 21, 1988 (53 FR 51394), is expected to be finalized in 1990. The revised section numbers will become effective on the effective date of the regulation.

the site and whether there are any immediate threats such as fire, explosion, or direct contact (see section 2.1 above) that should be addressed.

Therefore, effective immediately, all Regions must complete a removal site evaluation at each newly proposed NPL site within three months of the date the site is officially proposed for inclusion on the NPL (unless it is not appropriate; see next paragraph). The components of a removal site evaluation are described in section 300.410 of the proposed NCP. This evaluation should include review of any available PA/SI information and, except in extenuating circumstances, a site visit. When the evaluation is completed, the information should be entered into CERCLIS (see section 3.3 below) and a memorandum containing the statements in section 3.3 below should be sent from the appropriate Regional Division Director to the Regional Administrator, with a copy to the Director of the Emergency Response Division, Headquarters (ERD). Regions may choose to issue just one memorandum for all of the sites in the Region listed in a proposed update or may issue a separate memorandum for each site.

In some situations, Regional staff may believe that a removal site evaluation is not appropriate or cannot be completed within three months of the date of proposal. For example, if the Region has recently (e.g., in the twelve month period before proposal) conducted a removal site evaluation or a removal action and believes another site evaluation is not needed, it may not be appropriate to conduct another one. As another example, if a particular proposed NPL update includes a large number of sites in one Region, the Region may need more than three months to complete all the evaluations. In such situations, the Region should contact the Regional Coordinator in ERD. Then the appropriate Regional Division Director should send a memorandum to the Regional Administrator with a copy to the Director of ERD. The memo should briefly explain the reason that a removal site evaluation is not being conducted or is being delayed. For those that are delayed, the memo should explain when they will be completed.

2.3 <u>Reviewing The Status Of Final NPL Sites</u>

Because NPL sites can deteriorate while awaiting final remedial action, it is important to review all NPL sites periodically to ensure that there are no immediate threats. Such a periodic review also assists Regions in ensuring that the worst sites are addressed first.

Therefore, in keeping with commitments made to implement the Superfund Management Review, <u>all Regions must</u>, by <u>September 30, 1990</u>, <u>document that there</u> <u>are presently no immediate threats</u>, or that they are taking action, at all sites <u>on the final NPL</u> (see section 3.3 for information on documentation).

After this initial review and documentation, <u>subsequent documentation</u> <u>memoranda will be due on January 1 of each year starting with January 1, 1992</u>. <u>Regions must review and document the status of every final NPL site at least once</u> <u>every two years, reviewing half the final NPL sites each year</u> (using the procedures developed under section 3.1 and the statements in 3.3). This means

that after the initial review of all of the final NPL sites in 1990, Regions will review and document the status of half the final NPL sites again by January 1, 1992, the other half by January 1, 1993, and so on. This requirement recognizes that while sites must be reviewed routinely to achieve Agency goals, the reviews should not be so frequent that they create an excessive resource burden for the Regions. In addition, reviewing half of the sites one year and the other half the next should assist Regions in planning their workload and budget cycle. In keeping with the principle of addressing "worst sites first," Regions should consider the results of the 1990 review in deciding which half of the final NPL sites to review in 1991 (in order to document the results in 1992). If a proposed site makes the final NPL, generally it should be reviewed two years after the removal site evaluation (see section 2.2 above) or after it becomes final, whichever is later (unless a problem is suspected or an earlier review is deemed appropriate).

All documentation memoranda must be signed by a Regional Division Director and sent to the Regional Administrator with a copy to the Director of ERD. Section 3.0 describes procedures for accomplishing the review/documentation process.

Note: The purpose of the review and documentation is not to mandate that sites be visited but instead to ensure that site circumstances are considered at least every two years. Thus, specific site circumstances should dictate how each site will be reviewed for immediate threats. For example, some sites may be free from surface contamination, known to be very stable, or have extensive existing data. In these cases, there may be no concern in documenting the site as free from immediate threats without a recent site visit because weather, vandalism, etc., could not cause further harm. Regions may need to visit or use other methods to review other sites more frequently.

2.4 Mechanisms For Addressing Immediate Threats

Once the immediate threats have been identified, they should be addressed in a timely manner. The December 15, 1989, directive on accelerated response at NPL sites (OSWER Directive #9200.2-02) describes the removal and remedial mechanisms available for addressing NPL sites. Any of the mechanisms described there may be used for addressing immediate threats as well as for accelerating response. The type of threat found, the amount of time available before the threat must be addressed, and the resources available (e.g., personnel and contractor) will dictate whether remedial or removal authority should be used to address immediate threats. Generally, the removal program will be used to respond to immediate threats that must be addressed quickly (see the accelerated response directive for more information on the removal and remedial programs; e.g., the guidance explains the requirement for issuing an action memo for removal actions, etc.). The remedial program may be used to respond to some threats identified during the review process, especially when found during an on-going RI/FS and there is sufficient time to complete a Record of Decision and conduct an accelerated remedial action. As discussed in the directive on accelerated response, site managers must take advantage of enforcement authorities whenever possible.

3.0 <u>PROCEDURES FOR REVIEWING AND DOCUMENTING THAT NPL SITES DO NOT POSE</u> <u>INMEDIATE THREATS</u>

The following sections describe the types of procedures that Regions may use to review their NPL sites and document the findings. Section 3.1 describes optional procedures that the Regions may consider to review final NPL sites for immediate threats (the procedures do not apply to proposed NPL sites because Regions generally will conduct a formal removal site evaluation on proposed sites). Section 3.2 discusses options for structuring the review. Section 3.3 presents statements which all Regions must use to document that final NPL sites do not present an immediate threat or that action is being taken. Section 3.4 gives information on environmental indicators.

6

3.1 <u>Review Procedures For Final NPL Sites</u>

The Regions must develop internal procedures specifying how they will accomplish the review/documentation process. The procedures must explain which offices will be responsible for the initial review, how sites will be handled if further evaluation is needed, and who will sign the documentation memorandum (in some Regions, more than one Division Director may be involved). The Regions should establish their procedures for review of NPL sites as soon as possible, in order to allow sufficient time to complete the necessary work involved in meeting the September 30, 1990, deadline.

The Regions have wide latitude in the formulation of a review/documentation process. For example, some Regions may choose to have site managers (i.e., staff responsible for NPL sites on a day-to-day basis) conduct the initial review, referring questionable sites for further evaluation by the removal program, or the Regions may choose to have the removal program perform the entire evaluation. Alternatively, a Region may choose to establish a task force of staff from all Superfund programs to coordinate Regional reviews of NPL sites.

The Regions may choose to adopt any of the abovementioned options, or use any combination of approaches as appropriate, or may instead develop different procedures to accomplish the review. Flexibility is necessary in order to accommodate Regions with a small number of final NPL sites that may want to organize the review process differently from Regions with a large number of final NPL sites. In all cases, however, the final documentation consists of the statements and CERCLIS report discussed in section 3.3.

3.2 Structuring The Review Of Final NPL Sites

Regardless of the approach chosen and depending on the number of final NPL sites, the Regions may wish to conduct their reviews throughout the year. For example, if a Region has 200 final NPL sites (not counting Federal facilities; see footnote 2) the Region may wish to structure the review process so that 25 sites are reviewed each quarter (because the status of half of the sites will be reviewed and documented each year). In this case, the Region may wish to issue a documentation memo on a quarterly basis for each group of sites or may choose to wait and document all of the site reviews once a year, noting that the actual reviews were done on a quarterly basis and will continue on this basis in the coming year. Regional plans should also take into account the fact that additional sites will be finalized and/or proposed for the NPL during the twoyear review period. Regions will have to conduct removal site evaluations at these sites within three months of proposal (see section 2.2). In any case, the review (and findings if additional evaluation/action is needed) should be entered into CERCLIS when the review is completed.

3.3 <u>Documenting The Results Of The Reviews And Evaluations</u> (Final and Proposed NPL Sites)

Regions must use the following statements, as appropriate, to document that they have reviewed all final NPL sites (except Federal facilities; see footnote 2). The statements should also be used to document findings of the removal site evaluations conducted at newly proposed sites (see section 2.2).

Within the next several months Headquarters will add a new event type to CERCLIS and write new reports to assist the Regions in tracking their reviews and documenting the results. Headquarters will send guidance on the new event and reports to Regional offices in the near future. The CERCLIS reports will be the attachments indicated in the statements below.

<u>Documentation for sites where a response (remedial or removal) is ongoing or</u> <u>planned</u> (for planned actions, statement generally should be used only where response action is planned to be taken within twelve months of the date of the memorandum):

"The following sites have response action ongoing or are scheduled for response action to alleviate immediate threats at these sites. The attachment [CERCLIS report] shows the quarter and year when the work is scheduled to begin (for planned actions) or when work started (for ongoing actions)."

Documentation that sites do not have immediate threats:

"I have reviewed the available information on the sites listed in attachment ______ [CERCLIS report] and, based on this information, there are presently no immediate threats at these sites. The Agency, at its discretion, may take further action at these sites under CERCLA section 104, 106, or 122."

For final NPL sites, each Region is required to issue only one documentation memorandum per year (but may issue more than one as described in section 3.2 above). The memo must contain the statements and attachments described above and cover all of the final NPL sites in the Region that are being reviewed/documented that year. For proposed NPL sites, a Region may issue one memo for each site or

may group the sites. A Regional Division Director must sign the memorandum containing the appropriate statements and send it to the Regional Administrator with a copy to the Director, Emergency Response Division. As mentioned earlier, the first documentation memo (with the attachments) is due September 30, 1990, for all final NPL sites (except Federal facilities). Subsequent memos are due for half of the final NPL sites every year, starting on January 1, 1992. Memoranda for proposed NPL sites are due 3 months after proposal.

3.4 Environmental Indicators

Regions must complete an environmental indicator form for all removal actions completed at proposed or final NPL sites this fiscal year. i.e., FY 90. This will support the documentation and public explanation of what has been done to fulfill the Administrator's commitment on this Presidential objective. The environmental indicator forms and instructions will be sent to the Oil and Hazardous Material Coordinator in each Region in the near future.

4.0 **BIBLIOGRAPHY**

Guidance

"A Management Review of the Superfund Program," Report from U.S. EPA, Washington, DC (6/89)

OSWER Directive No. 9200.2-02, "Accelerated Response at National Priorities List Sites" (12/15/89)

Statutes and Regulations

The Comprehensive Environmental Response, Compensation, and Liability Act of 1980 as amended, 42 U.S.C. 9601-9657

The National Oil and Hazardous Substances Pollution Contingency Plan, 40 CFR Part 300 (11/20/85)

The National Oil and Hazardous Substances Pollution Contingency Plan, Proposed Rule, 40 CFR Part 300, 53 <u>Federal Register</u> 51394-51520 (12/21/88)

Guidance for State-Lead Removal Actions

•



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JL 10 1987

OSWER Directive # 9375.1-4-W

MEMORAHDUM

- SUBJECT: Addendum to the Manual: State Participation in the Superfund Program -- Appendix W, "Guidance for State Leff Removal Actions"
- FROM: Henry L. Longest II, Director Office of Emergency and Remedial Mesonhse
- TO: Waste Management Division Directors, Regions I-X Environmental Services Division Directors, Regions I, VI, and VII

Attached is interim final guidance on State-lead removal actions, issued as Appendix W to the <u>State Participation in the Superfund Program manual</u>. This guidance sets forth the policy and procedures for executing Cooperative Agreements with States for non-time-critical removal actions. The guidance is intended to provide the Regions with a new management tool for handling your workload and further delegating program responsibilities to States.

Appendix W has been developed by the Emergency Response Division in cooperation with the Hazardous Site Control Division, as well as Regional and State personnel who served on the workgroup. Two prior versions of this guidance document have been issued for Regional review and comment. This interim final guidance, issued as OSWER Directive 9375.1-4-W, has been revised to incorporate Regional comments where appropriate.

The final guidance is divided into four major sections:

•	Section	I:	Scope of State-Lead Removal Actions
9	Section	II:	Development of Cooperative Agreement Application Packages
•	Section	III:	Administering Cooperative Agreements
0	Section	IV	Closeout of Response Agreements

In addition, where applicable to removals, the final guidance incorporates existing procedures and requirements for executing Cooperative Agreements for remedial actions as contained in Chapters I-X of the <u>State Participation in</u> <u>the Superfund Program</u> manual. In order to streamline the guidance development process, sections of the manual relevant to the removal program have been crossreferenced in the State-lead removals guidance. Response personnel, therefore, must have access to the manual in order to execute Cooperative Agreements with States for removal actions. Questions or comments concerning implementation of this new program should be directed to Don Kraft, Special Assistant to the Director of the Emergency Response Division at (202) 382-2452.

•

Attachment

cc: OHM Coordinators, EPA Regions I-X Tim Fields Don Kraft

9375.1-4-W

APPENDIX W

GUIDANCE FOR STATE-LEAD REMOVAL ACTIONS

OSWER DIRECTIVE 9375.1-4-W

•

.

PREPACE

This-guidance sets forth the policy and procedures for awarding to States the authority and funds necessary to lead a CERCLA-funded removal action. It is intended to provide Regions with a new management tool for handling their workload and as a mechanism for further delegating program responsibilities to States. Under this program, States may lead non-time-critical removal actions at NPL and non-NPL sites. Authority to enter into a Cooperative Agreement with interested States rests with the Regional Administrator as set forth in Delegation 14-1-B (Superfund State Contracts and Cooperative Agreements for Removal Actions).

Provisions of this guidance may be subject to revision given CERCLA Reauthorization, proposed revisions to the National Contingency Plan and removal program policy/guidance development activities. In addition, it is anticipated that revisions to the operating procedures set forth in the guidance may be necessary once Regions have obtained experience in implementing Cooperative Agreements with States.

This guidance has been developed by OSWER'S Emergency Response Division (ERD) in cooperation with the Hazardous Site Control Division. Regional staff participating in the implementation of State-lead removal actions should contact their appropriate Regional grant personnel or HQ ERD Regional Coordinator if questions or problems arise when executing a Cooperative Agreement.

APPENDIX W

GUIDANCE FOR STATE-LEAD REMOVAL ACTIONS

I. SCOPE OF STATE-LEAD REMOVAL ACTIONS

A. Types of Cooperative Agreements

A State must enter into a Cooperative Agreement with EPA before beginning a response action using CERCLA funds. A Cooperative Agreement is the mechanism established by the Federal Grant and Cooperative Agreement Act that a Federal agency uses to provide States or political subdivisions with funding assistance while retaining significant involvement in the project. The Cooperative Agreement documents the respective responsibilities of the recipient of Federal funds and the agency providing the assistance. Cooperative Agreements are used to:

- . Transfer funds for specific project(s)
- . Document the State's statutory and regulatory responsibilities and assurances
- . Approve project-specific budgets and scopes of work
- . Identify any special program requirements related to the project
- . Document the Federal agency's role and responsibilities during the project.

There are two types of Cooperative Agreements: 1) a site-specific Cooperative Agreement: and 2) a multi-site Cooperative Agreement (MSCA). <u>State-lead removal actions may be executed via a site-specific</u> or multi-site Cooperative Agreement.

1. Site-specific Cooperative Agreements are appropriate mechanisms to fund response activities required at a single site. These agreements cover one removal action at one site and can be amended to include subsequent removal activities and to provide funds necessary to complete the action at that site.

2. A multi-site Cooperative Agreement is an "umbrella" Cooperative Agreement that, under one funding document, may include several response activities at more than one site within a State. MSCAs should be used in situations where sites are within close proximity. States requesting to lead removal actions at more than one site may choose to develop a MSCA or an existing MSCA may be amended to include a State-lead removal. If the State agency identified to lead Superfund response actions is different from the agency certified under an existing agreement with EPA, the State must submit to EPA a letter (signed by the Governor or Attorney General) indicating it has the authority to accept Federal funds and make the required assurances.

B. Types of Actions

1. <u>State-lead removal actions initially will be limited to</u> <u>non-time-critical removals at NPL and non-NPL sites</u>. All time-critical removal actions will be Federal-lead, including actions that are initially categorized as non-time-critical, but due to extenuating circumstances, the Regional Administrator (RA) has determined to be more appropriate for a Federal-lead response (e.g., a time-critical response becomes necessary, required response is more extensive than anticipated and exceeds State capabilities). Cooperative Agreements must contain a special condition to this effect.

(a) Non-time-critical removals appropriate for State-lead are actions where initiation of cleanup or stabilization efforts may be delayed for approximately six months or more from the time the threat is discovered.

(b) Non-time-critical removals include all activities formerly categorized as initial remedial measures (IRMs) under the remedial program and longer term removals that can be planned in advance.

(c) To date. State experience in leading non-time-critical actions has been limited to IRM-type activities previously conducted under the remedial program. Examples of State-lead IRM-type responses include fence construction. erosion control and off-site disposal of hazardous waste. Additional examples are provided in Exhibit 1.

2. The Superfund Comprehensive Accomplishments Plan (SCAP) process will be used as a planning/management tool for identifying removal actions appropriate for State-lead. Only removals that are listed on the approved or revised SCAP can be State-lead.

(a) Funds may not be obligated for State-lead removal actions that do not appear on the approved SCAP. Removal actions under consideration for State-lead must be listed on the SCAP at least one guarter in advance. However, these actions may

ECHIBIT 1

EXAMPLES OF STATE-LEAD IRM PROJECTS CONDUCTED UNDER THE REMEDIAL PROGRAM

- Excavation and off-site disposal of surface and buried waste materials and contaminated soil
- Fence construction

••

- Bank stabilization
- Water supply treatment of municipal well
- Installation of carbon filters on private wells
- Construction of lagoon perimeter dike
- Temporary cap for lagoons
- Posting of cautionary signs along creek
- Storm water control

be projected on the SCAP for more than one quarter in advance since, by definition, non-time-critical removals are actions where initiation of cleanup or stabilization efforts may be delayed for approximately <u>six months or more</u> from the time the threat is discovered.

(b) Removal actions identified for State-lead must be on the SCAP before the Action Memorandum and Cooperative Agreement can be approved. Preparation of the Action Memorandum and Cooperative Agreement application can begin, however, before the removal action is placed on the SCAP.

(c) Revisions to the approved SCAP require coordination with EPA Headquarters. Regional Oil and Hazardous Materials (OHM) Coordinators are responsible for initiating any necessary SCAP updates/revisions to ensure that sites identified for State-lead are placed on the SCAP, as required. OHM Coordinators should contact their designated Emergency Response Division (ERD) Regional Coordinator in EPA Headquarters when updates/revisions to the SCAP are required.

C. Scope of Activities

1. All CERCLA-funded preliminary assessment and section 104(b) activities undertaken to assess the extent of contamination and to determine whether the incident meets the NCP and CERCLA criteria for removal action will be Federal-lead. This does not preclude States from leading pre-remedial activities under a remedial Cooperative Agreement.

2. All Engineering Evaluations/Cost Analyses (EE/CAs) necessary to meet the requirements of the National Environmental Policy Act (NEPA) will be Federal-lead.

3. All enforcement activities, including Potentially Responsible Party (PRP) search, and notification and negotiation with PRPs will be Federal-lead. State-lead enforcement activities for removal actions may be considered once the program has been implemented.

4. Only those activities authorized in the initial or amended Action Memorandum will be State-lead, including any post removal site control (formerly operation and maintenance or O&M) that is CERCLA-funded.

(a) States will procure and lead all contractor cleanup and stabilization activities, including operation and maintenance activities authorized in the Action Memorandum.

(b) Once response begins, no party other than the State may direct the response activities being performed either by the State or its contractor(s). However, at the discretion of the RA, a response may be determined to be more appropriate for Federal-lead (see section I.B.1 of this guidance).

D. Approval of State-Lead Removals

1. The RA will determine, on a case-by-case basis, removal actions appropriate for State-lead. Factors the RA should consider when evaluating a State's request to lead a CERCLA-funded removal include, but are not limited to:

- State experience in leading activities conducted under the remedial program (e.g., IRM-type actions) that are similar to the response actions required to clean up or to stabilize the release at the site under evaluation for State-lead.
- . State experience in responding to hazardous substance spills/incidents independent of Federal involvement/funds.
- . Existence of a State Contingency Plan for hazardous substance release response.

2. <u>A Removal Action Memorandum and Cooperative Agreement are</u> required for all State-lead removal actions. The Action Memorandum will be an integral part of the Cooperative Agreement and must be approved before an Agreement can be awarded to the State. A copy of the approved Action Memorandum will be made available to the State and will be provided when the Agreement is awarded, if not before. All Cooperative Agreements, and amendments to the Agreement, will be negotiated at the Regional level and approved by the Regional Administrator. OSCs or RPMs. as appropriate, will be the State's primary EPA contact for developing and negotiating Cooperative Agreements.

(a) EPA will prepare the Action Memoranda in accordance with current program procedures, and in close cooperation/ consultation with the State. EPA will always select the response/activities to be taken at the site in consultation with the State. The Action Memorandum must document that the removal will be State-lead and identify what cleanup or stabilization actions must be taken within a specified cost and duration. The Action Memorandum must also identify activities that will be Federal-lead (e.g., enforcement activities).

- In accordance with program policy and procedures, the AA, OSWER must approve all Action Memoranda for removal actions initially or ultimately expected to exceed the statutory limitation on cost; otherwise, the RA will approve the Action Memoranda, including exemptions to the limitation on time, with OSWER concurrence required on proposed precedent-setting non-NPL removals.
- Changes in project scope and exemptions to the statutory limitations on cost and duration must be documented and approved in accordance with current removal program procedures before the Cooperative Agreement is amended. In order to expedite the approval process. State Project Officers (SPOs) should contact the assigned OSC/RPM as soon as it is known that an amendment to the Action Memorandum is necessary.

These procedures are discussed briefly in Section III of this guidance and in more detail in the <u>Superfund Removal</u> <u>Procedures</u> manual.

(b) States are responsible for preparing the Cooperative Agreement package, which must include a Cooperative Agreement Application, EPA Form 5700-33, and the attachments discussed in section II (Development of Cooperative Agreement Application Packages) of this guidance.

E. Funding Mechanism

1. State-lead removal actions will be funded via a Letter of Credit.

(a) Under this method, CERCLA funds are provided to a State through an existing Letter of Credit established at a Federal Reserve Bank chosen by the State. The State uses — or "draws down" — funds from the credit account to cover its immediate cash needs.

(b) Under a Letter of Credit, funds are obligated when the RA signs the Cooperative Agreement and sends it as an offer of award to the State; funds become accessible to the State on an as-needed basis upon execution of the Cooperative Agreement.

The State Participation Manual provides additional information on the Letter of Credit funding method.

F. Procurement Methods/Conditions

Under a removal Cooperative Agreement, States <u>must</u> award a fixed price subagreement (lump sum, unit price or a combination of the two) when procuring contractor support, regardless of the procurement method selected, unless it receives the Award Official's prior written approval.

1. Methods of procurement that States may use include small purchase, formal advertisement, competitive negotiation, or non-competitive negotiation, as appropriate.

2. All procurement subagreements must adhere to any Engineering Evaluations/Cost Analyses that may have been conducted by EPA for the project covered in the Cooperative Agreement. A special condition to this effect should be included in the Cooperative Agreement.

3. To conduct procurement activities under the Superfund program. a State must either have an internal procurement system that complies with the requirements of 40 CFR Part 33 (Procurement under Assistance Agreements) or must use Part 33 as its procurement regulation and allow EPA preaward review of proposed procurement actions. Additional information on this requirement is provided in section II.2(a) of this guidance and Volume II of the State Participation Manual.

G. State Assurances

In order to enter into a Cooperative Agreement for a removal action, EPA policy requires States to provide assurances for (1) sharing in the cost of cleanup at publicly operated sites, (2) providing a facility in compliance with the Resource Conservation and Recovery Act for off-site treatment, storage, or disposal of substances taken from a site, and (3) assuming responsibility for post removal site control (formerly termed operation and maintenance or O&M). OSCs/RPMs are responsible for monitoring State compliance with these requirements.

1. States are not required to share in the cost of a removal action, unless that removal is conducted at an NPL site that was publicly operated (either by a State or a political subdivision thereof) at the time of a release and a remedial action is ultimately undertaken at the site. In these situations, States are required to pay 50 percent of all removal costs. A State is not required to pay its cost share for the removal until the remedial action is funded by EPA.

9375.1-4-₩

2. States are statutorily required to ensure the availability of a hazardous waste treatment, storage or disposal facility, if a <u>remedial</u> action requires off-site treatment, storage, or disposal of hazardous substances. Such facilities must be in compliance with Subtitle C of RCRA and a RCRA compliance inspection must have been completed at the facility within the six-month period prior to the receipt of wastes. According to current EPA policy on off-site disposal (OSWER Directive #9330.2-3), <u>non-time-critical removal</u> actions are subject to these requirements. SPOs should contact their EPA-designated OSC/RPM or Regional RCRA Off-site Contact (RROC) for assistance in identifying disposal facilities and resolving issues pertaining to off-site disposal.

3. Removal program policy allows for payment of post removal site control within the twelve months time limit for the entire removal. The State must assume responsibility for post removal site control at the conclusion of the removal action. EPA may include funds for post removal site control for the project period covered in the Action Memorandum. However, a State must assume responsibility for post removal site control after the completion of the CERCLA-financed action. The Cooperative Agreement Application or a special condition must provide an assurance that the State will assume responsibility for all post removal site control as long as necessary once the action is complete.

II. DEVELOPMENT OF COOPERATIVE AGREEMENT APPLICATION PACKAGES

The State is responsibile for developing the Cooperative Agreement <u>Application package</u>. While the OSC/RPM will be the State's primary EPA contact for developing and negotiating Cooperative Agreements for removals, Regions may elect to assign administrative responsibilities to Regional staff other than the OSC/RPM. State officials responsible for response agreements. usually State Project Officers (SPOs)*, should work closely with an EPA-designated OSC/RPM when developing the Cooperative Agreement. Additional support to the OSC/RPM will be available through the ERD Regional Coordinators in EPA Headquarters. OSCs/RPMs and OHM Coordinators are responsible for reviewing the Cooperative Agreement Application for accuracy and completeness.

A. The Cooperative Agreement Application

The Cooperative Agreement Application package must include EPA Form 5700-33 and several attachments. The application form and the required attachments are briefly discussed below. Additional information is provided in EPA's manual entitled <u>State Participation in the Superfund</u> <u>Program</u>.

1. When applying for CERCLA funds, a State <u>must</u> complete EPA Form 5700-33, Application for Federal Assistance — State and Local Nonconstruction Programs. This form consists of five parts:

- Part I General Summary Information
- . Part II Project Approval Information
- . Part III Budget Information
- . Part IV Project Narrative Statement
- Part V Assurances.

General instructions for completing each part are included in the application form. The State should ensure that costs included in the application are allowable for payment under CERCLA. To be allowable, proposed costs must be consistent with section 111 of CERCLA and with Federal cost principles outlined in the OMB Circular λ -87, "Cost Principles for State and Local Governments." The State may seek assistance from the OSC/RPM in determining which costs may be allowable. Final determination of the reasonableness of the cost estimates in the application will be made by the EPA Award Official. Exhibit 2 presents the content and Exhibit 3 the appropriate level of detail required for completing Cooperative Agreement budget information.

2. Attachments to the Cooperative Agreement Application

A Cooperative Agreement Application for removal actions must include the following attachments:

- . State certification letter
- . Procurement System Certification, EPA Form 5700-49
- . Intergovernmental review comments
- . Community relations plan, if applicable
- . Quality assurance/quality control (QA/QC) plan
- . Site safety plan

These attachments are briefly discussed below. More detailed information is provided in the State Participation Manual.

While the term State Project Officer is used throughout the guidance to denote a State's counterpart to an EPA OSC/RPM, it is recognized that different terminology may be used among the States.

-EXHIBIT 2 OBJECT CLASS CATEGORIES CONTENT REQUIRED FOR COMPLETING COOPERATIVE AGREEMENT BUDGET SHEETS

CATEGORY	CONTENT - THE STATE MAY INCLUDE:
Personnel	 Costs of wages paid to State employees who are engaged in response activities. (Calculated either as a percentage of time or level of effort (LOE) basis.)
Fringe Benefits	 Fringe benefits for State employees, calculated as a fixed percentage of salary or by some other agreed-upon method.
Travel	 Costs incurred by State employees for travel necessary for response activities.
Equipment	 Purchase price of <u>necessary</u> equipment that the State furnishes, less its residual value after project completion. If equipment costs are based on usage rates, the costs are calculated by a standard depreciation usage method or in accordance with OMB Circular A-102, Attach. N.
Materials and Supplies	 Purchase price of any <u>necessary</u> materials and supplies the State furnishes.
Contractual Services	 Costs associated with reimbursing contractor services, including direct and indirect contractor costs and a reasonable profit for personal services and nonconstruction contracts. (See 40 CFR Part 33 and OMB Circular A-87.)*
Other Direct Costs	 Costs such as equipment rental, real property purchase (see 40 CFR Part 30), and miscellaneous costs.
Indirect Costs	The State may include indirect costs.

In accordance with the Prompt Payment Act (PL 97-177), Federal funds may not be used for payment of interest penalties to contractors when bills are paid late.

.

.

EXHIBIT 3 -OBJECT CLASS CATEGORIES APPROPRIATE LEVEL OF DETAIL FOR ITEMIZED COOPERATIVE AGREEMENT APPLICATION BUDGET

INFORMATION

Personnel Positions of staff . Number of hours -Salary of staff (annual or hourly rate) Estimates of personnel costs, by position Basis (percentage or other) upon which Fringe Benefits fringe benefits are calculated Estimates of fringe benefit costs, by position . Purpose and estimated number of trips Travel ٠ • Starting point and destination Transportation method • Per Diem while on travel • Number of persons traveling • Estimated cost of trips • Equipment Number and type(s) of equipment to . be purchased Price of each piece Type(s) of materials and supplies to be Materials and Supplies • furnished Total prices • Estimated number of personal services Contractual Services • or nonconstruction contracts Nature of contract services . Estimated total cost for each contract*

CATEGORY

* Estimates should allow for bid and activity contingencies.

(a) A certification letter must be included in the Cooperative Agreement Application package. This letter must be signed by the Governor or Attorney General and must indicate that the agency entering into the agreement has both the authority to do so and to make the assurances required by EPA as discussed in section I.G of this guidance. This letter may be a generic, one-time statement that covers all sites within the State, or it may be site-specific. If a State chooses to submit certification letters on a site-specific basis, a certification letter must be provided for each site covered under a multi-site Cooperative Agreement.

(b) In order to conduct procurement activities under the Superfund program, a State either must have an internal procurement system that meets the intent of EPA's regulations, Procurement Under Assistance Agreements (40 CFR Part 33), or it must follow Part 33 when it conducts procurement and allow EPA pre-award review. The procurement certification form, EPA Form 5700-48 (Procurement System Certification), demonstrates whether or not the State's procurement system is equivalent to EPA requirements. The certification is valid for two years or for the length of the project period established in the assistance agreement, which ever is greater. If the State has previously provided this certification to EPA, the State needs only to indicate in Part A of the form the date the certification was originally submitted.

(c) In accordance with Executive Order 12372, State 'est removal proposals are subject to intergovernmental reviebefore EPA will obligate funds. Intergovernmental review is implemented under 40 CFR Part 29, Intergovernmental Review of EPA Programs and Activities. Under this regulation, if the State has an established review process that includes the project in question, an applicant must formally notify its designated single point of contact, as well as any directly affected governmental entities and areawide or regional planning agencies that it is seeking Federal assistance and comply with the State's review process. A copy of the notification should be sent to the appropriate EPA Regional office. Any comments received in response to this notification must be attached to the Cooperative Agreement Application. Funds will not be obligated to the State until representatives of the State have had an opportunity to comment on the proposed project(s). The intergovernmental review process should be initiated at least one quarter prior to obligation of funds for response at a site. This lead time is necessary to meet the required 60-day State review period

and to allow time for EPA to respond to State and local comments. EPA must accommodate, or explain why it cannot accommodate, any comment received during the formal process before it can award funds to the State.

(d) Section 300.67(b) of the National Contingency Plan (NCP) requires a formal community relations plan (CRP) for all removal actions that last longer than 45 days (Note: proposed revisions to the NCP change this requirement to 120 days). States are therefore required to develop and implement a CRP for all removal actions that are expected to extend beyond this designated time frame. Additional information on CRPs and guidance for developing these plans is provided in the Superfund Removal Procedures manual and in EPA's manual entitled Community Relations in Superfund: A Handbook. States may submit the CRP separately from the Cooperative Agreement Application package. In that event, either the Cooperative Agreement Application or a special condition must provide that the State will submit the CRP prior to beginning on-site activities. The CRP is subject to EPA review and approval.

(e) As required by 40 CFR Part 30.302(d)(2), a State must submit within 30 days of receiving its Cooperative Agreement the QA/QC procedures it intends to use in environmental monitoring. These include any sample collection and analysis activities that may be necessary during the response. One generic QA/QC plan is sufficient for an MSCA. In addition, however, States are required to develop a site-specific Quality Assurance Project Plan (QAPP) to describe the measurement systems it intends to use on a site. When preparing the QA/QC plan and QAPP, States must meet Superfund program data requirements, including the following:

- . Data produced must be able to withstand the scrutiny of litigative proceedings. thus requiring appropriate chain-of-custody, document control and QA/QC documentation.
- . Data collection must be cost-effective. Costs of generating the data cannot significantly exceed costs associated with similar analyses provided by the EPA Contract Laboratory Program (CLP).
 - Data turnaround times must meet project needs.
The QA/QC plan must be reviewed by the EPA Regional QA Officer and the RPM/OSC and must be approved by the Award Official before any sampling can begin for a project.

(f) States are required to have a site safety plan in place before field activities can commence. A site-specific plan must be developed for each removal action. Each safety plan must provide for the protection of on-site personnel and area residents. Plans must be consistent with site conditions and must cover all phases of incident operations. Site safety plans must also comply with all applicable Federal. State and local Occupational Safety and Health (OSH) laws and Occupational Safety and Health Administration (OSHA) standards. Each site-specific safety plan must be submitted to the OSC/RPM for review to ensure removal program requirements are met. Additional information on site safety plans is provided in the Superfund Removal Procedures manual. Site safety plans may be submitted to EPA after the Cooperative Agreement has been signed providing that the application or a special condition provides that the plan will be submitted for EPA review before on-site activities begin.

III. ADMINISTERING COOPERATIVE AGREEMENTS

OSCs/RPMs and SPOs are responsible for ensuring that response activities are conducted according to the agreed-upon scope of work, budget, and schedule included in the Cooperative Agreement and Action Memorandum. The State official responsible for directing response activities, usually the SPO, must ensure that all on-site activities are consistent with CERCLA, the NCP and removal program policies and procedures. Daily on-site presence by SPOs is therefore required when response activities are underway. The OSC/RPM and SPO should maintain close communication at all times to monitor progress effectively. This can include meetings, phone calls, written correspondence, and review of contractor monthly progress reports. In addition, the OSC/RPM and SPO must notify each other immediately of any unscheduled or unanticipated events (e.g., a fire or explosion on site that may require an emergency response by EFA) that may have a direct impact on the project and/or on the terms of the response agreement.

This section briefly discusses State reporting and cost documentation requirements, monitoring State financial commitments and procedures for agreement adjustments. Additional information on response agreement administration and reporting and cost documentation requirements is provided in the State Participation Manual.

A. <u>Reporting Requirements</u>

When entering into a Cooperative Agreement, States are subject to a number of reporting requirements, including submittal of technical progress reports, financial status reports and other reports as required in the Cooperative Agreement. These reporting requirements are discussed briefly below.

1. For removal actions. States must submit quarterly technical progress reports within 30 days of the end of each Federal fiscal quarter and at the end of each response. These reports are submitted to the Region and present information on activities performed during the quarter and on total work to date. Exhibit 4 lists the minimum elements that State technical progress reports should contain for removal actions.

(a) The OSC/RPM is responsible for reviewing technical progress reports and providing any necessary direction or assistance to the State. as required. The content of State reports should be tailored to specific requirements of the response activities as covered in the Action Memorandum and the provisions of the Cooperative Agreement. All information should be site-specific and the OSC/RPM and SPO should reach an initial agreement on the content and format of these reports during negotiation of the Cooperative Agreement. For certain removal actions, EPA may require daily communication from the State, including submission of contractor progress reports. A special condition to this effect should be included in the Cooperative Agreement at the time the agreement is negotiated and may include a provision that the State require from its contractor submission of daily and/or weekly progress reports.

(b) In addition, the State must prepare and submit a final technical progress report to the OSC/RPM within 90 days after the completion of the removal action. Removal actions are complete when the scope of work in the Action Memorandum has been completed and the State has demobilized. The Cooperative Agreement is closed out when all administrative/financial reporting requirements are completed. See section IV of this guidance.

2. Once a removal action has been completed. States must submit a final Financial Status Report (FSR), Standard Form 269, and any ancillary reports. Ancillary reports include statistical or monitoring data, operation and maintenance manuals and other reports as required in the Cooperative Agreement. The State should

EXHIBIT 4 CONTENT OF STATE TECHNICAL PROGRESS REPORTS

- Description of activities and tasks completed to date, by site
- Estimates (percentages) of work completed for each activity, by site, including a basis for the estimates
- Itemization of expenditures by object class for each response included in the Cooperative Agreement, including both expenditures for the quarter and the cumulative expenditures to date
- Estimated variances in cost and time to complete the project
- Narrative explanation of any trends observed

••

 Description of any actions taken or planned to resolve problems or delays encountered submit a final FSR and all ancillary reports to the OSC/RPM within 90 calendar days after completion of the removal action.

(a) The State must provide EPA with the opportunity to verify that the removal has been successfully completed. At the conclusion of the removal, joint EPA/State final inspection may be required to confirm that the removal action has been implemented properly and that all outstanding action items are resolved. The State and EPA will jointly inspect the project to confirm that all outstanding action items are resolved. The State shall prepare a final inspection report describing any outstanding items and their resolutions. A copy of this report shall be submitted to the EPA OSC/RPM.

(b) In addition, a Final OSC Report must be prepared by the State at the conclusion of each removal action in accordance with the NCP and removal program policy and procedures. This report will be prepared by the SPO, in consultation with the OSC/RPM, and must be signed by the SPO. The Final OSC Report is subject to the approval of the OSC/RPM and therefore must be signed by the OSC/RPM.

B. Cost Documentation Requirements

When entering into a Cooperative Agreement, States must also adhere to certain cost documentation requirements as set forth in 40 CFR Parts 30 and 33 and all additional Superfund-specific requirements or procedures for documenting State expenditure of CERCLA funds.

- For removal actions. State accounting and recordkeeping activities must be detailed on a site-specific basis to ensure effective cost recovery. States must also track and report expenditures by object class category. Superfund procedures include documentation and recordkeeping measures intended to protect the integrity of site data, such as:
 - (a) Notifying EPA of archive research by third parties unless authorized in writing by EPA
 - (b) Requiring all microform copying of original documentation to be done in accordance with, or in a manner equivalent to, the technical requirements for copying Federal government records (36 CFR section 1230 <u>et seq.</u>).
- 2. In addition, the Superfund program adheres to the general Agency-wide policy that, when requested by EPA, States must provide documentation to support cost recovery litigation and

related efforts. Documentation must be available for use as evidence to answer questions such as what work was authorized, what work was completed, what charges were incurred for the work, and what payments were made for the work.

C. Monitoring Financial Commitments

OSCs/RPMs are responsible for ensuring that State expenditure of CERCLA funds corresponds to technical progress at the site and is within the cost ceiling in the Action Memorandum and Cooperative Agreement. Costs incurred by the State may not exceed the amount obligated for the response in the Cooperative Agreement.

1. The OSC/RPM should review State drawdowns on a monthly basis. using drawdown information available from the Regional financial management office. Key documents for this review include the Financial Management System (FMS) Outlay Report and the State's quarterly reports.

2. Funds awarded under a Cooperative Agreement are assigned an activity (e.g., removal action) and site code. The OSC/RPM should ensure that the State is drawing funds from the proper account for the site and for the action funded. The account from which drawdowns are made, identified in the FMS Outlay Report and the State quarterly reports, must match the action being conducted.

3. Under a multi-site Cooperative Agreement, when a removal at a particular site has been completed, remaining funds may be used to fund removal activities at another site covered by the same Agreement. Such funds, however, must be officially transferred to the appropriate account. A transfer of funds between sites under a MSCA requires a formal amendment to the Cooperative Agreement to move the funds from one account to another. Funds remaining in a site account at the completion of a removal should be deobligated and returned to EPA unless the State requires additional funds to support other removal actions covered in the same Cooperative Agreement. All amendments to Cooperative Agreements will be negotiated at the Regional level. This requirement will ensure that the statutory limitation on cost is adhered to when there is a need to transfer funds.

D. Agreement Adjustments

Agreement adjustments consist of alterations in the amount, terms, conditions, project period, project scope or some other administrative, technical, or financial aspect of the Action Memorandum or Cooperative Agreement. Depending upon the significance of the change, adjustments to the Agreement can be made, either through formal amendments or in writing, between the OSC/RPM and the SPO. Modifications to Action Memoranda are subject to current removal procedures and policy. All modifications to the initial Action Memorandum (e.g., ceiling increases, exemptions to the statutory limits on cost and duration) must be documented in an amended Cooperative Agreement, but only after the Action Memorandum requesting the modification has been approved. Circumstances where a Cooperative Agreement and Action Memorandum must be formally amended are presented below.

1. A Cooperative Agreement must be amended when it requires a significant change. These occur when:

- . Objectives and/or scope of the project, as funded, are altered
- . Funds obligated under one accounting code will be used for a different activity or a different site
- . There is any increase or <u>substantial</u> decrease in the project period or budget
- . A rebudgeting of funds occurs, such as a transfer between contractural services and another object class category or a change in the amount of indirect costs
 - Changes to special conditions to the Cooperative Agreement award document.

(a) Formal amendments are not required for minor project changes that are consistent with the project's objectives. Minor changes include shifts between object class categories (except as indicated above) and adjustments to the work plan within the scope and objectives of the funded project. Minor project changes should be approved in writing by the OSC/RFM or the Regional Administrator, as appropriate.

(b) Responsibility for requesting and obtaining approval of modifications to the Cooperative Agreement rests with the SPO and OSC/RPM. The SPO or the OSC/RPM identifies the need for an adjustment and the SPO initiates the approval process by submitting a request to the Region. The OSC/RPM then determines whether a formal amendment is needed, and if so, transmits the request to the Regional Administrator. If the RA signs the amendment, it is sent to the State for acceptance and is executed upon signature by an authorized State official. 2. According to standard removal program policy and procedures, special requests, in the form of Action Memoranda, must be prepared when the following are required after approval of the initial Action Memorandum:

- . An increase in the project's established cost ceiling
- . Exemptions to the statutory limitations on cost and duration
- . Changes in scope of work
- . Changes among cost categories where no change in total project ceiling is required.

Guidance on the procedures for requesting and obtaining approval for these changes is set forth in the <u>Superfund Removal Procedures</u> manual. OSCs/RPMs are responsible for preparing these requests and obtaining the necessary concurrence, as well as ensuring that the Cooperative Agreement is amended accordingly.

IV. CLOSEOUT OF RESPONSE AGREEMENTS

In order to close out a Cooperative Agreement, EPA must determine that the State has completed the removal action and complied with all applicable administrative requirements under the agreement. The general closeout responsibilities of the State and the OSC/RPM are briefly discussed below.

A. State Responsibilities

1. A State must complete the work specified in the Cooperative Agreement and the Action Memorandum and produce the required reports (e.g., final FSR and technical progress report), manuals, plans and other information. The State is responsible for ensuring that all products and reports necessary for closeout are submitted to EPA, as required.

2. The SPO is responsible for the completion of the final report as described in the Cooperative Agreement including the reports and activities as follows:

- . Draft and final technical report
- Final FSR
- . Ancillary products such as O&M manuals
- . Property/equipment report
- . Other required reports.

B. EPA OSC Responsibilities

1. OSCs/RPMs are responsible for ensuring that the State has provided all required reports and is in compliance with the provisions of the Cooperative Agreement and Action Memorandum.

- 2. Specific OSC/RPM responsibilities are as follows:
 - . Reviews/inspects work in the project and arranges for acceptance or correction
 - . Reviews and approves required reports
 - Follows through on delinguencies or deficiencies
 - Ensures that all disputes and audit exceptions are resolved
 - . Transmits final determinations on any outstanding issues
 - Documents the closure in the official file
 - . Maintains the official records.

The State Participation Manual provides detailed guidance on closing out response agreements. OSCs/RPMs should refer to the State Participation Manual for additional information on the requirements/procedures for closing out Cooperative Agreements.

Clarification on Allowability of Management Assistence to States for ERAs and Removals



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

OSWER DIRECTIVE 9375.1-13

JUN 1 C 1953

OFFICE OF SOLID WASTE AND EMERGENCY RESPONS

MEMORANDUM

- SUBJECT: Clarification on Allowability of Management Assistance to States for ERAs and Removals
- FROM: Henry L. Longest II, Director Office of Emergency and Remedia Response
- TO: Waste Management Division Directors Regions I - X

PURPOSE

This memorandum is intended to clarify Agency policy on the allowability of providing funds to States for management assistance activities for expedited response actions (ERAs) conducted during Federal-lead response actions and for Federallead removals.

BACKGROUND

Current guidance does not specifically address the allowability of funding management assistance for State participation in Federal-lead ERAs or removals. Several Regions have received requests from States for funding management assistance to support ERAs, and Regions have asked for clarification from Headquarters on this issue. In addition, some States have requested funding to support removal actions, and OERR believes that clarification is required on the allowability of funding State involvement in Federal-lead removals as well.

POLICY ON ALLOWABILITY

If an existing management assistance cooperative agreement does not appear to have sufficient funds to cover a State's role in supporting an ERA being conducted during Federal-lead remedial response, the Region and State can amend the cooperative agreement to include the unexpected increase in State support. If the State is not already receiving management assistance for a Federal-lead activity when an ERA becomes necessary, the Region can provide management assistance via a cooperative igreement if the State feels its role in the ERA warrants such 'inancial assistance. States may not receive funds for management assistance on Federal-lead time-critical removals due to their emergency nature and related time constraints. These constraints do not allow sufficient time to negotilate and award funding of State support. Further, management assistance will generally not be provided for Federal-lead non-time-critical removals. However, funding may be considered on a case-by-case basis subject to availability of fund's when removal actions are complex and State support is expected to be extensive. Prior to consideration of funding management assistance for State involvement in any removals, Regions must consult with Headquarters.

IMPLEMENTATION

Funding for ERA assistance must be shown on the budget sheet for the management assistance cooperative agreement as a separate activity and tasks associated with this activity should be described in the Statement of Work for the agreement. Requirements for State cost-share depend on the phase of the remedial project during which the ERA is conducted. State cost share is not required for management assistance if the ERA is conducted during the RI/FS at publicly operated sites but cost share is required if the ERA is conducted after a ROD has been signed. Management assistance provided for the remedial action phase of response must be cost shared by the State at the same percentage as the remedial action.

Management assistance for Federal-lead removal activities will be funded via site-specific cooperative agreements only and should not be incorporated into Multi-site/activity Cooperative Agreements.

CONTACTS

If you have any questions on funding management assistance for ERAs, please contact Jan Wine on 382-2443. Questions on removals should be directed to Cristina Griffin on 382-2307.

cc: Harvey Pippen, GAD Lisa Karpf, OIG

Issuance of Administrative Orders for Immediate Removal Actions

1	United States Em	ironmental Protection Agen		Interim Directive
SEDA	OSWED Direct	ington. UC 20460 inco Initiation E	,	9833.1
VLIA	USWEN Direct		request	1
Name of Contact Person	Mail Code			
John Cross	WH	-527	475-677	70
Lead Office	ST	App	roved for Review	
		Hice Carector .		Date ,
0 osw - 0 AA.	OSWER 71.G.	Kling		5-30-86
Title			•	
Issuance of Aca	unistrative Orders	for Innediate Ren	cval Actions	
	•			
Summary of Directive				
			.	
This sets forth gu	idance on issuing A	ministrative Orde	ers for immediat	te
removal actions un	der CERCLA. This gu	idance should be	used in conjunc	ction
with the recently	issued Guidance Meno	brandum on the Use	e and Issuance	
of Administrative	Orders under Section	n 106(a) of CERCLA	A dated September	
8, 1983.				
		•• • • • •		
Key Words: admini	strative orders, im	ediate, guidance	•••	
remova	l actions,	•		
· ·				
ype of Orrective (Manual, Policy D	prective, Announcement, etc.)		Status	
Cuidance Memorandu	m		🔲 Draft	New
Guicance Meloranii	111		Final	Revisio
and this Directive Supersede Pres		Y No Deet It Sugal	I Disectory	
"Yee" to Either Question What f			ninghi riguida Direguud	(am []] (63
183 to child Couston, mat	yn active (nanioer, uner			
				•••••
		X Other (Secold	•	
		Uner (Sperity)	•	•
		•	-	· ·
his Request Meets OSWER Direct	ives System Format	•		
		••••	: 0	
19. C. Kla	¥	-		5-22-51
ignature of OSWER Directives Of	licer		0	ate .
			. •	• • • • • • •
			[· · ··
PA Form 1315-17 (10-85)				· .
				. •



WASHINGTON, D.C. 20460

FEB 2 | 1984

- OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

HEMORANDUM

SUBJECT: Issuance of Administrative Orders for Immediate Removal Actions FROM: Lee M. Thomas Assistant Administrator

TO: Regional Administrators; Regions I-X Air & Waste Management Division Directors Regions III, IV, VI, VII, VIII, X Waste Management Division Directors, Regions I, V Director, Office of Emergency and Remedial Response, Region II Toxics and Waste Management Division Director, Region IX Environmental Services Division Directors, Regions I - X Regional Counsel, Regions I - X

This memorandum sets forth guidance on issuing Administrative Orders for immediate removal actions under CERCLA. This guidance should be used in conjunction with the recently issued Guidance. Memorandum on Use and Issuance of Administrative Orders under Section 106(a) of CERCLA dated September 8, 1983.

Since becoming the Assistant Administrator, OSWER, I have sought to implement a "balanced" CERCLA program which uses both the administrative and civil judicial enforcement provisions of the Act--as well as the Fund--to secure clean up of hazardous waste sites. One of my primary enforcement goals is to increase the use of Administrative Orders for immediate removals. Orders are particularly useful in immediate removal situations, since they can be issued quickly, can require discrete segments of work (e.g., surface cleanup) and carry the threat of additional damages and penalties in the event of non-compliance.

We estimate that Administrative Orders may be appropriate for a significant percentage of immediate removal situations. Increased resources will be provided to the Regions, and I expect the Regions to devote resources to accomplishing this goal of increased Administrative Orders for removals.

In addition, the Regions must develop a satisfactory organizational structure if the Administrative Order program is to succeed. The organization of enforcement personnel varies among the Regions. The majority of the Regions keep their "remedial" and "removal" personnel in different divisions. Since CERCLA enforcement has (until now) concentrated heavily on remedial sites, most regional technical enforcement personnel have been assigned to the remedial response units (generally, the Air and Hazardous Material Divisions). Personnel responsible for immediate and planned removals have usually been assigned to the Environmental Services Division which, as a general rule, has not been assigned enforcement personnel.

Obviously, the ability of a Region to implement this new policy requires both close coordination among the immediate response staff and their colleagues in the technical enforcement and regional counsel offices and an organizational structure capable of developing and issuing quality orders. Regions that do not currently dedicate technical enforcement staff to their immediate removal program should assure that appropriate personnel are in place in the technical enforcement office to implement this policy and to handle the workload.

I. BACKGROUND

CERCLA identifies two types of response actions for which the Fund can be used: 'removal actions and remedial actions. The National Contingency Plan (NCP) further refines the former category into "immediate" and "planned" removals and describes the process and procedures for proceeding with these forms of response. (See Federal Register 31180; July 16, 1982). Please refer to the attached appendix for "n outline of the relevant CERCLA and NCP provisions regarding remuval activity, Administrative Orders and enforcement.

Because of the large number of sites which pose a health hazard, the Office of Emergency and Remedial Response (OERR) defines the category of immediate removals according to the immediacy and severity of t a hazard to the public health or environment. These categories establish a guide for the purpose of assessing the length of time within which the Agency must respond to the event. Agency response to situations which require immediate response (e.g., threats of fire, explosion or spills) normally takes place in a matter of hours or one or two days at the most; Agency response to other situations (e.g., rusting barrels that have not yet begun to leak, holding ponds that may overflow with the advent of the rainy season) normally takes place during a period which may range from a week to a month.

This guidance is most applicable to the <u>latter</u> situation; i.e., the Regions should consider issuing Administrative Orders in situations when there is at least one week between the time the On-Scene <u>Coordinator (OSC) determines that an immediate, removal is warranted</u> and the time that actual on-site response must begin.

Administrative Orders are a useful enforcement tool in these types of immediate removals situations, for the following reasons. First, they encourage private party response, particularly since it

-2-

is CSWER policy to meet, if at all possible, with responsible parties after the Order is issued if a meeting is requested. The results of an OWPE analysis of 49 completed immediate removals indicate that the elapsed time between the request for funds and the start of site response ranged from eight days to more than three weeks for 24 of the sites. This clearly indicates that there is time to issue Administrative Orders in appropriate situations, and the process described in this memorandum can be implemented in as little time as a week, if necessary. Second, removals require discrete units of work (e.g., barrel or contaminated soil removal) which makes responsible party compliance and Agency compliance monitoring easier. Third, the costs of immediate removals are generally moderate; this increases the probability of private party compliance.

In the event of non-compliance with an Administrative Order, the Agency is prepared to quickly initiate a Fund-financed response and seek fines/treble damages from the responsible parties. Since the treble damages will be based on the Fund dollars expended, these situations are particularly amenable to establishing treble damage claims, which the Agency will seek to recover in its \$107 cost recovery actions. (The average obligation for 110 prior immediate removals undertaken by the Agency was approximately \$275,000). Issuance of Administrative Orders for these situations also may improve the equitable position of the Agency in subsequent cost recovery cases.

II. CRITERIA FOR ISSUING ADMINISTRATIVE ORDERS

First, of course, the Agency must meet the legal threshold that an imminent and substantial endangerment to public health or the environment may exist.¹ Information which can be used and evaluated by the OSC or his supervisor to make this determination include:

- 1. Notification in accordance with CERCLA \$103 (a), (b) or (c)
- 2. Investigations by government authorities conducted pursuant to CERCLA §104 (e) or other statutory authority.

The Agency must be able to properly document and justify both its assertion that an immediate and significant risk of harm to human life or health or to the environment exists and its choice of the ultimate response action at a site in order to be able to oppose a challenge to the Order and to successfully litigate any subsequent cost recovery action. Adequate documentation consists of photographs, samples, monitoring or other documented site analysis. The Agency should follow chain of custody procedures to maintain the integrity of samples taken at the site. Please refer to the Cost Recovery Guidance, issued August 26, 1983 for more detailed guidance. The Revised Superfund Removal Guidance to be issued in late February -1984 will also provide additional guidance on immediate removal assessments.

- 3. Notification of a release by a federal or state permit holder when required by the permit.
- 4. Inventory efforts or random/incidental observation by government agencies or the public.

If the facts reach the legal thresholds of CERCLA \$106, several policy criteria for deciding whether to issue an Order for an immediate removal should be considered. The first of these is the amount of time available before site response must begin. This determination will usually be made by the OSC. An Order may be appropriate if there is a minimum of one week available for issuing the Order and meeting with the recipients (see further below) between the time of the decision to seek funds for the immediate removal and the initiation of on-site response. (Of course if an order can be issued in less than a week the Regions are not bound by the "one week minimum". However, the Regions should always attempt to have 48 - 72 hours available for the recipients to request and conduct a conference.)

A second policy criterion is the number of potential recipients of the Order and their financial viability. There should be a "manageable" number of responsible parties and they should be collectively capable of undertaking site response. The Regions will use their best judgement to decide what constitutes a "manageable" number of responsible parties and assess the capability of the parties to undertake the response for any individual immediate removal situation. (For a more lengthy discussion of criteria to consider when issuing an Administrative Order, please refer to the Administrative Order guidance.) When there is a large number of potentially responsible parties, Orders need not be issued to all of the parties. In this type of situation the Region should issue the Orders only to those parties most likely to comply. The Region, however, is not precluded from issuing Orders to all the parties if it so desires.

These criteria are to be used as general guidelines for determining whether an Administrative Order should be issued for an immediate removal. The varying factual circumstances presented in any potential removal action mandate that each Region conduct this necessary factual analysis to decide the appropriateness of an Order.

III. PROCESS FOR ISSUING ADMINISTRATIVE ORDERS

The timely development and issuance of Administrative Orders for immediate removals will require effective coordination among the OSC, technical enforcement personnel and the legal counsel in both the Regions and Headquarters. OSWER will not dictate how the Regions must organize or adjust personnel in order to accomplish this task, but it will expect the Regions to have a system in place which is capable of implementing an administrative order program for immediate removals. The procedures for developing and issuing orders follow:

The decision by the OSC or his superior to request funds for an immediate remov.l also triggers the process for deciding whether to issue an Administrative Order.

The OSC will inform the technical enforcement branch (or other appropriate enforcement personnel if no separate branch exists) and the Regional Counsel that a request for a Fund-financed immediate removal is being developed. Appropriate personnel in OERR and OWPE should also be informed of this action. While the OSC and his staff prepare the 10-point document,² technical enforcement personnel and the Regional Counsel should begin to identify responsible parties and assess their financial ability to conduct site cleanup.

The OSC or the Regional Counsel will attempt to orally contact (with written follow-up) potentially responsible parties in order to secure private-party response in lieu of the Fund. While previous Agency policy was to proceed with Fund-financed response if the responsible parties refused to act, the Agency will now issue administrative orders in appropriate circumstances before initiating Fund action, so long as the site does not pose an unreasonable risk of harm to the public health, welfare or the environment.

Regardless of whether a responsible party agrees or not to undertake the removal, development of the 10-point document should proceed as usual. However, the OSC and technical enforcement staff (in consultation with the Regional Counsel) shall apply the criteria cutlined in Part A (above) to recommend to the Regional Administrator whether to issue an Administrative Order. The decision to issue the order rests with the Regional Administrator, subject to the current delegations.

If the Regional Administrator decides to issue an Administrative Order, the Order will be drafted by technical enforcement personnel with the advice of the Regional Counsel. The technical information contained in the 10-point document will normally provide the basis for the Order's "Findings of Fact" while the Agency's intended response actions will serve as the remedy the recipient is required to implement.

2Requests for less than \$250,000 can be approved by the Regional Administrator while requests for more than \$250,000 require the approval of OERR. (It is anticipated that within the month, the Regional Administrators will be delegated the authority to obligate up to \$1 million for removal actions.) The ten point document itself must justify its cost estimates and be consistent with the NCP. With the issuance of the Revised Superfund Removal Guidance, the 10 point document will become an Action Memorandum. Since Administrative Orders will normally be issued in situations in which site response is not required for at least one week, OSWER policy is to provide recipients when possible an opportunity to meet with Agency personnel to discuss the terms of the Order and the means for compliance. Therefore, the Order should include the following provisions:³

- A statement of the imminent and substantial danger pursuant to \$106 of CERCLA and the risk of harm under \$300.65 of the NCP.
- 2. A statement of the authority of the issuing official (normally the Regional Administrator) to issue the Order and why the recipient is liable under \$107.
- 3. The steps the recipient must take to comply with the order, (following the provisions of the
- ten-point document in order to be as specific as possible).
- 4. A mandatory timetable for performing and completing the response. (The timetable should include at least one short term interim deadline so the Agency will have the ability if necessary, to demonstrate non-compliance before the project completion date.)
- 5. A provision informing the recipient that his duty to obey the terms of the order takes effect 72 hours after he receives the order.
- 6. A provision informing the recipient that he may orally contact the Agency to request a conference on the Order. The recipient must follow up his oral request in writing.
- 7. A provision specifying a date certain by which responses (either oral or written) to the Order must be received.
- 8. A provision which states that EPA reserves the right to undertake the action if emergency circumstances dictate such action and that such action in no way relieves the parties of responsibility for the costs of such actions.
- 9. A provision which requires: proper chain of custody procedures to be followed for any testing and sampling,
- adequate recordkeeping of activities (so records may be used as evidence in any future enforcement case), cooperation from employees of any contractor who engages in site activity, and availability of such employees to the U.S. in preparation and trial of a subsequent enforcement case.

JRefer to the general Administrative Order Guidance for examples of model orders and conference procedures.

Under a separate delegations memorandum to the Regions, the concurrence requirement will be waived for all Administrative Orders for immediate removals with obligations of \$1,000,000 or less. Within two weeks of issuance of the Order, the Regions are to send a copy of the final Order to OWPE.

As a matter of policy, in order to increase the likelihood of compliance, the Agency encourages the convening of a conference with the recipients of an Administrative Order. Since Administrative Orders will generally be issued for immediate removal situations which do not require response in less than one week, the Agency will normally attempt to hold a meeting with the recipient, if requested by the recipient. The conference should be convened on an expedited basis (e.g., within 72 hours after the Order is issued) if the recipient orally requests the conference. However, the Agency retains the right to "waive" a conference if immediate response is warranted because of deteriorating conditions at the site. The Regional Administrator shall have the authority to decide whether to eliminate the conference prior to or following the issuance of the Administrative Order. If the Regional Administrator waives the opportunity for a personal conference, a regional representative, must at least give the parties an opportunity to be heard by telephone before the effective date of the Order. In ceneral, conferences concerning removal actions should be used to clarify the requirements of the Order rather than as an opportunity to negotiate the requirements.

The Agency must create a good administrative record of its meetings with the recipient of an Order for either enforcement of the Order or cost recovery after a Fund-financed cleanup. The Agency participants should prepare a written summary of the conference containing:

- 1. The date and parficipants.
- 2. A summary of the significant issues raised and arguments/ data used by the recipient to contest the Order.
- 3. The result of the conference (e.g. agreements reached with the recipient, indication from the recipient of an unwillingness to comply with the Order)

The presiding official, (designated by the Regional Administrator) must also prepare a statement which addresses any significant arguments raised by the recipient and recommends whether any modifications to the Order are warranted. (See the September 8, 1983 Administrative Order Guidance for a complete discussion of the procedures and "ground rules" for conducting the conference and the time frames for holding them.) If the recipient agrees to undertake the stipulated response measures, the agreement may be in the form of a Consent Order. The OSC will monitor compliance with the Order and recommend additional enforcement action if the terms of the Consent Order are breached. If the recipient does not agree to undertake the measures contained in the Order, the Agency will generally not refer a case to the Department of Justice to force compliance because of the time constraints presented by the emergency. Rather, the Fund will be used for site response and the recipient(s) will be sued for cost recovery--including punitive damages in appropriate cases.

IV. USE OF THE FUND WHILE THE ADMINISTRATIVE ORDER IS BEING ISSUED

Normally, once an Order has been deemed appropriate for an immediate removal situation, the CERCLA Fund shall not be used to undertake a federally-funded immediate removal during the time period in which the Agency develops the Order, issues it to the responsible party, and conducts the conference.

However, if site conditions deteriorate-- presenting a corresponding increase in the threat that the site presents-- the Fund can be used for response while the Administrative Order process continues. In such instances, the Regional Administrator can approve the use of Funds below \$250K and request the Assistant Administrator, OSWER, to release funds if the response work will be greater than \$250K.⁴ The Administrative Order process should continue since the parties may undertake site response at the next convenient break in activity.

Thus, if there are deteriorating conditions at the site, the OSC should continue all steps necessary for undertaking a Fundfinanced response while the Order is being developed. The 10-point document should be prepared and receive the concurrence of all officials up through the Regional Administrator or the Director, OERR.

However, no actual obligation of Funds for site response will normally occur until after the Order has been issued and the conference has been held. Since the Order will only be issued in situations where an immediate response can be delayed, there will normally be time to see the Administrative Order process through to conclusion. The conference must be held within the time period specified in the Order (which will correspond to the time the Agency has before the response activity needs to begin). Since

"If deteriorating conditions require the Fund to respond while the Order is still being issued, OSWER assumes that the Fund will take all response actions necessary at the site (e.g., remove all barrels, not merely those that may be about to leak).

·.

the timing of the obligation will vary according to the estimated time needed to mobilize equipment and personnel, the OSC should work closely with the technical enforcement and Regional Counsel staff during the drafting of the Order to assure that the time period established for issuing the Order is synchronized with the time requirements for site response.

If the conference does not result in private party response--or if changing conditions at the site require accelerated response--the Fund-financed immediate removal will take place. If Fund-financed activity does begin, the Order may be written to require the potentially responsible parties to undertake site activity at the next convenient break point in activity. If the parties still fail to undertake the site response activity, enforcement efforts will emphasize cost recovery with the additional imposition of fines/penalties as appropriate.

V. COST RECOVERY

The Agency will normally not initiate a civil action in the event of non-compliance with an Order but instead will seek to recover costs and damages after a Fund-financed response. Therefore, while enforcement personnel are carrying out the Administrative Order process, they should also be aware of the requirements for a successful cost recovery action. They must be able to document the following factors (some of which are the same ones necessary for the issuance of the Administrative Order itself).

- The need for the immediate removal (evidence of an imminent and substantial endangerment or threat of endangerment to public health, welfare or the environment)
- Liability of the responsible parties (evidence to support the contention that the parties meet the liability standard of \$107)
- 3. Proof that the Fund-financed response activity was "not inconsistent" with the requirements of the NCP.
- 4. Documentation of all eligible costs for site-specific Fund expenditures.

Enforcement personnel must assure sufficient documentation of these factors from the period in which the 10-point document is developed and Funds are obligated through the actual clean up of the site. These cost recovery requirements must be met regardless --of whether there will be a simple cost recovery action (if no Administrative Order is issued) or an action for response costs plus damages (if the Order is not complied with). The Agency must assure that evidence is preserved for any subsequent enforcement action. Proper chain of custody procedures must be used for any sampling or testing, and adequate records of site activity must be kept. Employees of any contractor used for site activity must cooperate with and be made available to the U.S. in preparation and trial of any subsequent enforcement action. Enforcement, program and legal offices should work together throughout the case development.

VI. FOLLOW-UP

This guidance represents a substantial departure from prior practice, and I expect that it will take some time to implement. For these reasons, I will be reviewing all immediate removals referred to Headquarters for compliance with this guidance. In addition, for immediate removals under \$250,000, I will ask the Directors, OWPE and OERR to review the compliance with this guidance guarterly, and to advise me accordingly.

Appendix

cc: Gene Lucero, CWPE William Hedeman, OERR Kirk Sniff, OECM Dan Berry, OGC

APPENDIX

Authority/Requirements/Enforcement of Administrative Orders for Removal Actions under CERCLA

Under §106(a) of CERCLA:

If, EFA, acting on behalf of the President:

determines that there may be an imminent and substantial endangerment to the public health or welfare or the environment because of

an actual or threatened release of a hazardous substance from a facility

may, after notice to the affected state,

issue such orders as may be necessary to protect public health and welfare and the environment.

Under §106(b) of CERCLA:

EPA may take action in the appropriate U.S. district court, against any person who willfully violates or fails or refuses to comply with any Order issued under \$106(a), to enforce such order and

may fine such person not more than \$5,000 for each day such violations occur or such failure to comply continues.

Under \$107(c)(3) of CERCLA:

Any person who is liable for a release or threat of release of a hazardous substance that:

fails without sufficient cause to properly provide removal action upon order of the President pursuant to \$106

may be liable to the United States for punitive damages in an amount at least equal to and not more than three times, the amount of any costs incurred by the Fund as a result of such failure to take proper action.

Any monies received in punitive damages shall be deposited in the Fund. National Contingency Plan Requirements for Immediate Removals

Under \$300.65 of the NCP:

Immediate Removal action is appropriate when the lead agency determines that:

• the initiation of the removal action will prevent or mitigate immediate and significant risk of harm to human life or health or to the environment from such situations as:

- 1. Human, animal, or food chain exposure to acutely toxic substances
- 2. Contamination of drinking water supply
- 3. Fire and/or explosion
 - 4. Similarly acute situations

Immediate removal action may include but are not limited to:

1. Collecting and analyzing samples to determine the source and dispersion of the hazardous substance

- 2. Providing alternative water supplies
- 3. Installing security fencing or other measures to limit access
- 4. Controlling the source of the release
- 5. Measuring and sampling

. .

- Moving hazardous substances off-site for storage, destruction, treatment or disposal
- 7. Placing physical barriers to deter the spread of the release
- 8. Controlling the water discharge from an upstream impoundment
- 9. Recommending to the appropriate authorities the evacuation of threatened individuals _
- 10. Using chemicals and other materials in accordance with Supart H to restrain the spread of the substance and mitigate its effects
- 11. Executing damage control or salvage operations

Procedures for Authorizing Application for Civil Search Warrants Under CERCLA





Under § 104(e) of CERCLA, as amended by SARA, the United States may seek access by warrant, administrative order, or court order. If access is obtained by administrative order, the appropriate documents are issued by relevant client agencies. If access is to be obtained by court order, then the Assistant Attorney General of the Land and Natural Resources Division must approve the complaint, upon referral from the relevant client agency according to ordinary procedures. For access to be sought through application on a civil CERCLA warrant,¹ the instant memorandum will confirm the procedures to be used by the Department of Justice.

Under $95.320-\lambda-2$ of the U.S. Attorney's Manual, application for warrant under CERCLA may not be handled unilaterally by the U.S. Attorneys. Applications for such warrants must be coordinated through the Environmental Enforcement Section.

Clearance through the Environmental Enforcement Section is important for a variety of reasons. First, the nature of the governmental activities involved under CERCLA civil warrants may be much broader and last considerably longer than an inspection under the other federal environmental regulatory statutes. Typically the latter require only a few days or weeks to conduct routine environmental sampling. Under CERCLA, access may be sought under a warrant for not only sampling, but even simple

¹ The memorandum does not cover procedures for seeking a <u>criminal</u> search warrant where a CERCLA violation may be involved. All such matters are to be referred to the Director, Environmental Crimes Unit, EES.

removal-type activity, e.g., security/fencing, limited drum removal. The greater relative complexity of the governmental activity involved can be expected to provoke more challenges to CERCLA civil warrants than those under other statutes and the issues raised by CERCLA warrants may be much more complex. Second, this is a relatively new and vital area of the law. We must ensure that maximum efforts are made to develop this critical area of the law in an excellent manner. EES lawyers must make all reasonable efforts to ensure that exercises of the civil warrant authority under CERCLA will be vindicated by the federal courts, through proper presentation of facts and legal arguments by Departmental attorneys with experience in this area. Finally, since our experience has shown that judicial challenges to civil CERCLA warrants tend to move very rapidly, sometimes on an emergency motion basis, EES needs to work closely with client agencies on these matters so that the Division's Appellate Section is advised and prepared with sufficient lead time to expeditiously address appellate proceedings.

Coordinating these warrant applications through EES must be done on an expedited basis so that client agencies' program objectives are achieved. Moreover, our resources must not be consumed by duplicative work. Balancing the needs for careful warrant application preparations with that for expeditious handling of these matters, we will use the following procedures:

1. The client agency will telephonically notify the relevant EES Assistant Chief or Senior Lawyer when the Agency plans to seek a civil warrant.

2. The client agency will follow-up the request by expeditiously transmitting a short memorandum concisely explaining why the warrant is needed with a draft copy of the warrant application and supporting affidavits.

3. Upon receipt of the telephonic notification or written request, whichever first occurs, the EES Assistant Chief or Sr. Lawyer will arrange for either an EES staff attorney or an AUSA to handle the review and prosecution of the application. Unless a dispute develops between EES/AUSA personnel and the client agency, the EES Assistant Chief or Sr. Lawyer may approve the application. If such a dispute develops, it must be brought to the attention of the Chief or Deputy Chief, EES for resolution. 4. Handling of these matters is to be afforded priority on our docket. Moreover, the Chief or Assistant Chief of the Appellate Section shall be advised of each application request by the EES Assistant Chief or Sr. Lawyer as soon as possible after notification by the client agency, so that Appellate can be prepared to handle expeditiously appeal matters.

5. All civil actions to <u>enforce</u> civil CERCLA warrants, by way of application for civil contempt or other judicial orders, shall be authorized in writing by the Assistant Attorney General. Such actions shall be afforded highest priority on the docket.

For general advice/guidance on handling CERCLA civil warrant matters, contact John Fleuchaus, ORCM-Waste, 382-3109.

Attachment

Entry and Continued Access Under CERCLA

•

•



JNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUN - 5 1987

OSWER DIRECTIVE

9829.2

OFFICE OF ENFORCEMENT AND COMPLIANCE MONITORING

MEMORANDUM

SUBJECT: Entry and Continued Access Under CERCLA

FROM: Thomas L. Adams, Jr. Assistant Administrator Showed h. Adama

TO: Regional Administrators I-X Regional Counsels I-X

I. INTRODUCTION

This memorandum sets forth EPA's policy on entry and continued access to facilities by EPA officers, employees, and representatives for the purposes of response and civil enforcement activities under CERCLA. 1/ In short, the policy recommends that EPA should, in the first instance, seek to obtain access through consent. Entry on consent is preferable across the full range of onsite activities. If consent is denied, EPA should use judicial process or an administrative order to gain access. The appropriate type of judicial process varies depending on the nature of the onsite activity. When entry is needed for short-term and non-intrusive activities, an <u>ex parte</u>, judicial warrant should be sought. In situations involving long-term or intrusive access, EPA should generally file suit to obtain a court order.

The memorandum's first section addresses the recently amended access provision in CERCLA. The memorandum then sets forth EPA policy on obtaining entry and the procedures which should be used to implement this policy, including separate discussions on consent, warrants, court orders, and administrative orders.

^{1/} This policy does not address information requests under Section 104(e)(2).

II. STATUTORY AUTHORITY

EPA needs access to private property to conduct investigations, studies, and cleanups. The Superfund Amendments and Reauthorization Act of 1986 (SARA) explicitly grants EPA 2/ the authority to enter property for each of these purposes. Section 104(e)(1) provides that entry is permitted for "determining the need for response, or choosing or taking any response action under this title, or otherwise enforcing the provisions of this title."

SARA also establishes a standard for when access may be sought and defines what property may be entered. EPA may exercise its entry authority "if there is a reasonable basis to believe there may be a release or threat of a release of a hazardous substance or pollutant or contaminant." \$ 104(e)(1). SARA, however, does not require that there be a release or threatened release on the property to be entered. 3/ Places and properties subject to entry under Section 104(e) include any place any hazardous substance may be or has been generated, stored, treated, disposed of, or transported from; any place a hazardous substance has or may have been released; any place which is or may be threatened by the release of a hazardous substance; or any place where entry is needed to determine the need for response or the appropriate response, or to effectuate a response action under CERCLA. \$ 104(e)(3). EPA is also authorized to enter any place or property adjacent to the places and properties described in the previous sentence. § 104(e)(1).

EPA is granted explicit power to enforce its entry authority in Section 104(e)(5). Under that provision EPA may either issue an administrative order directing compliance with an entry request or proceed immediately to federal district court for injunctive relief. Orders may be issued where consent to entry is denied. Prior to the effective date of the order, EPA must provide such notice and opportunity for consultation as is reasonably appropriate under the circumstances. If EPA issues an order, the order can be enforced in court. Where there is a "reasonable basis to believe there may be a release or threat of a release of a hazardous substance or pollutant or contaminant," courts are instructed to enforce an EPA request or order unless the EPA

2/ Although CERCLA and SARA confer authority upon the President that authority has been delegated to the EPA Administrator. Exec. Order No. 12580, § 2(g) and (i), 52 Fed. Reg. 1923 (1987).

3/ The House Energy and Commerce bill at one point contained this limitation. H.R. Rep. No. 99-253 Part 1, 99th Cong., 1st Sess., 158 (1985). This limitation, however, was dropped prior to introduction of the bill for floor debate. See H.R. 2817, 99th Cong., 1st Sess., 131 Cong. Rec. H10857 (December 4, 1985). "demand for entry or inspection is arbitrary and capricious, an abuse of discretion, or otherwise not in accordance with law." § 104(e)(5). The legislative history makes clear that courts should enforce an EPA demand or order for entry if EPA's finding that there is a reasonable basis to believe there may be a release or threat of release is not arbitrary and capricious. 132 Cong. Rec. S14929 (October 3, 1986) (Statement of Sen. Thurmond); 132 Cong. Rec. H9582 (October 8, 1986) (Statement of Rep. Glickman). See United States v. Standard Equipment, Inc., No. C83-252M (W.D. Wash. November 3, 1986). In addition, a penalty not to exceed \$25,000/day may be assessed by the court for failure to comply with an EPA order or the provisions of subsection (e).

Finally, Section 104(e)(6) contains a savings provision which preserves EPA's power to secure access in "any lawful manner." This broad savings provision is significant coming in the wake of the Supreme Court's holding that:

> When Congress invests an agency with enforcement and investigatory authority, it is not necessary to identify explicitly each and every technique that may be used in the course of executing the statutory mission.

. . . Regulatory or enforcement authority generally carries with it all the modes of inquiry and investigation traditionally employed or useful to execute the authority granted.

Dow Chemical Co. v. United States, 90 L.Ed. 2d 226, 234 (1986). <u>4</u>/ One lawful means of gaining access covered by this paragraph is use of judicially-issued warrants. <u>See</u> S. Rep. No. 99-11, 99th Cong. 1st Sess. 26 (1985).

In numerous instances prior to the passage of SARA, EPA obtained court rulings affirming its authority to enter property to conduct CERCLA activities. 5/ Following enactment of SARA,

^{4/} See also, Mobil Oil Corp. v. EPA, 716 F.2d 1187, 1189 (7th Cir. 1983), cert. denied, 466 U.S. 980 (1984) (EPA authority to sample effluent under Section 308 of the Clean Water Act broadly construed); CEDs, Inc. v. EPA, 745 F.2d 1092 (7th Cir. 1984), cert. denied, 471 U.S. 1015 (1985).

^{5/} United States v. Pepper Steel and Alloy, Inc., No. 83-1717-CIV-EPS (S.D. Fla. October 10, 1986); Bunker Limited Partnership v. United States, No. 85-3133 (D. Idaho October 21, 1985); United States v. Coleman Evans Wood Preserving Co., No. 85-211-CIV-J-16 (M.D. Fla. June 10, 1985); United States v. Baird & McGuire Co. No. 83-3002-Y (D. Mass. May 2, 1985); United States v. United Nuclear Corp., 22 ERC 1791, 15 ELR 20443 (D.N.M. April 18, 1985).

several courts have ordered siteowners to permit EPA access. <u>United States</u> v. <u>Long</u>, No. C-1-87-167 (S.D. Ohio May 13, 1987); <u>United States</u> v. <u>Dickerson</u>, No. 84-76-VAL (M.D. Ga. May 4, 1987); <u>United States</u> v. <u>Standard Equipment</u>, Inc., No. C83-252M (W.D. Wash. Nov. 3, 1986). Further, the one adverse ruling on EPA's right of access has been vacated by the Supreme Court. <u>Outboard</u> <u>Marine Corp. v. Thomas</u>, 773 F.2d 883 (7th Cir. 1985), <u>vacated</u>, 93 L. Ed. 2d 695 (1986).

III. EPA ACCESS POLICY

EPA needs access to sites for several types of activities, including:

- preliminary site investigations;
- ° removal actions;
- ° RI/FSs; and
- ° remedial actions.

Within each of these categories, the scope of the work and the time needed to complete that work may vary substantially. This memorandum sets Agency policy on what means should be used to gain access over the range of these various activities.

EPA may seek access through consent, warrant, administrative order, or court order. Consent is the preferred means of gaining access for all activities because it is consistent with EPA policy of seeking voluntary cooperation from responsible parties and the public. In certain circumstances, however, the Region should consider obtaining judicial authorization or issuing an administrative order in addition to obtaining consent. For example, where uncertainty exists whether a siteowner will continue to permit access over an extended period, reliance on consent alone may result in a substantial delay if that consent is withdrawn.

When consent is denied, EPA should seek judicial authorization or should issue an administrative order. If the judicial ---route is chosen, EPA may seek an <u>ex parte</u> warrant or a court order. Warrants are traditionally granted for short-term entries. Generally, warrants should not be used when the EPA access will involve long-term occupation or highly intrusive activities. Clearly, warrants are appropriate for preliminary site investigations. On the other hand, because of the long, involved nature of remedial actions, access for such projects should be sought through a request for a court order. Neither removals nor RI/FSs, however, can be rigidly matched with a given judicial access procedure. Depending on the activities to be undertaken and the circumstances at the site, either a warrant or a court order may be appropriate. In deciding whether to use a warrant or a court order when access is needed for a removal or to conduct a RI/FS, the following general principles should be considered. First, if the activity will take longer than 60 days a court order normally is appropriate. Second, even if the activity will take less than 60 days, when the entry involves removal of large quantities of soil or destruction of permanent fixtures, a court order may again be appropriate. Finally, warrants should not be used if EPA action will substantially interfere with the operation of onsite business activities. These issues must be resolved on a case-by-case basis.

If EPA needs to gain access for a responsible party who has agreed to undertake cleanup activities under an administrative order or judicial decree, EPA may, in appropriate circumstances, designate the responsible party as EPA's authorized representative solely for the purpose of access, and exercise the authorities contained in Section 104(e) on behalf of the responsible party. Such a procedure may only be used where the responsible party demonstrates to EPA's satisfaction that it has made best efforts to obtain access. A further condition on the use of this procedure is that the responsible party agree to indemnify and hold harmless EPA and the United States for all claims related to injuries and damages caused by acts or omissions of the responsible party. The responsible party should also be advised that the expenses incurred by the government in gaining access for the responsible party are response costs for which the responsible party is liable. Before designating any responsible party as an authorized representative, the Region should consult with the Office of Enforcement and Compliance Monitoring.

IV. ACCESS PROCEDURES

- A. Entry on Consent
 - 1. General Procedures

The following procedures should be observed in seeking consent:

<u>Initial Contact</u>. Prior to visiting a site, EPA personnel <u>6</u>/ should consider contacting the siteowner to determine if consent will be forthcoming. EPA personnel should use this opportunity to explain EPA's access authority, the purpose for which entry is needed, and the activities which will be conducted.

^{6/} As used in this guidance, the term "EPA personnel" includes contractors acting as EPA's authorized representatives.

<u>Arrival</u>. EPA personnel should arrive at the site at a reasonable time of day under the circumstances. In most instances this will mean during normal working hours. When there is a demonstrable need to enter a site at other times, however, arrival need not be limited to this timeframe. Entry must be reasonable given the exigencies of the situation.

Identification. EPA personnel should show proper identification upon arrival.

<u>Request for Entry</u>. In asking for consent, EPA personnel should state the purpose for which entry is sought and describe the activities to be conducted. EPA personnel should also present a date-stamped written request to the owner or person-in-charge. A copy of this request should be retained by EPA. Consent to entry must be sought from the owner 7/ or the person-in-charge at that time.

If practicable under the circumstances, consent to entry should be memorialized in writing. A sample consent form is attached. Although oral consents are routinely approved by the courts, a signed consent form protects the Agency by serving as a permanent record of a transaction which may be raised as a defense or in a claim for damages many years later. If a siteowner is unwilling to sign a consent form but nonetheless orally agrees to allow access, EPA should document this oral consent by a follow-up letter confirming the consent.

Since EPA contractors often are involved in gaining access in the first instance, the Regions should ensure that their contractors are acquainted with these procedures.

2. Denial of Entry

If consent is denied, EPA personnel or contractors, before leaving, should attempt to determine the grounds for the denial. EPA personnel, however, should not threaten the siteowner with penalties or other monetary liability or make any other remarks which could be construed as threatening. EPA personnel may explain EPA's statutory access authority, the grounds upon which this authority may be exercised, and that the authority may be enforced in court.

^{7/} If EPA's planned site activities will not have a physical effect on the property, EPA generally need not seek consent from the owner of leased property where the lessee is in possession. The proper person in those circumstances is the lessee. But where EPA entry will have a substantial physical effect on the property, both the lessee and the property-owner should be contacted since in this instance interests of both will be involved.
3. Conditions Upon Entry

Persons on whose property EPA wishes to enter often attempt to place conditions upon entry. EPA personnel should not agree to conditions which restrict or impede the manner or extent of an inspection or response action, impose indemnity or compensatory obligations on EPA, or operate as a release of liability. The imposition of conditions of this nature on entry should be treated as denial of consent and a warrant or order should be obtained. See U.S. EPA, General Counsel Opinions, "Visitors' Release and Hold Harmless Agreements as a Condition to Entry of EPA Employees on Industrial Facilities," Gen'l and Admin. at 125 (11/8/72). If persons are concerned about confidentiality, they should be made aware that business secrets are protected by the statute and Agency regulations. 42 U.S.C. § 9604(e); 40 C.F.R. § 2.203(b). EPA personnel should enter into no further agreements regarding confidentiality.

- B. Warrants
 - 1. General Procedures

To secure a warrant, the following procedures should be observed:

<u>Contact Regional Counsel</u>. EPA personnel should discuss with Regional Counsel the facts regarding the denial of consent or other factors justifying a warrant and the circumstances which give rise to the need for entry.

Contact Department of Justice. If after consultation with Regional Counsel a decision is made to seek a warrant, the Regional Counsel must contact directly the Environmental Enforcement Section in the Land and Natural Resources Division at the Department of Justice. 8/ The person to call at the Department is the Assistant Chief in the Environmental Enforcement Section assigned to the Region. The Assistant Chief will then arrange, in a timely manner, for the matter to be handled by either an Environmental Enforcement Section attorney or a U.S. Attorney. The Region must send to the Environmental Enforcement Section, by Magnafax or other

^{8/} This procedure is necessary to comply with internal Department of Justice delegations of authority. Referral to a local U.S. Attorney's office is not sufficient for CERCLA warrants. The Environmental Enforcement Section of the Department of Justice must approve all warrant applications. (See Memorandum from David T. Buente, Jr. to All Environmental Enforcement Attorneys, "Procedures for Authorizing Applications for Civil Search Warrants Under CERCLA" (4/3/87) attached).

expedited means, a draft warrant application and a short memorandum concisely stating why the warrant is needed.

<u>Prepare Warrant Application</u>. The warrant application must contain the following:

- a statement of EPA's authority to inspect; (see \$ II, supra)
- a clear identification of the name and location of the site and, if known, the name(s) of the owner and operator of the site;
- 3) a statement explaining the grounds for a finding of a reasonable basis for entry (i.e., a reasonable basis to believe that there may be a release or threatened release of a hazardous substance or pollutant or contaminant) and the purpose for entry (i.e., determining the need for response, or choosing or taking any response action, or otherwise enforcing CERCLA);
- affidavits supporting the asserted reasonable basis for entry and describing any attempts to gain access on consent, if applicable; and
- 5) a specific description of the extent, nature, and timing of the inspection;

Following preparation of the warrant application, the Justice Department attorney will file the application with the local U.S. Magistrate.

EPA may ask the Justice Department attorney to seek the assistance of the United States Marshals Service in executing the warrant where EPA perceives a danger to the personnel executing the warrant or where there is the possibility that evidence will be destroyed.

2. Reasonable Basis for Entry

A warrant for access on a civil matter may be obtained upon a showing of a reasonable basis for entry. This reasonable basis may be established either by presenting specific evidence relating to the facility to be entered or by demonstrating that the entry is part of a neutral administrative inspection plan.

A specific evidence standard is incorporated in SARA as a condition on EPA's exercise of its access authority: EPA must have "a reasonable basis to believe there may be a release or threat of a release of a hazardous substance or pollutant or contaminant." § 104(e)(1). SARA's express specific evidence standard is consistent with how courts have formulated the specific evidence test in the absence of statutory guidance. E.g., West Point-Pepperell, Inc. v. Donovan, 689 F. 2d 950, 958 (11th Cir. 1982) (there must be a "showing of specific evidence sufficient to support a reasonable suspicion of a violation").

In drafting a warrant application, conclusory allegations regarding the specific evidence standard under subsection 104(e)will not suffice. Courts generally have refused to approve warrants where the application contains mere boilerplate assertions of statutory violations. Warrant applications have been granted, on the other hand, where the application contained detailed attestations by government officials or third-party complaints which have some indicia of reliability. Ideally. EPA warrant applications should contain an affidavit of a person who has personally observed conditions which indicate that there may be a release or threat of a release of a hazardous substance. If they are available, sampling results, although not required, should also be attached. Warrant applications based on citizen, employee, or competitor complaints should include details that establish the complainant's credibility. 9/

C. Court Orders

The provisions in CERCLA authorizing EPA access may be enforced by court order. To obtain a court order for entry, the Region should follow the normal referral process. If only access is required, the referral package can obviously be much abbreviated. If timing is critical, EPA HQ will move expeditiously and will refer the case orally if necessary. The Regions, however, should attempt to anticipate the sites at which access may prove problematic and should allow sufficient lead time for the referral process and the operation of the courts. The Regions should also not enter lengthy negotiations with landowners over access. EPA and DOJ are prepared to litigate aggressively to establish EPA's right of access.

^{9/} If information gathered in a civil investigation suggests that a criminal violation may have occurred, EPA personnel should consult the guidance on parallel proceedings. (Memorandum from Courtney Price to Assistant Administrators et al., "Policy and Procedures on Parallel Proceedings at the Environmental Protection Agency" (1/23/84)). Use of CERCLA's informationgathering authority in criminal investigations is addressed in separate guidance. (Memorandum from Courtney M. Price to Assistant Administrators et al., "The Use of Administrative Discovery Devices in the Development of Cases Assigned to the Office of Criminal Investigations" (2/16/84)).

Prior to seeking a court order, EPA should request access, generally in writing, and assemble the record related to access. The showing necessary to obtain a court order is the same as for obtaining a warrant: EPA must show a reasonable basis to believe that there may be a release or a threat of a release of a hazardous substance or pollutant or contaminant. An EPA finding on whether there is reason to believe a release has occurred or is about to occur must be reviewed on the arbitrary and capricious standard. § 104(e)(5) (B)(i). If the matter is not already in court, EPA must file a complaint seeking injunctive and declaratory relief. Simultaneous to filing the complaint, EPA may, if necessary, file a motion, supported by affidavits documenting the release or threatened release, requesting an immediate order in aid of access. If the matter is already in litigation, EPA may proceed by motion to seek an order granting access. 10/

In a memorandum supporting EPA's request for relief it should be made clear that by invoking judicial process, EPA is not inviting judicial review of its decision to undertake response. action or of any administrative determinations with regard to the response action. Section 113(h) of SARA bars judicial review of removal or remedial action except in five enumerated circumstances. A judicial action to compel access is not one of the exceptions. Statements on the floor of the House and the Senate confirm that EPA enforcement of its access authority does not provide an opportunity for judicial review of response decisions. Senator Thurmond, chairman of the Judiciary Committee, remarked that when EPA requests a court to compel access "there is no jurisdiction at that time to review any response action . .

Parenthetically, it should be noted that the broad equitable 10/ power granted to courts in Section 106 can also be relied on to obtain a court order. An additional source of authority for courts in this regard is the All Writs Act, 28 U.S.C. \$ 1651. The Act authorizes federal courts to "issue all writs necessary or appropriate in aid of their respective jurisdictions . . . " 28 U.S.C. § 1651. This authority "extends under appropriate circumstances, to persons who, though not parties to the original... action or engaged in wrongdoing are in a position to frustrate the implementation of a court order " United States v. New York Telephone Co., 434 U.S. 159, 174 (1977). Thus, the All Writs Act may prove useful as a means of compelling persons not a party to a consent decree to cooperate with EPA and other settling parties in execution of the decree. The use of the All Writs Act, however, may be limited in light of the Supreme Court's interpretation of the Act in Pennsylvania Bureau of Correction v. United States Marshal Service, 88 L. Ed. 2d 189 (1985).

[T]he court may only review whether the Agency's conclusion that there is a release or threatened release of hazardous substances is arbitrary or capricious." 132 Cong. Rec. S14929 (October 3, 1986) (Statement of Sen. Thurmond); 132 Cong. Rec. 119582 (October 8, 1986) (Statement of Rep. Glickman); <u>see United States</u> v. <u>Standard Equipment, Inc.</u>, No. C83-252M (W.D. Wash. Nov. 3, 1986).

D. Administrative Orders

If a siteowner denies an EPA request for access, EPA may issue an adminstrative order directing compliance with the request. § 104(e)(5)(A). Each administrative order must include a finding by the Regional Administrator that there exists a reasonable belief that there may be a release or threat of release of a hazardous substance and a description of the purpose for the entry and of the activities to be conducted and their probable The order should indicate the nature of the prior duration. request for access. Further, the order should advise the respondent that the administrative record upon which the order was issued is available for review and that an EPA officer or employee will be available to confer with respondent prior to the effective The length of the time period during which date of the order. such a conferences may be requested should be reasonable under the circumstances. In deciding what is a reasonable time period, consideration should be given to the interference access will cause with onsite operations, the threat to human health and the environment posed by the site, and the extent of prior contacts with the respondent. The order should advise the respondent that penalties of up to \$25,000 per day may be assessed by a court against any party who unreasonably fails to comply with an order. \$ 104(e)(5). Following the time period for the conference and any conference. the issuing official should send a document to the respondent summarizing any conference. EPA's resolution of any objections. and stating the effective date of the order.

If, following issuance of an administrative order, the siteowner continues to refuse access to EPA, the order may be enforced in federal court. EPA should not use self-help to execute orders. Courts are required to enforce administrative orders where there is a reasonable basis to believe that there may be a release or threat of a release of a hazardous substance. EPA's determination in this regard must be upheld unless it is arbitrary and capricious. § 104(e)(5)(B)(i). EPA will seek penalties from those parties who unreasonably fail to comply with orders.

All administrative orders for access must be concurred on by the Office of Enforcement and Compliance Monitoring prior to issuance.

DISCLAIMER

The policies and procedures established in this document are intended solely for the guidance of government personnel. They are not intended, and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. The Agency reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

Attachments

CONSENT FOR ACCESS TO PROPERTY

Name:	
-------	--

Address of Property:

I consent to officers, employees, and authorized representatives of the United States Environmental Protection Agency (EPA) entering and having continued access to my property for the following purposes:

[the taking of such soil, water, and air samples as may be determined to be necessary;]

[the sampling of any solids or liquids stored or disposed of on site;]

[the drilling of holes and installation of monitoring wells for subsurface investigation;]

[other actions related to the investigation of surface or subsurface contamination;]

[the taking of a response action including . . .]

I realize that these actions by EPA are undertaken pursuant to its response and enforcement responsibilities under the Comprehensive Environmental Response, Compensation and Liability Act (Superfund), 42 U.S.C. § 9601 et seq.

This written permission is given by me voluntarily with knowlege of my right to refuse and without threats or promises of any kind.

Date

Signature

Interim Guidance on Notice Letters, Negotiations, and Information Exchange



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

983410

207 1 9 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Transmittal of Notice Letter Guidance

PROM: Gene Ducero, Director MUL WCWO Office of Waste Programs Enforcement

TO: Addressees

Attached is the "Interim Guidance on Notice Letters, Negotiations, and Information Exchange." Note that Appendix C containing model notice letters is not included in this package, but will be distributed under separate cover in the next couple of weeks.

Attachment

Addressees:

Directors, Waste Management Divisions, Regions I, IV, V, VII, VIII Director, Emergency and Remedial Response Division, Region II Director, Hazardous Waste Management Division, Region III Directors, Air and Waste Management Division, Regions II, VI Director, Toxics and Waste Management Division, Region IX Director, Hazardous Waste Division, Region X

9834,10

INTERIM GUIDANCE ON NOTICE LETTERS, NEGOTIATIONS, AND INFORMATION EXCHANGE

Table of Contents

- I. Introduction
- II. Purpose and Scope of Guidance
- III. Statutory Authority
 - A. Settlements
 - Special Notice Procedures and Information Release B .
- IV. Information Exchange
 - A. Information Requests
 - Information Release **B**.
- Notice Letters and Negotiation Moratorium for RI/FS and ۷. RD/RA
 - A. Purpose of Notice Letters
 - B. General Notice Letter
 - Whether to Issue General Notice 1.
 - Timing of General Notice 2.
 - 3. Recipients of General Notice
 - Contents of General Notice 4.
 - c. RI/FS and RD/RA Special Notice Letters
 - Whether to Issue RI/FS and RD/RA Special Notice 1.
 - Notifying PRPs When Not Appropriate to Issue 2.
 - RI/FS and RD/RA Special Notice 3. DOJ Role in RI/FS and RD/RA Negotiations
 - Timing of RI/FS Special Notice 4.
 - Timing of RD/RA Special Notice
 - 5.
 - 6. Recipients of RI/FS and RD/RA Special Notice 7. Contents of RI/FS and RD/RA Special Notice
 - Conclusion of Negotiation Moratorium and Deadline D. Management for RI/FS and RD/RA
- VI. Notice Letters and Negotiation Moratorium for Removal Actions
 - λ. Notice Letters
 - 1. Whether to Issue Notice for Removals
 - 2. When to Use Special Notice Procedures for Removals

- 3. Notifying PRPs When Not Appropriate To Utilize Special Notice Procedures for Removals
- 4. DOJ Role in Removal Negotiations
- 5. Timing of Notice for Removals
- 6. Recipients of Notice for Removals
- 7. Contents of Notice for Removals
- B. Conclusion of Negotiation Moratorium and Deadline Management for Removals
- C. Administrative Orders and Negotiation Moratorium for Removals
- VII. Disclaimer
- VIII. For Further Information

Appendices

Appendix	λ:	Timing of RD/RA Special Notice Letter
Appendix	B :	Settlement Process Timelines
Appendix	C:	Model Notice Letters (To be provided under
		separate cover)

•



OCT 1 9 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Interim Guidance on Notice Letters, Negotiations, and Information Exchange

FROM: J. Winston Porter Assistant Administrator

TO: Regional Administrators

I. INTRODUCTION

The Superfund Amendments and Reauthorization Act of 1986 (SARA), which amends the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA), maintains the importance of a strong Superfund enforcement program.¹ In particular, SARA emphasizes the importance of entering into negotiations and reaching settlements with potentially responsible parties (PRPs) to allow PRPs to conduct or finance response actions. SARA generally codified the Agency's Interim CERCLA Settlement Policy but also established some new authorities and procedures that were designed to facilitate settlements.

A fundamental goal of the CERCLA enforcement program is to facilitate voluntary settlements. EPA believes that such settlements are most likely to occur when EPA interacts frequently with PRPs. Frequent interaction is important because it provides the opportunity to share information about a site and may reduce delays in conducting response actions caused by the lack of communication. Important mechanisms for promoting interaction and facilitating communication between EPA and PRPs include issuing notice letters, entering into negotiations, and exchanging information with PRPs.

¹ CERCLA of 1980 as amended by SARA of 1986 is referred to in this guidance as CERCLA.

This guidance replaces the October 12, 1984 guidance on "Procedures for Issuing Notice Letters" and the October 9, 1985 guidance on "Timely Initiation of Responsible Party Searches, Issuance of Notice Letters, and Release of Information." ³ Although certain procedures and the timing of various activities have been modified, this guidance retains many fundamental aspects of the October 12, 1984 and October 9, 1985 guidances. In particular, this guidance re-emphasizes the importance of timely issuance of notice letters and the exchange of information between EPA and PRPs. In addition, this guidance incorporates a moratorium and "formal" period of negotiation (referred to as a negotiation moratorium) into the settlement process. EPA's commitment to carrying out these activities is crucial for supporting our fundamental goal of facilitating negotiated settlements.

II. PURPOSE AND SCOPE OF GUIDANCE

The purpose of this guidance is to assist the Regions in establishing procedures for the issuance of notice letters to PRPs, for the conduct of negotiations between EPA and PRPs, and for the exchange of information between EPA and PRPs.

This guidance addresses the use of both "general" and "special" notice letters for removal and remedial actions. Special notice letters differ from general notice letters because special notices trigger the negotiation moratorium. The negotiation moratorium is the period of time where a moratorium is imposed on certain EPA actions and a period of "formal" negotiations is established between EPA and PRPs.

Use of both general and special notice letters are discretionary. However, the Regions are expected to issue general and special notices for the vast majority of remedial actions. Such notice letters will be issued for remedial investigations/feasibility studies (RI/FSs) and remedial designs/remedial actions (RD/RAs). Although it is generally appropriate to issue a "removal notice" for all removal actions, the Regions are not expected to invoke the \$122(e) special notice procedures for most removals.

This guidance also addresses the timing, duration, and conclusion of the negotiation moratorium. Finally, this guidance discusses the process of information exchange between EPA and PRPs, including requests for and releases of site-specific information.

These guidances were issued under OSWER Directive Numbers 9834.1 and 9834.2, respectively.

III. STATUTORY AUTHORITY

λ. SETTLEMENTS

Sections 104(a), 122(a), and 122(e)(6) authorize settlements and establish certain conditions for allowing PRPs to conduct or finance response actions. Section 104(a) authorizes EPA to enter into an agreement with PRPs to allow PRPs to conduct or finance response actions in accordance with \$122 if EPA determines that the PRPs will conduct the response action properly and promptly. Under \$104'a), PRPs cannot conduct the RI/FS unless EPA determines that the PRP is qualified to perform the RI/FS, EPA contracts with or arranges for a qualified person other than the PRP to assist EPA in overseeing and reviewing the RI/FS, and the PRP agrees to reimburse the Fund for the costs EPA incurs in overseeing and reviewing the PRP's RI/FS.

Section 122(a) similarly authorizes EPA to enter into agreements with PRPs to perform response actions if EPA determines the action will be conducted properly. Section 122(a) also provides for EPA, when practicable and in the public interest, to facilitate settlements with PRPs to expedite effective remedial actions and to minimize litigation.

Section 122(e)(6) provides that no PRP may undertake any remedial action at a facility where EPA or a PRP pursuant to an administrative order or consent decree under CERCLA has initiated an RI/FS unless the remedial: action has been authorized by EPA.

B. SPECIAL NOTICE PROCEDURES AND INFORMATION RELEASE

Sections 122(e) and 122(a) contain provisions relating to the special notice procedures and the release of information to PRPs. Section 122(e) provides for EPA to utilize the special notice procedures if EPA determines that a period of negotiation would facilitate an agreement with PRPs and would expedite remedial actions. Section 122(e) also provides for EPA to release certain information to PRPs. Such information includes, to the extent available, the names and addresses of other PRPs, the volume and nature of substances contributed by each PRP, and a ranking by volume of the substances at the facility.³ In

³ Congress recognized that there may be limitations to the availability of information at early phases of the response action. In particular, Congress noted that the RI/FS special notice need not be accompanied by information on volume and nature of waste and ranking if this information is not available at the start of the RI/FS. A separate notice and information release should be provided for private parties who actually conduct the remedial action and information on volume, nature and ranking of wastes should be made available routinely at this

addition, this section provides for EPA to make such information available in advance of the special notice upon request by a PRP in accordance with procedures provided by EPA.

Issuance of a special notice triggers a moratorium on the commencement of certain actions by EPA under \$104 or \$106. The purpose of the moratorium is to provide for a period of negotiation between EPA and PRPs. The moratorium prohibits EPA from commencing any response action under \$104(a), and an RI/FS under \$104(b), or an action under \$106 for 60 days after receipt of the notice. If EPA determines that a "good faith offer" has been submitted by the PRP within 60 days after receipt of the special notice. EPA shall not commence an action under \$104(a) or take any action against any person under \$106 for an additional 60 days or commence an RI/FS under \$104(b) for an additional 30 days.

Under \$122(e)(2)(a), EPA may commence any additional other studies or investigations authorized under \$104(b), including the remedial design, during the negotiation period. Under \$122(e)(2)(C), if an additional PRP is identified during the negotiation period or after an agreement has been entered into, EPA may bring the additional party into the negotiation or may enter into a separate agreement with the PRP. Under \$122(e)(5), EPA is not prohibited from undertaking a response or enforcement action during the negotiation period when there is a significant threat to public health or the environment.

Section 122(a) provides that if EPA decides not to use the special notice procedures established under \$122(e), EPA is required to notify PRPs in writing of this decision along with an explanation why it is inappropriate to use such procedures. The decision by EPA to use or not to use the special notice procedures is not subject to judicial review.

IV. INFORMATION EXCHANGE

The exchange of information between EPA and PRPs is crucial for facilitating settlements. Information exchange should be an ongoing process of communication. EPA uses information obtained from PRPs to determine potential liability, to determine the need for response, and to support the selection of the remedy. PRPs use information obtained from EPA to organize among themselves and to develop a "good faith offer" to conduct or finance response actions.

time. See the Conference Report on the Superfund Amendments and Reauthorization λ ct of 1986, 99 Cong., 2d Sess. Report 99-962 pp. 253 (1986).

5

A. INFORMATION REQUESTS

EPA may request information from PRPs about various activities and conditions under \$104(e) of CERCLA and under \$3007(a) of the Resource Conservation and Recovery Act (RCRA). In addition, EPA may issue administrative subpoenas under \$122(e)(3)(b) of CERCLA. Information commonly requested includes details concerning waste operations and waste management practices, the type and amount of substances contributed by each PRP, as well as the name of other PRPs that contributed substances to the site.

Information requests should be issued as early as practicable and may be issued as a separate letter during the PRP search process, as part of the general notice letter, or through an administrative subpoena. A detailed discussion about the use of information request letters and administrative subpoenas is contained in the forthcoming "Guidance on Use and Enforcement of Information Requests and Administrative Subpoenas under CERCLA \$\$104(e) and 122(e)."

The Regions have the discretion to decide whether to issue an information request as a separate letter during the PRP search or as a component of a general notice letter. Issuing a separate information request letter in advance of the general notice may be advantageous in situations where information from PRPs is needed to determine whether it is appropriate to issue a notice letter to such parties.

Information requests should be developed in accordance with the forthcoming guidance on information requests and administrative subpoenas as mentioned above. An information request should also indicate that EPA plans to vigorously enforce information requests with the new enforcement tools authorized under SARA which include issuing orders under \$104(e)(5). Finally, the information request should indicate that it is the PRPs responsibility to inform EPA whether information they provide to EPA is confidential and subject to protection under \$104(e) of CERCLA.

B. INFORMATION RELEASE

It is important to gather and release site-specific information to PRPs as soon as reasonably practicable. Gathering and releasing such information early in the process will not only expedite response and enforcement activities but will help PRPs organize and negotiate among themselves as well.

As indicated, \$122(e)(1) provides for the release of certain information to PRPs to the extent such information is available. Such information includes the names and addresses of other PRPs, the volume and nature of substances contributed by each PRP, and a ranking by volume of the substances at the facility. This information is to be provided to PRPs in advance of the special notice in accordance with procedures developed by EPA.

Congress recognized the limitations to EPA's ability to make certain information available to PRPs, especially early in the response process. Therefore, this information can be released only to the extent such information is available. If the Regions have information on volume, the Regions should develop volumetric rankings and should make such information available to PRPs as soon as practicable. However, due to their preliminary and summary nature, EPA will not expend resources to explain or defend any list or ranking. Lists or rankings released to PRPs and others should always contain appropriate disclaimers.

The Regions are encouraged to release information to PRPs as soon as reasonably possible. The Regions may respond directly to individual PRP requests for information, may use the notice letters as vehicles to release such information to PRPs, or may establish alternative mechanisms in some situations as discussed below. The Regions are strongly encouraged to use the notice letters to release site-specific information. In particular, use of the general notice may provide a convenient opportunity to release information in advance of the special notice pursuant to the statutory provision that EPA release such information in advance of the special notice in accordance with procedures developed by EPA.

Although it is generally preferable to release information to individual PRPs through notice letters, alternative mechanisms may be used in unusual circumstances. For example, in instances where there are many PRPs and/or where there is a substantial amount of information to be released, the Regions may consider making the information available through a central mechanism (e.g. through a PRP steering committee if one has been formed and if the committee has agreed to be a clearinghouse for distributing information to other PRPs). An alternative would be to indicate in the notice letter that the Region has sitespecific information that will be made available to the PRPs in a manner specified in the letter.

V. NOTICE LETTERS AND NEGOTIATION MORATORIUM FOR RI/FS AND RD/RA

This guidance creates a systematic process for issuing three separate notice letters for remedial actions. The three notice letters are 1) the general notice, 2) the RI/FS special notice, and 3) the RD/RA special notice. Even though the RI/FS and RD/RA special notice letters are separate letters, they are discussed in the same section below since the content of these letters is basically the same. In instances where the content of the RI/FS and RD/RA special notices differ, separate sections are presented. Also, this guidance is written with the assumption that each notice letter will be issued in sequence. Consequently, the guidance has been structured so that certain information provided or requested in one letter is not repeated in a subsequent letter. The content of actual letters may, however, need to be modified in situations where this process is not followed.

For example, there may be a situation where site activities are already underway and where the Region is ready to issue the RI/FS special notice but has not issued a general notice. In this instance, it would not be necessary to wait to send the special notice until after a general notice is issued. However, it may be appropriate to include certain aspects of the general notice into the special notice.

A. <u>PURPOSE OF NOTICE LETTERS</u>

The purpose of the general notice is to inform PRPs of their potential liability for future response costs, to begin or continue the process of information exchange, and to initiate the process of "informal" negotiations. In addition, the general notice informs PRPs about the possible use of the \$122(e) special notice procedures and the subsequent moratorium and "formal" negotiation period.

The purpose of the special notice is similar to the general notice, except that the special notice is also used to invoke the statutory moratorium on certain EPA actions and to initiate the process of "formal" negotiations. Although the general notice does not trigger a moratorium on any EPA action and does not invoke a "formal" period of negotiation, the general notice is expected to initiate a dialogue between EPA and PRPs. Issuance of a general notice should be viewed as a mechanism for initiating negotiations whereas issuance of a special notice should be viewed as a mechanism for concluding negotiations.

The term "informal" negotiations does not mean that such negotiations are not serious efforts to reach a settlement. Rather "informal" negotiations refers to any negotiations that are not conducted as part of the negotiation moratorium triggered by issuance of a special notice under \$122(a). The terms "informal" and "formal" negotiations are used to draw a distinction between negotiations which are and are not covered by the \$122(e) moratorium.

B. <u>GENERAL NOTICE LETTER</u>

Agency notification procedures should provide PRPs with sufficient time to organize and develop a reasonable offer to conduct or finance the response action. Toward this end, the Regions should contact PRPs prior to issuing a \$122(e) special notice by issuing a general notice letter.

1. Whether to Issue General Notice

A general notice letter should be issued at the vast majority of sites that are proposed for or listed on the National Priorities List (NPL) where negotiations for the RI/FS and RD/RA have not yet been initiated. Circumstances where it may not be appropriate to issue the general notice include sites where a notice pursuant to previous guidance was issued prior to the reauthorization of CERCLA or where the Region is ready to issue a special notice at the site. These exceptions are important for minimizing any possible disruption to ongoing activities.

2. <u>Timing of General Notice</u>

The general notice letter should be sent to PRPs as early in the process as possible, preferably once the site has been proposed for inclusion on the NPL. Early receipt of the general notice will ensure that PRPs have adequate knowledge of their potential liability as well as a realistic opportunity to participate in settlement negotiations. When a separate information request letter has been sent to PRPs prior to the general notice, the information request should be sent as early as possible to avoid any delay in issuing the general notice.

3. <u>Recipients of General Notice</u>

General notice letters should be sent to all parties where there is sufficient evidence to make a preliminary determination of potential liability under \$107 of CERCLA. If there is doubt about whether available information supports issuance of the general notice, separate information request letters may be sent to such parties prior to issuing the notice. If a Federal agency has been identified as a generator at a facility not owned/operated by the Federal agency, such agency should be routinely notified like other PRPs.

If additional PRPs are identified after the general notice but before the RI/FS special notice is issued, the Regions should provide a general notice to those additional PRPs. If additional PRPs are identified after general and special notices are issued, the additional PRPs need not receive a general notice before receiving the appropriate special notice. However, relevant aspects of the general notice should be incorporated into the special notice.

Copies of the general notice should be provided to the Regional administrative record coordinator, the appropriate State

representative, the State or Federal trustee if a trustee for natural resources has been designated, and to EPA headquarters at the same time notices are sent to PRPs. The copies of notices to headquarters should be sent to the Information Management Section within the Program Management and Support Office of the Office of Waste Programs Enforcement (OWPE).

Providing copies to the administrative record coordinator is important for ensuring that the notice is placed in the administrative record.⁴ Providing copies to the State representative and the State or Federal trustee is important for ensuring that States are appropriately informed about possible future negotiations.⁵ Providing copies to OWPE is essential for permitting entry into the Superfund Enforcement Tracking System (SETS). Entry into sets will facilitate our efforts to track site activities and to respond to Congressional and other inquiries. Direct Regional input of data into SETS on notice letter recipients is planned for FY 1988.

It is not necessary to provide copies of each general notice to the administrative record coordinator, State representative, State or Federal trustee, or headquarters in instances where identical notices are provided to multiple PRPs. Where there are multiple PRPs at a site, a copy of one general notice with a list of other parties who have received the letter would suffice.

4. <u>Contents of General Notice</u>

The general notice letter should contain the following components: a) a notification of potential liability for response costs, b) a discussion about future notices and the possible future use of special notice procedures, c) a general discussion about site response activities, d) a request for information about the site (if appropriate), e) the release of certain site-specific information (where available), f) a discussion about the merits of forming a PRP steering committee, g) a notice regarding the development of an administrative record, and h) a deadline for response to the letter and information on the EPA representative to contact.

⁴ A discussion about placing notice letters in the administrative record is covered in the forthcoming "Guidance on the Administrative Record for Selecting a Response Action Under CERCLA" and in the preamble to the forthcoming revisions to the National Contingency Plan.

⁵ State participation in negotiations is covered in the forthcoming "Interim Guidance on EPA-State Relations in CERCLA Enforcement."

a. <u>Potential liability</u>: The letter should inform parties that they are potentially liable for response costs under \$107 of CERCLA, including the costs of conducting the RI/FS and RD/RA. The letter should define the scope of potential liability and should briefly explain why the parties have been identified as PRPs.

b. <u>Future notice under 5122(a) or 5122(e)</u>: The letter should indicate that EPA will notify the party at an appropriate point in the future. The letter should specify that this notice will either be a 5122(a) notice or a 5122(e) special notice and should explain what these notices are.

The letter should indicate that the \$122(a) notice is a notice which informs parties that EPA will not use the \$122(e) special notice procedures. The letter should indicate that the notice will provide an explanation for the decision not to use the special notice procedures.

The letter should also indicate that a \$122(e) special notice will invoke the negotiation moratorium. The letter should make clear that issuance of a \$122(e) special notice letter is discretionary and may be used if EPA determines that use of such procedures would facilitate an agreement and expedite remedial action. The letter should also explain the purpose of the special notice and the subsequent negotiation moratorium. Informing PRPs about the special notice procedures and the negotiation moratorium will alert PRPs to possible future negotiations and increase their awareness of their opportunities for participation in such negotiations.

c. <u>Site response activities</u>: The letter should generally discuss the activities EPA plans to undertake at the site. Where appropriate, such activities should include scheduled start or completion dates for the RI/FS or RD/RA. Instances where it may not be appropriate to provide start or completion dates include situations where the general notice is issued very early in the process and where specific dates have not yet been set, or where it is expected that target dates are likely to change significantly.

d. <u>Information request</u>: The letter should request information on substances sent to or present at the site and the names of other PRPs pursuant to \$104(e) of CERCLA and/or \$3007(a) of RCRA if a separate information request has not already been issued. The content of the information request should be consistent with the forthcoming "Guidance on Use and Enforcement of Information Requests and Administrative Subpoenas Under CERCLA \$104(e) and \$122(e)." e. <u>Information release</u>: At a minimum, the letter should release the names and addresses of other PRPs who have received the general notice letter. In addition, to the extent such information is available, the letter should include the volume and nature of substances contributed by each PRP and a ranking by volume of the substances at the facility if such information has not been previously released.

f. <u>PRP steering committee</u>: The letter should request that the PRPs identify a member of their organization who will represent their interests. In addition, the letter should recommend that PRPs form a steering committee to represent the group's interests in possible future regotiations. The letter should indicate that establishing a steering committee is important for facilitating negotiations with EPA.

g. <u>Administrative record</u>: The letter should be used as a vehicle for informing PRPs of the availability of an administrative record that will contain documents which form the basis for the Agency's decision on the selection of remedy. The letter should indicate that the record will be open to the public for inspection and comment. The letter should also provide information regarding the opening of the record and where it will be located.

h. <u>PRP response and EPA contact</u>: The letter should encourage PRPs to notify EPA by a specified date of their interest to participate in future negotiations. The letter should indicate that PRPs may respond as a group through a steering committee if one has been formed. The letter should also provide a cut off date for voluntary compliance with information requests (if a request for information is contained in the general notice). An appropriate time frame for the PRP response to an information request is generally thirty days from receipt of the letter. Finally, the letter should provide the name, phone number, and address of the EPA representative to contact.

C. RI/FS and RD/RA SPECIAL NOTICE LETTERS

Prior to EPA's conduct of the RI/FS and RD/RA, the Regions should either issue the special notice to PRPs or provide PRPs with an explanation why it was not appropriate to use the special notice procedures. Issuance of the special notice triggers a moratorium on EPA's conduct of the RI/FS and remedial action. While the statute does not impose a moratorium on EPA's conduct of the remedial design, the Agency will not generally conduct such activities during the moratorium. The purpose of the moratorium is to provide for a formal period of negotiation between EPA and PRPs where the PRPs will be encouraged to conduct or finance response activities.

9834.10

The negotiation moratorium may last a total of 90 days for the RI/FS and 120 days for the RD/RA if EPA receives a "good faith offer" from PRPs within the first 60 days of the moratorium. The negotiation moratorium would conclude after 60 days if the PRPs do not provide EPA with a "good faith offer."

The initial 60 day moratorium begins on the date the PRPs receive the special notice via certified mail. In instances where there is more than one PRP and PRPs are likely to receive the special notice on different days, the date the moratorium begins should be seven days from the date the letters are mailed to the PRPs. In either case, the special notice must make clear when the negotiation moratorium begins and ends.

1. Whether to Issue RI/FS and RD/RA Special Notice

EPA has the discretion to use the special notice procedures when EPA determines that a period of negotiation would facilitate an agreement with PRPs and would expedite remedial actions. The Agency believes entering into such negotiations would generally facilitate settlements and plans to utilize the RI/FS and RD/RA special notice procedures in the wast majority of cases.

There are, however, some circumstances where it would generally not be appropriate to use such procedures. Such circumstances include 1) where past dealings with the PRPs strongly indicate they are unlikely to negotiate a settlement, 2) where EPA believes the PRPs have not been negotiating in good faith, 3) where no PRPs have been identified at the conclusion of the PRP search, 4) where PRPs lack the resources to conduct response activities, 5) where there are ongoing negotiations, or 6) where notice letters were already sent prior to the reauthorization of CERCLA and onging negotiations would not benefit by issuance of a special notice.

Special notices may be issued for operable units of remedial actions. The test for determining whether to issue a special notice for an operable unit is generally the same as for fullscale remedial actions. The general expectation is that separate special notices will be issued for each separate operable unit as long as issuing the notice would facilitate an agreement and would expedite the remedial action. However, special notices may also be issued for only major operable units or may cover a series of operable units if appropriate under the circumstances at the site.

For example, if several operable units will be conducted at a site as relatively separate and distinct response actions, it may be appropriate to consider using separate special notices which would trigger separate negotiation moratoriums. If a series of operable units will make up a remedial action it may be appropriate to issue the special notice to cover only the major operable unit(s) or to cover several operable units.

2. <u>Notifying PRPs When Not Appropriate to Issue</u> <u>RI/FS and RD/RA Special Notice</u>

In instances where EPA decides it is inappropriate to issue the special notice, \$122(a) provides for EPA to notify PRPs in writing of that decision. The notice must indicate the reasons why the Region determined that issuing the special notice and entering into "formal" negotiations was not appropriate. The notice should be provided to all PRPs that have been identified to date as well as to the Regional administrative record coordinator for placement in the record. Such notices should be provided as soon as practicable. In instances where the RI/FS or RD/RA have not yet been initiated, the notice should be sent prior to the initiation of such activities if possible.

In addition, the \$122(a) notice should be used as a vehicle for informing PRPs that the Agency will establish or has established an administrative record containing technical documents supporting the Agency's decision on the selection of remedy. The notice should indicate that the record is open for public inspection and comment and should specify where the record will be or has been located.

3. DOJ Role in RIFS and RD/RA Negotiations

The Regions should notify the Chief of the Environmental Enforcement Section in the Department of Justice (DOJ) prior to issuing special notice letters where settlement by a consent decree is contemplated. A copy of this memorandum should also be provided to the Office of Waste Programs Enforcement and the Office of Enforcement and Compliance Monitoring in Headquarters.

The memorandum to DOJ should indicate when the Region intends to issue the special notice. Because most RI/FS negotiations involve consent orders, notice to DOJ on the RI/FS is not ordinarily necessary. However, where a site is in litigation or where settlement by consent decree is expected, DOJ should be notified at least 30 days prior to issuing the RI/FS special notice. In addition, where the resolution of the matter by an administrative order is expected to involve a compromise of past or future response costs and the total response costs will exceed \$500,000, DOJ is to be notified. DOJ's role will be to review the compromise of the claim pursuant to section 122(h)(1) but not to review the administrative order for the RI/FS. For RD/RA negotiations, the notice should be sent to DOJ at least 60 days prior to issuing the RD/RA special notice. The memorandum should also identify the EPA Regional representative DOJ should contact.

In addition, the Regions should consult with the Chief of the Environmental Enforcement Section prior to sending a copy of any draft consent decree or any outline of a draft consent decree to PRPs. The Regions are encouraged to include a draft consent decree with the RD/RA special notice or soon thereafter as discussed below.

4. Timing of RI/FS Special Notice

It is important that PRPs receive the RI/FS special notice letter as soon as practicable. Of greater importance, the letter must be sent sufficiently in advance of obligations for the RI/FS so that negotiations do not delay the initiation of the RI/FS by the Fund in the event the negotiations do not result in an agreement providing for the PRPs to conduct or finance the RI/FS. Timely receipt of the special notice will have a significant effect on the PRPs ability for meaningful participation in formal negotiations.

The RI/FS special notice letter should be sent to PRPs no later than 90 days prior to the scheduled date for initiating the RI/FS. The scheduled date for initiating the RI/FS refers to the date funds will be obligated to commence response activities. A minimum of 90 days is important for ensuring that the negotiation moratorium does not delay initiation of the RI/FS in the event negotiations do not result in a settlement. The time for service by mail should be taken into account.

5. <u>Timing of RD/RA Special Notice</u>

The timing of the RD/RA special notice letter will have a significant impact on both the success of negotiations and on EPA's ability to move forward with implementing a remedy without delay. As indicated earlier, "formal" negotiations pursuant to special notice are not the sole vehicle for reaching settlements. "Informal" negotiations must occur throughout the process and in advance of the special notice. To assure that "formal" negotiations are productive, EPA must initiate PRP search and information exchange activities as well as "informal" negotiations as early as possible.

The primary purpose of the special notice procedures is to facilitate settlements through negotiation. A primary concern in determining when to issue an RD/RA special notice is whether there is a likelihood that meaningful negotiations can be conducted at a given stage in the process. Another concern is that, to the extent practicable, the negotiations must be scheduled to minimize any delay in the remedial design and remedial action. A final concern is that negotiations be carried out in a way that does not undermine or have the appearance of undermining the public participation process. This guidance establishes an approach which identifies when the Regions must generally issue the RD/RA special notice letter. The Regions may, however, adopt an alternative approach under appropriate circumstances. Appendix λ contains illustrations of the three approaches discussed below.

a. <u>General Approach: Issue special notice when release</u> <u>draft FS and proposed plan for public comment.</u> The Regions generally must issue the RD/RA special notice when the draft feasibility study (FS) and proposed plan 7 are released to the public for comment. As shown in Appendix A, issuance of the special notice with the release of the draft FS and proposed plan triggers the initial 60 day negotiation moratorium. The initial 60 day negotiation moratorium begins at the start of the 30 day public comment period and, in conjunction with the first 30 days of the 60 day extended negotiation moratorium, is concurrent with the Record of Decision (ROD) review and approval process. The remaining 30 days of the extended negotiation moratorium is concurrent with the initial phases of the remedial design. EPA's ability to sign the ROD is not affected by the duration of the negotiation moratorium. The ROD may be signed at any point after the close of the public comment period and the preparation of the responsiveness summary for the public.

In most cases, commencing formal negotiations at the same time that the draft FS and proposed plan are released will properly balance the considerations stated earlier relating to EPA's ability to conduct meaningful negotiations, to minimize delay in implementing the RD/RA, and to maintain the integrity of the public participation process. Under this approach, formal opportunity for PRP involvement would begin at an early yet concrete stage in the process. Early participation may be especially advantageous in situations where PRPs have not been previously or substantially involved in RI/FS activities. In addition, PRPs and the public would have knowledge of the possible range of alternatives through the draft FS and proposed

• The time periods depicted in the following discussion and illustrated in Appendix A reflect "best case" scenarios where various response and enforcement activities are expected to be carried out without delay. For example, the public comment period lasts 30 days and does not take into account a possible extension.

⁷ The proposed plan refers to the public participation document developed pursuant to \$17(a). This is a non-legal, non-technical document that describes the alternatives in the FS, and specifies and provides a brief analysis of BPA's preferred alternative. A more detailed discussion of the proposed plan will be contained in the forthcoming "Guidance on Documenting Decisions at Superfund Sites" (referred to as the ROD Guidance). plan prior to "formal" negotiations. This information is important for assisting the PRPs in developing a meaningful "good faith offer" for conducting or financing the RD/RA.

b. <u>Alternative Approach:</u> <u>Issue special notice prior to</u> <u>release of draft FS and proposed plan for public comment.</u> Although the Regions generally will issue the RD/RA special notice when the draft FS and proposed plan are released to the public for comment, the Regions are encouraged to issue the special notice earlier in the process if this action would facilitate the prospects for reaching a settlement. If a Region chooses to follow this approach, the Region should include with the special notice a summary or fact sheet of the alternatives EPA has screened and the alternatives the Agency is currently considering. *

As shown in Appendix A, the RD/RA special notice may be issued prior to EPA's release of the draft FS and proposed plan. Issuance of the special notice triggers the initial 60 day negotiation moratorium. The initial negotiation moratorium is concurrent with the review and release of the draft FS and proposed plan. The initial negotiation moratorium is completed prior to the initiation of the public comment period. The public comment period is concurrent with the first 30 days of the extended negotiation moratorium. The remaining 30 days of the extended negotiation moratorium is concurrent with the ROD review and approval process. The ROD could be signed and the negotiation moratorium could be concluded at about the same time. EPA's ability to sign the ROD is not affected by the negotiation moratorium. The ROD may be signed at any point after the close of the public comment period and the preparation of the responsiveness summary for the public.

In many cases, providing special notice at this early stage may be inappropriate because too much uncertainty would exist. about the remedy to allow for meaningful negotiations. However, under other circumstances it may be appropriate to issue the

[•] Release of a summary or fact sheet on the alternatives that have been screened and the alternatives that are being considered is important for facilitating negotiations at this early stage in the remedial process. This information will be useful to PRPs in developing their "good faith offer" for conducting or financing a response action and will be important for informing PRPs about the alternatives the Agency is considering at the site. The Regions should include the summary of alternatives or fact sheet in the administrative record for each site.

special notice early in the process, especially in situations where there is a relatively small group of PRPs, it is clear what the remedy is likely to be, and the remedy is not likely to be controversial.

Where circumstances permit issuance of the special notice at this early stage, an advantage to this approach is that the ROD review and approval process and the negotiation moratorium could be concluded at about the same time. This would help assure that cleanup occurs as soon as possible whether through a negotiated settlement or Fund-financed action. In addition, there would be an early opportunity to inform PRPs of various remedial alternatives under consideration by EPA prior to EPA's identification of the proposed plan. Early participation may be advantageous where PRPs have not been previously or substantially involved in RI/FS activities.

c. <u>Alternative Approach: Issue special notice when the ROD</u> <u>is signed.</u> Although the Regions generally will issue the RD/RA special notice letter when the draft FS and proposed plan are released to the public for comment, there may be some limited circumstances where it is appropriate to issue the notice later in the process (i.e. when the ROD is signed). This approach may be followed, however, only where the Region can provide adequate justification and where the Region has obtained prior approval from Headquarters. Approval must be obtained in writing from the Directors of the Office of Waste Programs Enforcement and the Office of Emergency and Remedial Response.

As shown in Appendix A, under this approach the RD/RA special notice would not be issued until the ROD is signed. Thus, the entire 60 to 120 day negotiation moratorium would not occur until the remedial design phase.

An advantage to this approach is that since the ROD would be signed and the remedy would be selected at the start of the RD/RA negotiation moratorium, the PRPs would know precisely which remedy the "good faith offer" and the negotiations should focus on. In addition, since the negotiations would begin after the close of the public comment period, the PRPs and EPA would have the benefit of knowing the public comments.

The major disadvantage to this approach is that the negotiation moratorium would not occur until the end of the process (i.e. not until the beginning of the remedial design phase). Issuing the special notice at this point would create the greatest potential for a subsequent delay in implementing the remedy. Instances where it may, however, be appropriate to issue the special notice later in the process (i.e. not until the ROD is signed) may be where more time is needed to conduct informal negotiations, where the site is particularly complex, or where there is an extraordinarily large number of PRPs (e.g. hundreds of PRPs). Another example may be where there is little expectation that a Fund-financed remedial action will occur in the near future at an enforcement-lead site. If Fund-financed activities are not expected to occur and a later moratorium would facilitate cleanup, it may be less important to initiate and conclude negotiations early in the process.

6. <u>Recipients of RI/FS and RD/RA Special Notice</u>

The RI/FS and RD/RA special notice letters should be sent to all parties where there is sufficient evidence to make a preliminary determination of potential liability under \$107 of CERCLA. If there is doubt about whether available information supports issuance of the RI/FS and RD/RA special notices, separate information request letters may be sent to such parties prior to issuing such notice. If a Federal agency has been identified as a generator at a facility not owned/operated by the Federal agency, such agency should be routinely notified like other PRPs.

Section 122(e)(2)(C) authorizes EPA to bring additional parties into negotiations or to enter into a separate agreement with parties when additional PRPs are identified during the negotiation period or after an agreement has been entered into. The Regions may provide a special notice to additional parties if they are identified after issuance of the RI/FS special noticeletter. However, issuance of a special notice to additional parties would not change the duration of the negotiation moratorium. The special notice may invite PRPs to participate in remaining negotiations, but would not extend the pre-existing negotiation moratorium.

Copies of the special notices should be provided to the Regional administrative record coordinator, the appropriate State representative, the State or Federal trustee if a trustee for natural resources has been designated, and to EPA headquarters at the same time notices are sent to PRPs. The copies of notices to headquarters should be sent to the Information Management Section within the Program Management and Support Office of the Office of Waste Programs Enforcement (OWPE).

Providing copies to the administrative record coordinator is important for ensuring that the notice to be placed in the record. Providing copies to the State representative and the State or Federal trustee is important for ensuring that States are appropriately informed about possible future negotiations. Providing copies to OWPE is essential for permitting entry into the Superfund Enforcement Tracking System (SETS). Entry into sets will facilitate our efforts to track site activities and to respond to Congressional and other inquiries. Direct Regional input of data into SETS on notice letter recipients is plagned for FY 1988.

It is not necessary to provide copies of each special notice to the administrative record coordinator, State representative, State or Federal trustee, or headquarters in instances where identical notices are provided to multiple PRPs. Where there are multiple PRPs at a site, a copy of one special notice with a list of other parties who have received the letter would suffice.

7. Contents of RI/FS and RD/RA Special Notice

The RI/FS and RD/RA special notice letters should contain the following components: a) a notification of potential liability, b) a discussion about the special notice and subsequent negotiation moratorium, c) a discussion about the response activities to be conducted, d) a copy of a statement of work or workplan and a draft administrative order on consent for the RI/FS, e) a copy of a draft consent decree for the RD/RA (if possible), f) a discussion about what constitutes a "good faith offer" for the RI/FS, g) a discussion about what constitutes a "good faith offer" for the RD/RA, h) a release of certain sitespecific information (where available and appropriate), i) a demand for payment of EPA costs incurred to date, j) a notification about the administrative record, and k) a deadline for response to the letter and the name of the EPA representative to contact.

a. <u>Potential liability</u>: The letter should specify that PRPs are potentially liable for the costs of conducting the RI/FS or the RD/RA. A detailed discussion about potential liability is not necessary particularly if the RI/FS or RD/RA special notice references the general notice.

b. <u>Special notice and formal negotiations</u>: The letter should discuss the purpose of the special notice and the subsequent negotiation moratorium. The level of detail will depend upon whether the PRP has received the general notice and whether the general notice provided an adequate discussion. At a minimum, the letter should make clear that EPA is inviting PRPs to participate in "formal" negotiations for PRP conduct of the RI/FS or RD/RA and that this letter automatically triggers the formal negotiation period. In addition, it is important that the special notice indicate the date the negotiation moratorium will conclude in the absence of and in the event of a "good faith offer." Finally, the letter should explain that a consent order or consent decree should be finalized by the end of the moratorium. c. <u>Response actions to be conducted</u>: The letter should identify the response activities EPA plans to conduct at the site and provide scheduled dates for initiating such activities if appropriate.

d. <u>Statement of work or workplan and draft administrative</u> order on consent for RI/FS special notice: The letter should provide a statement of work or workplan and draft administrative order (AO) on consent. Such information is crucial to PRPs in their development of a "good faith offer" to EPA for conducting or financing the RI/FS and for ultimately facilitating settlements. The Regions are encouraged to provide the draft AO on consent with the notice letter if practicable. At a minimum, the letter should contain a copy of the statement of work with the expectation that the draft AO will follow as soon as practicable.

e. <u>Draft consent decree for RD/RA special notice</u>: The letter should contain a copy of the draft consent decree if possible. It is important that PRPs have the draft consent decree at the start of negotiations or soon thereafter since the decree contains important information which will assist PRPs in developing their "good faith offer" to EPA.

f. <u>"Good faith offer" for RI/FS</u>: The letter should indicate that a "good faith offer" is a written proposal which demonstrates the PRP's qualifications and willingness to conduct or finance the RI/FS. A "good faith offer" for the RI/FS should include the following:

- a statement of the PRPs willingness to conduct or finance the RI/FS which is generally consistent with EPA's statement of work or work plan and draft administrative order on consent or provides a sufficient basis for further negotiations;
- a paragraph-by-paragraph response to EPA's statement of work or workplan and draft administrative order on consent;
- o a detailed statement of work or workplan identifying how the PRPs plan to proceed with the work;
- a demonstration of the PRPs technical capability to undertake the RI/FS. This should include a requirement that PRPs identify the firm they expect will conduct the work or that PRPs identify the process they will undertake to select a firm;
- o a demonstration of the PRPs financial capability to finance the RI/FS;

- o a statement of the PRPs willingness to reimburse EPA for the costs EPA incurs in overseeing the PRP conduct of the RI/FS as required by \$104(a)(1); and
- o the name, address, and phone number of the party or steering committee who will represent the PRPs in negotiations.

g. <u>"Good faith offer" for RD/RA</u>: The letter should indicate that a "good faith offer" is a written proposal which demonstrates the PRPs qualifications and willingness to conduct or finance the RD/RA. A "good faith offer" for the RD/RA should include the following:

- a statement of the PRPs willingness to conduct or finance the RD/RA which is generally consistent with EPA's proposed plan or which provides a sufficient basis for further negotiations in light of EPA's proposed plan;
- a paragraph-by-paragraph response to EPA's draft consent decree, including a response to other documents that may have been attached to the decree such as a technical scope of work for the proposed plan or access or preauthorization agreements;
- o a detailed "statement of work" or "workplan" identifying how PRPs plan to proceed with the work;
- a demonstration of the PRPs technical capability to undertake the RD/RA. This should include a requirement that PRPs identify the firm they expect will conduct the work or that PRPs identify the process they will undertake to select a firm;
- a demonstration of the PRPs capability to finance the RD/RA;
- o a statement of the PRPs willingness to reimburse EPA for past response and oversight costs;
- a discussion about the PRPs position on releases from liability and reopeners to liability; and
- o the name, address, and phone number of the party or steering committee who will represent the PRPs in negotiations.

h. <u>Information release</u>: To the extent such information is available and to the extent such information has not been previously released, the letter should contain information on the names and addresses of other PRPs, the volume and nature of substances contributed by each PRP, and a ranking by volume of the substances at the facility. Note that the release of information with the RI/FS and RD/RA special notices is not intended to require the release of information previously _ provided to PRPs.

i. <u>Demand for payment</u>: The letter should include a demand that PRPs reimburse EPA for the costs the Agency has incurred in conducting response activities at the site pursuant to \$107(a). The letter should identify the action EPA undertook and the cost of conducting the action. The letter should also indicate that the Agency anticipates expending additional funds on activities covered by this notice and other specified future activities. Finally, the letter should demand payment of interest for past and future response costs incurred by EPA pursuant to \$107(a). Notice letters should not be delayed to obtain cost information where such information has not been previously collected.

j. <u>Administrative record</u>: The letter should be used as a vehicle for informing PRPs of the availability of an administrative record containing documents that form the basis for the Agency's decision on the selection of remedy. The letter should indicate that the record is open to the public for inspection and comment. The letter should also indicate where the record will be or has been located.

k. <u>PRP response and EPA contact person</u>: The letter should encourage PRPs to notify EPA⁻of their interest to participate in negotiations. The letter should indicate that PRPs may respond as a group through a steering committee if a committee has been formed. In addition, the letter should provide the name, phone number, and address of the EPA representative to contact.

D. <u>CONCLUSION OF NEGOTIATION MORATORIUM AND DEADLINE</u> <u>MANAGEMENT FOR RI/FS AND RD/RA</u>

At the conclusion of the \$122(e) negotiation moratorium, the Regions should have a fully negotiated administrative order on consent for the RI/FS and a fully negotiated consent decree for the RD/RA which has been signed by the PRPs. A signed document is necessary to show that an agreement has, in fact, been reached.

Pre-SARA guidance for drafting an administrative order is provided in "Superfund Administrative Order: Workshop and Guidance Materials" (1985) and for drafting a consent decree in "Guidance on Drafting Consent Decrees in Hazardous Waste Cases" (May 1, 1985). These guidances are being revised to include SARA's requirements.

At the conclusion of the 120 day moratorium for the RD/RA a determination must be made on whether to continue settlement activities, whether the site should be cleaned up using Superfund money, or whether to initiate a \$106 enforcement action. A continuation of settlement activities may include seeking an extension to the 120 day negotiation moratorium as discussed below, or sending a consent decree to the Department of Justice for lodging in the appropriate district court.

In instances where an agreement has been reached and fully negotiated but PRPs have not yet obtained signatures, it may be necessary to obtain an extension to the negotiation moratorium. Extensions may also be necessary where the agreement has not been fully negotiated but all major issues are resolved and outstanding issues are well defined and final language is imminent. Extensions to the negotiation moratorium can be obtained only in certain circumstances as discussed in the February 12, 1987 "Interim Guidance: Streamlining the CERCLA Settlement Decision Process." 19

The timing of special notice letters will have a significant affect on our ability to successfully conclude negotiations at the end of the moratorium period. The Streamlined Settlement Policy provides for two different processes for obtaining extensions for the RI/FS and RD/RA moratoriums. The policy indicates that the Regional Administrator has the discretion to terminate or extend negotiations for the RI/FS after 90 days. However, extension of negotiations beyond an additional 30 days should be authorized by the Regional Administrator only in limited cases.

Relating to the RD/RA moratorium. the Streamlined Settlement Policy provides for either Regional or Headquarters approval of an extension under certain circumstances. An extension to the 120 day RD/RA moratorium may be granted for an additional 30 days by the Regional Administrator when settlement is likely and imminent. An additional extension beyond the 30 days may be approved only by the Assistant Administrator for the Office of Solid Waste and Emergency Response (OSWER) and only in rare and extraordinary circumstances.

This guidance re-emphasizes the importance of meeting the 90 day moratorium for the RI/FS and the 120 day moratorium for the RD/RA. To aid that policy, this guidance identifies three circumstances where the Regional Administrator and Assistant Administrator for OSWER may consider granting such extensions for the RD/RA moratorium.

¹⁰ This guidance was issued under OSWER Directive #9832.9.

First, it may be appropriate for the Regional Administrator or the Assistant Administrator to extend the 120 day moratorium for the RD/RA if EPA selects a remedy in the ROD which is significantly different from the Agency's stated preference in the proposed plan. This could mean that the focus of negotiations could change significantly, requiring additional time to reach agreement with PRPs.

The second example applies to Fund-lead sites. It may be appropriate for the Regional Administrator or the Assistant Administrator to extend the 120 day negotiation moratorium for the RD/RA if non-enforcement activities at the site (e.g. an extended public comment period or an extended ROD review and approval process) cause a significant delay in the Agency's ability to move forward in implementing a Fund-financed remedy. An extension to the negotiation moratorium may be especially appropriate if there is reason to believe a negotiated settlement is imminent. In other words, if the Fund is not ready to move forward in implementing the remedy at the end of the 120 day negotiation moratorium there is no reason to conclude negotiations if there is reason to believe an agreement can be reached.

The third example applies to enforcement-lead sites. It may be appropriate for the Regional Administrator or the Assistant Administrator to extend the 120 day negotiation moratorium for the RD/RA after a \$106 litigation referral has been prepared and referred to the Department of Justice (DOJ) for action. In fact, the preparation and referral of a case to DOJ may be an important mechanism for providing the necessary impetus for reaching a voluntary settlement. In many cases it may be appropriate to issue a unilateral administrative order concurrent with the referral.

VI. NOTICE LETTERS AND NEGOTIATION MORATORIUM FOR REMOVAL ACTIONS

The notice letter process for removal actions differs from the notification process for remedial actions. As discussed above, the notification process for remedial actions involves issuance of three notice letters. The notification process for removals will involve only one notice letter which may or may not invoke the \$122(e) special notice procedures as discussed below.

A. NOTICE LETTERS

1. Whether to Issue Removal Notice

The Regions should attempt to contact PRPs prior to initiating a Fund-financed removal action to inform PRPs of their potential liability where EPA will incur response costs or
to secure a private party response. This guidance encourages the Regions to seek PRP response through a written notice letter but the Regions may contact PRPs verbally (with a written follow-up notice). This is consistent with the guidance on "Issuance of Administrative Orders for Immediate Removal Actions" (2/21/84).

The Regions should issue notice letters to readily identifiable PRPs for removal actions in the vast majority of cases. The content of the notice will vary depending whether the notice will be used simply to notify PRPs of their potential liability for an action EPA has already taken or is about to take, whether the notice will be used to encourage a private party response through "informal" negotiations (i.e. negotiations not triggered by the \$122(e) special notice procedures), or whether the notice will be used as a mechanism for invoking the \$122(e) special notice procedures which provide for "formal" negotiations between EPA and PRPs.

2. When to Use Special Notice Procedures for Removals

The Regions should consider using the \$122(e) special notice procedures only for those removals where the threat is of a nature that it is not necessary to initiate an onsite removal action for at least six months. The "six month planning time period" begins once the site evaluation is completed. This means that for the vast majority of removal actions the Regions will not be required to utilize the special notice procedures. It is not appropriate to utilize special notices for most removal actions because the subsequent moratorium may interfere with the Agency's ability to implement the remedy in a timely manner. In addition, it may not be worth expending the time and resources to enter into formal negotiations when a removal will be a relatively short term and inexpensive response action.

The Regions should include the following factors in their determination of whether it is appropriate to utilize the special notice procedures for removals with a six month planning lead time: 1) whether viable PRPs have been identified, 2) whether the PRPs are expected to respond favorably to the invitation to participate in negotiations and to conduct or finance the removal action, 3) whether issuance of the special notice could delay implementation of the removal action, and 4) whether it may be more appropriate to enter into "informal" negotiations in lieu of "formal" negotiations under \$122(e).

In determining the PRPs viability, the Region should inquire about the PRPs financial and technical capability for conducting and/or financing the removal action in an effective and timely manner. In determining the PRPs willingness to undertake or finance the removal action, the Region should, at a minimum, obtain a verbal agreement from the PRPs prior to issuance of the special notice. In determining whether the special notice may delay implementation of the remedy or in determining whether to enter into "informal" rather than "formal" negotiations, the Regions should consider whether the \$122(e) negotiation moratorium would interfere with other activities at the size.

3. <u>Notifying PRPs When Not Appropriate to Utilize</u> <u>Special Notice Procedures for Removals</u>

EPA's decision on whether to use the special notice procedures for any response action is clearly discretionary. However, \$122(a) requires the Agency to notify PRPs in writing when the Agency decides not to utilize such procedures. The removal notice provides a convenient vehicle for informing PRPs of EPA's decision not to utilize the special notice procedures. The notice should, therefore, inform PRPs of EPA's decision not to utilize such procedures when this determination has been made and should provide an explanation for that decision.

4. DOJ Role in Removal Negotiations

The Regions should consult with the Chief of the Environmental Enforcement Section of DOJ prior to issuing a special notice letter for removal actions where settlement by consent decree is contemplated, or where the settlement is expected to involve a compromise of past or future response costs and the total response costs will exceed \$500,000. The Regions should consult with DOJ prior to releasing a draft consent decree to PRPs.

5. <u>Timing of Removal Notice</u>

A removal notice that does not invoke the special notice procedures should be provided to PRPs as soon as practicable. For removal notices that invoke the special notice procedures, the notice should be issued as early as possible but no later than 120 days before the scheduled date for initiating the removal action. The scheduled date for initiating the removal action is the date removal extramural cleanup contractor funds will be obligated and onsite cleanup will begin.

The timing of a notice which invokes the special notice procedures is critical because issuance of the notice triggers the subsequent 60 to 120 day moratorium on EPA conduct of the removal action. (The moratorium would last only 60 days in instances where the PRPs do not provide EPA with a "good faith offer"). Issuing the special notice at least 120 days before EPA will begin the removal ensures that the subsequent 120 day moratorium does not affect EPA's ability to implement the removal action in the event negotiations do not result in an agreement for PRP conduct of the removal action.

6. Recipients of Removal Notice

The removal notice should be sent to all parties where there is sufficient evidence to make a preliminary determination.of potential liability under \$107 of CERCLA. If a Federal agency has been identified as a generator at a facility not owned/operated by the Federal agency, such agency should be routinely notified like other PRPs.

Copies of removal notices should be provided to the Regional administrative record coordinator, the appropriate State representative, and to headquarters. Providing copies to the administrative record coordinator is important for ensuring that the notice to be placed in the record. Providing copies to the State representative is important for ensuring that States are appropriately informed about possible future negotiations.

Providing copies to the Information Management Section within the Program Management and Support Office of the Office of Waste Programs Enforcement for entry into the Superfund Enforcement Tracking System (SETS). Copies should be sent to OWPE at the same time they are sent to PRPs. Providing copies to OWPE is essential for facilitating our efforts to track site activities and to respond to Congressional and other inguiries.

It is not necessary to provide copies of each removal notice to the administrative record coordinator, State representative, State or Federal trustee, or headquarters in instances where identical notices are provided to multiple PRPs. Where there are multiple PRPs at a site, a copy of one removal notice with a list of other parties who have received the letter would suffice.

7. <u>Contents of Removal Notice</u>

As indicated, the content of the removal notice will vary depending upon whether the purpose of the letter is to simply inform PRPs of their potential liability or whether the letter will also be used to provide an opportunity for PRP involvement in negotiations either through "informal" or "formal" negotiations. The following highlights the components that should be included in the three different types of removal notices. The specific content of each component of the removal notice should be essentially the same as described earlier for RI/FS and RD/RA general and special notices, except where otherwise specified.

a. Notice of potential liability: If the purpose of the removal notice is simply to inform PRPs of their potential liability and to provide notice that the Agency has or is about to take a response action, the notice should contain the following components: a notice of potential liability; a discussion about site response activities that have been or will

be conducted at the site; a notice on the availability of an administrative record; and a notice pursuant \$122(a) that the special notice procedures will not be used.

The notification under \$122(a) should inform PRPs that the Agency will not (or did not) use the \$122(e) special notice procedures for this particular response action and should provide an explanation for that decision. The letter should indicate that it is the Agency's policy not to use the special notice procedures for removals unless there is a six month planning lead time prior to the initiation of the response action. If the response action does involve a removal with a six month planning lead time but the Agency made a case-specific determination not to use the special notice procedures, the letter should provide an explanation why the use of such procedures was determined to be inappropriate for that particular response action.

b. Notice of potential liability and opportunity to enter into "informal" negotiations: If the purpose of the removal notice is to inform PRPs of their potential liability and to provide PRPs with an opportunity to enter into negotiations with EPA without invoking the \$122(e) special notice procedures, the notice should contain, the following components: a notice of potential liability, a discussion about site response activities that will be conducted at the site; a copy of the statement of work or workplan and draft administrative order on consent; a notification pursuant to \$122(a) that the special notice procedures will not be used; a request that PRPs notify EPA within a specified period of time of their interest to participate in negotiations; a notice on the availability of the administrative record: and information on the SPA representative to contact. The £122(a) notification should contain the same information discussed in the proceeding paragraph.

Notice of potential liability and opportunity to enter С. into "formal" negotiations pursuant to \$122(e) special notice procedures: If the purpose of the removal notice is to inform PRPs of their potential liability and to provide PRPs with an opportunity to enter into negotiations with EPA using the \$122(e) special notice procedures, the notice should contain the following components: a notice of potential liability; a discussion about site response activities that will be conducted at the site; a discussion about the special notice procedures and the negotiation moratorium; a copy of the statement of work or workplan and draft administrative order on consent; a discussion about what constitutes a "good faith offer"; a request that PRPs notify BPA within a specified period of time indicating their interest to participate in negotiations; a notice on the availability of the administrative record; and information on the EPA representative to contact. The "good faith offer" should contain essentially the same components as described above for the $RD/R\lambda$.

B. <u>CONCLUSION OF NEGOTIATION MORATORIUM AND</u> DEADLINE MANAGEMENT FOR REMOVALS

At the conclusion of the \$122(e) negotiation moratorium for removal actions, the Regions should have a fully negotiated administrative order on consent which has been signed by the PRPs. (Where appropriate, a signed consent decree should be provided). A signed administrative order on consent (or a consent decree) will show that the negotiations have been successfully completed.

The expectation is that the negotiations will be concluded at the end of the 120 day moratorium and the Regions are strongly encouraged to conclude the negotiations within this period of time. In instances where the negotiations do not result in an agreement, the Regions may seek an extension to the 120 day moratorium, issue an administrative order, or proceed with a Fund-financed removal. Note that the Regional Administrator may grant an extension to the 120 day moratorium only in limited and appropriate circumstances.

C. <u>ADMINISTRATIVE ORDERS AND NEGOTIATION MORATORIUM</u> <u>FOR REMOVALS</u>

In most instances, use of the special notice procedures for removal actions will not affect existing policy on issuing administrative orders for removals since the special notice procedures will be issued for only a small portion of removals. For details on the Agency's policy on administrative orders refer to the guidance on "Issuance of Administrative Orders for Immediate Removals" (2/21/84).

It is necessary, however, to modify existing policy in one respect. In instances where Regions use the special notice procedures for a removal action and where issuance of an administrative order is necessary and appropriate, the Regions should not issue the order until the end of the negotiation moratorium. This ensures that the negotiation moratorium will be used to negotiate voluntary settlements.

VII. DISCLAIMER

The policies and procedures established in this document are intended solely for the guidance of Government personnel. They are not intended and can not be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. The Agency reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

•

•

VIII. FOR FURTHER INFORMATION

For further information or questions concerning this guidance, please contact Kathy MacKinnon in the Office of Maste Programs Enforcement at FTS-475-6770.

983410

Appendix A

•

Timing of RD/RA Special Notice Letter

A. General Approach: Issue RD / RA Special Notice When Release Draft FS and Proposed Plan

Selection of Remedy Process

Conduct Review/ RI/FS FS/Propose Plan	Public Com- ment	Review ROD	Conduct RD)) }})
---	------------------------	---------------	---------------	----------	---

11

Notice and	Extended
Negotiation	Negotiation
Moratorium	Moratorium

Special Notice / Negotiation Moratorium



B. Alternative Approach: Issue RD / RA Special Notice Prior to Release of Draft FS and Proposed Plan

Selection of Remedy Process

Conduct RI/FSReview/ Release Screening of AlternativesReview/ Release FS/Proposed Plan	Public Com- ment	Review ROD	Conduct RD		
---	------------------------	---------------	---------------	--	--

..

Negotiation Negotiation Moratorium Moratorium	Notice and Negotiation Moratorium	Extended Negotiation Moratorium
--	---	---------------------------------------

Special Notice / Negotiation Moratorium



C. Alternative Approach: Issue RD / RA Special Notice Once ROD Signed

Selection of Remedy Process

..

RI/FS FS/Proposed Contraction RO RD

Notice and	Extended
Negotiation	Negotiation
Moratorium	Moratorium

Special Notice / Negotiation Moratorium



9834<u>,</u>i ù

Appendix B

•

PRP Settlement Process for RI/FS and RD/RA

PRP Settlement Process for Rt / FS



PRP Settlement Process for RD / RA

•



Draft Guidance on CERCLA 106 Administrative Orders for Removal Actions

•



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUN 1 9 1989

. . .

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Draft Guidance on CERCLA §106(a) Administrative Orders for Removal Actions

FROM: John Cross *H* Office of Waste Programs Enforcement

TO: Oil and Hazardous Materials Coordinators Regions I-X

We are circulating an early draft of the Guidance on CERCLA §106(a) Administrative Orders for Removal Actions at this time to facilitate your discussion of §106 orders at the upcoming Removal Managers' meeting. The guidance is in preliminary form, and will be subsequently circulated for comment to Regional Enforcement Branch Chiefs and Regional Counsel.

The guidance is designed to reflect statutory changes implemented by SARA and changes in Agency policy due to Agency experience. The guidance is also consistent with the Agency's 90 Day Review Report and concepts endorsed by the Settlement Incentives and Disincentives Work Group. An endorsed draft guidance is expected to be distributed formally by the end of this summer jointly by OWPE and OECM.

If you wish to submit written comments on the draft guidance, please send them by pouch mail to Kathryn Nolan at EPA Headquarters, OWPE, OS-510. If you have any questions concerning this guidance, she can be reached at (FTS) 382-2034. Thank you for your assistance in the review of this document.

DRAFT

MEMORANDUM

- **SUBJECT:** Guidance on Issuance of CERCLA §106(a) Administrative Orders for Removal Actions
- **FROM:** Jonathan Z. Cannon, Acting Assistant Administrator Solid Waste and Emergency Response

Edward E. Reich, Acting Assistant Administrator Enforcement and Compliance Monitoring

TO: Regional Administrators, Regions I-X

I. Introduction

This memorandum sets forth procedures regarding issuance o §106(a) administrative orders for removal actions under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended by the Superfund Amendments and Reauthorization Act of 1986 (CERCLA or Superfund).¹ This guidance applies to unilateral enforcement actions and settlement agreements for potentially responsible party (PRP) conduct of removal actions based on §106 of CERCLA.² It is designed for use by On-Scene Coordinators (OSC), Office of Regional Counsel (ORC), and Removal Program Managers. For additional guidance on legal issues related

For a discussion of general principles governir settlement agreements, see the Interim CERCLA Settlement Policy dated December 5, 1984 (OSWER Directive number 9835.0).

¹ For guidance on the general purposes and principles of the Agency's administrative order authority under §106(a) of CERCLA, and more detailed procedures on implementation of that authority, see the (date) memorandum, "Guidance on CERCLA §106(a) Administrative Orders for Remedial Actions" (OSWER Directive number XXX). These memoranda together supersede the September 8, 1983 "Guidance Memorandum on Use and Issuance of Administrative Orders Under §106(a) of CERCLA" (OSWER Directive number 9833.0) and the February 21, 1984 guidance on "Issuance of Administrative Orders for Immediate Removal Actions" (OSWER Directive number 9833.1A). Changes to the guidances are the result of statutory amendments and Agency experience.



to §106 administrative orders, see the memorandum entitled "Guidance on CERCLA §106(a) Administrative Orders for Remedial Actions" (OSWER Directive number XXX).

In order to successfully conduct the largest number of removals, the Agency must secure private party response actions. To accomplish this, in appropriate circumstances, the Agency negotiates settlement agreements embodied in §106(a) consent administrative orders (consent orders) with parties willing to do the work. The Agency issues §106(a) unilateral administrative orders (unilateral orders) when private parties are not willing to undertake work as part of a consent order.

Private party removals serve an important function in the Superfund response and enforcement process. Private party removals can free up the Fund for additional removal actions at sites without financially viable PRPs. PRPs are also introduced to the Superfund enforcement and settlement process, and therefore are more likely to cooperate if subsequent response actions are necessary. Finally, when private parties conduct the removal action, it eliminates the need for subsequent cost recovery actions, which frequently demand considerable time from Regional technical and legal staff, and frequently occur under statute of limitations deadlines.

II. General Procedures for Enforcement Removal Actions

Civil investigators working together with the OSC should conduct a PRP search immediately after the OSC determines the need for a removal action. While the OSC determines the scope of the removal, the OSC and/or ORC should develop an enforcement strategy. The exigencies of the circumstances, particularly in an emergency removal context, will affect the enforcement approach.

PRP Notification

Once PRPs have been identified, OSCs should notify them in writing that EPA is planning to conduct a response action at the site and that they are potentially liable. The notice should state that PRPs may agree to conduct the response action through a

³ Appendix A of this document defines the two types of §106 administrative orders and distinguishes them from their judicial counterparts. For guidance on the role of §106 judicial actions and procedures to follow for their implementation, see the memorandum, "Guidance on CERCLA Section 106 Judicial Actions," (Reich/Porter, 2/24/89) (OSWER Directive Number 9835.7). For information on issuing §106 unilateral administrative orders to Federal facilities, see "Enforcement Actions under RCRA and CERCLA at Federal Facilities," dated January 25, 1988 (OSWER Directive number XXX).

settlement agreement with the Agency. Although time may not permit settlement negotiations at sites where a true emergency exists, the Agency should attempt to notify all known PRPs prior to conducting any response action. OSCs may send this notice prior to fully delineating the scope of the removal. Advance notice to PRPs is particularly important at sites with multiple PRPs. To expedite the settlement process in time-critical situations, OSCs may initially contact PRPs orally. Oral notification should be followed by written confirmation. Written notice to PRPs at timecritical removal sites may take the form of a general notice letter or a CERCLA §122(a) letter. There is no need for oral notification at non-time-critical removal sites. At non-time-critical removal sites, OSCs should notify PRPs through special notice letters or CERCLA §122(a) letters. The Agency should issue CERCLA §122(a) letters when a decision is made not to issue special notice at a site. For examples of all three letters, see Appendix C of this For further information concerning special notice cuidance. procedures, see the "Interim Guidance on Notice Letters, Negotiations and Information Exchange" (Adams/Porter _/_/_) (OSWER directive number 9834.10), 53 Fed. Reg. 5298 (1988).

Action Memorandum

Following PRP notice, OSCs should finalize the scope of the removal, determine whether to enhance the PRP search, and begin PRP negotiations. At the start of negotiations, OSCs should provid PRPs with a document which summarizes the work required at the sit (a statement of work) and establish a negotiations deadline. The statement of work should be accompanied by a draft consent order prepared by the Superfund Enforcement personnel and reviewed and approved by ORC.

A signed Action Memorandum should be prepared during negotiations. An Action Memorandum serves several purposes in addition to its traditional function of authorizing Fund financing at sites. During negotiations, it notifies PRPs that the Agency has the finances to conduct the removal with the Fund, and sue for treble damages. A signed Action Memorandum also constitutes an important part of the administrative record for the removal action response decision, and will be critical in any enforcement case initiated to enforce a unilateral order, in the event that an order is issued. A confidential addendum to the Action Memorandum sets forth the enforcement strategy.

Unilateral Orders/Consent Orders

If PRPs agree by the negotiation deadline to conduct the removal, a consent order should be signed. If PRPs do not agree by the negotiation deadline to conduct the removal, the negotiations team should strongly consider issuing a unilateral order. Unilateral orders should routinely be issued in cases which meet the criteria set forth in Part V of this guidance. Issuanc



of a unilateral order should be considered prior to either Fund financing or a judicial referral.⁴ Unilateral orders give PRPs a final opportunity to participate in the cleanup process before a court compels them to do so. Under unilateral orders, PRPs incur liability for the cleanup, penalties and damages. Moreover, issuing a unilateral order prior to judicial referral should further support record review of the Agency's response action in any subsequent court proceeding.

If PRPs ask to settle with the Agency as they prepare the first deliverable under the unilateral order, OSCs and ORC may attempt to negotiate a consent order with the PRPs. Consent orders are beneficial to the Agency because EPA may recover past costs through a consent order.⁵ However, in most situations, PRPs have already been given an opportunity to settle with the Agency prior to this point. Therefore, if PRPs do not readily agree to sign a consent order during these negotiations,⁶ negotiations should be terminated and PRP conduct of the response action should continue under the unilateral order.

If PRPs agree to conduct the removal but not to sign a consent order, and the case does not meet the criteria for a unilateral order (i.e. no imminent and substantial endangerment), the Agency may conduct the removal with Fund finances. Where there are special circumstances, and where Fund financing is not available, PRPs may proceed with conduct of the removal under Agency oversight. At the outset of such PRP action, ORC must provide written notice to these PRPs. The written notice should notify PRPs that they will continue to be liable in the event that a subsequent response action is required at the site. This written notification protects the interests of the Agency in the event that the response action is not fully or adequately executed.

<u>Oversight</u>

Under both consent and unilateral orders, oversight should be conducted routinely to ensure PRP compliance with the terms of the order. OSCs must immediately bring work corrections and missed milestones to the PRPs' attention. Under a consent order, PRPs

⁴ See Appendix A, Part II for an explanation of a judicial referral.

⁵ Under-a unilateral order, past costs may be obtained through a demand letter or a cost recovery action.

⁶ Any compromise of past costs must be conducted under the authority of CERCLA §122(h)(1). If a past cost is compromised and total past additional response costs at the site (i.e. work and money) exceed \$300,000 excluding prejudgment interest, prior written approval of the compromise must be obtained from DOJ.

DRAFT

will be liable for stipulated penalties for noncompliance. The Agency should be prepared to obligate the Fund and/or refer the case to the Department of Justice (DOJ) should it be determined that the PRP does not intend to comply with the order. Through a judicial referral, the Agency may obtain stipulated penalties, §106 daily penalties, and/or PRP compliance with the order.

<u>III. Statutory Requirements for Issuing §106(a) Consent and</u> <u>Unilateral Administrative Orders</u>

Section 106(a) administrative orders for removal actions must meet several statutory requirements. These statutory requirements apply to both removal consent and unilateral administrative orders. The Action Memorandum should already contain the information needed to support these statutory requirements. OSCs should ensure that this information is adequately contained in the Action Memorandum. Superfund Enforcement Personnel and/or ORC should consult the Action Memorandum when drafting the removal consent and/or unilateral order.

The statutory requirements which apply to both consent and unilateral orders are described in numbers 1 - 4 below. The fifth statutory requirement set forth below applies only to unilateral orders.

1) Evidence of a Release or Threatened Release of a Hazardov Substance

A removal action may be funded by the Agency when there is information regarding release or threat of a release of a hazardous substance. Similarly, a unilateral order may be issued under section 106 where there is a release or a threatened release of a hazardous substance. The nature of the determination of the release or threat of release should be identified in the order.

The hazardous substances that are the subject of the release or threat of release should be documented in the Action Memorandum. At least some of the hazardous substances should be referenced in the order as well.

2) Evidence that the Release or Threatened Release is from a Facility

The order should specify the physical location of the release or threatened release. This establishes that the release or threatened release is from a facility as defined in CERCLA §101(9).

 $^{^7}$ See CERCLA section 101(22) for the definition of a release. CERCLA section 101(14) defines hazardous substances; see also 40 C.F.R. § 302.4.

DRAFT

3) Evidence of Imminent and Substantial Endangerment

The term imminent and substantial endangerment has a particular meaning in environmental statutes. An endangerment is a threatened or potential harm; actual harm need not be shown. An endangerment may be imminent if the conditions which give rise to it are present, even though they may not be realized for years. When §106(a) administrative orders are issued for removals, the data base available to support an endangerment finding may be limited. It will most likely consist of information from the inspection and preliminary sampling data. This information should be documented in the Action Memorandum and referenced in the unilateral order.

4) Notice to Affected State

Regions must notify the state prior to issuing a unilateral order. The affected state is interpreted to be the state where the facility with the release or threatened release is located. Written notification to the state should precede Federal action, if possible. When rapid response at a site is necessary, notice may be provided in a telephone call from the EPA Division Director to the director of the state lead agency responsible for the CERCLA program. Written confirmation of oral notice always must follow.

5) Persons who may Receive Unilateral Orders

The classes of persons who may receive unilateral orders include, at a minimum, the four classes of parties who are liable under section 107 of CERCLA. These classes generally are (1) present owners and operators, (2) past owners and operators at the time of disposal, (3) persons who arranged for treatment or disposal, and (4) transporters who selected the site. In addition, other persons may receive unilateral orders to assure relief. For example, unilateral orders may be issued to obtain the necessary cooperation of parties indispensible to completion of a response action.

See the previously mentioned guidance on §106 remedial administrative orders if greater detail is needed concerning these statutory requirements.

IV. Elements of Unilateral Administrative Orders

In addition to statutory prerequisites that must be satisfied before unilateral orders may be issued, other substantive elements are usually included in unilateral orders. These elements are necessary for the unilateral orders to be both enforceable and

Drime

effective A unilateral order should contain the following sections:

- o Introduction and Jurisdiction;
- o Parties Bound;
- o Findings of Fact;
- o Conclusions of Law and Determinations;
- o Notice to the State;
- o Work to be Performed;
- o Quality Assurance;
- o Modification of the Work Required;
- o Compliance with Other Applicable Laws;
- o Designated Project Coordinators and OSC Authority;
- o Progress Reports, Notice of Delay;
- o Access and Data/Document Availability;
- o Administrative Record, Record Preservation;
- o Reimbursement of Oversight Costs;
- o Further Enforcement, Reservations, and Disclaimers;
- o Effective Date/Subsequent Modification;
- o Opportunity to Confer; and

.

o Termination and Satisfaction.

The guidance on §106 remedial administrative orders contains a discussion of the following provisions: Findings of Fact; Conclusions of Law and Determinations; Work to be Performed; Effective Date; and Opportunity to Confer. See also the Model Unilateral Order for Removals, dated XXX (OSWER Directive number XXX).

⁸ Although a unilateral order for a removal action should include an assertion that the removal action is consistent with the NCP, it is recognized that the NCP expressly exempts from certain provisions of its coverage removals conducted by PRPs pursuant to §106 of CERCLA. See 40 C.F.R. Part 300.65(h).



V. Factors for Issuing Unilateral Orders in Removal Actions

The following factors should be considered when issuing unilateral orders for removal actions. These factors differ from §106 judicial action factors because unilateral orders serve many different purposes. For a discussion of the policy criteria relevant to §106 judicial actions, see the Guidance on CERCLA Section 106 Judicial Actions, mentioned previously.

A) Immediacy of the Need to Respond

Generally, it will not be possible to issue unilateral orders for true emergency removals. At time-critical removal sites, if there is sufficient time (e.g., two weeks) before on-site activity must begin, a unilateral order may be issued. Alternatively, the Regional office may wish to stabilize site conditions and issue a unilateral order for the remainder of the removal action.

There is sufficient time at non-time-critical removals to issue a unilateral order before on-site activity must begin. By definition, non-time-critical removals have a planning period of more than six months before on-site activity must begin.

B) PRPs are Liable and No Indication that they are not Financially Viable

Unilateral orders should only be issued to PRPs for whom the Agency has sufficient evidence as to their liability. Information documenting PRP liability should have been secured through PRP searches, including §104 information requests. Unless there is information to the contrary, the Regions should assume that PRPs have the financial capability to conduct the removal.

C) The Response Action should be Specifically Defined

The unilateral order should define the removal response action with enough specificity to instruct the PRPs as to what is expected of them. A clearly defined response action is also necessary for the Agency to determine whether the PRPs have complied with the order. Where possible, this information may be directly incorporated into the unilateral order from the Action Memorandum. Where site conditions necessitate a "decide as proceed" approach and the removal action cannot be specifically defined, a unilateral order generally should not be issued.

D) Unique Technical Problems/Oversight

Where a removal action presents unusual implementation difficulties for the PRPs, or unique technical problems which may present unusual oversight difficulties for the Agency, the site may be inappropriate for a unilateral order.

DRAFT

E) PRP Technical Inability/Lack of Trustworthiness

Due to the technical inability of the PRPs, and/or their lack of trustworthiness, the Agency's oversight of nonsettling PRPs may differ from that of settling PRPs, despite the fact that PRPs operating under unilateral orders are subject to daily penalties for failure to comply with the terms of the order. Therefore, Regional offices should consider the technical ability of the PRPs and/or their trustworthiness prior to issuing a unilateral order.

F) Very Low Cost/Low Priority

Removal actions estimated to cost relatively little to conduct, and/or removal actions which are low in priority, may not be appropriate for unilateral orders.

G) Resources

If critical legal staff are not available to assist in a removal enforcement action, OSCs may conduct the cleanup through Fund financing, or stabilize the site and postpone enforcement action until enforcement resources are available.

VI. Follow-Up Procedures for Unilateral Orders

Agency policy is to provide PRPs an opportunity to discus implementation of a unilateral order with the Agency. The conference is not an adversarial hearing and does not constitute part of a proceeding to challenge the order. Instead, the conference is designed to ensure that the order is based on complete and accurate information, and to facilitate implementation. See the guidance on §106 remedial administrative orders for further information on the conference. In the case of removals without much lead time, the Agency may provide less formal conference procedures than that described in the §106 remedial administrative order guidance.

In the event of noncompliance with the unilateral order, Regions have flexibility to take one of the following actions: seek penalties to compel compliance with the order, takeover the project and utilize Fund financing, or seek a court order compelling PRP conduct of the removal action. Where the OSC decides that site specific circumstances require an immediate response, or the ORC assesses our likelihood of success at court to be minimal, Regions should utilize Fund financing. Therefore,

DRAFT

site circumstances may dictate the proper course of action.⁹ The Agency may need first to stabilize the site before referring a case to DOJ. For further information, see the guidances on §106 remedial administrative orders and §106 judicial actions.

VII. Note on Purpose and Use of this Memorandum

The policy and procedures set forth herein, and internal office procedures adopted pursuant hereto, are intended solely for the guidance of attorneys and employees of the U.S. Environmental Protection Agency. They are not intended to, nor do they constitute rulemaking by the Agency, and may not be relied upon to create a right or benefit, substantive or procedural, enforceable at law or in equity by any person. The Agency may take any action which is at variance with the policies or procedures contained in this memorandum, or which is not in compliance with internal office procedures that may be adopted pursuant to these materials.

If you have any questions concerning any material contained herein, please call Kathryn Nolan (FTS) 202-382-2034 of the Office of Waste Programs Enforcement.

⁹ Under Agency policy, Regions have discretion to determine the proper course of action in the event of PRP noncompliance with unilateral orders. Regions have discretion to take courses of action other than Fund financing followed by cost recovery.

UNAFT

APPENDIX A

ADMINISTRATIVE AND JUDICIAL SETTLEMENT AND ENFORCEMENT TOOLS

I. <u>Administrative</u> Enforcement

Settlement: §106 Consent Administrative Orders

The Agency bases its removal settlement agreements on section 106 of CERCLA. Removal settlement agreements may be encompassed in §106 consent administrative orders or consent decrees. If PRPs do not adequately comply with §106 consent orders, the Agency may pursue stipulated penalties, §109 monetary penalties, and §106 daily penalties through a referral to DOJ (see below).

No Settlement: §106 Unilateral Administrative Orders

Where there is no settlement agreement, unilateral orders may be used to compel PRPs to conduct removals. Upon receipt of unilateral orders, PRPs may comply with the terms of the orders and conduct the removal, or they may decide to settle with the Agency. If PRPs decide to settle, the unilateral orders may be replaced with a consent order.

If PRPs do not comply with the unilateral order, a court maw impose daily fines under §106(b)(1), and/or punitive damages in amount up to three times that expended from the Fund. Puniti damages may be obtained only in the event of a cleanup financed by the Fund. Therefore, if a settlement agreement cannot be reached, Regions should consider issuing a unilateral order. This facilitates Agency collection of penalties or punitive damages for noncompliance.

II. Judicial Enforcement

Noncompliance with the Order: §106 Judicial Actions

Administrative orders are not self-enforcing, nor can the Agency enforce them without assistance from the pertinent Federal District Court. The Agency seeks enforcement of its administrative orders in court through the assistance of the Department of Justice If PRPs refuse to comply with a unilateral order directing (DOJ). them to conduct a removal, and circumstances at the site do not require immediate site cleanup with Fund finances, the Agency should refer the case to DOJ. DOJ will file the case in court for judicial enforcement. Even if the Agency cleans up the site with Fund finances, the Agency may refer a case to DOJ for the collection of penalties or damages, along with a cost recovery action for Fund expenditures, from recalcitrant PRPs. Referrals to DOJ are necessary whether the Agency seeks compliance with the order or penalties. Referrals to DOJ are made through a section 106 judicial action.

الم المناز الم

APPENDIX B

Removal Actions--Scope of Response

Section 101(23) of CERCLA defines the term "removal" to include a variety of activities. Removal activities include, among others: monitoring, assessing and evaluating the release or threat of release of hazardous substances; disposal of removed material; measures to limit access; provision of alternative water supplies; and temporary evacuation and housing of threatened individuals not otherwise provided for. The National Contingency Plan and the February 1988 "Superfund Removal Procedures, Revision Number Three" (OSWER Directive number 9360.0-03B) divide the statutory concept of removals into Classic Emergencies, Time-Critical, and Non-Time-Critical removal actions. These determinations are based upon the site evaluation which examines the immediacy and the severity of the hazard to public health and the environment. The categories establish a scale for assessing the length of time within which the Agency must respond to an event. Once a site has been categorized, it does not change categories.

Classic emergency removal actions are undertaken if a release or threat of release requires response within hours of the lead agency's determination that a removal action is necessary. In classic emergency removal actions, on-site activity often lasts less than 30 days. During classic emergencies, response personnel may need to base their decisions on relatively limited data and act quickly. As a result, it is often difficult to ensure an adequate PRP response. Common examples of classic emergency removal actions include road accidents and spills, or fencing of a contaminated area.

Time-critical removal actions are those where, based on the site evaluation, the lead agency determines that a removal action is appropriate and that less than six months is available before cleanup activities begin on-site. Examples of time-critical removal actions include removal and transport of drums, barrels, tanks, or other bulk containers that contain or may contain hazardous substances to a RCRA-approved facility, or containment of wastes until a more in-depth study of the site can be conducted. Once site activity has been begun within six months and a site has been categorized as time critical, it does not change category to non-time-critical regardless of when the response is completed.

Non-time-critical removal actions are those where, based on the site evaluation, the lead agency determines that a removal action is appropriate and that a planning period of more than six months is available before on-site activities must begin. For nontime-critical removals, extensive data collection and analysis is conducted to more completely document the actual or potential health and environmental threat. The lead agency for non-timecritical removals will undertake an engineering evaluation/cost analysis (EE/CA) or its equivalent. EE/CAs contain evaluations of

Unmi

possible alternative technologies, selection of the response, and documentation of the decision-making process. EE/CAs use a screening process and analysis of removal options based upon such factors as technical feasibility, institutional considerations, reasonableness of cost, timeliness of the option with respect to threat mitigation, environmental impacts, and the protectiveness of the option. This information will be subject to review and comment by the public prior to initiation of the affected removal. Non-time-critical removal actions include activities such as containment, treatment, disposal, or incineration of hazardous materials, or stabilization of berms, dikes, or impoundments.

Interim Final Guidance on Removal Action Levels at Contaminated Drinking Water Sites



UNITED STAFES ENVIRONMENTAL PROFECTION AGENCY WASHINGTON CC 20180

6 1987 **ACT**

UFFICE OF SOLID WASTE AND EWERGENC + RESPONSE

MEMORANDUM

- SURJECT: Interim Final Guidance on Removal Action Levels at Contaminated Drinking Water Sites (OSWER Directive 9360.1-10)
- FROM: Henry L. Longest II, Director Office of Emergency and Remedial
- TO: Waste Management Division Directors, Regions I-X Environmental Services Division Directors, Regions I, VI and VII

(s)DMe

Attached is the interim final guidance on removal action levels at drinking water contamination sites. This guidance is effective immediately. The EPA Office of Drinking Water (ODW) and the Agency for Toxic Substances and Disease Registry (ATSDR) will also be developing guidance documents related to drinking water contamination in the near future. OERR will be working with ODW and ATSDR in an attempt to ensure a consistent approach between the agencies.

Questions or comments regarding this guidance or appropriate removal action levels should be directed to Jean Schumann, Emergency Response Division, 44-5488, FTS 382-4671.

Attachment

Jack McGraw CC: Tom Devine Gene Lucero Marcia Williams Walt Kovalick Tim Fields, ERD Russ Wyer, HSCn Steve Lingle, HSED Art Weissman, PAS Michael Cook, ODW Mark Greenwood, OGC William Farland, CAG Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X Mark Bashor, ATSDR

REMOVAL ACTION LEVELS FOR CONTAMINATED DRINKING WATER SITES

Introduction

The purpose of this guidance is to establish "action levels" for providing alternate water supplies under Superfund removal authority at contaminated drinking water sites. The action level is the primary criterion that must be met for a site to qualify for removal response. The action levels established in this guidance must generally be satisfied before removal authority can be used at either National Priorities List (NPL) sites or non-NPL sites.

Under the 1982 National Contingency Plan (NCP), removal actions were taken in response to "immediate and significant" threats to human health or the environment. The removal program used the 10-Day Health Advisory as the principal benchmark to identify those drinking water contamination incidents that posed the most acute threats to human health. The November 1985 NCP broadened removal authority by authorizing response in situations that present a "threat" to human health or the environment. Therefore, removal actions may now be taken in less urgent situations than under the 1982 NCP.

In response to this expansion of removal authority, the Office of Emergency and Remedial Response (OERR) is revising removal program action levels for contaminated drinking water sites. This guidance expands the previous policy in a number of ways. First, the numeric action levels are now based on levels that are protective for a lifetime exposure rather than a 10-day exposure. Second, both carcinogenic and non-carcinogenic health effects are considered. Third, a reduction factor is used for volatiles to account for exposure due to inhalation. Finally, additional guidance is provided on the use of site-specific factors to trigger removal actions.

The action levels established in this guidance allow a site to qualify for removal response if either: 1) the numeric trigger is exceeded at the tap, or 2) site-specific factors otherwise indicate that a significant health threat exists. The guidance also discusses information sources on health threats from drinking water contamination, factors to consider in determining the extent of action, action levels vs. cleanup standards, prioritizing removal sites, and obtaining exemptions to the statutory limits for alternate water supply sites.

Action Level Based on Numeric Trigger

The numeric trigger is calculated using a model that establishes four different action levels, depending on whether the substance is also a potential human carcinogen and/or volatile. The model is explained below and summarized in <u>Exhibit 1</u>. Based on this model, <u>Exhibit 2</u> lists the numeric action level for various substances that may be found in drinking water at Superfund sites. A site may qualify for removal response if the numeric trigger for the drinking water contaminant is exceeded at the tap of at least one residence ("residence" includes schools, businesses, etc.). (Note that the decision to initiate a removal action is based on other factors as well, such as the availability of other response mechanisms to initiate action in a timely manner.) The first step in calculating the numeric trigger is determining whether the substance of concern is also a potential human carcinogen and/or volatile. For purposes of this guidance, a substance is a carcinogen if it falls into categories A, B, or C of EPA's carcinogen classification guidelines. (A substance should be considered a non-carcinogen if it is in categories D or E.) <u>Volatile</u> organic chemicals (VOCs) are generally of low molecular weight, high vapor pressure, and low solubility. For purposes of this guidance, VOCs include those chemicals identified as volatiles in the following documents: Test Methods for Evaluating Solid Waste, Vol. IA, SW-846, 3rd ed., November 1986 (Chapter 2); Contract Lab Program Statement of Work, October 1986 (Exhibit C); Methods for the Determination of Organic Compounds in Finished Drinking Water and Raw Source Water, September 1986 (available from Regional water program offices); and 40 CFR Part 264, Appendix IX (analytical methods 8010 and 8240 designate volatiles).

With the substance thus classified, the second step is to determine the appropriate action level in accordance with the categories below:

- <u>Non-volatile non-carcinogens</u> -- Action level equals the Drinking Water Equivalent Level (DWEL).*
- 2. Volatile non-carcinogens -- Action level equals 50 percent of the DWEL.
- Non-volatile carcinogens -- Action level is determined by comparing the DWEL to the 10⁻⁴ Lifetime Upperbound Cancer Risk Level, and choosing the lower of the two.
- Volatile carcinogens -- Action level is determined by comparing 50 percent of the DWEL to the 10⁻⁴ Lifetime Upperbound Cancer Risk Level, and choosing the lower of the two.

The action level for methylene chloride, for example, is calculated as follows. Methylene chloride is a volatile and a potential human carcinogen (classified as a "B2" under EPA guidelines). The DWEL for methylene chloride equals 1750 ppb and the 10^{-4} Cancer Risk Level equals 48 ppb. The action level is determined by comparing 50 percent of the DWEL, or 875 ppb, to the 10^{-4} Cancer Risk Level, or 48 ppb, and choosing the lower of the two, which is $\frac{48 \text{ ppb}}{\text{ppb}}$. If at least one residence has methylene chloride levels that exceed 48 ppb at the tap, the site may qualify for removal response.

This model will provide an action level for many of the substances commonly encountered in drinking water at Superfund sites, including many solvents. However, OERR is still working on establishing an appropriate action level for certain substances in the two situations described below. Until action levels are developed, most decisions regarding these substances will be made in OERR. The modifications discussed below have been incorporated into Exhibits 1 and 2.

The calculated action level for a substance is lower than or equal to the Maximum Contaminant Level (MCL) established under the Safe Drinking Water Act (SDWA). For example, for vinyl chloride, a volatile carcinogen, the calculated action level under this model is 1.5 ppb (1.5 ppb is the 10⁻⁴ Cancer Risk Level, which is lower than 50% of the DWEL). However, 1.5 ppb is lower than the MCL for vinyl chloride, which is 2 ppb. Given the limited scope of the removal program, it may not be appropriate for the removal program to trigger removal action at levels equal to or below the MCL. Therefore, OERR is currently examining whether it would be appropriate to establish an alternate action level for these substances that is above the MCL. Until an action level is established for these substances, removal action may be initiated if contaminant levels exceed the 10-Day Health Advisory. However, if contaminant levels are between the calculated action level and the 10-Day Health Advisory, OERR will review individual site conditions to determine if removal action should be taken.

The calculated action level is based on the DWEL, but the 10-Day Health Advisory is lower than the DWEL. For most substances, the 10-Day Health Advisory is higher than the DWEL. In some cases, however, the 10-day advisory is lower than the DWEL. (This situation occurs primarily where 10-day exposure data were not available, so the 10-Day Health Advisories were based on other studies.) For example, the action level for barium (a non-volatile non-carcinogen) is based on the DWEL of 1800 ppb, but the 10-Day Health Advisory for barium is 1500 ppb. OERR is currently examining whether it would be appropriate to use the lower 10-day advisories as the removal action level. Until OERR determines if an alternate action level is appropriate for these substances, removal action may be initiated if contaminant levels exceed the DWEL. However, if contaminant levels are between the (lower) 10-Day Health Advisory and the DWEL, OERR will review individual site conditions to determine if removal action should be taken.

Action Level Based on Site-Specific Factors

A significant health threat may exist even though the numeric action level has not been exceeded. A removal action may be initiated if the health risk at a site has been <u>analyzed in detail</u> and the analysis indicates that a serious health risk is present due to site-specific factors. Examples of such factors include evidence that a contaminated groundwater plume is moving, contaminant levels will likely increase (e.g., increased pumping from an aquifer anticipated during summer months), people have been drinking contaminated water for a long period of time, multiple contaminants are likely to result in synergistic effects, there are sensitive members in the population at risk, etc.

With regard to a threat based on future contamination, as a general rule, removal action may be warranted where it can be projected that the numeric action level will be exceeded within 6 months. It is important to note that this 6 month period is not related to the definition of time-critical/non-timecritical removal actions. For example, where contaminant levels will likely exceed the DWEL by a significant amount within 6 months, a time-critical removal action would be appropriate. However, if contaminant levels will only exceed the DWEL by a minimal amount within 6 months, a non-time-critical removal action may be more appropriate. Future threat may therefore warrant either a time-critical or non-time-critical removal action. When conditions such as those described above are present, the site may qualify for removal action even though a numeric indicator has not been exceeded. Decisions will be made on a case-by-case basis. OERR concurrence must be obtained before approving Action Memoranda for contaminated drinking water sites where the removal action decision is based solely on site-specific factors, even where site cost or time projections do not exceed the statutory limits on removal actions. However, if an <u>emergency</u> exists based on sitespecific factors, action may be initiated immediately and OERR should be contacted as soon as possible.

Information Sources

DWELS, as well as RfDs and other relevant standards and advisories, are available to the Regions through the Integrated Risk Information System (IRIS). IRIS can be accessed on-line through E-mail; type in "IRIS" at the prompt rather than "mail." The EPA Office of Drinking Water has also established a Safe Drinking Water Hotline, which can provide information about relevant standards and criteria, and treatment techniques for contaminated drinking water. The Hotline telephone number is 800-426-4791 (in the Washington D.C. area, 382-5533).

Additional advice and information on health assessments at drinking water contamination sites may be obtained from the Agency for Toxic Substances and Disease Registry (ATSDR) and the Superfund Public Health Evaluation Manual (OSWER Directive #9285.4-01). ATSDR may be particularly helpful in providing advice on threats posed by site-specific factors.

OERR should be contacted if a substance of concern does not have a DWEL, RfD, and/or cancer risk level.

Determining the Extent of Action

Once it has been determined that a site qualifies for removal response based on a numeric trigger or site-specific factors, the Region must determine how many residences (including businesses, schools, etc.) will receive alternate water supplies. First, the area of impact should be estimated (both extent and magnitude of the threat) by considering factors such as the hydrogeology of the site, plume movement, and the likelihood of contaminant levels increasing. For sites where removal action is warranted because the numeric trigger has been exceeded at certain residences, the area of impact may be defined to include neighboring residences which are at risk, but do not exceed the numeric trigger.

After the area of impact is defined, the number of residences to be provided with alternate water supplies must be determined by considering cost vs. benefits received, the statutory limits on removal actions, and the availability of other response mechanisms. For example, response to widespread low-level contamination may be too extensive for removal action, and therefore, may be addressed more appropriately by the remedial program. In another case, a contaminated aquifer may affect a public water supply system and private wells, but Superfund resources may only be needed to address the private wells.

Determining the appropriate extent of action therefore involves analysis of both the area of impact and programmatic factors.

Action Levels vs. Cleanup Standards

The numeric action levels established in this guidance are not intended to be used as cleanup standards. The MCL, if available, will generally be the appropriate cleanup standard. (For guidance on the use of MCLs and MCLGs as cleanup standards, see "Interim Guidance on Compliance with Applicable or Relevant and Appropriate Requirements," July 9, 1987, OSWER Directive 9234.0-05. Final guidance will be issued in the CERCLA Compliance with ARARs Manual.) This means that for any residence provided with an alternate water supply, the goal will generally be to meet MCLs. For example, if carbon filter units will be provided to treat drinking water contaminated with trichloroethylene (TCE), treated water should achieve 5 ppb TCE, the MCL.

Prioritizing Removal Sites

Sites may qualify for removal action under either the numeric indicator or site-specific factor approaches. For the purpose of <u>prioritizing</u> those sites that qualify for removal action, response should be initiated as soon as possible if contaminant levels exceed the 10-Day Health Advisory or site-specific factors otherwise indicate that an emergency exists.

Exemption to the Statutory Limits

To obtain an exemption to the \$2 million/12 month limits on removal actions based on a continuing <u>emergency</u>, it will generally not be adequate to show that contaminant levels exceed the numeric action level by some minimal amount. An exemption may be justified if contaminant levels exceed the 10-Day Health Advisory, significantly exceed the numeric action level, or an emergency exists based on site-specific factors. A finding that contaminant levels exceed the numeric action level by a minimal amount may be appropriate, however, in "non-emergency" situations where an exemption is based on the new <u>consistency</u> waiver.

Summary of Policy

A contaminated drinking water site may qualify for removal response if: 1) the numeric action level (based on the DWEL and/or the 10^{-4} Lifetime Upperbound Cancer Risk Level) is exceeded, or 2) site-specific factors otherwise indicate the presence of a serious health threat. In prioritizing those sites that qualify for response under this model, Regions should give priority to sites where contaminant levels exceed the 10-Day Health Advisory or site-specific factors otherwise indicate that an emergency exists. Exhibit 1: Summary of Action Level Decision Model

No contaminant levels exceed the NUMERIC action level?

Is the substance a volatile and/or potential human carcinogen?

- Non-volatile non-carcinogens -- Action level equals the DWEL.
- Volatile non-carcinogens -- Action level equals 50% of the DWEL.
- $^\circ$ Non-volatile carcinogens -- Action level is determined by comparing the DWEL to the 10^{-4} Lifetime Upperbound Cancer Risk Level, and choosing the lower of the two.
- $^{\circ}$ Volatile carcinogens -- Action level is determined by comparing 50% of the DWEL to the 10⁻⁴ Lifetime Upperbound Cancer Risk Level, and choosing the lower of the two.

Do either of the two modifications to the numeric action level apply?

Is the numeric action level lower than or equal to the MCL, if available? If yes:

- * If contaminant levels are between the numeric action level and the 10-Day Health Advisory, contact DERR to determine appropriate action.
- [°] If contaminant levels exceed the 10-Day Health Advisory, action may be taken if the site otherwise qualifies for removal response.

If the action level is based on the DWEL, is the 10-Day Health Advisory lower than the DWEL? If yes:

- * If contaminant levels are between the (lower) 10-Day Health Advisory and the DWEL, contact OERR to determine appropriate action.
- If contaminant levels exceed the DWEL, action may be taken if the site otherwise qualifies for removal response.

If contaminant levels do not exceed the numeric trigger, can the site qualify for removal response based on SITE-SPECIFIC FACTORS?

A site can qualify for removal response if the health risk at a site has been analyzed in detail and the analysis indicates that a serious health risk is present due to site-specific factors.

- ATSDR may be particularly helpful in providing advice on health risk due to site-specific factors.
- OERR concurrence must be obtained before approving Action Memoranda based on site-specific factors, even where the site will not exceed the statutory limits on removal actions.
Exhibit _

FOR CONTAMINATED DRINKING WATER SITES

(ug/L)

and the second							
Chemical	Volatile (Y/N)	EPA Carcinogen Group ^a	MCL	10-Nay . HA	DWELD	10-4 Cancer Risk Level	Removal Action Level
Alachlor	N	B2	None	100	350	15	15
Barium	N	D	1000	1500 ^C	1800	NA	1800 ^d
Benzene	Y	A	5	235	NA	120	120
Cadmium	N	D	10	43 ^c	17	NA	17
Carbofuran	N	. E	None	50 ^C	175	NA	17 5 ^d
Carbon tetrachloride	Y	В2	5	160	24	27	12
Chlordane	N	B2	None	63	1.6	2.7	1.6
Chlorobenzene	Y	D	None	4300 ^c	1505	NA	753
Chromium (total)	N	D	50	1400	168	NA	168
Cyanide	N	D	None	220 ^c	770	NA	770 ^d
o-Dichlorobenzene	Y	D	None ,	8930 ^c	3115	NA	1558
p-Dichlorobenzen e	Y	C	75.	16700 ^c	3500	175	175
1,2-Dichloroethane	Y	B2	5	740 ^c	None	38	38
l,1-Dichloroethylene	Y	c	7	1000 ^c	350	None	175
Cis-1,2-Dichloroethylene	Y	D	None	1000 ^c	350	NA	175
Trans-1,2-Dichloroethylene	Y	D	None	1430 ^c	350	NA	175
Dichloro ane/Methylene chlori	Y	B2	None	1500	1750	48	48

FOR CONTAMINATED DRINKING WATER SITES

(ug/L)

Chemical	Volatile (Y/N)	EPA Carcinogen Group ^a	MCĹ	10-Day HA	DWELD	10-4 Cancer Risk Level	Removal Action Level
Endrin	N	Ε	0.2	5	1.6	NA	1.6
Ethylbenzene	Y	D	None	3200 ^C	3395	NA	1698 ^d
Heptachlor	N	82	None	10	17	7.6	7.6
Lindane	N	C	4	1200	10	None	10 .
Mercury (inorganic)	Ne	. D	2	1.6 ^c	5.5	NA	5.5f
Methoxychlor	N	D	100	2000	1750	NA	1750
Methyl ethyl ketone (MEK)	Y	D	None	7500 ^c	864	NA	. 432
Nickel	N	D	None	1000	350	NA	350
Pentachlorophenol (PCP)	Y	D	None	300 ^c	1050	NA	525 ^d
Styrene	Y	C	None	2000 ^C	7000	None	3500 ^d
Tetrachloroethylene (PCE)	Y	B2/C	None	2000	500	66	66
Toulene	Y	D	None	3460 ^C	12100	NA	6050 ^d
Toxaphene	N	B2	5	40	None	3.1	409
1,1,1-Trichloroethane	Y	D	200	35000 ^c	1000	NA	500
Trichloroethylene	Y	R2	5	None	257	280	128
Vinyl chloride	Y	A	2	2600	None	1.5	1 300 ^h
Xylenes (total)	Y	D	None	7800c	2157	NA	1078

REMOVAL NUMERIC ALIION LEVELS

FOR CONTAMINATED DRINKING WATER SITES

(ug/L)

Chemical	Yolatile (Y/N)	EPA Carcinogen Group ^a	MCL	10-Nay HA	DWELD	10-4 Cancer Risk Level	Removal Action Level
							Hu.

^a Carcinogen group designation is from EPA carcinogen classification guidelines for effects from ingestion.

^b DWEL = RfD x $\frac{70 \text{ kg}}{2 \text{ 1/day}}$. (Note that the DWEL in health advisory documents produced by EPA's Office of Drinking Water 2 $\frac{1}{2 \text{ adv}}$ may be slightly different due to rounding.)

- ^C Because no suitable studies of appropriate duration were available, these 10-Day Health Advisories were based on Health Advisories of greater or lesser duration, e.g., 1-Day, Longer-term, and Lifetime Health Advisories.
- ^d Removal action level is an interim value. OERR is examining whether it would be appropriate to use the lower 10-Day Health Advisory (50% for volatiles) as the action level. Until that time, if contaminant levels levels exceed the action level shown in the table, removal action may be taken. If contaminant levels exceed the 10-day advisory (50% for volatiles), but not the DWEL (50% for volatiles), consult OERR.
- e Not soluble in water.
- f Removal action may be initiated if mercury levels exceed the DWEL of 5.5 ug/L. If mercury levels exceed the 10-day advisory of 1.6 ug/L, but not 5.5 ug/L, consult OERR.
- 9 Removal action may be initiated immediately if toxaphene levels exceed the 10-Day Health Advisory of 40 ug/L. If toxaphene levels exceed the 10⁻⁴ Cancer Risk Level of 3.1 ug/L, but not 40 ug/L, consult OERR.
- ^h Removal action may be initiated immediately if vinyl chloride levels exceed 1300 ug/L, which is 50% of the 10-Day Health Advisory. If vinyl chloride levels exceed the 10⁻⁴ Cancer Risk Level of 1.5 ug/l, but not 1300 ug/L, consult OERR.

NA = Not appropriate.

1

Information on Drinking Water Action Levels

DRUCE Engelber



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

APR 19 :coc

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

MEMORANDUM

SUBJECT: Information on Drinking Water Action Levels Timothy Fields, Jr., Director finally fully M. Emergency Response Division FROM:

TO: Superfund Branch Chiefs, Regions I-X OHM Coordinators, Regions I-X

The purpose of this memorandum is to to provide you with updated information on removal program drinking water action levels, as described in OSWER Directive 9360.1-10. The OSWER Directive will eventually be amended to include this information, but it is being sent to you now to ensure that you receive the data as soon as possible. It is important to note that the drinking water action level is only one of several factors to consider in deciding whether a removal action is appropriate.

- ^o <u>Dichloromethane</u> Change the 10⁻⁴ Cancer Risk Level to 480 ppb (not 48), and the removal action level to 480 ppb.
- Alachlor Change the 10⁻⁴ Cancer Risk Level to 44 ppb (not 15), and the removal action level to 44 ppb.
- Arsenic ERD has had several requests about arsenic action levels. At the moment, there is considerable controversy within EPA and the scientific community about the carcinogenicity of arsenic. For now, rather than establishing an official action level, the removal program may consider taking action when arsenic levels exceed 50 ppb (the current MCL), but the Regions should always consult Headquarters for arsenic sites. (You must also consider whether the arsenic is naturally occurring, since SARA prohibits Superfund response to such releases unless it is an emergency and no one else can respond.)
- <u>DBCP</u> The 10⁻⁴ Cancer Risk Level for DBCP is 2.5 ppb and the 10-Day Health Advisory is 50 ppb. The Agency is currently working on a proposed MCL for DBCP. The removal action level at this time will be 2.5 ppb, based on the cancer risk value, but it is important to keep in mind that Superfund should not be used to respond to releases that result from the lawful application of pesticides (unless an emergency exists). Many farming areas may show elevated levels of DBCP, but this may have been caused by normal use of the pesticide. Two items related to this are attached: 1) a July 1985 memo that addresses releases from lawfully applied pesticides and 2) a recent memo written by an OSC in Region 9, after consultation with Headquarters, regarding DBCP contamination.

Vinyl chloride - Many OSCs have been concerned because the action level guidance provides that the Regions should not implement action for vinyl chloride unless concentrations exceed 1300 ppb [50% of the 10-Day Health Advisory). This does not mean, however, that alternate water supplies cannot be provided at lower levels for vinyl chloride. To clarify the current policy for vinyl chloride, the action level for immediate Regional response is 1300 ppb, but Headquarters may authorize action at lower levels. Specifically, Headquarters will consider response if levels exceed 2 ppb, the MCL. In determining whether action is appropriate, Headquarters will consider factors such as the source of the contamination, the scope of the response needed, and the availability of other response mechanisms.

Headquarters consultation is required for these actions because the action level calculated under the general model (which would equal 1.5 ppb, the 10⁻⁴ Cancer Risk Level) is <u>lower</u> than the MCL of 2 ppb. The removal program needs to be careful about setting precedents for taking actions at or below the MCL, because any public water supply system in the country could then potentially qualify for removal response if contaminant concentrations exceeded the MCL. In general, public water supply systems are responsible for ensuring that their systems comply with MCLs.

- [•] Chloroform The DWEL is 350 ppb, the 10^{-4} Cancer Risk Level is 600 ppb, and the removal action level is 175 ppb (50% of the DWEL).
- List of removal alternate water supply sites ERD recently sent the Regions a table prepared by TAT listing sites where the removal program has provided alternate water supplies in the past. Please change the following information on the Region 5 Main Street Wellfield site: the number of residences affected is 301, not 40,000, and the methods used include bottled water, carbon filtration, and water main connection in addition to air stripping.
- Risk additivity At the recent "Removal Program Managers" meeting in Washington, D.C., ERD was requested to send the Regions information about adding risks from multiple chemicals in drinking water. In trying to collect this information from EPA sources, we found that there was some inconsistency in risk additivity policy within the Agency. The Superfund Toxics Integration Branch (TIB) will address this issue when they revise the Superfund Public Health Evaluation Manual this year. In the interim, risks from multiple contaminants in the drinking water should not be routinely summed in establishing action levels. However, if the Region is particularly concerned about possible synergistic effects or effects on the same target_organ from the contaminants present, the Regions may contact the Superfund TIB for further assistance (FTS 475-9486). Information on health effects from various chemicals is available from the Integrated Risk Information System (IRIS).

If you have any questions regarding this information, you may contact Jean Schumann of my staff at FTS 382-4671.

Attachments

cc: Mark McClannahan, ATSDR Hans Crump Paul Nadeau Dave Bennett John-Riley Bruce Engelbert Joe LaFornara

_

Policy on Flood Plains and Wetlands Assessment

•

03/19/87 United States En Wast	1. Directive Number						
EPA OSWER Dire	9280.0-02						
	2. Originator in	ormation					
Name of Contact Person SMITH	Mail Code	Office OERR/PAS	Telephone Number 382-3300				
3. THE POLICY ON FLOOD PL	1. The POLICY ON FLOOD PLAINS AND WETLANDS ASSESSMENTS						
4. Summary of Directive (include brief state Discusses situations the flood plains or wetland factors which should be assessment for response pursuant to section 104 (8/85, 12 pp)	hat require is assessment considered actions un con 106 of	preperation of at, and the l in preparing a dertaken CERCLA.	a n				
9280.0-1.	9280.0-1.						
5. Keywords SUPERFUND, CERCLA, RE	MEDIAL PROG	RAM, FLOOD PLAIN	ns management				
6a. Does this Directive Supercede Previous	Directive(s)?	yee X No	What directive (number, title)				
b. Does it Supplement Previous Directives(s)? X yes No What directive (number, title) 9280.0-01							
7. Draft Level A - Signed by AA/DAA B - Sig This Bequest Mests OSWER Directives Syste	ned by Office Direct	or C - For Review &	Comment in Development				
8. Signature of Laad Office Directives Coord)	netor .		Date				
9. Name and Title of Approving Official HEDEMAN/LUCERO			Date 08/06/85				
OSWER	OSW	'ER	OSWER				

DIRECTIVE DIRECTIVE



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460



MEMORANDUM

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE OSWER Directive 9280.0-02

SUBJECT: Policy on Floodplains and Wetland Assessments for CERCLA Actions

FROM: William N. Hedeman, Jr., Director JU Milliam N. Hedeman, Jr., Director JU Milliam M. Hedeman, Jr., Director Gene Lucero, Director Office of Waste Programs Enforcement

TO: Toxic and Waste Management Division Directors Regions I-X

Response to releases of hazardous substances is often affected by floodplain and wetland issues. Under this policy Superfund actions must meet the substantive requirements of the Floodplain Management Executive Order (E.O. 11988), and the Protection of Wetlands Executive Order (E.O. 11990) (see attached), and Appendix A of 40 CFR Part 6, entitled Statement of Procedures on Floodplain Management and Wetland Protection. The purpose of Appendix A of 40 CFR Part 6 is to set forth EPA's policy and guidance for carrying out the provisions of Executive Orders 11988 and 11990. This memo discusses situations that require preparation of a floodplains or wetlands assessment, and the factors which should be considered in preparing an assessment, for response actions undertaken pursuant to section 104 or 106 of CERCLA.

For removal actions, the on-scene coordinator (OSC) must consider, to the extent practicable, taking into account the exigencies of the situation, the effect the response action will have on floodplains and wetlands. For remedial actions, a floodplain/wetlands assessment must be incorporated into the analysis conducted during the planning of the remedial action.

I. BACKGROUND

A. Floodplains

Floodplains are relatively flat areas or lowlands adjoining the channel of a river, stream or water course which have been or may be covered by floodwater. A flood is a general and temporary condition of partial or complete inundation of normally dry land areas from the overflow of inland and/or tidal waters and/or the unusual and rapid accumulation or runoff of surface waters from any source. A reference to a floodplain should be accompanied by a modifier indicating the level of flooding, e.g., 100-year floodplain (one percent chance of flooding in any year).

Executive Order 11988 - Floodplain Management

Executive Order 11988 requires Federal agencies carrying out their responsibilities to take action to reduce the risk of flood loss, to minimize the impact of floods on human safety, health and welfare, and to restore and preserve the natural and beneficial values served by floodplains. To do this, Federal agencies must evaluate the potential effects of any actions they may take in a floodplain to ensure that their planning programs and budget requests reflect consideration of flood hazards and floodplain management, including the restoration and preservation of such land areas as natural undeveloped floodplains. This order emphasizes the importance of evaluating alternatives to avoid effects and incompatible development in the floodplains, of minimizing the potential harm to floodplains if the only practicable alternative reguires siting an action in a floodplain and providing early and adequate opportunities for public review of plans and proposals involving actions in floodplains.

B. Wetlands

Wetlands are land areas which, because of their frequent inundation by surface or ground water, can support vegetative or aquatic life that requires saturated soil conditions. Wetlands generally include but are not limited to swamps, marshes, bogs and similar areas such as sloughs, pot holes, wet meadows, river overflows, mud flats and natural ponds.

Executive Order 11990 - Protection of Wetlands

Executive Order 11990 requires Federal agencies in carrying out their responsibilities to take action to minimize the destruction, loss or degradation of wetlands, and to preserve and enhance the natural and beneficial values of wetlands. The order emphasizes the importance of avoiding undertaking new construction located in wetlands unless there is no practicable alternative to that construction, minimizing the harm to wetlands if the only practicable alternative requires construction in the wetland, and providing early and adequate opportunities for public review of plans and proposals involving new construction in wetlands.

C. Statement of Procedures on Floodplain Management and Wetlands Protection - Appendix A to 40 CFR Part 6

EPA has promulgated regulations implementing procedures on the National Environmental Policy Act (NEPA) at 40 CFR Part 6. Appendix A of Part 6 (Appendix A) deals with procedures on Floodplain Management and Wetland Protection. The purpose of Appendix A is to set forth Agency policy and guidance for carrying out the provisions of Executive Orders 11988 and 11990.

Appendix A provides that it is the intent of these Executive orders that, wherever possible, Federal agencies implement the floodplains/wetlands requirements through existing procedures, such as those internal procedures established to implement NEPA. In those instances where the environmental impacts of a proposed action are not significant enough to require an environmental impact statement (EIS) pursuant to section 102(2)(C) of NEPA, or where programs are not subject to the requirements of NEPA, alternative but equivalent floodplain/wetlands evaluation and public comment and notice procedures must be established. Furthermore, Appendix A prescribes the requirements for floodplain/wetlands review of proposed EPA actions.

II. POLICY

A. Removal Actions

Removal actions are exempt from compliance with section 102(2)(C) of NEPA because there is a fundamental conflict in statutory purpose between EIS requirements and EPA's removal authority. This conflict arises from the fact that it would be virtually impossible for EPA to follow the lengthy EIS process and at the same time expeditiously undertake removal actions.

1. Floodplain/Wetland Assessment

However, a floodplains/vetlands evaluation required by Appendix A would not be as lengthy as the EIS process. Therefore, the OSC or lead Agency should attempt to incorporate a floodplains/wetlands assessment into the preliminary assessment for the removal action. The floodplains/wetlands assessment must consider the following: whether or not the action will be located in or affect a floodplain or wetland; the impact of the action on the floodplain or wetland; the alternatives available; and measures to minimize potential harm to the floodplain or wetland if there is no practicable alternative to locating in or affecting the floodplain or wetland [for a more detailed explanation of these factors see Section III, Remedial Actions, of this policy}. However, because removal actions often involve situations requiring expeditious action to protect public health, welfare or the environment, it may not always be feasible to perform a floodplains/wetlands . assessment. In those circumstances where a floodplain/wetland assessment cannot be performed, the OSC report or other

-4-

documents should specify the reasons. At the OSC's discretion, considering the exigencies of the situation, the OSC should consult with the Regional 404 Staff where wetlands/floodplains are involved or suspected to be involved.

For all lead agency removal actions where a floodplain/wetland assessment is performed and is proposed to be in or affecting a floodplain/wetland the OSC shall document the decision in the OSC report. The decision shall be accompanied by a Statement of Findings, not to exceed three pages that includes (i) the reasons why the proposed action must be located in or affects the floodplain/wetlands; (ii) a description of significant facts considered in making the decision to locate in or to affect the floodplain or wetland including alternative sites and actions; (iii) a statement indicating whether the proposed action conforms to applicable State or local floodplain/wetland protection standards; (iv) a description of the steps taken to design or modify the proposed action to minimize potential harm to or within the floodplain or wetland; and (v) a statement indicating how the proposed action affects the natural or beneficial values of the floodplain or wetland.

2. Opportunity for Citizen Comment

Appendix A has two public notice requirements. One is public notice when it is apparent that a proposed or potential agency action is likely to impact a floodplain or wetland and the other is public notice of the selected decision.

Because of the expeditious nature of removal actions extending 45 days or less, no formal community relations plan must be developed. Instead, a spokesperson will be designated by the lead agency to inform the community of actions being taken, to respond to inquiries and to provide information concerning the release. If the exigencies of the situation permit the performance of a floodplain/wetland assessment, the assessment must be included in the spokesperson's presentation. This will provide early public notice as required by Appendix A.

The OSC report, which contains the selected decision or the reasons why a floodplain/wetland assessment cannot be performed, must also be made available to the public. The OSC report will provide public notice of the selected decision as required by Appendix A.

If the required removal action extends over 45 days, a formal community relations plan must be developed. If the exigencies of the situation allow for a floodplain/wetland assessment, this assessment must be made available for a three week public comment period. This will provide early public notice and an opportunity for participation in the decisionmaking process as required by Appendix A.

- 5 -

If it is known that a floodplain/wetland assessment will be conducted at the time of the preparation of the Community Relations Plan than the public comment period must be noted in the plan. The OSC report, which contains the selected decision or the reasons why a floodplain/wetland assessment cannot be done, is also required for the "longer" removals and must be made available to the public. This will provide public notice of the selected decision as required by Appendix A.

B. Remedial Actions

An EIS is unnecessary for remedial actions provided in that EPA meets the standards for a functional equivalent exception to the EIS requirements of section 102(2)(C) of NEPA. To comply with the functional equivalent exception, the agency must have expertise in environmental matters and meet the following criteria. First, the agency's authorizing statute must provide substantive and procedural standards that ensure full and adequate consideration of environmental issues. Second, the agency must afford an opportunity for public participation in the evaluation of environmental factors prior to arriving at a final decision.

1. Consideration of Environmental Issues

Remedial actions satisfy the first criterion for a functional equivalent exception because of the mandate for environmental assessment contained in section 104 of CERCLA and the procedural safeguards developed by EPA for the remedial planning process. The language in section 104, that directs that remedial actions be necessary to protect public health, welfare, and the environment, establishes a standard mandating consideration of environmental effects. Moreover, the procedures set forth in the National Contingency Plan (NCP) establish a process for conducting an analysis during the planning of remedial actions that is similar in content to the evaluation underlying an EIS. This analysis is contained in the remedial investigation/feasibility study (RI/FS). Therefore, for a remedial action to comply with the alternative but equivalent floodplain/wetland evaluation contained in Appendix A of 40 CFR Part 6, a floodplain/wetlands assessment must be incorporated into the analysis conducted during the planning of remedial actions which is established by the NCP.

During the scoping of remedial response actions, the Remedial Project Manager (RPM) or the lead Agency in conjunction with Regional 404 staff, should identify any floodplain or wetlands located within the site area or that could be affected by the response action. If the area is predominantly privately owned, the RPM or the lead agency shall consult with the Federal Insurance Administration of the Federal Emergency Management Agency which has two maps that will be useful in identifying floodplains. The Flood Insurance Rate Map shows the boundaries and elevations of the 100 and 50C years floodplains. The other map, Flood Hazard Boundary Map, shows the appropriate area of the 100 years zone. A copy of these maps can be obtained by calling 1-800-638-6620. For areas predominately State or Federally owned, consult with the controlling Federal or State agency. Maps are available for some wetland areas from the Fish and Wildlife Service (National Wetlands Inventory Maps) or from local and State planning agencies. Also, the Regional 404 staff has access to the most up to date wetlands area information.

If there are no floodplains/wetlands located within the site area or that could be affected by a response action, the feasibility study should so state, and the response action may proceed without further consideration of the procedures set forth below. However, if the site is located within a floodplain/wetland or if the proposed remedial action would affect a floodplain/wetland, the RPM or the lead agency must conduct a floodplain/wetland assessment which will be integrated into the feasibility study. In the RPM's discretion, the RPM should consult with the Regional 404 staff in cases that require a floodplains/wetlands assessment. Floodplain/Wetland assessments shall consist of a description of the proposed action, a discussion of its effect on the floodplain/wetlands, a description of the alternatives considered and their effects on the floodplains and wetlands, and measures to minimize potential harm to the floodplaih/ wetland if there is no practicable alternative to locating in or affecting floodplain/wetlands.

a. Floodplain Assessment Of Alternatives

In assessing the alternatives and their effects on the floodplain and floodplain protection, the RPM or lead agency should consider such factors as environmental effects, community welfare, cost and technology. All possible alternatives must be considered, including the no action alternative. If one or more of the alternatives will be located in a floodplain, those alternatives may not be selected unless a determination is made that no practicable alternatives exists outside the floodplain.

If no practicable alternatives exist outside the floodplain, and the RPM or lead agency has determined or proposes to allow a remedial action to be located in a floodplain, then the RPM or lead agency shall act to minimize potential harm or avoid adverse effects to the floodplain. This includes acting to restore and preserve the natural and beneficial values of floodplains. The benefits of preserving floodplains in their natural or relatively undisturbed state include not only reduction of flood hazards, but maintenance of water guality standards, replenishment of ground water, soil conservation, the fostering of fish, wildlife and plant resources and the provision of recreational areas.

The following are possible methods for minimizing potenti harm to floodplains. This list, however, does not preclude the RPM or lead agency from using other measures that minimize potential harm or avoid adverse effects to floodplains.

-7-

- 1. Use minimum grading requirements.
- 2. Return the site to natural contours.
- 3. Maintain floodplain vegetation to reduce sedimentation.
- 4. Regulate methods used for grading, filling, soil removal and replacement to reduce sedimentation.
- 5. Require topsoil protection program.
- 6. Raise the site above the floodplain.
- 7. Construct new structures or facilities in floodplains in accordance with accepted floodproofing and other flood protection measures and elevate structures above the base flood level rather than filling inland, wherever practicable.

b. Wetland Assessment Of Alternatives

In assessing the alternatives and their effects on wetlands, the RPM or lead agency in conjunction with the Regional 404 staff, should consider such factors as environmental effects, community welfare, cost and technology. All possible alternatives must be considered, including the no action alternative If one or more of the alternatives will be located in a wetland, those alternatives may not be selected unless a determination is made that no practicable alternative exists outside the wetlands.

If no practicable alternative exists outside the wetlands, and the RPM or lead agency has determined or proposes to allow a remedial action to be located in a wetlands, then the RPM or lead agency shall act to minimize potential harm or to avoid adverse effects to the wetlands. This includes action to allow restoration and preservation of the natural and beneficial values of the wetlands. The benefits of preserving wetlands in their natural or relatively undisturbed state include the control of flood and storm hazards, maintenance of water quality standards and water supply, maintenance of natural systems, natural pollution abatement, conservation and long term productivity of existing flora and fauna, species and habitat diversity and stability, hydrologic utility, fish, wildlife, timber and food resources, and other uses of wetlands in the public interest including recreational, scientific and cultural uses. All impacts caused by an action occurring in a wetland must be evaluated and mitigated according to the EPA mitigation policy (under authority of the Clean Water Act section 404) in effect at the time of the proposed action, including the effects on the wetlands natural or beneficial value.

C. Documentation of Decision

For all lead agency response actions proposed to be in or affecting a floodplain/wetland the RPM or lead agency shall document their decision in the Record of Decision (ROD). The decision shall be accompanied by a Statement of Findings which may be included in the ROD support document or attached as a separate appendix. This statement will not exceed three pages and will include: (i) The reasons why the proposed action must be located in or affect the floodplain or wetlands; (ii) a description of significant facts considered in making the decision to locate in or affect the floodplain or wetlands including alternative sites and actions; (iii) a statement indicating whether the proposed action conforms to applicable State or local floodplain/wetland protection standards; (iv) a description of the steps taken to design or modify the proposed act to minimize potential harm to or within the floodplain or wetlands; and (v) a statement indicating how the proposed action affects the natural or beneficial values of the floodplain or wetlands.

2. Opportunity for Response to Citizen Concerns

Remedial actions satisfy the second criterion for a functional equivalent exception because current Agency procedures for public comment on remedial actions and the proposed amendments to the NCP afford the public an ample opportunity for participation in the evaluation of environmental factors prior to arriving at a final decision. The proposed amendments to the NCP and the current Superfund Community Relations Policy provide for a minimum 21-day comment period on the feasibility study which outlines alternative remedial measures prior to selection of the final ' remedial response. This public involvement in the remedial planning process would enable remedial actions to meet the public participation requirement for the functional equivalent exception to NEPA.

Appendix A, however, appears to require two further public notice requirements. One is any early public notice when it is apparent that a proposed or potential agency action is likely to impact a floodplain or wetlands and the other is public notice of the selected decision.

Current Agency policy suggests that a fact sheet summarizing the feasibility study response alternatives and other issues, be provided to the public 2 weeks prior to the minimum 3 week public comment period for the feasibility study. The fact sheet will include a statement explaining whether a proposed or potential remedial action is likely to impact a floodplain or wetlands. This will provide early public notice as required by Appendix A.

Concerning the public notice of a selected decision, the Agency suggests that a public notice and updated fact sheet summarizing the ROD be provided to the public. In addition, when the ROD is signed, it becomes a public document. The public notice, fact sheet and the availability of the signed ROD in the information repositories will provide public notice of the selected decision as required by Appendix A. The updated fact sheet will contain the alternative selected, any effects the response will have on floodplain/wetlands, and the Statement of Findings described in the Documentation of Decision Section above.

In addition, the Agency suggests that public meetings and other community relations activities be held as specified in the community relations plan. - 9 -

D. Summary

1. Removal Actions

For removal actions, EPA's policy is to pursue actions that will meet applicable or relevant standards, and criteria of the other Federal environmental laws that deal with floodplains/ wetlands to the maximum extent practicable, considering the exigencies of the situation.

2. Remedial Actions

For remedial actions, EPA's policy is to pursue remedies that attain or exceed applicable and relevant standards of other Federal environmental laws that deal with floodplains/wetlands, unless specific circumstances exist as referenced in section 300.68(i)(5) of the NCP. CERCLA procedural and administrative requirements will be modified to provide safeguards similar to those provided under other laws. Applications for and receipt of permits is not required for on-site response actions taken under the Fund-financed or enforcement authorities of CERCLA (i.e., Clean Water Act 404 permits are not required).

III. COMPLIANCE WITH OTHER FLOODPLAIN/WETLAND LAWS

The Agency has concluded that cleanups pursuant to sections 104 and 106 of CERCLA should comply with other Federal environmental standards, as a matter of policy, but not as a matter of law, except in a limited set of circumstances. For example, Section 10 of the Rivers and Harbors Act of 1899 and section 404 of the Clean Water Act apply to dredge and fill activities and must be complied with except in very limited circumstances such as fund balancing. (See "CERCLA Compliance with other Environmental Statutes" 50 FR 5928). However, permits are not required for these actions. This policy has also been proposed in amendments to the NCP (50 FR 5862). In addition, Federal public health and environmental criteria and advisories and State standards shall be considered, with appropriate adjustment, in determining the appropriate response action. Therefore, the Agency should also consider State and local floodplain/wetland protection standards and other Federal guidance. If the Agency does not use applicable State and local standards, the reason why should be documented in the Record of Decision or the Statement of Findings prepared by the OSC.

IV. IMPLEMENTATION

This policy will apply to all removals and remedial investigations and feasibility studies that are initiated after August 1, 1985.

CERCLA Compliance with Other Environmental Statutes

United States	
Environmental	Protection
Agency	

Office of Solid Waste and Emergency Response

€PA	DIRECTIVE NUMBER: 9234.0-02 TITLE: CERCLA Compliance with Other Environmental Statutes
	APPROVAL DATE: October 2, 1985 EFFECTIVE DATE: October 2, 1985 ORIGINATING OFFICE: OERR/PAS XI FINAL
	STATUS: REFERENCE (other documents):

OSWER OSWER OSWER /E DIRECTIVE DIRECTIVE D



OCT - 2 1985

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

OSWER DIRECTIVE 9234.0-2

MEMORANDUM

SUBJECT: CERCLA Compliance With Other Environmental Statutes FROM: Winston Porter Assistant Administrator

TO: Regional Administrator Regions I-X

This memorandum sets forth the Environmental Protection Agency (EPA) policy on the applicability of the standards, criteria, advisories, and guidance of other State and Federal environmental and public health statutes to actions taken pursuant to sections 104 and 106 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA). This policy addresses considerations for on-site and off-site actions taken under CERCLA.

I. <u>Discussion</u>

The National Oil and Hazardous Substances Pollution Contingency Plan (NCP) establishes the process for determining appropriate removal and/or remedial actions at Superfund sites. In the course of this process, EPA will give primary consideration to the selection of those response actions that are effective in preventing or, where prevention is not practicable, minimizing the release of hazardous substances so that they do not migrate to cause substantial danger to present or future public health, welfare, or the environment. As a general rule, this can be accomplished by pursuing remedies that attain or exceed the requirements of applicable or relevant and appropriate Federal public health or environmental laws. However, because of unique circumstances at particular sites, there may be alternatives that do not meet the standards of other laws, but that still provide protection of public health, welfare, and the environment.

Although response actions that prevent hazardous substances from migrating into the environment are seen as the most effective under CERCLA, actions which minimize migration must also be considered since CERCLA primarily addresses inadequate past disposal practices and resulting unique site conditions. At certain sites, it may be technically impractice environmentally unacceptable, or excessively costly to implem a response action that prevents migration or restores the site to its original, uncontaminated condition.

II. Policy

Section 104 of CERCLA requires that for off-site remedial actions, storage, destruction, treatment or secure disposition, be in compliance with subtitle C of the Resource Conservation and Recovery Act (RCRA). CERCLA is silent, however, concerning the requirements of other laws with regard to all other response actions taken pursuant to sections 104 and 106.

As a general rule, the Agency's policy is to attain or exceed applicable or relevant and appropriate Federal environmental and public health requirements in CERCLA response actions unless one of the specifically enumerated situations is present. Where such a situation is present and a requirement is not followed, the Agency must document and explain the reasons in the decision documents. Other Federal criteria, advisories, guidances, and State standards also will be considered and may be used in developing remedial alternatives, with adjustments for site-specific circumstances. If EPA does not use, or uses and adjusts any pertinent standards in this category, EPA will fully document the reasons why in the decision documents.

A. On-site Response Actions

(1) For removal actions, EPA's policy is to pursue actions that will meet applicable or relevant and appropriate requirements of other Federal environmental and public health laws to the maximum extent practicable, considering the exigencies of the situation.

(2) For remedial actions, EPA's policy is to pursue remedies that attain or exceed applicable or relevant and appropriate requirements of other Federal public health and environmental laws, unless the specific circumstances identified below exist.

CERCLA procedural and administrative requirements will be modified to provide safeguards similar to those provided under other laws. Application for and receipt of permits is not required for on-site response actions taken under the Fund-financed or enforcement authorities of CERCLA.

B. Off-Site Response Actions

CERCLA removal and remedial activities that involve the removal of hazardous substances from a CERCLA site to offsite facilities for proper storage, treatment or disposal must be in compliance with all applicable or relevant standards of Federal environmental and public health statutes. Off-site facilities that are used for storage, treatment, or disposal of Superfund wastes must have all appropriate permits or authorizations.

If the facility or process that is being considered for receipt of the Superfund wastes has not been permitted or authorized, the State or responsible party will be required to obtain all appropriate permits. Furthermore, as stated in the Agency's off-site policy memorandum, "Procedures for Planning and Implementing Off-Site Response Actions", May 6, 1985, barring several exceptions enumerated in that memorandum, no CERCLA hazardous substances shall be taken off-site to a unit in a RCRA facility if the receiving Region's Administrator determines that the unit has significant RCRA violations or other environmental conditions that affect the satisfactory operation of the facility. A State's responsibility for obtaining any appropriate Federal, State or local permits (e.g., RCRA, TSCA, NPDES, UIC, Clean Air, etc.) will be specified in a contract or cooperative agreement with the State as part of its assurances required under section 104(c) of CERCLA.

III. Other Laws or Guidances That May Be Used to Determine the Appropriate Extent of Response Actions

Federal and State environmental and public health requirements, criteria, guidance and advisories fall into two categories:

- Federal requirements that are applicable or relevant and appropriate,
- Other Federal criteria, advisories, guidances, and State standards to be considered.

An initial list of both categories is attached.

A. Applicable or Relevant and Appropriate Federal Requirements

"Applicable" requirements are those Federal requirements that would be legally applicable, whether directly, or as incorporated by a federally authorized State program, if the response actions were not undertaken pursuant to CERCLA section 104 or 106.

"Relevant and Appropriate" requirements are those Federal requirements that, while not "applicable", are designed to apply to problems sufficiently similar to those encountered at CERCLA sites that their application is appropriate. Requirements may be relevant and appropriate if they would be "applicable" but for jurisdictional restrictions associated with the requirement.

For example, the RCRA 40 CFR Part 264 Subpart F Ground-Water Protection Standards would be applicable to the management or cleanup of hazardous wastes in ground water from hazardous waste management facilities if such actions were not taken pursuant to CERCLA sections 104 or 106. Yet RCRA Subtitle C regulations, while not applicable to hazardous wastes disposed of prior to the November 19, 1980, effective date of those regulations, could be relevant to CERCLA response actions regardless of when the wastes were disposed of or managed.

B. Other Federal Criteria, Advisories, Guidances and State Standards to Be Considered

This category includes other standards, criteria, advisories and guidance that may be useful in developing Superfund remedies. These criteria, advisories and guidances were developed by EPA, other Federal agencies and the States. The concepts and data underlying these requirements may be used at Superfund sites in an appropriate way.

IV. Implementation

A. Removal Actions

For both on and off-site Fund-financed removal actions, the lead agency should consult with the Regional Response Team within the framework of the Regional Contingency Plan to determine the most effective action.

(1) On-site

For on-site removal actions, the lead agency shall, as appropriate, attempt to <u>attain or exceed</u> all Federal applicable or relevant and appropriate public health or environmental requirements. The lead agency also shall, as appropriate, <u>consider</u> other Federal criteria, guidances, and advisories as well as State standards in formulating the removal action. However, because removal actions often involve situations requiring expeditious action to protect public health, welfare, or the environment, it may not always be feasible to fully meet them. In those circumstances where they cannot be attained, the decision documents, OSC reports, or other documents should specify the reasons.

(2) Off-site

Off-site facilities that are used for storage, treatment, or disposal of Superfund wastes must have all appropriate permits or authorizations and, barring certain exceptions, enumerated in the off-site policy, no hazardous substance shall be taken off-site to a unit in a RCRA facility if the Region determines that the unit has significant RCRA violations or other environmental conditions that affect the satisfactory operation of the facility.

B. Remedial Actions

1. Presentation and Analysis of Alternatives

To the extent that it is both possible and appropriate, at least one remedial alternative shall be developed as part of the feasibility study (FS) in each of the following categories:

(a) Alternatives for treatment or disposal in an offsite facility, as appropriate;¹

(b) Alternatives that <u>attain</u> applicable and relevant and appropriate Federal public health or environmental requirements;

(c) As appropriate, alternatives that <u>exceed</u> applicable and relevant and appropriate public health or environmental requirements²;

(d) As appropriate, alternatives that do not attain applicable or relevant and appropriate public health or environmental requirements but will reduce the likelihood of present or future threat from the hazardous substances and that provide significant protection to public health, welfare and environment. This must include an alternative that closely approaches the level of protection provided by the applicable or relevant and appropriate requirements;

(e) A no action alternative.

2. Selection of Remedy

The decisionmaker will consider all of the alternatives arrayed in the feasibility study and will give primary consideration to remedies that attain or exceed applicable or relevant and appropriate Federal public health and environmental requirements. Where the selected remedy involves an EPA standard, criterion, or advisory, the decisionmaker will ensure appropriate coordination with affected EPA programs.

In appropriate cases, the decisionmaker may select a remedial action that includes both on- and off-site components.

I These alternatives must be consistent with EPA's May 6, 1985 off-site policy, "Procedures for Planning and Implementing Off-Site Response Actions". In some cases, off-site disposal or treatment may not be feasible and this alternative may be eliminated during initial screening of alternatives. The decision documents should reflect this screening.

² For instance, the Agency might choose incineration as an alternative that exceeds what would be required by applicable standards because it is a more permanent and reliable solution than RCRA closure standards for land disposal facilities.

The decisionmaker may select an alternative that does not attain applicable or relevant standards in one of the five following circumstances:

(a) <u>Interim Remedy</u> - Where the selected alternative is not the final remedy and will become part of a more comprehensive remedy, the lead agency may select an interim remedy;

(b) Fund-Balancing - For Fund-financed responses only, the need for protection of public health, welfare and the environment at the facility under consideration for all of the alternatives that attain or exceed applicable or relevant and appropriate Federal requirements is, considering the amount of money available in the Fund, outweighed by the need for action at other sites that may present a threat to public health or welfare or the environment. In the event of Fund balancing, the lead agency shall select the alternative which most closely approaches the level of protection provided by applicable or relevant and appropriate Federal requirements, considering the specific Fund-balanced sum of money available for the immediate facility. Fund-balancing is not a consideration in determining the appropriate extent of remedy when the response will be performed by a potentially responsible party;

(c) <u>Technical Impracticality</u> - Where no alternative that attains or exceeds applicable or relevant and appropriate Federal public health or environmental requirements is technically practical to implement, the lead agency shall select the alternative that most closely approaches the level of protection provided by the applicable or relevant and appropriate requirements, and which is reasonable to implement from an engineering perspective;

(d) <u>Unacceptable Environmental Impacts</u> - Where all the alternatives that attain or exceed Federal public health or environmental requirements, if implemented, will result in significant adverse environmental impacts, the lead agency shall select the alternative that most closely approaches the level of protection provided by applicable or relevant and appropriate requirements, without resulting in significant adverse environmental impacts; or

(e) Overriding Public Interest Related to Enforcement -Where the remedy is to be carried out pursuant to CERCLA section 106, the Fund is unavailable, there is a strong public interest in expedited cleanup, and the litigation probably would not result in the desired remedy, the lead agency will select the alternative that most closely approaches applicable or relevant and appropriate Federal public health and environmental statutes in light of the need to invoke the exception. Where one of these situations is present, the decisionmaker <u>may</u> select an alternative which does not attain or exceed applicable or relevant and appropriate Federal public health or environmental requirements, yet still provides protection of the public health and welfare and the environment. The basis for not meeting the requirements must be fully documented and explained in the appropriate decision documents. The Agency anticipates that most final CERCLA remedial actions will attain or exceed applicable or relevant and appropriate public health or environmental requirements.

Other Federal criteria, advisories, guidances, and State standards also will be considered and may be used in developing remedial alternatives, with appropriate adjustments for site specific circumstances. If EPA does not use, or uses and adjusts any <u>pertinent standards</u> in this category, EPA will fully document the reasons why in the decision documents.

For Fund-financed actions, where State standards are part of the cost-effective remedy, the Fund will pay to attain those standards. Where the cost-effective remedy does not include those State standards, the State may pay the difference to attain them.

3. Administrative and Procedural Aspects

The following modifications will be made to the Superfund community relations program to ensure that it provides a similar level of public involvement to that provided by the permitting programs of other environmental laws:

A fact sheet should be included with the public notice and feasibility study which is provided to the public 2 weeks before the 3 week public comment period. The fact sheet will clearly summarize the feasibility study response alternatives and other issues, including which alternatives attain or exceed Federal public health and environmental reguirements. For those alternatives that do not attain applicable or relevant and appropriate requirements of other public health and environmental laws, the fact sheet shall identify how they do not attain the requirements and explain how they nonetheless meet the goals of CERCLA. The public notice should include a timetable in which a decision will be reached, any tentative determinations which the Agency has made, the location where relevant documents can be obtained, identification of community involvement opportunities, the name of an Agency contact, and other appropriate information.

• A public notice and updated fact sheet should be prepared upon (1) Agency selection of the final response action and (2) completion of the final engineering design. Prior to selecting the final engineering design, the Agency may hold a public meeting to inform the public of the design alternatives and to solicit comments. • If a remedy is identified that is materially different from those proposed during the feasibility study public comment period, a new 3 week public comment period may be required prior to amending the Record of Decision, taking into consideration the features of the alternatives addressed in the public comment period.

The CERCLA enforcement community relations program will also be modified to provide for an enhanced public participation program for both consent decrees and administrative orders. This program will be substantially equivalent to the revised program for Fund-financed actions. Furthermore, consent decrees and administrative orders will incorporate administrative requirements (i. e. recordkeeping, monitoring) similar to those mandated by other environmental programs.

V. Applicability of Policy

This policy applies to two situations:

- a site-specific FS has not yet been initiated; the FS must fully comply with this policy.
- the FS has been initiated, but the remedy has not yet been selected; the requirements of this policy shall be incorporated into the FS and Record of Decision (ROD) as practicable.

This policy does not apply to RODs signed before February 12, 1985, the date of proposal of this policy.

If you have any questions or comments, please contact James Lounsbury, Director, Policy Analysis Staff (202 382-2182) or Stephen M. Smith of his staff (202 382-2200).

Attachment

POTENTIALLY APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS

1. EPA's Office of Solid Waste administers, <u>inter</u> alia, the Resource Conservation and Recovery Act of 1976, as amended (Pub. L. 94-580, 90 Stat 95, 42 U.S.C. 6901 <u>et seq</u>.). Potentially applicable or relevant requirements pursuant to that Act are:

- Open Dump Criteria Pursuant to RCRA Subtitle D
 criteria for classification of solid waste disposal
 facilities (40 CFR Part 257).
 Note: Only relevant to nonhazardous wastes.
- b. In most situations Superfund wastes will be handled in accordance with RCRA Subtitle C requirements governing standards for owners and operators of hazardous waste treatment, storage, and disposal facilities: 40 CFR Part 264, for permitted facilities, and 40 CFR Part 265, for interim status facilities.
 - Ground Water Protection (40 CFR 264.90-264.109).
 - Ground-Water Monitoring (40 CFR 265.90-265.94).
 - Closure and Post Closure (40 CFR 264.110-264.120, 265.110-265.112).
 - Containers (40 CFR 264.170-264.178, 265.170-265.177).
 - Tanks (40 CFR 264.190-264.200, 265.190-265.199).
 - Surface Impoundments (40 CFR 264.220-264.249, 265.220-265.230).
 - Waste Piles (40 CFR 264.250-264.269, 265.250-265.258)

- Land Treatment (40 CFR 264.270-264.299, 265.270-265.282).
- Landfills (40 CFR 264.300-264.339, 265.300-265.316).
- Incinerators (40 CFR 264.340-264.999, 265.340-265.369).
- Dioxin-containing Wastes, (50 FR 1978). Includes the the final rule for the listing of dioxin containing waste.
- 2. EPA's Office of Water administers several potentially applicable or relevant and appropriate statutes and regulations issued thereunder:
 - a. Section 14.2 of the Public Health Service Act as
 amended by the Safe Drinking Water Act as amended
 (Pub. L. 93-523, 88 Stat 1660, 42 U.S.C. 300f et seq.)
 - Maximum Contaminant Levels (for <u>all</u> sources of drinking water exposure). (40 CFR 141.11-141.16)
 - Underground Injection Control Regulations. (40
 CFR Parts 144, 145, 146, and 147)
 - b. Clean Water Act as amended (Pub. L. 92-500, 86 Stat 816, 33 U.S.C. 1251 <u>et</u>. <u>seq</u>.)
 - Requirements established pursuant to sections 301, 302, 303 (including State water quality standards), 306, 307, (including Federal pretreatment requirements for discharge into a publicly owned treatment works), and 403 of the Clean Water Act. (40 CFR Parts 131, 400-469)

- c. Marine Protection, Research, and Sanctuaries Act (33 U.S.C. 1401).
 - Incineration at sea requirements. (40 CFR Part
 220-225, 227, 228. See also 40 CFR 125.120-125.124)
- 3. EPA's Office of Pesticides and Toxic Substances Toxic Substances Control Act (15 U.S.C. 2601).
 - PCB Requirements Generally: 40 CFR Part 761; Manufacturing Processing, Distribution in Commerce, and Use of PCBs and PCB Items (40 CFR 761.20-761.30); Markings of PCBs and PCB Items (40 CFR 761.40-761.45); Storage and Disposal (40 CFR 761.60-761.79). Records and Reports (40 CFR 761.180-761.185). See also 40 CFR 129.105, 750.
 - Disposal of Waste Material Containing TCDD. (40 CFR Part 775.180-775.197).
- 4. EPA's Office of External Affairs
 - Section 404(b)(1) Guidelines for Specification of Disposal Sites for Dredged or Fill Material (40 CFR Part 230).
 - Procedures for denial or Restriction of Disposal
 Sites for Dredged Material (\$404(c) Procedures, 40
 CFR Part 231).
- 5. EPA's Office of Air and Radiation administers several potentially applicable or relevant and appropriate statutes and regulations issued thereunder:
 - a. The Uranium Mill Tailings Radiation Control Act of 1978 (42 U.S.C. 2022).

- Uranium mill tailing rules Health and Environmental Protection Standards for Uranium and Thorium Mill Tailings, (40 CFR Part 192).
- b. Clean Air Act (42 U.S.C. 7401).
 - National Ambient Air Quality Standards for total suspended particulates (40 CFR Part 50.6-50.7)
 - National Ambient Air Quality Standards for ozone (40 CFR 50.9).
 - Standards for Protection Against Radiation high and low level radioative waste rule, (10 CFR Part 20). See also 10 CFR Parts 10, 40, 60, 61, 72, 960, 961.
 - National Emission Standard for Hazardous Air
 Pollutants for Asbestos, (40 CFR 61.140-61.156).
 See also 40 CFR 427.110-427.116, 763.
 - National Emission Standard for Hazardous Air
 Pollutants for Radionuclides (40 CFR Part 61, 10
 CFR 20.101-20.108).
- 6. Other Federal Requirements
 - a. OSHA requirements for workers engaged in response activities are codified under the Occupational Safety and Health Act of 1970 (29 U.S.C. 651). The relevant regulatory requirements are included under:
 - Occupational Safety and Health Standards (General Industry Standards) (29 CFR PArt 1910).
 - The Safety and Health Standards for Federal
 Service Contracts (29 CFR Part 1926).

- The Shipyard and Longshore Standards (29 CFR Parts 1915, 1918).
- Recordkeeping, reporting, and related regulations
 (29 CFR Part 1904).
- b. Historic Sites, Buildings, and Antiquities Act (16
 U.S.C. 461).
- c. National Historic Preservation Act, 16 U.S.C. 470. Compliance with NEPA required pursuant to 7 CFR Part 650. Protection of Archaelogical Resources: Uniform Regulations -- Department of Defense (32 CFR Part 229, 229.4), Department of the Interior (43 CFR Part 7, 7.4).

D.O.T. Rules for the Transportation of Hazardous Materials, 49 CFR Parts 107, 171.1-171.500. Regulation of activities in or affecting waters of the United States pursuant to 33 CFR Parts 320-329. The following requirements are also triggered by Fundfinanced actions:

- Endangered Species Act of 1973, 16 U.S.C. 1531. (Generally, 50 CFR Parts 81, 225, 402).
 Wild and Scenic Rivers Act, 16 U.S.C. 1271.
 Compliance with NEPA required pursuant to 36 CFR Part 297.
- Fish and Wildlife Coordination Act, 16 U.S.C. 661
 note.
- Fish and Wildlife Improvement Act of 1978, and
 Fish and Wildlife Act of 1956, 16 U.S.C. 742a not

- Fish and Wildlife Conservation Act of 1980, 16
 U.S.C. 2901. (Generally, 50 CFR Part 83).
- Coastal Zone Management Act of 1972, 16 U.S.C.
 1451. (Generally, 15 CFR Part 930 and 15 CFR 923.45 for Air and Water Pollution Control Requirements).
 OTHER FEDERAL CRITERIA, ADVISORIES, GUIDANCES,

AND STATE STANDARDS TO BE CONSIDERED

- 1. Federal Criteria, Advisories and Procedures
 - Health Effects Assessments (HEAs)
 - Recommended Maximum Concentration Limits (RMCLs)
 - Federal Water Quality Criteria (1976, 1980, 1984).
 Note: Federal Water Quality Criteria are not legally enforceable. State water quality standards are legally enforceable, developed using appropriate aspects of Federal Water Quality Criteria. In many cases, State water quality standards do not include specific numerical limitations on a large number of priority pollutants.
 When neither State standards nor MCLs exist for a given pollutant, Federal Water Quality Criteria are pertinent and therefore are to be considered.
 - Pesticide registrations.
 - Pesticide and food additive tolerances and action levels.
 Note: Germane portions of tolerances and action levels
 may be pertinent and therefore are to be considered in
 certain situations.
 - Waste load allocation procedures, EPA Office of Water.
 - * Federal sole source aquifer requirements.

- Public health basis for the decision to list pollutants as hazardous under section 112 of the Clean Air Act.
- EPA's Ground-water Protection Strategy.
- New Source Performance Standards for Storage Vessels for Petroleum Liquids.
- TSCA health data.
- Pesticide registration data.
- TSCA chemical advisories (2 or 3 issued to date).
- Advisories issued by FWS and NWFS under the Fish and Wildlife Coordination Act.
- Executive Orders related to Floodplains (11988) and Wetlands (11990) as implemented by EPA's August 6, 1985, ...
 Policy on Floodplains and Wetlands Assessments for CERCLA Actions.
- * TSCA Compliance Program Policy.
- OSHA health and safety standards that may be used to protect public health (non-workplace).
- Health Advisories, EPA Office of Water

2. State Standards

- State Requirements on Disposal and Transport of Radioactive wastes.
- State Approval of Water Supply System Additions or Developments.
- State Ground Water Withdrawal Approvals.
- Requirements of authorized (Subtitle C of RCRA) State

hazardous waste programs.

- State Implementation Plans and Delegated Programs
 Under Clean Air Act.
- All other State requirements, not delegated through EPA authority.
- * Approved State NPDES programs under the Clean Water Act.
- Approved State UIC programs under the Safe Drinking
 Water Act.

Note: Many other State and local requirements could be pertinent. Forthcoming guidance will include a more comprehensive list.

3. USEPA RCRA Guidance Documents

* Draft Alternate Concentration Limits (ACL) Guidance

- A. EPA's RCRA Design Guidelines
 - Surface Impoundments, Liners Systems, Final Cover and Freeboard Control.
 - 2. Waste Pile Design Liner Systems.
 - 3. Land Treatment Units.
 - 4. Landfill Design Liner Systems and Final Cover.
- B. Permitting Guidance Manuals
 - Permit Applicant's Guidance Manual for Hazardous Waste Land Treatment, Storage, Disposal Facilities.
 - Permit Writer's Guidance Manual for Hazardous Waste Land Treatment, Storage, and Disposal Facilities.
 - 3. Permit Writer's Guidance Manual for Subpart F.
 - Permit Applicants Guidance Manual for the General Facility Standards.
- 5. Waste Analysis Plan Guidance Manual.
- 6. Permit Writer's Guidance Manual for Hazardous Waste Tanks.
- 7. Model Permit Application for Existing Incinerators.
- Guidance Manual for Evaluating Permit Applications for the Operation of Hazardous Waste Incinerator Units.
- 9. A guide for Preparing RCRA Permit Applications for Existing Storage Facilities.
- 10. Guidance Manual on closure and post-closure Interim Status Standards.
- C. Technical Resource Documents (TRDs)
 - 1) Evaluating Cover Systems for Solid and Hazardous Waste.
 - 2) Hydrologic Simulation of Solid Waste Disposal Sites.
 - 3) Landfill and Surface Impoundment Performance Evaluation
 - 4) Lining of Water Impoundment and Disposal Facilities.
 - 5) Management of Hazardous Waste Leachate.
 - Guide to the Disposal of Chemically Stabilized and Solidified Waste.
 - 7) Closure of Hazardous Waste Surface Impoundments.
 - 8) Hazardous Waste Land Treatment.
 - Soil Properties, Classification, and Hydraulic
 Conductivity Testing.
- D. Test Methods for Evaluating Solid Waste
 - 1) Solid Waste Leaching Procedure Manual.
 - Methods for the Prediction of Leachate Plume Migration and Mixing.

- 3) Hydrologic Evaluation of Landfill Performance (HELP) Model Hydrologic Simulation on Solid Waste Disposal Sites.
- 4) Procedures for Modeling Flow Through Clay Liners to Determine Required Liner Thickness
- 5) Test Methods for Evaluating Solid Wastes
- A Method for Determining the Compatibility of Hazardous Wastes
- 7) Guidance Manual on Hazardous Waste Compatibility
- 4. USEPA Office of Water Guidance Documents
- A. Pretreatment Guidance Documents
 - 304(g) Guidance Document Revised Pretreatment Guidelines
 (3) Volumes)
- B. Water Quality Guidance Documents
 - Ecological Evaluation of Proposed Discharge of Dredged Material into Ocean Waters (1977)
 - 2) Technical Support Manual: Waterbody Surveys and Assessments for Conducting Use Attainability Analyses (1983)
 - 3) Water-Related Environmental Fate of 129 Priority Pollutants (1979)
 - 4) Water Quality Standards Handbook (1983)
 - 5) Technical Support Document for Water Quality-based Toxics Control.

- C. NPDES Guidance Documents
 - NPDES Best Management Practices Guidance Manual (June 1981)
 - 2) Case studies on toxicity reduction evaluation (May 1983).

D. Ground Water/UIC Guidance Document

- 1) Designation of a USDW
- 2) Elements of Aquifer Identification
- 3) Interim guidance for public participation
- 4) Definition of major facilities
- 5) Corrective action requirements
- 6) Requirements applicable to wells injecting into, through or above an aquifer which has been exempted pursuant to \$146.104(b)(4).
- 7) Guidance for UIC implementation on Indian lands.
- 5. USEPA Manuals from the Office of Research and Development
 - 1) EW 846 methods laboratory analytic methods
 - 2) Lab protocols developed pursuant to Clean Water Act §304(h).

Discharge of Wastewater from CERCLA Sites into POTWs

•

•



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

APR 1 5 1986

MEMORANDUM

SUBJECT: Discharge of Wastewater from CERCLA Sates into POTWS

FROM: Henry L. Longest II, Director

Office of Emergency and Remedian Resonance Rebecca Hanmer, Director Reference Hanne

Office of Water Enforcement and Permits

Gene A. Lucero, Director Che H: Lucero Office of Waste Programs Enforcement

TO: Waste Management Division Directors Regions I - X

> Water Management Division Directors Regions I - X

A number of emergency removals and remedial cleanup actions under CERCLA will involve consideration of publicly owned treatment works (POTWs) for discharge of wastewater. The current off-site policy (issued on May 6, 1985) does not address the set of concerns and issues unique to POTWs that must be evaluated during the Remedial Investigation and Feasibility Study (RI/FS) for discharge of CERCLA wastewater to POTWs.

Recently, we have had meetings with representatives of the Association of Metropolitan Sewerage Authorities (AMSA) to discuss technical and policy concerns related to the POTW/CERCLA issue. This memorandum is to highlight some of the major points under consideration which were shared with AMSA at their recent Winter Technical Conference. The Agency intends to develop policy on the use and selection of POTWs for CERCLA wastewater. Your comments are sought on the proposed criteria set forth herein. These criteria may be useful in evaluation of POTWs for response actions (fund financed or responsible party financed) to be taken in the interim.

Our position is that no CERCLA discharges to a POTW should occur unless handled in a manner demonstrated to be protective of human health and the environment. Full compliance with all applicable requirements of the Clean Water Act (CWA), the Resource Conservation and Recovery Act (RCRA), and any other relevant or appropriate environmental statutes will be necessary . The national pretreatment program, under the Clean Water Act, requires an analysis to determine whether the discharge of an industrial user of a POTW may pass through the POTW to cause receiving water quality problems or may interfere with POTW operations (including sludge disposal). If the analysis suggests that limits on the industrial user's discharge are needed to prevent pass through or interference, local limits or other safeguards, as necessary, must be established by the POTW and/or the NPDES permitting authority. The national pretreatment program requirements apply to the introduction of all non-domestic wastewater into any POTW, and include, among other things, the following elements:

- Prohibited discharge standards prohibit the introduction of pollutants to the POTW which are ignitable, corrosive, excessively high in temperature, or which may cause interference or pass through at the POTW.
- Categorical discharge standards include specific pretreatment standards which are established by EPA for the purpose of regulating industrial discharges in specific industrial categories.
- Local limits where no categorical standards have been promulgated or where more stringent controls are necessary.

POTWs under consideration as potential receptors of CERCLA wastewaters may include those POTWs either with or without an approved pretreatment program. POTWs with an approved pretreatment program are required to have the mechanisms necessary to ensure compliance by industrial users with applicable pretreatment standards and requirements.* POTWs without an approved pretreatment program must be evaluated to determine whether sufficient mechanisms exist to allow the POTW to meet the requirements of the national pretreatment program in accepting CERCLA wastewaters. As noted above, pass through and interference are always prohibited, regardless of whether a POTW has an approved pretreatment program. POTWs without an approved pretreatment program must therefore have mechanisms which are adequate to apply the requirements of the national pretreatment program to specific situations.

- 2 -

^{*}POTWs with approved pretreatment programs must, among other things, establish procedures to notify industrial users (IUs) of applicable pretreatment standards and requirements, receive and analyze self-monitoring reports from IUs, sample and analyze industrial effluents, investigate noncompliance, and comply with public participation requirements.

Determination of a POTW's ability to accept CERCLA wastewater as an alternative to on-site treatment and direct discharge to receiving waters must be made during the Remedial Investigation/ Feasibility Study (RI/FS) process. During the remedial alternatives analysis, the appropriateness of using a POTW must be carefully evaluated. Water Division officials and their state counterparts should participate in the evaluation of any remedial alternatives recommending the use of a POTW, and should concur on the selection of the POTW.

If an alternative considers the discharge of wastewater from a CERCLA site into a POTW, the following points should be evaluated in the RI/PS prior to the selection of the remedy for the site:

- o The quantity and quality of the CERCLA wastewater and its compatibility with the POTW (The constituents in the CERCLA wastewater must not cause pass through or interference, including unacceptable sludge contamination or a hazard to employees at the POTW; in some cases, control equipment at the CERCLA site may be appropriate in order to pretreat the CERCLA discharge prior to introduction to the POTW).
- The ability (i.e., legal authority, enforceable mechanisms, etc.) of the POTW to ensure compliance with applicable pretreatment standards and requirements, including monitoring and reporting requirements.
- o The POTW's record of compliance with its NPDES permit and pretreatment program requirements to determine if the POTW is a suitable disposal site for the CERCLA wastewater.
- o The potential for volatilization of the wastewater at the CERCLA site and POTW and its impact upon air quality.
- o The potential for groundwater contamination from transport of CERCLA wastewater or impoundment at the POTW, and the need for groundwater monitoring.
- o The potential effect of the CERCLA wastewaters upon the POTW's discharge as evaluated by maintenance of water quality standards in the POTW's receiving waters, including the narrative standard of "no toxics in toxic amounts".

- o The POTW's knowledge of and compliance with any applicable RCRA requirements or requirements of other environmental statutes (RCRA permit-by-rule requirements may be triggered if the POTW receives CERCLA wastewaters that are classified as "hazardous wastes" without prior mixing with domestic sewage, i.e., direct delivery to the POTW by truck, rail, or dedicated pipe; CERCLA wastewaters are not all necessarily considered hazardous wastes; case by case determinations have to be made).
- o The various costs of managing CERCLA wastewater, including all risks, liabilities, permit fees, etc. (It may be appropriate to reflect these costs in the POTW's connection fees and user charge system).

Based upon consideration of the above elements, the discharge of CERCLA wastewater to a POTW should be deemed inappropriate if the evaluation indicates that:

- o The constituents in the CERCLA discharge are not compatible with the POTW and will cause pass through, interference, toxic pollutants in toxic amounts in the POTW's receiving waters, unacceptable sludge contamination, or a hazard to employees of the POTW.
- The impact of the transport mechanism and/or discharging of CERCLA wastewater into a POTW would result in unacceptable impacts upon any environmental media.
- The POTW is determined to be an unacceptable receptor of CERCLA wastewaters based upon a review of the POTW's compliance history.
- o The use of the POTW is not cost-effective.

If consideration of the various elements indicates that the discharge of CERCLA wastewater to a POTW is deemed appropriate:

- There should be early public involvement, including contact with POTW officials and users, in accordance with the CERCLA community relations plan and public participation requirements.
- o The NPDES permit and fact sheet may need to be modified to reflect the conditions of acceptance of CERCLA wastewaters; permit modification may be necessitated by the need to incorporate specific pretreatment requirements, local limits, monitoring requirements and/or limitations on additional pollutants of concern in the POTW's discharge or other factors.

Policy to be developed in the future will apply to all removal, remedial, and enforcement actions taken pursuant to CERCLA and Section 7003 of RCRA. We would appreciate your feedback on this memorandum and any experience in the use of POTWs for CERCLA removal or remedial actions that you have to offer.

If you have any comments or questions on this issue, please submit written comments to the workgroup co-chairs: Shirley Ross (FTS-382-5755) from the Office of Emergency and Remedial Response, or Victoria Price (FTS-382-5681) from the Office of Water.

cc: Ed Johnson Russ Wyer Tim Fields Steve Lingle

Occupational Safety and Health Administration: Labor Federal Register Volume 51, No. 244

Friday December 19, 1986

Part IV

Department of Labor

Occupational Safety and Health Administration

29 CFR Part 1910 Hazardous Waste Operations and Emergency Response; Interim Final Rule

DEPARTMENT OF LABOR

Occupational Sefety and Health Administration

29 CFR Part 1910

[Docket No. S-750]

Hazardous Waste Operations and Emergency Response

AGENCY: Occupational Safety and Health Administration: Labor. ACTION: Interim final rule.

SUMMARY: This interum final rule amends the Occupational Safety and Health Administration (OSHA) standards for hazardous materials in Subpart H of 29 CFR Part 1910 by adding a new § 1910.120 containing employee protection requirements for workers engaged in hazardous waste operations including emergency response to hazardous substance incidents.

Coverage includes employees involved in responses covered by the Comprehensive Environmental **Response.** Compensation and Liability Act of 1980 as amended (CERCLA or "Superfund" Act) (Pub. L 95-510, 42 U.S.C. 9601 et seq. 94 Stat 2767 such as clean-up of hazardous waste sites. certain hazardous waste operations conducted under the Resource **Conservation and Recovery Act of 1978** as amended (RCRA) [Pub. L 94-580. 42 U.S.C. 6901 et seq. 90 Stat 2795], and emergency response to incidents involving the handling, processing and transportation of hazardous substances.

The issuance of this intern final rule is mandated by section 126(e) of the "Superfund Amendments and Reauthorization Act of 1986" (SARA) [Pub. L. 99-499]. The interim final rule will regulate employee safety and health at hazardous waste operations and during emergency response to hazardous substance incidents until a final standard, also mandated by section 125 of SARA, is issued by OSHA and becomes effective. The final OSHA standard also mandated by section 128 of SARA is the subject of a Notice of Proposed Rulemaking which will be published shortly.

DATER Interim rule effective December 19. 1988: various start-up dates havebeen established in paragraph (p) of the standard. The incorporation by reference of cartain publications listed in the regulations is approved by the Director of the Federal Register as of December 19. 1988.

FOR FURTHER INFORMATION CONTACT: Mr. James F. Foster. U.S. Department of Labor. Occupational Safety and Health Administration, Division of Consumer Alfairs. Room S-4220. 200 Constitution Avenue. NW., Washington, DC 20210, 202-523-0151.

This interim final rule was prepared by Michael B. Moore and Chappell D. Pierce. Directorate of Safety Standards. Office of Fire Protection Engineering and Systems Safety Standards. (202) 523-7225.

SUPPLEMENTARY INFORMATION

I. Background

On October 17, 1986, the President signed into law the "Superfund Amendments and Reauthorization Act of 1966" (SARA) [Pub. L. 99-499]. As part of SARA the Secretary of Labor ("Secretary") is directed to issue an interim final rule within 60 days after the date of enactment, which is to provide no lese protection for workers engaged in covered operations than the protections contained in the Environmental Protection Agency's (EPA), "Health and Safety Requirements for Employees Engaged in Field Activities" manual (EPA Order 1440.2) dated 1981 and the existing OSHA standards under Subpart C of 29 CPR Part 1928. SARA also directs the Secretary to issue, within one year. final standard under section 6(b) of the Occupational Safety and Health Act of 1970 for the health and safety of employees engaged in hazardous waste operations. SARA further indicates that certain specific areas of employee protection (i.e., medical surveillance personal protective equipment, training, and others) contained in section 125(b) are relevant to protect employees engaged in hazardous waete operations.

The interm final rule issued today becomes effective immediately and will remain in effect until one year after issuance of the final OSHA standard. which will be proposed shortly. Congress has clearly directed in section 126(e) that these interim final rules become effective upon issuance and the standard provides this. Implementation is to commence immediately, however. various start-up dates are set forth in paragraph (p) of the standard which recognize that full implementation cannot be completed immediately for some provisions. In addition OSHA will. of course, recognize greater feasibility constraints in the first three months of the standard and take those constraints into account in enforcement.

This intern final rule has been adopted from the language of the EPA manual entitled "Health and Safety Requirements for Employees Engaged in Field Activities" (1981) and the language of OSHA's safety and health standards in Subpart C of 29 CFR Part 1928. The interm final rule also contains language

taken from various documents i either jointly or solely by the EPA. OSHA, the U.S. Coast Guard, and the National Institute for Occupational Safety and Health (NIOSH). OSHA has specifically used the joint OSHA/EPA/ USCG/NIOSH document entitled. "Occupational Safety and Health Guidance Manual for Hazardaus Waste Site Activities" (Preamble Reference 6). as an outline in preparing this interim rule. This four agency manual has been developed as a result of the collaborative efforts of professionals representing the four agencies. These professionals, who are knowledgeable in hazardous waste operations, worked with over 100 experts and organizations in the development of the criteria contained in this manual. The manual was published in October 1985 and is public information. The manual is a gudance document for managers responsible for occupational safety and health programs at inactive hazardous waste sites. The manual is intended for use by government officials at all levels and contractors involved with hezardous waste operations. The manual provides general guidance and is intended to be used as a preliminary basis for developing a specific health and safety program for hazardou operations. Further the major su areas listed in SARA section 128. nearly identical to these major chapters listed in the manual.

Congress indicated that reasonably comprehensive protection was intended for employees at hazardous waste operations, as discussed below, covering more than the minimum requirements specified in the EPA manual (EPA Order 1440.2) and Subpart C of 29 CFR Part 1928. In light of the short period of time Congress directed for issuance of this standard. OSHA's utilization of recognized sources of guidance which have been created by experts in the area and utilizing the resources of relevant agencies is appropriate.

In view of the brief period given for the issuence of this document, it may be necessary to issue minor corrections in the near future.

II. Summary and Explanation of the Standard

Paragraph (a)—Scope, Application, and Definitions

In paragraph (a)(1). *Scope*. OSHA has defined the scope of the standard to include:

(i) Hazardous substance response operations under the Comprehensive Environmental Response. Compensation, and Liability Act c as amended (CERCLA) including initial investigations at CERCLA sites before the presence or absence of hazardous substances has been ascertained:

(ii) Major corrective actions taken in c!can-up operations under the Resource Conservation and Recovery Act of 1978 as amended (RCRA):

(lii) Operations involving hazardous waste storage, disposal and treatment facilities regulated under 40 CFR Parts 284 and 285 pursuant to RCRA except for small quantity generators and those employers with less than 90 days accumulation of hazardous wastes as defined in 40 CFR 282.34;

(1v) Hazardous waste operations sites that have been designated for clean-up by state or local governmental authorities: and

(v) Emergency response operations for releases of or substantial threats of releases of hazardous substances and post-emergency response operations to such releases.

Thus this standard will cover hazardous waste clean-up operations at CERCLA sites. RCRA sites. emergency response sites and those sites designated by State or local governments. It will also cover other hazardous waste operations. such as storage. disposal or treatment of hazardous waste at RCRA facilities.

OSHA believes that Congress intended the interim rule to have a broad scope and application. This is indicated by the legislative intent as reflected in the language of SARA. The language of section 128(e) explicitly states that the Secretary "shall issue interim final regulations under this section. . . " (emphasis supplied). "Under this section" refers to the entire section 128 of SARA. And, as previously noted. section 126(a) mandates safety and health standards for the protection of employees engaged in hazardous waste operations. Thus, OSHA believes Congress intended the interim final rule to mirror section 126 and provide protective provisions to employees engaged in hazardous waste operations.

The argument is buttressed further by the fact that section 126(e) states that the interim final rule shall provide no iess (emphasis added) protection for workers employed by contractors and emergency response workers than the protection contained in the Environmental Protection Agency Manual "Health and Safety Requirements for Employees Engaged in Field Activities" and exiting standards under Subpart C of 29 CFR Part 1928. The two sources cated in section 125(e) are not a limitation on the scope of the interim rule. Rather, this language establishes the minimum amount of

protective provisions, with the broad parameters of employee protection delineated by the remainder of section 128.

This interpretation is reinforced because SARA is a freestanding statutory provision and not an amendment to CERCLA. The clear Congressional intent then is to provide protection to employees whenever they deal with hazardous wastes.

The hazards an employee faces at a RCRA, CERCLA. or emergency response site are the same hazards. The risk of exposure is to the same types of hazardous substances. The scope of the regulation fulfills the Congressional mandate: to effectively provide for employee health and safety at hazardous waste operations and emergency response uncidents.

As indicated in the application provisions, different provisions of the standard apply to clean-up operations, regular hazardous waste operations and emergency response to take into account relevant differences.

Further the term "hazardous wasta operation" is used in section 126(s) of SARA. "Hazardous weste" is also a term used in RCRA and there is no indication from SARA or its legislative history that RCRA facilities were to be excluded from coverage by this interim rule. This is a further reason why OSHA has included RCRA hazardous waste operations under the coverage of this interim final rule. However, small quantity generators: employers who have less than 90 days of hazardous waste accumulation: and solid waste disposal operations which do not involve hazardous waste are not covered by this interim final rule. Also, employees at hazardous waate sites who will not be exposed or do not have the potential to be exposed to hazardous substances are not covered by this interm final rule.

Emergency response employees who respond or will respond to incidents involving hazardous substances are covered by this interim final rule. Public employees of states that have agreements with OSHA under section 18 of the OSH Act must issue regulations at least as effective as these to protect public employees.

Municipal or other sanitary landfills that handle domestic wastes are not covered. Similar waste paper or scrap metal operations are generally not covered because of the type of wastes they handle. But they could be covered if they have clean-ups for or handle hazardous wastes meeting the scope provisions of the standard.

Operations with no exposure to onsite hazardous substances, i.e., road building for site access. construction of on-site or the setting up of temporary facilities in the clean zone or the closure of a RCRA site involving the building of a clay cap over hazard wastes, are considered to be construction activities covered by the standards in 29 CFR Part 1928.

The scope and application provisions carry out the intent of Congress and are consistent with good occupational safety and health policy. Employees performing clean-up operations under CERCLA. RCRA (corrective actions) and post emergency response. generally those employees likely to have the highest exposures to hazardous substances over a longer period. are covered by virtually all the provisions of the rule. Employees exposed to hazardous westes in routine RCRA hazardous waste operations, who are regularly exposed to hazardous wastes but in a more controlled environment. are covered by the more limited requirements of paragraph (o) of the interim final rule. Emergency response workers, exposed usually for short periods to often unknown but possibly high levels of hazardous substances. have specific provisions directed towards this situation.

In paragraph (a)(2), Application: OSHA designates the requirements which apply to the specific work activities covered by this interim final rule. The requirements set forth in paragraph (1) of this section specifically apply to the work conducted by emergency response personnel, such as fire fighters, emergency medical system (EMS) employees and police, when they respond to hazardous substance incidents.

The requirements set forth in paragraph (o) of this section specifically apply to the hazardous waste operations at RCRA sites which are involved in disposal, treatment, storage and handling of hazardous weste. The exclusion of small quantity operators and less than 90-day accumulators excludes from coverage by the interim rule operators such as dry cleaners and gas stations which come within the purview of RCRA but are not hazardous waste operators in the normal meaning of the term. The approximately 4.000 RCRA sites where reasonably large quantities of hazardous wastes are requiarly handled, treated and stored are covered by the rule. This reflects the legislative intent, meets the normal meaning of hezardous waste operations and covers the type of safety and health hazards that this regulation is designed to control.

Most of the requirements of the interim rule apply to clean-up activities of hazardous substances or hazardous westes at CERCLA sites. corrective actions at RCRA sites. and clean-up operations of hazardous substances at emergency incidents after emergency response personnel have concluded their duties.

The employer must also comply with the standards in 29 CFR Parts 1910 and 1928, as well as with the requirements specifically covered in this interim rule. If there is a conflict or overlap, the more protective provisions are to apply. Since this interim rule does not cover all of the hazards present at hazardous waste operations, other OSHA standards in Parts 1910 and 1928 apply also. Other OSHA standards cover many other hazards, and OSHA wants to make clear that the other standards continue to apply. Also, hazardous waste operators who are not within the scope of this standard are covered by the Parts 1910 and 1928 standards.

In paragraph (a)(3), Definitions. OSHA has defined various terms used in this rulemaking. The definitions for hazardous substances and hazardous wastes have been taken from the U.S Environmental Protection Agency (EPA) and U.S. Department of Transportation (DOT) regulations. This has been done to assure consistency and compatibility between this interim rule and the rules and regulations of the EPA and DOT. The remaining definitions have been taken for the most part from SARA, the four agency manual (Reference 6) or existing OSHA standards.

The term "established permissible exposure limit" is defined to give direction as to the appropriate degree of protection needed to be achieved by personal protective equipment and other similar purposes.

Paragraph (b)-General Requirements

in paragraph (b). General requirements. OSHA sets forth for the most part a summary of requirements which are specified in detail in later paragraphs. The preamble discussion for later paragraphs sets forth the reasons for the various provisions. Many of these requirements are part of the minimum requirements which Congress directed OSHA to lesue in section 128(e) of SARA. The EPA manual (EPA Order 1440.2) referenced in section 126(e) requires extensive training and medical surveillance programs. Subpart C of 29 CFR Part 1928. also referenced, requires. in addition, accident prevention programs (§ 1928-20(b)), use of appropriate personal protective equipment (§ 1928.28), semitation and. illumination requirements (§§ 1926.26

and 1928.27), provisions on safe handling of toxic substances (§ 1928.21 (b)(5)), precautions in confined spaces (§ 1928.21(b)(6)) and similar provisions. Congress also directed additional provisions for the proposed regulation. which are considered relevant for the interm regulation. These include engineering controls. maximum exposure limits and monitoring. handling requirements. decontamination procedures and emergency response. Based on this comprehensive statutory direction OSHA believes that the intent of Congress is to have employers implement a safety and health program that will address the recognized serious hazards to employees involved in hazardous waste operations. Therefore. OSHA has incorporated the more important elements of section 126(b). along with the mandatory provisions of section 126(e) of SARA, into this rule. Each general requirement in paragraph (b) calls for employer action and directs the employer to the specific paragraph of this rule that contains the duties in greater detail.

OSHA believes that these requirements are necessary to assure adequate employee protection to the known hazards faced by employees. The language used in these requirements has been adapted from the various documents listed in the Reference section of this preamble.

Three of the subparagraphs in paragraph (b) do not reference other paragraphs in the regulation. Paragraph (b)(1) requires the employer to develop a safety and health program for hazardous waste operations. Such programs are part of the requirements mandated by SARA for the interim rule. Thus, Subpart C of 29 CFR Part 1928 requires such a program in § 1925.20(b) and EPA Order 1440.2 requires training in "safety plan development" (pg. 5). ÖSHA's experience also establishes that a safety and health program is necessary to protect employees so that hazards are assessed and control programs are systematically laid out. Prior OSHA section 6(b) health standards require a compliance plan to set forth a health program to protect employees from the bazard.

Paragraph (b)(14) requires compliance with Subpart P of 29 CFR Part 1928 which covers excavation. OSHA considers that those provisions already apply, but they are singled out because they are particularly important to monitor since much excavation activity occurs on hazardous waste sites.

Paragraph (b)(15) requires employers to noufy contractors and subcontractors of the hazards identified by the employer at hazardous waste operations. Sections 126(b)(2) and (e) of SARA indicate Congress's specific interest in protecting employees of contractors and in involving contractors in the safe operation of hazardous waste sites. This provision assists the contractor to become aware of the risks so that the contractor's employees may be better protected.

Paragraph (c)—Site Characterization and Analysis

For an effective safety and health program, which Congress clearly intends for employees, the employer needs to know the hazards faced by employees in order to develop and implement effective control measures. Site characterization provides the information needed to identify site hazards and to select employee protection methods. The more accurate. detailed, and comprehensive the information available about a site. the more the protective measures can be tailored to the actual hazards that the employees may encounter. Congress clearly intended that such a requirement be included. Subpart C of 29 CFR Part 1928 referenced in section 128(e) of SARA requires "frequent and regu inspections of the job site" (29 CFI. 1928.20(b)(2)). Also section 126(b)(1) or SARA provides for site analysis. Also item #9 of the EPA manual (EPA Order 1440.2) addresses this practice.

Site characterization generally proceeds in three phases:

1. Prior to site entry, gather information away from the site, conduct reconnaissance from the site perimeter and conduct offsite characterization.

 Conduct onsite surveys. During this phase, restrict site entry only to reconnaissance personnel.

3. Once the site has been determined safe for commencement of other activities, continue monitoring to provide an updated source of information about site conditions.

It is important to recognize that site characterization is a continuous process. At each phase of site characterization. information shall be obtained and evaluated to define the potential hazards of the site. This essessment shall be used to develop a safety and health plan for the next phase of work. In addition to the formal information gathering that takes place during the phases of site characterization described here, all site personnel should be constantly alert for new informs" about site conditions. Other requirements of this section have b adopted from reference 6.

Paragraph (d)-Site Cuntrol.

As part of the employers' site safety and health plan, this paragraph requires the employer to consider site control to minimize potential contamination of employees. Several items need to be considered, such as establishing work zones, so that employees know the hazards in different areas and will keep out of hazardous areas where the employees' presence is not required. Use of a buddy system and good site communications will assist in rescue of employees who become unconscious. !rapped or otherwise seriously disabled on site.

Site control is especially important in emergency situations. Paragraph (d) describes the basic components of a program to control the activities and movements of employees and equipment at a hazardous waste site.

Several site control procedures can be implemented to reduce employee exposure to chemical, physical, biological, and safety hazards. The degrees of site control necessary depends on site characteristics, site size. and the surrounding community. The site control program should be established in the planning stages of a project and modified based on new information and site assessments developed during site characterization. The appropriate sequence for implementing these measures should be determined on a site-specific basis. In many cases, it will be necessary to implement several measures simultaneously.

The text used in this paragraph has been adapted from Reference 6. Item 9 of the EPA manual (Order 1440.2) indicates the need for this. In addition Subpart C of 29 CFR Part 1928 provides for regular inspection of job sites so hazards on the site can be controlled.

Paragraph (e)—Training.

The intern final rule includes specific provisions for initial and review training of employees before they are permitted to engage in hazardous waste operations that could expose them to safety and health hazards. Both the EPA manuel and 29 CFR 1928.21 and 1928.22 referred to in section 128(e) of SARA have training and information requirements. The EPA manual has specific provisions for basic, intermediate and advanced training. It requires 40 hours training for employees managing uncontrolled hazardous waste sites. 24 hours for employees engaged in routine activities and 32 hours for intermediate activities. Additionally, section 125 generally has requirements for extensive training programs. The clear congressional intent of the interim final rule training provisions is to provide employees with the knowledge and skills necessary to perform hazardous waste clean-up operations with minimal risk to their safety and health.

The provisions for employees include a minimum of 40 hours of initial instruction off the site, and a minimum of 3 days of actual field experience under the direct supervision of a trained and experienced supervisor, at the time of job assignment. This amount of training is specifically directed by Congress for the interim final rule by its reference to the EPA manual which basically requires this amount of training for hazardous waste operators and Congress has specifically imposed these hour and day requirements under section 126(d) of SARA for the proposed final standard. There are slight differences between the EPA manual and section 126(c) of SARA. But they are sufficiently slight so that OSHA believes it appropriate to make the interim final rule consistent with what Congress directs for the proposed final rule so that employers need not make minor modifications to their training programs after two years.

In addition there are often many hazards at a waste site. The employee needs to be trained to recognize the hazards and appropriate work practices to minimize those hazards. The employee also needs to be well trained in the use of respirators and other forms of PPE. Without training those may not be used effectively and will not provide adequate protection. An extensive training program is necessary to achieve these objectives. The paragraph specifies these and the other items needed for effective training to avoid hazards.

Managers and supervisors directly responsible for hazardous waste site operations are to receive the same training as that of employees and at least eight additional hours of specialized training on managing hazardous waste operations. Since these people are responsible for directing others, it is necessary to enhance their ability to provide guidance and to make informed decisions. Both the EPA manual and section 125(e) of SARA direct eight hours of additional training for supervisors and managers.

The provisions also state that employees shall be retrained on an annual basis on relevant matters such as review of health hazards and use of personal protective equipment. Employees at hazardous waste operations face serious health and safety risks. Reminders are needed of this and of work practices to avoid hazards. Personal protective equipment provides much of their protection. If there is no retraining in the use, care and maintenance of said equipment, such equipment is unlikely to be utilized in a manner to provide adequate protection. The regulation provides for eight hours of annual retraining. The EPA manual for refresher training [liem =10] requires this amount of training.

In all areas of training, whether it be for general site employees, on-site supervisors or for the use of specific equipment, the level of training provided needs to be consistent with the worker's job function and responsibilities. The training information should be presented clearly and, as a further safeguard, refresher training should be supplied to reemphasize the initial training and to update employees on any new policies or procedures.

A less detailed training provision is provided for employees working at routine operation on RCRA sites. Those sites will have more stable working conditions and the hazards will be better identified and more carefully controlled. Therefore OSHA believed not as extensive training is needed for those employees for the interim rule. OSHA specifies 24 hours for the required training based on the EPA manual which specifies this as the basic level of training for most routine field activities. OSHA in the proposal document will request comment whether this or a greater amount of training is appropriate for the permanent rule.

Paragraph (!)-Medical Surveillance

The interim final rule both includes specific provisions for baseline and periodic medical examinations. The EPA manual referred to in section 126(e) of SARA has requirements for both initial or baseline and periodic medical examinations. The examinations are to be provided to those routinely exposed to hazardous substances, to those whose duties are physically taxing and those who routinely wear respirators. In addition section 125(b) provides that routine medical examinations are to be provided to workers engaged in hazardous waste operations. Although the language is slightly different, the clear intent is to provide a comprehensive medical surveillance program for employees engaged in hazardous waste operations where it is medically prudent.

The paragraph states medical surveillance is to be provided to employees who have been or are expected to be exposed to hazardous substances or health hazards above established permissible exposure limits

for 30 or more days in a 12-month period or who wear respirators 30 days during the year. These are the employees who will be at greater health risk and employees who wear respirators need to be examined to determine whether they can safely do so as a routine matter. Some dividing line is needed, because employees who might be present on a hazardous waste site only a few days a year or working in areas such as offices or the periphery where exposures are low would not normally benefit from medical surveillance as their likely cumulative exposures to toxic chemicals would be very low probably not significantly higher than the general population. The EPA manual indicates some dividing line is appropriate because it directs medical surveillance only for employees "routinely" exposed

Wearing respirators for any part of each of 30 days will require medical surveillance because it indicates routine exposure to toxic chemicals. There is no requirement that there be 240 hours of respirator use before medical surveillance is required. Similarly being exposed over established safe levels to several chemicals each for less than 30 days but totalling more than 30 days per year requires medical surveillance. This indicates routine exposures to hazardous substances and also combinations of chemicals may cause synergistic effects creating greater health hazards than an individual chemical.

OSHA has based many of the details of medical surveillance on its expenence in issung health standards under section 6(b) of the OSH Act and as directed by section 6(b)(7) of the Act. Congress would be knowledgable that medical surveillance requirements in these standards represent OSHA's expert judgement of what is an appropriate medical surveillance program.

The appropriate medical tests and examinations depend on the substance an employee is exposed to and whether the employee wears a respirator. As employees on hazardous waste sites will be exposed to differing substances, the paragraph can not specifically state the required tests. Consequently the paragraph states that the employer provide to the physician information on exposures, respirator use, and daties on the site. The physician is then to determine the appropriate medical surveillance protocol in terms of specific tests and examinations. By the employer specifying duties the physician also can judge whether the employee can bandle the arduousness of the work.

In situations where most of the employees on the site have similar

exposures the protocol may be similar for all employees. Where different groups of employees on the site have substantially different exposures, several different protocols may be appropriate for the site s workers depending on exposures.

There are a number of sources for guidance on specific medical examination protocols. Chapter 5 of Reference 6 provides such guidance by groups of chemicals likely to be present on a site. It references other authorities. The manual should be supplied to the physician. It is also a basis for the medical surveillance program required by this paragraph. In addition, the EPA medical monitoring program guidelines referenced by the EPA manual provides guidance on specific protocols.

The paragraph requires an initial or baseline medical examination either prior to the start up date for employees who are currently working at hazardous waste sites or prior to initial assignment to an area where medical examinations will be required. The purpose is to take a detailed medical history and where possible develop a health baseline prior to any exposures so as to be able to evaluate changes which may be connected to hazardous substance exposures. In addition the initial examination will permit evaluation of whether the employee can appropriately wear respirators and whether the employee has preexisting conditions which would make exposure to hazardous substances inappropriate. An initial examination has been required by other OSHA health standards and is recommended in Reference 6.

The physician must be informed of what type of respirators and personal protective equipment an employee is likely to wear. The medical examination is to include appropriate tests to evaluate the employee's ability to wear respirators and PPE.

The physician will also specify the protocol of the periodic examinations. These may be different from the initial examination, for example, only an updated medical history would be required. The periodic examinations are required yearly. OSHA's experience in other health standards has been that this is an appropriate period and it is also recommended by Reference 6. EPA's medical monitoring program guidelines cross referenced in the EPA manual recommends baseline annual examination generally and a termination examination. It is reasonable to determine periodically whether exposures have brought medical changes and to identify conditions caused by chemicals at an early stage to permit more effective treatment. In some

circumstances, the physician ma advise more frequent examinations. _

Examinations are also to be provided when the employee brings to the employer s attention signs or symptoms indicating possible overexposure to hazardous substances. The employee is to be trained in recognizing what symptoms may indicate substances to which the employee is exposed. Examples may be dizziness or rashes. Examinations are also required, when medically appropriate. during emergencies when exposure to higher levels is possible. Por example, a urinary phenol test is appropriate for employees exposed to high lavels of benzene in an emergency.

Finally, a medical examination is required for employees who have been required to have medical examinations upon termination of employment or reassignment to an area where medical examinations are not required. This is to detect conditions which have developed prior to departure and is recommended by the EPA program.

The medical examination is to be provided under the supervision of a licensed physician, i.e., the person must be qualified to make medical judgements. As provided by section 6(b)(7) of the OSH Act. the employ to pay the cost of the examination addition provisions are included so the employee is not discouraged from taking the examination. The exam is to be given at a reasonable time and place. If given during regular working hours the employee shall receive the employer's normal pay for that time. If the exam is given outside regular working hours, the employee shall be paid his regular wages for the time spent taking and waiting for the examination.

The physician shall make a report to the employer of medical conditions which may make the employee at increased risk to work at the site and any recommendations on limitations on use of respirators and other PPE as a result of the medical conditions. This will provide guidance for the safe employment of the employee at the site. The physician shall not reveal diagnoses or conditions unrelated to employment, but shall inform the employee directly of those conditions and any and all occupationally related conditions.

The medical paragraph requires that appropriate records be kept to assist in future evaluation of the employee's health. Secondarily, this information may assist in research on occupational related disease. Records should be kept pursuant to the provisions of 29 CFP 1910.20. Full consideration was give that standard to appropriate retention periods.

Paragraph (g)—Engineering Controle, Work Practices, and Personnel Protective Equipment

Anyone entering a hazardous waste site must be protected against potential hazards. The purpose of engineering controls, work practices, and PPE is to shield or isolate individuals from the chemical, physical, and biologic hazards that may be encountered at a hazardous waste site. Careful selection and use of adequate engineering controls, work practices and PPE should protect any employee from health and many other hazards including hazards to the respiratory system, skin, eyes, face, hands, feet, head, body, and hearing.

Requirements of both Subpart C of 29 CFR Part 1928 and the EPA manual mandated to be included in the standard by Congress cover the provision and use of personal protective equipment. Se for example. 29 CFR 1928.25 and items 7(a). 9(e)(7) and 9(b)(2) of the EPA manual is addition existing OSHA regulations which apply to bezardous waste operations. in 29 CPR Part 1910, Subpart Z require exposures to various toxic and hazardous substances to be controlled with engineering controls if feasible, otherwise with PPE. These requirements apply now to employers and workers on Superfund ates pursuant to EPA regulations in 40 CFR Part 300. Finally, Congress specified in section 125(b) that there should be both PPE and engineering control provisions for the permanent finel standerd.

Paragraph (g)(1) basically carries over the existing requirements of Subpart Z. OSHA regulated touc and hezerdous substances are to be controlled to the permissible exposure limit if feasible. If not feasible they are to be controlled with PPE.

Paragraph (g)(2) provides that to achieve established permusible exposures limits for substances not regulated by OSHA. the employer may use an appropriate combination of engineering controls, work practices, and PPE. As these are interim regulations. preference for engineering controls where not already required would not be appropriate because of the limited time frame of this regulation and the frequent inability to install such controis in a short period. In addition it is OSHA's experience that this is an appropriate approach based on the emergency temporary standards it has issued which are also in effect for a limited period. OSHA will ask for comment in these areas in the proposal document.

Examples of engineering controls which may be feasible are pressurized. cabe on materials handling equipment or pressurized control rooms in materials handling areas. However, in many cases personal protective equipment will be the only feasible means for providing protection to employees engaged in hazardous waste operations. The selection of personal protective equipment (PPE) must be based on the information obtained during the site characterization and analysis, as is required by paragraph (g)(3)(i) of this standard. Once an estimate of the types of hazards and their potential concentration has been obtained, the proper respirators and protective clothing can be selected based on the performance characteristics of the PPB relative to the onte hezards and work conditions, as is required by paragraph (g)(3)(ii) of the standard. These requirements are derived from Reference 6 and are also supported by a NIOSH document, "Personal Protective Equipment for Hazardova Materiala Incidents: A Selection Guide." These two document also support the requirements of paragraphs (g)(2)(iii) and (g)(2)(iv) which require positive pressure respirators with escape provisions to be used in IDLH atmospheres and totally-encapsulating chemical protective suits to be used where contact of the skin by the substance would be an IDLH situation.

Proper respirator selection, as required by this standard and 29 CFR 1910.134, involves providing a sufficient protection factor through the type of respirator used, respirator fitting, work site conditions, and respirator selection and use program. Proper protective clothing selection, as required by this standard, involves choosing protective clothing made of materials and construction which will prevent breakthrough of hazardous substances by permeation and penetration. or reduce the level of exposure to a safe level during the employee's duration of contact. Information on the performance characteristics of PPE is available in test reports and manufacturer's literature. Appendix B provides son-mandatory guidelines on classifying substance hazards as four levels (A. B. C. and D), and matching four levels of appropriate protection provided by different protective ensembles. These guidelines may be used as a basis for protective clothing selection, and the selection further refined when more information is obtained, as provided for in paragraph (g)(2)(v) of the standard. (In certain circumstances, this standard does specify the appropriate level of

protection. See paragraph (c)(4)(iii)). Paragraph (g)(3)(vi) cross reference the existing requirements to select and use PPE pursuant to the requirements of 23 CFR 1910. Subpart I.

Paragraph (g)(4) requires totallyencapsulating suit materials used for Level A protection (the highest level of protection) to provide protection from the specific hazards which have been identified as requiring that level of protection. The purpose of this requirement is to be certain that the suit selected is comprised of materials which will provide the necessary protection. since no one material will provide protection from all hazarda. Paragrapha (g)(4)(ii) and (g)(4)(iii) require totallyencapsulating suits to be capable of maintaining positive air pressure to beloprevent inward leakage of bazardous substances, and to be capable of preventing inward gas leakage of more than 0.5 percent. These requirements. which are based on testing of totallyencapsulating suits, are included to establish a minimum level of suit performance so that their level of protection can be quantified for proper selection. The example test methods in Appendix A for totally-encapsulating chemical protective suits were taken from draft American Society for Testing and Materials committee documents.

Paragraph (g)(5) requires a PPE. program to be established. This requirement is based upon reference 6. 29 CFR 1928.28, EPA menual items 4 and 7(g), and is included, since, in most cases. PPE will be the only protection. feasible (or employee protection, and because the amount of protection. afforded by PPE is dependent upon se many factors, such as selection, fit, work duration and conditions, and decontamination. The PPE program is required to insure that the level of protection afforded by PPE is sufficient and continues to be sufficient for employee safety during hazardous waste operations.

Paragraph (h)-Monitoring

It is essential that employers ba provided with accurate information on employee exposures in order to implament the correct PPE, engineering controls, and work practices. Airborns contaminants can present a significant threat to employee safety and health. Thus, identification and quantification of these contaminants through air monitoring is an essential component of a safety and health program at a hazardous waste site. Reliable measurements of airborne contaminants are useful for selecting personal protective equipment, determining whether engineering controls can achieve permissible exposure limits and which controls to use. delineating areas where protection is needed, assessing the potential health effects of exposure. and determining the need for specific medical monitoring. As mentioned above, section 128(e) of SARA mandates the use of PPE by its direction that at a minimum the requirements of the EPA manual and Subpart C be followed. Those include requirements for use of PPE. But PPE cannot be effectively used unless monitoring has identified the type of PPE to be used. This is a further reason to include this provision in the interm final rule.

The language of this paragraph was adapted from reference 6.

Paragraph (i)—Informational Programs

In paragraph (i), *Informational Programs.* OSHA is requiring employers. as part of their safety and health program. to develop and implement a site specific safety and health plan for each hazardous waste operation site.

The site safety and health plan shall be developed by the employer, utilizing the other parts of the organizational plan and the employer's safety and health program. The site safety and health plan will address the anticipated safety and health hazards of each work operation or activity and the means to eliminate the hazards or to effectively control them to prevent injury or illness.

This site safety and health plan is to include: [1] The names of those responsible for assuring that safe and healthful practices and procedures are followed on the whole site: (2) risk analysis or systems analysis for specific work tasks or operations on the site: (3) employee training assignments both off site and on-the-job-training on site; (4) the list of required personal protective equipment needed for each work task and operation on site: (5) the employer's medical surveillance program for the site; (6) the methods for identification and characterization of safety and health hazards on the site including the monitoring procedures that will be done throughout the work on site; (7) site control measures including those for establishing work zones on the site: (8) the necessary decontamination procedures which are matched to the kinds of anticipated contaminants to be cleaned from employees and equipment: (9) the standard operating procedures to be used by employees on site: and (10) the contingency plan for emergencies and confined space entry procedures. Safety meetings and briefings and site inspections shall also be mentioned in the plan as well as the procedures to be

followed in changing or modifying the plan.

The site safety and health plan is necessary to protect employee health. There are many hazards at a hazardous waste operation which need to be determined and addressed. The plan provides that this will be done in a systematic manner so that hazards will not be missed and so that needed protective action will not be overlooked. The approach used has be adapted from reference 6.

Paragraph (j)—Handling Drums and Containers

The handling of drums and containers at bazardous waste sites poses one of the greatest dangers to bazardous waste site employees. Hazards include detonations, fires, explosions, vapor generation, and physical injury resulting from moving heavy containers by hand and working around stacked drums. heavy equipment, and deteriorated drums. While these hazards are always present, proper work practices can minimize the risks to site personnel. Handling and storage of hazardous substances is addressed in item (a) of the EPA manual.

Containers are handled during characterization and removal of their contents and during other operations. Many of the hazards encountered during the handling of drums occur during the handling of containers. The relative size of a container when compared to the size of a drum is no indication of the degree of hazard posed by the container. They should be treated in accordance with the level of hazard posed by their contents not by their size. The language used in this paragraph was adapted from Reference 6.

Paragraph (k)—Decontamination

As part of the care of PPE required by this standard, decontamination is a necessary practice to properly protect those employees who may be exposed to hazardous substances. Decontamination provisions protect an employee from being exposed to hazardous substances which would otherwise be on the employee's PPE when it is removed. The standard requires that a decontamination plan be developed and implemented before any employees or equipment may enter areas on site where potential for exposure to hazardous substances exists.

As required by the standard, decontamination procedures and areas shall be developed to minimize hazardous exposures to employees whose equipment and PPE are being decontaminated, as well as to

employees who are assisting in thedecontamination of workers and equipment. These measures are required since without proper procedures and decontamination areas, employees may be unknowingly exposed to bazardous substances which have contacted, or otherwise adhered to equipment and clothing. The standard also requires that all employees, clothing, equipment and decontamination fluids and equipment be decontaminated or disposed of before leaving a contaminated area. These provisions are required so that contaminated persons and materials do not leave the "hot zone" and thereby expose other employees and persons to hazardous substances.

Decontamination methods and cleaning fluids must be matched to the particular bazardous substance at the site in order for the decontamination procedures to be effective in removing the hazards from PPE and other equipment. No one decontamination fluid will be effective for all bazardous substances. As required by the standard the decontamination program must be effective and it must be monitored by the site safety and health officer to maintain its effectiveness. These requirements are included so that employees are not exposed to hazart' substances by reusing PPE and other equipment which are still contaminate.

The language used in this paragraph was adapted from reference 6.

Paragraph (I)—Emergency Response

Section 126(e) of SARA specifically discusses protecting "emergency response workers." in addition in the EPA manual under items 4 and 9 and in 29 CFR 1928.23 and 1926.24 call for preparations and planning for emergencies. Congress made its intent clear that emergency planning and response is an important part of any employer's safety and health program and indicated that it is to be eddressed in the interim final rule.

In paragraph (1)(1), Emergency Response. General OSHA is requiring employers covered in paragraph (a)(2)(ii), who are involved in hazardous waste operations, as part of their on-site contingency planning to develop and implement an emergency response plan. These employers are to inform all their employees on the waste site about the emergency response plan. The plan is to be available for use prior to the start of work on the site. The plan will be a part of the site safety and health plan. The elements of the emergency response plan will include: (1) Recognition of emergencies: (2) methods or procedure for alerting employees on site; (3)

evacuation procedures and routes to places of refuge or safe distances away from the danger area: (4) means and methods for emergency medical treatment and first aid: (5) line of authority for employees: and (6) on-site decontamination procedures: site control means and methods for evaluating the plan.

Employers whose employees will be responding to hazardous substance emergency incidents from their regular work location or duty station, such as e fire department, fire brigade or emergency medical service, will also be required to have an emergency response plan. These employees which may be called upon to respond to bazardous substance emergency incidents involving a railroad tank car, motor carrier tank truck or to a plant location are considered off-site emergency response activities under this section. The emergency response plan is to include the incident command system required in paragraph (1)(3) of this section.

In paragraph (1)(2). Hozardous waste operations, on-site emergency response. OSHA is requiring the training of on-ente emergency response personnel to have the same basic training as for the other employees involved in on-site bazardous waste.operations plus the training needed to develop and retain the necessary skills for anticipated emergency response activities. Also, the procedures for handling bazardous substances on-site emergency incidents are to be oriented to the specific site and made a part of the emergency response plan.

The requirement of paragraphs [1](3] and (1)(4) apply more broadly to all employers whose employees respond to off-site emergency incidents. In paragraph (1)(3). Off-site emergency response. OSHA is mandating that employers. such as fire departments, emergency medical and first-aid squads. fire brigades, etc., conduct monthly training sessions for their employees totalling 24 hours annually.

Nota—OSHA does not have junadiction over state and local government employees. OSHA state plan states must insue regulations as effective as these to cover state and local government employees in the state.

Training activities, such as breathing apparatus use, training, hose handling and preplanung may be used as training subjects for the monthly sessions provided hazerdous substance incident operations are uncluded in the presentation, discussion or drill. These training sessions and drills must involve at least 24 hours of training on an annual basis.

The incident command system shall be established by these employers for the incidents that will be under their control and shall be interfaced with the other organizations or agencies who may respond to such an incident. The National Transportation Safety Board. as a result of its investigation of hazardous materials incidents, has consistently recommended that better state and local emergency response planning be done to reduce the loss of life and property and that a system using a command post and on-scene commander be unplemented. (See Special Investigation Report, On-scene Coordination Among Agencies at Hazardous Materials Accidents, NTSB-HZM-79-3. September 13, 1979; and Multiple Vehicle Collisions and Fire. Caldecott Tunnel near Oakland. California, NTSB/HAR-83/01, National Transportation Safety Board, Washington, DC. April 7, 1982, for further information.) Where available. state and local district emergency response plans shall be utilized in developing the incident command system and the emergency response plan to essure compatability with the other emergency responding agencies or employers.

in paragraph (1)(4), Hazardous materials teams. OSHA is requiring employers, who utilize specially trained teams involved in intimate contect with controlling or handling hazardous substances, to provide special training for the affected employees in such areas as care and use of chemical protective clothing, techniques and procedures for . stopping or controlling leaking containers and decontamination of clothing and equipment for anticipated hazardous substance incidents. The employer is to make available to each team member a physical examination by a licensed physician and to implement a medical surveillance program in accordance with the requirements of paragraph (f) of this section.

In paragraph (1)(5). OSHA is requiring employers covered in paragraphs (a)(2) (i) and (ii) of this section, who will be involved in cleaning up hazardons wasts after the emergency response activities are concluded, to comply with the same requirements that apply to others involved with hazardous wasts clean-up operations. These hazardous wasts clean-up operations will be typically done by special contractors and not by those agencies involved in responding to the instal emergency incident.

Paragraph (m)—Illumination

OSHA is required by SARA in section 128(e) to cover lighting of the worksite. In paragraph (m). *Illumination*. OSHA requires certain minimum illumination. levels for work areas that are occupied by employees. Section 126(e) of SARA requires as a minimum the inclusion of the requirements of Subpart C of 29 CFR. Part 1926. Section 1926.26 of that Subpart requires the amount of illumination set forth in this paragraph.

Paragraph (n)—Sanitation for Temporary Worksites

in paragraph (a). Sanitation for temporary workettes. OSHA sets minimum requirements for potable and non-potable water supplies, tailet (actives, and other areas related to sanitation at temporary workplaces. OSHA is mandated by SARA in section 126(e) to include sanitation requirements in the interim final rule since it requires the incorporation of provisions of Subpart C.

Paragraph (o)—Operations Conducted Under the Resource Conservation and Recovery Act of 1978 (RCRA)

OSHA is providing a separate paragraph for operations conducted at worksites involving hazardous waste storage, disposal and treatment operating under the Resource Conservation and Recovery Act of 1976 (RCRA). This separate paragraph of requirements is appropriate because RCRA site operations. (not including major corrective actions and their associated hazards which are like CERCLA sites and are covered by the main part of the standard) generally are different from the operations and hazards found on a CERCLA clean-up site. For example, RCRA sites covered. by this paragraph tend for the most part to be fixed on-going operations. involving the receiving, processing storage, treatment, and disposel of bazardous westes or substance from outside sources. CERCLA sites on the other hand are temporary emergency clean-up operations involving often undefined and substantial quantities of hazardous substances.

Consequently bezards should be better controlled and more routine and stable for the RCRA sites covered by this paragraph and so less extensive requirements are appropriate.

Paragraph (p)-Start-up Dates

Section 128(e) of SARA directs that these interim final regulations take effect on issuance. Consequently, these regulations do become effective on issuance. However, completion of implementation for some provisions is not feasible immediately. For these provisions, commencement of implementation must begin immediately but completion of full compliance is required as soon as possible or feasible but in no case later than a specified date, which is no longer than three months.

It is OSHA's judgment that all provisions can be fully implemented by the periods specified. OSHA also believes that the immediate effectiveness provisions specifically upply to the mandatory requirements.

OSHA does not believe that Congress intended that work at current hazardous waste operations stop until implementation of all requirements can be feasibly completed. This paragraph so indicates. However, for new sites, these requirements can be completed in advance. It is not OSHA's intention that emergency actions necessary to protect the public safety and health be prevented because in a particular circumstance it is not feasible to carry out particular requirements of this standard in the time needed to respond to the emergency

III. References

I. Superfund Amendments and Reauthorization Act of 1986 (SARA), Pub. L. Mailton

2. Comprehensive Environmental Response. Compensation and Liability Act of 1940 (CERCLA or "Superfund"), Pub. L 96-3:0. Dreember 11, 1980, 94 Stat, 2787.

3. Resource Conservation and Recovery Act of 1978 (RCRA), Pub. L 94-580, October 21, 1978, 90 Stut. 2795.

4. "Health and Safety Requirements for Employees Engaged in Field Activities". Fayronmental Protection Agency Order 1440 2. U.S. Environmental Protection Agency, July 12, 1961.

5. Subjects C and D of 29 CFR Part 1928. 6. "Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities.". Occupational Safety and Health Autoministration. Environmental Protection Agency, U.S. Coast Guard, and National Institute for Occupational Sufety and Health. D1HIS (NIOSH) Publication No. 85-115, Octoner 1965.

IV. Regulatory Impact Analysis. Regulatory Flexibility Analysis and Environmental Impact Analysis

OSHA anticipates that this interim final standard will have a significant impact upon employers and their employees who work at CERCLA sites and at some RCRA sites: and who respond to emergency clean-ups of hazardous substance spills. OSHA has had little time since the enactment of SARA to collect information concerning these industries. As a result, the currently available information is insufficient for OSHA to use to estimate the potential benefits and costs that would occur as a consequence of compliance with this interim final rule OSHA is collecting additional information to be used in conjunction with the information from the comments that will be received in response to publication of the proposed rule covering hazardous waste operations. This information will be sufficient for OSHA to provide a complete Regulatory Impact Analysis for the final rule that will govern hazardous waste operations.

Regulatory Flexibility Act Analysis. The requirements of the Regulatory Flexibility Act are not applicable to this interim final rule, under 5 U.S.C. 603(a). because notice and comment proposed rulemaking under the Administrative Procedures Act. or any other statute. is not required.

Environmental Impact Analysis. The National Environmental Policy Act (NEPA) of 1969 (42 U.S.C. 4321 et seq). as implemented by the regulations (40 CFR Part 1500) of the Council on Environmental Quality (CEQ), requires that federal agencies assess their regulatory actions to determine if there is a potential for a significant impact on the quality of the human environment and, if necessary, to prepare an environmental impact statement.

In accordance with these requirements and DOL NEPA Compliance Procedures (29 CFR Part 11. Subpart B. section 11.10(a)(4)), OSHA has determined that due to the compressed rulemaking schedule imposed by the Congress in issuing the interim regulation, no environmental impact statement will be prepared for this interim rule.

In similar situations, for example. when an emergency temporary standard (ETS) has been issued, the courts have held that NEPA does not require advance preparation of an environmental statement for an ETS (Dry Color Manufacturing Association v. U.S. Department of Labor: 486 F. 2d 98. 107 (3rd Cir. 1973)). This interim final standard is similar in nature to an ETS issued for relatively brief periods for short notice pursuant to section 6(c) of the Occupational Safety and Health Act of 1970 and section 101(b) of the Federal Mine Safety and Health Act of 1977. The DOL NEPA regulations set forth in 29 CFR Part 11. Subpart B. section 11.10(a)(4). provide that in these situations the regulations set forth in 40 CFR Parts 1500 et seg may not be strictly observable.

OSHA. however. will assess the environmental effects of the proposed permanent regulation of hazardous waste sites. The possibility that increased training related to empli safety and health protection will all affect and reduce inadvertent environmental releases of hazardous substances at waste sites will be analyzed. The results of this study will be available for review and comment prior to the hearing on the proposed permanent standard and will be an appropriate issue for discussion at the public hearings scheduled for the proceeding.

In the internm. OSHA welcomes any comments on any environmental effects that might occur as a result of promulgation of a rule on hazardous waste sites.

V. International Trade

OSHA has preliminarily concluded that this interm final rule will not significantly affect international trade. The firms that will be primarily affected by this interim final rule deal with hazardous waste products and are not involved in international trade. In addition, the hazardous wastes to be handled under this interim final rule are primarily by, products from previously manufactured goods and consequently. any potential costs would not be borne by the goods that are currently being traded. Nevertheless, the information that OSHA is collecting and the information that will be supplied in response to the publication of the proposed rule covering Hazardous Waste Operations will be carefully reviewed and analyzed to establish the potential impacts of the final rule upon international trade.

VI. State Plan States

This Federal Register document adds an interm final rule (section 1910.120. "Hazardous Waste Operations and Emergency Response") to existing Subpart H of 29 CFR Part 1910. OSHA's general industry standards on hazardous materials. The 25 States with their own OSHA approved occupational safety end heelth plans must develop a comperable standard applicable to both the private and public (State and local government employees) sectors within six months of the publication date of this interim final rule or show OSHA why there is no need for ection. e.g., because an existing state standard covering this area is elready "at least as effective" as the new Federal standard. These states are Alaska, Arizona, California. Connecticut (for state and local government employees only). Hawaii, Indiana, Iowa, Kentucky, Maryland, Michigan, Minnesota, Neveda, New Mexico, New York (for state and local government employee

only). North Carolina. Oregon. Puerto Rico. South Carolina. Tennessee. Utah. Vermont. Virginia. Virgin Islands. Washington, and Wyoming. Until such time as a state standard is promulgated. Federal OSHA will provide interim enforcement assistance. as appropriate. in these states.

List of Subjects in 29 CFR Part 1910

Containers. Drums. Emergency response. Flammable and combustible liquids. Hazardous materials. Hazardous substances. Hazardous wastes. Incorporation by reference. Materials handling and storage. Personal protective equipment. Storage areas. Training. Waste disposal.

VII. Immediate Effectiveness and Absence of Notice and Comment

Section 128(e) of SARA specifically provides that the "Secretary of Labor shall issue interim final regulations under this section within 60 days . . . after date of enactment. The express use of the phrase "interim final regulations." which in the rulemaking context commonly describes a rule issued without notice and comment, in connection with the extremely limited tune frame provided by this section. makes clear that Congress intended this rule to be issued without the timeconsuming process of notice and comment. The Agency, therefore. concludes that neither the notice and comment rulemaking provisions of the OSH Act nor those of the Administrative Procedures Act are applicable to the issuance of this interim final rule. The Agency also expressly finds that "good cause" exists under 5 U.S.C. 553(b)(B) for not providing notice and comment because notice and comment procedures. under these circumstances, would be impractical and contrary to the public interest.

Section 125(e) also expressly provides that "Such internm final regulations shall take effect upon issuance. . . ." OSHA finds this specific direction of law requires the Agency to issue this rule with an immediate effective date and, further, constitutes good cause not to delay the effective date of this rule until 30 days after publication under 5 U.S.C. 553(d).

Authority

This document has been prepared under the direction of John A. Pendergrass. Assistant Secretary of Labor for Occupational Safety and Health. U.S. Department of Labor, 200 Constitution Avenue NW., Washington. DC. Pursuant to section 126(e) of the Superfund Amendments and Reauthorization Act of 1986 (Pub. L. 99-499). Sections 6 and 8 of the Occupational Safety and Health Act of 1970 (22 U.S.C. 655, 657). Sections J and 4 of the Administrative Procedures Act (5 U.S.C. 552(a), 553), and Secretary of Labor s Order 9-83 (48 FR 35736), 29 CFR Part 1910 is amended by adding a new § 1910.120. Hazardous Waste

Operations, as set forth below, effective December 19, 1986.

Signed at Weshington. DC this 16th day of December 1986.

John A. Peadergrass,

Assistant Secretary of Labor.

PART 1910-OCCUPATIONAL SAFETY AND HEALTH STANDARDS

1. The Authority citation for Subpart H of Part 1910 is amended by adding the following:

Authority: * * Section 1910.120 issued under the authority of section 128(e) of the Superfund Amendments and Reauthorization Act of 1986 (Pub. L 99-498), Sections 6 and 6 of the Occupational Sefety and Health Act of 1970 (29 U.S.C. 855, 857), sections 3 and 4 of the Administrative Procedure Act (5 U.S.C. S52(a), 533) and Secretary of Labor's Order 9-83 (48 FR 35738).

2. Part 1910 of Title 29 of the Code of Federal Regulations is amended by adding a new § 1910.120 to read as follows:

§ 1910.120 Hazardous waste operatione and emergency response.

(a) Scope. application, and definitions.—(1) Scope. This section covers employers and employees engaged in the following operations:

(i) Hazardous substance response operations under the Comprehensive Environmental Response. Compensation. and Liability Act of 1930 as amended (42 U.S.C. 9601 *et seq*) (CERCLA), including initial investigations at CERCLA sites before the presence or absence of hazardous substances has been ascertained:

(ii) Major corrective actions taken in clean-up operations under the Resource Conservation and Recovery Act of 1975 as amended (42 U.S.C. 6901 *et seq*) (RCRA);

(iii) Operations involving hazardous waste storage. disposal and treatment facilities regulated under 40 CFR Parts 264 and 265 pursuant to RCRA. except for small quantity generators and those employers with less than 90 days accumulation of hazardous wastes as defined in 40 CFR 282.34;

(iv) Hazardous waste operations sites that have been designated for clean-up by state or local governmental authorities; and

(v) Emergency response operations for releases of or substantial threats of releases of hazardous substances and post-emergency response operations for such releases.

(2) Application. (1) All requirements of Part 1910 and Part 1928 of Title 29 of the Code of Federal Regulations apply pursuant to their terms to hazardous waste operations (whether covered by this section or not). In addition the provisions of this section apply to operations covered by this section. If there is a conflict or overlap, the provision more protective of employee safety and health shall apply. 29 CFR 1910.5(c)(1) is not applicable.

(ii) All paragraphs of this section except paragraph (o) apply to hazardous substance response operations under CERCLA. major corrective actions taken in clean-up operations under RCRA. post-emergency response operations. and hazardous waste operations that have been designated for clean-up by state or local governmental authorities.

(iii) Only the requirements of paragraph (o) of this section apply to those operations involving hazardous waste storage, disposal, and treatment facilities regulated under 40 CFR Parts 284 and 285, except for small quantity generators and those employers with less than 90 days accumulation of hazardous wastes as defined in 40 CFR 282.34.

(iv) Paragraph (I) of this section applies to emergency response operations for releases of or substantial threats of releases of hazardous substances.

(3) Definitions—"Buddy system" means a system of organizing employees into work groups in such a manner that each employee of the work group is designated to observe the activities of at least one other employee in the work group. The purpose of the buddy system is to provide quick assistance to those other employees in the event of an emergency.

"Decontamination" means the removal of hazardous substances from employees and their equipment to the extent necessary to preclude the occurrence of foreseeable adverse health effects.

"Emergency response" means response to any occurrence which results, or is likely to result, in a release of a hazardous substance due to an unforeseen event.

"Established permissible exposure limit" means the inhalation or dermal permissible exposure limit specified in 29 CFR Part 1910. Subpart Z. or if none is specified the exposure limits in "NIOSH Recommendations for Occupational Health Standards" dated September 1986 incorporated by reference. or if neither of the above is specified, the standards specified by the American Conference of Governmental Industrial Hygienists in their publication "Threshold Limit Values and Biological Exposure Indices for 1986–87" dated 1986 incorporated by reference. or if none of the above is specified, a limit based upon a published study or manufacturers' safety data sheet brought to the employer's attention. The two documents incorporated by reference are available for purchase from the following:

- NIOSH. Publications Dissemination. Division of Standards Development and Technology Transfer. National Institute for Occupational Safety and Health. 4676 Columbia Parkway. Cincinnat. OH 45228, (513) 841-4287
- American Conference of Covernmental Industrial Hygnenists, 6500 Glenway Ave., Building D-7, Cincinnati, OH, 45211-4438, (513) 661-7881

and are available for inspection and copying at the OSHA Docket Office. Docket No. S-780. Room N-3871, 200 Constitution Ave., NW., Washington. DC 20210.

"Hozardous substance" means any substance designated or listed under (I) through (iv) below. exposure to which results or may result in adverse effects on the health or safety of employees: (i) any substance defined under

section 101(14) of CERCLA.

(ii) any biological agent and other disease-causing agent as defined in section 104(a)(2) of CERCLA.

(iii) any substance listed by the U.S. Department of Transportation and regulated as bazardous materials under 49 CFR 172.101 and appendices, and

(iv) hazardous waste.

"Hozardous waste" means (i) a waste or combination of wastes as defined in 40 CFR 251.3. or (ii) those substances defined in 49 CFR 171.8.

Hazardous waste operation" means any operation involving employee exposure to hazardous wastes, hazardous substances, or any combination of hazardous wastes and hazardous substances that are conducted within the scope of this standard.

"Hazardous waste site" or "site" means any facility or location at which hazardous waste operations within the scope of this standard take place.

"Health hazard" means a chemical, mixture of chemicals or a pethogen for which there is statistically significant evidance based on at least one study conducted in accordance with established scientific principles that acute or chronic health effects may occur in exposed employees. The tarm "health hazard" includes chemicals which are carcinogens, toxic or highly toxic agents, reproductive toxins, irritants, corrosives, sensitizars, hepatotoxins, nephrotoxins, neurotoxins, agents which act on the hematopoietic system, and agents which damage the lungs, skin, eyes, or mucous membranes. Further definition of the terms used above can be found in Appendix A to 29 CFR 1910.1200.

"IDLH" or "Immediately dangerous to life or health" means any condition that poses an immediate threat to life, or which is likely to result in acute or immediate severe health effects. This includes oxygen deficiency conditions.

"Immediate severe health effects" means any acute clinical sign or symptom of a serious. exposure-related reaction manifested within 72 hours after exposure to a bazardous substance.

"Oxygen deficiency" means that concentration of oxygen by volume below which air supplying respiratory protection must be provided. It exists in atmospheres where the percentage of oxygen by volume is less than 19.5 percent oxygen.

"Site safety and health officer" means the individual located on a hazardous waste site who is responsible to the employer and has the authority and knowledge necessary to implement the site safety and health plan and verify compliance with applicable safety and health requirements.

(b) General requirements-{1) Safety and health program. Each employed shall develop and implement a safety and health program for its employees involved in hazardous wasta operations. The program, as a minimum, shall incorporate the requirements of this section and be provided, as appropriate. to any subcontractor or its representative who will be involved with the hazardous waste operation. The program shall be designed to identify, evaluate, and control safety and health hazards and provide for emergency response for hazardous waste operations.

(2) Site characterization and analysis. Hazardous waste sites shall be evaluated in accordance with paragraph (c) of this section to identify specific site hazards and to determine the appropriate safety and health control procedures needed to protect employees from the identified hazards.

(3) Site control. Site control procedures shall be implemented in accordance with paragraph (d) of this section before clean-up work begins to control employee exposure to hazardous substances.

(4) Training. Initial or refresher or review training meeting the requirements of paragraph (a) of this section shall be provided to employees before they are permitted to engage in hazardous waste operations that could expose them to hazardous substant safety. or health hazards.

(5) Medical surveillance. Medical surveillance shall be provided in accordance with paragraph (f) of this section for employees exposed or potentially exposed to hazardous substances or health hazards or who wear respirators.

(6) Engineering controls. work practices and personal protective equipment. Engineering controls. work practices, personal protective equipment, or a combination of these shall be implemented in accordance with paragraph (g) of this section to protect employees from exposure to hazardous substances and health hazards.

(7) Monitoring. Monitoring shall be performed in accordance with paragraph (h) of this section to assure proper selection of engineering controls, work practices and personal protective equipment so that employees are not exposed to levels which exceed established permissible exposure limits for hazardous substances.

(8) Informational program. Employees. contractors. and subcontractors or their representative shall be informed of the degree and nature of safety and health hazards specific to the work site by using the safety and health plan out! in paragraph (i) of this section.

(9) Material handling. Hazardous substances and contaminated soils, liquids, and other residues shall be handled, transported, labeled, and disposed of in accordance with paragraph (j) of this section.

(10) Decontamination. Procedures for all phases of decontamination shall be developed and implemented in accordance with paragraph (k) of this section.

(11) Emergency response. Emergency response to hazardous waste operation incidents shall be conducted in accordance with paragraph (1) of this section.

(12) *Illumination*. Areas accessible to employees shall be lighted in accordance with the requirements of paragraph (m) of this section.

(13) Sanitation. Facilities for employee sanitation shall be provided in ... accordance with paragraph (n) of this section.

(14) Site excavation. Site excavations created during initial site preparation or during hazardous waste operations shall be shored or sloped to prevent accidental collapse and conducted in accordance with Subpart P of 29 CFR Part 1928.

(15) Contractors and sub-contractors An employer who retains contractor (sub-contractor services for work in hazardous waste operations shall inform those contractors. sub-contractors. or their representatives of any potential fire. explosion. health or other safety hazards of the hazardous waste operation that have been identified by the employer.

(c) <u>Site characterization and analysis</u>. (1) A preliminary evaluation of a site's characteristics shall be performed prior to site entry by a trained person to aid in the selection of appropriate employee protection methods prior to site entry. During site entry, a more detailed evaluation of the site's specific characteristics shall be performed by a trained person to further identify existing site hazards and to further aid in the selection of the appropriate engineering controls and personal protective equipment for the tasks to be performed.

(2) All suspected conditions that may pose inhalation or skin absorption hazards that are immediately dangerous to life or health (IDLH) or other conditions that may cause death or senous harm shall be identified during the preliminary survey and evaluated during the detailed survey. Examples of such hazards include, but are not limited to, confined space entry, potentially explosive or flammable situations. visible vapor clouds, or areas where biological indicators such as dead animals or vegetation are located.

(3) The following information to the extent available shall be obtained by the employer prior to allowing employees to enter a site:

(i) Location and approximate size of the site.

(ii) Description of the response activity and/or the job task to be performed.

(iii) Duration of the planned employee activity.

(iv) Site topography.

(v) Site accessibility by air and roads. (vi) Pathways for hazardous

substance dispersion.

(vii) Present status and capabilities of emergency response teams that would provide assistance to on-site employees at the time of an emergency.

(viii) Hazardous substances and health hazards involved or expected at the site and their chemical and physical properties.

(4) Personal protective equipment (PPE) shall be provided and used during initial site entry in accordance with the following requirements:

(i) Based upon the results of the preliminary site evaluation, an ensemble of PPE shall be selected and used during initial site entry which will provide protection to a level of exposure below established permissible exposure limits for known or suspected hazardous substances and health hazards and will provide protection against other known and suspected hazards identified during the preliminary site evaluation.

(ii) An escape self-contained breaching apparatus of at-least five minutes duration shall be carried by employees or kept available at their immediate work station if positivepressure self-contained breaching apparatus is not used as part of the entry ensemble.

(iii) If the preliminary site evaluation does not produce sufficient information to identify the hazards or suspected hazards of the site an ensemble of Level B PPE shall be provided as minimum protection and direct reading instruments shall be cartied for identifying IDLH conditions. (See Appendix B for guidelines on Level B protective equipment.)

(iv) Once the hazards of the site have been positively identified, the appropriate PPE shall be selected and used in accordance with peragraph (g) of this section.

(5) The following monitoring shall be conducted during sits entry when the site evaluation produces information which show the potential for ionizing radiation or IDLH conditions, or when the site information is not sufficient to rule out these possible conditions:

(i) Monitoring for hezerdous levels of ionizing rediction.

(11) Monitoring the air with appropriate test equipment for IDLH and other conditions that may cause death or serious harm (combustible or explosive atmospheres, oxygen deficiency, toxic substances.)

(iii) Visually observe for signs of actual or potential IDLH or other dangerous conditions.

(6) Once the presence and concentrations of specific hazardous substances and health hazards have been established, the risks associated with these substances shall be identified. Employees who will be working on the site shall be informed of any risks that have been identified.

Note.-Risks to consider include, but are not limited to:

Exposures exceeding the appropriate Threshold Limit Values (TLVs), Permissible Exposure Limits (PELs), or Recommended Exposure Limits (RELs).

DLH Concentrations.

Potential Skin Absorption and Irritetion Sources.

Potential Eye Instation Sources.

Explosion Sensitivity and Flammability Ranges.

(7) Any information concerning the chemical, physical, and toxicologic

properties of each substance known or expected to be present on site that is available to the employer and relevant to the duties an employee is expected to perform shall be made available to all employees prior to the commencement of their work activities.

(8) An ongoing sir monitoring program in accordance with paragraph (h) of this section shall be implemented after site characterization has determined the site is safe for the start-up of operations.

(d) Site control. (1) A site control program for preventing contamination of employees shall be developed during the planning stages of a hazardous waste operation clean-up.

(2) The site control program shall, as a minimum, include: A site map; site work zones: the use of a "buddy system"; site communications: the standard operating procedures or safe work practices: and, identification of nearest medical assistance.

(e) Training. (1) All employees (such as equipment operators and general laborers) exposed to hazardous substances, health hazards, or safety hazards shall be thoroughly trained in the following:

(i) Names of personnel and alternates responsible for site safety and health:

(ii) Sefety, health and other hazards present on the site:

(iii) Use of PPE.

(iv) Work practices by which the employee can minimize risks from hazards;

(v) Safe use of engineering controls and equipment on the site:

(vi) Medical surveillance requirements including recognition of symptoms and signs which might indicate over exposure to hazards: and

(vii) Paragraphs (G) through (K) of the site safety and health plan set forth in paragraph (i)(2)(i) of this section.

(2) All employees shall at the time of job assignment receive a <u>minimum of 40</u> hours of initial instruction off the site. and a <u>minimum of three days of actual</u> field experience under the direct supervision of a trained, experienced supervision. Workers who may be exposed to unique or special hazards shall be provided additional training. The level of training provided shall be consistent with the employee's job function and responsibilities.

(3) On-site management and supervisors directly responsible for or who supervise employees engaged in hazardous waste operations shall receive training as provided in paragraph (e)(1) and (e)(2) of this section and at least eight additional hours of specialized training on managing such operations at the time of job assignment. (4) Trainers shall have received a level of training higher than and including the subject matter of the level of instruction that they are providing.

(5) Employees shall not participate in field activities until they have been trained to a level required by their job function and responsibility.

(6) Employees and supervisors that have received and successfully completed the training and field experience specified in paragraphs (e)(1), (e)(2) and (e)(3) of this section shall be certified by their instructor as having completed the necessary training. Any person who has not been so certified or meets the requirements of paragraph (e)(1) of this section shall be prohibited from engaging in hazardous waste operations after March 18, 1987.

(7) Employees who are responsible for responding to hazardous emergency situations that may expose them to hazardous substances shall be trained in how to respond to expected emergencies.

(8) Employees specified in paragraph (e)(1) and managers specified in paragraph (e)(3) of this section shall receive <u>eight hours of refresher training</u> <u>annually</u> on the items specified in paragraph (e)(1) of this section and other relevant topics.

(9) Employers who can show by an employee's work experience and/or training that the employee has had initial training equivalent to that training required in paragraphs (e)(1), (e)(2), and (e)(3) of this section shall be considered as meeting the initial training requirements of those paragraphs. Equivalent training includes the training that existing employees might have already received from actual, on-site experience.

(f) Medical surveillance---{1} Employees covered. A medical surveillance program shall be instituted by the employer for:

(i) all employees who are or may be exposed to hazardous substances or health hazards at or above the established permussible exposure limits for these substances, without regard to the use of respirators, for 30 days or more a year. or

(ii) all employees who wear a respirator for 30 days or more a year, or

(iii) HAZMAT employees specified in paragraph (1)(4) of this section while engaged in hazardous waste operations covered by this section.

(iv) The employer shall make medical examinations or consultations available to all employees who may have been exposed in an emergency situation to hazardous substances at concentrations above the permissible exposure limits. (2) Frequency of medical examinations and consultations. Medical examinations and consultations shall also be made available by the employer to each employee covered under paragraph (f)(1) of this section on the following schedules:

(1) Prior to assignment or for employees covered on the effective date of this standard as specified in paragraph (p) of this section.

(ii) At least once every twelve months for each employee covered.

(iii) At termination of employment or reassignment to an area where the employee would not be covered if the employee has not had an examination within the last six months.

(iv) As soon as possible, upon notification by an employee either that the employee has developed signs or symptoms indicating possible overexposure to hazardous substances or health hazards

(v) At more frequent times, if the examining physician determines that an increased frequency of examination is medically necessary.

(3) Content of medical examinations ond consultations. (i) Medical examinations required by paragraph (f)(2) of this section shall include a medical and work history with special emphasis on symptoms related to the handling of hazardous substances and to fitness for duty including the ability to wear any required PPE under conditions (i.e., temperature extremes) that may be expected at the work site.

(ii) The content of medical examinations or consultations made available to employees pursuant to paragraph (f) shall be determined by the examining physician.

(4) Examination by a physician and costs. All medical examinations and procedures shall be performed by or under the supervision of a licensed physician, and shall be provided without cost to the employee, without loss of pay, and at a reasonable time and place.

(5) Information provided to the physician. The employer shall provide the following information to the examining physician:

(i) A copy of this standard and its appendices.

(ii) A description of the employee's duties as they relate to the employee's exposures.

(iii) The employee's exposure levels or anticipated exposure levels.

(iv) A description of any personal protective equipment used or to be used, and

(v) Information from previous medical examinations of the employee which is not readily available to the examining physician. (6) Physician's written opinion. (i) T employer shall obtain and furnish the employee with a copy of a written opinion from the examining physician containing the following:

(A) The results of the medical examination and tests.

(B) The physician's opinion as to whether the employee has any detected medical conditions which would place the employee at increased risk of material impairment of the employee's health.

(C) The physician's recommended limitations upon the employees assigned work.

(D) A statement that the employee has been informed by the physician of the results of the medical examination and any medical conditions which require further examination or treatment.

(ii) The written opinion obtained by the employer shall not reveal specific findings or diagnoses unrelated to occupational exposure.

(7) Recordkeeping. (i) An accurate record of the medical surveillance required by paragraph (f)(1) of this. section shall be retained. This record shall be retained for the period specified and meet the criteria of 29 CFR 1910.20.

(ii) The record required in paragraph (f)(5)(i) of this section shall include at least the following information:

(A) The name and social security number of the employee:

(B) Physicians' written opinions: (C) Any employee medical complaints

related to exposure to hazardous substances:

(D) A copy of the information which shall be provided to the examining physician by the employer, with the exception of the standard and its appendices.

(iii) The employer shall ensure that this record is retained for the period specified in 29 CFR 1910.20.

(g) Engineering controls, work practices, and personal protective equipment for employee protection—{1} Engineering controls, work practices and PPE. (l) Engineering controls and work practices shall be instituted to reduce and maintain employee exposure to or below the permissible exposure limits of those hazardous substances regulated by 29 CFR Part 1910, Subpert Z, except to the extent that such controls and practices are not feasible.

Note.—Engineering controls which may be feasible are the use of pressurized cabs or control booths on equipment, and/or the use of remotely operated material handling equipment. Work practices which may be feasible are removing all nonessential employees from potential exposure during opening of drums, wetting down dusty aperations and locating employees upwind of pussible hazards.

(ii) Whenever engineering controls and work practices are not feasible. PPE shall be used to protect employees to reduce exposure to below established permissible exposure limits.

(iii) The employer shall not implement a schedule of employee rotation as a means of compliance with permissible exposure limits.

(2) Engineering controls. work practices, and personal protective equipment for substances not regulated in Subport Z An appropriate combination of engineering controls. work practices, and personal protective equipment shall be established to reduce and maintain employee exposure to or below the established permissible exposure limit for hazardous substances not regulated by 29 CFR Part 1910. Subpart Z and health hazards.

(3) Personal protective equipment selection. (i) Personal protective equipment (PPE) shall be selected and used which will protect employees from the hazards and potential hazards they are likely to encounter as identified during the site characterization and analysis.

(ii) Personal protective equipment selection shall be based on an evaluation of the performance characteristics of the PPE relative to the requirements and limitations of the site. the task-specific conditions and duration, and the hazards and potential hazards identified at the site.

(iii) Positive pressure self-contained breathing apparatus. or positive pressure air-line respirators equipped with an escape air supply shall be used in IDUH conditions.

(iv) Totally-encapsulating chemical protective suits (Level A protection) shall be used in conditions where contact of the skin by the hazardous substance may result in an IDLH situation.

(v) The level of protection provided by PPE selection shall be increased when additional information or site conditions show that increased protection is necessary to reduce employee exposure below established permissible exposure limits for hazardous substance and health hazards. (See Appendix B for guidance on selecting PPE ensembles.)

Note .- The level of protection provided may be decreased when additional information or site conditions show that decreased protection will not result in huzardous exposures to employees.

(vi) Personal protective equipment shall be selected and used to meet the requirements of 29 CFR Part 1910.

Subpart L and additional requirements specified in this section.

(4) Totally-encapsulating chemical protective suits. (i) Totallyencapsulating suit materials used for Level A protection shall protect employees from the particular hazards which are identified during site characterization and analysis.

(ii) Totally-encapsulating suits shall be capable of maintaining positive air pressure. (See Appendix A.)

(iii) Totally-encapsulating suits shall be capable of preventing inward test gas leakage of more than 0.5 percent. (See Appendix A.)

(5) Personal protective equipment (PPE) program. A personal protective equipment program shall be established for hazardous waste operations. The PPE program shall address the following elements:

(i) Site hezards.

(ii) PPE selection.

(iii) PPE use.

(iv) Work mission duration.

(v) PPE maintenance and storage.

(vi) PPE decontamination,

(vii) PPE training and proper fitting.

(viii) PPE donning and doffing procedures.

(ix) PPE inspection.

(x) PPE in-use monitoring. (xi) Evaluation of the effectiveness of

the PPE program, and

(xii) Limitations during temperature extremes.

(h) Monitoring. (1) Air monitoring shall be used to identify and quantify airborne levels of hazardous substances in order to determine the appropriate level of employee protection needed on site.

(2) As a first step, air monitoring shall be conducted to identify any IDLH and other dangerous situations, such as the presence of flammable atmospheres. oxygen-deficient environments, toxic levels of airborne contaminants, and radioactive materials.

(3) As a minimum, periodic monitoring shall be conducted when:

(i) Work begins on a different portion of the site.

(ii) Contaminants other than those previously identified are being handled.

(iii) A different type of operation is initiated (e.g., drum opening as opposed to exploratory weil drilling.)

(iv) Employees are handling leaking drums or containers or working in areas with obvious liquid contamination (e.g., a spill or lagoon.)

(4) High-risk employees, e.g., those closest to the source of contaminant generation. shall receive personal monitoring sufficient to characterize employee exposure.

(i) Informational programs-(1) General As part of the safety and health program required in paragraph (b)(1) of this section, the employer shall develop and implement a site safety and health plan meeting the requirements of paragraph (i)(2) of this section for each hazardous waste operation.

(2) Site safety and health plan. The site safety and health plan, which shall be available on the site for inspection by employees, their designated representatives, and OSHA personnel. shall address the safety and health hazards of each phase of site operation and include the requirements and procedures for employee protection.

(i) The site safety and health plan, as a minimum, shall address the following:

(A) Names of key personnel and alternates responsible for site safety and health and appointment of a site safety and health officer.

(B) A safety and health risk analysis for each site task and operation.

(C) Employee training assignments. (D) Personal protective equipment to be used by employees for each of the site tasks and operations being conducted.

(E) Medical surveillance requirements.

(F) Frequency and types of air monitoring, personnel monitoring, and environmental sampling techniques and instrumentation to be used. Methods of maintenance and calibration of monitoring and sampling equipment to be used.

(G) Site control measures.

(H) Decontamination procedures.

(I) Site's standard operating

procedures.

(]) A contingency plan meeting the requirements of paragraphs (1)(1) and (1)(2) of this section for safe and effective responses to emargencies including the necessary PPE and other equipment

(K) Confined space entry procedures.

(ii) Pre-entry briefings shall be held prior to initiating any site activity and at such other times as necessary to ensure that employees are apprised of the site safety and health plan and that it is being followed.

(iii) Inspections shall be conducted by the site safety and health officer or. in the absence of that individual, another individual acting on behalf of the employer as necessary to determine the effectiveness of the site safety and health plan. Any deficiencies in the effectiveness of the site safety and health plan shall be corrected by the employer.

(j) Handling drums and containers (1) General. (i) Drums and containers used during the clean-up shall meet the appropriate DOT. OSHA, and EPA regulations for the wastes that they contain.

(ii) Drums and containers shall be inspected and their integrity shall be assured prior to being moved. Drums or containers that cannot be inspected before being moved because of inaccessible storage conditions shall be moved to an accessible location and inspected prior to further handling.

(iii) Unlabeled drums and containers shall be considered to contain hazardous substances and handled accordingly until the contents are positively identified and labeled.

(:v) Site operations shall be organized to minimize the amount of drum or container movement.

(v) Prior to movement of drums or containers, all employees exposed to the transfer operation shall be warned of the potential hazards associated with the contents of the drums or containers.

(vi) U.S. Department of Transportation specified salvage drums or containers and suitable quantities of proper absorbent shall be kept available and used in areas where spills, leaks, or motures may occur.

(vii) Where major spills may occur, a spill containment program shall be implemented to contain and isolate the entire volume of the hazardous substance being transferred.

(viii) Drums and containers that cannot be moved without rupture. leakage. or spillage shall be emptied into a sound container using a device classified for the material being transferred.

(ix) A ground-penetrating system or other type of detection system or device shall be used to estimate the location and depth of drums or containers.

(x) Soil or covering material shall be removed with caution to prevent drum or container rupture.

(xi) Fire extinguishing equipment meeting the requirements of 29 CFR Part 1910. Subpart L shall be on hand and ready for use to control small fires.

(2) <u>Opening drums and containers</u>. The following procedures shall be followed in areas where drums or containers are being opened:

(i) Where an airline respirator system is used, connections to the bank of air cylinders shall be protected from contamination and the entire system shall be protected from physical damage.

(Ii) Employees not actually involved in opening drums or containers shall be kept a safe distance from the drums or containers being opened.

(iii) If employees must work near or adjacent to drums or containers being opened. a suitable shield that does not interfere with the work operation shall be placed between the employee and the drums or containers being opened to protect the employee in case of accidental explosion.

(iv) Controis for drum or container opening equipment, monitoring equipment, and fire suppression equipment shall be located behind the explosion-resistant barrier.

(v) Material handling equipment and hand tools shall be of the type to prevent sources of ignition.

(vi) Drums and containers shall be opened in such a manner that excess interior pressure will be safely relieved. If pressure cannot be relieved from a remote location, appropriate shielding shall be placed between the employee and the drums or containers to reduce the risk of employee injury.

(vii) Employees shall not stand upon or work from drums or containers.

(3) Electrical material handling equipment. Electrical material handling equipment used to transfer drums and containers shall:

(i) Be positioned and operated to minimize sources of ignition related to the equipment from igniting vapors released from ruptured drums or containers, or

(ii) Meet the requirements of 29 CFR 1910.307 and be of the appropriate electrical classification for the materials being handled.

(4) Radioactive wastes. Drums and containers containing radioactive wastes shall not be handled until such time as their hazard to employees is properly assessed.

(5) Shock sensitive wastes.

Caution: Shipping of shock sensitive wastes may be prohibited under U.S. Department of Transportation regulations. Employers and their shippers should refer to 49 CFR 173.21 and 173.50.

As a minimum, the following special precautions shall be taken when drums and containers containing or suspected of containing shock-sensitive wastes are handled:

(i) All non-essential employees shall be evacuated from the area of transfer.

(ii) Material handling equipment shall be provided with explosive containment devices or protective shields to protect equipment operators from exploding containers.

(iii) An employee alarm system capable of being perceived above surrounding light and noise conditions shall be used to signal the commencement and completion of explosive waste handling activities.

(iv) Continuous communications (Le., portable radios, hand signals, telephones, as appropriate) shall be maintained between the employee charge of the immediate handling and the site safety officer or commapost until such time as the handling operation is completed. Communication equipment or methods that could cause snock sensitive materials to explode snail not be used.

(v) Drums and containers under pressure. as evidenced by bulging or swelling, shall not be moved until such time as the cause for excess pressure is determined and appropriate containment procedures have been implemented to protect employees from explosive relief of the drum.

{vi} Drums and containers cuntaining packaged laboratory wastes shall be considered to contain shock-sensitive or explosive materials until they have been characterized.

(8) <u>Laboratory waste packs</u>. In addition to the requirements of paragraph (j)(5) of this section, the following precautions shall be taken, as a minimum, in handling laboratory waste packs (lab packs):

(i) Lab packs shall be opened only when necessary and then only by an individual knowledgeable in the inspection, classification, and segregation of the containers within the pack according the hazards of the wastes.

(il) If crystalline material is noted any container, the contents shall be handled as a shock-sensitive wasta be the contents are identified.

(7) <u>Sampling drums and containers</u>. Sampling of containers and drums shall be done in accordance with a sampling procedure which is part of the site safety and health plan developed for and available to employees and others at the specific worksite.

(8) Shipping and transport. (1) Drums and containers shall be identified and classified prior to packaging for shipment.

(ii) Drum or container staging areas shall be kept to the minimum number necessary to safely identify and classify materials and prepare them for transport.

(iii) Staging areas shall be provided with adequate access and egress routes.

(iv) Bulking of hazardous wastes shall be permitted only after a thorough characterization of the materials has been completed.

(9) Tank and vault procedures. (i) Tanks and vaults containing hazardous substances shall be handled in a manner similar to that for drums and containers, taking into consideration the size of the tank or vault.

(ii) Appropriate tank or vault entry procedures meeting paragraph

(i)(2)(i)(K) of this section shall be followed whenever employees must enter a tank or vault

(k) Decontamination. (1) A decontamination procedure shall be developed, communicated to employees and unplemented before any employees or equipment may enter areas on site where potential for exposure to hazardous substances exists.

(2) Standard operating procedures shall be developed to minimize employee contact with hazardous substances or with equipment that has contacted bazardous substances.

(3) Decontamination shall be performed in areas that will minimize the exposure of uncontaminated employees or equipment to contaminated employees or equipment.

(4) All employees leaving a contaminated area shall be appropriately decontaminated; all clothing and equipment leaving a contaminated area shall be appropriately disposed of or decontaminated.

(5) Decontamination procedures shall be monitored by the site safety and health officer to determine their effectiveness. When such procedures are found to be ineffective, appropriate steps shall be taken to correct any deficiencies.

(6) All equipment and solvents used for decontamination shall be decontaminated or disposed of properly.

(7) Protective clothing and equipment shall be decontaminated, cleaned, laundered, maintained or replaced as needed to maintain their effectiveness.

(8) Impermeable protective clothing which contacts or is likely to have contacted hazardous substances shall be decontaminated before being removed by the employee.

(9) Employees whose nonimpermeable clothing becomes wetted with hazardous substances shall immediately remove that clothing and proceed to shower. The clothing shall be disposed of or decontaminated before it is removed from the work zone.

(10) Unauthorized employees shell not remove protective clothing or equipment from change rooms.

(11) Commercial laundries or cleaning establishments that decontaminate protective clothing or equipment shall be informed of the potentially harmful effects of exposures to hazardous substances.

(12) Where the decontamination procedure indicates a need for showers and change rooms, they shall be provided and meet the requirements of 29 CFR 1910.141.

(1) Emergency response-(1) General. (i) An emergency response plan shall be developed and implemented to handle anticipated on-site emergencies prior to the commencement of hazardous waste operations. Emergency response activities to all other hazardous waste operations shall follow an emergency response plan meeting the requirements of this section.

(ii) Elements of an emergency response plan. The employer shall develop an emergency response plan for on-site and off-site emergencies which shall address, as a minimum, the following:

(A) Pre-emergency planning.

(B) Personnel roles. lines of authority. training, and communication.

(C) Emergency recognition and prevention.

(D) Safe distances and places of refuge.

(E) Site security and control.

(F) Evacuation routes and

procedures.

(G) Decontamination.

(H) Emergency medical treatment and first aid.

(I) Emergency alerting and response procedures.

(J) Critique of response and followup.

(K) PPE and emergency equipment. (2) On-site emergency response—(i)

Training for site emergency response shall be conducted in accordance with paragraph (e) of this section.

(ii) Procedures for handling site emergency incidents. (A) In addition to the elements for the emergency response plan required in paragraph (l)(1)(ii) above, the following elements shall be included for site emergency response plans:

(1) Site topography, layout, and prevailing weather conditions.

(2) Procedures for reporting incidents to local, state, and federal governmental agencies.

(B) The site emergency response plan shall be a separate section of the Site Safety and Health Plan.

(C) The site emergency response plan shall be compatible and integrated with the disaster, fire and/or emergency response plans of local, state, and federal agencies.

(D) The site emergency response plan shall be rehearsed regularly as part of the overall training program for site operations.

(E) The site emergency response plan shall be reviewed periodically and, as necessary, be amended to keep it current with new or changing site conditions or information.

(F) An employee alarm system shall be installed in accordance with 29 CFR 1910.165 to notify employees of an onsite emergency situation, to stop work activities if necessary, to lower background noise in order to speed ______ communication, and to begin emergency procedures.

(C) Based upon the information available at time of the emergency, the employer shall evaluate the incident and the site response capabilities and proceed with the appropriate steps to implement the on-site emergency response plan.

(3) Off-site emergency response—(i) Training. Training for handling emergency responses involving hazardous substances shall be conducted on a monthly basis and shall be at least 24 hours annually. The training snall include as a minimum recognition of hazards, selection, care, and use of personal protective equipment and safe operating procedures to be used at the incident scene.

(ii) Procedures for handling off-site emergency incidents. (A) The senior officer responding to an incident involving a hazardous substance or waste shail establish an Incident Command System (ICS). All emergency responders and their communications shall be coordinated and controlled through the individual in charge of the ICS.

(B) The individual in charge of the ICS shall identify. to the extent possible. all hazardous substances or conditions present.

(C) Based on the hazardous substances and/or conditions present. the individual in charge of the ICS shall implement appropriate emergency operations, and assure that the personal protective equipment worn is appropriate for the hazards to be encountered. However, personal protective equipment shall meet, at a minimum, the criteria contained in 29 CFR 1910.156(e) when worn while performing fire fighting operations beyond the incipient stage.

(D) Self-contained breathing apparatus shall be worn at all times during emergency operations involving exposure to hazardous substances or health hazards. After October 18, 1988 only positive pressure self-contained respirators shall be used.

(E) The individual in charge of the ICS shall limit the number of emergency response personnel at the emergency site to those who are actively performing emergency operations. However, operations in hazardous areas shall be performed using the buddy system in groups of two or more.

(F) Back-up personnel shall be standing by with equipment ready to provide assistance or rescue. Qualified basic life support personnel, as a minimum, shall also be standing by with medical equipment and transportation capability.

(G) The individual in charge of the ICS shall designate a safety officer, who is knowledgeable in fire fighting or rescue operations and hazardous substance handling procedures, with specific responsibility to identify and evaluate hazards and to provide direction with respect to the safety of operations for the emergency at hand.

(H) When activities are judged by the sufety officer to be unsafe and/or to involve an imminent danger condition, the safety officer shall have the authority to alter, suspend, or terminate those activities. The safety officer shall immediately inform the individual in charge of the ICS of any actions taken to correct these hazards at an emergency scene.

(I) After emergency operations have terminated, the individual in charge of the ICS shall implement appropriate decontamination procedures.

(4) Hazardous materials teams (HAZMAT). (i) Employees who are members of the HAZMAT team. employees designated by the employer to plug, patch or otherwise temporarily control or stop leaks from containers which hold hazardous substances or health hazards shall be given training in accordance with paragraph (1)(3) of this section that includes the care and use of chemical protective clothing and procedures to be followed when working on leaking drums, containers, tanks, or bulk transport vehicles.

(ii) Members of HAZMAT teams shall receive an annual physical examination by a licensed physician and be provided medical surveillance as required in paragraph (f) of this section.

(iii) Personal protective clothing and equipment to be used by HAZMAT team members shall meet the requirements of paragraph (g) of this section.

(iv) Approved self-contained compressed air breathing apparatus may be used with approved cylinders from other approved self-contained compressed air breathing apparatus provided that such cylinders are of the same capacity and pressure rating. All compressed air cylinders used with selfcontained breathing apparatus shall meet U.S. Department of Transportation and National Institute for Occupational Safety and Health criteria.

(5) Post-emergency response operations. Upon completion of the emergency response, if it is determined that it is necessary to remove hazardous substances, health hazards and materials contaminated with them such as contaminated soil or other elements of the natural environment, then such operations shall meet all the requirements of paragraphs (b) through (n) of this section.

(m) *lilum:nation*. Work areas shall be lighted to not less than the minimum illumination intensities listed in Table H-102.1 while any work is in progress:

TABLE H-102.1. -- MINIMUM ILLUMINATION INTENSITIES IN FOOT-CANOLES

Fool-candres	Area of costalions
-	
3	General the areas
3	Encaverage and weaks areas, accessivelys
	active storage event, loading platforms, re
	future, and hald maintenance areas.
S	WARTE HEREFOLDER, COTTONE, NEWHOLD, AND
•	Turners, shalls, and annexes underground
	and some flowers and a li
1	
	second onud Gazari wronai' sa noo
	and the second cars and the second cars and a
	shaf be acceptable for use in the turne
•	Personal Contraction of Contractiono
IG _	General shoos (e.g., mechanical and electric
	CAL ADVANTANT FOOTS, ACTIVA SECTIONS
	Corrects of both dustions, locket of disea
	at colds, data and, and other literal
•	
'	
×	HAR BE STREAM CHANGE BUT DUCOF

(n) Sanitation at temporary workplaces—(1) Potable water. (1) An adequate supply of potable water shall be provided on the site.

(ii) Portable containers used to dispense drinking water shall be capable of being tightly closed, and equipped with a tap. Water shall not be dipped from containers.

(iii) Any container used to distribute drinking water shall be clearly marked as to the nature of its contents and not used for any other purpose.

(iv) Where single service cups (to be used but once) are supplied, both a sanitary container for the unused cups and a receptacle for disposing of the used cups shall be provided.

(2) Nonpotable water. (i) Outlets for nonpotable water, such as water for industrial or firefighting purposes shall be identified to indicate clearly that the water is unsafe and is not to be used for drinking, washing, or cooking purposes.

(ii) There shall be no crossconnection, open or potential, between a system furnishing potable water and a system furnishing nonpotable water.

(3) *Toilets facilities.* (1) Toilets shall be provided for employees according to Table H-102.2.

TABLE H-102.2 --- TOHLET FACILITIES

Number of emergyane	Meanure number of leading
29 or lower	Che. Che tolet sett and 1 unnel per 40 employees.

TABLE	H-102.2 -	TOILET	FACILITIES
	Con	hrued	

	:	Mountum number of facilitie	•
More men 200	i	One lower seel and I umail 50 employees.	•

(ii) Under temporary field conditions. provisions shall be made to assure not less than one toilet facility is available.

(iii) Hazardous waste sites, not provided with a sanitary sewer, shall be provided with the following toilst facilities unless prohibited by local codes:

(A) Privies:

(B) Chemical toilets:

(C) Recirculating toilets: or

(D) Combustion toilets.

(iv) The requirements of this paragraph for sanitation facilities shall not apply to mobile crews having transportation readily available to nearby toilet facilities.

(4) Food handling. All employees' food service facilities and operations shall meet the applicable laws. ordinances. and regulations of the jurisdictions in which they are located.

(5) Temporary sleeping quarters. When temporary sleeping quarters are provided, they shall be heated. ventilated, and lighted.

(6) Washing facilities. The employ, shall provide adequate washing facilities for employees engaged in operations where hazardous substances may be harmful to employees. Such facilities shall be in near proximity to the worksite, within controlled access work zones and shall be so equipped as to enable employees to remove hazardous substances.

(o) Certain Operations Conducted under the Resource Conservation and Recovery Act of 1973 (RCRA). Employers conducting operations specified in paragraph (g)(2)(iii) of this section shall:

(1) Implement a hazard communication program meeting the requirements of 29 CFR 1910.1200;

(2) Implement a medical surveillance program meeting the requirements of paragraph (f) of this section;

(3) Develop and implement a safety and health program for employeee involved in hazardous waste operations. The program shall be designed to identify, evaluate and control safaty and health hazards and provide for emergency response to their facilities for the purpose of employee protection:

(4) Develop and implement a decontamination procedure in accordance with paragraph (k) of this section, and

(5) Develop and implement a training program for employees involved with hazardous waste operations to enable each employee to perform their assigned duties and functions in a safe and healthful manner so as not to endanger themselves or other employees. The initial training shall be for 24 hours and refresher training shall be for eight hours annually.

(p) Star: up dates—(1) Training and medical provisions. Initial training and medical surveillance as specified by paragraph (e) and (f) of this section shall be commenced on the effective date of this standard, and be fully implemented as soon as possible but no later than March 18, 1987. Employees may continue in their work assignments until March 16, 1987 though training and medical examinations have not been completed so long as all feasible training and examinations have been completed.

(2) Sufety and health program. The emoloyer shall develop and implement a safety and health program as required by paragraph (b)(1) of this section as suon as is feasible and have it completed and implemented no later than March 18, 1987.

(3) Engineering controls. work proctices. and personal protective equipment. (i) The engineering controls. work practices and personal protective equipment required by paragraph (g)(2) of this section shall be implemented as soon as feasible and implementation shall be completed no later than March 16, 1987.

(ii) The engineering controls, work practices and personal protective equipment required by paragraph (g)(1) of this section are existing requirements of other OSHA standards and continues to be required from the effective date of this standard.

(4) Site safety and health plan. The site safety and health plan required by paragraph (i)(2) of this section shall be completed as soon as feasible but no later than February 16, 1987.

(5) Certain operations conducted under RCRA. The requirements specified by paragraph (0) of this section shall be instituted by March 16, 1967.

(6) Other requirements. Requirements of this standard which do not have a separate start-up date and have not been required by other OSHA standards shall be carried out from the effective date of this standard.

(7) New operations. Operations covered by this section which are started after March 16. 1987, shall be in compliance with this section from the start of their operation.

Appendices to § 1910.120—Hazardous Weste Operations and Emergency Response

Note.—The following appendices serve as non-manaatory guidelines to assist employees and employees in complying with the appropriate requirements of this section.

Appendix A—Personal Protective Equipment Test Methods

This appendix sets forth the nonmandatory examples of tests which may be used to evaluate compliance with paragraphs 1910.120(g)(4) (ii) and (iii). Other tests and other challenge agents may be used to evaluate compliance.

A. Fully-Encapsulated Suit Pressure Test 1 0—Scope

1.1 This practice measures the ability of a gas tight totally-encapsulating chemical protective suit material, seems, and closures to maintain a fixed positive pressure. The results of this practice allow the gas tight integrity of a total-encapsulating chemical protective suit to be evaluated.

1.2 Resistance of the suit materials to permeation, penetration, and degradation by specific hazardous substances is not determined by this test method.

20-Description of Terms

2.1 Totally-encepsulated chemical protective suit (TECP suit)—A full body garment which is constructed of protective clothing materials: covers the wearer's torso, head, arms, and less may cover the wearer's hands and (set with tightly attached gloves and boots: completely encloses the wearer by itself or in combination with the wearer's respiratory equipment, gloves, and boots.

2.2 Protective clothing material—Any material or combination of materials used in an item of clothing for the purpose of isolating parts of the body from direct contact with a potentially hazardous liquid or gaseous chemicals.

2.3 "Gas tight"—for the purpose of this practice the limited flow of a gas under pressure from the inside of a TECP suit to atmosphere at a prescribed pressure and time interval.

2.4 "Shall"-This term indicates a mandatory requirement.

2.5 "Should"--This term indicates a recommendation or that which is advised but not required.

2.6 "May"—This term is used to state a permissive use or an alternative method to a specific requirement.

3.0---Summary of Practice

3.1 The TECP suit is visually inspected and modified for the test. The test apparatus is attached to the suit to permit inflation to the pre-test suit expansion pressure for removal of suit wrinkles and creases. The pressure is lowered to the test pressure and monitored for three minutes. If the pressure drop is excessive, the TECP suit fails the tests and is removed from service. After leak location and repair the test is repeated.

4.0—Required Supplies

4.1 Source of compressed air.

4.2 Test apparatus for suit testing including a pressure measurement device with a sensitivity of at least "s inch water gauge

4.3 Vent valve closure plugs or sealing tape.

4.4 Scapy water solution and soft brush. 4.5 Stop watch or appropriate timing device.

5.0-Salety Precautions

5.1 Care shall be taken to provide the correct pressure safety devices required for the source of compressed air used.

6.0-Test Procedure

6.1 Prior to each test, the tester shall perform a visual inspection of the suit. Check the suit for seam integrity by visually examining the seams and qently pulling on the seams. Ensure that all air supply lines. fittings, visor, zippers, and valves are secure and show no signs of deterioration.

6.1.1 Seal off the vent valves along with any other normal inlet or exhaust points (such as umbilical air line fittings or face piece opening) with tape or other appropriate means (caps, plugs, fixture, etc.). Care should be exercised in the sealing process not to damage any of the suit components.

6.1.2 Close all closure assemblies

6.1.3 Prepare the suit for inflation by providing an improvised connection point or the suit for connecting an airline. Attach the pressure test apparatus to the suit to permit suit inflation from a compressed air source equipped with a pressure indicating regulator. The leak tightness of the pressure test apparatus should be tested before and after each test by closing off the end of the tubing attached to the suit and assuring a pressure of three inches water gauge for three minutes can be maintained. If a component is removed for the test, that component shall be replaced and a second test conducted with another component removed to permit a complete tests of the ensemble.

8.1.4 The pre-test expansion pressure (A) and the suit test pressure (B) shall be supplied by the suit manufacturer but in no case shall they be less than: A=3 inches water gauge and B=2 inches water gauge. The ending suit pressure (C) shall be no less than 80% (%) of the test pressure (B); Le., the pressure drop shall not exceed 20% (%) of the test pressure (B).

6.1.5 Inflate the suit until the pressure inside is equal to pressure "A", the pre-test expansion suit pressure. Allow at least one minute to fill out the winkles in the suit. Release sufficient air to reduce the suit pressure to pressure "B", the suit test pressure to pressure "B", the suit test pressure. Begin timing. At the end of three minutes, record the suit pressure as pressure "C." the ending suit pressure. The difference between the suit test pressure and the ending suit test pressure (B-C) shall be defined as the suit pressure drop.

6.1.6 If the suit pressure drop is more than 20 percent (%) of the suit test pressure B during the three minute test period, the suit fails the test and shall be removed from service.

7.0-Retest Procedure

7.1 If the suit fails the test check for leaks by inflating the suit to pressure A and brushing or wiping the entire suit (including seams, closures, lens gaskets, glove-to-sleeve joints, etc.) with a mild soap and water solution. Observe the suit for the formation of soap bubbles, which is an indication of a leak, Recair all identified leaks.

7.2 Retest the TECP suit as outlined in Test procedure 6.0.

8.0-Report

8.1 Each TECP suit tested by this practice shall have the following information recorded.

8.1.1 Unique identification number identifying brand name, date of purchase, material of construction, and unique fit features: e.g., special breathing apparetus.

8.1.2 The actual values for test pressures, A. B. and C shall be recorded along with the specific observation times. If the ending pressure (C) is less than 80% of the test pressure (B) the suit shall be identified as failing the test. When possible, the specific leak location shall be identified in the test records. Retest pressure data shall be recorded as an additional test,

8.1.3 The source of the test apperatus used shall be identified and the sensitivity of the pressure gauge shall be recorded.

8.1.4 Records shall be kept for each pressure test even if repairs are being made at the test location.

Caution

Visually inspect all parts of the suit to be sure they are positioned correctly and secured lightly before putting the suit back into service. Special care should be taken to examine each exhaust valve to make sure it is not blocked.

Care should also be exercised to assure that the inside end outside of the suit is completely dry before it is put into storege.

B. Fully-Encapsulated Suit Qualitative Leak Test

1.0-Scope

1.1 This practice semi-qualitatively tests gas tight totally-encapsulating chemical protective suit integrity by detecting inward leakage of ammonia vapor. Since no modifications are made to the suit to carry out this test, the results from this practice provida a realistic test for the integrity of the entire suit.

1.2 Resistance of the suit materials to permeetion, penetration, and degradation is not determined by this test method.

2.0-Description of Terms

2.1 Totelly-encapsulated chemical protective suit (TECP suit)—A full body garmant which is constructed of protective clothing materials: covers the wearer's tarso, head, arms, and legs; may cover the wearer's hands and fast with tightly attached gloves and boots: completely encloses the wearer by itself or in combination with the wearer's respiratory equipment, gloves, and boots.

2.2 Protective clothing maternal—Any material or combination of materials used in an liem of clothing for the purpose of isolating parts of the body from direct contact with a potentially hazardous liquid or gaseous chemicals.

2.3 "Gas tight"—for the purpose of this practice the limited flow of e gas under pressure from the unside of a TECP suit to atmosphere at a prescribed pressure and time interval.

Li Shail"-This term indicates a mandatory requirement.

2.5 Should —This term indicates a recommendation or that which is advised but not required.

2.6 "May "—This term is used to state a permissive use or an alternative method to a specific requirement.

2.7 Intrusion Coefficient—A number expressing the level of protection provided by a gas tight totally-encapsulating chemical protective suit. The intrusion coefficient is calculated by dividing the test room chailenge agent concentration by the concentration of challenge agent found inside the suit. The accuracy of the intrusion coefficient is dependent on the challenge agent monitoring methods. The larger the intrusion coefficient the greater the protection provided by the TECP suit.

3.0-Summary of Recommended Practice

3.1 The volume of ammonia solution required to generate the test atmosphere is determined using the directions outlined in 6.1. The suit is donned by a person wearing the appropriate respiratory equipment (normally a self-contained breathing apparatus) and worn inside the enclosed test room. The ammonie solution is taken by the suited individual into the test room and poured into an open plastic per. A twominute evenoration period is observed before the test room concentration is measured using a high range ammonia length of stain detector tube. When the ammonia reaches a concentration of between 1000 and 1200 ppm. the suited individual starts a standardized exercise protocol to stress and flex the suit. After this protocol is completed the test room concentration is measured again. The suited individual exits the test room and his standby person measures the ammonia concentration inside the suit using a low range ammonia length of stain datector tube or other more sensitive ammonia detector. A stand-by person is required to observe the test individual during the test procedure, aid the person in donning and doffing the TECP suit and monitor the suit interior. The intrusion coefficient of the suit can be calculated by dividing the average test area concentration by the interior suit concentration. A colorimetric indicator strip of bromophenoi blue is placed on the inside of the suit face piece lens so that the suited individual is sole to detect a color change and know if the suit has a significant leak. If a color change is observed the individual should leave the test room immediately.

4.0-Required Supplies

4.1 A supply of concentrated ammonia (58 percent ammonium hydroxide by weight).

4.2 A supply of bromophanol/blue indicating paper, sensitive to 5-10 ppm ammonia or greater over a two-minute period of exposure.

4.3 A supply of high range (0.5-10 volume percent) and low range (5-700 ppm) detector tubes for anmonia and the corresponding sampling pump. More sensitive ammonia detectors can be subsultuted for the low range detector tubes to improve the sensitivity of this proctice. 44 A plastic pan (PVC) at least 12" 1. . and a half pint plastic container (PVC) with hightly closing lid.

4.5 Volumetric measuring device of at least 50 milliliters in volume with an accuracy of at least ± 1 milliliters.

50-Salety Precautions

5.1 Concentrated admonta is a corrosive volattle liquid requiring eye, skin, and respiratory protection.

5.2 Since the threshold limit value for ammonia is 25 ppm, only persons wearing the appropriate respirator protection shall be in the chamber. Normally only the person wearing the total-encapsulating suit will be inside the chamber. A stand-by person shall have a self-contained breathing apparatus, or equivalent breathing apparatus, available to enter the test area should the suited individual need assistance.

5.3 A method to monitor the suited individual must be used during this test. Visual contact is the simplest but other methods using communication devices are acceptable.

5.4 The test room shall be large enough to allow the exercise protocol to be carried out and ventilated to allow for easy exhaust of the ammonia test atmosphere after the test(s) are completed.

5.5 Individuals shall be medically screened for the use of respiratory protection and checked for silergies to ammonia before participating in this test procedure.

6.0-Test Procedure

6.1.1 Measure the test area to the nea. foot and calculate its volume in cubic feet. Multiply the test area volume by 0.2 multiliters of ammonia per cubic foot of test area volume to determine the approximate volume of ammonia required to generate 1000 ppm in the test area.

8.1.2 Measure this volume from the supply of concentrated ammonia and place it into a closed plastic container.

6.1.3 Place the jar, several high range emmonus detector tubes and the pump in the clean test pan and locate it near the test area entry door so that the suited individual has easy access to these supplies.

6.2.1 In a non-contaminated atmosphere, open a presented ammonia indicator strip and fasten one end of the strip to the inside of suit face shield lens where it can be seen by the wearer. Care shall be taken not to contaminate the detector part of the indicator paper by touching it. A small piece of masking tape or equivalent should be used to attach the indicator strip to the interior of the suit face shield.

6.2.2 If problems are encountered with this method of attachment the indicator strip can be attached to the outside of the respirator face piece being used during the test, assuming the face piece is worn within the TECP suit.

6.3 Don the respiratory protective device normally used with the suit and then don the TECP suit to be tested. Check to be sure all openings which are intended to be sealed (mopers, gloves, etc.) are completely seale DO NOT, however, plug off any venting valves. 6.4 Step into the enclosed test room such as a closet, bathroom, or test booth, equipped with an exnaust fan. No air should be exhausted from the chamber during the test because this will dilute the ammonia challenge concentrations.

6.5 Open the container with the premeasured volume of ammonia within the enclosed test room, and pour the liquid into the empty plastic test par. Wait two minutes to allow for adequate volatilization of the ammonia. A small mixing fan can be used near the evaporation pan to increase the evaporation rate of ammonia.

6.6 After two minutes a determination of the ammonia concentration within the chamber should be made using the high range colonmetric detector tube. A concentration of 1000 ppm ammonia or greater shall be generated before the exercises are started.

6.7 To test the integrity of the suit the following four minute exercise protocol should be followed:

6.7 1 Raising the arms above the head with at least 15 raising motions completed in one minute.

6.7.2 Walking in place for one minute with at least 15 raising motions of each leg in a one-minute period.

6.7.3 Touching the toes with a least 10 complete motions of the arms from above the head to touching of the toes in a one-minute period.

6.7.4 Deep knee bends with at least 10 complete standing and squatting motions in a one-minute period.

6.8 At any time during the test should the colorimetric indicating paper change colors the test should be stopped and section 6.10 and 6.12 initiated.

6.9 After completion of the test exercise. the test area concentration should be measured again using the high range colorimetric detector tube.

6.10 Exit the test area.

6.11 The opening created by the suit zipper or other appropriate suit penetration should be used to determine the ammonia concentration in the suit with the low range length of stain detector tube or other ammonia monitor. The internal TECP suit air snould be sampled far enough from the enclosed test area to prevent a false ammonia reading.

6.12 After completion of the measurement of the suit interior ammonia concentration the test is concluded and the suit is dolled and the respirator removed.

6.13 The ventilating fan for the test room should be turned on and allowed to run for enougn time to remove the ammonia gas.

6.14 Any detectable ammonus in the suit interior (5 ppm NHs or more for the length of stain datector tube) indicates the suit fails the test. When other ammonia detectors are used, a lower level of detection is possible and it should be specified as the pass fail criteria.

6.15 By following this practice an intrusion coefficient of approximately 300 or more can be measured with the suit in a completely operational condition.

7 0-Retest Procedures

7.1 If the suit fails this test check for lesks by following the pressure test in test A above. 7.2 Retest the TECP suit as outlined in the test procedure 6.0.

A O-Report

6.1 Each gas tight totally-encapsulating chemical protective suit tested by this practice shall have the following information recorded.

8.1.1 Unique identification number identifying brand name, date of purchase, material of construction, and unique suit features: e.g., special breathing apparatus, 8.1.2 Ceneral description of test room

used for test. 8.1.3 Brend name and purchase date of

ammonia delector strips.

6.1.4 Brand name, sampling range, and expiration date of the length of stain ammonia detector tubes. The brand name and model of the sampling pump should also be recorded. If another type of ammonia detector is used, it should be identified along with its minimum detection limit for ammonia.

6.1.5 Actual test results shall list the two test area concentrations, their average, the interior suit concentration, and the calculated intrusion coefficient. Retest data shall be recorded as an additional test.

8.2 The evaluation of the data shall be specified as 'suit passed" or "suit failed" and the date of the test. Any detectable ammonia (3 ppm or greater for the length of stain detector tube) in the suit interior indicates the suit fails this test. When other ammonia detectors are used, a lower level of detection is possible and it should be specified as the pass fail criteria.

Caution

Visually inspect all parts of the suit to be sure they are positioned correctly and secured tightly before putting the suit back into service. Special care should be taken to examine each exhaust valve to make sure it is not blocked.

Care should also be exercised to assure that the inside and outside of the out is completely dry before it is put into storage.

Appendix B—General Description and Discussion of the Lovels of Protection and Protective Gener

This appendix sets (orth information about personal protective equipment (PPE) protection levels which may be used to essust employers in complying with the PPE requirements of this section.

As required by the standard, PPE must be selected which will protect employees from the specific hazards which they are likely to encounter during their work on-site.

Selection of the appropriate PPE is a complex process which must take into consideration a variety of factors. Key factors involved in this process are identification of the hazards, or suspected hazards, their routes of potential bazard to employees (inhelation, skin absorption, ingestion, and eve or skin contact), and the performance of the PPE materials (and seems) in providing a barrier to these hazards. The amount of protection provided by PPE is materialhazard specific. That is, protective equipment materials will protect well against some hazardous substances and poorly, or not at atl, against others. In many instances. protective equipment materials cannot be found which will provide continuous protection from the particular hazardous substance. In these cases the breakthrough time of the protective material should exceed the work durations, or the exposure after breakthrough must not pose a bazardous level.

Other factors in this selection process to be considered are matching the PPE to the employee s work requirements and taskspecific conditions. The durability of PPE meterials, such as teer strength and seam strength, in relation to the employee s tasks must be considered. The effects of PPE in relation to heat stress and task duration are a factor in selecting and using PPE. In some cases layers of PPE may be necessary to provide sufficient protection, or to protect expensive PPE inner garments, suits or equipment.

The more that is known about the hazards at the site, the easier the job of PPE selection becomes. As more information about the hazards and conditions at the site becomes aveilable, the site supervisor can make decisions to up-grade or down-grade the level of PPE protection to match the tasks at hand.

The following are guidelines which an employer can use to begin the selection of the appropriate PPE. As noted above, the site information may suggest the use of compliant of PPE selected from the different protection levels (i.e., A. B. C. or D). as being more suitable to the hezards of the work. It should be cautioned that the listing below does not fully address the performance of the specific PPE material in relation to the specific hezards at the job site, and that PPE selection, evaluation and re-selection is an ongoing process until sufficient information about the hezards and PPE performance is obtained.

Part A. Personal protective equipment has been divided into four categories based on the degree of protection afforded and are as foilows (See Part B of this appendix for further explanation of Levels A. B. C. and D hazards):

L Lovel A-To be selected when the greatest level of skin, respiratory, and eye protection is required.

Level A equipment: used as appropriate

1. Pressure-demand, full face-piece soifcontained breathing apparatus (SCEA), or pressure-demand supplied air respirator with escape SCBA, approved by the National institute for Occupational Safety and Health (NIOSH).

2. Totally-encapsulating chemical-

- protective suit
 - 3. Coveralle.
 - 4. Long underwear."
 - S. Gloves, outer, chemical-resistant.

6. Gloves, inner, chemical-resistant.

7. Boots, chemical-resistant, steel toe and shank.

6. Hard hat (under suit)*

9. Disposable protective sut, gloves and boots (Depending on sut construction, may

be worn over totally-encapsulating suit). 10. Two-way redics (worn inside

encapsulating suit).

"Opnonal as applicable.

IL Level 6—The highest level of respiratory protection is necessary but a lesser level of skin protection is needed.

Level 8 equipment: used as appropriate

1. Pressure-demand, full-facepiece selfcontained breathing apparatus (SCBA), or pressure-demand supplied air respirator with escape SCBA (NIOSH approved).

2. Hooded chemical-resistant clothing (overails and long-sleeved jacket: coveralls: one or two-piece chemical-splash suit disposable chemical-resistant overalls).

3. Coversils*.

4. Cloves. outer. chemical-resistant.

5. Cloves, inner, chemical-resistant, 6. Boots, outer, chemical-resistant steel toe

and snank.

7 Boot-covers, outer, chemical-resistant (disposable)*.

8. Hard hat.

9. Two-way radios (worn inside encapsulating suit).

10. Face shield.

*Optronal, as applicable.

III. Level C.—The concentration(s) and type(s) of automme substance(S) is known and the criterie for using air purifying respirators are met.

Level C equipment used as appropriate

1. Full-face or balf-mask. eir purifying. canisterequipped respirators (NIOSH approved).

2. Hooded chemical-resistant clothing (overails: two-prece chemical-splash suit disposable chemical-resistant overalls).

- 3. Covereils*.
- 4. Gloves. outer. chemical-resistant.
- 5. Cloves, inner. chemical-resistant.

6. Boots (outer). chemical-resistant steel toe and shank".

7. Boot-covers. outer. chemical-resistant (disposable)".

8. Hard hat

9. Escape mask*

10. Two-way radios (worn under outside protective clothing).

11. Face sheld"

Optional, as applicable.

IV. Level D—A work uniform affording minimal protection: used for nuisance contamination only.

Level D equipment: used as appropriate

1. Coveralls.

2 Cloves*

3. Boots/shoes. chemical-resistant steel toe and shank.

4. Boots. outer. chemical-resistant (disposable)*.

5. Safety glasses or chemical splash goggies".

6. Hard bat.

7. Escape mask*.

8. Face shield*.

*Optional as applicable.

Part 8. The types of hezards for which levels A. B. C. and D protection are appropriate are described below:

L Level A-Level A protection should be used when:

1. The hazardous substance has been identified and requires the highest level of protection for skin. eyes, and the respiratory system based on either the measured (or potential for) high concentration of atmospheric vapors, gases, or particulates: or the site operations and work functions involve a high potential for splash, immersion, or exposure to unexpected vapors, gases, or particulates of materials that are narmful to skin or capable of being absorbed through the intact skin.

2. Substances with a high degree of hazard to the skin are known or suspected to be present, and skin contact is possible, or

3. Operations must be conducted in confined, poorly ventilated areas and the absence of conditions requiring Level A have not yet been determined.

IL Level B protection should be used when: 1. The type and atmospheric concentration of substances have been identified and require a high level of respiratory protection, but less skin protection.

Note.—This involves atmospheres with IDLH concentrations of specific substances that do not represent a severe skin bezard: or that do not meet the criteria for use of airpunfying respirators.

2. The atmosphere contains less than 19.5 percent oxygen, or

3. The presence of incompletely identified vapors or gases is indicated by a directreading organic vapor detection instrument, but vapors and gases are not suspected of containing high levels of chemicals harmful to skin or capable of being absorbed through the intact skin.

III. Level C protection should be used when:

1. The atmospheric contaminants, liquid spiashes, or other direct contact will not adversely affect or be absorbed through any exposed skin.

2. The types of eur contaminants have been identified, concentrations measured, and a canister respirator is available that can remove the contaminants, and

3. All criteria for the use of au-punfying respirators are met.

IV. Level D protection should be used when:

1. The atmosphere contains no known hazard, and

2. Work functions preclude splashes, immersion, or the potential for unexpected inhalation of or contact with hazardous levels of any chemicals.

Note—As stated before combinations of personal protective equipment other than those described for Levels A. B. G. and D protection may be more appropriate and may be used to provide the proper level of protection.

Appendix C-Compliance Guidelines

1. Occupational Safety and Health Program. Each hazardous wests site chan-op effort will require a site specific occupational safety and health program headed by the site coordinator or the employer's representative. The program will be designed for the protection of employees at the site. The program will need to be developed before work begins on the site and implemented as work proceeds. The program is to facilitate coordination and communication among personnel responsible for the various activities which will take place at the site. It will provide the overall means for planning and unplementing the needed safety and

health training and job orientation of employees, who will be working at the site. The program will provide the means for identifying and controlling worksite hezards and the means for monitoring program effectiveness. The program will need to cover the responsibilities and authority of the site coordinator for the safety and health of employees at the site, and the relationships with contractors or support services as to what each employer s safety and bealth responsibilities are for their employees on the site. Each contractor on the site needs to have its own safety and health program so structured that it will smoothly interface with the program of the site coordinator.

Each sue safety and health program will need to include the following: (1) Policy statements of the line of authority and accountability for implementing the program. the objectives of the program and the role of the site safety and health officer or manager and staff: (2) means or mathods for the development of procedures for identifying and controlling workplace hazards at the site: (3) means or methods for the development and communication to employees of the various plans, work rules, standard operating procedures and practices that pertain to individual employees and supervisors: (4) the training of supervisors and employees to develop the needed skills and knowledge to perform their work in a safe and healthful manner; (5) means to anticipate and prepare for emergency situations and: (6) information feedback to aid in evaluating the program and for improving the effectiveness of ti program. The management and employe. should be trying continually to improve the effectiveness of the program thereby ennancing the protection being afforded those working on the site.

Accidents on the site should be investigated to provide information on how such occurrences can be avoided in the future. When injunes or diaesses occur on the site. they will need to be investigated to determine what needs to be done to prevent this incident from occurring egain. Such information will need to be used as feedback on the effectiveness of the program and the information turned into positive steps to prevent any reoccurrence. Receipt of employee suggestions or complaints relating to safety and health issues involved with site activities is also a feedback mechanism that needs to be used effectively to improve the program and may serve in part as an evaluative tooi(s).

2. Training. The employer is encouraged to utilize those training programs that have been recognized by the National Institute of Environmental Health Sciences through its training grants programs. These training end educational programs are being developed for the employees who work directly with hazardous substances. For further information about these programs contact: National Institute of Environmental Health Sciences. P.O. Box 12233. Research Triangle Park, NC 2709.

Training programs for emergency service orgenzations are available from the U.S. National Fire Academy. Emittsburg, MD the various state fire training schools. Th International Society of Fire Service Instructors. Ashland. MA is another resource. 3 Decontomination. Decontamination

procedures should be tailored to the specific hazards of the site and will vary in complexity, and number of steps. depending on the level of hazard and the employee s exposure to the hazard. Decontamination procedures and PPE decontamination methods will vary depending upon the specific substance. since one procedure or method will not work for all substances. Evaluation of decontamination methods and procedures should be performed. as necessary, to assure that employees are not exposed to hazards by reusing PPE. References in Appendix D may be used for guidance in establishing an effective decontamination program.

4. Emergency response plans. States, along with designated districts within the states. will be developing or have developed emergency response plans. These district and state plans are to be utilized in the emergency response plans called for in this standard. Each employer needs to assure that its emergency response plan is compatible with the local plan. In addition, the CAER program of the Chemical Manufacturers Association (CMA) is another helpful resource in formulating an effective emergency response plan. Also the current Emergency Response Guidebook from the U.S. Department of Transportation, CMA's CHEMTREC and the Fire Service Emergency Management Handbook should be used as tésources as weil.

Appendix D-References to Appendix

The following references to the Appendix may be consulted for further information on the subject of this notice: 1. OSHA Instruction DFO CPL 270-January 29. 1988. Special Emphasis Program: Hazaraous Waste Sites.

2. OSHA Instruction DFO CPL 2-2.37 A---January 23. 1988. Technical Assistance and Guidelines for Superfund and Other

Hazardous Waste Site Activities.

3. OSHA Instruction DTS CPL 2.74-January 29. 1988. Hazardous Waste Activity Form. OSHA 173.

4. Hazardous Waste Inspections Reference Manual, U.S. Department of Labor. Occupational Safety and Health Administration, 1986.

5. Memorandum of Understanding Among the National Institute for Occupational Safety and Heelth. the Occupational Safety and Heelth Administration, the United States Coast Guard, and the United States Environmental Protection Agency. Guidance for Worker Protection During Hazardous Woste Sile Investigations and Clean-up and Hazardous Substance Emergencies. December 18, 1980.

6. National Priorities List, 1st Edition, October 1984: U.S. Environmental Protection Agency, Revised periodically.

7. The Decontamunation of Response Personnel. Field Standard Operating Procedures (F.S.O.P.) 7: U.S. Environmental Protection Agency, Office of Emergency and Remedial Response. Hazardous Response Support Division, December 1984.

8. Preparation of a Site Safety Plan, Field Standard Operating Procedures (F.S.O.P.) 9: U.S. Environmental Protection Agency, Office of Emergency and Remedial Response. Hazgrdous Response Support Division. April 1985.

9. Standard Operating Safety Guidelines: U.S. Environmental Protection Agency, Office of Emergency and Remedial Response. Hazardous Response Support Division. ~ Environmental Response Team: November 1984.

10. Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities. National Insuitute for Occupational Safety and Health (NIOSH). Occupational Safety and Health Administration (OSHA). U.S. Coast Guard (USCG), and Environmental Protection Agency (EPA): October 1985.

11. Protecting Health and Safety at Hazardous Waste Sites: An Overview, U.S. Environmental Protection Agency, EPA/825/ 9-85/008: September 1985.

12. Hazardous Waste Sites and Hazardous Substance Emergencies. NIOSH Worker Bullenn, U.S. Department of Health and Human Services. Public Health Service. Centers for Disease Control. Netional Lastitute for Occupational Safety and Health: December 1962.

13. Personal Protective Equipment for Hazardous Materials Incidents: A Selection Guide: U.S. Department of Health and Human Services. Public Health Service. Centers for Disease Control. National Institute for Occupational Selety and Health: October \tilde{i} 1984.

14. Fire Service Emergency Management Handbook, Federal Emergency Management Agency, Washington, DC. January 1985.

15. Emergency Response Guidebook, U.S. Department of Transportation, Washington, DC, 1983.

[FR Doc. 86-28471 Filed 12-16-68: 12:57 am]

Interim Guidance on Compliance with Applicable or Relevant and Appropriate Requirements

[U	1 Directive Number					
SEPA OSWE	9234.0-05					
	2. Origini	Itor Information				
Name of Contact Person	Mail Code	Office	Telephone Number			
Arthur Weissman	WH-548D	OSWER/OERR/OPM/PAS	382-2182			
Interim Guidance on Compliance with Applicable or Relevant and Appropriate Requirements						
4 Summary of Directive (Include brief statement of purpose) The guidance addresses the requirement in CERCLA, as amended by the Superfund Amendments and Reauthorization Act of 1986, that remedial actions comply with applicable or relevant and appropriate requirements (ARARs) of Federal laws and more stringent, promulgated State laws. The guidance describes how requirements are generally to be identified and applied, and discusses specifically compliance with State requirements and certain surface water and groundwater standards.						
5 Keywords Superfund, CERCLA, SARA, Other environemtnal requirements, Compliance						
5a. Does this Directive Supersede Previous Directive(s)? Yes X No What directive (number, title) b Does It Supplement Previous Directive(s)? X Yes No What Directive (number, title) 9234.0-02 CERCLA Compliance with other Environmental Statutes						
7 Draft Level A — Signed by AA/DAA B — Signed by Office Director C — For Review & Comment In Development						
This Request Meets OSWER Directives System F	ormat					
8. Signature of Lead Office Oprectives Coordinato	e		Dete 6/26/87			
9 Name and Title of Approving Official-7			Date 7/9/87			
J. ATTISLON POPLER, ASSISTANT A	aministrato)r	<u> </u>			

OSWER OSWER OSWER DIRECTIVE DIRECTIVE



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JL 9 1987

OFFICE OF SOLIO WASTE AND EMERGENCY RESPONSE

9234.0-05

MEMORANDUM •

SUBJECT: Interim Guidance on Compliance with Applicable or Relevant and Appropriate Bequirements FROM: J. Winston Porter

Assistant Administrator

TO: Addressees

Executive Summary

The guidance addresses the requirement in CERCLA, as amended by the Superfund Amendments and Reauthorization Act of 1986, that remedial actions comply with applicable or relevant and appropriate requirements (ARARs) of Federal laws and more stringent, promulgated State laws. The quidance describes how requirements are generally to be identified and applied, and discusses specifically compliance with State requirements and certain surface water and groundwater standards. "Applicable" and "relevant and appropriate" are defined, and the three types of ARARs (chemical-, location-, and action-specific) are described. Guidance is given on how and at what points ARARs are to be used in the remedial process. Eligible State requirements are defined, with particular reference to "promulgated," and direction is given on evaluating siting laws and on using the waiver regarding consistency of application. Finally, the guidance discusses the use of water standards specified in the law (MCLGs, FWQC, ACLs), and describes the use of MCLs as cleanup standards for surface water or groundwater that is or may be used for drinking.

Purpose

This memorandum provides interim guidance on compliance with other Federal and State environmental laws in conducting CERCLA remedial actions. The guidance is intended to help define the nature, scope, and use of applicable or relevant and appropriate requirements. The guidance is not intended to be comprehensive or exhaustive. The Agency is currently developing a guidance manual that provides detailed information on potential ARARs in the major Federal environmental statutes.
Background

Section 121(d) of CERCLA, as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA), requires that Fund-financed, enforcement, and Federal facility remedial actions comply with requirements or standards under Federal and State environmental laws. The requirements that must be complied with are those that are applicable or relevant and appropriate to the hazardous substances, pollutants, or contaminants at a site or to the circumstances of the release. Compliance is required at the completion of the remedial action for hazardous substances, pollutants, or contaminants that remain on-site. Any such requirements may be waived under six conditions provided that protection of human health and environment is still assured.

SARA essentially codified and expanded upon the Agency's Compliance Policy, which was included in the National Contingency Plan (revised November 20, 1985). The major difference between that policy and the new statutory requirement is that the latter includes more stringent, promulgated State environmental standards as potentially applicable or relevant and appropriate requirements, and Maximum Contaminant Level Goals and Federal Water Quality Criteria as potentially relevant and appropriate requirements.

GENERAL GUIDANCE ON IDENTIFYING AND USING ARARS

This section defines what ARARs are, describes the different types of ARARs, and discusses how they are applied to the remedial process.

Definition of ARARs

A requirement under other environmental laws may be either "applicable" or "relevant and appropriate" to a remedial action, but not both. A twotier test may be applied: first, to determine whether a given requirement is applicable; then, if it is not applicable, to determine whether it is nevertheless relevant and appropriate.

<u>Applicable requirements</u> means those cleanup standards, standards of control, and other substantive environmental protection requirements, criteria, or limitations promulgated under Federal or State law that specifically address a hazardous substance, pollutant, contaminant, remedial action, location, or other circumstance at a CERCLA site.

"Applicability" implies that the remedial action or the circumstances at the site satisfy all of the jurisdictional prerequisites of a requirement. For example, the minimum technology requirement for landfills under RCRA would apply if a new hazardous waste landfill unit (or an expansion of an existing unit) were to be built on a CERCLA site.

<u>Relevant and appropriate requirements</u> means those cleanup standards, standards of control, and other substantive environmental protection requirements, criteria, or limitations promulgated under Federal or State law that, while not "applicable" to a hazardous substance, pollutant, contaminant, remedial action, location, or other circumstance at a CERCLA site, address problems or situations sufficiently similar to those encountered at the CERCLA site that their use is well suited to the particular site.

The relevance and appropriateness of a requirement can be judged by comparing a number of factors, including the characteristics of the remedial action, the hazardous substances in question, or the physical circumstances of the site, with those addressed in the requirement. It is also helpful to look at the objective and origin of the requirement. For example, while RCRA regulations are not applicable to closing undisturbed hazardous waste in place, the RCRA regulation for closure by capping may be deemed relevant and appropriate.

A requirement that is judged to be relevant and appropriate must be complied with to the same degree as if it were applicable. However, there is more discretion in this determination: it is possible for only <u>part</u> of a requirement to be considered relevant and appropriate, the rest being dismissed if judged not to be relevant and appropriate in a given case.

Non-promulgated advisories or guidance documents issued by Federal or State governments do not have the status of potential ARARs. However, as described below, they may be considered in determining the necessary level of cleanup for protection of health or environment.

Types of ARARs

There are several different types of requirements that Superfund actions may have to comply with. The classification of ARARs below is offered for illustrative purposes.

• <u>Ambient or chemical-specific requirements</u> set health or riskbased concentration limits or ranges in various environmental media for specific hazardous substances, pollutants, or contaminants. Examples: Maximum Contaminant Levels, National Ambient Air Quality Standards.

These requirements may set protective cleanup levels for the chemicals of concern in the designated media, or else indicate an acceptable level of discharge (e.g., air emission or wastewater discharge taking into account water quality standards) where one occurs in a remedial activity. If a chemical has more than one such requirement, the more stringent ARAR should be complied with.

There are at present a limited number of actual ambient or chemicalspecific requirements. In order to achieve remedies that are protective of health and environment, it may frequently be necessary to use chemicalspecific advisory levels such as Carcinogenic Potency Factors or Reference Doses. While not actually ARARs, these chemical-specific advisory levels may factor significantly into the establishment of protective cleanup levels. Guidance for establishing such chemical-specific, health-based cleanup levels is given in the Superfund Public Health Evaluation Manual (EPA 540/1-86/060, Oct. 1986). • <u>Performance</u>, <u>design</u>, <u>or</u> <u>other action-specific requirements</u> set controls or restrictions on particular kinds of activities related to management of hazardous substances, pollutants, or contaminants. Examples: RCRA regulations for closure of hazardous waste storage or disposal units; RCRA incineration standards; Clean Water Act pretreatment standards for discharges to POIWs.

These requirements are triggered not by the specific chemicals present at a site but rather by the particular remedial activities that are selected to accomplish a remedy. Since there are usually several alternative actions for any remedial site, very different requirements can come into play. These action-specific requirements may specify particular performance levels, actions, or technologies, as well as specific levels (or a methodology for setting specific levels) for discharged or residual chemicals.

• Locational requirements set restrictions on activities depending on the characteristics of a site or its immediate environs. Examples: Federal and State siting laws for hazardous waste facilities; sites on National Register of Historic Places.

These requirements function like action-specific requirements. Alternative remedial actions may be restricted or precluded depending on the location or characteristics of the site and the requirements that apply to it.

Using ARARs

This section explains how and where requirements may be applied in the remedial planning process.

First, actual ARARs can be identified only on a site-specific basis. They depend on the specific chemicals at a site, the particular actions proposed as a remedy, and the site characteristics. Guidance is being developed on the potential ARARs under the major Federal environmental statutes for various activities, locations, and chemicals.

Where there are no specific ARARs for a chemical or situation, or where such ARARs are not sufficient to be protective, one should identify pertinent health advisory levels (such as Reference Doses or Carcinogenic Potency Factors) as described above in order to ensure that a remedy is protective.

The different ARARs that may apply to a site and its remedial action should be identified and considered at multiple points in the remedial planning process, namely:

- During scoping of the RI/FS, chemical-specific and location-specific ARARs may be identified on a preliminary basis.
- During the <u>site characterization phase</u> of the Remedial Investigation, when the public health evaluation is conducted to assess risks at a site, the chemical-specific ARARs and advisories and location-specific ARARs are identified more comprehensively and used to help determine the cleanup goals.

- During <u>development of remedial alternatives</u> in the Feasibility Study, action-specific ARARs are identified for each of the proposed alternatives and considered along with other ARARs and advisories.
- During <u>detailed analysis of alternatives</u> all the ARARs and advisories for each alternative are examined as a package to determine what is needed to comply with other laws and be protective.
- When an <u>alternative is selected</u> it must be able to attain all ARARs unless one of the six statutory waivers is invoked.
- During remedial design the technical specifications of construction must ensure attainment of ARARs.

Note that CERCLA §121(e) exempts any on-site response action from having to obtain a Federal, State, or local permit.

In general, <u>on-site actions need comply only with the substantive</u> <u>aspects of these requirements</u>, not with the administrative aspects. That is, neither applications nor other administrative procedures such as permitting or administrative reviews are considered ARARs for actions conducted entirely on-site, and therefore should not be pursued during the remedial planning or the remedial action. However, the RI/FS, Record of Decision, and design documents should demonstrate full compliance with all substantive requirements that are ARARs. Also, other Federal and State program offices should be consulted as appropriate to ensure that remedies are substantively compliant with identified ARARs.

GUIDANCE ON IDENTIFYING STATE ARARS

This section describes the basic factors to be considered in identifying State requirements for Superfund remedial actions.

States are required by CERCLA to identify State ARARs "in a timely manner," that is, in sufficient time to avoid inordinate delay or duplication of effort in the remedial process. Regions should expect to work closely with their States so that the appropriate ARARs are identified at critical stages in the process. At a minimum, chemical-specific and location-specific ARARs should be identified after site characterization, and action-specific ARARs should be identified after initial screening of alternatives (prior to detailed analysis) for alternatives that pass through the screening. To the extent possible, Regions and States should negotiate to try to resolve any differences of opinion about ARARs.

Eligible Requirements

The statute specifically limits the scope of potential requirements to those that are promulgated. "Promulgated" requirements are laws imposed by State legislative bodies and regulations developed by State agencies that are of general applicability and are legally enforceable. State advisories, guidance, or other non-binding policies, as well as standards that are not of general application, cannot be treated as requirements under CERCLA. However, as with their Federal counterparts, State advisories may still be considered in determining an appropriate, protective remedy.

General State goals that are duly promulgated (such as a nondegradation law) have the same weight as explicit, numerical standards, although the former have to be interpreted in terms of a site and therefore may allow more flexibility in approach. Similarly, State laws or regulations that prescribe methods for deriving numerical standards for specific cases may also be potential requirements.

<u>On-site actions need comply only with the substantive aspects of a</u> <u>State requirement</u>, not with the administrative aspects. Where the requirement involves review by a State board based on explicit criteria, the best approach is to incorporate the substantive criteria into the RI/FS and remedy selection process and to maintain close consultation with appropriate State representatives.

Limitations on State Siting Laws

CERCLA \$121(d)(2)(C) puts special limitations on the applicability of State requirements or siting laws for hazardous waste facilities that could result in a State-wide prohibition of land disposal. Specifically, in order to be treated as potentially applicable or relevant and appropriate requirements, such laws must:

- 1) be of general applicability and be formally adopted
- 2) be based on technical (e.g., hydrogeologic) or other relevant considerations
- 3) not be intended to preclude land disposal for reasons other than protection of health or environment.

In addition, the State must arrange and pay for additional costs for outof-State or other disposal necessitated by such a law.

The first criterion is similar to the criterion that a requirement be promulgated, as discussed above. The second criterion requires that such a law be based on sound scientific or technical considerations, such as groundwater flow, surficial geology, and engineering design. The third criterion requires some evidence that health or environmental protection motivates the prescribed restrictions; the introductory sections of a law, the nature of the technical considerations, or the legislative history can be used to make this determination.

Consistency of Application

CERCLA (d)(4)(E) allows a State requirement to be waived if it has not been consistently applied by the State in similar circumstances at other remedial actions. The waiver cannot be used if the State has demonstrated the intention to consistently apply the requirement. Consistency of application by a State may be determined by examining the following:

- Application of requirement at similar sites or in similar response circumstances (considering nature of contaminants or media affected, characteristics of waste and facility, degree of danger or risk, etc.)
- Proportion of cases (including enforcement actions) in which requirement was not applied out of total actions where it could have been applied
- Reason for non-application of requirement in past cases
- Intention to consistently apply requirement in future as shown by policy statements, legislative history, site remedial planning documents, or State responses to Federal-lead sites; newly promulgated requirements shall be presumed to embody this intention unless there is contrary evidence.

All previous actions by States since promulgation that relate to similar remedial actions may be considered in evaluating consistency.

GUIDANCE ON APPLYING SPECIFIED WATER STANDARDS

CERCLA §121(d)(2)(A) and (B) explicitly mention three kinds of surface water or groundwater standards with which compliance is potentially required - Maximum Contaminant Level Goals (MCLGs), Federal Water Quality Criteria (FWQC), and alternate concentration limits (ACLs) where human exposure is to be limited. This section describes these requirements and how they may be applied to Superfund remedial actions. The guidance is based on Federal requirements and policies; more stringent, promulgated State requirements (such as a stricter classification scheme for groundwater) may result in application of even stricter standards than those specified here.

Background

These three standards or criteria each derive from separate statutes and have different purposes and uses.

MCLGs are developed under the Safe Drinking Water Act as chemicalspecific health goals used in setting enforceable drinking water standards, known as Maximum Contaminant Levels (MCLs), for public water supply systems. MCLGs are based entirely on health considerations and do not take cost or feasibility into account. Moreover, as health goals MCLGs are set at levels where no known or anticipated health effects may occur, including an adequate margin of safety. MCLs are required to be set as close as feasible to the respective MCLGs, taking into consideration the best technology, treatment techniques, and other factors (including cost). However, as the standard for public water supplies, MCLs are fully protective of human health and (for carcinogens) fall within the acceptable risk range of 10⁻⁴ to 10⁻⁷. Furthermore, for non-carcinogens, which are the majority of contaminants, MCLs will nearly always be set at the same level as the respective MCLGs. Also, these standards assure that even sensitive populations will experience no adverse health effects. Thus, there will be no difference in the protectiveness of MCLGs and MCLs for most contaminants, and, as discussed above, MCLs provide a sufficient level of protectiveness even for carcinogens.

FWQC are developed under the Clean Water Act as guidelines from which States determine their water quality standards. Different FWQC are derived for protection of human health and protection of aquatic life.

ACLs are one of three possible standards available under the Subpart F Groundwater Protection Standards of RCRA. For setting both a trigger and a cleanup level for remediating groundwater contamination, an ACL, the background concentration, or for a small group of chemicals the MCL can be selected for a given site.

Statutory Mandate

CERCLA §121(d)(2) states that remedial actions shall attain applicable or relevant and appropriate requirements under the Safe Drinking Water Act, the Clean Water Act, and RCRA, and specifically shall attain MCLGs and FWQC where they are relevant and appropriate under the circumstances of the release or threatened release. It further states that for FWQC this determination will be based on the designated or potential use of the water, the media affected, the purposes of the criteria, and current information.

CERCLA §121(d)(2)(B)(ii) limits the use of ACLs that are set above health-based levels based on projections that health-based levels will be achieved at a likely point of human exposure. Such a point of exposure may not be beyond the Superfund facility boundary unless the groundwater discharges into surface water and does not cause a statistically significant increase of contaminants in the surface water. To apply such an ACL outside the facility, moreover, the remedial action must include enforceable measures to prevent use of any contaminated groundwater.

Application

In determining the applicable or relevant and appropriate requirements for remedial actions involving contaminated surface water or groundwater, the most important factors to consider are the uses and potential uses of the water and the purposes for which the potential requirements are intended.

The actual or potential use of water, and the manner in which it is used, will determine what kinds of requirements may be applicable or relevant and appropriate. For Class III-type groundwater that is not suitable for drinking because of high salinity or widespread contamination and that does not affect drinkable groundwater, drinking water standards are neither applicable nor relevant and appropriate. For Class I- and Class II-type groundwater or surface water that is or may be used for drinking, drinking water standards are applicable or relevant and appropriate, and the surface water or groundwater must ultimately be cleaned up to such levels.

For water that is or may be used for drinking, the Maximum Contaminant Levels (MCLs) set under the Safe Drinking Water Act are generally the applicable or relevant and appropriate standard. MCLs are applicable at the tap where the water will be provided directly to 25 or more people or will be supplied to 15 or more service connections. Otherwise, where surface water or groundwater is or may be used for drinking, MCLs are generally relevant and appropriate as cleanup standards for the surface water or the groundwater.

A standard for drinking water for a contaminant for which there is an MCL may be more stringent than the MCL to ensure adequate protection in special circumstances, such as where either multiple contaminants in groundwater or multiple pathways of exposure present extraordinary risks. In setting a level more stringent than the MCL in such cases, a site-specific determination should be made by considering MCLGs, the Agency's policy on the use of appropriate risk ranges for carcinogens, levels of quantification, and other pertinent guidelines. Prior consultation with Headquarters is encouraged in such cases.

When MCLs do not exist for contaminants identified at the site, cleanup levels should be set using chemical-specific advisory levels. Cleanup levels should be selected such that the total risk of all contaminants falls within the acceptable risk range of 10^{-4} to 10^{-7} . In cases where noncarcinogens are present, cleanup levels should be based on acceptable levels of exposure as determined by the Reference Dose, taking into account the effects of other contaminants at the site.

It should be noted that while MCLs are generally the cleanup standards, as described above, the treatment necessary to attain an MCL level for one chemical (or a protective level for a chemical without an MCL) may result in an actual level for another chemical that is below its respective MCL (or protective level).

A more stringent FWQC for aquatic life may be found relevant and appropriate when there are environmental factors that are being considered at a site, such as protection of aquatic organisms. The Agency is still formulating a position with respect to the use of FWQC for protection of human health.

Guidance on the use of ACLs based on limitations on exposure will be forthcoming.

* * *

Further Information .

For further information on the subject matter in this interim guidance, contact Steve Smith (FTS-382-2200) or Arthur Weissman (FTS-382-2182) of the Policy and Analysis Staff, Office of Emergency and Remedial Response.

Addressees

Regional Administrators, Regions I-X Regional Counsel, Regions I-X Director, Waste Management Division, Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division, Region II Director, Hazardous Waste Management Division, Regions III and VI Director, Toxics and Waste Management Division, Region IX Director, Hazardous Waste Division, Region X Environmental Services Division Directors, Region I, VI, and VII

Notification of Restrictions on Reimbursement of Private Party Costs for Removal Actions

Varies of Constit Person Originator Information Varies of Constit Person Mail Code Varies of Constit Person Mail Code Varies of Constit Person Wall Code Varies of Constit Person Wall Code Varies of Constit Person Varies Office Varies of Constit Person Output Varies of Constit Person Separature of Office Varies of Constit Person Separature of Office Varies of Constit Person Output Separature of Office Output Varies of Constit Person Output Separature of Office Output Varies of Office Output Varies of Overconse Notification of Restrictions on Reimbursement of Private Party Costs Notification of Restrictive provisions of CERCLA regarding private party reimbursement for renoval costs. Directs the Regions to ensure affected Communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Varies of Overconse Subscrease Previous Overconse Annual cercl Oright Policy directive Oright Varies of Unconse Subscrease Previous Overconse Annual cercl One Varies of Unconse Subscres P	SEPA	United States Environmen Washington. SM/ED Diractiva (ntal Protection Agency OC 20460		Interim Directive Num
Summary of Directive Mult Code WH-548 Transmort Humber 332-2452 Jack is Dirichen Directive Dust Signature of Office Director Director Downe Dust Signature of Office Director WH-548 Description Approved for Reverve Date Director WH-548 Director Downe Director WH-548 Description Director WH-548 Description Director Date Director WH-548 Description Date Director Date Director Date Director Directo Director Director Director Director Director Director Director D		Originate	or information		10,0,0.0
Jack 19 UZ 100001 INT-340 Account of Account	Name of Contact Person	Mail Code		Teleshone Numb	81
decar - Ouvre Separare at Other Director Notification of Restrictions on Reimbursement of Private Party Costs ror Removal Actions Notification of Restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) read Directore (Manual Policy Directore, Announcement, etc.) Policy directive resting Directore Streamed Directores (Inter- Separate Office Directores from the inter- readed Directores Streament Directores (Inter- Costs Outre (Secondry Outre Office Directores Contents Contents Outre (Secondry Outre Contents Outre Contents Outre Office Directores Contents Outre (Secondry Outre Contents	Jackie Uziuden	WN-340		382-2452	
Defail		Signature of Office Dir	rector	a lar verien	1 Oara
OSW ALOSWER IV. WMMSTBAL IM H. MMGAT. 11127185- Notification of Restrictions on Reimbursement of Private Party Costs for Removal Actions Winnery of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) The of Ourective Manual, Pancy Directore. Announcement, etc.] Policy directive Submy Directore Substance Previous Directore(sh) Yes Lef No Does It Supplement Previous Directore(sh) The of Ourective Substance Previous Directore(sh) The of Directore Substance Previous Directore(sh) The of Directore Substance Substance Previous Directore(sh) The of Directore Substance Prevenues Directore(sh) The of Directore Substance Previous Di			-1 · 1 · 11	(ulusice
Intermediation Image: Status Stat		IEA IV. WIMAS	STOME LANH.	Un cint.	11127185
Notification of Restrictions on Reimbursement of Private Party Costs for Removal Actions Unimmary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) rea of Directive MManual, Poincy Directive, Anniaumement, etc.] Policy directive Policy directive Policy Directives Supersede Previous Directive(s) Previous Previous Directives Supersede Previous Directive(s) Previous Previous Previous Directives Corective (number, tile) Previous Previous Directives Corectives (number, tile) Previous Directives Corectives (number, tile) Previous revious Previous Previous Previous Previous Previous	fille -	- /	l	U	•
for Removal Actions for Removal Actions Ummary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) read Orective Manual, Paker Directore, Announcement, etc. Policy directive Policy directive Policy directive Policy directives functional Orectives (I) Yes _ Ho Does It Supplement Previous Directives (I) Yes _ Ho Does It Supplement Previous Directives (I) Yes _ Ho Does It Supplement Previous Directives (I) Yes _ Other (Specify) Yes _ Directive Policy directives Gifter Policy directives Gifter Policy directives Gifter Previous Directives Gifter Previous Directives Gifter Previous Directives Gifter Previous Office Directives Gifter Policy directives Gifter Previous Directives Directives Gifter Previous Directives Gifter Previous Directives Gifter Previous Direct	Notification of A	Restrictions on Reimb	ursement of Pri	vate Party C	osts
Summery of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: Status of Outlines of Outl	for Removal Actio	ons		-	
Summary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985)	•	•			
Nummary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: the provision of the set provisions. (Signed H. Longest, November 25, 1985) Image: the provision of the set provision of the set provision of the set provision of the set provision. (Signed H. Longest, November 25, 1985) Image: the provision of the set provide of the set provide of the set of the set of the set of the set provide of the set of the set provide of the set of the set of the set provide of the set of the set provide of the set of the set provide of the set provide of the set provide of the set of the set provide of the set provide of the set of the set provide of the set provide of the set of the set provide of the set of the set provide of the set provide of the set provide of the set of the set provide of the set pr					
Jummary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) When the provision of the provisi					
Jummary of Directive Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: Signed Directive (Manual, Pater Directive, Announcement, etc.) Policy directive Image: Signed Directive Supersece Previous Directive(si) Image: Signed H. Longest, Supersece Previous Directive(si) Image: Signed H. Longest, Supersece Previous Directive(si) Image: Signe Core Supersece Previous Supersece Previous Corective(si) Image: Signe Core Supersece Previous Supersece Previous Corective(si) Image: Signe Core Supersece Previous Supersece Previous Supersece Previous Corective(si) Ima					
Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. (Signed H. Longest, November 25, 1985) Image: the restrictive provision of the provis	lummary of Directive			•	
Outlines the restrictive provisions of CERCLA regarding private party reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. (Signed H. Longest, November 25, 1985) Image: state informed of these provisions. Image: state informed provision	;	•		-	
reimbursement for removal costs. Directs the Regions to ensure affected communities are informed of these provisions. (Signed H. Longest, November 25, 1985)	⁷ Outlines the rest	rictive provisions of	CERCLA regardi	ng private p	arty
communities are informed of these provisions. (Signed H. Longest, November 25, 1985)	reimbursement for	removal costs. Dire	cts the Regions	to ensure a	ffected
(Signed H. Longest, November 25, 1985) The set of Oursetive (Manuel, Palicy Directive, Announcement, etc.) Policy directive Policy Po	communities are in	nformed of these prov	isions.		
(Signed R. Longest, November 23, 1985)	(Cienad V. Lange-			•	
rps of Directives (Manual, Palley Directive, Announcement, etc.) Status Policy directive Directive Policy directive Directive Bass this Directive Supersede Previous Directive(s)? Yes Privation Privation Bass this Directive Supersede Previous Directive(s)? Yes Privation Directive (number, title) Press to Either Question, What Directive (number, title) Other (Specify) Press to Either Question, What Directives (number, title) Other (Specify) Press to Either Question, What Directives (number, title) Other (Specify) Press to Either Question, What Directives System Format Oppe Bass Accuest Meets OSWEA Directives System Format Oppe Brature of Lead Office Directives Qflicer Date MMM WMASS Date Stature of OSWEA Directives Qflicer Date Stature of OSWEA Directives Qflicer Date Stature of OSWEA Directives Qflicer Date Date Date Stature of OSWEA Directives Qflicer Date	(Signea M. Longes)	C, NOVEMDER 25, 1985)	1		
rps of Directive (Manual, Policy Directive, Announcement, etc.) Status Policy directive Orsk Policy directive Orsk Policy directive Prinal Revision Des this Directive Supersede Previous Directive(s)? Yes Prinal Revision Des this Directive Supersede Previous Directive(s)? Yes Prinal Revision Des this Directive Supersede Previous Directive(s)? Yes Prinal Occ Des this Directive Supersede Previous Directive(s)? Yes Prinal Occ Tree" to Either Question, What Directive (number, title) Prinal Occ Prinal Occ Prinal Occ Prinal Occ Prinal Occ Prinal Occ OSW R Owpe OGC Occ OSW R Orectives Suscer Format Stature of Lead Office Directives Qflicer Date MMM MMMASTME Date Stature of OSWER Officer Date Stature of OSWER Officer Date <		٨			
Anoswer of Oswer Oswer Officer status of Oswer Oswer Officer status of Oswer Officer status Office Oswer Oswer Officer Man Man Status Status Man Man Status Status Man Man Status Status Man Man Status Status Man Man Status Sta		· · ·//			
rps of Directives (Manuel, Policy Directive. Announcement, etc.) policy directive Policy d		MAUL	and the		
Image: Status Image: Status Image: Status Image: Status Image: Status Image: Status Policy directive Image: Status Policy directive Image: Status Image: Status Image: Status Policy directive Image: Status Image: Status Image: Status Imat			c Ture		
Image: Status Image: Status Image: Status Image: Status Policy directive Image: Status Policy directive Image: Status Policy directive Image: Status Image: Status Image: Status Policy directive Image: Status Image: Status Image: Status I	•	IV. which	ר -		
Prove of Directive (Manual, Policy Directive, Announcement, etc.) Policy directive Policy Policy directive Policy Pol		·nl ' IN)	1		
rps of Directive (Manual, Policy Directive, Announcement, etc.) Policy directive Previous Previous Policy directive Policy directive Policy Policy directive Policy Policy directive Policy Policy directive Previous Previous Policy Policy Policy Policy Previous Previous Previous Previous Previous Policy Policy Policy Previous Previous Policy Policy Policy Previous Previous Previous Policy Policy Policy Previous Previous Previous Policy Policy Policy Policy Previous Previous Policy Poli			K.		
Image: Policy directive (Manual, Palicy Directive, Announcement, etc.) Status Policy directive Image: Directive (Manual, Palicy Directive, Announcement, etc.) Image: Directive Directive Policy directive Image: Directive (Manual, Palicy Directive(s)) Image: Policy directive (Manual, etc.) Dess this Directive Supersede Previous Directive(s) Image: Policy directive (Number, title) Image: Policy directive (Number, title) Image: Policy Directive (Duestion, What Directive (number, title) Image: Policy directive (Number, title) Image: Policy Directives (Duestion, What Directive (number, title) Image: Policy directive (Number, title) Image: Policy Directives (Duestion, What Directive (Number, title) Image: Policy directive (Number, title) Image: Policy Directives System Format Image: Policy directives System Format Image: Policy Directives Giftcer Image: Policy directives (Date Image: Policy directives Giftcer Image: Policy directives (Date <t< td=""><td></td><td>$(\mathcal{P}_{\mu\nu},\mathcal{V}_{\mu\nu})$</td><td>5.</td><td></td><td></td></t<>		$(\mathcal{P}_{\mu\nu},\mathcal{V}_{\mu\nu})$	5.		
rpe of Ourective (Manual, Policy Durective, Announcement, etc.) Status Policy directive Image: Construction Policy directive Image: Construction Des this Directive Supersede Previous Directive(s)? Yes Image: Previous Directive (number, title) Image: Previous Directives (number, title) Image: Previous Directive (number, title) Image: Previous Directives System Format Image: Previous Directives Constructives System Format Image: Previous Directives Constructives Constructives Previous Directives Constructives Image: Previous Directives Co	•	Janey 11,418	5.		
ppe of Ourective (Manual, Palicy Directive, Announcement, etc.) Policy directive Policy directive Status Urective Supersede Previous Directive(s)? Yes If No Does It Supplement Previous Directive(s)? Yes If No Does It Supplement Previous Directive(s)? Yes If No Policy directive Supersede Previous Directive(s)? Yes If No Does It Supplement Previous Directive(s)? Yes If No Policy directive Supersede Previous Directive(s)? Yes If No Does It Supplement Previous Directive(s)? Yes If No Policy directive Supersede Previous Directive(s)? Yes If No Does It Supplement Previous Directive(s)? If No Ocean If No If No Ocean If No If No <	h	Janey 11/27/8	5.		
Policy directive Policy directive Policy directive Policy directive Policy directive Policy directive Previous Directive(s) Previous	ł	Janey 11/27/8	5 .		
Image: Supersede Previous Ourective(s)/ Yes If No Oces It Supplement Previous Oirective(s)/ Yes "Yes" to Either Question, What Oirective (number, title) Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan Image: Plan	rpe of Oirective (Manual, Policy Direc	We Announcement etc.	5. 	Status	
Des Ithis Directive Supersede Previous Directive(s)? [Yes [PNo Does It Supplement Previous Directive(s)? [Yes ['Yes'' to Either Question, What Directive (number, title) ''Yes'' to Either Question, What Directives System Format ''Yes'' to Either Question, What Directives Cifficer ''Yes'' to Either Question, What Directives Gifficer ''Yes'' to Either Question, ''Yes'' to Either Question, ''Yes'''' to Either Question, ''Yes'	pe of Ourecuve (Manual, Policy Ourec	Jure Announcement etc.)	5. 	Status Oraft	
"Yes" to Either Question, What Directive (number, title) Image: Pfan	Policy directive	With Announcement. etc.)	5. 	Status Oraft · Final	4 New Revision
IviewPfan	ppe of Orrective (Manual, Policy Orrect Policy directive	S Directive(s)?	No Oces It Supplem	Status Oraft • G Final ent Previous Direct	Vetsij U Yes
IviewPfan I AA-OSWER OUST CECM Other (Specify) I CERR OWPE I CSW Regions OPPE Is Request Meets OSWER Directives System Format gnature of Lead Office Directives Officer Date II/124/85 gnature of OSWER Directives Gifficer Date II/124/85	pe of Orrective (Manual, Policy Orrect Policy directive Des this Orrective Supersede Previous "Yes" to Either Question, What Orrec	S Directive(s)? Yes If	No Oces It Supplem	Status Oraft Final ent Previous Oirect	Vetsi? Yes
Image: State of the state	rpe of Ourective (Manual, Policy Ourect Policy directive Des this Ourective Supersede Previous "Yes" to Either Question, What Direct	S Directive(s)? Yes If	No Oces It Supplem	Status Oraft Graft Final ent Previous Oirect	Vetsi? Yes
Image: Construction Image: Construction Imag	ype of Ourective (Manual, Policy Ourect Policy directive oes this Ourective Supersede Previous "Yes" to Either Question, What Directive	S Directive(s)? Yes Lit	No Oces It Supplem	Status Oraft Final ent Previous Direct	Vetsi? Yes
CEARR OWPE OGC CSW Regions OPPE is Request Meets OSWER Directives System Format Oate gnature of Lead Office Directives Officer Oate MMM WMASTME 1//24/85 gnature of OSWER Directives Gifficer Oate Shemic Directives Gifficer Oate	Policy directive (Manual, Policy Direct Policy directive Des this Directive Supersede Previous "Yes" to Either Question, What Directive	S Directive(s)? Yes It	No Oces It Supplem	Status Oraft · G Final ent Previous Oirect	VetSi7 Yes
LI CSW LI Regions LI OPPE is Request Meets OSWER Oirectives System Format gnature of Lead Office Oirectives Officer Oate MAMM WMASTME gnature of OSWER Oirectives Gificer Oate Shemic Dialdiha, for P. J. Brown and III 27.55	rpe of Directive (Manual, Policy Direct Policy directive Des this Directive Supersede Previous "Yes" to Either Question, What Direct Interwittan	S Directive(s)? [] Yes [] Clive (number, title)	No Oces It Supplem	Status Oraft · G Final ent Previous Direct	I Ves
Shewic Divectives Gillicer Date UID 2.55	Policy directive (Manual, Policy Direct Policy directive Des this Office Supersede Previous "Yes" to Either Question, What Directive IviewPlan AA-OSWER 0UST	S Directive(s)? Yes If CECM	No Does It Supplem	Status Oraft Final ent Previous Direct	I New Revision I Yes
MAMM WMASTME - Date 1/27/85 gnature of OSWER Directives Gifficer Date 1/27/85	Policy directive (Manual, Policy Direct Policy directive Des this Directive Supersede Previous "Yes" to Either Question, What Directive Diview Pfan AA-OSWER 0UST CERR 0WPE CSW Regions	S Directive(s)? Yes It CECM CECM OGC OPPE	No Oces It Supplem	Status Oraft Final ent Previous Oirect	Image: Provision Imag
NUMM WMASTIME - 1/127/85 produce of OSWER Directives Gifficer Date - 11/27/85	Policy directive (Manual, Policy Direct Policy directive Des this Directive Supersede Previous "Yes" to Either Question, What Directive Max-OSWER 00057 CERR 00075 CERR 00075 CERR 00075 CERR 00075	Svstem Format	No Oces It Supplem	Status Oraft - G Final ent Previous Direct	I I New Revision I Yes
gnature of OSWER Directives Gilicer Date -	Policy directive (Manual, Policy Direct Policy directive Des this Directive Supersede Previous "Yes" to Either Question, What Direct Maa-OSWER OUST CERR OWPE CSW Aegions is Accuest Meets OSWER Orectives Citi	Svstem Format	No Oces It Supplem	Status Oraft G Final ent Previous Direct	Gate
Sherry Fielding for P 16man 1 11/22/05-	rpe of Ourective (Manual, Policy Ourect Policy directive Des this Ourective Supersede Previous "Yes" to Either Question, What Oirect WiewPfan AA-OSWER OUST - QERR OWPE GSW Regions is Acquest Meets OSWER Ourectives insture of Lead Office Ourectives Office WMM WIM 6	Svstem Format	No Oces It Supplem	Status Oraft Final ent Previous Oirect	Vetsi? Ves Oate 1/172/85
	ype of Orrective (Manual, Policy Orrect Policy directive Des this Orrective Supersede Previous "Yes" to Either Question, What Direct Wiew/Plan AA-OSWER OUST CERB OWPE CSW Regions is Accuest Meets OSWER Orrectives gnature of Lead Office Orrectives Offic MAM WMAS posture of OSWER Directives Officer	Source: Versile Sourcement etc.)	No Oces It Supplem	Status Oraft Final ent Previous Oirect	Date Date Date



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20469

101 2 5 1955

:

OFFICE OF SOLID WASTE AND EWEPGENCY RESPONSE

MEMORANDUM

SUBJECT: Notification of Restrictions on Reimbursement of Private Party Costs for Removal Actions

FROM: Henry L. Longest II, Director

TO: Superfund Branch Chiefs, Regions I-X Oil and Hazardous Materials Coordinators, Regions I-X

The purpose of this memorandum is to request that you ensure that communities are informed of the restrictive provisions of CERCLA regarding reimbursement of private party response costs in carrying out the NCP. Attached is the statucory and regulatory language governing private party reimbursement. The requirement for prior approval conserves the Fund and ensures that actions by others do not create further health or environmental threats.

In several communities, residents paid the costs for hocking-up their homes to the public water supply when local well water was found to be contaminated. Since this action was taken without prior EPA approval, the residents could not be reimbursed from the Fund, even though the actions taken were approved in the scope of work for that removal.

To avoid such situations in the future, when a removal action that will affect private residences is approved, the USC shall attempt to notify ail residents involved that expenses incurred by residents are incurred at their risk and excense, and are not reimbursable by the Federal government. When time is sufficient for consideration of preauthorization requests, the OSC should advise residents of CERCLA and NCP provisions regarding private party reimbursement. Such notification might well involve printed statements that only preauthorized actions by private parties are eligible for reimbursement. Further, the OSC should be cautious in making statements that can be construed by community members as a promise by EPA to reimburse them for cleanup costs. Most Superfund cleanup actions should be undertaken by the responsible party, by a State under a duly authorized Superfund contract or cooperative agreement, or by EPA contractors. Very few private party preauthorizations are anticipated, and those that are granted will occur under extraordinary circumstances. Should Regional resconse personnel receive a request for preauthorization from a private citizen to undertake removal actions, however, the request must be forwarded to Headquarters within five working days of receipt. It should be sent to:

> William O. Ross Hazardous Site Control Division (4H-542E) U.S. Environmental Protection Agency Washington, D.C. 20460

If you have any questions regarding this memorandum, please contact Jackie Dziuban of the Emergency Response Division at (202) 392-2452. Thank you for your cooperation in this matter.

ATTACHMENT

Section 111(a)(2) of CERCLA provides that money in the Fund may be used for:

"payment of any claim for necessary response costs incurred by any other person as a result of carrying out the national contingency plan established under section 311(c) of the Clean Mater Act and amended by section 105 of this title: <u>Provided</u>, <u>however</u>, That such costs must be approved under said plan and certified by the responsible Federal official." (Emphasis in original)

Section 300.25(d) of the National Contingency Plan provides:

"If any person other than the Federal government or a State or person operating under contract or cooperative agreement with the United States takes response action and intends to seek reimbursement from the Fund, such actions, to be in conformity with this Plan for purposes of section 111(a)(2) of CERCLA, <u>may only be reimpursed</u> if such person notifies the Administrator of EPA or his/her designee prior to taking such action and <u>receives prior approval to take such action</u>." (Emphasis supplied)

Analytical Support for Superfund

.

United States Environmental Protection Agen Waspington, DC 20460	CI	Interim Directive Number			
SEPA OSWER Directive Initiation R	lequest	0240 0 2			
Originator Information					
Main Coce ULLSUR	Telegnone Numpe	<u>'</u>			
Etan Koven with Top	1 00 TOTOTO 1 382 700				
GUST Signature of Cifice Oirector		Gate			
Analytical Support for Sup	ærfund				
Summary of Directive A review of alternative Superfund sample a ESD, REM, FIT, TAT, ERCS, AND ESAT).	nalysis resourc	es (CLP,			
General guidance regarding the use of the above.	analysis resour	ce given			
Request that each Region manage and monito resources via an integrated management and track provides for:	r the use of sa king systems wh	id ich			
 a site project manager accountable for spec an interface with the lab services of the E a planning and scheduling function. maintenance of a data base which includes same, turnaround times, cost and QA. documented adherence to QA practices. 	ifying project of SD and the CLP. ample scurces, w	needs. where			
Typs of Sirective (Manual, Policy Directive, Announcement, etc.)	Status Oraft X Final	New Remsion			
Coes this Directive Supersade Previous Directive(s)? [] Yes [X] No Does it Supplement Previous Directive(s)? [] Yes [A No If "Yes" to Either Cuestion, What Directive (number, trile)					
Bannan Olan					
AA-ÖSWER OUST CECM Other (Specify) CESR OWPE CGC CSV Regions CPPE	-*				
· Aecuest Meets CSWER Directives System Format					
ture of Load Office Quectives Officer	0.	3/20/86			
Signature of OSWER Directives Officer	0	lie			
524 Form 1315-17 (10-85)					



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

WASHINGTON, D.C. 20460

NAR 20 585

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

SUBJECT: Analytical Support for Superfund

FROM: Henry L. Longest II, Director Office of Emergency and Remedia, Kemponse

TO: Waste Management Division Directors Regions I - X

> Environmental Services Division Directors Regions I, VI, and VII

The purposes of this memorandum are:

- to review the alternative Superfund sample analysis resources that are available to you;
- to provide some general guidance regarding the use of these resources; and
- to request that each Region manage and monitor the use of these resources.

The two principal sources of Superfund program analytical support have been the Regional laboratories and the Contract Laboratory Program (CL?). Additional contractor sources are Remedial (REM), Field Investigation Teams (FIT), Technical Assistance Teams (TAT) or Emergency Response Cleanup Services (ERCS) and their subcontractors, and the Environmental Services Assistance Teams (ESAT) Program. ESAT is currently being developed to supplement the Regional laboratory staffing levels with contractor employees. We expect to have the ESAT contracts awarded and operational before the end of the first quarter in FY '87.

As a general rule, the Regional labs should be used to analyze samples where responsiveness and flexibility are paramount requirements. Analyses requiring quick response, methodology fine-tuning, and close interaction between the analyst and the data user are best done in Regional labs. In FY '86, we are using a total of 109 EPA work years to analyze samples, review data, and manage sample workload in support of Superfund activities. In FY '87, we will supplement EPA resources with 100 ESAT work years. The CLP is the laboratory service that should be used for routine high volume sample analysis requiring consistency of methodology, 30 to 40 day turn-around times, and data of known and documented quality. Faster turn-around times can be obtained by using the Special Analytical Services (SAS) option of the CLP. SAS can also be used to analyze unusual matrices with non-standard methodology. In addition, SAS has frequently been used to enhance routine CLP analyses by incorporating a desired method or parameter change consistently across an analytical case. In FY '85, the CLP analyzed 70,400 samples at a total analysis cost of \$32,998,143. The CLP will be able to analyze over 80,000 samples in FY '87. It should be viewed as your primary resource for the above described services.

The analytical resources available from the remedial and removal contractors include both fixed laboratory support, and mobile laboratories and portable instruments for use in the field. The amount of support available from these sources is limited when compared to the CLP and must be effectively managed. Attached are descriptions of emergency response removal, remedial and enforcement program needs and some suggestions on how the remedial/removal contractor resources can be used to supplement the CLP and ESD activities.

The choice of which analytical service to use should be driven by the data requirements of each program activity. The Regional laboratories and the CLP have well established and recognized Quality Assurance programs. However, whenever they cannot meet program requirements, you may use the other contracting modes at your disposal as long as you assure that basic requirements are met. In particular, you should be sensitive to costs, to clear definition of work, to enforcement needs, and to quality assurance requirements.

Careful management of analytical support services is important because of the large commitment of both FTE and dollar resources to this effort. The need for good Regional management and coordination was emphasized in the May 17, 1985 report of the Superfund Laboratory Services Management Review Group established by the Administrator. Although several Regions have already taken important initiatives in this area, I want to emphasize the importance of including all analytical services in this management system. Specifically, I believe that it is essential that each Region have an integrated management and tracking system that meets the following requirements as a minimum:

 Site project manager accountability for specifying project needs and acquiring appropriate analytical services for the project.

OSWER Directive No. 9240.0-2

- 2 -

- A point of interface with the laboratory services of the ESD and CLP that can inform the project manager of the availability of these services and schedule sample analysis.
- 3. A planning and scheduling function closely linked to the overall site planning process and that will provide accurate projections of analytical needs and close week-to-week program contact on scheduled and actual sample shipments.
- 4. Maintenance of a data base to monitor costs and schedules for remedial, removal, and enforcement samples. At a minimum these data should include:
 - records on where samples were sent for analysis, i.e., ESD lab, CLP, or other (e.g., remedial or removal) contractor lab; this should be keyed to the source of the samples.
 - data on turn-around times, cost, and OA requirements for samples not analyzed through the CLP.
- Documented adherence to appropriate quality assurance practices and procedures.

I believe it should be each Region's choice as to what organizational units should provide these integrated management functions. My concern is simply that the functions exist in each Region and that they have the ability to both manage the process and provide useful information to Regional and Headquarters managers.

Attachment

cc: Environmental Services Division Directors Regions II, III, IV, V, VIII, IX, and X

> Carol Finch Office of Regional Operations

Gene Lucero Office of Waste Programs and Enforcement

- 3 -

REMEDIAL PROGRAM ANALYTICAL REGUIREMENTS

The primary requirement for analytical support under the remedial propram is the RI/FS. Specific requirements include a variety of analytical techniques and protocols tailored towards site specific requirements for quantity, quality, timeliness and cost. Four key phases of analytical support have been identified within the RI/FS process:

<u>PHASE 1- INITIAL CHARACTERIZATION-</u> The first phase is the up-front field analysis to characterize the problems at the site and the probable extent of contamination. The key requirements of this phase are the ability to take a fairly large number of samples and perform the analyses quickly and inexpensively. This may be accomplished through portable field instruments and mobile laboratories operated by the remedial contractors. The information provided by this process is used to design the second phase of sampling.

<u>PHASE 2- DETAILED SITE SAMPLING--</u> The second phase entails a more focused sampling to accurately define the extent of contamination. This information is used during the feasibility study to support the development and evaluation of alternative remedial actions, and during the ROD process to support key decisions on the appropriate extent of remedy and selection of the cost effective remedy. Therefore, QA/QC requirements are essential. The primary analytical resource for this phase is the CLP, with judicious use of ESD or remedial contractor laboratories to meet special program requirements.

<u>PHASE 3- FILLING IN CATA GAPS-</u> The third phase is best characterized as "filling in" the data gaps identified during the feasibility study. These are normally highly focused samples designed to answer particular questions regarding the analysis of alternatives or determining the extent of remedy. Quick turn-around is a key concern for these analyses to avoid delaying the completion of the project; however, data quality must be commensurate with the intended use of the data in the decision-making process. Sources available for these analyses include the remedial contractor in-house laboratories, special analytical services uncer the CLP, and the ESD lab.

<u>PHASE 4-</u> <u>PPOJECT ENGINEEPING-</u> The fourth phase involves special analyses to support bench and pilot scale testing and treatability studies to assess the technical performance of a particular technology or to provide engineering data for remedial design. Normally, this work is done by the remedial contractors.

Two efforts are currently underway which should better define these requirements. The first is the development of Data Quality Objectives (DQOs) for RI/FSs. The DGOs will define on a site specific basis data quality requirements for the various decisions that are required during the RI/FS. The second is the RI/FS pilot program which is intended to improve the quality and timeliness of the RI/FS through closer integration and phasing of data collection activities with the feasibility study components of remedial alternative development, screening and evaluation. These efforts will provide more detailed criterie on the various sources of analytical support, QA/QC requirements, end the best way of integrating these into the RI/FS process. The emergency response program requires analytical services throughout the response process. Phases of an emergency or removal action that may require analytical support include: 1) preliminary assessment to determine whether an emergency response is appropriate; 2) site investigation, including verification of the identity, concentration, and/or location of hazardous substances, including data to support selection of the appropriate removal action; 3) proper disposal of contaminated materials; and 4) verification that the removal action was effective, including site samples to determine full cleanup and long-term monitoring. Each of these activities requires that the Region specify the analytical needs consistent with the incident to be addressed. Considerations include analytical services, turnaround time, quality assurance (QA), and cost (see table below). These considerations can vary by the phase of the response, and whether the incident is a clear-cut, time-critical emergency or a removal action where timeliness is needed, but not critical.

Generally, either TAT or ERCS obtains analytical services with the approval of the OSC. Use of CLP Routine Analytical Services (RAS) is generally inappropriate if turnaround times of less than 30 days are required. However, CLP Special Analytical Services (SAS) may be able to provide turnaround times of 14 days or less if special requests are submitted. Support from Regional or State laboratories also should be considered. Regional laboratories may be able to provide cost-effective, rapid analyses through the ESAT program. When the CLP or Regional laboratory mechanisms are not utilized, TAT obtains laboratory support through TAT Special Projects. ERCS contractors often rely on OSC or TAT recommendations, or in some cases, access private laboratories based on pre-existing verbal arrangements.

	Preliminary Assessment For	II Site Invortigation	III Disposal	IV Verification/ •
Time	High, particularly if acute threat of human exposure to high-hazard sub- stances.	Medium. Reason- able turnaround time needed for removal decision process.	Medium. Reasonable turnaround time needed to proceed with removal action and disposal.	Medium/Low. Verification of cleanup during and after re- moval requires better turn- around time than does long- term monitoriny.
QA	Medium. Need usable data, but not high QA at expense of timeli- ness.	High. Need ver- ifiable data to make removal decisions, doc- ument for cost recovery.	Medium. Data quality does not need to be high for waste compatibility and disposal.	Medium. Need usable data, but not high QA at expense of cost con- siderations.
Cost Oswer dire	Low. Uther considerations are more important in a suspected emergency. ctive No. 9240.0-2	Medium. Stan- dard analyses and medium time frame allow more cost considera- tion.	High. Stan- dard analyses and medium time frame allow more cost consid- eration.	High. Stan- dard analysis and longer time frame allow more cost consid- eration.

CONSIDERATIONS FOR REMOVAL PROGRAM ANALYTICAL NEEDS (Importance of Time, QA, and Cost)

٠

ENFORCEMENT ANALYTICAL REQUIREMENTS

The enforcement analytical needs fall into two categories: 1) RI/FS and 2) case support. The majority of the analytical needs for enforcement are for RI/FS on enforcement lead sites. Since these studies are being done by the same contractors who conduct RI/FS for Fund lead sites, the remedial program analytical requirements also apply to these studies. For the second category, civil case support, analytical needs can arise long after the RI/FS is completed and limited data are needed to support a file case. In this instance the preference is to utilize the Regional and other EPA laboraories. However, each Region should decide on a case-by-case basis how to manage its resources, and judicious use of CLP and other contractor analytical services may be necessary. For those sites where samples are taken for criminal case support, only EPA laboratories should be used.

Guidance on Producing Superfund Guidance Documents

GUIDELINES FOR PRODUCING SUPERFUND DOCUMENTS

I. <u>ISSUE STATEMENT</u>: The purpose of these guidelines is to provide writers of Superfund documents with assistance in writing guidance documents that are more usable and accessible. Guidance documents which are well-written can be expected to result in greater program effectiveness.

II. EXECUTIVE SUMMARY

These guidelines address the need to improve Superfund guidance documents for the benefit of users. They are designed to assist writers of guidance to produce documents that are readable, concise, well-referenced, and to the point.

Quality in guidance is necessary because a significant number of guidance users are decentralized, receive quantities of information demanding their attention, and engage in a variety of tasks which require immediate access to information.

This document also addresses issues of availability, cross-referencing, indexing, and follow-up contacts. Writing techniques are suggested that can result in streamlined documents written in clear English, and that provide an appropriate level of detail. Formatting suggestions are made to facilitate condensation for use in field manuals or electronic indexing or filing.

III. INTRODUCTION AND BACKGROUND

Reauthorized Superfund will generate a considerable volume of additional guidance. Many Superfund guidance users are decentralized in Regional, State, or field offices and must address complex issues which require immediate answers under field conditions. They need documents organized for easy, quick access with emphasis on readability and conciseness. The reader should be able to ascertain where additional written information can be found, and which individuals within the Agency can provide additional information.

These problems can in part be addressed by the way in which the guidance is written. These guidelines suggest certain techniques which can assist the writer in obtaining the desired result. They also discuss cross-referencing, instructions for entry in the OSWER Directives System, and contacts for securing additional information. The end result should be to assist the writer in preparing a more usable document.

A. Purpose of Guidance

Superfund guidance is primarily intended to help users implement the program effectively and consistently. Individual documents may be designed to serve one or more of the following purposes:

*To provide information and direction regarding procedures, policies, and technical developments which assist program personnel in conducting daily business uniformly and consistently throughout the country. This establishes a pattern for solving specific types of problems.

•In lieu of regulation. Where regulations are silent, and formal policy documents are inappropriate, guidance documents can provide general direction to assist the user in solving particular types of problems.

*As a response to requests for assistance from persons implementing the program.

In Superfund, the actual writing of guidance usually occurs in Headquarters program offices, developed by workgroups which represent a crosssection of the users. However, it can originate from top-down or bottom-up, depending upon who perceives the need for the guidance and who initiates the process. Much of the Superfund Guidance has been developed through a bottom-up procedure, with significant contributions coming from Regions, States, and other user groups.

Guidance should be used for the same variety of reasons that it is written: to assure the consistent, effective implementation of the program. While guidance is intended to be flexible, e.g., to be applied to a situation consistent with best professional judgement, it can also be used to strengthen the manager's position in handling unfamiliar situations.

B. Types of Guidance

Currently, four types of non-regulatory Superfund documents are produced and generically referred to as "guidance":

- Guidance Documents they white bith grey hundre type
 Procedural Documents white bigs green burles + Type
 Technical Documents white wish Superfund Red burles Type
 Manuals Green with burles burles + Type

Each serves a different function, as discussed below. Individual documents may stand alone or make reference to others; taken together, they form a body of information which establishes pattern and practice as the program matures.

* Guidance Documents explain what can be done to fulfill the requirements of a regulation or policy.

Generally, they cover a subject broadly and even comprehensively, but

should not spell out specific steps of an approach. Instead, they explain ideas which might be considered rather than how a required analysis or test might be conducted; or they may describe an analytic process, but do not discuss the mechanics of carrying out the process.

EXAMPLE: Guidance on Remedial Investigations and Feasibility Studies Under CERCLA; Relationship of the Removal and Remedial Program Under the Revised NCP

• <u>Procedural Documents</u> describe specifically how to conduct an analysis or activity; they provide a sufficient level of detail to complete an action without need for additional guidelines. They often provide a step-by-step procedure for conducting an action, including how to fill out forms where appropriate. They may also deal with the concrete mechanics or methods of an operation.

EXAMPLE: ERCS Contracts Users' Manual; Quality Assurance/Field . Operations Methods Manual

• <u>Technical Documents</u> provide scientific or engineering information relevant to program activities. They act as reference documents for use in complying with requirements, but do not explicitly relate the technical information to requirements. They are often products of scientific research or development conducted for program support.

EXAMPLE: Remedial Action at Waste Disposal Sites; Mobile Treatment Technologies for Superfund Wastes

• <u>Manuals</u> combine features of all of the above categories, particularly guidance and procedural documents. They enable users to carry out the program activities or requirements discussed without need for other supporting program documents.

EXAMPLE: Public Health Evaluation Manual; Compliance Manual

IV. GUIDELINES FOR WRITING SUPERFUND DOCUMENTS

A. Keeping the User In Mind

Users of guidance represent a cross-section of individuals having a wide variety of responsibilities. Guidance is used in Headquarters to define the program and to measure the program's effect when assessing results. It is used by Regional and State administrative offices to provide direction to office and field staff and to contractors and PRPs. Some of those in the latter categories may have little or no knowledge of the inner workings of EPA or the Superfund program. Thus, the guidance, combined with direction or verbal information obtained from Agency personnel, may constitute the basis of their experience with the Agency.

In addition, field personnel who are away from their offices for long periods of time, with limited telephone access and limited storage and transporting capabilities, require documents that are accessible as well as brief, transportable, and concise. Insofar as possible, when creating a guidance document, writers should consider the audiences. The bottom line is that if the guidance is not (1) accessible, (2) readable (3) brief and to the point (4) clear and (5) transportable, it probably will not be read---and if it's not read, it will not be used.

B. Structuring a Guidance Document

Field and program personnel who work with a wide spectrum of problems need documents that are easily and quickly consulted to provide immediate information. They do not have time to search voluminous material to secure an answer to what appears to them to be a straightforward question. A consistent format saves many hours of research time for the user.

The following proposed structure allows the reader to become quickly oriented to the issues at hand, then proceed to in-depth material which can be absorbed as time allows.

• <u>Issue Statement</u> - The document should begin with a clear, concise statement of the issue to be discussed.

• <u>Executive Summary</u> - This section serves two purposes, and may also be serving several different audiences. First, it should enable the reader to quickly determine if the document is pertinent. It sketches the most important points, and indicates where in the document more in-depth information is contained. Secondly, it can convey a basic message to the policy manager who may never read the entire document. In many documents, it also acts as a briefing for upper level managers regarding the major policy issues embodied in the larger document.

• Introduction and Background - This section introduces the material to be presented, may provide historical reference, cites pertinent regulations, statutes or other existing documents and sources of related information. It can also provide lead-in for the prescriptive material which follows in the next segment.

 <u>Prescriptive Material</u>. This segment provides any prescriptive material applicable to the matter at hand, flagged or underlined. (Prescriptive material is defined as that which is mandated or required by law or regulation.)

• Expository Information. The narrative that constitutes the actual guidance is placed here, together with case studies, examples, and other appropriate illustrative material. This provides the next level of detail for the reader who has concluded that this document addresses the needs of the situation and now wishes in-depth information.

• Referencing Section. This segment includes cross-referencing to other documents, identified by their OSWER identification number, if available; or by their date and source where no OSWER number has been assigned. In addition, it should include the name of a contact with phone number for obtaining further information. Larger, more comprehensive documents might include glossaries and indexes.

C. Language and Streamlining Techniques

The choice of <u>language</u> affects the user's ability and/or willingness to implement the guidance. Because "guidance," by definition, is not prescriptive unless it cites specific regulatory or statutory requirements. the language style should be flexible rather than rigid. The most common illustration of flexibility is the use of "may," which is flexible, rather than "must" or "shall," which connote inflexibility.

When guidance contains prescriptive material, it should be referenced back to the appropriate regulatory or statutory citation. It is then advisable to separate or set off the prescriptive material by means of starring, flagging, underlining, or paragraphing so that it stands out.

<u>Streamlining</u> means coming immediately to the point or heart of the matter with a minimum of verbiage. It results from visualizing the finished product and producing a final document that mirrors this concept.

Below are a series of suggestions which can assist the writer in producing readable, usable guidance.

• <u>Outline the document</u>, and write from the outline. Outlines need not be formal; they can take the form of concise bullets that become the organizing points for the document, and can even be developed into topic sentences. Picture the user and the constraints under which that individual operates. Visualize the final document and its intended purpose for the program and for the user.

* Establish a style consistent with the purpose of the document. Remain with that style until the document is completed. While style can be individual, it is also true that each of the four document types has its own style, and that style has an effect on the reader.

For example, <u>technical documents</u> which inform the user of new technology or new information tend to be most effective when the style is concise and direct.

<u>Procedural documents</u> often set forth specific steps, tasks, or operational steps which the reader is to follow. These documents are sometimes the least flexible of the four types. The style should be simple and straightforward, with complete, direct sentences. Emphasis should be on an orderly, clear presentation.

<u>Guidance documents</u> frequently offer an array of options and case studies, to which the user may apply best professional judgment in adapting to the situation at hand, within the broad parameters of the guidance. The writing style can be somewhat more flexible and expository.

<u>Manuals</u>, in that they stand alone in implementing part of a program, frequently combine elements of the other documents. They may be the most lengthy of the four, and may include technical information with specific procedures and an array of options. Style will therefore emerge as a composite of the others.

In selecting the appropriate style, it is useful to remember that nouns and action verbs written in the active voice supply greater power and emphasis than writing with adjectives and adverbs and in passive voice. Further, impersonal language increases the rigidity or inflexibility of the document. The language used in a regulation or directive is more impersonal than that used in a simple memorandum.

For example:

<u>Rigid:</u> "To comply with this provision, the owner/operator must determine whether a bulk hazardous waste is a liquid or contains free liquid. EPA regulations define "free liquids" as "liquids which readily separate from the solid portion of a waste under ambient temperature or pressure." (40 CFR 260.10)...EPA believes that the Paint Filter Liquids Test is the appropriate test to be used to determine the absence or presence of free liquids in bulk and containerized waste."

(Note the prescriptive material quoted and referenced)

<u>Flexible:</u> "This step is designed to facilitate responses to clearcut, time-critical emergencies for which only limited data are available. In those cases, OSCs <u>may have to rely</u> primarily on findings of the preliminary assessment, without significant additional data collection....Where the OSC has determined that the incident is not time-critical, a more thorough analysis...<u>will</u> be possible before recommending a...response."

• Limit explanation. Select simple words and omit needless words. The reader has limited reading time. "Less is more" is often a useful concept in government writing.

For example:

Excessive Detail: "Implementing the notification program will require both staff time and the direct expenditure of funds. You will find guidelines on typical costs of various activities throughout this handbook. The available funding and staffing level will have an impact, for example, on how extensive the public education program can be and which methods of form distribution can be used. State budget allocations will also impact the types of activities possible. Some states will find that they have adequate staffing but little money to pay for printing and travel, while other States may experience the opposite."

<u>Concise:</u> The scope of the rulemaking will be limited to municipal landfills because, currently, reliable data on which to construct and defend sound Criteria exist only for municipal waste landfills. Moreover, by limiting the rule, the Agency can expect to promulgate the revisions within the Statutory deadline of March 31, 1988. <u>Avoid unexplained acronyms, unexplained technical terms, and program-</u> <u>specific language not part of the general experience of the reader.</u> The standard Agency practice of writing a phrase in full with the acronym in parentheses at the beginning of the article is helpful. Even so, assumptions that the reader has the same working knowledge of the subject as the writer can be unwarranted. Many readers find themselves intensely irritated at government writing which requires the constant translation of acronyms and jargon. In extensive documents requiring acronyms, inclusion of a glossary can be helpful.

For example:

- (1) "You should begin exploring mechanisms to implement CA's with OFA Regional counterparts early in the FY about sites in Regions targeted for CERCLA action. This should be accomplished by review of the NPL, the SMP, and the SCAP. The SCAP commitments should be reflected, as appropriate, in the SEA's for your Region."
- (2) "You should begin exploring mechanisms to implement cooperative arrangements with Office of Federal Activities (OFA) Regional counterparts early in the fiscal year. This should be accomplished by review of the National Priorities List (NPL), the Site Management Plan (SMP), and the Superfund Comprehensive Accomplishments Plan, (SCAP). The SCAP commitments should be reflected, as appropriate, in the State/EPA Agreements for your Region."

<u>Summarize when appropriate</u>. Use of summaries depends upon the degree of complexity of the material being presented. When the material is lengthy and complex, it may be useful to summarize at the end of sections or chapters, or it may be useful to present a final summary at the end of the document. Another effective technique is the use of bullet points at the beginning of sections to highlight main points. In addition, careful construction of the Executive Summary assists both writer and reader in keeping to the point.

"When revising and rewriting, look for redundancy, strive for clarity, and reduce detail when possible. Then recheck the editing to be sure the meaning hean't been lost. Short, complete sentences placed in logical order with logical paragraphing simplify the work of the reader. All paragraphs should have topic sentences at the beginning or the end. The final editing effort is the comparison of the finished product with the writer's initial concept and the mental picture of the user. How well do they match? Will the reader be able and willing to understand the writer's message? Only when the match is achieved, is it time for others to review the document.

D. Content.

Appropriate Level of Detail. There are two considerations with re-

to the document. As a rule, procedural documents require the most detail and specificity, technical documents somewhat less, and guidance documents least of the three. Manuals will be a composite and should have a level of detail appropriate to the subject.

Secondly, consideration should be given to the amount of detail needed by the various users. While the material may be applicable to a variety of situations, some users may need less detail. Where more detail is required, it should be presented and organized so that it does not distract such users.

It is the writer's job to determine the appropriate level of detail in planning and outlining the document.

<u>Technically Correct.</u> The writer is responsible for presenting a final document that is technically correct and to conduct whatever reviews and checks are necessary to assure that the document is in fact correct.

<u>Prescriptive Only When Required</u>. Guidance should be assumed to be "guidance" unless otherwise stated. "Best professional judgment" should be assumed unless the document specifically flags prescriptive (mandatory) material. When these flags are present, statutory or regulatory references should be cited, as well as appropriate back-up policy or precedential material. Otherwise, flexibility should be assumed and language should be constructed accordingly.

Filing, Indexing, and Follow-Up Information have been discussed in Section B. However, each document should end with a section that provides assistance in obtaining additional data, and in filing and cross-referencing.

E. Classification.

Documents have been classified for general use in the Superfund Program. The four classification types previously discussed are designed to be broad and flexible, while giving users a clear idea of the kind of information presented in each. As the system evolves, it is anticipated that each kind of document will be identified by a characteristic cover which makes it easily identifiable.

In addition to the Superfund classification system, each document will, of course, be assigned an OSWER Directives Number and be included in that indexing system. Beyond that, institutional users (Regional or State offices, others) may wish to develop their own internal filing or reference system. The classification system, plus the Issue Statement and Executive Summary at the beginning of each document, will facilitate the design of such a system.

(F) Considerations in Publishing a Document

Once a document text has been prepared and approved for publication, a variety of publication requirements and procedures are triggered. Since these are important in assuring that all agency requirements are met, they, will be the subject of a subsequent procedural guidance. Coordination of publication procedures is the responsibility of the Office of Program Management, OERR.

FOLLOW-UP INFORMATION OR CONTACTS

- For general writing information THE ELEMENTS OF STYLE, by William Strunk, Jr. and E.B. White, Third Edition, MacMillan Publishing Co., Inc., New York 1979.
- (2) For information about OSWER Directives System Contact Sherry Fielding, OSWER Directives Coordinator, Immediate Office, Assistant Administrator for Solid Waste and Emergency Planning - FTS 382-4483
- (3) For information about the Guidance Review Project or OERR Guidance, contact Arthur B. Weissman, Policy Analysis Staff, Superfund Program, FTS 382-2182.

Comparison of EPA and Commonwealth of Kentucky PCB Cleanup Standards

COMPARISON OF EPA AND COMMONWEALTH OF KENTUCKY PCB CLEANUP STANDARDS

The following paragraphs briefly summarize EPA and Commonwealth of Kentucky statutes applicable to PCB cleanup standards. In the case of the EPA regulations, some of the contaminant levels are regulatory standards, while others are simply recommended levels. EPA standards will soon change, as described below. The Kentucky cleanup standards for PCBs are based on general language in the Kentucky Hazardous Substances Statute as described below.

EPA

EPA has established various PCB cleanup standards or recommendations under the Toxic Substances Control Act (TSCA), the Clean Water Act (CWA), the Safe Drinking Water Act (SDWA), and the Comprehensive Environmental Response, Cleanup and Liability Act (CERCLA). While TSCA currently has the major role in PCB regulation, the EPA has announced its intent to further regulate PCB wastes under the Resource Conservation and Recovery Act (RCRA).

TSCA

Under Section 17 of TSCA, EPA considers spills, leaks, or other uncontrolled discharges of PBCs in concentrations equal to or greater than 50 ppm to be improper disposal of PCBs. TSCA Section 17 provides EPA with the authority to compel responsible parties to clean up spills at such levels of contamination.

A recent "TSCA PCB Cleanup Policy" (signed 3/20/87 and effective thirty days after publication in the Federal Register) will establish PCB cleanup standards for the majority of PCB spill situations. Certain spill situations described in that policy will not, however, be subject to these standards, based on site-specific conditions and the discretion of EPA Regional offices.

Spills, leaks or other discharges occurring before this policy becomes effective have been or will be cleaned up to levels ranging between background and 50 ppm depending on Regional discretion and site characteristics.

CWA

Under the CWA, EPA set contaminant levels, based on toxicity and risk assessments, for adoption under state programs. These are not regulatory levels by definiton, however, they will function as such under state authority. The states must set levels that are at least as strict or more strict. The CWA recommended maximum levels are 2 ug/l (24-hour average) for freshwater, and 0.03 ug/l (24-hour average) for saltwater based on acute toxicity to aquatic life; and 0.79 ng/l (based on 10^{-5} lifetime cancer risk), 0.079 ng/l (based on 10^{-6} lifetime cancer risk).

SDWA

. Under the SDWA, EPA has suggested levels of maximum allowable contamination. These health-based levels of maximum PC3 contamination include the 10-day health advisory levels of 100 ug/day (cnildren) and 700 ug/day (adult); and for Aroclor, 100 ug/l (child) and 350 ug/l (adult). The Agency will recommend a cleanup level of 0.005 ug/l for PCBs in drinking water based on a 10⁻⁵ lifetime cancer risk.

CERCLA

CERCLA recommended cleanup levels are based on the consideration of soil/air partition coefficients and the potential for ingestion, and dermal and inhalation exposure. There is currently insufficient data to develop advisory levels for one-day and long-term, non-cancer effects. The non-cancer 10-day advisory level is 42 ug/g on site, and 47 ug/g, based on inhalation exposure, at the perimeter of the site. The chronic intake advisory, 10^{-6} cancer risk (upper bound estimate) is 0.6 ug/g without cover, or 6 ug/g with 25 cm of clean soil cover (i.e., PCB < 0.1 ppm), on site; and 2 ug/g without cover, or 20 ug/g with 25 cm clean soil cover, at the site perimeter.

Site-specific factors, ARARs, and Regional discretion each play a role in CERCLA final cleanup levels. ARARs include TSCA, CWA, SDWA and the guidelines, standards and recommendations of the Food and Drug Administration, the American Conference of Governmental Industrial Hygienists, the National Institute of Occupational Safety and Health, the National Academy of Science, and the Occupational Safety and Health Administration. Current RCRA regulations pertain to PCB disposal requirements.

The Regions have discretionary authority to set more stringent cleanup standards on a site-by-site basis. Region IV, which includes Kentucky, adheres to TSCA cleanup standards.

Kentucky

Kentucky cleans up hazardous substances to background levels wherever practicable. This standard is based on Kentucky Revised Statutes (KRS) 224.877. "Definitions-Regulations-Notification-Minimization-Emergency plan-Powers of department," paragraph (4) as follows:

"(4) Persons having possession of or control over a hazardous substance being discharged or who caused the discharge shall take the actions necessary to restore the environment to the extent practicable and minimize the harmful effects from any discharge into the air, lands or waters of the Commonwealth."

As described by Alex Barber, Kentucky Department of Environmental Protection Division of Waste Management, Kentucky cleans up PCBs to background levels unless it is not practicable to do so. Such a situation may arise when soil contaminated above background is removed down to bedrock. Generally cleanup to background levels will result in lower cleanup levels than EPA's 50 ppm action level. Kentucky may allow contamination up to 10 ppm, if the site is covered with a state-approved cap.

PCB contamination in water is regulated by 401 Kentucky Administrative 'Regulations 5:031 Section 4. The state standards are based on CWA Section 304 criteria. The state standard for ambient warm water is .0014 ppb. For finished water the standard is .0013 ppb, based on SDWA.

Summary--

EPA's specific standards and recommendations provide for protection of human health and the environment at PCB cleanups particulary as ARARs are incorporated in CERCLA cleanups. The Kentucky cleanup to background may, in some cases, exceed EPA cleanup standards. Kentucky water standards for PCB contamination are based on the federal CWA and SDWA.

Concurrence on Payment of Relocation Costs for Business During Removal


UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON D C. 20460

JUN 1 5 1987

MEMORANDUM

OFFICE OF GENERAL COUNSEL

SUBJECT: Concurrence on Payment of Relocation Costs for Business During Removal

FROM: Henry L. Longest II, Director Office of Emergency and Remedia esponse J. Daniel Berry 71 10 - 1

Associate General Counsel Grants, Contracts, and General Law Division

TO: Stephen Luftig, Director Emergency and Remedial Response Division Region II

> Douglas B. Blazey, Regional Counsel Region II

You have asked for our concurrence in the use of the Hazardous Substance Superfund (the Superfund) to pay costs associated with the relocation of a business during the course of a removal action carried out under the Comprehensive Environmental Response, Compensation and Liability Act, as amended (CERCLA). Given the equities in this particular case, it is our view that it is appropriate to reimburse the business for temporary relocation costs. Any costs related to business losses are, of course, not reimbursable from the Superfund.

As we understand the facts, EPA took a removal action in a building at the Signo/Mt. Vernon, New York site in which RPM, a laminating company, conducted business. EPA advised RPM that it could not safely operate its business while the removal team was working. Subsequently, the Occupational Safety and Health Administration posted a notice of Alleged Imminent Hazard in the building and the county health department issued a Notice of Closure. The Agency for Toxic Substances and Disease Registry also recommended that RPM be excluded from the building because of the potential threat to its employees. RPM moved its operations to another location and is seeking reimbursement for its costs.

Section 101(23) of CERCLA defines "removal" to include such actions "as may be necessary to prevent, minimize, or mitigate damage to the public health or welfare or the environment, which may otherwise result from a release or threat of release. The term includes, in addition, without being limited to . . . temporary evacuation and housing of threatened individuals not otherwise provided for . . . You have suggested that this authority is sufficiently broad to include temporary relocation of businesses as well as residents. We agree that the statute can be read to support such an interpretation.

We intend to address the temporary relocation of businesses during a removal action under the relocation functions delegated to the Federal Emergency Management Agency (FEMA). We are currently working with FEMA to determine the appropriate reimbursement for RPM, and to develop guidance and regulations for any future temporary business relocation.

We understand that FEMA's regional office will be working directly with you to develop the information necessary for RPM's reimbursement. If there are any questions on this matter, please contact Tim Fields or Lisa Guarneiri at FTS 475-8110.

cc: Walt Kovalick Timothy Fields Hans Crump Linda Garczynski Mark Mjoness Lisa Guarneiri Michael Hirsch, FEMA Dennis W. Kwiatkowski, FEMA Charles Robinson, FEMA Fred kubel, Region II Charles Fitzsimmons, Region II

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY REGION II

DATE: MAR 2 3 1987

SUBJECT: Attached Memorandum Recarding Relocation Costs of a Business Christopher J. paggett FROM: Regional Administrator TO: Francis S. Blake and Winston J. Porter General Counsel Assistant Administrator Solid Waste and Emergency Response

> Enclosed is a copy of a memo that Doug Blazey and Steve Luftig have sent to their colleagues in your offices requesting assistance on a particularly novel question. I feel we need an answer quickly and wanted you to know that the principles in RPM have been in contact with Lee's and my office regarding their plight.

> I feel we should assist them and am sympathetic to many of the points they raise. It is likely that we will be sued if a financial settlement is not reached. I believe the general public expects Superfund to support some reimbursement for a displaced business assuming we are statutorily authorized to make such payments.

Your interest in this matter would be appreciated.

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

Relocation Costs of Business in Removal Action n C. A.ECT: Callah Stephen Lufzig, Director Douglas R. and Emergency and Remedial Regional Co FROM: Response Division TO: J. Daniel Berry, Acting Henry L. Longest II and Associate General Counsel Director, Office of Emergency

> This is to advise you that Region II would like to use Fund money to cover a business' relocation costs resulting from an EPA removal action. It is our understanding that the Agency has never done this before, therefore, we would like your concurrence on this action.

and Remedial Response

I. BACKGROUND

for Grants and General Law

RPM is a small family operated laminating company, which employs 22 people and is located in Mt. Vernon, Westchester County, New York. RPM rents a portion of the 2nd floor and all of the first floor of a building located at 200-208 S. 14th Avenue, Mt. Vernon, New York. EPA is undertaking a removal action on a portion of the second floor and the envire third floor of the same building. The removal action involves the removal of various shock-sensitive explosives, flammable ilquids, flammable solids, poisons, compustibles, oxidizers and acids.

EPA advised RPM that it could not operate its cusiness while EPA was physically conducting removal activities inside the building. EPA was concerned not only for RPM's safety during the physical removal activities in the building, but also for the removal team's safety while undertaking the clean-up operation. EPA further advised RPM that it had no objections it RPM worked during the evening when the removal team was not actually working in the Duilding.* Subsequent to this, the Occupational Safety and Health Administration ("OSHA") posted a notice of "Alleged Imminent Hazard" and the Westchester County Department of Health ("DOH") issued a Notice of Closure. The basis for DOH's Notice of Closure was the hazard RPM's presence might have on the general public. In addition, the Agency for Toxic Substances and Disease Registry ("ATSDR") recommended, initially in a January 1987 lecter and in at least two documents in February 1987, that RPM be excluded from the building due to the threat to RPM and the surrounding community. Unable to conduct its business in the building, RPM made arrangements to move some of its lighter equipment to a temporary location where the work that could be pertormed, i.e. cutting, was performed and the remaining asyscts of the work were subcontracted out.

^{*/}From the beginning, the OSC thought it would be appropriate to compensate RPM, if possible, for its losses and asked ORC to investigate this matter. As the removal action progressed, however, the issue crystillized into whether RPM'S relocation costs could be compensated.

After being excluded from the building for several weeks, RPM purchased an option to buy the building, at a cost of allegedly \$50,000, where it had temporarily located its cutting operation. Before RPM exercised its option on the new building, OSHA rescinded its Notice of Alleged Imminent Danger and the Westchester County Department of Health rescinded its Notice of Closure. Although ATSOR did not rescind its recommendation, RPM was permitted to re-enter the building.* Under an agreement reached with the On Scene Coordinator, RPM is permitted to work 1 day a weak and evenings, when EPA is not in the building working.

The Region feels that we can compensate RPM, using money from the Hazardous Substance Trust Fund, for its re-location costs to the temporary premises.**

RPM has requested payment for relocation expenses to the temporary location as well as expenses it will incur in a complete and permanent move to the "temporary" location. The issue of what costs are eligible for reimbursement is a distinct topic from the narrow question of whether we are authorized to pay any business relocation costs under CERCLA. As you may know, EPA has interagency agreement with the Federal Emergency Management Agency ("FEMA") to reimburse it for all relocation activities undertaken by FEMA pursuant to CERCLA response actions. FEMA has detailed accounting and eligibility requirements based at least, in part, upon the

*/Pursuant to RPM's concurrence, EPA is using a portion of RPM's leased premises as a storage area for some of the segregated chemicals.

**/Since R?M has already expended the money for the initial relocation and some of the rent, EPA would be reimbursing kPM for these expenses. We recognize that normally there needs to be EPA preauthorization (40 C.F.R. \$300.25) before EPA reimburses a person incurring response costs. We feel, however, that such a preauthorization is implied from the combination of the actions of EPA, OSHA, JOH and ATSDR. To require a more formal preauthorization (such as an Order or a written statement) would not be reasonable in circumstances like this. Specifically, rather than having parties voluntarily agree to do that which is requested by the federal and local governments in implementing a response action, parties will wait until they are required to co-operate, either through the judicial or administrative processes. All of this time and effort, in the long run, will slow down our removal actions. requirements of the Uniform Relocation Act. Whether all questions that arise as to the propriety of certain expenses are answered by FEMA regulations and policies is an open question. However, we see little efficiency in pursuing these issues until the threshold question of basic eligibility is resolved.* The Region has also been asked to consider RPM's relocation expenses to the new building as a response cost. The Region would welcome your opinion on this issue as well.

The pertinent dates of the events in this case are as follows:

December 8-17, 1986 - EPA conducted a preliminary assessment and inspection of the site. EPA advised RPM that the business can not be operated when the removal team was physically working inside the building.

December 17 - January 7, 1987 - RPM continued to work normal hours. (Appropriate for holiday season)

January 8, 1987 - EPA began removal activities inside the building. RPM advised by the OSC to stay out of the building because the ether was being sampled.

January 9, 1987 - Ether still being sampled. OSC advised RPM to stay out of the building.

January 9, 1987 - Agency for Toxic Substances and Disease Registry ("ATSDR"), based on a review of the videotape of the interior of the building and a partial list of the contents of the building, made an assessment that the building constituted an imminent fire and explosion hazard. ATSDR recommended that RPM be immediately excluded from the building.

*/For example, RPM contends that it was required to relocate because "permitted" night operations would not allow it to send out and receive shipments during the day.

Further, because the heavy equipment could not be easily dismantled and relocated in a temporary facility, RPM utilized subcontractors for certain operations because of its disrupted work place. Many financial issues arise from these "tangled" operations.

To further complicate matters, RPM is likely to permanently relocate to its "temporary" location and has alledgedly paid \$50,000 for an option to ouy the new facility. The potential relocation costs are, therefore, significantly higher because heavy equipment now located at the response site, must be dismantled, moved and re-aligned by experts. One of RPM's estimates of this move is \$150,000. Moreover, we understand that RPM's lease at the response site expires in Sectember 1987 and that a permanent relocation was contemplated by that time in any event. January 12 - 14, 1987 - RPM worked evenings when the removal team was not in the building. EPA hired a van service to transport the employees. EPA also accepted deliveries and made shipments for RPM during the day when RPM could not be in the building.

January 14, 1987 - Occupational Safety and Health Administration posted a notice of "Alleged Imminent Danger" (The notice is not a closure order. It serves as a warning to workers that there is an alleged dangerous condition in the workplace.)

January 15, 1987 - Dr. Anita Curran, Commissioner of the Westchester County Department of Health issued an Notice of Closure to RPM.

January 15, 1987 - RPM excluded from the building.

January 22 and 29, 1987 - Hearing held on the Notice of Closure. Agreement reached that RPM must stay out of the building until all of the explosives that have been identified as of January 26, 1987 are removed, provided, however, that the order would be reinstated if EPA disclosed the existence of additional explosives or materials that the Commissioner felt constituted an imminent hazard. Approximately 600 pounds of shock sensitive material were found on 1/27/87.

January 30, 1987 - ATSDR strongly recommended to the Westchester Department of Health that evacation of RPM be continued. (letter is attached)

February 3, 1987 - ATSDR recommended that RPM be excluded from the building until: The building is completely inventoried; all leaking containers are overpacked; all explosive or shock sensitive materials have been removed; incompatible substances have been segregated; a Fire Marshall is on site at all times; and EPA reviews and approves RPM's proposed activities in the building.

February 10, 1987 - RPM locates another building in which to temporarily operate.

February 17 - 27, 1987 - RPM began moving some of its equipment out of the building between the hours of 7 and 9 in the morning.

EPA goes in the building at 9. RPM continues to move material out each morning for subcontracting. They have set up shop at another temporary location. They do their own cutting, and then subcontract the work out.

February, 1987 - RPM purchases an option to buy another building from which to operate its business.

February 27, 1987 - Dr. Curran, after talking to the OSC, gave RPM permission to enter the building on February 28 and Marca 1 to manufacture some goods. EPA stopped its activities inside the building for those two days. March 10, 1937 - OSHA lifted its Notice of Alleged Imminent Danger. " Westchester County Department of Health lifted its Notice of Closure.

March 11, 1987 - RPM begins working 1 day and four nights in the building. The day work is conditioned on whether EPA is removing explosives from the building or sampling unknowns.

II. WHETHER RPM'S RELOCATION EXPENSES CONSTITUTE A RESPONSE COST IN THE CONTEXT OF A REMOVAL ACTION

Section 101(23) of the Comprehensive Environmental Response, Compensiton and Liability Act, as amended, 4240.S.C. §§ 9601 et seq. (hereinafter "CERCLA") defines a removal faction as:

"the clean up or removal of released hazardous substances from the environment, such actions as may be necessary taken in the event of a threat of release of hazardous substances into the environment, such actions as may be necessary to monitor, assess and evaluate the release or threat of release of hazardous substances, the disposal of removed material, or the taking of such other action as may be necessary to prevent minimize or mitigate damage to the public health, welfare or the environment, which may otherwise result from a release or threatened release ... (emphasis added)

EPA's initial basis for exluding RPM from the building was due to the fear that the removal team might do something that would cause a release and harm RPM as well as EPA. ATSDR and the Westchester County Department of Health, however, made recommendations of exclusion and excluded RPM from the building at all times because RFM might do something that would cause a release in the removal area and thereby threaten the public health and welfare. Indeed, RPM worked with various chemicals, including TCE and heat in its laminating business. There is no fire wall between the RPM premises and the premises where the removal action is taking place. The removal area contained, and to some extent still does contain, many shock sensitive, flammable and other dangerous chemicals*. Hence, excluding RPM from the building was an action necessary to "prevent, minimize of mitigate damage to the public health or welfare or to the environment, which may otherwise result from a release or threat of release".

> elives it has identified most of the shock sensitive is and has placed them either in magazines located on in a walk-in vault, also located on site.

Consequently, RPM's exclusion* from the building was a part of the removal action being undertaken at the site, for which RPM incurred, among other costs (i.e. business losses), relocation costs. The only issue remaining is whether those relocation costs be compensated from the fund.

II. WHETHER RPM'S RESPONSE COSTS ARE COMPENSABLE UNDER CERCLA

One of the many purposes of CERCLA is compensation of innocent victims of a release of hazardous substances or pollutants or contaminents which may affect the public health or welfare. In that regard, a person whose water supply is contaminated, through no fault of his own, is often provided bottled drinking water by EPA, paid for out of the fund. In many instances, businesses, such as restaurants, are provided bottled water by EPA. Hence, a legitimate use of the Fund is to compensate innocent victims of pollution.

Section 101(23), in defining (removal", is silent as to relocation of businesses without specifically excluding the possibility of compensation. Indeed, the closest the definition comes to addressing relocation at all is:

The term includes, in addition, without being limited to, ... temporary evacuation and housing of threatened individuals not otherwise provided for ...

Hence, a strong argument can be made that businesses can be compensated for their relocation costs.**

*/We collieve it is inconsequential that the County, rather than EPA, took the legal action to actually exclude RPM from the building. The National Contingency Plan, 40 C.F.R. §§ 300.22 and .24 requires the federal government to work together with state and local governments to respond expeditiously to emergency situations. To argue that the County's closure order was not intimately related to and a part of the total response action taking place at the site is inconsistant with the cooperative intent of CERCLA.

""/ Moreover, if the instant action was a remedial action, RPM would be compensated for its relocation expenses. Section 101(24) in defining "remedial" action provides in part:

"... The term includes the costs of permanent relocation of residents and business ... [which] the President determines that alone or in combination with other measures, ... may otherwise be necessary to protect the public health or welfare ...

It is clear that the definition of "removal" neither explicitly or implicitly prohibits compensation of a husiness for it. Felocation costs. The phrase the "taking of such other actions/ is way be necessary to prevent, minimize or mitigate damage to the public health or welfare or to the environment" 42 U.S.C. § 101(23), gives EPA very broad authority, to determine what actions are necessary in any given situation. Further, the list of actions that constitute removal actions contained in Section 101(23) is preceded by the term [without being limited to". Hence, just what constitutes a remeval action lies within the sound discretion of EPA.

Without making any broad assertions with respect to compansating businesses for the relocation costs in removal actions, the Region feels that the circumstances and equities of this case justity compensation. The only reason that RPM relocated from the site was because of EPA's removal action. Hence, any costs incurred by RPM should be considered a response cost and paid for by the fund.

IV. CONCLUSION

We have tried to set forth, as clearly as possible, the factual setting which underlies the conclusion that.RPM should be compensated for its relocation costs. We would appreciate your <u>concurrence with the Region's belief that RPM's relocation to its</u> temporary site, reasonable rent and the move back to the site, <u>constitute response costs which can be compensated from the rund</u>. We also invite your opinion on how to treat the relocation expenses that RPM may incur, should it decide, to exercise its option to buy the new building.*

Thank you for your cooperation on this matter.

cc: Timothy Fields William Ross

*/The Region is in the process of discussing with Region 7 the procedures it uses during the relocations it has been involved with and to find out whether it has either temporarily or permanently relocated businesses. We are also in the process of determining whether the FEMA has any regulations or procedures regarding which relocation costs can fairly be paid, as well as trying to determine ourselves, what costs constitute relocation costs and whether they are compensable, and if so, what portion is compensable.

Employee Occupational Health and Safety

United States Environmental Protection Agency Washington DC 20460	1 Grecove Number		
OSWER Directive Initiation Request	9285.3-02		
2 Originator information			
Name of Contact Person Mail Code Office	Telephore Code		
Rod Turpin EDIson NI ERD/ERT	340-6745		
3. Тше			
Employee Occupational Health & Safety			
4 Summary of Oirective (include one) statement of purpose;			
Provides instructions reminding EPA employees that they must c	omply with OSHA.		
5. Keywords Superfund, CERCLA, SARA			
64. Odes This Directive Supersede Previous Directive(s)? No . Yes What directive	s (number, tile)		
b. Does It Supplement Previous Directive(s)?			
Draft Level A - Signed by AA/DAA B - Signed by Office Director C - For Revew & Comment D - In Development			
8. Document to be distributed to States by Headquarters?			
This Request Mosts OSWER Directives System Format Standards.			
). Signature of Lead Office Directives Coordinator	Date		
Richard Hyde	7/7/87		
10. Name and Title of Approving Official	Date		
Win Porter, Assistant Administrator	7/7/87		
PA Form 1315-17 (Rev. 5-67) Previous editions are obsciete.			

OSWER OSWER OSWER O VE DIRECTIVE DIRECTIVE DIRECTIVE



JL 7 1987

OFFICE OF SOLID WASTE AND EMERGENCY RESPI OSWER Directive 9285.3-(

MEMORANDUM

SUBJECT: Employee Occupational Health and Safety FROM: A Winston Porter, Assistant Administrator Office of Solid Waste and Emergency Response

TO: Addressees

As we initiate our field activities under the Superfund Amendments and Reauthorization Act of 1986 (SARA), I wish to remind everyone that all EPA employees are required to comply with the Federal Occupational Health and Safety Act (OSHA). In addition to complying with such federal standards as 29 CFR 1910/1926, EPA employees must also adhere to the appropriate EPA orders, policies, and guidelines pertaining to employee occupational healt' and safety. For example, EPA Order 1440.3 requires all EPA employees using respiratory protection devices to participate in a medical monitoring program. I want to encourage all Regions and OSWER offices to continue to implement effective medical monitoring programs for its employees. EPA Order 1440.2 requires all EPA employees engaged in routine field activities to be trained and to receive training certification levels commensurate with the degree of anticipated hazards. EPA Order 1440 specifies the responsibilities of all Agency employees in this area.

More recently, section 126(a) of SARA requires the Secretary of Labor, within one year of the date of enactment, to promulgate standards (29 CFR 1910.120) for health and safety protection of employees engaged in hazardous waste operations. The Secretary is also required in Subsection 126(e) to issue interim final regulations within 60 days after enactment of SARA. The interim final rule was published in the <u>Federal</u> <u>Register</u> (Vol. 51, No. 244, pages 45654 - 45675) on December 19, 1986. This interim final rule took effect upon the date of issuance (December 19, 1986), and it was OSHA's judgment that all provisions could be fully implemented not later than 90 days after issuance (March 17, 1987). As with other OSHA Section 6 Standards, EPA is required to comply with 29 CFR 1910.120 per Executive Order 12196 (February 1980). In addition, section 126(f) of SARA requires the Administrator of EPA, within 90 days after the promulgation of final regulations under section 126(a), (January 17, 1988), to promulgate standards identical to 29 CFR 1910.120 (those promulgated by the Secretary of Labor under section 126(a)). EPA's Workgroup No. 2427 (Hazardous Waste Operations and Emergency Response Worker Protection Standards), chaired by Rod Turpin, Safety and Occupational Health Manager, Environmental Response Team (ERT), Edison, NJ, has been established and includes representatives of five (5) EPA Regions, two (2) States, and OSHA. In addition, the following EPA Headquarters offices are represented: Office of Policy, Planning and Evaluation, Office of Pesticides and Toxic Substances, Office of General Counsel, Office of Research and Development, Office of Solid Waste and Emergency Response, and the Occupational Health and Safety Staff. The objective of this Workgroup is to:

- Promulgate EPA Worker Health and Safety Standards identical to OSHA's standards (29 CFR 1910.120)for those 27
 States which do not have in effect an approved State Plan under the Occupational Safety and Health Act of 1970.
- Develop an implementation/enforcement strategy for these EPA standards.

In order to better implement both EPA and OSHA Occupational Health and Safety requirements, OSWER has established an Integrated Health and Safety Program for the sole purpose of assisting OSWER in providing a safe and healthy work environment for its field activities. This program is managed by Rod Turpin, ERT, Edison, NJ. Please feel free to call him at 201-321-6745 (FTS 340-6745) for any additional information.

Thank you in advance for your assistance and dedication in making our work environment a safe and healthy one.

Addressees:

Regional Administrators, Regions I - X Henry L. Longest II (WH-548) Marcia E. Williams (WH-562) Ronald Brand (WH-562A) Gene A. Lucero (WH-527)

cc: David Weitzman (PM-273F) Waste Management Division Directors, EPA Regions I-X Environmental Services Division Directors, EPA Regions I-X

EPA Interim Guidance on Indemnification of Superfund Response Action Contractors Under Section 119 of SARA

154

United States Environmental Protection Agency Office of Solid Waste and Emergency Response

€PA

DIRECTIVE NUMBER: 9835.5

TITLE: EPA Interim Guidance on Indemnification of Superfund Response Action Contractors under Section 119 of SARA

APPROVAL DATE: October 6, 1987

EFFECTIVE DATE: October 6, 1987

ORIGINATING OFFICE: OWPE

I FINAL

LEVEL OF DRAFT

- A Signed by AA or DAA
- B -- Signed by Office Director
- C Review & Comment

REFERENCE (other documents):

SWER OSWER OSWER DIRECTIVE DIRECTIVE DI

	nited States Environmental Prot Washington, DC 204	ection Agency 60	1 Directive Number
OSWER	Directive Initia	ation Re	quest
	2. Originator Informa	tion	
Name of Contact Person	Mail Code	Office	Telephone Code
Tom Gillis	WH 527	OWPE	382-4524
3 Tule EPA Interim Guidance on of SARA	Indemnification of	Superfund C	ontractors under Section 119
4 Summary of Directive (include brief state	ement of purpose)		
Provide Guidance to EPA	Regional Personnel	on EPA Supe	rfund RAC Indemnification
including EPA Interim Gu	idelines. Procedure	s for Broce	ssing Indemnification Requests
and Model Indemnification	D Contract Language		bring indemnitiention keydests,
	wentrat	-	
5 Keywords	·	·····	
Superfund, CERCLA, Indemn	ification, RAC, Cor	ntrator. Lial	bility. Contracts
6a Does This Directive Supersede Previou	Is Directive(s)?		
	X No	Yes	What directive (number, title)
b Does It Supplement Previous Directive	s)?	<u> </u>	
	XNO	Ves	What directive (number, title)
7 Draft Level			
Y A - Signed by AA/DAA	8 - Signed by Office Direct	or C-F	or Review & Comment D - In Developmen
8. Document to be d	istributed to States	by Headqua	rters?
This Request Meets OSWER Directives	System Format Standards.		
a aignature di Lead Onice Directives Cool			
14ar land M h.	10 uno		10/0/57
10 Name and Title of Approving Official			Date

OSWER OSWER OSWER O VE DIRECTIVE DIRECTIVE DIRECTIVE

EPA Form 1315-17 (Rev. 5-87) Previous editions are obsolete



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

OSWER Directive 9835.5

78ei 6 700

MEMORANDUM

SUBJECT: EPA Interim Guidance on Indemnification of Superfund Response Action Contractors Under Section 119 of SARA J. Winston Porter, Assistant Administrator FROM: Office/of Solid Waste and Emergency Response C. Morgan Ainghorn, Acting Assistant Administrator Office of Administration and Resources Management TO: Regional Administrator, Regions I-X Regional Counsel, Regions I-X Director, Waste Management Division Regions I, IV, V, VII, and VIII Director, Emergency and Remedial Response Division Region II Director, Hazardous Waste Management Division Region III and VI Director, Toxics and Waste Management Division Region IX Director, Hazardous Waste Division Region X Director, Environmental Services Division Regions I, VI, and VII

Purpose

Subject to certain restrictions, Section 119 of the Superfund Amendments and Reauthorization Act of 1986 (SARA) authorizes the Environmental Protection Agency (EPA)¹ to provide indemnification² to response action contractors (RACs) working at Superfund sites for States, potentially responsible parties (PRPs), and EPA (including RACs working for the U.S. Army Corps).

¹ Under Executive Order 12580, the President has also authorized other Federal agencies to indemnify RACs working for those agencies.

² "Indemnification" is an agreement whereby one party agrees to reimburse a second party for losses (in this case liability losses) suffered by the second party. of Engineers at EPA-lead sites)³. The purpose of this memo is to describe her EPA may provide indemnification to RACs using Section 119 authority.

Background

Response action contractors have traditionally relied on commercial liability insurance or indemnification to sufficiently offset their potential liability risks from participation in the Superfund program. During the Superfund reauthorization debate, the RAC community identified several factors which, the RACs contended, impaired their ability to adequately offset risk. These factors included:

- Potential subjection to strict, joint and several liability under Superfund and under some state laws; and
- o Inability of the commercial liability insurance market to provide liability insurance coverage to RACs involved in the Superfund cleanup program that is both adequate and affordable.

Prior to the reauthorization of CERCLA, EPA provided indemnification to RACs working for EPA through contract authority implementing CERCLA. EPA took this step in order to retain qualified contractors, given the absence of pollution liability insurance coverage. Under this old indemnification agreement, the Pederal government indemnified RACs above an initial \$1 million for third party liabilities and defense expenses. The indemnification agreement was void in cases of gross negligence or willful misconduct.

3 SARA Section 119(e)(2) defines "response action contractor" as:

- a. any person who enters into a response action contract (which is defined in part as any written contract or agreement to provide any CERCLA removal or remedial action at a facility listed on the NPL, or to provide any ancillary services related to such response) with respect to any release or threatened release of a hazardous substance or pollutant or contaminant from a facility and is carrying out such a contract; and
- b. any person retained or hired by the person who enters into a response action contract, to provide any services related to a response action; and
- c. any person, public or nonprofit private entity, conducting a field demonstration pursuant to SARA Section 311(b) (i.e., the "Alternative or Innovative Treatment Technology Research and Demonstration Program").

Section 119 of SARA responds to many of the concerns of the RAC community by:

- Establishing a standard of negligence for actions brought against RACs under Federal law;⁴
- Authorizing EPA to provide to RACs, on a discretionary basis, limited indemnification against pollution liability arising from RAC negligence; and
- Providing express statutory authority for indemnification and a funding mechanism.

The approach taken in Section 119 provisions is based on the following key points:

- A Federal liability standard of negligence, combined with RAC indemnification which is subject to limits and deductibles, provides adequate performance incentives for RACs working in the Superfund program;
- RAC indemnification provides an adequate substitute for insurance;
- Discretionary indemnification is an interim vehicle that will keep the Superfund program operative until the insurance industry returns to the RAC liability insurance market; and
- Discretionary indemnification does not create a Federally intrusive insurance program that interferes with private sector efforts to develop RAC liability insurance coverage.

⁴ The Federal standard of negligence under Section 119 applies only to Federal law. It does not preclude States from applying their own statutory law or common law liability standards, which may in some cases be strict liability. Response action contractors sued in Federal courts are under a "standard of care" defined by Federal law as negligence. However, if an action is brought under state law, a strict liability standard could apply.

EPA Task Foise on RAC Indemnification

To avoid program delays, a Task Force was established to determine how EPA will provide indemnification to RACs working in the Superfund program. The Task Force is composed of representatives from EPA's Office of Waste Programs Enforcement (OWPE), Office of Emergency and Remedial Response (OERR), Office of Solid Waste (OSW), Office of General Counsel (OGC), Office of the Comptroller (OC), Office of Administration (OA), and the U.S. Army Corps of Engineers. The primary goals of the Task Force are to:

- o Establish an EPA RAC indemnification program;
- o Develop Section 119 RAC final indemnification guidelines and regulations;
- Ensure a forum for adequate public comment on RAC indemnification; and
- Promote private sector provision of RAC pollution liability insurance in the future by providing technical assistance to the insurance industry.

The Task Force will attempt to reach these goals by producing several work products that: (1) carefully analyze and estimate the potential pollution liability risk to which RACs are exposed by participating in the Superfund cleanup program; (2) determine what the final EPA indemnification terms and conditions will be; (3) prepare the Agency for implementing an interim RAC indemnification program; and (4) develop the Section 119 regulations.

Interim EPA Indomnification Guidelines

1

SARA Section 119 now provides EPA's sole authority to extend indemnification to RACs working in the Superfund program. Delegation of authority from the President authorizing EPA to use Section 119 provisions was issued through Executive Order 12580 on January 26, 1987. The delegation authorizes EPA to use Section 119 indemnification authority from the date of enactment (DOE) of SARA. Consequently, EPA must adhere to Section 119 provisions from SARA DOE (October 17, 1986).

Section 119(c)(7) requires that EPA promulgate regulations for carrying out indemnification provisions and, prior to promulgation of the regulations, develop guidelines to carry out use of Section 119 indemnification authority. Because of the complexity of the issues, EPA is proceeding deliberately in establishing these guidelines and is seeking substantial public

¢

comment. Mainwhile, EPA is providing contractors with Section 119 coverage on an interim basis, using procedures outlined in this memorandum. Ultimately, this coverage will be amended to reflect guidance and regulations that will be developed in conformance with Section 119 requirements.

As further described in this memorandum, authorization to provide indemnification will be made by OSWER with concurrence from the Office of the Comptroller (OC). Authorization to indemnify will be made upon receipt of a recommendation from the Task Force. The OC will provide concurrence (or non-concurrence) with recommendations to indemnify within seven calendar days of receipt of a recommendation. Execution of indemnity agreements will be made by appropriate Agency administrative offices.

- Section 119(c)(4) mandates that RACs must meet the following requirements before they can receive Federal indemnification for potential pollution liability associated with Superfund response action activities:

- The RAC must make diligent efforts to obtain insurance coverage from non-Federal sources to cover pollution liability; and
- o In the case of a RAC contract covering more than one facility, the RAC agrees to continue to make such diligent efforts each time the RAC begins work under the contract at a new facility.

Section 119(c)(4) also requires that the following circumstances must exist before a RAC can receive Federal indemnification for potential pollution liability associated with Superfund response action activities:

- o At the time the response action contract is entered into, insurance is not available, at a "fair and reasonable price", in sufficient quantity to offset potential RAC pollution liability risk; and
- o Adequate insurance to cover such liability is not generally available at the time the response action contract is entered into.

In future guidance (i.e., the guidance which is to be published for public comment), EPA plans to include guidelines for determining whether insurance is "generally available" or is "fairly and reasonably priced". For the purpose of this interim guidance, EPA has determined, based on information currently available, that Superfund RACs are unable to obtain reasonably priced pollution liability insurance. Therefore, RACs are eligible to receive indemnification under Section 119 from DOE of SARA. However, EPA will require that RACs seeking Federal indemnification meet the following requirements:

- o Within 30-days of signing an indemnification agreement with EPA, RACs must submit to EPA (or to the appropriate State Contracting Officer) written documentation concerning the efforts they have made to date to secure pollution liability insurance coverage (e.g., a RAC could submit a written statement from an insurance broker stating that the RAC has attempted to secure pollution liability coverage from insurance carriers in the past six months).
- If the RAC has secured pollution liability coverage, it must submit to EPA (or to the State Contracting Officer) a copy of the policy and declaration page; and
- o Every twelve months (or more frequently, if EPA determines that there has been a significant change in circumstances concerning the availability of pollution liability insurance) the RAC must submit to EPA (or to the State Contracting Officer) written documentation addressing the additional efforts the RAC has made to secure pollution liability insurance coverage including:
 - Copies of applications submitted to three known underwriters of pollution liability insurance;
 - If pollution liability coverage was denied by an underwriter, a summary of the reasons why such coverage was denied;
 - A status report of any pollution liability insurance obtained. The report would include: 1) type of coverage; 2) premium charged; 3) limits of coverage; 4) deductible levels, and any other major terms and conditions of the insurance coverage. A copy of the actual policy and declaration page could be provided in lieu of a written status report;
 - If pollution liability coverage was offered by an underwriter, but not accepted by the RAC, a report on the insurance offered (such as the "status report" required above), and a summary of the reasons why such coverage was not accepted; and
 - A status report concerning the alternative pollution liability risk transfer mechanisms the RAC has pursued other than commercial pollution liability insurance (e.g., risk retention groups, purchasing groups, association captives).

This information should be forwarded to the appropriate EPA official (or State Contracting Officer). This information will be reviewed by the Task Force as needed.

As required under the interim guidelines listed above, EPA expects RACs to demonstrate the extent to which they have attempted to secure pollution liability insurance coverage. EPA also expects that RACs will continue to monitor the market for pollution liability insurance, and continue to seek and secure such insurance coverage (however limited) from commercial insurance carriers or through alternative risk transfer mechanisms (e.g., self-insurance pools).

Indemnification of BACs Working for EPA

Pre-SARA indemnification terms will apply to work performed at a site after the date of enactment (DOE) of SARA if response work at the site was initiated under an EPA contract prior to the DOE of SARA.

EPA will enter into new indemnification agreements (See Attachment A), subject to Section 119 authority, with:

- RACs who are currently working under contract with EPA, for work they will initiate at a new site after DOE of SARA; and
- RACs receiving new contracts (or new cooperative agreements, in the case of Site Demonstration projects) with EPA after DOE of SARA for Superfund response action activities.

RACs currently under contract with EPA have been alerted to the changes that will be forthcoming to their indemnification agreements with EPA. EPA headquarters personnel in the Procurement and Contracts Management Division of the Office of Administration have been trained on the use of Section 119 and, with the assistance of the Task Force, will administer Section 119 indemnification interim procedures for EPA contractors. Requests for indemnification of EPA contractors will be subject to the approval of OSWER and concurrence of OC.

Indemnification of RACs Working for States

Section 119(c)(2) authorizes the indemnification of RACs working for States or political subdivisions of States (pursuant to a Section 104(d)(1) agreement with EPA) for new work initiated at Superfund sites from DOE of SARA. EPA may indemnify RACs performing response action activities for a State at a State-lead Superfund site after DOE of SARA. EPA will offer indemnification to RACs working for a State only if:

- o The RAC's response action is part of new site work initiated at a Superfund site after DOE of SARA and it is related directly to cleanup of the site;
- RACs working for a State must meet all of the circumstances and issuance requirements set forth by Section 119(c)(4), as listed above; and
- RACs working for a State must meet all of EPA's interim guideline requirements, as listed previously on pages five and six.

EPA will not offer indemnification to RACs for site work they performed for States prior to DOE of SARA. Any EPA indemnification provided to a RAC(s) working for a State(s) will be subject to limits, deductibles, and other restrictions as required by Section 119(c)(5).

Until EPA issues final guidance and regulations, all requests for EPA indemnification of a RAC working for a State at a Superfund site will be processed via the Task Force. States should submit requests to both the Indemnification Task Force, c/o Director, Office of Emergency and Remedial Response (OERR), and to the Regional Superfund Branch Chief. Requests should identify the Regional Site Coordinator and State contact, and should include pertinent information regarding Section 119(c)(4) requirements as discussed previously. If the Task Force recommends approval of the indemnification request, the Office of the Comptroller will provide concurrence (or non-concurrence) within seven calendar days of receipt of the recommendation. Final approval for EPA indemnification of a State RAC will be made by the Director of the Office of Emergency and Remedial Response. If approval is authorized, then the Grants Administration Division will implement the approval through a special condition to be included in the State/EPA cooperative agreement (See Attachment A).

Indemnification of RACs working for Other Federal Agencies

Section 119(c)(2) authorizes the indemnification of RACs working for other Federal agencies at Superfund sites from DOE of SARA. A delegation of authority from the President authorizing other Federal Agencies to use Section 119 provisions was issued on January 26, 1987. Other Federal agencies follow all EPA guidance and regulations with respect to Section 119. Other Federal agencies that use Section 119 authority must provide their own source of funds (e.g., their agency appropriation) to pay all indemnification costs (e.g., claims and legal defense costs).

At some Superfund sites, the U.S. Army Corps of Engineers manages response actions pursuant to an interagency agreement with EPA. For Section 119 indemnification purposes, any RAC working as a contractor for the Corps of Engineers at such sites (and where, for remedial actions, the site is listed on the NPL) is considered to be working for EPA rather than for some "other Federal agency". EPA will offer the same indemnification to contractors procured by the Corps of Engineers that it offers to contractors procured by EPA.

Indemnification of RACs Working for PRPs

Under Section 119(c)(2) authority, EPA can, in limited circumstances and subject to strict financial tests, indemnify RACs performing response action activities for PRPs subject to a consent order or decree at Superfund sites after DOE of SARA. EPA will use its authority to indemnify RACs working for PRPs only in extremely limited cases, e.g., where EPA indemnification of the PRP RAC is the solution of last resort. EPA will offer indemnification to RACs working for PRPs <u>only if</u>:

- o The PRPs are unable to provide adequate indemnification, and as a result, are unable to obtain the services of a qualified RAC;
- o The RAC's response action is part of new site work initiated at a Superfund site after DOE of SARA, and the action is related specifically to the cleanup of the site;
- o RACs working for PRPs meet all of the issuance requirements set forth by Section 119(c)(4);
- o The circumstances set forth in Section 119(c)(4) exist; and
- RACs working for PRPs meet all of EPA's interim guideline requirements.

EPA will not offer indemnification to RACs for work performed for PRPs prior to DOE of SARA, nor for any PRP RAC response serivity that is not related specifically to a remedy at a Superfund site.

Further, Section 119(c)(5)(C) of SARA requires that, before EPA can enter into an indemnification agreement with a RAC performing work under contract with a PRP(s) at a Superfund site(s), EPA must determine the amount which the PRP(s) is able to indemnify the RAC. In making such a determination, BPA shall take into account the total net assets and resources of the PRP(s) with respect to the facility at the time of such determinations. If EPA determines that the amount which the PRP(s) is able to indemnify the RAC is inadequate, then EPA may enter into an indemnification agreement with the RAC to meet the anticipated shortfall. EPA will consider the combined capabilities of all the PRPs at a site to determine whether, as a group, they are capable of providing adequate coverage. In general, the Agency expects to use this provision only in cases where PRPs are small firms with few assets. Therefore, Regions should not make requests for Federal indemnification where PRPs are large corporations with substantial assets or where the PRPs, as a group, have substantial assets. As a result, EPA does not expect requests for Pederal indemnification to become an integral part of settlement negotiations.

EPA plans to provide additional guidance in the future concerning the determinations that need to be made as a prerequisite to indemnifying RACs working for PRPs (such as defining "net assets and resources" of the PRPs, and whether the PRPs are "unable to provide adequate indemnification"). Until EPA distributes this guidance, all such determinations will be made by the Task Force.

EPA indemnification of a RAC working for a PRP is a measure of last resort. If EPA does provide indemnification in these cases, the consent decree (or order) should specify terms and conditions, using the model EPA indemnification agreement for RACs working for PRPs shown in Attachment A. If EPA enters into an indemnification agreement with a RAC working for a PRP(s), the RAC must:

- Retain financial responsibility for a deductible amount if commercial pollution liability insurance is unavailable or unreasonably priced; and
- Exhaust all administrative, judicial, and common'law claims for indemnification against all PRPs participating in the cleanup of the facility before EPA can pay a claim.

If a BAC has received partial indemnification from a PRP(s), EPA may also provide indemnification in cases where the PRP indemnification is deemed insufficient, and in mixed funding cases. EPA may provide indemnification above the PRP indemnification. The consent decree should specify the terms and conditions using the model EPA indemnification agreement shown in Attachment A.

All requests for EPA indemnification of a RAC working for a PRP(s) at a Superfund site should be submitted to both the Indemnification Task Force, c/o Director, Office of Waste Programs Enforcement (OWPE), and to the Regional Superfund Enforcement Branch Chief. Please identify the Regional Site Coordinator and the Regional Counsel's Site Representative. Include pertinent information regarding the number of PRPs, financial profile of the PRPs, type of work to be performed, etc., such that the Task Force can make determinations per Section 119(c)(4) and Section 119(c)(5).

Upon determining that a RAC meets all of the circumstances and requirements set forth in Section 119 and in EPA interim guidelines, the Task Force will evaluate an amount to which the PRP(s) is able to indemnify the RAC and an amount to which EPA will indemnify the RAC in excess of the PRP indemnification amount. Any EPA indemnification provided to a RAC(s) working for PRP(s) will be subject to limits, deductibles, and other limitations as required by Section 119(c)(5). If the Task Force recommends approval of the indemnification request, the Office of the Comptroller will provide concurrence (or non-concurrence) within seven calendar days of receipt of the recommendation. Final approval for EPA indemnification of a PRP RAC will be made by the Director of OWPE.

RACS Working for PRPs Without BPA Indemnification

Those RACs working for PRPs at Superfund sites who do not receive indemnification from EPA may either receive no indemnification at all, or may receive indemnification from PRPs only. For those RACs working with no indemnification, PRPs should demonstrate that the RAC is qualified to perform the work adequately, has sufficient financial capability to complete the projected work, and demonstrates financial responsibility for potential third party liability costs. This can be ensured through a combination of adequate competition in the contract procurement process and a demonstration of financial responsibility. Such a demonstration can consist of purchase of performance bonds, letters of credit, insurance, maintenance of a trust fund, etc. A consent decree should specify the aforementioned. For these RACs receiving indemnification from PRPs only (and where EPA deems the indemnification to be adequate), RACs should be qualified to perform work adequately. This can be ensured through a combination of adequate competition in the contract procurement process, and through a demonstration of financial responsibility. The PRP indemnification is sufficient demonstration of financial responsibility; therefore, performance bonds, letters of credit, etc., are not required. The consent decree should specify the aforementioned as well as the indemnification terms and conditions.

Publicly Owned Treatment Works

Section 119(c)(5)(D) specifically prohibits EPA from indemnifying an owner or operator of a facility regulated under the Solid Waste Disposal Act. Therefore, publicly owned treatment works subject to permit-by-rule provisions cannot be indemnified (nor can any other permit-by-rule facility, such as an underground injection facility). The intent of this provision is to prohibit EPA from offering indemnification to off-site treaters or disposers of Superfund hazardous waste. Therefore, while POTWs not subject to RCRA regulation (i.e., POTWs without a permit-by-rule) are not explicitly prohibited from EPA indemnification authority under Section 119, the Agency has determined that an extension of indemnification authority to any POTW would not be consistent with Congressional intent in Section 119. Therefore, EPA will not provide indemnification to POTWs under Section 119 authority.

Summary

This memorandum describes the current Federal indemnification provisions for response action contractors working in the Superfund program as provided in Section 119 of SARA. The statute gives the Federal government the discretionary authority to indemnify RACs for liability arising out of negligence. Acts of gross negligence and willful misconduct are expressly excluded from the indemnity provision. The Section 119 indemnity provision does not preempt the rights of States to enforce a standard of strict liability.

Federal indemnification is meant to be an interim vehicle which will keep the Superfund program operative until the insurance industry returns to the market. It is not intended to create a Federally intrusive program that will interfere with private sector efforts to develop RAC liability insurance coverage.

13 OSWER Directive 9835.5

.

Please direct all questions and comments to Robert Mason at FTS 382-4015 or Tom Gillis at FTS 382-4524

Attachments A. Model Indemnification Agreements B. CERCLA (as amended) Section 119

•

.

cc: Administrator Deputy Administrator General Counsel Regional Grants Office, Regions I-X Regional Financial Management Office, Regions I-X Regional Superfund Branch Chiefs, Regions I-X Attachment A

MODEL INDEMNIFICATION AGREEMENTS

.

.

This attachment contains model EPA indemnification agreements for use by EPA, States, and PRPs when RACs seek indemnification from EPA. Any deviation from the model language must be approved by the EPA Indemnification Task Porce. Four models are attached:

- I. Model EPA/RAC Indemnification Agreement
- II. Model State Cooperative Agreement Indemnification Special Condition
- III. Model EPA/RAC Indemnification Agreement for RACs under Contract with PRPs
- IV. Model EPA/ SITES Program Technology Vendor Indemnification Agreement

I

.

MODEL EPA/RAC INDEMNIFICATION AGREEMENT

.

Insurance -- Liability to Third Persons --Commercial Organizations (EPAAR 1557.228-70) (APR 1984) (with deviation)

(a) This Clause H will be modified by the mutual agreement of the parties hereto within 180 days of the EPA's promulgation of final guidelines for carrying out the provisions of Section 119 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (CERCLA).

(b) The Contractor shall procure and maintain such insurance as is required by law or regulation, including that required by FAR Part 28, in effect as of the date of execution of this contract, and any such insurance as the Contracting officer may, from time to time, require with respect to performance of this contract.

(c) At a minimum, the Contractor shall procure and maintain the following types of insurance.

 (1) Workmen's compensation and occupational disease insurance in amounts to satisfy State law;

(2) Employer's liability insurance in the minimum amount of \$100,000 per occurrence;

(3) Comprehensive general liability insurance for bodily injury, death or loss of or damage to property of third persons in the minimum amount of \$1,000,000 per occurrence;

(4) When vessels are used in the performance of the contract, vessel collision liability and indemnity liability insurance in such amounts as the Contracting Officer may require or approve: provided, that the Contractor may, with the approval of the Contracting Officer, maintain a self-insurance program. All insurance required pursuant to the provisions of this paragraph shall be in such form and for such periods of time as the Contracting Officer may, from time to time, require or approve and with insurers approved by the Contracting Officer.

(d) The Contractor further agrees that it will make diligent efforts throughout contract performance in accordance with EPA guidelines to obtain adequate pollution liability insurance.

(e) The Contractor agrees, to the extent and in the manner required by the Contracting Officer, to submit for the approval of the Contracting Officer all insurance maintained by the. Contractor in connection with the performance of this contract $_{\rm fi}$ and for which the Contractor seeks reimbursement hereunder. The

Contractor's submission shall include documentation demonstrating its diligent efforts to obtain pollution liability insurance.

(f) The Contractor shall be reimbursed, for the portion allocable to this contract, the reasonable cost of insurance (including reserves for self-insurance) as required or approved pursuant to the provisions of this contract clause.

(g)(1) Pursuant to Section 119 of CERCLA, the EPA will hold harmless and indemnify the Contractor against any liability . (including the expenses of litigation or settlement) for negligence arising out of the Contractor's performance under this contract in carrying out response action activities. Such indemnification shall apply only to liability not compensated by insurance or otherwise and shall apply only to liability which results from a release of any hazardous substance or pollutant or contaminant if such release arises out of the response action activities of this contract. Further, any liability within the deductible amounts of the Contractor's insurance will not be covered under this contract clause H

(2) For purposes of this clause (g), if the Contracting
Officer has determined that the insurance identified in paragraph
(d) is not available at a reasonable cost, the Government will
hold harmless and indemnify the Contractor for liability to the
extent such liability exceeds \$100,000.00.

(3) The Contractor shall not be reimbursed for liabilities as defined in (g) (including the expenses of litigation or settlement) that were caused by the conduct of the Contractor (including any conduct of its directors, managers, staff, representatives or employees) which was grossly negligent, constituted intentional misconduct, or demonstrated a lack of good faith. Further, the Contractor shall not be indemnified for liability arising under strict tort liability, or any other basis of liability other than negligence.

(h) The Government may discharge its liability under this contract clause by making payments directly to the Contractor or directly to parties to whom the Contractor may be liable.

(i) With prior written approval of the Contracting Officer, the Contractor may include in any subcontract under this contract the same provisions in this clause whereby the Contractor shall indemnify the subcontractor. Such a subcontract shall provide the same rights and duties and the same provisions for notice, furnishings of evidence or proof, and the like, between the Contractor and the subcontractor as are established by this clause. Similar indemnification may be provided for subcontractors at any time upon the same terms and conditions. Subcontracts providing for indemnification within the purview of this contract clause shall provide for prompt notification to the Contractor which is covered by this contract clause, and shall entitle the Government, at its election, to control, or assist in the settlement of defense of any such claim or action. The Government will indemnify the Contractor with respect to his obligation to subcontractors under such subcontract provisions. The Government may discharge its obligations under this paragraph by making payments directly to subcontractors or to parties to whom the subcontractors may be liable.

(j) If insurance coverage required or approved by the Contracting Officer is reduced without the Contracting Officer's approval, the liability of the Government under this contract clause will not be increased by reason of such reduction.

(k) The Contractor shall:

(1) Promptly notify the Contracting Officer of any claim or action against the Contractor or any subcontractor which reasonably may be expected to involve indemnification under this contract clause;

(2) Furnish evidence or proof of any claim covered by this contract clause in the manner and form required by the Government; and

(3) Immediately furnish the Government copies of all pertinent papers received by the Contractor. The Government may direct, control, or assist the settlement or defense of any such claim or action. The Contractor shall comply with the Government's directions, and execute any authorizations required in regard to such settlement or defense.

(1) Reimbursement for any liabilities under this contract clause will not exceed appropriations available from CERCLA's Hazardous Substance Superfund (except to the extent that Congress may make appropriations to specifically fund any deficiencies) at the time such liabilities are represented by final judgments or by settlements approved in writing by the Government.
MODEL STATE COOPERATIVE AGREEMENT INDEMNIFICATION SPECIAL CONDITION

II

EPA INDEMNIFICATION

EPA will provide indemnification pursuant to Section 119 of CERCLA, as amended, to contractors carrying out response actions under this agreement provided that the State certifies to EPA that:

- 1. The contracts awarded under this agreement are defined in section 119(e) of CERCLA, as amended;
- The contracts awarded under this agreement include the following clause that exclusively governs EPA indemnification:

(see attached clause)

3. At the end of each calendar year and at the end of each project period, all statements and materials related to pollution liability insurance submitted by the Contractors to the State Contracting Officer will be transferred to EPA.

. Attachment

(1) Pursuant to Section 119 of CERCLA, the EPA will hold harmless and indemnify the Gontractor against any third party liability (including the expenses of litigation or settlement) for negligence arising out of the Contractor's performance under this contract in carrying out response action activities. Such indemnification shall apply only to liability not compensated by insurance or otherwise and shall apply only to liability which results from a release of any hazardous substance or pollutant or contaminant if such release arises out of the response action activities of this contract. Further, any liability within the deductible amounts of the Contractor's insurance required by this contract will not be covered by this clause. This Clause will be modified by the mutual agreement of the parties hereto within 180 days of the EPA's promulgation of final guidelines for carrying out the provisions of Section 119 (CERCLA).

- (A) The Contractor shall submit to the State Contracting Officer within 30 days of award a written statement from an insurance broker stating that the Contractor has attempted to secure pollution liability coverage from insurance carriers in the past six months;
- (B) If the Contractor has secured pollution liability coverage, it must submit a copy of the policy and declaration page to the State Contracting Officer; and
- (C) Every twelve months, or as directed by the EPA, the Contractor shall submit to the State Contracting Officer written documentation of the additional efforts made by the contractor to secure pollution liability insurance coverage, including:
 - Copies of applications to three known underwriters of pollution liability insurance;
 - o A status report of any pollution liability insurance obtained, to include type of coverage, premium charged, limits of coverage, deductibles and major terms and conditions of coverage (e.g., a copy of the actual declaration page could be provided in lieu of a status report);
 - o If pollution liability coverage was offered by an underwriter, but not accepted by the RAC, a report on the insurance offered (such as the "status report" required above), and a summary of the reasons why such coverage was not accepted;
 - If pollution liability coverage was rejected by the underwriter, a summary of the reasons why such coverage was denied; and

o A status report on what alternative pollution liability risk transfer mechanisms the contractor has pursued other than commercial pollution liability insurance (e.g., captives, letters of credit, group purchasing of insurance, etc.).

(2) For purposes of this clause, the EPA will hold harmless and indemnify the Contractor for liability described herein to the extent such liability exceeds \$100,000.00.

(3) The Contractor shall not be reimbursed for liabilities as defined herein (including the expenses of litigation or settlement) that were caused by the conduct of the Contractor (including any conduct of its directors, managers, staff, representatives or employees) which was grossly negligent, constituted intentional misconduct, or demonstrated a lack of good faith. Further, the Contractor shall not be indemnified for liability arising under strict tort liability, or any other basis of liability other than negligence.

(4) The EPA may discharge its liability under this contract clause by making payments directly to the Contractor or directly to parties to whom the Contractor may be liable.

(5) With prior written approval of the State Contracting Officer, the Contractor may include in any subcontract under this contract the same provisions in this clause whereby the Contractor shall indemnify the subcontractor. Such a subcontract shall provide the same rights and duties and the same provisions for notice, furnishings of evidence or proof, and the like, between the Contractor and the subcontractor as are established by this clause. Similar indemnification may be provided for subcontractors at any time upon the same terms and conditions. Subcontracts providing for indemnification within the purview of this contract clause shall provide for prompt notification to the Contractor which is covered by this contract clause, and shall entitle the EPA, at its election, to control, or assist in the settlement or defense of any such claim or action. The EPA will indemnify the Contractor with respect to his obligation to subcontractors under such subcontract provisions. The EPA may discharge its obligations under this paragraph by making payments directly to subcontractors or to parties to whom the subcontractors may be liable.

(6) If insurance coverage required or approved by the State Contracting Officer is reduced without the State Contracting Officer's approval, the liability of the EPA under this contract clause will not be increased by reason of such reduction.

(7) The Contractor shall:

- Promptly notify the Assistant Administrator,
 OSWER, EPA of any claim or action against the
 Contractor or any subcontractor which reasonably
 may be expected to involve indemnification under
 this contract clause.
- Furnish evidence or proof of any claim covered by this contract clause in the manner and form required by the EPA.
- o Immediately furnish the EPA copies of all pertinent papers received by the Contractor. The EPA may direct, control, or assist the settlement or defense of any such claim or action. The Contractor shall comply with the EPA's directions, and execute any authorizations required in regard to such settlement or defense.
- Submit any disagreements concerning EPA indemnification to the Assistant Administrator,
 OSWER, EPA for resolution. Decision by the Assistant Administrator will constitute final Agency action.

(8) Reimbursement for any liabilities under this contract clause is available exclusively from the EPA and will not exceed appropriations available from CERCLA's Hazardous Substance Superfund (except to the extent that Congress may make appropriations to specifically fund any deficiencies) at the time such liabilities are represented by final judgement or by settlements approved in writing by the EPA.

(9) Nothing in this clause shall be construed as an indemnification agreement between the State and the Contractor.

(10) Nothing in this contract shall be construed to create, either expressly or by implication, any contractual relationship between EPA and the Contractor except as specifically provided in this clause. EPA is not authorized to represent or act on behalf of the State in any manner relating to this contract and has no responsibility with regard to the mutual obligations of the State and the Contractor as provided herein. III

MODEL EPA/RAC INDEMNIFICATION AGREEMENT FOR RACS UNDER CONTRACT WITH PRPS

.

Sec. _____ Pollution Liability Insurance and Contractor Indemnification

A. Pollution Liability Insurance

(1) The Contractor shall obtain such pollution liability insurance (hereinafter insurance) as the EPA determines is available at a fair and reasonable price at the time of contract award. The cost of such insurance is an allowable contract cost.

(2) The Contractor shall report to EPA on its efforts to obtain pollution liability insurance.

- (A) Within 30 days of signing this agreement, the Contractor shall submit to the EPA a written statement from an insurance broker stating that the Contractor has attempted to secure pollution liability coverage from insurance carriers in the past six months;
- (B) If the Contractor has secured pollution liability coverage, it must submit a copy of the policy and declaration page to EPA; and
- (C) Every twelve months, or as directed by the EPA, the Contractor shall submit to the EPA written documentation of the additional efforts made by the contractor to secure pollution liability insurance coverage including:
 - Copies of applications to three known underwriters of pollution liability insurance;
 - o A status report of any pollution liability insurance obtained, to include type of coverage, premium charged, limits of coverage, deductibles and major terms and conditions of coverage (e.g., a copy of the actual declaration page could be provided in lieu of a status report);
 - o If pollution liability coverage was offered by an underwriter, but not accepted by the RAC, a report on the insurance offered (such as the "status report" required above), and a summary of the reasons why such coverage was not accepted;
 - o If pollution liability coverage was rejected by the underwriter, a summary of the reasons why such coverage was denied; and

 A status report on what alternative pollution liability risk transfer mechanisms the contractor has pursued other than commercial pollution liability insurance (e.g., captives, letters of credit, group purchasing of insurance, etc.).

(3) If, during the period of this contract, EPA determines that insurance or additional insurance is available, the contractor shall obtain such insurance.

B. PRP Indemnification

[The following are minimum clauses. PRPs may include additional, non-conflicting terms.]

(1) The PRPs will hold harmless and indemnify the Contractor against any third party liability (including the expense of litigation or settlement) for negligence arising out of the Contractor's performance of this contract in carrying out response action activities. Such indemnification shall apply only to liability which results from a release of a hazardous substance, pollutant, or contaminant if such release arises out of the response action activities in this contract. Indemnification under this paragraph will apply only to liability not compensated by insurance, not within the deductible amounts of the Contractor's insurance in paragraph A, above, nor within the deductible in paragraph D, below. Indemnification provided under this paragraph shall not exceed \$______(amount determined by EPA).

(2) Any liability subject to indemnification shall be presented first under this paragraph.

(3) The PRPs are individually and collectively responsible for the indemnification under this paragraph, unless otherwise specifically provided within.

(4) If the PRPs fail to satisfy the indemnification claim within 60 days of its presentation, the Contractor will notify the EPA of such failure.

C. EPA Indemnification

(1) Pursuant to Section 119 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (CERCLA), the EPA will hold harmless and indemnify the Contractor against any third party liability (including the expenses of litigation or settlement) for negligence arising out: of the Contractor's performance under this contract in carrying out response action activities. Such indemnification shall apply only to liability not compensated by insurance, indemnification provided in accGrdance with paragraph B, above, or otherwise and shall apply only to liability which results from a release of any hazardous substance or pollutant or contaminant if such release arises out of the response action activities of this contract. Further, any liability within the deductible amounts of the Contractor's insurance in paragraph A, above, or the deductible in paragraph D, below, will not be covered by this paragraph.

(2) This paragraph will be modified by the mutual agreement of the parties hereto within 180 days of the EPA's promulgation of final guidelines for carrying out the provisions of Section 119 of CERCLA.

(3) The Contractor shall not be reimbursed for liabilities as defined herein (including the expenses of litigation or settlement) that were caused by the conduct of the Contractor (including any conduct of its directors, managers, staff, representatives or employees) which was grossly negligent, constituted intentional misconduct, or demonstrated a lack of good faith. Further, the Contractor shall not be indemnified for liability arising under strict tort liability, or any other basis of liability other than negligence.

(4) The EPA may discharge its liability under this contract paragraph by making payments directly to the Contractor or directly to parties to whom the Contractor may be liable.

(5) With prior written approval of the EPA, the Contractor may include in any subcontract under this contract the same provisions in this clause whereby the Contractor shall indemnify the subcontractor. Such a subcontract shall provide the same rights and duties and the same provisions for notice, furnishings of evidence or proof, and the like, between the Contractor and the subcontractor as are established by this paragraph. Similar indemnification may be provided for subcontractors at any time upon the same terms and conditions. Subcontracts providing for indemnification within the purview of this paragraph shall provide for prompt notification to the Contractor which is covered by this paragraph, and shall entitle the EPA, at its election, to control, or assist in the settlement or defense of any such claim or action. The EPA will indemnify the Contractor with respect to his obligation to subcontractors under such subcontract provisions. The EPA may discharge its obligations under this paragraph by making payments directly to subcontractors or to parties to whom the subcontractors may be liable.

(6) If insurance coverage required in paragraph A, above, is reduced without the EPA's approval, the liability of the EPA under this paragraph will not be increased by reason of such reduction.

- (7) The Contractor shall:
 - Promptly notify the Assistant Administrator, OSWER, EPA of any claim or action against the Contractor or any subcontractor which reasonably may be expected to involve indemnification under this paragraph.
 - Furnish evidence or proof of any claim covered by this paragraph in the manner and form required by the EPA.
 - o Immediately furnish the EPA copies of all pertinent papers received by the Contractor. The EPA may direct, control, or assist the settlement or defense of any such claim or action. The Contractor shall comply with the EPA's directions, and execute any authorizations required in regard to such settlement or defense.
 - Submit any disagreements concerning EPA indemnification to the Assistant Administrator, OSWER, EPA for resolution. Decision by the Assistant Administrator will constitute final Agency action.

(8) The Contractor may present a claim for indemnification under this paragraph only after compliance with the provisions in paragraphs B, above, and C, below.

(9) If the PRPs fail to indemnify the Contractor in the amount provided in paragraph B, above, no indemnification for that amount will be paid under this paragraph until the Contractor demonstrates to EPA's satisfaction that it has exhausted all administrative and judicial claims for indemnification under paragraph B, above, and any common law claims for indemnification that it has against the PRPs. Evidence of exhaustion of claims may include a judicial order dismissing the Contractor's claims, documentation of the Contractor's unsuccessful efforts to enforce a judgement against the PRPs, or documentation of the Contractor's unsuccessful claims in a bankruptcy proceeding involving the PRPs.

(10) Reimbursement for any liabilities under this paragraph will not exceed appropriations available from CERCLA's Hazardous Substance Superfund (except to the extent that Congress may make appropriations to specifically fund any deficiencies) at the time such liabilities are represented by final judgement or by settlements approved in writing by the EPA. (11) Nothing in this contract shall be construed to create, either expressly or by implication, any contractual relationship between EPA and the Contractor except as specifically provided in this section. EPA is not authorized to represent or act on behalf of the (PRPs) in any manner relating to this contract and has no responsibility with regard to the mutual obligations of the (PRPs) and the Contractor as provided herein.

D. Contractor Deductible

The Contractor shall pay the first \$100,000.00 of any liability subject to indemnification under this contract before seeking indemnification under paragraphs B and C, above.

.

IV

MODEL EPA/ SITES PROGRAM TECHNOLOGY VENDOR

INDEMNIFICATION AGREEMENT

EPA_Indemnification

(1) Pursuant to Section 119 of CERCLA, the EPA will hold harmless and indemnify the Recipient against any liability (including the expenses of litigation or settlement) for negligence arising out of the Recipient's performance under this cooperative agreement in carrying out response action activities through the Superfund Innovative Technology Evaluation program under Section 311(b) of CERCLA. Such indemnification shall apply only to liability not compensated by insurance or otherwise and shall apply only to liability which results from a release of any hazardous substance or pollutant or contaminant if such release arises out of the response action activities of this cooperative agreement. Further, any liability within the deductible amounts of the Recipient's insurance will not be covered under this clause. If the recipient has secured pollution liability coverage, it must submit a copy of the policy and the declaration page to EPA.

(2) Every twelve months, or as directed by the EPA, the Recipient shall submit to the Contracting Officer written documentation of the additional efforts made by the recipient to secure pollution liability insurance coverage, including:

- Copies of applications to three known underwriters of pollution liability insurance;
- o A status report of any pollution liability insurance obtained, to include type of coverage, premium charged, limits of coverage, deductibles and major terms and conditions of coverage (e.g., a copy of the actual declaration page could be provided in lieu of a status report);
- If pollution liability coverage was rejected by the underwriter, a summary of the reasons why such coverage was denied; and

(3) For purposes of this clause, the Government will hold harmless and indemnify the Recipient for liability to the extent such liability exceeds \$100,000.00.

(4) The Recipient shall not be reimbursed for liabilities as defined herein (including the expenses of litigation or settlement) that were caused by the conduct of the Recipient (including any conduct of its directors, managers, staff, representatives or employees) which was grossly negligent, constituted intentional misconduct, or demonstrated a lack of good faith. Further, the Recipient shall not be indemnified for liability arising under strict tort liability, or any other basi's of liability other than negligence. (5) The Government may discharge its liability under this cooperative agréement clause by making payments directly to the Recipient or directly to parties to whom the Recipient may be liable.

(6) With prior written approval of the Contracting Officer, the Recipient may include in any subcontract under this cooperative agreement the same provisions in this clause whereby the Recipient shall indemnify the subcontractor. Such a subcontract shall provide the same rights and duties and the same provisions for notice between the Recipient and the subcontractor as are established by this clause. Similar indemnification may be provided for subcontractors at any time upon the same terms and conditions. Subcontracts providing for indemnification within the purview of this cooperative agreement clause shall provide for prompt notification to the Recipient which is covered by this cooperative agreement clause, and shall entitle the Government, at its election, to control, or assist in the settlement or defense of any such claim or action. The Government will indemnify the Recipient with respect to his obligation to subcontractors under such subcontract provisions. The Government may discharge its obligations under this paragraph by making payments directly to subcontractors or to parties to whom the subcontractors may be liable.

(7) If insurance coverage required or approved by the Contracting Officer is reduced without the Contracting Officer's approval, the liability of the Government under this cooperative agreement clause will not be increased by reason of such reduction.

(8) The Recipient shall:

(a) Promptly notify the Assistant Administrator, OSWER, EPA of any claim or action against the Recipient or any subcontractor which reasonably may be expected to involve indemnification under this cooperative agreement clause;

(b) Furnish evidence or proof of any claim covered by this cooperative agreement clause in the manner and form required by the Government;

(c) Immediately furnish the Government copies of all pertinent papers received by the Recipient. The Government may direct, control, or assist the settlement or defense of any such claim or action. The Recipient shall comply with the Government's directions, and execute any authorizations required in regard to such settlement or defense; and

(d) Submit any disagreements concerning EPA indemnification⁶ to the Assistant Administrator, OSWER, EPA for resolution. Decision by the Assistant Administrator will constitute final Agency action.

(9) Reimbursement for any liabilities under this cooperative agreement clause will not exceed appropriations available from CERCLA's Hazardous Substance Superfund (except to the extent that Congress may make appropriations to specifically fund any deficiencies) at the time such liabilities are represented by final judgement or by settlements approved in writing by the Government.

(10) This Clause will be modified by the mutual agreement of the parties hereto within 180 days of the EPA's promulgation of final guidelines for carrying out the provisions of Section 119 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (CERCLA). ATTACHMENT B

-

CERCLA (AS AMENDED)

.

SECTION 119

high priority to facilities where the release of hazardous substances or pollutants or contaminants has resulted in the closing of drinking water wells or has contaminated a principal drinking water supply.

SEC. 118. RESPONSE ACTION CONTRACTORS.

(a) LIABILITY OF RESPONSE ACTION CONTRACTORS -----

(1) RESPONSE ACTION CONTRACTORS.—A person who is a response action contractor with respect to any release or threatened release of a hazardous substance or pollutant or contaminant from a vessel or facility shall not be liable under this title or under any other Federal law to any person for injuries, costs, damages, expenses, or other liability (including but not limited to claims for indemnification or contribution and claims by third parties for death, personal injury, illness or loss of or damage to property or economic loss) which results from such release or threatened release.

(2) NEOLIGENCE, ETC.—Paragraph (1) shall not apply in the case of a release that is caused by conduct of the response action contractor which is negligent, grossly negligent, or which constitutes intentional misconduct.

(3) EFFECT ON WARRANTIES; EMPLOYER LIABILITY.—Nothing in this subsection shall affect the liability of any person under any warranty under Federal, State, or common law. Nothing in this subsection shall affect the liability of an employer who is a response action contractor to any employee of such employer under any provision of law, including any provision of any law relating to worker's compensation.

(1) GOVERNMENTAL EMPLOYEES.—A state employee or an employee of a political subdivision who provides services relating to response action while acting within the scope of his authority as a governmental employee shall have the same exemption from liability (subject to the other provisions of this section) as is provided to the response action contractor under this section. (b) SAVINGS PROVISIONS.—

(1) LIABILITY OF OTHER FERSONS.—The defense provided by section 107(b)(S) shall not be available to any potentially responsible party with respect to any costs or damages caused by any act or omission of a response action contractor. Except as provided in subsection (a)(4) and the preceding sentence, nothing in this section shall affect the liability under this Act or under any other Federal or State law of any person, other than a response action-contractor.

(2) BURDEN OF PLAINTIFF. --- Nothing in this section shall affect the plaintiff's burden of establishing liability under this title.

(c) INDEMNIFICATION. -

(1) IN OBNERAL.—The President may agree to hold harmless and indemnify any response action contractor meeting the re-

quirements of this subsection against any liability (including the expenses of litigation or settlement) for negligence arising out of the contractor's performance in carrying out response action activities under this title, unless such liability was caused by conduct of the contractor which was growly negligent or which constituted intentional misconduct.

(2) Applicability.—This subsection shall apply only with respect to a response action carried out under written agreement with—

(A) the President;

(B) any Federal agency;

(C) a State or political subdivision which has entered into a contract or cooperative agreement in accordance with section 104(d)(1) of this title; or

(D) any potentially responsible party corrying out any agreement under section 125 (relating to abatement).

(3) Source or running.—This subsection shall not be subject to section 1301 or 1241 of title 31 of the United States Code or section 3732 of the Revised Statutes (41 U.S.C. 11) or to section 3 of the Superfund Amendments and Resutherinstion Act of 1936. For purposes of section 111, amounts expanded pursuant to this subsection for indemnification of any ruponse acticontractor (except with respect to federally sound or operated facilities) shall be considered governmental ruponse costs incurred pursuant to section 104. If sufficient funds are unavailable in the Hazardous Substance Superfund established under subchapter A of chapter 98 of the Internal Revenue Code of 1354 to make payments pursuant to such indemnification or if the Fund is repealed, there are authorised to be appropriated such amounts as may be necessary to make such payments.

(1) Requirements.—An indemnification agreement may be provided under this subsection only if the President determines that each of the following requirements are met:

(A) The liability covered by the indemnification agreement exceeds or is not covered by insurance available, at a fair and reasonable price, to the contractor at the time the contractor enters into the contract to provide response action, and adequate insurance to cover such liability is not generally available at the time the response action contract is entered into.

(B) The response action contractor has made diligent efforts to obtain insurance coverage from non-Pederal sources to cover such liability.

(C) In the case of a response action contract covering more than one facility, the response action contractor agrees to continue to make such diligent efforts each time the contractor begins work under the contract at a new facility.

(5) LIMITATIONS ----

(A) LEASTLETY COVERED.—Indemnification under this subsection shall apply only to response action contractor liability which results from a release of any hazardous substance or pollutant or contaminant if such release arises out of response action activities.

(B) DEDUCTIOLES AND LIMITS.—An indemnific agreement under this subsection shall include deductions and shall place limits on the amount of indemnification to be made available.

(C) CONTRACTS WITH POTENTIALLY ANDVONUULE PAR

enter lato an indemnification agreement with a re-sponse action contractor carrying out a written contract or agreement with any potentially responsible party, the President shall determine an anount which the po-tentially responsible party is able to indemnify the con-tractor. The President may only in the sub-tractor agreement only if the President determines that such amount only if the President determines that such amount only if the President determines that such amount of indemnification is incodequate to cover any reasonable potential liability of the contrac-tor arising out of the contractor's againsmost in per-forming the contract or agreement with such party. The President shall make the determinations in the preced-ing sentences (with respect to the amount and the ada-quacy of the amount) taking into account the total net ameth and resources of potentially responsible parties with respect to the facility of the time of such determi-(i) Decision to inderivity. In deciding whether to

(ii) Conversions.-The Provident may pay a claim (ii) Conversion agreement referred to the claube (i) for the amount determined under clause (i) only if the contractor has exhausted all administration judicial, and common has chausted all administration against all potentially responsible parties participating in the clean-up of the facility with respect to the liabi-ingence in performing the contractor's may define the contractor arising out of the contractor's ma-such party. Such indemnification agreement with such party. Such indemnification agreement shall re-quire such contractor to pay any deductible established under subparagraph (B) before the contractor may re-cover any amount from the potentially responsible party or under the indemnification agreement. (D) RCRA succurrans.-No owner or operator of a facility regulated under the subsection with respect to each facility demnified under this subsection with respect to each facility

(B) Pressons astrained or ansen -A person related or hird by a person described in subsection (cl2018) shall be aligible for indemnification under this subsection only if the President specifically approves of the relating of the president specifically approves of the relating of hiring of such person.
(6) Over ancovrant to this subsection for indemnification of any persons who is a response action contractor with respect to any release or threatened release shall be considered a cast of response incurred by the United States Gavernment with respect to such release.
(7) Racutations. -The President shall promulgate regulations for carrying out the provisions of this subsection. Before promulgation of the regulations, the President shall develop remulations to carry out this section. Development of such guidelines to carry out this action. Development of such guidelines to carry out this section.

(1) Bruy. - The Comptonler General shall conduct a study in the freed year onding Bayemeter 20, 1989, on the application index such a such as a first index and the such of the freed of the freedom of the freedom of the such and the such of the freedom
(3) Reservers Actron contraterios. The term "response action contractor" means—

(4) any—
(5) person who enters late a response action contract with respect to any release or threatened release of a hosendows substance or pollutant or contaminant from a facility and is carrying out such contract; and
(6) person, public or nonprofit private entity, conducting a fold demonstration pursuant to section \$11(b) and

(B) any person who is retained or hired by a person described in subpurgraph (A) to provide any services relating to a response action.
(3) Insurance. The term "insurance" means liability insurance which is fair and reasonably priced, as determined by the President, and which is made available at the time the contractor enters into the response action contract to provide response. action.

for program management meetmotin-() Comparition. -- Response action contractors and subcontinuition be selected in accordance with title IX of the Federal Property and Administrative Services Act of 1949. The Federal selection procedures shall apply to appropriate contracts negotiated by all Federal governmental agencies involved in carrying out this Act. Such procedures shall be followed by response action contractors and subcontractors.

SEC. IM FEDERAL PACILITIES.

(a) APPLICATION OF ACT TO FEDERAL GOVERNMENT.

(1) IN GENERAL. -- Each department, agency, and instrumentality of the United States (including the executive, legislative, and judicial branches of government) shall be subject to, and comply with, this Act in the same manner and to the same extent, both procedurally and substantively, as any nongovernmental entity, including liability under section 107 of this Act. Nothing in this section shall be construed to affect the liability of any person or entity under sections 108 and 107.

(8) APPLICATION OF REQUIREMENTS TO FEDERAL FACILITIES .---All guidelines, rules, regulations, and criteria which are applicable to preliminary assessments carried out under this Act for facilities at which hazardous substances are located, applicable to evaluations of such facilities under the National Contingency Plan, applicable to inclusion on the National Priorities List, or applicable to remedial actions at such facilities shall also be applicable to facilities which are owned or operated by a department, agency, or instrumentality of the United States in the same manner and to the extent as such guidelines, rules, regulations, and criteria are applicable to other facilities. No department, agency, or instrumentality of the United States may adopt or utilize any such guidelines, rules, regulations, or criteria which are inconsistent with the guidelines, rules, regulations, and criteria established by the Administrator under this Act

(3) EXCEPTIONE.—This subsection shall not apply to the extent otherwise provided in this section with respect to applicable time periods. This subsection shall also not apply to any requirements relating to bonding, insurance, or financial responsibility. Nothing in this Act shall be construed to require a State to comply with section 104(cXS) in the case of a facility which is owned or operated by any department, agency, or instrumentality of the United States.

(4) STATE LAWE. -State laws concerning removal and remedial action, including State laws regarding enforcement, shall apply to removal and remedial action at facilities owned or aperated by a department, agency, or instrumentality of the United States when such facilities are not included on the National Priorities List. The preceding sentence shall not apply to the extent a State law would apply any standard or requirement to such facilities which is more stringent than the standards, and requirements applicable to facilities which are not owned or operated by any such department, agency, or instrumentality.

(b) * ** CE. -Each department, agency, and instrumentality of the Uni tes shall add to the inventory of Federal agency hazardous waste facilities required to be submitted under section 3016 of the Solid Waste Disposal Act (in addition to the information reguired under section 3016(a)(3) of such Act) information on contamination from each facility owned or operated by the department, agency, or instrumentality if such contamination affects contiguous or adjacent property owned by the department, agency, or instrumentality or by any other person, including a description of the monitoring data obtained.

(c) FEDERAL AGENCY HAMARDOUR WASTE CONFLIANCE DOCKES. — The Administrator shall establish a special Federal Agency Heserdous Waste Compliance Docket (hereinafter in this section referred to as the "docket") which shall contain each of the following: (4) States

as the "dochet") which shall contain each of the following: (1) All information submitted under section 3016 of the Waste Disposal Act and subsection (b) of this section regarding any Federal facility and notice of each subsequent action inhen under this Act with respect to the facility.

(2) Information submitted by each department, agency, or instrumentality of the United States under section 3005 or 3010 of such Act.

(3) Information submitted by the department, agency, or instrumentality under section 103 of this Act.

The docket shall be available for public inspection at reasonable times. Six months after establishment of the docket and every 6 months thereafter, the Administrator shall publish in the Federal Register a list of the Federal facilities which have been included in the docket during the immediately preceding 6-month period. Such publication shall also indicate where in the appropriate regional office of the Environmental Protection Agency additional information may be obtained with respect to any facility on the docket. The Administrator shall establish a program to provide information to the public with respect to facilities which are included in the docket under this subsection.

(d) ASSESSMENT AND EVALUATION.-Not later than 18 months after the enactment of the Superfund Amendments and Reauthorization Act of 1986, the Administrator shall take steps to assure that a preliminary assessment is conducted for each facility on the docket. Following such preliminary assessment, the Administrator shall, where appropriate-

(1) evoluate such facilities in accordance with the criteria established in accordance with esction 105 under the National Contingency Plan for determining priorities among releases; and

(1) include such facilities on the National Priorities List maintained under such plan if the facility meets such criteria. Such criteria shall be applied in the same manner as the criteria are applied to facilities which are owned or operated by other persone. Evaluation and listing under this subsection shall be completed not later than 50 months after such date of enactment. Upon the receipt of a petition from the Governor of any State, the Administrator shall make such an evaluation of any facility included in the docket.

'. REQUISED ACTION BY DEPARTMENT.-

(1) RIPS.-Not later than 6 months after the inclusion of any facility on the National Priorities List, the department.

Significant New Legislation-Federal Employees Liability Reform and Tort Compensation Act of 1988



UNITED STATES ENVIRONMENTAL PROTECTION AGENC WASHINGTON D.C. 20166

DEC 30 1933

JEFICE OF VENERAL US UNSEL

MEMORANDUM

SUBJECT:	,Significant New Legislation - Federal Employees
	Liability Reform and Tort Compensation Act of 1988
FROM :	Craig Annear
	Associate General Counsel
	Grants, Contracts and General Law Division
то:	General Counsel
	Deputy General Counsels
	nehari generat cognaeta

Deputy General Counsels Regional Counsels Associate General Counsels Assistant General Counsels

On November 18, 1988, the President signed into law the "Federal Employees Liability Reform and Tort Compensation Act of 1988." P.L. 100-694 (copy attached). The purpose of this Act is to protect Federal employees from personal liability for common law torts committed within the scope of their employment while providing persons injured by such acts with a remedy against the United States.

This Act was in response to the United States Supreme Court's decision in <u>Westfall</u> v. <u>Erwin</u>, 108 S. Ct. 580, 98 L. Ed 2d 519, 56 U.S.L.W. 4081 (1988). In We<u>stfall</u> the Court held that Federal employees have absolute immunity from state-law tort suits only to the extent that their actions were within the scope of employment <u>and</u> were discretionary. The Act changes the We<u>stfall</u> standards by requiring that a Federal employee only has to show that he/she was acting within the scope of employment to be absolutely immune from the tort suit.

The Act provides that sult against the United States under the Federal Tort Claims Act. 28 U.S.C. 2671 et <u>seq</u>. is the exclusive remedy for a person injured by the tortious conduct of a Federal employee who is acting within the scope of his/her employment. The Act applies only to common law or state-law tort suirs. It does not cover an alleged violation of the Constitution of the United States or a violation of a statute of the United States under which an action against the individual is otherwise authorized.

The Act authorizes the Attorney General to determine and certify that the employee was acting within the scope of employment. Where such certification is made, an action in Federal court is deemed an action against the United States with the United States substituted for the named employee. Where the action is in State court, following certification the Attorney General is authorized to remove the suit to Federal court and to substitute the United States for the employee. Additionally, the employee is given the right to have the issue of scope of employment determined by the court where the Attorney General refuses to certify. The Act is effective for all claims, civil actions. and proceedings pending on, or filed on or atter N:wember 18, 1988.

If you have questions or require additional information, Ray Spears. of my staff. is available to assist you. Ray can be reached at FTS 382-4548.

Attachment

OSHA Final Rule, Hazardous Waste Operations and Emergency Response (29 CFR 1910)



Monday March 6, 1989

Part III

Department of Labor

.

Occupational Safety and Health Administration

29 CFR Part 1910 Hazardous Waste Operations and Emergency Response; Final Rule

DEPARTMENT OF LABOR

Occupational Safety and Health Administration

29 CFR Part 1910

[Docket No. S-760A]

Hazardous Waste Operations and Emergency Response

AGENCY: Occupational Safety and Health Administration; Labor. Action Final nule.

summary: The Occupational Safety and Health Administration (OSHA) is amending the OSHA standard for hazardous waste operations and emergency response found in 29 CFR 1910.120. This final rule will replace the existing interim final rule required by **Congress** in the Superfund Amandments and Reauthorization Act of 1986 (as amended) (SARA) (Pub. L. 99-499, 29 U.S.C. 655 note). When this final rule becomes effective one year from today, the interim final rule promulgated December 19, 1966 (51 FR 45654) will be revoked. The interim final rule remains in effect until then. The Notice of Proposed Rulemaking for this final rule was published in the Federal Register on August 10, 1967 (52 FR 29620).

This rule will regulate the safety and health of employees involved in cleanup operations at uncentrolled hazardous wasts size being cleaned-up under government mandate, in certain hazardous waste treatment, storage, and disposal (TSD) operations conducted under the Resource. Conservation and Recovery Act of 1978 as amended (RCRA) [42 U.S.C. 6901 et seq], and in any emergency response to incidents involving hazardous substances.

This standard provides for employee protection during initial site characterization and analysis. monitoring activities, materials handling activities, training, and emergency response.

OATES: This final rule will become effective March 6, 1990.

Paperwork authorization has been granted by the Office of Management and Budget (OMB) under control number 1218-0139

ADORESS: In compliance with 28 U.S.C. 2112(a), the Agency designates for receipt of petitions for review of the standard, the Associate Solicitor for Occupational Safety and Health. Office of the Solicitor. Room S-4004. U.S. Department of Labor. 200 Constitution Avenue NW, Washington, DC 20210. FOR FURTHER INFORMATION CONTACT: Mr. James F. Foster, U.S. Department of Labor, Occupational Safety and Health Administration, Division of Consumer Affairs, Room N-3847, 200 Constitution Avenue NW., Washington, DC 20210. 202-523-8151.

SUPPLEMENTARY INFORMATION:

L Background

The U.S. Environmental Protection Agency estimates that approximately 57 million metric tons of hazardous waste are produced each year in the United States.1 These wastes must be treated and stored or disposed in a manner that protects the environment from the adverse affects of the various constituents of those wastes.

In response to the need to protect the environment from the improper disposal of these hazardous wastes. Congress. over the years, has enacted several pieces of legislation intended to control the nation's hazardous waste problem. Federal laws passed in 1965 * and 1970 * initially addressed solid waste disposal. Several other pieces of legislation have been enacted by Congress that have ultimately led to the development of thisrule and they are discussed below.

A. The Resource Conservation and Recovery Act of 1976

.....

The first comprehensive, federal effort to deal with the solid waste problem in general, and hazardous waste specifically, came with the passage of the Resource Conservation and . Recovery Act of 1975 (RCRA) 4. The actprovides for the development of federal and state programs for otherwise unregulated land disposal of waste materials and for the development of resource recovery programs. It regulates anyone engaged in the creation. transportation, treatment, and disposal of "hazardous wastes." It also regulates facilities for the disposal of all solid wastes and prohibits the use of open dumps for solid wastes in favor of requiring sanitary landfills.

There are, however, many hazardous waste disposal sites that were created prior to the passage of RCRA. These sites are often abandoned and contain unknown quantities of unknown westes.

B. The Comprehensive Environmental Response. Compensation and Liability Act of 1980

In response to the need to clean-up and properly reclaim these pre-RCRA

sites. Congress enacted the Comprehensive Environmental Response. Compensation, and Liability Act of 1980 (CERCLA) * commonit known as "Superfund." Superfund established two related funds to be used for the immediate removal of hazardons substances released into the environment. Superfund is intended to establish a mechanism of response for the immediate clean-up of hazardona waste contamination from accidental spills and from chronic environmental damage such as is associated with abandoned hazardous waste disposal sites.

The treatment and disposal of hazardous wastes under RCRA and CERCLA creates a significant risk to the safety and health of employees who work in treatment and disposal operations. Exposure to hazardous wastes through skin contact, skin absorption, and inhalation pose the most significant risks to employees. Employee exposure to these risks occurs when employees respond to hazardous substance or waste emergencies, when they work with hazardous wastes daring storage, treatment and disposal operations or when they participate in clean-up of abandoned-waste site

This risk of exposure and the need for protecting employees exposed to hazardous wastes is addressed in the "Superfund Amendments and Reputherization Act of 1986" (SARA).

C. Superfund Amendments and Regultorization Act of 1988

On October 17, 1986, the President signed into law the "Superfund Amendments and Resuthorization Act of 1986" (SARAL." As part of SARA, in section 126 of Title L Congress addressed the nsk of injury to employees by providing that the Secretary of Labor ("Secretary") issue interim final worker protection regulations within 60 days after the date of enactment of SARA that would provide no less protection for workers engaged in hazardous wasts operations than the protections contained in the U.S. Environmental Protection Agency's (EPA) "Health and Safety Requirements for Employees Engaged in Field Activities" manual (EPA Order 1440.2) dated 1981, and the existing OSHA standards under Subpart C of 29 CFR Part 1928. OSHA published those interim final regulations in the Federal Register on December 19, 1986 (51 FR 45854). A correction notice was published on May 4, 1987 (52 FR 16241).

Environmental Protection Agency. dy's Problem Hazardous Waste at 1 (1980). . U.S. Environ Everybody's Problem Hanaroous www. ¹ Solid Waste Disposel Act. Pub. L. No. **89-372**, 79

^{*} Resource Recovery Act. Pub. L. No. 41-512. M Stat 1427 and Pub. L. 93-14. 87 Stat II.

^{4 42 13.}S.C. 6901 at seq.

^{+ 42} U.S.C. 9001 et seg.

⁺ Pub. L. 99-499.

With the exception of a few provisions that had delayed start-up dates. OSHA's interim final regulations became effective on December 19, 1988 in accordance with section 126(e) of SARA. and apply to all regulated workplaces until the final rule developed under sections 126 (a)-(d) becomes effective.

Section 126(a) of SARA provides that the Secretary shall " * * pursuant to section 6 of the Occupational Safety and Health Act of 1970, promulgate standards for the health and safety of employees engaged in hazardous waste operations." These standards must be promulgated within one year after the date of enactment of SARA. This notice completes the development of those standards by issuing a final rule based upon the proposed regulations as indicated in sections 126(a) and 126(b) of SARA.

Pursuant to section 126(c) of SARA. the final regulations issued today are to take effect in one year. Section 126(c) also provides that the final regulations are to include each of the worker protection provisions listed in section 126(b), unless the Secretary determines that the evidence in the public record developed during this rulemaking and considered as a whole does not support inclusion of any such provision. A discussion of the public record for this rulemaking and the changes made to the proposed regulations issued August 10, 1987 follows.

This final rule has been adapted from the language of the proposed rule. Changes have been made to address more fully the provisions which Congress directed the Agency to cover and the comments made in the public record. OSHA utilized several sources for the proposal. These included the EPA manual entitled "Health and Safety **Requirements for Employees Engaged in** Field Activities" (1981), the language of OSHA's safety and health standards in Subpart C of 29 CFR Part 1928 and various documents issued either jointly or separately by the EPA, OSHA, the U.S. Coast Guard, and the National Institute for Occupational Safety and Health (NIOSH).

OSHA specifically used the joint OSHA/EPA/USCG/NIOSH manual entitled. "Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities" (Preamble Reference 6), as an outline in preparing the interim rule and the proposed rule. This manual was developed as a result of the collaborative efforts of professionals representing the four agencies. These professionals, who are knowledgeable in hazardous waste operations, worked with over 100 experts and organizations in the development of the criteria contained in this manual. The manual was published in October 1985 and is public information. The manual is a guidance document for managers responsible for occupational safety and health programs at inactive hazardous waste sites. The manual is intended for use by government officials at all levels and contractors involved in hazardous waste operations. The manual provides general guidance and is intended to be used as a preliminary basis for developing a specific health and safety program for hazardous waste operations. Further, the major subject areas listed in section 126(b) of SARA are nearly identical to the major chapters in the manual.

Based upon the extensive public comments and hearing testimony. OSHA has modified the proposal. The final rule takes into account the entire record. In addition, the language of this final rule clarifies some areas of confusion in the interim rule that OSHA has identified during the public comment period and since the promulgation of the interim final rule. The final rule also reorganizes some of the sections to clarify the standard.

D. Regulatory History

The Superfund Amendments and Reauthorization Act of 1986 (SARA) gave the Secretary of Labor 60 days to issue interim final regulations which would provide no less protection for workers employed by contractors and emergency response workers than the protections contained in the **Environmental Protection Agency** Manual (1981) "Health and Safety **Requirements for Employees Engaged in** Field Activities" and existing standards under the Occupational Safety and Health Act of 1970 found in Subpart C of Part 1926 of the Code of Federal **Regulations. Those interim final** regulations were to take effect upon issuance and would apply until final regulations became effective (SARA, § 126(e)). OSHA issued its interim final regulations on December 19, 1988 (51 FR 45654).

SARA also instructed the Secretary of Labor to promulgate, within one year after the date of the enactment of section 126 of SARA and pursuant to section 6 of the Occupational Safety and Health Act of 1970, standards for the health and safety protection of employees engaged in hazardous waste operations (SARA, section 126(a)). On August 10, 1987 OSHA issued a Notice of Proposed Rulemaking and Public Hearings (52 FR 29820). That Notice set forth OSHA's proposed language for its final rule and announced public hearings that would be held to gather further information to aid the agency in developing its permanent final rule.

Informal public hearings on the subject of this rulemaking were scheduled and held to afford interested parties the opportunity to comment on OSHA's proposals. The hearings were held October 13–16 and 20–21. 1987 in Washington, DC and October 27–28. 1987 in Seattle, Washington. The hearings originally scheduled for San Francisco. CA in the August 10. 1987 Notice of Proposed Rulemaking were rescheduled for Seattle, WA in an October 13, 1987 announcement (52 FR 37973).

Testimony from over 40 witnesses was presented at the hearings. Further, over 30 post hearing comments were submitted to the record of this rulemaking. In addition to the public hearings and the testimony received in response to those hearings, OSHA received over 125 written comments on its proposed language for a final rule.

II. Summary and Explanation of the Standard

Paragraph (a)—Scope, Application, and Definitions

1. Scope. OSHA proposed to define the scope of this final rule in paragraphs (a)(1) and (a)(2). "Scope" defines the specific worker populations to be covered by this rule.

The scope of this rulemaking has been an issue during the development and _ promulgation of the final rule. OSHA requested specific comment on whether the proposed rule was appropriate.

Eastman Kodak's comment (10-38) states. "The preamble of the proposed standard at page 29622 requested 'specific comment on whether [OSHA's] interpretation of scope is too broad or too narrow.' The scope of applicability of the standard, especially with regard to ongoing operations at hazardous waste management facilities regulated under RCRA and/or corresponding state programs, appears to be appropriate."

While the language of the final rule is somewhat different from the language of the proposed rule, the four major areas of scope remain essentially the same. These four areas of scope include (1) clean-up operations at uncontrolled hazardous waste disposal sites that have been identified for clean-up by a governmental health or environmental agency, (2) routine operations at hazardous waste treatment, storage and disposal facilities or those portions of any facility regulated by 40 CFR Parts 264 and 265, (3) emergency response operations at sites where hazardous substances have been or may be released, and (4) corrective actions at RCRA sites. In addition OSHA has clarified that the agency intends to cover voluntary clean-ups at government identified sites.

OSHA's proposal addressed the three specific populations of workers at the above operations. First, it was proposed to regulate those operations where employees are engaged in the clean-up of uncontrolled hazardous waste sites. These operations include those hazardous substance response operations under the Comprehensive Environmental Response. Compensation, and Liability Act of 1980 as amended (CERCLA), including initial investigations at CERCLA sites before the presence or absence of hazardous substances has been ascertained, those major corrective actions taken in cleanup operations under the Resource **Conservation and Recovery Act of 1978** as amended (RCRA), and those hazardous waste operations at sites that have been designated for clean-up by state or local governmental authorities.

The second worker population proposed to be covered included those employees engaged in operations involving hazardous waste treatment, storage, and disposal (TSD) facilities regulated under 40 CFR Parts 264 and 265 pursuant to RCRA, except for small quantity generators and those employers with less than 90 days accumulation of hazardous wastes as defined in 40 CFR 262.34.

The third and final worker population proposed to be covered were those employees engaged in emergency response operations for releases or substantial threats of releases of hazardous substances, and postemergency response operations to such releases at all workplaces.

in paragraph (a)(1)(i) of the final rule OSHA is regulating all government mandated clean-up operations at uncontrolled hazardous waste disposal sites. These operations were included in paragraphs (a)(1)(i) and (a)(1)(iii) of the proposal. For the purposes of this final rule, "Superfund" and other uncontrolled hazardous waste disposal sites include hazardous substance response operations at sites regulated under 40 CFR Part 300, Subpart F: RCRA closure activities conducted under 40 CFR Part 285, Subpart G; and those similar uncontrolled hazardous waste disposal sites that have been designated for clean-up by Federal, state or local governmenis.

OSHA intends and the change in language clarifies that all government mandated clean-ups are covered. These include not only sites on the various . "Superfund" lists, but also all other government mandated clean-ups as well. The changed language makes clear that such clean-ups are covered whether or not they are financed by the government. The language further clarifies that clean-ups mandated by any level of government are covered.

In paragraph (a)(1)(ii) of the final rule, OSHA is regulating corrective actions at RCRA facilities. This paragraph adopts the language proposed in paragraph (a)(1)(ii) of the proposal with one change. The term 'major' has been deleted as a modifier of "corrective action." Several commenters requested clarification of the term "major corrective action." International Technologies, a major hazardous waste clean-up contractor, requested in their comment (10-44), "Please clarify major corrective actions conducted under RCRA.' What distinguishes 'major' corrective actions from other corrective actions?" The State of Indiana commented (10-23). "There is no definition of what constitutes a 'major corrective action' under RCRA." In addition, the term "major" is not used in EPA terminology. "Corrective action" is a term unique

to RCRA and has been defined for use with RCRA. OSHA's addition of the modifier "major" raised many definitional questions. Therefore OSHA. in the final rule, is deleting the word "major" to be consistent with EPA terminology and eliminate confusion. Rather than define "major corrective action." OSHA is amending the language of the proposal to include a phrase describing the level of corrective action that is to be regulated in the scope of this rule. OSHA will be regulating those corrective actions that potentially expose employees to a 'safety or health hazard." OSHA is not concerned with those corrective actions that are intended to abate environmental risks without exposing employees to safety or health hazards The phrase "safety or health hazard" in the introductory language is the phrase that OSHA has used to differentiate the type of releases that this standard regulates versus those release that may pose only environmental threats rather than safety or health threats to employees.

OSHA has decided to add a new paragraph (a)(1)(iii) to the final rule that would include within the scope of this rule those voluntary clean-up operations conducted at sites recognized by governmental bodies as uncontrolled hazardous waste disposal sites. All other voluntary clean-ups would be exempt from 29 CFR 1930.120. OSHA does not have the statutory responsibility to identify hazardo waste sites. It will leave to agencie. with that authority the responsibility to identify those sites. Those voluntary sites that are not recognized by the government as uncontrolled hazardous waste disposal sites would be exempt from 29 CFR 1910.120, however, they would still be regulated by the other OSHA general industry or construction Industry standards applicable to the work being performed at the site.

OSHA did not propose to cover voluntary clean-ups of hazardous substances in its proposed rule. Many comments suggested this, however, the Agency has concluded that individuals involved in voluntary clean-ups may be exposed to the same safety and health risks at voluntary sites identified by the government whether or not the government is compelling action. However, it would be difficult to k...w whether or not sites not identified by the government are hazardous waste sites without a structured evaluation system for such potential sites.

OSHA raised an issue on the scope in the preamble to the proposal that generated several comments. On page 29622 of the preamble to the proposal, OSHA listed several TSD facilities thr would not be covered by the final rul The exemptions were taken from a lia published by the U.S. EPA that are not directly regulated by U.S. EPA. However, the proposed standard's language did not grant these exemptions. Comments did not support the exemptions and OSHA did not believe that they were appropriate.

The particular exemption that generated the most comment exempted those TSD facilities which operate under a state bazardous waste program pursuant to RCRA section 3006. These state hazardous waste programs are recognized by U.S. EPA in a similar fashion to the OSHA state plan states under section 18 of the OSH Act. A number of commenters, such as the State of Indiana (10-23), objected to this type of exemption by OSHA as not being appropriate. They stated OSHA jurisdiction should not be impacted by U.S. EPA state agreements, but only those state agreements provided in the OSH Act. OSHA agrees with these commenters and therefore OSHA jurisdiction will be delegated to only those states which OSHA has formal agreements with under the OSH Act. However, it should be noted that the U.S. EPA jurisdictions under SARA section 125 may make use of their state agreements.

9296

Other commentars. EXXON (10-33) and CONOCO (10-32), suggested that OSHA incorporate the exceptions on page 29822 as a separate paragraph in the final rule.

Typical TSD facilities range from the hazardous waste generator with a hazardous waste storage area to the large, complex hazardous waste disposal facility. SPA estimates that approximately 80 percent of all generators also treat, store, or dispose of their hazardous wastes and thereby qualify as a TSD facility. Over 30.000 TSD facilities notified EPA in 1980 that they would qualify for regulation under section 3004 of RCRA.

OSHA continues to regulate RCRA TSD facilities in paragraph (a)(1)(iv) of the final rule as it was proposed in the regulatory language of the proposal. The list of exemptions on page 29622 will not be incorporated into the final rule. OSHA believes that such a list would create too great a gap in the protection of workers. For example, with respect to workers at TSD facilities operating under a state hazardous waste program pursuant to RCRA section 3008, OSHA agrees with a comment made by the State of Indiana (10-23) that it is possible that the workers in those 42 authorized states identified by Indiana could be without the protections. mandated by Congress

In paragraph (a)(1)(v) OSHA would continue to regulate emergency response operations for releases of, or substantial threats of releases of, bazardous substances without regard to the location of the operation as proposed in paragraph (a)(2) of the proposal. Such emergency response operations are not limited to those responses at uncontrolled hazardous wasts disposal sites or RCEA TSD facilities. With respect to transportation incidents, responders to the scene are covered but operators (i.e., track drivers and train crews) are not covered unless they become actively involved in the response action.

OSHA is making major revisions to proposed paragraph (1). These revisions have been made in response comments concerning OSFIA's involvement in regulating emergency response at every site involving hazardous substance release or potential release. Some of the comments were in favor of OSHA's continued involvement with emergency response (i.e., American Chemical Society, 10-44) and others were opposed to continued involvement (i.e., ECOLAB, 10-64). Others supported OSHA involvement in emergency response activities at uncontrolled hazardous waste sites and certain RCRA facilities but opposed the

agency's involvement with non-waste clean-up or non-RCRA facilities (i.e., The Chlorine Institute, 10-24). Yet others called for two separate areas in the rule: one for hazardons waste operations, and one for emergency response (i.e., Allied Signal, 10-38). Others opposed coverage of emergency response to petroleum spills (CONOCO, Ex. 10-32).

OSHA after reviewing all the comments, continues to believe that it is the clear intent of Congress that any employees participating in an emergency response to the release or potential for release of hazardous substance be covered by this rulemaking. This Congressional intent applies to all such emergency responses including those both off and on hazardous waste sites.

The statutory language indicates that all emergency responses where the threat of hazardous substance spills exist are to be covered.

Section 128(b)(11) of SARA specifically provides that "requirements for emergency response" are to be included and is not limited to hazardous waste sites.

In addition, section 126(d)(4) states:

Training of Emergency Response Personnel.—Such training standards shall set forth requirements for the training of workers who are responsible for responding to hazardous emergency situations who may be exposed to toxic substances in carrying out their responsibilities. (emphasis added)

This is very broad language that is not limited to hazardous wasts operations or hazardous wastes or substances on CERCLA or RCRA sites. It covers all "hazardous emergency situations" for all "toxic substances" which would clearly cover all types of emergency response for chemical spills including chemical tanker spills and the like. It should also be noted that once a tank truck spills a toxic chemical in an emergency it creates a hazardous waste in the very real sense.

Further, the grant provision of the statute clearly indicates that grants can be made to train workers for emergency response at any location, not just on hazardous wasts sites.

Section 128(g)(1) states:

Grant Purposes.—Grants for the training and education of workers who are or may be engaged in activities related to hazardous waste removal or containment or emergency response may be under this section. (emphasis added)

Other statutory sections also indicate the legislative intent to cover all emergency responses where hazardous chemical spills are possible.

In addition to the statutory language, the documents cited by Congress as the minimum guides for OSHA to use in developing this rule refer to all emergency responses. The EPA manuel and the OSHA construction standards referred to in the statute require preparations and planning for emergencies generally, not just for hazardous wasts site emergencies.

In addition the legislative history indicates that Congress intended Section 126 to cover emergency response to all situations where spills of hazardous chemicals were a possibility and not just emergency response on hazardous waste sites. For example, Senator Hatch stated:

This amendment will address the concerns that have been reised that the Department of Labor issue standards for employees engaged in hazardous waste operations, as well as emergency response. (9/26/88 Cong. Rec. pg. S-12031)

As discussed elsewhere in this preamble OSHA believes there is a clear need for training and other provisions to protect workers engaged in all emergency responses when there is the possibility of hazardous substance spills. This is needed whether or not the emergency occurs on a hazardous waste site. The agency believes that the hazards are the same in these cases.

Pinally, other parts of SARA, in particular Title III, address emerge 1000 response actions and planning by communities and local governm employers outside of the hezardous waste clean-up operation. The Congressional concerns on toxic emergencies also discussed in Task Force on Toxic Emergencie Environmental and Energy Study **Conference Special Report, Sep** 18, 1988. This report stresses the need for training of emergency response personnel as well as emergency response planning and related areas. This was part of the legislative research which led to the passage of section 128 of SARA

OSHA's final rule rulemaking divides emergency response into three separate areas. First, OSHA is regulating emergency response by employees at uncontrolled hazardous wasta sites in paragraph (I) of the final rule. This paragraph contains the requirements that were in paragraphs (1)(1) and (1)(2) of the proposel and the interim rule. These regulations applied to the "onsite" operations of the interim rule. Second, OSHA is regulating emergency response at RCRA facilities in paragraph (p)(8). This paragraph contains the requirements that were in paragraphs (1)(1) and (1)(3) of the proposal and interim rule. These regulations applied to the "off-sits"

operations of the interim rule. Third, OSHA is regulating emergency response to harandone substance missage by simployees not covered by yaragespie (1) and (p)(6) in paragraph (q). Peragraph (q) contains the requirements proposed in peragraphs (1)(1), (3), (3), (4), and (1)(5) of the proposal and interim rule. These regulations were directed toward emergency response barns, industrial fire brigades, and hazardous materials

emergency response to releases of hazardous substances. The agency did not propose to limit emergency response to uncontrolled hazardous waste sites but decided instead to propose to cover all emergency response whether it was done at uncontrolled hazardous waste sites or anywhere else, including petro-chemical and similar manufacturing aciities. In its proposal OSHA covered

OSERV's decision to propose coverage of all emergency response was based upon the high tisk associated with emergency response by uniralized and unprotected employees and the need for proper training and equipment to be provided for emergency response to baserdous substance releases. In testimony during the public hearings on the released for emergency response to baserdous substance releases. In testimony during the public hearings on this release of baserings on the pervens who respond to spills and accidental release of baserings chemicals; and these personals, have had the least protection in terms of chamical emergency response safety and health plane, training and equipment." (Tr. pgs. 24–25). Mc. Busser goes on to stata. The real strength of 25 CFR 1910.120 is that it not only prevides for a more consistent and thercogh approach to protecting workers involved in hazardone weste operations, but also for personnel who face extremely high risk to life and health that's associated with chemical emergency response." (Tr. **be** 251

Another witness, Mr. Ray Simpson, one of OSHA's expect witnesses on fix suppression. fire inspection, and training, testified. "I like to support any concept that advocates properly equipping, training, and supporting emergency responders. When I talk about an emergency responder. I'm not talking simply about fire fighters although that's basically my supertise. I'm talking about the emergency modical technicians, the people who handle the victims. I'm talking about the police officer who, many times, is first on the scene before any of us get there: the many who really must make, in some situations, the initial decision about

what's going to happen. I have learn over those many years that the two greatest dangers that face us as 8

anagement areas areas areas in proceedures that we're is facing and the lack of plan or any proceedures that we're is properly. Associates the optimization of Labor Competence of Industrial Organizations (AFL-GIO) also used base and the participation of the hours of the AFL-GIO is bearing of the Generalizero on European and Houring of the Generalizero on European and Houring of the AFL-GIO is bearing before the Hours of Labor Competitions on the issues of comparison on the issues of comparison on the issues of the AFL-GIO is bearing on the issues of the AFL-GIO is bearing of the AFL-GIO is bearing of the AFL-GIO is bearing on the issues of the issues of the Houring of the AFL-GIO issues that had nearly and bear explored on the issues of the problems further and plane. New vertices is both Supediad opperstores multices and BGMA operations and the issues of the problems further and plane. They beer provide a such that the issues of our findiphter's under worker that the issues of our findiphter's under social and the issues are also and other accidential releases. They beer provide all which the issues is a substantiate of the issues are also as a substant of the issues and experiment is the issues of the issues and other accidential releases. They were explicitly is the issues are also as a substant of the issues are also as a substant of the issues are also as a substant of the issues and experiment is the issues are also as a substant of the issues and experiment is the issues are also as a substant of access and experiment is the issues of SARA, as well as its is and experiment is the issues of SARA, as well as indicating experiment is a substant of the proposed rule. The issues of a substant of accessing of the issue of the provection of accessing and issues that the issues of the issues of the provection of accessing and issues that the issues of the issues area and experiment is the issues of the provection of a substant of the issues and experiment is the issues aread accessing the issue and any

once it is not contained) is in the consect sense meaning a harmadous waste operation." "This interpretation is reinforced by the fact that SARA is a free-standing statutory provision and not an amendment to CERCLA. The class Congressional intent then is to provide protection to employees whenever they deal with harmadows wastes."

In addition section 128(6)(2) discussing training for emergency response personnel utilizes the very broad term "hazardous emergency response training. Section 128(6)(2) indicates that training gasarts may be given independent concept separate from basardous waste neuronal operation. Fo these and other resons OSHA believes entergency response to basardous response to the resons OSHA believes action 128 is intended to corre-emergency response to basardous response and other resons OSHA believes action 128 is intended to corre-emergency response to basardous response and other resons OSHA believes action 128 is intended. In a construc-possibility of engaging in emergency in a sector the resonance." A composed to basardous who respon-response supported by this final rule to the action makes in atom substances are covered by this final rule to the action that they are exposed to basardous substances. State and local government employees in atom substances are covered by the final rule to the action that they are exposed to basardous substances. State and local government employees in atom substances to be commute the instant of Congress with the acops of the proposed rule. Many of these comments that OSHA has ecceeded to the train and to of the final rule. There is no premulgate standards for the health to promulgate standards for the health to promulgate standards for the health to apprece native of some of the sense of the section of employees engaged in hazardous waste operation-trad(a) of SARA is to directive to OSHA to permissive the permulgate standards for with respect to emergency response in directive to promulgate standards for-with respect to emergency response of permission. "However, other com-rectivities onuside of hazardous a operation." However, other com-tained from the periodemical

industry support, on a limited basis. OBHA's decision to cover emergency response with the scope of the standard CONOCCO's comment (10-32) is representative of this point of view. CONOCCO states. "Connoco's primary concern with the proposed rule centers on the extremely broad scope of employee coverage under this standard and compared to Congress' intent to cover "hezardons waste operations and emergency response.' We believe that to cover employees engaged in hazardous waste operations and emergency response to these operations on a full-time basis." While this comment would seem to support OSHA's coverage of employees engaged in this and the numers operations. Based upon public testimony and written comments received into the record of this rulemaking. OSHA has concluded, that because of the high risk associated with emergency response and the number of these incidents occurring that coverage of workers conducting such appropriate and necessary.

highest exposures to hazardona substances over a longer period—would be covered by virtually all the previsions of this final rule. Employees exposed to hazardona waste operations, who are regularly exposed to hazardons wastes but in a more controlled environment, would be covered by the more limited requirements of peragraphs (p) and (q). Emergency response workers, exposed usually for short periods to often unknown but possibly high levels of hazardons substances, would be regulated by paragraph (q). 2. Application, OSHA proposed to define the application of this final rule in paragraph (e)(3) of OSHA's Notice of Proposed Rulemaking (NPRM) published on August 10, 1987 (52 FR 29620). "Application" establishes which regulations within this rule apply to the specific worker populations to be protected by this rule. In paragraph (a)(3)(1) OSHA proposed that the employer would have to comply OSHA believes that the scope of this final rule carries out the integet of Congress and is consistent with good occupational safety and health policy. Employees performing clean-up operations under CERCLA, RCEA (corrective actions) and state or local government designated sites—generally those employees likely to have the highest exposures to baserdous

should apply to ensure employee safety and health. Other OSHA standards regulate many other hazards, and OSHA wants to make clear that the other standards continue to apply. Also, OSHA proposed that hazardons waste operators who are not within the scope of this standard should continue to be regulated by the Parts 1910 and 1923 standards. OSHA is keeping those provisions in the final rule for the reasons stated with minor editorial changes for clarification. In paragraph (a)(3)(3), OSHA proposed that all paragraphs of section 1910.120 except paragraph (c) would apply to hazardons waste operations at CERCLA sites, at major corrective action at RCRA sites, and at sites designated for clean-up by stats and local governments. Paragraph (c) of the proposal addressed cartain operations conducted under the Resource (RCRA). requirements specifically covered in the proposed rule. If there were a conflict or overlap between standards, it was proposed that the more protective provisions would apply. Since this rule does not cover all of the bazards present at hazardous waste operations, other OSHA standards in Parts 1910 and 1925 with the standards in 29 CFR Parts 1910 and 1928, as well as with the

("According to an application of the types of sites that the second from the second from the second from the speaker than the second sequence of the types of the day operations of an EPA licensed TSD facility. OSHA has made two editorial changes in its proposed language in paragraph (a)(3)(3) without changing the intent of the paragraph. First, rather than referring to each of the types of sites individually, OSHA is making references to the scope paragraph (a)(1)(1) through (a)(1)(11) to identify the sites that this application paragraph addressed. The sites to be addressed remain the scope paragraph because the codification of paragraph bas changed in this final rule down to changes made to the proposed. Second because the codification of paragraph bas changed in the proposed. Second because the codification of paragraph bas changed in the final rule the paragraph (c) will apply specifically to hazardous waste operations at RCRA sites which are involved in trastment, storage, disposed and handling of hazardous waste (p) are discussed later in the preamble.

In paragraph (a)(3)(iii), OSHA proposed that the requirements set forth in paragraph (o) of section 1910.120 would apply specifically to the

of the term. The exclusion was available to these operations depending upon the employer's decision to provide or not provide emergency response by employees to releases of, or substantial threats of releases of, barardous regulations for cartain small quantity generators and less than 90-day accumulators, such as dry cleaners and gas stations, which come within the purview of RCRA, but are not bazardous waste operations in the normal meaning sites which are involved in treatment storage, disposal and handling of hazardous waste. The proposal contained a limited exclusion from these hazardous waste operations at RCRA

substances. OSHA proposed to exempt small quantity generators and less than 90 day accumulators from all parts of the rule of they did not provide emergency response by their employees to releases of, or substantial threats of releases of, hazardows substances. OSHA further hazardows substances. OSHA further

proposed to eccempt small quantity generature and less than 50 day accumulators from all parts of the rule emergency response by their employees to releases of, or substantial threats of releases of, hazardous substances. OSHA recognized that many small quantity generators are smaller balaneses with limited employee populations. Since most of these establishments rely on the emergency response services of local fire and record departments. OSHA is providing a complete eccemption from these proposed standards when the employer can show that employees are not required or encouraged to engage in energency response, but are directed in the case of emergency spills of hazardous substances to maintain a safe distance and to call local fire or other emergency response organizations. In cases where such establishments do provide emergency resiblishments do provide emargency response by employees, and thereby expose employees to bazardous substances. OSHA proposed that such employers meet the emergency response requirements of paragraph [1] of this proposed rule. OSHA concludes its proposed rule. These businesses are not proposed rule. These businesses are not proposed rule. The osha rule. The o

engaged in hazardous waste operations as that term is conceived of normally. In addition, it is not believed that Congress intended such businesses to be covered. They do not present the relatively high

expression to a sampley of hearedous that the expression of the typically difficulty oppily to the work conducted by encrypting apply to the work conducted by encrypting response performed in the discussion of encrypting terms and perform a work conducted in the perspect performance of the encrypting of the perspective local perspective of the encrypting of the perspective should be and an encrypting of the perspective of the encrypting of the perspective of the encrypting of the perspective of the encrypting of the perspective should be and an encrypting of the encrypting of the perspective should be and an encrypting of the encrypting of the perspective should be and an encrypting of the en

amergency response operations. 3. *Definitions*. In puregraph (a)(4). efinitions. OSLIA proposed to identify ad define the various terms used in this

relevanting that may cause confusion.
However, the following new ordinations have been acided as a vesit of communication of resulting another in the neuron of resulting another in the following new been acided as a result of the following proves and the following of resulting another of communication of whether a result of the following in th

RELA. Mr. Thomas Seymour-directed the following questi Captain Lemen (Tr. pg. 195). stion 10

[Mr. Seymour:] We have necessed some feedback in our record shout the order of 1 hierarchy theit we have used for permussion exposure listit. There have been statement made that the RELs are us per-verviewed, developed. I wonder if you might describe us how the RELs have been developed by 1 National lawiture for Occupational Safety and Health.

Captain Lemen's response (Tr. pgs 195–197) to Mr. Seymour war

Okray. In suppose to the personner productive many presenters. The first process of developing an RE. at Netdonaid has the process of developing an RE. at Netdonaid has the process of developing an RE. at Netdonaid has the process of developing and Handh's to to protocomparations but a process of the proc

In Hight of these consumers OSHA has concluded that the NIOSH REL's have undergone the necessary paer review to be included in the standard's hierarchy of limits. The term "established exposure levels" was defined in the proposal to indicate the levels which would try medical surveillance of the exposi employees. The term included not OSHA established PELs. but also

exposure limits suggested by NIOSH and ACCIH. After review of these and other comments. OSHA concludes that it is appropriate to go beyond the OSHA established PELs in triggering modical surveillance. First, medical surveillance is appropriate for workers exposed to toxic chemicals other than these covered by the PEL's. Second, because of the broadly-worded language in section 128(b)(3), which requires medical surveillance for workers engaged in hazardous waste operations which would expose than to toxic substances." Some of these "toxic substances." are not included in the OSHA PEL. When OSHA completes its rulemaking on the at contamination proposal (PEL's project), there will be fewer toxic substances not covered by PEL's. But in light of Congressional language and the large number of hazardous chemicals present in an uncontrolled basardous warts esite, OSHA concludes that this definition is appropriate to protect employee safety and health.

The term "permissible exposure limits" was defined in the proposal as the inhalation or darmal permissible exposure limit specified in 20 CFR Part 1910. Subpart Z. As a result of the comments received is the record. OSHA has annended its definition for "permissible exposure limits." OSHA has amended the definition for "permissible exposure limits" to include a reference to Subpart C of Part 1910. It now includes both Subpart Z hasili hazards and those requirements in Subpart G of Part 1910. "Free, OSHA has changed the term "established exposure level" to the term "published exposure level" to the term "published exposure level" to the term "published exposure level" to the specified the term "WOSH Recommendations for Occupational Health Standards" dated 1968. Incorporated by reference, or if none is specified, the termediate published in the standards specified by the American Conference of Governmental Industrial Hygismists in their publication "Threshold Limit Values and Biological Exposure indices for 1967–86" dated 1967, incorporated by reference. Third, the provisions of (10,2) on medical surveillance have been changed to cover ovenexposures to both PEL's and if none, then over-exposure to published exposure limits. OSHA concludes that with these changes the definitions are clear, comprehensive and carry out both statutory directives and appropriate medical criteria in determining whether medical

surveillance is required. Some commenters stated a broader guide is necessary for respirator use and that is discussed under paragraph (g). OSHA requested comment on the appropriateness of its definitions of bazardous weste, bealth hazard and bazardous substance and whether they were consistent with EPA and DOT practice. Several comments were received on these issues. One set of comments criticized OSHA's incorporation of petroleum and petroleum products in its definition of

bazardons substances. A typical comment was made by EXXON (10-53). In their comments EXXON presented the following discussion:

Perhaps the most fundamental mainterpretation contained is this rule is the inclusion of petroleum and petroleum substance. As discussed in Comment EAAs, below at pages 11 to 16 [Internal ECCION comment references]. Congress, the Environmental Protection Agency (EPA), and the Department of Transportation (DOOT) have uniformly recognized the inappropriatement of characterizing petroleum is a hararchose tubetance. There is no indication in SARA Section 126 that Congress intended to charge the petroleum outdation or to subject petroleum releases to emergency response regulation. EXXON further stated:

It is EXCOM's understanding that e situation is not an ensurgency response subject to the requirements of paragraph (1) unless there is a relaxest of paragraph (1) unless the second second second second definition of "hararches substance" to accurate and correct: The proposed definitions of "hararches materials" under GCR 1972. By so deing perceives and paradeum products have been inducted as hararches for energyner response operations. Congress, is the very CERCA sections the federal definitions of "hararches substance". The products are subject to the bardeneous requirements for energyner response operations. Congress, is the very CERCA sections the federal definition of "hararches substance," EPA regulations under CERCA have theoryperated this congressional directive. See 60 CER Part 301 and discussion at 30 FR 19348, 19469 (April 4 1946), DOT has specifically recognized this Federal petroleum exclusion and incomponented the substance," See 60 CER Part 301 and discussion at 30 FR 19348, 19469 (April 4 1946), DOT has specifically recognized this Federal petroleum exclusion and incomponented the substance," See 60 CER Part 301 and discussion at 30 FR 19348, 19469 (April 4 1946), April 4 1946). As and, the proposed OSHA definition is inconstituent with the CERCIA, EPA and DOT definitions of "hararchese enditions at 40 CFR 1974. The stude DOT definitions at 60 CFR 1974. The stude DOT definitions at a final barachese the DOT definitions at a final ba

hazardom wastes. As noted above, the DOT definition of hazardom substance at 49 CFR 171.8 should properly be incorporated in the proposed OSHA definition of substance. It is not a waste definition. Therefore, the proposed definition of hazardom waste should be limited to waste materialic and, the DOT definition of hazardom substance should be dearly excluded.

COSTA does not egree with thee argument. Section 120 of SARA is directed to protecting workers from the basards, and parameters equilibration perroleum products creats equilibration perroleum products.
During the question of petroleum and petroleum products.
During the question of petroleum and petroleum products.
During the question of petroleum and petroleum products.
During the question (Tr. pg. 83); 'To you feel that medical monitoring for these types of products [petroleum products] is appropriate?
Dr. Chase responded. 'Petroleum products is just too bread a term for me to answer that in a general very. Cartain petroleum derivatives are more toxic than others. Some about chronic toxicity, others enhances toxicity, and others, the concern is more about chronic toxicity of the individuals who teerified if petroleum and petroleum products) assertion was the testimenty of the petroleum and petroleum.
Mc. Gregory Noll, the Heisardou of SHA panel advected on the testimenty of the futurenational Association of Fire petroleum and petroleum products of SHA panel advected on the tester of basertia. Weshington Fire Department, the International Association of the petroleum and petroleum products of the prime of the responses and to this question for the perspective of the international Association of the petroleum and petroleum products of the prime of the responses are involved with flammable diversed for. Noll the searchous outpetroleum and petroleum products of pase 3, that you indicate that st least 50 percent of your responses are involved with flammable liquids end gases you find. The auting in this ruleusing for hazardous outpetroleum in the ruleusing for hazardous outpetroleum in the ruleusing for hazardous outpetroleum and petroleum gases you find.
Mc. Noll responded. 'I think realistically, with HAZMATs being the busi-sent or yve been successfully

commodities have been thrown into the hazardous materials field.

"We now regard them in the hazardous materials field from a practical perspective."

Mr. Thomas Seymour of the OSHA panel asked Mr. Richard Duffy of the International Association of Fire Fighters (Tr. pg. 110), "Mr. Duffy, we have had some previous commenters who have advocated that petroleum and petroleum products be excluded from the scope of the standard.

The example that you just gave about the propane tank inside the building exploding and killing fire fighters, what is your opinion about whether we should exclude petroleum products from this standard?"

Mr. Duffy responded: "I don't know how we would classify them. I would object to that. I mean. I don't know how to better qualify-I could talk to you for days about incidents involving petroleum products. I don't see any reason to exclude them any more than excluding the oxidizers or any group. I mean, you could pick lots of products and ask to exclude them. And I'm sure a lot of the lobbying entities can establish reasons for it. But I can't see any in terms for fire fighters."

Mr. Charles Gordon of the Department of Labor's Office of the Solicitor and a member of the OSHA panel asked Captain Richard A. Lemen, Director of the Division of Standards Development and Technology Transfer of, NIOSH the following question (Tr. pg. 200-201): "In the case of spills of petroleum or petroleum products in either an emergency response situation or as a hazardous waste dump were there are petroleum products as one of the major contaminants, is it appropriate for all the provisions of the OSHA standard or the recommendations to apply in those circumstances?"

Captain Lemen responded, "We believe it is appropriate and they should apply in those circumstances, as well."

Mr. Seymour also asked Deputy Chief Roger Ramsey of the Seattle Fire Department (Tr. pg. 142): "I gather from what you have also said that the definition we have, including the DOT hazardous material definition for hazardous substance and materials is appropriate, and that we should not exclude petroleum products from the coverage of this standard?"

Deputy Chief Ramsey responded,

"Absolutely not." Many spills and emergency response to these spills involve petroleum products. These spills present both health and safety risks. Training is necessary to protect employees who respond to petroleum spills as with

other spills. In fact, these are usually the same employees.

OSHA concludes that it is crucial to cover responses to petroleum spills as well as all other spills because petroleum products constitute a substantial threat to employees responding to accidental releases of these substances. Many petroleum products present health hazards as well as fire and explosion hazards. In addition they often contain fractions which present high health hazards. For example, many contain benezene, a carcinogen to which employees may be exposed.

Therefore, OSHA is not amending its definition for "hazardous substance" to include the petroleum exclusion referenced by some of the commenters.

The other definitions are discussed in the preamble to the proposal for this rulemaking. There were no major comments. OSHA concludes that those definitions are appropriate for the reasons stated in the proposal preamble.

Paragraph (b)-Safety and Health Program

Paragraph (b) of the proposal has been reorganized for clarity as a result. of the public comment. Basic requirements remain the same. Specific changes are discussed below. This paragraph basically requires that a written safety and health program cover safety and health organization and specific work practices to assure employee safety and health. OSHA has concluded that it is crucial for employee safety and health to have a written safety and health program that would force the systematic identification of site hazards and identify employee response to those hazards. The written plan is necessary to communicate hazards to employees for their awareness and protection. (See preamble discussion at 52 FR 29824.)

OSHA received many comments supporting the requirement for a written safety and health program (i.e., State of Wyoming, 10-9; James T. Dufour, 10-78; International Association of Fire Fighters Local 291, 10-12); other commenters have made suggestions for changes to the proposed language.

OSHA concludes that for the reasons stated a written program is necessary. The following discussion covers specific changes.

OSHA has included a non-mandatory note at the beginning of new paragraph (b) that explains the acceptability of safety and health programs developed and implemented to meet other Federal, state, or local regulations in meeting the requirements of this paragraph. Some commenters believed that OSHA's

requirements for a safety and health program were somewhat duplicative of the contingency plans and emergency response plans required by the E.P.A. for its permit requirements (i.e., Tennes-Valley Authority, 10-43; National F and Coating Association, 10-72; Joh Wax, 10-84). OSHA will permit existen programs that have been designed to meet other government or corporate requirements. For example, contingency plans developed under 40 CFR 265.50 are acceptable in meeting this requirement if they are supplemented with the provisions established by the OSHA standard. OSHA does not intend to require the duplication of efforts made to meet other governmental regulations. Therefore, any plan containing all of the elements required for the OSHA plan will be acceptable in meeting this requirement without the need for developing a separate OSHA plan.

In paragraph (b)(1) of the final rule OSHA has taken the language proposed in paragraphs (b)(1)(i), (b)(2), and (b)(3) of the proposal and subdivided it into paragraphs (b)(1)(i), (b)(1)(ii), (b)(1)(iii). and (b)(1)(iv). Paragraph (b)(1)(i) contains the first two sentences of the proposal along with two new sentences that clarify what the safety and health program shall include. OSHA has included the new sentences and the new note to this paragraph to provide further guidance to employers who may need assistance in developing their safety and health program.

In paragraph (b)(1)(ii) of the final r. OSHA is using the last sentence and the list of chapters proposed in paragraph (b)(1)(i) and subparagraphs (A) through (C). There are no changes made to the language as proposed other than a recodification of the paragraphs.

In paragraph (b)(1)(iii) of the final rule OSHA is using the exact language proposed in paragraph (b)(2). The proposed language has been moved to this paragraph because it contains a requirement that is of a general nature

In paragraph (b)(1)(iv) of the final rule OSHA is using the language proposed in paragraph (b)(3)(i) with one exception. A new phrase would require the employer to inform contractors and subcontractors of the site emergency response procedures in addition to the proposed information. One commenter. CDM Federal Programs Corporation (10-83), suggested revised language to the proposal that would assure that the contractors and subcontractors received the site specific safety and health plan as well as the safety and health programs. OSHA agrees with the suggestion of the commenter and that

the new language accomplishes the recommended change suggested by

recommended change anggested by CDM Federal Program. In paragraph (b)(1)(v) of the final rule OSHA is using the exact language of proposed paragraphs (b)(3), and (b)(4) of the final rule OSHA is using the exact language of paragraphs (b)(1), (ii), (b)(1)(iii), and (b)(1)(iv) of the proposal One commenter, James T. Dufour (10-78), while supporting the use of safety and health plans as an appropriate communication tool for identifying sits bazards, suggested that OSHA should require a more comprehensive review and control of the plan to assure its professional quality. OSHA believes that the language of paragraph (b)(4)(iv) would provide for this type of overstight and control. Therefore, the only change to paragraphs (b)(1)(ii) through (b)(1)(iv)

Parograph (c)—Site Characterization and Analyzie.

The employer needs to know the hazards faced by employees in order to develop and implement effective control measures. Site characterization provides the information methods. The more accurate, detailed, and comprehensive the information available about a site than more the protection methods. The more accurate, detailed, and comprehensive the information available about a site, than more the protective measures can be salored to the archiver accurate, detailed. Section 128(b)(1) of SARA provides that the proposal include that such a requirement be included. Section 128(b)(1) of SARA provides that the proposal include that site ' · · ' I is important to recognize that site ' is analysis of the site · · · ' I. is important to recognize that site thereare the potentiation is a continuous process. At each phase of site characterization is obtained and evaluated to develop a safety and health plan for the sext phase of work. In addition to the formal information gathering that takes place during the phase of site characterization described above, all site prevented information approach information accurate about a bout at the source information should be constantly alart for new information about allowed to the store of the site formation is formation to the formation approach information approach is taken place of site of the site

conditions. In paragraph (c) of the final rule OSHA has used most of the language in paragraph (c) of the proposal. New headmotes have been added to the major paragraphs to make reading the requirements easter. In paragraphs (c)(1) through (c)(4) of the final rule, OSHA has used the language of paragraphs (c)(1) through (c)(3) of the proposal. The reason for the one additional paragraph in the final rule is that OSHA has numbered the

initial unnumbered paragraph in the proposal, and renumbered the rest. This is an editorial change and does not change any of the proposed requirements.

In paragraph (c)(2) of the final rule. OSFA is using the language of paragraph (c)(2) of the proposal still requires the use of a five minite seamp soft-contained breaching apparents commenters, the Sata of Wyoning (10-9) and CDM Federal Programs Corporation (10-43), suggested that OSFA, revise this requirement to recognize that the neorogenetic the social backwork and the nature of the backth hazards and the nature of the work to be performed. OSFA association movel that the conditions where possible health hazards may occur about num be required to any SCRA. Therefore OSFA has an another of the proposal as follows. Two conditions will contained by the sature of the provided to set conditions will may be preformed. OSFA agrees and a performed to carrent be conditions where explores population that must be provided course to SCRA. Therefore OSFA has an encoded its proposal as follows. Two conditions will contained heat the employee populations that must be provided for set of the set of the set of conditions where sets presental to populations where a set are used as the protocodies are as at and as a mended its protocodies where the constituted for the provided with BCCRA. The propulations where these two conditions are agreed by for the final rule. OSFA is using the language from paragraph (c)(0)(0), (c)(50)(01), and c)(50)(10), call the proposal with some clauses in proposal OSFA has required that direct reading instruments, where they are available, in the present at a particular for analysis. In some expective two states are for analysis. In some situations, employees and samples out to a laboratory for analysis. In some situations, employees in the dates then direct reading instruments, may require a language tread for analysis than the employee's actual for analysis than the employee's actual

exposure time on the job. Therefore, OSHA is amending its proposal by recognizing direct reading instruments as an alternative to standard testing procedures. OSHA has added the phrase "appropriate direct reading test equipment." In place of "appropriate equipment."

(c)(5)(13). response to use the final rule after site characterization has determined that the site is safe for extr-up of operations. This is not a new requirement eince it uses the same language as that proposed in paragraph (c)(6) of the proposed. OSHA has moved the paragraph from its position in the proposal to paragraph (c)(6)(1v) of the final rule because it is related to the subject matter of paragraph (c)(6). OSHA considers this to be an editorial change because there is no change in the proposed language. Puragraphs (c)(6)(1) and (c)(6)(11) continue to use the language of proposed paragraphs (c)(5)(1) and (c)(5)(13). OSHA has also added a paragraph (c)(0)(iv) that would require that an

In paragraph (c)(7) of the final rule OEHA is using the language of proposed paragraph (c)(6) with one change. In the note which describes risks to be considered, OSHA has amended paragraph (c)-by-changing the language to reflect the exponent latits and lawds to be asset in the final rule. Direct reference to Permissible Exposure Limits (RELa) has been deleted and a reference is made to permissible exposure limits and published exposure limits and published exposure limits change is made since those terms incorporate PELs, TLVs, and RELs by definition.

Paragraph (d)-Site Control

In paragraph (d) of the final rule OSHA is using the language of paragraph (d) of the proposal. Minor editorial changes have been made for clarity without changing the proposed requirements. The need for requirements for site control is discussed at 52 FR 29525 in the presemble to our proposal. There were few substantive comments. OSHA concludes that these provisions are uccessary as discussed in the proposal.

Paragraph (e)—Training

The proposed rule included specific provisions for faitisl and routine training of employees before they would be permitted to engage in hazardous waste operations that could expose them to

safety and leadify homored. Societ recurrent training to be included in the final rule. The interact of the final version pervisions with minimum rule to their societ of unperpendent requirements for rules the final or interaction for rules of the state or interaction of these days of actual field and requirements of the site, and a schemen of these days of actual field of SAMA for the proposed requiriences has and emperiment. Compares has a complexitient of the site, and a schemen of these days of actual field of SAMA for the proposed these hore confid on perpendent where works 126(6) of SAMA for the proposed these hore confid on the state or work in SAGA of the proposed requirement. The proposed requiriences has a experiment. Compares has a scheme work in the state of these days of actual field comparison of these days of actual field of SAMA for the proposed these hore confid on the state state of the state of the state state state state of the state state state state of the state state state of the state of the state state state state

Managers and supervisors at regulated hadiities, who are directly responsible for the elity's operations, must have the same training as that of also employees and editional time for specialized training as managing harmedees wants operations. Since these managers and expervisors are responsible for directing others, it is necessary to enhance their shility to provide guidance and to make informed decisions. Section 125(d)(2) of SARA provides that there shall be eight hours of additional training for supervisors and managers.

usi mangen. The provisions also prop minimums be retrained on basis on relevant metters : no also proposed that patrained on an annual nat mathers such as revi

of health baseneds and the use of personal protective equipment. Employees at hearndow verses of this each of versit prectices unservery is avoid hearndown were bearded of this each of versit prectices unservery is equipment. Remaining, whether it is provided to provide the use of protective provide the second provide to provide the second the provide to provide the second provide to provide the second to provide the second provide to provide the second to be provide to provide the second to provide the second to provide the second to provide the second to be provide to provide the second to be provide to the second to the second to be provide to the second to the second to the second to the second to be provide to the second to the sec

In paragraph (e)(3) of the OSHA is revealed the proposed and proposed comments addressed the tr all employees who work on homeward the soluting requirement of resting for some employees at the type of site way because free the proferming for some employees at the type of site way because free the proferming for some employees at the proposed as any because free the proposed as the solution operand is out of who accurate an unspecsed site of the type of equipment they will be training and the solution been operand in which they accurate a supposed with resonance from additional balancing because of the type of equipment they will be training requirement is searches and any role additional training because of the type of equipment they will be training requirement is exceeded any will be commented as a significant to which they or the of hour minimum intribute the of hour minimum intribute to the of hour minimum intribute requirement is escensive for any log standards to which they will be the of hour minimum intribute to the of hour minimum intribute requirement is escensive for any proposed and a significant burdle of the of hour minimum intribute to the of productions are required to the standard an explorement of the any proposed and a significant burdle of the of hour minimum is an explore to the solution of the any proposed is the the standard of the standard for any proposed is were the standard of the proposed is the the standard of the standard to any the any the appropriate. For example, these works how works the standard to any of another the standard work to any of another the standard the training is any to any be appropriate. For example, how works a non-the-job training is any the standard of the standard is any the standar

indicating that exposures are under both permissible exposure limits and published exposure limits and that respirators are not necessary.

In paragraph (e)(4) of the final rule OSHA is using the language proposed in paragraph (e)(3).

In paragraph (e)(5) of the final rule OSHA is using the language proposed in paragraph (e)(4) with the addition of a new sentence. Some commenters thought that the proposed language for the qualification of trainers was too broad and ambiguous. The State of Indiana (10-23) offered a representative comment: "Knowledge or training equivalent to (redundant phrase removed) a level of training higher than the level that they are presenting is no assurance that an employee is capable of providing adequate training to other employees."

Another commenter, the International Union of Operating Engineers (10-58), stated. "We believe it irresponsible to summarily state that trainers must be 'qualified.' without defining the term other than to suggest that one who knows more than the person he trains may be a qualified trainer."

Subsequent to the receipt of posthearing briefs. Congress amended section 126(d) of SARA to require the Secretary of Labor to develop requirements for the certification of training programs offered to employees and employers who must meet the training requirements of this standard. OSHA will soon be publishing a Notice of Proposed Rulemaking to carry out this Congressional direction. The requirements of that rulemaking will expand on the provisions stated in this rulemaking.

In order to provide interim guidance to employees and employers in determining the competency of trainers and their qualifications. OSHA has added two sentences to the proposed language. These sentences require the use and demonstration of training, credentials and experience to show competency as a trainer.

In paragraph (e)(6) of the final rule OSHA is using the language of proposed paragraph (e)(6) with one minor change. In addition to permitting certification to be given by the classroom instructor. OSHA will also recognize certifications given by the head or supervisory instructor of the training facility. This change recognizes the fact that some training certificates are signed by the head instructor upon recommendation of the classroom instructor. rather than by the individual classroom instructor.

In paragraph (e)(7) of the final rule OSHA is using the exact language of proposed paragraph (e)(7).

In paragraph (e)(8) of the final rule OSHA is using the language of proposed paragraph (e)(8) with the addition of an example of the type of refresher training that OSHA would consider acceptable. OSHA considers, and has now suggested, that critiques of prior emergency response performance can serve as a means of refresher training. Critiques of performance during an emergency response can give employees a training experience in which they have actual knowledge of the acceptable or nonacceptable actions taken during the response. Such critiques can also provide employees with the experience they may need to perform in a more appropriate manner during their next response. The proposed requirement for annual refresher training has not been changed.

In paragraph (e)(9) of the final rule OSHA is using the exact language of paragraph (e)(9) in the proposal.

Paragraph (f)—Medical Surveillance

The proposed rule included specific provisions for baseline, periodic and termination medical examinations. Section 126(b)(3) of SARA provides that this rule include requirements for medical examinations of workers engaged in hazardous waste operations. In addition, the EPA manual referred to in section 126(e) of SARA has more detailed requirements for initial or baseline, periodic and termination medical examinations. The clear Congressional direction is to provide a comprehensive medical surveillance program for employees engaged in hazardous waste operations where it is medically prudent.

In paragraphs (f)(1) and (f)(2) OSHA is making some changes for clarity. In addition, OSHA is using the new term "permissible exposure limits or published exposure levels" instead of the term "established exposure levels." The reasoning for this change has been discussed under the paragraph of this preamble addressing definitions.

OSHA would like to clarify an issue concerning who is covered by medical surveillance under paragraph (f)(2) that has cause confusion since the promulgation of the interim final rule. After reviewing the record of comments addressing medical surveillance. it seems that several commenters. in particular from the fire service (i.e., 10-1, 10-3. 10-4. 10-12, 10-32 10-79), believe that all firefighters must have the medical surveillance protections of paragraph (f) since they may wear respirators 30 days or more a year. Firefighters responding to structural fires will typically wear self-contained breathing apparatus when they enter

burning structures or other hazardous locations and they may make such responses 30 days or more a year. OSHA is not requiring all firefighters who wear respirators 30 days or more a year to have medical surveillance. Paragraph (f) applies only to individuals within the scope of paragraph (a)(1)(i) through (a)(1)(iii) as set forth in paragraph (a)(2)(ii). Typical firefighters from local fire departments do not fall within this scope. These firefighters are normally covered by the requirements of paragraph (q) as specified in paragraph (a)(2)(iv). Paragraph (q) does not contain requirements for medical surveillance of firefighters unless they are members of an organized and designated hazardous materials response team, are hazardous materials specialists, or have been injured due to an overexposure to health hazards during an emergency incident involving hazardous substances as established in paragraphs (e)(9) (i) and (ii) of the final rule.

In paragraph (f)(3) of the final OSHA is using the language proposed in paragraph (f)(2) with some changes. In new paragraph (f)(3)(i)(B). OSHA is adding the phrase "unless the attending physician believes a longer interval is appropriate" to the proposed language of paragraph (f)(2)(i)(B). Several commenters (State of Wyoming, 10-9; American Society of Safety Engineers. 10-29; Union Carbide Corporation, 10-56) suggested that an annual medical examination may be excessive for some employees, particularly when an attending physician can make a recommendation for a less frequent schedule. The American Society of Safety Engineers (10-29) stated, "This reviewer concurs in the approach that OSHA has outlined in this comment area that the practical health benefit of annual medical examination for hazardous waste operation workers is indeed uncertain. This is a broad area that requires input from the attending physician, the employee and the employer. It is recommended that annual medical examination not be required rigidly, that this be a flexible time frequency."

Wyoming (10-9) stated. "Periodic occupational health physical examination on an annual basis may not be warranted under all conditions." They go on to state, "It seems reasonable that a good occupational health program requiring physical examination would be based upon documented personal exposure levels and a medical physician's recommendation rather than on an arbitrary administrative decision to require personal to undergo annual
periodic physicals if they fit into the categories under § 1910.120 (f){1)(i) and (f)(1)(ii)."

Union Carbids (10-56) said. "The frequency of medical examinations and consultations in this proposed rule has been redefined and the proposed change clarifies the issue of medical surveillance but retains the annual requirement for 'all employees who wear a respirator " ".' This frequency of examination is arbitrary. There is not medically-supportable rational for this annual requirement."

There were also comments in support of OSHA's annual physical examination requirement. The Occupational Health Nurses (10-30) stated, "AAOHN supports pre-exposure, annual, and exit examinations with provision of additional exams if over-exposure or signs or symptoms develop." Lockheed (10-45) responded to OSHA's question on whether examinations should be performed yearly, or at other intervals by stating, "Medical exams should be performed at least yearly."

performed at least yearly." GSX Chemical Services, Inc. (10-63) stated, "(12) Paragraph (f) describes medical surveillance requirements. The general program described by OSHA for pre-employment, annual, post-exposure, and termination medical examinations is excellent."

BP America, Inc. (10-85) stated, "The need for medical surveillance of workers who would be covered under the provisions of the proposed regulation is appropriate and is supported." They further state, "The proposed requirement to examine workers exposed in emergency situations, but not continue periodic surveillance simply because of the single episode. *per se*, is logical, and is strongly supported. Having such employees continue under periodic medical surveillance on the basis of the findings of the medical examination is, of course. appropriate."

Because of variations in employee exposures due to work schedules, annual physicals may not be medically necessary. OSHA concludes that annual medical examinations may not always be appropriate. Accordingly the standard is amended to permit the physician to reduce the frequency to not less than bi-annually if the physician believes it is appropriate. The physician may also increase the frequency if it is medically appropriate.

OSHA has also replaced the term "established exposure limits" with the phrase "permissible exposure limits or published exposure levels" in new paragraph (f)(3)(1)(D) since the terms have been redefined as previously explained. The rest of the language in new paragraph (f)(3) remains as it was proposed in paragraph (f)(2).

In paragraph (f)(4)(i) of the final rule OSHA is using the exact language proposed in paragraph (f)(3)(i).

In paragraph (f)[4](ii) of the final OSHA is using the language of proposed paragraph (f)[3](ii) with one change. OSHA is still requiring that the content of medical examination and consultations be determined by the attending physician. However, OSHA has added language that would direct the employee, employer, and physician to Appendix D for guidelines in developing the examination.

Several commenters requested guidance on the content of the medical examinations required by the proposal. The Okolona Fire District (10-1) commented. "As, written the current document is rather vague." They continued, ". . . the document should give guidance on what the physical examination should entail." The American Association of Occupational Health Nurses (10-30), suggested, "At least minimum content of the physical examination should be specified. An 'exam" may be no more than visual inspection of an individual's eyes, ears and throat and have no relevance to the exposure situation.

Other commenters supported OSHA's proposel for the employer and the physician to determine examination protocols. Eastman Kodak (10-88) commented, "We support OSHA's position that the physician is best able to determine an appropriate medical surveillance protocol. As noted by OSHA, employees may be exposed to differing substances and may be required to use differing levels of personal protective equipment, such as respirators. In view of the particular circumstances presented, the physician is in the best position to formulate and follow an appropriate medical protocol. OSHA should not include a detailed protocol for medical surveillance." Lockheed (10-45) responded to OSHA's . issue on protocols, "No. As with training, differences in amounts, kinds and combinations of exposures in different working situations require that protocol for medical surveillance be left to the discretion of the attending physician."

Dr. james Melius testified. "I'd like to direct most of my testimony to discussions of medical surveillance programs for hazardous waste and emergency response workers. I'd like to begin be saying that programs for both of these sets of workers are extremely important." (Ir. pg. 107) He goes on to say. "The medical surveillance program for the workers, therefore, đ with initially assessing they. work at the site and their capability conducting that work. It should inch. an assessment that focuses through . medical history and initial physical examination on their cardiovescular respiratory system, also looking for signs of other major medical problem Selective testing may also be useful 1 these instance, including pulmonary function testing, chest x-rays and electrocardiograms. However. the workers may differ in their benefits f. this testing depending on their age an other risk factors." (Tr. pgs. 110-111)

OSHA believes both sides of the argument can be addressed by placin recommended criteria for medical examination protocols in the Appendi to this section. Some commenters hav suggested protocols that OSHA considered for placement in the Appendix. The St. Petersburg Fire Department (10-4) suggested. "A full physical examination: height, weight. eyesight, pulse, blood pressure. respiratory, skin examination. neurological examination, heart and lungs, medical history, and any other aspects determined by the physician. Also included are: Pulmonary fu test, chest X-ray, urine analys blood test, and hearing examine The chapter on medical survei found in the OSHA/NIOSH/EPA/Coa: Guard manual in Appendix F also provides guidance. OSHA also believe that the language of Appendix F will provide guidance for developing the examination protocol.

In paragraph (f)(5) of the final rule OSHA is using the language of paragraph (f)(4) in the proposal with or change. OSHA has added a recommendation that a physician licensed in occupational medicine be used to supervise or administer the examination. Several commenters suggested that the use of such a physician would assure a more complet occupation-oriented examination than one offered by a physician licensed in another field.

Representative of these comments was the suggestion of the American Association of Occupational Health Nurses (10-30). The AAOHN (10-30) stated, "The nature of the potential exposures in hazardous waste operations requires specialized knowledge in toxicology—knowledge of signs and symptoms and effects of exposure to various substances—r common in basic health professioncurricule. This is information th occupational health nurses and physicians may have via advan. education degrees or continuing education, certification and experience." The AAOHN recommended that OSHA change its proposed language to require the examination to be performed "by a registered professional nurse or licensed physician with training and expertise in evaluating exposures to hazardous substances."

In recognition of AAOHN's comments. OSHA has added the recommendation for the use of a physician from the field of occupational health. The language of the final rule, while it does not preclude the use of occupational nurses, does not specifically call for the use of an occupational nurse. The final language requires that the examination be conducted under the supervision of a licensed physician and that would certainly allow the use of occupational nurses if the attending physician permits.

In paragraphs (f)(0). (f)(7) and (f)(8) of the final rule OSHA is using the exact language proposed in paragraphs (f)(5). (f)(6) and (f)(7)

Paragraph (g)—Engineering controls, work practices, and personal protective equipment for employee protection

OSHA is using the same opening paragraph for paragraph (g) that was in the opening paragraph for paragraph (g) in the proposal.

In paragraph (g)(1)(i) of 'be final rule OSHA is using the language of paragraph (g)(1)(i) of the proposal.

In paragraphs (g)(1)(ii) and (g)(1)(iii) of the final rule OSHA is using the exact larguage of paragraphs (g)(1)(ii) and (g)(1)(iii) of the proposal, except that the reference to Subpart G is deleted. A new paragraph (g)(1)(iv) is added to cross reference the requirements of Subpart G for clarity.

In paragraph (g)(2) of the final rule OSHA is using the language proposed in paragraph (g)(2) with some editorial modifications.

In paragraphs (g)(3), (g)(4) and (g)(5) of the final rule OSHA is using the language of paragraphs (g)(3), (g)(4) and (g)(5) in the proposal with minor editorial corrections to be consistent with the terms and language of the final rule.

Paragraph (h)-Monitoring

In paragraph (h)(1) of the final rule OSHA has combined the proposed language in the opening paragraph and paragraph (h)(1) of the proposal with a clarification. The new paragraphs are designated (h)(1)(i) and (h)(1)(ii).

In paragraph (h)(1)(1), OSHA has modufied its proposed language by edding the phrase, "where it is not obvious that an exposure does or does

not exist." OSHA is adding this phrase to clarify that monitoring is not necessary where the site environment or safety precautions taken by the employer prevent employee exposure to hazardous levels of chemical exposure. OSHA is only requiring monitoring where there may be a question as to an employee's exposure. When there is a question then the employer should monitor. Where there is no question of exposure, then monitoring is not necessary. For example, if it is obvious through site characterization and analysis that there are no exposures at the worksite, monitoring need not be performed unless worksite conditions or work practices change to the extent that workers could be potentially exposed to hazardous concentrations of chemical exposure. If an employer decides that employees should wear level B protection in an area where exposure will most probably be below the PEL's, then during initial entry monitoring will not be necessary because the employees

are more than adequately protected. In paragraphs (h)(2) and (h)(3) of the final rule. OSHA is using the language proposed in paragraphs (h)(2) and (h)(3) except for two changes. First, OSHA is adding language to clarify that monitoring should be used to determine exposure above permissible exposure limits which are not immediately dangerous to life or health. Second. OSHA is deleting proposed subparagraph (h)(3)(v) because it is too general in nature and the previous four subparagraphs adequately cover the hazard.

In paragraph (h)(4) OSHA is using the exact language proposed in paragraph (h)(4) with one addition. If employees with the highest exposure are overexposed, then representative samples of other employees who may be overexposed must be taken to determine if controls or PFE are needed.

Paragraph (i)-Informational programs

In paragraph (i) of the final rule OSHA is using the language of paragraph (i) of the proposal. Minor editorial changes have been made for clarity without changing the proposed requirements. The need for requirements for informational programs is discussed at 52 FR 29628 in the preamble to our proposal. There were few substantive comments. OSHA concludes that these provisions are necessary as discussed in the proposal.

Paragraph (j)—Handling drums and containers

In paragraph (j) of the final rule OSHA is using the language proposed in paragraph (j). Minor editorial changes have been made for clarity without changing the proposed requirements. The need for requirements for handling drums and containers is discussed at S2 FR 29629 in the preamble to our proposal. There were few substantive comments. OSHA concludes that these provisions are necessary as discussed in the proposal.

Paragraph (k)-Decontamination

In paragraph (k) of the final rule OSHA is using the language of paragraph (k) in the proposal. However, the agency has reorganized the paragraph and provided headnotes to make the reading of the paragraph easier. The need for requirements for decontamination is discussed at 52 FR 29629 in the preamble to our proposal. There were few substantive comments. OSHA concludes that these provisions are necessary as discussed in the proposal.

Paragraph (I)—Emergency response by employees at uncontrolled hazardous waste sites

In paragraph (1)(1) OSHA is using the exact language from proposed paragraph (1)(1)(i).

In paragraphs (1)(2)((i) through (1)(2)(xi) OSHA is using the exact text from paragraph (1)(1)(ii)(A) through (1)(1)(ii)(K).

In paragraph (1)(3) OSHA is using the language of proposed paragraph (1)(2)(i)(A) with some modification. The modifications are considered editorial and are made because of OSHA's reorganization of the overall proposed paragraph (I). In paragraph (I)(3) OSHA will require that employees performing emergency response at uncontrolled hazardous waste sites be trained in accordance with paragraph (e) of this section. This requirement is the same as proposed in the first part of proposed paragraph (1)(2)(i)(A). The portion of proposed paragraph (1)(2)(i)(A) that addresses training at RCRA sites is moved to the discussion of training in paragraph (p) of this rulemaking because of OSHA's reorganization of this paragraph.

The language proposed in paragraph (1)(2)(i)(B) has been moved to paragraph (e)(9) of this final rule. This move is considered editorial since it does not change any duties imposed on the employer, it only reflects the reorganization of proposed paragraph (1).

In paragraphs (1)(4)(i) through (1)(4)(vii) OSHA is using the exact language from paragraphs (1)(2)(ii)(A) through (1)(2)(ii)(G).

in summary, peragraphs (1)(1) through (1)(4) of the final rule use the language of paragraphs (1)(1) and (1)(2) of the proposal with some modifications due to the reorganization of the emergency response requirements of the proposal.

Paragraph (m)--Illumination

In paragraph (m) and Table H-120.2 of the final rule OSHA is using the language of paragraph (m) and Table H-102.1 of the proposal with one minor change. OSHA has combined the language of the opening paragraph and paragraph (m)(1) of the proposal into one paragraph designated paragraph (m). Minor editorial changes have been made for clarity without changing the proposed requirements. OSHA has combined the language of the opening paragraph and paragraph (m)(1) of the proposal into one paragraph designated paragraph and paragraph designated paragraph (m). The meet for requirements for illumination is discussed at 52 FR 29651 in the presable to our proposal. There were few substantive comments. OSHA concludes that these previsions are necessary as discussed in the proposal.

Paragraph (n)—Sanitation at temporary wortplaces

In paragraph (n) of the final rule OSHA is using the language of paragraph (n) in the proposal with some minor editorial changes. The opening paragraph of proposed paragraph (n) has been deleted because it is not a requirement, and Table H-102.2 has been resumbered Table H-102.3. Minor editorial changes have been made for clarity without changing the proposed requirements. The need for requirements for illumination is discussed at S2 FR 20031 in the presumble to our proposal. There were few substantive comments. OSHA concludes that these provisions are necessary as discussed in the proposal

Paragraph (o)—New technology programe

the Resource Conservation and Recovery Act of 1976 (RCRA). Proposed paragraph (o) has been moved to paragraph (p) of the final rule. In paragraph (o)(1) of the final rule OSHA is using the exact language that was proposed in paragraph (p)(1). In paragraph (o)(2) of the final rule OSHA has used dis language of paragraph (p)(2) with some changes. In paragraph (o) of the final rule OSHA is using the language of proposed paragraph (p). This change is necessary due to the reorganization of the emargency response requirements and the moving of proposed paragraph (o). Certain Operations Conducted Under

OSHA has revised the paragraph to include some additional examples of acceptable means of suppression. The agency has also added additional information to provide guidance to the employer in making evaluations of products and new technologies. These changes are considered to be editorial since the requirement of the proposal has not changed.

Paragraph (p)—Certain operations conducted under the Resource . Conservation and Recovery Act of 1978 (RCRA)

In paragraph (p) of the final rule OSI-A is using the language proposed in paragraph (c) with some changes. OSI-IA has revised the opening paragraph of the proposal to include large quantity generators of hazardous wasta that store those wastes less than 80 days within the scope of this beragraph-

In paragraphs (p)(1), (p)(2), (p)(3), and (p)(4) of the final rule OSHA has used the proposed language of paragraphs (p)(1), (o)(2), (o)(3), and (o)(4) with some nince editorial changes. The proposed requirements for each individual paragraph remain the same. OSHA is adding two new paragraphs (p)(5) and (p)(6). OsHA is address new technology programs and material handling programs and material procedures for using new technologies and equipment. Congress in the SARA legislation, directed OSHA to address new technology programs to its rule. The language of the proposal limited rew technology programs to unconscilled hassaclose wasts etcas programs at RCRA TSD facilities. In paragraph (p)(7) OSHA is adding the peragraph (p)(7) OSHA is adding the started above. In paragraph (p)(7) OSHA is using the language of proposal with some changes. In the same from paragraph (o)(5) of the proposal with some changes. In paragraph (p)(7) OSHA is using the language of proposed paragraph (o)(5)(1) with one change. OSHA has moved a requirement to paragraph (p)(7) from the last sentence of proposed paragraph (o)(5)(1) that requires employers to provide employees with a cartificate indicating that they have successfully completed the training required in the paragraph (o)SHA believes that the issuance of this cartificate will make it eense for anyloyees to devening in the paragraph (o)SHA believes that the issuance of this

employers to determine if new employees have completed the necessary training and are .eady for employment.

In paragraph (p)? OSHA is using the langua. A paragraph (o)(5)(1) of the prop-two exceptions. First, the last s of proposed paragraph (o)(5)(1) been moved to paragraph (o)(5)(1) been moved to paragraph (o)(5)(1) been moved to paragraph (o)(5)(1) inal rule as discussed above. S the requirement for eight hours annual refresher training is add paragraph. OSHA has added th requirement to this paragraph b-the new format of the final rule 1 addresses training for new empl and current employees separatel proposal there was no distinction between the two groups of empli. In paragraph (p)(7)(111) OSHA is address. the two groups of empli-ter providing the required training to employees. OSHA received many comments on trainers' qualification facilities did not address these qualifications. Therefore OSHA is requiring that trainers be properly trained and qualified to conduct it: of twinning that they are expected is provide.

In paragraph (p)(8) of the final O is addressing emergency response RCRA facilities. Paragraph 6 addresses the subject may is paragraph (1) of the pro-paragraph applied to RCRA TSD facilities. Most of the language used this paragraph (1)(1)(1). If paragraph (1)(2)(1) of the final r OSHA has used some of the languag from paragraph (1)(2)(1) of the propos-The basic requirement for the development and implementation of written emergency action plan that addresses site procedures for handlij emergency response is the same un th final rule as it was in the proposal. OSHA will still permit an exemption from this paragraph if the employer totally evacuates the facility at the ti of the emergency and has an emergen action plan meeting the requirements 29 CFR 1910.36(a). OSHA considers the changes made in this paragraph to be editorial since the proposed obligator of the employer remain the same. In paragraph (p)(0)(1)(1). This paragraph contains the munimum elements that must be addressed in the employers emergency to addressed a since the same paragraph employers employer remains the munimum

employers emergency response plan. The basic elements of the required -1-a remain the same as proposed. As stated before, training ar-

cartification of training were ' many tames discussed during rulemaking for this final rule. Several z

commenters indicated that there was a need for more specific training criteria for the courses to be offered and the quality of the instructors presenting the courses. In light of those comments. OSHA has added a new paragraph (p)(6)(iii) that addresses emergency response training on RCRA TSD facilities. The language that is used in the final rule was developed from that suggested in the comments made to the record of this proceeding.

Basically OSHA is requiring that all employees who are expected to perform emergency response at RCRA TSD facilities be trained in how to safely perform emergency response duties prior to being called upon to perform those duties [See paragraph (p)(8)(iii)(A).] Examples of the types of training to be provided have been given. Exemptions are provided in Exception =1 and Exception #2 when employee exposure is reduced through preemergency planning that includes development of employee awareness of hazards. OSHA is also requiring that employees who have attended and successfully completed the training that is required in paragraph (p)(8) be certified as having done so. Employers would also have to certify the continued competency of employees on an annual basis (See paragraph (p)(8)(Hi)(C)].

In paragraph (p)(8)(iv) of the final rule OSHA is addressing the procedures to be used for handling emergency incidents. The language in the final rule has been taken from paragraph (1)(2)(ii) and the requirements remain the same as proposed.

Paragraph (g)—Emergency response to hazardous substance releases not previously covered

in paragraph (q) OSHA is covering those emergency response situations that occur at locations other than uncontrolled hazardous waste sites and RCRA TSD facilities. The typical site covered by this paragraph would be a transportation accident where hazardous substances are or have the potential for leaking into the environment. Other sites covered by this paragraph would include hazardous substance releases at chemical manufacturing facilities such as the release that occurred at the Union Carbide plants in Buphol, India, and Institute, WV.

A typical scenario where this paragraph would be applicable would be the emergency response to a derailed tank car containing a hazardous substance that has begun to leak its contents into the atmosphere. The emergency response to this type of accident would usually include the first responders (i.e., witnesses, police, amployees on the train), the first dispatched-responsers (i.e., the first due rescue and fire apparatus), any multiple alarm dispatches (i.e., additional fire and rescue apparatus, HAZMAT teams, state fire marshal, Coast Guard or Federal E.P.A. national response teams), and the clean-up crew (i.e., initial response employees of the site owner who clean-up the release). Employees of outside clean-up contractors would be covered by paragraphs (b) through (p).

As the clean-up scenario proceeds towards completion, the various employees on the scene will need different levels of training and protective equipment required in this paragraph.

In paragraph (q)(1) of the final rule OSHA is using the language taken from paragraph (1)(1)(i) with some minor editorial changes. OSHA wants to emphasize that employers who will evacuate their employers who will evacuate their employers from the workplace when an emergency occurs and who do not permit any of their employees to assist in handling the emergency are exempt from the requirements of this paragraph if they provide an emergency action plan in accordance with § 1910.38(a).

In paragraph (q)(2) of the final rule OSHA is using the exact language of paragraph (l)(1)(ii).

paragraph (1)(1)(1). In paragraph (q)(3) of the final rule OSHA is using the language proposed in paragraph (1)(3)(ii) with the following changes. In paragraph (q)(3)(1) OSHA has used the language proposed in paragraph (1)(3)(ii)(A) with some change. OSHA has deleted the requirement that the senior official responding to an hazardous substance emergency establish the incident Command System (ICS). As a result of other requirements in this final rule, the incident Command System should already be established prior to an emergency. The senior official responding to an incident' scene should only need to take charge of the incident and begin to implement the preplanned ICS.

In paragraph (q)(3)(iv) OSHA has used the proposed language of paragraph (l)(3)(ii)(D) with a change. The proposed language required all employees engaged in emergency response and exposed to hazardous substances in any way to wear positive pressure self-contained breathing apparatus while engaged in emergency response. The final rule will require only those employees engaged in emergency response and exposed to hazardous substances "presenting an inhelation hazard or potential inhelation hazard" to wear positive pressure self-contained breathing apparatus. OSHA has made this change since several comments suggested that some individuals engaged in emergency response may be exposed to hazardous substances that do not pose an inhalation hazard and. therefore, would negate the need for respiratory protection. Such protection would become a burden to those employees engaged in operations not requiring the use of such equipment.

In paragraph (q)(3)(vi) of the final rule OSHA has used the language of paragraph (1)(3)(ii)(F) with the following change. In the proposal OSHA called for 'qualified basic life support" personnel to be present at the site. In some emergency medical service (EMS) systems the term "basic-life support (BLS)" identifies a unique group of trained individuals who have received an established level of specialized training. Typically emergency medical response begins at the first-responder level, and progresses through basic-first aid and basic-life support to advancedlife support (ALS). The amount of training and expertise increases asindividuals progress through the system. As a result of several comments, OSHA has decided to reduce the level of training required for a minimum standby capability at a hazardous waste sites. Employees trained and qualified in basic first aid have the basic skills such as initial patient assessment. maintenance of airway, control of bleeding, immobilization of fractures. and possibly cardiopulmonary resucitation (CPR) to control injuries until a higher level responder arrives. If response time for BLS or ALS is long enough that it is necessary for this level of training to be at the site in case of an emergency, this rule does not prohibit the stationing of this level at the site. However. OSHA believes that if BLS or ALS service is available within a reasonable time, a qualified basic first aider can provide the necessary interim Care.

The rest of the language in paragraph (q)(3) contains the language that was proposed in paragraph (l)(3)(ii) without change.

In paragraph (q)(4) of the final rule OSHA has used the language from paragraph (l)(3)(1)(C) with some minor editorial changes to reflect the changes made to other paragraphs in this rule. The basic requirement for the use and training of skilled support personnel remains the same as it was proposed.

In paragraph (q)(5) of the final rule OSHA has used the language from paragraph (1)(3)(1)(B) with one major change. OSHA has eliminated the requirement for 24 hours of training for specialist employees and has replaced it with a requirement for annual training or demonstration of competency in their area of specialization. The required minimum hours of training was deleted because some employees may need more or may need less than 24 hours for their area of specialization. Specialized employees are by definition individuals specialized in their area of expertise and should only require whatever level of training is necessary to maintain their level of competency. OSHA considers the other changes made to the language of this paragraph to be editorial.

In paragraph (q)(6) of the final rule OSHA addresses the training requirements for employees who will be responding to bazardous materials incidents. In paragraph (q)(6) (i), (ii), (iii), and (iv) OSHA has provided tiered training criteria for those employees who may be designated as members of an emergency response team. The various levels of response and the required competency levels are based upon recognized levels of response being discussed in the hazardous materials response industry as recommended in several of the comments made during this rulemaking.

To illustrate OSHA's tiered approach to training, the following scenario describes a possible emergency response call.

A state trooper is on routine patrol along a highway passing through a residential and light industrial area of a large metropolitan city. Ahead in his path of travel, the trooper notices a multi-vehicle accident involving a large overturned tank truck. Immediately the trooper uses his radio to contact his dispatcher to report the accident. After letting the dispatcher know the location and type of accident, the trooper places his vehicle across the travel lanes of the highway approaching the accident site to stop traffic. While he is doing this the dispatcher is alerting the fire and rescue companies in the immediate area and dispatching an established number of fire and rescue vehicles. The trooper then surveys the accident scene from his vehicle trying to identify the type of cargo on the overturned truck. Seeing three different U.S. DOT placards on the vehicle the trooper makes note of the four digit numbers and checks his DOT **Emergency Response Guide for a** summary of actions to be taken for the chemicals identified on the placards. After determining his next on-site responsibility, he recontacts his dispatcher with the additional information and secures the scene. He stays away from the immediate accident site and does not become involved in rescue or site mitigation.

While the trooper has been securing the scene, the fire and rescue units dispatched after his first radio call begin to arrive on the scene with the additional information from the trooper's second call. The officer-incharge (OIC) of the fire/rescue response stops his vehicles in a safe location and contacts the state trooper. After determining the type of accident and vehicles involved, the OIC takes control of the scene and directs his crews to take a predetermined defensive action in controlling a leak that has begun on the tanker. The OIC then contacts the dispatcher and reports his assessment of the accident scene including the fact that the tanker is now leaking. He requests the dispatcher to send him the closest hazardous materials response team. He also asks for representatives from the shipper of the liquid and the liquid's manufacturer.

In the meantime, firefighters have established a perimeter defense of the accident scene using fire hose lines and proper personal protective equipment. They begin to evacuate surrounding homes and businesses as indicated in the Emergency Response Guide in case the leaking tanker should explode. They construct dikes and diversion pits to contain water and chemical run-off from the fire hose lines. Rescue personnel, including emergency medical technicians, have made a preliminary assessment of the accident scene and have determined whether any individuals in the spill area are trapped in their vehicles or need immediate assistance. They report their observations to the OIC.

A decision is made by the OIC, based upon the reports of the police officer, the emergency response crew, and the data on the DOT placards, that no rescue attempts can be made safely until such time as the leaking liquid is positively identified and controlled by the HAZMAT team. The proper local suthorities are notified under the requirements of SARA Title III.

As firefighters continue to provide defensive protection of the scene and as emergency medical technicians establish a triage area for the treatment of injured passengers, the HAZMAT team arrives and begins to take control of the accident scene. Hazardous materials technicians and specialists assess the scene and plan their attack on the leaking tanker.

After equiping themselves properly. the HAZMAT team makes a final, preattack evaluation of the scene, including a scan of the area with appropriate monitoring equipment, and reports its findings to the fire and rescue personnel. Based upon the results of the pre-a' evaluation and a determination by HAZMAT team members using monitoring equipment that the spill area is non-hazardous. rescue personnel now enter the area of the accident to provide emergency medical treatment to injured passengers and to extricate those passengers who may have been trapped in their vehicles. The HAZMAT team proceeds to the point of release and secures the leak.

After all the injured have been cared for and after the leak has been stopped, the firefighters and HAZMAT team begin to clean-up the accident scene in accordance with pre-planned procedures.

All four levels of hazardous materials response have played a role in this scenario. The state trooper, the first on the scene, is the first responder awareness level. The first responding fire and rescue companies who provided the defensive strack are the first responder operations level. The responding HAZMAT team had both hazardous materials technicians and hazardous materials specialists. In this scenario the state trooper would have to have a sufficient amount of training, the first responding fire/rescue companies would need eight hours of training, and the HAZMAT team would need 24 hot of training. The tiered training schedu is based upon the duties and responsibilities of the individuals involved in the various levels of response illustrated in the scenario.

In paragraph (q)(7) of the final rule OSHA is addressing the competency of the trainers who will be providing the training necessary for those employees responding to hazardous materials incidents. As discussed before, several commenters were concerned that OSHA's proposal for the qualifications of trainers was too weak.

In paragraph (q)(8) of the final rule OSHA is addressing refresher training for those employees who have been trained in accordance with paragraph (q)[6]. In paragraph (1)(3)(i)[A) of the proposal OSHA addressed the training of employees who perform emergency response at non-bazardous wasts cleanup sites. OSHA is using this proposed language in paragraphs (q)(8)(i) and (q)(8)(ii) because the language of the proposal was intended to cover the type of emergency response now regulated by paragraph (q).

In paragraph (q)(9) of the final rule OSHA is using the language of paragraph (!)(4)(ii) of the proposal with some editorial change. The basic requirement that employees who are members of an organized or designated

9310

HAZMAT team and hazardous materials specialists receive a baseline physical examination in accordance with paragraph (?) of this section remains the same as proposed.

In paragraph (q)(10) of the final rule OSHA is using the proposed language of paragraph (1)(4)(11).

• In paragraph (q)(11) of the final rule OSHA is using the exact language as proposed in paragraph (l)(5). In paragraph (l)(5) OSHA regulated postemergency clean-up and the language used in that paragraph has caused some confusion. Rather than change the basic requirement, OSHA is offering the following clarification of the intent of paragraph (q)(11):

Post-emergency response can be performed by two basic groups of employees: employees of the site, or employees from off of the site. Postemergency clean-up begins when the individual in charge of the initial emergency response declares the site to be under control and ready for clean-up. For the purposes of this rule, paragraph (q)(11) will apply to those employees who come from other employers located off-of-the-site to perform postemergency clean-up. Employees of the employer at the site where the release occurred, and who perform postemergency clean-up, are considered, under this rule, to be part of the initial emergency response and not subject to paragraph (q)(11). The reason for this distinction is that employees at the site are more familiar with the types of emergencies that may occur and the types of clean-up operations that may have to take place. The more hazardous exposure to employees occurs when outside contractors or other off-site employees are brought into a strange environment and are expected to cleanup the residue from a release. With this clarification, OSHA concludes that no change to the proposed language is Decessary.

III. Summary of the Preliminary Regulatory Impact and Regulatory Flexibility Analysis and Environmental Impact Assessment

Introduction

Executive Order 12291 (46 FR 13197, February 19, 1981) requires that a regulatory impact analysis be conducted for any rule having major economic consequences for the national economy, individual industries, geographical regions, or levels of government. In addition, the Regulatory Flexibility Act of 1980 (Pub. L 96-353, 94 Stat. 1164 (5 U.S.C. 601 *et seq.*)) requires the Occupational Safety and Health Administration (OSHA) to determine whether a regulation will have a significant economic impact on a substantial number of small entities, and the National Environmental Policy Act (NEPA) of 1969 (42 U.S.C. 4321, et seq.) requires the agency to assess the environmental consequences of regulatory actions.

In order to comply with these requirements. OSHA has prepared a Regulatory impact and Regulatory Flexibility Analysis (RIA) for the hazardous waste operations and emergency response standard. This analysis includes a profile of the industries that will be affected, the estimated number of employees who are at risk from occupational exposures to hazardous wastes, technological feasibility, costs, benefits, and an overall economic impact of the standard. The RIA is available in the OSHA Docket Office.

Data Sources

The primary sources of information used for this analysis are: an April 1987 report by the Eastern Research Group (ERG) entitled, "Preparation of Data To Support a Regulatory Analysis and **Environmental Assessment of the** Proposed Standard for Working at Hazardous Waste Sites;" and the comments supplied in response to the Notice of Proposed Rulemaking, the comments made during the public hearings, and the post-hearing comments and submissions. The information contained in the ERG report was gathered from the Environmental Protection Agency sources, industry sources, experts in the area of hazardous waste management, etc. Consequently, OSHA believes that it has given due notice to all responsive parties and that the data used are the best available data for this final **Regulatory Impact Analysis (RIA).**

Industry Profile

The standard will affect about 20.000 uncontrolled hazardous waste sites. about 4.000 hazardous waste operations conducted under the Resource **Conservation and Recovery Act (RCRA)** of 1976, about 13,600 spills of hazardous materials that occur annually outside a fixed facility, and about 11.000 spills of hazardous material that occur annually inside a fixed facility. The firms that will be affected by this standard are as follows: about 100 contractors that perform hazardous waste site clean-ups. about 50 engineering or technical services firms that perform hazardous waste preliminary assessments or site investigations and remedial investigations or feasibility studies for hazardous waste sits cleanups, about

300 RCRA-regulated commercial treatment, storage and disposal facilities: about 3.700 RCRA-regulatec facilities that are operated by a hazardous waste generator: about 19.4 state and local police departments: about 28.000 fire departments; about 7 private hazardous materials (HAZMA response teams; and about 22.000 manufactures that use in-hours personnal to respond to emergency spills of hazardous materials within th facility.

Population at Risk

As many as 1.758 million employees police officers, and firefighters may be at risk from exposure to hazardous waste or to hazardous materials during an emergency response to a hazardour material spill. Of these employees, abc 14.000 work at uncontrolled hazardous waste site cleanups, 52,700 at RCRAregulated facilities. 563,200 are police officers, 944,500 are firefighters, 7,500 are private HAZMAT members, and 178,000 are members of industrial fire brigades that provide in-plant emergency responses to hazardous material spills. Most of these employee however, do not work fulltime around hazardous waste. In fact, most police officers will not face a hazardous material emergency response and most fire fighters and industrial fire brigade personnel, who are at risk, are annuall exposed to hazardous materials for on a few hours.

Feasibility

The standard does not require the us of any large-scale capital equipment th is not currently used in normal work operations. In addition, each provision requires equipment and work practices that are currently available. Thus, OSHA has determined that the standar is technologically feasible.

Benefits

This standard will protect 1.757 million employees and firefighters from health and safety hazards caused by their exposure to hazardous wastes. Th benefits of this standard are quantified in Chapter 3 of the Final Regulatory Analysis (FRA). The FRA indicates the this standard will prevent 20 cancer deaths per year and from 6 to 20 deaths per year from cardiovascular. neurological, renal and liver disorders. The standard will also prevent 1.925 injuries per year involving 18.700 lost work days. The FRA also estimates that 6 fatalities that are not illness related will be prevented. This last figure 18 likely to be an underestimate. Individua incidents which are discussed in

Chapter 3 and which may have been prevented by following the standard have sometimes led to more than 6 deaths. Also, the FRA does not take into account the benefits to the surrounding, non-worker community derived from the better handing of hazardous waste and emergency response incidents by the more qualified, properly trained and equiped response teams that are likely to result from compliance with this standard.

Chapter 3 of the FRA also presents risk rates. For example, the 17 excess cancer deaths per 1000 exposed hazardous waste workers for an occupational lifetime of exposures is likely to be reduced by 75 per cent.

OSHA concludes therefore, that this standard will substantially reduce the significant risk of material impairment of health which results from exposure to hazardous waste either at hazardous waste operations or from emergency response.

However, section 126 of SARA gives OSHA clear statutory directions to issue this standard and is reasonably explicit about what type of provisions should be included. Section 126 is also a free standing provision and not an amendment to the OSH Act. Accordingly, it evidences a legislative intent to issue these regulations without the specific need to quantify benefits and reach significant risk conclusions.

Cost of Compliance

OSHA used current work practices as its baseline for estimating the cost of full compliance with the standard. This estimated cost does not include any cost that is currently being incurred by employers as part of their work practices because those work practices, and therefore those costs, would continue whether or not the final standard were promulgated.

OSHA estimated that the total annualized incremental cost of full compliance with the standard will be about \$153.422 million, of which \$27.968 million will be spent by contractors on government-mandated clean-ups of uncontrolled hazardous waste sites \$18.372 million will be spent by RCRAregulated facility cleanups and operations, \$17.332 million will be spent by police departments, \$50.553 million will be spent by fire departments, \$4.226 million will be spent by private HAZMAT teams, and \$29.179 million will be spent by industrial fire brigades. The provision with the largest annual cost of compliance is the employee training provision (\$92.978 million), followed by the medical surveillance provision (\$11.293 million), the use of escape self-contained breathing

apparatus (\$9.507 million), and the written plan to minimize employee exposure to hazardous materials during postemergency cleanups of hazardous materials spills (\$8.381 million).

Economic Impacts

Most of the incremental cost of compliance will be paid by the government or the private firm responsible for the hazardous waste cleanup. OSHA calculated that it is economically feasible for every affected industry or group to comply with the standard. There may be an impact upon some labor markets as a consequence of the provision that only sufficiently experience employees, or employees certified to have received the necessary training at an appropriate training facility, will be allowed to work on hazardous waste sites. This provision will effectively curtail the current practice of using local subcontractors to provide short-term employees for hazardous waste site cleanups and limit the number of employees eligible to work at hazardous waste sites. This in turn, may increase future wage rates and the cost of hazardous waste site cleanups.

Regulatory Flexibility Analysis

Pursuant to the Regulatory Flexibility Act of 1980, the Assistant Secretary has assessed the expected impacts of the standard on small entities. Based on the available information. OSHA determined that the standard may have some impact upon some small entities. The cost of adequately training an employee off-site prior to working at a hazardous waste site cleanup will substantially reduce the use of subcontractor labor on a one-time basis. Thus, some local subcontractors face a potential reduction in hazardous waste site cleanup work. The majority of this subcontracted work will probably be performed by those subcontractors who concentrate upon this type of work. Subcontractors who have performed cleanup work but who do not elect to train employees needed to qualify for future work will probably be excluded from working in this market.

In addition, there could be an economic impact upon some small local fire departments depending upon the amount of financial resources available to them for additional training. With the allowance for different amounts of trairing hours depending upon the expected extent of involvement with hazardous materials spills. OSHA believes that this economic impact will not significantly affect a substantive number of local fire departments.

Environmental Impact Assessm Finding of No Significant Impact

OSHA reviewed the final standard and concluded that no significant environmental impacts are likely to result from its promulgation. In OSHA's December 19, 1986, interim final rule for the protection of workers engaged in hazardous waste and emergency response operations, information was solicited from the public on various issues, including possible environmental impacts of the regulation. On the basis of the review detailed below, and in accordance with the requirements of the National Environmental Policy Act (NEPA) of 1969 (42 U.S.C. 4321 et seq.). the Council on Environmental Quality (CEQ) NEPA regulations (40 CFR Part 1500 et seq.), and the Department of Labor's implementing regulations for NEPA compliance (29 CFR Part 11), the Assistant Secretary determined that the standard will not have a significant impact on the external environment.

In most OSHA regulatory actions, two environments may be affected: (1) The workplace environment, and (2) the general human environment external to the workplace, including impacts on air and water pollution, solid waste, and energy and land use. The hazard waste standard, however, is uni that it focuses on the external environment because during these operations, the workplace and the external environment are usually one and the same. The standard is also unusual in that it is the first regulation since the passage of the Occupational Safety and Health Act of 1970 (the Act) to be mandated specifically by Congress under section 126 of the Superfund Amendments and Reauthorization Act (SARA). As indicated in the earlier sections of this Notice, the provisions of section 126 detail those protections that OSHA must include for workers at hazardous waste and emergency response operations. For example, section 128 requires that provisions for site analysis, training, and medical surveillance, among others, be included in the standard. In addition, there is a wide range of OSHA, EPA, and other standards that already apply to some activities that occur at hazardous waste sites and during emergency response operations. For example, there are existing OSHA standards that cover construction activities, onsite machiner, and equipment, selection and use of ... personal protective equipment, har of toxic and explosive materials, a general environmental and safe such as walking-working surfa and illumination. Moreover, the

standard. in many instances, either reflects OSHA regulations, procedures adopted by other federal agencies (e.g., EPA), or practices that are commonly used by those knowledgeable in hazardous wasts and emergency response operations. To the extent that existing standards, rules, or standard operating procedures are incorporated into this rule, no significant change in the environment is anticipated.

Potential Positive Environmental Effects

While OSHA does not anticipate any significant environmental effects as a result of this standard, there is a potential for some beneficial impacts. In general, as the work practices and procedures requirements of the standard reduce the incidence of employee injury. an indirect result should be a reduction in the likelihood of environmental releases of hazardous materials. (Virtually all provisions of the standard can be categorized in this manner, because once they are implemented. they will have a positive influence on worker safety.) As these requirements also provide guidance for routine reactions to situations encountered in emergencies, they may help to reduce the severity of such emergencies. Additional potentially positive impacts might be categorized as follows: (1) Direct benefits associated with reduced incidences in, or the severity of, the reisese of hazardous materials, and (2) indirect benefits associated with the improved flow of information and increased worker awareness of hazardous materials or with improved worker preparedness (either for normal

potentially beneficial environmental effects.

Monitoring (h). The requirements of this provision will increase the amount of monitoring for airborns hazardous substances at uncontrolled hazardous waste sites. In some cases, hazardous materials will be detected, and steps will be taken to more quickly control the release to the atmosphere, thereby providing an environmental benefit.

Handling drums and containers (j). A number of specific requirements of this paragraph will result in potentially positive environmental impacts. Relevant subsections include: Inspecting drums and containers: making salvage drums or absorbents available; initiating a spill containment program: emptying unsound drums and containers; requiring ground penetrating redar; and decontaminating equipment. These are discussed briefly in the following sections.

Inspection of drums/containers before moving (j)(1)(iii). This section requires that drums and containers be inspected for their integrity prior to handling and moving. Under current practices at hazardous waste cleanup sites, drums and containers are often handled with mechanized equipment (e.g., a barrel grapple on a backhoe arm) before being inspected, if unsound drums rupture or leak, any solid contaminated by the rupture or leak is removed for disposal upon completion of drum handling operations. This provision will, through worker awareness, increase the probability of averting ruptures and leakage. In addition, any hazardous materials in containers that cannot be moved without rupturing will have to be transferred to safe containers (as required in paragraph (j)(1)(ix)), with obvious positive environmental effect. These procedures will reduce the volume of contaminated soil requiring disposal and will also lower the possibility that leachets or runoff will carry contaminants offsite. This requirement does not have an impact on emergency response actions because the routines outlined are already standard procedure.

Availability of salvage drums/ absorbents (j)(1)(vii). This provision specifies that selvage drums or . . containers as well as suitable amounts of proper absorbent be kept available for use in areas where spills, leaks, or ruptures might occur. This requirem and t will result in increased availability of salvage drums and spill absorbents at uncontrolled hexardous waste sites and in emergency response situations where spills are imminent, thereby reducing the environmental consequences related to spills of hezerdous materials. In those instances where salvage drums/ absorbents would have been inadequate without this requirement, there is a potential benefit to the environment.

implement a spill containment program (j)(1)(viii). The purpose of this provision is to develop a program to be implemented, in the event of a major spill, that would contain and isolate hazardous materials being transferred into containers and drums. To the extent that this program is implemented, there will be a potential for reducing the negative environmental effects that occur as a result of spills, leakage, etc. This requirement will reduce the environmental impact of potential spills at cleanup sites.

Empty unsound drum/containers (j)(1)(ix). Unsound containers often rupture during handling operations. This provision requires that drums and containers that cannot be moved without spillage, leakage, or rupture emptied into a sound container. This requirement will reduce the incidenc drum and container rupture and will provide concomitant environmental benefits.

Use of a ground penetrating syster. estimate depth and location of containers ())(1)(x). At present, when preliminary investigations at hazarde waste sites indicate that buried drun or containers may be present, ground penetrating systems are frequently uto determine the depth and location (the drums. The requirements of this provision will very likely cause an increase in the use of these systems. thereby reducing the number of instances in which buried containers would go undetected or where undetected containers would be accidentally ruptured during excavat activities. Where it applies, the requirement will help prevent accide ruptures and spills, improve the thoroughness of remedial actions, an benefit the site environment.

Develop Decontamination Proced. (k). The requirement to clean and decontaminate equipment, personne and personal protective equipment v prevent the migration of hazardous substances officite, thereby benefittir, the surrounding environment. It will eliminate or minimize the contamina of personnel. Decontamination is already standard practice at most cleanup sites.

Inform Contractors of Existing Hazards (b)(1)(iv). Under this provise contractors are to be informed of any "fire, explosion, health or other safet hazards" that are present. By ensure, that contractors know the location as nature of site hazards, this requirems will reduce the possibility that contractor activities will result in inadvertent releases or spills of hazardous materials.

Gather Information Before Site En. (c)(4). Among the various requirement for site evaluation are those for information to be gathered regarding (a) pathways for hezardous substanc dispersion, and (b) status and capabi of emergency response teams. These procedural requirements will result in increased ability to predict and preve movement offsite of hazardous materials, will mitigate emergency situations quickly and effectively, and will reduce the possibility or seventy contaminant release. As the requirements of the section mirror current practices, compliance will be accomplished with little difficulty.

Provide Worker Training (e). The training requirement will assure that

•

activities will be carried out by qualified personnel, with the knowledge and ability to fulfill their job functions in a safe and responsible manner. To the extent that this occurs, there will be a potential benefit to the environment (in emergency-response situations, similar benefits accrue from emergency response training and RCRA-regulated facility employee training.) For example, worker training will result in a more careful bandling of materials accompanied by a reduction in the potential for inadvertent spills, improper disposal, etc. In emergency situations this training will essure a more efficient and effective cleanup of hazardous materials or a quicker response to avert further hazardous material releases.

Informational Programs (i). These provisions include requirements for a site safety and health plan, pre-entry briefings, and site inspections. These requirements will not directly affect the existing environment; their purpose is to provide workers with the information necessary to carry out their activities safely. To the extent that this occurs, there will be a potential benefit to the environment. For example, implementing comprehensive site plans will reduce the incidence of accident releases of hazardous materials. Similarly, requiring pre-entry briefings will reduce the likelihood of employees unknowingly encountering contaminants or allowing their improper release or disposal.

Emergency Response Plan (1) and (r). The development and implementation of a response plan for on-site and off-site emergencies will provide for greater worker preparedness. In emergencies. workers will be able to respond more quickly and effectively, thereby benefitting the environment.

Potentially Negative Impacts

In some situations, there may be a potential for negative effects on the environment as a result of the standard. Any potential negative impacts, however, are not expected to be significant. To illustrate this, negative impacts may occur if there is an increase in the time required to implement specific cleanup and spill response activities, or to implement safe work practices or procedures required by the standard. Any such effects are likely to be negligible, however since response teams already have established operating procedures similar to those in OSHA's standard.

Another potential negative impact may result from the requirement that salvage drums and ebsorbents be readily available. This may increase the number of repacked hazardous waste drums and the amount of spent absorbent used, which could add to the amount of material that would require safe disposal. Similarly, the requirements for implementation of proper decontamination procedures for all equipment, personal protective gear, and personnel at hazardous waste emergencies, cleanup sites, and RCRA sites may result in an increase in the frequency and use of decontamination materials. This, in turn, could generate a larger volume of spent decontamination fluids which would then require proper handling and disposal. Again, any such impact should be negligible since. decontamination is largely standard procedure for most hazardous waste operations. A possible exception may be during activities that take place in the early stages of site evaluation before cleanup, or at spill response, where decontamination procedures are not yet standardized.

Conclusion

To the extent that the work practices and procedures are implemented, increased worker awareness and preparedness will result in a safer and more healthful work environment, which may indirectly benefit the environment. Any negative impacts that may occur as a result of the implementation of these work practices or procedures are expected to be negligible. Based on this assessment and the information presented earlier in the preamble, OSHA concludes that no significant environmental changes are anticipated as a result of the standard.

IV. International Trade

OSHA has evaluated the potential impact that this final standard would have upon international trade. OSHA has determined that the final standard would have a minimal potential impact upon the prices of products, so that there would be no effective change in the level of exported or imported products.

V. OMB Approval Under the Paperwork Reduction Act

This section contains a collection of information pertaining to the preparation of a written safety and health plan site characterization and analysis, site control, training, medical surveillance, emergency controls, work practices, PPE, monitoring, informational programs, handling drums and containers, decontamination, emergency response planning, and emergency response planning, and emergency response drills. OMB has reviewed these collections and has approved them under approval number 1218-0139.

VI. Public Reporting Burden

Public reporting burden for the collection of information identified in paragraph IV above is estimated to average 3.7 hours per response, including the time for reviewing instruction, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other suggestions for reducing this burden to the Director, **Directorate of Safety Standards** Programs, OSHA Room N-3605, U.S. Department of Labor, Washington, DC 20210; and to the Office of Information and Regulatory Affairs. Office of Management and Budget, Washington, DC 20503.

VII. State Plan States

This Federal Register document amends an interim final rule (section 1910.120, "Hazardous Waste Operations and Emergency Response") in Subpart H of 29 CFR Part 1910, OSHA's general industry standards on hazardous materials. The 25 states with their own OSHA approved occupational safety and health plans must develop a comparable standard applicable to both . the private and public (state and local government employees] sectors within six months of the publication date of this permanent final rule or show OSHA why there is no need for action, e.g., because an existing state standard covering this area is already "at least as effective" as the new Federal standard. These states are Alaska, Arizona, California (for state and local government employees only). Connecticut (for state and local government employees only), Hawaii, Indiana, Iowa, Kentucky, Maryland, Michigan, Minnesota, Nevada, New Mexico, New York (for state and local government employees only), North Carolina, Oregon, Puerto Rico, South Carolina, Tennessee, Utah, Vermont, Virginia, Virgin Islands, Washington, and Wyoming. After the effective date of this final rule, until such time as a state standard is promulgated. Federal OSHA will provide interim enforcement assistance, as appropriate, in these states.

VIII. Federal and State Coverage of the Public Sector and Volunteers

Federal OSHA is specifically precluded by section 3(5) of the Occupational Safety and Health Act from covering employees of any State or political subdivision thereof. However, States that elect to have their own occupational safety and health program under a plan approved and monitored 'by OSHA under section 18(b) of the Act are required to extend their coverage to these employees (see section VII of this preamble for a list of these states). Thus, a State hazardous waste operations standard that is either identical to or at least as effective as this Federal OSHA standard will apply to public sector as well as private sector employees in these States. Public sector employees in States without State plans will be protected from exposure to hazardous weste under Title L section 126(f) of the Superfund Amendments and Reauthorization Act of 1986 (SARA), administered by the U.S. Environmental Protection Agency (EPA). This section requires EPA to promulgate, within 90 days of the promulgation date of this Federal OSHA standard, an identical standard that applies to employees of State and local governments in each State which does not have an OSHAapproved State plan.

'

OSHA's hazardous waste operations standard and the identical or equivalent standards which will be promulgated by States with OSHA-approved State plans apply under certain circumstances to volunteer firefighters and other volunteers engaged in emergency response operations or hazardous waste operations within the scope of these standards (see paragraphs (a) (1) and (2) of this standard). In many communities, fire and other emergency response services are provided by volunteer companies. In some cases, these companies are established as independent, private sector entities. In others, they are considered a component of State or local government (see 29 CFR 1875.5 for factors to consider in ising whether or not an entity is a public agency). A volunteer working for a public or private entity in a State with an OSHA-approved State plan must be considered an employee under State law in order to be covered by the State's hezardous waste operations and emergency response standard-for example, because of an employeremployee relationship or because of pay, retirement benefits, health insurance coverage, workers' compensation benefits, etc. This determination is made by each State as part of its standards promulgation process. In a State without an OSHAapproved State plan, a private entity fire company with one or more paid employees would be covered under this Federal standard (29 CFR 1975.4).

IX. Federalism

This final regulation has been reviewed in accordance with Executive Order 12812 (52 FR 41685; October 30. 1967) regarding Federalism. Executive Order 12812 requires that agencies, to the extent possible, refrain from limiting state policy options, consult with states prior to taking any actions that would restrict state policy options, and take such actions only when there is clear constitutional authority and the presence of a problem of national scope. The Executive Order provides for preemption of state law only if there is a clear Congressional intent for the Agency to do so. Any such preemption is to be limited to the extent possible.

During the development of this rule. OSHA has, to the extent possible. refrained from limiting state policy options by developing a rule that permits flexibility on the part of the States through the use of performance language. We have also consulted with the States, in particular those states with approved state OSHA plans, during the public hearings and comment period called for in the notice of propose rulemaking for this rule. We will continue to work with the States that have state occupational safety and health plans approved under section 18 of the OSHA Act to encourage those states to develop their own policies to achieve program objectives and continue to work with appropriate state officials as they present their state standards for approval.

This rulemaking is directed by Congress under the Superfund Amendments and Resuthorization Act of 1906 (SARA). The Constitutional authority and Congressional intent for Federal action in the area of worker protection standards for employees engaged in hazardous waste operations is mandated clearly in section 120 of SARA. Congress therefore has identified the protection of employees engaged in hazardous waste operations and emergency response as a problem of national scope through the enactment of SARA.

Section 18 of the Occupational Safety and Health Act (OSH Act), permits any state to develop its own independent state occupational safety and health program. Any state may develop and submit to OSHA, for approval and use. a state occupational safety and health program that provides, among other things, worker protection "at least as effective as" that protection provided under the Federal program.

With respect to Section 4 of Executive Order 12812, Section 18 of the OSH Act also expresses Congress' clear intent to preempt state laws relating to issues with respect to which Federal OSHA has promulgated occupational safety or health standards. Under the OSH Act, a state can avoid preemption only if it submits, and obtains Federal OSHA approval of, a plan for the developme of such standards and their enforcemas mentioned above. Occupational safety and health standards develope by such approved Plan-States must, among other things, be as least as effective in providing safe and health: employment and places of employmeas the Federal standards.

OSHA has used its regulatory preemption of State law to the minimlevel necessary to achieve the object of the OSH Act and section 125 of SARA.

Section 126 of SARA, under paragra (f), requires that the U.S. Environmen Protection Agency (EPA) provide the state and local government workers v are not covered by the protections of approved OSHA state plans with protection that is identical to thet provided under the Federal OSHA standards. Non-state and local government employees would be regulated by the Federal OSHA standard. State and local governmen workers, employed in 25 non-OSHA state plan states, would not normally covered by standards promulgated under Federal OSHA or approved st. OSHA programs. OSHA has worked with EPA in the development of this final rule to assure that the protectio provided to all state and local government employees is consistent with that provided by the Federal Ot standard and the OSH Act. SPA as t regulatory authority for the 08 state plan states will address their actions with respect to worker protection policies that have federal implications in their relemaking.

This final rule is written so that employees engaged in hazardous we operations and related emergency response operations in every state. including those state and local government employees in states regulated by EPA, would be protecte by general, performance oriented standards. To the extent that there s state or regional pecultarities caused the types of bazardous wasts operations, including the types of related emergency response provide states with occupational safety and health plans approved by OSHA unsection 18 of the OSH Act would be to develop their own state standard address any special problems. This would assure the compatibility of at or local emergency response plans developed independently by state o local emergency planning committe under Title III of SARA with Federe worker protection standards issued by OSHA and EPA.

9316

And, under the OSH Act, if a state develops its own OSHA approved state program, it could make additional requirements in its standards. States that will be covered by regulations issued by EPA under paragraph 126(f) of SARA will be provided the same option. Moreover, the performance nature of this final rule, of and by itself, allows for flexibility by states and owners or operators of hazardous wastes sites or providers of emergency response to provide as much safety as possible using varying methods consonant with the conditions in each state.

In summary, there is a clear national problem, identified by Congress, related to occupational safety and health in hazardous waste operations and related emergency response. While the individual states, if all acted collectively, might be able to deal with the safety problems involved, most have not elected to do so in the seventeen years since the enactment of the OSH Act. Those states which have elected to participate under section 18 of the OSH Act. would not be preempted by this final regulation and would be able to address special, local conditions within the framework provided by this performance oriented standard while ensuring that their standards are at least as effective as the Federal standard. Stata comments were invited on the proposal and those that were submitted to the record were fully considered prior to promulgation of this Final Rule.

The agency certifies that this document has been assessed in light of the principles, criteria, and requirements stated in sections 2 through 5 of Executive Order 12621. There are no provisions of this rulemaking that are inconsistent with the principles, criteria, and requirements stated in sections 2 through 5 of Executive Order 12821. States which have approved state occupational safety and health plans may incur additional costs associated. with standards development and enforcement as a result of this rulemaking. Funding for these approved state plan programs is available from OSHA under section 18 of the OSH Act. This rulemaking would not change the State's ability to discharge traditional State governmental functions or other aspects of State sovereignty.

An outline of § 1910.120 is included for the convenience of the reader as foilows:

Table of Contents

Hazardous Waste Operations and Emergency Response

§ 1910.120 Hazardous waste operations and emergency response.

- (a) Scope, application, and definitions.
 - (1) Scope.
 - (2) Application.
 - (3) Definitions
- (b) Safety and health program.
 - (1) General.
- (2) Organizational structure chapter of the site program.
- (3) Comprehensive workplan chapter of the site program.
- (4) Site-specific safety and health plan. (i) General
- (ii) Elements
- (iii) Pre-entry briefing.
- (iv) Effectiveness of site safety and health plan.
- (c) Site characterization and analysis.
 - (1) General
 - (2) Preliminary evaluation.
- (3) Hazard identification.
- (4) Required information.
- (5) Personal protective equipment.
- (6) Monitoring.
- (7) Risk identification.
- (8) Employee notification.
- (d) Site control.
 - (1) General.
 - (2) Site control program.
 - (3) Elements of the site control program.
- (e) Training.
 - (1) General
- (2) Elements to be covered.
- (3) Frequency of training.
- [4] Management and supervisor training.
- (5) Qualifications for trainers.
- (6) Training certification.
- (7) Emergency response.(8) Refresher training.
- (9) Equivalent training.
- (f) Medical surveillance.
 - (1) General
 - (2) Employees covered.
 - (3) Frequency of medical examinations and consultations.
- (4) Content of medical examinations and consultations
 - (5) Examination by a physician and costs.
 - (6) Information provided to the physician.
 - (7) Physician's written opinion.
- (8) Recordkeeping.
- (g) Engineering controls, work practices, and personal protective equipment for employee protection.
 - (1) Engineering controls, work practices and PPE for substances regulated in Subparts G and Z.
 - (2) Engineering controls, work practices, and PPE for substances not regulated in Subparts G and Z.
 - (3) Personal protective equipment selection.
 - (4) Totally-encapsulating chemical protective suits.
 - (5) Personal protective equipment (PPE) program.

- (h) Monitoring.
 - (1) General
 - (2) Initial entry.
 - (3) Periodic monitoring.
- (4) Monitoring of high-risk employees.
- (i) Informational programs.
- (j) Handling drums and containers.
 - (1) General.
 - (2) Opening drums and containers.
 - (3) Material handling equipment.
 - (4) Radioactive wastes.
 - (5) Shock sensitive wastes.
 - (6) Laboratory waste packs.
- (7) Sampling of drum and container contents.
- (8) Shipping and transport.
- (9) Tank and vault procedures.
- (k) Decontamination.
- (1) General.
- (2) Decontamination procedures.
- (3) Location
- (4) Equipment and solvents.
- (5) Personal protective clothing and equipment
 - (6) Unauthorized employees. .
 - (7) Commercial laundries or cleaning es-
 - tablishments.
- (8) Showers and change rooms.
- (1) Emergency response by employees at uncontrolled hazardous waste sites.
 - (1 Emergency response plan.
- (2) Elements of an emergency response
- plan. (3) Procedures for handling emergency incidents.
- (m) Illumination.
- (n) Sanitation at temporary workplaces.
- (1) Potable water.
- (2) Nonpotable water.
- (3) Toilets facilities.
- (4) Food handling.
- (5) Temporary sleeping quarters.

(1) Safety and health program.

(4) Decontamination program.

(5) New technology program.

(7) Training program.

* (iii) Trainers.

plan.

(mi) Training.

incidents.

ously covered.

plan.

SDORSE.

(i) New employees.

(ii) Current employees.

(6) Material handling program.

(6) Emergency response program.

(i) Emergency response plan.

(1) Emergency response plan.

(ii) Elements of an emergency response

(iv) Procedures for handling emergency

(q) Emergency response to hazardous sub-

stance releases by employees not previ-

(2) Elements of an emergency response

(3) Procedures for handling emergency "

- (6) Washing facilities.
- (7) Showers and change rooms.
- (o) New technology programs.
- (p) Certain Operations Conducted Under the **Resource Conservation and Recovery Act** of 1976 (RCRA).

(2) Hazard communication program.

(3) Medical surveillance program.

(4) Stilled support parameters
(5) Specialist amployees
(6) Trubulag
(8) Trubulag
(8) Their support over a second seco 130 tst haponder evernaes level. Bet responder operations level. Heastdow autorich technicies. Heastdow autorich specialist. Heastdow autorich specialist.

(7) Thetawa.
(3) Reductor Uniting.
(4) Reductor Uniting.
(5) Medical curvelliness and consets
(8) Charateral protective clothing.
(14) Part-energy supports open open.

Aquesdian in (1988-128 – Filmention Weak Operation and Europeany Requests Aquesdia A. – Personnel Presentive Equipment Two Medicelie Appendix B. – Consort Description and Discontinue of the Lowels of Presention and Presentive Gene. Aquesdia D. – Compliance Coldalisms Aquesdia D. – Relevance.

List of Subjects in 28 CFR Part 1928

Contribuers, Brunns, Emergenzy response, Flammable and combucilitie liquids, Phoserdows maturinite, Hacardows substances, Hacardows wustes, Incorporation by reference, Maturials handling and storage, Personal protective equipment, Storage areas, Trabalog, Waste disposed.

This document has been prepared under the direction of John A. Penderpress, Austictual Sciencetary of Labor for Occupational Science and Heads, U.S.: Department of Labor, 200 Constitution Avenue NW., Washington, DC 20078, Personnet to section 228 of the Superfusit Amendments and Rescatherbards Act of 1968 as amended (Pub. L. 99-468, 199 Stat. 1969 as amended by Pub. L. 190-372, section 1076), 101 Stat. 1329-378, 29 U.S.C. 605 note), sections 2 of the Occupational Sciency and Health Act of 1976 (29 U.S.C. 681, 697), section 4 of the Administrative Procedures Act (5 U.S.C. 553), 29 CFR Part 1971 and Sectorary of Labor's Order 9-68 (49 FR 50739), it is proposed to amond 39 CFR Part 1900 by revising § 1910.120, Hazardous Warts Operations and Reservany Response, as set forth balow.

Signed at Weshington, DC this 20th day of shruury 1988.

A Performance

er a Labor.

PART 1910—OCCUPATIONAL SAFETY AND HEALTH STANDARDS

The enthonity citation for Subpart H

of Part 1910 is amend following paragraph: mded by adding the

Authentity: Section 1280120 instead under the embority of section 128 of the Separitical Assessments and Resultantical Act of 1988 as assessed (28 U.S.C. 605 acts), excitence 6 and 8 of the Occupational Section and Health Act of 1970 (28 U.S.C. 605, 607), exciten 6 of the Administrative Procedure Act (8 U.S.C. SSS), 29 C.FR Part 1811 and Secretary of Labor's Order 6-65 (48 PR 1973).

2. Section 1910.120 of Title 29 of the Code of Federal Regulations is revised to read as follows:

§ 1910,130 Hamindous waste operational and emergency response.

(a) Scope, application, and definition—(1) Scape. This section covers the following systematics, unlike the employer can demonstrate the possibility for employers and poperations required by a presentions required by a presention required by a presention required by a presention required by a substances that are conducted after the lask, whether Forders, state, hours of because that are conducted after the lask, whether for the FFA's National Phototy Site List (NFL), even a substances that are conducted after the lask, whether for the FFA's National Phototy Site List (NFL), even a substances that are conducted after the lask, whether for the FFA's National Phototy Site List (NFL), even a substance of the starting data and phototy site are conducted for the SFA NFL, and tables substances to a supervise substance of the starting data are conducted by the SFA NFL, and tables substances to a supervise substance of the starting data are considered.
(ii) Carrenties at after are required by the supervise and tables to be a supervised by the supervise substance of the second by Packets, state, hout or other governmental bodies as a uncontrolled by Packets, and Recovers the supervised by Andread Packet are tool and the start of SFA NFL, and the second state are constanted at treatment, storage, and disposed (TSD) facilities and the supervise and the start of the board to a supervise and the second state are supervised by Packets, and the start of the start of the board to a supervise and the start of the start of the board to a supervise and the second start of the sta

hazard.

(2) Application (1) All requirements of Part 1910 and Part 1928 of Title 29 of the Code of Federal Regulations apply

ivante and emergency response operations whether covered by this section or not. If them is a conditic or overlap, the pervision mere pertactive of amployee safety and basist shell apply without regard to 15 CFR MALS(e)(1).
(1) Harmetons within the same of peratises within the same of peragraphs (a)(1)(1) through (a)(1)(14) of this section music comply with all peragraphs (a)(1)(1) through (a)(1)(14) of this section music paragraphs (b) this section except paragraphs (c)(1)(1) of this section paragraph (c)(1)(1) of this section paragraph (c) of this section. pursuant to their textus to becardo waste and emergency response

Encoptions: For large quantify generation of humericus wave who enve these waves is then 60 days and for much weeks generations of humericus waves, who they compare of humericus there is the respond to a non-set of humericus threads of relations of humericus substantial threads of relations of humericus and threads are compt from the requirements of this works.

the replements of the worlds.
[14] Energy reports a quantities in a substantial insulation of the sectors are substantial insulation of the purpose of the sectors are an operating or the sectors.
(3) Definition—"Fusicly quantities are an operating or the sector and a sector of the work group is designated to be observed by at base one other employees in the work group is designated to be observed by at base one other employees in the work group is designated to be observed by at base one other employees in the work group is an other employee in the work group is an other employee in the work group is an other employee in the work group is an error of an emergency.
"Clean-up operation" means an operation where heats does not be observed a substances to any other mammer processed or bandlad with the ultimate goal of malting the site safer for people or the exvitorment.
"Decontraning involution" means the removal of hazardous substances that employees and their equipment to the extent necessary to preclude the occurrence of foreseeable adverse is the safer for people or the explorement to the extent necessary to preclude the occurrence of foreseeable adverse is a strained affects.

"Emergency response" or "responding to emergencies" means a response effort by employees from outside the

- Trialit Regions f Yol. Sr. No. 42 / Monday, March 6, 1989 / Jules and Regulations.

tamediate release ever e by other designated responders (i.e., mutual-tid groups, local fire departments, etc.) to an eccumence which results, er is likely to result, in an uncontrolled release of a hazardone substance. Responses to incidential releases of hazardom substances where the substance can be absorbed, neutralized, or otherwise controlled at the time of release by employees in the tamedists release area, or by maintenance personnel av not considered to be emergency responses within the scope of this standard. Responses to releases of hazardows substances where is no potential safety or hashith hazard (i.e., fire, explosion, or chemical exposure) are not considered to be emergency are not considered to be emergency

"Facility" means (A) any building, "Facility" means (A) any building, structure, installation, equipment, pipe or pipeline (including any pipe into a server or publicly owned treatment works), well, pit, pond, lagoon, impoundment, ditch, storage container, notice vehicle, rolling stock, or sincraft, or (B) any site or area where a hezardous substance has been deposited, stored, disposed of, or placed, or otherwise come to be located; but does not include any commer product in commercies or any water-borns

(a) "Transretow motor of scenario of organization of any organization of any organization of the scenario of any organization of any organization of any organization of any organization of the scenario organization of the scenario organization of the bright of any organization of the bright of a separate component of a five bright of this definition, exponence to which results or the bright of CERCLA:
(B) Any biological agent and other discuss of the scenario of a section 101(18) of CERCLA:
(C) Any substance is the bright bright or any any organication and the state of the scenario of the bright of CERCLA:
(D) Hazardow materials under 40 CFR (D) (D) Hazardow materials under 40 CFR (D) (D) Hazardow materials under 40 CFR (D) (D) Hazardow weste "means"

(A) A vesse or combination of vesses as defined in 40 CTR 2013, or
(B) These substances defined as a constructed within the according set operation conducted within the according manual end within the according a standard end within the according and and appears the transmitter principles that act on the terms and according to the terms and a depend which are carcinogens, and quents which are carcinogenes. It also includes a stransphere for a subling of the terms and a depend which are carcinogenes, in also includes a stransphere of the terms and a depend which are the active of the terms and a depend which are the active of the terms and a depend which are the according of the terms and a depend which are the according to a backlift. These are according to a standard of the terms and a depend to according the active of the terms and the according the according the active of the terms and the according the active of the terms and the active of the terms and the term at the according the active of the terms and the active of the terms and the according the active of the terms and the active of the terms and the according the terms and the term at the

reference. "Post emer

"Post emergency response" means that portion of an emergency response performed after the immediate threat of a release has been stabilized or

eliminated and clean-q it begun. If post emergency, performed by an employer's own employees who were part of the i. emergency response, it is conside be part of the initial response and post emergency response. Howevy group of an employer's own employees response, performs the clean-up operation, then the separate group employees would be considered to performing post-emergency respor-and subject to paragraph (g)(11) of section.

"Chalified person" means a per "Chalified person" means a per accessing training, knowledge a corporation to the area for which the person has the responsibility and t authority to combrol. "Site soferty and health supervise official/" means the individual locs on a hararrdous waste site who is responsible to the employer and health requirements. "Small quartity compliance with applicable activy and health requirements. "Small quartity convertor" mean generator of harardous wastes who any calender month generates or a then 1.000 kilogeness (2.305 | heardons waste in that an "Upcontrolled harardous wastes who are a an inter where an accumulation of harardous waste in that an "Upcontrolled harardous disest to hasardous waste a took. Rease sites found on public lands, such as those created by former states of such states taken place. Other sites are found on priva-place disposed harardous to prov-place of the same sound on priva-property, often belonging to generator of former generators of harardous waste. Examples of such sites includ-but are not limited to, surface importante families to, surface importantes are not covered by this definition.

(b) Sofety and health program. Note to (b) Safety and health programs developed and implemented to meet othe Federal, stata, or local regulations are considered acceptable in meeting this requirement if they cover or are modified cover the topics required in this paragraph An additional or separate safety and heal program is not required by this paragraph.

(1) General. (i) Employers shall develop and implement a written saft and health program for their emp' involved in bazardous waste opt The program shall be designer' identify, evaluate, and contro and health hazarda, and provi.

emergency zerponse for hezerdous waste operations.

(ii) The written safety and health program shall incorporate the follow

(A) An organizational structure; (B) A comprehensive workplan;

(C) A site-specific safety and health plan which need not repeat the employer's standard operating ocedures required in paragraph (b)(1)(i)(F) of this section;

(D) The safety and health training program:

(E) The medical surveillance program: (F) The employer's standard operating

procedures for safety and health; and (G) Any necessary interface between general program and site specific

activities.

(iii) Sits excevation. Site excevations created during taitial site preparation or during hexardous waste operations shall be shared or sloped as appropriate to prevent accidental collepse in accordance with Subpert P of 29 CFR Part 1020

(iv) Contractors and sub-contractors. An employee who retains contractor or sub-contractor services for work in hexardous waste operations shall inform those contractors, sub-contractors, or their representatives of the site emergency sesponce procedures and any potential firs, explosion, health, safety or other hazards of the hazardous waste operation that have been identified by the amployer, including those identified in the employer's information program.

(v) Program availability. The written safety and health program shall be made evailable to any contractor or subcontractor or their representative who will be involved with the herendous waste operation; to employees; to employee designated representatives; to OSHA personnel, and to personnel of other Pederal, state, or iocal equacies with regulatory authority over the eite.

(2) Organizational structure part of the site program.--(i) The organizational structure part of the program shall establish the specific chain of command and specify the overall responsibilities of supervisors and employees. It shall include, at a minimum, the following elements:

(A) A general supervisor who has the responsibility and authority to direct all hazardous waste operations

(B) A site safety and health supervisor who has the responsibility and authority to develop and implement the site safety

and health pien and verify compliance. (C) All other personnel needed for hazardons waste atta operations and emergency seeponse and their general functions and responsibilities.

(D) The lines of authority. responsibility, and communication.

(ii) The organizational structure shall be reviewed and updated as necessary to reflect the current status of waste site operations.

(3) Comprehensive workplan part of the site program. The compres workplan part of the program shall address the tasks and objectives of the site operations and the logistics and resources required to reach those tasks and objectives. (i) The comprehensive workplan shall

address anticipated clean-up activities as well as normal operating procedures which need not repeat the employer's procedures available elsewhere.

(ii) The comprehensive workplan shall define work tasks and objectives and identify the methods for accomplishing those tasks and objectives.

(iii) The comprehensive worknism shall establish personnel requirements for implementing the plan.

(iv) The comprehensive workplan shall provide for the implementation of the training required in paragraph (e) of this section.

(v) The comprehensive workpien shall provide for the implementation of the required informational programs required in paragraph (i) of this section.

(vi) The comprehensive worknlan shall provide for the implementation of the medical surveillance program described in paragraph (f) of this . section.

(4) Sile-specific safety and beaith plan part of the program.-(1) General The site safety and health plan, which must be kept on site, shall address the safety and health hazards of each phase of site operation and include the requirements and procedures for employee protection. [ii] Elements. The site safety and

health plan, as a minimum, shall address the following:

(A) A safety and health risk or bazard analysis for each site task and operation found in the workplan.

(B) Employee training assignments to assure compliance with paragraph (e) of this section.

(C) Personal protective equipment to be used by employees for each of the site tasks and operations being conducted as required by the personal protective equipment program in paragraph (g)(5) of this section.

(D) Medical surveillance requirements in accordance with the program in paragraph (f) of this section.

(E) Frequency and types of air monitoring, personnel monitoring, and environmental sampling techniques and instrumentation to be used, including methods of maintenance and calibration of monitoring and sampling equipment to be used.

(P) Site control measures in accordance with the site control program required in paragraph (this section.

(G) Decontamination procedu accordance with paragraph (k) c section.

(H) An emergency response pl meeting the requirements of part (1) of this section for safe and eff responses to emergencies, inclus necessary PPE and other equipm

(I) Confined space entry proce

() A spill containment program meeting the requirements of pare (j) of this section.

(iii) Pre-entry briefing. The site specific safety and health plan si provide for pre-entry briefings to prior to initiating any site activit at such other times as necessary ensure that employees are appris the site safety and health plan ar this plan is being followed. The information and data obtained fr characterization and analysis we required in paragraph (c) of this : shall be used to prepare and upd site safety and health plan.

(Iv) Effectiveness of site safet. health plan. Inspections shall be conducted by the site safety and supervisor or, in the absence of t individual, another individual w: knowledgeable in occupational s and health, acting on behalf of th employer as necessary to determ effectiveness of the size safety as health plan. Any deficiencies in t effectiveness of the site selety an health plan shall be corrected by employer.

(c) Site characterisation and analysis-(1) General Hexardor: sites shall be evaluated in accurwith this peregraph to identify sy site hezards and to determine the appropriate safety and health cor procedures needed to protect emu from the identified hexards.

(2) Preliminary evaluation. A preliminary evaluation of a site's characteristics shall be performed to site entry by a qualified person order to aid in the selection of appropriate employee protection methods prior to site entry. imme after initial site entry, a more deta evaluation of the site's specific characteristics shall be performed qualified person in order to furthe identify existing site hazards and further aid in the selection of the appropriate engineering controls + personal protective equipment for tasks to be performed.

(3) Hazard identification. All suspected conditions that may no

inhelectors or edds shearption hazards that are immediately dangerous to life or health (IDLH), or other conditions that may cause death or serious harm, shall be identified during the preliminary survey and evaluated during the detailed survey. Examples of such hezards include, but are not limited to, confined space entry, potentially explosive or flammable situations, visible super clouds, or areas where tiological indicators such as dead azimals or vegetation are located. (4) *Required information*. The following information to the extent available shall be obtained by the employer prior to allowing employees to enter a site: (1) Location and approximate size of the site.

(ii) Description of the response activity and/or the job teak to be performed.

(itt) Duration of the planned employee

(v) Salety and parameter employer activity.
(v) Salety and health barantic supercised at the site.
(v) Salety and health barantic supercised at the site.
(vi) Present status and capabilities of anexymoury response teams that would provide an emergency.
(vii) Present status and physical supercises and health barantic supercises and physical protective equipment. The source of the site, and their characterise equipment (TTE) shall be provided and used during initial site entry in accordance with the filleways for health barantic and physical protective equipment (TTE) shall be provided and used during initial site entry in accordance with the filleways are entry which will provide and used during initial site entry which will provide and used during initial site entry which will provide and used during the provided that and used during the provide and used during the protection against other known and published exposure larks and used during the provide that and used during the provide that are any which will provide protection against other known and published exposure larks and used during the provide that and which will provide protection against other known and supercise hazards identified during the provide that and which will provide and there is no permissible exposure larks and information as a guide to appropriate personal information as a guide to appropriate personal information of the entry emsemble, and if respiratory protection is warrauted by the potential hazards identified during the potential is and which and provide and an and information as a guide to appropriate personal information as a sput to constant is a or used as part of the entry ensemble, and if respiratory protection is warrauted by the potential hazards identified during the sole as part to formation is a sput the response of the section.

least five minute's duration shall be carried by employees during initial site

eatry. (iii) If the preliminary site evaluation does not produce sufficient information to identify the baserds or suspected bazards of the site, an ensemble providing protection equivalent to Lavel B PPE shall be provided as minimum protection, and direct reading instruments shall be used as appropriate for identifying IDLH conditions. (See Appendix B for a description of Lavel B bazards and the recommendations for Lavel B protective equipment.) (iv) Once the bazards of the site bave

been identified, the appropriate PPS shall be selected and used in accordance with paragraph (g) of this Bectlop.

(e) Measivering. The following monitoring shall be conducted during initial site entry when the site evaluation produces information that above the potential for ioniding radiation or IDLH conditions, or when the site information is not sufficient reasonably to eliminate these possible conditions:

instruments for becardons le ionizing rediation. (ii) Monitoring the air with (1) Monitoring with direct reading struments for bezurdons levels of

appropriate direct reading test equipment (i.e., combustible ges meters; detector tubes) for IDLH and other conditions that may cause death or serious harm (combustible or explosive atmospheres, oxygen deficiency, toxic substances).

substances). (iii) Visually observing for signs of actual or petential IDLH or other dangerves conditions. (iv) An ongoing at menitoring program is accordance with peragraph (b) of this section shall be implemented after eite characterisation has determined the sits is safe for the start-up of operations. (7) Risk identification. Once the presence and concentrations of specific bazardene substances and health bazards have been established, the risks associated with these substances shall be identified. Employees who will be working on the site shall be informed of any risks that have been identified. In situations covered by the Hazard Communication Standard, 29 CFR 1910.1200, training required by that standard need not be displicated.

Note to (c)(7).—Riels to consider include, but are not limited to: (a) Exposures exceeding the permissible exposure limits and published exposure lovels.

(b) IDLH concentrations. (b) Potential skin absorption and uritation

(d) Potential eye irritation ev (e) Explosion sensitivity and

ţ,

anges. (I) Oxygen deficiency.

(8) Employee notification. Any

information concerning the chemical physical, and toxicologic properties (each substance known or expected to present on sits that is evailable to the employer and relevant to the duties -employee is expected to perform sha be made available to the affected employees prior to the commencemen of their work activities. The employee may utilize information developed for the baseard communication standard 1

(d) Sile control—(1) General. Appropriate aits control procedures shall be implemented to control employee exposure to hexardous substances before clean-up work beg this purpose.

(2) Sile control program. A site control program for protecting employees while part of the employee's site safety an health program required in paragraph (b) of this section shall be developed during the planning stages of a hazardons waste clean-up operation a modified as necessary as new information becomes available.

(3) Elements of the cite control program. The site control program is a subshame, include: A site a subshame, include: A site over a subshame, include: A site over a subshame is a subshame in the converse sector of a back over a subshame in the precision of a subshame in the precision of the subshame is a subshame in the precision of the subshame is a subshame in the subshame is a subshame is a subshame in the subshame is a subshame is a subshame is a subshame in the subshame is a subshame

(2) Elements to be covered. The training shall thoroughly cover the following:

(i) Names of personnel and altr responsible for site safety and l

(ii) Safety, issuith and other hexards present on the size;

(iii) Use of passenal protective

equipment: (iv) Work practices by which the employee can minimize risks from hexards:

(v) Safe use of angineering controls and equipment on the site;

(vi) Medical surveillance requirements, including recognition of symptome and signs which might

indicate everyconsure to hazards; and (vif) The contents of paragraphs (G) through (]) of the site safety and health plan set forth in paragraph (b)(4)(ii) of this section.

(3) Initial training. (1) General site workers (such as equipment operators, general laborars and supervisory personnel) engaged in hazardous substance removal or other activities which expose or potentially expose workers to hemerican substances and health hemerican of these days actual field experience, under the direct supervision of a trained, experienced supervision.

(ii) Workers on site only occasionally for a specific limited task (such as, but not limited to, ground water monitoring, land surveying, or geo-physical surveying) and who are unlikely to be exposed over permissible exposure limits and published exposure limits shall receive a minimum of 24 hours of instruction of the site, and the minimum of one day actual field experience under the direct supervision of a trained, experienced supervisor.

derly as site whe (iii) Wetters regularly on site wh work in arous which have been --monitored and fully characterized liceting that exposures are under missible exposure limits and ind published expos ne limite wh respirators are not necessary, and the characterization indicates that there are no bealth hazards or the possibility of an emergency developing, shall receive n of 24 hours of instruction off a minimu the site and the minimum of one day actual field experience under the direct supervision of a trained, experienced relear.

(iv) Workers with 24 hours of training whe are asvered by paragraphs (a)(3)(ii) and (a)(3)(iii) of this section, and who become general site workers or who are required to wear respirators, shall have the additional 16 hours and two days of training necessary to total the training specified in paragraph (a)(3)(i).

(4) Management and supervisor training. On-site management and supervisors directly responsible for, or who supervise employees engaged in, hamsdows waste operations shell receive 40 hours initial training, and three days of supervised field experience (the training may be reduced to 24 hours and one day if the only area of their responsibility is employees covered by peregraphs (e)[3](ii) and (e)[3](iii)) and at isset eight additional hours of specialized training at the time of job assignment on such topics as, but not limited to, the employer's safety and health program and the associated employee training program, personal protective equipment program, spill containment program, and health hazard monitoring procedure and techniques.

(5) Qualifications for trainers. Trainers shall be qualified to instruct employees about the subject matter that is being presented in training. Such trainers shall have satisfactorily completed a training program for teaching the subjects they are expected to teach, or they shall have the academic credentials and instructional experience necessary for teaching the subjects. Instructors shall demonstrate competent instructional skills and knowledge of the applicable subject matter.

(6) Training certification. Employees and supervisors that have received and successfully completed the training and field experience specified in paragraphs (e)[1] through [a][4] of this section shall be certified by their instructor or the head-instructor and trained supervisor as having successfully completed the necessary training. A written certificate shall be given to each paragraph (e)[9] of this section shall be prohibited from experiments of paragraph (e)[9] of this section shall be prohibited from experiments of paragraph (e)[9] of this section shall be prohibited from experiments of paragraph (e)[9] of this

enging the bases doue waste operations. (F) Encourse angency responding to hexardius energency situations at hexardius energency situations at hexardius wuste clean-up sites that may express these to hexardices substances shall be trained in how to respond to such expected emergencies.

such expected emergencies. (8) Refresher training. Employees specified in paragraph (e)(1) of this section, and managers and supervisors specified in paragraph (e)(4) of this section, shall receive eight hours of refresher training ennually on the items specified in paragraph (e)(2) and/or (e)(4) of this section, any critique of incidents that have occurred in the past year that can serve as training examples of related work, and other relevant topics.

(9) Equivalent training. Employers who can show by documentation or certification that an employee's work experience and/or training has resulted in training equivalent to that training required in paragraphs (e)(1) throw (e)(4) of this section shall not be required to provide the initial trair requirements of those paragraphs such employees. However, certificemployees new to a site shall receappropriate, site specific training b site entry and have appropriate supervised field experience at the i site. Equivalent training includes a academic training or the training th existing employees might have alive received from actual bazardous wa site work experience.

(f) Medical surveillance-(1) Gene Employers engaged in operations specified in paragraphs (a)(1)(i) thr (a)(1)(iv) of this section and not cov by (a)(2)(iii) exceptions and employ of employees specified in paragraph (q)(9) shall institute a medical surveillance program in accordance this paragraph.

(2) Employees covered. The medi surveillance program shall be institby the employeer for the following employees:

(i) All employees who are or may exposed to hazardous substances or health hazards at or above the permissible exposure limits or, if the no permissible exposure limit, abovpublished exposure levels for these substances, without regard to the un respirators, for 30 days or more a ye

(ii) All employees who were a respirator for 30 days or more a yea. as required by § 1919.134;

(ill) All employees who are injured due to overexposare from an emerge incident involving hexardow substa or health héxards: et

(iv) Members of HAZMAT teams.

(3) Frequency of medical examinations and consultations. Medical examinations and consultat shall be made available by the emplto each employee covered under paragraph (f)(2) of this section on the following schedules:

(1) For employees covered under paragraphs (1)(2)(1), (1)(2)(11), and (1)(2)(11);

(A) Prior to assignment.

(B) At least once every twelve mor for each employee covered unless the attending physician believes a longer interval (not greater than bienmally) appropriate;

(C) At termination of employment (reassignment to an area where the employee would not be covered if the employee has not had an examination within the last six months;

(D) As soon as possible upon notification by an employee that the employee has developed signs or symptoms indicating possible overexposure to hazardous substances or health hazards, or that the employee has been injured or exposed above the permissible exposure limits or published exposure levels in an emergency situation:

(B) At more frequent times, if the examining physician determines that an increased frequency of examination is medically necessary.

(ii) For employees covered under paragraph (f)(2)(iii) and for all employees including those of employers covered by paragraph (a)(1)(v) who may have been injured. received a health impairment, developed signs or symptoms which may have resulted from exposure to hazardous substance resulting from an emergency incident, or exposed during an emergency incident to hazardous substances at concentrations above the permissible exposure limits or the published exposure levels without the necessary personal protective equipment being used:

(A) As soon as possible following the emergency incident or development of signs or symptoms;

(B) At additional times, if the examining physician determines that follow-up examinations or consultations are medically necessary.

(4) Content of medical examinations and consultations. (1) Medical examinations required by paragraph (f)(3) of this section shall include a medical and work history (or updated history if one is in the employee's file) with special emphasis on symptome related to the handling of hexardous substances and health hexards, and to fitness for duty including the ability to wear any required PPE under conditions (i.e., temperature extremes) that may be expected at the work site.

(ii) The content of medical examinations or consultations made available to employees pursuant to paragraph (f) shall be determined by the attending physician. The guidelines in the Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities (See Appendix D, Reference #10; should be consulted.

(5) Examination by a physician and costs. All medical examinations and procedures shall be performed by or under the supervision of a licensed physician, preferably one knowledgeable in occupational medicine, and shall be provided without cost to the employee, without loss of pay, and at a reasonable time and place.

(6) Information provided to the physician. The employer shall provide one copy of this standard and its appendices to the attending physician.

and in addition the following for each employee:

(i) A description of the employee's duties as they relate to the employee's exposures.

(ii) The employee's exposure levels or anticipated exposure levels.

(iii) A description of any personal protective equipment used or to be used.

(iv) Information from previous medical examinations of the employee which is not readily available to the examining physician.

(v) Information required by § 1910.134. (7) Physician's written opinion. (i) The employer shall obtain and furnish the employee with a copy of a written opinion from the attending physician

containing the following: (A) The physician's opinion as to whether the employee has any detected medical conditions which would place the employee at increased risk of material impairment of the employee's health from work in hazardous waste operations or emergency response, or from respirator use.

(B) The physician's recommended limitations upon the employee's assigned work.

(C) The results of the medical examination and tests if requested by the employee.

(D) A statement that the employee has been informed by the physician of the results of the medical examination and any medical conditions which require further examination or treatment.

(ii) The written opinion obtained by the employer shall not reveal specific findings or diagnoses unrelated to occupational exposures.

(8) Record keeping. (1) An accurate record of the medical surveillance required by paragraph (f) of this section shall be retained. This record shall be retained for the period specified and meet the griteria of 29 GFR 1910.20.

meet the criteria of 29 CFR 1910.20. (ii) The record required in paragraph (f)(8)(1) of this section shall include at least the following information:

(A) The name and social security number of the employee;

(B) Physician's written opinions, recommended limitations, and results of examinations and tests;

(C) Any employee medical complaints related to exposure to hazardous substances;

(D) A copy of the information provided to the examining physician by the employer, with the exception of the standard and its appendices.

(g) Engineering controls. work practices, and personal protective equipment for employee protection. Engineering controls, work practices, personal protective equipment, or a combination of these shall be (1) Engineering controls, work practices and PPE for substances regulated in Subparts G and Z. (1) Engineering controls and work practic shall be instituted to reduce and maintain employee exposure to or belo the permissible exposure limits for substances regulated by 29 CFR Part 1910, to the extent required by Subpart Z, except to the extent that such controls and practices are not feasible.

Note to (g)(1)(i): Engineering controls whimay be feasible include the use of pressurized cabs or control booths on equipment, and/or the use of remotely operated material handling equipment. Wor practices which may be feasible are removu all non-essential employees from potential exposure during opening of drums, wetting down dusty operations and locating employees upwind of possible hazards.

(ii) Whenever engineering controls and work practices are not feasible. PPI shall be used to reduce and maintain employee exposures to or below the permissible exposure limits or dose limits for substances regulated by 29 CFR Part 1910, Subpart Z.

(iii) The employer shall not imp) a schedule of employee rotation means of compliance with persons a coposare limits or dose limits on when there is no other feasible way of complying with the airborne or dermal dose limits for ionizing radiation.

(iv) The provisions of 29 CFR, Subpart G, shall be followed.

(2) Engineering controls, work practices, and PPE for substances not regulated in Subparts G and Z. An appropriate combination of ensir he controls, work practices and personal protective equipment shall be used to reduce and maintain employee exposure to or below published exposure levels for hazardous substances and health hazards not regulated by 29 CFR Part 1910. Subparts G and Z. The employer may use the published literature and MSDS as a guide in making the employer's determination as to what level of protection the employer believes is appropriate for hazardous substances and health hazards for which there is no permissible exposure limit or published exposure limit.

(3) Personal protective equipment selection. (i) Personal protective equipment (PPE) shall be selected and used which will protect employees from the hazards and potential hazards t¹ are likely to encounter as identified during the site characterization r enalysis.

9822

(ii) Personal protective equipment selection shall be based on an evaluation of the performance: characteristics of the FFE selective to the requirements and limitations of the site, the task-specific conditions and duration, and the lanards and potential hazards identified at the site.

(iii) Positive pressure self-contained breathing apparatus, or positive pressure air-line respirators equipped with an escape air supply, shall be used when chemical exposure levels present will create a substantial possibility of immediate death, immediate serious illness or injury, or impair the ability to escape.

(iv) Totaily-encapsulating chemical protective suits (protection equivalent to Level A protection as recommended in Appendix B) shall be used in conditions where skin absorption of a hazardous substance may result in a substantial possibility of immediate death, immediate serious illness or injury, or impair the shility to escape.

(v) The level of protection provided by PPE selection shall be increased when additional information on site conditions indicates that increased protection is necessary to reduce employee exposures below permissible exposure limits and published exposure levels for hazardous substances and health hazards. (See Appendix B for guidance on selecting PPE ensembles.)

Note to (g)(3): The level of employee protection provided may be decreased when additional information or vite conditions show that decreased protection will not result in hannedeus exposures to employees.

(vi) Personal protective equipment shall be exected and used to exect the requirements of 29 CFR Part 1916, Subpart I, and additional requirements specified in this section.

(4) Totally-encopsulating chemical protective suits. (1) Totallyencopsulating suits shall protect employees from the particular kasards which are identified during site cherecterization and analysis.

(ii) Totally-encapsulating suits shall be capable of maintaining positive air pressure. (See Appendix A for a test method which may be used to evaluate this requirement.)

(iii) Totally-encapsulating suits shall be capable of preventing inward test gas leakage of more than 0.5 percent. (See Appendix A for a test method which may be used to evaluate this requirement.)

(5) Personal protective equipment (PPE) program. A written personal protective equipment program, which is part of the employer's safety and health program required in paragraph (b) of this section or required in paragraph (p)(1) of this section and which is also a part of the site-specific safety and health plan shall be established. The PPE program shall address the elements listed below. When elements, such as donning and doffing procedures, are provided by the manufacturer of a piece of equipment and are attached to the plan, they need not be rewritten into the plan as long as they adequately address the procedure or element.

(i) PPE selection based upon site hazards.

(ii) PPE use and limitations of the equipment.

(iii) Work mission duration.

(iv) PPE maintenance and storage.

(v) PPE decontamination and disposal.

(vi) PPE training and proper fitting,

(vii) PPE dozung and dolling

procedures.

(viii) PPE inspection procedures pror to, during, and after use.

(ix) Evaluation of the effectiveness of the PPE program, and

(x) Limitations during temperature extremes, best stress, and other appropriate medical considerations.

appropriate medical considerations. (h) *Monitoring*—(1) *General.* (i) Monitoring shall be performed in accordance with this paragraph where there may be a question of employee exposure to hazardous concentrations of hazardous substances in order to assure proper selections of engineering controls. work practices and personal protective equipment so that employees are not exposed to levels which exceed permissible exposure limits or published exposure levels for hazardous substances.

(ii) Air monitoring shall be used to identify and quantify airborne levels of hampious substances and safety and health hampids in order to determine the appropriate level of employee protection needed on site.

[2] Initial estry. Upon initial entry. representative air monitoring shall be conducted to identify any IDLH condition, exposure over permissible exposure limits or published exposure levels, exposure over a radioactive material's dose limits or other dangerous condition such as the presence of flammable atmospheres or oxygendeficient environments.

(3) Periodic monitoring. Periodic monitoring shall be conducted when the possibility of an IDLH condition or flammable atmosphere has developed or when there is indication that exposures may have risen over permissible exposure limits or published exposure levels since pror monitoring. Situations where it shall be considered whether the possibility that exposures have risen are as follows: (i) When work begins on a different portion of the site.

(ii) When contaminants other than those previously identified are being handled.

(iii) When a different type of operation is initiated (e.g., drum opens as opposed to exploratory well druling

(iv) When employees are handling leaking drums or constainers or workin in areas with obvious liquid contamination (e.g., a spill or lagoon).

(4) Monitoring of high-risk employee After the actual clean-up phase of any hazardous waste operation commence for example, when soil, surface water containers are moved or disturbed: the employer shall monitor those employe likely to have the highest exposures to hazardous substances and health hazards likely to be present above permissible exposure limits or publish exposure levels by using personal sampling frequently enough to characterize employee exposures. If th employees likely to have the highest exposure are over permissible exposu limits or published exposure limits, the monitoring shall continue to determine all employees likely to be above those limits. The employer may utilize a representative sampling approach by documenting that the employees and chemicals chosen for monitoring are based on the criteria stated above.

Note to (b): it is not required to monitor employees engaged in site characterisation operations covered by pangruph (c) of this section.

(i) Informational programs. Joyn shall develop and implement a progra which is part of the employer's safety and health program required in paragraph (b) of this section, to inform employees, contractors, and subcontractors (or their representativactually engaged in hazardous waste operations of the nature, level and degree of exposure likely as a result o participation in such hazardous waste operations. Employees, contractors ar subcontractors working outside of the operations part of a site are not cover by this standard.

(j) Handling drams and containers-(1) General. (i) Hazardous substances and contaminated soils, liquids, and other residues shall be handled, transported, labeled, and disposed of accordance with this paragraph.

(ii) Drums and containers used duri the clean-up shall meet the appropria DOT. OSHA, and EPA regulations for the wastes that they contain.

(iii) When practical, drums and containers shall be inspected and the integrity shall be assured prior to beu maved. Brans or containers that cannot be inspected before being moved:

because of storage conditions (i.e., buried beneath the earth, stacked behind other drums, stacked several tiers high in a pile, etc.) shall be moved to an accessible location and inspected prior to further handline

(iv) Unlabelled drums and containers shall be considered to contain hexardous substances and handled accordingly until the contents are positively identified and labeled.

(v) Sits operations shall be organized to minimize the amount of drum or container movement.

(vi) Prior to movement of drums or containers, all employees exposed to the transfer operation shall be warned of the potential hazards associated with the contents of the drums or containers.

(vii) U.S. Department of Transportation specified salvage drums or containers and suitable quantities of proper absorbent shall be kept available and used in areas where spills, leaks, or

ruptures may occur. (viii) Where major spills may occur. a spill containment program, which is part of the employer's safety and health program required in paragraph (b) of this section, shall be implemented to contain and isolate the entire volume of the hazardous substance being transferred.

(bx) Drums and containers that cannot moved without supture, loakage, or be g the interest without reprint, issuinge, or spillage shall be emptied into a sound container using a device classified for the material being transferred. (x) A grand-penetrating system or other type of detection system or device shall be used to estimate the location

and depth of buried drame or contain

(xi) Soil or covering material shall be removed with castion to prevent dram or container supture

(xii) Fire extinguishing equipment meeting the requirements of 29 CFR Part 1910. Subpart L. shall be on hand and ready for use to control incipient fires.

(2) Opening drums and containers. The following procedures shall be followed in areas where drums or containers are being opened: (i) Where an airline respirator system

is used, connections to the source of air supply shall be protected from contamination and the entire system shall be protected from physical damag

(ii) Employees not actually involved in opening drums or containers shall be kept a safe distance from the drums or containers being opened.

(iii) If employees must work near or adjacent to drams or containers being opened, a suitable shield that does not interfere with the work operation shall

be placed between the employee and the drums or containers being opened to protect the employee in case of accidental emission.

(iv) Controls for drum or container opening equipment, monitoring equipment, and fire suppression equipment shall be located behind the explosion-resistant barrier.

(v) When there is a reasonable possibility of flammable atmospheres being present, material handling equipment and hand tools shall be of the type to prevent sources of ignition

(vi) Drums and containers shall be opened in such a manner that excess interior pressure will be safely relieved. If pressure can not be relieved from a remote location, appropriate shielding shall be placed between the employee and the drums or containers to reduce the risk of employee injury.

(vii) Employees shall not stand upon or work from drums or containers.

(3) Material handling equipment. Material handling equipment used to transfer drums and containers shall be selected, positioned and operated to minimize sources of ignition related to the equipment from igniting vapors released from ruptured drums or containers.

[4] Radioactive wastes. Drums and containers containing radioactive wastes shall not be handled until such time as their hazard to employees is property assessed

(5) Shock consitive wastes. As a minimum, the following special preceptions shall be taken when drum and containers containing or suspected of containing shock-consitive wastes are handlade

(i) All non-essential employees shall be evacuated from the area of transfer.

(II) Meterial handling equipment shall be provided with explosive containment devices or protective shields to protect equipment operators from exploding containers.

(iii) An employee alarm system capable of being perceived above surrounding light and noise conditions shall be used to signal the commencement and completion of explosive waste handling activities.

(iv) Continuous communications (i.e., portable radios, hand signals, telephones, as appropriate) shall be maintained between the employee-incharge of the immediate handling area and both the site safety and health supervisor and the command post until such time as the handling operation is completed. Communication equipment or methods that could cause shock sensitive materials to explode shall not be used.

(v) Drums and containers pressure. as evidenced by by swelling, shall not be moved untri suc time as the cause for excess pressure determined and appropriate containment procedures have been implemented to protect employees fro explosive relief of the drun.

(vi) Drums and containers containing packaged laboratory wastes shall be considered to contain shock-sensitive explosive materials until they have be characterized.

Caution: Shipping of about semative wester may be prohibited under U.S. Department of Transportation regulations. Employers and their shippers should refer | 40 CFR 173.21 and 173.50.

(6) Laboratory waste packs. in addition to the requirements of paragraph (j)(5) of this section. the following precautions shall be taken, a a minimum, in handling laboratory waste packs (lab packs):

(i) Lab packs shall be opened only when necessary and then only by an individual knowledgeable in t inspection, classification, and segregation of the containers within th pack according to the hazards of the wester

(ii) If crystalline material is not 1 any container, the contents sh handled as a shock-sensitive 1 iana a the contents are identified.

(7) Sampling of dram and postainer contants. Sampling of containers and drums shall be done in accordance with a sampling procedure which is part of the site safety and health plan developed for and available to employees and others at the specific worksits.

(6) Shipping and transport. (1) Drums and containers shall be identified and classified prior to packaging for shipment.

(ii) Drum or container staging areas shall be kept to the minimum number necessary to identify and classify materials safely and prepare them for transport.

(iii) Staging areas shall be provided with adequate access and egress routes

(iv) Bulking of hazardous wastes shall be permitted only after a thorough characterization of the materials has been completed.

(9) Tank and vault procedures. (i) Tanks and vaults containing hazardous substances shall be handled in a manne similar to that for drums and containers taking into consideration the size of " tank or vault.

(ii) Appropriate tank or vault or procedures as described in the employer's safety and bealth pla

9.9362.08

be followed whenever employees must enter a tank or vault. (k) Decontamination-(1) General.

Procedures for all phases of decontamination shall be developed and implemented in accordance with this

paragraph.

paragrapha
(2) Decontamination procedures shall be developed, communicated to employees and implement may enter excess on site where potential for exposure to hazardous substances exists.
(ii) Standard operating procedures thall be developed to minimize employees contacted hazardous substances.
(iii) All employees leaving a contacted hazardous substances.
(iii) All employees leaving a contacted area shall be appropriately decontaminated area shall be appropriately decontaminated area shall be appropriately disposed of or decontaminated.
(iv) Decontamination procedures shall be monitored by the site safety and health superviser to determine their effectiveness. When such procedures are found to be theffective. appropriate shall be taken to correct any deficiencies.
(3) Location. Decontamination and the monitored by the site safety of the shall be taken to correct any deficiencies.

b) forcessive of geographical event will minimize the exposure of uncontaminated employees or equipment to contaminated employees or equipment to contaminated employees or equipment and solvents used for decontaminated and solvents used for decontaminated exclusions shall be decontaminated equipment shall be decontaminated or replaced as needed to maintained or replaced as needed to maintain their effectivenes.
(5) Permonal protective clothing and equipment shall be decontaminated or fisposed of properly.
(5) Permonal protective clothing and equipment shall be decontaminated or replaced as needed to maintain their effectivenes.
(ii) Employees whose non-impermeable clothing becomes aveilating and proceed to shower. The clothing and proceed to shower. The clothing shall be disposed of or decontaminated before it is removed from the work zona.
(6) Unruthorized employees shall not remove protective clothing or equipment from the work zona.

from change rooms. (7) Commercial laundries or cleaning establishments. Commercial laundries

or cleaning artabilishments that decontaminate protective clothing or equipment shall be informed of the potentially harmful effects of exposures to hazardous substances. (8) Showers and change rooms. Where the decontamination procedure

temperature conditions prevent the effective use of water, then other effective means for cleansing shall be provided and used. indicates a need for regular showers and change rooms outside of a contaminated area, they shall be provided and meet the requirements of 29 CFR 1910.141. If

(1) Emergency response by employees at uncontrolled hazardous wasts stas-(1) Emergency response plan shall be developed and implemented by all employers within the scope of this section to handle anticipated emergencies prior to the commencement of hazardous wasts operations. The plan shall be in writing and available for inspection and copying by employees, their representatives. OSIA personnal and other governmental agencies with relevant responsibilities.

in bandling the emergency, are exempt from the requirements of this paragraph if they provide an emergency action plan complying with section 1910.36(a) (ii) Employers who will evacuate their employees from the workplace when an emergency occurs, and who do not permit any of their employees to assist permit any of their employees to assist

of this part. (2) Elements of an amergency response plan. The employer shall develop an emergency response plan for emergencies which shall address, as a minimum, the following minimum, the following

(i) Pre-emergency planning.
(ii) Personnel roles, lines of authority, and communication.

(III) Emergency recognition and prevention.
(IV) Safe distances and places of refuge.
(V) Site security and control.
(V) Site security and control.
(VI) Execution routes and procedures.
(VI) Decontamination procedures which are not covered by the site safety and health plan.

(viii) Energency medical treatment and first aid.

(ix) Emergency elerting and response procedures.
(x) Critique of response and follow-up.
(xi) PPE and emergency equipment.
(3) Procedures for handling emergency incidents. (i) In addition to the elements for the emergency response plan required in paragraph (1)(2) of this section, the following elements shall be included for emergency response plans: (A) Site topography. layout, and prevailing weather conditions.
(B) Procedures for reporting incidents to local, state, and federal governmental

(ii) The emergency response plan shall be a separate section of the Site Safety and Health Plan.

(iii) The emergency response plan shall be compatible and integrated with the disaster. fire and/or emergency response plans of local, state, and federal agencies. (iv) The emergency response plan shall be reheared regularly as part of the overall training program for site

operations.

(v) The site emergency response plan shall be reviewed periodically and, as necessary, be amended to keep it current with new or changing site conditions or information.

(vi) An employee alarm system shall be installed in accordance with 29 CFR 1910.165 to notify employees of an emergency situation; to stop work activities if necessary; to lower background noise in order to speed background noise in order to speed procedures.

(vii) Based upon the information available at time of the emergency, the employer shall evaluate the incident an⁴ the site response capabilities and proceed with the appropriate steps to implement the site emergency response

(m) *Illumination*. Areas accessible to employees shall be lighted to not less than the minimum illumination intensities listed in the following Table H-120.1 while any work is in progress. plan.

TABLE H-120.1.--Nünimum ülumination Intensities in Foot-Candles

(n) Sani workpiace adequata be provid (ii) Port dispense (8	5 ;		ал ; ;	5	3	
tation at temporary (1) Potable water. (1) An supply of potable water shall ad on the site. able containers used to trinking water shall be	drung areas, and index tokens and worknooms.) First and subscript, information, and of- fices.	rg) General shops (s.g. mechanical and sourcourt suppriser rooms, schwa supercourt, barness or hereg guar- turn, locker or desard rooms.	num of 10 too-candles a required at turnel and shuft heading during offing muching, and coaking. Mine Solety and Headin Administration Specored cap tights that he ac-	Turnels, shafts, and general under-	Interest Warbours, contors, hel-	Garard en even. Excendin and want aven. acces-	Area or operations

9325

capable of being tightly closed, and equipped with a tap. Water shall not be dipped from containers.

(iii) Any container used to distribute drinking water shall be clearly marked as to the nature of its contents and not used for any other purpose.

(iv) Where single service cups (to be used but once) are supplied, both a sanitary container for the unused cups and a receptacle for disposing of the used cups shall be provided.

(2) Nonpotable water. (i) Outlets for nonpotable water, such as water for firefighting purposes, shall be identified to indicate clearly that the water is unsafe and is not to be used for drinking, washing, or cooking purposes.

(ii) There shall be no crossconnection, open or potential, between a system furnishing potable water and a system furnishing nonpotable water.

(3) *Toilet facilities*. (i) Toilets shall be provided for employees according to the following Table H-120.2.

TABLE H-120.2.-TOILET FACILITIES

Number of employees	Minimum number of , facilities
20 or lewer More than 20, tewer than 200. More than 200	One. One tollet-seet and one unnal per 40 employees. One tolet seet and one unnal per 50 employees.

(ii) Under temporary field conditions, provisions shall be made to assure that at least one toilet facility is available.

(iii) Hazardous waste sites not provided with a sanitary sewer shall be provided with the following toilet facilities unless prohibited by local codes:

(A) Chemical toilets:

- (B) Recirculating toilets:
- (C) Combustion toilets; or
- (D) Flush toilets.

(iv) The requirements of this paragraph for sanitation facilities shall not apply to mobile crews having transportation readily available to nearby toilet facilities.

(v) Doors entering toilet facilities shall be provided with entrance locks controlled from inside the facility.

(4) Food handling. All food service facilities and operations for employees shall meet the applicable laws, ordinances, and regulations of the jurisdictions in which they are located.

(5) Temporary sleeping quarters. When-temporary sleeping quarters are provided, they shall be heated, ventilated, and lighted.

(6) Washing facilities. The employer shall provide adequate washing

facilities for employees engaged in operations where hazardous substances may be harmful to employees. Such facilities shall be in near proximity to the worksite; in areas where exposures are below permissible exposure limits and published exposure levels and which are under the controls of the employer; and shall be so equipped as to enable employees to remove hazardous substances from themselves.

(7) Showers and change rooms. When hazardous waste clean-up or removal operations commence on a site and the duration of the work will require six months or greater time to complete, the employer shall provide showers and change rooms for all employees exposed to hazardous substances and health hazards involved in hazardous waste clean-up or removal operations.

(i) Showers shall be provided and shall meet the requirements of 29 CFR 1910.141(d)(3).

(ii) Change rooms shall be provided and shall meet the requirements of 29 CFR 1910.141(e). Change rooms shall consist of two separate change areas
separated by the shower area required in paragraph (n)(7)(i) of this section. One change area, with an exit leading off the worksite, shall provide employees with a clean area where they can remove, store, and put on street clothing. The second area, with an exit to the worksite, shall provide employees with an area where they can put on, remove and store work clothing and personal protective equipment.

(iii) Showers and change rooms shall be located in areas where exposures are below the permissible exposure limits and published exposure levels. If this cannot be accomplished, then a ventilation system shall be provided that will supply air that is below the permissible exposure limits and published exposure levels.

(iv) Employers shall assure that employees shower at the end of their work shift and when leaving the hazardous waste site.

(o) New technology programs. (1) The employer shall develop and implement procedures for the introduction of effective new technologies and equipment developed for the improved protection of employees working with hazardous waste clean-up operations. and the same shall be implemented as part of the site safety and health program to assure that employee protection is being maintained.

(2) New technologies, equipment or control measures available to the industry, such as the use of foams, absorbents, adsorbents, neutralizers, or other means to suppress the level of air contaminates while excavating the site or for spill control. shall be evaluat. employers or their representatives. Suc... an evaluation shall be done to determine the effectiveness of the new methods, materials, or equipment before implementing their use on a large scale for enhancing employee protection. Information and data from manufacturers or suppliers may be used as part of the employer's evaluation effort. Such evaluations shall be made available to OSHA upon request.

(p) Certain Operations Conducted Under the Resource Conservation and Recovery Act of 1978 (RCRA). Employers conducting operations at treatment, storage, and disposal (TSD) facilities specified in paragraph (a)(1)(iv) of this section not exempted by paragraph (a)(2)(iii) of this section shall provide and implement the programs specified in this paragraph.

(1) Safety and health program. The employer shall develop and implement a written safety and health program for employees involved in hazardous waste operations that shall be available for inspection by employees, their representatives and OSHA personnel. The program shall be designed to Identify, evaluate and control safety and health hazards in their facilities for purpose of employee protection, to provide for emergency response me the requirements of paragraph (p)(8) of this section and to address as appropriate site analysis, engineering controls, maximum exposure limits, hazardous waste handling procedures and uses of new technologies.

(2) Hazard communication program. The employer shall implement a hazard communication program meeting the requirements of 29 CFR 1910.1200 as part of the employer's safety and program.

Note to 1919.120.—The exemption for hazardous waste provided in § 1910.1200 is applicable to this section.

(3) Medical surveillance program. The employer shall develop and implement a medical surveillance program meeting the requirements of paragraph (f) of this section.

(4) Decontamination program. The employer shall develop and implement a decontamination procedure meeting the requirements of paragraph (k) of this section.

(5) New technology program. The employer shall develop and implement procedures meeting the requirements of paragraph (o) of this section for introducing new and innovative equipment into the workplace.

(6) Material handling program. W employees will be handling drums o. containers, the employer shall develop and implement procedures meeting the requirements of paragrephs (j)(1) (ii) through (viii) and (xi) of this section, as well as (j)(3) and (j)(3) of this section prior to starting such work.

(7) Training program—(1) New employees The employer shall develop and implement a training program, which is part of the employer's safety and bealth program, for employees involved with hazardous waste operations to enable employees to perform their assigned duties and functions in a safe and healthful manner so as not to endanger themselves or other employees. The initial training shall be for 24 hours and refresher training shall be for eight hours annually. Employees who have received the initial training required by this paragraph shail be given a written certificate attesting that they have successfully completed the necessary training.

(ii) Current employees. Employers who can show by an employee's previous work expenence and/or training that the employee has had training equivalent to the initial training required by this paragraph, shall be considered as meeting the initial training requirements of this paragraph as to that employee. Equivalent training includes the training that existing employees might have already received from actual site work experience. Current employees shall receive eight hours of refresher training annually.

(iii) Zroiners. Trainers who teach initial training shall have satisfactorily completed a training course for teaching the subjects they are expected to teach or they shall have the academic credentials and instruction experience necessary to demonstrate a good command of the subject matter of the courses and competent instructional skills.

(8) Emergency response program-(i) Emergency response plan. An emergency response plan shall be developed and implemented by ail employers. Such plans need not duplicate any of the subjects fully addressed in the employer's contingency planning required by permits, such as those issued by the U.S. Environmental Protection Agency, provided that the contingency plan is made part of the emergency response plan. The emergency response plan shall be a written portion of the employers safety and health program required in paragraph (p)(1) of this section. Employers who will evacuate their employees from the worksite location when an emergency occurs and who do not permit any of their employees to assist in handling the emergency are

exempt from the requirements of paragraph (p)(8) if they provide an emergency action plan complying with § 1910.38(a) of this part.

(ii) Elements of an emergency response plan. The employer shall develop an emergency response plan for emergencies which shall address, as a minimum, the following areas to the extent that they are not addressed in any specific program required in this paragraph:

(A) Pre-emergency planning and coordination with outside parties.

 (B) Personnel roles, lines of authority, and communication.
 (C) Emergency recognition and

prevention.

(D) Safe distances and places of refuge.

(E) Site security and control.

(F) Evacuation routes and procedures.

(G) Decontamination procedures.

(H) Emergency medical treatment and first aid.

(I) Emergency alerting and response procedures.

(J) Critique of response and follow-up. (K) PPE and emergency equipment.

(iii) Training. (A) Training for emergency response employees shall be completed before they are called upon to perform in real emergencies. Such training shall include the elements of the emergency response plan, standard operating procedures the employer has established for the job, the personal protective equipment to be worn and procedures for handling emergency incidents.

Exception #1: An employer need not train all employees to the degree specified if the employer divides the work force in a manner such that a sufficient number of employees who have responsibility to control emergencies have the training specified, and all other employees, who may first respond to an emergency incident, have sufficient awareness training to recognize that an emergency response situation exists and that they are instructed in that case to summon the fully trained employees and not attempt control activities for which they are not trained.

Exception #2: An employer need not train all employees to the degree specified if arrangements have been made in advance for an outside fully-trained emergency response team to respond in a reasonable period and all employees, who may come to the incident first, have sufficient awareness training to recognize that an emergency response situation exists and they have been instructed to call the designated outside fullytrained emergency response team for assistance.

(B) Employee members of TSD facility emergency response organizations shall be trained to a level of competence in the recognition of health and safety hazards to protect themselves and other employees. This would include training in the methods used to minimize the risk from safety and health hazards; in the safe use of control equipment; in the selection and use of appropriate personal protective equipment; in the safe operating procedures to be used at the incident scene: in the techniques of coordination with other employees to minimize risks; in the appropriate response to over exposure from health hazards or injury to themselves and other employees; and in the recognition of subsequent symptoms which may result from over exposures.

(C) The employer shall certify that each covered employee has attended and successfully completed the training required in paragraph (p)(8)(iii) of this section, or shall certify the employee's competency at least yearly. The method used to demonstrate competency for certification of training shall be recorded and maintained by the employer.

(iv) Procedures for handling emergency incidents. (A) In addition to the elements for the emergency response plan required in paragraph (p)(8)(1) of this section, the following elements shall be included for emergency response plans to the extent that they do not repeat any information already contained in the emergency response plans

(1) Site topography. Layout, and prevailing weather conditions.

(2) Procedures for reporting incidents to local, state, and federal governmental agencies.

(B) The emergency response plur shall be compatible and integrated with the disaster, fire and/or emergency response plans of local, state, and federal agencies.

(C) The emergency response plan shall be rehearsed regularly as part of the overall training program for site operations.

(D) The site emergency response plan shall be reviewed periodically and, as necessary, be amended to keep it current with new or changing site conditions or information.

(E) An employee alarm system shall be installed in accordance with 29 CFR 1910.165 to notify employees of an emergency situation: to stop work activities if necessary; to lower background noise in order to speed communication; and to begin emergency procedures.

(F) Based upon the information available at time of the emergency, the employer shall evaluate the incident and the site response capabilities and proceed with the appropriate steps to implement the site emergency response plan.

(q) Emergency response to hazardous substance releases. This paragraph covers employers whose employees are engaged in emergency response no matter where it occurs except that it does not cover employees engaged in operations specified in paragraphs (a)(1)(i) through (a)(1)(iv) of this section. Those emergency response organizations who have developed and implemented programs equivalent to : this paragraph for handling releases of hazardous substances pursuant to section 303 of the Superfund Amendments and Reauthorization Act of 1986 (Emergency Planning and Community Right-to-Know Act of 1986. 42 U.S.C. 11003) shall be deemed to have met the requirements of this paragraph.

(1) Emergency response plan. An emergency response plan shall be developed and implemented to handle anticipated emergencies prior to the commencement of emergency response operations. The plan shall be in writing and available for inspection and copying by employees, their representatives and OSHA personnel. Employers who will evacuate their employees from the workplace when an emergency occurs, and who do not permit any of their employees to assist in handling the emergency. are exempt from the requirements of this paragraph if they provide an emergency action plan in accordance with § 1910.38(a) of this part.

(2) Elements of an emergency response plan. The employer shall develop an emergency response plan for emergencies which shall address, as a minimum, the following to the extent that they are not addressed elsewhare:

(i) Pre-emergency planning and coordination with outside parties.

(ii) Personnel roles, lines of authority, training, and communication.

(ili) Emergency recognition and prevention.

(iv) Safe distances and places of refuge.

(v) Site security and control.

(vi) Evacuation routes and procedures.

(vii) Decontamination.

(viii) Emergency medical treatment and first aid.

(ix) Emergency alerting and response procedures.

(x) Critique of response and follow-up.

(xi) PPE and emergency equipment.

(xii) Emergency response

organizations may use the local emergency response plan or the state emergency response plan or both, as part of their emergency response plan to avoid duplication. Those items of the emergency response plan that are being properly addressed by the SARA Title III plans may be substituted into their emergency plan or otherwise kept together for the employer and employee's use.

(3) Procedures for handling emergency response. (i) The senior emergency response official responding to an emergency shall become the individual in charge of a site-specific Incident Command System (ICS). All emergency responders and their communications shall be coordinated and controlled through the individual in charge of the ICS assisted by the senior official present for each employer.

Note to (q)(3)(1).—The "senior official" at an emergency response is the most senior official on the sute who has the responsibility for controlling the operations at the site. Initially it is the senior officer on the first-due piece of responding emergency apparatus to arrive on the incident scene. As more senior officers arrive (i.e., battalion chief, fire chief, state law enforcement official, site coordinator, etc.) the position is passed up the line of authority which has been previously established.

(ii) The individual in charge of the ICS shall identify, to the extent possible, all hazardous substances or conditions present and shall address as appropriate site analysis, use of engineering controls. maximum exposure limits, hazardous substance handling procedures, and use of any new technologies.

(iii) Based on the hazardous substances and/or conditions present. the individual in charge of the ICS shall implement appropriate emergency operations, and assure that the personal protective equipment worn is appropriate for the hazards to be encountered. However, personal protective equipment shall meet, at a minimum, the criteria contained in 29 CFR 1910.156(e) when worn while performing fire fighting operations beyond the incipient stage for any incident or site.

(iv) Employees engaged in emergency response and exposed to hazardous substances presenting an inhalation hazard or potential inhelation hazard shall wear positive pressure selfcontained breathing apparatus while engaged in emergency response. until such time that the individual in charge of the ICS determines through the use of air monitoring that a decreased level of respiratory protection will not result in hazardous exposures to employees.

(v) The individual in charge of the ICS shall limit the number of emergency response personnel at the emergency site. in those areas of potential or actual exposure to incident or site hazards. to those who are actively performing emergency operations. However, operations in hazardous areas shall L performed using the buddy system in groups of two or more.

(vi) Back-up personnel shall stand by with equipment ready to provide assistance or rescue. Advance first aid support personnel, as a minimum. shall also stand by with medical equipment and transportation capability.

(vii) The individual in charge of the ICS shall designate a safety official, who is knowledgable in the operations being implemented at the emergency response site, with specific responsibility to identify and evaluate hazards and to provide direction with respect to the safety of operations for the emergency at hand.

(viii) When activities are judged by the safety official to be an IDLH condition and/or to involve an imminent danger condition, the safety official shall have the authority to alter. suspend, or terminate those activities. The safety official shall immediately inform the individual in charge of the ICS of any actions needed to be taken to correct these hazards at an emergency scene.

(ix) After emergency operations have terminated, the individual in charge of the ICS shall implement appropriate decontamination procedures.

(x) When deemed necessary for meeting the tasks at hand. approved self-contained compressed air breathing apparatus may be used with approved cylinders from other approved selfcontained compressed air breathing apparatus provided that such cylinders are of the same capacity and pressure rating. All compressed air cylinders used with self-contained breathing apparatus shall meet U.S. Department of Transportation and National Institute for Occupational Safety and Health criteria.

(4) Skilled support personnel. Personnel, not necessarily an employer's own employees. who are skilled in the operation of certain equipment. such as mechanized earth moving or digging equipment or crane and hoisting equipment, and who are needed temporarily to perform immediate emergency support work that cannot reasonably be performed in a timely fashion by an employer's own employees, and who will be or may be exposed to the hazards at an emergency response scene, are not required to meet the training required in this paragraph for the employer's regular employees. However, these personnel shall be given an initial briefing at the site prior to their participation in any emergency response. The initial briefing shall -

include instruction in the wearing of appropriate personal protective equipment, what chemical hazards are involved, and what duties are to be performed. All other appropriate safety and health precautions provided to the employer's own employees shall be used to assure the safety and health of these personnel.

(5) Specialist employees. Employees who, in the course of their regular job duties, work with and are trained in the hazards of specific hazardous substances, and who will be called upon to provide technical advice or assistance at a hazardous substance release incident to the individual in charge, shall receive training or demonstrate competency in the area of their specialization annually.

(6) Training. Training shall be based on the duties and function to be performed by each responder of an emergency response organization. The skill and knowledge levels required for all new responders, those hired after the effective date of this standard, shall be conveyed to them through training before they are permitted to take part in actual emergency operations on an incident. Employees who participate, or are expected to participate, in emergency response, shall be given training in accordance with the following paragraphs:

(i) First responder awareness level. First responders at the awareness level are individuals who are likely to witness or discover a hazardous substance release and who have been trained to initiate an emergency response sequence by notifying the proper authorities of the release. They would take no further action beyond notifying the authorities of the release. First responders at the awareness level shall have sufficient training or have had sufficient experience to objectively demonstrate competency in the following areas:

(A) An understanding of what hazardous materials are, and the risks associated with them in an incident.

(B) An understanding of the potential outcomes associated with an emergency created when hazardous materials are present.

(C) The ability to recognize the presence of hazardous materials in an emergency.

(D) The ability to identify the hazardous materials, if possible.

(E) An understanding of the role of the first responder awareness individual in the employer's emergency response plan including site security and control and the U.S. Department of Transportation's Emergency Response Guidebook. (F) The ability to realize the need for additional resources, and to make appropriate notifications to the communication center.

(ii) First responder operations level. First responders at the operations level are individuals who respond to releases or potential releases of hazardous substances as part of the initial response to the site for the purpose of protecting nearby persons, property, or the environment from the effects of the release. They are trained to respond in a defensive fashion without actually trying to stop the release. Their function is to contain the release from a safe distance, keep it from spreading, and prevent exposures. First responders at the operational level shall have received at least eight hours of training or have had sufficient experience to objectively demonstrate competency in the following areas in addition to those listed for the awareness level and the employer shall so certify:

(A) Knowledge of the basic bazard and risk assessment techniques.

(B) Know how to select and use proper personal protective equipment provided to the first responder operational level.

(C) An understanding of basic bazardous materials terms.

(D) Know how to perform basic control, containment and/or confinement operations within the capabilities of the resources and personal protective equipment available with their unit.

(E) Know how to implement basic decontamination procedures.

(F) An understanding of the relevant standard operating procedures and termination procedures.

(iii) Hazardous materials technician. Hazardous materials technicians are individuals who respond to releases or potential releases for the purpose of stopping the release. They assume a more aggressive role than a first responder at the operations level in that they will approach the point of release in order to plug, patch or otherwise stop the release of a hazardous substance. Hezardous materials technicians shall have received at least 24 hours of training equal to the first responder operations level and in addition have competency in the following areas and the employer shall so certify:

(A) Know how to implement the employer's emergency response plan. (B) Know the classification.

(b) Know the classification, identification and verification of known and unknown materials by using field survey instruments and equipment.

(C) Be able to function within an assigned role in the Incident Command System. (D) Know how to select and use . proper specialized chemical personal protective equipment provided to the hazardous materials technician.

(E) Understand hazard and risk assessment techniques.

(F) Be able to perform advance control, containment, and/or confinement operations within the capabilities of the resources and personal protective equipment available with the unit.

(G) Understand and implement decontamination procedures.

(H) Understand termination procedures.

(I) Understand basic chemical and toxicological terminology and behavior.

(iv) Hazardous materials specialist. Hazardous materials specialists are individuals who respond with and provide support to hazardous materials technicians. Their duties parallel those of the hazardous materials technician. however, those duties require a more directed or specific knowledge of the various substances they may be called upon to contain. The hazardous materials specialist would also act as the site lisison with Federal, state, local and other government authorities in regards to site activities. Hazardous materials specialists shall have received at least 24 hours of training equal to the technician level and in addition have competency in the following areas and the employer shall so certify:

(A) Know how to implement the local emergency response plan.

(B) Understand classification. identification and verification of known and unknown materials by using advanced survey instruments and equipment.

(C) Know of the state emergency response plan.

(D) Be able to select and use proper specialized chemical personal protective equipment provided to the hazardous materials specialist.

(E) Understand in-depth hazard and risk techniques.

(F) Be able to perform specialized control, containment, and/or confinement operations within the capabilities of the resources and personal protective equipment available.

(G) Be able to determine and implement decontamination procedures. (H) Have the ability to develop a site

safety and control plan.

(I) Understand chemical, radiological and toxicological terminology and behavior.

(v) On scene incident commander. Incident commanders, who will assume control of the incident scene beyond the

first responder awareness lavel, shall receive at least 24 hours of training equal to the first responder operations lavel and in addition have competency in the following areas and the employer shall so cartify: (A) Know and be able to implement the employer's incident command

(B) Know how to implement the employer's energency response plan.
(C) Know and understand with employees vorting in chemical protective clothing.
(D) Know and understand the barands and risks associated with employees of decourse plan.
(E) Know of the state emergency response plan and of the Pederal Regional Response Team.
(F) Know and understand the incompression of decourtains and of the Pederal Regional Response Team.
(F) Know and understand the incompression of decourtains and of the Pederal Regional Response of decourtains the procedures.
(F) Know and understand the incompression of decourtains and of the Pederal Regional Response training autifiects they are composed to teach any of the above training the subject take any of the above training the subject takes and a good command of the subject matter of the powerse they are to teach.
(A) Refresher training (J) Those employees who are trained in account refresher to the competent their competency to demonstrate of the subject s and a good command of the subject matter of the powerse they are to teach.
(B) Rofreeter training refresher training refresher to the competence of the subject matter of the powerse they are to teach.
(B) Refreeter training or competence of the subject matter of the powerse they are to teach.
(B) A seasement shall be made of the training or competency easel to account their other or shall be made of the training or competency is made, the employee who are the subject or the training or competency easel of the subject shall be made of the training or competency or the state or the training or competency or the state of the training or compe

(9) Medical surveillance and consultation. (1) Members of an organized and designated HAZMAT team and baserdows materials specialists shell receive a baseline physical examination and be provided with medical surveillance as required in paragraph (f) of this section.
(li) Any emergency response any physical examination and be provided to a supployees who exhibits signs or symptoms which may have resulted from exponents to be another signs or subsequently, shall be provided with medical consultation as required in paragraph (f)(3)(li) of this section.
(10) Chemical protective clothing and equipment to be used by organized and designated HAZDAT team members, or the section of the section.

to be used by hazardous materials specialists, shall mast the requirements of puragraphs (g) (3) through (5) of this section. (11) Post-smergency response operations. Upon completion of the smergency response, if it is determined that it is necessary to remove hazardous substances, health hazards, and materials contaminated with them (such as contaminated soil or other elements of the natural surviconment) from the sits of the incident, the employer conducting the cleans-up shall comply with one of paragraphs (b) through (c) of this section; or (ii) Where the clean-up is done on plant property using plant or workplace suployees, such employees shall have completed the training requirements of the following; 30 CFR 1300. and other spropriate safety and basks training made necessary by the tasks that they are expected to be performed each as personal protective equipment and decontamination procedures. All equipment to be used in the performance of the clean-up work shall be in serviceable conditions and shall be in serviceable conditions and isall

Affindices to Juness - Halandous WASTE OFFICATIONS AND DECIGRADICE RESPONSE

Make The following appendiant servers: was mandatory guidelines to server to apployees and employees to server employees and employees to server the appropriate suggerments of this section. However persympt Shi0.130(g) makes randidary in certain dimensiones the use of Level A and Level S IFE protocoles. ni Totatio Linipara

Appendix A. Tenned Protective Equipment Test Michaels This appendix sets forth the non-mondatory examples of tests which may be used to evaluate compliance with 5 1930.120 (g)(4) (3) and (31). Other tasks and other challenge agents may be used to evaluate compliance.

A Totally-encouperintly chemical protective and preserve and 1.0-Boope
1.1 This precise measures the ability of a gas tight totally-encouperinting chemical protective and materials are chemical protective and materials and chemical protective and materials are tight integrity of a totally-encouperinting chemical protective and the precise allow the gas tight integrity of a totally-encouperinting chemical protective and the precise allow the gas tight integrity of a totally-encouperinting chemical protective and the precise allow the gas tight integrity of a totally-encouperint malerials to permeasion protective and the weak archive in the precise of the section of the protective dependence of the sectores in not dependence in the permeasurement of the sectores in the protective axis (TECP exil)" means a chill body gurmant which is constructed of protective clothing materials corvers the weak weak's tores, head, arms, lags and.

respirator; may cover the weard fact with tightly attached gloves and completely encloses the wearer and respirator by itself or in combine tion with wearer's gloves and boots. 2.1 "Protective clocking metarror" means any material or combination of m.urnals used in an item of clothing for the purpose isoleting parts of the body from direct conti-with a potentially besardons liquid or

error a promercury management liquid or the presence the limit of the vessel of a presence of the limit of the vessel of a presence of the limit of the vessel of a presence of a presence of a second between the presence of a presence of the presence

case shall they be less than: (A) = three inches water gauge; and (B) = two inches water gauge. The ending suit pressure (C) shall be no less than 80 percent of the test pressure (B); i.e., the pressure drop shall not exceed 20 percent of the test pressure (B).

8.1.5 Inflate the suit until the pressure inside is equal to pressure (A), the pre-test expansion suit pressure. Allow at least one minute to fill out the wrinkles in the suit. Release sufficient air to reduce the suit pressure to pressure (B), the suit test pressure to pressure (B), the suit test pressure. Begin timing. At the end of three minutes, record the suit pressure as pressure (C), the ending suit pressure. The difference between the suit test pressure and the ending suit test pressure (B-C) shall be defined as the suit pressure drop. 8.1.8 If the suit pressure drop is more than

6.1.8 If the suit pressure drop is more than 20 percent of the suit test pressure (B) during the three-minute test period, the suit fails the test and shall be removed from service.

7.0-Retest Procedure

7.1 If the suit fails the test check for leaks by inflating the suit to pressure (A) and brushing or wiping the entire suit (including seams, closures, lens gaskets, glove-to-sleeve joints, etc.) with a mild soap and water solution. Observe the suit for the formation of soap bubbles, which is an indication of a leak. Repair all identified leaks.

7.2 Retest the TECP suit as outlined in Test procedure 6.0.

8.0-Report

8.1 Each TECP suit tested by this practice shall have the following information recorded:

8.1.1 Unique identification number. identifying brand name, date of purchase, material of construction, and unique fit features, e.g. apecial breathing apparents.

features, e.g., special breathing apparatus. 8.1.2 The actual values for test pressures (A), (B), and (C) shall be recorded along with the specific observation times. If the ending pressure (C) is less than 80 percent of the test pressure (B), the suit shall be identified as failing the test. When possible, the specific leak location shall be identified in the test records. Retest pressure data shall be recorded as an additional test.

6.1.3 The source of the test apparatus used shall be identified and the sensitivity of the pressure gauge shall be recorded.

8.1.4 Records shall be kept for each pressure test even if repairs are being made at the test location.

Caution

Visually inspect all parts of the suit to be sure they are positioned correctly and secured tightly before putting the suit back into service. Special care should be taken to examine each exhaust valve to make sure it is not blocked.

Care should also be exercised to assure that the inside and outside of the suit is completely dry before it is put into storage.

B. Totally-encapsulating chemical protective suit qualitative leak test

1.0-Scope

1.1 This practice semi-qualitatively tests gas tight totally-encapsulating chemical protective suit integrity by detecting inward leakage of ammonia vapor. Since no modifications are made to the suit to carry out this test, the results from this practice provide a realistic test for the integrity of the entre suit.

1.2 Resistance of the suit materials to permeation, penetration, and degradation is not determined by this test method. ASTM jest methods are available to test suit materials for these characteristics and the tests are usually conducted by the manufacturers of the suits.

2.0-Definition of terms

2.1 "Totally-encapsulated chemical protective suit (TECP suit) means a full body garment which is constructed of protective clothing materials; covers the wearer's torso, head, arms, legs and respirator; may cover the wearer's hands and feet with tightly attached gloves and boots; completely encloses the wearer and respirator by itself or in combination with the wearer's gloves, and boots.

2.2 "Protocuve clothing material" means any material or combination of materials used in an item of clothing for the purpose of isolating parts of the body from direct contact with a potentially hazardous diquid or gaseous chemicals.

2.3 "Gas tight" means, for the purpose of this test method, the limited flow of a gas under pressure from the inside of a TECP suit to atmosphere at a prescribed pressure and time interval.

2.4 "Intrusion Coefficient" means a number expressing the level of protection provided by a gas tight totally-encapsulating chemical protective suit. The intrusion coefficient is calculated by dividing the test room challenge agent concentration by the concentration of challenge agent found inside the suit. The accuracy of the intrusion coefficient is dependent on the challenge agent monitoring methods. The larger the intrusion coefficient the greater the protection provided by the TECP suit.

3.0--Summary of recommended practice 3.1 The volume of concentrated aqueous ammonia solution (ammonia hydroxide NHLOH) required to generate the test atmosphere is determined using the directions outlined in 6.1. The suit is donned by a person wearing the appropriate respiratory equipment (either a positive pressure self-contained breathing apparatus or a positive pressure supplied air respirator) and worn inside the enclosed test room. The concentrated aqueous ammonia solution is taken by the suited individual into the test room and poured into an open plastic pan. A two-minute evaporation period is observed before the test room concentration is measured, using a high range ammonia length of stain detector tube. When the ammonia. vapor reaches a concentration of between 1000 and 1200 ppm, the suited individual starts a standardized exercise protocol to stress and flex the suit. After this protocol is completed, the test room concentration is measured again. The suited individual exits the test room and his stand-by person measures the ammonia concentration inside the suit using a low range ammonia length of stain detector tube or other more sensitive ammonia detector. A stand-by person is required to observe the test individual during the test procedure; aid the person in donning and doffing the TECP suit: and monstor the

suit interior. The intrusion coefficient of the suit can be calculated by dividing the average test area concentration by the interior suit concentration. A colorimetric ammonia indicator strip of bromophenol blue or equivalent is placed on the inside of the suit face piece lens so that the suited individual is able to detect a color change and know if the suit has a significant leak. If a color change is observed the individual shall leave the test room immediately.

4.0-Required supplies

4.1 A supply of concentrated aqueous (58 percent ammonium hydroxide by weight).

4.2 A supply of bromophenol/blue indicating paper or equivalent, sensitive to 5-10 ppm ammonia or greater over a twominute period of exposure. [pH 3.0 (yellow) to pH 4.6 (blue)]

4.3 A supply of high range (0.5-10 volume percent) and low range (5-700 ppm) detector tubes for ammonia and the corresponding sampling pump. More sensitive ammonia detectors can be substituted for the low range detector tubes to improve the sensitivity of this practice.

4.4 A shallow plastic pan (PVC) at least 12':14':1' and a half pint plastic container (PVC) with tightly closing lid.

4.5 A graduated cylinder or other volumetric measuring device of at least 50 milliliters in volume with an accuracy of at least ± 1 milliliters.

5.0--Safety precautions

5.1 Concentrated aqueous ammoniumhydroxide, NH₄OH, is a corrosive volatile hquid requiring eye, skin, and respiratory protection. The person conducting the test shall review the MSDS for squeous ammonia.

5.2 Since the established permissible exposure limit for ammonia is 50 ppm, only persons wearing a positive pressure selfcontained breathing apparatus or a positive pressure supplied air respirator shall be in the chamber. Normally only the person wearing the totally-encapsulating suit will be inside the chamber. A stand-by person shall have a positive pressure self-contained breathing apparatus, or a positive pressure supplied air respirator available to enter the test area should the suited individual need assistance.

5.3 A method to monitor the suited individual must be used during this test. Visual contact is the simplest but other methods using communication devices are acceptable.

5.4 The test room shall be large enough to allow the exercise protocol to be carried out and then to be ventilated to allow for easy exhaust of the ammonia test atmosphere after the test(s) are completed.

S.5 Individuals shall be medically screened for the use of respiratory protection and checked for allergies to ammonia before participating in this test procedure.

6.0-Test procedure

6.1.1 Measure the test area to the nearest foot and calculate its volume in cubic feet. Multiply the test area volume by 0.2 multilities of concentrated equeous:ammonia solution per cubic foot of test area volume to determine the approximate volume of

connectivited equences animated in required to generate 1000 pype is the test area...
a.2.7. Measure this volume from the empty of connectivite query animated equery and animated is and place it is a connective animated animated animated animated is an test area and begin to be contained, are realing to the contained animated is an test area and begin to be a the emitted animated animated is any access to these supplies.
a.2.7. The near constant is the emitted in the animated animated animated is an experiment in the animated animated animated is an test area and begin to be a the animated animated animated is an experiment in the animated animated is an test area to be a the animated animated animated in the animated is an test and the animated animated is an test and the animated animated is an test and the animated is an test and the animated is an test and the indicator of the animated is the experiment is bedien to a the indicator with the indicator with the animated is the experiment of the animated is the experiment is being used with the animate is being used in the animate is an indicator with the animate is an indicator is an indicator in the animate is the animate is an indicator in the animate is a

4.4 Step into the enclosed test room such as a chosel, betthroom, or two booth, equipped with an excloser the should be character with the second of the summaries during the second statement of the summaries of the sum of th

one minute. 6.7.2 Walking in place for one minute with at least 15 ruising motions of each lag in a one-minute period. 6.7.2 Touching the tase with a least 10 complete motions of the arms from above the based to touching of the tase in a one-minute

period. 8.7.4 Knee bands with at least 10 complete standing and equatting motions in a one-summe period. 6.8 If all any time during the test the colorimetric indicating paper should change colors, the test should be coupped and section 6.10 and 5.12 initiated (Bee §4.2). 6.3 After completion of the test councies, the test area concentration should be

measured again using the high range colorization denotes table.
6.10 Exit the test area.
6.11 The opening created by the set single of the suppropriate sull penetration about the card to determine the iner mage length of state detector table or other summaria manifur. The thermal TBCP sull algorithe test uses to prove at a failer anomalie reacting.
6.13 After completion of the measurement of the sective completion of the section of the summarial reacting.
6.14 After completion of the measurement of the section reacting.
6.15 After completion of the measurement of the section reacting.
6.14 After completion of the measurement of the section reacting.
6.15 The ventificting fam the failed suit the section of the section is an other section of the secti

7.0 — Retract proceederee
7.1 If the sout fails this test, check far leads by following the presence test is test A obsyre.
7.2 Retract the TRLP out as collined in the test procedure 6.0.
8.0 —Report
8.1 Each gas tight totally encopresisting chemical protective suit tested by this practice deal larve the following bibranches.
8.1.1 Unique therefore the following bibranches.
8.1.2 Unique therefore so of parcheses.
8.1.3 Unique therefore and unique sets of parcheses.
8.1.4 Unique therefore and unique sets of parcheses.
8.1.3 Unique therefore and unique sets of parcheses.
8.1.4 Unique therefore and unique sets of an observation of the test.
8.1.3 Bread same and parchase date of annotation of the test.

6.1.4 Brend name, sampling range, and copiration data of the length of stain annonies detector tables. The brend series and model of the sampling pump should also be recorded. If another type of annonies detectors hash for annonies detectors hash for the transmitter present area concentrations, their arrange, the intrasion coefficient. Review shall list the two test area concentrations, their arrange, the intrasion coefficient. Review data shall be recorded as an additional tast.
1.2 The evaluation of the data shall be recorded as an additional tast.
1.2 The evaluation of the data shall be submitted as an additional tast.
and the data of the tast. Any detectable, and the data of the tast. Any detectable, and the data the file optic data the static for the mathematic file press the static tast. When other emmonis detectors are used, a lower

level of detection is possible and (t shoul specified as the pass fail criteria.

Castle

Vigually inspect all parts of the suit to be sure they are positioned correctly and secured tightly before putting the suit back into envior. Special care about he suits to cramine each otherst valve to make sure it is not blocked. Care should also be exercised to assure that the inside and outside of the suit is completely dry before it is put into storage.

Apprendix B – General Descriptions and Differentiate of the Lorent of Protocolins and Protocoline Gase This approach: each farch information about personal protocolive equipments (PTR) protocolins for this version. As required by the standard. FPR must be advanted which will protocol coupleynes from the specific bearest which they are likely to comments of this section. Marcada which will protocol coupleynes from the specific bearest which they are likely to comments the standard. FPR must be advanted by the standard of the pro-cessive symphotic starts from the base of potential bearest is the start of the Basered is a waiting of Eacture. Kay Eacture to base of potential bearest is second to the FPR measuries (and second is second a spectrate to these baserption, ingention, and oppe or aids contactly and the performance of the FPR measuries (and second is performance of the FPR measuries) (and the performance of the FPR measuries) (and the performance of the FPR measuries) (and second is most of all optime of period by PPR is measured baseries a substance and poorthy, or not of all optime of these taxes and poorthy, or not of all optime of the period and baseries be protocoline from the period baseries be advanted which will provide the exposure of the over description, or the exposure of the protocoline period only the exposure of the over description, or the exposure of the protocoline period of the protocoline taxes of the protocoline taxes of the protocoline of the protocoline taxes of the taxes of the protocoline of the protocoline

break through may not prove a hasardown area. There factors in this selection process to be considered are matching the FPS to this suppoper is work requirements and task-specific conditions. The durability of FPS naterials, such as twar strength and essen strength, should be considered in relation to the employee's tasks. The effects of FPS in relation to best stress and task duration are provide sufficient protocolog, or to protocol compensate FPS take germents, sufts or equipment. The more that is known about the basards at the site, the easer the job of FPS selection becomes. As more information about the basards and conditions at the tasks at hand. The following are guidelines which an employer can use to begin the tasks at hand. The following are guidelines which an employer can use to be first estimated at first FPE. As noted above, the tasks at hand. The following are guidelines which an employer can use to be first estimated at the star area and the tasks of the statementics may aggest the tasks at hand. The following are guidelines which an employer can use to be above, the site information may aggest the tasks of the statement protection is readed from the different protection is reals to be above the statement protection is reals of the space prior can use to be above the site information may aggest the tasks of the statement protection is reals from the statement protection is reals from the statement protection is reals of the states of the statement protection is reals from the statement protection is reals of the states of the statement protection is reals of the states of the statement protection is reals and the states of the statement protection is reals and the states of the statement protection is reals and the states of the statement protection is reals and th

work. It should be cautioned that the listing below does not fully address the performance of the specific PPE material in relation to the specific hazards at the job site, and that PPE selection, evaluation and re-selection is an ongoing process until sufficient information about the hazards and PPE performance is obtained

Part A. Personal protective equipment is divided into four categories based on the degree of protection afforded. (See Part B of this appendix for further explanation of Levels A. B. C. and D hazards.)

I. Level A-To be selected when the greatest level of skin, respiratory, and eye protection is required.

The following constitute Level A

equipment; it may be used as appropriate; 1. Positive pressure, full face-piece selfcontained breathing apparetus (SCBA), or positive pressure supplied air respirator with escape SCBA, approved by the National Institute for Occupational Safety and Health (NIOSH).

2. Totally-encapsulating chemical-protective suit.

- 3. Coveralls.¹
- 4. Long underwear.¹
- 5. Gloves, outer, chemical-resistant.
- 6. Gloves, inner, chemical-resistant.
- 7. Boots, chemical-resistant, steel toe and shank.

8. Hard hat (under suit)." 9. Disposable protective suit, gloves and boots (depending on suit construction, may -

be worn over totally-encapsulating suit). II. Level B-The highest level of respiratory protection is necessary but a lesser level of skin protection is needed.

The following constitute Level B equipment; it may be used as appropriate.

1. Positive pressure, full-facepiece selfcontained breathing apparatus (SCBA), or positive pressure supplied air respirator with escape SCBA (NIOSH approved). 2. Hooded chemical-resistant clothing

(overalls and long-sleeved jacket; coveralls; one or two-piece chemical-splash suit disposable chemical-resistant overalls).

3. Coveralis, 1

4. Gloves, outer, chemical-resistant.

5. Gloves, inner, chemical-resistant.

6. Boots, outer, chemical-resistant steel toe and shank.

7. Boot-covers, outer, chemical-resistant (disposable), 1

- 6. Hard bat. 1
- 9. [Reserved]
- 10. Face shield.1

III. Level C-The concentration(s) and type(s) of airborne substance(s) is known and the criteria for using air purifying respirators are met.

The following constitute Level C

equipment; it may be used as appropriate. 1. Full-face or half-mask, air purifying

respirators (NIOSH approved). 2. Hooded chemical-resistant clothing

(overalls: two-piece chemical-splash suit; disposable chemical-resistant overalls). 3. Coveralle. 1

4. Gloves. outer, chemical-resistant.

5. Gloves, inner, chemical-resistant.

6. Boots (outer), chemical-resistant steel toe and shank."

- 7. Boot-covers, outer, chemical-resistant (disposable) 1.
- 8. Hard hat.1
- 9. Escape mask.1
- 10. Face shield."
- IV. Level D-A work uniform affording
- minimal protection, used for nuisance contamination only.
- The following constitute Level D
- equipment; it may be used as appropriate:
- 1. Coveralls.
- 2. Gloves. 1
- 3. Boots/shoes, chemical-resistant steel toe and shank.
- 4. Boots, outer, chemical-resistant
- (disposable). 1 5. Safety glasses or chemical splash
- goggles*.
 - 6. Hard bat.
 - 7. Escape mask.¹
 - 8. Face shield.
- Part B. The types of hazards for which levels A. B. C. and D protection are appropriate are described below:
- L Level A-Level A protection should be

used when:

1. The hazardous substance has been identified and requires the highest level of protection for skin, eyes, and the respiratory system based on eithef the measured (or potential for) high concentration of atmospheric vapors, gases, or particulates; or the site operations and work functions involve a high potential for splash, immersion, or exposure to unexpected vapors, gases, or particulates of materials that are harmful to skin or capable of being absorbed through the skin;

2. Substances with a high degree of hazard to the skin are known or suspected to be present, and skin contact is possible; or

3. Operations are being conducted in confined, poorly ventilated areas, and the absence of conditions requiring Level A have not yet been determined.

IL Level B-Level B protection should be used when:

1. The type and atmospheric concentration of substances have been identified and require a high level of respiratory protection. but less skin protection;

2. The atmosphere contains less than 19.5 percent oxygen; or

3. The presence of incompletely identified vapors or gases is indicated by a directreading organic vapor detection instrument, but vapors and gases are not suspected of containing high levels of chemicals harmful to skin or capable of being absorbed through the skin.

Note: This involves atmospheres with IDLH concentrations of specific substances that present severe inhalation hazards and that do not represent a severe skin hazard; or thet do not meet the criteria for use of air-purifying respirators.

III. Level C-Level C protection should be used when:

1. The atmospheric contaminants, liquid splashes, or other direct contact will not

adversely affect or be absorbed through any exposed skin:

2. The types of air contaminants have been. identified, concentrations measured, and an air-purifying respirator is available that can remove the contaminants; and

3. All criteria for the use of air-purifying respirators are met.

IV. Level D-Level D protection should be used when:

1. The atmosphere contains no known hazard; and

2. Work functions preclude splashes, immersion, or the potential for unexpected inhalation of or contact with hazardous levels of any chemicals.

Note: As stated before, combinations of personal protective equipment other than those described for Levels A, B, C, and D protection may be more appropriate and may be used to provide the proper level of protection.

As an aid in selecting suitable chemical protective clothing, it should be noted that the National Fire Protection Association is developing standards on chemical protective clothing. These standards are currently undergoing public review prior to adoption. including

NFPA 1991-Standard on Vapor-Protective Suits for Hazardous Chemical Emergencies (EPA Level A Protective Clothing)

NFPA 1991-Standard on Liquid Splash-**Protective Suits for Hazardous Chemical Emergencies (EPA Level B Protective** Clothing)

NFPA 1993-Standard on Liquid Splash- . Protective Suits for Non-emergency, Nonflammable Hazardous Chemical Situations (EPA Level B Protective Clothing)

These standards would apply documentation and performance requirements to the manufacture of chemical protective suits. Chemical protective suits meeting these requirements would be labelled as compliant with the appropriate standard. When these standards are adopted by the National Fire Protection Association, it is recommended that chemical protective suits which meet these standards be used.

Appendix C-Compliance Guidelines

1. Occupational Safety and Health Program. Each hazardous waste site clean-up effort will require an occupational safety and health program headed by the site coordinator or the employer's representative. The purpose of the program will be the protection of employees at the site and will be an extension of the employer's overall safety and health program. The program will need to be developed before work begins on the site and implemented as work proceeds as stated in paragraph (b). The program is to facilitate coordination and communication of safety and health issues among personnel responsible for the various activities which will take place at the site. It will provide the overall means for planning and implementing the needed safety and health training and job orientation of employees who will be working at the site. The program will provide the means for identifying and controlling worksite hazards and the means for

^{&#}x27; Optional, as applicable.

maniforing program effectiveness. The program will need to corver the coordinates or the employee's manager on the site for the safety and health of employees at the site, and the initial markets with contractors or support services as to what each employee's safety and health employees on the site. Each contractor on the site meeds to have its own asfety and health program so atroctured that it will smoothly interface with the program of the site coordinator or principal contractor. Also these employees involved with treating a contractor. Also these employees involved with treating a contractors wasts as covered in paragraph (p) must have inplemented, e acting cold basistic program for their employees. This program is to include the bases of communication programs is to include the paragraphs (p)(7) and (b)(4) as parts of the employeers comproductive overall acting required in paragraphs (p)(7) and the treating required is paragraphs (p)(7) and the treating required is paragraphs (p)(8) as parts of the employeers comproductive overall acting and health program. This program is to be in

Tank etc or variplace safety and basits program will need to include the following (1) Policy stammars of the program will need to include the following is approximate and controlling workplace and the file of the size safety and basits program the objectives of the program or antibod for the the development and communication to employee of the various place, work rule, and basits and provides and supply and communication to supply and the rule of the various place, work rule, and the file of the size safety and basits or provide and the file operating workplace and supply and communication to supply and the rule of the various place, work rule, and and power for training or provide the program and the program and the program the objectives and of the program the file of explores the officitives of the program the size or vertiples a should be trying on the size or vertiples a should be trying on the size or vertiples a should be trying and communication to be the rule of the program the size or vertiples a should be trying and the individed for a vertiple or the size of the state of the program the size or vertiples a should be trying on the size or vertiples a should be trying and the individe or ultravely the size of the program and the individe or ultravely to be work to be more or the size or vertiples a should be trying and the individe or ultravely will need to be investiged to determine what mersed is the program and the individe or ultravely will need to be work to be program to be the more of the program the size or vertiples a should be trying to a freetback method in the individe of the program to be the more of the size or t

Explaners are pool examples of professional statustication of the employee's properties and product to the sequences of programs of the other sequences and products and the end of the employee's regulations are discribed account of the OSIA employee's regulations are discribed account of the oSIA employee's regulations are discribed account of the oSIA employee's regulations are also also product account of the oSIA employee's requirements of programs the employee's and basility programs and the employee's requirement program. The training productions approaching the employee account of the OSIA employees the analysis of the requirement program is a subsequent program. The training programs for employees the analysis of the requirement of the osial and others and the end of the requirement is program. The training programs for employees the analysis of the requirement of the osial and others and the end of the requirement of the ossial and others and the end of the requirement of the ossial and others and the end of the requirement of the ossial and others and the end of the requirement of the ossial and others and the end of the requirement of the ossial and others and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the end of the requirement of the ossial and the ossial of the requirement of the ossial and th

response plan and the state emergency response plan.
Specialist employees and as technical coperts that work with basardous materials a their require plan to the subport sector and use of personal protective equipment latitudian to an analytic the care and use of personal protective equipment latitudian command gravers and have their provide the care and use of personal protective equipment latitudian command gravers and have the latitudian command gravers and have the care and use of personal protective equipment latitudian command gravers and have the care and use of personal protective equipment latitudian common to here the access and use of personal, who work for public works dependent occurs of the based to the incident rooms of the harande they have a substance, and search be accessed by any be called to the incident rooms of the harande they have a substance, and a support personal, who have a base a part of the accessed by the state of the search of the search of the state of the search of the state of the search of the state of the search of the search of the search of the state of the search of the search of the search of the state of the search of the search of the state of the search of the state of the search
A. Energency response plans. States, slong with designated districts within the states, will be developing or have developed local emergency response plans. These state and district plans should be utilized in the stategy response plans called for in the stategy response plans called for in the stategy response plans is compatible with the local plan. The major reference being used to aud in developing the state and bocal district plans is the *Hazardour Materials Energyency Plannung Caude*. NRT-1. The current Energy Plans is the *Hazardour Materials Energy Plannes* (CAA's CHEMTREC, and the Play Service Energency Management) Haadbook may also be used as resources.

And on the second secon

proposare pairs can use property encoded in the configurary plan may be solutioned in the conceptanty response plan required is spin 120 or otherwise lay laydles for exployer and anyways us. 5. Pernonal protocolve equipered programs. The purpose of personal productive desking and equipment (PTE) is to solid or include individuals from the character, physical, and biologic barants that may be encountered at a barantown minimum entry.

As elemented in Appendix R, we single combined as of presentive equipment and clothing is expected of presenting equipment and conjunction with other presenting equipment and and its effectiveness evaluated periodically. The use of PPE can itself create eigenfactuat worker basands, such as bast stress, pipelacual and psychological stress, and imputed young, mobility, and communication. For any given situation, equipment and clothing about be selected that provide an adequate large of presented. All provide an adequate as well as under-protection, can be basardous and should be avoided where possuble.

Two basic objectives of any PPE program should be to protect the wearer from safety and bealth hazards, and to prevent injury to the wearer from incorrect use and/or mailunction of the PPE. To accompliab these goals, a comprehensive PPE program should include hazard identification, modical monitoring, environmental surveillance, selection, use, maintenance, and decontamination of PPE and the associated training.

The written FFS program should include policy extensests, procedures, and guidelines. Copies sheald be made available to all employees, and a reference copy should be made evailable at the worksite. Technical data on equipment, maintenance manuals, relevant regulations, and other essential information should also be collected and maintained

9. Incident command system (ICS). Paragreph 1910.120(q)(3)(ii) requires the implementation of an ICS. The ICS is an organized approach to effectively control and manage operations at an emergency incident. The individual is charge of the ICS is the sensior official responding to the incident. The individual is charge of the ICS is the sensior official responding to the incident. The individual is charge of the ICS is the service. During large complex fires involving several companies and many pieces of spheretus, a command post would be established. This enabled one individual to be in charge of managing the incident, rather than having several officers from different companies making separate, and sometimes conflicting, decisions. The individual is charge of the command post versul delegate responsibility for performing various usuls to subordinate officers. Additionally, all communications were routed through the command post to reduce the number of radio transmissions and eliminate confusion.

However, stalings, testin, and sil decisions were made by one individual. The ICS is a very similar spaces, except it is implemented for emergency response to all individuals, both longs and mail: that involve heathers, both longs and mail: that involve heathers, both longs and mail: that involve

For a small basishest, the basilvitiani in charge of the RTB may perform many that of the KCS. There may not be any or Fith, diagonizes of units to estronizates. For example, in response to a small betchest, the hullvideal is charge of the RCS, is address to normal command activities, may because the activity officer and may designed us a beat complayer (with proper equipment) as a beat up to provide estimations of messed. ORBM does recommend, however, that a baset two employees be designated as beat.up percented into the societance useded any include recom

To filmitus the operation of the KCS, the following accesses might develop during a small buddent, such as an overtamed task truck with a small lask of filmmable liquid. The first responding senior afficer would implement and take so accessed of the KCS. That person would size up the incident and determine if additional personnal and apparents were accessed by would defend what actions to take to control the basis and determine the proper lavel of personnal protective equipment. If additional assistance is not needed, the individual in charge of the ICS would implement actions to stop and control the leak using the favorest number of personnel that can affectively accompliab the tasks. The individual in charge of the ICS then would designate himself as the safety officer and two other employees as a back-up in case reacte may become necessary. In this scenario, decommand and procedure would not be necessary.

A large complex incident may require many employees and difficult, time-consuming efforts to control. In these situations, the individual in charge of the IC3 will wont to delegate different tasks to subordinates to order to maintum a span of control that will heap the number of subordinates, that are reporting, to a manageable larvel. Delegation of task at large incidents may be by location, where the incident econe is divided into sectors, and subordinate different coordinate activities within the sector that they have been assigned.

Delegation of tasks can also be by function. Some of the functions that the individual in charge incident are: medical services; evacuation: water supply; resources (equipment, apparatus); media relations; selety; and, site control (Integrate activities with police for crowd and traffic control). Also for a large incident, the individual in charge of the ICS will designate several omployees as back-up personnel: and a number of safety officers to monitor conditions and recommend safety precautions.

Therefore, no matter what size or complexity an incident may be, by implementing an ICS there will be one individual in charge who makes the decisions and gives directions; and, all actions, and communications are coordinated through one

cannot be one of communed. Reals a cynamic should reduce an and contrast a science and spectra and spectr

Appendix D-Referen

The following references may be consulted for further information on the subject of this

standard: 1. OSHA Instruction DFO CPL 2.70-January 29, 1996, Special Emphasis Program: Hazardow: Waste Siles. 2. OSHA Instruction DFO CPL 3-137A-January 29, 1996, Technical Assistance and Guidelines for Superfund and Other Hazardow Waste Sile Activities. 3. OSHA Instruction DTS CPL 2.76-Jenuary 29, 1998, Hazardow Waste Activity Form, OSHA 173.

A. Hozardous Wasts .nspections Reference
 A. Hozardous Wasts .nspections Reference
 Nanual, U.S. Department of Labor.
 Occupational Safety and Health
 Administration, 1988.
 Amministration, 1988.
 Amministration, 1988.
 Administration of Understain...ing Among
the National Institute for Occupational Safety
and Haalth, the Occupational Safety and
 Health Administration, the United States
 Coast Guard, and the United States
 Environmental Protection Agency, Guidance
 for Worker Protection During Heardous

Standard Operating Sofety Guidelines:
 Standard Operating Sofety Guidelines:
 U.S. Environmental Protection Agency, Office of Emergency and Remodul Response, of Emergency and Remodul Response, Hanardova Response Support Division, Environmental Response Team: November

10. Occupational Safety and Health Guidance Manual for Hearriors Wate Site Activities, Natural Institute for Occupational Safety and Health (NIOSH), Occupational Safety and Health Administration (OSHA), U.S. Coast Geard (USCC), and Environmental Protection Agency (EPA): October 1988.

Protecting Health and Safety et Hamardous Waste Siles: An Overview, U.S. Eavironmental Protection Agency, EPA/023/ 9-65/000: September 1982.
 Hamardous Wante Sile and Hamardous Substance Emergencies. NIOSH Worker Balletin. U.S. Department of Health and Human Service. Public Health Service. Centers for Disease Control. Netional Institute for Occupational Safety and Health December 1982.
 Personal Protective Equipment for Hanardous Materials Incidence: A Selection Guider U.S. Department of Health Conter Services. Public Health Service. Centers for Disease Control. Netional Institute for Occupational Safety and Health: Octobe 1984.
 Fire Service Emergency Management Handbook International Association of Pite Chiefs Foundation, 101 East Holly Avenue. Unit 105. Sterling. VA 22170, Jamuary 1983.
 Emergency Response on Hamardoux U.S. Department of the Congress on Hamardoux Mategement Agency. Washington, DC, July 1994.
 Workbook for Fire Commond Alan V. Brunacial and J. David Beageron, National

Fire Protection Association, Battarymarch Park, Quincy, MA 02283, 1988. 18. Fire Commond. Alan V. Brunsetial, National Fire Protection, Batterymarch Park, Quincy, MA 02283, 1983. 19. Incident Commond System, Fire Protection Publications, Oklahoma State University, Stillwater, OK 74078, 1983. 20. Site Emergency Response Flanning, Chemical Manufacturers Association. Washington, DC 20037, 1986. 21. Hazardous Materials Emergency Planning Guide, NRT-1, Environmental Protection Agency, Washington, DC, March 1987.

1987.

22. Community Teamwork: Working Tagether to Promote Hazardous Adstends Transportation Safety. U.S. Department of Transportation, Weshington, DC, May 1983. 23. Discator, Plannag Guide for Business and Industry, Federal Emergency Management Agency, Publication No. FEMA 141. August 1987.

(The Office of Management and Budget has approved the information collection requirements in this section under control requirements in thi number 1218-0138)

[FR Doc. 69-6962 Filed 5-1-69; 11:54 am]

Treatability Studies Contractor Work Assignments



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D C 20460

JUL 1 2 1989

OFFICE OF SOLID WASTE AND EMERGENCY RESPONS OSWER Directive # 9380.3-01

£

MEMORANDUM

- SUBJECT: Treatability Studies Contractor Work Assignments
- FROM: Henry L. Longest II, Director Waltan Office of Emergency and Remedial Response

TO: Superfund Branch Chiefs, Regions I-X

Purpose

The purpose of this memo is to require that all future remedial and removal work assignments involving treatability – studies contain a provision requiring the contractor to send a copy of the treatability study to the Agency's Superfund Treatability Data Base which is being developed by the Office of Research and Development (ORD). In addition, you are also directed to send a copy of all treatability studies performed to date and which are readily available, to this central repository.

Background

The Agency has initiated a treatability study program to facilitate the performance of and improve the quality of treatability studies performed in support of the Superfund program. The establishment of a Treatability Data Base is an important part of this program if we are to utilize this information to aid in the selection of remedies and the planning of future treatability studies. A centralized repository for treatability studies is not currently in place and knowledge gained from treatability studies is not efficiently communicated. ORD is developing a repository for the studies to aid us in this area.

This repository will provide information to aid in remedy selections on a site-specific basis, improve future planning for treatability studies, and further our knowledge of technologies on a national basis. It is our intention to minimize Regional resources required to maintain the data base in the future by requiring the contractors to assume responsibility for sending treatability studies to the central repository. The treatability studies collected as a result of this effort will ensure that information available reflects current Superfund experience. The treatability study information as well as other pertinent technical information, will be available to the Regions and contractors through the Alternative Treatment Technology Information Center (ATTIC) in FY 1990. Please contact Mike Mastracci at FTS 475-8933 (mail code RD-681 at the U.S. EPA HQ).

Implementation

Work assignment managers and project officers for removal and the remedial projects are to include a provision in all future work assignments requiring that copies of treatability studies be sent to the following address:

> Attn: Ken Dostle U.S. Environmental Protection Agency Superfund Treatability Data Base ORD/REEL 26 W. Martin Luther King Drive Cincinnati, Ohio 45268

The work assignment should also require that the treatability study report provided to ORD be a separate and complete document which is a camera-ready master copy. We are also collecting treatability studies retroactively as well. You are directed to send copies of all treatability studies that are readily available to the address identified above.

The Agency is also developing detailed guidance on planning and performing treatability studies with the first of these planned for distribution in early FY-90. Today's memo will be updated in the future to require that contractors comply with these guidances as well. Your assistance with the development and implementation of this program is appreciated. Please contact Robin Anderson at FTS 382-2446 or Scott Maid at FTS 382-4671 if you have question or comments on the application of this requirement to the remedial or removal program respectively.

cc: OHM Coordinators, Regions I-X ARC Project Officers, Regions I-X ERCS Project Officers, Region I-X REM Project Officers (OERR) Russ Wyer (OERR/HSCD) Tim Fields (OERR/HSCD) Scott Maid (OERR/ERD) Robin Anderson (OERR/HSCD) Mike Mastracci (ORD) Ken Dostle (ORD) Betti Van Epp (OERR/OPM) Joseph Lafornara (OERR/ERT)

Final Guidance on Adminstrative Records for Selecting CERCLA Response Actions



CEC 3 Igno

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE OSWER Directive * 9833.3A-1

MEMORANDUM

SUBJECT: Final Guidance on Administrative Records for Selecting CERCLA Response Actions

FROM:

Don R. Clay Assistant Administ

TO: Regional Administrators, Regions I-X

This memorandum transmits to you our "Final Guidance on Administrative Records for Selecting CERCLA Response Actions." This document replaces the "Interim Guidance on Administrative Records for Selection of CERCLA Response Actions," previously issued on March 1, 1989.

The guidance sets forth the policy and procedures governing the compilation and establishment of administrative records for selecting response actions under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA), as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA). This guidance is also consistent with and expands on Subpart I of the National Oil and Hazardous Substances Pollution Contingency Plan, 55 Fed. Reg. 8859 (March 8, 1990).

This guidance reflects input received from the Regions, Headquarters and the Department of Justice. There have been several drafts of this guidance and comments have been incorporated. I thank you for your assistance.

Attachment

cc: Director, Waste Management Division, Regions I, IV, V, and VII Director, Emergency and Remedial Response Division, Region II Director, Hazardous Waste Management Division, Regions III, VI, VIII, and IX Director, Hazardous Waste Division, Region X Director, Environmental Services Division, Regions I, VI, and VII Regional Counsel, Regions I-X Administrative Record Coordinators, Regions I-X

Privied on Recycled Paper
OSWER Directive No. 9833.3A-1

.

FINAL GUIDANCE ON ADMINISTRATIVE RECORDS FOR SELECTING CERCLA RESPONSE ACTIONS

U.S. Environmental Protection Agency Office of Solid Waste and Emergency Response Washington, D.C. 20460

.

.

	н.	Privileged Documents	34		
	I.	Guidance Documents	37		
	J.	Technical Literature	38		
	к.	Legal Sources	39		
	L.	NPL Rulemaking Docket Information	39		
	м.	RCRA Documents	39		
	Ν.	Post-Decision Information	40		
IV.	INVO	LVEMENT OF OTHER PARTIES	42		
	A.	States	42		
		1. State Involvement in Federal-Lead Sites	42		
		2. Federal Involvement in State-Lead Sites	43		
	в.	Federal Facilities	44		
	c.	ATSDR	45		
	D.	Natural Resources Trustees	45		
v.	DISC	LAIMER	46		
VI.	FURT	HER INFORMATION	46		
GLOSSARY					
APPENDICES					
App	endix	A. Sections 113(j)-(k) of CERCLA	50		
Арр	endix	B. Model File Structure	52		
App	endix	C. Model Index	56		
App	endix	D. Model Position Description for Administrative			
		Record Coordinator	57		
Арр	endix	E. Compendium of Response Selection Guidance	59		
App	endix	F. Model Transmittal Cover Letter	85		
Арр	endix	G. Nodel Document Transmittal Acknowledgement	86		
Арр	endix	H. Nodel Fact Sheet	87		
App	endix	I. Model Notice of Public Availability	88		
Арр	endix	J. Microform Approval Memorandum	89		
App	endix	K. Model Certification	90		
App	endix	L. Preamble to Subpart I of NCP	91		
App	endix	M. Subpart I of the NCP	L01		

-

.

TABLE OF CONTENTS

I.	INTR	ODUCTION
	A.	Purpose and Scope of the Administrative Record 1
	в.	Judicial Review
	c.	Public Participation 4
II.	PROC	EDURES FOR ESTABLISHING THE ADMINISTRATIVE RECORD 4
	A.	Administrative Record Coordinator 4
	в.	Multiple Response Actions 6
	c.	Compilation
	D.	Index
	Ε.	Location
		1. General
		2. Special Documents 9
	F.	Public Availability
		1. General
		2. Remedial Actions
		3. Removal Action
	G.	Maintaining the Record
	H.	Confidential File
	I.	Copying
	J.	Micrographics
	K.	Certification
III	. CONT	ENTS OF THE ADMINISTRATIVE RECORD
	λ.	Remedial Actions
	в.	Removal Actions
	c.	Imminent and Substantial Endangerment
	D.	Public Comments
	E.	Enforcement Actions
		1. Negotiation Documents
		2. PRP-Lead RI/FS
		3. Administrative Orders and Consent Decrees 32
	F.	Excluded Documents
	G.	Draft Documents and Internal Memoranda

•

.

I. INTRODUCTION

A. Purpose and Scope of the Administrative Record

This guidance addresses the establishment of administrative records under Section 113 of the Comprehensive Environmental Response, Compensation, and Liability Act of 1980 (CERCLA), as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA).¹ Section 113(k)(1) of CERCLA requires the establishment of administrative records upon which the President shall base the selection of a response action (see Appendix A for the complete statutory language).

Chapter I of this guidance introduces the purpose and scope of the administrative record. Chapter II reviews procedures for compiling and maintaining the administrative record. Chapter III examines the various types of documents which should be included in the administrative record. Chapter IV discusses how agencies outside EPA are involved in establishing the record. Finally, this guidance includes a glossary of frequently used terms and acronyms as well as several appendices.

Although this guidance is written for use by the United States Environmental Protection Agency (EPA), it can be adapted for use by state and federal agencies required to establish administrative records for the selection of CERCLA response actions. As used in this guidance the term "lead agency" means either EPA, a state or other federal agency, which is responsible for compiling and maintaining the administrative record. As used in this guidance, the term "support agency" means the agency or agencies which furnish necessary data to the lead agency, reviews response data and documents and provides other assistance as requested by the OSC or RPM. This guidance reflects the revisions to the National Oil and Hazardous Substances Pollution Contingency Plan (NCP) published on March 8, 1990, 55 Fed. Reg. 8859 (see Appendices L and M).

The administrative record established under Section 113(k) of CERCLA serves two primary purposes. First, the record contains those documents which form the basis for selection of a response action and under Section 113(j), judicial review of any issue concerning the adequacy of any response action is limited to the record. Second, Section 113(k) requires that the administrative record act as a vehicle for public participation

¹ 42 U.S.C. §9613. References made to CERCLA throughout this memorandum should be interpreted as meaning "CERCLA, as amended by SARA."

in selecting a response action. This guidance document discusses procedures developed to ensure that the lead agency's administrative records meet these twin purposes.

The administrative record is the body of documents that "forms the basis" for the selection of a particular response at a site. This does not mean that documents which only support a response decision are placed in the administrative record. Documents which are included are relevant documents that were relied upon in selecting the response action, as well as relevant documents that were considered but ultimately rejected (e.g., documents "considered or relied on").

This document uses the phrase "considered or relied on" in discussing which documents should be included in the administrative record to indicate that it is EPA's general policy to be inclusive for placing documents in the administrative record. However, this term does not mean that drafts or internal documents are normally included in the administrative record. Lead or support agency draft or internal memoranda are generally not included in the administrative record, except in specific circumstances (see section III.G. at page 33). Thus, the record will include final documents generated by the lead and support agency, as well as technical and site-specific information. Information or comments submitted by the public or potentially responsible parties (PRPs) during a public comment period (even if the lead agency does not agree with the information or comments) are also included in the administrative record (see section III.D. at page 30).

The following principles should be applied in establishing administrative records:

- The record should be compiled as documents relating to the selection of the response action are generated or received by the lead agency;
- The record should include documents that form the basis for the decision, whether or not they support the response selection; and
- o The record should be a contemporaneous explanation of the basis for the selection of a response action.

The effort to establish adequate administrative records encompasses a vast array of people including: Administraive Record Coordinators, Remedial Project Managers (RPMs), On-Scene Coordinators (OSCs), enforcement staff, records management staff, Regional Counsel staff, Community Relations Coordinators (CRCs), other federal agencies, states, CERCLA contractors, and the

public.² This guidance will discuss the roles and responsibilities of these people and how they interact with one another.

B. Judicial Review

Section 113(j)(1) of CERCLA provides that judicial review of any issues concerning the adequacy of any response action shall be limited to the administrative record.

Judicial review based on an administrative record provides numerous benefits. Under Section 113(j) of CERCLA and general principles of administrative law, when the trial court reviews the response action selected, the court is limited to reviewing the documents in the administrative record. As a result, facts or arguments related to the response action that challenging parties present for the first time in court will not be considered.

Record review saves time by limiting the scope of trials, thereby saving the lead agency's resources for cleanup rather than litigation. Courts will not allow a party challenging a decision to use discovery, hearings, or additional fact finding to look beyond the lead agency's administrative record, except in very limited circumstances. In particular, courts generally will not permit persons challenging a response decision to depose, examine, or cross-examine EPA, state or other federal agency decisionmakers, staff, or contractors concerning the selection of the response action.

Furthermore, the administrative record may be cited long after officials responsible for the response decisions have moved into different positions or have left the lead or support agency. Judicial review limited to the record saves time involved in locating former employees who may not remember the facts and circumstances underlying decisions made at a much earlier time.

Moreover, in ruling on challenges to the response action decision, the court will apply the highly deferential "arbitrary and capricious" standard of review set forth in Section 113(j)(2)of CERCLA. Under this standard, a court does not substitute its judgment for that of the decisionmaker. The reviewing court does not act as an independent decisionmaker, but rather acts as a reviewing body whose limited task is to check for arbitrary and capricious action. Thus, the court will only overturn the response selection decision if it can be shown on the

² As used hereinafter in this guidance the term "public" includes potentially responsible parties (PRPs).

administrative record, that the decision was arbitrary and capricious or otherwise not in accordance with the law. However, the extent to which EPA benefits from having judicial review limited to the record depends on the quality and completeness of each record.

C. Publi zicipation

Section 113(k)(2) of CERCLA requires that the public have the opportunity to participate in developing the administrative record for response selection. Section 117 of CERCLA also includes provisions for public participation in the remedial action selection process.³ Both sections reflect a statutory emphasis on public participation. Participation by interested persons will ensure that the lead agency has considered the concerns of the public, including PRPs, during the response selection process. In addition, for purposes of administrative and judicial review, the record will contain documents that reflect the participation of the public and the lead agency's consideration of the public's concerns.

If the lead agency does not provide an opportunity for involvement of interested parties in the development of the administrative record, persons challenging a response action may argue that judicial review should not be limited to the record. The lead agency must, therefore, make the information considered or relied on in selecting a response action available to the public, provide an appropriate opportunity for public comment on this information, place comments and information received from the public in the record, and reflect in the record the lead agency's consideration of this information.

II. PROCEDURES FOR ESTABLISHING THE ADMINISTRATIVE RECORD

A. Administrative Record Coordinator

Each region should have an Administrative Record Coordinator. The Record Coordinator generally has the duty of ensuring that the administrative record files are compiled and maintained according to Subpart I of the NCP and this guidance."

⁶ The "administrative record file" should be distinguished from the "administrative record." The administrative record file refers to the documents as they are being compiled. Until a response action decision has been selected, there is no complete administrative record for that decision. Thus, to avoid creating the impression that the record is complete at any time prior to

4

³ 42 U.S.C. §9617.

OSWER Directive No. 9833.3A-1

The Record Coordinator will not be responsible for deciding which documents are included in a record file. Those decisions should be made by the OSC or RPM, with appropriate consultation of ORC staff. The Record Coordinator's duties ordinarily include:

- Developing procedures for creating record files;
- Ensuring that the public is notified that the record files are available for inspection;
- o Ensuring that the records are available at or near the site:
- Ensuring that the records are available at the regional office or other central location;
- o Coordinating efforts to obtain the necessary documents;
- o Indexing the record files;
- Updating the record files and indices on a regular basis (e.g., quarterly);
- o Ensuring availability of the record file for copying;
- Ensuring that sampling and testing data, quality control and quality assurance documentation, and chain of custody forms are available for public inspection, possibly at a location other than that of the record files;
- Coordinating with ORC staff on questions of relevance and confidentiality of documents submitted for the record files;
- Arranging for production and presentation of the record to court when necessary for judicial review;
- Maintaining the confidential portion of the record files, if necessary;
- o Maintaining the "Compendium of CERCLA Response Selection Guidance Documents";
- Coordinating with states and federal agencies on record files compiled by them; and

the final selection decision, the set of documents is referred to as the administrative record file rather than the administrative record.

 Notifying appropriate personnel of the timing for review of state and federal record files.

Appendix D contains a model position description for an Administrative Record Coordinator.

The Record Coordinator must work closely with RPMs, OSCs, enforcement staff, records management staff, Regional Counsel staff, community relations staff, and the Department of Justice (DOJ) (for cases in litigation).

If the way the record was compiled and maintained is questioned in litigation, the Record Coordinator may be called upon to prepare an affidavit or testify about those procedures. Therefore, the Record Coordinator should be familiar with the procedures associated with the record, and be qualified to fulfill the responsibilities outlined above.

B. Multiple Response Actions

In general, every decision document (e.g., Record of Decision (ROD) or Action Memorandum) must be supported by an administrative record. Under CERCLA, cleanups are often broken up into distinct response actions. At a given site this may include several removal actions, and/or remedial actions known as operable units. For every removal action or operable unit, a separate administrative record must be compiled.

Information relevant to more than one response decision, such as a site inspection report or a preliminary assessment report may be placed in the record file for an initial response action and incorporated by reference in the indexes of subsequent record files for that site.

C. Compilation

The administrative record file should be compiled as relevant documents on the response action are generated or received. Thus, all documents which are clearly relevant and non-privileged should be placed in the record file, entered into the index, and made available to the public as soon as possible. For example, the remedial investigation/feasibility study (RI/FS) work plan, summaries of quality assured data, the RI/FS released for public comment, the proposed plan, and any public comments received on the RI/FS and proposed plan should be placed in the record file as soon as they are generated or received.

When there are questions whether particular documents should be included in the record file, such documents can be segregated and reviewed at regular intervals (e.g., quarterly). For

OSWER Directive No. 9833.3A-1

example, draft documents or documents subject to claims of ' rrivilege should be set aside for review by ORC and other appropriate staff. At critical times, such as prior to the public comment period, the issues regarding these documents should be completely resolved and the documents included in the record file, if appropriate.

The record file should be updated while it is available for public inspection. The additional documents should be placed in the record file and entered in the index. Any updates to the record file should be made to all copies of the record file.

All documents considered or relied on in selecting the response action should be in the record file when a decision document (e.g., a record of decision) is signed. Documents relevant to the response selection but generated or received after the decision document is signed should be placed in a postdecision document file and may be added to the administrative record file in certain circumstances (see section III.N. at page 40).

D. Index

Each administrative record file must be indexed. The index plays a key role in enabling both lead agency staff and members of the public to help locate and retrieve documents included in the record file. In addition, the index can be used for public information purposes or identifying documents located elsewhere, such as those included in the compendium of guidance documents (see Appendix E). The index also serves as an overview of the history of the response action at the site.

The index also provides the lead agency with a degree of control over documents located at or near the site. The creation of an index will prevent persons from altering the record simply by physically adding or removing documents from the record file.

The index should include the following information for each document:

- o Document Number;
- o Document Date date on the document;
- o Document Title one or two line identification. Identify the actual document, not a transmittal memo or other less relevant document. Include sufficient information so the document cannot be confused with another (e.g., the title "report" may be insufficient);

- Author Name and affiliation;
- o Recipient Name and affiliation; and
- o Document Location.

The index can be organized either by subject or in chronological order. If documents are customarily grouped together, as with sampling data and chain of custody documents, they may be listed as a group in the index to the administrative record file. Appendix C contains a model index organized by subject. Computer databases have been helpful in generating and updating the index.

The index should be updated when the record file is updated. It is preferable to update the record file when documents are received, or at least quarterly. Such updates should coincide with the periodic updating of the record file and review of material for which there are questions about relevance or privilege (see section II.C. at page 6). The index hould also be updated before any public comment period commencat. The index should be labeled "draft index" until all relevant documents are placed in the record file. When the decision document is signed, the draft index should be updated and labeled "index."

E. Location

E.1. General

Section 113(k)(1) of CERCLA requires that the administrative record be available to the public "at or near the facility at issue."⁵ Duplicates of the record file may be kept at any other location. A copy of the record file must be located at the regional office or other central location. Both copies of the record file should be available for public inspection at reasonable times (e.g., 9-4, Monday-Friday). In the case of an emergency removal, unless requested, the record file needs to be available for public inspection only at the central location (see section II.F.3. at page 14).⁶

The record file located at or near the site should be placed in one of the information repositories which may already exist for community relations purposes. These are typically located in a library, town hall, or other publicly accessible place. If there is no existing information repository, or if the repository

⁵ See 40 C.F.R. §300.805.

⁶ 40 C.F.R. §§300.805(a)(5) and (b).

does not have sufficient space for the record file, any other publicly accessible place may be chosen to house the record file. When a Superfund site is located at or near an Indian reservation, the centrally located copy of the record file may be located at the Indian tribal headquarters. The Community Relations Coordinator (CRC) should be consulted on the location of the information repository and record file.

The record file should be transmitted to the local repository in coordination with the CRC. The CRC should make the initial contact to establish the local repository and request housing for the record file. The Record Coordinator should make arrangements for delivering the record file to the local repository.

The record file should include an introductory cover letter addressed to the librarian or repository manager (see Appendix F). In addition, a transmittal acknowledgement form should be included to ensure receipt of the record file (see Appendix G). Finally, an administrative record fact sheet should accompany the record to answer questions from the public (see Appendix H). Updates to the record file should be handled in a similar fashion (see section II.C. at page 6).

In addition to the publicly available record file, if feasible, a master copy of the record file should be kept at the regional office or other central location of the lead agency. To preserve the integrity of the master copy of the record file, it should not be accessible to the public. If not feasible to establish a master copy, the lead agency will need to establish an effective security system for the publicly available record file. The master copy of the record file may be maintained in microform to conserve storage space (see section II.J. at page 21).

E.2. Special Documents

Certain documents which are included in the record file do not have to be maintained at or near the site or, in some cases, at the regional office or other central location, because of the nature of the documents and the burden associated with maintaining such documents in multiple locations. These documents, however, must be incorporated in the record file by reference (e.g., in the index but not physically in the record

⁷ If the site is located at a federal facility which requires security clearance, the administrative record file for that site must be located where security clearance is not required. The public must have free access to the record file.

file), and the index must indicate where the documents are publicly accessible. Where a document is listed in the index but 'not located at or near the site, the lead agency must, upon request, include the document in the record file at or near the site.⁸ This applies to verified sampling data, chain of custody forms, and guidance and policy documents. It does not apply to documents in the confidential file.

Unless requested, the following types of documents do not have to be located in multiple locations:

Verified Sampling Data⁹

Verified sampling data do not have to be located in either administrative record file. The sampling data may be left in its original storage location (e.g., Environmental Services Division (ESD) or contract laboratory). Data summary sheets, however, must be located in the record file. The index must list the data summary sheets, reference the underlying verified sampling data, and indicate where the sampling data can be found.

Chain of Custody Forms¹⁰

As with verified sampling data, chain of custody forms do not have to be located in either administrative record file. The chain of custody forms may be left in the original storage location. The index must reference the chain of custody forms and indicate their location.

¹⁰ 40 C.F.R. §300.805(a)(1).

⁸ 40 C.F.R. §300.805(b).

⁹ 40 C.F.R. §300.805(a)(1). "Verified sampling data" are data that have undergone the quality assurance and quality control process. "Invalidated sampling data" have been incorrectly gathered or analyzed and will not be part of the record file. "Unvalidated sampling data" are data which has not yet undergone the quality assurance and quality control process. Because it is superseded by verified data, the unvalidated data are not generally part of the record files. However, such data may in some cases be relied on in selecting a response action, such as an emergency removal where there is no time for verification. Unvalidated sampling data which are relied on in selecting a response action should be included in the record file.

Confidential and Privileged Documents¹¹

When a confidential or privileged document is included in the record file, it should be kept in a confidential portion of the record file. The confidential file should be kept in a locked cabinet at the regional office or other central location. It should not be located at or near the site. The index should identify the title and location of the document, and describe why the lead agency considers it confidential or privileged. Furthermore, the lead agency should summarize or redact the document to make available, to the extent feasible, factual information (especially if such information is not found elsewhere in the record file and is not otherwise available to the public). This summary or redaction should be performed as soon as possible after the determination that a document is privileged or confidential, and inserted in the portion of the record file available to the public and included in the index. See also section III.H. at page 34.

Guidance and Policy Documents¹²

Guidance and policy documents that are not site specific are available in a compendium located in the regional office. ("Compendium of CERCLA Response Selection Guidance Documents," Office of Waste Programs Enforcement, May 1989.) This eliminates the need for reproducing copies of frequently used documents for each site record file. The documents in the compendium need not be physically included in the record file, but the guidance and policy documents considered or relied on in selecting the response action must be listed in the record file index along with their location and availability. See also section III.I. at page 37 and Appendix E.

Technical Literature¹³

Publicly available technical literature that was not generated for the site at issue (e.g., an engineering textbook), does not have to be located in the regional office or other central location or at or near the site. The document must be clearly referenced in the index. However, technical literature not publicly available must be physically included in the record file at the regional office or other central location and at or near the site. See also section III.J. at page 38.

¹¹ 40 C.F.R. §300.805(a)(4).

 12 40 C.F.R. §300.805(a)(2).

¹³ 40 C.F.R. §300.805(a)(3).

F. Public Availability

F.1. General

Section 113(k) of CERCLA specifies that the administrative record "shall be available to the public." In satisfying this provision, the lead agency must comply with all relevant public participation procedures outlined in Sections 113(k) and 117 of CERCLA. The NCP (see Appendices L and M) contains additional requirements on public availability (see also "Community Relations in Superfund: A Handbook," October 1988 - OSWER Directive No. 9230.0-3A; "Community Relations During Enforcement Activities," November 3, 1988 - OSWER Directive No. 9836.0-1A).

The availability of the record file will vary depending upon the nature of the response action. Different procedures are outlined below for remedial and removal response actions.

In all cases, the lead agency should publish a notice of availability of the record file when the record file is first made available for public inspection in the vicinity of the site at issue.¹⁶ The notice should explain the purpose of the record file, its location and availability, and how the public may participate in its development.

The notice should be published in a major local newspaper of general circulation. The newspaper notices should be distributed to persons on the community relations mailing list. These notices should also be sent to all known PRPs if they are not already included on the community relations mailing list. As PRPs are discovered, the lead agency should add their names to the community relations mailing list and mail them all the notices sent to the other PRPs. Publication of the notice should be coordinated with the community relations staff. A copy of the notice of availability and list of recipients should be included in the record file. Appendix I contains a model notice of availability.

This public notice may be combined with other notices for the same site, such as a notice of availability of the community relations information repository, if they occur at the same time. In addition to the required newspaper notice, the public can be informed of the availability of the record file through existing mechanisms (e.g., general and special notice letters, Section 104(e) information requests, and the community relations mailing list). In addition, Headquarters will publish notices in the

¹⁴ See 40 C.F.R. \$300.815(a) and \$\$300.820(a)(1) and (b).

Federal Register. They will be published quarterly and will list sites where remedial activity is planned.

F.2. Remedial Actions

The administrative record file for a remedial action must be available for public inspection when the remedial investigation begins.¹⁵ For example, when the remedial investigation/ feasibility study (RI/FS) work plan is approved, the lead agency must place documents relevant to the selection of the remedy generated up to that point in the record file. Documents generally available at that time include the preliminary assessment (PA), the site investigation (SI), the RI work plan, inspection reports, sampling data, and the community relations plan. The lead agency must continue to add documents to the record file periodically after they are generated or received during the RI/FS process.

The record file must be publicly available both at a regional office or other central location and at or near the site (see section II.E. at page 8).¹⁰ In addition, the notice of availability should be sent to persons on the community relations mailing list, including all known PRPs.

With the completion of the RI/FS, the lead agency should undertake the following public participation procedures:

- Prepare a proposed plan which briefly analyzes the remedial alternatives evaluated in the detailed analysis of the RI/FS and proposes a preferred remedial action alternative;
- Make the RI/FS report and proposed plan available in the record files both at a regional office or other central location and at or near the site;
- Publish in a major local newspaper of general circulation a notice of availability and brief analysis of the RI/FS report and proposed plan. The notice should include the dates for submission of public comments;
- Mail the notice or copy of the notice to all PRPs on the community relations mailing list;
- o **Provide a formal comment period of not less than 30 calendar** days for submission of comments on the proposed plan. Upon

¹⁶ 40 C.F.R. §300.805(a).

¹⁵ 40 C.F.R. §300.815(a).

timely request the lead agency will extend the public. comment period by a minimum of 30 additional days.¹⁷ [Note: The lead agency is encouraged to consider and respond to significant comments that were submitted before the public comment period. Considering early comments provides practical benefits both substantively and procedurally. Early comments may provide important information for the selection decision, and early consideration provides the public (and, particularly, PRPs) with additional informal opportunities for participating in the decisionmaking process.];

- Provide the opportunity for a public meeting(s) in the affected area during the public comment period on the RI/FS and proposed plan;
- Keep a transcript of the public meeting(s) on the RI/FS and proposed plan held during the comment period and include a copy of the transcript in the record file;
- Prepare a discussion (to accompany or be part of the decision document) of any significant changes to the proposed plan which occurred after the proposed plan was made available for public comment which are reflected in the ROD;
- Prepare a response to each of the significant comments submitted during the public comment period to accompany the ROD (see section III.D. at page 30); and
- Publish in a major local newspaper of general circulation a notice of the availability of the ROD and make the ROD available to the public before beginning any remedial action, as required under Section 117(b) of CERCLA.

Comments received after signing the ROD should be placed in a post-decision document file and may be added to the record file in certain situations (see section III.N. at page 40).

F.3. Removal Actions

Section 113(k)(2)(A) of CERCLA requires that the EPA establish procedures for the appropriate participation of interested persons in the development of the administrative record for the selection of a removal action. "Appropriate" participation depends on the nature of the removal, as outlined below.

¹⁷ 40 C.F.R. §300.430(f)(3)(i)(c).

Time-critical Removal Actions

A time-critical removal action is a removal action for which, based on the site evaluation, the lead agency determines that a period of less than six months exists before on-site removal activities must be initiated. This category includes emergency removal actions which are described in greater detail below.

The administrative record file for these actions must be available for public inspection no later than 60 days after the initiation of on-site removal activity. Where possible, the record file should be made available earlier. The record file must be available both at the regional office or other central location and at or near the site at issue.

If, however, on-site cleanup activity is initiated within hours of the verification of a release or threat of a release and on-site cleanup activities cease within 30 days (emergency actions), the record file need only be available at the regional office or other central location, unless it is requested that a copy of the record file be placed at or near the site.¹⁸

For all time-critical removals, a notice of the availability of the record file must be published in a major local newspaper and a copy of the notice included in the record file. This notice should be published no later than 60 days after initiation of on-site removal activity.¹⁹

A public comment period of not less than 30 days should be held in appropriate situations.²⁰ In general, a public comment period will be considered appropriate if cleanup activity has not been completed at the time the record file is made available to the public and if public comments might have an impact on future action at the site. If a public comment period is considered appropriate, it should begin at the time the record file is made available for public inspection. Note, however, that even if an action is completed before the record file is available, the record file should be made available to the public. The notice for the public comment period may be combined with the notice of availability of the record file if they occur at the same time. The notice should be mailed to all PRPs on the community

²⁰ 40 C.F.R. §300.415(m)(2)(ii).

¹⁸ 40 C.F.R. §300.805(b).

¹⁹ 40 C.F.R. §300.415(m)(2)(i).

.

relations mailing list. The notice should also be sent to all known PRPs if they are not already on the community relations mailing list.

The lead agency must respond to all significant comments received during the public comment period and place the comments and the responses to them in the record file (see section III.D. at page 30).²¹ Whether or not the lead agency holds a public comment period, comments received by the lead agency before the decision document is signed and related to the selection of the removal action must be placed in the record file. For information, including comments, generated or received after the decision document is signed, see section III.N. at page 40.

Non-Time-Critical Removal Actions

A non-time-critical removal action is a removal action for which, based on the site evaluation, the lead agency determines that a planning period of at least six months exists before onsite removal activities must be initiated.

The administrative record file for a non-time-critical removal action must be made available for public inspection when the engineering evaluation/cost analysis (EE/CA) is made available for public comment.²² The record file must be available at the regional office or other central location and at or near the site. A notice of the availability of the record file must be published in a major local newspaper and a copy of the notice included in the record file. The notice should be published in a major local newspaper of general circulation. In addition, Headquarters will publish these notices in the Federal Register. They will be published quarterly and will list sites where non-time critical removal activity is planned. The newspaper notice should be distributed to persons on the community relations mailing list and placed in the record file. These notices should also be sent to all known PRPs if they are not already on the community relations mailing list. As PRPs are discovered, the lead agency should add their names to the community relations mailing list and mail them all the notices sent to the other PRPs. Publication of the notice should be coordinated with the community relations staff. A copy of the notice of availability should be included in the record file. Appendix I contains a model notice of availability.

²² 40 C.F.R. §300.415(m)(4).

²¹ 40 C.F.R. §300.415(m)(2)(iii).

OSWER Directive No. 9833.3A-1

A public comment period on the EE/CA of not less than 30 days must be held so that interested persons may submit comments on the response selection for the record file. Upon timely notice, the lead agency will extend the public comment period by a minimum of 15 days.²³ A notice of the public comment period may be combined with the notice of availability of the record file if they occur at the same time. The lead agency must respond to all significant comments received during the public comment period and place the comments and the responses to them in the record file (see section III.D. at page 30).²⁴

The lead agency is encouraged to consider and respond to significant comments that were submitted before the public comment period. Considering early comments provides practical benefits both substantively and procedurally. Early comments may provide important information for the selection decision, and early consideration provides the public (and, particularly, PRPs) with additional informal opportunities for participating in the decision making process.

Comments generated or received after the decision document is signed should be kept in a post-decision document file. They may be added to the record file in certain situations (see section III.N. at page 40).

G. Maintaining the Record

Document room procedures should be established to ensure orderly public access to the record files. In establishing public access procedures, the security and integrity of the record files must be maintained at all times.

Each regional office or other central location should have a reading area where visitors are able to review the record files. The record file must be available during reasonable hours (e.g., 9-4, Monday-Friday). The public reading area should include, wherever feasible:

- o Administrative record files;
- o Guidance Compendium (see section III.I. at page 37);
- o Access to a copier; and
- o Sign-in book.

23 40 C.F.R. §300.415(m)(4)(iii).

²⁴ 40 C.F.R. §300.415(m)(4)(iv).

Controlled access to the files is accomplished by use of a 'visitor sign-in book. Sign-in books help minimize instances in which documents are lost or damaged. They also provide documentation of the lead agency's efforts to provide public access to the record files. Pertinent information recorded in the book should include:

- o Date of visit;
- o Name;
- o Affiliation;
- o Address;
- o Phone number;
- o Site documents viewed; and
- Cost of copied materials (if applicable).

The lead agency may choose not to use sign-in books if the books deter the public from reviewing the record files.

Since documents in the record file should be complete, properly organized and legible, the integrity of the record file must be maintained. If possible, storage and reading areas should be supervised to maintain proper security. Documents should not leave the document room or be left unattended. To the extent feasible, the Administrative Record Coordinator should check the order of the documents after being viewed by the public to be certain all documents have been returned intact. The documents in the record file should be kept secure, either in a locked room or in locked cabinets.

The record file located at or near the site should be handled with similar care. If possible, the record file should be treated as a non-circulating reference; it should not leave the local repository except under supervision. The phone number of a record file contact should be provided to record file users and to the manager of the local repository so that problems can be identified and resolved. This information can be included in an informational fact sheet accompanying the record file (see Appendix H). In addition, the Record Coordinator should plan periodic reviews of the local record files.

Where the site is a fund-lead or PRP-lead, EPA should retain (in addition to the publicly available record file) a master copy of the record file at the regional office or other central

location, if feasible. Where a state or other federal agency is the lead agency at a site, EPA should assure that the state or other federal agency maintains (in addition to the publicly available record file) a master copy of the record file. The record files are permanent records that must be retained.

As to the local repository, the statute and regulations are silent concerning the duration of public availability of the record file. The lead agency's primary concern is public participation in development of the administrative record. Following initiation of the response action, public interest in background information other than the Record of Decision or RI/FS may wane. In any event, the statutory provisions for judicial review and deadlines for filing cost recovery actions provide useful references for keeping the record file publicly available. See Sections 113(g) and (h) of CERCLA.

Where there is ongoing (or possible) litigation, the record file in the regional or other central location should be available at least until the litigation is over.

The record file continues to serve as a historical record of the response selection, even after the statute of limitations for cost recovery action has passed. Where there is considerable public interest, the local repository may wish to keep the record file available for public viewing.

H. Confidential File

In certain situations, documents in the record file may be subject to an applicable privilege (see section III.H. at page 34). To the extent feasible, information relevant to the response selection which is contained in a privileged document should be summarized or redacted as to make the document disclosable and then included in the publicly accessible portion of the record file. The privileged document should be included in a confidential portion of the record file.

The Administrative Record Coordinator should maintain a confidential portion of the record file for privileged documents. These documents should be listed in the index to the entire record file and identified as "privileged." The index should identify the title and location of the privileged document, and describe the basis for the asserted privilege.

The confidential portion of the record file should be stored in locked files at the regional office or other central location

²⁵ See 40 C.F.R. §300.810(d).

OSWER Directive No. 9833.3A-1

and should not be located at or near the site. The confidential portion of the record file should be separate from the publicly available record file to protect against inadvertent disclosure. Each privileged document should be stamped "confidential" at the bottom of each page of the document. Where the material is not a written document (such as a computer disk or cassette tape) the jacket should be stamped "confidential." A complete list of all materials contained in the confidential portion of the record file should be maintained by the Record Coordinator. The Record Coordinator should also maintain a log which will include the time, date, document name, and will identify persons checking out and returning materials to the confidential file.

As soon as a new record file is established, a routine access list for the confidential file should be prepared for each record file. When EPA is the lead agency, this routine access list must be approved by the Waste Management Division Director or the Environmental Services Division Director, and ORC. Once approval is given, persons on the list will be able to access the confidential files through the Record Coordinator. No one should have access to the confidential files other than those identified on the routine access list. For state or other federal agencylead sites, the Regions should take steps to insure that state or other federal agencies develop routine confidential file access list procedures.

This policy and procedure for privileged materials does not supersede any policy and procedures established under the Freedom of Information Act (FOIA), 5 U.S.C. §552, and EPA regulations implementing FOIA at 40 C.F.R. Part 2. Upon receipt of requests for the administrative record file pursuant to FOIA, if the requester is in close proximity to the record file, the lead agency may respond to FOIA requests by telling a requester the location and availability of the record file. Decisions regarding disclosures of materials under FOIA should be coordinated among the various lead agency officials with access to such materials.

I. Copying

Section 117(d) of CERCLA requires that each document developed, received, published, or made available to the public under Section 117 be made available for public inspection and copying at or near the site. Under Section 113(k)(2)(B) of CERCLA, these documents must also be included in the administrative record file. Under these provisions of CERCLA, the lead agency must ensure that documents in the record file are available for copying, but does not bear responsibility for copying the documents themselves. Therefore, it is preferable

that the record file should be located in a facility which contains a copying machine (e.g., a public library).

When the administrative record file is available at a facility at or near the site and copying facilities are available there, the lead agency may encourage the requester to make use of the copying facilities at that location. If copying of the record file located at or near the site is difficult for a requesting party, the lead agency may arrange for copying on behalf of a requester at the regional or other central location. The lead agency may ask that requesters arrange for copying by contractors or commercial copy centers who then bill the requester directly.

The lead agency should follow the FOIA regulations at 40 C.F.R. Part 2, in determining the appropriate charge for copying. Copying fees should be waived for other federal agencies, EPA contractors or grantees, and members of Congress. The EPA currently charges \$.20 a page for paper copies as provided in 40 C.F.R. Part 2. Reproduction of photographs, microfilms or magnetic tapes, and computer printouts should be charged at the actual cost to the lead agency.

J. Micrographics

The lead agency may make the administrative record file available to the public in microform.²⁰ Use of micrographics can significantly reduce the space required to store administrative record files. In addition, micrographics can simplify the tasks of reproducing copies of the record file and transmission of the record files to the local repositories. Any use of micrographics should be conducted in an orderly manner consistent with records management procedures. If using micrographics to maintain the record files, the lead agency must provide a micrographic reader at the regional office or other central location to ensure public access to the record file. If a record file is located at or near the site and micrographics are used, the lead agency must ensure that a micrographic reader at that location is available.

Microform copies of original documents are admissable in court if created in an organized fashion. The Business Records as Evidence Act (28 U.S.C. §1732) specifies that copies of records, which are made "in the regular course of business" and copied by any process which accurately reproduces the original, are "as admissible in evidence as the original itself." See also Federal Rules of Evidence 1003. Since the NCP provides for use of microform, microform copies of administrative record documents

²⁶ See 40 C.F.R. §300.805(C).

that are produced in the regular course of business are likely to be admissible in court.

The Office of Information Resources Management (OIRM) has granted approval for the use of micrographics in establishing administrative records (see Appendix J). Any use of micrographics should still comply with the remaining provisions of Chapter 6 of the EPA Records Management Manual (7/13/84).

K. Certification

A certification as to the completeness of the administrative record must be performed when the record is filed in court. Appendix K contains a model court certification.

When EPA is the lead agency such certification should be signed by the Regional Administrator's designee, after consultation with ORC. Any certification of the record should be made by program staff and not legal staff. The region may also choose to have the Administrative Record Coordinator certify that the record was compiled and maintained in accordance with applicable agency regulations and guidance. Such certification would attest that the record was compiled in accordance with current agency procedures and would not address the completeness of the record file.

If a state or other federal agency is the lead agency that agency must certify that the record was compiled and maintained in accordance with applicable EPA regulations and guidance. After the state or federal agency provides this certification, the Regional Administrator's designee should certify as to the completeness of the record, as provided in Appendix K.

III. CONTENTS OF THE ADMINISTRATIVE RECORD

A. Remedial Actions

The administrative record for selection of a remedial action should consist of:

- o documents which were considered or relied on to select the remedial action; and
- o documents which demonstrate the public's opportunity to participate in and comment on the selection of the remedial action."

²⁷ See 40 C.F.R. §§300.810 and 300.815.

OSWER Directive No. 9833.3A-1

Below is a list of documents that are usually generated when a remedial response action is selected. These documents should be included in the administrative record file if they are generated and considered or relied on in selecting the remedial response action. Documents that demonstrate the public's opportunity to participate in and comment on selecting the remedial response action should also be included in the record file. Documents not listed below, but meeting the above criteria, should be included.

Factual Information/Data

- o Preliminary Assessment (PA) report;
- o Site Investigation (SI) report;
- o Remedial Investigation/Feasibility Study (RI/FS) work plan;
- o Amendments to the final work plan;
- Sampling and Analysis Plan (SAP): consisting of a quality assurance project plan (QAPP) and a field sampling plan;
- o Sampling data: verified data during the RI/FS, or any data collected for previous actions such as RCRA or removal actions which are considered or relied on in selecting the remedial action. Unvalidated data should be included only if relied on in the absence of validated data (see note 9 at page 10);
- o Chain of custody forms;
- o Inspection reports;
- o Data summary sheets;
- o Technical studies performed for the site (e.g., a groundwater study);
- o Risk evaluation/endangerment assessment and underlying documentation (see section III.C. at page 29);
- Fact sheet or summary information regarding remedial action alternatives generated if special notice letters are issued to PRPs at an early stage of the RI/FS (see "Interim Guidance on Notice Letters, Negotiations, and Information Exchange," October 19, 1987 - OSWER Directive No. 9834.1);
- RI/FS (as available for public comment and as final, if different); and

o Data submitted by the public, including PRPs.

Policy and Guidance

- Memoranda on site-specific or issue-specific policy decisions. Examples include memoranda on off-site disposal availability, special coordination needs (e.g., dioxin), applicable or relevant and appropriate requirements (ARARs) (to the extent not in the RI/FS), cost effectiveness and utilization of permanent solutions and alternative treatment technologies;
- o Guidance documents (see section III.I. at page 37); and
- o Technical literature (see section III.J. at page 38).

Public Participation (Include the documents that show the public was notified of site activity and had an opportunity to participate in and comment on the selection of response action)

- o Community relations plan;
- Newspaper articles showing general community awareness;
- o Proposed plan;
- Documents sent to persons on the community relations mailing list and associated date when such document was sent;
- Public notices: any public notices concerning response action selection such as notices of availability of information, notices of meetings and notices of opportunities to comment;
- The community relations mailing list (including all known PRPs);²⁵
- o Documentation of informal public meetings: information generated or received during meetings with the public and

²⁸ Individual names and addresses of members of the general public which are on the community relations mailing list should not be included in the public record file. Disclosure of such information may result in a Privacy Act violation (see also section III.H. at page 34) or inhibit the general public from requesting information about the site. The lead agency should then place individual names and addresses in the confidential portion of the record file.

memoranda or notes summarizing significant information submitted during such meetings;

- Public comments: complete text of all written comments submitted (see also section III.D. at page 30);
- Transcripts of formal public meetings: including meetings held during the public comment period on the RI/FS, proposed plan, and any waiver of ARARs under Section 121(d)(4) of CERCLA;
- Responses to significant comments: responses to significant comments received from the public concerning the selection of a remedial action; and
- Responses to comments from the state and other federal agencies.

Enforcement Documents (Include if the document contains information that was considered or relied on in selecting the response selection or shows that the public had an opportunity to participate in and comment on the selection of response action. Do not include enforcement documents solely pertaining to liability)

- o Administrative orders;
- o Consent decrees;
- Affidavits containing relevant factual information not contained elsewhere in the record file;
- o Notice letters to PRPs;
- o- Responses to notice letters;
- Section 104(e) information request letters and Section 122(e) subpoenas; and
- Responses to Section 104(e) information request letters and Section 122(e) subpoenas.

Other Information

- o Index (see section II.D. at page 7);
- Documentation of state involvement: documentation of the request and response on ARARs, Section 121(f)(1)(G) notices and responses, a statement of the state's position on the proposed plan (concurrence, nonconcurrence, or no comment at

the time of publication), opportunity to concur in the selected remedy and be a party to a settlement (see section IV.A. at page 42);

- health assessments, health studies, and public health advisories issued by the Agency for Toxic Substances and Disease Registry (ATSDR) (see section IV.C. at page 45); and
- Natural Resource Trustee notices and responses, findings of fact, final reports and natural resource damage assessments (see section IV.D. at page 45)

Decision Documents

- o Record of decision (ROD): remedial action decision document (including responsiveness summary);
- Explanations of significant differences (under Section 117(c)) and underlying information; and
- o Amended ROD and underlying information.

The administrative record serves as an overview of the history of the site and should be understandable to the reader. Appendix B provides a model file structure for organizing the record file. Appendix C contains a model index.

B. Removal Actions

The administrative record for selection of a removal action should consist of:

- documents which were considered or relied on to select the removal action; and
- o documents which demonstrate the public's opportunity to participate in and comment on the selection of the removal action, when appropriate.

Below is a list of documents that are usually generated when a removal response action is selected. These documents should be included in the administrative record file if they are generated and considered or relied on when selecting the removal action. Documents that demonstrate the public's opportunity to participate in and comment on the removal response action should also be included in the record file. Documents not listed below, but meeting the above criteria, should be included.

²⁹ See 40 C.F.R. §§300.810 and 300.820.

Factual Information/Data

- o Preliminary assessment (PA) report;
- o Site evaluation (SI) report;
- o EE/CA (for a non-time-critical removal action);
- o Sampling plan;
- Sampling data: verified data obtained for the removal action, or any data collected for previous actions such as RCRA or other response actions which are considered or relied on in selecting the removal action. Unvalidated data should be included only if relied on in the absence of validated data (see note 9 at page 10);
- Chain of custody forms;
- o Inspection reports;
- o Technical studies performed for the site (e.g., a ground water study);
- Risk evaluation/endangerment assessment and underlying documentation; and
- o Data submitted by the public, including PRPs.

Policy and Guidance

.

,•

- Memoranda on site-specific or issue-specific policy decisions. Examples include memoranda on off-site disposal availability, compliance with other environmental statutes, special coordination needs (e.g., dioxin);
- o Guidance documents (see section III.I. at page 37); and
- o Technical literature (see section III.J. at page 38).

Public **Participation** (Include the documents that show the public was notified of site activity and had an opportunity to participate in the response selection.)

- o Community relations plan;
- Newspaper articles showing general community awareness;
- Documents sent to persons on the community relations mailing list and associated date when such documents was sent;

- Public notices: any public notices concerning response action selection such as notices of availability of information, notices of meetings, and notices of opportunities to comment;
- The community relations mailing list (including all known PRPs);³⁰
- Documentation of public meetings: information generated or submitted during meetings with the public (including PRPs) and memoranda or notes summarizing significant information submitted during such meetings;
- Public comments: complete text of all written comments submitted (see section III.D. at page 30);
- Responses to significant comments: responses to significant comments received from the public concerning the selection of a removal action; and
- Responses to comments from states and other federal agencies.

Enforcement Documents (Include if the document contains information that was considered or relied on in selecting the response selection or shows that the public had an opportunity to participate in and comment on the selection of response action. Do not include enforcement documents solely pertaining to liability)

- o Administrative orders;
- o Consent decrees;
- Affidavits containing relevant factual information not contained elsewhere in the record file;
- o Notice letters to PRPs;

³⁰ Individual names and addresses of members of the general public which are on the community relations mailing list should not be included in the public record file. Disclosure of such information may result in a Privacy Act violation (see also section III.H. at page 34) or inhibit the general public from requesting information about the site. The lead agency should then place individual names and addresses in the confidential portion of the record file.

- o Responses to notice letters;
- - Responses to Section 104(e) information request letters and Section 122(e) subpoenas.

Other Information

- o Index (see section II.D. at page 7);
- o Documentation of state involvement (see section IV.A. at page 42);
- ATSDR health assessments, health studies, and public health advisories (see section IV.C. at page 45); and
- Natural Resource Trustee notices and responses, findings of fact, final reports and natural resource damage assessments (see IV.D. at page 45).

Decision Documents

- o EE/CA Approval Memorandum;
- o Action Memorandum;
- Amended Action Memorandum; and
- Other documents which embody the decision for selection of a removal action.

The administrative record serves as an overview of the history of the site and should be understandable to the reader. Appendix B provides a model file structure for organizing the record file. Appendix C contains a model index.

C. Imminent and Substantial Endangerment

Under Section 106 of CERCLA, the EPA may find the existence of an imminent and substantial endangerment to the public health or welfare or the environment because of an actual or threatened release of a hazardous substance.

Determining the existence of an imminent and substantial endangerment is an important component in selecting the response action. Therefore, all documents considered or relied on in making that determination, including any risk assessment, and its supporting documentation, must be included in the administrative record file.³¹ If there is proper documentation of the determination of an imminent and substantial endangerment in the record file, judicial review of that determination in an action under Section 106 of CERCLA should be limited to the administrative record.

D. Public Comments

The administrative record file should document the public's opportunity to be involved in selecting a response action. This can be accomplished by including in the record file all documents related to the opportunity to participate (e.g., notices and fact sheets), and relevant written comments and information submitted by the public (e.g., reports and data).

Public requests for information (e.g., Freedom of Information Act (FOIA) requests for copies of reports), need not be included in the record file.

The lead agency should request that substantive oral comments (either in person or over the phone) be put in writing by the commenter and submitted to the record file. The commenter should be advised that the obligation to reduce the comment to writing rests with the commenter. The lead agency, however, may reduce it to writing where the lead agency will want to rely on the comment.

The lead agency may respond to comments received prior to a public comment period in various ways, depending on the nature and relevance of a particular comment. The lead agency's consideration of such a comment may be in the form of a written response, or reflected by documented actions taken after receiving the comment, or even by changes in subsequent versions of documents. If the lead agency prepares a written response to a comment, the comment and response should be included in the record file.

The lead agency may notify commenters that comments submitted prior to a formal public comment period must be resubmitted or specifically identified during the public comment period in order to receive formal response by the lead agency. Alternatively, the lead agency may notify a commenter that the lead agency will respond to the comment in a responsiveness summary prepared at a later date. The lead agency, however, has

³¹ See "Guidance on Preparing Superfund Decision Documents: The Proposed Plan, The Record of Decision, Explanation of Significant Differences, ROD Amendment," OSWER Directive No. 9355.3-02, June 1989.

no duty to respond to any comments received before the formal public comment period, or to respond to comments during the public comment period until the close of the public comment period.

The lead agency, however, is encouraged to consider, respond to and include in the record file significant comments that were submitted before the public comment period. Considering early comments provides practical benefits both substantively and procedurally. Early comments may provide important information for the selection decision, and early consideration provides the public (and, particularly, PRP's) with additional informal opportunities for participating in the decision making process.³²

All comments received by the lead agency during the formal public comment period are to be included in the record file in their original form, or if not feasible, an explanation should be placed in the record file explaining why such comments were not included. Comments received during the formal public comment period must be addressed in the responsiveness summary (included with the ROD in remedial response actions). The responses may be combined by subject or other category in the record file.

Comments which are received after the formal comment period closes and before the decision document is signed should be included in the record file but labeled "late comment." Such comments should be handled as post-decision information (see section III.N. at page 40).

Comments received after the decision document is signed should be placed in a post-decision document file. They may be added to the record file in limited circumstances (see section III.N. at page 40).

E. Enforcement Actions

The same procedures should be used for establishing an administrative record whether or not a response action is selected in the context of an enforcement action. The following additional information, however, may assist the lead agency where there is enforcement activity.

E.1. Negotiation Documents

During negotiations with the lead agency, a potentially responsible party (PRP) may produce documents and claim that they

 $^{^{32}}$ See 40 C.F.R. §§300.815(b), 300.825(a)(2) and (b)(2).

constitute confidential business information (CBI) or offers of settlement subject to Rule 408 of the Federal Rules of Evidence.

Generally, those documents are not part of the administrative record for response selection unless they are submitted by PRPs for consideration in selecting a response action and are considered or relied on in selecting the response action. A privileged document which was considered or relied on in selecting the response action should be placed in the confidential portion of the record file. Such a document should be summarized and the summary included in the publicly accessible portion of the record file (see section II.H. at page 19). If the information cannot be summarized in a disclosable manner, the information should be placed in the confidential portion of the record file only and listed in the index to the file.

E.2. PRP-Lead RI/FS

Where a PRP is conducting the RI/FS, the PRP must submit all technical information on selection of the remedial action generated during the RI/FS to the lead agency. Technical information includes work plans, sampling data, reports, and memoranda. The lead agency, and not the PRP, will establish and maintain the administrative record file (see "Interim Guidance on Potentially Responsible Party Participation in Remedial Investigations and Feasibility Studies," May 16, 1988, OSWER Directive No. 9835.1a and "Model Administrative Order on Consent for Remedial Investigation and Feasibility Study," January 30, 1990, OSWER Directive No. 9835.10.)

PRPs may be delegated responsibility for some record file maintenance activities, such as housing the files at or near the site. PRPs cannot, however, be responsible for decisions on what documents comprise the record file, because of, among other things, the potential for a conflict of interest.

E.3. Administrative Orders and Consent Decrees

Final administrative orders and consent decrees issued prior to selection of the response action (e.g., ordering a PRP to conduct the RI/FS), should be included in the administrative record file. Administrative orders or consent decrees issued after the signing of the ROD or the action memorandum should not be included in the record file, unless the consent decree or administrative order meets the criteria for the inclusion of post-decision documents in the record file (see section III.N. at page 40). Drafts of administrative orders and consent decrees should not be included in the record file, unless the drafts contain factual information that was considered or relied on and is not found elsewhere in the record file.

The issues relating to administrative records for administrative orders and de minimis settlements are not addressed by this guidance.

F. Excluded Documents

Certain documents should not be included in the administrative record file because they are irrelevant to the selection of the response action. Documents should be excluded from the record file if they were not considered or relied on in selecting the response action.

Material beyond the scope of the record file should be kept in separate files maintained at the regional office or other central location. These files need not be made publicly available, although many of the documents in the files may be available to the public if requested under FOIA.

Examples of documents that are irrelevant to the decision on selecting a response action may include Hazard Ranking System (HRS) scoring packages, contractor work assignments, cost documentation (as opposed to cost effectiveness information), and National Priorities List (NPL) deletion information. If, however, these documents contain information that is considered or relied on in the response action selection and is not contained elsewhere in the record file, then the documents should be included in the record file.

Information regarding PRP liability is generally not included in the record file for selection of the response action except to the extent such information (typically substance specific) is considered or relied on in selecting the response action. Documents relating to PRP liability, however, should be compiled and maintained in the regional office or other central location so that they are available at the time of notice to PRPs or referral of any litigation.

G. Draft Documents and Internal Memoranda

In general, only final documents should be included in the administrative record file. The record file should not include preliminary documents such as drafts and internal memoranda. Such documents are excluded from the record file because drafts and internal memoranda are often revised or superseded by subsequent drafts and memoranda prior to the selection of the response action. The preliminary documents are, therefore, not considered or relied on in making the response action decision.

Drafts (or portions of them) and internal memoranda should be included, however, in three instances. First, if a draft
document or internal memorandum is the basis for a response decision the draft document or internal memorandum should be placed in the record file. This may occur if the draft contains factual information which was relied on but is not included in a final document, a final document does not exist, or a final document did not exist when the response decision was made.

Second, if a draft document or internal memorandum is circulated by the lead agency to other persons (e.g., the support agency, PRPs or the general public) who then submit comments which the decisionmaker considers or relies on when making a response action decision, relevant portions of the draft document or the memorandum and comments on that document should be included in the record file.

Third, if a draft document or internal memorandum explains or conveys decisions on the procedures for selecting the remedy or the substantive aspects of a proposed or selected remedy (e.g., the scope of a site investigation or the identification of potential ARARs), the document should be placed in the record file, even though the document was signed by a person other than the Regional Administrator and generated long before the decision document was signed.

Examples of internal memoranda and staff notes which should not be included in the record file are documents that express tentative opinions or internal documents that evaluate alternative viewpoints. Recommendations of staff to other staff or management should also not be included in the record file, except for those staff recommendations which ultimately embody a final decision relevant to response selection. Drafts and internal memoranda may also be subject to claims of privilege (see section III.H., below).

H. Privileged Documents

Some documents in the administrative record file may be protected from public disclosure on the basis of an applicable privilege.³³ Any documents which are considered or relied on in a response action selection, but withheld from the public portion of the record file based on privilege, must be placed in a confidential portion of the record file (see section II.H. at page 19).

If a document is excluded from the public portion of the record file based on privilege, the relevant information should, to the extent feasible, be extracted and included in the public

³³ See 40 C.F.R. \$300.810(c).



record file. This can often be accomplished by deleting or .redacting the privileged information from the document.

The privileges discussed below may be asserted with respect to documents that are considered or relied on in the selection of a response action. The head of the office responsible for developing the document in question should assert the privilege. In all cases, the official asserting a privilege should consult with ORC.

Public disclosure of a privileged document may result in waiver of the privilege, although the nature and extent of the waiver will depend on the privilege asserted and the circumstances of the disclosure. If the privilege is waived and the document becomes a public document, it must be disclosed to any requester. In light of the potential for waiver, it is important that personnel not release potentially privileged documents to any party without consulting with ORC.

Deliberative Process

The deliberative process privilege applies to predecisional, deliberative communications that express opinions, advice, and recommendations of staff to other staff or management. The privilege functions to encourage the honest and free expression of opinion, suggestions and ideas among those formulating policy for government agencies (see "Guidance for Assertion of Deliberative Process Privilege," 10/3/84).

In general, if a document contains factual information forming the basis for the selection of the response action, the factual portion should be included in the record file.

Use of the deliberative process privilege should be balanced with the statutory mandate of including the public in the response action selection process. The privilege should be asserted if disclosure of the document will have an inhibiting effect on frank and open discussion among government staff and decisionmakers. Documents should not be withheld solely because they would reveal flaws in the case or information embarrassing to the government. Specific procedures exist for assertion of the deliberative process privilege, which include consulting with ORC.

Confidential Business Information (CBI)

The EPA must withhold from the public record trade secrets and commercial and financial information that is subject to protection under 40 C.F.R. Part 2. However, Section 104(e)(7) of CERCLA greatly restricts the assertions of confidentiality claims

OSWER Directive No. 9833.3A-1

by PRPs at CERCLA sites. The decisionmaker should attempt to avoid using CBI in making response action decisions and can do so in most cases by using other information instead.³⁴ Where the decisionmaker must use CBI in making its decision, 40 C.F.R. Part 2 and Section 104(e)(7) of CERCLA will apply and such information should be placed in the confidential portion of the administrative record file.

Attorney Work Product

This exclusion applies to documents prepared in anticipation of possible litigation. The work product privilege covers all documents prepared by an attorney or under an attorney's supervision, including reports prepared by a consultant or program employee. Litigation need not have commenced but it must be reasonably contemplated. These documents generally relate t enforcement or defensibility of a decision and are not conside: or relied on in selecting a response action. These documents should not, therefore, be in the administrative record file.

Attorney-Client Communication

The attorney-client privilege applies to confidential communications made in connection with securing or rendering legal advice. The privilege is limited to communications where there was an intention to keep the information confidential.

Personal Privacy

This exemption covers information about individuals in personnel, medical, and similar files, the disclosure of which would constitute a clearly unwarranted invasion of personal privacy. The records must pertain to an individual, and not a business, to be excluded from the public portion of the administrative record file under this exemption. Often, information subject to the protection under the personal privacy privilege can be redacted from the document and the redacted version can be placed in the public portion of the record file.

State Secrets

The lead agency is authorized to exclude from public scrutiny information which, if released, would harm national security or interfere with the government's ability to conduct foreign relations. This privilege could be particularly important where the PRP is a federal agency or a contractor for a federal agency. In the case of a federal facility cleanup, an

³⁴ See 40 C.F.R. §300.810(d).

Inter-Agency Agreement should spell out procedures for asserting this privilege.

Confidential Informant

Statements obtained from witnesses who have been granted confidentiality may be privileged.

Information Exempted by Other Statutes

Information specifically exempted from disclosure by a federal statute need not be part of the public record. The statute in question must leave no discretion as to the requirement that matters be withheld from the public, or it must establish particular criteria for withholding or refer to particular types of matters to be withheld.

I. Guidance Documents

Guidance documents, or portions of guidance documents, that are considered or relied on in selecting a response action should be included in the administrative record file for that response action. Any guidance documents generated to address issues that specifically arise at the site for which the record file is being compiled should be physically included in the record file. Certain guidance documents, however, do not have to be kept in the record file. Guidance documents not generated for the particular site for which the record is being compiled may be kept in a compendium of guidance documents maintained at the regional office or other central location.³⁵

Each Region should maintain a compendium of guidance documents which are frequently used in selecting response actions. As with an administrative record file, the compendium of guidance documents must be available to the public, but only at the regional office or other central location. The record file located at or near the site should contain an index to the compendium of guidance documents. The Administrative Record Coordinator should maintain and update the compendium of guidance documents. If a guidance document maintained in the compendium is considered or relied on when making a response action decision, the index to the record file must list the document and indicate its location and availability. See also Appendix E.

If a guidance document is listed in a bibliography to a document included in the record file (e.g., listed in the bibliography to the RI/FS), it need not be listed again in the

³⁵ See 40 C.F.R. §300.805(a)(2).

index to the record file. In this case, however, the index must state that documents listed as bibliographic sources might not be listed separately in the index.

If a guidance document which is not included in the guidance compendium is considered or relied on in selecting the response action, the document should be physically included in the record file.

J. Technical Literature

Technical literature generated for the site at issue should be physically included in the administrative record file for that site, whether or not it is publicly available.

Similarly, technical literature not specifically generated for the site which is not publicly available should also be included in the site-specific record file. Such documents include technical journals and unpublished documents that are not available through the Library of Congress or not circulated to technical libraries.

Publicly available technical literature not generated for the site, however, need not be located at or near the site or at the regional office or other central location if the documents are referenced in the index to the record file.⁵⁶ These documents do not have to be physically included in the record file, unless requested, because they are already available to the public. Copying such documents creates a significant burden to the lead agency and copyright laws may pose additional barriers to such copying. Examples of publicly available technical literature include engineering manuals, groundwater monitoring or hydrogeology textbooks, ATSDR toxicological profiles, and articles from technical journals.

If technical literature is listed in a bibliography to a document included in the record file (e.g., listed in the bibliography to the RI/FS), it need not be listed again in the index to the record file. In this case, however, the index must state that documents listed as bibliographic sources might not be listed separately in the index.

Computer models and technical databases need not be physically included in the record file but should be referenced in the index to the record file and made available upon request. Printouts or other documents produced from the models and databases should be physically included in the record file if

³⁶ See 40 C.F.R. §300.805(b)(3).

such documents contain information which was considered or relied on in selecting the response action.

K. Legal Sources

Copies of statutes and regulations cited in documents included in the record file need not be included in the record file 1f they are readily available to the public. For example, the NCP and other regulations are easily accessible since they are published in the Federal Register and the Code of Federal Regulations (C.F.R.).

Copies of the actual standards (statutes or regulations) comprising federal and state ARARs should be physically included in the record file if they are not easily accessible. Also, other federal and state criteria, advisories, and guidance documents pertinent to the site (e.g., what the EPA refers to as "TBCs," or standards "to be considered"), may not be easily accessible. If such documents are cited in an RI/FS, appendix to the RI/FS, EE/CA, or ROD, those advisories which are not readily available should be included in the record file.

L. NPL Rulemaking Docket Information

Generally, information included in the National Priorities List (NPL) rulemaking docket, such as the Hazard Ranking System (HRS) scoring package and comments received on the listing, need not be included in the record file for selection of a response action. The NPL docket contains information relevant to the decision to list a site, which may be irrelevant to the decision on response action selection.

Documents in the NPL docket which contain sampling data or other factual information which was considered or relied on in selecting a response action should be included in the record file if the information is not available already in the record file. Such information may include early sampling data taken by parties other than the lead agency or its contractors (e.g., a State).

M. RCRA Documents

If an action is taken under CERCLA at a site with a history of Resource Conservation and Recovery Act (RCRA) activity, much of the information relating to those RCRA activities may be considered or relied on in making the CERCLA response action selection. Any relevant RCRA information, particularly information on waste management and RCRA corrective action at the site, should be included in the administrative record file (e.g., RCRA permit applications, inspection reports, RCRA Facility Assessment (RFA), RCRA Facility Investigation (RFI), Corrective

Measures Studies (CMS), or responses to RCRA information _ requests).

Not all pre-existing RCRA information will be considered or relied on in selecting a CERCLA response action, but information on types of wastes, quantity of wastes, and observations of potential threats gathered during RCRA investigations generally will be considered and thus should be included in the record file.

N. Post-Decision Information

In all cases, documents generated or received after signing the decision document should be kept in a post-decision document file. This file is not part of the administrative record file and should be maintained only at the regional office or other central location.

In general, post-decision documents should not be added to the administrative record file. Since the record file contains the information which was considered or relied on in selecting the response action, documents generated or received after selecting the response action are not relevant to that response decision and should not be included in the record file. Such documents may, however, be relevant to later response selection decisions and, if so, should be included in the record file pursuant to Section 300.825 of the NCP.

Documents kept in the post-decision document file may be added to the record file in the situations described below:

- Where a decision document does not address or reserves a portion of the decision to be made at a later date.³⁷ For example, a decision document that does not resolve the type of treatment technology. In such cases, the lead agency should continue to add documents to the record file which form the basis for the unaddressed or reserved portion of the decision;
- Where there is a significant change in the selected response action. Changes that result in a significant difference to a basic feature of the selected remedial action (e.g., timing, ARARs), with respect to scope, performance, or cost

³⁷ 40 C.F.R. §300.825(a)(1).

³⁸ 40 C.F.R. §300.825(a)(2). See 40 C.F.R. §300.435(c)(2)(1).

may be addressed in an explanation of significant differences. Section 117(c) of CERCLA states:

[a]fter adoption of a final remedial action plan -(1) if any remedial action is taken, (2) if any enforcement action under section 106 is taken, or (3) if any settlement or consent decree under section 106 or section 122 is entered into, and if such action, settlement, or decree differs in any significant respects from the final plan, the President or the State shall publish an explanation of the significant differences and the reasons such changes were made.

The record file should include the explanation of significant differences, underlying documentation for the response action changes, any significant comments from the public, and the lead agency responses to any significant comments. A formal public comment period is not required for an explanation of significant differences;

o Where the changes are so significant that they fundamentally alter the very nature or basis of the overall response action. Such changes will require an amended decision document.³⁹ The Region will decide whether a change to a response action is considered a significant or a fundamental change for purposes of addressing the change (see Chapter 8 of "Interim Final Guidance on Preparing Superfund Decision Documents: The Proposed Plan and Record of Decision," June 1989, OSWER Directive No. 9355.3-02).

When the decision document is amended, the amended decision document, the underlying documentation, any significant comments from the public, and the lead agency's responses to any significant comments, should be included in the record file. ROD amendments will require a formal public comment period;⁴⁰

 Where comments containing significant information are submitted by interested persons after the close of the public comment period. The lead agency must consider such comments only to the extent that the comments contain significant information not contained elsewhere in the record file which could not have been submitted during the public comment period and which substantially support the

⁴⁰ 40 C.F.R. §300.435(c)(2)(ii).



 $^{^{39}}$ 40 C.F.R. §300.825(a)(2).

need to significantly alter the response action.⁴¹ Documents meeting this test should be included in the record file, along with the lead agency's responses to the significant comments, whether or not such information results in a change to the selected decision. In this case, the comments and the lead agency responses to such comments, including any supporting documents, should be included in the record file; and

- Where the lead agency holds public comment periods after the selection of the response action.⁴² The lead agency may hold additional public comment periods or extend the time for submission of public comment on any issue concerning response selection. Such comment should be limited to the issues for which the lead agency requested additional comment. All comments responsive to the request submitted during such comment periods, along with any public notices of the comment period, transcripts of public meetings, and lead agency responses to the comments, should be placed in the record file.
- IV. INVOLVEMENT OF OTHER PARTIES
- A. States
- A.1. State Involvement in Federal-Lead Sites

The administrative record for a federal-lead site must reflect the state's opportunity to be involved in selecting the response action. The record for a remedial action should include documents that reflect at least the following state participation or the opportunity for state participation:⁴³

- Letter to state requesting identification of ARARs and the final response from state identifying ARARs (and certification from the state);
- Comments, or the opportunity to comment, on a proposed finding or decision to select a response action not attaining a level or standard of control at least equivalent to a state ARAR;
 - ⁴¹ 40 C.F.R. §300.825(c).

⁴³ See also Section 121(f) of CERCLA

⁴² 40 C.F.R. §300.825(b).

- Comments, or the opportunity to comment, on the final draft RI/FS, the proposed plan and EPA responses to the comments;
- Significant post-decision comments by the state and EPA responses to the comments (place in the post-decision document file for possible inclusion in the record file see section III.N. at page 40).

The administrative record for a removal action should reflect any state participation, especially any state comments and EPA responses to the comments.

The record file should only include final state comments, unless the comments explain or convey decisions on substantive aspects of a proposed or selected remedy (e.g., the scope of a proposed action or the identification of potential ARARS). Any preliminary deliberations between the state and EPA relevant to the response selection need not be part of the record file if superseded by documentation of the state's final position.

The governing body of an Indian tribe should be afforded the same treatment as a state in accordance with Section 126 of CERCLA.

A.2. Federal Involvement in State-Lead Sites

Where a state has been officially designated the lead agency for a CERCLA site, the state must compile and maintain the administrative record for that site in accordance with Section 113(k) of CERCLA and Section 300.800 of the NCP. Since EPA has ultimate responsibility for both the selection of a response action (e.g., EPA signs the ROD) and the record on which that response action is based, EPA must participate in compiling and maintaining the record. In such cases, EPA must assure that the record file forms a complete basis for the selection of the response action.

The state as lead agency must maintain the record file at a state office (e.g., the state's central environmental agency office) and at or near the site. At a minimum, the state as lead agency also must transmit a copy of the index, the RI/FS work plan, the RI/FS released for public comment, the proposed plan, and any public comments received on the RI/FS and the proposed plan to the appropriate EPA Regional office. These documents should be transmitted to EPA as they are generated or received. Transmittal of the index will not suffice. In addition, other documents may be requested by EPA on a case-by-case basis.

⁴⁴ See 40 C.F.R. §300.800(c).

The Superfund Memorandum of Agreement (SMOA), or Cooperative Agreement (CA), must address the administrative record requirements. The following language should be included in the SMOA or CA where the state has been officially designated the lead agency for a CERCLA site:

The state must compile and maintain the administrative record upon which the selection of the [remedial, removal] action is based. The compilation and maintenance of the record must follow 40 C.F.R. Part 300, Subpart I and EPA guidance on the administrative record. The administrative record must be located at the state [environmental agency] office, and at or near the site. In addition, the state must submit copies of the index. the RI/FS workplan, the RI/FS released for public comment, the proposed plan, and any public comments received on the RI/FS and proposed plan to the EPA Regional office, as they are added to the administrative record file. In addition, the state must submit other documents that are requested by EPA. The state shall comply with Section 113 of CERCLA and any applicable regulations. EPA may require the retention of other documents for cost recovery purposes.

The record file compiled by the state should reflect EPA's participation, comments, concurrence, and disagreements at the same stages as are required for state involvement in a federallead site. The state must place in the record file any documents submitted by EPA for inclusion in the record file.

B. Federal Facilities

Federal agencies have the responsibility, pursuant to Executive Order 12580, to establish the administrative record for federal facilities under their jurisdiction, custody, or control where using CERCLA authority for a response action. The record file for a federal facility must include all documents considered or relied on in selecting a response action, including documents submitted by EPA on the selection of the response action. The federal agency must comply with all NCP (see Appendix M) and CERCLA requirements in compiling and maintaining the record, including the minimum public participation requirements in Sections 113 and 117 of CERCLA.⁴⁵

⁴⁵ See 40 C.F.R. §300.800(b).

The federal agency must maintain the record file at or near the site and ensure easy public access to the record file. If, for example, a site is a Department of Defense facility, the record file should be housed in a location which does not require military clearance for access. The federal agency should keep a complete copy of the record file at a location within the federal agency office comparable to an EPA Regional office.

At NPL sites and any other site where EPA is involved in selecting a response action at a federal facility, EPA must participate in compiling and maintaining the record. In such cases, EPA must assure that the record file forms a complete basis for the selection of the response action. At a minimum, the federal agency must transmit a copy of the index, the RI/FS workplan, the RI/FS released for public comment, the proposed plan, and any public comments received on the RI/FS and proposed plan to the appropriate EPA Regional office. These documents should be transmitted to EPA as they are generated. Transmittal of the index will not suffice. In addition, other documents may be requested by EPA on a case-by-case basis. Inter-Agency Agreements (IAGs) should spell out procedures for compiling and maintaining the record.

C. ATSDR

Participation in the selection of a response action by the Agency for Toxic Substance and Disease Registry (ATSDR) should be reflected in the administrative record. The record file must include the initial and subsequent health assessments and any other information EPA solicits and obtains from ATSDR which EPA considers or relies on in its selection of a response action.

Draft versions of the health assessment and other draft documents upon which ATSDR comments should not be included in the record file. If, however, EPA solicits comments from ATSDR on a draft document such as a draft work plan or RI report, and receives formal comments from ATSDR which EPA considers or relies on in selecting a response action, then the document and comments should be included in the record file.

In the event that the ATSDR health assessment and EPA's risk assessment appear inconsistent, a document explaining the difference should be generated and placed in the record file.

D. Natural Resources Trustees

Section 122(j)(1) of CERCLA requires that the EPA give notice to the Natural Resources Trustee of a release or threatened release of any hazardous substance which may have resulted in damages to natural resources. The administrative record file must include the notice to the Natural Resources Trustee, and any subsequent final communications (e.g., a release or final report). In addition, any factual information provided by the Natural Resources Trustee which is considered or relied on in selecting a response action should be included in the record file.

In the event that the Natural Resources Trustee's damage assessment and EPA's risk assessment appear inconsistent, a document explaining the difference should be generated and placed in the record file.

V. DISCLAIMER

The policies and procedures established in this document are intended solely for the guidance of employees of the U.S. Environmental Protection Agency. They are not intended and cannot be relied upon to create any rights, substantive or procedural, enforceable by any party in litigation with the United States. EPA reserves the right to act at variance with these policies and procedures and to change them at any time without public notice.

VI. FURTHER INFORMATION

For further information concerning this memorandum, please contact Gary Worthman in the Office of Waste Programs Enforcement at FTS (202) 382-5646.

GLOSSARY

Administrative Record: as used in this guidance, the body of documents that were considered or relied on which form the basis for the selection of a response action.

<u>Administrative Record File</u>: as used in this guidance, the ongoing collection of documents which are anticipated to constitute the administrative record when the selection of response action is made.

<u>ARAR</u>: applicable or relevant and appropriate requirements (see Section 121(d) of CERCLA).

ATSDR: Agency for Toxic Substance and Disease Registry.

<u>CA</u>: cooperative agreement (entered into with a state or local government to transfer funds to conduct response activities).

CBI: confidential business information.

<u>CERCLA</u>: Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended by the Superfund Amendments and Reauthorization Act of 1986 (also known as Superfund).

<u>C.F.R.</u>: Code of Federal Regulations.

<u>CMS</u>: corrective measure study (RCRA corrective action document, equivalent to an FS).

CRC: Community Relations Coordinator.

CRP: community relations plan.

<u>Document</u>: as used in this guidance, includes writings, drawings, graphs, charts, photographs, and data compilation from which information can be obtained. It does not, however, include physical samples.

DOJ: Department of Justice.

EE/CA: engineering evaluation/cost analysis (removal document).

EPA: United States Environmental Protection Agency.

ESD: Environmental Services Division.

Explanation of Significant Differences: post-ROD document described in Section 117(c) of CERCLA.

FOIA: Freedom of Information Act.

FSP: field sampling plan.

HRS: Hazard Ranking System.

IAG: inter-agency agreement (made with a federal agency).

Lead Agency: the agency that provides the OSC or RPM to plan and implement a response action under the NCP.

NCP: National Oil and Hazardous Substances Pollution Contingency Plan, as revised on March 8, 1990 (55 FR 8859).

NPL: National Priorities List.

OE: EPA Office of Enforcement.

OERR: EPA Office of Emergency and Remedial Response.

OIRM: EPA Office of Information Resources Management.

<u>Operable Unit</u>: a discrete action that comprises an incremental step toward comprehensively addressing site problems (see section 300.5 of the NCP).

ORC: EPA Office of Regional Counsel.

OSC: On-Scene Coordinator (project manager for a removal action)

OSWER: EPA Office of Solid Waste and Emergency Response.

OWPE: EPA Office of Waste Programs Enforcement.

PA: preliminary assessment.

PRP: potentially responsible party.

OAPP: quality assurance project plan.

RA: remedial action.

<u>RCRA:</u> the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act.

RD: remedial design.

<u>RI/FS</u>: remedial investigation/feasibility study.

<u>RFA:</u> RCRA facility assessment (RCRA document, equivalent to a PA/SI).

<u>RFI</u>: RCRA facility investigation (RCRA corrective action document, equivalent to an RI).

<u>ROD</u>: Record of Decision (documents the selection of a remedial action).

<u>RPM</u>: remedial project manager (project manager for a remedial action).

SAP: sampling and analysis plan.

SARA: Superfund Amendments and Reauthorization Act of 1986 (see CERCLA above).

Site File: the file containing all site documentation.

SI: site investigation.

×.

SMOA: Superfund memorandum of agreement (made with a state).

<u>Support Agency</u>: the agency that provides the support agency coordinator to furnish necessary data to the lead agency, review response data and documents, and provide other assistance as requested by the lead agency. The support agency may also concur on decision documents.

:

J. JUDICIAL REVIEW -

(1) LIMITATION —In any judicial action under this Act. judicial review of any issues concerning the adequacy of any response action taken or ordered by the President shall be imited to the administrative record. Otherwise applicable principles of administrative law shall govern whether any supplemental materials may be considered by the court.

(2) STANDARD.—In considering objections raised in any judicial action under this Act, the court shall uphold the President's decision in selecting the response action unless the objecting party can demonstrate, on the administrative record, that the decision was arbitrary and capticious or otherwise not in accordance with law.

(3) REMERY.—If the court finds that the selection of the response action was arbitrary and capricious or otherwise not in accordance with law, the court shall award (A) only the response costs or damages that are not inconsistent with the national contingency plan, and (B) such other relief as is consistent with the National Contingency Plan.

(4) PROCEDURAL ERRORS.—In reviewing alleged procedural errors, the court may disailow costs or damages only if the errors were so serious and related to matters of such central relevance to the action that the action would have been significantly changed had such errors not been made.

SECTION 113 (R) OF CERCLA

K. ADMINISTRATIVE RECORD AND PARTICIPATION PROCEDURES.

(1) ADMINISTRATIVE RECORD — The President shall establish an administrative record upon which the President shall base the selection of a response action. The administrative record shall be available to the public at or near the facility at issue. The President also may place duplicates of the administrative record at any other location.

2) PARTICIPATION PROCEDURES -

(A) REMOVAL ACTION.—The President shall promulgate regulations in accordance with chapter 5 of title 5 of the United States Code establishing procedures for the appropriate participation of interested persons in the development of the administrative record on which the President will base the selection of removal actions and on which judicial review of removal actions will be based.

(B) REMEDIAL ACTION. — The President shall provide for the participation of interested persons, including potentially responsible parties, in the development of the administrative record on which the President will base the selection of remedial actions and on which judicial review of remedial actions will be based. The procedures developed under this subparagraph shall include, at a minimum, each of the following:

(i) Notice to potentially affected persons and the public, which shall be accompanied by a brief analysis of the plan and alternative plans that were considered.

(ii) A reasonable opportunity to comment and provide information regarding the plan.

(iii) An opportunity for a public meeting in the affected area. in accordance with section 117(a)(2) (relating to public participation).

(iv) A response to each of the significant comments, criticisms, and new data submitted in written or oral presentations.

(v) A statement of the basis and purpose of the selected action.

For purposes of this subparagraph, the administrative record shall include all items developed and received under this subparagraph and all items described in the second sentence of section 117(d). The President shall promulgate regulations in accordance with chapter 5 of title 5 of the United States Code to carry out the requirements of this subparagraph.

(C) INTERNA ESCORD.—Until such regulations under subparagraphs (A) and (B) are promulgated, the administrative record shall consist of all items developed and received pursuant to current procedures for selection of the response action, including procedures for the participation of interested parties and the public. The development of an administrative record and the selection of response action under this Act shall not include an adjudicatory hearing.

(D) POTENTIALLY ESSPONSIBLE PARTIE.—The President shall make reasonable efforts to identify and notify potentially responsible parties as early as possible before selection of a response action. Nothing in this paragraph shall be construed to be a defense to liability.

APPENDIX 8

MODEL FILE STRUCTURE

This model file structure may be used to compile an administrative record file for a remedial action, a removal action, or a combination of both remedial and removal actions. If the record documents a remedial action decision, section 2 of the file will contain only those removal action documents which (a) predate the remedial record of decision and (b) are relevant to the selection of the remedial action. If the record documents a removal action decision, sections 3, 4, and 5 of the file will contain only those remedial action documents which (a) predate the removal action memorandum and (b) are relevant to the selection of the removal action.

Justification is unnecessary for file categories without any documents. Those categories should be left out of the index.

A document should be filed in only one category, even if it falls into more than one category. It may be referenced in another category. If necessary, additional subcategories may be developed to accommodate documents not falling in any of the defined subcategories. Avoid adding categories of miscellaneous documents.

The correspondence subcategory can include comments and responses specific to the category. If the comments and responses are general in nature or address more than one category, they may be included in the public participation category.

INDER (FIRST DOCUMENT)

1.0 SITE IDENTIFICATION

- 1.1 Background RCRA and other information
- 1.2 Notification/Site Inspection Reports
- 1.3 Preliminary Assessment (PA) Report
- 1.4 Site Investigation (SI) Report
- 1.5 Previous Operable Unit Information

2.0 REMOVAL RESPONSE

- 2.1 Sampling and Analysis Plans
- 2.2 Sampling and Analysis Data/Chain of Custody Forms
- 2.3 EE/CA Approval Memorandum (for non-time-critical ren
- 2.4 EE/CA
- 2.5 Action Memorandum
- 2.6 Amendments to Action Memorandum

3.0 REMEDIAL INVESTIGATION (RI)

- 3.1 Sampling and Analysis Plan (SAP)
- 3.2 Sampling and Analysis Data/Chain of Custody Forms
- 3.3 Work Plan
- 3.4 RI Reports

4.0 FEASIBILITY STUDY (PS)

- 4.1 ARAR Determinations
- 4.2 FS Reports
- 4.3 Proposed Plan
- 4.4 Supplements and Revisions to the Proposed Plan

5.0 RECORD OF DECISION (ROD)

- 5.1 ROD
- 5.2 Amendments to ROD
- 5.3 Explanations of Significant Differences

6.0 STATE COORDINATION

6.1 Cooperative Agreements/SNOAs

7.0 ENFORCEMENT

- 7.1 Enforcement History
- 7.2 Endangerment Assessments
- 7.3 Administrative Orders
- 7.4 Consent Decrees
- 7.5 Affidavits
- 7.6 Documentation of Technical Discussions with PRPs on Response Actions
- 7.7 Notice Letters and Responses

8.0 HEALTH ASSESSMENTS

- 8.1 ATSDR Health Assessments
- 8.2 Toxicological Profiles

9.0 NATURAL RESOURCE TRUSTEES

- 9.1 Notices Issued
- 9.2 Findings of Fact
- 9.3 Reports

10.0 PUBLIC PARTICIPATION

- 10.1 Comments and Responses
- 10.2 Community Relations Plan
- 10.3 Public Notice(s) (Availability of the Administrative Record File, Availability the Proposed Plan, Public Meetings)
- 10.4 Public Meeting Transcripts
- 10.5 Documentation of Other Public Meetings
- 10.6 Fact Sheets and Press Releases
- 10.7 Responsiveness Summary
- 10.8 Late Comments

11.0 TECHNICAL SOURCES AND GUIDANCE DOCUMENTS

- 11.1 EPA Headquarters Guidance
- 11.2 EPA Regional Guidance
- 11.3 State Guidance

11.4 Tecnnical Sources

.

APPENDIX C

MODEL INDEX

Attached is an excerpt of the Index of documents included in the Administrative Record for the Love Canal site. The Index lists the documents according to the EPA file structure (category number). The Index includes the following information fields:

- DOCUMENT NUMBER.... indicates the first and last page numbers of the document. Both page numbers will be the same for one-page documents. In this particular index, the document number consists of a three letter site code followed by microfilm reel and frame numbers.
- TITLE..... indicates the title or an enhanced description of the document in parentheses.
- AUTHOR..... indicates the author or primary originator and the author's corporate affiliation.
- RECIPIENT..... indicates the addressee or primary recipient and the addressee's corporate affiliation.
- DATE..... indicates document date by month/day/year. // means no date was available.
- TYPE..... indicates the document type.
- CATEGORY indicates the EPA file structure number.

APPENDIX D

MODEL POSITION DESCRIPTION FOR ADMINISTRATIVE RECORD COORDINATOR

INTRODUCTION

The incumbent serves as an Administrative Record Coordinator in one of the Regional offices of the Environmental Protection Agency (EPA). [Each Region may want to add an introduction to Superfund and the Regional office here.] The incumbent is responsible for compiling and maintaining administrative record files for CERCLA (Superfund) response action decisions.

Section 113(k) of CERCLA requires the establishment of an administrative record upon which the selection of a response action is based. Such a record is a compilation of all documents which the Agency considered or relied on in making its response action decision. Judicial review of any issues concerning the adequacy of any response action decision is limited to the administrative record. Public participation in the development of the record is required by law.

Establishment of thorough and complete administrative records is essential to EPA's Superfund program. Administrative records which include public participation and withstand judicial scrutiny allow EPA to meet its goals and objectives.

The incumbent will be responsible for compiling and maintaining administrative records for large numbers of Superfund sites. Each record requires coordination with many people including: Federal staff, state and local officials, private contractors, the general public and potentially responsible parties. Further responsibilities include deliberations over which materials to include in each record and requirements for dealing with privileged materials.

MAJOR DUTTES AND RESPONSIBILITIES

- 1. The incumbent is responsible for compiling and maintaining all of the administrative records for selection of CERCLA response actions for a Regional office of the EPA. The incumbent must have complete knowledge of all rules and procedures governing development of the administrative record files.
- 2. Receives and reviews all documents submitted by the Remedial Project Manager (RFM), On-Scane Coordinator (OSC), Office of Regional Counsel (ORC) and other appropriate staff for inclusion in the administrative record files. The incumbent will coordinate with staff responsible for deciding what documents are included in the record and will arrange for adding documents to the record file.

- 3. Compiles the administrative record file for each CERCIA response action. This includes logging the receipt of each document, maintaining a central master file of documents, redacting information from privileged documents as directed by ORC, maintaining any privileged portions of each record using Agency security measures, arranging for copying of documents in each record and transmitting the documents to appropriate repositories.
- 4. Coordinates the compilation of the administrative record files with state and federal agencies. This includes receiving records maintained by state and federal agencies and notifying appropriate personnel of these records for their review.
- Maintains and updates (monthly) an index of each administrative record file in conformance with Agency guidelines.
- 6. Ensures public access to administrative record files. This includes notifying the public of the availability of the record, making the record available for public inspection, coordinating with personnel at the facility where the record is located, maintaining an adequate copying facility and maintaining a log of persons reviewing documents. The incumbent will have to respond to phone calls and visitors wanting information on and from the record. These functions will be coordinated with the Office of Public Affairs and Superfund Community Relations Coordinators.
- 7. Maintains the Regional Superfund Central Library of guidance documents and technical references.

CONTROLS OVER NORE

The incumbent works under the general supervision of the [Hazardous Waste Branch Chief]. An administrative record is reviewed and certified for litigation by a person designated by the Regional Administrator.

APPENDIX E

COMPENDIUM OF CERCLA RESPONSE SELECTION GUIDANCE DOCUMENTS

USERS MANUAL

U. S. ENVIRONMENTAL PROTECTION AGENCY OFFICE OF WASTE PROGRAMS ENFORCEMENT

MAY 1989

.

TABLE OF CONTENTS

Sec. 20

- : NTRODUCTION
- 10 OVERVIEW OF COMPENDIUM USE
 - 11 USE BY EPA PERSONNEL
 - LSE BY THE PUBLIC
- 3.9 STRUCTURE OF THE COMPENDIUM
 - 31 FILE STRUCTURE FILE STRUCTURE
 - 12
- 40 UPDATING THE COMPENDIUM
 - REGIONAL INPUT 41 REGIONAL INPUT KEEPING THE COMPENDIUM CURRENT 42

LIST OF TABLES

Table

Pize

·

COMPENDIUM CATEGORIES AND NUMBER SERIES 3-1

Appendix

.

- (A) REGIONAL COMPENDIUM LOCATIONS AND ADMINISTRATIVE RECORD COORDINATORS
- (3) COMPENDIUM OF CERCLA RESPONSE SELECTION GUIDANCE DOCUMENTS INDEX

10 INTRODUCTION

This manual describes how to use the Compendium of CERCLA Response Selection Guidance Documents' (Compendium) Each U.S. Environmental Protection Agency EPA) Regional Office maintains a compendium of guidance documents frequently used during development and selection of response actions under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA).

EPA Headquarters used several sources to develop the initial Compendium. These sources included a pamphlet titled "Selected Technical Guidance for Superfund Projects" (OSWER Directive 9200.7-01); the OSWER Directive System; the Superfund, Resource Conservation and Recovery Act (RCRA), and Enforcement dockets; the Hazardous Waste Collection Database, and any existing regional compendiums. The documents in the Compendium are referenced in administrative records for decisions on selection of response actions.

The administrative record described here is the body of documents that form the basis for selection of a CERCLA response action. Establishment of the administrative record is required by §113(k) of CERCLA. An administrative record is the compilation of documents considered or relied on by EPA in making a decision. Documents that EPA anticipates will be included in the administrative record when the decision on a response action selection is made, are referred to as the "administrative record file." Guidance documents, or portions of guidance documents, that are considered or relied on in selecting a CERCLA response action should be part of an administrative record file.

Certain frequently used guidance documents may be referenced in the index to an administrative record but not physically included in the administrative record file. The reference should indicate the title and location of any documents included in the administrative record but maintained in the Compendium, which is kept at a central regional location. If a guidance document that is not listed in the Compendium is considered or relied on in selecting the response action, the document must be physically included in the administrative record file. The Compendium here reduce the burden of copying and storing multiple copies of frequently used guidance documents.

Section 2.0 of this manual briefly discusses use of the Compendium by EPA personnel and the public. Section 3.0 discusses the Compendium's file and index structure. Documents in the Compendium are filed in three-ring binders and listed on an index which is generated by and maintained on a computer database. Procedures for updating the Compendium are presented in Section 4.0

10 OVERVIEW OF COMPENDIUM LSE

The Compendium is intended for use by two groups: EPA personnel, during the process of response action selection and administrative record development, and the public, for review prodocuments referenced in the index to an administrative record.

The user should note that although the term "guidance" is often used in discussing the Compendium, it does not imply that only guidance documents are included. The documents π_3 , also be policies, memoranda, clarifications, case studies, manuals, handbooks, reports, and other documents used in the selection of CERCLA response actions.

2.1 USE BY EPA PERSONNEL

EPA personnel use the Compendium primarily to reference frequently used guidance documents that may be maintained in the Compendium rather than physically included in each administrative record file. The index must indicate which documents are physically located in the Compendium and must specify the location and accessibility of the Compendium. The index should also reference only the specific documents in the Compendium that were considered or relied on for the site for which the record is being compiled. The index should not reference the entire Compendium.

2.2 USE BY THE PUBLIC

As with any unrestricted document included in a record, the Compendium documents are accessible for public review. When EPA publishes a notice of availability of an administrative record file, that notice will include the location of the Compendium. The Compendium will be available for public viewing at a central regional establishment (for example, the EPA Regional Office), and unlist or near the sits for which the record is being compiled. (See Appendix A for a list of the location of each regional copy of the Compendium and the names of the Regional Administrative Record Coordinators.)

(2)

3.0 STRUCTURE OF THE COMPENDIUM

Currently, the Compendium is organized into 10 categories. An overview of the file tructure is presented below as well as a discussion of the index that identifies the documents included in the Compendium. This section also discusses the data elements identified in the index. The data elements provide vital information on the documents included in the Compendium and are contained in a database used to compile the Compendium and generate the index.

3.1 FILE STRUCTURE

The Compendium is structured according to 10 major categories that generally reflect the various components of a response action selection under CERCLA. Table 3-1 lists the current Compendium categories. The documents are further grouped into subcategories that indicate their more specific nature, when applicable. For example, the remedial investigation/feasibility study (RI/FS) section of the Compendium is broken down into more specific subcategories to identify the wide range of RI/FS documents available. When the documents apply to multiple categories, secondary references are provided in the Compendium index.

Each document has been assigned a unique four-digit document number. The bound documents contained in each category are arranged numerically. When a user wants to access a document, he or she will find the document filed according to the assigned number. The four-digit number series assigned to each category are also listed in Table 3-1.

3.2 INDEX STRUCTURE

- When an administrative record index refers to a document contained in the Compendium, that document is also identified in the Compendium index. The index, contained as the first document in the Compendium, provides the information necessary to identify and locate the desired documents (For a copy of the current Compendium index, see Appendix B.)

Because in most cases the user will know the title of the document rather than the number assigned, the index lists the documents under each category in alphabetical order. An alphabetical listing of secondary references follows the primary documents listed under each category.

(3)

TABLE J-1

COMPENDIUM CATEGORIES AND NUMBER SERIES

CATEGORIES	NUMBER SERIES
Index	00 00
Pre-Remedial	0001-0999
Removal Action	1000-199 9
Remedial forestigation/ Feasibility Study	2000-2999
General	2000-2099
RI Data Quality/Site & Waste Assessment	2100-2199
Land Disposal Facility Technology	2200-2299
Other Technologies	2300-2399
Groundwater Monitoring & Protection	2400-24 99
ARARs ¹	300 0- 39 99
Water Quality	4000-4999
Rick Assessment	5000-5999
Cost Analysis	60 00-6999
Community Relations	7 000-7999
Esforcement	8000-8 999
Selection of Remody/Decision Decamation	9000-9999

¹ Applicable or Relevant and Appropriate Requirements

⁽⁴⁾

Maintaining the index in a database allows the information to be organized in different have. For example, should the Region need an index that is sorted entirely in alphabetical order by title, chronologically by document date, numerically by the number assigned each document etc., EPA Headquarters can generate and forward such an index. The data elements of the Compendium database, as identified on the index, are included in Appendix B.

4.0 UPDATING THE COMPENDIUM

The Compendium is designed to allow for the periodic addition of newly developed policy or guidance documents. Updates to the Compendium are necessary in the following cases: (1) EPA releases relevant new guidance, policy, reports, etc.; (2) regional staff find additional documents that should be included in the Compendium; and (3) existing documents are revised or superseded. EPA Headquarters will continue to monitor the information sources used to develop the initial Compendium for new or revised documents that may qualify for inclusion in the Compendium.

Guidance documents identified for addition to the Compendium will be reviewed and relevant information will be entered into the existing database. After the database is updated, a new index will be generated and sent to each Regional Office. This new index will replace any previous indices. Hard copies of the additional documents will be sent to each region for inclusion in the Compendium. The revised index will indicate the category for each new document.

4.1 **REGIONAL INPUT**

Partice involved in the response action selection process, as well as Administrative Record Coordinators, may find documents that are frequently included in administrative records but are not referenced in the Compendium. In such cases it may be desirable to include the documents in the Compendium as part of the updating process. However, since the Compendium is designed to be nationally applicable, only documents used frequently in different regions will be included. Any region-specific document should be maintained in separate regional files and not in the Compendium.

(5)

4.2 KEEPING THE COMPENDIUM CURRENT

The a document is included in the Compendium it Alid remain in the Compendium maintain the integrity of any record that refers to it. However, documents contained in the Compendium may be revised in the future to reflect changes, for example, changes in policy, rechnology, or law. The most current version of these documents will be added to the Compendium, as appropriate, so that they will be available for the administrative record process.

Although no document included in the Compendium will ever be replaced or removed once an administrative record index refers to it, those documents that are superseded will be flagged and identified on a separate index (superseded index) attached to the Compendium's $\neg a \neg$ index. The superseded index will also identify the corresponding revised version added to $\neg e$ Compendium to indicate the new document that should be used.

Response action selections frequently rely on technical data generated at Superfund sites across the country. Such data is often maintained on national databases. Depending on their use and availability, certain of these databases may be included in the Compendium. For example, the Public Health Risk Evaluation Database (PHRED) is part of the Compendium. PHRED is stored on two floppy diskettes that are regularly updated as additional information becomes available. Whenever updated PHRED diskettes are generated, they will be added to the Compendium. Those diskettes that were previously included will also remain in the Compendium and will be identified on the superseded index.

(6)

(APPENDIX A)

.

REGIONAL COMPENDIUM LOCATIONS AND ADMINISTRATIVE RECORD COORDINATORS

Coordinator/PH *

Pegio	n Adress	1. Remedial
VEGTO	M Address	
I	90 Canal Street Boston, MA 02203	<pre>1. Brenda Haslett (617)573-1759 FTS 833-1759</pre>
	60 Westview Street * Lexington, MA 02173	2. Pam Bruno (617)860-4309
II	26 Federal Plaza New York, NY 10278	1. Jenny Delcimento (212)264-8676 FTS 264-8676
	Woodbridge Avenue * Raritan Depot - Bldg 10 Edison, NJ 08837	2. Norman Vogelsang (201)321-6657 FTS 340-6657
III	841 Chestnut Street Philadelphia, PA 19107	1. Margaret Leva (215)597-3037 FTS 597-3037
		2. Joan Henry (215)597-2711 FTS 597-2711
IV	345 Courtland Street, N.E. Atlanta, GA 30365	1. Debbie Jourdan (404)347-2930 FTS 257-2930
		2. Same
v	230 South Dearborn Street Chicago, IL 60604	1. Jamie Bell FTS 353-7446
		2. Jan Pfundheller FTS 353-7626
VI	1445 Ross Avenue 12th Floor, Suite 1200 Dallas, TX 75270	1. Karen Witten (214)655-6720 FTS 255-6720
		2. Joann Woods (214)655-2270 FTS 255-2270
*	The Compendium was initially d Administrative Record Coordinators	iistributed to reme dial s only. Copies may be

located at this address.

•

, <u>Pegion</u>	Address	Coordinator/PH = 1. Remedial 2. Removal
VII	726 Minnesota Avenue Kansas City, KS 66101	1. Barry Thierer FTS 276-7052
	25 Funston Road * Kansas City, KS 66115	2. Helen Bennett (913)236-3881 FTS 757-3881
VIII	999 18th Street Suite 500 Denver, CO 80202	1. Carole Macy FTS 330-1281
		2. Tina Ardemus FTS 330-7039
IX	215 Fremont Street San Francisco, CA 94105	1. Tom Mix FTS 484-1960 Don Briggs FTS 556-6637
		2. Holly Hadlock (415)768-1354
x	1200 Sixth Avenue Seattle , WA 98101	1. Lynn Williams (206)442-2121 FTS 399-2121
		2. Same

.

* The Compendium was initially distributed to remedial Administrative Record Coordinators only. Copies may not be located at this address.

(APPENDIX B)

•

COMPENDIUM OF CERCLA RESPONSE SELECTION GUIDANCE DOCUMENTS

INDEX
TABLE OF CONTENTS

CLERCEN - Subcategory	1	<u> 2- ; e</u>
Pre-Remedial	1001-3002	
Removal Action	· 300 - 1008	
RI FS - General	2000-2012	:
RI. FS - RI Data Quality, Site & Waste Assessment	2100-2119	:
RI. FS - Land Disposal Facility Technology	2200-2212	1
RI/FS - Other Technologies	2300-2320	5
RI, FS - Ground-Water Monstoring & Protection	2400-2408	7
ARARs	3000-3005	\$
Water Quality	4000-4003	7
Risk Assessment	5000-5015	ç
Cost Analysis	6000-600 I	11
Community Relations	7000-7000	11
Eaforcement	8000-800 i	12
Selection of Rannedy/Decision Documents	9000-9001	12

Data Element Definitions

List of Organizational Abbreviations and Acronyms Identified in the Index

"The range for each number series identified represents the numbers assigned to those documents currently in the Compendium.

••	Ravis - campai							
3000) CASE SILDIES 1-33 BENEDONE RESPONSE AT HURREDUE WARK SUIES	03/01/01	- CHOVER & LINERS	8 anna	8.34			11 A 340 / at .
			- Cartin Alama					
3044) LPA CLARE FOR IMPROVING DE JONRES DIVINDADENA EFFECTS OF	06/01/05	- ANY MEDGENIAL RESEARCHELANDER	f freet	250			
	CLEARLY OF LACED CALLED HAZARDILL&-INSITE SINES							
, 100		10/01/05	COMER/CEAR	d custi	21012	L		
	SADIS LICE CHICA							
2003	i jowi ginyiniya gagwal		CIENC. PAS	e kana k	48			
1004	4 ACELLING BENERAL ACTIONS AT LACENDERLED INLACED BULLE VILLE SITES	84/84/85	- BD.INELL \$ 20	d same	150			
	(NE 1-1V)		t and the second second second second					
			- Canch Align					
Junt.	I THE HER OF THE REAL PRANE AND AN AN ADDRESS OF THE CONCLARCTICS	48/01/85	· COLLEGE B IS IN ALLER	8 sand 8	٠			
			- ILQIND & A001					
800.00	4 DEAL FOR DESIDER AT HILFERE THERE SAVES TANKIN DEPORT	03/01/64		8 64408	85	٠		- 10 - 200-2
2003	4 GLVISED PROCEDURE HER INFLUENCE OFF-SITE RESPONSE ACTION	11/11/47	- Mikiak, j w Alibak	t inst	30			
2000	4 BIJS MENDERAS	01/13/61	· CONCEST IN C ACUM	e tenet		8	B B BEB / B S	
2000	4 BINS WEDDINGS FRUDER	04/35/88	· CALLST, HT & AGUN	t court	16		B3 BB785 BAB MING AND AND ALL AND ALL	
							and memory ways and a summary state a list of a sum	
							AL FIVE 18 Y	
2010	4 BANKING HERBAL-LEAD MANDLAL PROJECT MONCEMENT HOLEKOK	13/01/86	- 4.16.14ph	ة الدردة	179	•		
2011	S BARRAD MARCHAE DESIGN NO MARCHAE ACTION GARDING	06/01/86	- citing	e email	190	•		
3013	S BARBARA SIAN LEAD BEALLIA PROJECT MOREARD MOLICIX	13/01/86		f canad	1 10	•		
••	R1/85 - At Cuta (gastily/Sile & music Astestment							
1100	s a claradicul ce superior e filio cruitatica al sola	11401101			2.20	٠		LEWER B.L.
			um					
1 601	6 DATA QUELITY OBJECTIVES FOR MUNDIAL REPORTE ACTIVITIES	01/01/87	CEM FELLINAL MICHANE CEM-	8 oran 8	1 20			
	Lave Labraiced Philass							

1100 No voi title Unie Aulius Status Pages Dies - Attactments -----.

77

45/14 100

-

-

CONFIDENCE OF CERCEA AS STUDIE SEARCTICE CARLINES ALLERED S

		HULL
-		

•

CLAMBORIA OF CARCIA REPORT SELECTION CLAUNES ELEMANTS

ND	VD4 Fille	inte	Authors	Status #	ages 41	C.		
•								
2145	A PRACTICAL GLADE AND CHILAD-UNITE SAUFLINE	08/01/85	SANCELEDA MED ET AL VILLINGES SI	f une f	175			
			- SCALF IN R ARDAN					
1146	S SERVICE SAFENG QUELITY ASSANCE LESS'S QUEE	87/81/85	- BARDI D'S & STARLS I'S AMEY OF	i inal	140			
	• • • • • • • • • • • • • • • • • • • •		NEV LAS VICAS			•		11 47 44 4 1 1
			- 400004 6 9 /1/400					
2117	A SCHL SAME HE GAME IV AND ANCE LEW'S GAME			f a the l	104			
••••			Minibe Lat MCat			•		
2114	- MAL MERGER HER FUNLATIONS SEAD WALK, LANDAREN ANALAL	11/01/04	- Calut	1.041	in the second			
						•		
				1.041	1 80			
					1.00	•		10000
••	na sta - a und Daarnash Anrikkin indonésian							
••	BIND - Care to Barten Lections, service all							
	11 (THERE AND CATHORNELED METALINE APRIL 2415				4/5	•		
1.001		(1/0)/00			2010			114/381 0
2201		97/91/62	· MINN I LAISA CLEARS	1 1041	54 1	•		
			· SACREDS N & AREAL					•
1701		00/01/78		11011	61 1	•		
120	I I LOD DISTURE ASSNICTION	ag/11/67	· LUNCESE H & ACLAIN	i tasi	41 4	1	the second to the second calle should and second	
			MORE CANAL				LAG HELEMAN A 1150 HEREMAN (HELEMAN)	
							At the stants with the stants and the stants and	

74

PAG 10

COX

٠

							8 882 - 488 c	
3385	Is I wind it wills and an and and any approximate facturities	09/01/0 0	militation and	i conditi	470			
			LANDERS & ADDRESS BELLING					
			#6.#648.#6 L#					
3396	IS CHARGE OF THESE ANTRACTORY AND DESTRICT FACILITIES	03/01/83	LACRED & ALACRE	i smail		,		
3387	IS PRESENTS HE REALING FIGH DIRACH CAR LINES IN COMME	01/01/01	- 680	in all	145			
	ALGARID LINER REQUESS					-		ICALN PVID
2,006		47/01/02	tPA	in al t	ha			
						•		
3300		03/04/05		t such				
			GINNEL PA		•	•		41 A games .
3310		08/03/06	Children in a state of state					
					-	•		strate data a
		10-01-01		4				
3314		10/01/05	Fühlunder j. G. Antilis /Lanis (Child alles	i cont	66			
3914	IS NUMBER OUTDING COLUMNS CONSTRUCTION QUILITY ASSAULT HIS INCASELLS WISH LIND DISTRICT FACILITIES	10/01/00	FARMON J C. Anthe Andrew Market	4 cana 8	60			48 mi n <i>4</i> - 4
3344		10/01/05	FARMON (G AND AND ALLIAND CLARE DOV CLARE DOV	f court	**	•		45 WEN (7.4
3915	15 MUNICA CARDICI COLARD COSTACTION QUALITY ASSAULT HIS INCARLA VISIT AND DEVICE FACILITIES 15 MIANDER OF BLACTING MIGHTS AT INCARDAE WISH LOUPINGS	10/01/05	JARNANE J C. ANDA J.AND PLEINIGH CLIMER DOV CLIMER SEAMS (J. 53 4). /ADDLE D. 11541 	f 1966 F	••			-1 min #
9819 9914	IS MUTHERS OF DESCRIPTION OF DESCRIPTION OF AN INFORMATION OF DESCRIPTION OF DESCRIPTION OF A DESCRIPTION OF	10/01/05	HARMONE & G. Anthe /Lodd HLE (LA (De LANGE DOV LINGE SEAME (D. 55 % /ATHER D. 115); MC.	f 1966) f 1966)	••			18 WIN <i>9.4</i>
9999 9999 9999		14/01/05	HARDINGS & G. ANNES /Look +LL LA (De LANNE DOV LINNE BERNIE (D. 1994) BERNIE (D. 1994) ANDER ANDER (D. A. ANNES DELATE	d soudd d soudd	••	,		18 48 14 17 1 7 1
3313 3313	IS MUSICAL CARDING COLUMN COMMENDE COMMENDE QUELTY ASSAULT HIS INCARDED TO BE DE DIDING FACILITIES IS MEANING TO BEACHING MIGHTS AN INCOMENDA WORK LOUPINES INCARD TO BE IGN MINICAL MONICAL MORELEOUS	14/61/65 61/61/65 01/01/85	FARMANNE & G. ANNES /Look +LE (LA (De ALMERE DOV ARME BERMER (D. 53 40. /ARMER D) (L1943 MC ANARCE & ANNE Shadde & Alme	f court f court f court	••	• •		18 48 14 19 14 14 1 14 14 14 14 14 14 14 14 14 14 14 14
7040 7315 7717	15 MURIELE QUEDEL COLUMN COERCIGE QUELTY ASSAULT HE INCARENE WILL AND DEVELOR FACILITIES 13 MEANING OF REACTING MERIES AN INCOMENCE WILL COMPLETE MURIELE BANKEY 13 MY ICOMETER OF ION MINISTER MODIFICE AND MERICAN ASSAULT ELEMENT OF ION MINISTER MODIFICE AND MERICAN ASSAULT ELEMENT OF ION MINISTER OF STREAM (Secondary	81/81/85 81/81/84 84/81/85	FARMINGE & G. ANNES /Look +LE (LA (De ALMERE BOW ARME BERNER (S. 55 46. /ARMERE D) & (15)() MC ANOR (S. ANNE RECARDER (ANNES RECARDER (ANNES	d court d court d court d court	••	• •		18 48 14 19 14 14 4 14 14 18 46 14 19 14
70m0 9915 9916	15 MURIELE QUEDEL COLUMN CONSTRUCTION QUELITY ASSAULT HIS INCARELE VISIE LIND DEVICES FACELULES 13 MEANING OF REACTING MIGHTS AN INCOMENDATION OF REACTING MURICLE DAMAGE 23 MINING ANNUEL DE INFORMATION MONOCOL MIG.MEMORY ASSICTING EXAMINE OF FACEMENT COLLECTION SUSTAINS (Secondary Meteromy)	81/81/85 81/61/84 84/81/85	FARMANNE & G. ANNES /Look +LE (LA (De ALMERE BOW ARME BERMER (S 55 46 /ARMER D) 61514 MC ANORER (ANNE Reader (ANNE	6 1166) 6 11669 6 11669	••			18 48 14 19 14 14 4 14 14 18 46 14 19 14
0407 9819 9917	15 MURIELE QUEDECE COLUMNE - COLEMENT QUELITY ASSAULT HIS INCARELE WITH LIND DEVICES FACILITIES 13 MEANING OF REACTING MIGHTS AN INCOMENTAL MIGHT LINE HUNGLE BANKEY 13 MYLICARENTY OF DE INFORMATION MONOCH MIGHTANTS ALSYCONG ENGEN AD LINDING COLLECTION SUSTAINS (Secondary Reference)	61/61/85 G1/61/85	HARMONDE & G. ANNEL /LAND HELLER (DE CEPHILE DOV LINNE SERVEL (J. 55 4). /AEDLE D. 61514 MC LINNEL & ANNEL SERVER & ANNEL	44 44 44	••	•		18 48 14 19 14 14 4 14 14 16 46 14 19 14 14
3915 3813 3813	 15 Million Ca. Oxford COLINDE COMBINE COMBILITY ASSAULT HIS INCARELA VISIT AND DEVELOR FACTORIES WITH ASSAULT HIS INCARED OF REACTING MIGHTS AN INCOME WISH ENGLISHED IN HUNCLE BANKEY 15 AVVICABLE IV OF DIG ION ANNALIS MONOCH MIG.Mistors AN SYNCHOL ENGLISH OD STADIME CELLECTION SYNCHAE (Secondary Reference) 11/15 Athen Incomplete 	61/61/85 61/61/85	HARMONE & G. Andra /Lands HER (LA (De CEMINE 2007 CEMINE 2007 SERVICES \$5.46 /ARTICE 20 (1016) MC COCINENS & ANDR SERVICE & ANDR	44 44 44	•			18 98 9 9 11 11 4 112 18 96 9 9 11 -
3314 3813 3000	15 Millioners Guidener Grounder Grounder Guilitt Albande Hib 19248212 Wild Lind Grundes Facilities 13 Milandre Grundelin mildes al Indeficies Wild Lindefills 14 Milandre Grundelin mildes Al Indeficies Wilde Mildefills 15 Milandre Grundelin Milandre Milandre Milandre 15 Milandre Grundelin Milandre Milandre 15 Milandre Grundelin Milandre Milandre 15 Milandre Grundelin Milandre 15 Milandre Grundelin Milandre 15 Milandre Grundelin Milandre 15 Milandre Grundelin Milandre 16 Milandre Grundelin Milandre 16 Milandre Grundeline Milandre Milandre 16 Milandre Grundeline Milandre Milandre 17 Milandre Grundeline Milandre Milandre 18 Milandre Grundeline Milandre Milandre Milandre 18 Milandre Grundeline Milandre Milandre Milandre Milandre 19 Milandre Milandre Milandre Milandre Milandre Milandre 19 Milandre Milandre Milandre Milandre Milandre Milandre Milandre 19 Milandre Milandr	61/61/85 91/01/85	FARMANNE & G. Andik /Look ett (Link) (1)-Mikk (kov (1)-Mikk (kov Statistik (kov Statistik (kov (1)-Mikk (k	44 44 94	•	•		18 48 14 19 14 14 4 14 14 18 46 14 19 14 14
 3311 	 15 Milliones Galdines COLANDE COLEMENT CONSTRUCTION QUILITY ASSAULT INB INVALUES VISIT COD DI UNION FACTIONES UNDER LOUFFILLS INVALUES AND DI UNION FACTIONES UNDER LOUFFILLS INVALUE BANNET 25 MYLICORETTY OF DE ION ANNALS MONOCH MIGHTANTS ACTIVITIES AND DE LADORE GELICITOR SIGNAL (Sociality Reference) 16 A GLOUDDING OF HIGHER LIS LOOP AN BE COLANDER OF INVALUES 	10-01-05	fillunes g G Andia (Lodo elizita) tient Stand Stand Stand Stand Andia to stan (Andia 2) tient MC Andia to stan Stand to control a Andia Standa g Alian	66 66 76	66 • •	•		• • • • • • • • • • • •

inte autors Status Pages tare - Attactoris ••

(an ees wat tille 10.000 to 10.000

101.63 COMPOSITION OF COMPA ALCOHOM SELECTION CLAUMER ELECTIONS

41/16/88

75

•

PAGE 10 5

PAGE NO 6

-

CONTROLING OF CENCLA RESPONDE SELECTION CLARINE CUILLANDES

£004 МФ	nai litte 	Cinte -		Status P	eges (s	er at tas fommed s	k ∆antji terke e rosto
\$ 101	te Course distante de esta sin	04/01/80		8 88288		4	48 ~ (na. 11
3 703	17 INCIDE BLOCE PROCEEDING FOR PROCEEDING HIGHE INCIDENTATION	88/ 61/81	- SCHOR TA EL AL ARDRANDU MELLARDI COMP	f small	445	,	4 606 1 4 <i>0</i> Ph 14
2 303	IT ITA GADE FOR IDDIFTING GLEALP ALUDATINES AT INZIDUS WELL SI US AD STILLS BIOLOGICAL DEARDS		- CRIMALIAN D. A. ACHEL - PACLESC DERIDUESE LARCHAUMY - RANNERE & C. ACERVALLES GRANNLANDALA REFLECTE LAN	1 1141	1 40		Etta data y by
2304	IT IFA GARE FOR INFECTIOLS WILL'E INTRIMO	05/01/06	- Charles Aligne	1 3440	л	,	
1 101	IT WARNES BULLISH ICE GLENNE IS SALACE WEGNINGH SINS		Gandel Double-La Val Ally 1 - 14 5 K 54	d freel	10	•	+ F 100 M
3100		05/36/85	BARRI E ADDR GRANNLINDE-GLALAND E VESTORCE JOHRIDE	8 80an 1	115	•	(fail) (****
1161	18 1991 M.C.I. C.B. STALLALING BLADIAS ACTION MIDIOLOGY #1.48	86/8 1/83		4 11141	e 34	•	81 A 6021 / M
3 308	ng same malas. Felix Strates start allow mills allos a cartellas est southallitas mais st	06/01/86	- Partin Hill Andin - Classond pi in a sta and s Clames	4 44444	415	•	H 4/301 (
3 300	19 I JOLANIA MANGDAA AGANDA AS WALIS DASPORAL SA DES GREVA SADA	10/01/85	- HANNERS - GRANNES CANADRA	t inst	160	•	H 4417 ()
1110	an Landania Pulat annociate	11/01/05	NERO E AND MUS C / MM ASSECTATES	f soul	540	•	LI AFTON & A
3344	20 aCB114 URANIDE IECONDUCIES ICH BURNIND WISHS	09/01/86	- BARLEY IN /EPA CAMP CRISSIN AND ACRES INC. GALUE & D. ANDO	4 1 1 10 1	1 163 1	•	61 Ri 744 ()
3363	IN PRACTICE CLADE-MIAL BADS FOR INZADDLE WISH WCHARARDS	84/01/86		e crant	•)	1	dd Ardenae y 11
4343	28 PRACTISCAL GLAUX - BLAL BLOOK FOR FOLIABLES WISTE WEWEEARDS - PRODUCT BLOWRY	61/01/86		t famil		I	

76

	3114		66/11/86		t i mat	33	•	BE MEMER ALL SAME SERVICE FROM MILLERANS	CONST AVER 1
		FINCE IT F. STVETUELA MARINE LIVE GEADING						a t Alba	
	3343	In alwise of several minimal transferring of children and the	11/01/04	SAME B.C. ET AL / JAD ASSULTANDS	d Lond	110	•		
				- Ballitety of August					
	3396	It serves a worked thinks thereas the champelie should	88/19/84 ·		i ical	183	•		41 Ar 340 F & 45 -
		STRIS-VOL I MICHAGAL EMELATION							
	2317	3) Sectors March & Charles Charles Han Particle and Anther Charles	#1/D1/\$4		l cont	130	٠		
	3316	22 SISHING TO ALLEI MANE IN SITE START IZATION OF MADE DEPOSITS	09/01/86	AND DEPENDENT OF A STAR ADDRESS OF A DEPENDENCE OF		167			
				- CRUBE III ANDA					
1	3310	2) MILINERER SINGERIC LLACE HER MEANING & CRIECA SOLS AD	08/81/88		f treat	1.00			
4		MULLES .							
	1120	22 MEANING MARKAGER AND A MOUNTARY STRATEGIES WITH	07/01/06		t small	13			
		5 MGH 115 B							
	•4	As ris County water was to ing & Pealecters							
	1400	3) CRIMERIA SCR ACTIVITIES AND AS OF MANAGEMENT MANDER COT LICER	87/81/86		l cruel	950	J		
		REAL STARTER INCOMENCE CLARAGE							
	3481	to the REAL CONTRACTOR COLOR TANK INCOMENCE (MILLING (CH.)	13/19/85	LICEND G & ANPE	i smal	55		B B BBLE AF FLERBE BE' SH BERSERNI AL	
		GADTOL BURADE						BORRES IN 155 MI LORD POL MATCH	
								F381188894 8 \1491 488 5	
	1011	Is there while she that, at the called her at individual set		PURINE & M PURINE	t scal				
		undia para uno 75							
	2002	A CHEAN WALK PRESCIENCE SMARCY	00/01/04	GRANE OF CHILDRED WATER PRIMECTARY	t coat	43			
	Jaka	IN CARELINES FOR COLOURNER CLASSIFICATION MAR BE IPA	12/02/00	AND THE OF CHILDRED MATCHING THE PARTY	is at t	فطنان			
		CILLAR MAR (CONCLUDE EMANGE							
	1485	A COMPANY OF MANAGEMENTE AND CARD MERA CHIAND WITH	01/10/06	INTERVESTING OF CONTRACTOR PROPERTY	l the	30		FE MANDAMETERS AND AND THE SAME METALLING	
		athe all soft Striklarks							

ND	vat state	(inte	Author s	Sialus Pages Liei	At Las Jan est s	
		••				

MAR COMPOSITION OF CORCEA DESPENSE SECTION CONTINUES (DECEMBER)

t ind

and a second and have been been and the second second

.

PALE NO 1 81/16/00

The is mindeline of the function of the Light sections will be derive same.

.....

PAGE HD 8

.

.....

- 1043834

CLAPSICHAM OF CONTA AS STORE SELECTION CARDINES (ESLANDES)

	ND	val title	(ate	Auditure a	Status (Pages	lize	Al factoment s	
			•••	•••••					
	2406	to Million (CR CROAD-WALK CAMARING	05/01/86	INCOMENTS IN SIL CHILDER MALLE FASE	f inal	JUU			15-56
				KINCE			-		
	han a		60/01/00	- 124	d annal				
							•		
	100				e sent e	•	•		the second second
	••								
	1000	25 APRICARIES OF THE STAN MARKED LEGANCIE REGARDINGS	04/01/85	- 3400468 3 7050	i indi		1		t Cardina al estas
	•	ALENGING LINES AD LEADAN COLLECTION STREET							
	, 10 0 i	75 CARGA CAPELARCE IN BIORIS BURGENDAL STATISES	10/03/03	· FCA18A.) W /Contin	t inst	19	•	ED PROVENESSES AND COMPANY COMPANY OF THE SECOND	
								A COLORADOR DIA DIA CONTRACTOR	
Ď	1001			- 0000	in al i	345			
			es/31/87	· Brūnis, L. II. /IPA	t inat	•	1		
	-		-	- BOLGERIAL SCINCINGS BC	1 stat	150			
				· (m)			•		
			a		t apet		,		
						•	•		
					4				
	3400						•		• • • • • • • • • • • • • •
		ROMA SIMPLADIA HOUSING LINE CONDUCT (SECONDARY INFORMATICS)							•
	2404	HIN I HAL AGA GUARD BONE GID LO HUN IDA KANG EMILAHAN ADE I	11/11/00	ACCHIDI G A AURA	B 10103	>>	8	at the attraction of the part of	COMPLETE COLOR
		Calinati Chiladed (Secondary Saferance)						1944 4 4 4 4 4 5 5 11 3 14 5 5 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
								PERFERENCE & STATEMENTS	
	2485	M CREATION AD MARGINGE INSPECTION GADE INCRA CREAD INTER	07130186	CENERALISE AND A BOOMERADE DEVELOPED	6 (ma)	20		A B BRANCHARD & S AND AND AND AND AND AN AN AN AN AN AN AN AN	· · · · · · · · · · · · · · · · · · ·
		ADA KAAG SISTAA) (Secondary helerence)							
	240	23 AGA GRAD-BANK ADE KRING NOMCAL OFGIGLON GADICE	00/01/86	8PA	8 4 mm 8	170			
		ann an (that) (training priorant)							
	2004	IS ALLA CIDAD-MILE ADE DEUE TIGHACA DECIMINE CADECE	07/01/87	ILLERD G & ADME	d famil	٠			
		and the and the state of the st							

78

	-										
					604 W.H.						
			a	ingeneration and	Calle & Respired Street				45		
	(Jul										
		nd III.	Upic			•	laius I	19 1 91 1	8164	AT Bar tan int s	6.8.08.04 (1) H (4)
		AN DERE ANALISET ATTRACTOR ANALISET AND ADDRESS AND A DESIDER OF A DESI		154			ا ام ہ	-			
		Chat Hermine Internet						-	•		
		LA ACTIVITY A DECISION AND ON ADDITION ADDITION (BEAUTION)									
		televants)						-	-		
	••	when Quality									
		an an informality canceled and a limit of a constant of the state of t	47/6 1/63	(Breed)				14			4.0.000 Million - 1.1.1
	-		03/01/00		-	9	Bank I	•4			12,0031 (Pro)
	-						6.0 0	•	4		15 36 1 0
			-								
3	-								•		•• •• • • •
•		the country and and also show that the state of the state					t mail				
				- 00444. (-			•		
	-	t mit Upphilipe the Cheveland antes activite t Evite 5 (Secondary Antesance)			a tananarahi		laud	17			
		,								P8 5844 44 2	
										2.2 All all 2.2.2.2. all bel apper and as to	
										55 18 46 199 8 ALM 8 18 8 1 4 8 4 4 4 4 4 4 4 4 4 4 4 4 4	
										EN ING. BILL E LAND	
	••										
		All store and a statistic field. Balls	Ge/ 16/84			6	144		,		
	2000			MANUELAN		•			•		
	-	A DEMON PRESS & BURGERCH PROPERTY OF COMPLEXES PRISING		Q. LANK A	SEGLIANS INL		****	1.00			
									-		
	100	If the GADAGE ICA DE CERDINATION CO ALER 144 DI ALELLINA	83/14/87	-				48		bb bemit Balad sintgas argarat	
		ACTIVE BIES ON DE DE SANDARAD RENDER ANNE SE		A12.0							

termit to a

00/24/00 IPA

PALA 143

Marts wild be 1006 p \$20025

.

PALE HG 10

MARK

ODIPHIOILIN OF CORD & MENONE SELECTION GAUMES INCLUDES

•

HD	voi	111e	úsie 	Authors	Status (Pages	lici	Alter mails	1.600,02.00 A 2.000 A
300-1)) }	CLARKLINES FOR DECORAS ADDESIDED	00/ <u>24</u> /86	· 694	i inal	14	ł		
1003	. 17		01/34/65	- LPA	i insi	14	ł		
1004	10		00/34/86	67A	ê însê		ı		
3003		neprisen, M. P. 1995) Gaultines file de march ener annende de genera andres		· 674	i i)	••	ł		
-	. 26-	(1-83-86-6-86-6-86-6-86-6-86-6-6-6-6-6-6-6	05/01/84	00/0464/6640	f small	1710	,		
		VOL 20 AGENDE, ASBNG, AMESICS, BARAM, BRADAJPARDE, Galana, Carron Managande, Genardere, Genarde,		Canali Josef					El Ar Yeur El an Inn
		General Con 146, Corre Chills, Carrol, Cor, 1,1405-002,500, 1,1-005-002,500, VG. 20							
		I LOCALDEDINGE, I LOCALDEDINGE.							
		realized, and, analy an a							
		POLYCEORIOLOGI BURGHELS (POLS), POLYCELLC HUMLIC ISENCCHERGE (PURS), PERSE, SELBALE LAD (CEPOLOS), SCHLM							
		CLARGE, BARREL ACID, 3,3,7,6-15 MAD-CROMBIND-D-CHOLMI 1,1,3 2-15 MAD-CROSTING, 35 MAD-CROSD-101.001, MILLING,							
		1 1 1-MIDSCHEIME, 1 1,1-MIDSCHULDING MIDSCHEIMER 2 4 5-MIDSCHEIMER, 3 4 6-MIDSCHEIMER, MINSCHE ÖRMUNN							
		was Ordenes, mand, and too cordinas)							
5000			-	484. 1	inst	-	2		

80

•

00

PALA PÜ 11

.

18

-

CONTRACTOR OF CONTRACTOR STATES AND A STATES

		•							
		at title	(ste	Auditors s	Status P	ages I	lici	nd Las Fallent s	
		•• ••••							
-		te mellem all alle talls statistical desire an Tallande Andre Bare			4 4				
-							•	and the second	
								ALLES ASSERTABLE REPORTED FOR THE STATE	
								444.8.441 5 BU/ BB	
- 94	11	IN PLENC HEAR IN BUDY EVERALISIN DELIGENCE OF ALLOW S MANAGE AND	00/14/88		ê êndê	-			
		NO DISATTLE COMMONG DE COMEINE PAR SIGNE AL MOLDIN							
		IN ADLE OF ADAM STRUCTUY BIOINSAIS IN DIE MONTHAL ACTION (MICHELS	46/01/67	ABEY LA LEA PACIFIC MURRIEST	f inci	-			
		AL INCOME AND SOUS		L ALBANDY					
-			04/01/00	- (her	ليوزز	144			
_				Cana	t tool	-			
	194	11 D. Contrad. Activity of Activity of Constraints on Activity					•		114410 0114
_	_								
	115	1) ALE SCHOOL SPECIAL			64.911	1.00	•		11441
				· MEASH, 1 & ANPI					
		h) increasing represented and according to a second of the	11/13/03		d taugt		1		
		analysis							
			10/01/07	PHE ASSELLAND SALEMENTELL	t ingt	56			
				CREAMEN					
			84/94/88		f coal	a daa	•		
					1 1 1 1 1				
-		I Beenbelberte mit and and and an and an and a second formant.				•	•		
		Aples and S							
••		Contracts by Relations							
		A COMPACTY OFFICIAL OF A PARTY OF	46/01/86		1 inni	100		ti there will star with the the	STATES BARR

PACE ND 12

- 101 63

CANNELLA DE CANCER DE SPORTE SECTION CERENCE CEREMINES

.

рик Ма , .	st Mile	ünte	Audikars 	Status Pa	iges I	laei	al 14+105-1+1 1	Liberli i Fran	
••	(ates cannal								
	N MARINE MESSION CARACTERSTONIC PORTICIPATION IN MARINE CARACTERS (CONTRACT CONTRACTOR PORTICIPATION IN MUNICIPAL INCLUSION OF AN FEASION IN STATIS	45/36/89 45/36/89) - Porlar, 5 10 Addah) - Porlar 5 10 Addah	t inst Linst	11 17	1		1 (1966)) 1 (1966))	₽16 a+ 8 +6 1 - 1
••	Selection of Result/Decision Discussion								
9008 9885	al mailin Galance (n susanan talacin a singh 15 milancinga bicashi mai dabigin silacada	13/34/86 86/34/85	- Pantan, y w Admin - Kilifallida, m Administrati Massis dat	i inal i mai	10 3	4 1		(BAQ H	e 11

.

82

.

DATA ELEMENT DEFINITIONS

The data elements of the Compendium database, as identified on the index are soon below:

DATA ELEMENT	DEFINITION
Dec No	Unique four-digit number assigned to a document included in the Compendium according to category
Vol	Volume number of the binder in which the hard copy of the document is contained.
Title	Title of the document. Secondary Reference is identified following the title when a document relates to more than one category. The document itself is filed under the number series assigned to its primary category.
Date	The date the document was published by or released from the issuing office or entity.
Authors	Author(s) and affiliation(s). Also includes identification of the EPA Project Officer and issuing office, where applicable.
Status	Indicates the status of a document, either draft or final version.
Pages	Total number of printed pages of the document, including any attachments.
Tier	Tier 1 or Tier 2. Tier 1 documents are the core documents of the Compendium as listed in the pamphlet titled "Selected Technical Guidance for Superfund Projects," compiled by OERR. Tier 2 documents are all other documents included in the Compendium.
Attachments	Attachments to a document by complete or abbreviated title.
OSWER/EP/HERiter	EPA report or OSWER Directive System numbers, where applicable.

•

•

LIST OF ORGANIZATIONAL ABBREVIATIONS AND ACRONYMS IDENTIFIED IN THE INDEX.

21220.0000	<u>A:::20::71</u>
Agency for Toxic Substances and Disease Registry	ATSDR
Center for Environmental Research Information	CERI
Contract Laboratory Program	CLP
U.S. Corps of Engineers	COE
Exposure Assessment Research Division	EARD
Environmental Criteria and Assessment Office	ECAO
Environmental Monitoring Systems Laboratory	EMSL
Emergency Response Division	ERD
Environmental Research Laboratory	ERL
Hazardous Response Support Division	HRSD
Hazardous Site Control Division	HSCD
fazardous Site Evaluation Division	HSED
Hazardous Waste Engineering Research Laboratory	HWERL
Municipal Environmental Research Laboratory	MERL
Office of Environmental Engineering and Technology	OEET
Office of Emergency and Remedial Response	OERR
Office of Health Effects Assessment	OHEA
Office of Research and Development	ORD
Office of Solid Water	OSW
Office of Solid Wass and Emergency Response	OSWER
Office of Weste Programs Enforcement	OWPE
Policy Agalysis Staff	PAS
Waterways Experiment Station	WES
Waste Management Division	WMD

.

APPENDIX P

MODEL TRANSMITTAL COVER LETTER

[Name of Contact] [Address;

Sear [Name of Contact]:

The U.S. Environmental Protection Agency is required by law to establish administrative records "at or near a facility at issue." This administrative record consists of information upon which the Agency bases its selection of response action for a particular Superfund site.

By providing the public with greater access to these records, it is our hope that they will be better equipped to comment constructively on site activities and to understand the issues relating to the selection of the response action at the site.

We appreciate having the [Name of local repository] as the designated administrative record facility for the [Name of site] Superfund site. The enclosed record files, along with any future documents relating to technical activities at the site should be placed in the [Name of local repository] and be available for public review. The record files should be treated as a non-circulating reference - it should not be removed from your facility.

Also enclosed is a fact sheet to assist you and your staff in answering questions posed by the public concerning administrative records for selection of response actions at Superfund sites. Please feel free to distribute this guide to the public.

To ensure the receipt of the administrative record file, I would appreciate your completion of the attached Document Transmittal Acknowledgment form. Please return this form in the enclosed selfaddressed, stamped envelope.

Again, I would like to thank you for your cooperation with the U.S. EPA in serving as a Field Repository. If you have any questions or comments, please contact [Name of EPA contact] at [Phone No.].

Sincerely,

[Name] Administrative Record Coordinator

APPENDIX G

HODEL DOCUMENT TRANSHITTAL ACKNOWLEDGMENT

From: [Regional Office Address]

To: [Field Repository Address]

I acknowledge that I have received the following documents from the U.S. EPA Region _____ Office, pertaining to [Site Name] Superfund site.

Administrative	Record	Name -			[Site Name]
Administrative	Record	Document	Numbers	-	

Signed _____

Date _____

Please return this form to: [Regional Office Address]

APPENDIX H

PACT SHEET

Administrative Records in Local Repositories

The "administrative record" is the collection of documents which form the basis for the selection of a response action at a Superfund site. Under section 113(k) of the Comprehensive Environmental Response, Compensation, and Liability Act, as amended by the Superfund Amendments and Reauthorization Act (CERCLA), EPA is required to establish an administrative record for every Superfund response action and to make a copy of the administrative record available at or near the site.

The administrative record file must be reasonably available for public review during normal business hours. The record file should be treated as a non-circulating reference document. This will allow the public greater access to the volumes and also minimize the risk of loss or damage. Individuals may photocopy any documents contained in the record file, according to the photocopying procedures at the local repository.

The documents in the administrative record file may become damaged or lost during use. If this occurs, the local repository manager should contact the EPA Regional Office for replacements. Documents may be added to the record file as the site work progresses. Periodically, EPA may send supplemental volumes and indexes directly to the local repository. These supplements should be placed with the initial record file.

The administrative record file will be maintained at the local repository until further notice. Questions regarding the maintenance of the record file should be directed to the EPA Regional Office.

The Agency velcomes comments at any time on documents contained in the administrative record file. Please send any such comments to [name and address]. The Agency may hold formal public comment periods at certain stages of response process. The public is urged to use these formal review periods to submit their comments.

For further information on the administrative record file, contact [name and phone no. of Administrative Record Coordinator].

.

APPENDIX I

MODEL NOTICE OF PUBLIC AVAILABILITY

THE UNITED STATES ENVIRONMENTAL PROTECTION AGENCY ANNOUNCES THE AVAILABILITY OF THE ADMINISTRATIVE RECORD XYZ SITE, [Locality, State]

The U.S. Environmental Protection Agency (EPA) announces the availability for public review of files comprising the administrative record for the selection of the (remedial, removal) action at the XYZ site, [Locality, State]. EPA seeks to inform the public of the availability of the record file at this repository and to encourage the public to comment on documents as they are placed in the record file.

The administrative record file includes documents which form the basis for the selection of a [remedial, removal] action at this site. Documents now in the record files include [preliminary assessment and site investigation reports, validated sampling data, RI/FS work plan, and the community relations plan]. Other documents will be added to the record files as site work progresses. These additional documents may include, but are not limited to, the RI/FS report, other technical reports, additional validated sampling data, comments and new data submitted by interested persons, and EPA responses to significant comments.

The administrative record file is available for review during normal business hours at:

[Repository Name] [Address and Phone	#]	and	U.S.EPA - Region 2 [Address and Phone #]

"Additional information is available at the following locations:

Verified sampling data and documentation	•	Contract laboratory, [Address and Phone #]
Guidance documents and	-	EPA-Region 2

termical literature [Address and Phone #]

Written comments on the administrative record should be sent to:

(Name), Office of Public Affairs U.S. EPA - Region 3 [Address and Phone #] APPENDIX J

MICROPORN APPROVAL MEMORANDUM



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

OCT 2 | MIR

STA OF ST STATE

MENCRANDUM

- SUBJECT: Microfilming t e Admin.strative Record FRCM: Edward J. Hanley, Director
- TO: Asa R. Frost, Jr., Director OSWER Information Management Staff

In accordance with EPA Records Management Manual, Chapter 6, dated 7/13/84, I approve OSWER's request for an administrative record micrographic system for regional hazardous waste management programs.

The feasibility study prepared for OWPE, entitled "Assessment to the Suitability and Costs of Alternatives for the Administrative Record" (June 30, 1988), satisfactorily documents and justifies the need for converting the administrative record to microform. In particular, the requirement under SARA to make the administrative record publicly available at or near each hazardous waste site makes microform a cost-effective storage medium.

Please inform each regional hazardous waste program of ay approvel of OSWER's request and of the need to comply with the remaining provisions of Chapter 5 of the EPA Records Manual should the region proceed with implementing an administrative record micrographic system.

ce: SIRMOs, Region 1 - X

:

APPENDIX K

MODEL CERTIFICATION

IN THE (NAME OF COURT)

UNITED STATES OF AMERICA,	:	
Plaintiff,	:	
	:	
۷.	:	
	:	
[NAMES OF DEFENDANTS]	:	
	:	
Defendants,	:	CIVIL ACTION NO.
[number]		
•	:	
ν.	:	
	:	
[NAMES OF THIRD PARTY	:	
DEFENDANTS]	:	
	1	
Third Party Defendants	:	

CERTIFICATION OF DOCUMENTS COMPRISING THE ADMINISTRATIVE RECORD

The United States Environmental Protection Agency (EPA) hereby certifies that the attached documents constitute the administrative record for selection of response actions under the Comprehensive Environmental Response Compensation and Liability Act or 1980, as amended, for the [name of site] site in [City or County], [State].

By the United States Environmental Protection Agency:

.

.

In vitness whereof I name this day o in (gity]	have subscribed my
feignatur	<u>.</u>

.

ftyped name

.

PREAMBLE TO SUBPART I OF NCP

Support I of the NCP is entirely new It .mpiements CERCLA requirements concerning the establishment of an administrative record for selection of a response action. Section 113(k)(1) of CERCLA requires the establishment of 'an administrative record upon which the President snall base the selection of a response action. "Thus. today s rue requires the estabushment of an administrative record that contains documents that form the basis for the selection of a CERCLA response action. La addition. section 113(k)(2) requires the promuigation of regulations establishing procedures for the participation of interested persons in the development of the administrative record

These regulations regarding the administrative record include procedures for public participation. Because one purpose of the administrative record is to facilitate public involvement, procedures for end of the proposed NCP are addressed elsewhere in this preamble

The following sections discuss the major comments received on the proposed suppart I and EPA's responses

Scme General comments.

Proposed rate Subpart I details how the administrative record is assembled, maintained and made available to the public.

Response to comments Comments on the administrative record regulations included the suggestion that the preamble provide a general statement differentiating between the administrative record and the information repository.

EPA agrees that while subpart I includes ample information on the requirements of the administrative record, a brief clarification would help to differentiate the record from the information repository.

The information repository includes a diverse group of documents that relate to a Superfund site and to the Superfund program in general, including documents on site activities, information about the site location, and background program and policy guides. EPA requires an information repository at all remedial action eites and any site where a removal action is likely to extend beyond 120 days. The purpose of the information repository is to allow open and convenient public access to documents explaining the actions taking place at a site.

The administrative record discussed in this subpart, by contrast, is the body of documents that forms the basis of the agency's selection of a particular response at a site, i.e., documents relevant to a response selection that the lead agency relies on as well as relevant comments and information that the lead agency considers but may reject in the ultimate response selection decision. Thus, the recerd will include documents the lead and support agency generate. PRP and public comments, and technical and site-specific information. ments. and technical and site-speafic information. These documents essentionally overlap with those included in the information repository. The administrative record includes such information as eitespecific data and comments, guidance documents and technical references used in the selection of the response action. The information repository may include guides to the Superfund process. background information. fact shoets press releases, maps, and other information to aid public understancing

11 15 e response regardiess of whether the information has bearing on the eventual response selection at that site.

One commenter felt that there was no mechanism for PRPs to perticipate in the development of the edministrative record. In response. PRPs are given a chance to participate in the development of the administrative record throughout its compilation. EPA will make available information considered in selecting the response action to PRPs and others through the administrative record file. Interested persons may peruse the record file, submit information to be included in the administrative record file. or may comment on its contents during the ensuing public comment period.

Name: Section 300.800(a). Establishment of an administrative record. Section 300.810(a). Contents of the administrative record.

Proposed rule: Section 113(k)(1) of **CERCLA** states that the "President shall establish an administrative record upon which the President shall base th selection of a response action." EPA used similar language in § 300.800(a) of the proposed rule: "The lead agency shall establish an edministrative record that contains the documents that form the basis for the selection of a response action." (Emphasis added.) Section 300.810(a) states that the administrative record file for selection of a response action typically, but not in all cases, will contain the following types of documents " "." followed by an enumeration of those documents.

Response to comments: EPA's choice of the phrase "form the besis" in § 300.000(a) drew many comments. The comments expressed concern that the lead agency would have the discretion to include in the administrative record only those documents that support EPA's selected remody.

These comments appear to be based on a misunderstanding of what the phrase "forms the basis of" means as it was used in the proposed rule. The statute defines the edministrative record as the "record upon which the President shall base the selection of a response ection." EPA's intent is defining the record as the file that "contains the documents that form the basis for the selection of a response ection" was simply to reflect the statutory language. For example, an administrative record will contain the public comments submitted on the proposed action, even if the lead agency is required to consider these comments and respond to eignificant comments in making a final decision. Thus, these comments also form the basis of the first section decision. EPA intends that he regulatory language defining the administrative record file emoody general principles of administrative taw concerning what, documents are included in an "administrative record for an agency decision. As a result, contrary to the suggestion of the commenters, the proposed definition of the administrative record does not mean that the record will contain only those

documents supporting the selected

response action. A commenter asked that the onrase "but not in ell cases ' be deleted from \$ 300.810(a), or specify the cases where documents are excluded from the administrative record. EPA believes it is better not to attempt to list excluded documents in the NCP since EPA cannot possibly anticipate all the types of documents that will be generated for a site or for future sites, and which of these documents should be excluded except as generally described in 3 300.810(b). It should be noted, for example, that although a health accessment done by ATSDR would normally be included in the admunistrative record, it would not be if the assessment was generated by ATSDR after the response is selected.

Others commented that certain documents should always be included in the administrative record. EPA believes that only a small group of documents will always be generated for every type of CERCLA site, since each site is unique. Other documents may or may not be generated or relevant to the selection of a particular response action at a site. EPA understands that a definitive list of required documents would assist parties m trying to easess the completeness of the administrative record, but such a list would not be practical. Different sites require different documents.

A related group of comments asked that the administrative record alweys include certain documents, including. specifically, "verified sampling data." draft and "predecisional" documents. and technical studies. One comment stated that "invalidated" sampling data and drafts must be part of the administrative record in some situations. Verified sampling data. 1.e., data that have gone through the quality assurance and quality control process. will be included in the record when they have been used in the selection of e one action. "Invalidated" data. 1 c.. data which have been found to be incorrectly gathered, are not used by EPA in selecting the response action and should therefore not be included in the

DATA ELEMENT DEFINITIONS

.

The data elements of the Compendium database, as identified on the index are shown below

.

DATA ELEMENT	DEFINITION
Doc No	Unique four-digit number assigned to a document instance in the Compendium according to category
Vol	Volume number of the binder in which the hard copy of the document is contained.
Title	Title of the document. Secondary Reference is identified following the title when a document relates to more than one category. The document itself is filed under the number series assigned to its primary category.
Date	The date the document was published by or released from the usuing office or entity.
Authors	Author(s) and affiliation(s). Also includes identification of the EPA Project Officer and issuing office, where applicable.
Status	Indicates the status of a document, either draft or final version.
Pages	Total number of printed pages of the document, including any attachments.
Tier -	Tier 1 or Tier 2. Tier 1 documents are the core documents of the Compendium as listed in the pumphlet titled "Selected Technical Guidance for Superfund Projects," compiled by OERR. Tier 2 documents are all other documents included in the Compendium.
Attachments	Attachments to a document by complete or abbreviated title.
OSWER/EPAtellahor	EPA report or OSWER Directive System numbers, where applicable.

per od upline apprins straf ve record regardless of their significance. When the lead agency considers comments submitted after the decision document has been signed, the "significance" of e comment has a bearing on whether it will be included in the administrative record, as specified in \$ 300.825(c). In addition, while EPA is under no legal obligation to place in the record or consider comments submitted prior to the comment period. EPA will generally. as a matter of policy consider significant comments submitted prior to the comment period, place them into the record, and respond to them at an appropriate time. However, persons who wish to ensure that the comments they submitted prior to the comment period are included in the record must resubmit such comments during the comment period.

Final rule: Section 300.800(a) is promulgated as proposed.

Name: Section 300.800(b). Administrative record for federal facilities.

Proposed rule: Section 300.800(b) states that the lead agency for a federal facility, whether EPA. the U.S. Coast Guard, or any other federal agency. shall compile and maintain an administrative record for that facility: When federal agencies other than EPA are the lead at a federal facility site. they must furnish EPA with copies of the record index. in addition to other specified documents included in the record. The preamble to the proposed NCP discussion of \$ 300.800(b) (53 FR 51464) states that EPA will establish procedures for interested parties to participate in the administrative record development, and that EPA may furnish documents which the federal agency is required to place in the record.

Response to comments: One comment stated that EPA should be the custodian for administrative records for federal facilities, especially where the federal facility is a PRP, to avoid any conflict of interest in questions of Hability or litigation. Another comment stated that the requirements in § 350.000(b) of the proposed rule would be burdensome to federal agencies in compiling and maintaining the record. Executive Order 12500 grants federal

Executive Order 12500 grants federal agencies the authority to "establish the edministrative record for selection of response actions for federal facilities under their jurisdiction, custody or control." To avoid the potential for conflicts of interest by federal agencies who are PRPs and in charge of compiling and maintaining the record, ZPA retains record by subplementing the record and by requiring an accounting of what is in the record through a report of the indexed contents. EPA believes that these requirements represent sufficient Agency oversight to avoid potential conflicts of interest at federal facilities while ensuing that federal facilities while ensuing that federal leed agencies remain responsible for compiling and maintaining their own administrative record.

EPA is making a minor editorial change in § 300.800(b)(1) to reflect that the federal agency compiles and maintains an administrative record for a facility, and not of a facility, since § 300.800(a) already provides that the record will be located at or near that facility.

Final rule: EPA is promulgating the rule as proposed, except for the following minor editorial change in the first sentence of § 300.800(b)(1): "If a federal agency other than EPA is the lead agency for a federal facility, the federal agency shall compile and maintain the edministrative record for the selection of the response action for that facility in accordance with this subpart."

Name: Section 300.600(c). Administrative record for state-lead sites.

Proposed rule: Section 113(k) of CERCLA states that the President "shall establish an administrative record upon which the President shall base the selection of a response action." Section 300.800(c), entitled "Administrative record for state-lead sites," requires that states compile administrative records. for state-lead sites in accordance with the NCP.

Response to comments: Several commenters believe that the new administrative record procedures place an onerous burden on the state, and as that state requirements such as Open Records Acts should be allowed as a substitute for compliance with subpart L Another commenter recommended that states be allowed to determine whether a complete administrative record is needed at or near the site when a site is state-lead. Where a response is taken unde CERCLA at a state-lead site. EPA is ultimately responsible for the selection of a response action. Therefore, under section 113(k), EPA must establish an administrative record for the CERCLA response action at the site, and must, at a minimum, comply with subpart L There may be many different ways of compiling administrative records and involving the public in the development

111 A Lead agent as not an intermediate additional public may provide additional public involvement opportunities at a site in response to whether or not states should maintain a complete edministrative record at or near the site. EPA believes that states must have such a record in order to meet CERCLA section 113 k) requirements.

EPA has included a minor editor all change in § 300.800(c) to reflect that a state compiles and maintains an administrative record for rather than at a given site.

Final rule: EPA is promuigating § 300.800(c) as proposed, except for a minor editorial change in the first sentence as follows: "If a state is the leed agency for a site, the state shall compile and maintain the administrative record for the selection of the response action for that site in accordance with this subpart."

Name: Sections 300.800(d) and 300.800(e). Applicability.

Proposed rule: Section 300.600(d) etates that the provisions of subpart I apply to all remedial actions where the remedial investigation began after the promulgation of these rules, and for all removals where the action memorandum is signed after the promulgation of these rules. Section 300.800(d) also proposes that "[T]his subpart applies to all response actions taken under section 104 of CERCLA or sought secured or ordered administratively or judicially under section 106 of CERCLA." Section 300.800(e) states that the lead agency will epply subpart I to all response actions not included in § 300.800(d) "to the extent practicable."

Response to comments: One commenter argued that the applicable provisions of subpart I should be amended to require agencies to comply with the subpart for all sites where the remedy selection decision was made more than 80 days after proposal of the revised NCP for comment. Another comment stated that § 300.600(e) be revised to state that § 300.600(e) be revised to state that is any future actions they take, and that all lead agency actions must comply with subpart I "to the maximum extent practicable."

in response. EPA will adhere as closely as possible to subpart I for sites where the remedial investigation began before these regulations are promulgated. EPA will not, however. require that these sites comply with requirements which, because of the

pelachered o Porlexanto e under the final rule the administrative record file. must be available at the beginning of the remedial investigation phase. If these regulations are promulgated when a site is in the middle of the remedial investigation process, and the administrative record is not yet available. the lead agency cannot at this point comply with these regulations. Additionally EPA believes that adding language to proposed NCP § 200 800(e) to state that lead agencies will comply with provisions of subpart I in any future actor after promulgation of the new rule is unnecessary and redundant computance will be leasily required, and applicability to all future response actions is implicit in the rule. Likewise. insertion of the word "maximum" before the phrase "extent practicable" is unnecessary since it would give additional emphasis but would got substantively change the requirement or the meaning of the rule.

One comment agreed with EPA's interpretation that subpart I applies to all response actions "sought secured or ordered administratively or judicially, but others disagreed. Several stated that the term "judicially" should be deleted from \$ 300.800(d) because they argue that response actions ordered judicially would receive de novo adjudication. instead of advainistrative record review. CERCLA section 113([/[1] states: "In say judicial action under this Act. judicial review of any usues concerning the adequacy of any response action taken or ordered by the President shall be limited to the administrative record." Commenters contend that this section does not apply to injunctive actions under CERCLA section 108 because them are not actions "taken or order by the President." To the contrary, the selection of a response action is a "response action taken * * * by the President" Accordingly. section 113(1)(1) requires that judicial review of the response action selected by the a is "limited to the administrative re a that. Further, section 113(j)(2) sti "in any judicial action under this chapter"-whether for injunctive relief. enforcement of an administrative orde or recovery of response costs or damages --- a party objecting to "the President's decision in selecting the response action" must demonstrate. "on the edministrative record, that th decision was arbitrary or capricious or otherwise not in accordance with law."

EPA received several comments objecting to EPA's determination that judicusi review of an andangerment ALT - 3.13 . 4 record They have a rail as a matter of acmidistrative and constitutional law a finding of molineat and substantial endangement is not an issue concerning the adequacy of the response action. As stated in CERCLA section 113(), and therefore must receive de novo review by a court. A second comment requested that EPA state in the regulation that review of EPA's expenditures in the

mplementation of a remedy is de novo. An assessment of endangerment at a site is a factor highly relevant to the selection of e response action, and is in fact part of the remedial investigation (RH process central to the decision to select a response action. Therefore, the determination of endangerment (which will generally be included in the decision document) will be included in the administrative record for selection of a response action and should be reviewed as part of that record. (EPA notes that the term "endangerment assessment" document has been superseded by the term "Tisk assessment" document, and while assessments of endangerment at a site are still conducted during the RL it is the "nak assessment" document that becomes part of the record.) In response to the comment that Accury expeditures on a response action should receive de novo review. EPA notes that this issue was not raised in the proposed NCP, and is therefore not addressed in the final rule.

Final rule: EPA is promulgating the rule as proposed.

Namer Section 300.808. Location of the administrative record file.

Proposed rule: Section 113[k](1] of CERCLA states that "the edministrative record shall be available to the public at or near the facility at issue. The President also may place duplicates of the administrative record at any other location." Section 200,005 of the proposed NCP provides five exemptions for isformation which need not be placed at or near the facility at issues Sampling and testing data, guidance documents, publicity available technical literature, documents in the confidential perion of the file, and emergency removal actions lasting less than 30 days.

days. Bespanse to comments: One commenter supported limiting the amount of information which must be logged at or near the site, but many commenters stated that every document contributing to decision-making, including confidential documents which are part of the record, should be located at or near the site and agency

-1. asserred a ACTIVE DUDI are contrar comment st administrat places, at a near the sit statutory re record only Issue." One acknowleds headquarte keep the ac Superfund : Indian rese requested t regulatory | records can other record and have th Daber recor Recourse

documents both unner very costly. often kent a space is iim accommode for large, oc the data an which is loc addition, re technical lit ----additional : Superfund (experience coble view file at or se of the same documents file.

However specify the review a de bet not ave. near the sit confidential a inclusion in the individual will Preedom of Inf ion to have t available for a sute. IPA belle documents negaset ser CERCLA # be "availat addition. th LOUDGAS T information weating for information practical, a informetics

 Se in centand of the mormal on a central to the response selection decision.

The confidential portion of the file need not be located at or near the site. and will not be available upon request either at the site or at the central location, since the information is not available for public review.

EPA believes that requiring that the record be located in two places is recessary to ensure both adequate public access to the record files and cetter lead-agency control over the record documents. The statutory requirement in CERCLA section 113(k)(1) states that the President may also place duplicates of the administrative record at any other location. This section clearly provides authority to maintain a second administrative record at a central location. Section 300.805 of the proposed NCP (33 FR 51515) reflects EPA's decision to make this statutory option a regulatory requirement A centrally lucated record may offer easier access to interested parties located far from the response site.

EPA agrees with the commenter that housing the centrally located copy of the record at Indian tribal headquarters may be appropriate when a Superfund site is located at or near an Indian reservation. In the 1966 amendments to CERCIA. Indian tribes are accorded status equivalent to states, and can be designated lead agencies for response actions, in which case they would also be required to compile and maintain the administrative record at or near the site.

as follows:

1. Section 300.005(b) is added to the rule as follows: "Where documents are placed in the central location but not in the file located at or near the site. such documents shall be added to the file located at or near the site upon request. except for documents included in paragraph (s)(4) of this section." The as follows. The ead agency may make the administrative record file available to the public in microform." 3. The section has been renumbered accordingly.

Name: Sections 300.810(a)-(d). Documents not included in the administrative record file.

Proposed rule. Section 300 810(b) discusses which documents may be excluded from the edministrative record. Section (c) discusses privileged information that is not included in the administrative record. Section 300.810(d) discusses confidential information that is placed in the confidential portion of the administrative record.

Response to comments: One commenter argued that § 300.810 should specifically include an exemption for classified documents related to national security. While the NCP currently does not address the potential conflict between national security concerns and the requirement to establish a publicly accessible administrative record, it is not clear that such an exemption could be adequately specified by rule or that an exemption would appropriately resolve this conflict. Section 121()) provides a national security waiver by Presidential order of any requirements under CERCLA, which can be invoked in certain circumstances. Under this provision, protection of national security interests requires case-by-case review under section 121(j) and not a blanket exemption in the NCP. Nothing in the NCP limits the availability of this waiver.

Another comment received by EPA stated that the treatment of privilege and confidential documents in the records is unfair, because it denies access to documents that may be critical to the selection of a remody. EPA has provided for a confidential portion of the administrative record where documents containing, for example, trade secrets of companies that have developed patented cleanup technologies being considered as a response-selection alternative can be t confidential. To maintain a fair balance between the need for confidentiality and the public's right of review of the record, the lead agency must summarize or redact a document containing confidential information to make available to the greatest extent possible critical factual information relevant to the selection of a respo action in the nonconfidential portion of the record.

A final comment proposed that an index to the privileged documents should be included in the-

montany wern a compressive

edministrative record EPA agross believing that an index will et interested parties know in generationwhat documents are included in the record without compromising the confidential nature of the information contained in those documents.

Finally. EPA is adding a sentence 'D § 300 810(a)(6) to clarify that the incex can include a reference to a group of documents. if documents are customanly grouped. This will simp: ', EPA s task without compromising the integrity of the record.

Final rule. 1. EPA is promulgating \$\$ 300.810(b). (c) and (d) as proposed with a minor editorial change to clarify the first sentence of \$ 300.810(d).

2. The following language is added 'o § 300.810(a)(8) to provide for listing grouped documents in the administrative record file index: 'If documents are customarily grouped together, as with sampling data chain of custody documents, they may be listed as a group in the index to the administrative record file."

Name: Section 300.815. Administrative record file for a remedial action.

Proposed rule: The term "administrative record file" is used throughout the proposed NCP Section 300.815(a) proposes that the administrative record file be made available for public inspection at the beginning of the remedial investigation phase.

Response to comments: EPA received several comments objecting to the concept of an administrative record file. They objected because there is no statutory authority for establishing a file, and because they were concerned that the lead agency could edit the file, specifically by deleting public and PEP comments and information that do not support the response action ultimately chosen by EPA, and that these comments and information would not remain a part of the final administrative record.

The statute requires the President to establish an administrative record. Under subpart I of the NCP, the administrative record file is the mechanism for compiling, and will contain, the administrative record required by section 113(k). One reason EPA adopted the concept of an administrative record file is that EPA felt that it may be confusing or misleading to refer to an ongoing compilation of documents as an "administrative record" until the compilation is complete. Until the response action has been selected, there The filt of a void creaking rempression that the record is complete at any time prior to the final selection decision, the set of documents is referred to as the administrative record the rather than the administrative record.

However this does not mean, as the comments appear to suggest, that the lead agency may edit the administrative record file in a manner that removes comments and rechnical data simply because they are not supportive of the final selection decision. Any comments and technical information placed in the record file for a proposed response action and relevant to the selection of that response action. whether in support of, or in opposition to, the selected response action, become part of the administrative record for the final response selection decision. Such materials will remain in the administrative record file, and will become part of the final administrative record. However. EPA believes that as a matter of law documents that are erroneously placed in the administrative record file (e.g., documents that have no relevance to the response selection or that pertain to an entirely different site) would not necessarily become part of the final administrative record.

EPA received additional comments stating that the administrative record file should be available before the beginning of the remedial investigation phase. These comments suggested that the file be available: When a site is entered into the CERCLIS data base: when the HRS score is calculated: when proposed for inclusion on the NPL: after the preliminary assessment report and after the remedial site investigation.

EPA believes that the point at which a site is entered into the CERCLIS data base is too early to put any information which would be relevant to a selection of a response action into a record file because at this point there has been no site evaluation and therefore little factual information about the site upon which to base a response decision. Interested parties can already find any information on a site that we id be included at the point of the HRS scoring and placement on the NPL in the NPL docket, which is publicly available. The preliminary assessment and remedial investigation stages of a response are premature for making the administrative cord available: at these points there is little information relevant to response selection on which to comment or to review. Once the RI/FS work plan is epproved, and the RI/FS study beginsincluding such activities as project

ACCO F2 13 3 CO 4C _F FAR ASSESSMET and analysis of alternatives— rere s a concrent body of site-specific information with relevance to the response selection upon which to comment. EPA believes that the beginning of the RL/FS phase is the point in the process when it makes sense to start a publicly available record of information relevant to the response selection.

One comment suggested that interested persons would have no chance to comment on the formation of the RI/FS work plan. The comment suggested that the record file should be available before the RI/FS work plan is approved, e.g., with a draft work plan or statement of work. EPA disagrees. Approved work plans are often amended. An interested person may comment on the scope or formation of the work plan, and such comments can be taken into account by the lead agency and incorporated into a final or amended work plan. Such comments must be considered if submitted during the comment period on the proposed ACTION.

Final rule: EPA is promulgating § 300.815(a) as proposed. -

Name: Section 300.815. Administrative record file for a remedial action. Section 300.820(a). Administrative record file for a removal action.

Proposed rule: Subpart I requires that the administrative record for a remedial action be available for public review when the remedial investigation begins. Thereafter, relevant documents are placed in the record as generated or received. The proposed regulations also require that the lead agency publish a aswepaper notice announcing the evaluability of the record files. and a second notice announcing that the proposed plan has been issued. A public comment period of at least 30 days is required on the proposed plan. Section 300.620(a) outlines the steps for the evailability of the record and public at for a non-time-critical remo action. IPA solicited comments on a proposal currently under consideration to require quarterly or semi-annual notification of record availability and the initiation of public comment in the Federal Register.

Response to comments: Some commenters suggested that the use of the Federal Register to announce the evailability of the administrative record is too costly or of little or no benefit. Several commenters requested clarification on how and when the lead agency should respond to comments. Another stated that lead agencies should be encouraged—though not

EPA chose not to require a notice availability of the administrative return in the Federal Register in this milemaking because it is still untient whether the benefits of this additional hotice butweigh its costs EPA may decide in the future to require this additional notice if it determines that such notice would improve hotification

EPA agrees with commenters that clanfication is needed as to when the lead agency should respond to comments. We also agree that the ead asency should be encouraged to respond to comments submitted before the public comment period. EPA generality will consider any timely comments containing significant information even if they are not received during the formal comment period, and encourages other lead agencies to do so. EPA w : strive to respond to comments it receives as early as possible. and ; encourage other lead agencies 'o 'o..ow sut However, any lead agency is required to consider and respond to on y those comments submitted during a formal comment period. Any other comments are considered at the lead agency's discretion. EPA has revised the runge of these sections to reflect the policy on consideration of public comments submitted prior to public comment periods.

One comment recommended that the regulations should provide how long the administrative record must be available. and suggested EPA coordinate efforts with the National Archives about retaining the record as a historical record. Another feit that materials were not always placed into the record in a timely manner, and that the record was not always available to the working public during evenings and weekends or accompanied by a copying machine. Similarly. one commenter feit that documents should be placed in the record when they are generated or in a senbed timeframe of two weeks. Another asked that free copies of key documents be included in the record.

EPA believes that the length of time a record must be available at or near the site will be dependent on site-specific considerations such as ongoing activity, pending litigation and community interest. EPA also believes that difficulties sometimes encountered by the working public require resolution on a site-by-site basis and do not ment a change in the proposed NCP language. Special provisions may have to be made by the records coordinator, with the aid of other site team members, including

regionalistie manager to ensure that the -scota location chosen is convenient to The public and that copying facilities are made available. Using public libraries to house the record should promote better availability of the record during apaworking hours and on weekends. In "response to mandating deadlines for lead agencies to place documents into the administrative record file. Agency guidance aiready directs record compilers to place documents into the record file as soon as they are received. Agency prucy additionally prescribes a suggested uneframe for placing documents in the record file. EPA believes that mandatory deadlines in the NCP would do little to increase the rate at which records are already compiled. The decision to place free copies of key documents in the record at or near the site will be a site-specific decision based on the level of community interest in these documents. Those who wish to make comes of key documents of any document contains in the administrative record file should aiready have access to copying facilities.

EPA received a comment requesting that it publish a joint nonce of evaluability of the administrative record with a nonce of evaluability of Technical Assistance Grants. Another comment stated that the removal site evaluation and engineering evaluation/cost analysis (EE/CA) must be included in the record for a non-time-critical removal action.

Publishing notice of the availability of the record in tandem with ennouncements of the availability of Technical Assistance Grants (TAGe) is a good idea where TAGs are available for a removal action. The TAGe. however. are generally designed to support citizen involvement in technical issues for sites undergoing remedial schons. The one-year. S2 million limitations on removals and the limited number of alternatives usually reviewed make further expense on a u advisor less beneficial than & might be for a long-term remedial action. As for placing the removal arts evaluation a sai EE/CA in the administrative reco rd. EPA agrees that generally suc documents would be part of the administrative record for the removal Action

Finally. EPA is making a minor change to the language of § 300.520(a)(4). EPA is substituting the term "decision document" in place of action memorandum to allow for situations where the agency's decision document ac: on memorandum

Final rule: 1. The second sentences of §§ 300.815(b). 300 820(a)(2) and 300.820(b)(2) are revised to reflect the new language on responding to comments as follows: "The lead agency is encouraged to consider and respond, as appropriate, to significant comments that were submitted prior to the public comment period."

2. In § 300.820(a)(4), the term "decision document" is substituted for "action memorandum."

3. The remainder of § 300.820(a) is promulgated as proposed.

Name: Section 300.820(b). Administrative record file for a removal action—time-critical and emergency.

Proposed rule: Section 300.620(b) outlines steps for public participation and administrative record availability for time-critical and emergency removal responses (S3 FR \$1516): "Documents included in the administrative record file shall be made available for public inspection no later than 60 days after initiation of on-site removal activity," at which point notification of the availability of the record must be published. The lead agency than, as appropriate, will provide a public comment period of not less than 30 days on the selection of the response action.

Response to comments: Several comments suggested that public comment requirements up § 300.820(b) were unnecessary and burdensome, especially the requirement to publish a notice of the availability of the record. One commant argued that requiring public notification of both cord availability and of a site's inclusion on the NPL was unnece and duplicative. Another comment stated that the requirements for public notification and public comment are not appropriate for all time-critical removal ctions, and recommended that the dministrative record be available for review only for those time-critical removal actions that do require public notice and comment. A related comme stated that the requirement to publish a notice of availability of the administrative record for all time-critical removal actions be eliminated in favor of making the record available but not requiring an advertisement or comment period, since some time-critical removal actions are completed before a public comment period could be held. Others asked that the public commant period become mandatory, or at least mandatory for removal activities not already completed at the time the reco is made svallable. Another comment requested that the record become

initiation of on-site remove activity because the current 60-day period prevented the consideration of any prework comments. A second comment supported the 60-day period. Finally, a commenter argued that it made little sense to make the record aveilable after 60 days for an emergency response because the on-scene coordinator (OSC) report containing most of the response information isn't required to be completed until one year following the response action.

in general, the public perticipation requirements under \$ 300.820(b) are designed to preserve both the flexibility and discretion required by the lead agency in time-critical removal action situations as well as EPA's commument to encouraging public perucipetion and to keeping an affected community wellinformed. EPA believes the notification and comment periods required in \$ 300.820(b) provide for both Agency flexibility and meaningful public stvolvement. The regulatory language stating that "The lead agency shall, as appropriate, provide a public comment period of not less than 30 days provides the lead agency needed flexibility when the emergency sature of circumstances makes holding a comment pened infeasible.

While EPA believes that it is necessary to announce the availability of the administrative record for nmecritical and emergency removal actions as well as non-time-critical actions, EPA believes that requiring establishment of the administrative record and publishing a notice of its availability 30 days after initiating a removal action in all cases. instead of "no later than 60 days after initiating a removal action." as proposed, would be somewhat nature. It has been EPA's expenses that it aften takes 60 days to stabilize a site (La., those activities that bein to reduce, retard or prevent the spread of a hezardous substance release and help to eliminate an immediate threat). EPA believes that the overriding task of nergency response teams during this critical period must be the undertaking of necessary stabilization, rather than administrative duties. Compiling and edvertising the record before a site has teven because stabilized would divert emergency response teams from devoting their full attention to a e. IPA believes that such administrative procedures are better left for after site stabilization.

Public sotice requirements for announcing the availability of the administrative record and for s site's inclusion on the NPL are not duplicative.

aren' Leo sions. Removal actions to "DI JIWEVS Take place at siles on the NPL therefore, the notice requirements are obviously not duplicative for these removal actions. For remedial sites that are on the NPL the administrative record need not be established for some time efter listing on the NPL so publishing a notice of the availability of the record would be essential to make ine affected public coenizant of site progress and their opportunity for review of documents included in the record.

Lastly the procedures specified in § 300 820(b) are applicable to an emergency removal that starts and finishes within 60 days. However, as provided in § 300.820(b)(2), a comment period :s held only where the lead agency deems it appropriate. But because the administrative record is an avenue for public information as well as for public comment. EPA also believes that even if the action is completed before the record file is made available. it is still appropriate to make the record available to the public. There is also no inherent contradiction in the OSC report being available one year after completion of the response action while the administrative record becomes available 60 days after initiation of onsite activities. Since the OSC report is a summary of the site events and is not a document which is considered in the selection of response action, it is not generally included in the administrative meand.

Final rule: EPA is promulgating \$ 300.820(b) as proposed, except that:

1. The second sentence of § 300.820(b)(2) is revised on responding. to public comments as described above.

2. Secura 300.820(b)(3) is revised consistent with § 300.820(a)(4); the term "action memorandum" is changed to "decision document."

Name: Section 300.825. Record requirements after decision document is

signed. Proposed rule: Section 308.686 describes situations where d ate. may be added to the add record after the decusion does عا اد signed. Documents may be added to a record in the following of When the document add set & perties of the decision which the decision document does not address or reserves for later; when the response acti changes and an explanation of significant differences or an amended decision document is issued: when the agency holds additional public com periods after the decision is signed; and when the agency receives comments

- . ton a nec e selvinere la ine recordi vin ch touid not have been submitted during The public comment period which substantially support the need to significantly alter the response action " (53 FR 51516). In addition, subpart E of the proposed NCP discusses RODamendments and Explanations of Significant Differences. Explanations of Significant Differences may be used for significant changes which do not fundamentally change the remedy, and do not require public comment, ROD amendments must be used for fundamental changes, and require a public comment period.

Response to commente: One commenter asked that subpart I reflect the factors consistently applied by courts when determining whether the record should be supplemented. including such criteria as Agency reliance on factors not included in the record. an incomplete record, and strong evidence that EPA engaged in improper benevior or acted in bad faith. A related comment stated that since general principles of administrative law apply to administrative record restrictions and supplementing the record, language limiting supplementing the record should be deleted from the NCP. EPA believes that including specific tenets of administrative law governing supplementing of the record in the NCP itself is unnecessary. These tenets apply to record review of response actions whether or not they are included in the NCP. The requirements of § 300.825(c) do not supplant principles on supplementing administrative records.

Another comment reco ended that EPA permit the record to be supplemented with any issue contested by a PRP, while granting an objective third party the ability to accept or reject record supplements. EPA already requires that any documents concernit remody selection submitted by PRPs within the public commont period be included in the record. All significant eening evidence submitted after the decision document is complete is already included in the record, so long as it meets the requirements of § 300.825(c), is not included elsewhere in the record. could not have been submitted during the public comment period, and supports the need to significantly alter the ponse action. IPA believes these 2 criteria ere ressonable and do not require the use of a third-party arbitrator.

One comment stated that all PRP submissions must be placed in the record in order to protect a party's due process right to be beard. EPA disagrees that all PRP submissions to the lead

-.. ·. inter plattiet ne par volue proments. The process provided in the F-.es--- netuding the tot ce of availability of the proposed plan 1-5 -e Administrative record for reviewe availability of all documents under . -a the response selection decision for review throughout the decision-max.-2 process, the opportunity to comment on "he proposed plan and all documents in the administrative record file, the requirement that the lead agency consider and respond to all significant PRP comments raised during the comment period, the notice of significant changes to the response selection, and the opportunity to submit, and requirement that the lead agency consider. any new significant information that may substantially support the need to significantly after the response selection even after the selection decision—is sufficient to satisfy due process. Moreover, the opportunity provided for PRP and public involvement in response selection exceeds the minimum public participation requirements set forth by the statute. Placing a reasonable limit on the length of time in which comments must be submitted, and providing for case-by-case acceptance of late comments through § 300.825(c). does no infringe upon procedural rights of PRPs.

• .

One commenter asked that the permussive "may" in § 300.825(e) be changed so there is as lead-agency discretion over whether to add to the administrative record documents submitted after the remedy selection. and stated that additional public comment periods as outlined in \$ 300.825(b) should not be only at EPA a option. A related comment stated that the multiple qualifiers in § 300.825(c). including the phrases "substantially" support the need" and "significantly alter the response action" (53 FR 51518). grant IPA overly broad discretionary powers over what documents may be added to the record. The commenter suggests deleting the word "substantially," as well as stating that all comments, even these disregarded by EPA. should be included in the record for the purpose of judicial review. EPA disagrees that the word "may" in either § 300.425(a) or § 300.425(b) is too permissive. Section 300.425(b) of the osal was sumply intended to clarify the lead agency's implicit authority to hold additional public comment periods. in addition to these required under subpart E for ROD amendmenta. whenever the lead agency decides it would be appropriate. Because these additional comment periods are not

Hermiss ver language simply reliects -- ead agency & discretion with respect to inese additional public involvement opportunities. Similarly, lead-egency discretion to add to the administrative record documents submitted after a decision document has been staned provides the lead agency the option to go beyond the minunum requirements for public perticipation outlined in the statute in response to requests to delete the qualifiers in § 300.825(c). this language is intentionally designed to define carefully ine circumstances in which EPA must consider commenta submitted after the response action has been selected. This standard recognizes CERCLA's manuale to proceed expeditious to implement selected response actions, but also recognizes that there will be certain instances in which significant new information warrants reconsideration of the selected response action. Section 300.825(c) is intended to provide a reasonable limit on what comments EPA must review or consider after a decision has been made.

Several commenters requested that PRPs not identified until after the close of the public comment period should be allowed an opportunity to comment on the record within 60 days of EPA's noufication of potential liability. EPA makes simulicant efforts to involve PRPs as early in the process as possible. When PRPs are identified late in the process. they may provide EPA with comments at that time. EPA will consider comments which are submitted after the decision document is sumed in accordance with the criteria of 1 300.825(c). This is true no matter wh the PRP is identified in the process. EPA believes that the current rule is sufficient for granting these late-Identified PRPs the opportunity for submitting late comments for the record.

One commenter stated that new information that confirms or substantiates pror public comment should be made part of the record, even after a ROD is signed. EPA is not required by statute or regulation to consider these comments, although a lead agency may, and frequently does, consider post-ROD comments it considers to be significant—in which case both the comment and the lead agency's response are part of the record.

Finally, EFA is making a minor change to § 300.225(b) on additional public comment periods to clarify that, in addition to comments and responses to comments, documents supporting the request for an additional comment period, and any decision documents would be placed in the administrative niended in the proposal a clart "cation is necessary to ensure consistency.

Final rule. EPA is promulgeing § 300 825 as proposed except for an addition to the last sentence of section (b) as follows: "All additional comments submitted during such comment periods that are responsive to the request, and any response to these comments, along with documents supporting the request and any final decision with respect to the usue, shall be placed in the administrative record file."

Subpart **j—Use of Dispersants and** Other Chemic**ais**

The following sections discuss comments received on subpart J and EPA's responses.

Name: Sections 300.900-300.920. General

Existing rule: Section 300.81 described the purpose and applicability of existing subpart H (now subpart /), and § 300.82 defines the key terms used in the regulation. Section 300.83 provides that EPA shall maintain a schedule of dispersants and other chemical or biological products that may be authorized for use on oil discharges called the "NCP Product Schedule."

Section 300.84 sets forth the . procedures by which an OSC may authorize the use of products listed on the NCP Product Schedule. The section provides that an OSC, with concurrence of the EPA representative to the RKT and the concurrence of the state(e) with jurisdiction over the navigable waters (as defined by the CWA) polluted by the oil discharge, may authorize the use of dispersant, surface collecting egenin, and biological additives listed on the NCP Product Schedule.

This section also provides that if the CSC determines that the use of a dispensant, surface collecting agent, or biological additive is necessary to prevent or substantially reduce a hexard to human life, and there is insufficient time to obtain the aceded concurrences, the OSC may unilaterally sutherize the use of any product, including a product not on the NCP Product Schedule. In such instances, the OSC must inform the EPA RRT representative and the affected states of the use of a product as soon as possible and must obtain their concurrence for the continued use of the product once the threat to human life has subsided. This provision eliminates delays in potentially life-threatening situations, such as spills of highly flammable petroleum products in harbors or near inhabited artees. Although they will not be listed on the Schedula, this section also provides for stents on a case-by-case casis. The use of sinking agents is pronibiled.

Section 300.84 explicitly encourages advance planning for the use of dispersants and other chemicals. The OSC is authorized to approve the use of dispersants and other chemicals without the concurrence of the EPA representative to the RRT and the affected states if these parties have previously approved a plan identifying the products that may be used and the particular circumstances under which their use is presuthorized.

Section 300.85 details the data that must be submitted before a dispersant, surface collecting agent, or biological additive may be placed on the NCP Product Schedule. Section 300.86 describes the procedures for placing a product on the Product Schedule and also sets forth requirements designed to avoid possible misrepresentation or misinterpretation of the meaning of the placement of a product on the Schedule, including the wording of a disclaimer to be used in product advartisements or technical literature referring to placement on the Product Schedule.

Appendix C details the methods and types of apparatus to be used in carrying out the revised standard dispersant effectiveness and equatic toxicity tests. Appendix C also sets forth the format required for summary presentation of product test data.

Proposed rule: Proposed subpart J is very similar to subpart H and contains only minor revisions. Section numbers and references to other sections and subparts have been changed where appropriate. Technical changes and minor wording changes to improve clarity have also been made.

Definitions formerly presented in subpart H have been moved to subpart A, and a new definition has been added for misosilaneous all spill control agents. Accordingly, a list of data requirements for miscellaneous spill control agents is proposed to be added to § 300.918. The definition for navigable waters is as defined in 40 CFR 110.1.

Section 300.010, which addressed "Authorization of use," was modified slightly in the proposed regulation to emphasize the importance of obtaining concurrence for the use of dispersants and other chemicals from the appropriate state representatives to the Regional Response Team (RRT) and the DOC/DOI natural resource trustees "as appropriate."

Response to comments-1. Involvement of DOC/DOI trustees. Many commenters opposed the inclusion of the DOC/DOI trustees in

APPENDIX M

SUBPART I OF NCP

Subpart I-Administrative Record for Selection of Response Action

§ 300.800 Establishment of an administrative record.

'a) General requirement. The lead agency spani establish an administrative record that contains the documents that form the basis for the selection of a response action. The lead agency shall compile and maintain the subdivistrative record in accordance with this subpart.

(b) Administrative records for federal facilities. (1) If a federal agency other than EPA is the lead agency for a federal facility, the federal agency shall compile and mentals the administrative record for the selection of the response action for that facility is accordance with this support. EPA may formsh documents which the federal agency shall place in the administrative record includes all documents that form the basis for the selection of the response action.

(2) EPA or the U.S. Coast Guard shall compile and meantain the administrative record when it is the lead agreety for a federal factity.

(3) if EPA is involved in the selection of the response action at a faderal facility on the NPL, the faderal agencyacting as the land agency shall provide EPA with a copy of the states of

cionuments included in the administrative recert file, the RI/PS workplan, the RI/PS released for public comments the proposed plan, any public comments received on the RI/PS and proposed plan, and any other documents EPA may request on a case-by-case builds.

ic: idministrative acord for stateead stres. If a state is the lead agency for a site, the state mail compile and זה החשר שעת בשנותוהנסי שלי מובזתובה the selection of the response action for that site in accordance with this SUDDARL EPA may require the state to place additional documents .n the administrenve record file to ensure that the edministrative record includes all documents which form the basis for the selection of the response action. The state shall provide EPA with a copy of the index of documents included in the edministrative record file, the RI/PS workplan, the RI/FS released for pupile comment the proposed plan any public comments received on the RI/FS and proposed plan, and any other documents EPA may request on a case-oy-case bame.

(d) Applicability. This subpart applies to all response actions taken under section 106 of CERCLA or sought, secured, or ordered administratively or judicially under section 108 of CERCLA, as follows:

(1) Remedial actions where the remedial investigation commences after the promulgation of these regulations: and

(2) Removal actions where the action memorandum is sugged after the promugation of these regulations.

(e) For these response actions not included in paragraph (d) of this section, the lead agency shall comply with this subpart to the extent practicable.

§ 300.005 Location of the administrative reserve file.

(a) The load agency shall establish a docket at an office of the load agency or other control location at which dominants included in the editionstive restort file shall be located and a copy of the documents included in the administrative record file shall also be made available for public surportion at or near the site at locate an appreciate balance

(1) Sampling and using data quality crowel and quality accurace documentation, and claim of caredy forms, need not be located at or user the sin at issue or at the canted location, provided that the index to the administrative record file indicates the location and evaluability of this information, 3. P.Duc y available technical versione not generated for the site at usue, such as engineering textbooks, articles from technical journals, and toxicological profiles, need not be located at or near the site at issue or at the central location, provided that the hiersture is listed in the index to the administrative record file or the literature is cited in a document in the record.

4. Documents included in the confidential portion of the administrative record file shall be located only in the central location.

(5) The administrative record for a removal action where the release or threat of release requires that on-site removal activities be initiated within hours of the lead agency's determination that a removal is appropriate and onsite removal activities cease within 30 days of initiation, need be available for public inspection only at the central location.

(b) Where documents are placed in the central location but not in the file located at or near the site, such documents shall be added to the file located at or near the site upon request, except for documents included in paragraph (a)(4) of this section.

(c) The lead agency may make the administrative record file available to the public in microform.

§ 300.810 Contents of the administrative record file.

(a) Contents. The administrative record file for selection of a response action typically, but not in all cases, will contain the following types of documents:

(1) Documents containing factual information, data and analysis of the factual information, and data that may form a basis for the selection of a response action. Such documents may include verified sempling data. quality control and quality assurance documentation, chain of custody forms, site inspection reports, preliminary assessment and site evaluation reports. ATSDR health assessments, documents supporting the lead againty's determination of im it and ablic besith nt. pub ni and substantial enden eveluations, and to

**- evaluations. In addition, for uons, such documents may roved workplans for the Such documents may include guidance on conducting remedial investigations and feasibility studies, guidance on determining applicable or relevant and appropriate requirements, guidance on nsk/exposure easessments, engineering handbooks, articles from technical journals, memorands on the application of a specific regulation to a site, and memorands on off-site disposel

capacity: (3) Documents received, published, or made available to the public under § 300.815 for remedial actions, or § 300.820 for removal actions. Such documents may include nonce of availability of the administrative record file, community relations pian, iroposed plan for remedial action, nonces of public comment periods, public comments and information received by the lased agency, and responses to similicant comments

(4) Decision documents. Such documents may include action memoranda and records of decision:

(5) Enforcement orders. Such . documents may include administrative orders and consent decrees; and

(6) An index of the documents included in the edministrative record file. If documents are customarily grouped together, as with sampling data chain of custody documenta, they may be listed as a group in the index to the administrative record file.

(b) Documents not included in the administrative record file. The lead agency is not required to include documents in the administrative record file which do not form a basis for the selection of the response action. Such documents include but are not limited to draft documents, internal memoranda, and day-to-day notes of staff unless such documents contain information that forms the basis of selection of the response action and the information is not included in any other document in the administrative record file.

(c) Privileged documents. Privileged documents shall not be included in the record file except as provided in paragraph (d) of this section or where such privilege is waived. Privileged documents include but are not limited to documents subject to the attorney-client, attorney work product, deliberative process, or other applicable privilege.

process, or other applicable privilege. (d) Confidential file. If information which forms the basis for the selection

102

SELL DE DIACEG IN THE DIDIC - A -A -A -A portion of the administrative record The The confidential or privileged document itself shall be placed in the confidential portion of the administrative record file if information, such as confidential business information, cannot be summerized in a disclosable manner the information shall be placed only in the confidential portion of the administrative record file. All documents contained in the confident al portion of the administrative record file shall be listed in the index to the file.

§ 300.815 Administrative record file for a remedial action.

(a) The administrative record file for the selection of a remedial action shall be made svailable for public inspection at the commencement of the remedial investigation phase. At such time, the lead agency shall publish in a major local newspaper of general circulation a notice of the availability of the administrative record file.

(b) The lead agency shall provide a public comment period as epecified in § 300.430(f)(3) so that interested persons may submit comments on the selection of the remedial action for inclusion in the administrative record file. The lead agency is encouraged to consider and respond as appropriate to significant comments that were submitted prior to the public comment period. A written response to significant comments submitted during the public comment period shall be included in the administrative record file.

(c) The load agency shall comply with the public participation procedures required in § 300.430(f)(3) and shall document such compliance in the administrative record.

(d) Documents generated or received after the record of decision is signed shall be added to the administrative record file only as provided in § 300.825.

§ 300.030 Administrative record file for a removal action.

(a) II based on the site evaluation, the lead agency determines that a removal action is appropriate and that a planning period of at least six months exists before on-site removal activities must be initiated:

(1) The administrative record file shall be made available for public inspection when the engineering evaluation/cost anaivsis EE/CA) is made available for public comment. At such time, the lead agency shall publish in a major local newspaper of general circulation a notice of the availability of the administrative record file.

12: The ead agency shall provide a public comment period as specified in § 200 415 so that interested persons may submit comments on the selection of the removal action for inclusion in the administrative record file. The lead agency is encouraged to consider and respond, as appropriate, to significant comments that were submitted prior to the public comment period. A written response to significant comments agency the public comment period shall be included in the administrative record file.

(3) The lead agency shall comply with the public participation procedures of § 300.415(m) and shall document compliance with § 300.415(m)(3)(i) through (iii) in the administrative record file.

(4) Documents generated or received after the dacision document is signed shall be added to the administrative record file only as provided in § 300.825.

(b) For all removal actions not included in paragraph (a) of this section:

(1) Documents included in the administrative record file shall be made available for public inspection no later than 60 days after initiation of on-site removal activity. At such time, the lead agency shall publish in a major local newspaper of general circulation a notice of aveilability of the administrative record file.

(2) The lead agency shall, as appropriate, provide a public comment period of not less than 30 days beginning at the time the administrative record file is made available to the public. The lead egency is encouraged to consider and respond, as appropriate, to significant comments that were submitted price to the public comment period. A written response to significant comments submitted during the public comments period shall be included in the administrative record file.

(3) Documents generated or received after the decision document is signed shall be added to the edministrative record file only as provided in § 300.825.

§ 206.825 Record requirements after the desisten document is signed.

(a) The lead agency may add documents to the administrative record file after the decision document selecting the response action has been signed if:

(1) The documents concern a portion of a response action decision that the

.

decision document does not address or reserves to be decided at a later date: or

(2) An explanation of significant differences required by § 300 433(c), or an amended decision document is issued, in which case, the explanation of significant differences or amended decision document and all documents that form the basis for the decision to modify the response action shall be added to the administrative record file.

(b) The lead agency may hold additional public comment periods or extend the time for the submission of public comment after a decision document has been signed on any issues concerning selection of the response action. Such comment shall be limited to the issues for which the lead agency has requested additional comment. All additional comments submitted during such comment periods that are responsive to the request, and any response to these comments, along with documents supporting the request and any final decision with respect to the issue, shall be placed in the administrative record file.

(c) The lead agency is required to consider comments submitted by interested persons after the close of the public comment period only to the extent that the comments contains significant information not contained elsewhere in the administrative record file which could not have been submitted during the public comment period and which substantially support the need to significantly alter the response action. All such comments and any responses thereto shall be placed in the administrative record file.

Subpart J—Use of Dispersants and Other Chemicals

| JULIU General

(a) Section 312(c)(2)(G) of the Clean Water Act requires that EPA prepare a schedule of dispersants and other chemicals, if any, that may be used in carrying out the NCP. This subpart makes provisions for such a schedule.

(b) This subpart applies to the navigable waters of the United States and adjoining shorelines, the waters of the contiguous zone, and the high seas beyond the contiguous zone in connection with activities under the Outer Continental Shelf Lands Act. activities under the Deepwater Port Act of 1874, or activities that may affect natural resources belonging to. appertaining to, or under the exclusive management authority of the United States, including resources under the Magnuson Fishery Conservation and Management Act of 1878. C: This suppart applies of the stores of the suppart applies of the store store and as defined in suppart A of this park has may be used to remove or control of discharges.

300.905 NCP Product Schedule.

(a) Oil Discharges (1' EPA sna.. maintain a schedule of dispersants and other chemical or biological products that may be authorized for use on oil discharges in accordance with the procedures set forth in § 100 910. This schedule, called the NCP Product Schedule, may be obtained from the Emergency Response Division (OS-200) U.S. Environmental Protection Agenci-Washington, DC 20460. The telephone number is 1-202-382-2190.

(2) Products may be added to the NCP Product Schedule by the process specified in § 300.920.

(b) Hazardous Substance Releases [Reserved].

§ 300.910 Authorization of use.

(a) The OSC, with the concurrence of the EPA representative to the RRT and. as appropriate, the concurrence of the RRT representatives from the states with jurisdiction over the navigable weters threatened by the release or discharge, and in consultation with the DOC and DOL natural resource trastees. when precucable, may authorize the use of dispersents. surface collecting agents. biological additives. or miscellaneous oil spill control agents on the oil discharge. provided that the dispersants, surface collecting agents, biological additives, or miscellaneous oil spill control agents are listed on the NCP Product Schedule.

(b) The OSC, with the concurrence of the EPA representative to the RRT and, as appropriate, the concurrence of the RRT representatives from the states with jurisdiction over the navigable waters threatened by the release or discharge, and in consultation with the DOC and DOI natural resource trustees, when precticable, may authorize the use of burning agents on a case-by-case basis.

(c) The OSC may authorize the use of any dispersant, surface collecting agent, other chemical agent, burning agent, biological additive, or miscellaneous of spill control agent, including products not listed on the NCP Product Schedule, without obtaining the concurrence of the EPA representatives to the RRT, the RRT representatives from the states with jurnediction over the navigable waters threatened by the release or discharge, when, is the judgment of the OSC, the use of the product is necessary to prevent or substantially reduce a hazard to human life. The OSC is to inform the

Furthering, the Use of Innovative Treatment Technologies in OSWER Programs



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460

JUN 1 0 1991

OFFICE OF SOLID WASTE AND EMERGENCY RESPONSE

> OSWER Directive 9380.0-17

MEMORANDUM

SUBJECT :	Furthering the Use of Innovative Treatment Technologies in OSWER Programs
FROM:	Don R. Clay Assistant Administrator
TO:	Director, Waste Management Division, Regions I, IV, V, VII, and VIII Director, Environmental Services Division, Regions I, VI, and VII Director, Emergency and Remedial Response Division, Region II
	Director, Hazardous Waste Management Division, Regions III, VI and IX Director, Hazardous Waste Division, Region X Director, Water Management Division, Regions IV and X

I want to exercise further leadership in the use of innovative technologies--by creating additional incentives for affected groups such as potentially responsible parties, facility owners/operators, consulting engineers, technology vendors and the public and by using tools currently at our disposal. EPA and responsible parties or facility owners/operators, should be exploring and promoting more effective and less costly technologies to solve the considerable problems we face. Consulting engineers and new technology vendors are essential partners in this process as well.

While I believe our clean-up partners can and will promote the implementation of innovative technology, we need to inject a sense of responsible urgency to prevent the expenditure of dollars in pursuing less effective or more costly remedies. We have made some important progress to date, and now is the time to broaden our efforts and expand into additional program areas. Furthermore, we have a responsibility to provide technological leadership to the other major environmental clean-up programs society will be pursuing beyond those administered by OSWER. This leadership will not only improve the quality and efficiency of cleanups, but will also help make U.S. firms leaders in the international marketplace for waste treatment and site remediation.



Each of the affected groups sees some risk tied to an effort to "push on the envelope" of technology application. However, these risks are directly related to potential benefits -- both short-term at a particular site and long-term benefits which will accrue from knowledge gained by our experiences. Only if some of us are willing to work constructively with our uncertainty is there reason to expect significant progress toward more applications of technologies that are truly innovative.

I understand innovation requires a sense of creativity and may be accompanied by false starts, second attempts, intensively reengineered solutions, and (despite best efforts) some equipment failures. I recognize that while most will agree with the need for new and better approaches, the inherent risks associated with early technology use serve as very serious impediments. The extensive review and criticism of our programs from both outside and inside the Agency may have tended to make us averse to unnecessary risks. It should be recognized that however welldesigned and carefully planned our efforts may be, they may not meet contract specifications on many first attempts and may need refinement before routine application can be expected. Indeed, information gained from a first-time application that fails to perform as designed may be viewed as a form of success.

In addition, this definition of innovation needs to be recognized by EPA regional and headquarters managers. Remedial Project Managers (RPMs) and On-Scene Coordinators (OSCs) must have support from their managers if an innovative technology does not work as expected. The program should recognize and assume the risks inherent in using new technologies. The challenges these projects present will usually require great efforts from our most competent and experienced RPMs and OSCs. They should view these challenges as career opportunities rather than as career risks.

Innovative treatment technologies should be routinely considered as an option in engineering studies where treatment is appropriate. They should not be eliminated from consideration solely because of uncertainties in their performance and cost. These technologies may be found to be cost-effective, despite the fact the their costs are greater than conventional options, after consideration of potential benefits which could include increased protection, superior performance, and greater community acceptance. In addition, future sites will benefit by information gained from the field experience.

The attached directive is designed to increase field applications of innovative technologies for cleaning up contaminated sites. It also encourages expanded application of existing OSWER policies and emphasizes the value of existing, support activities in this area. It is intended to sharpen the focus and level of attention by EPA staff and managers on their mission to provide technological leadership by implementing existing authorities under the Superfund, Resource Conservation and Recovery Act (RCRA), Underground Storage Tank (UST), and Oil Pollution Act programs. Furthermore, this guidance is intended to integrate the continual search for improved remedies with the use of new technologies and to make this objective a permanent feature of EPA's clean-up programs. It is intended to create an atmosphere which recognizes that reasonable risk-taking, which is protective of human health and the environment, is necessary to achieve this end.

The statement consists of seven major initiatives. The first four initiatives concern the Superfund program. The first one addresses some impediments to the full-scale use of new equipment funding of remedial encourages expedited design and and construction projects. This initiative also provides contract flexibility in the start-up phase of selected remedial and removal actions to assist vendors in establishing a pattern of reliable operation in order to satisfy contract performance standards. The second initiative is intended to ensure that innovative alternatives are thoroughly evaluated for PRP-lead sites that are early in the planning process. This provision encourages EPA regions to fund treatability studies and engineering analyses for promising treatment technologies that might otherwise be considered unproven by the PRPs and too early in the development process. The third initiative provides a capability to rapidly evaluate the efficacy of a PRP-proposed innovative remedy that is offered in addition to the primary one approved in the Record of Decision (ROD). This provision entails direct technical support to evaluate innovative remedies, while moving the remediation process forward. The fourth initiative seeks to utilize the potential of the removal program for expanding our experience with the field application of new technologies. The directive clarifies OSWER's position that the removal program is an important and viable means for furthering the use of these treatment alternatives.

Another provision in the guidance is designed to encourage studies on the potential use of new technologies for RCRA corrective action. Regions should consider promoting the pilot testing of promising innovative technologies at a limited number of sites. In the past, land ban considerations have sometimes discouraged owners/operators or regions from pursuing such approaches. This guidance encourages the use of soil and debris treatability variances, where necessary, to allow innovative technology studies to proceed. This authority was recently delegated to the regions.

The sixth initiative recognizes unique opportunities presented by Federal facilities. We are exploring the potential use of these facilities for developing and applying new technologies, and regional offices are encouraged to work with Federal facility managers to further this objective.

The final provision encourages expanded use of the Federal Technology Transfer Act as an opportunity for joint technology assessments with industry. PRPs and owners/operators may sign cooperative agreements with EPA for services to support innovative
technology treatability or pilot studies. This procedure offers the prospect of non-adversarial engagement, outside the regulatory context, to allow the development of third-party data on remediation technologies.

I know there is a tension created by the desire to promote new technology developments within existing management tracking systems and program commitments and goals. I recognize that these goals may also be statutory in origin. Issues are certain to arise concerning the selection and use of new treatment technologies because of the rapid pace of development in this area. These issues cannot be resolved by this guidance and must be addressed through common sense and judgement on a case-by-case basis. There may be circumstances where program goals and commitments must be adjusted in order to achieve better clean-up solutions.

Although not specifically discussed in the attached guidance, EPA is also strongly committed to using innovative technologies in cleaning up oil spills under the Oil Pollution Act. We have embarked on an aggressive research program with other Federal agencies and the private sector to examine clean-up technologies and remediation techniques. We anticipate this work will lead to new and improved technologies in this area as well.

This directive is a call for your attention to exploring and exploiting opportunities for using innovative remediation technologies. It reflects my personal commitment and belief that we must invest the necessary resources and take the risks now to develop the technologies necessary to fulfill the long-term needs of our hazardous waste clean-up programs.

GUIDANCE FOR INCREASING THE APPLICATION OF INNOVATIVE TREATMENT TECHNOLOGIES FOR CONTAMINATED SOIL AND GROUND WATER

INTRODUCTION

The Office of Solid Waste and Emergency Response (OSWER) is seeking to further the use of innovative treatment technologies in order to (1) better pursue its statutory and regulatory mandates to promote treatment to the maximum extent practicable, (2) speed the availability of performance data regarding newly developed treatment technologies to many constituencies facing mandates to clean contaminated sites, (3) broaden the inventory of accepted treatment-based solutions, and (4) increase the likelihood that remediation costs can be lowered in the near term through the demonstration of a larger number of engineering options to solve site remediation problems.

Both SARA and HSWA give us the framework to consider treatment as an essential element in our clean-up decisionmaking. Our record of accomplishment since SARA in selecting treatment technologies for Superfund remedial and removal projects is very good. However, our experience in implementing remedies is limited, and we face a large future obligation to cleanup sites in the RCRA and UST programs. For example, the large number of cleanups expected under the RCRA corrective action program may encompass up to 4,000 facilities and 64,000 waste management units.

Section 121(b) of CERCLA requires EPA to select remedies that "utilize permanent solutions and alternative treatment technologies or resource recovery technologies to the maximum extent practicable" and to prefer remedial actions in which treatment "permanently and significantly reduces the volume, toxicity, or mobility of hazardous substances, pollutants, and contaminants as a principal element." This objective of permanent treatment-based remedies should be applied to RCRA and UST cleanups, within their respective legislative contexts. To achieve this goal, EPA must encourage new or innovative treatment technologies that are capable of treating contaminated soils/sludges and ground water more effectively, less expensively, and in a manner more acceptable to the public than existing conventional methods.

Innovative treatment technologies are newly developed technologies whose lack of sufficient full-scale application blocks routine consideration for site remediation. They may be new technologies, or may be available and in use for various industrial applications other than hazardous waste remediation. As such, innovative technologies are not part of standard engineering practice or the competitive market process where available alternatives are routinely presented to the government and private sector. In functional terms, we define as "innovative" those treatment technologies for source control <u>other than</u> incineration and solidification/stabilization and pumping with conventional treatment for ground water. Innovative technologies inherently require extra effort to gather information and analyze options and extra engineering and financial risk in adapting them for specific site applications. In addition, there is extra uncertainty for people developing such solutions who work in organizations focused on performance outcomes with high levels of certainty and known costs.

Existing directives and guidance contain a number of that encourage the consideration of innovative references Policy for the Superfund program was originally technologies. outlined in a February 21, 1989 memorandum on "Advancing the Use of Treatment Technologies for Superfund Remedies." This memorandum reaffirmed the use of treatment technologies and summarized quidance documents and activities that supported the use of innovative technologies. It cited the need to search for new technologies that can improve performance and reduce cost. The importance of innovative technologies was further emphasized in the Superfund Management Review (90-Day Study) which primarily contained recommendations concerning technical support and More recently, the National Contingency Plan expects research. that treatment will be used for highly toxic and highly mobile waste and encourages the consideration of innovative methods.

As a result of SARA and this guidance, the selection of innovative technologies in the remedial program has increased dramatically. For the last three fiscal years, almost half of the selected treatment technologies for source control have been innovative. However, few full-scale innovative remedies have actually been implemented. As a result, we are not benefiting from actual clean-up experience or developing the equipment necessary to fulfill long-term program needs. This directive seeks to preserve our momentum with the selection of these technologies, to expedite their use in remedial actions, to expand the application of new technologies to other OSWER programs, and to realize the potential for development and technology application at Federal facilities.

This directive sets forth several initiatives and new procedures that will help provide incentives for broader use of innovative technology. Some of these initiatives are directed toward potential responsible parties and owners/operators, since they will be assuming a larger share of the remedial projects in the future. Other new initiatives are intended to remove impediments to the first-time use of new equipment. The directive, also encourages wider application of available resources and tools. In addition, Attachment A highlights some important ongoing program efforts that deserve mentioning.

STATEMENT OF INTENT

•

Innovative treatment technologies are to be routinely considered as an option in feasibility studies for remedial sites and engineering evaluations for removals in the Superfund program, where treatment is appropriate commensurate with the National Contingency Plan (NCP) expectations. In addition, innovative treatment-based remedies should be pursued to the extent practicable for cleanup of RCRA and UST units that pose significant health and environmental threats similar to those at Superfund sites. EPA should exercise leadership with state UST programs to encourage similar approaches for underground tanks. Innovative technologies considered in the remedy selection process for Superfund, RCRA, and UST should <u>not</u> be eliminated solely on the grounds that an absence of full-scale experience or treatability study data makes their operational performance and cost less certain than other forms of remediation.

When assessing innovative technologies, it is important to fully account for their benefits. Despite the fact that their costs may be greater than conventional options, innovative technologies may be found to be cost-effective, after accounting for such factors as increased protection, superior performance, and greater community acceptance. In addition, experience gained from the application of these solutions will help realize their potential benefits at other sites with similar contaminants.

NEW INITIATIVES

This directive prescribes six new initiatives affecting Superfund and RCRA programs to encourage and further enable the field application of innovative technologies and their evaluation for potential further use. It also affirms the use of a relatively little-used opportunity for joint EPA work with PRPs and owners/operators to evaluate new technologies.

1. Superfund Innovative Technology Start-Up Initiative.

Designed for Fund-lead projects, this initiative consists of two efforts to assist the early application of new technology. First, we need to encourage the expedited funding of remedial design and construction projects that involve innovative treatment technologies. OERR will be revising its Remedial Action funding priority-setting procedures to give more consideration to innovative technologies. Earlier funding of these projects will help achieve the technology development goals of the Superfund program and will provide EPA with significant data to support future Records of Decisions (RODs).

Second, this initiative provides contract flexibility in the start-up phase of selected remedial and removal actions to assist vendors in establishing a pattern of reliable operation that satisfies performance standards. This is intended to address some of the impediments to the use of new full-scale equipment; it will support initial start-up and shake-down costs and modifications necessary to effectively evaluate whether the selected technology can perform to specifications prior to beginning actual remediation. In the remedial program, the Corps of Engineers (COE) will provide separate contract provisions that will aid in the commencement of operations of a unit process or integrated set of processes and will be available only for some proportion of the whole site remedy (e.g., processing the first 1,000 cu. yds. of a 30,000 cu. yd. site). Funds are not targeted at making the technology work at any cost, but to aid in clearly establishing the likely performance adequacy of the technology prior to the onset of the contracted clean-up effort. Contracting strategies are being considered to compensate vendors, regardless of whether they successfully achieve performance limits. Further implementation guidance for the remedial and removal programs will be issued later this year.

2. Dual Track RI/FS Initiative (Superfund)

This initiative is designed for PRP-lead sites that are early in the planning process where there is an opportunity to conduct engineering evaluations of remedies through the RI/FS This initiative is intended to ensure that process. innovative technologies are thoroughly evaluated and that needed treatability studies are conducted for potential This provision should help encourage EPA to take remedies. risks (when faced with reluctant PRPs) that it would not otherwise take by encouraging a comprehensive evaluation of technologies. EPA regions may fund additional treatability studies and engineering analyses for promising treatment technologies that would otherwise be considered unproven and too early in the development process. The purpose of this initiative is to encourage treatability studies to ensure that alternative remedies that the government believes may have merit are thoroughly evaluated and considered in the ROD. Data from EPA treatability studies and the evaluation of additional innovative technologies have intrinsic value to the Agency. Therefore, even if, in a particular case, there may be some doubt as to EPA's ability to cost recover for these additional studies (although, in general, the Agency would expect such costs to be subject to cost recovery), these studies should be pursued based on their value to the overall program.

3. Tandem ROD Evaluation Initiative (Superfund)

As in the previous initiative, this provision is primarily designed for PRP-lead sites, although it may also be applicable for some Fund-financed situations. This program will provide a capability to rapidly evaluate the efficacy of a PRP-proposed innovative remedy that is offered in tandem with the primary one approved in the ROD. Both of the remedies would be part of the proposed plan. Typically, such an alternate solution would be approved on a contingent basis in the ROD based on acceptable treatability studies, but it would need further development and pilot testing during the design period for the primary technology. Tandem RODs (or contingent RODS based on formal evaluation) are a decision vehicle designed to move the process of cleanup toward expeditious closure, while leaving room for PRPs with a decided interest in innovative technologies to pursue additional pilot tests to demonstrate an alternate approach that is both innovative and potentially cost-effective. This program is based on direct technical support for regional project management teams to help resolve technical issues posed by alternate approaches; it is designed to lift the burden from the regional project manager of bearing the risks of evaluating and trying something "new."

Technical support will be provided for focused evaluation of the PRP work so as to support expedient regional decisions about the acceptability of the alternate technology. The work will be carried out with and through the appropriate OSWER/ORD Technical Support Centers or the SITE demonstration program and will be conducted as a mini-evaluation of the proposed alternative so that the data will be available for future applications. When considering whether to proceed with a tandem ROD, regions should first consult with ORD concerning the scope of effort required for the evaluation.

In the case in which the secondary innovative technology is chosen for implementation (after the completion of pilot testing) but significant delays to the original schedule have occurred, the region may consider the engineering problems of making the full-scale unit operational in assessing stipulated penalties. That is, in limited cases, stipulated penalties should not be imposed if the delays are the unavoidable result of being innovative.

4. Removal Program Initiative (Superfund)

The removal program represents an important and viable means for expediting the field application of innovative technologies. The relatively small volumes frequently requiring response and streamlined contracting procedures provide an opportunity to complete clean-up projects and provide documentation on lessons learned relatively quickly. Smaller waste volumes at some sites may also allow the use of pilot-scale technologies under some circumstances.

Although there have been more innovative projects actually constructed through the removal program than the remedial program, its potential has not been fully realized. This is because time constraints often favor excavation and off-site disposal or treatment and also because of the absence of clear legislated goals regarding the use of new technology. This subject was one of the issues addressed in a 1988 audit report by the Inspector General of Region IV removal sites. The report has had the undesirable effect of discouraging OSCs from using these technologies.

This directive is meant to clarify EPA's position on this issue. It is OSWER policy to further the use of innovative technologies through the removal program. This includes all actions, including time-critical actions, where feasible. These projects are expected to fulfill an important role in adding to our knowledge base on promising new technologies. Further guidance will be included in an upcoming document, "Administrative Guidance for Removal Program Use of Alternatives to Land Disposal" (OSWER Directive 9380.2-1), which provides guidelines promoting the use of alternatives to land disposal.

5. RCRA Corrective Action and Closure Innovative Technology Initiative

We are currently engaged in efforts to develop best demonstrated available technology (BDAT) treatment standards for contaminated soil and debris at CERCLA and RCRA corrective action and closure sites. These sites present unique treatment problems that were not generally considered in developing the current BDAT standards, which were based on data from the treatment of industrial process wastes. There is general agreement that wide scale use of incineration is not appropriate for soil and debris, and there is a need to explore alternative approaches. The current schedule is to promulgate a rule for debris in May 1992 and soil in April Prior to publication of these final rules, a site-1993. specific treatability variance process (40 CFR 268.44 (h)) is available for contaminated soil and debris to establish an alternative standard for specified waste at individual sites. The variance process, along with applicable guidance treatment levels, is described in Superfund LDR Guide #6A (OSWER Directive: 9347.3-06FS, July 1989), and is intended to be used as an interim approach until final standards are established.

This initiative encourages the regions to use treatability variances at corrective action and closure sites

to conduct treatability or technology demonstration studies to gain additional information on the use of innovative treatment for contaminated soil and debris. The regions should select appropriate pilot-scale projects with cooperative owners/operators that can provide data on the capability of technologies and the treatability of different wastes. The information from this work should help to expedite corrective action and closures after the final BDAT rule is published for soils. It is also possible that early data from this effort could be available for consideration in the final rule.

Projects should be carefully selected to maximize the utility of data and likelihood of success. Regional corrective action staff and regional Superfund staff should communicate regarding the history of use of treatability variances in the Superfund program to identify site factors that require consideration when selecting an appropriate site.

Authority for issuing site-specific variances for contaminated soil and debris has recently been delegated to the regions (Decision Memorandum: "Delegation of Authority to Grant Treatability Variances," from Charles L. Grizzle to the Administrator, April 12, 1991). The facility and EPA, in collaboration with the state, can implement variances for onsite demonstrations through two mechanisms: temporary authorization under the Permit Modification Rule, or 3008(h) orders for interim-status facilities.

 Demonstration Projects at Federal Facilities (Superfund, RCRA, and UST)

Federal facilities offer unique opportunities for both developing and applying innovative approaches to hazardous waste remediation. Desirable attributes include their often sizable areas and isolated locations, controlled access, numerous contamination problems, and increasingly active environmental restoration programs.

EPA headquarters is exploring the use of Federal facilities for both site-specific technology demonstrations and as test locations for evaluation of more widely applicable technologies. Equally important is the establishment of mechanisms to ensure timely sharing of information. Regions are encouraged to suggest innovative approaches <u>and</u> to be receptive to proposals for innovation from Federal facility managers, e.g., by building timing and performance flexibility into compliance agreements in acknowledgment of current uncertainties associated with innovation.

The Office of Federal Facilities Enforcement (OFFE) will work with the regions to identify locations for sponsoring potential test and evaluation activities. With assistance from the Technology Innovation Office, OFFE will develop necessary policies and guidance to ensure that support for innovation is congruent with other program and environmental objectives.

7. Joint Technology Assessment Opportunities with Industry under the Federal Technology Transfer Act

During the clean-up planning and implementation process, PRPs or owners/operators should be reminded of the opportunity to engage EPA in evaluation studies and other arrangements at their expense to determine whether innovative technology concepts would be operative in the situation they are facing or other similar situations. Under the Federal Technology Transfer Act (FTTA) of 1986 and Executive Order 12591, cooperative agreements related to research, development, and technology transfer can be expeditiously executed (i.e., in less than 60 days) between industry and government. In this case, such arrangements would allow the PRP to reimburse EPA for facilities, support services, and staff time spent in joint evaluation of early technology treatability or pilot studies. As projects progress into the later planning stages, careful judgement needs to be exercised to avoid new work that will result in unproductive delay, while remaining sensitive to important new technology developments.

Since this program is conducted in the research and development arena, it offers the prospects of non-adversarial engagement, outside the regulatory context, to allow the joint development of credible data about remediation technologies. <u>This opportunity should be especially advantageous to</u> (1) PRPs and owners/operators capable of early planning for technology options at a few sites and desirous of early EPA input, as well as (2) PRPs and owners/operators faced with a number of similar waste sites in the future-- under Superfund, RCRA Corrective Action, and the UST program--who want to develop more uniform, cost-effective technology proposals for such sites. Basic information about the FTTA is described further in Attachment B.

IMPLEMENTATION

The first six initiatives involve field testing new technologies that may benefit by technical assistance from the Office of Research and Development (ORD). ORD represents an objective third party that can be easily accessed through the existing OSWER/ORD support structure. This structure consists of five laboratories, which constitute the Technical Support Centers (both for Superfund and newly established for RCRA), the Superfund Technical Assistance Response Team (START) program, the Bioremediation Field Initiative, and the Superfund Innovative Technology Evaluation (SITE) program. Several of these programs are discussed later in this memorandum, and Regional offices are encouraged to use them. OSWER has asked ORD to give priority to requests for technical assistance under this directive, and we will use our existing priority-setting systems to accommodate needs articulated pursuant to this directive.

BROADER APPLICATION OF AVAILABLE RESOURCES AND TOOLS

In addition to these new initiatives, the application of other important existing policies and efforts should be broadened.

o Furthering Innovative Remediation at Leaking UST Sites

State and local UST programs have identified 100,000 confirmed leaks, and this number may triple in the next several years. The majority of sites currently undergoing corrective action are being remediated through pumping and treating ground water and excavation and off-site disposal of contaminated soil. The national UST program has established corrective action streamlining as one of its top priorities. The program's strategy includes promoting the use of improved technologies that will produce better and faster cleanups at lower cost than traditional methods.

The UST/LUST program has worked closely with the Office of Research and Development and private companies to foster the development of innovative site assessment and cleanup technologies, such as field measurement techniques, soil vapor surveying, vacuum-enhanced free-product recovery, active and passive bioremediation, and vacuum extraction. These technologies now must be moved from demonstrations to routine Regional offices should increase their use in the field. efforts to make state and local managers and staff, as well as cleanup consultants and contractors, more familiar with these non-traditional but proven technologies. Headquarters will continue fostering the development of even newer tools and techniques and should increase its support of regional efforts to achieve broader use of improved technologies.

o Further Enabling State Innovative Technology Leadership

First, the CERCLA core funding program provides an opportunity to assist states in establishing innovative technology advocates. Core program cooperative agreements help support state response programs to ensure involvement in CERCLA implementation activities. This may be a vehicle for promoting new technologies where the state and region agree it is appropriate. This approach is currently being utilized with success in Minnesota. The advocates can serve an important role of promoting the development and use of innovative technologies in the state CERCLA programs, with obvious spinoff benefits for their RCRA and UST programs. Some states have shown a strong interest in new technologies, and we should do everything possible to support their efforts and encourage initiatives at the state level.

Second, last year's <u>RCRA Implementation Study</u> highlighted the opportunity to empower a few states interested in furthering technology development. Regions should be open and encouraging of state applications for authority for RCRA R&D permitting, permit modification, treatability exclusion, and Subpart X permitting. States not authorized for RD&D permitting may consider a cooperative effort with the region for issuing these permits. The RD&D activities could involve treatability studies for a site or activities to help develop and commercialize a technology. This package of authorities will allow new technology developers and users to flourish in selected states.

In addition to the Federal Facilities Initiative above, states may want to work directly with Federal facilities in developing pilot sites for innovative technology. These activities do not have to be limited to final remedies, but may also include treatability tests, site stabilization, and demonstrations. Federal facilities under both CERCLA and RCRA authority may be particularly well suited for integrating clean-up activities with innovative treatment technologies.

Model RI/FS Work Plan and PRP Notice Letter Demand for Innovative Options

Some regions have issued special notices containing a Statement of Work and administrative order language requiring the responsible party to evaluate the use of innovative technologies at a particular site. This procedure should receive broader use at Superfund sites where alternatives for remediation are being considered for analysis in the RI/FS and where prerequisite treatability studies are required. This requirement in the special or general notice letters will help facilitate the development and use of innovative treatment technologies by the private sector. Specific language for this approach could be developed from OWPE's guidance document titled "Model Statement of Work for RI/FSs conducted by PRPs" (OSWER Directive 9835.8).

o Advocacy and Funding of Treatability Studies

Superfund program policy (Directive 9380.3-02FS, Treatability Studies Under CERCLA: An Overview, December 1989) requires that treatability studies should be conducted to generate data needed to support the implementation of treatment technologies. For sites where an innovative

0

technology is being considered, these studies will help provide performance information that should assist in the engineering evaluations. Funds are budgeted annually in the SCAP based on expected need for conducting treatability studies. Data and reports from these studies should be forwarded to Glen Shaul at ORD's Risk Reduction Engineering Lab. The appropriate protocol and format for these reports can be found in the "Guide for Conducting Treatability Studies Under CERCLA" (EPA/540/2-89/058). Information contained in these reports will be available through the Alternative Treatment Technology Information Center (ATTIC).

Every effort should be made to conduct or, as appropriate, to evaluate the PRP's treatability study. In planning for this activity, oversight funding should be requested through the SCAP budget process. Oversight of PRPlead treatability studies may be funded through the enforcement budget. In situations where PRPs recommend use of innovative treatment technologies at a site, but where treatability study data are insufficient, EPA policy allows fund and conduct technology-specific Agency to the treatability studies. The costs associated with the conduct of these treatability studies are recoverable under Section 107 of CERCLA.

0

Tracking and Expediting SITE Demonstrations

A recent Inspector General audit of the SITE program focused on delays in matching Superfund sites with This has contributed to overall delays in technologies. completing demonstration projects and technology assessments. In response, OSWER is encouraging greater participation in the SITE program and will begin tracking regional site nominations as a reporting measure in STARS (see "Implementation of an OSWER Recommendation from the Office of Inspector General Audit Report on the Superfund Innovative Technology Evaluation (SITE) Program"--memorandum dated January 2, 1991). OSWER will support the designation of additional regional FTE for support of SITE program demonstrations and recognizes the for time delays in RI/FSs at sites with potential demonstration projects. ORD management has also agreed that SITE demonstration projects must be more responsive to regional needs for treatability data.

Recently, ORD completed an internal management review of the SITE program. The purpose of the review was to evaluate the program's impact on Superfund remediation activities and to identify any changes needed to improve the program. Several changes already adopted are directed at making the program a more integral component of regional office Superfund site activities. The SITE program will make the design of technology evaluations sufficiently flexible to meet the regional offices' needs for treatability studies before remedy selection is made. SITE demonstration data will be presented to the RPM or OSC on a fast turnaround basis so that the data are available to be factored into the remedy selection decision. The SITE program will take advantage of ongoing remediation activities as a source of technology evaluations and technology transfer where possible. In addition, the program will use sites that are being evaluated under the START program and projects that are identified pursuant to this directive, as potential test locations for SITE evaluations.

ATTACHMENT A

Existing Program Efforts to Further Innovative Technologies

OSWER has several other ongoing efforts directed toward furthering the application of innovative alternatives through the acquisition and efficient use of data, reduction of technical uncertainties, and elimination of contracting impediments. These programs represent important resources that should continue to be used. The first two resources, that are of interest to the UST, RCRA, and Superfund Programs, concern the collection and use of data:

o Technical Support and Information Management

Readily accessible information on innovative technologies is a major priority of the Superfund program. This objective is being met through the utilization of on-line computer systems, direct expert technical assistance, and support for field activities to evaluate the performance of a given technology. Currently, EPA maintains several computer databases that may be accessed for information on treatment technologies. These databases include the Alternative Treatment Technology Information Center (ATTIC), the OSWER Bulletin Board (CLU-IN), the ROD Database, the Hazardous Waste Collection Database, and the Computerized On-line Information These systems include information on the System (COLIS). application of innovative technologies and may be used to aid networking among OSCs and RPMs. Due to the general shortage of cost and performance data on new technologies, use of these databases is important to provide the most current information available.

Technical assistance is available to Superfund and RCRA staff through ORD's Technical Support Centers and the Environmental Response Branch, OERR. Part of this effort involves networking among project managers through the engineering and ground water forums. In addition, as part of an initiative to provide direct technical support to OSCs and RPMs, the Superfund Technical Assistance Response Team(START) has been established to help evaluate the potential use of technologies. Currently, technical experts from EPA's Office and Development are providing long-term of Research consultation and support at 35 sites with complex treatment technologies issues. In addition, ORD is assisting the Superfund program in developing protocols for conducting treatability studies, so technologies can be evaluated using standardized parameters. ORD is also providing a staff person in each Regional office to serve as a liaison with their engineers and scientists.

o Bioremediation Field Initiative

Begun in the 4th quarter of FY 90, this program is intended to provide more real-time information on the field application of biotechnology for treating hazardous waste. Currently, over 131 CERCLA, RCRA, and UST sites have been identified as considering, planning, or operating full-scale biotreatment systems. The major focus of this initiative is to furnish direct support in evaluating full-scale cleanup assistance for conducting operations and technical treatability and pilot-scale studies. Several sites have already been selected for participation in the program. Performance, cost, and reliability information generated from these bioremediation studies will be used to further develop a treatability study database that will be made available to regional staff.

o Procurements for Innovative Technologies

Over the past several months, OSWER has been working with the Procurement and Contracts Management Division (PCMD) to address particular issues associated with the procurement of innovative technologies. As these issues are resolved, regions are encouraged to use the new provisions to the extent The first issue concerns the contracting for possible. treatability Under the Federal Acquisition studies. Regulations (FAR), firms are restricted from performing both the design and construction of a project. EPA has determined that this prohibition applies only to the prime contractor responsible for the overall design, and not to subcontractors The EPA Acquisition performing treatability studies. Regulations are being amended to clarify this point and to allow possible exceptions for contractors to work on both design and construction on a case-by-case basis.

A second issue concerns constraints on contractors working for both EPA and later working for a potentially responsible party (PRP) at the same site. This constraint was originally imposed on contractors to avoid conflicts of interest. Innovative technology is a special exception within these general guidelines. Rather than automatically assuming a contractor should first be precluded from working for a PRP after working for EPA, it is EPA's intent and commitment to first permit contractors and/or subcontractors performing evaluations of innovative technologies for the Agency to later work for the PRPs in as many instances as possible. Only in rare instances would EPA envision not permitting such work to be performed for the PRP. EPA and PRPs often work together in the spirit of cooperation and site work may be divided The Agency has therefore determined not to accordingly. preclude PRPs from using EPA contractors to perform such work as treatability studies. In addition, we want to ensure that vendors who perform treatability studies for EPA may also remain eligible to support PRP-lead design or construction work. This position is reflected in the final conflict of interest provisions for Superfund contracts which are currently being prepared and were initially published in the <u>Federal Register</u> as a proposed rule.

.

•

•

United States Environmental Protection Agency Office of Research and Development Washington, DC 20460 EPA/600/9-90/050 November 1990

GEPA Opportunities for Cooperative R&D with EPA: The Federal Technology Transfer Act

Both the U.S. Environmental Protection Agency (EPA) and private industry seek new, cost-effective technologies to prevent and control pollution. In the past, however, legal and institutional barriers have prevented government and industry from collaborating in developing and marketing these technologies. Also, the efforts of many companies to develop new technologies have been stymied by a lack of resources, such as scientific experts in particular fields or highly specialized equipment. The Federal Technology Transfer Act of 1986 (FTTA) removes some of these barriers to the development of commercial pollution control technologies.

The FITA makes possible cooperative research and development agreements (CRDAs) between federal laboratories, industry, and academic institutions. CRDAs set forth the terms of government/industry collaboration to develop and commercialize new technologies. According to the Act, these agreements will foster the technological and industrial innovation that is "central to the economic, environmental, and social well-being of citizens of the United States."

What Can Industry Gain from Signing a CRDA with EPA?

Access to High-Quality Science

EPA's 12 research laboratories employ over 600 scientists and engineers. Many of these laboratories combine world-class expertise with state-of-the-art equipment and fully permitted testing facilities. Certain types of environmental research, such as development of innovative technologies for treating hazardous wastes, require the collaboration of experts in many different fields. This type of interaction is easily adapted at EPA laboratories, because they are inter-disciplinary in nature.

Expanded Communication Channels Between Government and the Private Sector

CRDAs build working selationships between the government and the private sector. All parties benefit from the different perspectives that government and private sector scientists bring to an R&D project.



Exclusive Agreements for Developing New Technologies

Until recently, industry had little incentive to cooperate with federal laboratories because any technologies developed during joint research remained in the public domain for all to use. Now, under some CRDAs, companies are given exclusive rights to market and commercialize new technologies that result from the collaboration.

Licensing and Research Agreements: How Do They Work?

The procedure for setting up a cooperative R&D or licensing agreement under the FTTA is designed to encourage collaboration between industry and EPA laboratories. For industry, the key advantage of the process is the speed and ease with which the agreements can be negotiated and signed. CRDAs are not subject to federal contracting or grant requirements. In addition, each laboratory director has the authority to establish CRDAs for that particular lab, and this decentralization of the decision-making process reduces the administrative procedures involved.

Another important advantage is that CRDAs are flexible enough to fit the goals of many different sizes and types of companies. For example, under the FTTA, a company can support applied research at an EPA laboratory while reserving first rights to involvement in any technology that results. Or, if the scientific mechanism that makes a company's product work is unknown, the company can cooperate with an EPA laboratory to identify this mechanism. A company can also share space and equipment with EPA in a combined effort to develop an innovative technology.

Interested?

Several companies already have CRDAs with EPA, including Exxon, Shell Oil, Ford Motor Company, Dow-Corning, Hewlett-Packard, and CH₂M Hill, as well as several small businesses.

For further information about this program please write to:

Mr. Larry Pradkin, FTTA Coordinator Office of Technology Transfer and Regulatory Support Office of Research and Development U.S. Environmental Protection Agency 26 West Martin Luther King Drive Cincinnati, OH 45268.